Posebna izdanja, broj 25, 2020.

Marko Kostić

Abstract Degenerate Volterra Integro-Differential Equations



Mathematical Institute of the Serbian Academy of Sciences and Arts

Marko Kostić

marco.s@verat.net, markokostic121@yahoo.com

ABSTRACT DEGENERATE VOLTERRA INTEGRO-DIFFERENTIAL EQUATIONS

Издавач: Математички институт САНУ, Београд, Кнеза Михаила 36 Серија: Посебна издања, књига 25 Рецензенти: Теодор Атанацковић, Градимир Миловановић и Стеван Пилиповић Примљено за штампу одлуком Научног већа Математичког института САНУ За издавача: Владан Ђорђевић, главни уредник Технички уредник: Драган Аћимовић Штампа: "Академска издања", Земун Штампање завршено септембра 2020. Издавање је подржало Министарство просвете, науке и технолошког развоја Србије MSC (2010): Primary: 47D06, 34G25; Secondary: 47D60, 47D62, 47D99

CIP - Каталогизација у публикацији Народна библиотека Србије, Београд

517.968.22

KOSTIĆ, Marko, 1977-

Abstract degenerate Volterra integro-differential equations / Marko Kostić. -Београд : Математички институт САНУ, 2020 (Земун : Академска издања). -V, 516 str. ; 24 cm. - (Посебна издања / Математички институт САНУ ; књ. 25) Tiraž 60. - Registar. - Bibliografija: str. 495-516.

ISBN 978-86-80593-71-5

а) Интегралне једначине

COBISS.SR-ID 21272073

Contents

PREFACE
NOTATION
INTRODUCTION
Chapter 1. PRELIMINARIES 28 1.1. Selected topics on vector-valued functions and closed linear operators 28 1.2. Multivalued linear operators 37 1.2.1. Fractional powers 43 1.2.2. Hypercyclic and disjoint hypercyclic multivalued linear operators 44 1.3. Hypercyclic and disjoint hypercyclic MLO extensions 53 1.4. Laplace transform of functions with values in SCLCSs 56 1.5. Operators of fractional differentiation, Mittag-Leffler and Wright functions 63
Chapter 2. ABSTRACT DEGENERATE VOLTERRA INTEGRO- DIFFERENTIAL EQUATIONS IN LOCALLY CONVEX SPACES
2.1. (C, B) -resolvents
spaces
2.2.2. Differential and analytical properties of degenerate (a, k) -regularized <i>C</i> -resolvent families
2.2.3. Degenerate time-fractional equations associated with abstract differential operators
2.2.4. Degenerate second order equations associated with abstract differential operators
2.2.5. Semilinear degenerate relaxation equations associated with abstract differential operators
2.3. Degenerate multi-term fractional differential equations in locally convex spaces
2.3.1. Abstract Cauchy problem [(90)–(91)]

CONTENTS

2.3.2. Exponentially equicontinuous k-regularized C-resolvent (i, j) -
propagation families $\dots \dots \dots$
2.3.3. Exponentially equicontinuous (a, k) -regularized <i>C</i> -resolvent
families generated by A, B ; exponentially equicontinuous $(k; C)$ -regularized resolvent (i, j) -propagation families 119
2.3.4. Degenerate k-regularized (C_1, C_2) -existence and uniqueness
propagation families for (112)
2.3.5. Degenerate k-regularized (C_1, C_2) -existence and uniqueness
families
2.4. Abstract degenerate multi-term fractional differential equations with
Riemann–Liouville derivatives
2.5. The existence and uniqueness of solutions of abstract degenerate
fractional differential equations: ultradistribution theory
2.6. Entire and analytical solutions of abstract degenerate Cauchy problem
(PN)
2.7. Abstract incomplete degenerate differential equations
2.8. Abstract degenerate fractional differential equations in locally convex
spaces with a σ -regular pair of operators $\ldots \ldots \ldots \ldots \ldots \ldots 209$
2.9. Abstract degenerate non-scalar Volterra equations
2.10. Hypercyclic and topologically mixing properties of abstract degenerate
fractional differential equations
2.10.1. Hypercyclic and topologically mixing properties of solutions of the
equations $\mathbf{D}_t^{\alpha} Bu(t) = Au(t)$ and $B\mathbf{D}_t^{\alpha} u(t) = Au(t)$ ($\alpha > 0$) . 223
2.10.2. Hypercyclic and topologically mixing properties of certain classes of
degenerate abstract multi-term fractional differential equation 229
2.10.3. Hypercyclic and topologically mixing properties of abstract degenerate first and second order equations
2.10.4. <i>D</i> -Hypercyclic and <i>D</i> -topologically mixing properties of degenerate
fractional differential equations
2.11. The existence of distributional chaos in abstract degenerate fractional
differential equations
2.11.1. Distributional chaos for single operators
2.11.2. Distributionally chaotic properties of abstract degenerate
fractional differential equations
2.12. Appendix and notes
Chapter 3. MULTIVALUED LINEAR OPERATORS APPROACH 285
3.1. Abstract degenerate Volterra integro-differential inclusions
3.2. Multivalued linear operators as subgenerators of various types of
(a, k) -regularized C-resolvent operator families $\ldots \ldots \ldots \ldots 293$
3.2.1. Differential and analytical properties of (a, k) -regularized C-resolvent
families
3.2.2. Non-injectivity of regularizing operators C_2 and C
3.2.3. Degenerate K-convoluted C-semigroups and degenerate K-convoluted
C -cosine functions in locally convex spaces $\ldots \ldots 329$

CONTENTS

3.3. Degenerate <i>C</i> -distribution semigroups and degenerate <i>C</i> -ultradistribution semigroups in locally convex spaces
3.4. Degenerate C -distribution cosine functions and degenerate C -
ultradistribution cosine functions in locally convex spaces
3.5. Subordinated fractional resolvent families with removable singularities
at zero
3.5.1. Semilinear degenerate Cauchy inclusions
3.5.2. Purely fractional case
3.6. Hypercyclic and topologically mixing properties of abstract degenerate
(multi-term) time-fractional inclusions
3.6.1. Hypercyclic and topologically mixing properties of problem $(DFP)_{\alpha,\beta}$
3.6.2. Hypercyclic and topologically mixing properties of abstract degenerate
Cauchy problems of first and second order
3.6.3. Hypercyclic and topologically mixing properties of certain
classes of abstract degenerate multi-term fractional differential
inclusions
3.6.4. \mathcal{D} -Hypercyclic and \mathcal{D} -topologically mixing properties of abstract
degenerate multi-term fractional differential inclusions 393
3.7. Perturbation results for abstract degenerate Volterra integro-differential
equations
3.8. Approximation and convergence of degenerate (a, k) -regularized
C-resolvent families
3.8.1. Laguerre expansions of degenerate (a, k) -regularized C-resolvent
families $\dots \dots \dots$
3.8.2. Laguerre expansions of solutions to abstract non-degenerate
differential equations of first order
3.9. The existence and uniqueness of solutions of abstract incomplete
differential inclusions
3.9.1. Complex powers of multivalued linear operators with polynomially
bounded C-resolvent
3.9.2. Abstract incomplete differential inclusions
3.10. Inverse generator problem
3.10.1. Applications to degenerate time-fractional equations with abstract
1
3.11.1. Asymptotically almost periodic type functions, asymptotically almost automorphic type functions and their generalizations 443
3.11.2. Evolution systems and Green's functions
*
3.11.3. Quasi-asymptotically almost periodic functions and their
generalizations
3.11.4. Quasi-asymptotically almost periodic functions depending on two
parameters and composition principles
3.11.5. Invariance of quasi-asymptotical almost periodicity under the action
of convolution products

CONTENTS

3.11.6. Applications to abstract nonautonomous differential equations of
first order
3.11.7. Semilinear Cauchy problems
3.12. Almost periodic and asymptotically almost periodic type solutions
with variable exponents $L^{p(x)}$
3.12.1. Lebesgue spaces with variable exponents $L^{p(x)}$
3.12.2. Generalized almost periodic and generalized asymptotically almost
periodic functions in Lebesgue spaces with variable exponents
$L^{p(x)}$
3.12.3. Generalized two-parameter almost periodic type functions and
composition principles $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots 473$
3.12.4. Generalized (asymptotical) almost periodicity in Lebesgue spaces
with variable exponents $L^{p(x)}$: action of convolution products 475
3.12.5. Some applications
3.13. Appendix and notes
Index
Bibliography

PREFACE

The theory of linear degenerate Volterra integro-differential equations is in its very early stage. The present monograph is probably the first one that is specifically dedicated to abstract degenerate Volterra integro-differential equations and abstract degenerate (multi-term) fractional differential equations. It is a logical continuation of my previously published books entitled "Generalized Semigroups and Cosine Funtions" (Mathematical Institute SANU, Belgrade, 2011; [291]) and "Abstract Volterra Integro-Differential Equations" (CRC Press, Boca Raton, FL, 2015; [292]). Decomposition of material into chapters, sections and subsections, as well as the style used for presenting bibliographic information are almost the same as those already employed in [292].

Fractional calculus and fractional differential equations are rapidly growing fields of research, having invaluable importance in modeling of various problems appearing in physics, engineering, chemistry, biology, medicine and many other sciences. Degenerate fractional differential equations have not attracted the attention of a large number of authors working in the field of applied science; fractional models of some well known degenerate equations of mathematical physics considered in this monograph still do not have reasonable interpretations in the world of real phenomena. Although formulated in the general setting of (infinite-dimensional) sequentially complete locally convex spaces, a great part of our results on degenerate Volterra equations and degenerate fractional differential equations seems to be new even in the setting of Banach spaces. Throughout the book, we use four different kinds of fractional derivatives: the Caputo derivative, which is the most commonly used, the Riemann–Liouville derivative, the modified Liouville rightsided fractional derivative and the Weyl fractional derivative. The (multivalued) linear operators employed in our analyses need not be densely defined, in general.

Although the often large, we are sure that list of references is not complete. The book is not compulsively detailed and there is a large number of important topics that will not be analyzed here. Selection is based on the personal views and aspirations of the author, who pays special attention on examining possibilities to apply various types of convoluted or *C*-regularized families of solution operators in the analysis of abstract degenerate equations under consideration.

The target audiences are researchers and experts interested in getting to know the basic methods of linear theory of abstract degenerate Volterra integro-differential equations; the book can be also useful for graduate students in mathematics, physics or engineering science. The required mathematical preparation is no higher than basic functional analysis, complex analysis in one variable, integration theory and elementary partial differential equations.

There is a long list of people deserving my thanks. First of all, I would like to express sincere thanks to my family, closest friends and colleagues for permanent support of my work. I would also like to extend my sincere appreciation to Prof. V. Fedorov (Chelyabinsk, Russia), Prof. D. Velinov (Skopje, Macedonia), Prof. P. J. Miana, L. Abadias (Zaragoza, Spain), Prof. M. Murillo-Arcila, J. A. Conejero, A. Peris, J. Bonet (Valencia, Spain), Prof. D. Sidorov (Moscow State University, Russia), Prof A. Favaron (Milano, Italy), Prof. B. Chaouchi (Khemis Miliana, Algeria), Prof. M. Li, C. Chen, C.-G. Li (Chengdu, China), Prof. R. Ponce (Talca, Chile), Prof. C. Bianca (Paris, France), Prof. V. Keyantuo (Rio Piedras Campus, Puerto Rico, USA), Prof. T. Diagana (Huntsville, USA) and G. M. N'Guérékata (Baltimor, USA), Prof. E. M. A. El-Sayed (Alexandria, Egypt), Prof. M. S. Moslehian (Mashhad, Iran), Prof. C.-C. Kuo (New Taipei City, Taiwan) and C.-C. Chen (Taichung, Taiwan), for many stimulating discussions during the research. Special thanks go to Prof. S. Pilipović (Novi Sad, Serbia).

Loznica/Novi Sad December, 2019

Marko Kostić

NOTATION

 $\mathbb{N}, \mathbb{Z}, \mathbb{O}, \mathbb{R}, \mathbb{C}$: the natural numbers, integers, rationals, reals, complexes. For any $s \in \mathbb{R}$, we denote $|s| = \sup\{l \in \mathbb{Z} : s \ge l\}$ and $[s] = \inf\{l \in \mathbb{Z} : s \le l\}$. Re z. Im z: the real and imaginary part of a complex number $z \in \mathbb{C}$; |z|: the modul of z, $\arg(z)$: the argument of a complex number $z \in \mathbb{C} \setminus \{0\}$. $\mathbb{C}_{+} = \{ z \in \mathbb{C} : \operatorname{Re} z > 0 \}.$ $B(z_0, r) = \{ z \in \mathbb{C} : |z - z_0| \leq r \} \ (z_0 \in \mathbb{C}, r > 0).$ $\Sigma_{\alpha} = \{ z \in \mathbb{C} \setminus \{0\} : |\arg(z)| < \alpha \}, \ \alpha \in (0, \pi].$ $\operatorname{card}(G)$: the cardinality of G. $\mathbb{N}_0 = \mathbb{N} \cup \{0\}.$ $\mathbb{N}_n = \{1, \cdots, n\}.$ $\mathbb{N}_{n}^{0} = \{0, 1, \cdots, n\}.$ \mathbb{R}^n : the real Euclidean space, $n \ge 2$. The Euclidean norm of a point $x = (x_1, \ldots, x_n) \in \mathbb{R}^n$ is denoted by $|x| = (x_1^2 + x_2^2)$ $\cdots + x_n^2)^{1/2}$. If $\alpha = (\alpha_1, \ldots, \alpha_n) \in \mathbb{N}_0^n$ is a multi-index, then we denote $|\alpha| = \alpha_1 + \cdots + \alpha_n$. $x^{\alpha} = x_1^{\alpha_1} \dots x_n^{\alpha_n}$ for $x = (x_1, \dots, x_n) \in \mathbb{R}^n$ and $\alpha = (\alpha_1, \dots, \alpha_n) \in \mathbb{N}_0^n$. $f^{(\alpha)} := \partial^{|\alpha|} f / \partial x_1^{\alpha_1} \dots \partial x_n^{\alpha_n}; D^{\alpha} f := (-i)^{|\alpha|} f^{(\alpha)}.$ In (X,τ) is a topological space and $F \subset X$, then the interior, the closure, the boundary, and the complement of F with respect to X are denoted by int(F) (or F°), \overline{F} , ∂F and F^{c} , respectively.

Let Γ be a Jordan curve in the Eucliean plane \mathbb{R}^2 . Then we denote by $int(\Gamma)$ $(ext(\Gamma))$ the bounded (unbounded) connected component of $\mathbb{R}^2 \smallsetminus \Gamma$.

If X is a vector space over the field $\mathbb{F} \in \{\mathbb{R}, \mathbb{C}\}$, then for each non-empty subset F of X by span(F) we denote the smallest linear subspace of X which contains F.

 \circledast : the abbreviation for the fundamental system of seminorms which defines the topology of a sequentially complete locally convex space E.

SCLCS: shorthand used to denote a sequentially complete locally convex space.

L(E, X): the space of all continuous linear mappings from E into another SCLCS X, L(E) = L(E, E).

 \mathcal{B} : the family of bounded subsets of E.

 E^* : the dual space of E.

 E^{**} : the bidual of E.

A: a linear operator on E.

 \mathcal{A} : a multivalued linear operator on E (MLO).

C: a continuous linear operator on E.

If F is a subspace of E, then we denote by $\mathcal{A}_{|F}$ the part of \mathcal{A} in F.

 \mathcal{A}^* : the adjoint operator of \mathcal{A} .

 $D(\mathcal{A}), R(\mathcal{A}), \rho(\mathcal{A}), \sigma(\mathcal{A})$: the domain, range, resolvent set and spectrum of \mathcal{A} .

 $N(\mathcal{A})$ or Kern (\mathcal{A}) : the null space of \mathcal{A} .

 $\overline{\mathcal{A}}$: the closure of \mathcal{A} .

 $n(\mathcal{A})$: the stationarity of \mathcal{A} .

 $\sigma_p(\mathcal{A})$: the point spectrum of \mathcal{A} .

 $\rho_C(\mathcal{A})$: the *C*-resolvent set of \mathcal{A} .

Let A be a closed linear operator on E. Then [D(A)] denotes the sequentially complete locally convex space D(A) equipped with the following system of seminorms $p_A(x) = p(x) + p(Ax), x \in D(A), p \in \circledast$.

$$D_{\infty}(A) = \bigcap_{n \ge 1} D(A^n).$$

 $\chi_{\Omega}(\cdot)$: the characteristic function, defined to be identically one on Ω and zero elsewhere.

 $\Gamma(\cdot)$: the Gamma function.

If $\alpha > 0$, then $g_{\alpha}(t) = t^{\alpha-1}/\Gamma(\alpha), t > 0$; $g_0(t) \equiv$ the Dirac delta distribution. $\mathcal{D} = C_0^{\infty}(\mathbb{R}), \ \mathcal{E} = C^{\infty}(\mathbb{R})$: the Schwartz spaces of test functions.

 $\mathcal{S}(\mathbb{R}^n)$: the Schwartz space of rapidly decreasing functions $(n \in \mathbb{N})$; $\mathcal{S} \equiv \mathcal{S}(\mathbb{R})$.

 \mathcal{D}_0 : the subspace of \mathcal{D} which consists of those functions whose support is contained in $[0, \infty)$.

 $\mathcal{D}'(E) = L(\mathcal{D}, E), \ \mathcal{E}'(E) = L(\mathcal{E}, E), \ \mathcal{S}'(E) := L(\mathcal{S}, E)$: the spaces of continuous linear functions $\mathcal{D} \to E, \ \mathcal{E} \to E$ and $\mathcal{S} \to E$, respectively.

 $\mathcal{D}'_0(E), \mathcal{E}'_0(E), \mathcal{S}'_0(E)$: the subspaces of $\mathcal{D}'(E), \mathcal{E}'(E)$ and $\mathcal{S}'(E)$, respectively, containing the elements whose support is contained in $[0, \infty)$.

If $1 \leq p < \infty$, $(X, \|\cdot\|)$ is a complex Banach space, and $(\Omega, \mathcal{R}, \mu)$ is a measure space, then $L^p(\Omega, X, \mu)$ denotes the space which consists of those strongly μ -measurable functions $f: \Omega \to X$ such that $\|f\|_p := (\int_{\Omega} \|f(\cdot)\|^p d\mu)^{1/p}$ is finite; $L^p(\Omega, \mu) \equiv L^p(\Omega, \mathbb{C}, \mu).$

 $L^{\infty}(\Omega, X, \mu)$: the space which consists of all strongly μ -measurable, essentially bounded functions.

 $||f||_{\infty} = ess \sup_{t \in \Omega} ||f(t)||$, the norm of a function $f \in L^{\infty}(\Omega, X, \mu)$.

 $L^p(\Omega : X) \equiv L^p(\Omega, X) \equiv L^p(\Omega, X, \mu)$, if $p \in [1, \infty]$ and $\mu = m$ is the Lebesgue measure; $L^p(\Omega) \equiv L^p(\Omega : \mathbb{C})$.

 $L^p_{loc}(\Omega:X)$: the space which consists of those Lebesgue measurable functions $u(\cdot)$ such that, for every bounded open subset Ω' of Ω , one has $u_{|\Omega'} \in L^p(\Omega':X)$; $L^p_{loc}(\Omega) \equiv L^p_{loc}(\Omega:\mathbb{C}) \ (1 \leq p \leq \infty)$.

 $C_0(\mathbb{R}^n)$: the space consisted of those functions $f \in C(\mathbb{R}^n)$ for which

 $\lim_{|x|\to\infty} |f(x)| = 0$, topologized by the norm $|f| := \sup_{x\in\mathbb{R}^n} |f(x)|$.

 $C_b(\mathbb{R}^n)$ (BUC(\mathbb{R}^n)): the space of bounded continuous functions (bounded uniformly continuous functions) on \mathbb{R}^n , topologized by the norm $|f| := \sup_{x \in \mathbb{R}^n} |f(x)|$.

NOTATION

 $C^{\sigma}(\mathbb{R}^n)$: the space of bounded Hölder continuous functions on \mathbb{R}^n , topologized by the norm $|f|_{\sigma} := \sup_{x \in \mathbb{R}^n} |f(x)| + \sup_{x,y \in \mathbb{R}^n, x \neq y} \frac{|f(x) - f(y)|}{|x - y|^{\sigma}}$ $(0 < \sigma < 1)$.

If X is a Banach space, then the abbreviation $AC_{loc}([0,\infty):X)$ $(BV_{loc}([0,\infty):X))$ stands for the space of all X-valued functions that are absolutely continuous (of bounded variation) on any closed subinterval of $[0,\infty)$.

 $AC_{loc}([0,\infty)) \equiv AC_{loc}([0,\infty):\mathbb{C}), BV_{loc}([0,\infty)) \equiv BV_{loc}([0,\infty):\mathbb{C}).$

 $BV[0,T], BV_{loc}([0,\tau)), BV_{loc}([0,\tau):X)$: the spaces of functions of bounded variation.

 $C^k(\Omega: E)$: the space of k-times continuously differentiable functions $(k \in \mathbb{N}_0)$ from a non-empty subset $\Omega \subseteq \mathbb{C}$ into E; $C(\Omega: E) \equiv C^0(\Omega: E)$.

If $k \in \mathbb{N}$, $p \in [1, \infty]$ and Ω is an open non-empty subset of \mathbb{R}^n , then we denote by $W^{k,p}(\Omega : X)$ the Sobolev space which consists of those X-valued distributions $u \in \mathcal{D}'(\Omega : X)$ such that, for every $i \in \mathbb{N}^0_k$ and for every $\alpha \in \mathbb{N}^n_0$ with $|\alpha| \leq k$, one has $D^{\alpha}u \in L^p(\Omega : X)$; $H^k(\mathbb{R}^n : X) \equiv W^{k,2}(\mathbb{R}^n : X)$.

 $W_{loc}^{k,p}(\Omega:X)$: the space of those X-valued distributions $u \in \mathcal{D}'(\Omega:X)$ such that, for every bounded open subset Ω' of Ω , one has $u_{|\Omega'} \in W^{k,p}(\Omega':X)$.

 $S^{\alpha,p}(\mathbb{R}^n)$: the fractional Sobolev space of order $\alpha > 0$ $(p \in [1, \infty])$.

 $\mathcal{L}, \mathcal{L}^{-1}$: the Laplace transform and its inverse transform; $\tilde{f}(\lambda) \equiv \mathcal{L}f(\lambda)$.

 $\mathcal{F}, \mathcal{F}^{-1}$: the Fourier transform and its inverse transform.

LT - E: we say that a function $h(\cdot)$ belongs to the class LT - E if there exists a function $f \in C([0, \infty) : E)$ such that for each $p \in \circledast$ there exists $M_p > 0$ satisfying $p(f(t)) \leq M_p e^{at}, t \geq 0$ and $h(\lambda) = (\mathcal{L}f)(\lambda), \lambda > a$.

If a function K(t) satisfies the condition (P1) stated in Section 1.2, then we denote $abs(K) = inf\{\operatorname{Re} \lambda : \tilde{K}(\lambda) \text{ exists}\}.$

 $L^1_{loc}([0,\infty))$, resp. $L^1_{loc}([0,\tau))$: the space of scalar valued locally integrable functions on $[0,\infty)$, resp. $[0,\tau)$.

 J_t^{α} : the Riemann–Liouville fractional integral of order $\alpha > 0$.

 D_t^{α} : the Riemann–Liouville fractional derivative of order $\alpha > 0$.

 \mathbf{D}_t^{α} : the Caputo fractional derivative of order $\alpha > 0$.

 W^{α}_{\pm} : the Weyl fractional derivative of order α .

 $D_{-}^{\beta}u(s)$: the modified Liouville right-sided fractional derivative of order $\beta > 0$.

 $E_{\alpha,\beta}(z)$: the Mittag-Leffler function $(\alpha > 0, \beta \in \mathbb{R})$; $E_{\alpha}(z) \equiv E_{\alpha,1}(z)$.

 $\Psi_{\gamma}(t)$: the Wright function $(0 < \gamma < 1)$.

 $\wp(R)$: the set which consists of all subgenerators of an (a, k)-regularized C-resolvent family $(R(t))_{t \in [0,\tau)}$.

 $a^{*,n}(t)$: the *n*-th convolution power of function a(t).

 $\delta_{j,l}$: Kronecker's delta.

supp(f): the support of function f(t).

The notation of various spaces of generalized (asymptotically) almost periodic functions will be explained in Section 3.11 and Section 3.12.

The main purpose of this monograph is to provide an overview of the recent research on abstract degenerate Volterra integro-differential equations and abstract degenerate fractional differential equations in sequentially complete locally convex spaces. Considering only linear equations, we also contribute to the theories of abstract degenerate differential equations of first and second order. Some results of ours seem to be new even for abstract non-degenerate differential equations.

The organization and main ideas of this monograph, which is composed of three chapters, are given as follows. In order to make it useful for self-study, in Chapter 1 we have collected together in one place the mathematical preliminaries and diverse tools required for reading the material from Chapter 2 and Chapter 3. In Section 1.1, we remind ourselves of basic things concerning sequentially complete locally convex spaces (SCLCSs), closed linear operators, analytical functions with values in SCLCSs, integration of functions with values in SCLCSs, function spaces used, and complex powers of almost C-nonnegative operators. We present the most important definitions and results from the theory of multivalued linear operators (MLOs) in Section 1.2; in a separate subsection, we give some new results about hypercyclic and disjoint hypercyclic classes of multivalued linear operators. In Section 1.3, we explore a new theoretical approach to the Laplace transform of functions with values in SCLCSs [**312**] and collect various properties of vector-valued Laplace transform needed for our further work. The operators of fractional differentiation, Mittag-Leffler and Wright functions are investigated in Section 1.4.

The second chapter is consisted from twelve sections. One of the main subjects considered in this chapter is the following abstract degenerate Cauchy problem:

(1)
$$Bu(t) = f(t) + \int_0^t a(t-s)Au(s)ds, \quad t \in [0,\tau),$$

where $0 < \tau \leq \infty$, $t \mapsto f(t)$, $t \in [0, \tau)$ is a continuous mapping with values in a Hausdorff sequentially complete locally convex space E over the field of complex numbers, $a \in L^1_{loc}([0, \tau))$ and A, B are closed linear operators with domain and range contained in E. The reader may consult the monograph [459] by J. Prüss, the author's one [292] and the references cited there for the general theory of abstract non-degenerate Volterra equations in Banach and sequentially complete locally convex spaces, i.e., the theory of various types of resolvent (sometimes also called solution) families for (1), with B = I. Compared with non-degenerate case, increasingly less has been said about the well-posedness of abstract degenerate

Cauchy problem (1). The study of (1) starts presumably with the papers [195,196] by A. Favini and H. Tanabe (for some other references on degenerate integrodifferential equations, one may refer e.g. to [91, 92, 173, 175, 182, 183, 188, 189, 207, 280, 305, 309, 311, 319, 321, 327, 397] and [524]), who have analyzed the wellposedness of equation (1) in the setting of Banach spaces, considering separately the so-called hyperbolic case

(2)
$$\sup_{s>0, k\in\mathbb{N}} \|(B(sB+A)^{-1})^k\| < \infty$$

and the parabolic case

(3)
$$\sup_{\operatorname{Re}\lambda \ge 0} (1+|\lambda|)^{-1} \|B(\lambda B+A)^{-1}\| < \infty.$$

Blank hypothesis in [195] is that the operator $T = BA^{-1}$ is a bounded linear operator on E, as well as that the space E has a direct decomposition representation $E = N(T) \oplus \overline{R(T)}$ (similar assumptions have been used in [424, Sections 1.1.5–1.1.6], where the authors have investigated degenerate integrated semigroups, as well as in many other research papers concerning abstract degenerate differential equations).

EXAMPLE 0.0.1. [195]

- (i) Let Ω be a bounded domain in \mathbb{R}^n with smooth boundary, and let $m \in C^2(\overline{\Omega})$ be a nonnegative superharmonic function. Set $E := L^2(\Omega)$, $A := -\Delta$ with the Dirichlet boundary conditions, $D(B) := \{f; mf \in E\}$ and $Bf := mf \ (f \in D(B))$. Then (2) holds, with the corresponding supremum being less than or equal to 1.
- (ii) Let Ω be a bounded domain in \mathbb{R}^n with smooth boundary, let $m \in C(\overline{\Omega})$ be a nonnegative function, and let B be a multiplication operator by m(x), acting in $E := H^{-1}(\Omega)$. Set $D(A) := H_0^1(\Omega)$ and $A := -\Delta$. Then (3) holds.

Section 2.1 is devoted to the study of (C, B)-resolvents of closed linear operators. The material from Section 2.2 is taken from [**306**] and [**319**]. Following the approach of T.-J. Xiao and J. Liang (cf. [**542**, Definition 1.4] for the case a(t) = k(t) = 1, and [**543**, Definition 2.3] for the case $a(t) = g_n(t)$, k(t) = 1), we introduce the notion of an exponentially equicontinuous (a, k)-regularized Cresolvent family for (1). Generally, in our approach, the resolvent set of A does not contain 0 and can be even the empty set, which clearly implies that the operator T need not be defined. Although providing only partial information about the C-wellposedness of the problem (1), it is worth noting that our method has some advantages compared with other existing because we do not use any assumption on the decomposition of the state space E. For the introduced class, we analyze Hille–Yosida type theorems, subordination principles and perturbations in Subsection 2.2.1, as well as differential and analytical properties in Subsection 2.2.2. Abstract degenerate time-fractional differential equations of Caputo order $\alpha \in (0, 2]$ are investigated in Subsection 2.2.3–Subsection 2.2.4, while semilinear degenerate

8

relaxation equations associated with abstract differential operators are investigated in Subsection 2.2.5.

Unless specified otherwise, by E we denote a Hausdorff sequentially complete locally convex space over the field of complex numbers and by $A := A_0, A_1, \ldots, A_{n-1}, A_n := B$ we denote closed linear operators acting on E. Let $n \in \mathbb{N} \setminus \{1\}, 0 \leq \alpha_1 < \cdots < \alpha_n, f : [0, \infty) \to E$ be a continuous function, and let \mathbf{D}_t^{α} denote the Caputo fractional derivative of order α [61, 292]. The well-posedness of the following multi-term fractional differential equation has been analyzed in a series of recent papers (cf. [292, Section 2.10] for an extensive survey of results on abstract multi-term fractional differential equations with Caputo fractional derivatives):

$$\mathbf{D}_t^{\alpha_n} u(t) + \sum_{i=1}^{n-1} A_i \mathbf{D}_t^{\alpha_i} u(t) = f(t), \ t \ge 0; \quad u^{(j)}(0) = u_j, \ j = 0, \dots, \lceil \alpha_n \rceil - 1.$$

Define $\mathbb{N}_{n-1} := \{1, \ldots, n-1\}, \mathbb{N}_{n-1}^0 := \{0, 1, \ldots, n-1\}, m_i := \lceil \alpha_i \rceil, i \in \mathbb{N}_{n-1}, T_{i,L}u(t) := A_i \mathbf{D}_t^{\alpha_i}u(t), \text{ if } t \ge 0, i \in \mathbb{N}_{n-1} \text{ and } \alpha_i > 0, \text{ and } T_{i,R}u(t) := \mathbf{D}_t^{\alpha_i}A_iu(t), \text{ if } t \ge 0 \text{ and } i \in \mathbb{N}_{n-1}.$ For every $t \ge 0$ and $i \in \mathbb{N}_{n-1}$, we denote by $T_iu(t)$ either $T_{i,L}u(t)$ or $T_{i,R}u(t)$. In Section 2.3, the principal object of investigation is the following abstract degenerate multi-term problem:

(4)
$$\sum_{i=1}^{n-1} T_i u(t) = f(t), \quad t \ge 0$$

Set $\mathcal{I} := \{i \in \mathbb{N}_{n-1} : \alpha_i > 0 \text{ and } T_{i,L}u(t) \text{ appears on the left hand side of } (4)\}, Q := \max \mathcal{I}, \text{ if } \mathcal{I} \neq \emptyset \text{ and } Q := m_Q := 0, \text{ if } \mathcal{I} = \emptyset. We subject the following initial conditions to the equation } (4); \text{ cf. } [308] \text{ for more details:}$

(5)
$$u^{(j)}(0) = u_j, \quad 0 \le j \le m_Q - 1 \text{ and } (A_i u)^{(j)}(0) = u_{i,j} \text{ if } m_i - 1 \ge j \ge m_Q.$$

If $T_{n-1}u(t) = T_{n-1,L}u(t)$, then (5) reads as follows:

$$u^{(j)}(0) = u_j, \quad 0 \le j \le m_{n-1} - 1.$$

From a theoretical point of view, the problem [(4)-(5)] is not most general and later we will see how it can be further generalized. The analysis of problem [(4)-(5)] occupies a great deal of our attention and, for the sake of better exposition and understanding, we will be forced sometimes to repeat the basic things from its definition (sometimes the summation index on the left hand side of (4) runs differently and sometimes the time-variable will be denoted by s). It is our strong belief that this will not cause any form of plagiarism and inconsistency.

The most important subcases of problem [(4)-(5)] are the following fractional Sobolev equations:

$$(\mathrm{DFP})_R : \begin{cases} \mathbf{D}_t^{\alpha} Bu(t) = Au(t) + f(t), & t \ge 0, \\ Bu(0) = Bx; & (Bu)^{(j)}(0) = 0, \ 1 \le j \le \lceil \alpha \rceil - 1, \end{cases}$$

and

$$(\mathrm{DFP})_L : \begin{cases} B\mathbf{D}_t^{\alpha} u(t) = A u(t) + f(t), & t \ge 0, \\ u(0) = x; & u^{(j)}(0) = 0, \ 1 \leqslant j \leqslant \lceil \alpha \rceil - 1, \end{cases}$$

where $\alpha > 0$. These problems are generalizations of the usual Sobolev linear degenerate equations of first order:

$$B\frac{d}{dt}u(t) = Au(t) + f(t), \ u(0) = u_0 \ (t \ge 0)$$

and

$$\frac{d}{dt}Bu(t) = Au(t) + f(t), \ Bu(0) = Bu_0 \ (t \ge 0)$$

For further information concerning the wellposedness of Sobolev first order degenerate equations, the reader may consult the monographs by G. V. Demidenko, S. V. Uspenskii [140], A. Favini, A. Yagi [199], S. G. Krein [361], R. W. Carroll, R. E. Showalter [95], I. V. Melnikova, A. Filinkov [424], A. G. Sveshnikov, A. B. Al'shin, M. O. Korpusov, Yu. D. Pletner [503] and G. A. Sviridyuk, V. Fedorov [210], as well as the papers [6, 25, 49, 50, 90, 180, 197, 202, 219, 247, 258, 396, 420, 453, 454, 507, 517, 518] and [542]. The well-posedness of various types of degenerate second-order Sobolev equations has been analyzed in [14, 95, 199, 203, 306, 317, 421, 431, 490, 525, 543] and [555]. The corresponding results on degenerate Sobolev equations with integer higher-order derivatives can be found in [16, 18, 43, 210, 305, 515, 543] and [199, Section 5.7]; concerning abstract degenerate fractional differential equations, we may refer to [42, 210, 214, 271, 306, 317, 321, 325, 379] and [558].

We continue by explaining the organization of material in Section 2.3; cf. [307, 311, 314] and [327]. Various types of solutions of abstract Cauchy prob- $\lim [(4)-(5)]$ and its integral analogues are presented in Subsection 2.3.1. For the purpose of study of abstract multi-term problem [(4)-(5)], we introduce the classes of exponentially equicontinuous k-regularized C-resolvent (i, j)-propagation families (Subsection 2.3.2) and exponentially equicontinuous (k; C)-regularized resolvent (i, j)-propagation families (Subsection 2.3.3). We investigate subordination principles, regularity properties, existence and uniqueness of solutions of problem [(4)-(5)]and related Cauchy problems obtained by integration; following the approach of N. H. Abdelaziz and F. Neubrander [6] for abstract degenerate equations of first order, in Subsection 2.3.3 we introduce and analyze the classes of exponentially equicontinuous (a, k)-regularized C-resolvent families generated by a pair of operators and exponentially equicontinuous (k; C)-regularized resolvent (i, j)-propagation families for [(4)-(5)], clarifying also some new results on the C-wellposedness of the equation $(DFP)_L$ with abstract differential operators. In Theorem 2.3.20, we reconsider the assertion of [292, Theorem 2.3.3] for systems of abstract degenerate differential equations involving polynomial matrices of abstract differential operators. Later on, we explain how we can use this result in the analysis of existence and uniqueness of entire solutions of some well known degenerate differential equations of mathematical physics, like the Rossby wave equation, the Sobolev equation, the internal wave equation in the Boussinesq approximation, the gravity-gyroscopic wave equation, and the equation of small amplitude oscillations of a rotating viscous fluid. In Subsection 2.3.4–Subsection 2.3.5, we take up the study of the following special case of problem [(4)-(5)]:

(6)
$$B\mathbf{D}_{t}^{\alpha_{n}}u(t) + \sum_{i=1}^{n-1} A_{i}\mathbf{D}_{t}^{\alpha_{i}}u(t) = A\mathbf{D}_{t}^{\alpha}u(t) + f(t), \quad t \ge 0;$$
$$u^{(j)}(0) = u_{j}, \quad j = 0, \dots, \lceil \alpha_{n} \rceil - 1,$$

where $0 \leq \alpha < \alpha_n$. Following the method employed in our joint papers with C.-G. Li and M. Li [346, 347], we introduce the class of degenerate k-regularized (C_1, C_2) -existence and uniqueness propagation families for (6), the class of degenerate k-regularized (C_1, C_2) -existence and uniqueness families, and the class of (k, C_2) -uniqueness families for (6). We present some applications to abstract degenerate equations analyzed by M. V. Falaleev and S. S. Orlov in [174], as well as an illustrative example of abstract degenerate Cauchy problem (6) whose solution can be defined only locally. We investigate subordination principles and perturbation results for the introduced classes. In the remainder of Subsection 2.3.5, we analyze entire and analytical solutions of (6), and apply our theoretical results in the study of abstract Boussinesq-Love equation, which is important in the modeling the longitudinal waves in an elastic bar with the transverse inertia, and the abstract Barenblatt-Zheltov-Kochina equation, which is important in the study of fluid filtration in fissured rocks, as well as in the studies of moisture transfer in soil and the process of two-temperature heat conductivity. We also analyze some fractional analogues of these equations.

Denote by D_t^{α} the Riemann-Liouville fractional derivative of order α [61, 315, 345]. In Section 2.4, we analyze the following abstract multi-term fractional differential equation:

(7)
$$BD_t^{\alpha_n}u(t) + \sum_{j=1}^{n-1} A_j D_t^{\alpha_j}u(t) = AD_t^{\alpha}u(t) + f(t), \quad t \in (0,\tau)$$

in a complex Banach space E. If $\alpha_n > 1$ ($\alpha_n = 1$, $\alpha_n < 1$), then we say that (7) is of type (SC1) ((SC2), (SC3)). Depending on type of (7), we subject initial conditions to (7) and carry out further analyses. The main result of Section 2.4 is Theorem 2.4.9, in which we analyze some applications of certain subclasses of k-regularized (C_1, C_2) -existence and uniqueness families in the study of unique solvability of (7).

Section 2.5 conveys results of a joint research study [213] with V. E. Fedorov. Suppose that $n \in \mathbb{N}$, $\zeta \in (0, 1]$, p_0, p_1, \ldots, p_n and q_0, q_1, \ldots, q_n are given nonnegative integers satisfying $p_0 = q_0 = 0$ and $0 < p_1 + q_1 \leq p_2 + q_2 \leq \ldots \leq p_n + q_n$. Set $T_i u(s) := (\mathbf{D}_s^{\zeta})^{p_i} A_i (\mathbf{D}_s^{\zeta})^{q_i} u(s), s \geq 0, i \in \mathbb{N}_n^0, S_l := \{i \in \mathbb{N}_n : q_i \geq 1\},$ $S_r := \{i \in \mathbb{N}_n : p_i \geq 1\}$, and conventionally, $\max(\emptyset) := \emptyset, \mathbb{N}_{\emptyset}^0 := \emptyset$. We analyze the following abstract degenerate Cauchy problem:

(8)
$$\sum_{i=0}^{n} T_i u(s) = 0, \quad s \ge 0,$$

accompanied with the following initial conditions:

(9)
$$((\mathbf{D}_{s}^{\zeta})^{j}u(s))_{s=0} = u_{j}, \quad j \in \mathbb{N}_{\max\{q_{i}-1:i \in S_{l}\}}^{0}, \text{ and}$$

$$((\mathbf{D}_{s}^{\zeta})^{j}A_{i}(\mathbf{D}_{s}^{\zeta})^{q_{i}}u(s))_{s=0} = u_{i,j} \ (i \in S_{r}, \ j \in \mathbb{N}_{p_{i}-1}^{0})$$

Although not visible at first glance, the study of problem [(8)–(9)] taken up in this section leans heavily on the method used in our previous paper [**303**] on regularization of ultradistribution semigroups and ultradistribution sines of Beurling class. The crucial thing is the existence and polynomial boundedness of the operators P_{λ}^{-1} , where

$$P_{\lambda} := \lambda^{(p_n + q_n)\zeta} B + \sum_{i=0}^{n-1} \lambda^{(p_i + q_i)\zeta} A_i, \quad \lambda \in \mathbb{C} \smallsetminus \{0\},$$

on an appropriate ultra-logarithmic region $\Lambda_{\alpha,\beta,l}$ (cf. J. Chazarain [99], 1971).

Assume now that $n \in \mathbb{N}$, $0 < \zeta \leq 2$, q_0, q_1, \ldots, q_n are given non-negative integers satisfying $q_0 = 0$ and $0 < q_1 \leq q_2 \leq \ldots \leq q_n$, as well as that $p_i = 0$ for all $i \in \mathbb{N}_n^0$. Hence, $T_i u(s) = A_i(\mathbf{D}_s^{\zeta})^{q_i} u(s)$, $s \geq 0$, $i \in \mathbb{N}_n^0$ and $P_{\lambda} = \lambda^{q_n \zeta} B + \sum_{i=0}^{n-1} \lambda^{q_i \zeta} A_i$, $\lambda \in \mathbb{C} \setminus \{0\}$. In Section 2.6, we investigate the abstract degenerate multi-term Cauchy problem (8) accompanied with the initial conditions of the following form:

$$((\mathbf{D}_{s}^{\zeta})^{j}u(s))_{s=0} = u_{j}, \ j \in \mathbb{N}_{q_{n}-1}^{0}, \ \text{if} \ \zeta \in (0,1], \ \text{resp.},$$

$$(10) \ ((\mathbf{D}_{s}^{\zeta})^{j}u(s))_{s=0} = u_{j}, \ j \in \mathbb{N}_{q_{n}-1}^{0}; \ \left(\frac{d}{ds}(\mathbf{D}_{s}^{\zeta})^{j}u(s)\right)_{s=0} = v_{j}, \ j \in \mathbb{N}_{q_{n}-1}^{0}, \ \text{if} \ \zeta \in (1,2].$$

In Definition 2.6.1, we introduce the notion of an entire (analytic) solution of problem [(8), (10)]. Denote by $\mathfrak{W}(\mathfrak{W}_e)$ the subspace of E^{q_n} , resp. E^{2q_n} , consisting of all initial values $(u_0, \ldots, u_{q_n-1}) \in E^{q_n}$, resp. $(u_0, \ldots, u_{q_n-1}; v_0, \ldots, v_{q_n-1}) \in E^{2q_n}$, subjected to some analytical solution $u(\cdot)$ of problem (8) on the region $\mathbb{C} \setminus (-\infty, 0]$ (entire solution $u(\cdot)$ of problem (8)). The main result of Section 2.6 is Theorem 2.6.2, which asserts that the set \mathfrak{W} is dense in $(C(\bigcap_{i=0}^{n} D(A_j)))^{q_n}$ for the topology of E^{q_n} , provided that $0 < \zeta < 1$, resp. $(C(\bigcap_{i=0}^n D(A_j)))^{2q_n}$ for the topology of E^{2q_n} , provided that $1 < \zeta < 2$. Similar statements hold for the equations with integer order derivatives, when we have that the set \mathfrak{W}_e is dense in $(C(\bigcap_{i=0}^n D(A_i)))^{q_n}$ for the topology of E^{q_n} , provided that $\zeta = 1$, resp. $(C(\bigcap_{i=0}^n D(A_i)))^{2q_n}$ for the topology of E^{2q_n} , provided that $\zeta = 2$. The class of C-regularized semigroups of growth order r > 0 has recently been introduced in [103] following the ideas of G. Da Prato [126] (1966). In Section 2.7, we continue our previous research studies [101, 103] by investigating abstract incomplete Cauchy problems degenerate in time. For that purpose, we introduce the concept of degenerate (B, C)-regularized semigroups of growth order r > 0 and clarify their most important structural properties. In Theorem 2.7.4, we deal with the following abstract incomplete degenerate differential equations with modified Liouville right-sided fractional derivatives:

$$(FP_{\alpha_1,\beta_1,\theta}): \begin{cases} u \in C^{\infty}((0,\infty):E), \\ D_{-}^{\alpha_1}BD_{-}^{\beta_1}u(s) = e^{i\theta/\gamma}Au(s), \quad s > 0, \\ \lim_{s \to 0+} Bu(s) = Cx, \\ \text{the set } \{Bu(s):s > 0\} \text{ is bounded in } E \end{cases}$$

and

$$(FP_{\alpha_1,\beta_1,\theta})': \begin{cases} u \in C^{\infty}((0,\infty):E), \\ D_{-}^{\alpha_1}BD_{-}^{\beta_1}u(s) = e^{i\theta/\gamma}Au(s), \quad s > 0, \\ \lim_{s \to 0+} Bu(s) = Cx, \\ \text{the sets } \{(1+s^{-(q+1)/\gamma})^{-1}u(s):s > 0\} \\ \text{and } \{(1+s^{-(q+1)/\gamma})^{-1}Bu(s):s > 0\} \text{ are bounded in } E; \end{cases}$$

here, $0 < \gamma < 1/2$, $\alpha_1 \ge 0$, $\beta_1 \ge 0$, $\alpha_1 + \beta_1 = 1/\gamma$, $\theta \in (\gamma \pi - (\pi/2), (\pi/2) - \gamma \pi)$, $C \in L(E)$ is injective, A belongs to the class $\mathcal{M}_{B,C,q}$ defined later, and q > -1, resp. $-1 - \gamma < q \leq -1$, in the case of consideration of problem $(FP_{\alpha_1,\beta_1,\theta})$, resp. $(FP_{\alpha_1,\beta_1,\theta})'$. In Theorem 2.7.5, we treat the following abstract incomplete degenerate Cauchy problem of second order

$$(P_{2,q,B}): \begin{cases} u \in \mathcal{A}(\Sigma_{(\pi/2)-(\omega/2)}:E), \ Bu \in \mathcal{A}(\Sigma_{(\pi/2)-(\omega/2)}:E), \\ Bu''(z) = \frac{d^2}{dz^2}Bu(z) = Au(z), \ z \in \Sigma_{(\pi/2)-(\omega/2)}, \\ \lim_{z \to 0, z \in \Sigma_{\delta}}Bu(z) = Cx, \ \text{for every } \delta \in (0, (\pi/2) - (\omega/2)) \\ \text{the sets } \{(1+|z|^{-(2q+2)})^{-1}u(z): z \in \Sigma_{\delta}\} \text{ and} \\ \{(1+|z|^{-(2q+2)})^{-1}Bu(z): z \in \Sigma_{\delta}\} \text{ are bounded in } E \\ \text{for every } \delta \in (0, (\pi/2) - (\omega/2)), \end{cases}$$

where (-3)/2 < q < (-1)/2, $\mathcal{A}(\Sigma_{(\pi/2)-(\omega/2)} : E)$ denotes the set consisting of all analytic functions from the sector $\Sigma_{(\pi/2)-(\omega/2)} = \{\lambda \in \mathbb{C} \setminus \{0\} : |\arg(\lambda)| < 0\}$ $(\pi/2)-(\omega/2)$ into E, and the condition (H) specified later holds with some number $\omega \in [0,\pi)$. We raise a question of finding some sufficient conditions guaranteeing the uniqueness of solutions of problem $(P_{2,q,B})$. The main aim of Section 2.8 is to present the most important results from our second joint research study with V. E. Fedorov [214]. In this section, we follow the methods employed in the monograph G. A. Sviridyk–V. E. Fedorov [509], which are maybe insufficiently reconsidered for abstract degenerate Volterra integro-differential equations so far (cf. Section 2.12) for more details). The notion of a regular resolvent set of a closed linear operator A acting on E plays an important role in this section: let us recall that the regular resolvent set of A, $\rho^r(A)$ shortly, is defined as the union of those complex numbers $\lambda \in \rho(A)$ for which $(\lambda - A)^{-1} \in R(E)$, where R(E) denotes the set consisting of all regular bounded linear operators $D \in L(E)$, i.e., the operators $D \in L(E)$ for which there exists a positive constant c > 0 such that for each seminorm $p \in \mathbb{R}$ there exists another seminorm $q \in \mathfrak{B}$ such that $p(D^n x) \leq c^n q(x), x \in E, n \in \mathbb{N}$ (infimum of such constants is said to be constant of regularity of D). Here, \circledast is the abbreviation for the fundamental system of seminorms which defines the topology of E. Assume now that X, Y are two SCLCSs as well as that A and B are two closed linear operators acting between the spaces X and Y. Set $R^B_{\lambda}(A) := (\lambda B - A)^{-1}B$ and $L^B_{\lambda}(A) := B(\lambda B - A)^{-1}$. By a regular B-resolvent set of the operator A, $\rho^B_r(A)$ for short, we mean the set

$$\rho_r^B(A) := \{ \lambda \in \mathbb{C} : (\lambda B - A)^{-1} \in L(Y, X), \ R_\lambda^B(A) \in L(X), \ L_\lambda^B(A) \in L(Y) \}.$$

It is said that the operator A is (B, σ) -regular, or equivalently, that a pair of operators (A, B) is σ -regular, iff there exists a finite number a > 0 such that, for every $\lambda \in \mathbb{C}$ with $|\lambda| > a$, we have $\lambda \in \rho_r^B(A)$. We also define the notion of a (B, p)-regular operator A (a p-regular pair of operators (A, B)), where $p \in \mathbb{N}_0$. The main result of Section 2.8 is Theorem 2.8.6, where we give some sufficient conditions for the existence of a unique strong solution of the following inhomogeneous degenerate Cauchy problem

$$(\mathrm{DF})_f: \begin{cases} \mathbf{D}_t^{\alpha} B u(t) = A u(t) + f(t), & t \ge 0, \\ u^{(k)}(0) = x_k, & 0 \le k \le \lceil \alpha \rceil - 1, \end{cases}$$

under the assumption on (B, p)-regularity of operator A. In [459, Chapter II], J. Prüss has studied abstract non-scalar Volterra equations. Applications have been given in the analysis of viscoelastic Timoshenko beam model, Midlin–Timoshenko plate model and viscoelastic Kirchhoff plate model, with the corresponding materials being non-synchronous, as well as in the analysis of some problems appearing in the theories of linear thermoviscoelasticity and electrodynamics. Section 2.9 is written in an expository manner (we do have obligation to say that Subsection 2.3.5, Section 2.4, Section 2.8 and Subsection 3.6.4 are also written without giving the proofs of structural results; the only exception is Theorem 2.4.6) and its aim is to show how the techniques established in [459] and [299] can be helpful in the analysis of a substantially large class of abstract degenerate Volterra integral equations of non-scalar type [318]. More precisely, we treat the following linear degenerate Volterra equation:

(11)
$$Bu(t) = f(t) + \int_0^t A(t-s)u(s)ds, \quad t \in [0,\tau),$$

where X and Y are two complex Banach spaces satisfying that Y is continuously embedded in X, B is a closed linear operator with domain and range contained in $X, \tau \in (0, \infty], f \in C([0, \tau) : X)$ and $A \in L^1_{loc}([0, \tau) : L(Y, X))$. In Definition 2.9.1, we introduce the notion of a strong (mild) solution of problem (11) as well as the notion of (kC)-wellposedness of (11); here, k(t) is a scalar-valued kernel on $[0, \tau)$ and the operator $C \in L(X)$ is injective. For the purpose of research of (11), we introduce the classes of (weak) (A, k, B)-regularized C-pseudoresolvent families and (A, k, B)-regularized C-resolvent families in Definition 2.9.2. After that, we investigate the generation of (A, k, B)-regularized C-pseudoresolvent families, the main solution concept considered, as well as their analytical properties and hyperbolic perturbation results.

The main aim of Section 2.10 is to continue our previous researches of hypercyclic and topologically mixing properties of abstract non-degenerate (multi-term) fractional differential equations with Caputo derivatives (cf. [292, Chapter 3] for a comprehensive survey of results). It is our strong belief that our study of (disjoint) hypercyclic and (disjoint) topologically mixing extensions of multivalued linear operators (binary relations) will receive some attention of the authors working in the field of linear topological dynamics. Concerning the other subjects, we would like to note that we are not primarily concerned with studying new concepts in the

theory of hypercyclicity and that some of our results have origins in the theory of hypercyclic single valued operators. The most important assertions, from the view-point of possible applications, will be those in which we reconsider the well known Desch–Schappacher–Webb and Banasiak–Moszyński criteria for chaos of strongly continuous semigroups.

Suppose, for the time being, that E is a separable infinite-dimensional Fréchet space over the field $\mathbb{K} \in \{\mathbb{R}, \mathbb{C}\}$, whose topology is induced by the fundamental system $(p_n)_{n \in \mathbb{N}}$ of increasing seminorms. Then the translation invariant metric $d: E \times E \to [0, \infty)$, defined by

$$d(x,y) := \sum_{n=1}^{\infty} \frac{1}{2^n} \frac{p_n(x-y)}{1+p_n(x-y)}, \quad x,y \in E,$$

satisfies, among many other properties, the following:

$$d(x+u, y+v) \leq d(x, y) + d(u, v)$$
 and $d(cx, cy) \leq (|c|+1)d(x, y)$,

for any $c \in \mathbb{K}$ and $x, u, u, v \in E$. A linear mapping $T: E \to E$ is said to be hypercyclic, resp. cyclic, iff there exists an element $x \in E$ whose orbit Orb(x,T) := $\{T^n x : n \in \mathbb{N}_0\}$ is dense in E, resp. iff there exists an element $x \in E$ such that the linear span of Orb(x, T) is dense in E, while T is said to be topologically transitive iff for for any pair of open non-empty subsets U. V of E there exists $n \in \mathbb{N}$ such that $T^n(U) \cap V \neq \emptyset$. Using a simple Baire Category argument, it readily follows that T is hypercyclic iff T is topologically transitive (G. D. Birkhoff [77], 1929). Furthermore, a linear mapping $T: E \to E$ is said to be chaotic (supercyclic, resp. positively supercyclic) iff T is hypercyclic and the set of periodic points of T, defined by $\{x \in E : \text{ there exists } n \in \mathbb{N} \text{ such that } T^n x = x\}$, is dense in E (iff there exists an element $x \in E$ whose projective orbit $\{cT^n x : n \in \mathbb{N}_0, c \in \mathbb{K}\}$, resp. positive projective orbit $\{cT^n x : n \in \mathbb{N}_0, c \ge 0\}$, is dense in E). Chronologically, the first example of a hypercyclic operator was constructed by G. D. Birkhoff in the afore-mentioned paper [77]. In actual fact, there were proved that the translation operator $f \mapsto f(\cdot + a), f \in H(\mathbb{C}), a \in \mathbb{C} \setminus \{0\}$ is hypercyclic in $H(\mathbb{C})$; the hypercyclicity of the derivative operator $f \mapsto f', f \in H(\mathbb{C})$ was proved by G. R. MacLane [403] in 1952. The first example of a hypercyclic operator acting on a Banach space was constructed by S. Rolewicz [464] in 1969; the state space in his analysis is chosen to be $l^2(\mathbb{N})$. The most commonly used criterion for proving hypercyclicity of single operators, the so-called Hypercyclicity Criterion, was discovered independently by C. Kitai [284] (1982) and R. M. Gethner, J. Shapiro [224] (1987). From the period 1987 onwards, the study of hypercyclicity of single valued operators and various generalizations of this concept have experimented a great development. For further information concerning dynamical properties of single operators, we refer the reader to the monographs [238] by K.-G. Grosse-Erdmann, A. Peris and [60] by F. Bayart, E. Matheron.

Although the first examples of chaotic semigroups were given by C. R. MacCluer [402] and V. Protopopescu, Y. Azmy [458] already in 1992, the hypercyclic and chaotic properties of strongly continuous semigroups were studied in a systematic way for the first time in the paper [143] by W. Desch, W. Schappacher and G. F.

Webb (1997). The study of hypercyclicity of second order non-degenerate equations starts with the paper [85] by A. Bonilla and P. J. Miana (2008), while the study of hypercylicity of non-degenerate fractional differential equations goes back to author's paper [300] (2012).

The non-existence of an appropriate reference which treats the hypercyclicity of abstract degenerate PDEs may has been a strong motivational factor that influenced us to write the papers [**308**, **309**], from which the material of Section 2.10 is taken; our results seem to be new even for abstract degenerate differential equations of first and second order. The analysis we have carried out indicates that the class of hypercyclic abstract degenerate equations is, by all means, substantially larger than the corresponding class of non-degenerate equations. Hypercyclic and topologically mixing properties of solutions of abstract (multi-term) fractional differential equations with Riemann–Liouville derivatives can be analyzed similarly, and we will only refer the reader to the papers [**345**] and [**382**] for more details about this topic.

Distributional chaos is a very popular field of research in the theory of topological dynamics of linear operators. Let us recall that the notion of distributional chaos for interval maps was introduced by B. Schweizer and J. Smítal [481] in 1994. Distributional chaos was firstly considered in the setting of linear operators when studying a quantum harmonic oscillator [159] (1999) and [441] (2006). A systematic study of distributional chaos for backward shifts operators was initiated in [414], while an example of a backward shift operator with a full scrambled set appeared in [415]. For some other relevant references on distributional chaos, one may refer e.g. to [46,66,68,440] and [499]. A linear continuous operator T acting on a Fréchet space E is said to be distributionally chaotic iff there exist an uncountable set $S \subseteq E$ (scrambled set) and $\sigma > 0$ such that for each $\varepsilon > 0$ and for each pair $x, y \in S$ of distinct points we have that

$$\overline{dens}(\{k \in \mathbb{N} : d(T^k x, T^k y) \ge \sigma\}) = 1 \text{ and}$$
$$\overline{dens}(\{k \in \mathbb{N} : d(T^k x, T^k y) < \varepsilon\}) = 1,$$

where $d(\cdot, \cdot)$ denotes the metric on E and the upper density of a set $D \subseteq \mathbb{N}$ is defined by

$$\overline{dens}(D) := \limsup_{n \to +\infty} \frac{\operatorname{card}(D \cap [1, n])}{n}.$$

If we can choose S to be dense in E, then we say that T is densely distributionally chaotic. The notion of a (densely) distributionally chaotic strongly continuous semigroup on Fréchet space has recently been introduced in [112] (joint work with J. A. Conejero, P. J. Miana and M. Murillo-Arcila; cf. also [13,55,56] and [116] for further information concerning distributionally chaotic strongly continuous semigroups on Banach spaces) as follows: A strongly continuous semigroup $(T(t))_{t\geq 0} \subseteq L(E)$ is said to be distributionally chaotic iff there are an uncountable set $S \subseteq E$ and $\sigma > 0$ such that for each $\varepsilon > 0$ and for each pair $x, y \in S$ of distinct points we have that

$$Dens(\{t \ge 0 : d(T(t)x, T(t)y) \ge \sigma\}) = 1$$
 and

$$\overline{Dens}(\{t \ge 0 : d(T(t)x, T(t)y) < \varepsilon\}) = 1,$$

where the upper density of a set $D \subseteq [0,\infty)$ is defined now by

$$\overline{Dens}(D) := \limsup_{t \to +\infty} \frac{m(D \cap [0,t])}{t}$$

with $m(\cdot)$ being the Lebesgue measure on $[0, \infty)$. If, moreover, we can choose S to be dense in E, then $(T(t))_{t\geq 0}$ is said to be densely distributionally chaotic. The question of whether an operator $T \in L(X)$ or a strongly continuous semigroup $(T(t))_{t\geq 0} \subseteq L(X)$ is distributionally chaotic or not is closely connected with the existence of distributionally irregular vectors, i.e., those elements $x \in X$ such that for each $\sigma > 0$

$$\overline{dens}(\{k \in \mathbb{N} : d(T^k x, 0) > \sigma\}) = 1 \text{ and}$$
$$\overline{dens}(\{k \in \mathbb{N} : d(T^k x, 0) < \sigma\}) = 1,$$

respectively,

$$\overline{Dens}(\{t \ge 0 : d(T(t)x, 0) > \sigma\}) = 1 \text{ and}$$
$$\overline{Dens}(\{t \ge 0 : d(T(t)x, 0) < \sigma\}) = 1.$$

For the basic information concerning distributionally chaotic properties of ill-posed abstract non-degenerate equations of first order, we refer the reader to [112]. Distributionally chaotic properties of abstract non-degenerate fractional differential equations in Banach spaces has recently been analyzed in [320], where it has been pointed out that the notion of distributional chaos is much more appropriate for dealing with fractional equations than that of the usually considered Devaney chaos. The main purpose of Section 2.11 is to analyze a class of distributionally chaotic abstract degenerate (multi-term) fractional differential equations. For further information about (distributional) chaos of general binary relations, multivalued linear operators and abstract integro-differential equations, we refer the reader to the recent monograph of M. Kostić [294].

In this monograph, we will not discuss linear topological dynamics of abstract degenerate differential equations associated with the use of backward shift operators. For more details concerning discrete case, we refer the reader to the doctoral dissertation of Ö. Martin [409] and references cited therein.

We believe that our researches will enjoy reading Section 2.10 and Section 2.11. In order to make our monograph a convenient reference, we have concluded the second and third chapter with a concise summary and further guidance notes. Appendices to the second chapter are given in Section 2.12.

A large number of research papers, starting presumably with that of A. Yagi [549], written over the last twenty five years, have concerned applications of multivalued linear operators to abstract degenerate differential equations (cf. [199] for the basic source of information on this subject). In the third chapter of monograph, we investigate the abstract degenerate Volterra integro-differential equations in sequentially complete locally convex spaces by using multivalued linear operators and vector-valued Laplace transform. We follow the method which is based on the use

of (a, k)-regularized *C*-resolvent families generated by multivalued linear operators (cf. [366, 367, 421, 423, 424] and [482] for some special cases of this notion) and which suggests a very general way of approaching abstract Volterra equations. We also introduce and analyze the class of (a, k)-regularized (C_1, C_2) -existence and uniqueness families. The results presented in Chapter 3, which is composed of thirteen sections, are completely new even for abstract degenerate Caputo fractional inclusions in Banach spaces.

In Section 3.1, we analyze the following abstract degenerate Volterra inclusion:

(12)
$$\mathcal{B}u(t) \subseteq \mathcal{A} \int_0^t a(t-s)u(s)ds + \mathcal{F}(t), \quad t \in [0,\tau),$$

where $a \in L^1_{loc}([0, \tau))$, $a \neq 0$, $\mathcal{A}: X \to P(Y)$ and $\mathcal{B}: X \to P(Y)$ are given multivalued linear operators acting between sequentially complete locally convex spaces Xand Y, and $\mathcal{F}: X \to P(Y)$ is a given mutivalued mapping, as well as the following fractional Sobolev inclusions:

$$(DFP)_{\mathbf{R}} : \begin{cases} \mathbf{D}_t^{\alpha} Bu(t) \in \mathcal{A}u(t) + \mathcal{F}(t), & t \ge 0, \\ (Bu)^{(j)}(0) = Bx_j, & 0 \le j \le \lceil \alpha \rceil - 1, \end{cases}$$

where we assume that $B = \mathcal{B}$ is single-valued, and

$$(DFP)_{\mathbf{L}} : \begin{cases} \mathcal{B}\mathbf{D}_t^{\alpha}u(t) \subseteq \mathcal{A}u(t) + \mathcal{F}(t), & t \ge 0, \\ u^{(j)}(0) = x_j, & 0 \le j \le \lceil \alpha \rceil - 1. \end{cases}$$

We define various types of solutions of problems (12), $(DFP)_{\mathbf{R}}$ and $(DFP)_{\mathbf{L}}$. In Theorem 3.1.3 and Theorem 3.1.5, we reconsider the main results of research of M. Kim [280], while in Theorem 3.1.6 we prove an extension of [285, Theorem 3.5] for abstract degenerate fractional differential inclusions. Subordination principles are clarified in Theorem 3.1.8 and Theorem 3.1.9 following the methods proposed by J. Prüss [459, Section 4] and E. Bazhlekova [61, Section 3] (cf. [210, 314] and [327] for similar results known in the degenerate case).

Following the ideas of R. deLaubenfels [132], in Section 3.2 we introduce and analyze the class of (a, k)-regularized (C_1, C_2) -existence and uniqueness families (see [292, Section 2.8] for non-degenerate case). Later on, we single out the class of (a, k)-regularized C-resolvent families for special considerations. We focus our attention on the analysis of Hille–Yosida's type theorems for (a, k)-regularized Cresolvent families generated by multivalued linear operators (as in all previous researches of non-degenerate case, we introduce the notion of a subgenerator of an (a, k)-regularized C-resolvent family and investigate the most important properties of subgenerators). It is well known (see e.g. [199, Theorem 2.4], [285, Theorem 3.6] and [280, p. 169]) that Hille–Yosida's type estimates for the resolvent of a multivalued operator \mathcal{A} immediately implies that \mathcal{A} is single-valued in a certain sense. In part (ii) of Theorem 3.2.12, we will prove a similar assertion provided that the Hille–Yosida condition (288) below holds. For the validity of Theorem 3.2.12(ii), we have found the condition $k(0) \neq 0$ very important to be satisfied; in other words, the existence of above-mentioned single-valued branch of \mathcal{A} can be proved exactly in non-convoluted or non-integrated case, so that we have arrived

to a diametrically opposite conclusion to that stated on l. 7-13, p. 169 of [280]. Nevertheless, the existence of non-existence of such a single-valued branch of \mathcal{A} is not sufficient for obtaining a fairly complete information on the well-posednesss of inclusion (12) with $\mathcal{B} = I$ (the reading of papers [280] and [285] has strongly influenced us to write [312], which contains a great part of our results presented in Section 3.1–Section 3.2: compared with the results of [280], here we do not need the assumption that a(t) is a normalized function of local bounded variation). In the remainder of Section 3.2, we enquire into the possibility to extend the most important results from [292, Section 2.1, Section 2.2] to (a, k)-regularized C-resolvent families generated by multivalued linear operators, and present several examples and possible applications of our abstract theoretical results. We clarify the complex characterization theorem for the generation of exponentially equicontunuous (a, k)-regularized C-resolvent families, the generalized variation of parameters formula, and subordination principles; in a separate subsection, we analyze differential and analytical properties of (a, k)-regularized C-resolvent families. Furthermore, we investigate the following degenerate Volterra integral inclusion:

$$0 \in \mathcal{B}u(t) + \sum_{j=0}^{n-1} \mathcal{A}_j(a_j * u)(t) + \mathcal{F}(t), \quad t \in [0, \tau),$$

where $n \in \mathbb{N}, 0 < \tau \leq \infty, \mathcal{F}: [0, \tau) \to P(Y), a_0, \cdots, a_{n-1} \in L^1_{loc}([0, \tau))$, and $\mathcal{A} \equiv \mathcal{A}_0, \cdots, \mathcal{A}_{n-1}, \mathcal{B} \equiv \mathcal{A}_n$ are multivalued linear operators acting between the sequentially complete locally convex spaces X and Y ([327]). In Subsection 3.2.2, we analyze the situation in which some of the regularizing operators C_2 or C is not injective, and introduce the notion of an (a, k, C)-subgenerator of arbitrary strongly continuous operator family $(Z(t))_{t\in[0,\tau)} \subseteq L(X)$, where $0 < \tau \leq \infty$. The main aim of Subsection 3.2.3 is to clarify the most important results about degenerate K-convoluted C-semigroups and degenerate K-convoluted C-cosine functions in locally convex spaces. In this subsection, degenerate operator families under examination are defined locally or globally and their subgenerators are allowed to be multivalued linear operators. We thus provide a new unification concept in the theory of abstract degenerate differential equations of first and second order, paying special attention to explain, in a brief and concise manner, how we can improve our structural results from the second chapter of monograph [291] to degenerate operator families. We analyze the basic properties of subgenerators, extension and adjoint type theorems, real and complex characterization theorems, as well as generation of local degenerate K-convoluted C-semigroups (K-convoluted C-cosine functions). Following P. C. Kunstmann [363], we introduce the notion of stationarity of a multivalued linear operator and give some estimates on the upper bounds for stationarity of generators of degenerate fractionally integrated semigroups and cosine functions. The study of degenerate K-convoluted C-groups is without the scope of this monograph.

It is well known that the class of distribution semigroups in Banach spaces was introduced by J. L. Lions [391] in 1960 as an attempt to seek for the solutions of abstract first order differential equations that are not well-posed in the usual sense,

i.e., whose solutions are not governed by strongly continuous semigroups of linear operators. From then on, distribution semigroups have attracted the attention of a large number of mathematicians. The class of distribution semigroups with not necessarily densely defined generators has been introduced independently by P. C. Kunstmann [364] and S. W. Wang [530], while the class of C-distribution semigroups has been introduced by the author in [332]. Ultradistribution semigroups in Banach spaces, with densely or non-densely defined generators, and abstract Beurling spaces have been analyzed in the papers of R. Beals [64, 65]. J. Chazarain [99], I. Ciorănescu [109], H. A. Emami-Rad [167] and H. Komatsu [289] (cf. also [291, 351, 365] and [424]). On the other hand, the study of distribution semigroups in locally convex spaces has been initiated by R. Shiraishi, Y. Hirata [488], T. Ushijima [522] and M. Ju Vuvunikjan [528]. In our recent joint research study with S. Pilipović and D. Velinov [354], we have introduced and systematically analyzed the classes of C-distribution semigroups and C-ultradistribution semigroups in locally convex spaces. The main aim of Section 3.3 is to explore the classes of degenerate C-distribution semigroups and degenerate C-ultradistribution semigroups in locally convex spaces [352,353]. Section 3.4 is devoted to the study of degenerate C-distribution cosine functions and degenerate C-ultradistribution cosine functions in locally convex spaces (cf. [331, 333, 348] and [428] for single-valued case). Even in non-degenerate case, with C being the identity operator and the pivot space being one of Banach's, the results presented in Section 3.4 are completely new in ultradistributional case [523].

In Section 3.5, we continue the analysis of A. Favini, A. Yagi [199, Chapter III] and numerous other authors by investigating subordinated fractional resolvent families with removable singularities at zero and semilinear degenerate fractional differential inclusions [324, 325]. Our main contributions are contained in Subsection 3.5.2, where we also analyze a new class of abstract relaxation differential equations that are not degenerate in time.

The main ideas and organization of Section 3.6, where we reconsider our previously established results on hypercyclic and topologically mixing properties of abstract degenerate (multi-term) time-fractional equations by using the multivalued linear operators approach [321], will be explained within themselves. Let us only note, for now, that this section is consisted from four subsections and that an interesting application has been made in the study of topologically mixing properties of the Poisson heat equation in $C^2(\mathbb{R})$. Chronologically, G. A. Sviridyuk and N. A. Manakova were the first to investigate perturbations of a class of abstract degenerate differential equations of first order [510], 2003. Using the perturbation theory for strongly continuous semigroups and the theory developed by G. A. Sviridyuk [507], V. E. Fedorov and O. A. Ruzakova have analyzed in [219] the unique solvability for the Cauchy problem and Showalter problem for a class of perturbations of abstract degenerate differential equations of first order. The paper [219] contains a great number of applications to initial boundary value problems and we can freely say that this is the first systematic study of perturbations of abstract degenerate differential equations. Recently, A. Favini [186] has looked

into inverse problems of degenerate differential equations by using perturbation results for linear relations (cf. also M. A. Horani, A. Favini [15] and A. Favaron, A. Favini, H. Tanabe [184]). The main aim of Section 3.7 is to reconsider perturbation results for abstract non-degenerate Volterra integro-differential equations [292, Section 2.6] from the point of view of the theory of multivalued linear operators [323]. We provide several illustrative applications of our results, primarily to degenerate fractional differential equations with Caputo derivatives. We also provide a few instructive examples emphasizing that certain perturbation properties of abstract degenerate Volterra integro-differential equations can be analyzed by using the results from the perturbation theory for non-degenerate equations.

Approximation theory is an established field of mathematics whose focus is primarily on the approximation of real-valued continuous functions by some simpler class of functions. Concerning the theory of strongly continuous semigroups, of particular interest is the well-known Trotter–Kato theorem, which makes a relationship between the convergence of a sequence of infinitesimal generators (their resolvents) to the convergence of associated semigroups of operators. Briefly told, the main aim of Section 3.8 [**316**] is to prove some Trotter–Kato type formulae for degenerate (a, k)-regularized *C*-resolvent families in locally convex spaces (cf. [**25**, **50**, **79**, **95**, **187**, **199**, **370**, **509**] and [**519**] for the basic references on approximation of abstract degenerate differential equations), as well as to investigate the Laguerre expansions of degenerate (a, k)-regularized *C*-resolvent families (cf. [**2**] for the study of Laguerre expansions of non-degenerate strongly continuous semigroups in Banach spaces).

The first results about fractional powers of non-negative multivalued linear operators was given by El H. Alaarabiou [11,12] in 1991. In these papers, he extended the well known Hirsch functional calculus to the class \mathcal{M} of non-negative multivalued linear operators in a complex Banach space. Unfortunately, the method proposed in [11, 12] had not allowed one to consider the product formula and the spectral mapping theorem for powers. Nine years later, in 2000, C. Martínez, M. Sanz and J. Pastor [411] improved a functional calculus established in [11, 12], providing a new definition of fractional powers. A very stable and consistent theory of fractional powers of the operators belonging to the class \mathcal{M} has been constructed, including within itself the above-mentioned product formula, spectral mapping theorem, as well as almost all other fundamental properties of fractional powers of non-negative single-valued linear operators. Some later contributions have been given by J. Pastor [444], who considered relations between the multiplicativity and uniqueness of fractional powers of non-negative multivalued linear operators.

In order to motivate our research in Section 3.9, let us first look into the class consisting of multivalued linear operators \mathcal{A} , acting on a complex Banach space $(X, \|\cdot\|)$, for which $(-\infty, 0] \subseteq \rho \mathcal{A}$ and there exist finite numbers $M_1 \ge 1, \beta \in (0, 1]$ such that

(13)
$$||R(\lambda : \mathcal{A})|| \leq M_1 (1+|\lambda|)^{-\beta}, \quad \lambda \leq 0.$$

Assuming that (13) is true, we can apply the usual von Neumann's expansion in order to see that there exist positive real constants c > 0 and M > 0 such that

the resolvent set of \mathcal{A} contains an open region $\Omega_{c,M} \supseteq \Omega'_{c,M} := \{\lambda \in \mathbb{C} : |\operatorname{Im} \lambda| \leq (2M)^{-1}(c - \operatorname{Re} \lambda)^{\beta}, \operatorname{Re} \lambda \leq c\}$, where we have the estimate $||R(\lambda : \mathcal{A})|| = O((1 + |\lambda|)^{-\beta}), \lambda \in \Omega_{c,M}$. Let Γ' be the upwards oriented curve $\{\xi \pm i(2M)^{-1}(c - \xi)^{\beta} : -\infty < \xi \leq c\}$. In [199], cf. also Subsection 1.2.1, A. Favini and A. Yagi define the fractional power $\mathcal{A}^{-\theta}$, for $\operatorname{Re} \theta > 1 - \beta$, by

$$\mathcal{A}^{-\theta} := \frac{1}{2\pi i} \int_{\Gamma'} \lambda^{-\theta} (\lambda - \mathcal{A})^{-1} d\lambda,$$

 $\begin{aligned} \mathcal{A}^{\theta} &:= (\mathcal{A}^{-\theta})^{-1} \; (\operatorname{Re} \theta > 1 - \beta); \; \operatorname{then} \; \mathcal{A}^{-\theta} \in L(E) \; \operatorname{for} \; \operatorname{Re} \theta > 1 - \beta, \; \operatorname{and} \; \operatorname{the} \\ &\operatorname{semigroup} \; \operatorname{properties} \; \mathcal{A}^{-\theta_1} \mathcal{A}^{-\theta_2} = \mathcal{A}^{-(\theta_1 + \theta_2)}, \; \mathcal{A}^{\theta_1} \mathcal{A}^{\theta_2} = \mathcal{A}^{\theta_1 + \theta_2} \; \operatorname{of} \; \operatorname{powers} \; \operatorname{hold} \; \operatorname{for} \\ &\operatorname{Re} \theta_1, \operatorname{Re} \theta_2 > 1 - \beta. \; \operatorname{The} \; \operatorname{case} \; \beta \in (0, 1) \; \operatorname{occurs} \; \operatorname{in} \; \operatorname{many} \; \operatorname{applications} \; \operatorname{and} \; \operatorname{then} \; \operatorname{we} \\ &\operatorname{cannot} \; \operatorname{define} \; \operatorname{satisfactorily} \; \operatorname{the} \; \operatorname{power} \; \mathcal{A}^{\theta} \; \operatorname{for} \; | \operatorname{Re} \theta | \leq 1 - \beta. \; \operatorname{As} \; \operatorname{explained} \; \operatorname{in} \; \operatorname{the} \\ &\operatorname{introductory} \; \operatorname{part} \; \operatorname{of} \; \operatorname{paper} \; [\mathbf{184}] \; \operatorname{by} \; \operatorname{A}. \; \operatorname{Favaron} \; \operatorname{and} \; \operatorname{A}. \; \operatorname{Favini}, \; \operatorname{the} \; \operatorname{method} \; \operatorname{of} \; \operatorname{closed} \\ &\operatorname{extensions} \; \operatorname{used} \; \operatorname{in} \; \operatorname{the} \; \operatorname{pioneering} \; \operatorname{works} \; [\mathbf{44}] \; \operatorname{by} \; \operatorname{A}. \; \operatorname{Balakrishnan} \; \operatorname{and} \; [\mathbf{290}] \; \operatorname{by} \; \operatorname{H}. \\ &\operatorname{Komatsu} \; \operatorname{cannot} \; \operatorname{be} \; \operatorname{used} \; \operatorname{here} \; \operatorname{for} \; \operatorname{construction} \; \operatorname{of} \; \operatorname{power} \; \mathcal{A}^{\theta} \; (\operatorname{Re} \; \theta \in (0, 1 - \beta)). \; \operatorname{In} \\ &\operatorname{this} \; \operatorname{place}, \; \operatorname{we} \; \operatorname{would} \; \text{like} \; \operatorname{to} \; \operatorname{observe} \; \operatorname{that} \; \operatorname{the} \; \operatorname{method} \; \operatorname{proposed} \; \operatorname{by} \; \operatorname{F}. \; \operatorname{Periago}, \; \operatorname{B}. \\ &\operatorname{Straub} \; [\mathbf{447}] \; \operatorname{and} \; \operatorname{C}. \; \operatorname{Martinez}, \; \operatorname{M}. \; \operatorname{Sanz}, \; \operatorname{A}. \; \operatorname{Redondo} \; [\mathbf{413}] \; (\operatorname{cf.} \; \operatorname{also} \; [\mathbf{101}]) \; \operatorname{cannot} \; \operatorname{be} \\ &\operatorname{of} \; \operatorname{any} \; \operatorname{help} \; \operatorname{for} \; \operatorname{construction} \; \operatorname{of} \; \operatorname{power} \; \mathcal{A}^{\theta} \; \; \operatorname{has} \; \operatorname{been} \; \operatorname{constructed} \; &\operatorname{for} \; | \operatorname{Re} \; \theta | \leq 1 - \beta, \; \operatorname{provided} \; \\ \\ &\operatorname{validity} \; \operatorname{of} \; \operatorname{condition} \; [\mathbf{184}, \; (\operatorname{H3})]. \; \operatorname{In} \; \\ \\ &\operatorname{poh} \; \operatorname{has} \; \operatorname{been} \; \operatorname{constructed} \; \\ \\ &\operatorname{strue} \; \beta \in (0, 1), \; \\ \\ &\operatorname{the} \; \operatorname{condition} \; (\operatorname{H3}) \; \\ \\ \\ &\operatorname{the} \; \operatorname{and} \; \operatorname{condition} \; (\operatorname{H3}) \; \\ \\ \\ &\operatorname{the} \; \operatorname{the} \; \operatorname{the} \; \operatorname{condition} \; (\operatorname{H3}) \; \\ \\ \\ &\operatorname{the} \; \operatorname{the} \; \\ \\ \\ &\operatorname{the} \; \operatorname{the} \; \operatorname{the$

Assume now that $\alpha \ge -1$ and a closed multivalued linear operator \mathcal{A} satisfies:

 $(\Diamond) \ (0,\infty) \subseteq \rho(\mathcal{A})$ and

 $(\Diamond \Diamond) \sup_{\lambda > 0} (1 + |\lambda|)^{-\alpha} ||R(\lambda : \mathcal{A})|| < \infty.$

Given $\beta \ge -1$, $\varepsilon \in (0, 1]$, $d \in (0, 1]$, $c' \in (0, 1)$ and $\theta \in (0, \pi]$, put $B_d := \{z \in \mathbb{C} : |z| \le d\}$, $\Sigma_{\theta} := \{z \in \mathbb{C} : z \ne 0, \arg(z) \in (-\theta, \theta)\}$ and $P_{\beta,\varepsilon,c'} := \{\xi + i\eta : \xi \ge \varepsilon, \eta \in \mathbb{R}, |\eta| \le c'(1+\xi)^{-\beta}\}$. Then is checked at once that the hypotheses $(\diamondsuit) - (\diamondsuit \diamondsuit)$ imply the existence of numbers $d \in (0, 1]$, $c \in (0, 1)$, $\varepsilon \in (0, 1]$ and M > 0 such that:

- (§) $P_{\alpha,\varepsilon,c} \cup B_d \subseteq \rho(\mathcal{A}), \ (\varepsilon, c(1+\varepsilon)^{-\alpha}) \in \partial B_d$ and
- (§§) $||R(\lambda : \mathcal{A})|| \leq M(1+|\lambda|)^{\alpha}, \ \lambda \in P_{\alpha,\varepsilon,c} \cup B_d.$

Suppose now that X is a Hausdorff sequentially complete locally convex space over the field of complex numbers, SCLCS for short. Keeping in mind the above analysis (the notion will be explained later), it seems reasonable to introduce the following condition:

(H)₀: Let $C \in L(X)$ be not necessarily injective, let \mathcal{A} be closed, and let $C\mathcal{A} \subseteq \mathcal{A}C$. There exist real numbers $d \in (0,1]$, $c \in (0,1)$, $\varepsilon \in (0,1]$ and $\alpha \ge -1$ such that $P_{\alpha,\varepsilon,c} \cup B_d \subseteq \rho_C(\mathcal{A})$, the operator family $\{(1 + |\lambda|)^{-\alpha}(\lambda - \mathcal{A})^{-1}C : \lambda \in P_{\alpha,\varepsilon,c} \cup B_d\} \subseteq L(X)$ is equicontinuous, the mapping $\lambda \mapsto (\lambda - \mathcal{A})^{-1}C$ is strongly analytic on $\operatorname{int}(P_{\alpha,\varepsilon,c} \cup B_d)$ and strongly continuous on $\partial(P_{\alpha,\varepsilon,c} \cup B_d)$.

The first aim of Subsection 3.9.1 is to construct the complex power $(-\mathcal{A})_b$, $b \in \mathbb{C}$ of a multivalued linear operator \mathcal{A} satisfying the condition $(H)_0$. Although very elegant and elementary, our construction has some serious disadvantages because the introduced powers behave very badly (for example, we cannot expect

the additivity property of powers clarified in [103, Remark 2.11]) in the case that the regularizing operator C_1 , defined in this subsection, is not injective (since the resolvents and *C*-resolvents of a really multivalued linear operator are not injective, this is the main case in our examinations). The method proposed for construction of power $(-\mathcal{A})_b$ is different from that already employed in the single-valued linear case [334]; in Subsection 3.9.1, we first apply regularization with the operator C_1 and follow after that the approach from our joint research paper with C. Chen, M. Li and M. Žigić [103]. In particular, we define any complex power of a multivalued linear operator satisfying (13) and not the above-mentioned condition (H3).

The following sectorial analogue of (H) is most important in applications:

(HS)₀ : Let $C \in L(X)$ be not necessarily injective, let \mathcal{A} be closed, and let $C\mathcal{A} \subseteq \mathcal{A}C$. There exist real numbers $d \in (0,1]$, $\vartheta \in (0,\pi/2)$ and $\alpha \ge -1$ such that $\Sigma_{\vartheta} \cup B_d \subseteq \rho_C(\mathcal{A})$, the operator family $\{(1+|\lambda|)^{-\alpha}(\lambda-\mathcal{A})^{-1}C : \lambda \in \Sigma_{\vartheta} \cup B_d\} \subseteq L(X)$ is equicontinuous, the mapping $\lambda \mapsto (\lambda - \mathcal{A})^{-1}C$ is strongly analytic on $\operatorname{int}(\Sigma_{\vartheta} \cup B_d)$ and strongly continuous on $\partial(\Sigma_{\vartheta} \cup B_d)$.

The construction of power $(-\mathcal{A})_b, b \in \mathbb{C}$ of a multivalued linear, non-sectorial, operator \mathcal{A} for which $0 \notin \operatorname{int}(\rho_C(\mathcal{A}))$ is not trivial and we will not discuss this theme here. For some other approaches concerning the construction of fractional powers, the reader may consult [101, 138] and [502].

In Section 3.9, we will also see that a great number of resolvent equations and generalized resolvent equations clarified in Section 1.2 holds for *C*-resolvents of multivalued linear operators, where *C* is non-injective, in general. The third and, probably, the main aim of Section 3.9 is to continue our researches of abstract incomplete fractional degenerate differential equations with modified Liouville right-sided fractional derivatives [**292**] and abstract incomplete degenerate differential equations of second order (cf. Section 2.7). We investigate fractionally integrated C_1 -regularized semigroups generated by the negatives of introduced powers, and provide a few relevant applications of our theoretical results to abstract incomplete degenerate PDEs.

Chronologically speaking, R. deLaubenfels proved in 1988 that any injective infinitesimal generator A of a bounded analytic C_0 -semigroup in Banach space Ehas the property that the inverse operator A^{-1} also generates a bounded analytic C_0 -semigroup of the same angle [133]. In this paper, the author asked whether any injective infinitesimal generator A of a bounded C_0 -semigroup in E has the property that the inverse operator A^{-1} generates a C_0 -semigroup. As we know today, the answer is negative in general: a simple counterexample is given by H. Komatsu already in 1966, who constructed an injective infinitesimal generator of a contraction semigroup on the Banach space c_0 , for which the inverse operator is not an infinitesimal generator of a C_0 -semigroup [233,290]. Concerning the inverse generator problem, it should be noted that A. Gomilko, H. Zwart and Y. Tomilov proved in 2007 that the answer to R. deLaubenfels's question is negative in the state space l^p , where $1 and <math>p \neq 2$ (see [233]), as well as that S. Fackler proved in 2016 that the answer to this question is negative in the state space $L^p(\mathbb{R})$, where $1 and <math>p \neq 2$ (see [172]). We do not yet know whether there exists

an injective infinitesimal generator A of a bounded C_0 -semigroup in a Hilbert space H such that A^{-1} does not generate a C_0 -semigroup in H (it is well known that for any injective infinitesimal generator A of a contraction C_0 -semigroup in H, the inverse operator A^{-1} likewise generates a contraction C_0 -semigroup in H by the Lumer-Phillips theorem; see the paper [**394**] by R. Liu for the fractional analogue of this result). For further information about the inverse generator problem, we refer the reader to the papers [**135**] by R. deLaubenfels, [**162**] by T. Eisner, H. Zwart, [**563**, **564**] by H. Zwart and the recent survey [**232**] by A. Gomilko.

We would like to note that the complexity of inverse generator problem lies also in the fact that the use of real or complex representation theorems for the Laplace transform does not take a satisfactory effect. To explain this in more detail, assume that A is injective and generates a bounded C_0 -semigroup in the Banach space Eequipped with the norm $\|\cdot\|$. Then a simple calculation involving the Hille–Yosida theorem yields that, for every $\lambda > 0$ and $n \in \mathbb{N}$, we have

$$\begin{aligned} \frac{d^n}{d\lambda^n} [(\lambda - A^{-1})^{-1}] &= (-1)^n n! [\lambda^{-1} - \lambda^{-2} (\lambda^{-1} - A)^{-1}]^{n+1} \\ &= (-1)^n n! \sum_{k=0}^{n+1} (-1)^k \binom{n+1}{k} \lambda^{-(n+1+k)} (\lambda^{-1} - A)^{-k}, \end{aligned}$$

so that

$$\left\|\frac{d^n}{d\lambda^n}[(\lambda-A^{-1})^{-1}]\right\| \leqslant \frac{Mn!}{\lambda^{n+1}} \cdot 2^{n+1}.$$

Since the multiplication with number 2^{n+1} has appeared above, this estimate is completely useless if one wants to prove that the operator A^{-1} generates an exponentially bounded r-times integrated semigroup for some real number $r \ge 0$ (see also [232, Proposition 3.1, Theorem 3.2, Theorem 3.4]). On the other hand, a simple computation shows that the resolvent of A^{-1} is bounded in norm by $\text{Const} \cdot |\lambda|$ on any right half plane $\{z \in \mathbb{C} : \text{Re } z > a\}$, where a > 0, so that the complex characterization theorem for the Laplace transform immediately yields that the operator A^{-1} generates an exponentially bounded r-times integrated semigroup for any real number r > 2. In 2007, S. Piskarev and H. Zwart proved that the operator A^{-1} generates an exponentially bounded once integrated semigroup [449], while M. Li, J. Pastor and S. Piskarev improved this result in 2018 by showing that the operator A^{-1} generates a tempered r-times integrated semigroup for any real number r > 1/2. Moreover, they formulated a corresponding result for tempered fractional resolvent operator families of order $\alpha \in (0, 2]$; see [383] for more details.

Up to now, we do not have any relevant reference which treats the inverse generator problem for abstract degenerate Volterra integro-differential equations, even for abstract degenerate differential equations of first order. In contrast to non-degenerate differential equations, we have found the inverse generator problem much more important from the point of view of possible applications. The main aim of Section 3.10 is to analyze the inverse generator problem for the classes of mild (a, k)-regularized (C_1, C_2) -existence and uniqueness families, where the operator C_2 is not necessarily injective, and (a, k)-regularized C-resolvent families, where the operator C is not necessarily injective [**343**]; see, especially, Subsection 3.10.1,

where we investigate degenerate time-fractional equations with abstract differential operators. We reconsider some results from [134] and [383] for abstract degenerate Volterra integro-differential equations in Theorem 3.10.9 and Theorem 3.10.11. We also investigate the situation in which we do not assume the existence of C-resolvent of a corresponding multivalued linear operator \mathcal{A} on some right half plane. Moreover, we observe that the existence of C-resolvent set of \mathcal{A} at infinity does not play any role for the generation of certain classes of (a, k)-regularized C-resolvent operator families, C-(ultra)distribution semigroups and C-(ultra)distribution cosine functions by the inverse of a closed multivalued linear operator \mathcal{A} ; sometimes it is crucial to investigate the behaviour of C-resolvent set of \mathcal{A} around zero, only (see e.g. Proposition 3.10.5 and Example 3.10.8).

The theory of almost periodic and almost automorphic type functions is unavoidable nowdays. The existing literature on various types of almost periodic (automorphic) properties and asymptotically almost periodic (automorphic) properties of abstract non-degenerate Volterra integro-differential equations in Banach spaces is enormous. For a basic source of information in this direction, we refer the reader to [20, 87, 93, 119, 256, 274, 434, 435, 478] and [501]. Concerning degenerate case, mention should be made of papers by V. Barbu-A. Favini [49], A. Favini-G. Marinoschi [191], S. Q. Bu [91, 92], N. T. Lan [371], C. Lizama-R. Ponce [396, 397] and R. Ponce [455].

The most relevant details of research studies [49,191] and [396] are described as follows. In [49], V. Barbu and A. Favini have considered 1-periodic solutions of abstract degenerate differential equation $(d/dt)(Bu(t)) = Au(t), t \ge 0$, subjected with initial condition (Bu)(0) = (Bu)(1), by using P. Grisvard's sum of operators method and some results from investigation of J. Prüss [461] in non-degenerate case (here, A and B are closed linear operators acting on a Banach space X). The authors have used the multivalued linear approach to degenerate differential equations (cf. the next chapter for further information) and reduce the above problem to $v'(t) \in \mathcal{A}v(t), t \ge 0, v(0) = v(1)$, where the multivalued linear operator \mathcal{A} is defined by $\mathcal{A} = AB^{-1}$. The main problem in the whole analysis is the question of whether the inclusion $1 \in \rho(\mathcal{A})$ holds or not; in connection with this, we would like to remind ourselves [461] that $1 \in \rho(A)$ iff $2\pi i \mathbb{Z} \subseteq \rho(A)$ and $\sup\{\|(2\pi i n - A)^{-1}\| : n \in \mathbb{Z}\} < \infty$, provided that A generates a non-degenerate strongly continuous semigroup. Applications are given to the Poisson heat equation in $H^{-1}(\Omega)$ and $L^{2}(\Omega)$, as well as to some systems of ordinary differential equations. In [191], A. Favini and G. Marinoschi have continued the analysis from [49] by assuming the possible non-linearity of (multi-valued) operator A. C. Lizama and R. Ponce [396] have applied the Fourier multipliers techniques for establishment the necessary and sufficient conditions for the existence of 2π -periodic solutions to the following abstract inhomogeneous linear equation

(14)
$$\frac{d}{dt}(Bu(t)) = Au(t) + \int_{-\infty}^{t} a(t-s)Au(s)ds + f(t), \quad t \ge 0,$$

subjected with the initial condition $(Bu)(0) = (Bu)(2\pi)$. Here, $f: [0, \infty) \to X$ is a 2π -periodic function and $a \in L^1([0, \infty))$ is a scalar-valued kernel on $[0, \infty)$. The authors also showed some results on the maximal regularity of (14) in periodic Besov, Triebel–Lizorkin and Lebesgues vector-valued function spaces.

For a basic source of information about almost automorphic solutions of abstract non-degenerate integro-differential equations, the reader may consult the monograph [293], as well as the papers [7, 9, 10, 124, 125, 146, 147] and [239]. In [293], I have investigated the almost periodic and almost automorphic type solutions of various classes of abstract degenerate Volterra integro-differential equations and abstract degenerate fractional differential equations in Banach spaces, as well. This is probably the first research monograph regarding this problematic. After the final release of the monograph [293], I have written several new research papers in this field. Two of them, [148] and [149], have been completed in co-autorship with Professor T. Diagana. The material of [148] is presented in Section 3.12, while the material of [149] will appear as a chapter in the forthcoming edited book "Recent Studies in Differential Equations" by Nova Science Publishers, Inc. (New York). It is worth noting here that the concept of almost periodicity (respectively, almost automorphy, pseudo-almost periodicity, and pseudo-almost automorphy) in the Lebesgue space with variable exponent $L^{p(x)}(I, X)$ was first introduced and studied by T. Diagana and M. Zitane [150, 151]. However, the translation-invariance of these newly introduced spaces depends heavily upon the function $p \in C([0,\infty))$. To remove such a restriction, in Section 3.12 we introduce some new concepts so that the obtained almost periodic (respectively, asymptotically almost periodic) in $L^{p(x)}(I,X)$ are automatically translation-invariant. Among other things, it will be shown that these new functions generalize in a natural fashion the classical notion of almost periodicity (respectively, asymptotic almost periodicity). Many properties of the new functions are analyzed including their compositions. Further, we will make extensive use of these new functions to study some abstract Volterra integro-differential equations in Banach spaces including multi-valued ones.

In Section 3.11, we present the material from my recent paper [344] concerning quasi-asymptotically almost periodic functions in Banach spaces and related applications. The organization and main ideas of this section are given as follows. The concept introduced by H. Weyl [534] suggests a very general way of approaching almost periodicity. To the best knowledge of the author, the question whether the class of asymptotically Stepanov p-almost periodic functions, introduced by H. R. Henríquez [249], is contained in the class of Weyl-p-almost periodic functions taken without any ergodic components, has not been examined elsewhere by now. In this section, we introduce the class of Stepanov p-quasi-asymptotically almost periodic functions and prove later that this class contains all asymptotically Stepanov palmost periodic functions and make a subclass of the class consisting of all Weyl *p*-almost periodic functions (taken in the sense of A. S. Kovanko's approach [356], which is also followed in the definition of a quasi-asymptotically almost periodic function [336]). In such a way, we initiate the study of generalized (asymptotical) almost periodicity that intermediate Stepanov and Weyl concept. Further on, in Subsection 3.11.1 and Subsection 3.11.2, we recall the basic definitions and results about asymptotically almost periodic type functions, asymptotically almost automorphic type functions and evolution systems, Green's functions, respectively. In

Definition 3.11.9, we recall the notion of a quasi-asymptotically almost periodic (q-aap., for short) function, defined on the interval I, where $I = \mathbb{R}$ or $I = [0, \infty)$. After providing some observations and illustrative examples, in Theorem 3.11.13 we prove that any asymptotically almost automorphic (aaa.) function which is also q-aap. needs to be asymptotically almost periodic (aap.). We present a simple example of a q-aap, function that is uniformly continuous and whose range is not relatively compact in X (this is a simple modification of [251, Example 3.1]); we also show that there exists a q-aap. function that is uniformly continuous and not aap.. The notion of a Stepanov p-q-aap. function is introduced in Definition 3.11.17, while an analogue of Theorem 3.11.18 for Stepanov class has been proved in Theorem 3.11.18. The (Stepanov) class of S-asymptotically ω -periodic functions, introduced by H. R. Henríquez, M. Pierri and P. Táboas in [251], is a subclass of the class consisting of the (Stepanov) class of q-aap. functions (see Proposition 3.11.15). The main structural properties of (Stepanov) q-aap. functions are proved in Theorem 3.11.21 and Proposition 3.11.23. In Example 3.11.24 and Example 3.11.25, we verify that the class of (Stepanov) q-aap. functions is not closed under pointwise products with bounded scalar-valued (Stepanov) q-aap. functions, while in Example 3.11.26 we show that (Stepanov) q-aap. functions do not form vector spaces equipped with the usual operations of addition and multiplication with scalars, unfortunately. Subsection 3.11.4 is devoted to the analysis of (Stepanov) q-aap. functions depending on two parameters and related composition principles. In Theorem 3.11.28 and Theorem 3.11.29, we analyze the composition principles for q-aap, functions depending on two parameters following the approach presented in the monograph of T. Diagana [144] for aap. functions. The main objective in Theorem 3.11.31 and Theorem 3.11.32 is to prove corresponding results for Stepanov q-aap. functions; at these places, we follow the approach of W. Long and S.-H. Ding from [399].

Concerning applications, our main results are given in Subsection 3.11.5, where we analyze the invariance of quasi-asymptotical almost periodicity under the action of convolution products. We do not require the non-degeneracy of solution operator families in Subsection 3.11.5, so that the results established in this subsection can be simply incorporated in the analysis of certain classes of abstract degenerate inhomogeneous fractional inclusions and abstract degenerate inhomogeneous Volterra integro-differential inclusions in Banach spaces. With a view to motivate our researchers for the analyses of evolution systems and abstract quasi-linear differential equations of first order, in Subection 3.11.6 we investigate the existence and uniqueness of q-aap. solutions of abstract (semilinear) nonautonomous differential equations of first order. In Subection 3.11.6, we analyze the abstract nonautonomous differential equations

$$u'(t) = A(t)u(t) + f(t), \quad t \in \mathbb{R}, u'(t) = A(t)u(t) + f(t), \quad t > 0; \ u(0) = x$$

and their semilinear analogues; here, the operator family $A(\cdot)$ is consisted of closed linear operators with domain and range contained in X, the condition (H1) clarified below holds and the evolution system $U(\cdot, \cdot)$ generated by $A(\cdot)$ is hyperbolic, i.e. the condition (H2) clarified below holds. In Theorem 3.11.37 (Theorem 3.11.39), the inhomogenity $f(\cdot)$ is Stepanov *p*-q-aap. and the associated Green's function satisfies the condition (451) ((453)). In contrast to this, in our investigations of semilinear analogues of the abstract Cauchy problems (449) and (450) carried out in Subsection 3.11.7, we assume that the corresponding function $F(\cdot, \cdot)$ is q-aap.. This is essentially caused by the fact that the Stepanov *q*-q-aap. of function $F(\cdot, x(\cdot))$, established in composition principles Theorem 3.11.31 and Theorem 3.11.32, holds only if we additionally assume that the range of function $x(\cdot)$ is relatively compact, which need not be true for q-aap. functions and their Stepanov generalizations. Further study of semilinear nonautonomous differential equations with forcing term $F(\cdot, \cdot)$ belonging to Stepanov class of q-aap. functions is without scope of this book. Finally, in Example 3.11.44, we provide an instructive example of applications of our abstract theoretical results obtained, continuing thus the analyses raised by T. Diagana in [145, Section 4] and the author [338, Example 3.1].

Some notes and appendices to the third chapter are given in Section 3.13.

Some failures of the monograph are described as follows. Subordination principles are very actual and important theme in the theory of abstract Volterra integro-differential equations, degenerate or non-degenerate in time, and we will be imperatively forced to reconsider the statements like Theorem 2.2.6 and Theorem 2.2.13 several times throughout the book; some kind of duplicating, which we have tried to reduce to the minimum level, can be also easily recognized while reading the parts about differential and analytical properties of various types of degenerate operator families of solving operators, analytical properties of C-resolvents of multivalued linear operators, and while introducing the fundamental definitions of hypercyclic and topologically mixing properties of abstract degenerate equations under our consideration. In order to ensure better readability, we have decided to repeat some of the equations mentioned in the introductory part once more, but with different labels.

For the sake of brevity and better exposition, and because of some similarity with our previous researches of non-degenerate case, we have been forced to write the third chapter of monograph in a half-expository manner, including only the most relevant details of proofs of our structural results.

CHAPTER 1

PRELIMINARIES

1.1. Selected topics on vector-valued functions and closed linear operators

Vector-valued functions, closed operators. Unless specified otherwise, by E we denote a Hausdorff sequentially complete locally convex space over the field of complex numbers, SCLCS for short. The abbreviation \circledast stands for the fundamental system of seminorms which defines the topology of E; if E is a Banach space and A is linear operator on E, then the norm of an element $x \in E$ is denoted by ||x||. Assuming that X is another SCLCS, then by L(E, X) we denote the space consisting of all continuous linear mappings from E into X; $L(E) \equiv L(E, E)$. Let \mathcal{B} be the family of bounded subsets of E, let \circledast_X denote the fundamental system of seminorms which defines the topology of X, and let $p_B(T) := \sup_{x \in B} p(Tx)$, $p \in \circledast_X, B \in \mathcal{B}, T \in L(E, X)$. Then $p_B(\cdot)$ is a seminorm on L(E, X) and the system $(p_B)_{(p,B)\in \circledast_X \times \mathcal{B}}$ induces the Hausdorff locally convex topology on L(E, X). Consequently, the Hausdorff locally convex topology on E^* , the dual space of E, defines the system $(|\cdot|_B)_{B\in\mathcal{B}}$ of seminorms on E^* , where $|x^*|_B := \sup_{x\in B} |\langle x^*, x \rangle|$, $x^* \in E^*, B \in \mathcal{B}$. Here \langle , \rangle denotes the duality bracket between E and E^* , sometimes we shall also write $\langle x, x^* \rangle$ or $x^*(x)$ to denote the value of $\langle x^*, x \rangle$. It is well known that the spaces L(E, X) and E^* are sequentially complete provided that E is barreled [419]. By E^{**} we denote the bidual of E. If X and Y are two SCLCSs such that Y is continuously embedded in X, then we write $Y \hookrightarrow X$. A linear operator $A: D(A) \to E$ is said to be closed if the graph of the operator A, defined by $G_A := \{(x, Ax) : x \in D(A)\}$, is a closed subset of $E \times E$. We identify A with its graph if there is no risk for confusion. The resolvent set, spectrum and range of A are denoted by $\rho(A)$, $\sigma(A)$ and R(A), respectively. The null space of A is denoted by either N(A) or Kern(A). As is well known, a necessary and sufficient condition for a linear operator $A: D(A) \to E$ to be closed is that, for every net $(x_{\tau})_{\tau \in I}$ in D(A) such that $\lim_{\tau \to \infty} x_{\tau} = x$ and $\lim_{\tau \to \infty} Ax_{\tau} = y$, the following holds: $x \in D(A)$ and Ax = y [419]. A linear operator A is said to be closable iff there exists a closed linear operator B such that $A \subseteq B$. Let us recall that the closability of the operator A is equivalent to say that, for every net $(x_{\tau})_{\tau \in I}$ in D(A) such that $\lim_{\tau\to\infty} x_{\tau} = x$ and $\lim_{\tau\to\infty} Ax_{\tau} = 0$, we have y = 0 [419]. We assume the closedness of A in the sequel of this section, if not stated otherwise. We introduce the Hausdorff sequentially complete locally convex topology on D(A) $(\overline{D(A)})$ by the following system of seminorms: $p_A(x) =: p(x) + p(Ax), x \in D(A),$

$$\rho_C(A) := \{ \lambda \in \mathbb{C} : \lambda - A \text{ is injective and } (\lambda - A)^{-1} C \in L(E) \}.$$

By the closed graph theorem [419], the following holds: If E is a webbed bornological space (this, in particular, holds if E is a Fréchet space), then any mapping $A: E \to E$ whose graph G(A) is closed in $E \times E$ must be continuous on E, and the C-resolvent set of A consists of those complex numbers λ for which the operator $\lambda - A$ is injective and $R(C) \subseteq R(\lambda - A)$. If F is a linear submanifold of E, then the part of A in F, denoted by $A_{|F}$, is a linear operator defined by $D(A_{|F}) := \{x \in D(A) \cap F : Ax \in F\}$ and $A_{|F}x := Ax, x \in D(A_{|F})$. Suppose $A: D(A) \to E$ is a linear operator. The power A^n of A is defined usually $(n \in \mathbb{N}_0)$. Put $D_{\infty}(A) := \bigcap_{n \ge 1} D(A^n)$. For a closed linear operator A acting on E, we introduce the subset A^* of $E^* \times E^*$ by

$$A^* := \{ (x^*, y^*) \in E^* \times E^* : x^*(Ax) = y^*(x) \text{ for all } x \in D(A) \}.$$

If A is densely defined, then A^* is also known as the adjoint operator of A and it is a closed linear operator on E^* . If $\alpha \in \mathbb{C} \setminus \{0\}$, A and B are linear operators, then we define the operators αA , A + B and AB in the usual way. A family Λ of continuous linear operators on E is said to be equicontinuous if for each $p \in \circledast$ there exist $c_p > 0$ and $q_p \in \circledast$ such that

$$p(Ax) \leq c_p q_p(x), \quad x \in E, \ A \in \Lambda.$$

The Gamma function will be denoted by $\Gamma(\cdot)$ and the principal branch will be always used to take the powers. Set, for every $\alpha > 0$, $g_{\alpha}(t) := t^{\alpha-1}/\Gamma(\alpha)$, t > 0, $g_0(t) \equiv$ the Dirac delta distribution and, by common consent, $0^{\zeta} := 0$. The *n*-th convolution power of a locally integrable function $a(t) \in L^1_{loc}([0,\infty))$ is denoted by $a^{*,n}(t)$. Then $a^{*,n} \in L^1_{loc}([0,\infty))$. Given two numbers $s \in \mathbb{R}$ and $n \in \mathbb{N}$ in advance, set $\lfloor s \rfloor := \sup\{l \in \mathbb{Z} : s \ge l\}, \lceil s \rceil := \inf\{l \in \mathbb{Z} : s \le l\}, \mathbb{N}_n := \{1, \ldots, n\}$ and $\mathbb{N}_n^0 := \{0, 1, \ldots, n\}.$

Function spaces. The Schwartz spaces of test functions $\mathcal{D} = C_0^{\infty}(\mathbb{R})$ and $\mathcal{E} = C^{\infty}(\mathbb{R})$ are equipped with the usual inductive limit topologies; the topology of the space of rapidly decreasing functions \mathcal{S} defines the following system of seminorms

$$p_{m,n}(\psi) := \sup_{x \in \mathbb{R}} |x^m \psi^{(n)}(x)|, \quad \psi \in \mathcal{S}, \ m, n \in \mathbb{N}_0.$$

If $\emptyset \neq \Omega \subseteq \mathbb{R}$, then we denote by \mathcal{D}_{Ω} the subspace of \mathcal{D} consisting of those functions $\varphi \in \mathcal{D}$ for which $\operatorname{supp}(\varphi) \subseteq \Omega$; $\mathcal{D}_0 \equiv \mathcal{D}_{[0,\infty)}$. If $\varphi, \psi \colon \mathbb{R} \to \mathbb{C}$ are measurable functions, then we define the convolution products $\varphi * \psi$ and $\varphi *_0 \psi$ by

$$\varphi * \psi(t) := \int_{-\infty}^{\infty} \varphi(t-s)\psi(s)ds \text{ and } \varphi *_{0} \psi(t) := \int_{0}^{t} \varphi(t-s)\psi(s)ds, \quad t \in \mathbb{R}.$$

Sometimes we will use the symbol * to denote the finite convolution product, if no confusion seems likely. Notice that $\varphi * \psi = \varphi *_0 \psi$, provided that $supp(\varphi)$ and $\operatorname{supp}(\psi)$ are subsets of $[0,\infty)$. Given $\varphi \in \mathcal{D}$ and $f \in \mathcal{D}'$, or $\varphi \in \mathcal{E}$ and $f \in \mathcal{E}'$, we define the convolution $f * \varphi$ by $(f * \varphi)(t) := f(\varphi(t - \cdot)), t \in \mathbb{R}$. For $f \in \mathcal{D}'$, or for $f \in \mathcal{E}'$, define \check{f} by $\check{f}(\varphi) := f(\varphi(-\cdot)), \varphi \in \mathcal{D} \ (\varphi \in \mathcal{E})$. Generally, the convolution of two distributions $f, q \in \mathcal{D}'$, denoted by f * q, is defined by $(f * q)(\varphi) := q(\check{f} * \varphi)$, $\varphi \in \mathcal{D}$. Then we know that $f * q \in \mathcal{D}'$ and $\operatorname{supp}(f * q) \subseteq \operatorname{supp}(f) + \operatorname{supp}(q)$.

The spaces $\mathcal{D}'(E) := L(\mathcal{D}, E), \ \mathcal{E}'(E) := L(\mathcal{E}, E)$ and $\mathcal{S}'(E) := L(\mathcal{S}, E)$ are consisted of all continuous linear functions $\mathcal{D} \to E, \mathcal{E} \to E$ and $\mathcal{S} \to E$, respectively; $\mathcal{D}'_{\Omega}(E)$, $\mathcal{E}'_{\Omega}(E)$ and $\mathcal{S}'_{\Omega}(E)$ denote the subspaces of $\mathcal{D}'(E)$, $\mathcal{E}'(E)$ and $\mathcal{S}'(E)$, respectively, containing E-valued distributions whose supports are contained in Ω ; $\mathcal{D}'_0(E) \equiv \mathcal{D}'_{[0,\infty)}(E), \ \mathcal{E}'_0(E) \equiv \mathcal{E}'_{[0,\infty)}(E), \ \mathcal{S}'_0(E) \equiv \mathcal{S}'_{[0,\infty)}(E).$ In the case that $E = \mathbb{C}$, then the above spaces are also denoted by $\mathcal{D}', \mathcal{E}', \mathcal{S}', \mathcal{D}'_{\Omega}, \mathcal{E}'_{\Omega}, \mathcal{S}'_{\Omega}, \mathcal{D}'_{0}, \mathcal{D}''_{0}, \mathcal{D}'_{0},$ \mathcal{E}'_0 and \mathcal{S}'_0 . Let G be an E-valued distribution, and let $f: \mathbb{R} \to E$ be a locally integrable function (cf. [292, Definition 1.1.4, Definition 1.1.5]). As in the scalarvalued case, we define the *E*-valued distributions $G^{(n)}$ $(n \in \mathbb{N})$ and hG $(h \in \mathcal{E})$; the regular *E*-valued distribution **f** is defined by $\mathbf{f}(\varphi) := \int_{-\infty}^{\infty} \varphi(t) f(t) dt$ $(\varphi \in \mathcal{D})$.

Suppose that $0 < \tau \leq \infty$, $n \in \mathbb{N}$. If $f: (0, \tau) \to E$ is a continuous function and

$$\int_0^\tau \varphi^{(n)}(t) f(t) dt = 0, \quad \varphi \in \mathcal{D}_{(0,\tau)},$$

then we know that there exist elements x_0, \ldots, x_{n-1} in E such that $f(t) = \sum_{j=0}^{n-1} t^j x_j$, $t \in (0, \tau)$. Let $\tau > 0$, and let X be a general Hausdorff locally convex space (not necessarily sequentially complete). Following L. Schwartz [479,480], it will be said that a distribution $G \in \mathcal{D}'(X)$ is of finite order on the interval $(-\tau, \tau)$ iff there exist an integer $n \in \mathbb{N}_0$ and an X-valued continuous function $f: [-\tau, \tau] \to X$ such that

$$G(\varphi) = (-1)^n \int_{-\tau}^{\tau} \varphi^{(n)}(t) f(t) dt, \quad \varphi \in \mathcal{D}_{(-\tau,\tau)}$$

In the case that X is a quasi-complete (DF)-space, then we know from [479, Corollarie 2, p. 90] that each X-valued distribution is of finite order on any finite interval $(-\tau,\tau)$. Furthermore, if $G(\varphi) = 0$ for all $\varphi \in \mathcal{D}_{(-\infty,\tau_1)}$, where $0 < \tau_1 < \tau$, then f(t) can be chosen so that f(t) = 0 for $t < \tau_1$. As it is well known, the above holds not only in quasi-complete (DF)-spaces but also in Banach spaces.

If $(M_p)_{p \in \mathbb{N}_0}$ is a sequence of positive real numbers with $M_0 = 1$, then we use the following conditions from the theory of ultradistributions (Komatsu's approach):

 $\begin{array}{l} (M.1): \ M_p^2 \leqslant M_{p+1}M_{p-1}, \ p \in \mathbb{N}, \\ (M.2): \ M_p \leqslant AH^p \min_{p_1, p_2 \in \mathbb{N}, p_1 + p_2 = p} M_{p_1}M_{p_2}, \ n \in \mathbb{N}, \ \text{for some } A > 1 \ \text{and } H > 1, \\ (M.3)': \ \sum_{p=1}^{\infty} \frac{M_{p-1}}{M_p} < \infty, \ \text{and} \\ (M.3): \ \sup_{p \in \mathbb{N}} \sum_{q=p+1}^{\infty} \frac{M_{q-1}M_{p+1}}{pM_pM_q} < \infty. \end{array}$

Recall that the condition (M.3) is slightly stronger than (M.3)' and that, for every s > 1, the Gevrey sequence $(p!^s)$ satisfies (M.1)-(M.3).

We assume a priori that (M_p) satisfies only the first of these conditions, (M.1); any employment of conditions (M.2), (M.3)' or (M.3) will be explicitly emphasized. The associated function of sequence (M_p) is defined by $M(\rho) := \sup_{p \in \mathbb{N}} \ln \frac{\rho^p}{M_p}$, $\rho > 0$; M(0) := 0, $M(\lambda) := M(|\lambda|)$, $\lambda \in \mathbb{C} \smallsetminus [0, \infty)$. Set $m_p := M_p/M_{p-1}$, $p \in \mathbb{N}$. Then the condition (M.1) implies that the sequence (m_p) is increasing. It is worth noting that the function $t \mapsto M(t)$, $t \ge 0$ is increasing as well as that $\lim_{\lambda \to \infty} M(\lambda) = \infty$ and the function $M(\cdot)$ vanishes in some open neighborhood of zero. Denote by $m(\lambda)$ the number of $m_p \le \lambda$. Since (M_p) satisfies (M.1), it follows that (cf. [**286**, p. 50]) $M(t) = \int_0^t \frac{m(\lambda)}{\lambda} d\lambda$, $t \ge 0$. Hence, the mapping $t \mapsto M(t)$, $t \ge 0$ is absolutely continuous and the mapping $t \mapsto M(t)$, $t \in [0, \infty) \smallsetminus \{m_p : p \in \mathbb{N}\}$ is continuously differentiable with $M'(t) = \frac{m(t)}{t}$, $t \in [0, \infty) \smallsetminus \{m_p : p \in \mathbb{N}\}$. Suppose now that (M_p) satisfies (M.1), (M.2) and (M.3)'. Let us recall that the

Suppose now that (M_p) satisfies (M.1), (M.2) and (M.3)'. Let us recall that the spaces of Beurling, respectively, Roumieu ultradifferentiable functions are defined by $\mathcal{D}^{(M_p)} := \mathcal{D}^{(M_p)}(\mathbb{R}) := \operatorname{indlim}_{K \Subset \Subset \mathbb{R}} \mathcal{D}_K^{(M_p)}$, respectively, $\mathcal{D}^{\{M_p\}} := \mathcal{D}^{\{M_p\}}(\mathbb{R}) := \operatorname{indlim}_{K \Subset \Subset \mathbb{R}} \mathcal{D}_K^{\{M_p\}}$, where $\mathcal{D}_K^{(M_p)} := \operatorname{projlim}_{h \to \infty} \mathcal{D}_K^{M_p,h}$, respectively, $\mathcal{D}_K^{\{M_p\}} := \operatorname{indlim}_{h \to 0} \mathcal{D}_K^{M_p,h}$,

$$\mathcal{D}_{K}^{M_{p},h} := \{ \phi \in C^{\infty}(\mathbb{R}) : \operatorname{supp}(\phi) \subseteq K, \ \|\phi\|_{M_{p},h,K} < \infty \} \text{ and } \\ \|\phi\|_{M_{p},h,K} := \operatorname{sup}\Big\{ \frac{h^{p} |\phi^{(p)}(t)|}{M_{p}} : t \in K, \ p \in \mathbb{N}_{0} \Big\}.$$

Henceforth the asterisk * stands for the Beurling case (M_p) or for the Roumieu case $\{M_p\}$. Let $\emptyset \neq \Omega \subseteq \mathbb{R}$. The space of vector-valued ultradistributions of *-class $\mathcal{D}^{*}(E) := L(\mathcal{D}^*, E)$ is consisted of all continuous linear mappings from \mathcal{D}^* into E; \mathcal{D}^*_{Ω} denotes the subspace of \mathcal{D}^* containing ultradifferentiable functions of *-class whose supports are compact subsets of Ω $(\mathcal{D}^*_0 \equiv \mathcal{D}^*_{[0,\infty)})$, while the symbol \mathcal{E}'^*_{Ω} denotes the space consisting of all scalar valued ultradistributions of *-class whose supports are compact subsets of Ω $(\mathcal{E}'^*_0 \equiv \mathcal{E}'^*_{[0,\infty)})$. Similarly we define the spaces $\mathcal{D}'^*_{\Omega}(E)$ and $\mathcal{D}'^*_0(E)$. An entire function of the form $P(\lambda) = \sum_{p=0}^{\infty} a_p \lambda^p$, $\lambda \in \mathbb{C}$ is of class (M_p) , respectively, of class $\{M_p\}$, if there exist l > 0 and C > 0, respectively, for every l > 0 there exists a constant C > 0, such that $|a_p| \leq Cl^p/M_p$, $p \in$ \mathbb{N} ; cf. [**286**] for further information. The corresponding ultradifferential operator $P(D) = \sum_{p=0}^{\infty} a_p D^p$ is of class (M_p) , respectively, of class $\{M_p\}$. Since (M_p) satisfies (M.2), the ultradifferential operator P(D) of *-class

$$\langle P(D)G,\varphi\rangle := \langle G, P(-D)\varphi\rangle, \quad G \in \mathcal{D}'^*(E), \ \varphi \in \mathcal{D}^*,$$

is a continuous linear mapping from $\mathcal{D}'^*(E)$ into $\mathcal{D}'^*(E)$. The multiplication of a vector-valued ultradistribution G of *-class by a function $a \in \mathcal{E}^*(\Omega)$ is defined as in the scalar case. For every $t \in \mathbb{R}$, we define the scalar-valued (ultra-)distribution $\delta_t \in \mathcal{D}'$ ($\delta_t \in \mathcal{D}'^*$) by $\delta_t(\varphi) := \varphi(t), \varphi \in \mathcal{D} (\varphi \in \mathcal{D}^*)$. For more details about vector-valued (ultra-)distribution spaces used henceforth, see [291, Section 1.3], [292,354, Section 1.1] and the references cited therein.

A function $f: [0,T] \to X$, where $0 < T < \infty$, is said to be Hölder continuous with the exponent $r \in (0,1]$ iff for each $p \in \bigotimes_X$ there exists $M \ge 1$ such that $p(f(t) - f(s)) \le M |t - s|^r$, provided $0 \le t, s \le T$, while a function $f: [0,\infty) \to X$ is said to be locally Hölder continuous with the exponent r iff its restriction on any finite interval [0, T] is Hölder continuous with the same exponent. Define $C^r([0, T])$: X) to be the vector space consisting of Hölder continuous functions $f: [0,T] \to X$ with the exponent r; if $r' \in (0,\infty) \setminus \mathbb{N}$, then we define $C^{r'}([0,T]:X)$ as the vector space consisting of those functions $f: [0,T] \to X$ for which $f \in C^{\lfloor r' \rfloor}([0,T]:X)$ and $f^{\lfloor r' \rfloor} \in C^{r' - \lfloor r' \rfloor}([0,T] : X)$. Assume that X is a Banach space. Then the space of all X-valued functions that are absolutely continuous (of bounded variation) on any closed subinterval of $[0,\infty)$ will be denoted by $AC_{loc}([0,\infty):X)$ $(BV_{loc}([0,\infty):X))$. By $C^k(\Omega:E)$ we denote the space of k-times continuously differentiable functions $(k \in \mathbb{N}_0)$ from a non-empty subset $\Omega \subset \mathbb{C}$ into a general sequentially complete locally convex space $E, C(\Omega : E) \equiv C^0(\Omega : E)$. If $X = \mathbb{C}$, then we also write $AC_{loc}([0,\infty))$ $(BV_{loc}([0,\infty)))$ in place of $AC_{loc}([0,\infty) : X)$ $(BV_{loc}([0,\infty):X))$; the spaces BV[0,T], $BV_{loc}([0,\tau))$, $BV_{loc}([0,\tau):X)$, as well as the space $L^p_{loc}(\Omega: X)$ for $1 \leq p \leq \infty$ are defined in a very similar way $(T, \tau > 0)$; $L^p_{loc}(\Omega) \equiv L^p_{loc}(\Omega : \mathbb{C})$. Let $k \in \mathbb{N}$, let $p \in [1, \infty]$, and let Ω be an open nonempty subset of \mathbb{R}^n . Then the Sobolev space $W^{k,p}(\Omega : X)$, sometimes also denoted by $H^{k,p}(\Omega : X)$, consists of those X-valued distributions $u \in \mathcal{D}'(\Omega : X)$ such that, for every $i \in \mathbb{N}_k^0$ and for every multi-index $\alpha \in \mathbb{N}_0^n$ with $|\alpha| \leq k$, we have $D^{\alpha}u \in L^p(\Omega, X)$. It is needless to say that the derivative D^{α} is taken in the sense of distributions. The subspace of $\mathcal{D}'(\Omega:X)$ consisting of all X-valued distributions of the form

(15)
$$u = \sum_{|\alpha| \leqslant k} u_{\alpha}^{(\alpha)},$$

where $u_{\alpha} \in L^{p}(\Omega : X)$, is denoted by $W^{-k,p}(\Omega : X)$ $(H^{-k,p}(\Omega : X))$. In this space, we introduce the norm

$$||u||_{-k,p,X} := \inf \left\{ \left(\sum_{|\alpha| \leq k} ||u_{\alpha}||_{L^{p}(\Omega,X)}^{p} \right)^{1/p} \right\}$$

where the infimum is taken over all representations of distibution u of form (15); $W^{-k}(\Omega:X) \equiv W^{-k,2}(\Omega:X) \ (H^{-k}(\Omega:X) \equiv H^{-k,2}(\Omega:X))$. It is worth noting that $W^{-k,p}(\Omega:X)$ is a Banach space and $W^{-k}(\Omega:X)$ is a Hilbert space. By $W^{k,p}_{loc}(\Omega:X) \ (H^{k,p}_{loc}(\Omega:X))$ we denote the space of those X-valued distributions $u \in \mathcal{D}'(\Omega:X)$ such that, for every bounded open subset Ω' of Ω , one has $u_{|\Omega'} \in W^{k,p}(\Omega':X)$.

Integration of functions with values in locally convex spaces. By Ω we denote a locally compact and separable metric space, and by μ we denote a locally finite Borel measure defined on Ω .

DEFINITION 1.1.1. (i) It is said that a function $f: \Omega \to E$ is simple iff there exist $k \in \mathbb{N}$, elements $z_i \in E$, $1 \leq i \leq k$ and pairwise disjoint Borel measurable subsets Ω_k , $1 \leq i \leq k$ of Ω , such that $\mu(\Omega_i) < \infty$, $1 \leq i \leq k$ and

(16)
$$f(t) = \sum_{i=1}^{k} z_i \chi_{\Omega_i}(t), \quad t \in \Omega.$$

- (ii) It is said that a function $f: \Omega \to E$ is (strongly) μ -measurable, (strongly) measurable for short, iff there exists a sequence (f_n) in E^{Ω} such that, for every $n \in \mathbb{N}$, $f_n(\cdot)$ is a simple function and $\lim_{n\to\infty} f_n(t) = f(t)$ for a.e. $t \in \Omega$.
- (iii) A function $f: \Omega \to E$ is said to be weakly μ -measurable, weakly measurable for short, iff for every $x^* \in E^*$, the function $t \mapsto x^*(f(t)), t \in \Omega$ is measurable.
- (iv) A function $f: \Omega \to E$ is said to be μ -measurable by seminorms, measurable by seminorms for short, iff for every $p \in \circledast$ there exists a sequence (f_n^p) in E^{Ω} such that $\lim_{n\to\infty} p(f_n^p(t) f(t)) = 0$ a.e. $t \in \Omega$.

It is clear that every strongly measurable function is also weakly measurable and that the converse statement is not true in general. We define the Bochner integral of a simple function $f: \Omega \to E$, given by (16), as follows $\int_{\Omega} f \, d\mu := \sum_{i=1}^{k} z_i \mu(\Omega_i)$. Let us observe that this definition does not depend on the representation (16).

Let $1 \leq p < \infty$, let $(X, \|\cdot\|)$ be a complex Banach space, and let $(\Omega, \mathcal{R}, \mu)$ be a measure space. Then the space $L^p(\Omega, X, \mu)$ consists of all strongly μ -measurable functions $f: \Omega \to X$ such that $\|f\|_p := (\int_{\Omega} \|f(\cdot)\|^p d\mu)^{1/p}$ is finite, we also use the abbreviation $L^p(\Omega, \mu)$ in the case that $X = \mathbb{C}$. The space $L^{\infty}(\Omega, X, \mu)$ consists of all strongly μ -measurable, essentially bounded functions and is equipped with the norm $\|f\|_{\infty} := \operatorname{ess} \sup_{t \in \Omega} \|f(t)\|, f \in L^{\infty}(\Omega, X, \mu)$. The functions that are equal μ -almost everywhere on Ω will be identified; furthermore, if μ is the Lebesgue measure on the real line, then, for every $p \in [1, \infty]$, the space $L^p(\Omega, X, \mu)$ will be also denoted by $L^p(\Omega: X)$. The Riesz–Fischer theorem states that $(L^p(\Omega, X, \mu), \|\cdot\|_p)$ is a Banach space for all $p \in [1, \infty]$; as is well known, $(L^2(\Omega, X, \mu), \|\cdot\|_2)$ is a Hilbert space. Assuming that the Banach space X is reflexive, the space $L^p(\Omega, X, \mu)$ is reflexive for all $p \in (1, \infty)$ and its dual is isometrically isomorphic to $L^{\frac{p}{p-1}}(\Omega, X, \mu)$.

DEFINITION 1.1.2. (C. Martinez, M. Sanz [410, pp. 99–102]; cf. [292, Section 1.2] for more details)

(i) Let $K \subseteq \Omega$ be a compact set, and let a function $f: K \to E$ be strongly measurable. Then it is said that $f(\cdot)$ is $(\mu$ -)integrable iff there is a sequence $(f_n)_{n \in \mathbb{N}}$ of simple functions such that $\lim_{n \to \infty} f_n(t) = f(t)$ a.e. $t \in K$ and for all $\varepsilon > 0$ and each $p \in \circledast$ there is a number $n_0 = n_0(\varepsilon, p)$ such that

(17)
$$\int_{K} p(f_n - f_m) d\mu \leqslant \varepsilon \quad (m, n \ge n_0).$$

In this case, we define

$$\int_{K} f \, d\mu := \lim_{n \to \infty} \int_{K} f_n d\mu.$$

- (ii) A function $f: \Omega \to E$ is said to be locally μ -integrable iff, for every compact set $K \subseteq \Omega$, the restriction $f_{|K}: K \to E$ is μ -integrable.
- (iii) A function $f: \Omega \to E$ is said to be μ -integrable iff it is locally integrable and

(18)
$$\int_{\Omega} p(f) d\mu < \infty, \quad p \in \circledast.$$

If this is the case, we define

$$\int_{\Omega} f \, d\mu := \lim_{n \to \infty} \int_{K_n} f \, d\mu$$

with $(K_n)_{n\in\mathbb{N}}$ being an expansive sequence of compact subsets of Ω with the property that $\bigcup_{n\in\mathbb{N}} K_n = \Omega$.

The definition in (ii) makes sense and does not depend on the choice of a sequence $(K_n)_{n \in \mathbb{N}}$. Moreover,

(19)
$$p\left(\int_{\Omega} f \, d\mu\right) \leqslant \int_{\Omega} p(f) d\mu, \quad p \in \circledast$$

and the μ -integrability of a function $f: K \to X$, resp. $f: \Omega \to X$, implies that for each $x^* \in X^*$, one has:

(20)
$$\left\langle x^*, \int_K f \, d\mu \right\rangle = \int_K \langle x^*, f \rangle d\mu, \text{ resp. } \left\langle x^*, \int_\Omega f \, d\mu \right\rangle = \int_\Omega \langle x^*, f \rangle d\mu.$$

Any continuous function $f: \Omega \to E$ satisfying (18) is μ -integrable and the following holds.

- THEOREM 1.1.3. (i) (The Dominated Convergence Theorem) Suppose that (f_n) is a sequence of μ -integrable functions from E^{Ω} and (f_n) converges pointwisely to a function $f: \Omega \to E$. Assume that, for every $p \in \circledast$, there exists a μ -integrable function $F_p: \Omega \to [0, \infty)$ such that $p(f_n) \leq F_p$, $n \in \mathbb{N}$. Then $f(\cdot)$ is a μ -integrable function and $\lim_{n\to\infty} \int_{\Omega} f_n d\mu = \int_{\Omega} f d\mu$.
- (ii) Let Y be an SCLCS, and let $T: X \to Y$ be a continuous linear mapping. If $f: \Omega \to X$ is μ -integrable, then $Tf: \Omega \to Y$ is likewise μ -integrable and

(21)
$$T\int_{\Omega} f \, d\mu = \int_{\Omega} Tf \, d\mu.$$

(iii) Let Y be an SCLCS, and let $T: D(T) \subseteq X \to Y$ be a closed linear mapping. If $f: \Omega \to D(T)$ is μ -integrable and $Tf: \Omega \to Y$ is likewise μ integrable, then $\int_{\Omega} f d\mu \in D(T)$ and (21) holds.

Assume that $\mu = dt$ is the Lebesgue measure on $\Omega = [0, \infty)$ and $f: [0, \infty) \to E$ is a locally Lebesgue integrable function. As in the Banach space case, we will denote the space consisting of such functions by $L^1_{loc}([0,\infty):E)$; similarly we define the space $L^1([0,\tau]:E)$ for $0 < \tau < \infty$. It is clear that (20) implies $\langle x^*, f(\cdot) \rangle \in L^1_{loc}([0,\infty))$ for $x^* \in E^*$. The first normalized antiderivative $t \mapsto f^{[1]}(t) := F(t) := \int_0^t f(s) ds, t \ge 0$ of $f(\cdot)$ is continuous for $t \ge 0$, and we have that $\int_0^t p(f) d\mu < \infty$ for any $p \in \circledast$ and $t \ge 0$. Set $f^{[n]}(t) := \int_0^t g_n(t-s)f(s) ds, t \ge 0$. The formula for partial integration in the third part of subsequent theorem will be of crucial importance in our analysis of operational properties of Laplace transform of non-continuous functions with values in SCLCSs.

THEOREM 1.1.4. (i) Suppose that
$$g \in C([0,\infty))$$
 and $f \in L^1_{loc}([0,\infty) : E)$.

- (ii) If $g \in L^1_{loc}([0,\infty))$ and $f \in C([0,\infty): E)$, then $gf \in L^1_{loc}([0,\infty): E)$.
- (iii) (The partial integration) Suppose that $g \in AC_{loc}([0,\infty))$ and

 $f \in L^1_{loc}([0,\infty): E)$. Then, for every $\tau \ge 0$, we have

(22)
$$\int_0^\tau g(t)f(t)dt = g(\tau)F(\tau) - \int_0^\tau g'(t)F(t)dt.$$

PROOF. Fix a number $\tau \in (0, \infty)$. Let $(f_n)_{n \in \mathbb{N}}$ be a sequence of simple functions in $E^{[0,\tau]}$ such that $\lim_{n\to\infty} f_n(t) = f(t)$ a.e. $t \in K = [0,\tau]$ and for all $\varepsilon > 0$ and each $p \in \circledast$ there is a number $n_0 = n_0(\varepsilon, p)$ such that (17) holds. Then $\int_0^{\tau} f(t)dt = \lim_{n\to\infty} \int_0^{\tau} f_n(t)dt$ and the sequence $(p(f_n))_{n\in\mathbb{N}}$ is convergent in $L^1[0,\tau]$. By the proof of [410, Proposition 4.4.1], there exists a sequence $(s_n)_{n\in\mathbb{N}}$ of simple functions in $\mathbb{C}^{[0,\tau]}$ such that $\lim_{n\to\infty} ||s_n - g||_{L^{\infty}[0,\tau]} = 0$, $\sup_{n\in\mathbb{N}} ||s_n||_{L^{\infty}[0,\tau]} \leq ||g||_{L^{\infty}[0,\tau]}$ and that for all $\varepsilon > 0$ and $p = |\cdot|$ there is a number $n_0 = n_0(\varepsilon, p)$ such that (17) holds with the functions $f_n(\cdot)$ and $f_m(\cdot)$ replaced respectively with $s_n(\cdot)$ and $s_m(\cdot)$. Clearly, $(s_n f_n)_{n\in\mathbb{N}}$ is a sequence of simple functions in $E^{[0,\tau]}$ such that $\lim_{n\to\infty} s_n(t)f_n(t) = g(t)f(t)$ a.e. $t \in [0,\tau]$. Furthermore, it can be easily seen that

$$\begin{split} \int_{0}^{t} p(s_{n}(t)f_{n}(t) - s_{m}(t)f_{m}(t))dt \\ &\leqslant \|s_{n}\|_{L^{\infty}[0,\tau]} \int_{0}^{t} p(f_{n}(t) - f_{m}(t))dt + \|s_{n} - s_{m}\|_{L^{\infty}[0,\tau]} \int_{0}^{t} p(f_{m}(t))dt \\ &\leqslant \|g\|_{L^{\infty}[0,\tau]} \int_{0}^{t} p(f_{n}(t) - f_{m}(t))dt \\ &+ (\|s_{n} - g\|_{L^{\infty}[0,\tau]} + \|s_{m} - g\|_{L^{\infty}[0,\tau]}) \int_{0}^{t} p(f_{m}(t))dt, \quad m, n \in \mathbb{N}. \end{split}$$

This proves (i). To prove (ii), observe first that using Definition 1.1.2 we can directly prove that a function $g_1f_1(\cdot)$ belongs to the space $L^1([0,\tau] : E)$, provided that $f_1: [0,\tau] \to E$ is a simple function and $g_1 \in L^1[0,\tau]$. By the proof of [410, Proposition 4.4.1] once more, we can find a sequence $(f_n)_{n\in\mathbb{N}}$ of simple functions in $E^{[0,\tau]}$ such that, for every $p \in \circledast$, $\lim_{n\to\infty} p(f_n - f)_{L^{\infty}[0,\tau]} = 0$, $\sup_{n\in\mathbb{N}} p(f_n)_{L^{\infty}[0,\tau]} \leq p(f)_{L^{\infty}[0,\tau]}$ and that for all $\varepsilon > 0$ there is a number $n_0 = n_0(\varepsilon, p)$ such that (17) holds. Therefore, $(gf_n)_{n\in\mathbb{N}}$ is a sequence in $L^1([0,\tau] : E)$ and $\lim_{n\to\infty} g(t)f_n(t) = g(t)f(t)$ a.e. $t \in [0,\tau]$. Making use of the dominated convergence theorem (Theorem 1.1.3(i)), we get that $gf \in L^1_{loc}([0,\infty) : E)$, as claimed. By (i) and (ii), the both integrals in (22) are well-defined. Let $x^* \in E^*$. Using the partial integration in the Lebesgue integral and (20), we get that

$$\int_0^\tau g(t) \langle x^*, f(t) \rangle dt = g(\tau) \langle x^*, F(\tau) \rangle - \int_0^\tau g'(t) \langle x^*, F(t) \rangle dt$$

Since x^* was arbitrary, it readily follows on account of (20) that (22) holds. The proof of the theorem is thereby complete.

Analytical properties of functions with values in SCLCSs. A function $f: \Omega \to E$, where Ω is an open subset of \mathbb{C} , is said to be analytic iff it is locally expressible in a neighborhood of any point $z \in \Omega$ by a uniformly convergent power

series with coefficients in E. It is well known that the analyticity of $f(\cdot)$ is equivalent with the weak analyticity of $f(\cdot)$; in other words, the mapping $\lambda \mapsto f(\lambda), \lambda \in \Omega$ is analytic iff the mapping $\lambda \mapsto \langle x^*, f(\lambda) \rangle, \lambda \in \Omega$ is analytic for every $x^* \in E^*$. If the mapping $f: \Omega \to E$ is analytic, then the mapping $\lambda \mapsto f(\lambda), \lambda \in \Omega$ is infinitely differentiable and

$$f^{(n)}(z) = \frac{n!}{2\pi i} \oint_{\Gamma} \frac{f(\lambda)}{(\lambda - z)^{n+1}} d\lambda, \quad z \in \Omega \smallsetminus \Gamma, \text{ Ind}_{\Gamma}(z) = 1, \ n \in \mathbb{N}_0,$$

which simply implies that the equality

$$f(z) = \sum_{n=0}^{\infty} (z - z_0)^n \frac{f^{(n)}(z_0)}{n!}$$

holds in a neighborhood of point $z_0 \in \Omega$. The identity theorem for analytic functions [27, Proposition A.2, p. 456] continues to hold in the case that X is a general locally convex space.

We need the following extension of Weierstrass theorem.

LEMMA 1.1.5. (E. Jordá [261, Theorem 3, p. 742]) Let $\emptyset \neq \Omega \subseteq \mathbb{C}$ be open and connected, and let $f_n: \Omega \to E$ be an analytic function $(n \in \mathbb{N})$. Assume further that, for every $z_0 \in \Omega$, there exists r > 0 such that the set $\bigcup_{n \in \mathbb{N}} f_n(B(z_0, r))$ is bounded and the set $\Omega_0 := \{z \in \Omega : \lim_{n \to \infty} f_n(z) \text{ exists}\}$ has a limit point in Ω . Then there exists an analytic function $f: \Omega \to E$ such that (f_n) converges locally uniformly to f.

Complex powers of almost *C***-nonnegative operators.** Let $m \in \mathbb{R}$, let *A* be a closed linear operator with domain and range contained in *E*, and let $C \in L(E)$ be an injective operator satisfying $CA \subseteq AC$. Then it is said that the operator *A* belongs to the class $\mathcal{M}_{C,m}$ iff $(-\infty, 0) \subseteq \rho_C(A)$ and the family

$$\{(\lambda^{-1} + \lambda^m)^{-1}(\lambda + A)^{-1}C : \lambda > 0\} \subseteq L(E)$$

is equicontinuous. Furthermore, it is said that A is almost C-nonnegative iff there exists $m \in \mathbb{R}$ such that A belongs to the class $\mathcal{M}_{C,m}$. It will be necessary to remind us of the following facts concerning the fractional powers of almost C-nonnegative operators; for further information, see [292, Section 2.9]. Set $p_n(x) := \sum_{i=0}^n p(A^i x)$ $(x \in D_{\infty}(A), p \in \mathfrak{B}, n \in \mathbb{N}_0), A_{\infty} := A_{|D_{\infty}(A)}$ and $C_{\infty} := C_{|D_{\infty}(A)}$. Then the system $(p_n)_{p \in \mathfrak{B}, n \in \mathbb{N}_0}$ induces a Hausdorff sequentially complete locally convex topology on $D_{\infty}(A), A_{\infty} \in L(D_{\infty}(A))$ and $C_{\infty} \in L(D_{\infty}(A))$ is injective.

In [292, Definition 2.9.11], we have generalized the notion of Balakrishnan's operators [44] as follows: Let $\alpha \in \mathbb{C}_+$ and $A \in \mathcal{M}_{C,-1}$. Then:

(i) If $0 < \operatorname{Re} \alpha < 1$, $D(J_C^{\alpha}) := D(A)$ and

$$J_C^{\alpha} x := \frac{\sin \alpha \pi}{\pi} \int_0^\infty \lambda^{\alpha - 1} (\lambda + A)^{-1} C A x \, d\lambda, \quad x \in D(A).$$

(n) If
$$\operatorname{Re} \alpha = 1$$
, $D(J_{\alpha}^{\alpha}) := D(A^2)$ and
 $J_{C}^{\alpha}x := \frac{\sin \alpha \pi}{\pi} \int_{0}^{\infty} \lambda^{\alpha-1} \Big[(\lambda+A)^{-1}C - \frac{\lambda C}{\lambda^2+1} \Big] Ax \, d\lambda + \sin \frac{\alpha \pi}{2} CAx,$

Then we know, for every $\alpha \in \mathbb{C}_+$, the operator J_C^{α} is closable. Set $A_{C,\alpha} := C^{-1}\overline{J_C^{\alpha}}$ $(\alpha \in \mathbb{C}_+)$.

Consider now the case in which A belongs to the class $\mathcal{M}_{C,m}$ for some $m \ge -1$. Due to [**292**, Proposition 2.9.20], we have that the operator A_{∞} is C_{∞} -nonnegative in the space $D_{\infty}(A)$, i.e., that $A_{\infty} \in \mathcal{M}_{C_{\infty},-1}$. Therefore, we can construct the power $A_{\infty,\alpha} \equiv (A_{\infty})_{C_{\infty},\alpha}$ in the space $D_{\infty}(A)$ ($\alpha \in \mathbb{C}_+$). In [**292**, Proposition 2.9.23], we have proved that $C^2(D(A^{2p+n})) \subseteq D(\overline{A_{\infty,\alpha}})$, provided that $\alpha \in \mathbb{C}_+$ satisfies $0 < \operatorname{Re} \alpha < n$ for some $n \in \mathbb{N}$ ($p \equiv \lfloor m + 2 \rfloor$). This is the most important auxiliary result which enables us to introduce the power A_{α} :

DEFINITION 1.1.6. Suppose $m \ge -1$, $n \in \mathbb{N}$, $p = \lfloor m + 2 \rfloor$, $A \in \mathcal{M}_{C,m}$, $\alpha \in \mathbb{C}_+$ and $0 < \operatorname{Re} \alpha < n$. Then we define the power A_{α} as follows

$$A_{\alpha} := C^{-2} ((1+A)^{-1}C)^{-n(p+1)-p} \overline{A_{\infty,\alpha}} ((1+A)^{-1}C)^{n(p+1)+p} C^2.$$

It is worth noting that Definition 1.1.6 does not depend on the particular choice of numbers m and n, as well as that A_{α} is a closed linear operator on E. Furthermore, the injectiveness of A implies the injectiveness of A_{α} . If this is the case, we define:

$$A_{-\alpha} := (A_{\alpha})^{-1}, \quad A_0 := I \quad \text{and}$$
$$A_{i\tau} := C^{-3} ((1+A)^{-1}C)^{-(3p+3)} A^{-1} A_{1+i\tau} ((1+A)^{-1}C)^{3p+3} C^3.$$

The construction of fractional powers of the operator A which belongs to the class $A \in \mathcal{M}_{C,m}$ for some real number m < -1 strongly depends on the injectiveness of A: it is still an open problem to construct the power A_{α} ($\alpha \in \mathbb{C}_+$) in the case that $A \in \mathcal{M}_{C,m}$ for some real number m < -1, and A is not injective. On the other hand, if the operator A is injective, then the operator A^{-1} belongs to the class $\mathcal{M}_{C,-m-2}$ and we can simply define the power A_{α} ($\alpha \in \mathbb{C}$) by setting $A_{\alpha} := (A^{-1})_{-\alpha}$ ($\alpha \in \mathbb{C}$). Observe only that -m - 2 > -1 if m < -1.

1.2. Multivalued linear operators

In this section, we present a brief overview of the necessary definitions and properties of multivalued linear operators that will be necessary for our further work. In a separate subsection, we exhibit our original contributions on hypercyclic and disjoint hypercyclic multivalued linear operators (this is a part of a joint research study with C.-C. Chen, J. A. Conejero and M. Murillo-Arcila [107]; see also [108]). For more details about multivalued linear operators, we refer the reader to the monographs [120, 139, 199] as well as to the papers [21, 30] and [200].

Let X and Y be two SCLCSs. A multivalued map (multimap) $\mathcal{A}: X \to P(Y)$ is said to be a multivalued linear operator (MLO) iff the following holds:

- (i) $D(\mathcal{A}) := \{x \in X : \mathcal{A}x \neq \emptyset\}$ is a linear subspace of X;
- (ii) $Ax + Ay \subseteq A(x+y), x, y \in D(A)$ and $\lambda Ax \subseteq A(\lambda x), \lambda \in \mathbb{C}, x \in D(A)$.

If X = Y, then we say that \mathcal{A} is an MLO in X. An almost immediate consequence of definition is that $\mathcal{A}x + \mathcal{A}y = \mathcal{A}(x + y)$ for all $x, y \in D(\mathcal{A})$ and $\lambda \mathcal{A}x = \mathcal{A}(\lambda x)$ for all $x \in D(\mathcal{A}), \lambda \neq 0$. Furthermore, for any $x, y \in D(\mathcal{A})$ and $\lambda, \eta \in \mathbb{C}$ with $|\lambda| + |\eta| \neq 0$, we have $\lambda \mathcal{A}x + \eta \mathcal{A}y = \mathcal{A}(\lambda x + \eta y)$. If \mathcal{A} is an MLO, then $\mathcal{A}0$ is a linear manifold in Y and $\mathcal{A}x = f + \mathcal{A}0$ for any $x \in D(\mathcal{A})$ and $f \in \mathcal{A}x$. Set $R(\mathcal{A}) :=$ $\{\mathcal{A}x : x \in D(\mathcal{A})\}$. The set $\mathcal{A}^{-1}0 = \{x \in D(\mathcal{A}) : 0 \in \mathcal{A}x\}$ is called the kernel of \mathcal{A} and it is denoted henceforth by $N(\mathcal{A})$ or Kern (\mathcal{A}) . The inverse \mathcal{A}^{-1} of an MLO is defined by $D(\mathcal{A}^{-1}) := R(\mathcal{A})$ and $\mathcal{A}^{-1}y := \{x \in D(\mathcal{A}) : y \in \mathcal{A}x\}$. It is checked at once that \mathcal{A}^{-1} is an MLO in X, as well as that $N(\mathcal{A}^{-1}) = \mathcal{A}0$ and $(\mathcal{A}^{-1})^{-1} = \mathcal{A}$. If $N(\mathcal{A}) = \{0\}$, i.e., if \mathcal{A}^{-1} is single-valued, then \mathcal{A} is said to be injective. It is worth noting that $\mathcal{A}x = \mathcal{A}y$ for some two elements x and $y \in D(\mathcal{A})$, iff $\mathcal{A}x \cap \mathcal{A}y \neq \emptyset$; moreover, if \mathcal{A} is injective, then the equality $\mathcal{A}x = \mathcal{A}y$ holds iff x = y. For any mapping $\mathcal{A} : X \to P(Y)$ we define $\check{\mathcal{A}} := \{(x, y) : x \in D(\mathcal{A}), y \in \mathcal{A}x\}$. Then \mathcal{A} is an MLO iff $\check{\mathcal{A}}$ is a linear relation in $X \times Y$, i.e., iff $\check{\mathcal{A}}$ is a linear subspace of $X \times Y$.

If $\mathcal{A}, \mathcal{B}: X \to P(Y)$ are two MLOs, then we define its sum $\mathcal{A}+\mathcal{B}$ by $D(\mathcal{A}+\mathcal{B}) := D(\mathcal{A}) \cap D(\mathcal{B})$ and $(\mathcal{A}+\mathcal{B})x := \mathcal{A}x + \mathcal{B}x, x \in D(\mathcal{A}+\mathcal{B})$. It can be simply verified that $\mathcal{A}+\mathcal{B}$ is likewise an MLO.

Let $\mathcal{A}: X \to P(Y)$ and $\mathcal{B}: Y \to P(Z)$ be two MLOs, where Z is an SCLCS. The product of \mathcal{A} and \mathcal{B} is defined by $D(\mathcal{B}\mathcal{A}) := \{x \in D(\mathcal{A}) : D(\mathcal{B}) \cap \mathcal{A}x \neq \emptyset\}$ and $\mathcal{B}\mathcal{A}x := \mathcal{B}(D(\mathcal{B}) \cap \mathcal{A}x)$. Then $\mathcal{B}\mathcal{A}: X \to P(Z)$ is an MLO and $(\mathcal{B}\mathcal{A})^{-1} = \mathcal{A}^{-1}\mathcal{B}^{-1}$. The scalar multiplication of an MLO $\mathcal{A}: X \to P(Y)$ with the number $z \in \mathbb{C}$, $z\mathcal{A}$ for short, is defined by $D(z\mathcal{A}) := D(\mathcal{A})$ and $(z\mathcal{A})(x) := z\mathcal{A}x, x \in D(\mathcal{A})$. It is clear that $z\mathcal{A}: X \to P(Y)$ is an MLO and $(\omega z)\mathcal{A} = \omega(z\mathcal{A}) = z(\omega\mathcal{A}), z, \omega \in \mathbb{C}$. Suppose that X' is a linear subspace of X, and $\mathcal{A}: X \to P(Y)$ is an MLO. Then we define the restriction of operator \mathcal{A} to the subspace X', $\mathcal{A}_{|X'}$ for short, by $D(\mathcal{A}_{|X'}) := D(\mathcal{A}) \cap X'$ and $\mathcal{A}_{|X'}x := \mathcal{A}x, x \in D(\mathcal{A}_{|X'})$. Clearly, $\mathcal{A}_{|X'}: X \to P(Y)$ is an MLO. It is well known that an MLO $\mathcal{A}: X \to P(Y)$ is injective (resp., single-valued) iff $\mathcal{A}^{-1}\mathcal{A} = I_{|D(\mathcal{A})}$ (resp., $\mathcal{A}\mathcal{A}^{-1} = I_{|R(\mathcal{A})}^{Y}$).

The integer powers of an MLO $\mathcal{A}: X \to P(X)$ is defined recursively as follows: $\mathcal{A}^0 =: I$; if \mathcal{A}^{n-1} is defined, set

$$D(\mathcal{A}^n) := \{ x \in D(\mathcal{A}^{n-1}) : D(\mathcal{A}) \cap \mathcal{A}^{n-1} x \neq \emptyset \},\$$

and

$$\mathcal{A}^n x := (\mathcal{A}\mathcal{A}^{n-1})x = \bigcup_{y \in D(\mathcal{A}) \cap \mathcal{A}^{n-1}x} \mathcal{A}y, \quad x \in D(\mathcal{A}^n).$$

We can prove inductively that $(\mathcal{A}^n)^{-1} = (\mathcal{A}^{n-1})^{-1}\mathcal{A}^{-1} = (\mathcal{A}^{-1})^n =: \mathcal{A}^{-n}, n \in \mathbb{N}$ and $D((\lambda - \mathcal{A})^n) = D(\mathcal{A}^n), n \in \mathbb{N}_0, \lambda \in \mathbb{C}$. Moreover, if \mathcal{A} is single-valued, then the above definitions are consistent with the usual definition of powers of \mathcal{A} . If $\mathcal{A}: X \to P(Y)$ and $\mathcal{B}: X \to P(Y)$ are two MLOs, then we write $\mathcal{A} \subseteq \mathcal{B}$ iff $D(\mathcal{A}) \subseteq D(\mathcal{B})$ and $\mathcal{A}x \subseteq \mathcal{B}x$ for all $x \in D(\mathcal{A})$. Assume now that a linear singlevalued operator $S: D(S) \subseteq X \to Y$ has domain $D(S) = D(\mathcal{A})$ and $S \subseteq \mathcal{A}$, where $\mathcal{A}: X \to P(Y)$ is an MLO. Then S is called a section of \mathcal{A} ; if this is the case, we have $\mathcal{A}x = Sx + \mathcal{A}0$, $x \in D(\mathcal{A})$ and $R(\mathcal{A}) = R(S) + \mathcal{A}0$. We say that an MLO operator $\mathcal{A}: X \to P(Y)$ is closed if for any nets (x_{τ}) in $D(\mathcal{A})$ and (y_{τ}) in Y such that $y_{\tau} \in \mathcal{A}x_{\tau}$ for all $\tau \in I$ we have that the suppositions $\lim_{\tau \to \infty} x_{\tau} = x$ and $\lim_{\tau \to \infty} y_{\tau} = y$ imply $x \in D(\mathcal{A})$ and $y \in \mathcal{A}x$.

Following C. Knuckles and F. Neubrander [285], we introduce the notion of a relatively closed MLO as follows. We say that an MLO $\mathcal{A}: X \to P(Y)$ is relatively closed iff there exist auxiliary SCLCSs $X_{\mathcal{A}}$ and $Y_{\mathcal{A}}$ such that $D(\mathcal{A}) \subseteq X_{\mathcal{A}} \hookrightarrow X$, $R(\mathcal{A}) \subseteq Y_{\mathcal{A}} \hookrightarrow Y$ and \mathcal{A} is closed in $X_{\mathcal{A}} \times Y_{\mathcal{A}}$; i.e., the assumptions $D(\mathcal{A}) \ni x_{\tau} \to x$ as $\tau \to \infty$ in $X_{\mathcal{A}}$ and $\mathcal{A}x_{\tau} \ni y_{\tau} \to y$ as $\tau \to \infty$ in $Y_{\mathcal{A}}$ implies that $x \in D(\mathcal{A})$ and $y \in \mathcal{A}x$. A relatively closed operator will also be called $X_{\mathcal{A}} \times Y_{\mathcal{A}}$ -closed. By way of illustration, let $\mathcal{A}, B: D \subseteq X \to Y$ be closed linear operators with the same domain D. Then the operator $\mathcal{A} + B$ is not necessarily closed but it is always $[D(\mathcal{A})] \times Y$ -closed (cf. [280, p. 170]). Examples presented in [285] can be simply reformulated for operators acting on locally convex spaces, as well:

- EXAMPLE 1.2.1. (i) If $\mathcal{A}: X \to P(Y)$ is an MLO, then $\overline{\mathcal{A}}: X \to P(Y)$ is likewise an MLO. This shows that any MLO has a closed linear extension, in contrast to the usually considered single-valued linear operators.
- (ii) Let $A: D(A) \subseteq X \to Y$ be a single-valued linear operator that is $X_A \times Y_A$ closed, let $\mathcal{B}: X \to P(Y)$ be an MLO that is $X_{\mathcal{B}} \times Y_{\mathcal{B}}$ -closed, and let $Y_A \hookrightarrow Y_{\mathcal{B}}$. Then the MLO $S = A + \mathcal{B}$ is $X_S \times Y_{\mathcal{B}}$ -closed, where $X_S := D(A) \cap X_{\mathcal{B}}$ and the topology on X_S is induced by the system $(s_{p,q,r})$ of fundamental seminorms, defined as follows: $s_{p,q,r}(x) =: p(x) + p(Ax) + q(x) + r(Ax)$, $x \in X_S$ $(p \in \circledast_X, q \in \circledast_{X_{\mathcal{B}}}, r \in \circledast_{Y_A})$.
- (iii) Let $A: D(A) \subseteq X \to Y$ be a single-valued linear operator that is $X_A \times Y_A$ closed, let $\mathcal{B}: Y \to P(Z)$ be an MLO that is $Y_{\mathcal{B}} \times Z_{\mathcal{B}}$ -closed, and let $Y_{\mathcal{B}} \hookrightarrow Y_A$. Then the MLO $C = \mathcal{B}A: X \to P(Z)$ is $X_C \times Z_{\mathcal{B}}$ -closed, where $X_C := \{x \in D(A) : Ax \in Y_{\mathcal{B}}\}$ and the topology on X_C is induced by the system $(s_{p,q})$ of fundamental seminorms, defined as follows: $s_{p,q}(x) =:$ $p(x) + p(Ax) + q(Ax), x \in X_C \ (p \in \circledast_X, q \in \circledast_{Y_{\mathcal{B}}}).$
- (iv) Let $A: D(A) \subseteq X \to Y$ and $B: D(B) \subseteq X \to Y$ be two single-valued linear operators. Set

$$\mathcal{A} := B^{-1}A = \{(x, y) : x \in D(A), y \in D(B) \text{ and } Ax = By\}.$$

Then \mathcal{A} is an MLO in X, and the following holds:

- (a) If one of the operators A, B is bounded and the other closed, then \mathcal{A} is closed.
- (b) If A is closed and B is $X_B \times Y$ -closed, then \mathcal{A} is $[D(A)] \times X_B$ -closed.
- (c) If B is closed and A is $X_A \times Y$ -closed, then A is $X_A \times [D(B)]$ -closed.
- (d) If A is $X_A \times Y_A$ -closed and B is $X_B \times Y_B$ -closed, where $Y_B \hookrightarrow Y_A$, then \mathcal{A} is $X_C \times X_B$ -closed, where X_C is defined as in (iii).

If $\mathcal{A}: X \to P(Y)$ is an MLO, then we define the adjoint $\mathcal{A}^*: Y^* \to P(X^*)$ of \mathcal{A} by its graph

$$\mathcal{A}^* := \{ (y^*, x^*) \in Y^* \times X^* : \langle y^*, y \rangle = \langle x^*, x \rangle \text{ for all pairs } (x, y) \in \mathcal{A} \}.$$

It is simply verified that \mathcal{A}^* is a closed MLO, and that $\langle y^*, y \rangle = 0$ whenever $y^* \in D(\mathcal{A}^*)$ and $y \in \mathcal{A}0$. Furthermore, the equations [199, (1.2)–(1.6)] continue to hold for adjoints of MLOs acting on locally convex spaces.

The following important lemma can be proved by means of the Hahn–Banach theorem and the argumentation from [47].

LEMMA 1.2.2. Suppose that $\mathcal{A}: X \to P(Y)$ is an MLO and \mathcal{A} is $X_{\mathcal{A}} \times Y_{\mathcal{A}}$ closed. Assume, further, that $x_0 \in X$, $y_0 \in Y$ and $\langle x^*, x_0 \rangle = \langle y^*, y_0 \rangle$ for all pairs $(x^*, y^*) \in X^*_{\mathcal{A}} \times Y^*_{\mathcal{A}}$ satisfying that $\langle x^*, x \rangle = \langle y^*, y \rangle$ whenever $y \in \mathcal{A}x$. Then $y_0 \in \mathcal{A}x_0$.

With Lemma 1.2.2 in view, we can simply prove the following extension of Theorem 1.1.3(iii) for relatively closed MLOs in locally convex spaces. Here, by Ω we denote a locally compact and separable metric space and by μ we denote a locally finite Borel measure defined on Ω .

THEOREM 1.2.3. Suppose that $\mathcal{A}: X \to P(Y)$ is an MLO and \mathcal{A} is $X_{\mathcal{A}} \times Y_{\mathcal{A}}$ closed. Let $f: \Omega \to X_{\mathcal{A}}$ and $g: \Omega \to Y_{\mathcal{A}}$ be μ -integrable, and let $g(x) \in \mathcal{A}f(x)$, $x \in \Omega$. Then $\int_{\Omega} f \, d\mu \in D(\mathcal{A})$ and $\int_{\Omega} g \, d\mu \in \mathcal{A} \int_{\Omega} f \, d\mu$.

Now we will analyze the C-resolvent sets of MLOs in locally convex spaces. Our standing assumptions will be that \mathcal{A} is an MLO in X, as well as that $C \in L(X)$ is injective and $C\mathcal{A} \subseteq \mathcal{A}C$ (this is equivalent to say that, for any $(x, y) \in X \times X$, we have the implication $(x, y) \in \mathcal{A} \Rightarrow (Cx, Cy) \in \mathcal{A}$; by induction, we immediately get that $C\mathcal{A}^k \subseteq \mathcal{A}^k C$ for all $k \in \mathbb{N}$). Then the C-resolvent set of \mathcal{A} , $\rho_C(\mathcal{A})$ for short, is defined as the union of those complex numbers $\lambda \in \mathbb{C}$ for which

- (i) $R(C) \subseteq R(\lambda \mathcal{A});$
- (ii) $(\lambda A)^{-1}C$ is a single-valued linear continuous operator on X.

The operator $\lambda \mapsto (\lambda - \mathcal{A})^{-1}C$ is called the *C*-resolvent of \mathcal{A} ($\lambda \in \rho_C(\mathcal{A})$); the resolvent set of \mathcal{A} is defined by $\rho(\mathcal{A}) := \rho_I(\mathcal{A}), R(\lambda : \mathcal{A}) \equiv (\lambda - \mathcal{A})^{-1} \ (\lambda \in \rho(\mathcal{A})).$ We can almost trivially construct examples of MLOs for which $\rho(\mathcal{A}) = \emptyset$ and $\rho_C(\mathcal{A}) \neq \emptyset$: Let Y be a proper closed linear subspace of X, let \mathcal{A} be an MLO in Y, and let $\lambda \in \mathbb{C}$ so that $(\lambda - \mathcal{A})^{-1} \in L(Y)$. Taking any injective operator $C \in L(X)$ with $R(C) \subseteq Y$, and looking $\mathcal{A} = \mathcal{A}_X$ as an MLO in X, it is clear that $\lambda \in \rho_C(\mathcal{A}_X)$ and $\rho(\mathcal{A}_X) = \emptyset$. In general case, if $\rho_C(\mathcal{A}) \neq \emptyset$, then for any $\lambda \in \rho_C(\mathcal{A})$ we have $\mathcal{A}0 = N((\lambda I - \mathcal{A})^{-1}C)$, as well as $\lambda \in \rho_C(\bar{\mathcal{A}}), \ \bar{\mathcal{A}} \subseteq C^{-1}\mathcal{A}C$ and $((\lambda - \mathcal{A})^{-1}C)^k(D(\mathcal{A}^l)) \subseteq D(\mathcal{A}^{k+l}), k, l \in \mathbb{N}_0$; here it is worth noting that the equality $\mathcal{A} = C^{-1}\mathcal{A}C$ holds provided, in addition, that $\rho(\mathcal{A}) \neq \emptyset$ (see e.g. the proofs of [138, Proposition 2.1, Lemma 2.3]). The basic properties of C-resolvent sets of single-valued linear operators [291,292] continue to hold in our framework (observe, however, that there exist certain differences that we will not discuss here). For example, if $\rho(\mathcal{A}) \neq \emptyset$, then \mathcal{A} is closed; it is well known that this statement does not hold if $\rho_C(\mathcal{A}) \neq \emptyset$ for some $C \neq I$ (cf. [138, Example 2.2]). Arguing as in the proofs of [199, Theorem 1.7-Theorem 1.9], we can deduce the validity of the following theorem.

THEOREM 1.2.4. (i) We have

$$(\lambda - \mathcal{A})^{-1}C\mathcal{A} \subseteq \lambda(\lambda - \mathcal{A})^{-1}C - C \subseteq \mathcal{A}(\lambda - \mathcal{A})^{-1}C, \quad \lambda \in \rho_C(\mathcal{A}).$$

The operator $(\lambda - \mathcal{A})^{-1}C\mathcal{A}$ is single-valued on $D(\mathcal{A})$ and $(\lambda - \mathcal{A})^{-1}C\mathcal{A}x = (\lambda - \mathcal{A})^{-1}Cy$, whenever $y \in \mathcal{A}x$ and $\lambda \in \rho_C \mathcal{A}$.

(ii) Suppose that $\lambda, \mu \in \rho_C(\mathcal{A})$. Then the resolvent equation

$$\begin{split} (\lambda - \mathcal{A})^{-1}C^2x - (\mu - \mathcal{A})^{-1}C^2x &= (\mu - \lambda)(\lambda - \mathcal{A})^{-1}C(\mu - \mathcal{A})^{-1}Cx, \quad x \in X \\ holds \ good. \ In \ particular, \ [(\lambda - \mathcal{A})^{-1}C][(\mu - \mathcal{A})^{-1}C] &= [(\mu - \mathcal{A})^{-1}C][(\lambda - \mathcal{A})^{-1}C]. \end{split}$$

(iii) Suppose that \mathcal{A} and \mathcal{B} are two given MLOs, as well as that $C\mathcal{B} \subseteq \mathcal{B}C$ and $0 \in \rho_C(\mathcal{A}) \cap \rho_C(\mathcal{B})$. Then we have

$$(\lambda - \mathcal{A})^{-1}C^3x - (\mu - \mathcal{B})^{-1}C^3x = [\lambda(\lambda - \mathcal{A})^{-1}C - C] \\ \times [\mathcal{B}^{-1}C - \mathcal{A}^{-1}C][\lambda(\lambda - \mathcal{B})^{-1}C - C], \quad \lambda \in \rho_C(\mathcal{A}) \cap \rho_C(\mathcal{B}).$$

The following proposition can be proved by induction.

PROPOSITION 1.2.5. Suppose that $\lambda \in \rho_C(\mathcal{A}), n \in \mathbb{N}, x \in D(\mathcal{A}^n) = D((\lambda - \mathcal{A})^n)$ and $y \in (\lambda - \mathcal{A})^n x$. Then we have

$$((\lambda - \mathcal{A})^{-1}C)^{n}(\lambda - \mathcal{A})^{n}x = \{((\lambda - \mathcal{A})^{-1}C)^{n}y\} = \{C^{n}x\}.$$

By Theorem 1.2.4(i), it readily follows that the operator $\lambda(\lambda - \mathcal{A})^{-1}C - C \in L(X)$ is a bounded linear section of the MLO $\mathcal{A}(\lambda - \mathcal{A})^{-1}C$ ($\lambda \in \rho_C(\mathcal{A})$). Inductively, we can prove that, for every $x \in X$, $n \in \mathbb{N}_0$ and $\lambda \in \rho_C(\mathcal{A})$, we have $\operatorname{card}((\lambda - \overline{\mathcal{A}})^{-n}Cx) \leq 1$. Having in mind this fact, as well as the argumentation already seen many times in our previous research studies of *C*-resolvents of single-valued linear operators, we can prove the following extension of [292, Proposition 2.1.14] for MLOs in locally convex spaces.

PROPOSITION 1.2.6. Let $\emptyset \neq \Omega \subseteq \rho_C(\mathcal{A})$ be open, and let $x \in X$.

- (i) The local boundedness of the mapping λ → (λ − A)⁻¹Cx, λ ∈ Ω, resp. the assumption that X is barreled and the local boundedness of the mapping λ → (λ − A)⁻¹C, λ ∈ Ω, implies the analyticity of the mapping λ → (λ − A)⁻¹C³x, λ ∈ Ω, resp. λ → (λ − A)⁻¹C³, λ ∈ Ω. Furthermore, if R(C) is dense in X, resp. if R(C) is dense in X and X is barreled, then the mapping λ → (λ − A)⁻¹Cx, λ ∈ Ω is analytic, resp. the mapping λ → (λ − A)⁻¹C, λ ∈ Ω is analytic.
- (ii) Suppose that R(C) is dense in X. Then the local boundedness of the mapping λ → (λ − A)⁻¹Cx, λ ∈ Ω implies its analyticity as well as Cx ∈ R((λ − Ā)ⁿ), n ∈ N and

(23)
$$\frac{d^{n-1}}{d\lambda^{n-1}}(\lambda - \mathcal{A})^{-1}Cx = (-1)^{n-1}(n-1)!(\lambda - \bar{\mathcal{A}})^{-n}Cx \quad n \in \mathbb{N}.$$

Furthermore, if X is barreled, then the local boundedness of the mapping $\lambda \mapsto (\lambda - \mathcal{A})^{-1}C$, $\lambda \in \Omega$ implies its analyticity as well as $R(C) \subseteq R((\lambda - \mathcal{A}))$

(24)
$$\bar{\mathcal{A}}^{(n)}, n \in \mathbb{N} \text{ and}$$
$$\frac{d^{n-1}}{d\lambda^{n-1}} (\lambda - \mathcal{A})^{-1} C = (-1)^{n-1} (n-1)! (\lambda - \bar{\mathcal{A}})^{-n} C \in L(X), \quad n \in \mathbb{N}$$

(iii) The continuity of mapping $\lambda \mapsto (\lambda - \mathcal{A})^{-1}Cx$, $\lambda \in \Omega$ implies its analyticity and (23). Furthermore, if X is barreled, then the continuity of mapping $\lambda \mapsto (\lambda - \mathcal{A})^{-1}C$, $\lambda \in \Omega$ implies its analyticity and (24).

It is well known that $\rho_C(A)$ need not be an open subset of \mathbb{C} if $C \neq I$ and A is a single-valued linear operator in X (cf. [138, Example 2.5]) and that $\rho(\mathcal{A})$ is an open subset of \mathbb{C} , provided that X is a Banach space and \mathcal{A} is an MLO in X (cf. [199, Theorem 1.6]). The regular C-resolvent set of \mathcal{A} , $\rho_C^r(\mathcal{A})$ for short, is defined as the union of those complex numbers $\lambda \in \rho_C(\mathcal{A})$ for which $(\lambda - \mathcal{A})^{-1}C \in R(X)$, where R(X) denotes the set of all regular bounded linear operators $A \in L(X)$, i.e., the operators $A \in L(X)$ for which there exists a positive constant c > 0 such that for each seminorm $p \in \circledast$ there exists another seminorm $q \in \circledast$ such that $p(A^n x) \leq c^n q(x), x \in X, n \in \mathbb{N}$; the regular resolvent set of \mathcal{A} , $\rho^r(\mathcal{A})$ for short, is then defined by $\rho^r(\mathcal{A}) \coloneqq \rho_I^r(\mathcal{A})$. By the proof of [199, Theorem 1.6], it readily follows that $\rho^r(\mathcal{A})$ is always an open subset of \mathbb{C} . If \mathcal{A} is an MLO, then for each complex numbers $z_1, z_2, \ldots, z_n \in \mathbb{C}$ we have

$$(\mathcal{A} - z_1)(\mathcal{A} - z_2)\dots(\mathcal{A} - z_n) = \mathcal{A}^n + \sum_{k=1}^n a_{n-k}\mathcal{A}^{n-k}$$

where $a_{n-k} = \sum_{1 \leq i_1 < i_2 < \cdots < i_k \leq n} z_{i_1} z_{i_2} \cdots z_{i_k}$. The following polynomial spectral mapping theorem holds for multivalued linear operators in locally convex spaces (cf. [537, Theorem 7.5.3] for the case that C = I and X is a Banach space):

THEOREM 1.2.7. Suppose that $P_n(z) = \sum_{k=0}^n a_k z^k$ is a non-zero complex polynomial of degree $n \in \mathbb{N}_0$, and

$$\lambda - P_n(z) = (-1)^{n+1} a_n (f_1(\lambda) - z) (f_2(\lambda) - z) \dots (f_n(\lambda) - z), \quad \lambda, z \in \mathbb{C}.$$

Let $\emptyset \neq \Omega \subseteq \rho_C(\mathcal{A})$, and let $\lambda \in \mathbb{C}$ be such that $f_1(\lambda), f_2(\lambda), \ldots, f_n(\lambda) \in \Omega$. Then $\lambda \in \rho_{C^n}(P_n(\mathcal{A}))$ and

$$(\lambda - P_n(\mathcal{A}))^{-1}C^n = (-1)^{n+1}a_n(f_1(1) - \mathcal{A})^{-1}C(f_2(\lambda) - \mathcal{A})^{-1}C\dots(f_n(\lambda) - \mathcal{A})^{-1}C.$$

We continue by observing that the generalized resolvent equations hold for *C*-resolvents of multivalued linear operators. More precisely, we have the following theorem which can be proved by induction.

THEOREM 1.2.8. (i) Let $x \in X$, $k \in \mathbb{N}_0$ and $\lambda, z \in \rho_C(\mathcal{A})$ with $z \neq \lambda$. Then the following holds:

$$(z - \mathcal{A})^{-1}C((\lambda - \mathcal{A})^{-1}C)^{k}x = \frac{(-1)^{k}}{(z - \lambda)^{k}}(z - \mathcal{A})^{-1}C^{k+1}x + \sum_{i=1}^{k}\frac{(-1)^{k-i}((\lambda - \mathcal{A})^{-1}C)^{i}C^{k+1-i}x}{(z - \lambda)^{k+1-i}}.$$

(ii) Let $k \in \mathbb{N}_0$, $x, y \in X$, $y \in (\lambda_0 - \mathcal{A})^k x$ and $\lambda_0, z \in \rho_C(\mathcal{A})$ with $z \neq \lambda_0$. Then the following holds:

$$(z - \mathcal{A})^{-1}C^{k+1}x = \frac{(-1)^k}{(z - \lambda_0)^k}(z - \mathcal{A})^{-1}C^{k+1}y + \sum_{i=1}^k \frac{(-1)^{k-i}((\lambda_0 - \mathcal{A})^{-1}C)^iC^{k+1-i}y}{(z - \lambda_0)^{k+1-i}}$$

1.2.1. Fractional powers. In this subsection, we assume that X is a Banach space, $(-\infty, 0] \subseteq \rho(\mathcal{A})$ as well as that there exist finite numbers $M \ge 1$ and $\beta \in (0, 1]$ such that

$$||R(\lambda : \mathcal{A})|| \leq M(1+|\lambda|)^{-\beta}, \quad \lambda \leq 0.$$

Then there exist two positive real constants c > 0 and $M_1 > 0$ such that the resolvent set of \mathcal{A} contains an open region $\Omega = \{\lambda \in \mathbb{C} : |\operatorname{Im} \lambda| \leq (2M_1)^{-1}(c - \operatorname{Re} \lambda)^{\beta}, \operatorname{Re} \lambda \leq c\}$ of complex plane around the nonpositive half-line $(-\infty, 0]$, and we have the estimate $||R(\lambda : \mathcal{A})|| = O((1 + |\lambda|)^{-\beta}), \lambda \in \Omega$.

Let Γ' be the upwards oriented curve $\{\xi \pm i(2M_1)^{-1}(c-\xi)^{\beta} : -\infty < \xi \leq c\}$. We define the fractional power

$$\mathcal{A}^{-\theta} := \frac{1}{2\pi i} \int_{\Gamma'} \lambda^{-\theta} (\lambda - \mathcal{A})^{-1} d\lambda \in L(E)$$

for $\theta > 1 - \beta$. Set $\mathcal{A}^{\theta} := (\mathcal{A}^{-\theta})^{-1}$ $(\theta > 1 - \beta)$. Then the semigroup properties $\mathcal{A}^{-\theta_1}\mathcal{A}^{-\theta_2} = \mathcal{A}^{-(\theta_1+\theta_2)}$ and $\mathcal{A}^{\theta_1}\mathcal{A}^{\theta_2} = \mathcal{A}^{\theta_1+\theta_2}$ hold for $\theta_1, \theta_2 > 1 - \beta$ (it is worth noting here that the fractional power \mathcal{A}^{θ} need not be injective and that the meaning of \mathcal{A}^{θ} is understood in the MLO sense for $\theta > 1 - \beta$).

We endow the vector space $D(\mathcal{A})$ with the norm

$$\|\cdot\|_{[D(\mathcal{A})]} := \inf_{y \in \mathcal{A}} \|y\|.$$

Then $(D(\mathcal{A}), \|\cdot\|_{[D(\mathcal{A})]})$ is a Banach space and, since $0 \in \rho(\mathcal{A})$, the norm $\|\cdot\|_{[D(\mathcal{A})]}$ is equivalent with the following one $\|\cdot\| + \|\cdot\|_{[D(\mathcal{A})]}$. Since $0 \in \rho(\mathcal{A}^{\theta})$, $(D(\mathcal{A}^{\theta}), \|\cdot\|_{[D(\mathcal{A}^{\theta})]})$ is likewise a Banach space and we have the equivalence of norms $\|\cdot\|_{[D(\mathcal{A}^{\theta})]}$ and $\|\cdot\| + \|\cdot\|_{[D(\mathcal{A}^{\theta})]}$ for $\theta > 1 - \beta$ (cf. the proof of [199, Proposition 1.1]).

For any $\theta \in (0, 1)$, the vector space

$$E_{\mathcal{A}}^{\theta} := \left\{ x \in E : \sup_{\xi > 0} \xi^{\theta} \| \xi(\xi + \mathcal{A})^{-1} x - x \| < \infty \right\}$$

becomes one of Banach's when endowed with the norm

$$\|\cdot\|_{E^{\theta}_{\mathcal{A}}} := \|\cdot\| + \sup_{\xi>0} \xi^{\theta} \|\xi(\xi+\mathcal{A})^{-1} \cdot - \cdot\|.$$

It is clear that $E^{\theta}_{\mathcal{A}}$ is continuously embedded in E. We refer the reader to [181, 411, 444] and Section 2.12 for further information concerning interpolation spaces and fractional powers of multivalued linear operators. In Section 2.12, we will see that the main properties of C-resolvent sets of multivalued linear operators hold even if the injectivity of regularizing operator C is disregarded.

1.2.2. Hypercyclic and disjoint hypercyclic multivalued linear operators. The basic facts about topological dynamics of linear continuous operators in Banach and Fréchet spaces can be obtained by consulting the monographs [60] by F. Bayart, E. Matheron and [238] by K.-G. Grosse-Erdmann, A. Peris. Hypercyclicity and disjoint hypercyclicity of multivalued linear operators are relatively new topics in the field of linear topological dynamics. In this subsection, we also inquire into the topological transitivity, topologically mixing property and chaoticity of multivalued linear operators, as well as their disjoint analogues (cf. [67, 69, 71, 73, 106, 108, 137, 292, 390, 409, 470, 471] and [489] for further information on single-valued linear case).

In this subsection, X and Y will be two separable SCLCSs. Set $SL_p(x,\varepsilon) := \{y \in X : p(x-y) < \varepsilon\}, \varepsilon > 0, x \in X, p \in \circledast$. Assume that $C \in L(X)$ is injective. Put $p_C(x) := p(C^{-1}x), p \in \circledast, x \in R(C)$. Then $p_C(\cdot)$ is a seminorm on R(C) and the calibration $(p_C)_{p\in \circledast}$ induces a locally convex topology on R(C); we denote the above space by $[R(C)]_{\circledast}$. Notice that $[R(C)]_{\circledast}$ is a separable SCLCS, and $[R(C)]_{\circledast}$ is a (Fréchet, Banach space) provided that X is. Set $\mathfrak{S}_1 := \{z \in \mathbb{C} : |z| = 1\}$. Suppose that \mathcal{A} is an MLO in X. Then we say that a point $\lambda \in \mathbb{C}$ is an eigenvalue of \mathcal{A} iff there exists a vector $x \in X \setminus \{0\}$ such that $\lambda x \in \mathcal{A}x$; we call x an eigenvector of operator \mathcal{A} corresponding to the eigenvalue λ . Observe that, in purely multivalued case, a vector $x \in X \setminus \{0\}$ can be an eigenvector of operator \mathcal{A} corresponding to the eigenvalue λ . Then we define the MLO $\mathcal{A} \oplus \cdots \oplus \mathcal{A}$

on
$$\underbrace{X \oplus \cdots \oplus X}_{k}$$
 by $D(\underbrace{\mathcal{A} \oplus \cdots \oplus \mathcal{A}}_{k}) := \underbrace{D(\mathcal{A}) \oplus \cdots \oplus D(\mathcal{A})}_{k}$ and
 $\underbrace{\mathcal{A} \oplus \cdots \oplus \mathcal{A}}_{k}(x_1, x_2, \dots, x_k) := \{(y_1, y_2, \dots, y_k) : y_i \in \mathcal{A}x_i \text{ for all } i = 1, 2, \dots, k\}.$

We would like to propose the following definitions (for the sake of clearness and better exposition of material, we will not treat the subspace dynamical properties of MLOs here; cf. J. Banasiak, M. Moszyński [48] and [292, Chapter 3] for more details):

DEFINITION 1.2.9. Let $(\mathcal{A}_n)_{n \in \mathbb{N}}$ be a sequence of multivalued linear operators acting between the spaces X and Y, let \mathcal{A} be an MLO in X, and let $x \in X$. Then we say that:

- (i) x is a hypercyclic vector of the sequence $(\mathcal{A}_n)_{n\in\mathbb{N}}$ iff $x\in\bigcap_{n\in\mathbb{N}}D(\mathcal{A}_n)$ and for each $n\in\mathbb{N}_0$ there exists an element $y_n\in\mathcal{A}_n x$ such that the set $\{y_n:n\in\mathbb{N}\}$ is dense in Y; $(\mathcal{A}_n)_{n\in\mathbb{N}}$ is said to be hypercyclic iff there exists a hypercyclic vector of $(\mathcal{A}_n)_{n\in\mathbb{N}}$;
- (ii) \mathcal{A} is hypercyclic iff the sequence $(\mathcal{A}^n)_{n\in\mathbb{N}}$ is hypercyclic; x is said to be a hypercyclic vector of \mathcal{A} iff x is a hypercyclic vector of the sequence $(\mathcal{A}^n)_{n\in\mathbb{N}}$;
- (iii) x is a periodic point of \mathcal{A} iff $x \in D_{\infty}(\mathcal{A})$ and there exists an integer $n \in \mathbb{N}$ such that $x \in \mathcal{A}^n x$;

(iv) \mathcal{A} is topologically transitive iff for every two open non-empty subsets U, V of X there exists $n \in \mathbb{N}$ such that

(25)
$$U \cap \mathcal{A}^{-n}(V) \neq \emptyset;$$

- (v) \mathcal{A} is topologically mixing iff for every two open non-empty subsets U, V of X there exists $n_0 \in \mathbb{N}$ such that (25) holds for $n \ge n_0$.
- (vi) \mathcal{A} is chaotic iff \mathcal{A} is topologically transitive and the set constituted of all periodic points of \mathcal{A} is dense in X.

DEFINITION 1.2.10. Suppose that $N \in \mathbb{N}$, $(\mathcal{A}_{j,n})_{n \in \mathbb{N}}$ is a sequence of multivalued linear operators acting between the spaces X and Y $(1 \leq j \leq N)$, \mathcal{A}_j is an MLO in X $(1 \leq j \leq N)$ and $x \in X$. Then we say that:

- (i) x is a d-hypercyclic vector of the sequences $(\mathcal{A}_{1,n})_{n \in \mathbb{N}}, \ldots, (\mathcal{A}_{N,n})_{n \in \mathbb{N}}$ iff for each $n \in \mathbb{N}$ there exist elements $y_{j,n} \in \mathcal{A}_{j,n}x$ $(1 \leq j \leq N)$ such that the set $\{(y_{1,n}, y_{2,n}, \ldots, y_{N,n}) : n \in \mathbb{N}\}$ is dense in Y^N ; the sequences $(\mathcal{A}_{1,n})_{n \in \mathbb{N}}, \ldots, (\mathcal{A}_{N,n})_{n \in \mathbb{N}}$ are called d-hypercyclic iff there exists a d-hypercyclic vector of $(\mathcal{A}_{1,n})_{n \in \mathbb{N}}, \ldots, (\mathcal{A}_{N,n})_{n \in \mathbb{N}};$
- (ii) x is a d-hypercyclic vector of the operators $\mathcal{A}_1, \ldots, \mathcal{A}_N$ iff x is a d-hypercyclic vector of the sequences $(\mathcal{A}_1^n)_{n \in \mathbb{N}}, \ldots, (\mathcal{A}_N^n)_{n \in \mathbb{N}}$; the operators $\mathcal{A}_1, \ldots, \mathcal{A}_N$ are called d-hypercyclic iff there exists a d-hypercyclic vector of $\mathcal{A}_1, \ldots, \mathcal{A}_N$.

DEFINITION 1.2.11. Suppose that $N \in \mathbb{N}$, $(\mathcal{A}_{j,n})_{n \in \mathbb{N}}$ is a sequence of multivalued linear operators acting between the spaces X and Y $(1 \leq j \leq N)$, and \mathcal{A}_j is an MLO in X $(1 \leq j \leq N)$. Then we say that:

(i) the sequences $(\mathcal{A}_{1,n})_{n \in \mathbb{N}}, \ldots, (\mathcal{A}_{N,n})_{n \in \mathbb{N}}$ are d-topologically transitive iff for every open non-empty subset U of X and for every open non-empty subsets V_1, \ldots, V_N of Y, there exists $n \in \mathbb{N}$ such that

(26)
$$U \cap \mathcal{A}_{1,n}^{-1}(V_1) \cap \cdots \cap \mathcal{A}_{N,n}^{-1}(V_N) \neq \emptyset;$$

- (ii) the sequences $(\mathcal{A}_{1,n})_{n \in \mathbb{N}}, \ldots, (\mathcal{A}_{N,n})_{n \in \mathbb{N}}$ are d-topologically mixing iff for every open non-empty subset $U \subseteq X$ and for every open non-empty subsets V_1, \ldots, V_N of Y, there exists $n_0 \in \mathbb{N}$ such that, for every $n \ge n_0$, we have that (26) holds;
- (iii) the operators $\mathcal{A}_1, \ldots, \mathcal{A}_N$ are d-topologically transitive (d-topologically mixing) iff the sequences $(\mathcal{A}_1^n)_{n \in \mathbb{N}}, \ldots, (\mathcal{A}_N^n)_{n \in \mathbb{N}}$ are d-topologically transitive (d-topologically mixing).

DEFINITION 1.2.12. Given $N \ge 2$, the multivalued linear operators $\mathcal{A}_1, \ldots, \mathcal{A}_N$ acting on X are said to be d-chaotic iff $\mathcal{A}_1, \ldots, \mathcal{A}_N$ are d-topologically transitive and the set of periodic elements, denoted by $\mathcal{P}(\mathcal{A}_1, \mathcal{A}_2, \ldots, \mathcal{A}_N) := \{(x_1, x_2, \ldots, x_N) \in X^N : \exists n \in \mathbb{N} \text{ with } x_j \in \mathcal{A}_j^n x_j, j \in \mathbb{N}_N\}$, is dense in X^N .

REMARK 1.2.13. (i) Let the sequences $(\mathcal{A}_{1,n})_{n \in \mathbb{N}}, \ldots, (\mathcal{A}_{N,n})_{n \in \mathbb{N}}$ be dhypercyclic (d-topologically transitive, d-topologically mixing). Then for each $j \in \mathbb{N}_N$ the single operator sequence $(\mathcal{A}_{j,n})_{n \in \mathbb{N}}$ is hypercyclic (topologically transitive, topologically mixing).

- (ii) Suppose that x is a periodic point of \mathcal{A} . Then $x \in D_{\infty}(\mathcal{A})$ and there exists an integer $n \in \mathbb{N}$ such that $x \in \mathcal{A}^n x$. This implies the existence of elements $y_j \in X$ $(1 \leq j \leq n-1)$ such that $(x, y_1) \in \mathcal{A}, (y_1, y_2) \in \mathcal{A}, \ldots, (y_{n-1}, x) \in \mathcal{A}$. Repeating this sequence, we easily get that $x \in \mathcal{A}^{kn} x$ for all $k \in \mathbb{N}$, so that the periodic points of \mathcal{A} form a linear submanifold of X. Similarly, the set $\mathcal{P}(\mathcal{A}_1, \mathcal{A}_2, \ldots, \mathcal{A}_N)$ is a linear submanifold of X^N .
- (iii) Suppose that Ω is an open connected subset of $\mathbb{K} = \mathbb{C}$ satisfying $\Omega \cap \mathfrak{S}_1 \neq \emptyset$, as well as that $N \ge 2$ and the multivalued linear operators $\mathcal{A}_1, \ldots, \mathcal{A}_N$ act on X. Let $f: \Omega \to X \smallsetminus \{0\}$ be an analytic mapping such that $\lambda f(\lambda) \in \mathcal{A}_j f(\lambda)$ for all $\lambda \in \Omega$ and $j \in \mathbb{N}_N$. Set $\tilde{X} := \overline{span}\{f(\lambda): \lambda \in \Omega\}$. Then it is very simple to show that, for every $l \in \mathbb{N}$, the set $\mathcal{P}((\mathcal{A}_{1|\tilde{X}})^l, \ldots, (\mathcal{A}_N|_{\tilde{X}})^l)$ is dense in \tilde{X}^N ; cf. also the proof of Theorem 1.2.17 below.

REMARK 1.2.14. In Definition 1.2.9–Definition 1.2.12, our assumptions on Xand Y, as well as on $(\mathcal{A}_n)_{n\in\mathbb{N}}$, \mathcal{A} , $(\mathcal{A}_{j,n})_{n\in\mathbb{N}}$ and \mathcal{A}_j can be substantially relaxed $(1 \leq j \leq N)$. It suffices to suppose that X and Y are topological spaces as well as that \mathcal{A}_n and $\mathcal{A}_{j,n}$ (\mathcal{A} and \mathcal{A}_j) are binary relations from X to Y (binary relations on X). To make this precise, we need to recall some basic facts about binary relations and their compositions. Suppose, for the time being, that X, Y, Z and T are given non-empty sets. The notion of reflexivity, symmetry, anti-symmetry and transitivity of a binary relation $\rho \subseteq X \times Y$ is defined in the usual way; the classes of RST relations and partial order relations on the set X are well-known. If $\rho \subseteq X \times Y$ and $\sigma \subseteq Z \times T$ with $Y \cap Z \neq \emptyset$, then we define $\rho^{-1} \subseteq Y \times X$ and $\sigma \circ \rho \subseteq X \times T$ by $\rho^{-1} := \{(y, x) \in Y \times X : (x, y) \in \rho\}$ and

 $\sigma \circ \rho := \{ (x,t) \in X \times T : \exists y \in Y \cap Z \text{ such that } (x,y) \in \rho \text{ and } (y,t) \in \sigma \},\$

respectively. Domain and range of ρ are defined by $D(\rho) := \{x \in X : \exists y \in Y \text{ such that } (x,y) \in X \times Y\}$ and $R(\rho) := \{y \in Y : \exists x \in X \text{ such that } (x,y) \in X \times Y\}$, respectively; $\rho(x) := \{y \in Y : (x,y) \in \rho\}$ $(x \in X), x, \rho, y \Leftrightarrow (x,y) \in \rho$.

Assuming ρ is a binary relation on X and $n \in \mathbb{N}$, we define $\rho^n := \rho \circ \circ \circ \rho$ *n*-times inductively; $\rho^{-n} := (\rho^n)^{-1}$ and $\rho^0 := \Delta_X := \{(x, x) : x \in X\}.$

Set $D_{\infty}(\rho) := \bigcap_{n \in \mathbb{N}} D(\rho^n)$ and observe that the integer powers of an MLO \mathcal{A} in an SCLCS are introduced, actually, by using the above definition for powers of the associated linear relation \mathcal{A} . Keeping in mind these notions, it is almost trivial to restate Definition 1.2.9–Definition 1.2.12 for (sequences of) binary relations between the topological spaces X and Y. In general, we do not assume that X or Y has a linear vector space structure; unquestionably, this assumption is not so pleasant because we are no longer in a position to clarify various types of (d-)hypercyclic and (d-)spectral criteria ensuring some kind of hypercyclic or topologically mixing behaviour of considered binary relations (cf. the continuation of this subsection for more details). In what follows, we will present a few illustrative examples:

(i) Let ρ be an RST relation on X. Then it is clear that ρ is hypercyclic iff there exists an element $x \in X$ such that $C_x := \{y \in X : x \rho \}$ contains a sequence that is dense in X. If so, then any element of C_x is a hypercyclic element of ρ .

- (ii) Let X = G be finite and equipped with the discrete topology, let |G| > 1, and let ρ be a symmetric relation on G such that, for every $g \in G$, we have $(g,g) \notin \rho$. As is well-known, (G,ρ) is said to be a simple graph [448]. It can be simply proved that the graph G is connected iff G is hypercyclic iff G is chaotic; if this is the case, then any element of G is a hypercyclic element of ρ . Choosing some different topologies on G, as in part (iii) below, Definition 1.2.9 can be used to generalize the notion of connectivity of simple graphs. It is also worth noting that ρ need not be topologically mixing if it is topologically transitive; for example, if G is a square $x_1x_2x_3x_4$, then there does not exist an odd number $n \in \mathbb{N}$ such that $x_3 \in \rho^n(x_1)$; cf. Definition 1.2.9(v) with $U = \{x_3\}$ and $V = \{x_1\}$.
- (iii) Let X = G be finite, let (G, ρ) be a simple graph without isolated vertexes, and let τ be the topology on G. Denote by G_1, \ldots, G_k the connected components of graph (G, ρ) , where $k \in \mathbb{N}$. Then ρ is hypercyclic iff there exists a number $i \in \mathbb{N}_k$ such that G_i is dense in (G, τ) , when any element of G_i is a hypercyclic element of ρ . For example, let $G = \{a, b, c, d\}$, and let $\tau_1 = \{\emptyset, \{a\}, \{b\}, \{c\}, \{a, b\}, \{a, c\}, \{b, c\}, \{a, b, c\}, \{a, b, c, d\}\}$ and $\tau_2 = \{\emptyset, \{a, b\}, \{a, b, c, d\}\}$ (according to Wikipedia, there exist 355 distinct topologies on G but only 33 inequivalent). Set

$$\rho := \{ (a, d), (d, a), (b, c), (c, b) \}.$$

Then ρ is not hypercyclic in (G, τ_1) and ρ is hypercyclic in (G, τ_2) . Observe, finally, that ρ is topologically transitive iff for every two open nonempty subsets U, V of X there exists $i \in \mathbb{N}_k$ such that $U \cap G_i \neq \emptyset$ and $V \cap G_i \neq \emptyset$.

Hypercyclic and disjoint hypercyclic extensions of binary relations, introduced here for multivalued linear operators, can be also analyzed; cf. our research study [108] for further information concerning this question. In that paper, we will scrutinize hypercyclic and disjoint hypercyclic properties of operators on Cayley graphs, as well.

Before proceeding further, it should be worthwhile to mention that, in singlevalued linear case, J. Bés and A. Peris called the notion introduced in Definition 1.2.10 by d-universality (cf. also [236, Definition 2]).

EXAMPLE 1.2.15. It is clear that any multivalued linear extension of a hypercyclic (chaotic) single-valued linear operator is again hypercyclic (chaotic). It is also worth noting that Definition 1.2.9 prescribes some very strange situations in which even zero can be a hypercyclic vector: Let $\mathcal{A} := \{0\} \times W$, where W is a dense linear submanifold of X. Then \mathcal{A} is hypercyclic, zero is the only hypercyclic vector of \mathcal{A} and there is no single-valued linear restriction of \mathcal{A} that is hypercyclic (in particular, a hypercyclic MLO need not be densely defined and the inverse of a hypercyclic MLO need not be hypercyclic, in contrast to the single-valued linear case, see also [73, Problem 1]); furthermore, Definition 1.2.10 is very strange in multivalued linear operators setting because the operators $\mathcal{A}, \ldots, \mathcal{A}$ are all the same and d-hypercyclic with zero being its only d-hypercyclic vector (as the analysis performed by J. Bés and A. Peris on p. 299 of [73] shows, a single-valued linear operator A and its constant multiple cA cannot be d-hypercyclic, $c \in \mathbb{K}$). Observe, finally, that the non-triviality of submanifold A0 is not so directly and essentially connected with the hypercyclicity of A: Let $A := X \times W$, where W is a non-dense linear submanifold of X. Then A cannot be hypercyclic (topologically transitive).

Let $(O_n)_{n \in \mathbb{N}}$ be an open base of the topology of X, and let $O_n \neq \emptyset$ for every $n \in \mathbb{N}$. Then the set consisting of hypercyclic vectors of an arbitrary multivalued linear operator \mathcal{A} , denoted shortly by $\mathrm{HC}(\mathcal{A})$, can be computed by

(27)
$$\operatorname{HC}(\mathcal{A}) = \bigcap_{n \in \mathbb{N}} \bigcup_{k \in \mathbb{N}} \mathcal{A}^{-k}(O_n).$$

Similarly, the set consisting of all d-hypercyclic vectors of arbitrary multivalued linear operators $\mathcal{A}_1, \ldots, \mathcal{A}_N$, denoted shortly by $\mathrm{HC}(\mathcal{A}_1, \ldots, \mathcal{A}_N)$, can be computed by

(28)
$$\operatorname{HC}(\mathcal{A}_1,\ldots,\mathcal{A}_N) = \bigcap_{j_1,j_2,\ldots,j_N \in \mathbb{N}} \bigcup_{k \in \mathbb{N}} \bigcap_{l=1}^N \mathcal{A}_l^{-k}(O_{j_l}).$$

The formulae (27)–(28) become most important in the case that the powers of operator \mathcal{A} (operators $\mathcal{A}_1, \ldots, \mathcal{A}_N$) are continuous; cf. [120, Section II.3] for the notion, and [236] for more details about this subject and universal families of continuous linear operators. The study of some important subcases of universality, like supercyclicity and positive supercyclicity, is also without the scope of our analysis (cf. [291,292,390,462] and references cited therein for more details on the subject).

Suppose now that \mathcal{A} is an MLO in X, as well as that Y is another SCLCS and $\phi: X \to Y$ is a linear topological homeomorphism (cf. also [236, Proposition 4] and [409, Proposition 1.3.10]). Then we define the operator \mathcal{A}_Y in Y by

$$D(\mathcal{A}_Y) := \phi(D(\mathcal{A}))$$
 and $\mathcal{A}_Y(\phi(x)) := \phi(\mathcal{A}x), x \in D(\mathcal{A}).$

It is checked at once that \mathcal{A}_Y is an MLO in Y. Keeping this fact in mind, we can formulate a great number of (d-)hypercyclic comparison principles for multivalued linear operators. We leave these questions to the interesed reader.

It is an elementary fact that the point spectrum of adjoint of a hypercyclic continuous single-valued linear operator has to be empty [284]. As the next proposition shows, the same thing holds for hypercyclic multivalued linear operators:

PROPOSITION 1.2.16. Let \mathcal{A} be a hypercyclic MLO in X. Then $\sigma_p(\mathcal{A}^*) = \emptyset$.

PROOF. Let x be a hypercyclic vector of \mathcal{A} ; hence, for every $n \in \mathbb{N}$, there exists an element $y_n \in \mathcal{A}^n x$ such that the set $\{y_n : n \in \mathbb{N}\}$ is dense in X. Suppose to the contrary that there exist $\lambda \in \mathbb{K}$ and $x^* \in X^* \setminus \{0\}$ such that $\lambda x^* \in \mathcal{A}^* x^*$, i.e., that $\langle x^*, y \rangle = \lambda \langle x^*, x \rangle$, whenever $y \in \mathcal{A}x$. It is clear that $\langle x^*, y_n \rangle = \lambda^n \langle x^*, x \rangle$ for all $n \in \mathbb{N}$, so that the assumption $\langle x^*, x \rangle = 0$ implies $\langle x^*, y_n \rangle = 0$ for all $n \in \mathbb{N}$, and therefore, $x^* = 0$. So, $\langle x^*, x \rangle \neq 0$. If $|\lambda| \leq 1$, then $|\langle x^*, y_n \rangle| = |\lambda^n \langle x^*, x \rangle| \leq |\langle x^*, x \rangle|$ for all $n \in \mathbb{N}$; this would imply $|\langle x^*, u \rangle| \leq |\langle x^*, x \rangle|$ for all $u \in X$, which is a contradiction since $|\langle x^*, nu \rangle| \to +\infty$ for any $u \in X$ such that $\langle x^*, u \rangle \neq 0$. If $|\lambda| > 1$, then $|\langle x^*, y_n \rangle| = |\lambda^n \langle x^*, x \rangle| \ge |\langle x^*, x \rangle|$ for all $n \in \mathbb{N}$; this would imply $|\langle x^*, u \rangle| \ge |\langle x^*, x \rangle|$ for all $u \in X$, which is a contradiction since $|\langle x^*, u/n \rangle| \to 0$ for any $u \in X$, and $\langle x^*, x \rangle \neq 0$.

The following theorem extends the well-known result of R. M. Aron, J. B. Seoane-Sepúlveda and A. Weber [**31**, Theorem 2.1]. This is, actually, a kind of Godefroy–Shapiro and Dech–Schappacher–Webb Criterion (continuous version) for multivalued linear operators.

THEOREM 1.2.17. Suppose that Ω is an open connected subset of $\mathbb{K} = \mathbb{C}$ satisfying $\Omega \cap \mathfrak{S}_1 \neq \emptyset$. Let $f: \Omega \to X \setminus \{0\}$ be an analytic mapping such that $\lambda f(\lambda) \in \mathcal{A}f(\lambda)$ for all $\lambda \in \Omega$. Set $\tilde{X} := \overline{\operatorname{span}\{f(\lambda) : \lambda \in \Omega\}}$. Then the operator $\mathcal{A}_{|\tilde{X}}$ is topologically mixing in the space \tilde{X} and the set of periodic points of $\mathcal{A}_{|\tilde{X}}$ is dense in \tilde{X} .

PROOF. The proof of theorem is very similar to that of Theorem 2.10.3 and we will only outline the most relevant details. Without loss of generality, we may assume that $\tilde{X} = X$. If Ω_0 is an arbitrary subset of Ω which admits a cluster point in Ω , then the (weak) analyticity of mapping $\lambda \mapsto f(\lambda), \lambda \in \Omega$ shows that $\Psi(\Omega_0) := \operatorname{span}\{f(\lambda) : \lambda \in \Omega_0\}$ is dense in X. Further on, it is clear that there exist numbers $\lambda_0 \in \Omega \cap \mathfrak{S}_1$ and $\delta > 0$ such that any of the sets $\Omega_{0,+} := \{\lambda \in \Omega : |\lambda - \lambda_0| < 0\}$ $\delta, |\lambda| > 1$ and $\Omega_{0,-} := \{\lambda \in \Omega : |\lambda - \lambda_0| < \delta, |\lambda| < 1\}$ admits a cluster point in Ω . Suppose that U and V are two open non-empty subsets of X. Then there exist $y, z \in X$, $\varepsilon > 0$, $p, q \in \circledast$ such that $SL_p(y, \varepsilon) \subseteq U$ and $SL_q(z, \varepsilon) \subseteq V$. We may assume that $y \in \Psi(\Omega_{0,-}), z \in \Psi(\Omega_{0,+}), y = \sum_{i=1}^{n} \beta_i f(\lambda_i), z = \sum_{j=1}^{m} \gamma_j f(\tilde{\lambda_j}),$ where $\alpha_i, \beta_j \in \mathbb{C} \setminus \{0\}, \lambda_i \in \Omega_{0,-}$ and $\tilde{\lambda_j} \in \Omega_{0,+}$ for $1 \leq i \leq n$ and $1 \leq j \leq m$. Put $z_t := \sum_{j=1}^m \frac{\gamma_j}{\tilde{\lambda_j}^t} f(\tilde{\lambda_j})$ and $x_t := y + z_t, t \geq 0$. Then $\{x_t, y, z_t\} \subseteq D_{\infty}(\mathcal{A}), t \geq 0$ and it can be easily seen that $z \in \mathcal{A}^n z_n$, $n \in \mathbb{N}$ and $\omega_n := z + \sum_{i=1}^n \beta_i \lambda_i^n f(\lambda_i) \in \mathcal{A}^n x_n$, $n \in \mathbb{N}$, as well as that there exists $n_0(\varepsilon) \in \mathbb{N}$ such that, for every $n \ge n_0(\varepsilon)$, $x_n \in SL_p(y,\varepsilon)$ and $w_n \in SL_q(z,\varepsilon)$. Therefore, \mathcal{A} is topologically mixing. Since the set $\Omega \cap \exp(2\pi i \mathbb{Q})$ has a cluster point in Ω , the proof that the set of periodic points of \mathcal{A} is dense in X can be given as in that of [31, Theorem 2.1].

REMARK 1.2.18. It is clear that the validity of implication:

 $\langle x^*, f(\lambda) \rangle = 0, \ \lambda \in \Omega \text{ for some } x^* \in X^* \Rightarrow x^* = 0$ $\tilde{X} - X$

yields that $\tilde{X} = X$.

Now we would like to illustrate Theorem 1.2.17 with a concrete example that is very similar to Example 2.10.6. It provides the existence of a substantially large class of topologically mixing multivalued linear operators with dense set of periodic points.

EXAMPLE 1.2.19. Suppose that A is a closed linear operator on X satisfying that there exist an open connected subset $\emptyset \neq \Lambda$ of \mathbb{C} and an analytic mapping $g: \Lambda \to X \setminus \{0\}$ such that $Ag(\nu) = \nu g(\nu), \nu \in \Lambda$. Let P(z) and Q(z) be nonzero complex polynomials, let $R := \{z \in \mathbb{C} : P(z) = 0\}, \Lambda' := \Lambda \setminus R$, and let $\tilde{X} := \overline{span\{g(\lambda) : \lambda \in \Lambda\}}$. Suppose that

$$\frac{Q}{P}(\Lambda') \cap S_1 \neq \emptyset.$$

Then Theorem 1.2.17 implies that the parts of multivalued linear operators $Q(\mathcal{A})P(\mathcal{A})^{-1}$ and $P(\mathcal{A})^{-1}Q(\mathcal{A})$ in \tilde{X} are topologically mixing in the space \tilde{X} , and that the sets of periodic points of these operators are dense in \tilde{X} .

It is worth noting that L. Bernal-González investigated in [67, Theorem 4.3] a disjoint analogue of Theorem 1.2.17 for continuous linear operators. Now we will state the following version of afore-mentioned theorem for multivalued linear operators without going into full details concerning the case in which there exists an integer $p_0 \in \mathbb{N}_N^0$ for which the set D_{p_0} is not total in X. The proof goes similarly and therefore omitted.

THEOREM 1.2.20. Suppose that $N \in \mathbb{N}$ and $\mathcal{A}_1, \ldots, \mathcal{A}_N$ are given MLOs acting on X. Let for each $p \in \mathbb{N}_N^0$ there exists a total set D_p (that is, the linear span of D_p is dense in X) such that the following conditions hold:

- (i) Any element of the set D_p is an eigenvector of any operator \mathcal{A}_j $(p \in \mathbb{N}_N^0, j \in \mathbb{N}_N)$; if $e \in D_p$, then there exists an eigenvalue $\lambda_{j,p}(e)$ of the operator \mathcal{A}_j for which $\lambda_{j,p}(e)e \in \mathcal{A}_je$ and (ii)–(iii) hold, where:
- (ii) $\lambda_{j,0}(e) \in int(\mathfrak{S}_1), j \in \mathbb{N}_N, e \in D_0 \text{ and } \lambda_{j,j}(e) \in ext(\mathfrak{S}_1), j \in \mathbb{N}_N, e \in D_j;$
- (iii) Suppose $i, j \in \mathbb{N}_N$ and $i \neq j$. Then, for every $e \in D_i$, we have $|\lambda_{j,i}(e)| < |\lambda_{i,i}(e)|$.

Then the operators $\mathcal{A}_1, \ldots, \mathcal{A}_N$ are d-topologically mixing.

Now we will illustrate Theorem 1.2.20 with a simple example pointing out that the continuity of operators can be neglected from the formulation of [67, Theorem 4.3] (the metrizability of state space X is also inrelevant in our approach), as well as that there exists a great number of Banach function spaces where this extended version of afore-mentioned theorem can be applied (cf. also [67, Final questions, 2.]). Numerous other examples involving multivalued linear operators can be given similarly, by using the analysis from Example 1.2.19; cf. also [35], Example 2.11.6 and [457] for certain unbounded differential operators that we can employ here.

EXAMPLE 1.2.21. [259] Suppose that p > 2. Let X be a symmetric space of non-compact type and rank one, let P_p be the parabolic domain defined in the proof of [259, Theorem 3.1], and let $c_p > 0$ be the apex of P_p . Then we know that $int(P_p) \subseteq \sigma_p(\Delta_{X,p}^{\natural})$, where $\Delta_{X,p}^{\natural}$ denotes the corresponding Laplace– Beltrami operator acting on $L_{\natural}^{p}(X)$; furthermore, there exists an analytic function $g: int(P_p) \to L_{\natural}^{p}(X)$ such that $\Delta_{X,p}^{\natural}g(\lambda) = \lambda g(\lambda), \lambda \in int(P_p)$ and the set $\Psi(\Omega) :=$ $\{g(\lambda): \lambda \in \Omega\}$ is total in X for any open, non-empty subset Ω of $int(P_p)$. Suppose that $N \ge 2, -1 - c_p < a_1 < a_2 < \cdots < a_N < 1 - c_p$ and, for every $i \in \mathbb{N}_N$, there exists a point $\lambda_i \in int(P_p)$ such that $|\lambda_i + a_i| > \max(1, |\lambda_i + a_j|)$ for all $j \in \mathbb{N}_N \setminus \{i\}$ (this case can really occur). An application of Theorem 1.2.20 (with $D_i = \Psi(\Omega_i)$, Ω_0 is a small ball around c_p + and Ω_i is a small ball around $\lambda_i, i \in \mathbb{N}_N$) shows that the operators $\Delta_{X,p}^{\natural} + a_1, \Delta_{X,p}^{\natural} + a_2, \dots, \Delta_{X,p}^{\natural} + a_N$ are d-topologically mixing; furthermore, by Remark 1.2.13, the set $\mathcal{P}(\Delta_{X,p}^{\natural} + a_1, \dots, \Delta_{X,p}^{\natural} + a_N)$ is dense in $(L_{\natural}^p(X))^N$. Hence, the operators $\Delta_{X,p}^{\natural} + a_1, \Delta_{X,p}^{\natural} + a_2, \dots, \Delta_{X,p}^{\natural} + a_N$ are d-chaotic.

It is almost straigtforward to state a continuous version of Theorem 1.2.20 for abstract degenerate fractional inclusions in separable SCLCSs (see also Theorem 3.6.3 below).

The Hypercyclicity Criterion for single-valued linear operators has been discovered independently by C. Kitai [284] and R. M. Gethner, J. H. Shapiro [224]. It is worth noting that this criterion can be formulated, in a certain way, for multivalued linear operators and that we do not need any type of continuity or closedness of operators under examination for its validity; the proof of next proposition can be obtained by the methods of Theorem 2.10.7 (continuous version), which is omitted here.

PROPOSITION 1.2.22. (Hypercyclicity Criterion for MLOs) Let \mathcal{A} be an MLO in X. Suppose that $(a_n)_{n\in\mathbb{N}}$ is a strictly increasing sequence of positive integers. Let the set X_0 , consisting of those elements $y \in X$ for which there exists a sequence $(y_n)_{n\in\mathbb{N}}$ in X such that $y_n \in \mathcal{A}^{a_n}y$, $n \in \mathbb{N}$ and $\lim_{n\to\infty} y_n = 0$, be dense in X. Furthermore, let the set X_{∞} , consisting of those elements $z \in X$ for which there exist a null sequence $(\omega_n)_{n\in\mathbb{N}}$ in X and a sequence $(u_n)_{n\in\mathbb{N}}$ in X such that $u_n \in \mathcal{A}^{a_n}\omega_n$, $n \in \mathbb{N}$ and $\lim_{n\to\infty} u_n = z$, be also dense in X. Then $\underbrace{\mathcal{A} \oplus \cdots \oplus \mathcal{A}}_k$ is topologically transitive $(k \in \mathbb{N})$.

The question of whether any continuous hypercyclic single-valued linear operator on Banach space satisfies the Hypercyclicity Criterion was open for a long time; as is well-known, the negative answer was given by M. De La Rosa and C. Read in [129]. Now we will state a version of d-Hypercyclicity Criterion for MLOs; see [73, Proposition 2.6, Theorem 2.7, Remark 2.8] for single-valued linear case.

PROPOSITION 1.2.23. (d-Hypercyclicity Criterion for MLOs) Let $N \in \mathbb{N}$, and let \mathcal{A}_j be an MLO in X $(1 \leq j \leq N)$. Suppose that $(a_n)_{n \in \mathbb{N}}$ is a strictly increasing sequence of positive integers. Let the set X_0 , consisting of those elements $y \in X$ satisfying that for each $j \in \mathbb{N}_N$ there exists a sequence $(y_{n,j})_{n \in \mathbb{N}}$ in X such that $y_{n,j} \in \mathcal{A}_j^{a_n} y$, $n \in \mathbb{N}$ and $\lim_{n\to\infty} y_{n,j} = 0$, be dense in X. Furthermore, let for each $j \in \mathbb{N}_N$ there exists a dense set $X_{\infty,j}$, consisting of those elements $z \in X$ for which there exist elements $\omega_{n,i}(z)$ and $u_{n,i,j}(z)$ in X $(n \in \mathbb{N}, 1 \leq i \leq N)$ such that $(\omega_{n,j}(z))_{n \in \mathbb{N}}$ is a null sequence in X, $u_{n,i,j}(z) \in \mathcal{A}_j^{a_n} \omega_{n,i}(z)$, $n \in \mathbb{N}$ and $\lim_{n\to\infty} u_{n,i,j} = \delta_{i,j} z$ $(1 \leq i \leq N)$. Then the operators $\mathcal{A}_1 \oplus \cdots \oplus \mathcal{A}_1, \ldots, \mathcal{A}_N \oplus \cdots \oplus \mathcal{A}_N$ are d-topologically transitive $(k \in \mathbb{N})$.

PROOF. We will only outline the main details of proof. Since for any multivalued linear operator \mathcal{A} we have

$$\left(\underbrace{\mathcal{A} \oplus \cdots \oplus \mathcal{A}}_{k}\right)^{n} = \underbrace{\mathcal{A}^{n} \oplus \cdots \oplus \mathcal{A}^{n}}_{k}, \quad k, n \in \mathbb{N},$$

 $j \leq N$). The remainder of the proof can be given by repeating verbatim the

we may assume without loss of generality that k = 1 (otherwise, we can pass to the subsets X_0^k , $X_{\infty,j}^k$ $(1 \leq j \leq N)$, as well as to the tuples $(z, \ldots, z), (y_{n,j}, \ldots, y_{n,j}),$ $(\omega_{n,i},\ldots,\omega_{n,i}), (u_{n,i,j},\ldots,u_{n,i,j})$, each of which have exactly k components $(1 \leq 1)$

Observe that the assertions of Proposition 1.2.22 and Proposition 1.2.23 can be formulated for the sequences of multivalued linear operators (cf. **[73**, Remark 2.8 and the proof of Theorem 1.2.24 below), as well as that Proposition 1.2.22 and Proposition 1.2.23 give sufficient conditions for (d-)topological transitivity of operators. A sufficient condition for hypercyclicity of linear operators in Banach spaces, the real Hypercyclicity Criterion for linear operators, if we could say something like that, was stated for the first time by J. Bès, K. C. Chen and S. M. Seubert [69, Theorem 2.1] (cf. also R. deLaubenfels, H. Emamirad and K.-G. Grosse-Erdmann [137, Theorem 2.3]). As mentioned on p. 49 of [137], neither of the assertions [69, Theorem 2.1] and [137, Theorem 2.3] is not contained in the other. We continue by stating the following d-hypercyclic analogue of [137, Theorem 2.3] for single-valued linear operators in Fréchet spaces:

THEOREM 1.2.24. Suppose that X is a Fréchet space, T_1, \ldots, T_N are singlevalued linear operators on X $(1 \leq j \leq N)$ and $C \in L(X)$ is injective. Suppose that there exists a subset X_0 of $D_{\infty}(T_1) \cap \cdots \cap D_{\infty}(T_N)$, dense in X, as well as dense subsets X_1, \ldots, X_N of X and mappings $S_{j,n} \colon X_j \to D_{\infty}(T_1) \cap \cdots \cap D_{\infty}(T_N)$ $(1 \leq i \leq N, n \in \mathbb{N})$ such that:

- (i) $\lim_{n \to \infty} T_i^n x_0 = 0, \ x_0 \in X_0,$
- (ii) $\lim_{n\to\infty} S_{j,n} x_j = 0, x_j \in X_j, 1 \leq j \leq N$,

corresponding part of proof of [73, Proposition 2.6].

- (iii) $\lim_{n\to\infty} [T_j^n S_{i,n} x_j \delta_{j,i} x_j] = 0, x_j \in X_j, 1 \leq i, j \leq N,$ (iv) $R(C) \subseteq D_{\infty}(T_1) \cap \cdots \cap D_{\infty}(T_N)$ and $T_j^n C \in L(X), 1 \leq j \leq N,$
- (v) $CT_j x = T_j Cx, x \in D_{\infty}(T_j), 1 \leq j \leq N$,
- (vi) R(C) is dense in X.

Then the operators T_1, \ldots, T_N are d-hypercyclic.

PROOF. Define the operators $\mathfrak{T}_{j,n} \in L([R(C)]_{\circledast}, X)$ by $\mathfrak{T}_{j,n}(Cx) := T_j^n Cx$, $x \in X$ (cf. (iv)). It suffices to show that the sequences $(\mathfrak{T}_{1,j})_{j \in \mathbb{N}}, \ldots, (\mathfrak{T}_{N,j})_{j \in \mathbb{N}}$ are d-hypercyclic. Since the final conclusions of [73, Remark 2.8] also hold for sequences of continuous linear operators acting between different spaces, we need to prove the existence of a dense subset X'_0 of $[R(C)]_{\circledast}$, dense subsets X'_1, \ldots, X'_N of X and mappings $S'_{i,n}: X'_i \to [R(C)]$ $(1 \leq j \leq N, n \in \mathbb{N})$ such that the following holds:

- (a) $\lim_{n\to\infty} \mathfrak{T}_{j,n} x'_0 = 0, \ x'_0 \in X'_0,$ (b) $\lim_{n\to\infty} S'_{j,n} x'_j = 0, \ x'_j \in X'_j, \ 1 \leq j \leq N, \text{ and}$ (c) $\lim_{n\to\infty} [\mathfrak{T}_{j,n} S'_{i,n} x'_j \delta_{j,i} x'_j] = 0, \ x'_j \in X'_j, \ 1 \leq i, j \leq N.$

Set $X'_j := C(X_j), j \in \mathbb{N}^0_N$ and $S'_{j,n} \colon X'_j \to R(C)$ by $S'_{j,n}(Cx_j) := CS_{j,n}x, x_j \in X_j$ $(1 \leq j \leq N, n \in \mathbb{N})$. By (vi) and the density of X_j in X, we get that X'_0 is dense in $[R(C)]_{\circledast}$ and X'_{i} is dense in X $(1 \leq i \leq N)$. The property (b) follows immediately from (ii) and definition of $S'_{j,n}$. The property (a) follows by making use the fact

that X_0 belongs to $D_{\infty}(T_1) \cap \cdots \cap D_{\infty}(T_N)$, as well as from (v), (i) and definition of $\mathfrak{T}_{i,n}$. By (iii), (v) and inclusion $R(S_{i,n}) \subseteq D_{\infty}(T_1) \cap \cdots \cap D_{\infty}(T_N)$, we have

$$\lim_{n \to \infty} [\mathfrak{T}_j^n S'_{i,n} x'_j - \delta_{j,i} x'_j] = \lim_{n \to \infty} [T_j^n C S_{i,n} x_j - C \delta_{j,i} x_j]$$
$$= \lim_{n \to \infty} C[T_j^n S_{i,n} x_j - \delta_{j,i} x_j]$$
$$= C \lim_{n \to \infty} [T_j^n S_{i,n} x_j - \delta_{j,i} x_j] = 0,$$

provided that $x'_j = Cx_j \in X'_j$, $1 \leq j \leq N$. The proof of the theorem is thereby complete.

Now we would like to provide an illustrative application of Theorem 1.2.24. Let the requirements of Example 1.2.21 hold. Then we can simply prove with the help of Theorem 1.2.24 that the operators $\Delta_{X,p}^{\natural} + a_1, \Delta_{X,p}^{\natural} + a_2, \ldots, \Delta_{X,p}^{\natural} + a_N$ are d-hypercyclic, as well, by taking the injective operator

$$C := \frac{1}{2\pi i} \int_{\Gamma} e^{-(-\lambda)^{b}} \left(\lambda + \omega + \Delta_{X,p}^{\natural}\right)^{-1} d\lambda, \ \omega > 0 \text{ suff. large, } b \in (0, 1/2) \text{ fixed}$$

under advisement; here, Γ is the upwards oriented boundary of $\Sigma_{\alpha} \cup \{z \in \mathbb{C} : |z| \leq d\}$ with suitable chosen numbers $\alpha \in (0, \pi/2)$ and $d \in (0, 1]$. Then it is well known that (iv)–(vi) hold, see [292]; (i)–(iii) follows from the analyses already carried out in Example 1.2.21.

Now we want to turn our attention to the following theme:

1.3. Hypercyclic and disjoint hypercyclic MLO extensions

Let \mathcal{A} be an MLO in X. In what follows, we will often identify \mathcal{A} and its associated linear relation $\check{\mathcal{A}}$, which will be denoted by the same symbol.

It is clear that \mathcal{A} is contained in $X \times X$, which is hypercyclic, chaotic and topologically mixing (transitive). Denote

 $S(\mathcal{A}) := \{ Z : Z \text{ is a linear subspace of } X \times X \text{ and } \mathcal{A} \subseteq Z \}.$

We say that an MLO \mathcal{B} is a hypercyclic (chaotic, topologically mixing, topologically transitive) extension of \mathcal{A} iff $\mathcal{B} \in S(\mathcal{A})$ and \mathcal{B} is hypercyclic (chaotic, topologically mixing, topologically transitive).

Further on, let $N \in \mathbb{N}$, and let $\mathcal{A}_1, \ldots, \mathcal{A}_N$ be given MLOs in X. Then the MLOs $X \times X, \ldots, X \times X$, totally counted N times, are d-hypercyclic, d-chaotic and d-topologically mixing (transitive). Denote

$$S(\mathcal{A}_1, \dots, \mathcal{A}_N) := \{ (\mathcal{B}_1, \dots, \mathcal{B}_N) : \mathcal{B}_i \text{ is a linear subspace of } X \times X \\ \text{and } \mathcal{A}_i \subseteq \mathcal{B}_i \text{ for all } i \in \mathbb{N}_N \}.$$

We say that the tuple $(\mathcal{B}_1, \ldots, \mathcal{B}_N)$ of MLOs in X is a d-hypercyclic (d-chaotic, d-topologically mixing, d-topologically transitive) extension of $(\mathcal{A}_1, \ldots, \mathcal{A}_N)$ iff $(\mathcal{B}_1, \ldots, \mathcal{B}_N) \in S(\mathcal{A}_1, \ldots, \mathcal{A}_N)$ and the operators $\mathcal{B}_1, \ldots, \mathcal{B}_N$ are d-hypercyclic (dchaotic, d-topologically mixing, d-topologically transitive).

We proceed with a simple example.

- EXAMPLE 1.3.1. (i) Suppose that $X := \mathbb{C}^n$ and $\mathcal{A} := A \in L(X)$. Then \mathcal{A} is not hypercyclic, and $\mathbb{C}^n \times \mathbb{C}^n$ is the only hypercyclic (chaotic, topologically mixing, topologically transitive) extension of \mathcal{A} ; a similar statement holds for finite-dimensional spaces.
- (ii) Suppose that X is infinite-dimensional, W is a non-dense linear subspace of X and $\mathcal{A} = X \times W$. Then any hypercyclic MLO extension of \mathcal{A} has the form $X \times W'$, where W' is a dense linear subspace of X containing W.

Consider now the following MLO extension of \mathcal{A} :

$$\tilde{\mathcal{A}} := \bigcap \{ Z \in S(\mathcal{A}) : Z \text{ is hypercyclic} \}.$$

We call $\tilde{\mathcal{A}}$ the quasi-hypercyclic extension of \mathcal{A} and similarly define the quasi-chaotic (quasi-topologically transitive, quasi-topologically mixing) extension of \mathcal{A} .

EXAMPLE 1.3.2. Suppose that X is infinite-dimensional, $\{y_k : k \in \mathbb{N}\}$ is a dense subset of X, and $A \in L(X)$. Then any MLO extension \mathcal{A} of A has the form $\mathcal{A}x = Ax + W$, where $W = \mathcal{A}0$ is a linear submanifold of X. Inductively,

(29)
$$\mathcal{A}^n x = A^n x + \sum_{j=0}^{n-1} A^j(W), \quad n \in \mathbb{N}, \ x \in X.$$

Denote by \mathfrak{T} the set consisting of all linear manifolds W' of X such that there exists an element $x \in X$ with the property that, for every $n \in \mathbb{N}$, there exist elements $z_{n,j}$ in W' $(0 \leq j \leq n-1)$ such that the set $\{A^n x + \sum_{j=0}^{n-1} A^j z_{n,j} : n \in \mathbb{N}\}$ is dense in X. Then \mathfrak{T} is non-empty because $X \in \mathfrak{T}$ (with $x = 0, z_{n,j} = 0, 1 \leq j \leq n-1,$ $z_{n,0} = y_n$), and $\tilde{\mathcal{A}} = A + \bigcap \mathfrak{T}$. Now we will scrutinize some particular cases:

(i) Let X be an infinite-dimensional complex Hilbert space with the complete orthonormal basis {e_n : n ∈ N}. Define A ∑_{n=1}[∞] x_ne_n := ∑_{n=1}[∞] x_ne_{n+1}, for any x = ∑_{n=1}[∞] x_ne_n ∈ X. Then ||A|| = 1 and therefore A is not hypercyclic. It can be easily seen that any linear manifold W' belonging ℑ has to contain the linear span of {ω}, for some element ω of X such that ⟨ω, e₁⟩ ≠ 0. Let W be the linear span of {e₁}. In what follows, we will prove that A + W is a hypercyclic extension of A, with zero being the corresponding hypercyclic vector. It is clear that there exists a strictly increasing sequence (n_k)_{k∈ℕ} of positive integers such that

$$\left\| y_k - \sum_{j=0}^{n_k - 1} \alpha_{n_k, j} e_{j+1} \right\| < 2^{-k},$$

for some scalars $\alpha_{n_k,j}$ $(0 \leq j \leq n_k - 1)$. Clearly, the linear span of $\{e_1, \ldots, e_l\}$ is contained in $\sum_{j=0}^{l-1} A^j(W)$, for any $l \in \mathbb{N}$. Plugging $z_{n,j} := 0, 0 \leq j \leq n-1$, if $n \neq n_k$ for all $k \in \mathbb{N}$, and $z_{n,j} := \alpha_{n_k,j}e_1$ $(0 \leq j \leq n_k - 1)$, if $n = n_k$ for some $k \in \mathbb{N}$, we can simply deduce that zero is a hypercyclic vector of A + W, as claimed.

- (ii) Any hypercyclic MLO extension \mathcal{A} of the identity operator I on X has the form $\mathcal{A}x = x + W$, $x \in X$, where $W = \mathcal{A}0$ is a dense linear submanifold of X (when any element $x \in X$ is a hypercyclic vector of \mathcal{A}).
- (iii) Let $X := l_2(\mathbb{Z})$, let $(a_n)_{n \in \mathbb{Z}}$ be a bounded subset of $(0, \infty)$, and let

(30)
$$A\sum_{n=-\infty}^{\infty} x_n e_n := \sum_{n=-\infty}^{\infty} x_n a_n e_{n+1} \text{ for any } x = \sum_{n=-\infty}^{\infty} x_n e_n \in X;$$

here, $\{e_n : n \in \mathbb{Z}\}$ denotes the complete orthonormal basis of X. The hypercyclicity of bilateral weighted shift A has been characterized by H. Salas in [470, Theorem 2.1]: Given $\varepsilon > 0$ and $q \in \mathbb{N}$, there exists a sufficiently large integer $n \in \mathbb{N}$ such that $\prod_{s=0}^{n-1} a_{s+j} < \varepsilon$ and $\prod_{s=1}^{n} a_{j-s} >$ $1/\varepsilon$. Suppose that this condition does not hold. Then it can be simply verified with the help of (29) that span($\{e_{n_k} : k \in \mathbb{N}\}$) $\in \mathfrak{T}$ for any strictly decreasing sequence $(n_k)_{k\in\mathbb{N}}$ of negative integers. Hence, $\tilde{A} = A$.

- (iv) Suppose that W is a dense linear submanifold of X, and $\mathcal{A}x = Ax + W$, $x \in X$. Let U and V be two arbitrary open non-empty subsets of X. Then (29) implies that, for every $n \in \mathbb{N}$ and $u \in U$, there exists an element $\omega \in W$ such that $A^n u + \omega \in V \cap \mathcal{A}^n u$; in particular, \mathcal{A} is topologically mixing.
- (v) Suppose that W is a linear submanifold of X, Ax = Ax + W, $x \in X$, and for each $\omega \in W$ there exist an integer $n \in \mathbb{N}$ and elements $\omega_j \in W$ $(1 \leq i \leq n-1)$ such that $\sum_{j=1}^{n-1} A^j \omega_j = -A^n \omega$ (this, in particular, holds provided that there exists an integer $n \in \mathbb{N}$ such that $A^n(W) \subseteq W$). Then (29) implies that $\omega \in \mathcal{A}^n \omega$, $w \in W$, so that W is consisted solely of periodic points of \mathcal{A} . Suppose now that R(A) is dense in X, hence $R(A^j)$ is dense in X for all $j \in \mathbb{N}$. Using this fact and (iv), we get that $\mathcal{A}x = Ax + R(A^j), x \in X$ is a chaotic extension of A for all $j \in \mathbb{N}$; this is no longer true if R(A) is not dense in X, even if dimension of $X \setminus \overline{R(A)}$ equals 1, cf. (i). In the case that the operator A is nilpotent and W is a dense submanifold of X, then it can be easily seen that $\mathcal{A}x = Ax + W$, $x \in X$ is a chaotic extension of A.

We can similarly introduce the notion of a d-quasi-hypercyclic extension $(\mathcal{A}_1, \ldots, \mathcal{A}_N)$ (d-quasi-topologically transitive extension, d-quasi-topologically mixing extension) of any MLO tuple $(\mathcal{A}_1, \ldots, \mathcal{A}_N)$:

$$(\mathcal{A}_1,\ldots,\mathcal{A}_N) := \bigcap \{ (\mathcal{B}_1,\ldots,\mathcal{B}_N) \in S(\mathcal{A}_1,\ldots,\mathcal{A}_N) : \mathcal{B}_1,\ldots,\mathcal{B}_N \}.$$

are d-hypercyclic

The following is a continuation of Example 1.3.2:

EXAMPLE 1.3.3. Suppose that X is infinite-dimensional and $A_1, \ldots, A_N \in L(X)$. Then any MLO extension of tuple (A_1, \ldots, A_N) has the form $(A_1 + W_1, \ldots, A_N + W_N)$, where W_i is a linear submanifold of X $(1 \leq i \leq N)$. Using (29), we have that $(A_1 + W_1, \ldots, A_N + W_N)$ is d-hypercyclic iff there exists an element $x \in X$ with the property that, for every $n \in \mathbb{N}$, there exist elements $z_{n,j,l}$ in W_l $(0 \leq j \leq n-1, N)$.

 $1 \leq l \leq N$) such that the set $\{(A_1^n x + \sum_{j=0}^{n-1} A_1^j z_{n,j,1}, \ldots, A_N^n x + \sum_{j=0}^{n-1} A_N^j z_{n,j,N}) : n \in \mathbb{N}\}$ is dense in X^N ; for example, $(A_1 + X, \ldots, A_N + X)$ is d-hypercyclic with x = 0 being the corresponding d-hypercyclic vector. This immediately implies the profilation of $(\mathcal{A}_1, \ldots, \mathcal{A}_N)$. Arguing as in Example 1.3.2, it can be easily seen that the following holds:

- (i) Let A be the operator examined in Example 1.3.2(i). Then $(A + W_1, A^2 + W_2, \ldots, A^N + W^N)$ is a d-hypercyclic extension of tuple (A, A^2, \ldots, A^N) , where W_i is the linear span of $\{e_1, e_2, \ldots, e_i\}$ $(1 \le i \le N)$.
- (ii) Any d-hypercyclic extension of the tuple (I, \ldots, I) has the form $(I + W_1, \ldots, I + W_N)$, where W_i is a dense linear submanifold of X $(1 \le i \le N)$.
- (iii) Let $X := l_2(\mathbb{Z})$, let $(a_n)_{n \in \mathbb{Z}}$ be a bounded subset of $(0, \infty)$, and let A be defined through (30). Suppose that $(n_{k,i})_{k \in \mathbb{N}}$ is any strictly decreasing sequence of negative integers possessing the property that, for every $s \in \mathbb{N}$ and $j \in \mathbb{N}_{i-1}^0$, there exists $l \in \mathbb{Z}$ such that l < -s and $n_{l,i} \equiv j \pmod{i}$, $1 \leq i \leq N$. Set $W_i := \operatorname{span}(\{e_{n_{k,i}} : k \in \mathbb{N}\}), 1 \leq i \leq N$. Then $(A+W_1, A^2+W_2, \ldots, A^N+W_N)$ is a d-hypercyclic extension of the tuple (A, A^2, \ldots, A^N) .
- (iv) Let $(\mathcal{A}_1, \ldots, \mathcal{A}_N) = (A_1 + W_1, \ldots, A_N + W_N)$, where W_i is a dense linear submanifold of X $(1 \leq i \leq N)$. Then $(\mathcal{A}_1, \ldots, \mathcal{A}_N)$ is a d-topologically mixing extension of (A_1, \ldots, A_N) .
- (v) Suppose that the range of A_i is dense in X $(1 \le i \le N)$. Then $(A_1 + R(A_1^{j_1}), \ldots, A_N + R(A_N^{j_N}))$ is a d-chaotic extension of (A_1, \ldots, A_N) for all $j_1, \ldots, j_N \in \mathbb{N}$. In the case that the operator A_i is nilpotent and W_i is a dense submanifold of X $(1 \le i \le N)$, then it can be easily seen that $(A_1 + W_1, \ldots, A_N + W_N)$ is a d-chaotic extension of (A_1, \ldots, A_N) .

Further analysis of hypercyclic and disjoint hypercyclic extensions of multivalued linear operators is without scope of this book. Fairly precise specification of chaotic and disjoint chaotic extensions of multivalued linear operators seems to be a more delicate problem.

We close this section with the observation that disjoint hypercyclic and disjoint topologically mixing properties of degenerate fractional differential equations have recently been considered in a joint paper with V. Fedorov [215].

1.4. Laplace transform of functions with values in SCLCSs

Without any doubt, two most important monographs on the Laplace transform of scalar valued functions are written by G. Doetsch [158] (1937) and D. V. Widder [536] (1941). The following conditions on a scalar valued function k(t) will be used in the sequel:

- (P1): k(t) is Laplace transformable, i.e., it is locally integrable on $[0, \infty)$ and there exists $\beta \in \mathbb{R}$ such that $\tilde{k}(\lambda) := \mathcal{L}(k)(\lambda) := \lim_{b \to \infty} \int_0^b e^{-\lambda t} k(t) dt := \int_0^\infty e^{-\lambda t} k(t) dt$ exists for all $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \beta$. Put $\operatorname{abs}(k) := \inf\{\operatorname{Re} \lambda : \tilde{k}(\lambda) \text{ exists}\}$, and denote by \mathcal{L}^{-1} the inverse Laplace transform.
- (P2): k(t) satisfies (P1) and $\tilde{k}(\lambda) \neq 0$, Re $\lambda > \beta$ for some $\beta \ge abs(k)$.

For the basic theory of Laplace transform of Banach space valued functions, we refer the reader to the monograph [27] by W. Arendt, C. J. K. Batty, M. Hieber and F. Neubrander. As mentioned in [292, Section 1.2], the Laplace transform of functions with values in sequentially complete locally convex spaces has attracted much less attention so far. The main purpose of this section is to introduce a relatively simple and new theoretical concept useful in the analysis of operational properties of Laplace transform of non-continuous functions with values in sequentially complete locally convex spaces. This concept extends the corresponding one introduced by T.-J. Xiao and J. Liang [546], 1997, and coincides with the classical concept of vector-valued Laplace transform in the case that the state space X is one of Banach's [27].

We are concerned with the existence of Laplace integral

$$(\mathcal{L}f)(\lambda):=\tilde{f}(\lambda):=\int_0^\infty e^{-\lambda t}f(t)dt:=\lim_{\tau\to\infty}\int_0^\tau e^{-\lambda t}f(t)dt,$$

for $\lambda \in \mathbb{C}$. If $\tilde{f}(\lambda_0)$ exists for some $\lambda_0 \in \mathbb{C}$, then we define the abscissa of convergence of $\tilde{f}(\cdot)$ by

$$\operatorname{abs}_X(f) := \inf\{\operatorname{Re}\lambda : \tilde{f}(\lambda) \text{ exists}\};\$$

otherwise, $\operatorname{abs}_X(f) := +\infty$. It is said that $f(\cdot)$ is Laplace transformable, or equivalently, that $f(\cdot)$ belongs to the class (P1)-X, iff $\operatorname{abs}_X(f) < \infty$. Assuming that there exists a number $\omega \in \mathbb{R}$ such that for each seminorm $p \in \circledast$ there exists a number $M_p > 0$ satisfying that $p(f(t)) \leq M_p e^{\omega t}$, $t \geq 0$, we define $\omega_X(f) \in [-\infty, \infty)$ as the infimum of all numbers $\omega \in \mathbb{R}$ with the above property; if there is no such a number $\omega \in \mathbb{R}$, then we define $\omega_X(f) := +\infty$. Further on, we abbreviate $\omega_X(f)$ $(\operatorname{abs}_X(f))$ to $\omega(f)$ $(\operatorname{abs}(f))$, if no confusion seems likely. Define

$${}_{w}\operatorname{abs}(f) := \inf \bigg\{ \lambda \in \mathbb{R} : \sup_{t > 0} \bigg| \int_{0}^{t} e^{-\lambda s} \langle x^{*}, f(s) \rangle ds \bigg| < \infty \text{ for all } x^{*} \in X^{*} \bigg\},$$

 $F_{\infty} := \lim_{\tau \to \infty} F(\tau)$, if the limit exists in X, and $F_{\infty} := 0$, otherwise.

Keeping in mind Theorem 1.1.4, we can repeat verbatim the argumentation from [27, Section 1.4, pp. 27-30] in order to see that the following theorem holds good (the only essential difference occurs on l. 6, p. 29, where we can use [419, Mackey's theorem 23.15] in place of the uniform boundedness principle):

THEOREM 1.4.1. Let $f \in L^1_{loc}([0,\infty):X)$. Then the following holds:

- (i) The Laplace integral $\tilde{f}(\lambda)$ converges if $\operatorname{Re} \lambda > \operatorname{abs}(f)$ and diverges if $\operatorname{Re} \lambda < \operatorname{abs}(f)$. If $\operatorname{Re} \lambda = \operatorname{abs}(f)$, then the Laplace integral may or may not be convergent.
- (ii) $_{w} \operatorname{abs}(f) = \operatorname{abs}(f)$.
- (iii) Suppose that $\lambda \in \mathbb{C}$ and the limit $\lim_{t \to +\infty} \int_0^t e^{-\lambda s} p(f(s)) ds$ exists for any $p \in \circledast$. Then $\tilde{f}(\lambda)$ exists, as well.
- (iv) We have

$$\operatorname{abs}(f) \leq \operatorname{abs}(p(f)) \leq \omega(f), \quad p \in \mathfrak{B}.$$

In general, any of these two inequalities can be strict.

(31)
$$\operatorname{abs}(f) = \omega(F - F_{\infty})$$

(32)
$$\tilde{f}(\lambda) = F_{\infty} + \lambda \int_0^\infty e^{-\lambda t} (F(t) - F_{\infty}) dt, \quad \operatorname{Re} \lambda > \omega (F - F_{\infty}),$$

(33)
$$\tilde{f}(\lambda) = \lambda \tilde{F}(\lambda), \quad \operatorname{Re} \lambda > \max(\operatorname{abs}(f), 0)$$

and

 $\operatorname{abs}(f) \leq \omega \Leftrightarrow \omega(F) \leq \omega \quad (if \ \omega \geq 0).$

In particular, $f(\cdot)$ is Laplace transformable iff $\omega(F) < \infty$.

Recall [541], a function $h(\cdot)$ belongs to the class LT - X iff there exist a function $g \in C([0,\infty) : X)$ and a number $\omega \in \mathbb{R}$ such that $\omega(g) \leq \omega < \infty$ and $h(\lambda) = (\mathcal{L}g)(\lambda)$ for $\lambda > \omega$; as observed in [292, Section 1.2], the assumption $h \in LT - X$ immediately implies that the function $\lambda \mapsto h(\lambda), \lambda > \omega$ can be analytically extended to the right half plane $\{\lambda \in \mathbb{C} : \operatorname{Re} \lambda > \omega\}$. In the sequel, the set of all originals $g(\cdot)$ whose Laplace transform belongs to the class LT - X will be abbreviated to $LT_{or} - X$. Keeping the above observation and the equations (31)– (32)) in mind, we can simply prove that the mapping $\lambda \mapsto \tilde{f}(\lambda)$, $\operatorname{Re} \lambda > \operatorname{abs}(f)$ is analytic, provided that $f \in (\operatorname{P1}) - X$. If this is the case, the following formula holds:

(34)
$$\frac{d^n}{d\lambda^n}\tilde{f}(\lambda) = (-1)^n \int_0^\infty e^{-\lambda t} t^n f(t) dt, \quad n \in \mathbb{N}, \ \lambda \in \mathbb{C}, \ \operatorname{Re} \lambda > \operatorname{abs}(f).$$

In the following theorem, we will collect various operational properties of vector-valued Laplace transform.

THEOREM 1.4.2. Let $f \in (P1)$ -X, $z \in \mathbb{C}$ and $s \ge 0$.

- (i) Put $g(t) := e^{-zt} f(t), t \ge 0$. Then $g(\cdot)$ is Laplace transformable, $\operatorname{abs}(g) = \operatorname{abs}(f) \operatorname{Re} z$ and $\tilde{g}(\lambda) = \tilde{f}(\lambda + z), \lambda \in \mathbb{C}$, $\operatorname{Re} \lambda > \operatorname{abs}(f) \operatorname{Re} z$.
- (ii) Put $f_s(t) := f(t+s), t \ge 0, h_s(t) := f(t-s), t \ge s \text{ and } h_s(t) := 0, s \in [0,t].$ Then $\operatorname{abs}(f_s) = \operatorname{abs}(h_s) = \operatorname{abs}(f), \tilde{f}_s(\lambda) = e^{\lambda s}(\tilde{f}(\lambda) \int_0^s e^{-\lambda t} f(t) dt)$ and $\tilde{h}_s(\lambda) = e^{-\lambda s} \tilde{f}(\lambda) \ (\lambda \in \mathbb{C}, \operatorname{Re} \lambda > a).$
- (iii) Let $T \in L(X, Y)$. Then $T \circ f \in (P1)$ -Y and $T\tilde{f}(\lambda) = (T \circ f)(\lambda)$ for $\lambda \in \mathbb{C}$, $\operatorname{Re} \lambda > \operatorname{abs}(f)$.
- (iv) Suppose that $\mathcal{A}: X \to P(Y)$ is an MLO and \mathcal{A} is $X_{\mathcal{A}} \times Y_{\mathcal{A}}$ -closed, as well as $f \in (P1) X_{\mathcal{A}}$, $l \in (P1) Y_{\mathcal{A}}$ and $(f(t), l(t)) \in \mathcal{A}$ for a.e. $t \ge 0$. Then $(\tilde{f}(\lambda), \tilde{l}(\lambda)) \in \mathcal{A}$, $\lambda \in \mathbb{C}$ for $\operatorname{Re} \lambda > \max(\operatorname{abs}(f), \operatorname{abs}(l))$.
- (v) Suppose, in addition, $\omega(f) < \infty$. Put

$$j(t) := \int_0^\infty \frac{e^{-s^2/4t}}{\sqrt{\pi t}} f(s) ds := \lim_{\tau \to \infty} \int_0^\tau \frac{e^{-s^2/4t}}{\sqrt{\pi t}} f(s) ds, \quad t > 0$$

and

$$k(t) := \int_0^\infty \frac{s e^{-s^2/4t}}{2\sqrt{\pi}t^{\frac{3}{2}}} f(s) ds := \lim_{\tau \to \infty} \int_0^\tau \frac{s e^{-s^2/4t}}{2\sqrt{\pi}t^{\frac{3}{2}}} f(s) ds, \quad t > 0.$$

Then $j(\cdot)$ and $k(\cdot)$ are Laplace transformable,

$$\max(\operatorname{abs}(j), \operatorname{abs}(k)) \leqslant (\max(\omega(f), 0))^2, \ \tilde{j}(\lambda) = \frac{\tilde{f}(\sqrt{\lambda})}{\sqrt{\lambda}} \ and \ \tilde{k}(\lambda) = \tilde{f}(\sqrt{\lambda})$$

for all $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > (\max(\omega(f), 0))^2$.

(vi) Let $f \in (P1)-X$, $a \in L^1_{loc}([0,\infty))$ and $abs(|a|) < \infty$. Suppose, in addition, that $f \in C([0,\infty) : X)$. Then the mapping $t \mapsto (a * f)(t) = \int_0^t a(t - s)f(s)ds$, $t \ge 0$ is continuous, $a * f \in (P1)-X$, and

$$\hat{a} * f(\lambda) = \tilde{a}(\lambda)\tilde{f}(\lambda), \ \lambda \in \mathbb{C}, \ \operatorname{Re} \lambda > \max(\operatorname{abs}(|a|), \operatorname{abs}(f)).$$

PROOF. Factorizing in Theorem 1.1.4, Theorem 1.1.3(ii) and Theorem 1.2.3, the assertions (i)–(iv) can be proved as in the Banach space case (cf. [27, Proposition 1.6.1–Proposition 1.6.3] for more details). Consider now part (v). Let $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > (\max(\omega(f), 0))^2$ be fixed. Then $\operatorname{Re}(\sqrt{\lambda}) > \max(\omega(f), 0) \ge \max(\omega(F), 0)$ so that [292, Theorem 1.2.1(v)] implies in combination with (33) that $\tilde{f}(\sqrt{\lambda})$ exists, as well as that

$$\tilde{f}(\sqrt{\lambda}) = \frac{\tilde{F}(\sqrt{\lambda})}{\sqrt{\lambda}} = \int_0^\infty e^{-\lambda t} \int_0^\infty \frac{e^{-s^2/4t}}{\sqrt{\pi t}} f(s) ds \, dt.$$

On the other hand, we can simply prove with the help of dominated convergence theorem that the mapping $t \mapsto k(t)$, t > 0 is continuous as well as that for each seminorm $p \in \circledast$ there exists a finite number $m_p > 0$ such that $p(k(t)) \leq m_p t^{(-1)/2}$, $t \in (0, 1]$. This simply implies $k \in L^1_{loc}([0, \infty) : X)$. Since

(35)
$$\int_0^\infty e^{-\lambda t} \langle x^*, k(t) \rangle dt = \int_0^\infty e^{-\sqrt{\lambda}t} \langle x^*, f(t) \rangle dt, \quad x^* \in X^*,$$

we obtain that

$$\lim_{\tau \to \infty} \left\langle x^*, \int_0^\tau e^{-\lambda t} k(t) dt \right\rangle = \left\langle x^*, \tilde{f}(\sqrt{\lambda}) \right\rangle, \quad x^* \in X^*.$$

By Theorem 1.1.4(i), we get that the mapping $\tau \mapsto \int_0^{\tau} e^{-\lambda t} k(t) dt$, $\tau \ge 0$ is continuous so that the previous equality implies $\sup_{\tau \ge 0} |\langle x^*, \int_0^{\tau} e^{-\lambda t} k(t) dt \rangle| < \infty$ for all $x^* \in X^*$. Therefore, Theorem 1.4.1(ii) shows that $\lambda >_w \operatorname{abs}(k) = \operatorname{abs}(k)$ and $\tilde{k}(\lambda)$ exists. Using again (35), it readily follows that $\tilde{k}(\lambda) = f(\sqrt{\lambda})$, as claimed. Similarly we can prove that $\tilde{j}(\lambda) = f(\sqrt{\lambda})/\sqrt{\lambda}$. Suppose, finally, that the requirements of (vi) hold. Then it is very simple to prove that the mapping $t \mapsto (h * f)(t)$, $t \ge 0$ is continuous as well as that $\omega(1 * h * f) = \omega(h * (1 * f)) < \infty$. An application of Theorem 1.4.1(v) yields that $h * f \in (\text{P1})$ -X. Fix now a number $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \max(\operatorname{abs}(|h|), \operatorname{abs}(f))$. Since $\operatorname{abs}(\langle x^*, f(\cdot) \rangle) \leqslant \operatorname{abs}(f)$ for all $x^* \in X^*$, [27, Proposition 1.6.4] implies that $(\mathcal{L}(h * \langle x^*, f(\cdot) \rangle))(\lambda)$ exists. Using this fact, it readily follows that

$$\sup_{t>0} \left| \int_0^t e^{-s\operatorname{Re}\lambda} (h * \langle x^*, f(\cdot) \rangle)(s) ds \right| < \infty, \quad x^* \in X^*.$$

By Theorem 1.4.1(ii), we get that $\widetilde{h*f}(\lambda)$ exists. The equality $\widetilde{h*f}(\lambda) = \widetilde{h}(\lambda)\widetilde{f}(\lambda)$ can be proved in a routine manner.

The non-possibility of establishing Fubini–Tonelli theorem in this concept of integration does not able us to fully transfer some assertions from the Banach space case to the general locally convex space case; for example, in Theorem 1.4.2(vi) we consider the Laplace transform of finite convolution product and there it is almost inevitable to impose the condition that the function f(t) is continuous (this is not the case in Fréchet space, when we can use the well-known extension of Bochner concept of integration; see [410, p. 100]). In a great number of our structrural results stated in the monograph [292], we have incorrectly applied Theorem 1.4.2(vi) by assuming only that the function a(t) satisfies (P1). Strictly speaking, in all these assertions, we have to assume that $abs(|a|) < \infty$. For the sequel, we need the notion of a Lebesgue point of a function $f \in L^1_{loc}([0, \infty) : X)$. A point $t \ge 0$ is said to be a Lebesgue point of $f(\cdot)$ iff for each seminorm $p \in \circledast$, we have

(36)
$$\lim_{h \to 0} \frac{1}{h} \int_{t}^{t+h} p(f(s) - f(t)) ds = 0.$$

It is clear that any point of continuity of function $f(\cdot)$ is one of Lebesgue's points of $f(\cdot)$, as well as that the mapping $t \mapsto F(t), t \ge 0$ is differentiable at any Lebesgue's point of $f(\cdot)$. Furthermore, a slight modification of the proof of [27, Proposition 1.2.2; a)/b)] shows that the following holds:

(Q1) For each seminorm $p \in \circledast$ there exists a set $N_p \subseteq [0, \infty)$ of Lebesgue's measure zero such that

$$\lim_{h \to 0} p\left(\frac{1}{h} \int_{t}^{t+h} f(s)ds - f(t)\right) = 0, \quad t \in [0,\infty) \smallsetminus N_p$$

and (36) holds for $t \in [0, \infty) \setminus N_p$.

In the case that X is a Fréchet space, (Q1) immediately implies that almost every point t > 0 is a Lebesgue point of $f(\cdot)$. Using the proof of [27, Theorem 1.7.7], Theorem 1.1.4(iii), as well as the equations (19) and (34), we can simply prove that the Post–Widder inversion formula holds in our framework:

THEOREM 1.4.3. (Post–Widder) Suppose $f \in (P1) - X$ and t > 0 is a Lebesgue point of $f(\cdot)$. Then

$$f(t) = \lim_{n \to \infty} (-1)^n \frac{1}{n!} \left(\frac{n}{t}\right)^{n+1} \tilde{f}^{(n)}\left(\frac{n}{t}\right).$$

The situation is much more complicated if we analyze the Phragmén–Doetsch inversion formula for the Laplace transform of functions with values in SCLCSs. The following result of this type will be sufficiently general for our purposes:

THEOREM 1.4.4. Let $f \in (P1)$ -X and $t \ge 0$. Then the following holds:

$$f^{[2]}(t) = \lim_{\lambda \to \infty} \sum_{n=1}^{\infty} (-1)^{n-1} n!^{-1} e^{n\lambda t} \frac{\tilde{f}(n\lambda)}{n\lambda}.$$

PROOF. In view of Theorem 1.4.1(v), we have $F \in C([0,\infty) : X)$ and $\omega(F) < \infty$. The result now follows easily from [292, Theorem 1.2.1(ix)].

Now we will state and prove the following uniqueness type theorem for the Laplace transform.

THEOREM 1.4.5. (The uniqueness theorem for the Laplace transform) Suppose $f \in (P1) - X$, $\lambda_0 > \operatorname{abs}(f)$ and $\tilde{f}(\lambda) = 0$ for all $\lambda > \lambda_0$. Then F(t) = 0, $t \ge 0$, f(t) = 0 if t > 0 is a Lebesgue point of $f(\cdot)$, and for each seminorm $p \in \mathfrak{B}$ there exists a set $N_p \subseteq [0,\infty)$ of Lebesgue's measure zero such that p(f(t)) = 0, $t \in [0,\infty) \setminus N_p$. In particular, if X is a Fréchet space, then f(t) = 0 for a.e. $t \ge 0$.

PROOF. The function $t \mapsto F(t)$, $t \ge 0$ is continuous and by Theorem 1.4.1(v) we get that $\omega(F) < \infty$ and $\tilde{F}(\lambda) = 0$, $\lambda > \max(\lambda_0, 0)$. Now we can apply Theorem 1.4.3 in order to see that F(t) = 0, $t \ge 0$. The remainder of the proof is simple and therefore omitted.

REMARK 1.4.6. Suppose that $f \in L^1_{loc}([0,\infty):X)$ and for each seminorm $p \in \circledast$ there exists a set $N_p \subseteq [0,\infty)$ of Lebesgue's measure zero such that p(f(t)) = 0, $t \in [0,\infty) \smallsetminus N_p$. Then $\operatorname{abs}(f) = \operatorname{abs}(p(f)) = -\infty$ $(p \in \circledast)$ and $\tilde{f}(\lambda) = 0$ for all $\lambda \in \mathbb{C}$.

The following converse of Theorem 1.4.2(iv) simply follows from an application of Theorem 1.4.4.

PROPOSITION 1.4.7. Suppose that $\mathcal{A}: X \to P(Y)$ is an MLO and \mathcal{A} is $X_{\mathcal{A}} \times Y_{\mathcal{A}}$ closed, as well as $f \in (P1) - X_{\mathcal{A}}$, $l \in (P1) - Y_{\mathcal{A}}$ and $(\tilde{f}(\lambda), \tilde{l}(\lambda)) \in \mathcal{A}$, $\lambda \in \mathbb{C}$ for $\operatorname{Re} \lambda > \max(\operatorname{abs}(f), \operatorname{abs}(l))$. Then $l^{[1]}(t) \in \mathcal{A}f^{[1]}(t)$, $t \ge 0$ and $l(t) \in \mathcal{A}f(t)$ for any t > 0 which is a Lebesgue point of both functions f(t) and l(t).

Now we would like to briefly explain how we can extend the definition of Laplace transformable functions to the multivalued ones. Let $0 < \tau \leq \infty$ and $\mathcal{F}: [0, \tau) \rightarrow P(X)$. A single-valued function $f: [0, \tau) \rightarrow X$ is called a section of \mathcal{F} iff $f(t) \in \mathcal{F}(t)$ for all $t \in [0, \tau)$. We denote the set of all sections, resp., all continuous sections, of \mathcal{F} by $\sec(\mathcal{F})$, resp., $\sec_c(\mathcal{F})$. Suppose now that $\tau = \infty$ and any function $f \in$ $\sec(\mathcal{F})$ belongs to the class (P1)-X. Then we define $abs_X(\mathcal{F}) := sup\{abs_X(f) : f \in$ $\sec(v)\}; \mathcal{F}(\cdot)$ is said to be Laplace transformable iff $abs_X(\mathcal{F}) < \infty$.

The method proposed by T.-J. Xiao and J. Liang in [546] provides a sufficiently enough framework for the theoretical study of real and complex inversion methods for the Laplace transform of functions with values in SCLSCs, as well as for the studies of analytical properties and approximation of Laplace transform (see e.g. [541, Section 1.1.1] and [292, Section 1.2] for more details); this method can be successfully applied in the analysis of subordination principles for abstract timefractional inclusions, as well (cf. Theorem 3.1.8 below). It is also worth noting that there exists a great number of theoretical results from the monograph [27], not mentioned so far, which can be reconsidered for the Laplace transformable functions with values in SCLSCs; for example, all structural results from [27, Section 4.1] continue to hold in our framework. Due primarily to the space limitations, we will not consider here numerous important questions concerning the vector-valued Laplace transform of functions with values in SCLCSs.

The complex inversion theorem for the vector-valued Laplace transform reads as follows.

THEOREM 1.4.8. Assume $a > 0, r \in \mathbb{R}, q : \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda > a\} \to E$ is analytic, and for each $p \in \circledast$ there exists $M_p > 0$ such that

$$p(q(\lambda)) \leq M_p |\lambda|^r$$
, $\operatorname{Re} \lambda > a$.

Then for each $\alpha > 1$ there exists a function $f_{\alpha} \in C([0,\infty):E)$ with $f_{\alpha}(0) = 0$ and

$$p(h_{\alpha}(t)) \leq M_{\alpha}M_{p}e^{at}, \quad p \in \circledast, \ t \geq 0,$$
$$q(\lambda) = \lambda^{\alpha+r} \int_{0}^{\infty} e^{-\lambda t}h_{\alpha}(t)dt, \quad \operatorname{Re} \lambda > a$$

where M_{α} is independent of p and $q(\cdot)$.

The following extension of Arendt–Widder theorem has been proved by T.-J. Xiao and J. Liang (see e.g. [541]).

THEOREM 1.4.9. Let $a \ge 0$, $r \in (0,1]$, $\omega \in (-\infty,a]$, $M_p > 0$ for each $p \in \circledast$, and let $q: (a, \infty) \to E$ be an infinitely differentiable function. Then we have the equivalence of statements (i) and (ii), where:

- (i) One has $p((\lambda \omega)^{k+1} \frac{q^{(k)}(\lambda)}{k!}) \leq M, p \in \circledast, \lambda > a, k \in \mathbb{N}_0.$ (ii) There exists a function $F_r \in C([0,\infty) : E)$ satisfying $F_r(0) = 0$,

$$q(\lambda) = \lambda^r \int_0^\infty e^{-\lambda t} F_r(t) dt, \quad \lambda > a,$$

$$p\left(\int_0^{t+h} \frac{(t+h-s)^{-r}}{\Gamma(1-r)} F_r(s) ds - \int_0^t \frac{(t-s)^{-r}}{\Gamma(1-r)} F_r(s) ds\right) \leqslant M_p h e^{\omega t} \max(e^{\omega h}, 1),$$

$$p(t) = 0, \quad h \ge 0, \text{ and } m \in \mathbb{R}, \text{ if } n \in (0, 1), \text{ and}$$

for any $t \ge 0$, $h \ge 0$ and $p \in \circledast$, if $r \in (0, 1)$, and

$$p(F_r(t+h) - F_r(t)) \leqslant M_p h e^{\omega t} \max(e^{\omega h}, 1), \quad t \ge 0, \ h \ge 0, \ p \in \circledast,$$

if r = 1. Moreover, in this case,

$$p(F_r(t+h) - F_r(t)) \leqslant \frac{2M_p}{r\Gamma(r)} h^r \max(e^{\omega(t+h)}, 1), \quad t \ge 0, \ h \ge 0, \ p \in \circledast.$$

Recall that $\Sigma_{\alpha} = \{z \in \mathbb{C} \setminus \{0\} : |\arg(z)| < \alpha\} \ (\alpha \in (0, \pi])$. The most important analytical properties of Laplace transform of functions with values in SCLCSs are collected in the following theorem.

THEOREM 1.4.10. [296]

- (i) Let $\alpha \in (0, \frac{\pi}{2}]$, $\omega \in \mathbb{R}$ and $q: (\omega, \infty) \to E$. Then the following assertions are equivalent:
 - (a) There exists an analytic function $f: \Sigma_{\alpha} \to E$ such that $q(\lambda) = \tilde{f}(\lambda)$, $\lambda \in (\omega, \infty)$ and the set $\{e^{-\omega z}f(z) : z \in \Sigma_{\beta}\}$ is bounded for all $\beta \in (0, \alpha).$

(b) The function $q(\cdot)$ admits an analytic extension $\tilde{q}: \omega + \sum_{\frac{\pi}{2} + \alpha} \to E$ which satisfies that the set $\{(\lambda - \omega)\tilde{q}(\lambda): \lambda \in \omega + \sum_{\frac{\pi}{2} + \alpha}\}$ is bounded for all $\gamma \in (0, \alpha)$.

If this is the case, then we have that, for every $k \in \mathbb{N}$ and $\beta \in (0, \alpha)$, the set $\{z^k e^{-\omega z} f^{(k)}(z) : z \in \Sigma_\beta\}$ is bounded.

- (ii) Let α ∈ (0, π] and let f: Σ_α → E be an analytic function which satisfies that, for every β ∈ (0, α), the set {f(z) : z ∈ Σ_β} is bounded. Let x ∈ E. Then the following holds:
 - (a) The assumption $\lim_{t \to +\infty} f(t) = x$ implies $\lim_{z \to \infty, z \in \Sigma_{\beta}} f(z) = x$ for all $\beta \in (0, \alpha)$.
 - (b) The assumption $\lim_{t\to 0+} f(t) = x$ implies $\lim_{z\to 0, z\in\Sigma_{\beta}} f(z) = x$ for all $\beta \in (0, \alpha)$.
- (iii) Assume $x \in E$, $\alpha \in (0, \frac{\pi}{2}]$, $\omega \in \mathbb{R}$, $q: (\omega, \infty) \to E$ and let (i)(a) of this theorem hold. Then:
 - (a) $\lim_{t\to 0+} f(t) = x$ iff $\lim_{\lambda\to\infty} \lambda q(\lambda) = x$.
 - (b) Let $\omega = 0$. Then $\lim_{t \to +\infty} f(t) = x$ iff $\lim_{\lambda \to 0} \lambda q(\lambda) = x$.

We also need the following theorem on approximation of Laplace transform.

THEOREM 1.4.11. [296] Let $f_n \in C([0,\infty) : E)$, $n \in \mathbb{N}$, let the set $\{e^{-\omega t} f_n(t) : n \in \mathbb{N}, t \ge 0\}$ be bounded for some $\omega \in \mathbb{R}$, and let $\lambda_0 \ge \omega$. Then the following assertions are equivalent:

- (i) The sequence (f
 _n) converges pointwise on (λ₀,∞) and the sequence (f_n) is equicontinuous at each point t ≥ 0.
- (ii) The sequence (f_n) converges uniformly on compact subsets of $[0,\infty)$.

Assuming (ii) holds and $\lim_{n\to\infty} f_n(t) = f(t), t \ge 0$, one has $\lim_{n\to\infty} \tilde{f}_n(\lambda) = \tilde{f}(\lambda), \lambda > \lambda_0$.

Finally, it would be worthwhile to record the following slight extension of [459, Proposition 0.1, Theorem 0.4, p. 10–12].

Theorem 1.4.12. [296]

- (i) Assume $g: \mathbb{C}_+ \to E$ is analytic and satisfies that the sets $\{\lambda g(\lambda) : \lambda \in \mathbb{C}_+\}$ and $\{\lambda^2 g'(\lambda) : \lambda \in \mathbb{C}_+\}$ are bounded. Then the set $\{n!^{-1}\lambda^{n+1}g^{(n)}(\lambda) : \lambda \in \mathbb{C}_+, n \in \mathbb{N}_0\}$ is bounded as well.
- (ii) Assume $k \in \mathbb{N}_0$, $g: \mathbb{C}_+ \to E$ is analytic and satisfies that the set $\{\lambda^{n+1}g^{(n)}(\lambda): \lambda \in \mathbb{C}_+, 0 \leq n \leq k+1\}$ is bounded. Then there exists a function $u \in C^k((0,\infty): E)$ such that $g(\lambda) = \tilde{u}(\lambda), \lambda \in \mathbb{C}_+$ and the sets $\{t^n u^{(n)}(t): t > 0, 0 \leq n \leq k\}$ and $\{(t-s)^{-1}(1+\ln\frac{t}{t-s})^{-1}(t^{k+1}u^{(k)}(t) s^{k+1}u^{(k)}(s)): 0 \leq s < t < \infty\}$ are bounded.

1.5. Operators of fractional differentiation, Mittag-Leffler and Wright functions

The first congress on fractional calculus was held at the University of New Haven, in 1974 [131]. From then on, several applications of fractional calculus and fractional differential equations have emerged in engineering, physics, chemistry,

63

biology and other sciences. For the basic information about fractional calculus and non-degenerate fractional differential equations, the reader may consult the monographs by M. H. Annaby, Z. S. Mansour [23], T. M. Atanacković, S. Pilipović, B. Stanković, D. Zorica [36,37], D. Baleanu, K. Diethelm, E. Scalas, J. Trujillo [45], K. Diethelm [153], R. Hilfer [255], A. A. Kilbas, H. M. Srivastava, J. J. Trujillo [279], V. Kiryakova [281], F. Mainardi [405], K. B. Oldham, J. Spanier [439], I. Podlubny [452], S. G. Samko, A. A. Kilbas, O. I. Marichev [472], V. E. Tarasov [516] and the author [291,292]; we also hope that the doctoral dissertations of E. Bazhlekova [61] and Yu. V. Bogacheva [82], as well as the references [8,33,34,102, 105,157,165,166,169,227,231,270,271,274,277,345,382,477,529,550], and those from author's ones [295,355], may be of some help to the reader. For the basic information on the history of fractional calculus, the reader may consult [160], [292, Section 1.3], [439] and [472].

Suppose that $\alpha > 0$, $m = \lceil \alpha \rceil$ and I = (0, T) for some $T \in (0, \infty]$. Then the Riemann-Liouville fractional integral J_t^{α} of order α is defined by

$$J_t^{\alpha} f(t) := (g_{\alpha} * f)(t), \quad f \in L^1(I:E), \ t \in I,$$

while the Caputo fractional derivative $\mathbf{D}_t^{\alpha}u(t)$ is defined for those functions $u \in C^{m-1}([0,\infty):E)$ for which $g_{m-\alpha} * (u - \sum_{k=0}^{m-1} u_k g_{k+1}) \in C^m([0,\infty):E)$, by

$$\mathbf{D}_t^{\alpha} u(t) = \frac{d^m}{dt^m} \left[g_{m-\alpha} * \left(u - \sum_{k=0}^{m-1} u_k g_{k+1} \right) \right].$$

The existence of Caputo fractional derivative $\mathbf{D}_t^{\zeta} u$ for $t \ge 0$ implies $u \in C^{\lceil \zeta \rceil}((0, \infty) : E) \cap C^{\zeta}([0, T] : X)$, for each finite number T > 0. Suppose now, just for a few moments, that E is a Banach space. Then the Sobolev space $W^{m,1}(I : E)$ can be introduced in the following way (see e.g. [61, p. 7]):

$$W^{m,1}(I:E) := \left\{ f \mid \exists \varphi \in L^1(I:E) \exists c_k \in \mathbb{C} \ (0 \leqslant k \leqslant m-1) \right.$$
$$f(t) = \sum_{k=0}^{m-1} c_k g_{k+1}(t) + (g_m * \varphi)(t) \text{ for a.e. } t \in (0,\tau) \right\}.$$

If so, then we have $\varphi(t) = f^{(m)}(t)$ in the distributional sense, and $c_k = f^{(k)}(0)$ $(0 \leq k \leq m-1)$. The Riemann–Liouville fractional derivative D_t^{α} of order α is defined for those functions $f \in L^1(I:E)$ for which $g_{m-\alpha} * f \in W^{m,1}(I:E)$, by

$$D_t^{\alpha}f(t) := \frac{d^m}{dt^m} J_t^{m-\alpha} f(t), \quad t \in I.$$

Due to [61, Theorem 1.5], the Riemann–Liouville fractional integrals and derivatives satisfy the following equalities:

$$J^{\alpha}_t J^{\beta}_t f(t) = J^{\alpha+\beta}_t f(t), \quad D^{\alpha}_t J^{\alpha}_t f(t) = f(t),$$

for $f \in L^1(I:E)$ and

(37)
$$J_t^{\alpha} D_t^{\alpha} f(t) = f(t) - \sum_{k=0}^{m-1} (g_{m-\alpha} * f)^{(k)}(0) g_{\alpha+k+1-m}(t)$$

for any $f \in L^1(I : E)$ with $g_{m-\alpha} * f \in W^{m,1}(I : E)$. We are returning to the case in which E is a general SCLCS. Let $\beta > 0$ and $\beta \notin \mathbb{N}$. Then the Liouville right-sided fractional derivative of order β (see [279, (2.3.4)] for the scalar-valued case) is defined for those continuous functions $u: (0, \infty) \to E$ for which $\lim_{T\to\infty} \int_s^T g_{\lceil\beta\rceil-\beta}(t-s)u(t)dt = \int_s^\infty g_{\lceil\beta\rceil-\beta}(t-s)u(t)dt$ exists and defines a $\lceil\beta\rceil$ -times continuously differentiable function on $(0,\infty)$, by

$$\mathbf{D}_{-}^{\beta}u(s) := (-1)^{\lceil\beta\rceil} \frac{d^{\lceil\beta\rceil}}{ds^{\lceil\beta\rceil}} \int_{s}^{\infty} g_{\lceil\beta\rceil-\beta}(t-s)u(t)dt, \quad s > 0$$

We define the modified Liouville right-sided fractional derivative of order β , $D^{\beta}_{-}u(s)$ shortly, for those continuously differentiable functions $u: (0, \infty) \to E$ for which $\lim_{T\to\infty} \int_s^T g_{\lceil\beta\rceil-\beta}(t-s)u'(t)dt = \int_s^\infty g_{\lceil\beta\rceil-\beta}(t-s)u'(t)dt$ exists and defines a $\lceil\beta-1\rceil$ -times continuously differentiable function on $(0,\infty)$, by

$$D_{-}^{\beta}u(s) := (-1)^{\lceil \beta \rceil} \frac{d^{\lceil \beta - 1 \rceil}}{ds^{\lceil \beta - 1 \rceil}} \int_{s}^{\infty} g_{\lceil \beta \rceil - \beta}(t-s)u'(t)dt, \quad s > 0;$$

if $\beta = n \in \mathbb{N}$, then $\mathbf{D}_{-u}^{n} u$ and $D_{-u}^{n} u$ are defined for all *n*-times continuously differentiable functions $u(\cdot)$ on $(0, \infty)$, by $\mathbf{D}_{-u}^{n} u := D_{-u}^{n} u := (-1)^{n} d/d^{n}$, where d/d^{n} denotes the usual derivative operator of order n (cf. also [**279**, (2.3.5)]).

Let $\alpha \in (0, \infty) \setminus \mathbb{N}$, $f \in \mathcal{S}$ and $n = \lceil \alpha \rceil$. Let us recall that the Weyl fractional derivative W^{α}_{\pm} of order α is defined by

$$W_+^{\alpha}f(t) := \frac{(-1)^n}{\Gamma(n-\alpha)} \frac{d^n}{dt^n} \int_t^{\infty} (s-t)^{n-\alpha-1} f(s) ds, \quad t \in \mathbb{R}.$$

If $\alpha = n \in \mathbb{N}_0$, then we set $W^n_+ := (-1)^n \frac{d^n}{dt^n}$. It is well known that the following equality holds: $W^{\alpha+\beta}_+ f = W^{\alpha}_+ W^{\beta}_+ f$, $\alpha, \beta > 0$, $f \in \mathcal{S}$.

As already mentioned, in this book we will use the Caputo fractional derivatives, the Riemann–Liouville fractional derivatives, the modified Liouville rightsided fractional derivatives and Weyl fractional derivatives; for the brief summary of definitions of fractional order derivatives and integrals that usually appear in fractional calculus, we refer the reader to the recent paper [131] by E. C. de Oliveira and J. A. T. Machado.

The basic properties of Caputo fractional derivatives can be obtained by consulting [61] and [292, Preliminaries, Section 1.3]. For the later use, it would be very important to remind ourselves of the following facts. Suppose that $\beta > 0$, $\gamma > 0$ and $\mathbf{D}_t^{\beta+\gamma}u(t)$ is defined. Then the Caputo fractional derivative $\mathbf{D}_t^{\zeta}u(t)$ is defined for any number $\zeta \in (0, \beta + \gamma)$ but the equality $\mathbf{D}_t^{\beta+\gamma}u = \mathbf{D}_t^{\beta}\mathbf{D}_t^{\gamma}u$ does not hold in general (cf. [292, Preliminaries, Section 1.3, p. 14] and the equation (92) below). The validity of this equality can be proved in the following cases:

(1)
$$\gamma \in \mathbb{N}$$

- (2) $\left[\beta + \gamma\right] = \left[\gamma\right]$,
- (3) $u^{(j)}(0) = 0$ for $\lceil \gamma \rceil \leq j \leq \lceil \beta + \gamma \rceil 1$.

If $u \in C([0,\infty): E)$, resp. $u \in C^{m-1}([0,\infty): E)$ and $g_{m-\alpha} * (u - \sum_{k=0}^{m-1} u_k g_{k+1}) \in C^m([0,\infty): E)$, then the following equality holds:

(38)
$$\mathbf{D}_{t}^{\alpha}J_{t}^{\alpha}u(t) = u(t), \quad t \ge 0, \text{ resp. } J_{t}^{\alpha}\mathbf{D}_{t}^{\alpha}u(t) = u(t) - \sum_{k=0}^{m-1} u^{(k)}(0)g_{k+1}(t), \quad t \ge 0.$$

The Laplace transform of function $\mathbf{D}_{t}^{\alpha}u(t)$ is computed by

(39)
$$\int_0^\infty e^{-\lambda t} \mathbf{D}_t^\alpha u(t) dt = \lambda^\alpha \tilde{u}(\lambda) - \sum_{k=0}^{m-1} u^{(k)}(0) \lambda^{\alpha-1-k};$$

cf. the identity [292, (16)] for precise formulation.

The Mittag-Leffler and Wright functions are known to play a fundamental role in seeking of solutions of fractional differential equations. Let $\alpha > 0$ and $\beta \in \mathbb{R}$. Then the Mittag-Leffler function $E_{\alpha,\beta}(z)$ is defined by

$$E_{\alpha,\beta}(z) := \sum_{n=0}^{\infty} \frac{z^n}{\Gamma(\alpha n + \beta)}, \quad z \in \mathbb{C}.$$

Here we assume that $1/\Gamma(\alpha n + \beta) = 0$ if $\alpha n + \beta \in -\mathbb{N}_0$. Set, for short,

$$E_{\alpha}(z) := E_{\alpha,1}(z), \quad z \in \mathbb{C}.$$

Like the function $E_1(z) = e^z$, for which the differential relation $(d/dt)e^{\omega t} = \omega e^{\omega t}$ holds, the function $E_{\alpha}(z)$ satisfies that $\mathbf{D}_t^{\alpha} E_{\alpha}(\omega t^{\alpha}) = \omega E_{\alpha}(\omega t^{\alpha})$. For $\alpha = 1/2$, $E_{1/2}(z)$ is the error function: $E_{1/2}(z) = \exp(z^2) \operatorname{erfc}(-z)$, and for $\alpha = 2$, $E_2(z)$ is the hyperbolic cosine: $E_2(z) = \cosh(\sqrt{z})$. The asymptotic expansion of the entire function $E_{\alpha,\beta}(z)$ is given in the following important theorem (see e.g. [539, Theorem 1.1]):

THEOREM 1.5.1. Let $0 < \sigma < \frac{1}{2}\pi$. Then, for every $z \in \mathbb{C} \setminus \{0\}$ and $m \in \mathbb{N} \setminus \{1\}$,

$$E_{\alpha,\beta}(z) = \frac{1}{\alpha} \sum_{s} Z_s^{1-\beta} e^{Z_s} - \sum_{j=1}^{m-1} \frac{z^{-j}}{\Gamma(\beta - \alpha j)} + O(|z|^{-m}),$$

where Z_s is defined by $Z_s := z^{1/\alpha} e^{2\pi i s/\alpha}$ and the first summation is taken over all those integers s satisfying $|\arg(z) + 2\pi s| < \alpha(\frac{\pi}{2} + \sigma)$.

If $\alpha \in (0,2) \setminus \{1\}$, $\beta > 0$ and $N \in \mathbb{N} \setminus \{1\}$, then we have the following special cases of Theorem 1.5.1:

(40)
$$E_{\alpha,\beta}(z) = \frac{1}{\alpha} z^{(1-\beta)/\alpha} e^{z^{1/\alpha}} + \varepsilon_{\alpha,\beta}(z), \quad |\arg(z)| < \alpha \pi/2,$$

and

(41)
$$E_{\alpha,\beta}(z) = \varepsilon_{\alpha,\beta}(z), \quad |\arg(-z)| < \pi - \alpha \pi/2,$$

where

(42)
$$\varepsilon_{\alpha,\beta}(z) = \sum_{n=1}^{N-1} \frac{z^{-n}}{\Gamma(\beta - \alpha n)} + O(|z|^{-N}), \quad |z| \to \infty.$$

Perhaps the most interesting property of the Mittag-Leffler function $E_{\alpha,\beta}(z)$ is that

(43)
$$\int_0^\infty e^{-\lambda t} t^{\beta-1} E_{\alpha,\beta}(\omega t^\alpha) dt = \frac{\lambda^{\alpha-\beta}}{\lambda^\alpha - \omega}, \quad \operatorname{Re} \lambda > \omega^{1/\alpha}, \ \omega > 0.$$

Recall that the function $t \mapsto E_{\alpha,\beta}(-t)$, $t \ge 0$ is completely monotonic (i.e., that $(-1)^n (d^n/dt^n) E_{\alpha,\beta}(-t) \ge 0$, $t \ge 0$, $n \in \mathbb{N}_0$) provided that $\alpha \in (0,1]$ or $\beta \ge \alpha$ [405]. Among a huge variety of identities for the Mittag-Leffler functions, we will single out the following ones:

(i) For every $j \in \mathbb{N}$ and $\alpha > 0$, there exist uniquely determined real numbers $c_{l,j,\alpha}$ $(1 \leq l \leq j)$ such that:

(44)
$$E_{\alpha}^{(j)}(z) = \sum_{l=1}^{j} c_{l,j,\alpha} E_{\alpha,\alpha j - (j-l)}(z), \quad z \in \mathbb{C}.$$

- (ii) Let $\alpha > 0, \beta > 0$ and $r \in \mathbb{N}$. Then $z^r E_{\alpha,\beta+r\alpha}(z) = E_{\alpha,\beta}(z) + \sum_{n=0}^{r-1} \frac{z^n}{\Gamma(\beta+n\alpha)}$ for all $z \in \mathbb{C}$.
- (iii) If $m \in \mathbb{N}$ and $n \in \mathbb{N}$, then $E_{\frac{m}{n}}(z) = \frac{1}{m} \sum_{r=1}^{m-1} E_{\frac{1}{n}}(z^{1/m} \exp(2\pi i r/m) \text{ for all } z \in \mathbb{C}.$

The Mittag-Leffler function $E_{\alpha,\beta}(z)$ has the integral representation in the form

$$E_{\alpha,\beta}(z) = rac{1}{2\pi i} \int_{G} rac{\lambda^{lpha - eta} e^{\lambda}}{\lambda^{lpha} - z} d\lambda, \quad z \in \mathbb{C},$$

where G is a contour (the Hankel path) which starts and ends at $-\infty$ and encircles the disc $|\lambda| \leq |z|^{1/\alpha}$ counter-clockwise.

Let $\gamma \in (0,1)$. Then the Wright function $\Phi_{\gamma}(\cdot)$ is defined by

$$\Phi_{\gamma}(t) := \mathcal{L}^{-1}(E_{\gamma}(-\lambda))(t), \quad t \ge 0.$$

The Wright function $\Phi_{\gamma}(\cdot)$ can be analytically extended to the whole complex plane by the formula

$$\Phi_{\gamma}(z) = \sum_{n=0}^{\infty} \frac{(-z)^n}{n! \Gamma(1-\gamma-\gamma n)}, \quad z \in \mathbb{C},$$

and the following holds:

(i) $\Phi_{\gamma}(t) \ge 0, t \ge 0,$ (ii) $\int_{0}^{\infty} e^{-\lambda t} \gamma s t^{-1-\gamma} \Phi_{\gamma}(t^{-\gamma}s) dt = e^{-\lambda^{\gamma}s}, \lambda \in \mathbb{C}_{+}, s > 0, \text{ and}$ (iii) $\int_{0}^{\infty} t^{r} \Phi_{\gamma}(t) dt = \frac{\Gamma(1+r)}{\Gamma(1+\gamma r)}, r > -1.$

The asymptotic expansion of the Wright function $\Phi_{\gamma}(\cdot)$, as $|z| \to \infty$ in the sector $|\arg(z)| \leq \min((1-\gamma)3\pi/2, \pi) - \varepsilon$ is given by

$$\Phi_{\gamma}(z) = Y^{\gamma - 1/2} e^{-Y} \bigg(\sum_{m=0}^{M-1} A_m Y^{-M} + O(|Y|^{-M}) \bigg),$$

where $Y = (1 - \gamma)(\gamma^{\gamma} z)^{1/(1-\gamma)}$, $M \in \mathbb{N}$ and A_m are certain real numbers (see e.g. [61]). The Wright function $\Phi_{\gamma}(\cdot)$ can be integrally represented by the formula

$$\Phi_{\gamma}(z) = \frac{1}{2\pi i} \int_{G} \lambda^{\gamma-1} \exp(\lambda - z\lambda^{\gamma}) d\lambda, \quad z \in \mathbb{C},$$

where G is the Hankel path mentioned above.

We also need the following class of Wright type functions

$$\phi(\rho,\nu;z) := \sum_{n=0}^{\infty} \frac{z^n}{n!\Gamma(\rho n + \nu)}, \quad z \in \mathbb{C} \ (\rho > -1, \ \nu \in \mathbb{C}),$$

which is well known because of the following Laplace transform identity:

(45)
$$\int_0^\infty e^{-\lambda t} t^{\nu\rho} \phi(\rho, 1+\rho\nu, -st^\rho) dt = \lambda^{-1-\rho\nu} e^{-s\lambda^{-\rho}},$$

which is valid for s > 0, Re $\lambda > 0$ and $1 + \rho v \ge 0$. If $0 < \rho < 1$, then we know that there exist two finite real constants c > 0 and L > 0 such that

(46)
$$|\phi(\rho,\nu;-r)| \leqslant Le^{-cr^{1/(1+\rho)}}, \quad r \ge 0;$$

if $\rho = 1/2$, then the function $\phi(\rho, \nu; -r)$ can be represented in terms of the well known special functions $\operatorname{erf}(r)$, $\operatorname{erfc}(r)$ and $\operatorname{daw}(r)$. For more details about the Wright functions, see [234] and [401].

In the continuation, we will also use the Bessel functions of first kind. Let us recall that the Bessel function of order $\nu > 0$, denoted by J_{ν} , is defined by

$$J_{\nu}(z) := \left(\frac{z}{2}\right)^{\nu} \sum_{n=0}^{\infty} \frac{(-1)^k (z/2)^{2k}}{k! \Gamma(\nu+k+1)}, \quad z \in \mathbb{C}.$$

Then for each $\nu > 0$ we have the existence of a finite real constant M > 0 such that $\lim_{r \to +\infty} |r^{1/2} J_{\nu}(r)| = 0$. Then the following Laplace transform identity holds true for each $\beta \ge 0$:

(47)
$$\int_0^\infty e^{-\lambda t} J_{1+\beta} \left(2\sqrt{st} \right) s^{(1+\beta)/2} ds = t^{(1+\beta)/2} \lambda^{-2-\beta} e^{-t/\lambda}, \quad \text{Re}\,\lambda > 0, \ t > 0.$$

CHAPTER 2

ABSTRACT DEGENERATE VOLTERRA INTEGRO-DIFFERENTIAL EQUATIONS IN LOCALLY CONVEX SPACES

In this chapter, we investigate abstract (non-scalar) degenerate Volterra integro-differential equations and abstract (multi-term) fractional differential equations in sequentially complete locally convex spaces. We use different methods to achieve what we are aiming for. Special attention is paid to the analysis of hypercyclic, topologically mixing and distributionally chaotic classes of abstract degenerate integro-differential equations.

Unless stated otherwise, throughout this chapter we assume that E is an infinite-dimensional sequentially complete locally convex space over the field \mathbb{C} , SCLCS for short, and that A and B are two closed linear operators acting on E. The abbreviation \circledast stands for the fundamental system of seminorms which induces the topology on E; I denotes the identity operator on E, and $C \in L(E)$ denotes an injective operator satisfying $CA \subseteq AC$. In some sections, we need to have two different pivot spaces, so that we sometimes use the symbols $X, Y, Z \ldots$ in place of E. We start our work by investigating the basic properties of (C, B)-resolvents of closed linear operators.

2.1. (C, B)-resolvents

In this section, we would like to inscribe a few noteworthy facts about the main structural properties of (C, B)-resolvents. We assume that:

- 1. $A: D(A) \subseteq E \to E$ and $B: D(B) \subseteq E \to E$ are closed linear operators;
- 2. $C \in L(E)$ is an injective operator satisfying $CA \subseteq AC$ and $CB \subseteq BC$;
- 3. The closed graph theorem holds for mappings from E into E.

Then the set

$$\rho_C^B(A) := \{ \lambda \in \mathbb{C} : (\lambda B - A)^{-1} C \in L(E) \}$$

is called the (C, B)-resolvent set of A; the (C, B)-spectrum of A is defined by $\sigma_C^B(A) := \mathbb{C} \setminus \rho_C^B(A)$. Sometimes we also write $\rho_C(A, B)$ ($\sigma_C(A, B)$) for $\rho_C^B(A)$ ($\sigma_C^B(A)$); $\rho^B(A) \equiv \rho_I^B(A)$ and $\sigma^B(A) \equiv \sigma_I^B(A)$. If $C \neq I$, then the (C, B)-resolvent set of the operator A need not be open (for a counterexample of this type, with B = I, E being the Hardy space $H^2(\{z \in \mathbb{C} : |z| \leq 1\}), A \in L(E)$ being injective and C = A, see [138, Example 2.5]). For any $\lambda \in \rho_C^B(A)$, we define the right (C, B)-resolvent of A, $R_{\lambda}^{C,B}(A)$ for short, and the left (C, B)-resolvent of A, $L_{\lambda}^{C,B}(A)$ for

short, by

$$R_{\lambda}^{C,B}(A) := (\lambda B - A)^{-1}CB$$
 and $L_{\lambda}^{C,B}(A) := B(\lambda B - A)^{-1}C \in L(E).$

It is checked at once that the existence of operator $B^{-1} \in L(E)$ implies the closedness of the operator AB^{-1} , with domain and range contained in E, as well as that $\rho_C^B(A) \subseteq \rho_C(AB^{-1})$ and

(48)
$$(\lambda - AB^{-1})^{-1}C = B(\lambda B - A)^{-1}C, \quad \lambda \in \rho_C^B(A).$$

Now we would like to draw our attention to the case in which $\rho_C^B(A) \neq \emptyset$, the operator *B* is injective and $B^{-1} \notin L(E)$. Fix temporarily a number $\lambda \in \rho_C^B(A)$. Suppose that (x_{τ}) is a net in E as well as that $x_{\tau} \to x$ as $\tau \to \infty$ and $AB^{-1}x_{\tau} \to y$ as $\tau \to \infty$. This simply implies $B(\lambda B - A)^{-1}Cx_{\tau} \to B(\lambda B - A)^{-1}Cx$ as $\tau \to \infty$ and $B(\lambda B - A)^{-1}C(\lambda - AB^{-1})x_{\tau} = B(\lambda B - A)^{-1}C(\lambda B - A)B^{-1}x_{\tau} = Cx_{\tau} \rightarrow B(\lambda B - A)B^$ $(A)^{-1}C(\lambda x - y)$ as $\tau \to \infty$. Hence, $Cx = B(\lambda B - A)^{-1}C(\lambda x - y), Cx \in D(AB^{-1})$ and $AB^{-1}Cx = AB^{-1}B(\lambda B - A)^{-1}C(\lambda x - y) = A(\lambda B - A)^{-1}C(\lambda x - y)$. Further on, $B(\lambda B - A)^{-1}Cy = \lambda B(\lambda B - A)^{-1}Cx - Cx = A(\lambda B - A)^{-1}Cx, (\lambda B - A)^{-1}Cy = A(\lambda B - A)^{-1}Cy$ $B^{-1}A(\lambda B - A)^{-1}Cx = -B^{-1}Cx + \lambda(\lambda B - A)^{-1}Cx$, whence it easily follows that $Cy = -(\lambda B - A)B^{-1}Cx + Cx$ and $Cy = AB^{-1}Cx$. Hence, the operator AB^{-1} is closable and the supposition $C^{-1} \in L(E)$ implies that the operator AB^{-1} is closed: before proceeding further, we want to observe that the operator AB^{-1} need not be closed if the above requirements hold and $C^{-1} \notin L(E)$ (let A = B = C, and let R(C) be a proper dense subspace of E; then $\overline{CC^{-1}} = I \neq CC^{-1}$, see [138, Example 2.2]). It is not problematic to verify that the operator $\overline{AB^{-1}}$ commutes with the operator $B(\lambda B - A)^{-1}C$ ($\lambda \in \rho_C^B(A)$), and that the operator $\lambda + \overline{AB^{-1}}$ is injective $(\lambda \in \rho_C^B(A))$. By the foregoing, we have $\rho_C^B(A) \subseteq \rho_C(\overline{AB^{-1}})$ and the following modification of (48):

(49)
$$(\lambda - \overline{AB^{-1}})^{-1}C = B(\lambda B - A)^{-1}C, \quad \lambda \in \rho_C^B(A).$$

If the operator B is not injective, then AB^{-1} is an MLO in E and, in this case, we can simply prove that $\rho_C^B(A) \subseteq \rho_C(AB^{-1})$ and (48) continues to hold. Therefore, we have arrived at the following propositions.

PROPOSITION 2.1.1. Suppose that $\rho_C^B(A) \neq \emptyset$ and the operator B is injective.

- (i) If $B^{-1} \in L(E)$ or $C^{-1} \in L(E)$, then the operator AB^{-1} is closed, $\rho_C^B(A) \subseteq \rho_C(AB^{-1})$ and (48) holds.
- (ii) Suppose $B^{-1} \notin L(E)$ and $C^{-1} \notin L(E)$. Then the operator AB^{-1} is closable, $\rho_C^B(A) \subseteq \rho_C(\overline{AB^{-1}})$ and (49) holds.

PROPOSITION 2.1.2. Suppose that the operator B is not injective. Then AB^{-1} is an MLO in E, $\rho_C^B(A) \subseteq \rho_C(AB^{-1})$ and (48) holds in the sense of multivalued linear operators.

The inclusion $\rho_C(AB^{-1}) \subseteq \rho_C^B(A)$ also holds in some cases (for example, if $B \in L(E)$), but we will not go into further details concerning this question here. Using the trivial identities

$$(\lambda B - A)(\mu B - A)^{-1}C = C + (\lambda - \mu)B(\mu B - A)^{-1}C, \quad \mu \in \rho_C^B(A), \ \lambda \in \mathbb{C},$$

$$\begin{aligned} (\mu B-A)^{-1}C(\lambda B-A)x &= Cx + (\lambda-\mu)B(\mu B-A)^{-1}Cx, \\ \mu &\in \rho^B_C(A), \ \lambda \in \mathbb{C}, \ x \in D(A) \cap D(B), \end{aligned}$$

and observing that for each $\lambda \in \rho_C^B(A)$ we have $B(\lambda B - A)^{-1}C^2 = CB(\lambda B - A)^{-1}C$, the following version of Hilbert resolvent equation readily follows:

(50)
$$(\lambda B - A)^{-1}C^2 - (\mu B - A)^{-1}C^2 = (\mu - \lambda)(\mu B - A)^{-1}CB(\lambda B - A)^{-1}C,$$

for any $\lambda, \mu \in \rho_C^B(A)$. From this, we may conclude the following:

(RE1) Suppose $\lambda, \mu \in \rho_C^B(A)$. Then

(51)
$$L_{\lambda}^{C,B}(A)C - L_{\mu}^{C,B}(A)C = (\mu - \lambda)L_{\mu}^{C,B}(A)L_{\lambda}^{C,B}(A)$$

and

$$L^{C,B}_{\mu}(A)L^{C,B}_{\lambda}(A) = L^{C,B}_{\lambda}(A)L^{C,B}_{\mu}(A).$$

Hence, the nonemptiness of set $\rho_C^B(A)$ implies that the function $\lambda \mapsto L_{\lambda}^{C,B}(A) \in L(E), \lambda \in \rho_C^B(A)$ is a C-pseudoresolvent in the sense of [384, Definition 3.1] and the following holds [384]:

- (RE1)' The spaces $N(L_{\lambda}^{C,B}(A))$, $C^{-1}(R(L_{\lambda}^{C,B}(A)))$, $N(C \lambda L_{\lambda}^{C,B}(A))$ and $C^{-1}(R(C \lambda L_{\lambda}^{C,B}(A)))$ are independent of $\lambda \in \rho_{C}^{B}(A)$. (RE1)'' Suppose, additionally, that $N(L_{\lambda}^{C,B}(A)) = \{0\}$ for some $\lambda \in \rho_{C}^{B}(A)$. Then
- we can define the closed linear operator W on E by

$$D(W) := C^{-1}(R(L_{\lambda}^{C,B}(A))) \text{ and } Wx := (\lambda - (L_{\lambda}^{C,B}(A))^{-1}C)x \text{ for } x \in D(W);$$

observe that (RE1)' implies that the definition of W is independent of $\lambda \in \rho_C^B(A)$. Then $C^{-1}WC = W$, $\rho_C^B(A) \subseteq \rho_C(W)$ and $L_{\lambda}^{C,B}(A) = (\lambda - W)^{-1}C$, $\lambda \in \rho_C^B(A)$.

It is well known that the existence of operator W from (RE1)" cannot be proved in the case that there exists $\lambda \in \rho_C^B(A)$ such that the kernel space of operator $L_{\lambda}^{C,B}(A)$ is non-trivial (cf. also Example 2.1.6 below; then (RE1)" holds). It is not difficult to prove that

$$A(\lambda B - A)^{-1}CBx = B(\lambda B - A)^{-1}CAx, \quad x \in D(A) \cap D(B), \ \lambda \in \rho_C^B(A),$$

and (see the second equality in [509, Lemma 2.1.2] with C = I):

$$N(L_{\lambda}^{C,B}(A)) = C^{-1}[\{Ax : x \in D(A) \cap N(B)\}], \quad \lambda \in \rho_{C}^{B}(A).$$

The proof of following resolvent equation follows from (regeful and the fact that, for every $x \in D(B)$, one has $B(\lambda B - A)^{-1}CBCx = CB(\lambda B - A)^{-1}CBx$:

(RE2) Suppose $\lambda, \mu \in \rho_C^B(A)$ and $x \in D(B)$. Then

$$R_{\lambda}^{C,B}(A)Cx - R_{\mu}^{C,B}(A)Cx = (\mu - \lambda)R_{\mu}^{C,B}(A)R_{\lambda}^{C,B}(A)x$$

and

$$R^{C,B}_{\mu}(A)R^{C,B}_{\lambda}(A)x=R^{C,B}_{\lambda}(A)R^{C,B}_{\mu}(A)x$$

Taking into account (RE2) and proceeding as in the proofs of [384, Lemma 3.2, Lemma 3.3, we can deduce the following:

(RE2)' The spaces
$$N(R_{\lambda}^{C,B}(A)), C^{-1}(R(R_{\lambda}^{C,B}(A))) \cap D(B), N(C - \lambda R_{\lambda}^{C,B}(A))$$

and $C^{-1}(R(C - \lambda R_{\lambda}^{C,B}(A))) \cap D(B)$ are independent of $\lambda \in \rho_{C}^{B}(A)$.

Furthermore, if $B \in L(E)$ is injective, then it is not difficult to show that the operator $B^{-1}A$ is closed, as well as that $\rho_C(B^{-1}A) = \rho_C^B(A)$ and

$$(\lambda - B^{-1}A)^{-1}Cx = (\lambda B - A)^{-1}CBx, \quad x \in E;$$

cf. also (RE1)", [**384**, Theorem 3.4] and [**199**, Theorem 1.15]. Making use of [**296**, Lemma 3.3], (50) and the argumentation from [**138**, Section 2] (cf. [**138**, Proposition 2.6, Remark 2.7]), we can prove the following:

PROPOSITION 2.1.3. Let $\emptyset \neq \Omega \subseteq \rho_C^B(A)$ be open, and let $x \in E$.

- (i) The local boundedness of mapping λ → B(λB A)⁻¹Cx, λ ∈ Ω, resp. the assumption that E is barreled and local boundedness of mapping λ → B(λB A)⁻¹C ∈ L(E), λ ∈ Ω, implies the analyticity of mappings λ → (λB A)⁻¹C³x, λ ∈ Ω and λ → B(λB A)⁻¹C³x, λ ∈ Ω, resp. λ → (λB A)⁻¹C³ ∈ L(E), λ ∈ Ω and λ → B(λB A)⁻¹C³ ∈ L(E), λ ∈ Ω and λ → B(λB A)⁻¹C³ ∈ L(E), λ ∈ Ω and L → B(λB A)⁻¹Cx, λ ∈ Ω and λ → B(λB A)⁻¹C ∈ L(E), λ ∈ Ω are analytic, resp. the mappings λ → (λB A)⁻¹C ∈ L(E), λ ∈ Ω and λ → B(λB A)⁻¹C ∈ L(E), λ ∈ Ω are analytic.
- (ii) The continuity of mapping λ → B(λB A)⁻¹Cx, λ ∈ Ω implies its analyticity. The continuity of mappings λ → B(λB A)⁻¹Cx, λ ∈ Ω and λ → (λB A)⁻¹Cx, λ ∈ Ω implies the analyticity of mapping λ → (λB A)⁻¹Cx, λ ∈ Ω; the strong continuity of mapping λ → (λB A)⁻¹C ∈ L(E), λ ∈ Ω (λ → (λB A)⁻¹CB, λ ∈ Ω; with the meaning clear) implies the analyticity of mapping λ → (λB A)⁻¹CBx, λ ∈ Ω, provided that x ∈ D(B)), as well. Furthermore, if E is barreled, then the continuity of mapping λ → (λB A)⁻¹C ∈ L(E), λ ∈ Ω (λ → B(λB A)⁻¹C ∈ L(E), λ ∈ Ω) implies its analyticity; the same conclusion holds for the mapping λ → (λB A)⁻¹CB ∈ L(E), λ ∈ Ω, provided that E is barreled and B ∈ L(E).

For clarity's sake, we will prove parts (i) and (ii) of the following extension of [138, Corollary 2.8].

PROPOSITION 2.1.4. Let $\emptyset \neq \Omega \subseteq \rho_C^B(A)$ be open, and let $x \in E$.

(i) Suppose that the mapping $\lambda \mapsto (\lambda B - A)^{-1}Cx$, $\lambda \in \Omega$ is analytic. Then

$$\begin{aligned} (\lambda B - A)\frac{d^n}{d\lambda^n}(\lambda B - A)^{-1}Cx &= (-n)B\frac{d^{n-1}}{d\lambda^{n-1}}(\lambda B - A)^{-1}Cx, \quad n \in \mathbb{N}, \ \lambda \in \Omega, \\ Cx &\in D(((\lambda B - A)^{-1}B)^{n-1}(\lambda B - A)^{-1}), \ n \in \mathbb{N}, \ \lambda \in \Omega, \ and \ for \ each \\ n \in \mathbb{N} \ and \ \lambda \in \Omega, \end{aligned}$$
$$\frac{d^{n-1}}{d\lambda^{n-1}}(\lambda B - A)^{-1}Cx &= (-1)^{n-1}(n-1)!((\lambda B - A)^{-1}B)^{n-1}(\lambda B - A)^{-1}Cx. \end{aligned}$$

If, in addition, the mapping $\lambda \mapsto B(\lambda B - A)^{-1}Cx$, $\lambda \in \Omega$ is analytic, then for each $n \in \mathbb{N}$ and $\lambda \in \Omega$, $Cx \in D(B((\lambda B - A)^{-1}B)^{n-1}(\lambda B - A)^{-1})$, and

$$\frac{d^{n-1}}{d\lambda^{n-1}}B(\lambda B - A)^{-1}Cx = (-1)^{n-1}(n-1)!B((\lambda B - A)^{-1}B)^{n-1}(\lambda B - A)^{-1}Cx.$$

(ii) Suppose that the mapping $\lambda \mapsto B(\lambda B - A)^{-1}Cx$, $\lambda \in \Omega$ is analytic. Then for each $n \in \mathbb{N}$ and $\lambda \in \Omega$, $((\lambda B - A)^{-1}B)^{n-1}(\lambda B - A)^{-1}C^2x \in R(C)$, $n \in \mathbb{N}, \lambda \in \Omega$ and

$$\frac{d^{n-1}}{d\lambda^{n-1}}B(\lambda B-A)^{-1}Cx = C^{-1}(-1)^{n-1}(n-1)!B((\lambda B-A)^{-1}B)^{n-1}(\lambda B-A)^{-1}C^2x.$$

(iii) Suppose that E is barreled, and the mapping $\lambda \mapsto (\lambda B - A)^{-1}C \in L(E)$, $\lambda \in \Omega$ is analytic, resp., $B \in L(E)$ and the mapping $\lambda \mapsto (\lambda B - A)^{-1}CB \in L(E)$, $\lambda \in \Omega$ is analytic. Then for each $n \in \mathbb{N}$ and $\lambda \in \Omega$, $R(C) \subseteq D(((\lambda B - A)^{-1}B)^{n-1}(\lambda B - A)^{-1})$, resp., $R(CB) \subseteq D(((\lambda B - A)^{-1}B)^{n-1}(\lambda B - A)^{-1})$, and

$$\frac{d^{n-1}}{d\lambda^{n-1}}(\lambda B - A)^{-1}C = (-1)^{n-1}(n-1)!((\lambda B - A)^{-1}B)^{n-1}(\lambda B - A)^{-1}C \in L(E),$$

resp.,

$$\frac{d^{n-1}}{d\lambda^{n-1}}(\lambda B - A)^{-1}CB = (-1)^{n-1}(n-1)!((\lambda B - A)^{-1}B)^{n-1}(\lambda B - A)^{-1}CB \in L(E).$$

(iv) Suppose that E is barreled and the mapping $\lambda \mapsto B(\lambda B - A)^{-1}C \in L(E)$, $\lambda \in \Omega$ is analytic. Then $R(((\lambda B - A)^{-1}B)^{n-1}(\lambda B - A)^{-1}C^2) \subseteq R(C)$, $n \in \mathbb{N}, \lambda \in \Omega$ and

$$\frac{d^{n-1}}{d\lambda^{n-1}}B(\lambda B - A)^{-1}C = C^{-1}(-1)^{n-1}(n-1)! \times B((\lambda B - A)^{-1}B)^{n-1}(\lambda B - A)^{-1}C^2 \in L(E), \quad n \in \mathbb{N}, \ \lambda \in \Omega.$$

PROOF. Let $n \in \mathbb{N}$, let $\lambda \in \Omega$, and let Γ be a positively oriented circle around λ that is contained in Ω . Making use of the Cauchy integral formula, we get that

$$\frac{d^n}{d\lambda^n}(\lambda B - A)^{-1}Cx = \frac{n!}{2\pi i} \oint_{\Gamma} \frac{(zB - A)^{-1}Cx}{(z - \lambda)^{n+1}} dz.$$

Since the operators A and B are closed, we get from the above that $\frac{d^n}{d\lambda^n}(\lambda B - A)^{-1}Cx \in D(A) \cap D(B)$. Applying again the Cauchy integral formula, and taking into account that the operators $C^{-1}(\lambda B - A)C$ and B are closed, we get that

$$\begin{aligned} (\lambda B - A)\frac{d^n}{d\lambda^n}(\lambda B - A)^{-1}Cx &= C^{-1}(\lambda B - A)C\frac{n!}{2\pi i}\oint_{\Gamma}\frac{(zB - A)^{-1}Cx}{(z - \lambda)^{n+1}}dz \\ &= \frac{n!}{2\pi i}\oint_{\Gamma}\frac{C^{-1}(\lambda B - A)C(zB - A)^{-1}Cx}{(z - \lambda)^{n+1}}dz \\ &= \frac{n!}{2\pi i}\oint_{\Gamma}\frac{(\lambda B - A)(zB - A)^{-1}Cx}{(z - \lambda)^{n+1}}dz \end{aligned}$$

$$\begin{split} &= \oint_{\Gamma} \frac{Cx}{(z-\lambda)^{n+1}} dz - \oint_{\Gamma} \frac{B(zB-A)^{-1}Cx}{(z-\lambda)^{n+1}} dz \\ &= -\oint_{\Gamma} \frac{B(zB-A)^{-1}Cx}{(z-\lambda)^{n+1}} dz \\ &= -B \oint_{\Gamma} \frac{(zB-A)^{-1}Cx}{(z-\lambda)^n} dz \\ &= (-n)B \frac{d^{n-1}}{d\lambda^{n-1}} (\lambda B - A)^{-1}Cx, \end{split}$$

which proves the first equality in (i). This implies

$$\frac{d^{n}}{d\lambda^{n}}(\lambda B - A)^{-1}Cx = (-n)[(\lambda B - A)^{-1}B]\frac{d^{n-1}}{d\lambda^{n-1}}(\lambda B - A)^{-1}Cx,$$

and now the remainder of (i) simply follows by induction. To prove (ii), suppose that $\lambda_0 \in \Omega$. Since the mapping $\lambda \mapsto B(\lambda B - A)^{-1}Cx$, $\lambda \in \Omega$ is analytic, the Hilbert resolvent equation (50) shows that the mapping $\lambda \mapsto [(\lambda_0 B - A)^{-1}C]B(\lambda B - A)^{-1}Cx = (1/(\lambda_0 - \lambda))[(\lambda B - A)^{-1}C^2x - (\lambda_0 B - A)^{-1}C^2x], \lambda \in \Omega \setminus \{\lambda_0\}$ is analytic, as well. From this, we may conclude that the mapping $\lambda \mapsto (\lambda B - A)^{-1}C^2x$, $\lambda \in \Omega$ is analytic. By our assumption, the mapping $\lambda \mapsto B(\lambda B - A)^{-1}C^2x$, $\lambda \in \Omega$ is likewise analytic so that part (ii) follows almost directly from (i). The proofs of (iii) and (iv) are simple and therefore omitted. \Box

Summa summarum, Proposition 2.1.3 and Proposition 2.1.4 taken together provide a generalization of [296, Proposition 2.16] for degenerate (C, B)-resolvents.

REMARK 2.1.5. In the case that C = I and E is a Banach space, it is well known that the (I, B)-resolvent set of A is open, as well as that the (I, B)-resolvent, right (I, B)-resolvent and left (I, B)-resolvent of the operator A are analytic in $\rho_B(A)$ [509]. The corresponding statement in locally convex spaces has recently been analyzed in [214, Theorem 1].

Let us consider again the following abstract degenerate Volterra integral equation:

(52)
$$Bu(t) = f(t) + \int_0^t a(t-s)Au(s)ds, \quad t \in [0,\tau),$$

where $0 < \tau \leq \infty$, $t \mapsto f(t)$, $t \in [0, \tau)$ is a continuous mapping with values in a complex SCLCS E, $a \in L^1_{loc}([0, \tau))$ and A, B are closed linear operators with domain and range contained in E.

EXAMPLE 2.1.6. Let the function a(t) be a kernel on $[0, \tau)$ and let the operator W generate a (local) (a, k)-regularized C-resolvent family $(R(t))_{t \in [0, \tau)}$ satisfying $W \int_0^t a(t-s)R(s)x \, ds = R(t)x - k(t)Cx, \, x \in X, \, t \in [0, \tau)$ (the use of operator W here seems to be much better than the use of operator $\overline{AB^{-1}}$, provided that B is injective and $C \neq I$; cf. [292] for the notion and (49)). Then a simple computation involving the definition of operator W shows that for each element $y \in E$ such that

the element $x = C^{-1}L_{\lambda}^{C,B}(A)y$ is well-defined, we have

$$R(t)C^{-1}L_{\lambda}^{C,B}(A)y - k(t)L_{\lambda}^{C,B}(A)y = \int_{0}^{t} a(t-s)R(s)[C^{-1}A(\lambda B - A)^{-1}Cy]ds,$$

for any $t \in [0, \tau)$. In particular, if $y = (\lambda B - A)z$ for some $z \in D(A) \cap D(B)$, then the above requirements hold and we get

$$R(t)Bz - k(t)CBz = \int_0^t a(t-s)R(s)Az \, ds, \quad t \in [0,\tau).$$

Assuming additionally that R(t) commutes with A and B for all $t \in [0, \tau)$, the above implies that the function u(t) := R(t)z, $t \in [0, \tau)$ is a unique strong solution of the abstract degenerate Volterra equation (52), with f(t) = k(t)CBz, $t \in [0, \tau)$.

REMARK 2.1.7. Assume that the functions k(t) and |a|(t) satisfy the condition (P1), as well as that the operator B is injective. Using the definition of operator W, properties stated in (RE1)" and [**292**, Theorem 2.1.5], we have that W generates a global exponentially equicontinuous (a, k)-regularized C-resolvent family $(R(t))_{t\geq 0}$ (cf. [**292**] for the notion) provided that there exists a sufficiently large real number $\omega > 0$ such that the family $\{e^{-\omega t}R(t) : t \geq 0\} \subseteq L(E)$ is equicontinuous as well as that for each $\lambda \in \mathbb{C}$ with $\tilde{k}(\lambda)\tilde{a}(\lambda) \neq 0$ and $\operatorname{Re} \lambda > \omega$, the operator $B - \tilde{a}(\lambda)A$ is injective and

$$\tilde{k}(\lambda)B(B-\tilde{a}(\lambda)A)^{-1}Cx = \int_0^\infty e^{-\lambda t}R(t)x\,dt, \quad x \in E.$$

Combined with the conclusions clarified in the above example, we are in a position to prove Theorem 2.2.8(ii) in a much simpler way provided the injectiveness of B. It is also worth noting that we can use [**292**, Theorem 2.1.6, Proposition 2.1.7, Theorem 2.1.29, Proposition 2.1.32] here.

2.2. Degenerate (a, k)-regularized C-resolvent families in locally convex spaces

The main aim of this section is to present our recent results on degenerate (a, k)regularized *C*-resolvent families in locally convex spaces and semilinear degenerate
relaxation differential equations with abstract differential operators [**306**, **319**].

Before starting our work, we need to remind ourselves of some well known facts and definitions. A nonnegative infinitely differentiable function $\varphi: (0, \infty) \to \mathbb{R}$ is called a Bernstein function iff the function $\varphi'(\cdot)$ is completely monotonic, i.e., $(-1)^n \varphi^{(n+1)}(t) \ge 0, n \in \mathbb{N}_0, t > 0$. Following [459, Definition 4.4], it will be said that a function $a: (0, \infty) \to \mathbb{R}$ is a creep function iff a(t) is nonnegative, nondecreasing and concave. A creep function a(t) has the standard form

$$a(t) = a_0 + a_\infty t + \int_0^t a_1(s) ds$$

where $a_0 = a(0+) \ge 0$, $a_{\infty} = \lim_{t\to\infty} a(t)/t = \inf_{t>0} a(t)/t \ge 0$, and $a_1(t) = a'(t) - a_{\infty}$ is nonnegative, nonincreasing and $\lim_{t\to\infty} a_1(t) = 0$. Recall that, for every $j \in \mathbb{N}$ and $\alpha > 0$, there exist uniquely determined real numbers $c_{l,j,\alpha}$ (1 \le

 $l \leq j$) such that the equation (44) holds for the Mittag-Leffler function $E_{\alpha}(z)$. A function $a \in L^1_{loc}([0,\infty))$ is said to be completely positive iff for any $\eta \geq 0$, the solutions of the convolution equations

$$s(t) + \eta(a * s)(t) = 1$$
 and $r(t) + \eta(a * r)(t) = a(t)$

satisfy $s(t) \ge 0$ and $r(t) \ge 0$ on [0, T]; cf. also [459, Definition 4.5, p. 96] and [395, Remark 3.6, (3.3)].

Let $0 < \tau \leq \infty$ and $a \in L^{1}_{loc}([0,\tau))$. Then we say that the function a(t) is a kernel on $[0,\tau)$ iff for each $f \in C([0,\tau))$ the assumption $\int_{0}^{t} a(t-s)f(s)ds = 0$, $t \in [0,\tau)$ implies f(t) = 0, $t \in [0,\tau)$. If $\tau = \infty$ and $a \neq 0$ in $L^{1}_{loc}([0,\infty))$, then the famous Titchmarsh theorem [**520**, Theorem VII] implies that the function a(t) is automatically a kernel on $[0,\infty)$; the situation is quite different in the case that $\tau < \infty$, then we can apply the Titchmarsh–Foiaş theorem [**40**, Theorem 2.1] (cf. also [**291**, Theorem 3.4.40]) in order to see that the function a(t) is a kernel on $[0,\tau)$ iff $0 \in \text{supp}(a)$. In this place, it is worth noting that, for any function a(t) satisfying (P1), the condition $0 \in \text{supp}(a)$ is necessary and sufficient for the equality $\limsup_{\lambda\to\infty} \lambda^{-1} \ln |\tilde{a}(\lambda)| = 0$ to be true, or equivalently, for the convolution mapping $\mathcal{K}: f \mapsto a * f$ to be an injective operator on $C([0,\infty))$ with dense range in the Fréchet space $C_*([0,\infty))$ of all continuous functions $g: [0,\infty) \mapsto \mathbb{C}$ such that g(0) = 0, equipped with the seminorms $||g||_n := \sup_{t\in[0,n]} |g(t)| \ (n \in \mathbb{N})$; see e.g. [**27**, p. 106].

In this section, we assume that $a \neq 0$ in $L^1_{loc}([0,\infty))$ and $k \neq 0$ in $C([0,\infty))$, so that the functions a(t) and k(t) will be kernels on $[0,\infty)$.

2.2.1. The main structural properties of degenerate (a, k)-regularized *C*-resolvent families. We start this subsection by introducing the following definition (cf. [292, Subsection 2.1.1] and [459] for the case B = I):

DEFINITION 2.2.1. Let $0 < \tau \leq \infty$. A function $u \in C([0, \tau) : E)$ is said to be:

- (i) a (mild) solution of (52) iff $(a * u)(t) \in D(A)$, $t \in [0, \tau)$, A(a * u)(t) = Bu(t) f(t), $t \in [0, \tau)$ and the mapping $t \mapsto Bu(t)$, $t \in [0, \tau)$ is continuous,
- (ii) a strong solution of (52) iff the mapping $t \mapsto Au(t), t \in [0, \tau)$ is continuous, $(a * Au)(t) = Bu(t) - f(t), t \in [0, \tau)$ and the mapping $t \mapsto Bu(t), t \in [0, \tau)$ is continuous,
- (iii) a weak solution of (52) iff for every $(x^*, y^*) \in A^*$ and for every $t \in [0, \tau)$, one has $\langle x^*, Bu(t) \rangle = \langle x^*, f(t) \rangle + \langle y^*, (a * u)(t) \rangle, t \in [0, \tau)$.

It is clear that any strong solution of (52) is also a mild solution of the same problem; the converse statement is not true in general. Since [295, Lemma 2.4] continues to hold in SCLCSs, the concepts mild and weak solution of (52) coincide actually.

We introduce the notion of an exponentially equicontinuous (a, k)-regularized C-resolvent family for (52) as follows.

DEFINITION 2.2.2. Suppose that the functions a(t) and k(t) satisfy (P1), as well as that $R(t): D(B) \to E$ is a linear mapping $(t \ge 0)$. Let $C \in L(E)$ be injective, and let $CA \subseteq AC$. Then the operator family $(R(t))_{t\ge 0}$ is said to be an exponentially equicontinuous (a, k)-regularized *C*-resolvent family for (52) iff there exists $\omega \ge \max(0, \operatorname{abs}(a), \operatorname{abs}(k))$ such that the following holds:

- (i) The mapping $t \mapsto R(t)x, t \ge 0$ is continuous for every fixed element $x \in D(B)$.
- (ii) The family $\{e^{-\omega t}R(t): t \ge 0\}$ is equicontinuous, i.e., for every $p \in \circledast$, there exist c > 0 and $q \in \circledast$ such that

(53)
$$p(e^{-\omega t}R(t)x) \leq cq(x), \quad x \in D(B), \ t \geq 0.$$

(iii) For every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\tilde{k}(\lambda) \neq 0$, the operator $B - \tilde{a}(\lambda)A$ is injective, $C(R(B)) \subseteq R(B - \tilde{a}(\lambda)A)$ and

(54)
$$\tilde{k}(\lambda)(B - \tilde{a}(\lambda)A)^{-1}CBx = \int_0^\infty e^{-\lambda t}R(t)x\,dt, \quad x \in D(B).$$

If $k(t) = g_{r+1}(t)$ for some $r \ge 0$, then it is also said that $(R(t))_{t\ge 0}$ is an exponentially equicontinuous *r*-times integrated (a, C)-regularized resolvent family for (52); an exponentially equicontinuous 0-times integrated (a, C)-regularized resolvent family for (52) is also said to be an exponentially equicontinuous (a, C)-regularized resolvent family for (52).

- REMARK 2.2.3. (i) If B = I, then the above simply means that A is a subgenerator of the exponentially equicontinuous (a, k)-regularized Cresolvent family $(R(t))_{t \ge 0}$ in the sense of [296, Definition 2.1]; cf. also [296, Theorem 2.7]. The case $B \ne I$ is more difficult to deal with; for example, the validity of some very simple equalities, like R(t)Ax = AR(t)x, $t \ge 0$ or R(t)Bx = BR(t)x, $t \ge 0$, cannot be proved without making some new assumptions. Furthermore, it is not clear how one can define, by using a method similar to that employed in Definition 2.2.2, the notion of an exponentially equicontinuous (a, k)-regularized (C_1, C_2) -existence and uniqueness family for (52) in a satisfactory way.
- (ii) In contrast to [542,543], we do not assume in Definition 2.2.2 that $CB \subseteq BC$ or $R(C) \subseteq R(B \tilde{a}(\lambda)A)$ (Re $\lambda > \omega$, $\tilde{k}(\lambda) \neq 0$).
- (iii) The uniqueness theorem for Laplace transform implies that there exists at most one exponentially equicontinuous (a, k)-regularized C-resolvent family for (52).
- (iv) If *E* is complete and *B* is densely defined, then [**419**, Lemma 22.19] combined with (ii) of Definition 2.2.2 implies that, for every $t \ge 0$ and for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\tilde{k}(\lambda) \ne 0$, there exist two operators $\hat{R}(t), G(\lambda) \in L(E)$ such that $\hat{R}(t)x = R(t)x, x \in D(B)$ and $G(\lambda)x =$ $(B - \tilde{a}(\lambda)A)^{-1}CBx, x \in D(B)$. The operator family $(\hat{R}(t))_{t\ge 0} \subseteq L(E)$ is strongly continuous and, for every $p \in \circledast$, there exist c > 0 and $q \in \circledast$ such that (53) holds for all $x \in E$ and $t \ge 0$, with $(R(t))_{t\ge 0}$ replaced by $(\hat{R}(t))_{t\ge 0}$. Furthermore, $\tilde{k}(\lambda)G(\lambda)x = \int_0^\infty e^{-\lambda t}R(t)x \, dt, x \in E$.
- (v) The notion of a (local) (a, k)-regularized *C*-resolvent family $(R(t))_{t \in [0,\tau)}$ for (52) can be defined in many different ways, but it seems that there is no satisfactory option that would provide us a general approach to the

Volterra problems of the kind (52). Observe also that we can simply construct a great number of examples of strongly continuous operator families $(R(t))_{t \in [0,\tau)} \subseteq L(E)$ for which neither $R(t)A \subseteq R(t)A, t \in [0,\tau)$ nor $R(t)B \subseteq R(t)B, t \in [0,\tau)$, or only $R(t)B \nsubseteq R(t)B, t \in [0,\tau)$, but the solution of (52) exists and has the form $u(t) = R(t)x, t \in [0,\tau)$ for some $x \in E$; see e.g. [292, Example 2.8.11] and [295, Example 2.31]. Because of that, in the sequel of this section we tend to pay attention primarily to the notion introduced in Definition 2.2.2.

The proof of following proposition is omitted for the sake of brevity (cf. [296, Proposition 2.4] for more details).

- PROPOSITION 2.2.4. (i) Let $(R(t))_{t \ge 0}$ be an exponentially equicontinuous (a, k)-regularized C-resolvent family for (52), and let $b \ne 0$ in $L^1_{loc}([0, \infty))$. If there exist $M \ge 1$ and $\omega \ge 0$ such that $\int_0^t |b(s)| ds \le M e^{\omega t}$, $t \ge 0$, then $((b * R)(t))_{t \ge 0}$ is an exponentially equicontinuous (a, k * b)-regularized C-resolvent family for (52).
- (ii) Let $(R_i(t))_{t\geq 0}$ be an exponentially equicontinuous (a, k_i) -regularized C-resolvent family for (52), i = 1, 2. Then $(k_2 * R_1)(t)x = (k_1 * R_2)(t)x$, $t \geq 0, x \in D(B)$.
- (iii) Let $(R(t))_{t\geq 0}$ be an exponentially equicontinuous (a, k)-regularized C-resolvent family for (52). Suppose that $k(0) \neq 0$, b(t) satisfies (P1), $(b * k)(t) + k(t)k(0)^{-1} = 1$, $t \geq 0$ and the function $t \mapsto \int_0^t |b(s)| ds$, $t \geq 0$ is exponentially bounded. Then $(S(t) \equiv k(0)^{-1}R(t) + (b*R(\cdot))(t))_{t\geq 0}$ is an exponentially equicontinuous (a, C)-regularized resolvent family for (52).

It should be noted that our analysis covers many important subjects that have not been considered in [542, 543]. For example, we are in a position to clarify Hille–Yosida's type theorems for degenerate exponentially equicontinuous (a, k)regularized *C*-resolvent families.

THEOREM 2.2.5. (cf. [292, Theorem 2.1.6] for the case B = I)

- (i) Let $\omega_0 > \max(0, \operatorname{abs}(a), \operatorname{abs}(k))$, and let a(t) and k(t) satisfy (P1). Assume that, for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega_0$ and $\tilde{k}(\lambda) \neq 0$, the operator $B \tilde{a}(\lambda)A$ is injective and $C(R(B)) \subseteq R(B \tilde{a}(\lambda)A)$. If for each $x \in D(B)$ there exists a function $\Upsilon_x : \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda > \omega_0\} \to E$ which satisfies:
 - (a) $\Upsilon_x(\lambda) = \tilde{k}(\lambda)(B \tilde{a}(\lambda)A)^{-1}CBx$, $\operatorname{Re} \lambda > \omega_0$, $\tilde{k}(\lambda) \neq 0$,
 - (b) the mapping $\lambda \mapsto \Upsilon_x(\lambda)$, $\operatorname{Re} \lambda > \omega_0$ is analytic, and
 - (c) there exists $r \ge -1$ such that for each $p \in \circledast$ there exist $M_p > 0$ and $q_p \in \circledast$ satisfying

$$p(\Upsilon_x(\lambda)) \leqslant M_p q_p(x) |\lambda|^r$$
, $\operatorname{Re} \lambda > \omega_0, x \in D(B)$,

then, for every $\alpha > 1$, there exists an exponentially equicontinuous $(a, k * g_{\alpha+r})$ -regularized C-resolvent family $(R_{\alpha}(t))_{t \ge 0}$ for (52), and there exists a constant $c_{\alpha} > 0$ such that

$$p(R_{\alpha}(t)x) \leq c_{\alpha}M_pq_p(x)e^{\omega_0 t}, \quad p \in \circledast, \ x \in D(B), \ t \ge 0.$$

(ii) Let ω ∈ ℝ, ω₀ > max(0, ω, abs(a), abs(k)), and let a(t) and k(t) satisfy (P1). Assume that for each x ∈ D(B) there exists an infinitely differentiable function Υ_x: {λ ∈ C : Re λ > ω₀} → E which satisfies the item (i)(a) for real values of parameter λ, as well as that for each p ∈ ⊛ there exist c_p > 0 and q_p ∈ ⊛ such that

$$p\left(n!^{-1}(\lambda-\omega)^{n+1}\frac{d^n}{d\lambda^n}\Upsilon_x(\lambda)\right) \leqslant c_p q_p(x),$$

provided $\tilde{k}(\lambda) \neq 0, \ \lambda > \omega_0, \ x \in D(B), \ n \in \mathbb{N}_0$. Then, for every $r \in (0,1]$, there exists an exponentially equicontinuous $(a, k * g_r)$ -regularized C-resolvent family $(R_r(t))_{t\geq 0}$ for (52), and

$$p(R_r(t+h)x - R_r(t)x) \leqslant \frac{2c_p q_p(x)}{r\Gamma(r)} \max(e^{\omega(t+h)}, 1)h^r,$$

provided $p \in \circledast$, $t \ge 0$, h > 0, $x \in D(B)$. Furthermore, if B is densely defined and the mapping $t \mapsto R_1(t)x$, $t \ge 0$ is continuously differentiable for all $x \in D(B)$, then there exists an exponentially equicontinuous (a, k)regularized C-resolvent family for (52).

In the subsequent theorem (cf. [1, 61, 63, 395, 459, 460] and [292] for more details), we analyze subordination principles for degenerate (a, k)-regularized *C*-resolvent families.

THEOREM 2.2.6. (i) Let a(t), b(t) and c(t) satisfy (P1) and let $\int_0^\infty e^{-\beta t} |b(t)| dt < \infty$ for some $\beta \ge 0$. Let

$$\alpha = \tilde{c}^{-1} \Big(\frac{1}{\beta} \Big) \ if \ \int_0^\infty c(t) dt > \frac{1}{\beta}, \quad \alpha = 0 \ otherwise,$$

and let $\tilde{a}(\lambda) = \tilde{b}(\frac{1}{\tilde{c}(\lambda)}), \lambda \ge \alpha$. Assume that there exists an exponentially equicontinuous (b, k)-regularized C-resolvent family $(R_b(t))_{t\ge 0}$ for (52), with a(t) replaced by b(t), satisfying that the family $\{e^{-\omega_b t}R_b(t):t\ge 0\}$ is equicontinuous for some $\omega_b \ge 0$ (with the meaning clear). Assume, further, that c(t) is completely positive and there exists a function $k_1(t)$ satisfying (P1) and

$$\widetilde{k_1}(\lambda) = \frac{1}{\lambda \widetilde{c}(\lambda)} \widetilde{k}\left(\frac{1}{\widetilde{c}(\lambda)}\right), \quad \lambda > \omega_0, \ \widetilde{k}\left(\frac{1}{\widetilde{c}(\lambda)}\right) \neq 0, \ \text{for some } \omega_0 > 0.$$

Let

$$\omega_a = \tilde{c}^{-1} \left(\frac{1}{\omega_b}\right) if \int_0^\infty c(t) dt > \frac{1}{\omega_b}, \quad \omega_a = 0 \text{ otherwise.}$$

Then, for every $r \in (0,1]$, there exists an exponentially equicontinuous $(a, k_1 * g_r)$ -regularized C-resolvent family $(R_r(t))_{t\geq 0}$ for (52), satisfying that the family $\{e^{-\omega_a t}R_r(t) : t \geq 0\}$ is equicontinuous, if $\omega_b = 0$ or $\omega_b \tilde{c}(0) \neq 1$, resp., for every $\varepsilon > 0$, there exists $M_{\varepsilon} \geq 1$ such that the family $\{e^{-\varepsilon t}R_r(t) : t \geq 0\}$ is equicontinuous, if $\omega_b > 0$ and $\omega_b \tilde{c}(0) = 1$.

- (ii) Suppose α ≥ 0 and there exists an exponentially equicontinuous (1, g_α)-regularized C-resolvent family for 52. Assume, further, that a(t) and k(t) satisfies (P1), as well as that k̃(λ) = ã(λ)^α for λ sufficiently large, and a(t) is completely positive. Then, for every r ∈ (0,1], there exists an exponentially equicontinuous (a, k * g_r)-regularized C-resolvent family for (52) ((a, a^{*,n} * g_r)-regularized C-resolvent family for (52) if α = n ∈ N, resp. (a, g_{r+1})-regularized C-resolvent family if α = 0).
- (iii) Suppose α ≥ 0 and there exists an exponentially equicontinuous (g₂, g_α)-regularized C-resolvent function for (52). Let L¹_{loc}([0,∞)) ∋ c be completely positive and let a(t) = (c * c)(t), t ≥ 0. (Recall that for any function a ∈ L¹_{loc}([0,∞)) given in advance, such a function c(t) always exists provided a(t) is completely positive or a(t) ≠ 0 is a creep function and a₁(t) is log-convex). Assume k(t) satisfies (P1) and k̃(λ) = c̃(λ)^α/λ, λ sufficiently large. Then, for every r ∈ (0,1], there exists an exponentially equicontinuous (a, k * g_r)-regularized C-resolvent family for (52) ((a, c^{*,n} * g_r)-regularized C-resolvent family for (52) if α = n ∈ N, resp. (a, g_{r+1})-regularized C-resolvent family for (52) if α = 0).

REMARK 2.2.7. In the case that B = I and A is densely defined, the mapping $t \mapsto R_1(t)x, t \ge 0$, which appears in the formulation of Theorem 2.2.5(ii), is continuously differentiable for all $x \in E$ and, in the situation of Theorem 2.2.6(i), there exists an exponentially equicontinuous (a, k_1) -regularized C-resolvent family $(R(t))_{t\ge 0}$ for (52) satisfying that the family $\{e^{-\omega_a t}R_r(t):t\ge 0\}$ is equicontinuous, resp., for every $\varepsilon > 0$, the family $\{e^{-\varepsilon t}R_r(t):t\ge 0\}$ is equicontinuous (similar statements hold in the case of Theorem 2.2.6(ii)–(iii), cf. [292, Theorem 2.1.8] for further information). It is not clear whether the above results can be reformulated for abstract degenerate Volterra equations.

The following theorem provides an extension of [542, Theorem 1.6] and [543, Theorem 3.1].

THEOREM 2.2.8. Let $\tau = \infty$, let the functions |a|(t) and k(t) satisfy (P1), and let $(R(t))_{t\geq 0}$ be an exponentially equicontinuous (a, k)-regularized C-resolvent family for (52), satisfying (ii) of Definition 2.2.2 with $\omega \geq \max(0, \operatorname{abs}(|a|), \operatorname{abs}(k))$. (i) Suppose that $v_0 \in D(B)$ and the following condition holds:

(i.1) for every $x \in D(B)$, there exist a number $\omega_0 > \omega$ and a function $h(\lambda; x) \in LT - E$ such that $h(\lambda; x) = \tilde{k}(\lambda)B(B - \tilde{a}(\lambda)A)^{-1}CBx$ provided Re $\lambda > \omega_0$ and $\tilde{k}(\lambda) \neq 0$.

Then the function $u(t) = R(t)v_0$, $t \ge 0$ is a mild solution of (52) with $f(t) = k(t)CBv_0$, $t \ge 0$. The uniqueness of mild solutions holds if we suppose additionally that $CB \subseteq BC$ and the function k(t) satisfies (P2).

- (ii) Suppose that $v_0 \in D(A) \cap D(B)$, $CB \subseteq BC$, and the following condition holds:
 - (ii.1) for every $x \in E$, there exist a number $\omega_1 > \omega$ and a function $h(\lambda; x) \in LT E$ such that $h(\lambda; x) = \tilde{k}(\lambda)B(B \tilde{a}(\lambda)A)^{-1}Cx$ provided $\operatorname{Re} \lambda > \omega_1$ and $\tilde{k}(\lambda) \neq 0$.

Then the function $u(t) = R(t)v_0$, $t \ge 0$ is a strong solution of (52) with $f(t) = k(t)CBv_0$, $t \ge 0$. The uniqueness of strong solutions holds if we suppose additionally that the function k(t) satisfies (P2).

PROOF. Let $v_0 \in D(B)$. Then, for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\tilde{k}(\lambda) \neq 0$, we have

(55)
$$\tilde{a}(\lambda)\tilde{k}(\lambda)A(B-\tilde{a}(\lambda)A)^{-1}CBv_0 = \tilde{k}(\lambda)[-CBv_0 + B(B-\tilde{a}(\lambda)A)^{-1}CBv_0].$$

Taking into account (i.1), the equation (54), as well as the closedness of B and the uniqueness theorem for Laplace transform, it can be simply proved that $R(t)v_0 \in D(B)$ for all $t \ge 0$, as well as that the mapping $t \mapsto BR(t)v_0, t \ge 0$ is continuous and

$$\tilde{k}(\lambda)B(B-\tilde{a}(\lambda)A)^{-1}CBv_0 = \int_0^\infty e^{-\lambda t}BR(t)v_0dt, \quad \operatorname{Re}\lambda > \omega_0, \ \tilde{k}(\lambda) \neq 0.$$

The previous equality in combination with (55) and the uniqueness theorem for Laplace transform implies that

(56)
$$A(a * R)(t)v_0 = BR(t)v_0 - k(t)CBv_0, \quad t \ge 0,$$

so that the function $t \mapsto u(t) \equiv R(t)v_0, t \ge 0$ is a mild solution of (52) with $f(t) = k(t)CBv_0, t \ge 0$. In order to prove (ii), fix an element $v_0 \in D(A) \cap D(B)$. Since we have assumed that $CB \subseteq BC$, it readily follows that

$$\tilde{k}(\lambda)B(B-\tilde{a}(\lambda)A)^{-1}CBv_0-\tilde{k}(\lambda)BCv_0=\tilde{k}(\lambda)\tilde{a}(\lambda)B(B-\tilde{a}(\lambda)A)^{-1}CAv_0,$$

for any $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $k(\lambda) \neq 0$. By (ii.1), we know that there exists a continuous function $t \mapsto G(t), t \geq 0$ such that

(57)
$$\int_0^\infty e^{-\lambda t} G(t) dt = \tilde{k}(\lambda) B(B - \tilde{a}(\lambda)A)^{-1} C A v_0,$$

for any $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega_1$ and $k(\lambda) \neq 0$. With the help of (56)–(57) we can show that:

$$\int_0^\infty e^{-\lambda t} A \int_0^t a(t-s)R(s)v_0 ds \, dt = \int_0^\infty e^{-\lambda t} [BR(t)v_0 - k(t)CBv_0] dt$$
$$= \int_0^\infty e^{-\lambda t} \int_0^t a(t-s)G(s) ds \, dt,$$

for any $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega_1$. By the uniqueness theorem for Laplace transform, we get that $A \int_0^t a(t-s)R(s)v_0 ds = \int_0^t a(t-s)G(s)ds, t \ge 0$. Let $(x^*, y^*) \in A^*$. Then $\langle x^*, (a * G)(t) \rangle = \langle y^*, (a * R(\cdot)v_0)(t) \rangle, t \ge 0$, i.e., $\int_0^t a(t-s)\langle x^*, G(s) \rangle ds = \int_0^t a(t-s)\langle y^*, R(s)v_0 \rangle ds, t \ge 0$. Since the function a(t) is a kernel on $[0, \infty)$, we obtain that $\langle x^*, G(t) \rangle = \langle y^*, R(t)v_0 \rangle, t \ge 0$. Because of that, the equality $AR(t)v_0 = G(t)$ holds for any $t \ge 0$, and the function $u(t) = R(t)v_0, t \ge 0$ is a strong solution of (52) with $f(t) = k(t)CBv_0, t \ge 0$. It remains to be proved the uniqueness of mild solutions under the additional assumptions $CB \subseteq BC$ and the function k(t) satisfies (P2). Towards this end, suppose that the function $t \mapsto u(t),$ $t \ge 0$ is a mild solution of (52) with $f(t) \equiv 0$. Put $v(t) := Cu(t), t \ge 0$. Since $CB \subseteq BC$, the function v(t) is also a mild solution of (52) with $f(t) \equiv 0$. Since $\operatorname{abs}(|a|) < \infty$, we have that there exist constants $M \ge 1$ and $\omega_2 \ge 0$ such that $\int_0^t |a(r)| dr \le M e^{\omega_2 t}, t \ge 0$. Then, for every $\sigma > 0$, we have

$$(58) \quad \int_0^t \int_s^\infty e^{\lambda(s-r-\sigma)} |a(r)| dr \, ds = e^{-\lambda\sigma} \int_0^t \int_0^\infty e^{-\lambda\eta} |a(s+\eta)| d\eta \, ds$$
$$= e^{-\lambda\sigma} \int_0^\infty \int_0^t e^{-\lambda\eta} |a(s+\eta)| ds \, d\eta = e^{-\lambda\sigma} \int_0^\infty e^{-\lambda\eta} \int_\eta^{t+\eta} |a(r)| dr \, d\eta$$
$$\leqslant M e^{\omega_2 t - \lambda\sigma} \int_0^\infty e^{-(\lambda - \omega_2)\eta} d\eta = M e^{\omega_2 t - \lambda\sigma} / (\lambda - \omega_2), \quad t \ge 0, \ \lambda \text{ suff. large.}$$

Moreover,

$$(B - \tilde{a}(\lambda)A)(e^{\lambda \cdot} * a * v)(t) = B(e^{\lambda \cdot} * a * v)(t) - \tilde{a}(\lambda)(e^{\lambda \cdot} * Bv)(t)$$
$$= -\left(Bv(\cdot) * \int_{\cdot}^{\infty} e^{\lambda(\cdot - s)}a(s)ds\right)(t), \quad \operatorname{Re} \lambda > \omega, \ t \ge 0.$$

Combined with (54), the above implies:

(59)
$$(e^{\lambda \cdot} * a * v)(t) = -\left(\int_0^\infty e^{-\lambda r} R(r) u(\cdot) dr * \int_{\cdot}^\infty e^{\lambda(\cdot - s)} a(s) ds\right)(t),$$

for $\operatorname{Re} \lambda > \omega$ and $t \ge 0$. By (58)–(59), and (ii) of Definition 2.2.2, we get that for each $p \in \mathfrak{B}$ there exist c > 0 and $q \in \mathfrak{B}$ such that

(60)
$$e^{-\lambda\sigma}p((e^{\lambda\cdot} * a * v)(t)) \leq \frac{ce^{-\lambda\sigma}}{|\tilde{k}(\lambda)|(\lambda-\omega)} \int_0^t \int_s^\infty e^{\lambda(s-r)} |a(r)|q(u(t-s))dr ds,$$

for $\lambda > \omega$, $t \ge 0$ and $\sigma \ge 0$. On the other hand, we can always find constants $\sigma_0 > 0$ and $M \ge 1$ such that

(61)
$$\frac{e^{-\lambda\sigma_0}}{|\tilde{k}(\lambda)|} \leqslant M, \quad \lambda > \omega + 1.$$

If not so, then there exists a sequence $(\lambda_n)_{n\in\mathbb{N}}$ in $(\omega + 1, \infty)$ such that $|\tilde{k}(\lambda_n)| \leq e^{-n\lambda_n}$, $n \in \mathbb{N}$. Making use of the condition (P2) and the Bolzano–Weierstrass theorem, it can be easily seen that the sequence $(\lambda_n)_{n\in\mathbb{N}}$ must be unbounded; hence, $\limsup_{\lambda\to+\infty} (\ln |\tilde{k}(\lambda)|/\lambda) = -\infty$ and [27, Proposition 2.4.3] implies that $k(t) = 0, t \geq 0$, which is a contradiction. Applying now (60)–(61), we obtain that $\lim_{\lambda\to+\infty} e^{-\lambda\sigma}p((e^{\lambda} * a * v)(t)) = 0, t \geq 0, p \in \circledast$. Proceeding as in the proof of [542, Theorem 1.6], it readily follows that $(a*v)(t) = 0, t \geq 0$. Since the function a(t) is a kernel on $[0,\infty)$ and C is injective, we get that $0 = v(t) = Cu(t) = u(t), t \geq 0$.

REMARK 2.2.9. (i) Suppose $v_0 \in D(B)$ and, instead of (i.1), a slightly stronger condition

(i.1)' for every $x \in D(B)$, there exist a number $\omega_0 > \omega$ and two functions $h_1(\lambda; x), h_2(\lambda) \in LT - E$ such that $h_1(\lambda; x) = \tilde{k}(\lambda)\tilde{a}(\lambda)^{-1}B(B - \delta)$

 $\tilde{a}(\lambda)A)^{-1}CBx$ and $h_2(\lambda) = \tilde{k}(\lambda)\tilde{a}(\lambda)^{-1}$, provided $x \in D(B)$, $\operatorname{Re} \lambda > \omega_0$ and $\tilde{k}(\lambda) \neq 0$.

Then we can simply prove with the help of closedness of A and resolvent equation that there exists a function $h_3(\lambda; x) \in LT - E$ such that $h_3(\lambda; x) = \tilde{k}(\lambda)A(B - \tilde{a}(\lambda)A)^{-1}CBx$, provided $x \in D(B)$, $\operatorname{Re} \lambda > \omega_0$ and $\tilde{k}(\lambda) \neq 0$. Keeping in mind the uniqueness theorem for Laplace transform, it readily follows that the mapping $t \mapsto AR(t)v_0, t \geq 0$ is well defined and continuous. In conclusion, the function $u(t) = R(t)v_0, t \geq 0$ is a strong solution of (52) with $f(t) = k(t)CBv_0, t \geq 0$.

- (ii) Observe that we must impose the condition $abs(|a|) < \infty$ here because we need to apply Theorem 1.4.2(vi).
- (iii) Regarding the question of whether the function k(t) satisfies (P2) or not, the following comment should be made: Suppose that the assertion of [445, Lemma 4.1.1, p. 100] continues to hold with the sequence $\lambda_n = n$ replaced by any strictly increasing sequence $(\lambda_n)_{n \in \mathbb{N}}$ of positive real numbers tending to infinity (yet unproven or unsworn hypothesis, unknown in the existing literature to the best knowledge of the author). Then the uniqueness of solutions clarified in Theorem 2.2.8 holds even if the function k(t) does not satisfy (P2), and this can be proved by using the estimate (60) and the fact that there exist a positive real number $\sigma' > 0$ and a strictly increasing sequence $(\lambda_n)_{n \in \mathbb{N}}$ of positive real numbers such that $\lim_{n\to\infty} \lambda_n = \infty$ and $|\tilde{k}(\lambda_n)| \ge e^{-\sigma'\lambda_n}$, $n \in \mathbb{N}$ (cf. [27, Proposition 2.4.3]).

In the subsequent proposition, we state a rescaling result for degenerate K-convoluted C-semigroups in locally convex spaces; observe, however, that it is very difficult to say something more about perturbation properties of exponentially equicontinuous (a, k)-regularized C-resolvent families introduced in this section (cf. [296, Theorem 4.2(ii)] and [292, Section 2.6] for further information concerning non-degenerate case).

PROPOSITION 2.2.10. (i) Suppose $z \in \mathbb{C}$, K(t) satisfy (P1), F(t) is exponentially bounded, $k(t) = \int_0^t K(s) ds$, $t \ge 0$, there exists $\omega_0 > 0$ such that

$$\frac{\tilde{K}(\lambda) - \tilde{K}(\lambda + z)}{\tilde{K}(\lambda + z)} = \int_0^\infty e^{-\lambda t} F(t) dt, \quad \operatorname{Re} \lambda > \omega_0, \ \tilde{K}(\lambda + z) \neq 0,$$

and there exists an exponentially equicontinuous K-convoluted C-semigroup $(S_K(t))_{t\in[0,\tau)}$ for (52), i.e., (a,k)-regularized C-resolvent family for (52) with a(t) = 1. Then there exists an exponentially equicontinuous K-convoluted C-semigroup $(S_{K,z}(t))_{t\in[0,\tau)}$ for (52), with A replaced by A-zB. Furthermore,

$$S_{K,z}(t)x = e^{-tz}S_K(t)x + \int_0^t F(t-s)e^{-zs}S_K(s)x\,ds, \quad t \ge 0, \ x \in D(B).$$

(ii) Suppose z ∈ C, α > 0 and there exists an exponentially equicontinuous α-times integrated C-semigroup (S_α(t))_{t∈[0,τ)} for (52), i.e. (a, k)-regularized C-resolvent family for (52) with a(t) = 1 and k(t) = g_{α+1}(t). Then there exists an exponentially equicontinuous α-times integrated C-semigroup (S_{α,z}(t))_{t∈[0,τ)} for (52), with A replaced by A - zB. Furthermore,

$$S_{\alpha,z}(t)x = e^{-zt}S_{\alpha}(t)x + \int_{0}^{t} \sum_{n=1}^{\infty} {\alpha \choose n} \frac{z^{n}t^{n-1}}{(n-1)!} e^{-zs}S_{\alpha}(s)x \, ds, \quad t \ge 0, \ x \in D(B).$$

2.2.2. Differential and analytical properties of degenerate (a, k)-regularized *C*-resolvent families. In this subsection, we clarify the most important differential and analytical properties of degenerate (a, k)-regularized *C*-resolvent families. Let us recall that $\Sigma_{\alpha} = \{z \in \mathbb{C} \setminus \{0\} : |\arg(z)| < \alpha\}$ ($\alpha \in (0, \pi]$).

DEFINITION 2.2.11. (cf. [296, Definition 3.1] for the case B = I) Suppose that the functions a(t) and k(t) satisfy (P1), as well as that $C \in L(E)$ is an injective mapping satisfying $CA \subseteq AC$. Let $(R(t))_{t\geq 0}$ be an exponentially equicontinuous (a, k)-regularized C-resolvent family for (52), and let $0 < \alpha \leq \pi$. Then it is said that $(R(t))_{t\geq 0}$ is an exponentially equicontinuous (equicontinuous), analytic (a, k)-regularized C-resolvent family for (52), of angle α , iff there exists $\omega \geq \max(0, \operatorname{abs}(a), \operatorname{abs}(k))$ ($\omega = 0$) such that the following holds:

- (i) For every $x \in D(B)$, the mapping $t \mapsto R(t)x$, t > 0 can be analytically extended to the sector Σ_{α} ; since no confusion seems likely, we shall denote the extension by the same symbol.
- (ii) For every $x \in D(B)$ and $\beta \in (0, \alpha)$, one has $\lim_{z \to 0, z \in \Sigma_{\beta}} R(z)x = R(0)x$.
- (iii) The family $\{e^{-\omega z}R(z): z \in \Sigma_{\beta}\}$ is equicontinuous for all $\beta \in (0, \alpha)$, i.e., for every $p \in \circledast$, there exist c > 0 and $q \in \circledast$ such that

$$p(e^{-\omega z}R(z)x) \leq cq(x), \quad x \in D(B), \ z \in \Sigma_{\beta}.$$

Before going any further, we would like to observe that the assertion of [296, Theorem 3.6] cannot be transferred to degenerate (a, k)-regularized C-resolvent families without imposing some restrictive assumptions, including by all means the injectivity of the operator B. This is not the case with the assertion of [296, Theorem 3.7], as the following theorem shows.

THEOREM 2.2.12. Assume that |a|(t) and k(t) satisfy (P1), A and B are closed linear operators, $\omega \ge \max(0, \operatorname{abs}(|a|), \operatorname{abs}(k)), \alpha \in (0, \pi/2], C \in L(E)$ is injective and satisfies $CA \subseteq AC$. Assume, further, that for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\tilde{k}(\lambda) \ne 0$, we have that the operator $B - \tilde{a}(\lambda)A$ is injective and $C(R(B)) \subseteq R(B - \tilde{a}(\lambda)A)$. Let for each $x \in D(B)$ there is an analytic function $q_x \colon \omega + \Sigma_{\frac{\pi}{2} + \alpha} \to E$ such that

$$q_x(\lambda) = \tilde{k}(\lambda)(B - \tilde{a}(\lambda)A)^{-1}CBx, \quad \operatorname{Re}\lambda > \omega, \ \tilde{k}(\lambda) \neq 0$$

Suppose that, for every $\beta \in (0, \alpha)$ and $p \in \mathfrak{B}$, there exist $c_{p,\beta} > 0$ and $r_{p,\beta} \in \mathfrak{B}$ such that $p((\lambda - \omega)q_x(\lambda)) \leq c_{p,\beta}r_{p,\beta}(x), x \in D(B), \lambda \in \omega + \Sigma_{\beta+(\pi/2)}$ and that, for every $x \in D(B)$, there exists the limit $\lim_{\lambda \to +\infty} \lambda q_x(\lambda)$ in E. Then

there exists an exponentially equicontinuous, analytic (a, k)-regularized C-resolvent family $(R(t))_{t\geq 0}$ for (52), of angle α , and for each $\beta \in (0, \alpha)$ the family $\{e^{-\omega z}R(z) : z \in \Sigma_{\beta}\}$ is equicontinuous.

Subordination principle [292, Theorem 2.4.2] can be reformulated for degenerate (a, k)-regularized C-resolvent families, as well:

THEOREM 2.2.13. Assume that $k_{\beta}(t)$ satisfies (P1), $0 < \alpha < \beta$, $\gamma = \alpha/\beta$ and there exists an exponentially equicontinuous (g_{β}, k_{β}) -regularized C-resolvent family $(R_{\beta}(t))_{t \ge 0}$ for (52), with $a(t) = g_{\beta}(t)$ and $k(t) = k_{\beta}(t)$, satisfying that the family $\{e^{-\omega t}R_{\beta}(t): t \ge 0\}$ is equicontinuous for some $\omega \ge \max(0, \operatorname{abs}(k_{\beta}))$. Assume that there exist a function $k_{\alpha}(t)$ satisfying (P1) and a number $\eta > 0$ such that $k_{\alpha}(0) = k_{\beta}(0)$ and $\tilde{k_{\alpha}}(\lambda) = \lambda^{\gamma-1}\tilde{k_{\beta}}(\lambda^{\gamma}), \lambda > \eta$. Then there exists an exponentially equicontinuous (g_{α}, k_{α}) -regularized C-resolvent family $(R_{\alpha}(t))_{t\ge 0}$ for (52), with $a(t) = g_{\alpha}(t)$ and $k(t) = k_{\alpha}(t)$, satisfying that the family $\{e^{-\omega^{1/\gamma}t}S_{\alpha}(t): t \ge 0\}$ is equicontinuous and

$$R_{\alpha}(t)x = \int_0^{\infty} t^{-\gamma} \Phi_{\gamma}(st^{-\gamma}) R_{\beta}(s) x \, ds, \quad x \in D(B), \ t > 0.$$

Furthermore, for every $\zeta \ge 0$, the equicontinuity of the family $\{e^{-\omega t}(1+t^{\zeta})^{-1}R_{\beta}(t): t \ge 0\}$, resp. $\{e^{-\omega t}t^{-\zeta}R_{\beta}(t): t > 0\}$, implies the equicontinuity of the family $\{e^{-\omega^{1/\gamma}t}(1+t^{\gamma\zeta})^{-1}(1+\omega t^{\zeta(1-\gamma)})^{-1}R_{\alpha}(t): t \ge 0\}$, resp. $\{e^{-\omega^{1/\gamma}t}t^{-\gamma\zeta}(1+\omega t^{\zeta(1-\gamma)})^{-1}R_{\alpha}(t): t > 0\}$, and the following holds:

- (i) The mapping $t \mapsto R_{\alpha}(t)x, t > 0$ admits an analytic extension to the sector $\sum_{\min((\frac{1}{\alpha}-1)\frac{\pi}{2},\pi)}$ for all $x \in D(B)$.
- (ii) If $\omega = 0$ and $\varepsilon \in (0, \min((\frac{1}{\gamma} 1)\frac{\pi}{2}, \pi))$, then the family $\{R_{\alpha}(z) : z \in \Sigma_{\min((\frac{1}{\gamma} 1)\frac{\pi}{2}, \pi) \varepsilon}\}$ is equicontinuous and $\lim_{z \to 0, z \in \Sigma_{\min((\frac{1}{\gamma} 1)\frac{\pi}{2}, \pi) \varepsilon}} R_{\alpha}(z)x = R_{\alpha}(0)x$ for all $x \in D(B)$.
- $\begin{array}{l} R_{\alpha}(z)x = R_{\alpha}(0)x \ for \ all \ x \in D(B).\\ \text{(iii)} \ If \ \omega > 0 \ and \ \varepsilon \in (0, \min((\frac{1}{\gamma} 1)\frac{\pi}{2}, \frac{\pi}{2})), \ then \ there \ exists \ \delta_{\gamma,\varepsilon} > 0 \ such \ that \ the \ family \ \{e^{-\delta_{\gamma,\varepsilon} \operatorname{Re} z} R_{\alpha}(z) \ : \ z \in \Sigma_{\min((\frac{1}{\gamma} 1)\frac{\pi}{2}, \frac{\pi}{2}) \varepsilon}\} \ is \ equicontinuous.\\ Moreover, \ \lim_{z \to 0, z \in \Sigma_{\min((\frac{1}{\gamma} 1)\frac{\pi}{2}, \frac{\pi}{2}) \varepsilon} R_{\alpha}(z)x = R_{\alpha}(0)x \ for \ all \ x \in D(B). \end{array}$

Concerning differential properties of degenerate (a, k)-regularized *C*-resolvent families, the following statements can be verified to be true based on the information provided in the final part of the proof of [**291**, Theorem 3.2.15] (cf. also [**296**, Theorem 3.18], [**296**, Theorem 3.20] with m = 2, the proof of [**291**, Theorem 2.4.8]), and Theorem 2.2.5(i); recall only that for each sequence (M_n) of positive real numbers satisfying $M_0 = 1$, (M.1), (M.2) and (M.3)', we define the function $\omega_L(\cdot)$ by $\omega_L(t) := \sum_{n=0}^{\infty} \frac{t^n}{M_n}$, $t \ge 0$.

THEOREM 2.2.14. Suppose A and B are closed linear operators, |a|(t) and k(t)satisfy (P1), $r \ge -1$ and there exists $\omega \ge \max(0, \operatorname{abs}(|a|), \operatorname{abs}(k))$ such that, for every $z \in \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda > \omega, \tilde{k}(\lambda) \ne 0\}$, the operator $B - \tilde{a}(z)A$ is injective and $C(R(B)) \subseteq R(B - \tilde{a}(z)A)$. Suppose, additionally, that for every $\sigma > 0$ and $x \in D(B)$, there exist a number $c_{\sigma,x} > 0$, an open neighborhood $\Omega_{\sigma,x,\omega}$ of the region

 $\Lambda_{\sigma,x,\omega} := \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda \leqslant \omega, \operatorname{Re} \lambda \geqslant -\sigma \ln |\operatorname{Im} \lambda| + c_{\sigma,x}\} \cup \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda \geqslant \omega\},\$ and an analytic function $h_{\sigma,x} \colon \Omega_{\sigma,x,\omega} \to L(E)$ such that

$$h_{\sigma,x}(\lambda) = \tilde{k}(\lambda)(B - \tilde{a}(\lambda)A)^{-1}CBx, \quad \operatorname{Re}\lambda > \omega, \ \tilde{k}(\lambda) \neq 0$$

and the set

$$\{|\lambda|^{-r}h_{\sigma,x}(\lambda):\lambda\in\Lambda_{\sigma,x,\omega}, \operatorname{Re}\lambda\leqslant\omega\}$$

is bounded. If, for every $\sigma > 0$ and $p \in \circledast$, there exist $c_p > 0$ and $q_p \in \circledast$ such that $p(h_{\sigma,x}(\lambda)) \leq c_p |\lambda|^r q_p(x)$, $\operatorname{Re} \lambda > \omega$, $x \in D(B)$, then, for every $\zeta > 1$, there exists an exponentially equicontinuous $(a, k * g_{\zeta+r})$ -regularized C-resolvent family $(R_{\zeta}(t))_{t\geq 0}$ for (52), satisfying that the mapping $t \mapsto R_{\zeta}(t)x$, t > 0 is infinitely differentiable for all $x \in D(B)$.

THEOREM 2.2.15. (i) Suppose that |a|(t) and k(t) satisfy (P1), there exists an exponentially equicontinuous (a, k)-regularized C-resolvent family $(R(t))_{t \ge 0}$ for (52), satisfying (ii) of Definition 2.2.2 with some number $\omega \ge \max(0, \operatorname{abs}(|a|), \operatorname{abs}(k))$. Let $\omega_0 > \omega$. Denote, for every $x \in D(B)$, $\varepsilon \in (0, 1)$, and a corresponding $K_{\varepsilon,x} > 0$,

$$F_{\varepsilon,\omega_0,x} := \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda \ge -\ln \omega_L(K_{\varepsilon,x}|\operatorname{Im} \lambda|) + \omega_0\}.$$

Assume that, for every $x \in D(B)$ and $\varepsilon \in (0,1)$, there exist $K_{\varepsilon,x} > 0$, an open neighborhood $O_{\varepsilon,\omega_0,x}$ of the region $G_{\varepsilon,\omega_0,x} := \{\lambda \in \mathbb{C} : Re\lambda \ge \omega_0, \tilde{k}(\lambda) \neq 0\} \cup \{\lambda \in F_{\varepsilon,\omega_0} : Re\lambda \le \omega_0\}$ and the analytic mappings $h_{\varepsilon,x} : O_{\varepsilon,\omega_0,x} \to E$, $f_{\varepsilon,x} : O_{\varepsilon,\omega_0,x} \to \mathbb{C}$, $g_{\varepsilon,x} : O_{\varepsilon,\omega_0,x} \to \mathbb{C}$ such that:

- (a) $f_{\varepsilon,x}(\lambda) = \tilde{k}(\lambda)$, $\operatorname{Re} \lambda > \omega_0$; $g_{\varepsilon,x}(\lambda) = \tilde{a}(\lambda)$, $\operatorname{Re} \lambda > \omega_0$,
- (b) for every $\lambda \in F_{\varepsilon,\omega_0,x}$, the operator $B g_{\varepsilon,x}(\lambda)A$ is injective and $C(R(B)) \subseteq R(B g_{\varepsilon,x}(\lambda)A)$,
- (c) for every $x \in D(B)$, $h_{\varepsilon,x}(\lambda) = f_{\varepsilon,x}(\lambda)(B g_{\varepsilon,x}(\lambda)A)^{-1}CBx$, $\lambda \in G_{\varepsilon,\omega_0,x}$,
- (d) the set $\{(1 + |\lambda|)^{-m} e^{-\varepsilon |\operatorname{Re} \lambda|} h_{\varepsilon,x}(\lambda) : \lambda \in F_{\varepsilon,\omega}, \operatorname{Re} \lambda \leq \omega_0\}$ is bounded.

Then, for every $x \in D(B)$, the mapping $t \mapsto R(t)x$, t > 0 is infinitely differentiable and, for every compact set $K \subseteq (0, \infty)$, there exists $h_K > 0$ such that the set $\{\frac{h_K^n \frac{d^n}{dtn} R(t)x}{M_n} : t \in K, n \in \mathbb{N}_0\}$ is bounded; furthermore, if $K_{\varepsilon,x}$ is independent of $x \in D(B)$ and if for each $p \in \circledast$ there exist $c_p > 0$ and $q_p \in \circledast$ such that $p((1+|\lambda|)^{-m}e^{-\varepsilon|\operatorname{Re}\lambda|}h_{\varepsilon,x}(\lambda)) \leq c_p q_p(x), x \in D(B),$ $\lambda \in F_{\varepsilon,\omega_0}$, $\operatorname{Re}\lambda \leq \omega_0$, then the family $\{\frac{h_K^n \frac{d^n}{dtn} R(t)}{M_n} : t \in K, n \in \mathbb{N}_0\}$ is equicontinuous.

(ii) Suppose that |a|(t) and k(t) satisfy (P1), there exists an exponentially equicontinuous (a, k)-regularized C-resolvent family $(R(t))_{t\geq 0}$ for (52), satisfying (ii) of Definition 2.2.2 with $\omega \geq \max(0, \operatorname{abs}(|a|), \operatorname{abs}(k))$. Let $\omega_0 > \omega$. Denote, for every $x \in D(B)$, $\varepsilon \in (0, 1)$, $\rho \in [1, \infty)$ and a corresponding $K_{\varepsilon,x} > 0$,

$$F_{\varepsilon,\omega_0,\rho,x} := \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda \ge -K_{\varepsilon,x} | \operatorname{Im} \lambda |^{1/\rho} + \omega_0 \}.$$

Assume that, for every $x \in D(B)$ and $\varepsilon \in (0,1)$, there exist $K_{\varepsilon,x} > 0$, an open neighborhood $O_{\varepsilon,\omega_0,x}$ of the region $G_{\varepsilon,\omega_0,\rho,x} := \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda \geq \omega_0, \tilde{k}(\lambda) \neq 0\} \cup \{\lambda \in F_{\varepsilon,\omega_0,\rho,x} : \operatorname{Re} \lambda \leq \omega_0\}$, and analytic mappings $h_{\varepsilon,x} : O_{\varepsilon,\omega_0,x} \to E$, $f_{\varepsilon,x} : O_{\varepsilon,\omega_0,x} \to \mathbb{C}$ and $g_{\varepsilon} : O_{\varepsilon,\omega_0,x} \to \mathbb{C}$ such that the conditions (i)(a)–(d) of this theorem hold with $F_{\varepsilon,\omega_0,x}$, resp. $G_{\varepsilon,\omega_0,x}$, replaced by $F_{\varepsilon,\omega_0,\rho,x}$, resp. $G_{\varepsilon,\omega_0,\rho,x}$. Then, for every $x \in D(B)$, the mapping $t \mapsto R(t), t > 0$ is infinitely differentiable and, for every compact set $K \subseteq (0,\infty)$, there exists $h_K > 0$ such that the set $\{\frac{h_K^n \frac{dn}{dtn} R(t)x}{n!^{\rho}} : t \in K, n \in \mathbb{N}_0\}$ is equicontinuous; furthermore, if $K_{\varepsilon,x}$ is independent of $x \in D(B)$ and if for each $p \in \circledast$ there exist $c_p > 0$ and $q_p \in \circledast$ such that $p((1 + |\lambda|)^{-m} e^{-\varepsilon |\operatorname{Re}\lambda|} h_{\varepsilon,x}(\lambda)) \leq c_p q_p(x), x \in D(B), \lambda \in F_{\varepsilon,\omega_0,\rho,x}$, $\operatorname{Re}\lambda \leq \omega_0$, then the family $\{\frac{h_K^n \frac{dn}{dtn} R(t)}{n!^{\rho}} : t \in K, n \in \mathbb{N}_0\}$ is equicontinuous.

THEOREM 2.2.16 (The abstract Weierstrass formula). (i) Assume that a(t) and k(t) satisfy (P1), and there exist M > 0 and $\omega > 0$ such that $|k(t)| \leq Me^{\omega t}, t \geq 0$. Assume, further, that there exist a number $\omega' \geq \omega$ and a function $a_1(t)$ such that $abs(a_1) < \infty$ and $\tilde{a_1}(\lambda) = \tilde{a}(\sqrt{\lambda})$, $\operatorname{Re} \lambda > \omega'$ (Let us recall that the above holds if a(t) is exponentially bounded; in this case, $a_1(t) = \int_0^\infty s \frac{e^{-s^2/4t}}{2\sqrt{\pi t^{3/2}}} a(s) ds, t > 0$.) Let there exist an exponentially equicontinuous (a, k)-regularized C-resolvent family $(C(t))_{t\geq 0}$ for (52). Then there exists an exponentially equicontinuous, analytic (a_1, k_1) regularized C-resolvent family $(R(t))_{t\geq 0}$ for (52), with a(t) replaced by $a_1(t)$, of angle $\frac{\pi}{2}$, where:

(62)
$$k_1(t) := \int_0^\infty \frac{e^{-s^2/4t}}{\sqrt{\pi t}} k(s) ds, \quad t > 0, \ k_1(0) := k(0), \ and$$

(63)
$$R(t)x := \int_0^\infty \frac{e^{-s^2/4t}}{\sqrt{\pi t}} C(s)x \, ds, \quad t > 0, \ x \in D(B), \ R(0)x := C(0)x, \ x \in D(B).$$

(ii) Assume that k(t) satisfy (P1), β > 0 aw well as there exist M > 0 and ω > 0 such that |k(t)| ≤ Me^{ωt}, t ≥ 0. Let there exist an exponentially equicontinuous (g_{2β}, k)-regularized C-resolvent family (C(t))_{t≥0} for (52). Then there exists an exponentially equicontinuous, analytic (g_β, k₁)-regularized C-resolvent family (R(t))_{t≥0} for (52), with a(t) replaced by a₁(t), of angle π/2, where k₁(t) and R(t) are defined through (62)-(63).

REMARK 2.2.17. As Theorem 2.2.8 shows, the existence of an exponentially equicontinuous (a, k)-regularized *C*-resolvent family $(R(t))_{t\geq 0}$ for (52) does not automatically imply the existence of mild (strong) solutions of this problem; we need to impose the conditions like (i.1), (i.1)' or (ii.1). Regarding this question, the following facts should be stated:

- 1. Proposition 2.2.4(i): If the function b(t) satisfies the prescribed assumptions and $(R(t))_{t\geq 0}$ satisfies any of the conditions (i.1), (i.1)' or (ii.1), then $((b * R)(t))_{t\geq 0}$ satisfies the same condition as well, with the function k(t) replaced by (b * k)(t); Proposition 2.2.4(iii): If the functions k(t) and b(t) satisfy the prescribed assumptions and $(R(t))_{t\geq 0}$ satisfies any of the conditions (i.1), (i.1)' or (ii.1), then $(S(t))_{t\geq 0}$ satisfies the same condition as well, with the function k(t) replaced by 1.
- 2. In the formulations of Theorem 2.2.5(i)–(ii), as well as Theorem 2.2.12 and Theorems 2.2.14-2.2.15, we must add some very natural conditions ensuring the validity of (i.1), (i.1)' or (ii.1) for the corresponding resolvent families.
- 3. The conditions (i.1) and (ii.1) are invariant under the action of subordination principles stated in Theorem 2.2.6, while some additional assumptions must be imposed for the invariance of the condition (i.1)'.
- 4. The rescaling of degenerate K-convoluted C-semigroups (cf. Proposition 2.2.10) preserves the conditions (i.1), (i.1)' and (ii.1).
- 5. Due to the proof of [61, Theorem 3.1], it is not difficult to verify that the conditions (i.1), (i.1)' and (ii.1) are invariant under the action of subordination principles stated in Theorem 2.2.13 and Theorem 2.2.16.

Now we want to illustrate our results with a concrete example.

EXAMPLE 2.2.18. (cf. [291, Example 2.8.3(iii)] and [292, Example 2.6.10]) Let s > 1,

$$E := \left\{ f \in C^{\infty}[0,1] \; ; \; \|f\| := \sup_{p \ge 0} \frac{\|f^{(p)}\|_{\infty}}{p!^s} < \infty \right\}$$

and

$$A := -d/ds, \ D(A) := \{ f \in E \ ; \ f' \in E, \ f(0) = 0 \}.$$

If $f \in E$, $t \in [0,1]$ and $\lambda \in \mathbb{C}$, set $f_{\lambda}^{1}(t) := \int_{0}^{t} e^{-\lambda(t-s)} f(s) ds$ and $f_{\lambda}^{2}(t) := \int_{0}^{t} e^{\lambda(t-s)} f(s) ds$. Then $f_{\lambda}^{1}(\cdot)$ and $f_{\lambda}^{2}(\cdot) \in E$, $\lambda \in \mathbb{C}$; moreover, there exist b' > 0 and $M \ge 1$, independent of $f(\cdot)$, such that

(64)
$$||f_{\lambda}^{1}(\cdot)|| \leq M ||f|| e^{b'|\lambda|^{1/s}}, \quad \operatorname{Re} \lambda \geq 0, \ f \in E.$$

Furthermore, for each $\eta > 1$ there exists $M_{\eta} \ge 1$, independent of $f(\cdot)$, such that

(65)
$$||f_{\lambda}^{2}(\cdot)|| \leq M_{\eta}||f||e^{\eta|\lambda|}, \quad \operatorname{Re} \lambda \geq 0, \ f \in E.$$

Let $P_1(z) = \sum_{j=0}^{N_1} a_{j,1} z^j$, $z \in \mathbb{C}$, $a_{N_1,1} \neq 0$ be a complex non-zero polynomial, and let $P_2(z) = \sum_{j=0}^{N_2} a_{j,2} z^j$, $z \in \mathbb{C}$, $a_{N_2,2} \neq 0$ be a complex non-zero polynomial so that $N_1 = dg(P_1) > 1 + dg(P_2) = 1 + N_2$. For any complex non-zero polynomial P(z), we define the operator P(A) in the obvious way; then

(66)
$$\rho(P(A)) = \mathbb{C} \text{ and } R(\lambda : A)f = f_{\lambda}^{1}, \quad f \in E, \ \lambda \in \mathbb{C}.$$

Set $P_{\lambda}(z) := \lambda P_2(z) - P_1(z), z \in \mathbb{C} \ (\lambda \in \mathbb{C})$. Let $\{z_1, \ldots, z_s\}$ be the set which consists of joint multiple roots of polynomials $P_1(z)$ and $P_2(z)$. Then there exist

uniquely determined integers $k_1, \ldots, k_s \in \mathbb{N}$ such that

$$P_{\lambda}(z) = (z - z_1)^{k_1} \dots (z - z_s)^{k_s} (\lambda Q_2(z) - Q_1(z)), \quad z \in \mathbb{C}, \ \lambda \in \mathbb{C},$$

with $Q_1(z)$ and $Q_2(z)$ being two non-zero complex polynomials without joint multiple roots, satisfying additionally that $dg(Q_1) > dg(Q_2) + 1$. This implies that there exists d > 0 such that, for every $\lambda \in \mathbb{C}$ with $|\lambda| \ge d$, the polynomial $Q_{\lambda}(z) := \lambda Q_2(z) - Q_1(z)$ is square-free. Denote by $z_{1,\lambda}, \ldots, z_{N_1,\lambda}$ the roots of $P_{\lambda}(z)$ ($\lambda \in \mathbb{C}$). Using [252, Corollary 5.6] (this is an elementary result on root localization of complex polynomials), we get the existence of a positive real number $\vartheta \ge 1$ such that

(67)
$$|z_{i,\lambda}| \leqslant \vartheta (1+|\lambda|)^{\frac{1}{N_1-N_2}}, \quad 1 \leqslant i \leqslant N_1 \ (\lambda \in \mathbb{C}).$$

It is quite easy to prove that the operator $\lambda P_2(A) - P_1(A)$ has the bounded inverse for all $\lambda \in \mathbb{C}$, as well as that

(68)
$$(\lambda P_2(A) - P_1(A))^{-1} = (-1)^{N_1+1} a_{N_1,1}^{-1} R(z_{1,\lambda} : A) \dots R(z_{N_1,\lambda} : A), \quad \lambda \in \mathbb{C}.$$

Observe further that, for every $\lambda \in \mathbb{C}$ with $|\lambda| \geq d$, the discriminant of polynomial $Q_{\lambda}(z)$, for which it is well known that can be represented by a homogenous polynomial of degree $2(dg(Q_1) - 1)$ in the coefficients of $Q_{\lambda}(z)$, is a complex non-zero polynomial in λ . Hence, there exist numbers $d_1 \geq d$ and $\eta > 0$ such that $|D(Q_{\lambda}(z))| \geq \eta, |\lambda| \geq d_1$. Making use of this fact and [467, Theorem 1], we obtain the existence of a sufficiently small number $\zeta > 0$ such that, for every $\lambda \in \mathbb{C}$ with $|\lambda| \geq d_1$, and for every two distinct roots $z_{i,\lambda}, z_{j,\lambda}$ of polynomial $Q_{\lambda}(z)$, we have $|z_{i,\lambda} - z_{j,\lambda}| \geq \zeta$. Then the calculation contained in the analysis made in [292, Example 2.6.10], combined with the equality (68) and the above fact, shows that the norm of operator $(\lambda P_2(A) - P_1(A))^{-1}$ does not exceed $M \sum ||R(z_{j,\lambda} : A)||$, where the summation is taken over all roots $z_{j,\lambda}$ of polynomial $Q_{\lambda}(z)$. Taken together with (64)–(67) and the generalized resolvent equation, the above implies that there exist numbers b > 0, c > 0 and $\zeta > 0$ such that

$$\|(\lambda P_2(A) - P_1(A))^{-1}\| = O(e^{b|\lambda|^{1/(N_1 - N_2)s} + c|\lambda|^{1/(N_1 - N_2)}}), \quad \lambda \in \mathbb{C}.$$

and

(69)
$$\|(\lambda P_2(A) - P_1(A))^{-1} P_2(A) f\| \leq \zeta \|f\| e^{b|\lambda|^{1/(N_1 - N_2)s} + c|\lambda|^{1/(N_1 - N_2)s}}$$

for all $\lambda \in \mathbb{C}$ and $f \in D(P_2(A))$. Before proceeding further, it should be noted that the above estimates can be used in proving the existence of convoluted solutions of fractional analogues of the linearized Benney–Luke equation (sometimes also called Dzektser equation)

$$(\lambda - \Delta)u_t = \alpha \Delta - \beta \Delta^2 \quad (\alpha, \beta > 0, \ \lambda \in \mathbb{R}),$$

in contrast with the assertions of Theorem 2.2.20–Theorem 2.2.21 below, which can be applied only in the case that $\lambda > 0$ (cf. [**307**] for more details); as is well known, this equation is important in evolution modeling of some problems appearing in the theory of liquid filtration, see e.g. [**199**, p. 6]. Suppose $N_1 - N_2 > \alpha \ge 1$, $\delta \in (0, \pi/2], (\pi/2 + \delta)\alpha/(N_1 - N_2) < \pi/2, \rho > c/\cos((\pi/2 + \delta)\alpha/(N_1 - N_2))$ and $k(t) = \mathcal{L}^{-1}(e^{-\rho\lambda^{\alpha/n}})(t), t \ge 0$. By Theorem 2.2.12 and (69), there exists an exponentially bounded, analytic (g_{α}, k) -regularized resolvent family $(R_{\alpha}(t))_{t \geq 0}$ for the corresponding problem (52), of angle δ ; it is clear that the conditions (i.1)' and (ii.1) stated in Theorem 2.2.8 and Remark 2.2.9(i) holds for $(R_{\alpha}(t))_{t\geq 0}$. Observe finally that the case $N_1 = N_2 + 1$ is critical and that we always have the existence of integrated solution families in the case $N_2 \geq N_1$ (this follows from the above analysis and the fact that, for every $\lambda \in \mathbb{C}$ with $|\lambda| \geq d$, the roots $z_{1,\lambda}, \ldots, z_{N_2,\lambda}$ of polynomial $P_{\lambda}(z)$ belong to a compact set $K \subseteq \mathbb{C}$ which does not depend on λ ; see e.g. [252, Theorem 5.4]). Computing the optimal rate of integration is a non-trivial problem which will not be discussed here.

2.2.3. Degenerate time-fractional equations associated with abstract differential operators. With the exceptions of Remark 2.2.23 and Remark 2.2.26(ii), we assume in this subsection, and the next one, that $n \in \mathbb{N}$ and $iA_j, 1 \leq j \leq n$ are commuting generators of bounded C_0 -groups on a Banach space E. Denote by $\mathcal{S}(\mathbb{R}^n)$ the Schwartz space of rapidly decreasing functions on \mathbb{R}^n . Put $k := 1 + \lfloor n/2 \rfloor$, $A := (A_1, \ldots, A_n)$ and $A^\eta := A_1^{\eta_1} \ldots A_n^{\eta_n}$ for any $\eta = (\eta_1, \ldots, \eta_n) \in \mathbb{N}_0^n$. Denote by \mathcal{F} and \mathcal{F}^{-1} the Fourier transform on \mathbb{R}^n and its inverse transform, respectively. For every $\xi = (\xi_1, \ldots, \xi_n) \in \mathbb{R}^n$ and $u \in \mathcal{F}L^1(\mathbb{R}^n) = \{\mathcal{F}f : f \in L^1(\mathbb{R}^n)\}$, we set $|\xi| := (\sum_{j=1}^n \xi_j^2)^{1/2}$, $(\xi, A) := \sum_{j=1}^n \xi_j A_j$ and

$$u(A)x := \int_{\mathbb{R}^n} \mathcal{F}^{-1}u(\xi)e^{-i(\xi,A)}x\,d\xi, \quad x \in E.$$

Then $u(A) \in L(E)$, $u \in \mathcal{F}L^1(\mathbb{R}^n)$ and there exists a finite constant $M \ge 1$ such that

 $||u(A)|| \leq M ||\mathcal{F}^{-1}u||_{L^1(\mathbb{R}^n)}, \quad u \in \mathcal{F}L^1(\mathbb{R}^n).$

Let $N \in \mathbb{N}$, and let $P(x) = \sum_{|\eta| \leq N} a_{\eta} x^{\eta}$, $x \in \mathbb{R}^n$ be a complex polynomial. Then we define $P(A) := \sum_{|\eta| \leq N} a_{\eta} A^{\eta}$ and $E_0 := \{\phi(A)x : \phi \in \mathcal{S}(\mathbb{R}^n), x \in E\}$. We know that the operator P(A) is closable and the following holds (cf. [292,375,561] and [304] for further information):

$$(\triangleright) \ \overline{E_0} = E, \ E_0 \subseteq \bigcap_{\eta \in \mathbb{N}_0^n} D(A^\eta), \ P(A)|_{E_0} = P(A) \text{ and } \\ \phi(A)P(A) \subseteq P(A)\phi(A) = (\phi P)(A), \ \phi \in \mathcal{S}(\mathbb{R}^n).$$

Assuming that E is a function space on which translations are uniformly bounded and strongly continuous, the obvious choice for A_j is $-i\partial/\partial x_j$ (notice also that Ecan be consisted of functions defined on some bounded domain [132,375,561], [560, pp. 101-103]). If $P(x) = \sum_{|\eta| \leq N} a_{\eta} x^{\eta}$, $x \in \mathbb{R}^n$ and E is such a space (for example, $L^p(\mathbb{R}^n)$ with $p \in [1, \infty)$, $C_0(\mathbb{R}^n)$ or $BUC(\mathbb{R}^n)$), then $\overline{P(A)}$ is nothing else but the operator $\sum_{|\eta| \leq N} a_{\eta}(-i)^{|\eta|} \partial |x_1^{\eta_1} \dots \partial x_1^{\eta_n} \equiv \sum_{|\eta| \leq N} a_{\eta} D^{\eta}$, acting with its maximal distributional domain. Recall that P(x) is called *r*-coercive ($0 < r \leq N$) if there exist M, L > 0 such that $|P(x)| \geq M|x|^r$, $|x| \geq L$; by a corollary of the Seidenberg–Tarski theorem, the equality $\lim_{|x|\to\infty} |P(x)| = \infty$ implies in particular that P(x) is *r*-coercive for some $r \in (0, N]$ (cf. [27, Remark 8.2.7]). In the sequel of this subsection, M > 0 denotes a generic constant whose value may change from line to line. Let $p \in [1, \infty]$. Following L. Hörmander [257], it will be said that a function $u \in L^{\infty}(\mathbb{R}^n)$ is a Fourier multiplier on $L^p(\mathbb{R}^n)$ iff $\mathcal{F}^{-1}(u\mathcal{F}\phi) \in L^p(\mathbb{R}^n)$ for all $\phi \in \mathcal{S}(\mathbb{R}^n)$ and

$$||u||_{\mathcal{M}_p} := \sup\{||\mathcal{F}^{-1}(u\mathcal{F}\phi)||_{L^p(\mathbb{R}^n)} : \phi \in \mathcal{S}(\mathbb{R}^n), \ ||\phi||_{L^p(\mathbb{R}^n)} \leqslant 1\} < \infty.$$

We use the abbreviation \mathcal{M}_p for the space of all Fourier multipliers on $L^p(\mathbb{R}^n)$; cf. [254] for more details. Then \mathcal{M}_p is a Banach algebra under pointwise multiplication and $\mathcal{F}L^1(\mathbb{R}^n)$ is continuously embedded in \mathcal{M}_p . We need the following lemma (see e.g. [541, Lemma 5.2, Lemma 5.4, pp. 20-22]).

LEMMA 2.2.19. (i) Let $1 \leq p \leq \infty$, $j, n \in \mathbb{N}$, j > n/2 and $\{f_t\}_{t \geq 0}$ be a family of $C^j(\mathbb{R}^n)$ -functions. Assume that for each $x \in \mathbb{R}^n$, $\eta \in \mathbb{N}_0^n$ with $|\eta| \leq j, t \mapsto D^{\alpha} f_t(x), t \geq 0$ is continuous and that there exist $a > 0, r > n|\frac{1}{p} - \frac{1}{2}|$ and $M_t > 0$ (M_t is bounded on compacts of $t \geq 0$) such that

$$|D^{\eta}f_t(x)| \leq M_t^{|\eta|} (1+|x|)^{(a-1)|\eta|-ar}, \quad |\eta| \leq j, \ x \in \mathbb{R}^n, \ t \ge 0.$$

Then, for any $t \ge 0$, p = 1, ∞ (resp. $1), we have <math>f_t \in \mathcal{F}L^1(\mathbb{R}^n)$ (resp. $f_t \in \mathcal{M}_p$), $t \mapsto f_t$, $t \ge 0$ is continuous with respect to $|| \cdot ||_{\mathcal{F}L^1(\mathbb{R}^n)}$ (resp. $|| \cdot ||_{\mathcal{M}_p}$) and there exists a constant M > 0 independent of $t \ge 0$ such that

$$||f_t||_{\mathcal{F}L^1(\mathbb{R}^n)}$$
 (resp. $||f_t||_{\mathcal{M}_p}) \leq MM_t^{n|\frac{1}{p}-\frac{1}{2}|}, \quad t \ge 0.$

(ii) Let $1 , <math>j, n \in \mathbb{N}$, j > n/2 and $f \in C^j(\mathbb{R}^n)$. Assume that there exist $a \ge 0$, $r \ge n |\frac{1}{n} - \frac{1}{2}|$, $M_f \ge 1$ and $L_f > 0$ such that

$$|D^{\eta}f(x)| \leq L_f M_f^{|\eta|} (1+|x|)^{(a-1)|\eta|-ar}, \quad |\eta| \leq j, \ x \in \mathbb{R}^n, \ t \ge 0.$$

Then $f \in \mathcal{M}_p$ and there exists a constant M > 0 independent of $f(\cdot)$ such that

$$\|f\|_{\mathcal{M}_p} \leqslant ML_f M_f^{n|\frac{1}{p}-\frac{1}{2}|}.$$

Suppose now that $P_1(x)$ and $P_2(x)$ are non-zero complex polynomials in n variables and $0 < \alpha \leq 2$; put $N_1 := dg(P_1(x)), N_2 := dg(P_2(x))$ and $m := \lceil \alpha \rceil$. We investigate the generation of some very specific classes of (g_α, C) -regularized resolvent families associated with the following fractional degenerate abstract Cauchy problem

$$(\text{DFP}): \begin{cases} \mathbf{D}_t^{\alpha} \overline{P_2(A)} u(t) = \overline{P_1(A)} u(t), & t \ge 0, \\ u(0) = Cx; & u^{(j)}(0) = 0, \ 1 \le j \le \lceil \alpha \rceil - 1. \end{cases}$$

Convoluting both sides of (DFP) with $g_{\alpha}(t)$, and using the equality (38) it follows that every solution of (DFP) is, in fact, a strong solution of problem (52) with $B = \overline{P_2(A)}, \tau = \infty, a(t) = g_{\alpha}(t)$ and $f(t) \equiv \overline{P_2(A)}Cx$. It can be easily checked that any such a solution of problem (52) is also a strong solution of problem (DFP); cf. Definition 2.3.1 below. A continuous *E*-valued function $t \mapsto u(t), t \ge 0$ is said to be a mild solution of (DFP) iff $\overline{P_1(A)}(g_{\alpha} * u)(t) = \overline{P_2(A)}u(t) - \overline{P_2(A)}Cx, t \ge 0$. We start by stating the following extension of [**304**, Theorem 2.1]; observe only that we do not assume here the coercivity of $P_1(x)$ or $P_2(x)$, and that in the formulation of the afore-mentioned theorem we have that $P_1(x) = Q(x) = P(x)$ and $P_2(x) = 1$ $(x \in \mathbb{R}^n)$.

THEOREM 2.2.20. Suppose $0 < \alpha < 2$, $\omega \ge 0$, $P_1(x)$ and $P_2(x)$ are non-zero complex polynomials $N_1 = dg(P_1(x))$, $N_2 = dg(P_2(x))$, $N \in \mathbb{N}$ and $r \in (0, N]$. Let Q(x) be an r-coercive complex polynomial of degree N, $a \in \mathbb{C} \setminus Q(\mathbb{R}^n)$, $\gamma > \frac{n \max(N, \frac{N_1+N_2}{\min(1,\alpha)})}{2r}$ (resp. $\gamma = \frac{n}{r} |\frac{1}{p} - \frac{1}{2}| \max(N, \frac{N_1+N_2}{\min(1,\alpha)})$, if $E = L^p(\mathbb{R}^n)$ for some $1), <math>P_2(x) \neq 0$, $x \in \mathbb{R}^n$ and

(70)
$$\sup_{x \in \mathbb{R}^n} \operatorname{Re}\left(\left(\frac{P_1(x)}{P_2(x)}\right)^{1/\alpha}\right) \leqslant \omega.$$

Set

$$R_{\alpha}(t) := \left(E_{\alpha} \left(t^{\alpha} \frac{P_1(x)}{P_2(x)} \right) (a - Q(x))^{-\gamma} \right) (A), \quad t \ge 0.$$

Then $(R_{\alpha}(t))_{t \ge 0} \subseteq L(E)$ is a global exponentially bounded $(g_{\alpha}, R_{\alpha}(0))$ -regularized resolvent family for (DFP), $(R_{\alpha}(t))_{t \ge 0}$ is norm continuous provided $\gamma > \frac{n \max(N, \frac{N_1 + N_2}{\min(1, \alpha)})}{2r}$, and the following holds: (71) $||R_{\alpha}(t)|| \le M(1 + t^{\max(1, \alpha)n/2})e^{\omega t}, \quad t \ge 0, \text{ resp.},$ $||R_{\alpha}(t)|| \le M(1 + t^{\max(1, \alpha)n|\frac{1}{p} - \frac{1}{2}|})e^{\omega t}, \quad t \ge 0.$

PROOF. Put $C := R_{\alpha}(0)$. Then C is injective and it can be easily proved with the help of (>) that $\overline{CP(A)} \subseteq \overline{P(A)}C$ for any complex polynomial P(x); see e.g. [304]. Furthermore, $\sup_{x \in \mathbb{R}^n} |P_2(x)|^{-1} < \infty$ and, for every multi-index $\eta \in \mathbb{N}_0^n$ with $|\eta| > 0$, there exists $c_{\eta} > 0$ such that

(72)
$$\left| D^{\eta} \left(\frac{P_1(x)}{P_2(x)} \right) \right| \leq c_{\eta} (1+|x|)^{|\eta|(N_1+N_2-1)}, \quad x \in \mathbb{R}^n.$$

By induction, one can prove that, for every multi-index $\eta \in \mathbb{N}_0^n$ with $|\eta| > 0$, the following holds:

(73)
$$D^{\eta} E_{\alpha} \left(t^{\alpha} \frac{P_{1}(x)}{P_{2}(x)} \right) = \sum_{j=1}^{|\eta|} t^{\alpha j} E_{\alpha}^{(j)} \left(t^{\alpha} \frac{P_{1}(x)}{P_{2}(x)} \right) R_{\eta,j}(x), \quad t \ge 0, \ x \in \mathbb{R}^{n},$$

where $R_{\eta,j}(x)$ can be represented as a finite sum of terms like $\prod_{w=1}^{s_j} D^{\eta_{j,w}}(\frac{P_1(x)}{P_2(x)})$ with $|\eta_{j,w}| > 0$ $(1 \le w \le s_j)$ and $|\eta_{j,1}| + \cdots + |\eta_{j,s_j}| \le |\eta|$. Consider now the assertion of Theorem 1.5.1. Taking the number $\sigma > 0$ sufficiently small, and keeping in mind that $0 < \alpha < 2$, we obtain that, for every $m \in \mathbb{N} \setminus \{1\}$ and for every $t \ge 0, x \in \mathbb{R}^n$ with $|t^{\alpha}P_1(x)/P_2(x)| \ge 1$, the term

$$\begin{aligned} \left| E_{\alpha,\alpha j - (j-l)} \left(t^{\alpha} \frac{P_1(x)}{P_2(x)} \right) &- \frac{1}{\alpha} \left(t^{\alpha} \frac{P_1(x)}{P_2(x)} \right)^{(1 - (\alpha j - (j-l)))/\alpha} e^{(t^{\alpha} P_1(x)/P_2(x))^{1/\alpha}} \\ &- \sum_{j=1}^{m-1} \frac{(t^{\alpha} P_1(x)/P_2(x))^{-j}}{\Gamma(\alpha j - (j-l) - \alpha j)} \right| \end{aligned}$$

can be majorized by $M|t^{\alpha}P_1(x)/P_2(x)|^{-m}$. Clearly, the function $E_{\alpha,\alpha j-(j-l)}(\cdot)$ is bounded on compacts of \mathbb{C} , and (70) implies that $\operatorname{Re}((t^{\alpha}(P_1(x)/P_2(x))^{1/\alpha}) \leq \omega t, t \geq 0, x \in \mathbb{R}^n$. As in the proof of [**296**, Theorem 2.1], the above implies that, for every $t \geq 0, x \in \mathbb{R}^n$ and $1 \leq l \leq j \leq k$,

(74)
$$|E_{\alpha,\alpha j-(j-l)}(t^{\alpha}P_1(x)/P_2(x))| \leq M \left[1 + t^{1-(\alpha j-(j-l))}|P_1(x)/P_2(x)|^{\frac{1-(\alpha j-(j-l))}{\alpha}}e^{\omega t}\right]$$

By (44), (72)–(73) and the boundedness of derivatives of the Mittag-Leffler function $E_{\alpha}(\cdot)$ on compacts of \mathbb{C} , we obtain that, for every $t \ge 0$ and $x \in \mathbb{R}^n$ with $|t^{\alpha}P_1(x)/P_2(x)| \le 1$:

(75)
$$\left| D^{\eta} E_{\alpha} \left(t^{\alpha} \frac{P_1(x)}{P_2(x)} \right) \right| \leq M (t^{\alpha} + t^{\alpha |\eta|}) (1 + |x|)^{|\eta|(N_1 + N_2 - 1)}, \quad 0 < |\eta| \leq k.$$

If $0 < |\eta| \leq k, t \geq 0, x \in \mathbb{R}^n$ and $|t^{\alpha}P_1(x)/P_2(x)| \geq 1$, then the following holds (cf. (74) and [**304**, (2.6)–(2.7)]):

$$\begin{split} \left| D^{\eta} E_{\alpha} \left(t^{\alpha} \frac{P_{1}(x)}{P_{2}(x)} \right) \right| \\ &\leqslant M \sum_{j=1}^{|\eta|} t^{\alpha j} \sum_{l=1}^{j} \left[1 + |t^{\alpha} P_{1}(x) / P_{2}(x)|^{\frac{1 - (\alpha j - (j - l))}{\alpha}} e^{\omega t} \right] (1 + |x|)^{|\eta|(N_{1} + N_{2} - 1)} \\ &\leqslant M \sum_{j=1}^{|\eta|} t^{\alpha j} \sum_{l=1}^{j} [1 + e^{\omega t} (1 + t^{1 - (\alpha j - (j - l))})] (1 + |x|)^{|\eta|(\frac{N_{1} + N_{2}}{\min(1, \alpha)} - 1)} \\ &\leqslant M (1 + t^{\max(1, \alpha)|\eta|}) e^{\omega t} (1 + |x|)^{|\eta|(\frac{N_{1} + N_{2}}{\min(1, \alpha)} - 1)}. \end{split}$$

Taking into account (75), we obtain from the previous estimate that, for every $t \ge 0$ and $x \in \mathbb{R}^n$,

(76)
$$\left| D^{\eta} E_{\alpha} \left(t^{\alpha} \frac{P_1(x)}{P_2(x)} \right) \right| \leq M (1 + t^{\max(1,\alpha)|\eta|}) e^{\omega t} (1 + |x|)^{|\eta| (\frac{N_1 + N_2}{\min(1,\alpha)} - 1)}, \ 0 < |\eta| \leq k;$$

observe that the inequality $\operatorname{Re}((t^{\alpha}P_1(x)/P_2(x))^{1/\alpha}) \leq \omega t, t \geq 0, x \in \mathbb{R}^n$ and Theorem 1.5.1 together imply that the previous estimate also holds in the case that $|\eta| = 0$. Set $f_t(x) := E_{\alpha}(t^{\alpha}P_1(x)/P_2(x))(a - Q(x))^{-\gamma}, t \geq 0, x \in \mathbb{R}^n$. It is clear that there exists L > 0 such that $|Q(x)| \geq M|x|^r, |x| \geq L$ and $|a - Q(x)| \geq M|x|^r,$ $|x| \geq L$. Using [**380**, (3.19)], (76) and the product rule, it readily follows that, for every $t \geq 0, x \in \mathbb{R}^n$, and for every $\eta \in \mathbb{N}_0^n$ with $|\eta| \leq k$,

(77)
$$\left| D^{\eta} \left(E_{\alpha} \left(t^{\alpha} \frac{P_{1}(x)}{P_{2}(x)} \right) (a - Q(x))^{-\gamma} \right) \right|$$

$$\leq M (1 + t^{\max(1,\alpha)|\eta|}) e^{\omega t} (1 + |x|)^{|\eta| (\max(N, \frac{N_{1} + N_{2}}{\min(1,\alpha)}) - 1) - r\gamma}.$$

We obtain similarly that, for every $t \ge 0$, $x \in \mathbb{R}^n$, and for every $\eta \in \mathbb{N}_0^n$ with $|\eta| \le k$,

(78)
$$\left| D^{\eta} \left(P_2(x)^{-1} E_{\alpha} \left(t^{\alpha} \frac{P_1(x)}{P_2(x)} \right) (a - Q(x))^{-\gamma} \right) \right|$$

$$\leq M (1 + t^{\max(1,\alpha)|\eta|}) e^{\omega t} (1 + |x|)^{|\eta| (\max(N, \frac{N_1 + N_2}{\min(1,\alpha)}) - 1) - r\gamma}.$$

Suppose first that $\gamma > \frac{n \max(N, \frac{N_1+N_2}{\min(1,\alpha)})}{2r}$. By performing the Laplace transform and using (43) we get that for each $t \ge 0$ and $x \in \mathbb{R}^n$,

$$(a - Q(x))^{-\gamma} P_2(x) = P_2(x) E_\alpha \left(t^\alpha \frac{P_1(x)}{P_2(x)} \right) (a - Q(x))^{-\gamma} - \int_0^t g_\alpha(t - s) P_1(x) E_\alpha \left(s^\alpha \frac{P_1(x)}{P_2(x)} \right) (a - Q(x))^{-\gamma} ds.$$

Keeping in mind this equality, Lemma 2.2.19 and the fact that $R_{\alpha}(t)\overline{P(A)} \subseteq \overline{P(A)}R_{\alpha}(t), t \ge 0$ for any complex polynomial P(x), we can repeat verbatim the arguments used in the proof of [**304**, Theorem 2.1] so as to conclude that

$$\overline{P_1(A)} \int_0^t g_\alpha(t-s) R_\alpha(s) x \, ds = R_\alpha(t) \overline{P_2(A)} x - C \overline{P_2(A)} x, \quad t \ge 0, \ x \in D(\overline{P_2(A)}),$$

which clearly implies by Theorem 2.2.8 that for each $x \in D(\overline{P_2(A)})$ $(x \in D(\overline{P_1(A)}) \cap D(\overline{P_2(A)}))$ the function $t \mapsto u(t) \equiv R_{\alpha}(t)x, t \ge 0$ is a mild (strong) solution of (DFP). In general case, the proof of theorem can be completed by performing the Laplace transform once more. The proof is quite similar if $E = L^p(\mathbb{R}^n)$ for some $1 , and the only non-trivial thing in this case is to show the strong continuity of the operator family <math>(R_{\alpha}(t))_{t\ge 0}$. The arguments used in the proof of [**304**, Theorem 2.1] show that the mapping $t \mapsto R_{\alpha}(t)f, t \ge 0$ is continuous for every $f \in R(\overline{P_2(A)})$. But, $\overline{P_2(A)}_{|\mathcal{S}(\mathbb{R}^n)}$ is a linear topological homeomorphism of the space $\mathcal{S}(\mathbb{R}^n)$, which along with the exponential boundedness of $(R_{\alpha}(t))_{t\ge 0}$ implies the continuity of mapping $t \mapsto R_{\alpha}(t)f, t \ge 0$ for any $f \in L^p(\mathbb{R}^n)$.

We can prove in a similar way the following extension of [304, Theorem 2.2]; let us only note that the choice of regularizing operator C is slightly different now.

THEOREM 2.2.21. Suppose $0 < \alpha < 2$, $\omega \ge 0$, $P_1(x)$ and $P_2(x)$ are non-zero complex polynomials, $N_1 = dg(P_1(x))$, $N_2 = dg(P_2(x))$, $\beta > \frac{n}{2} \frac{(N_1+N_2)}{\min(1,\alpha)}$ (resp. $\beta \ge n |\frac{1}{p} - \frac{1}{2}| \frac{(N_1+N_2)}{\min(1,\alpha)}$, if $E = L^p(\mathbb{R}^n)$ for some $1), <math>P_2(x) \ne 0$, $x \in \mathbb{R}^n$ and (70) holds. Set

$$R_{\alpha}(t) := \left(E_{\alpha} \left(t^{\alpha} \frac{P_{1}(x)}{P_{2}(x)} \right) (1 + |x|^{2})^{-\beta/2} \right) (A), \quad t \ge 0.$$

Then $(R_{\alpha}(t))_{t\geq 0} \subseteq L(E)$ is a global exponentially bounded $(g_{\alpha}, R_{\alpha}(0))$ -regularized resolvent family for (DFP), $(R_{\alpha}(t))_{t\geq 0}$ is norm continuous provided $\beta > \frac{n}{2} \frac{(N_1+N_2)}{\min(1,\alpha)}$, and (71) holds.

REMARK 2.2.22. (i) The assumption $P_2(x) \neq 0, x \in \mathbb{R}^n$ implies that the operator $\overline{P_2(A)}$ is injective so that the character of degeneracy of problems considered in Subsection 2.2.3 is very mild. In actual fact, the assumption $\overline{P_2(A)}f = 0$, in combination with (\triangleright) and the fact that $P_2(\cdot)^{-1}\phi(\cdot) \in \mathcal{S}(\mathbb{R}^n), \phi \in \mathcal{S}(\mathbb{R}^n)$ implies that $\phi(A)f = 0, \phi \in \mathcal{S}(\mathbb{R}^n)$; hence, f = 0. Consider now the situation of Theorem 2.2.20, with E being a general

space and $\gamma > \frac{n \max(N, \frac{N_1+N_2}{2r})}{2r}$. Set $G_{\alpha}(t) := (P_2(\cdot)^{-1}f_t(\cdot))(A), t \ge 0$. Applying again (\triangleright), we get that $\overline{P_2(A)}G_{\alpha}(t)x = R_{\alpha}(t)x, t \ge 0, x \in E_0$. By the closedness of $\overline{P_2(A)}$, the above equality holds for any $x \in E$ so that $G_{\alpha}(t) = \overline{P_2(A)}^{-1}R_{\alpha}(t), t\ge 0$; furthermore, $(G_{\alpha}(t))_{t\ge 0} \subseteq L(E)$ is a strongly continuous operator family. Then the Laplace transform and the identity $\overline{P_1(A)}(g_{\alpha} * G_{\alpha})(t)x = R_{\alpha}(t)x - Cx, t\ge 0, x \in E$ can be used to prove that $\lambda^{\alpha-1}(\lambda^{\alpha}B - A)^{-1}Cx = \int_0^{\infty} e^{-\lambda t}G_{\alpha}(t)x \, dt$ for any $x \in E$ and $\lambda > 0$ sufficiently large; therefore, the condition (ii.1) stated in Theorem 2.2.8 holds, with $a(t) = g_{\alpha}(t)$ and k(t) = 1, which continues to hold in any case set out in Theorem 2.2.20–Theorem 2.2.21 and Remark 2.2.23 below. It should be also observed that $(G_{\alpha}(t))_{t\ge 0}$ is an exponentially equicontinuous (g_{α}, C) -regularized resolvent family generated by $\overline{P_1(A)}$, $\overline{P_2(A)}$ (cf. the next section for the notion and more details), and that for each $f \in D(\overline{P_1(A)}) \cap D(\overline{P_2(A)})$, the function $u(t) := R_{\alpha}(t)x, t \ge 0$ is a unique solution of the following Cauchy problem:

$$(P)_L: \begin{cases} u \in C([0,\infty) : [D(\overline{P_1(A)})]) \cap C([0,\infty) : [D(\overline{P_2(A)})]), \\ \overline{P_2(A)} \mathbf{D}_t^{\alpha} u(t) = \overline{P_1(A)} u(t), \quad t \ge 0, \\ u(0) = Cx; \ u^{(j)}(0) = 0, \quad 1 \le j \le \lceil \alpha \rceil - 1. \end{cases}$$

A similar result holds for the second order degenerate equations in the next subsection, and for the first order degenerate equations in the case that the requirements of Theorem 2.2.21 hold; cf. [307].

- (ii) It is worth noting that Theorem 2.2.20 and Theorem 2.2.21 can be strengthened in the following way. Suppose that the estimate (72) holds with the number $N_1 + N_2$ replaced by some other number $\sigma \ge 0$, and $|P_1(x)/P_2(x)| \le M(1+|x|)^{\sigma_1}, x \in \mathbb{R}^n$ for some $\sigma_1 \in [0, N_1]$. Put $W := \sigma + \chi_{(0,1)}(\alpha)\sigma_1(\alpha^{-1} - 1)$. Based on the evidence used in proving the estimate (76), along with the inequality [**380**, (3.19)] and the first estimate appearing in the proof of [**380**, Theorem 4.3], implies that the following holds:
- Theorem 2.2.20: The assertion of this theorem continues to hold for any number $\gamma > n \max(N, W)/2r$ (resp. $\gamma = \frac{n}{r} |\frac{1}{p} \frac{1}{2}| \max(N, W)$, if $E = L^p(\mathbb{R}^n)$ for some 1).
- Theorem 2.2.21: The assertion of this theorem continues to hold for any number $\beta > \frac{nW}{2}$ (resp. $\beta = n | \frac{1}{p} \frac{1}{2} | W$, if $E = L^p(\mathbb{R}^n)$ for some 1). $If <math>V_2 \ge 0$ and if for each $\eta \in \mathbb{N}_0^n$ there exists $M_\eta > 0$ such that

$$|D^{\eta}(P_2(x)^{-1})| \leq M_{\eta}(1+|x|)^{|\eta|(V_2-1)}, \quad x \in \mathbb{R}^n$$

(this holds with $V_2 = N_2$), then similarly as in the first part of this remark we can prove that the condition (ii.1) stated in Theorem 2.2.8 holds, with $a(t) = g_{\alpha}(t)$ and k(t) = 1, if: $\gamma > \frac{n \max(N, V_2, W)}{2r}$, resp. $\gamma \ge \frac{n}{r} |\frac{1}{p} - \frac{1}{2}| \max(N, V_2, W)$ (Theorem 2.2.20); $\beta > \frac{n \max(V_2, W)}{2}$, resp. $\beta \ge n |\frac{1}{p} - \frac{1}{2}| \max(V_2, W)$ (Theorem 2.2.21). Observe, finally, that there is a great number of concrete examples where we can further refine the obtained theoretical results by using direct calculations [291].

- (iii) The estimate (70) is very restrictive in the case that $\alpha \in (0, 1)$. If $1 < \alpha < 2$ and $\omega \ge 0$, then by the proof of [**380**, Theorem 4.2], cf. also [**304**, Remark 2.1(i)], the condition $\frac{P_1}{P_2}(\mathbb{R}^n) \subseteq \mathbb{C} \smallsetminus (\omega + \Sigma_{\alpha\pi/2})$ implies the validity of (70).
- (iv) Let $t \mapsto u(t), t \ge 0$ be a mild solution of the problem (52) with the operators A and B replaced respectively by $\overline{P_1(A)}$ and $\overline{P_2(A)}$. Then it can be simply proved that the variation of parameters formula $(R_{\alpha} * f)(t) = (C\overline{P_2(A)} * u)(t)$ holds for any $t \ge 0$. This implies that we can look into the C-wellposedness of the inhomogeneous degenerate Cauchy problem $(DFP)_f$, obtained by adding the term f(t) on the right hand side of (DFP). All this has been seen many times and we shall skip details for the sake of brevity; the interested reader may consult [**292**, Subsection 2.1.1] for further information concerning the C-wellposedness of abstract fractional Cauchy problems.
- (v) Let $0 < \alpha_0 < \alpha < 2$, and let the assumptions of Theorem 2.2.20 (Theorem 2.2.21) hold. Using Theorem 2.2.13(ii), it readily follows that there exists an exponentially bounded, analytic $(g_{\alpha_0}, R_{\alpha_0}(0))$ -regularized resolvent family $(R_{\alpha_0}(t))_{t \ge 0} \subseteq L(E)$ for (DFP), of angle min($((\alpha/\alpha_0) 1)\pi/2, \pi/2$); furthermore, if $E = L^2(\mathbb{R}^n)$ and $\omega = 0$, then the angle of analyticity equals min($((\alpha/\alpha_0) 1)\pi/2, \pi)$ and can be strictly greater than $\pi/2$.

REMARK 2.2.23. In this remark, we would like to explain how one can reformulate the assertions of Theorem 2.2.20 and Theorem 2.2.21 in E_l -type spaces. Let E be one of the spaces $L^p(\mathbb{R}^n)$ $(1 \leq p \leq \infty)$, $C_0(\mathbb{R}^n)$, $C_b(\mathbb{R}^n)$, $BUC(\mathbb{R}^n)$ and let $0 \leq l \leq n$. Let us recall that the space E_l is defined by $E_l := \{f \in E : f^{(\alpha)} \in I\}$ E for all $\alpha \in \mathbb{N}_0^l$. The totality of seminorms $(q_\alpha(f) := ||f^{(\alpha)}||_E, f \in E_l; \alpha \in \mathbb{N}_0^l)$ induces a Fréchet topology on E_l . Let $\mathbf{T}_l\langle\cdot\rangle$ possess the same meaning as in [542], let $a_{\eta} \in \mathbb{C}, 0 \leq |\eta| \leq N_1$, and let $b_{\eta} \in \mathbb{C}, 0 \leq |\eta| \leq N_2$. Assume that the operators $P_1(D)f \equiv \sum_{|\eta| \leq N_1} a_{\eta} D^{\eta} f$ and $P_2(D)f \equiv \sum_{|\eta| \leq N_2} b_{\eta} D^{\eta} f$ act with their maximal distributional domains. Then $P_1(D)$ and $P_2(D)$ are closed linear operators on E_l ; in the sequel, we assume that $P_1(D) \neq 0$ and $P_2(D) \neq 0$. Let $\omega \ge 0$ be such that (70) holds. Then it can be easily seen that P(D) generates an exponentially equicontinuous (g_{α}, I) -regularized resolvent family $(S_{\alpha}(t))_{t \ge 0}$ in the space E_n and the condition (ii.1) stated in Theorem 2.2.8 holds, with $a(t) = g_{\alpha}(t)$ and k(t) = 1. Let γ (β) have the same value as in the formulation of Theorem 2.2.20 (Theorem 2.2.21). Then the estimates (77)–(78) continue to hold, and slight modifications of the proofs of [542, Theorem 2.2, Theorem 2.4] show that the following holds (in our opinion, the proofs of Theorem 2.2.20 and Theorem 2.2.21 are much simpler than those of [542, Theorem 2.2, Theorem 2.4] in the case that l = 0 and $E \neq L^{\infty}(\mathbb{R}^n), E \neq C_b(\mathbb{R}^n)$:

(i) Theorem 2.2.20: Set $R_{\alpha}(t) =: \mathbf{T}_1 \langle E_{\alpha}(t^{\alpha} P_1(x)/P_2(x))(a-Q(x))^{-\gamma} \rangle, t \ge 0$. Then $(R_{\alpha}(t))_{t\ge 0}$ is an exponentially equicontinuous $(g_{\alpha}, R_{\alpha}(0))$ -regularized resolvent family for (DFP), $(R_{\alpha}(t))_{t\ge 0}$ is 'norm continuous' provided $\gamma > \frac{n \max(N, \frac{N_1+N_2}{\min(1, \alpha)})}{2r}$, in the sense that, for every bounded subset B of E_l and for every $\eta \in \mathbb{N}_0^l$, the mapping $t \mapsto \sup_{f \in B} q_\eta(R_\alpha(t)f), t \ge 0$ is continuous. The estimate (71) reads as follows:

(79)
$$q_{\eta}(R_{\alpha}(t)f) \leq M(1 + t^{\max(1,\alpha)n/2})e^{\omega t}q_{\eta}(f), \quad t \geq 0, \ f \in E_l, \ \eta \in \mathbb{N}_0^l, \text{ resp.},$$

 $q_{\eta}(R_{\alpha}(t)f) \leq M(1 + t^{\max(1,\alpha)n|\frac{1}{p} - \frac{1}{2}|})e^{\omega t}q_{\eta}(f), \quad t \geq 0, \ f \in E_l, \ \eta \in \mathbb{N}_0^l,$

with M being independent of $f \in E_l$ and $\eta \in \mathbb{N}_0^l$.

(ii) Theorem 2.2.21: Set $R_{\alpha}(t) =: \mathbf{T}_{1}\langle E_{\alpha}(t^{\alpha}P_{1}(x)/P_{2}(x))(1+|x|^{2})^{-\beta/2}\rangle, t \geq 0$. Then $(R_{\alpha}(t))_{t\geq 0}$ is an exponentially equicontinuous $(g_{\alpha}, R_{\alpha}(0))$ -regularized resolvent family for (DFP), $(R_{\alpha}(t))_{t\geq 0}$ is 'norm continuous' provided $\beta > \frac{n}{2} \frac{(N_{1}+N_{2})}{\min(1,\alpha)}$ and (79) holds.

Notice also that $(g_{\alpha}, R_{\alpha}(0))$ -regularized resolvent families for (DFP), constructed in this remark (Theorem 2.2.20–Theorem 2.2.21), satisfy $R_{\alpha}(t)R_{\alpha}(s) = R_{\alpha}(s)R_{\alpha}(t)$, $t, s \ge 0$, as well as that $R_{\alpha}(t)P(D) \subseteq P(D)R_{\alpha}(t)$, $t \ge 0$ ($R_{\alpha}(t)\overline{P(A)} \subseteq \overline{P(A)}R_{\alpha}(t)$, $t \ge 0$) for any complex polynomial P(x). The final conclusions of Remark 2.2.22(ii) remain true for $(g_{\alpha}, R_{\alpha}(0))$ -regularized resolvent families for (DFP) in the setting of E_{l} -type spaces.

Before proceeding to the next subsection, it would be worthwhile to note that Theorem 2.2.20 and Theorem 2.2.21 can be successfully applied in the analysis of the fractional analogues of the Barenblatt–Zheltov–Kochina equation in infinite domains (cf. [140, Example 1.6, p. 50] and [52]):

$$(\eta \Delta - 1) \mathbf{D}_t^{\alpha} u(t) + \Delta u = 0 \quad (\eta > 0),$$

where $0 < \alpha < 2$ and $\cos(\pi/\alpha) \leq 0$. Details can be left to the reader.

2.2.4. Degenerate second order equations associated with abstract differential operators. The main objective in this subsection is to prove some results on the *C*-wellposedness of the following abstract degenerate Cauchy problem of second order

$$(DFP)_2:\begin{cases} \frac{d^2}{dt^2}\overline{P_2(A)}u(t)=\overline{P_1(A)}u(t), & t \ge 0, \\ u(0)=Cx, \ u'(0)=0. \end{cases}$$

Keeping in mind the results clarified by now, as well as the analyses contained in the papers [559] and [542,543], the consideration of degenerate second order equations is similar to that of degenerate fractional equations of order $\alpha \in (0, 2)$; because of that, we shall only outline the main details and omit the proofs. As before, we assume that $P_1(x)$ and $P_2(x)$ are non-zero complex polynomials in n variables, as well as that $N_1 = dg(P_1(x))$ and $N_2 = dg(P_2(x))$. Set $F_t(z) := E_2(t^2z), t \ge 0, z \in \mathbb{C}$, and $\Omega(\omega) := \{\lambda^2 : \operatorname{Re} \lambda > \omega\}$, if $\omega > 0$ and $\Omega(\omega) := \mathbb{C} \setminus (-\infty, -\omega^2]$, if $\omega \le 0$. Given $l \ge 0$ and $t \ge 0$ in advance, set

$$Q_{l}(t) := \begin{cases} (1+t^{l})e^{\omega t}, & \text{if } \omega > 0, \\ 1+t^{2l}, & \text{if } \omega = 0, \\ 1+t^{l}, & \text{if } \omega < 0. \end{cases}$$

Suppose now that $P_2(x) \neq 0$, $x \in \mathbb{R}^n$ and $P_1(x)/P_2(x) \notin \Omega(\omega)$, $x \in \mathbb{R}^n$. Then, for every $\eta \in \mathbb{N}_0^n$ with $|\eta| > 0$, the equation (73) reads as follows:

(80)
$$D^{\eta} E_2\left(t^2 \frac{P_1(x)}{P_2(x)}\right) = \sum_{j=1}^{|\eta|} F_t^{(j)}\left(\frac{P_1(x)}{P_2(x)}\right) R_{\eta,j}(x), \quad t \ge 0, \ x \in \mathbb{R}^n,$$

where $R_{\eta,j}(x)$ is a finite sum of terms like $\prod_{w=1}^{s_j} D^{\eta_{j,w}}(\frac{P_1(x)}{P_2(x)})$ with $|\eta_{j,w}| > 0$ $(1 \leq w \leq s_j)$ and $|\eta_{j,1}| + \cdots + |\eta_{j,s_j}| \leq |\eta|$. Due to the computation established in [559, Lemma 2.1], we have that, for every $l \in \mathbb{N}_0$, $|F_t^{(l)}(P_1(x)/P_2(x))| \leq MQ_l(t)$, $t \geq 0, x \in \mathbb{R}^n$. Combining this estimate with (80), and repeating verbatim the arguments given in the proof of Theorem 2.2.20 (cf. also Remark 2.2.22), it can be easily seen that the following two theorems hold good.

THEOREM 2.2.24. Suppose that $P_1(x)$ and $P_2(x)$ are non-zero complex polynomials, $N_1 = dg(P_1(x)), N_2 = dg(P_2(x)), P_2(x) \neq 0, x \in \mathbb{R}^n, \omega \in \mathbb{R}, P_1(x)/P_2(x) \notin \Omega(\omega), x \in \mathbb{R}^n, N \in \mathbb{N} \text{ and } r \in (0, N].$ Let Q(x) be an r-coercive complex polynomial of degree N, $a \in \mathbb{C} \setminus Q(\mathbb{R}^n)$, let for each $\eta \in \mathbb{R}^n$ with $|\eta| > 0$ the estimate (72) hold with the number $N_1 + N_2$ replaced by $\sigma \geq 0$ (the choice $\sigma = N_1 + N_2$ is always possible), and let $\gamma > \frac{n \max(N, \sigma)}{2r}$ (resp. $\gamma = \frac{n}{r} |\frac{1}{p} - \frac{1}{2}| \max(N, \sigma)$, if $E = L^p(\mathbb{R}^n)$ for some 1). Set

$$R_2(t) := \left(E_2\left(t^2 \frac{P_1(x)}{P_2(x)}\right) (a - Q(x))^{-\gamma} \right) (A), \quad t \ge 0.$$

Then $(R_2(t))_{t \ge 0} \subseteq L(E)$ is a global exponentially bounded $(g_2, R_2(0))$ -regularized resolvent family for $(DFP)_2$, $(R_2(t))_{t \ge 0}$ is norm continuous provided $\gamma > \frac{n \max(N, \sigma)}{2r}$, and the following holds:

(81)
$$||R_2(t)|| \leq MQ_{n/2}(t), \quad t \ge 0, \text{ resp., } ||R_2(t)|| \leq MQ_{n|\frac{1}{p}-\frac{1}{2}|}(t), \quad t \ge 0.$$

THEOREM 2.2.25. Suppose that $P_1(x)$ and $P_2(x)$ are non-zero complex polynomials, $N_1 = dg(P_1(x)), N_2 = dg(P_2(x)), P_2(x) \neq 0, x \in \mathbb{R}^n, \omega \in \mathbb{R}$ and $P_1(x)/P_2(x) \notin \Omega(\omega), x \in \mathbb{R}^n$. Let for each $\eta \in \mathbb{R}^n$ with $|\eta| > 0$ the estimate (72) hold with the number $N_1 + N_2$ replaced by $\sigma \ge 0$ (the choice $\sigma = N_1 + N_2$ is always possible), and let $\beta > \frac{n\sigma}{2}$ (resp. $\beta = n|\frac{1}{p} - \frac{1}{2}|\sigma$, if $E = L^p(\mathbb{R}^n)$ for some 1). Set

$$R_2(t) := \left(E_2\left(t^2 \frac{P_1(x)}{P_2(x)}\right) (1+|x|^2)^{-\beta/2} \right) (A), \quad t \ge 0.$$

Then $(R_2(t))_{t\geq 0} \subseteq L(E)$ is a global exponentially bounded $(g_2, R_2(0))$ -regularized resolvent family for $(DFP)_2$, $(R_2(t))_{t\geq 0}$ is norm continuous provided $\beta > \frac{n\sigma}{2}$, and (81) holds.

REMARK 2.2.26. (i) Suppose that

$$\operatorname{Re}(P_1(x)/P_2(x)) \leqslant -\zeta |x|^r + \zeta_1, \quad x \in \mathbb{R}^n,$$

for some positive real numbers $r, \zeta, \zeta_1 > 0$. Then the Lagrange mean value theorem for vector-valued functions implies that $\sigma \ge r$, and by

the proof of [559, Lemma 2.1], we have that there exist numbers $L \ge 1$ and $M \ge 1$ such that, for every $j \in \mathbb{N}_0$ with j < n/2, the following holds $|F_t^{(j)}(P_1(x)/P_2(x))| \le MQ_j(t)(1+|x|)^{-jr/2}, t \ge 0, |x| \ge L$. Unfortunately, the above does not guarantee that we can refine the results clarified in Theorem 2.2.24 and Theorem 2.2.25 by replacing the number σ with $\sigma - (r/2)$, unless $P_2(x) \equiv 1$. Observe, however, that the refinement of this type is possible if $\omega \le 0$ (in this case we can estimate the derivatives of function $E_2(t^2P_1(x)/P_2(x))$ by using the formula appearing in the second line of the proof of [543, Theorem 4.1]).

- (ii) Let $V_2 \ge 0$. If for each $\eta \in \mathbb{N}_0^n$ there exists $M_\eta > 0$ such that $|D^{\eta}(P_2(x)^{-1})| \le M_{\eta}(1+|x|)^{|\eta|(V_2-1)}, x \in \mathbb{R}^n$ (recall that the choice $V_2 = N_2$ is always possible), and if we replace, in the formulations of Theorem 2.2.24 and Theorem 2.2.25, the number σ with $\max(\sigma, V_2)$, then condition (ii.1) stated in Theorem 2.2.8 holds, with a(t) = t and k(t) = 1.
- (iii) The assertions of Theorem 2.2.24 and Theorem 2.2.25, as well as the conclusions stated in the first and second part of this remark, continue to hold with suitable modifications in the setting of E_l -type spaces; cf. also [543, Theorem 4.1-Theorem 4.2].

Before closing this subsection with an illustrative example, it is our duty to say that the results from Subsection 2.1.3–Subsection 2.1.4 are inapplicable in the analysis of a great number of equations of mathematical physics that are not solvable relative to the highest-order time-derivative (cf. Example 2.3.22 for further information in this direction).

EXAMPLE 2.2.27. (i) Let $1 , <math>0 < \alpha < 2$, $l \in \mathbb{N}$, $E = L^p(\mathbb{R}^2)$, and let the fractional Sobolev space $S^{\alpha,p}(\mathbb{R}^2)$ be defined in the sense of [410, Definition 12.3.1, p. 297]; that is,

$$\mathbf{S}^{\alpha,p}(\mathbb{R}^2) := D((1 - \Delta_p)^{\alpha/2}),$$

where Δ_p acts on $L^p(\mathbb{R}^2)$ with its maximal distributional domain. Consider the following degenerate fractional Cauchy problem:

$$(P): \begin{cases} \mathbf{D}_{t}^{\alpha}[u_{xx} + u_{xy} + u_{yy} - u] = e^{-i\alpha\frac{\pi}{2}} \left[(-1)^{l+1} \frac{\partial^{2l}}{\partial x^{2\alpha}} u + u_{yy} \right], & t \ge 0, \\ u(0,x) = \phi(x); \ u_{t}(0,x) = 0 & \text{if } \alpha \ge 1, \end{cases}$$

cf. Theorem 2.2.21 with $P_1(x, y) = e^{-i\alpha \frac{\pi}{2}} (x^{2l} + y^2)$, $P_2(x, y) = x^2 + xy + y^2 + 1$ and $\omega = 0$. Then it can be easily seen that the conditions stated in Remark 2.1.23(ii) hold with $\sigma = \sigma_1 = 2l - 2$, so that for each $\beta \ge n |\frac{1}{p} - \frac{1}{2}|(2l-2)(1+\chi_{(0,1)}(\alpha)(\alpha^{-1}-1))$ there exists a global exponentially bounded $(g_\alpha, R_\alpha(0))$ -regularized resolvent family for the corresponding problem (DFP), obeying the property (ii.1) of Theorem 2.2.8 with $a(t) = g_\alpha(t)$ and k(t) = 1. Hence, there exists a unique strong solution of problem (P) provided that $\phi \in S^{2l+\beta,p}(\mathbb{R}^2)$.

(ii) Let $1 , <math>E = L^p(\mathbb{R}^n)$ and $Q \in \mathbb{N} \setminus \{1\}$. Consider the following degenerate second order Cauchy problem:

$$(P_2): \begin{cases} \frac{\partial^2}{\partial t^2} (\Delta u(t,x) - u(t,x)) = \sum_{|\eta| \leqslant Q} a_{\eta} D^{\eta} u(t,x), & t \ge 0, \ x \in \mathbb{R}^n, \\ u(0,x) = \phi(x), \ u_t(0,x) = 0; \end{cases}$$

then $P_1(x) = \sum_{|\eta| \leq Q} a_{\eta} x^{\eta}$ and $P_2(x) = -1 - |x|^2$ $(x \in \mathbb{R}^n)$. Assuming that the polynomial $P_1(x)$ is positive, as well as that the estimate (72) holds with some number $\sigma \geq 0$ and the condition stated in Remark 2.2.26(ii) holds with some number $V_2 \geq 0$, then there exists a unique strong solution of problem (P_2) provided that

$$\phi \in \mathbf{S}^{Q+n|\frac{1}{p}-\frac{1}{2}|\max(\sigma, V_2), p(\mathbb{R}^n)}$$

2.2.5. Semilinear degenerate relaxation equations associated with abstract differential operators. In this subsection, we shall present our recent results from [310] and [319] concerning the existence and uniqueness of mild solutions of the following semilinear degenerate relaxation equation

$$(\text{DFP})_{sl}: \begin{cases} \mathbf{D}_t^{\alpha} \overline{P_2(A)} u(t) = \overline{P_1(A)} u(t) + f(t, u(t)), & t \ge 0, \\ u(0) = x, \end{cases}$$

where $0 < \alpha < 1$, the function $f(\cdot, \cdot)$ satisfies certain properties and $iA_j, 1 \leq j \leq n$ are commuting generators of bounded C_0 -groups on a Banach space E.

We need the following definition.

DEFINITION 2.2.28. Let $0 < \alpha < 1$, let $C \in L(E)$ be injective and let $C^{-1}\overline{P_1(A)}C = \overline{P_1(A)}, C^{-1}\overline{P_2(A)}C = \overline{P_2(A)}$. A strongly continuous operator family $(P_{\alpha}(t))_{t>0} \subseteq L(E)$ is said to be an $(\alpha, \alpha, \overline{P_1(A)}, \overline{P_2(A)}, C)$ -resolvent family iff there exist $M \ge 1$ and $\omega \ge 0$ such that the mapping $t \mapsto ||t^{1-\alpha}P_{\alpha}(t)||, t \in (0, 1]$ is bounded, $||P_{\alpha}(t)|| \le Me^{\omega t}, t \ge 1$ and

$$(\lambda^{\alpha}\overline{P_2(A)} - \overline{P_1(A)})^{-1}Cx = \int_0^{\infty} e^{-\lambda t} P_{\alpha}(t) x \, dt, \quad \operatorname{Re} \lambda > \omega, \ x \in E.$$

The following theorem is backbone of this subsection.

THEOREM 2.2.29. Suppose $0 < \alpha < 1$, $\omega \ge 0$, $P_1(x)$ and $P_2(x)$ are non-zero complex polynomials, $N_1 = dg(P_1(x))$ and $N_2 = dg(P_2(x))$.

(i) Let $N \in \mathbb{N}$, $r \in (0, N]$, let Q(x) be an r-coercive complex polynomial of degree N, $a \in \mathbb{C} \setminus Q(\mathbb{R}^n)$, $\gamma > \frac{n \max(N, \frac{N_1+N_2}{\alpha})}{2r}$ (resp. $\gamma = \frac{n}{r}|\frac{1}{p} - \frac{1}{2}|\max(N, \frac{N_1+N_2}{\alpha})$, if $E = L^p(\mathbb{R}^n)$ for some 1), $<math>\gamma' > \frac{\max(N, \frac{N_1+N_2}{\alpha})}{r}\frac{n}{2} + \frac{N_1+N_2}{r\alpha}(1-\alpha)$ ($\gamma' = n|\frac{1}{p} - \frac{1}{2}|\frac{\max(N, \frac{N_1+N_2}{\alpha})}{r} + \frac{N_1+N_2}{r\alpha}(1-\alpha)$, if $E = L^p(\mathbb{R}^n)$ for some $1), <math>P_2(x) \neq 0$, $x \in \mathbb{R}^n$, and let (70) hold. Set

$$R_{\alpha,\gamma}(t) := \left(E_{\alpha} \left(t^{\alpha} \frac{P_1(x)}{P_2(x)} \right) (a - Q(x))^{-\gamma} \right) (A), \quad t \ge 0,$$

$$\begin{split} C &:= R_{\alpha,\gamma'}(0) \ and \\ P_{\alpha,\gamma'}(t) &:= t^{\alpha-1} \Big(P_2(x)^{-1} E_{\alpha,\alpha} \Big(t^{\alpha} \frac{P_1(x)}{P_2(x)} \Big) (a - Q(x))^{-\gamma'} \Big) (A), \quad t > 0. \end{split}$$

Then $(R_{\alpha,\gamma}(t))_{t\geq 0} \subseteq L(E)$ is a global exponentially bounded $(g_{\alpha}, R_{\alpha,\gamma}(0))$ regularized resolvent family for (DFP), $(P_{\alpha,\gamma'}(t))_{t>0} \subseteq L(E)$ is a global $(\alpha, \alpha, \overline{P_1(A)}, \overline{P_2(A)}, C)$ -resolvent family, $(R_{\alpha,\gamma}(t))_{t\geq 0}$ is norm continuous provided $\gamma > \frac{n \max(N, \frac{N_1+N_2}{\alpha})}{2r}$, $(P_{\alpha,\gamma'}(t))_{t>0}$ is norm continuous provided $\gamma' > \frac{\max(N, \frac{N_1+N_2}{\alpha})}{r} \frac{n}{2} + \frac{N_1+N_2}{r\alpha} (1-\alpha), (71) \text{ holds with } (R_{\alpha}(t))_{t \ge 0} \text{ replaced} by (R_{\alpha,\gamma}(t))_{t \ge 0} \text{ therein, and}$

$$||P_{\alpha,\gamma'}(t)|| \leq M t^{\alpha-1} (1+t^{1-\alpha+\frac{n}{2}}) e^{\omega t}, \quad t > 0, \ resp.,$$
$$||P_{\alpha,\gamma'}(t)|| \leq M t^{\alpha-1} (1+t^{1-\alpha+n|\frac{1}{p}-\frac{1}{2}|}) e^{\omega t}, \quad t > 0.$$

(ii) Suppose $\beta > \frac{n}{2} \frac{(N_1+N_2)}{\alpha}$ (resp. $\beta \ge n |\frac{1}{p} - \frac{1}{2}| \frac{(N_1+N_2)}{\alpha}$, if $E = L^p(\mathbb{R}^n)$ for some $1), <math>\beta' > (1 - \alpha + \frac{n}{2}) \frac{(N_1+N_2)}{\alpha}$ (resp. $\beta' \ge (1 - \alpha + n |\frac{1}{p} - \alpha)$) $(\frac{1}{2}|) \frac{(N_1+N_2)}{2}$, if $E = L^p(\mathbb{R}^n)$ for some $1 , <math>P_2(x) \neq 0$, $x \in \mathbb{R}^n$ and (70) holds. Set

$$R_{\alpha,\beta}(t) := \left(E_{\alpha} \left(t^{\alpha} \frac{P_1(x)}{P_2(x)} \right) (1+|x|^2)^{-\beta/2} \right) (A), \quad t \ge 0$$

$$C := R_{\alpha,\beta'}(0), and$$

ъ

$$P_{\alpha,\beta'}(t) := t^{\alpha-1} \Big(P_2(x)^{-1} E_{\alpha,\alpha} \Big(t^{\alpha} \frac{P_1(x)}{P_2(x)} \Big) (1+|x|^2)^{-\beta'/2} \Big) (A), \quad t > 0.$$

Then $(R_{\alpha,\beta}(t))_{t\geq 0} \subseteq L(E)$ is a global exponentially bounded $(g_{\alpha}, R_{\alpha,\beta}(0))$ regularized resolvent family for (DFP), $(P_{\alpha,\beta'}(t))_{t>0} \subseteq L(E)$ is a global $(\alpha, \alpha, \overline{P_1(A)}, \overline{P_2(A)}, C)$ -resolvent family, $(R_{\alpha,\beta}(t))_{t \ge 0}$ is norm continuous provided $\beta > \frac{n}{2} \frac{(N_1+N_2)}{\alpha}$, $(P_{\alpha,\beta'}(t))_{t>0}$ is norm continuous provided $\beta' > (1-\alpha+\frac{n}{2})\frac{(N_1+N_2)}{\alpha}$, (71) holds with $(R_{\alpha}(t))_{t\geq0}$ replaced by $(R_{\alpha,\beta}(t))_{t\geq0}$ therein, and (82) holds with $(P_{\alpha,\gamma'}(t))_{t\geq 0}$ replaced by $(P_{\alpha,\beta'}(t))_{t\geq 0}$ therein.

PROOF. Recall that $k = 1 + \lfloor n/2 \rfloor$. The results for $(R_{\alpha,\gamma}(t))_{t \ge 0}$ and $(R_{\alpha,\beta}(t))_{t\geq 0}$ have been proved in Theorem 2.2.20 and Theorem 2.2.21, respectively. In either choice of the regularizing operator C, we have $C^{-1}\overline{P_1(A)}C = \overline{P_1(A)}$ and $C^{-1}\overline{P_2(A)}C = \overline{P_2(A)}$ [304]. Furthermore, for every $j \in \mathbb{N}$, there exist uniquely determined real numbers $c_{l,j,\alpha}$ $(1 \leq l \leq j)$ such that $E'_{\alpha}(z) = \alpha^{-1} E_{\alpha,\alpha}(z), z \in \mathbb{C}$, as well as that $E_{\alpha}^{(j)}(z) = \sum_{l=1}^{j} c_{l,j,\alpha} E_{\alpha,\alpha j - (j-l)}(z), z \in \mathbb{C}$ (cf. (44)). Using these facts and the proof of Theorem 2.2.20, we may conclude that:

1. For every multi-index $\eta \in \mathbb{N}_0^n$ with $|\eta| > 0$, there exists $c_\eta > 0$ such that

$$\left| D^{\eta} \left(\frac{P_1(x)}{P_2(x)} \right) \right| \leq c_{\eta} (1+|x|)^{|\eta|(N_1+N_2-1)}, \quad x \in \mathbb{R}^n.$$

2. For every multi-index $\eta \in \mathbb{N}_0^n$ with $|\eta| > 0$, for every $t \ge 0$ and for every $x \in \mathbb{R}^n$, we have:

(83)
$$D^{\eta}E_{\alpha,\alpha}\left(t^{\alpha}\frac{P_{1}(x)}{P_{2}(x)}\right) = \sum_{j=1}^{|\eta|} t^{\alpha j}E_{\alpha,\alpha}^{(j)}\left(t^{\alpha}\frac{P_{1}(x)}{P_{2}(x)}\right)R_{\eta,j}(x)$$

(84) $= \sum_{j=1}^{|\eta|} t^{\alpha j}\sum_{l=1}^{j+1} \alpha c_{l,j+1,\alpha}E_{\alpha,\alpha j-j+l+\alpha-1}\left(t^{\alpha}\frac{P_{1}(x)}{P_{2}(x)}\right)R_{\eta,j}(x),$

where $R_{\eta,j}(x)$ can be represented as a finite sum of terms like

 $\prod_{q=1}^{s_j} D^{\eta_{j,q}}(\frac{\dot{P}_1(x)}{P_2(x)}) \text{ with } |\eta_{j,q}| > 0 \ (1 \leqslant q \leqslant s_j) \text{ and } |\eta_{j,1}| + \dots + |\eta_{j,s_j}| \leqslant |\eta|.$

In the remainder of the proof, by M we denote a generic constant whose value may vary at each occurrence. Owing to [1./2.], we get that

(85)
$$|R_{\eta,j}(x)| \leq M(1+|x|)^{|\eta|(N_1+N_2-1)}$$
, provided $1 \leq j \leq |\eta| \leq k$ and $x \in \mathbb{R}^n$.

Arguing as in the proof of [296, Theorem 2.1], we can prove that, for every $t \ge 0$, $x \in \mathbb{R}^n$, and for every $j, l \in \mathbb{N}$ such that $1 \le j \le k$ and $1 \le l \le j + 1$, the following holds:

(86)
$$|E_{\alpha,\alpha j-(j-l)+\alpha-1}(t^{\alpha}P_1(x)/P_2(x))|$$

 $\leq M \Big[1 + t^{1-(\alpha j-(j-l)+\alpha-1)} |P_1(x)/P_2(x)|^{\frac{1-(\alpha j-(j-l)+\alpha-1)}{\alpha}} e^{\omega t} \Big].$

If $t \ge 0$ and $x \in \mathbb{R}^n$ satisfies $|t^{\alpha}P_1(x)/P_2(x)| \le 1$, then the equation (83) yields:

$$|D^{\eta}(E_{\alpha,\alpha}(t^{\alpha}P_{1}(x)/P_{2}(x)))| \leq M(t^{\alpha} + t^{\alpha|\eta|})(1+|x|)^{|\eta|(N_{1}+N_{2}-1)}, \quad |\eta| \leq k.$$

Suppose now that $1 \leq l \leq j+1$, $1 \leq j \leq |\eta| \leq k$, $t \geq 0$, $x \in \mathbb{R}^n$ and $|t^{\alpha}P_1(x)/P_2(x)| \geq 1$. Then it can be easily seen that the supposition $1 - (\alpha j - (j - l) + \alpha - 1) \geq 0$ implies

$$\begin{split} (N_1 + N_2) \frac{1 - (\alpha j - (j - l) + \alpha - 1)}{\alpha} + |\eta| (N_1 + N_2 - 1) \\ \leqslant |\eta| \Big(\frac{N_1 + N_2}{\alpha} - 1 \Big) + (N_1 + N_2) \frac{1 - \alpha}{\alpha}. \end{split}$$

Using this estimate and (85), it readily follows that

$$\begin{aligned} |t^{\alpha}P_{1}(x)/P_{2}(x)|^{\frac{1-(\alpha j-(j-l)+\alpha-1)}{\alpha}}(1+|x|)^{|\eta|(N_{1}+N_{2}-1)} \\ \leqslant Mt^{1-(\alpha j-(j-l)+\alpha-1)}(1+|x|)^{|\eta|(\frac{N_{1}+N_{2}}{\alpha}-1)+(N_{1}+N_{2})\frac{1-\alpha}{\alpha}}, \end{aligned}$$

provided $1 - (\alpha j - (j - l) + \alpha - 1) \ge 0$. On the other hand, it is clear that

$$(87) |t^{\alpha}P_{1}(x)/P_{2}(x)|^{\frac{1-(\alpha j-(j-l)+\alpha-1)}{\alpha}}(1+|x|)^{|\eta|(N_{1}+N_{2}-1)} \leqslant M(1+|x|)^{|\eta|(N_{1}+N_{2}-1)},$$

provided $1 - (\alpha j - (j - l) + \alpha - 1) \leq 0$. Then, for $t \geq 0, x \in \mathbb{R}^n$ and $0 < |\eta| \leq k$, the following holds (cf. (84), (86)–(87) and [**304**, (2.6)–(2.7)]):

(88)
$$\left| D^{\eta} E_{\alpha,\alpha} \left(t^{\alpha} \frac{P_1(x)}{P_2(x)} \right) \right| \leq M(1+t^{1-\alpha})(1+t^{|\eta|}) e^{\omega t} (1+|x|)^{|\eta|(\frac{N_1+N_2}{\alpha}-1)+(N_1+N_2)\frac{1-\alpha}{\alpha}};$$

observe that the inequality $\operatorname{Re}((t^{\alpha}P_1(x)/P_2(x))^{1/\alpha}) \leq \omega t, t \geq 0, x \in \mathbb{R}^n$ and Theorem 1.5.1 together imply that the previous estimate also holds in the case that $|\eta| = 0$. Set

$$G_{\alpha,\gamma'}(t) := \left(P_2(x)^{-1} E_\alpha \left(t^\alpha \frac{P_1(x)}{P_2(x)} \right) (a - Q(x))^{-\gamma'} \right) (A), \quad t \ge 0,$$

in the case of examination of (i), resp.

$$G_{\alpha,\beta'}(t) := \left(P_2(x)^{-1} E_\alpha \left(t^\alpha \frac{P_1(x)}{P_2(x)}\right) (1+|x|^2)^{-\beta'/2}\right) (A), \quad t \ge 0,$$

in the case of examination of (ii). Then the identity

$$E_{\alpha}(t^{\alpha}P_{1}(x)/P_{2}(x)) = \int_{0}^{t} g_{1-\alpha}(t-s)s^{\alpha-1}E_{\alpha,\alpha}(s^{\alpha}P_{1}(x)/P_{2}(x))ds, \quad t > 0, \ x \in \mathbb{R}^{n}$$

(cf. [529, p. 212, 1.4] and the proof of [310, Theorem 2.1]), shows that

$$G_{\alpha,\gamma'}(t) = (g_{1-\alpha} * P_{\alpha,\gamma'})(t), \quad t > 0 \text{ and } G_{\alpha,\beta'}(t) = (g_{1-\alpha} * P_{\alpha,\beta'})(t), \quad t > 0.$$

The proof can be completed routinely by using the estimate (88), the above equalities and the argumentation used in the proofs of Theorem 2.2.20 and [**304**, Theorem 2.1]; see also Remark 2.1.23(i). \Box

It is worth noting that the assertion of Theorem 2.2.29 continues to hold, with appropriate technical modifications, in the case that $E = C_b(\mathbb{R}^n)$ or $E = L^{\infty}(\mathbb{R}^n)$, and that the additional refinement of lower bounds for the numbers γ , γ' , β , β' can be proved following the approach from Remark 2.1.23(ii). Suppose now that T > 0, the requirements of Theorem 2.2.29(i) or Theorem 2.2.29(ii) hold, $C := R_{\alpha,\beta'}(0)$ $(C := R_{\alpha,\beta'}(0))$ in the case of consideration Theorem 2.2.29(i) (Theorem 2.2.29(ii)) and $x \in R(C)$. Following the analyses of non-degenerate case from [**529**] and [**277**] (another way to see that the subsequent definition of a mild solution of the problem $(DFP)_{sl}$ is correct in the case that f(t, u(t)) = f(t) satisfies (P1) is to take the Laplace transform of both sides of the equality $\mathbf{D}_t^{\alpha} \overline{P_2(A)}u(t) = \overline{P_1(A)}u(t) + f(t)$ by making use of the formula (39), on the one hand, and to compute the Laplace transform of the right hand side of equality (89) below, on the other hand), it will be said that a continuous function $t \mapsto u(t), t \in [0, T]$ is a mild solution of the semilinear abstract degenerate Cauchy problem $(DFP)_{sl}$ on [0, T] iff the mapping $t \mapsto C^{-1}f(t, u(t)), t \in [0, T]$ is well-defined and continuous, as well as

(89)
$$u(t) = R_{\alpha}(t)C^{-1}x + \int_{0}^{t} P_{\alpha}(t-s)C^{-1}f(s,u(s))ds, \quad t \in [0,T].$$

Define the operator $Q_{\alpha} \colon C([0,T]:E) \to C([0,T]:E)$ by

$$(Q_{\alpha}u)(t) := R_{\alpha}(t)C^{-1}x + \int_{0}^{t} P_{\alpha}(t-s)C^{-1}f(s,u(s))ds, \quad t \in [0,T].$$

The most common technique to proving existence and uniqueness of mild solutions of semilinear fractional evolution equations is to apply some of the fixed point theorems; in our concrete situation, we must prove that the mapping $Q_{\alpha}(\cdot)$ has a unique fixed point. Not aspiring completeness of analysis here, we shall only state and prove the following adaptation of [445, Theorem 1.2, p. 184] to close the whole section.

THEOREM 2.2.30. Let T > 0, let $x \in R(C)$, and let the requirements of Theorem 2.2.29(i) or Theorem 2.2.29(ii) hold. Put $C := R_{\alpha,\gamma'}(0)$, in the case of Theorem 2.2.29(i), and $C := R_{\alpha,\beta'}(0)$, in the case of Theorem 2.2.29(ii). Suppose that the mapping $C^{-1}f : [0,T] \times E \to E$ is continuous in t on [0,T] and uniformly Lipschitz continuous (with constant L) on E. Then the semilinear fractional Cauchy problem $(DFP)_{sl}$ has a unique mild solution $u \in C([0,T] : E)$. Moreover, the mapping $x \to u(\cdot)$ is Lipschitz continuous from R(C) (endowed with the norm $\|\cdot\|_{R(C)} := \|C^{-1} \cdot\|$, R(C) becomes a Banach space) into C([0,T] : E).

PROOF. Set $M := \max_{t \in (0,T]} (t^{1-\alpha} \Gamma(\alpha) || P_{\alpha}(t) ||)$. Arguing as in the proof of [445, Theorem 1.2, p. 184], we get that, for every $u, v \in C([0,T] : E)$,

$$\|(Q_{\alpha}^{n}u)(t) - (Q_{\alpha}^{n}v)(t)\|_{C([0,T]:E)} \leq \frac{(MLT^{\alpha})^{n}}{\Gamma(n\alpha+1)} \|u - v\|_{C([0,T]:E)}, \quad n \in \mathbb{N}, \ t \in [0,T].$$

For a sufficiently large number $n \in \mathbb{N}$, one has

$$\frac{(MLT^{\alpha})^n}{\Gamma(n\alpha+1)} < 1$$

so that a well known extension of the Banach contraction principle implies that the mapping $Q_{\alpha}(\cdot)$ has a unique fixed point, finishing the proof of existence and uniqueness of mild solutions of problem $(\text{DFP})_{sl}$ on [0,T]. Keeping in mind a Gronwall-type inequality [153, Lemma 6.19, p. 111], the remainder of the proof follows similarly as in that of [445, Theorem 1.2, p. 184].

2.3. Degenerate multi-term fractional differential equations in locally convex spaces

Suppose that $n \in \mathbb{N} \setminus \{1\}$, $0 \leq \alpha_1 < \cdots < \alpha_n$, let A_1, \ldots, A_{n-1} are closed linear operators on a Hausdorff sequentially complete locally convex space E, and $f: [0, \infty) \to E$ is a continuous function. Let us recall (cf. the introductory part) that $m_i := \lceil \alpha_i \rceil$, $i \in \mathbb{N}_{n-1}$, $T_{i,L}u(t) := A_i \mathbf{D}_t^{\alpha_i} u(t)$, if $t \geq 0$, $i \in \mathbb{N}_{n-1}$ and $\alpha_i > 0$, and $T_{i,R}u(t) := \mathbf{D}_t^{\alpha_i} A_i u(t)$, if $t \geq 0$ and $i \in \mathbb{N}_{n-1}$. Henceforth it will be assumed that, for every $t \geq 0$ and $i \in \mathbb{N}_{n-1}$, $T_i u(t)$ denotes either $T_{i,L}u(t)$ or $T_{i,R}u(t)$. As already announced, we will deal with the following degenerate multi-term problem:

(90)
$$\sum_{i=1}^{n-1} T_i u(t) = f(t), \quad t \ge 0.$$

Set

$$P_{\lambda} := \sum_{i=1}^{n-1} \lambda^{\alpha_i - \alpha_{n-1}} A_i, \quad \lambda \in \mathbb{C} \smallsetminus \{0\},$$

 $\mathcal{I} := \{i \in \mathbb{N}_{n-1} : \alpha_i > 0 \text{ and } T_{i,L}u(t) \text{ appears on the left hand side of (90)}\}, Q := \max \mathcal{I}, \text{ if } \mathcal{I} \neq \emptyset \text{ and } Q := m_Q := 0, \text{ if } \mathcal{I} = \emptyset. \text{ Assume that } \alpha > 0 \text{ and } m = \lceil \alpha \rceil. \text{ The following facts can be proved by using the equality (38), induction and closedness of A:}$

- (a) Suppose that $l \in \mathbb{N}$ and $u, Au \in C^{l}([0,\infty) : E)$. Then $u^{(j)}(t) \in D(A)$, $t \ge 0$ and $Au^{(j)}(t) = (Au)^{(j)}(t), t \ge 0$ $(0 \le j \le l)$.
- (b) Suppose that the Caputo fractional derivatives $\mathbf{D}_t^{\alpha} u$ and $\mathbf{D}_t^{\alpha} A u$ are defined. Then $u^{(j)}(t) \in D(A), t \ge 0, A u^{(j)}(t) = (A u)^{(j)}(t), t \ge 0 \ (0 \le j \le m-1), \mathbf{D}_t^{\alpha} u(t) \in D(A), t \ge 0,$

$$A\mathbf{D}_t^{\alpha}u(t) = \mathbf{D}_t^{\alpha}Au(t), \quad t \ge 0,$$

and

$$J_t^{\alpha} A \mathbf{D}_t^{\alpha} u(t) = J_t^{\alpha} \mathbf{D}_t^{\alpha} A u(t) = A u(t) - \sum_{j=0}^{m-1} A u^{(j)}(0) g_{j+1}(t), \quad t \ge 0.$$

Taking into account (a)–(b), it seems reasonable to consider the equation (90) with the initial conditions

(91)
$$u^{(j)}(0) = u_j, \quad 0 \le j \le m_Q - 1 \text{ and } (A_i u)^{(j)}(0) = u_{i,j} \text{ if } m_i - 1 \ge j \ge m_Q.$$

If $T_{n-1}u(t) = T_{n-1,L}u(t)$, then (91) takes the following simple form: $u^{(j)}(0) = u_j$, $0 \leq j \leq m_{n-1} - 1$. If this is not the case, then the choice (91) may be nonoptimal, the index $i \in \mathbb{N}_{n-1}$ has to satisfy the inequality $m_i - 1 \geq m_Q$ in the second equality appearing in (91), and we cannot expect the existence of solutions of problem [(90)-(91)], in general (consider, for example, the case n = 3, $A_2 = A_1$ and $u_{2,0} \neq u_{1,0}$); furthermore, for any index $i \in \mathbb{N}_{n-1}$ satisfying the inequality $m_i - 1 \geq m_Q$ and for every non-negative integer $k \in [m_Q, m_i - 1]$, we need to introduce exactly one initial value $u_{i,k}$. Throughout the section, we assume that the function k(t) is a scalar-valued continuous kernel on $[0, \infty)$. We will use the following fact about Caputo fractional derivatives: Assume that $\alpha > 0$, $m = \lceil \alpha \rceil$, $\beta \in (0, \alpha)$ and the Caputo fractional derivative $\mathbf{D}_t^{\alpha} u(\cdot)$ is defined. Then the Caputo fractional derivative $\mathbf{D}_t^{\beta} u(\cdot)$ is also defined and the following equality holds:

(92)
$$\mathbf{D}_t^{\beta} u(t) = (g_{\alpha-\beta} * \mathbf{D}_t^{\alpha} u(\cdot))(t) + \sum_{j=\lceil\beta\rceil}^{m-1} u^{(j)}(0)g_{j+1-\beta}(t), \quad t \ge 0.$$

2.3.1. Abstract Cauchy problem [(90)-(91)]. Let us recall that $n \in \mathbb{N} \setminus \{1\}, 0 \leq \alpha_1 < \cdots < \alpha_{n-1}$, as well as that A_1, \ldots, A_{n-1} are closed linear operators on E and $f: [0, \infty) \to E$ is a continuous function. Let the set \mathcal{I} and number Q be defined as above.

Suppose, for the time being, that the initial values $u_j \in E$ $(0 \leq j \leq m_Q - 1)$ satisfy $u_j \in D(A_i)$, provided $i \in \mathbb{N}_{n-1}$, $T_i u(t) = T_{i,R} u(t)$ and $0 \leq j \leq m_i - 1$ (put $u_{i,j} := A_i u_j$ in this case), and $u_{i,j} \in E$, provided $i \in \mathbb{N}_{n-1}$, $T_i u(t) = T_{i,R} u(t)$ and $m_i - 1 \geq j \geq m_Q$. We start this section by introducing the notion of a strong solution of problem [(90)-(91)].

DEFINITION 2.3.1. A function $u \in C([0,\infty): E)$ is said to be a strong solution of problem [(90)-(91)] iff the term $T_i u(t)$ is well defined and continuous for any $t \ge 0, i \in \mathbb{N}_{n-1}$, and [(90)-(91)] holds identically on $[0,\infty)$. Now we would like to observe the following fact. If Q > 0, then we can consider the problem obtained from the problem (90) by replacing some of terms $T_{i,R}(t)$, for $1 \leq i \leq Q$, with the corresponding terms of form $T_{i,L}(t)$. By (92) and (b), it readily follows that a strong solution of problem [(90)–(91)] is also a strong solution of the problem described above, when endowed with the initial conditions (91).

Define, for every $i \in \mathbb{N}_{n-1}$ and $t \ge 0$,

$$\mathcal{T}_{i,L}u(t) := g_{\alpha_{n-1}-\alpha_i} * A_i \bigg[u(\cdot) - \sum_{j=0}^{m_i-1} u_j g_{j+1}(\cdot) \bigg](t),$$

if $T_i u(t) = T_{i,L} u(t)$, and

$$\mathcal{T}_{i,R}u(t) := g_{\alpha_{n-1}-\alpha_i} * \left[A_i u(\cdot) - \sum_{j=0}^{m_i-1} u_{i,j} g_{j+1}(\cdot) \right](t),$$

if $T_i u(t) = T_{i,R} u(t)$. Let $\mathcal{T}_i u(t)$ denote exactly one of terms $\mathcal{T}_{i,L} u(t)$ or $\mathcal{T}_{i,R} u(t)$. Integrating the equation (90) α_{n-1} times, the foregoing arguments imply that any strong solution $t \mapsto u(t), t \ge 0$ of problem [(90)–(91)] satisfies the following integral equation

(93)
$$\sum_{i=1}^{n-1} \mathcal{T}_i u(t) = (g_{\alpha_{n-1}} * f)(t), \quad t \ge 0.$$

This motivates us to introduce the following definition.

DEFINITION 2.3.2. Let $u_j \in E$ $(0 \leq j \leq m_Q - 1)$, let $u_{i,j} \in E$, provided $i \in \mathbb{N}_{n-1}, T_i u(t) = T_{i,R} u(t)$ and $0 \leq j \leq m_i - 1$, and let $\mathcal{V} \subseteq \mathbb{N}_{n-1}$. Then a continuous *E*-valued function $t \mapsto u(t), t \geq 0$ is said to be a \mathcal{V} -mild solution of (93) iff (i)–(v) hold, where:

- (i) $g_{\alpha_{n-1}-\alpha_i} * [u(\cdot) \sum_{j=0}^{m_i-1} u_j g_{j+1}(\cdot)](t) \in D(A_i)$ for all $t \ge 0$ and $i \in \mathcal{I} \cap \mathcal{V}$, the mapping $t \mapsto A_i(g_{\alpha_{n-1}-\alpha_i} * [u - \sum_{j=0}^{m_i-1} u_j g_{j+1}])(t), t \ge 0$ is welldefined and continuous for all $i \in \mathcal{I} \cap \mathcal{V}$,
- (ii) the mapping $t \mapsto (g_{\alpha_{n-1}-\alpha_i} * A_i[u \sum_{j=0}^{m_i-1} u_j g_{j+1}])(t), t \ge 0$ is continuous for all $i \in \mathcal{I} \smallsetminus \mathcal{V}$,
- (iii) $(g_{\alpha_{n-1}-\alpha_i}*u)(t) \in D(A_i)$ for all $t \ge 0$ and $i \in (\mathbb{N}_{n-1} \setminus \mathcal{I}) \cap \mathcal{V}$, the mapping $t \mapsto A_i(g_{\alpha_{n-1}-\alpha_i}*u)(t), t \ge 0$ is continuous for all $i \in (\mathbb{N}_{n-1} \setminus \mathcal{I}) \cap \mathcal{V}$,
- (iv) the mapping $t \mapsto (g_{\alpha_{n-1}-\alpha_i} * A_i u)(t), t \ge 0$ is well-defined and continuous for all $i \in (\mathbb{N}_{n-1} \smallsetminus \mathcal{I}) \smallsetminus \mathcal{V}$,
- (v) for every $t \ge 0$, the following holds:

$$(94) \quad \sum_{i \in \mathcal{I} \cap \mathcal{V}} A_i \left(g_{\alpha_{n-1}-\alpha_i} * \left[u(\cdot) - \sum_{j=0}^{m_i-1} u_j g_{j+1}(\cdot) \right] \right)(t) \\ + \sum_{i \in \mathcal{I} \smallsetminus \mathcal{V}} \left(g_{\alpha_{n-1}-\alpha_i} * A_i \left[u(\cdot) - \sum_{j=0}^{m_i-1} u_j g_{j+1}(\cdot) \right] \right)(t)$$

$$+\sum_{i\in(\mathbb{N}_{n-1}\smallsetminus\mathcal{I})\smallsetminus\mathcal{V}}(g_{\alpha_{n-1}-\alpha_{i}}\ast A_{i}u)(t)+\sum_{i\in(\mathbb{N}_{n-1}\smallsetminus\mathcal{I})\cap\mathcal{V}}A_{i}(g_{\alpha_{n-1}-\alpha_{i}}\ast u)(t)$$
$$=\sum_{i\in\mathbb{N}_{n-1}\smallsetminus\mathcal{I}}\sum_{j\in\mathbb{N}_{m_{i-1}}^{0}}g_{\alpha_{n-1}-\alpha_{i}+1+j}(t)u_{i,j}+(g_{\alpha_{n-1}}\ast f)(t), \quad t\ge 0.$$

If $\mathcal{V} = \emptyset$ ($\mathcal{V} = \mathbb{N}_{n-1}$), then we also say that u(t) is a strong (mild) solution of (93).

Any strong solution of problem [(90)-(91)] is also a strong solution of problem (93), and any \mathcal{V} -mild solution of problem (93) is also a \mathcal{V} -mild solution of (93) provided that $\mathcal{V}, \mathcal{V}' \subseteq \mathbb{N}_{n-1}$ and $\mathcal{V} \subseteq \mathcal{V}'$. As already observed in [**308**] for the problem $(DFP)_L$, a sufficiently smooth strong solution of the problem (93) need not be a strong solution of problem [(90)-(91)] in the case that $\mathcal{I} \neq \emptyset$. The situation is quite intricate even if $\mathcal{I} = \emptyset$ because then we can only prove that a strong solution of problem (93) satisfies the equation

$$\sum_{i \in \mathbb{N}_{n-1}} \left(g_{m_{n-1}-m_i} \ast g_{m_i-\alpha_i} \ast \left[A_i u(\cdot) - \sum_{j=0}^{m_i-1} u_{i,j} g_{1+j}(\cdot) \right] \right)(t) = (g_{\alpha_{n-1}} \ast f)(t), \quad t \ge 0,$$

which does not imply, in general, that the function

$$t \mapsto g_{m_i - \alpha_i} * \left[A_i u - \sum_{j=0}^{m_i - 1} u_{i,j} g_{1+j} \right](t), \quad t \ge 0$$

is m_i -times continuously differentiable for $i \in \mathbb{N}_{n-1}$ (the problem (DFP)_R is an exception, cf. [308]). Because of that, we shall primarily consider degenerate integral equation (93) in the sequel.

REMARK 2.3.3. It should be observed that we can further generalize the abstract form of problem (90) by assuming that some of terms $T_iu(t)$ can be expressed as sums of terms like $A'_i \mathbf{D}_t^{\alpha_i}(B'_i \mathbf{D}_t^{\beta_i}u(t))$ and $\mathbf{D}_t^{\alpha_i}A''_i(\mathbf{D}_t^{\beta_i}B''_iu(t))$, with $A'_i, B'_i, A''_i,$ B''_i being closed linear operators on E and $\beta_i \ge 0$ (cf. [490, Chapter VI] for a great number of such examples of degenerate equations of second order possessing a certain physical meaning). It would take too long to go into further details concerning this topic here.

2.3.2. Exponentially equicontinuous k-regularized C-resolvent (i, j)propagation families. Following the method employed in the papers [542, 543]
and [306], we introduce the notion of an exponentially equicontinuous k-regularized
C-resolvent propagation family for problem [(90)–(91)] as follows (cf. the problem
(93) with $\mathcal{I} = \emptyset$, $x = u_{i,j}$, the other initial values being zeroes, and then apply the
formula (39) for the Laplace transform of Caputo derivatives of the α^{th} order).

DEFINITION 2.3.4. Suppose that the function k(t) satisfies (P1), as well as that $1 \leq i \leq n-1$, $0 \leq j \leq m_i - 1$ and $R_{i,j}(t) \colon D(A_i) \to E$ is a linear mapping $(t \geq 0)$. Let the operator $C \in L(E)$ be injective. Then the operator family $(R_{i,j}(t))_{t\geq 0}$ is said to be an exponentially equicontinuous k-regularized C-resolvent (i, j)-propagation family for problem [(90)–(91)] iff there exists $\omega \geq \max(0, \operatorname{abs}(k))$ such that the following holds:

- (i) The mapping $t \mapsto R_{i,j}(t)x, t \ge 0$ is continuous for every fixed element $x \in D(A_i)$.
- (ii) The family $\{e^{-\omega t}R_{i,j}(t):t \ge 0\}$ is equicontinuous, i.e., for every $p \in \circledast$, there exist c > 0 and $q \in \circledast$ such that

$$p(e^{-\omega t}R_{i,j}(t)x) \leq cq(x), \quad x \in D(A_i), \ t \geq 0.$$

(iii) For every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\tilde{k}(\lambda) \neq 0$, the operator P_{λ} is injective, $C(R(A_i)) \subseteq R(P_{\lambda})$ and

(95)
$$\lambda^{\alpha_i - \alpha_{n-1} - j} \tilde{k}(\lambda) P_{\lambda}^{-1} C A_i x = \int_0^\infty e^{-\lambda t} R_{i,j}(t) x \, dt, \quad x \in D(A_i).$$

If $k(t) = g_{r+1}(t)$ for some $r \ge 0$, then it is also said that $(R_{i,j}(t))_{t\ge 0}$ is an exponentially equicontinuous *r*-times integrated *C*-regularized resolvent (i, j)-propagation family for [(90)-(91)]; an exponentially equicontinuous 0-times integrated *C*-regularized resolvent (i, j)-propagation family for [(90)-(91)] is also said to be an exponentially equicontinuous *C*-regularized resolvent (i, j)-propagation family for [(90)-(91)].

Before we state the following important extension of [543, Theorem 3.1], it is worth noting that we do not use here the condition $CA_i \subseteq A_iC$, in contrast to the corresponding definitions from [542,543] and [306], and that the existence of an exponentially equicontinuous k-regularized C-resolvent (i, 0)-propagation family for problem [(90)–(91)] implies the existence of an exponentially equicontinuous k-regularized C-resolvent (i, j)-propagation family for problem [(90)–(91)] $(j \in \mathbb{N}_{m_i-1}^0)$; if this is the case, we have $R_{i,j}(t)x = (g_j * R_{i,0}(\cdot)x)(t), t \ge 0,$ $j \in \mathbb{N}_{m_i-1}^0, x \in D(A_i)$. Observe also that the uniqueness theorem for Laplace transform implies that there exists at most one exponentially equicontinuous kregularized C-resolvent (i, j)-propagation family for problem [(90)–(91)] and that the assertions of [306, Remark 2.3(iv), Proposition 2.4, Theorem 2.5] can be reformulated in our context.

THEOREM 2.3.5. Suppose that $1 \leq i \leq n-1$, $0 \leq j \leq m_i - 1$ and there exists an exponentially equicontinuous k-regularized C-resolvent (i, j)-propagation family $(R_{i,j}(t))_{t\geq 0}$ for problem [(90)-(91)].

- (i) Assume that there exists $l \in \mathbb{N}_{n-1}$ such that the following condition:
 - (C.1) For every $v \in \mathbb{N}_{n-1} \setminus \{l\}$ and $x \in D(A_i)$, there exist a number $\omega_0 > \omega$ and a continuous E-valued function $t \mapsto f_{i,j,v}(t;x)$, $t \ge 0$ such that, for every $p \in \circledast$, there exists $M_p > 0$ with $p(f_{i,j,v}(t;x)) \leqslant M_p e^{\omega t}$, $t \ge 0$ ($v \in \mathbb{N}_{n-1} \setminus \{l\}$) and that, for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega_0$ and $\tilde{k}(\lambda) \ne 0$.

$$\int_0^\infty e^{-\lambda t} f_{i,j,v}(t;x) dt = \lambda^{\alpha_i - \alpha_{n-1} - j + \alpha_v - \alpha_{n-1}} \tilde{k}(\lambda) A_v P_\lambda^{-1} C A_i x$$

holds. Then for each $v_0 \in D(A_i)$ the function $u(t) := R_{i,j}(t)v_0, t \ge 0$ is a mild solution of the integral equation

(96)
$$\sum_{v=1}^{n-1} A_v(g_{\alpha_{n-1}-\alpha_v} * u)(t) = (g_{\alpha_{n-1}-\alpha_i+j} * k)(t)CA_iv_0, \quad t \ge 0,$$

defined in the same way as in Definition 2.3.2(ii).

- (ii) Let $\emptyset \neq \mathcal{K} \subseteq \mathbb{N}_{n-1}$. If the conditions:
 - (C.2) For every $l \in \mathcal{K}$ and $x \in D(A_i)$, and for every $v \in \mathbb{N}_{n-1} \setminus \{l\}$, there exist a number $\omega_{l,v} > \omega$ and a continuous *E*-valued function $t \mapsto g_{i,j,l,v}(t;x), t \ge 0$ such that, for every $p \in \mathfrak{B}$, there exists $M_{p,l,v} > 0$ with $p(g_{i,j,l,v}(t;x)) \le M_{p,l,v}e^{\omega_{l,v}t}, t \ge 0$ $(l \in \mathcal{K}, v \in \mathbb{N}_{n-1} \setminus \{l\})$ and that, for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega_{l,v}$ and $\tilde{k}(\lambda) \ne 0$,

$$\int_0^\infty e^{-\lambda t} g_{i,j,l,v}(t;x) dt = \lambda^{\alpha_i - \alpha_{n-1} - j + \alpha_v - \alpha_l} \tilde{k}(\lambda) A_v P_\lambda^{-1} C A_i x$$

and

(C.3) For every $l \in \mathcal{K}$, there exist a number $\omega_l > \omega$ and a continuous function $h_l: [0, \infty) \to \mathbb{C}$ satisfying (P1) and

$$\widetilde{h}_l(\lambda) = \widetilde{k}(\lambda)\lambda^{\alpha_i - \alpha_l - j}, \quad \operatorname{Re} \lambda > \omega_l,$$

holds, then for each $v_0 \in D(A_i)$ the function $u(t) = R_{i,j}(t)v_0, t \ge 0$ satisfies that the mappings $t \mapsto A_l u(t), t \ge 0$ are well-defined, continuous and that for each $p \in \circledast$ there exist $M_p > 0$ and $\omega_0 > \omega$ with $p(A_l u(t) - h_l(t)CA_i v_0) \le M_p e^{\omega_0 t}, t \ge 0$ ($l \in \mathcal{K}$). Furthermore, for every $t \ge 0$,

(97)
$$\sum_{l\in\mathcal{K}} (g_{\alpha_{n-1}-\alpha_l} * A_l u)(t) + \sum_{l\in\mathbb{N}_{n-1}\smallsetminus\mathcal{K}} A_l (g_{\alpha_{n-1}-\alpha_l} * u)(t)$$
$$= (g_{\alpha_{n-1}-\alpha_i+j} * k)(t)CA_i v_0.$$

- (iii) Suppose that (C.1) holds. Let $v_0 \in \bigcap_{i=1}^{n-1} D(A_i)$, let $\emptyset \neq \mathcal{K} \subseteq \mathbb{N}_{n-1}$, and let $CA_p \subseteq A_pC$ for all $p \in \mathbb{N}_{n-1}$. If the condition:
 - (C.4) For every $l \in \mathcal{K}$ and for every $v \in \mathbb{N}_{n-1} \setminus \{i\}$, there exist a number $\omega_{l,v} > \omega$ and a continuous function $h_{l,v} : [0, \infty) \to E$ satisfying that, for every $p \in \circledast$, there exists $M_{p,l,v} > 0$ with $p(h_{l,v}(t)) \leq M_{p,l,v} e^{\omega_{l,v}t}$, $t \geq 0$ and

$$\widetilde{h_{l,v}}(\lambda) = \tilde{k}(\lambda)\lambda^{\alpha_v - \alpha_{n-1} - j}A_l P_{\lambda}^{-1}CA_v v_0, \quad \operatorname{Re} \lambda > \omega_{l,v}, \ \tilde{k}(\lambda) \neq 0$$

holds, then the function $u(t) = R_{i,j}(t)v_0$, $t \ge 0$ satisfies that the mappings $t \mapsto A_l u(t)$, $t \ge 0$ are well-defined, continuous and that for each $p \in \circledast$ there exist $M_p > 0$ and $\omega_0 > \omega$ with $p(A_l u(t) - (g_j * k)(t)CA_l v_0) \le M_p e^{\omega_0 t}$, $t \ge 0$ $(l \in \mathcal{K})$. Furthermore, for every $t \ge 0$, (97) holds.

(iv) Suppose that $CA_p \subseteq A_pC$, $p \in \mathbb{N}_{n-1}$ and k(t) satisfies (P2), as well as that n = 3 or that $n \ge 4$ and the following condition holds:

(C.5) For every $p \in \circledast$ and $l \in \mathbb{N}_{n-1} \setminus \{i\}$, there exist numbers $\lambda_{p,l}, \sigma_{p,l} > 0$, a seminorm $q_{p,l} \in \circledast$ and a function $h_{p,l}: (\lambda_{p,l}, \infty) \to (0, \infty)$ such that:

$$p(P_{\lambda}^{-1}CA_{l}x) \leq [q_{p,l}(x) + q_{p,l}(A_{l}x)]h_{p,l}(\lambda), \quad \lambda > \lambda_{p,l}, \ x \in D(A_{l}),$$

and

$$\lim_{\lambda \to +\infty} e^{-\lambda \sigma_{p,l}} h_{p,l}(\lambda) = 0.$$

Then the function $u(t) = R_{i,j}(t)v_0$, $t \ge 0$ is a unique mild solution of the integral equation (96), provided that $v_0 \in D(A_i)$ and the assumptions of (i) hold. Furthermore, the function $u(t) = R_{i,j}(t)v_0$, $t \ge 0$ is a unique function satisfying that the mapping $t \mapsto A_lu(t)$, $t \ge 0$ is well-defined, continuous $(l \in \mathcal{K})$ and that (97) holds, provided that $v_0 \in D(A_i)$ and the assumptions of (ii) hold, resp. $v_0 \in \bigcap_{i=1}^{n-1} D(A_i)$ and the assumptions of (iii) hold.

PROOF. Let $v_0 \in D(A_i)$. Due to the condition (C.1) and the uniqueness theorem for Laplace transform, we have that the function $t \mapsto A_v(g_{\alpha_{n-1}-\alpha_v} * R_{i,j}(\cdot)v_0)(t), t \ge 0$ is well-defined, continuous and for each $p \in \circledast$ there exist $M'_p > 0$ and $\omega' > \omega$ with $p(A_v(g_{\alpha_{n-1}-\alpha_v} * R_{i,j}(\cdot)v_0)(t)) \le M'_p e^{\omega' t}, t \ge 0$ ($v \in \mathbb{N}_{n-1} \setminus \{l\}$); furthermore,

$$\int_0^\infty e^{-\lambda t} A_v(g_{\alpha_{n-1}-\alpha_v} * R_{i,j}(\cdot)v_0)(t)dt = \tilde{k}(\lambda)\lambda^{\alpha_i-\alpha_{n-1}-j+\alpha_v-\alpha_{n-1}}A_vP_\lambda^{-1}CA_iv_0,$$

for any $v \in \mathbb{N}_{n-1} \setminus \{l\}$ and for any $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega'$ and $\tilde{k}(\lambda) \neq 0$. Using the identity

(98)
$$\tilde{k}(\lambda)\lambda^{\alpha_i-\alpha_{n-1}-j+\alpha_l-\alpha_{n-1}}A_lP_{\lambda}^{-1}CA_iv_0$$
$$=\tilde{k}(\lambda)\lambda^{\alpha_i-\alpha_{n-1}-j}\bigg[CA_iv_0-\sum_{v\in\mathbb{N}_{n-1}\smallsetminus\{l\}}\lambda^{\alpha_v-\alpha_{n-1}}A_vP_{\lambda}^{-1}CA_iv_0\bigg],$$

for any $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega'$ and $\tilde{k}(\lambda) \neq 0$, and the uniqueness theorem for Laplace transform, it readily follows that the function $t \mapsto A_l(g_{\alpha_{n-1}-\alpha_l} * R_{i,j}(\cdot)v_0)(t), t \geq 0$ is well-defined, continuous and

$$\begin{aligned} A_{l}(g_{\alpha_{n-1}-\alpha_{l}}*R_{i,j}(\cdot)v_{0})(t) &= (g_{\alpha_{n-1}-\alpha_{i}+j}*k)(t)CA_{i}v_{0} \\ &- \sum_{v \in \mathbb{N}_{n-1} \smallsetminus \{l\}} A_{l}(g_{\alpha_{n-1}-\alpha_{v}}*R_{i,j}(\cdot)v_{0})(t), \quad t \ge 0, \end{aligned}$$

proving that the function $u(t) = R_{i,j}(t)v_0, t \ge 0$ is a mild solution of the integral equation (96). Suppose now that the conditions (C.2)–(C.3) hold, as well as $v_0 \in$ $D(A_i)$ and $\emptyset \ne \mathcal{K} \subseteq \mathbb{N}_{n-1}$. Clearly, (C.2) implies (C.1) with any $l \in \mathcal{K}$. Similarly as in the proof of (i), the conditions (C.2)–(C.3) in combination with the equation (98), multiplied by $\lambda^{\alpha_{n-1}-\alpha_l}$, imply that there exists a sufficiently large number $\omega_1' > \omega$ such that

$$A_l \int_0^\infty e^{-\lambda t} R_{i,j}(t) v_0 dt = \widetilde{h}_l(\lambda) C A_i v_0 - \int_0^\infty e^{-\lambda t} \sum_{v \in \mathbb{N}_{n-1} \smallsetminus \mathcal{K}} g_{i,j,l,v}(t;v_0) dt,$$

for any $l \in \mathcal{K}$ and for any $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega'_l$. Then we can use the assertion (i) and the uniqueness theorem for Laplace transform to complete the proof of (ii). In order to prove (iii), observe first that the assumptions $v_0 \in \bigcap_{i=1}^{n-1} D(A_i)$ and $CA_p \subseteq A_pC, p \in \mathbb{N}_{n-1}$ imply

$$(99) P_{\lambda}^{-1}C(\lambda^{\alpha_{i}-\alpha_{n-1}}A_{i}v_{0}) + \sum_{v \in \mathbb{N}_{n-1} \smallsetminus \{i\}} P_{\lambda}^{-1}C(\lambda^{\alpha_{v}-\alpha_{n-1}}A_{v}v_{0}) = P_{\lambda}^{-1}CP_{\lambda}v_{0} = Cv_{0},$$

provided $\operatorname{Re} \lambda > \omega$ and $\tilde{k}(\lambda) \neq 0$. Making use of (95) and (99), we obtain that, for any such a value of complex parameter λ , the following holds:

$$A_l \int_0^\infty e^{-\lambda t} R_{i,j}(t) v_0 dt = \lambda^{-j} \tilde{k}(\lambda) A_l P_\lambda^{-1} C(\lambda^{\alpha_i - \alpha_{n-1}} A_i v_0)$$
$$= \lambda^{-j} \tilde{k}(\lambda) A_l \bigg[C v_0 - \sum_{v \in \mathbb{N}_{n-1} \smallsetminus \{i\}} \lambda^{\alpha_v - \alpha_{n-1}} P_\lambda^{-1} C A_v v_0 \bigg].$$

Keeping in mind the last equation, as well as the condition (C.4) and the uniqueness theorem for Laplace transform, the proof of (iii) follows instantly. We will prove the uniqueness of solutions in (iv) only in the case that $v_0 \in D(A_i)$ and the assumptions of (i) hold. Let $t \mapsto u(t)$, $t \ge 0$ be a mild solution of the integral equation (96) with $v_0 = 0$. Convoluting the function $u(\cdot)$ with $g_{\xi}(\cdot)$, for a sufficiently large number $\xi > 0$, we may assume without of generality that, for every $v \in \mathbb{N}_{n-1}$, the mapping $t \mapsto A_v u(t)$, $t \ge 0$ is well-defined and continuous. Set, for every $t \ge 0$ and $\zeta > 0$, $v_{t,\zeta}(\lambda) := (g_{\zeta} * e^{\lambda \cdot})(t) - \lambda^{-\zeta} e^{t\lambda}, \lambda > 0$; $v_{t,0}(\lambda) := 0$ ($t \ge 0, \lambda > 0$). Then the mapping $t \mapsto v_{t,\zeta}(\lambda)$ is continuous in $t \ge 0$, for any fixed numbers $\zeta \ge 0$ and $\lambda > 0$, and by [**541**, Lemma 1.5.5, p. 23], there exists $M \ge 1$ such that the mapping $\lambda \mapsto v_{t,\zeta}(\lambda), \lambda > 0$ satisfies

(100)
$$|v_{t,\zeta}(\lambda)| \leq M[(1+t)^{\zeta-1}\lambda^{-1}(1+\lambda^{1-\zeta})+t^{\zeta-1}\lambda^{-1}], \quad \lambda > 0, \ t > 0, \ \zeta > 0.$$

Keeping in mind that $CA_p \subseteq A_pC$, $p \in \mathbb{N}_{n-1}$, we have that, for every $t \ge 0$ and $\lambda > 0$,

$$\begin{split} \lambda^{\alpha_i - \alpha_{n-1}} \int_0^t e^{\lambda(t-s)} A_i Cu(s) ds &+ \int_0^t v_{t-s,\alpha_{n-1} - \alpha_i}(\lambda) A_i Cu(s) ds \\ &= C \int_0^t e^{\lambda(t-s)} (g_{\alpha_{n-1} - \alpha_i} * A_i u)(s) ds \\ &= (-C) \sum_{v \in \mathbb{N}_{n-1} \smallsetminus \{i\}} \int_0^t e^{\lambda(t-s)} (g_{\alpha_{n-1} - \alpha_v} * A_v u)(s) ds \\ &= -\sum_{v \in \mathbb{N}_{n-1} \smallsetminus \{i\}} \left[\lambda^{\alpha_v - \alpha_{n-1}} \int_0^t e^{\lambda(t-s)} A_v Cu(s) ds + \int_0^t v_{t-s,\alpha_{n-1} - \alpha_v}(\lambda) A_v Cu(s) ds \right], \end{split}$$

which clearly implies that, for every $\lambda > \omega$, $\sigma > 0$ and $t \ge 0$, the following holds:

(101)
$$\lambda^{\alpha_{i}-\alpha_{n-1}-j}\tilde{k}(\lambda)e^{-\lambda\sigma}\int_{0}^{t}e^{\lambda(t-s)}Cu(s)ds$$
$$=-\lambda^{\alpha_{i}-\alpha_{n-1}-j}\tilde{k}(\lambda)e^{-\lambda\sigma}P_{\lambda}^{-1}C\int_{0}^{t}v_{t-s,\alpha_{n-1}-\alpha_{i}}(\lambda)A_{i}u(s)ds$$
$$-\lambda^{\alpha_{i}-\alpha_{n-1}-j}\tilde{k}(\lambda)e^{-\lambda\sigma}\sum_{v\in\mathbb{N}_{n-1}\smallsetminus\{i\}}P_{\lambda}^{-1}C\int_{0}^{t}v_{t-s,\alpha_{n-1}-\alpha_{v}}(\lambda)A_{v}u(s)ds.$$

By (95) and (101), we obtain that, for every $\lambda > \omega$, $\sigma > 0$ and $t \ge 0$, the following holds:

$$e^{-\lambda\sigma} \int_0^t e^{\lambda(t-s)} Cu(s) ds$$

= $-\frac{\lambda^{\alpha_{n-1}+j-\alpha_i}e^{-\lambda\sigma}}{\tilde{k}(\lambda)} \int_0^\infty e^{-\lambda s} R_{i,j}(s) \left(\int_0^t v_{t-r,\alpha_{n-1}-\alpha_i}(\lambda)u(r)dr\right) ds$
 $-e^{-\lambda\sigma} \sum_{v \in \mathbb{N}_{n-1} \smallsetminus \{i\}} P_{\lambda}^{-1} CA_v \int_0^t v_{t-s,\alpha_{n-1}-\alpha_v}(\lambda)u(s) ds.$

For the estimation of the first addend on the right hand side of the above equality, we can use the already employed fact that there exist numbers $\sigma_0 > 0$ and $M' \ge 1$ such that

(102)
$$\frac{e^{-\lambda\sigma_0}}{|\tilde{k}(\lambda)|} \leqslant M', \quad \lambda > \omega + 1.$$

Keeping in mind (100) and (102), it can be simply proved that, for every $\sigma > \sigma_0$ and for every $p \in \mathfrak{B}$, we have

(103)
$$\lim_{\lambda \to +\infty} p\left(\frac{\lambda^{\alpha_{n-1}+j-\alpha_i}e^{-\lambda\sigma}}{\tilde{k}(\lambda)} \int_0^\infty e^{-\lambda s} R_{i,j}(s) \left(\int_0^t v_{t-r,\alpha_{n-1}-\alpha_i}(\lambda)u(r)dr\right)ds\right) = 0.$$

If $n \ge 4$, then the condition (C.5) in combination with the previous equality and (100) shows that, for every $p \in \circledast$, there exists a sufficiently large number $\sigma_p > 0$ such that $\lim_{\lambda \to +\infty} e^{-\lambda \sigma_p} p((e^{\lambda} * Cu)(t)) = 0, t \ge 0$; the same holds in the case that n = 3 because then we can use, instead of condition (C.5), the equation (99) and the arguments already seen in proving the equation (103), to conclude that

$$\lim_{\lambda \to +\infty} p\left(e^{-\lambda\sigma} \sum_{v \in \mathbb{N}_{n-1} \smallsetminus \{i\}} P_{\lambda}^{-1} C A_v \int_0^t v_{t-s,\alpha_{n-1}-\alpha_v}(\lambda) v(s) ds\right) = 0,$$

for any $\sigma > \sigma_0$ and $t \ge 0$. In such a way, we obtain that for each $p \in \circledast$ the following holds: $\lim_{\lambda \to +\infty} \int_0^t e^{\lambda(t-s-\sigma)} Cu(s) ds = 0, t \ge 0, \sigma > \sigma_p$. By the dominated convergence theorem, it readily follows that for each $p \in \circledast$ we have: $\lim_{\lambda \to +\infty} p(\int_0^{t-\sigma} e^{\lambda(t-s-\sigma)} Cu(s) ds) = 0, t \ge \sigma > \sigma_p$. Therefore,

$$\lim_{\lambda \to +\infty} \int_0^t e^{\lambda(t-s)} C u(s) ds = 0, \quad t \ge 0.$$

Since C is injective, we can apply [**292**, Lemma 2.1.33(iii)] (cf. [**445**, Lemma 1.4.4, p. 100] for the Banach space case) to complete the proof. \Box

The uniqueness of solutions of integral equation (96), resp. (97), can be proved even in the case of non-existence of an exponentially equicontinuous k-regularized C-resolvent (i, j)-propagation family for problem [(90)-(91)]. Strictly speaking, the proof of Theorem 2.3.5 implies the following uniqueness type theorem for degenerate multi-term problems (cf. [197, Theorem 3.1] for a pioneering result on the uniqueness of solutions to abstract degenerate first-order equations):

THEOREM 2.3.6. Suppose that $CA_p \subseteq A_pC$ for all $p \in \mathbb{N}_{n-1}$, $\mathcal{V} \subseteq \mathbb{N}_{n-1}$ and the requirements in (C.5) hold for every seminorm $p \in \mathfrak{B}$ and for every number $l \in \mathbb{N}_{n-1}$. Then there exists at most one mild solution $t \mapsto u(t), t \ge 0$ of the integral equation (96) with $v_0 = 0$, resp. there exists at most one continuous *E*valued function $t \mapsto u(t), t \ge 0$ satisfying that the mapping $t \mapsto A_l u(t), t \ge 0$ is well-defined, continuous $(l \in \mathcal{K})$ and (97) holds with $v_0 = 0$. In particular, there exists at most one \mathcal{V} -mild solution of problem (93) and there exists at most one strong solution of problem [(90)-(91)].

REMARK 2.3.7. Suppose again that the general assumptions of Theorem 2.3.5 hold, i.e., that $1 \leq i \leq n-1$, $0 \leq j \leq m_i - 1$ and there exists an exponentially equicontinuous k-regularized C-resolvent (i, j)-propagation family $(R_{i,j}(t))_{t\geq 0}$ for problem [(90)-(91)].

- (i) Suppose that k(t) satisfies (P2) and, for every $l \in \mathbb{N}_{n-1} \setminus \{i\}$, there exists $j_l \in \mathbb{N}_{m_l-1}^0$ such that there exists an exponentially equicontinuous k-regularized C-resovent (l, j_l) -propagation family for problem [(90)–(91)]. By the proof of Theorem 2.3.5(iv), we have that the condition (C.5) automatically holds.
- (ii) The uniqueness of solutions of non-degenerate integral equations has recently been considered in [347]. It ought to be observed that we must impose the additional condition $CA_p \subseteq A_pC$, $p \in \mathbb{N}_{n-1}^0$ in the formulation of Theorem 3.2 in [347] in order for its proof to work.
- (iii) Let $\emptyset \neq \mathcal{K} \subseteq \mathbb{N}_{n-1}$. Suppose that $v_0 \in D(A_i)$ and the conditions (C.2)–(C.3) hold, or $v_0 \in \bigcap_{i=1}^{n-1} D(A_i)$, $CA_p \subseteq A_pC$ for all $p \in \mathbb{N}_{n-1}$, and the conditions (C.1) and (C.4) hold. Let u(t) be the solution of (97), satisfying the properties stated above. Consider now the equation (93) and the notion introduced in Definition 2.3.2 with indexes i, j replaced by i', j'. Then the following holds:
 - (a) If $i \in \mathbb{N}_{n-1} \setminus \mathcal{I}$, k(t) = 1, $u_{j'} = 0$ $(0 \leq j' \leq m_{i'} 1)$, $u_{i',j'} = CA_i v_0$, provided i' = i and j' = j, and $u_{i',j'} = 0$, otherwise, then u(t) is a $(\mathbb{N}_{n-1} \setminus \mathcal{K})$ -mild solution of (93) with f(t) = 0.
 - (b) If $i \in \mathcal{I}$, k(t) = 1, $u_{i',j'} = 0$ $(i' \in \mathbb{N}_{n-1}, j' \in \mathbb{N}_{m_{i'}-1}^0)$, $u_{j'} = Cv_0$, provided j' = j, $u_{j'} = 0$, otherwise, and $CA_i \subseteq A_iC$, then u(t) is a $(\mathbb{N}_{n-1} \smallsetminus \mathcal{K})$ -mild solution of (93) with f(t) = 0, provided that for each $i' \in \{s \in \mathcal{I} \smallsetminus \{i\} : m_s - 1 \ge j\}$ one has $A_{i'}Cv_0 = 0$.

(iv) Making use of [541, Theorem 1.1.9], the uniqueness theorem for Laplace transform and the formula (39), we can clarify some sufficient conditions for the existence of terms $A_p \mathbf{D}_t^{\alpha_p} u(t)$ and $\mathbf{D}_t^{\alpha_p} A_p u(t)$ $(p \in \mathbb{N}_{n-1})$. Unfortunately, it is very hard to verify these conditions in practical situations because we do not know the precise values of elements $R_{i,j}(0)x$, $R'_{i,j}(0)x, \ldots, (x \in D(A_i))$.

The notion of an exponentially equicontinuous (equicontinuous), analytic k-regularized C-resolvent (i, j)-propagation family $(R_{i,j}(t))_{t\geq 0}$ for problem [(90)–(91)] is introduced in the following definition.

DEFINITION 2.3.8. Suppose that $1 \leq i \leq n-1$, $0 \leq j \leq m_i - 1$, $0 < \alpha \leq \pi$ and there exists an exponentially equicontinuous k-regularized C-resolvent (i, j)propagation family $(R_{i,j}(t))_{t\geq 0}$ for problem [(90)-(91)]. Suppose, further, that the function k(t) satisfies (P1), as well as $C \in L(E)$ is an injective mapping. Then it is said that $(R_{i,j}(t))_{t\geq 0}$ is an exponentially equicontinuous (equicontinuous), analytic k-regularized C-resolvent (i, j)-propagation family $(R_{i,j}(t))_{t\geq 0}$ for problem [(90)-(91)], of angle α , iff the following holds:

- (i) For every $x \in D(A_i)$, the mapping $t \mapsto R_{i,j}(t)x$, t > 0 can be analytically extended to the sector Σ_{α} ; since no confusion seems likely, we denote the extension by the same symbol.
- (ii) For every $x \in D(A_i)$ and $\beta \in (0, \alpha)$, one has $\lim_{z\to 0, z\in\Sigma_{\beta}} R_{i,j}(z)x = R_{i,j}(0)x$.
- (iii) For every $\beta \in (0, \alpha)$, there exists $\omega_{\beta} \ge \max(0, \operatorname{abs}(k))$ ($\omega_{\beta} = 0$) such that the family $\{e^{-\omega_{\beta}z}R_{i,j}(z): z \in \Sigma_{\beta}\}$ is equicontinuous, i.e., for every $p \in \mathfrak{B}$, there exist c > 0 and $q \in \mathfrak{B}$ such that

$$p(e^{-\omega_{\beta}z}R_{i,j}(z)x) \leqslant cq(x), \quad x \in D(A_i), \ z \in \Sigma_{\beta}.$$

The proof of following theorem can be deduced similarly as that of [296, Theorem 3.7].

THEOREM 2.3.9. Assume that the function k(t) satisfies (P1), $1 \leq i \leq n-1$, $0 \leq j \leq m_i - 1$, $\omega \geq \max(0, \operatorname{abs}(k))$, $\alpha \in (0, \pi/2]$ and the operator $C \in L(E)$ is injective. Assume, further, that for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\tilde{k}(\lambda) \neq 0$, we have that the operator P_{λ} is injective and $C(R(A_i)) \subseteq R(P_{\lambda})$. Let for each $x \in D(A_i)$ there is an analytic function $q_x \colon \omega + \Sigma_{\frac{\pi}{2} + \alpha} \to E$ such that

$$q_x(\lambda) = \lambda^{\alpha_i - \alpha_{n-1} - j} \tilde{k}(\lambda) P_{\lambda}^{-1} C A_i x, \quad \operatorname{Re} \lambda > \omega, \ \tilde{k}(\lambda) \neq 0$$

Suppose that, for every $\beta \in (0, \alpha)$ and $p \in \circledast$, there exist $c_{p,\beta} > 0$ and $r_{p,\beta} \in \circledast$ such that $p((\lambda - \omega)q_x(\lambda)) \leq c_{p,\beta}r_{p,\beta}(x), x \in D(A_i), \lambda \in \omega + \Sigma_{\beta+(\pi/2)}$ and, for every $x \in D(A_i)$, there exists the limit $\lim_{\lambda \to +\infty} \lambda q_x(\lambda)$ in E. Then there exists an exponentially equicontinuous k-regularized C-resolvent (i, j)-propagation family $(R_{i,j}(t))_{t\geq 0}$ for problem [(90)-(91)], of angle α , and for each $\beta \in (0, \alpha)$ the family $\{e^{-\omega z}R_{i,j}(z): z \in \Sigma_{\beta}\}$ is equicontinuous.

Differential properties of exponentially equicontinuous (analytic) k-regularized C-resolvent (i, j)-propagation families in locally convex spaces can be clarified following the method employed in Section 2.2. In the following theorem, we state the

subordination principle for exponentially equicontinuous k-regularized C-resolvent (i, j)-propagation families.

THEOREM 2.3.10. Suppose that $1 \leq i \leq n-1$, $0 \leq j \leq m_i - 1$, $0 < \gamma < 1$, $0 \leq j' \leq m_i - 1$, and there exists an exponentially equicontinuous k-regularized C-resolvent (i, j)-propagation family $(R_{i,j}(t))_{t\geq 0}$ for problem [(90)-(91)] satisfying that the family $\{e^{-\omega t}R_{i,j}(t):t\geq 0\}$ is equicontinuous for some $\omega \geq \max(0, \operatorname{abs}(k))$. Assume that k(t) satisfies (P1) and there exist a scalar-valued continuous kernel $k_{\gamma}(t)$ on $[0, \infty)$, satisfying (P1), and a positive real number $\eta > 0$ such that

$$\widetilde{k_{\gamma}}(\lambda) = \lambda^{\gamma - 1 + j' - \gamma j} \tilde{k}(\lambda^{\gamma}), \quad \lambda > \eta.$$

Then there exists an exponentially equicontinuous k-regularized C-resolvent (i, j')propagation family $(R_{i,j',\gamma}(t))_{t\geq 0}$ for problem [(90)-(91)], with α_i replaced by $\gamma\alpha_i$ $(i \in \mathbb{N}_{n-1})$, and $(R_{i,j',\gamma}(t))_{t\geq 0}$ is given by $R_{i,j',\gamma}(0) := R_{i,j}(0)$,

$$R_{i,j',\gamma}(t)x := \int_0^\infty t^{-\gamma} \Phi_\gamma(st^{-\gamma}) R_{i,j}(s) x \, ds, \quad x \in D(A_i), \ t > 0.$$

Furthermore, the family $\{e^{-\omega^{1/\gamma t}}R_{i,j',\gamma}(t):t \ge 0\}$ is equicontinuous and, for every $\zeta \ge 0$, the equicontinuity of the family $\{e^{-\omega t}(1+t^{\zeta})^{-1}R_{i,j}(t):t\ge 0\}$, resp. $\{e^{-\omega t}t^{-\zeta}R_{i,j}(t):t>0\}$, implies the equicontinuity of the family $\{e^{-\omega^{1/\gamma t}}(1+t^{\gamma\zeta})^{-1}(1+\omega t^{\zeta(1-\gamma)})^{-1}R_{i,j',\gamma}(t):t\ge 0\}$, resp. $\{e^{-\omega^{1/\gamma t}}t^{-\gamma\zeta}(1+\omega t^{\zeta(1-\gamma)})^{-1}R_{i,j',\gamma}(t):t\ge 0\}$, and the following holds:

- (i) The mapping $t \mapsto R_{i,j',\gamma}(t)x$, t > 0 admits an analytic extension to the sector $\sum_{\min((\frac{1}{x}-1)\frac{\pi}{2},\pi)}$ for all $x \in D(A_i)$.
- (ii) If $\omega = 0$ and $\varepsilon \in (0, \min((\frac{1}{\gamma} 1)\frac{\pi}{2}, \pi))$, then the family $\{R_{i,j',\gamma}(z) : z \in \Sigma_{\min((\frac{1}{\gamma} 1)\frac{\pi}{2}, \pi) \varepsilon}\}$ is equicontinuous and $\lim_{z \to 0, z \in \Sigma_{\min((\frac{1}{\gamma} 1)\frac{\pi}{2}, \pi) \varepsilon}} R_{i,j',\gamma}(z)x = R_{i,j',\gamma}(0)x$ for all $x \in D(A_i)$.
- (iii) If $\omega > 0$ and $\varepsilon \in (0, \min((\frac{1}{\gamma} 1)\frac{\pi}{2}, \frac{\pi}{2}))$, then there exists $\delta_{\gamma,\varepsilon} > 0$ such that the family $\{e^{-\delta_{\gamma,\varepsilon} \operatorname{Re} z} R_{i,j',\gamma}(z) : z \in \Sigma_{\min((\frac{1}{\gamma} 1)\frac{\pi}{2}, \frac{\pi}{2}) \varepsilon}\}$ is equicontinuous. Moreover, $\lim_{z \to 0, z \in \Sigma_{\min((\frac{1}{\gamma} 1)\frac{\pi}{2}, \frac{\pi}{2}) \varepsilon}} R_{i,j',\gamma}(z)x = R_{i,j',\gamma}x$ for all $x \in D(A_i)$.

REMARK 2.3.11. Using the proof of [61, Theorem 3.1], it can be simply verified that any of conditions (C.1)–(C.5) is invariant under the action of subordination principle described in Theorem 2.3.10.

EXAMPLE 2.3.12. (cf. also [346, Example 5.1(i)]) Suppose that $c_l \in \mathbb{C} \setminus 0$ $(1 \leq l \leq n-1)$, as well as A and B are closed linear operators on E, and $A_l = c_l B$ for $1 \leq l \leq n-1$. We consider the following degenerate multi-term problem:

(104)
$$\mathbf{D}_t^{\alpha_n} Bu(t) + \sum_{l=1}^{n-1} c_l \mathbf{D}_t^{\alpha_l} Bu(t) = \mathbf{D}_t^{\alpha} Au(t), \quad t \ge 0,$$

equipped with the initial conditions of the form (91). Here $0 \leq \alpha_1 < \cdots < \alpha_n$, $0 \leq \alpha < \alpha_n$, and

$$P_{\lambda} = \sum_{l=1}^{n-1} c_l \lambda^{\alpha_l - \alpha_n} B - \lambda^{\alpha - \alpha_n} A + B, \quad \lambda \in \mathbb{C} \smallsetminus \{0\}.$$

(i) (a) Suppose $0 < \delta \leq 2, \ \sigma \geq 1, \ \frac{\pi\delta}{2(\alpha_n - \alpha)} - \frac{\pi}{2} > 0, \ 0 \leq j \leq \lceil \alpha_n \rceil - 1,$ and there exists an exponentially equicontinuous $(\sigma - 1)$ -times integrated *C*-resolvent propagation family $(R(t))_{t\geq 0}$ for problem (52), with $a(t) = g_{\delta}(t)$. Put $\sigma' := \max(1, 1 - j + (\alpha_n - \alpha)(\sigma - 1)\delta^{-1})$ and $\theta := \min(\frac{\pi}{2}, \frac{\pi\delta}{2(\alpha_n - \alpha)} - \frac{\pi}{2})$. Then, for every sufficiently small number $\varepsilon > 0$, there exists $\omega_{\varepsilon} > 0$ such that $C(R(B)) \subseteq R(P_{\lambda})$ for all $\lambda \in \omega_{\varepsilon} + \Sigma_{\frac{\pi}{2}\delta - \varepsilon}$ and the family $\{|\lambda|^{\frac{\delta - \sigma}{\delta}}(1 + |\lambda|^{\frac{1}{\delta}})(\lambda B - A)^{-1}CBx : \lambda \in \omega_{\varepsilon} + \Sigma_{\frac{\pi}{2}\delta - \varepsilon}, \ x \in D(B)\}$ is equicontinuous. Noticing also that

$$\arg\left(\lambda^{\alpha_n-\alpha}+\sum_{l=1}^{n-1}c_l\lambda^{\alpha_l-\alpha}\right)\approx(\alpha_n-\alpha)\arg(\lambda),\quad\lambda\to\infty,\ \arg(\lambda)<\frac{\pi}{\alpha_n-\alpha},$$

our choice of θ implies that, for every sufficiently small number $\varepsilon > 0$, there exists $\omega_{\varepsilon}' > 0$ such that, for every $\lambda \in \omega_{\varepsilon}' + \Sigma_{\frac{\pi}{2} + \theta - \varepsilon}$, one has:

$$\lambda^{\alpha_n - \alpha} + \sum_{l=1}^{n-1} c_l \lambda^{\alpha_l - \alpha} \in \omega_{\varepsilon} + \sum_{\frac{\pi}{2}\delta - \varepsilon}.$$

Put now, for every $x \in D(B)$ and $\lambda \in \omega_{\varepsilon}' + \sum_{\frac{\pi}{2} + \theta - \varepsilon}$,

$$q_x(\lambda) := \lambda^{-j-\sigma'} P_{\lambda}^{-1} CBx.$$

Then $q_x: \omega'_{\varepsilon} + \sum_{\frac{\pi}{2}+\theta-\varepsilon} \to E$ is an analytic function and, for every $\beta \in (0,\theta)$ and $p \in \circledast$, there exist $c_{p,\beta} > 0$ and $r_{p,\beta} \in \circledast$ such that $p((\lambda - \omega'_{\varepsilon})q_x(\lambda)) \leqslant c_{p,\beta}r_{p,\beta}(x), x \in D(B), \lambda \in \omega'_{\varepsilon} + \sum_{\frac{\pi}{2}+\theta-\varepsilon}$. By the proof of [27, Proposition 4.1.3, p. 248], we have that $\lim_{\operatorname{Re}\lambda\to+\infty} \lambda^{\delta-\sigma+1}(\lambda^{\delta}B - A)^{-1}CBx = R_{i,j}(0)x, x \in D(B)$, which simply implies that, for every $x \in D(B)$, there exists the limit $\lim_{\lambda\to+\infty} \lambda q_x(\lambda)$ in E. Therefore, Theorem 2.3.9 implies that there exists an exponentially equicontinuous, analytic $(\sigma'-1)$ -times integrated C-resolvent (n, j)-propagation family $(R_{n,j}(t))_{t\geq 0}$ for problem (104), of angle θ (with the clear meaning).

(b) Suppose $0 < \delta \leq 2$, $\sigma \geq 1$, $0 \leq j \leq \lceil \alpha_n \rceil - 1$, $\gamma \in (0, \frac{\pi}{2}]$ and $\frac{\delta(\frac{\pi}{2}+\gamma)}{(\alpha_n-\alpha)} - \frac{\pi}{2} > 0$. Put $\sigma_1 := \sigma'$ and $\theta_1 := \min(\frac{\pi}{2}, \frac{\delta(\frac{\pi}{2}+\gamma)}{(\alpha_n-\alpha)} - \frac{\pi}{2})$. Arguing similarly as in (a), one can prove the following: Suppose that for each $\varepsilon \in (0, \frac{\pi}{2} + \gamma)$ there exists $\omega_{\varepsilon} > 0$ such that for each $x \in D(B)$ there exists an analytic function $q_x : \omega_{\varepsilon} + \Sigma_{\frac{\pi}{2}+\gamma-\varepsilon} \to E$ satisfying that

$$q_x(\lambda) = \lambda^{\delta - \sigma} (\lambda^{\delta} B - A)^{-1} C B x, \quad \lambda \in \omega_{\varepsilon} + \sum_{\frac{\pi}{2} + \gamma - \varepsilon}, \ x \in D(B)$$

and for each $p \in \circledast$ there exist $c_p > 0$ and $q_p \in \circledast$ so that

$$p(q_x(\lambda)) \leqslant c_p \frac{q_p(x)}{1+|\lambda|}, \quad \lambda \in \omega_{\varepsilon} + \sum_{\frac{\pi}{2}+\gamma-\varepsilon}, \ x \in D(B).$$

Then the existence of limit $\lim_{\mathrm{Re}\,\lambda\to+\infty}\lambda^{\delta-\sigma+1}(\lambda^{\delta}B-A)^{-1}CBx$ in E, for all $x \in D(B)$, implies that there exists an exponentially equicontinuous, analytic $(\sigma_1 - 1)$ -times integrated C-resolvent (n, j)propagation family $(R_{n,j}(t))_{t\geq 0}$ for problem (104), of angle θ_1 ; if there is an element $x \in D(B)$ such that the limit $\lim_{\mathrm{Re} \lambda \to +\infty} \lambda^{\delta - \sigma + 1}$ $(\lambda^{\delta}B - A)^{-1}CBx$ does not exist in E, then the above holds with any number $\sigma_2 > \sigma_1$. For the purpose of illustration of obtained results, assume now that $n \in \mathbb{N}$ and $iA'_l, 1 \leq l \leq n$ are commuting generators of bounded C_0 -groups on a Banach space E. Put $A' := (A'_1, \ldots, A'_n)$; cf. the previous section for the definition of a closable operator P(A'), where P(x) is a complex polynomial in *n* variables. Suppose $0 < \delta < 2$, $\omega \ge 0$, $P_1(x)$ and $P_2(x)$ are non-zero complex polynomials, $N_1 = dg(P_1(x)), N_2 = dg(P_2(x)),$ $\beta > \frac{n}{2} \frac{(N_1 + N_2)}{\min(1,\delta)} \text{ (resp. } \beta \ge n | \frac{1}{p} - \frac{1}{2} | \frac{(N_1 + N_2)}{\min(1,\delta)} \text{, if } E = L^p(\mathbb{R}^n) \text{ for some}$ $1 , <math>P_2(x) \neq 0, x \in \mathbb{R}^n$ and (70) holds with the number α replaced by δ therein. Set

$$R_{\delta}(t) := \left(E_{\delta} \left(t^{\delta} \frac{P_1(x)}{P_2(x)} \right) (1 + |x|^2)^{-\beta/2} \right) (A'), \quad t \ge 0.$$

(105)

By Theorem 2.2.21, we know that $(R_{\delta}(t))_{t\geq 0} \subseteq L(E)$ is a global exponentially bounded $(g_{\delta}, R_{\delta}(0))$ -regularized resolvent family for the problem (52) with $B = \overline{P_2(A')}$, $A = \overline{P_1(A')}$ and $a(t) = q_{\delta}(t)$. By the conclusion stated in (a), it readily follows that there exists an exponentially equicontinuous, analytic C-resolvent (n, j)-propagation family $(R_{n,j}(t))_{t\geq 0}$ for problem (104), of angle $\theta = \min(\frac{\pi}{2}, \frac{\pi\delta}{2(\alpha_n - \alpha)} - 1)$ $\frac{\pi}{2}$). Since the condition (ii.1) given in the formulation of Theorem 2.2.8 holds, with $a(t) = g_{\delta}(t)$ and k(t) = 1, it is not problematic to show, with the help of our previous consideration and the results concerning the Laplace transform of vector-valued analytic functions (see e.g. Theorem 1.4.10 and [296, Section 3]) that the conditions (C.1) and (C.5) hold for (104), as well as that the condition (C.4)holds for (104) provided that $\alpha_{n-1} \leq \alpha$; we need the last condition because the inclusion $\lambda^{\alpha_{n-1}-\alpha_n-1}AP_{\lambda}^{-1}Cx \in LT-E$ has to be satisfied $(x \in E)$. It is also worth noting that we do not need the condition $\alpha_{n-1} \leq \alpha$ for the existence of solutions of the integral equation

$$Bu(t) + \sum_{l=1}^{n-1} c_l (g_{\alpha_n - \alpha_l} * Bu)(t) = A(g_{\alpha_n - \alpha} * u)(t) + CBv_0, \quad t \ge 0;$$

cf. (97). It is also worth noting that we can refine the results on C-wellposedness of equation (104) by using the estimates quoted in

Remark 2.1.23(ii) and that we can similarly consider the equation (104) in E_l -type spaces.

(ii) (cf. [346, Example 5.1(i)–(b)] and Example 2.2.18 for more details) Let $s > 1, 0 \leq j \leq \lceil \alpha_n \rceil - 1, k_{a,b}(t) := \mathcal{L}^{-1}(\exp(-a\lambda^b))(t), t \geq 0$ $(a > 0, b \in (0, 1)),$

$$E := \left\{ f \in C^{\infty}[0,1] \; ; \; \|f\| := \sup_{p \ge 0} \frac{\|f^{(p)}\|_{\infty}}{p!^s} < \infty \right\}$$

and

$$A' := -d/ds, \ D(A') := \{f \in E \ ; \ f' \in E, \ f(0) = 0\}.$$

Let $P_1(z) = \sum_{l=0}^{N_1} a_{l,1} z^l$, $z \in \mathbb{C}$, $a_{N_1,1} \neq 0$ be a complex non-zero polynomial, and let $P_2(z) = \sum_{l=0}^{N_2} a_{l,2} z^l$, $z \in \mathbb{C}$, $a_{N_2,2} \neq 0$ be a complex non-zero polynomial so that $N_1 = dg(P_1) \ge 1 + dg(P_2) = 1 + N_2$. Set $A := P_1(A')$ and $B := P_2(A')$. We have proved so far that there exist numbers b > 0 and c > 0 such that

(106)
$$\|(\lambda B - A)^{-1}\| = O(e^{b|\lambda|^{1/(N_1 - N_2)s} + c|\lambda|^{1/(N_1 - N_2)}}), \quad \lambda \in \mathbb{C},$$

and, for every complex non-zero polynomial P(z) with $dg(P) \leq N_1$, there exists $\zeta > 0$ such that

(107)
$$\|(\lambda B - A)^{-1} P(A') f\| \leq \zeta \|f\| e^{b|\lambda|^{1/(N_1 - N_2)s} + c|\lambda|^{1/(N_1 - N_2)}},$$

for all $\lambda \in \mathbb{C}$ and $f \in D(P(A'))$. Let $\theta \in (0, \pi/2]$, $b' = (\alpha_n - \alpha)/(N_1 - N_2)$ and let $b' \leq \pi/(\pi + 2\theta)$. Owing to (106)–(107) and Theorem 2.3.9, we obtain that there is a sufficiently large number a' > 0 such that there exists an exponentially equicontinuous, analytic $k_{a',b'}$ -regularized *I*-resolvent (n, j)-propagation family $(R_{n,j}(t))_{t\geq 0}$ for problem (104), of angle θ , satisfying the conditions (C.1)–(C.5). Denote, as before, $T_{l,L}u(t) = B\mathbf{D}_t^{\alpha_l}u(t), t \geq 0$ if $l \in \mathbb{N}_n$ and $\alpha_l > 0$, $T_{l,R}u(t) = \mathbf{D}_t^{\alpha_l}Bu(t), t \geq 0$ if $l \in \mathbb{N}_n, T_{0,L}u(t) = A\mathbf{D}_t^{\alpha}u(t), t \geq 0$ if $\alpha > 0$, and $T_{0,R}u(t) = \mathbf{D}_t^{\alpha}Au(t), t \geq 0$. Let $T_lu(t)$ be either $T_{l,L}u(t)$ or $T_{l,R}u(t)$ $(l \in \mathbb{N}_n^0)$. Then it can be easily seen that for each $x \in D(B)$ the function $u(t) = R_{n,j}(t)x, t \geq 0$ is a unique strong solution of problem

$$T_n u(t) + \sum_{l=1}^{n-1} c_l T_l u(t) = T_0 u(t) + \left(k_{a',b'}^{(m_n)} * g_{j+m_n-\alpha_n} \right)(t), \quad t \ge 0,$$

with all initial values chosen to be zeroes. Observe, finally, that the analysis contained in [**346**, Example 5.4] can be used for the construction of hypoanalytic exponentially equicontinuous k-regularized I-resolvent (n, j)-propagation families for the problem

$$\sum_{l \in \mathcal{K}} P_l(A') \mathbf{D}_t^{\alpha_l} u(t) + \sum_{l \in \mathbb{N}_{n-1} \smallsetminus \mathcal{K}} \mathbf{D}_t^{\alpha_l} P_l(A') u(t) = 0, \quad t \ge 0,$$

where $P_1(z), \ldots, P_{n-1}(z)$ are complex non-zero polynomials satisfying certain properties, and $\mathcal{K} \subseteq \mathbb{N}_{n-1}$.

2.3.3. Exponentially equicontinuous (a, k)-regularized *C*-resolvent families generated by *A*, *B*; exponentially equicontinuous (k; C)-regularized resolvent (i, j)-propagation families. In this subsection, our main task will be to investigate the *C*-wellposedness of problem $(DFP)_L$ with *A* and *B* being closed linear operators on *E*. Following the examination from [6, Section 2], we introduce the following definition.

DEFINITION 2.3.13. Suppose that the functions a(t) and k(t) satisfy (P1), as well as $R(t) \in L(E, [D(B)])$ for all $t \ge 0$. Let $C \in L(E)$ be injective, and let $CA \subseteq AC$ and $CB \subseteq BC$. Then the operator family $(R(t))_{t\ge 0}$ is said to be an exponentially equicontinuous (a, k)-regularized C-resolvent family generated by A, B iff there exists $\omega \ge \max(0, \operatorname{abs}(a), \operatorname{abs}(k))$ such that the following holds:

- (i) The mappings $t \mapsto R(t)x, t \ge 0$ and $t \mapsto BR(t)x, t \ge 0$ are continuous for every fixed element $x \in E$.
- (ii) The family $\{e^{-\omega t}R(t) : t \ge 0\} \subseteq L(E, [D(B)])$ is equicontinuous, i.e., for every $p \in \circledast$, there exist c > 0 and $q \in \circledast$ such that

$$p(e^{-\omega t}R(t)x) + p(e^{-\omega t}BR(t)x) \leqslant cq(x), \quad x \in E, \ t \ge 0.$$

(iii) For every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $k(\lambda) \neq 0$, the operator $B - \tilde{a}(\lambda)A$ is injective, $R(C) \subseteq R(B - \tilde{a}(\lambda)A)$ and

$$\tilde{k}(\lambda)(B - \tilde{a}(\lambda)A)^{-1}Cx = \int_0^\infty e^{-\lambda t} R(t)x \, dt, \quad x \in E.$$

If $k(t) = g_{r+1}(t)$ for some $r \ge 0$, then it is also said that $(R(t))_{t\ge 0}$ is an exponentially equicontinuous *r*-times integrated (a, C)-regularized resolvent family generated by A, B; an exponentially equicontinuous 0-times integrated (a, C)-regularized resolvent family generated by A, B is also said to be an exponentially equicontinuous (a, C)-regularized resolvent family generated by A, B.

Before going any further, it should be noticed that we have already constructed some examples of (g_{α}, k) -regularized *C*-resolvent families generated by *A*, *B* in Example 2.3.12(ii).

REMARK 2.3.14. Suppose that the functions a(t) and k(t) satisfy (P1), as well as that $CA \subseteq AC$ and $CB \subseteq BC$.

- (i) It is clear that an exponentially equicontinuous (a, k)-regularized C-resolvent family generated by A, B, if exists, must be unique.
- (ii) If for each $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\tilde{k}(\lambda) \neq 0$ the operator B commutes with $(B - \tilde{a}(\lambda)A)^{-1}C$, then the operator family $(BR(t))_{t\geq 0}$ is an exponentially equicontinuous (a, k)-regularized C-resolvent family for (52), and the condition (ii.1) stated in the formulation of Theorem 2.2.8 holds. Furthermore, for each $t \geq 0$ the operator BR(t) can be continuously extended from D(B) to the whole space E.

(iii) Assume $(R(t))_{t \ge 0}$ is an exponentially equicontinuous (a, k)-regularized Cresolvent family for (52) and there exists a strongly continuous operator
family $(\hat{R}(t))_{t \ge 0} \subseteq L(E)$ such that $\hat{R}(t)x = R(t)x, t \ge 0, x \in D(B)$ (the last condition automatically holds provided that E is complete and B is densely defined). If $B^{-1} \in L(E)$ and $BR(t) \subseteq R(t)B, t \ge 0$,
then $(R(t)B^{-1})_{t\ge 0}$ is an exponentially equicontinuous (a, k)-regularized C-resolvent family generated by A, B.

The proof of following theorem can be deduced by using slight modifications of the proofs of [6, Proposition 2.1, Lemma 2.2] and the fact that the assertion of [295, Lemma 2.4] continues to hold in SCLCSs.

THEOREM 2.3.15. Let $(R(t))_{t\geq 0}$ be an exponentially equicontinuous (a, k)-regularized C-resolvent family generated by A, B, and let $\operatorname{abs}(|a|) < \infty$. Then the following holds:

(i) For every x ∈ E and for every λ ∈ C with Re λ > ω and k̃(λ) ≠ 0, we have:

$$\tilde{k}(\lambda)B(B-\tilde{a}(\lambda)A)^{-1}Cx = \int_0^\infty e^{-\lambda t}BR(t)x\,dt.$$

- (ii) $R(t)Bx = k(t)Cx + \int_0^t a(t-s)R(s)Ax\,ds, t \ge 0, x \in D(A) \cap D(B).$
- (iii) $\int_0^t a(t-s)R(s)x\,ds \in D(A) \cap D(B), \ t \ge 0, \ x \in E.$
- (iv) $BR(t)x = k(t)Cx + A \int_0^t a(t-s)R(s)x \, ds, t \ge 0, x \in E.$
- (v) $R(t)B(D(A) \cap D(B)) \subseteq D(A) \cap D(B), t \ge 0.$
- (vi) $B(B \tilde{a}(\lambda)A)^{-1}CAx = A(B \tilde{a}(\lambda)A)^{-1}CBx$ for every $x \in D(A) \cap D(B)$ and for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\tilde{k}(\lambda) \neq 0$; AR(t)Bx = BR(t)Ax, $t \ge 0, x \in D(A) \cap D(B)$.
- (vii) Suppose that the function k(t) is differentiable in a point $t_0 \ge 0$ and $a \in AC_{loc}([0,\infty))$. If $\lambda \in \mathbb{C}$ satisfies $\operatorname{Re} \lambda > \omega$ and $\tilde{k}(\lambda) \ne 0$, then for every $j \in \mathbb{N}_0$, $z \in \mathbb{C}$ and for every complex polynomial $P(\cdot)$, we have:

(108)
$$\left(\frac{d}{dt}[(z(B-\tilde{a}(\lambda)A)^{-1}C-P(C))^{j}R(t)Bx]\right)_{t=t_{0}}$$

= $(z(B-\tilde{a}(\lambda)A)^{-1}C-P(C))^{j}\left(\frac{d}{dt}R(t)Bx\right)_{t=t_{0}}$.

(viii) Let $x \in D(A) \cap D(B)$. Then the function $t \mapsto u(t), t \ge 0$, defined by $u(t) := R(t)Bx, t \ge 0$ satisfies $u \in C([0,\infty) : [D(A)]) \cap C([0,\infty) : [D(B)])$ and

$$Bu(t) = k(t)CBx + \int_0^t a(t-s)Au(s)ds, \quad t \ge 0.$$

REMARK 2.3.16. (i) Suppose that $x \in D(A) \cap D(B)$, $\alpha > 0$ and there exists an exponentially equicontinuous (g_{α}, C) -regularized resolvent family $(R(t))_{t \ge 0}$ generated by A, B. Using the identity $R(t)Bx = Cx + \int_0^t g_{\alpha}(t-s)R(s)Ax\,ds, t \ge 0$, it readily follows that the mapping $t \mapsto R(t)Bx, t \ge 0$

is (m-1)-times continuously differentiable on $[0, \infty)$, where $m = \lceil \alpha \rceil$. Furthermore, it can be easily verified that the Caputo derivative $\mathbf{D}_t^{\alpha} R(t) Bx$ is well defined as well as that $\mathbf{D}_t^{\alpha} R(t) Bx = R(t) Ax$, $t \ge 0$. Keeping in mind Remark 2.3.7(ii) and Proposition 2.3.15(vi), we get that the function u(t) := R(t)Bx, $t \ge 0$ is a unique solution of the following Cauchy problem:

$$\begin{cases} u \in C([0,\infty): [D(A)]) \cap C([0,\infty): [D(B)]) \cap C^{m-1}([0,\infty): E), \\ B\mathbf{D}_t^{\alpha} u(t) = Au(t), \quad t \ge 0, \\ u(0) = Cx; \ u^{(j)}(0) = 0, \quad 1 \leqslant j \leqslant m-1. \end{cases}$$

In Theorem 2.3.18, we shall extend this result to the class of exponentially equicontinuous $(g_{\alpha}, g_{\alpha l+1})$ -regularized *C*-resolvent families generated by $A, B \ (l \in \mathbb{N}).$

(ii) Now we would like to illustrate the conclusion deduced in the first part of this remark to degenerate fractional equations associated with the abstract differential operators. For the sake of simplicity, we shall only consider the equations of order $\alpha \in (0, 2)$; the case $\alpha = 2$ has been considered in [**306**, Subsection 4.1] and here we only want to point out that the results from Subsection 2.1.4 and this remark can be also applied in the analysis of equation

$$\frac{\partial^2}{\partial t^2}(u_{zz} - \beta^2$$

$$u) + \omega_0^2 u_{yy} = 0$$
 ($\beta > 0, \ \omega_0 > 0$ is the Väisälä–Brunt frequency),

which is important in the linear theory of internal waves in stratified fluid ([483]), and the Boussinesq equation (see e.g. [140, Example 1.5, p. 50])

$$(\sigma^2 \Delta - 1)u_{tt} + \gamma^2 \Delta u = 0 \quad (\sigma > 0, \ \gamma > 0).$$

Assume that $n \in \mathbb{N}$ and $iA_j, 1 \leq j \leq n$ are commuting generators of bounded C_0 -groups on a Banach space E. Suppose again that $0 < \alpha < 2$, $\omega \geq 0$, $P_1(x)$ and $P_2(x)$ are non-zero complex polynomials, $N_1 = dg(P_1(x)), N_2 = dg(P_2(x)), \beta > \frac{n}{2} \frac{(N_1+N_2)}{\min(1,\alpha)}$ (resp. $\beta \geq n |\frac{1}{p} - \frac{1}{2}| \frac{(N_1+N_2)}{\min(1,\alpha)}$, if $E = L^p(\mathbb{R}^n)$ for some $1), <math>P_2(x) \neq 0, x \in \mathbb{R}^n$ and (70) holds with δ replaced by α . Define $(R_\alpha(t))_{t\geq 0}$ as in (105), with δ replaced by α ; $C \equiv R_\alpha(0)$. Then we know that $(R_\alpha(t))_{t\geq 0} \subseteq L(E)$ is a global exponentially bounded $(g_\alpha, R_\alpha(0))$ -regularized resolvent family for the problem

$$(P)_R: \begin{cases} \mathbf{D}_t^{\alpha} \overline{P_2(A)} u(t) = \overline{P_1(A)} u(t), & t \ge 0, \\ u(0) = Cx; & u^{(j)}(0) = 0, & 1 \le j \le \lceil \alpha \rceil - 1. \end{cases}$$

Furthermore, the analysis contained in Remark 2.1.23(i) implies that there exists an exponentially bounded, strongly continuous operator family $(G_{\alpha}(t))_{t\geq 0}$ such that $G_{\alpha}(t)x = \overline{P_2(A)}^{-1}R_{\alpha}(t)x, t \geq 0, x \in E$ and $\lambda^{\alpha-1}(\lambda^{\alpha}B - A)^{-1}Cx = \int_0^{\infty} e^{-\lambda t}G_{\alpha}(t)x dt$ for any $x \in E$ and $\lambda > 0$ sufficiently large. Hence, $(G_{\alpha}(t))_{t\geq 0}$ is an exponentially equicontinuous (g_{α}, C) -regularized resolvent family generated by $\overline{P_1(A)}$, $\overline{P_2(A)}$, as announced before. The consideration is quite similar in the case that the requirements of Theorem 2.2.21 hold.

We employ the following auxiliary lemma in the proof of Theorem 2.3.18 mentioned above.

LEMMA 2.3.17. (cf. [292, Corollary 2.1.20]) Suppose $\alpha > 0$, $l \in \mathbb{N}$, $z \in \mathbb{C}$, A is a subgenerator of an exponentially equicontinuous $(g_{\alpha}, g_{l\alpha+1})$ -regularized Cresolvent family $(S_{l,\alpha}(t))_{t\geq 0}$ on E, z - A is injective, $R(C) \subseteq R((z - A)^l)$ and $(z - A)^{-1}C \in L(E), \ldots, (z - A)^{-l}C \in L(E)$. Set, for every $x \in E$ and $t \geq 0$,

$$S_{\alpha}(t)x := (-1)^{l}S_{l,\alpha}(t)x + \sum_{j=0}^{l-1} (-1)^{j+1} \binom{l}{j} z^{l-j} \Big[\mathcal{L}^{-1} \Big(\frac{r^{\alpha j}}{(r^{\alpha} - z)^{l}} \Big) * S_{l,\alpha}(\cdot)x \Big](t)$$
$$+ \sum_{j=1}^{l} (-1)^{l-j} \mathcal{L}^{-1} \Big(\frac{r^{\alpha-1}}{(r^{\alpha} - z)^{l+1-j}} \Big)(t)(z-A)^{-j} Cx.$$

Then $(S_{\alpha}(t))_{t\geq 0}$ is an exponentially equicontinuous $(g_{\alpha}, (z-A)^{-l}C)$ -regularized resolvent family with a subgenerator A.

Now we state the following important extension of [6, Theorem 2.2].

THEOREM 2.3.18. Suppose that $\alpha > 0$, $l \in \mathbb{N}$, $z \in \mathbb{C}$, there exists an exponentially equicontinuous $(g_{\alpha}, g_{l\alpha+1})$ -regularized C-resolvent family $(S_{l,\alpha}(t))_{t\geq 0}$ generated by A, B, the operator zB - A is injective and $x \in D(A) \cap D(B) \cap D(((zB - A)^{-1}B)^{l}C))$. Define

$$u(t) := (-1)^{l} S_{l,\alpha}(t) Bx + \sum_{j=0}^{l-1} (-1)^{j+1} {l \choose j} z^{l-j} \Big[\mathcal{L}^{-1} \Big(\frac{r^{\alpha j}}{(r^{\alpha} - z)^{l}} \Big) * S_{l,\alpha}(\cdot) Bx \Big](t)$$

+ $\sum_{j=1}^{l} (-1)^{l-j} \mathcal{L}^{-1} \Big(\frac{r^{\alpha - 1}}{(r^{\alpha} - z)^{l+1-j}} \Big) (t) ((zB - A)^{-1}B)^{j} Cx, \quad t \ge 0.$

Then the function u(t) is a unique solution of the problem $(DFP)_L$ with $f(t) \equiv 0$ and the initial value x replaced by $((zB-A)^{-1}B)^l Cx$ (we will designate this problem by $(DFP)_{L,l}$ in the sequel).

PROOF. The uniqueness of solutions follows similarly as in Remark 2.3.16(i) and we will only prove that the function u(t) is a solution of the problem $(DFP)_{L,l}$. Denote $x_j := ((zB - A)^{-1}B)^j Cx$ $(j \in \mathbb{N}_l^0)$, $F_{j,l}(t) := \mathcal{L}^{-1}(\frac{r^{\alpha_j}}{(r^{\alpha}-z)^l})(t)$, t > 0 $(0 \leq j \leq l-1)$ and $G_{j,l}(t) := \mathcal{L}^{-1}(\frac{r^{\alpha-1}}{(r^{\alpha}-z)^{l+1-j}})(t)$, $t \geq 0$ $(1 \leq j \leq l)$. Then the function $F_{j,l}(t)$ is continuous on $(0, \infty)$, locally integrable on $[0, \infty)$ and exponentially bounded on $[1, \infty)$ $(0 \leq j \leq l-1)$, while the function $G_{j,l}(t)$ is continuous and exponentially bounded on $[0, \infty)$ $(1 \leq j \leq l)$; cf. [292]. Set $m := \lceil \alpha \rceil$. By Theorem 2.3.15(ii), we have that the mapping $t \mapsto S_{l,\alpha}(t)Bx$, $t \geq 0$ is (m-1)-times continuously differentiable and

$$\frac{d^{m-1}}{dt^{m-1}}S_{l,\alpha}(t)Bx = g_{\alpha l+2-m}(t)Cx + \int_0^t g_{\alpha+1-m}(t-s)S_{l,\alpha}(s)Ax\,ds, \quad t \ge 0;$$

hence $(\frac{d^j}{dt^j}S_{l,\alpha}(t)Bx)_{t=0} = 0, \ 0 \le j \le m-1$. This simply implies that the mapping $t \mapsto [F_{j,l} * S_{l,\alpha}(\cdot)Bx](t), \ t \ge 0$ is (m-1)-times continuously differentiable as well as

$$\frac{d^{m-1}}{dt^{m-1}}[F_{j,l}*S_{l,\alpha}(\cdot)Bx](t) = \left[F_{j,l}*\frac{d^{m-1}}{dt^{m-1}}S_{l,\alpha}(\cdot)Bx\right](t), \quad t \ge 0$$

provided $0 \leq j \leq l-1$; hence, $(\frac{d^p}{dt^p}[F_{j,l} * S_{l,\alpha}(\cdot)Bx](t))_{t=0} = 0, 0 \leq p \leq m-1$ $(0 \leq j \leq l-1)$. Now it is very simple to show that

$$\mathbf{D}_{t}^{\alpha}S_{l,\alpha}(t)Bx = g_{l\alpha+1-\alpha}(t)Cx + S_{l,\alpha}(t)Ax, \quad t \ge 0,$$

and

$$\mathbf{D}_t^{\alpha}[F_{j,l} * S_{l,\alpha}(\cdot)Bx](t) = g_{l\alpha+1-\alpha}(t)Cx + S_{l,\alpha}(t)Ax, \quad t \ge 0 \ (0 \le j \le l-1).$$

Suppose, for the time being, that the assumptions of Lemma 2.3.17 hold. Since for each $x \in D(A)$ the function $v(t) := S_{\alpha}(t)x, t \ge 0$ is a unique solution of the problem

$$\begin{cases} \mathbf{D}_t^{\alpha} v(t) = A v(t), & t \ge 0, \\ v(0) = C x; & v^{(j)}(0) = 0, & 1 \le j \le m - 1 \end{cases}$$

we may conclude from the above (by plugging $l = 1, 2, \ldots$ successively in Lemma 2.3.17) that for each $j \in \mathbb{N}_l$ the function $G_{j,l}(t)$ is (m-1)-times continuously differentiable on $[0, \infty)$ as well as that $(\frac{d^p}{dt^p}G_{j,l}(t))_{t=0} = 0, 1 \leq p \leq m-1$ $(1 \leq j \leq l)$ and the Caputo derivative $\mathbf{D}_t^{\alpha}G_{j,l}(t)$ is well defined $(1 \leq j \leq l)$. Since $G_{l,l}(t) = E_{\alpha}(zt^{\alpha}), t \geq 0$, it readily follows that the function u(t) satisfies $u(0) = x_l$ and $u^{(j)}(0) = 0, 1 \leq j \leq m-1$. It remains to be proved that $B\mathbf{D}_t^{\alpha} = Au(t), t \geq 0$. Carrying out a straightforward computation, it can be easily seen that this equality holds iff

$$(-1)^{l}g_{\alpha l+1-\alpha}(t)BCx + \sum_{j=0}^{l-1} (-1)^{j+1} \binom{l}{j} z^{l-j} [F_{j,l} * g_{\alpha l+1-\alpha}](t)BCx + \sum_{j=1}^{l} (-1)^{l-j} \mathbf{D}_{t}^{\alpha} G_{j,l}(t)Bx_{j} = \sum_{j=1}^{l} (-1)^{l-j} G_{j,l}(t)Ax_{j}, \quad t \ge 0$$

iff

$$(-1)^{l}g_{\alpha l+1-\alpha}(t)BCx + \sum_{j=0}^{l-1} (-1)^{j+1} \binom{l}{j} z^{l-j} [F_{j,l} * g_{\alpha l+1-\alpha}](t)BCx + \sum_{j=1}^{l} (-1)^{l-j} \mathbf{D}_{t}^{\alpha} G_{j,l}(t)Bx_{j} = \sum_{j=1}^{l} (-1)^{l-j} G_{j,l}(t) [zBx_{j} - Bx_{j-1}], \quad t \ge 0.$$

This is true because the coefficients of Bx_j , for every fixed number $j \in \mathbb{N}_l^0$, on both sides of previous equality are equal (cf. also the proof of [292, Theorem 2.1.19]). \Box

Suppose that the operator B is injective, $x \in D(AB^{-1})$, $\alpha > 0$ and there exists an exponentially equicontinuous (g_{α}, C) -regularized resolvent family $(R(t))_{t \ge 0}$ generated by A, B. Then it is readily seen that the function $u(t) := R(t)x, t \ge 0$ is a unique solution of problem $(DFP)_L$ with $f(t) \equiv 0$ and the initial value x replaced by $CB^{-1}x$. We leave to the interested reader the problem of transferring this conclusion, as well as the others from [6, Remark 2.4], to degenerate fractional equations whose solutions are governed by $(g_{\alpha}, g_{\alpha l+1})$ -regularized C-resolvent families generated by A, B $(l \in \mathbb{N})$.

Assume now that $n \in \mathbb{N} \setminus \{1\}$, $0 \leq \alpha_1 < \cdots < \alpha_{n-1}$, and A_1, \ldots, A_{n-1} are closed linear operators on E. In the analysis of existence and uniqueness of integral equations associated with the problem [(90)-(91)], we can also use the notion of an exponentially equicontinuous (analytic) (k; C)-regularized resolvent (i, j)-propagation family:

DEFINITION 2.3.19. (cf. Definition 2.3.4 and Definition 2.3.8) Suppose that the function k(t) satisfies (P1), as well as $1 \leq i \leq n-1$, $0 \leq j \leq m_i - 1$ and $R_{i,j}(t) \in L(E, [D(A_i)])$ for all $t \geq 0$. Let the operator $C \in L(E)$ be injective.

- (i) Then the operator family (R_{i,j}(t))_{t≥0} is said to be an exponentially equicontinuous (k; C)-regularized resolvent (i, j)-propagation family for problem [(90)-(91)] iff there exists ω ≥ max(0, abs(k)) such that the following holds:
 - (a) The mappings $t \mapsto R_{i,j}(t)x, t \ge 0$ and $t \mapsto A_i R_{i,j}(t)x, t \ge 0$ are continuous for every fixed element $x \in E$.
 - (b) The family $\{e^{-\omega t}R_{i,j}(t):t \ge 0\} \subseteq L(E, [D(A_i)])$ is equicontinuous, i.e., for every $p \in \mathfrak{B}$, there exist c > 0 and $q \in \mathfrak{B}$ such that

$$p(e^{-\omega t}R_{i,j}(t)x) + p(e^{-\omega t}A_iR_{i,j}(t)x) \leq cq(x), \quad x \in E, \ t \geq 0.$$

(c) For every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\tilde{k}(\lambda) \neq 0$, the operator P_{λ} is injective, $R(C) \subseteq R(P_{\lambda})$ and

$$\lambda^{\alpha_i - \alpha_{n-1} - j} \tilde{k}(\lambda) P_{\lambda}^{-1} C x = \int_0^\infty e^{-\lambda t} R_{i,j}(t) x \, dt, \quad x \in E.$$

- (ii) Let $(R_{i,j}(t))_{t \ge 0}$ be an exponentially equicontinuous (k; C)-regularized resolvent (i, j)-propagation family for problem [(90)-(91)]. Then it is said that $(R_{i,j}(t))_{t \ge 0}$ is an exponentially equicontinuous (equicontinuous), analytic (k; C)-regularized resolvent (i, j)-propagation family for problem [(90)-(91)], of angle α , iff the following holds:
 - (a) For every $x \in E$, the mappings $t \mapsto R_{i,j}(t)x$, t > 0 and $t \mapsto A_i R_{i,j}(t)x$, t > 0 can be analytically extended to the sector Σ_{α} ; since no confusion seems likely, we denote these extensions by the same symbols.
 - (b) For every $x \in E$ and $\beta \in (0, \alpha)$, one has $\lim_{z \to 0, z \in \Sigma_{\beta}} R_{i,j}(z)x = R_{i,j}(0)x$ and $\lim_{z \to 0, z \in \Sigma_{\beta}} A_i R_{i,j}(z)x = A_i R_{i,j}(0)x$.
 - (c) For every $\beta \in (0, \alpha)$, there exists $\omega_{\beta} \ge \max(0, \operatorname{abs}(k))$ ($\omega_{\beta} = 0$) such that the family $\{e^{-\omega_{\beta}z}R_{i,j}(z) : z \in \Sigma_{\beta}\} \subseteq L(E, [D(A_i)])$ is

equicontinuous, i.e., for every $p \in \circledast$, there exist c > 0 and $q \in \circledast$ such that

$$p(e^{-\omega_{\beta}z}R_{i,j}(z)x) + p(e^{-\omega_{\beta}z}A_{i}R_{i,j}(z)x) \leqslant cq(x), \quad x \in E, \ z \in \Sigma_{\beta}.$$

Exponentially equicontinuous (analytic) (k; C)-regularized resolvent (i, j)-propagation families yield results very similar to those obtained by k-regularized C-resolvent (i, j)-propagation families. Without going into a deeper analysis, we shall only observe that the assertions of Theorem 2.3.5(i)–(iii), Remark 2.3.7(i),(iii), Theorem 2.3.9 and Theorem 2.3.10 can be restated for exponentially equicontinuous (k; C)-regularized resolvent (i, j)-propagation families. Details can be left to the interested reader.

Before we move to the next subsection, it would be worthwhile to reconsider the assertion of [**292**, Theorem 2.3.3] for systems of abstract degenerate differential equations here. In order to do that, assume that $n \in \mathbb{N}$ and $iA_j, 1 \leq j \leq n$ are commuting generators of bounded C_0 -groups on a Banach space E, as well that $A = (A_1, \ldots, A_n)$; cf. Subsection 2.2.3 for more details. Denote by $\mathbb{C}^{m,m}$ the ring of $m \times m$ matrices over \mathbb{C} , and by I_m the identity matrix of format $m \times m$ $(m \in \mathbb{N})$. If $P(x) = [p_{ij}(x)]$ is an $m \times m$ matrix of polynomials of $x \in \mathbb{R}^n$, then there exist $d \in \mathbb{N}$ and matrices $P_\eta \in \mathbb{C}^{m,m}$ such that $P(x) = \sum_{|\eta| \leq d} P_\eta x^\eta$, $x \in \mathbb{R}^n$. Then we know that the operator $P(A) := \sum_{|\eta| \leq d} P_\eta A^\eta$ is closable on E^m (cf. [132, Theorem 14.1]).

Now we are ready to formulate the following theorem.

THEOREM 2.3.20. Let $(E, \|\cdot\|)$ be a complex Banach space and let $iA_j, 1 \leq j \leq n$ be commuting generators of bounded C_0 -groups on E. Suppose $\alpha > 0, d \in \mathbb{N}$ and $P_i(x) = \sum_{|\eta| \leq d} P_{\eta,i} x^{\eta} \ (P_{\eta,i} \in \mathbb{C}^{m,m}, x \in \mathbb{R}^n, i = 1, 2)$ are two given polynomial matrices. Suppose that for each $x \in \mathbb{R}^n$ the matrix $P_2(x)$ is regular. Then there exists a dense subset $E_{\alpha,m}$ of E^m such that, for every $\vec{x} \in E_{\alpha,m}$, there exists a unique solution of the following abstract Cauchy problem:

$$(\mathrm{DFP})': \begin{cases} \mathbf{D}_t^{\alpha} \overline{P_2(A)} \vec{u}(t) = \overline{P_2(A)} \mathbf{D}_t^{\alpha} \vec{u}(t) = \overline{P_1(A)} \vec{u}(t), & t \ge 0, \\ \vec{u}(0) = \vec{x}; & \vec{u}^{(j)}(0) = 0, & 1 \le j \le \lceil \alpha \rceil - 1. \end{cases}$$

Furthermore, for every $\vec{x} \in E_{\alpha,m}$, the mapping $t \mapsto \vec{u}(t)$, $t \ge 0$ can be extended to the whole complex plane (the extension of this mapping will be denoted by the same symbol in the sequel), and the following holds:

- (i) The mapping $z \mapsto \vec{u}(z), z \in \mathbb{C} \setminus (-\infty, 0]$ is analytic.
- (ii) The mapping $z \mapsto \vec{u}(z), z \in \mathbb{C}$ is entire provided that $\alpha \in \mathbb{N}$.

PROOF. Let us recall that $k = 1 + \lfloor n/2 \rfloor$. Suppose that $P_1(x) = [p_{ij;1}(x)]_{1 \leq i,j \leq m}$ and $P_2(x) = [p_{ij;2}(x)]_{1 \leq i,j \leq m}$ $(x \in \mathbb{R}^n)$, and d is the maximal degree of all non-zero polynomials $p_{ij;1}(x)$ and $p_{ij;2}(x)$ $(1 \leq i, j \leq m)$. Then $\sup_{x \in \mathbb{R}^n} (\det(P_2(x)))^{-1} < \infty$ and we can inductively prove that there exist numbers $M_1 \ge 1$ and $M_2 \ge 1$ such that for each $l \in \mathbb{N}_0$ there exist polynomials $R_{ij;l}(x)$ $(1 \leq i, j \leq m)$ of degree $\leq lmd$ satisfying that

$$(P_2(x)^{-1}P_1(x))^l = \frac{[R_{ij;l}(x)]_{1 \le i,j \le m}}{(\det(P_2(x)))^l}, \quad x \in \mathbb{R}^n$$

and the following holds:

(109)
$$\left| D^{\eta} \left(\frac{R_{ij;l}(x)}{(\det(P_{2}(x)))^{l}} \right) \right| + \left| D^{\eta} \left(p_{ij;1}(x) \frac{R_{ij;l}(x)}{(\det(P_{2}(x)))^{l}} \right) \right|$$

 $+ \left| D^{\eta} \left(p_{ij;2}(x) \frac{R_{ij;l}(x)}{(\det(P_{2}(x)))^{l}} \right) \right| \leq M_{1}^{l} (1 + |x|)^{lmdM_{2}},$

provided $l \in \mathbb{N}_0$, $x \in \mathbb{R}^n$, $0 \leq |\eta| \leq k$ and $1 \leq i, j \leq m$. It is very simple to prove that there exists a sufficiently large natural number k' satisfying 2|k' and

(110)
$$\lim_{l \to +\infty} \frac{\left(\Gamma(\frac{2M_2(l+1)md+n}{k'd})\right)^{1/2l}}{(\Gamma(\alpha l+1))^{1/l}} = 0.$$

Let a > 0 be fixed. Set $C := (e^{-a|x|^{k'd}})(A)$ and $E_{\alpha,m} := (R(C))^m$. Then $C \in L(E)$, C is injective and $D_{\infty}(A_1^2 + \cdots + A_n^2) \supseteq R(C)$ is dense in E [132]. Define

(111)
$$W_{\alpha}(z) := \left[\sum_{l=0}^{\infty} \frac{z^{\alpha l}}{\Gamma(\alpha l+1)} \left(\frac{R_{ij;l}(x)}{(\det(P_2(x)))^l} e^{-a|x|^{k'd}}\right)(A)\right]_{1 \le i,j \le m}, \ z \in \mathbb{C}.$$

Using (109)–(110) and the proof of [292, Theorem 2.3.3], it readily follows that $W_{\alpha}(z) \in L(E^m)$ for all $z \in \mathbb{C}$, as well as that the expressions

$$\left[\sum_{l=0}^{\infty}\sum_{v=1}^{m}\frac{z^{\alpha l}}{\Gamma(\alpha l+1)}\Big(p_{iv;2}(x)\frac{R_{vj;l+1}(x)}{(\det(P_{2}(x)))^{l+1}}e^{-a|x|^{k'd}}\Big)(A)\right]_{1\leqslant i,j\leqslant m}$$

and

$$\left[\sum_{l=0}^{\infty}\sum_{v=1}^{m}\frac{z^{\alpha l}}{\Gamma(\alpha l+1)}\left(p_{iv;1}(x)\frac{R_{vj;l}(x)}{(\det(P_{2}(x)))^{l}}e^{-a|x|^{k'd}}\right)(A)\right]_{1\leqslant i,j\leqslant m}$$

define the bounded linear operators on E^m $(z \in \mathbb{C})$. Furthermore, the mapping $z \mapsto W_{\alpha}(z), z \in \mathbb{C} \setminus (-\infty, 0]$ is analytic, and the mapping $z \mapsto W_{\alpha}(z), z \in \mathbb{C}$ is entire provided that $\alpha \in \mathbb{N}$. Suppose now $\vec{x} \in E_{\alpha,m}$. Then there exists $\vec{x'} \in E^m$ such that $\vec{x} = C_m \vec{x'}$, where $C_m = CI_m$. Setting $\vec{u}(z) := W_{\alpha}(z)\vec{x'}, z \in \mathbb{C}$, we immediately obtain that (i) and (ii) hold. It is not difficult to prove that $\mathbf{D}_t^{\alpha}(t)\vec{u}(t)$ is well-defined, as well as that

$$\mathbf{D}_{t}^{\alpha}(t)\vec{u}(t) = \left[\sum_{l=1}^{\infty} \frac{t^{\alpha(l-1)}}{\Gamma(\alpha(l-1)+1)} \left(\frac{R_{ij;l}(x)}{(\det(P_{2}(x)))^{l}} e^{-a|x|^{k'd}}\right)(A)\right]_{1 \leq i,j \leq m} \vec{x'}, \quad t \geq 0,$$

and $\vec{u}(0) = \vec{x}$, $\vec{u}^{(j)}(0) = 0$, $1 \leq j \leq \lceil \alpha \rceil - 1$. Since $\overline{P_1(A)}$ and $\overline{P_2(A)}$ are closed, we can prove with the help of (>) that $\vec{u}(t) \in D(\overline{P_1(A)}) \cap D(\overline{P_2(A)})$, $\mathbf{D}_t^{\alpha} \vec{u}(t) \in D(\overline{P_2(A)})$, the term $\mathbf{D}_t^{\alpha} \overline{P_2(A)} \vec{u}(t)$ is well defined, with

$$\overline{P_2(A)}\mathbf{D}_t^{\alpha}\vec{u}(t) = \mathbf{D}_t^{\alpha}\overline{P_2(A)}\vec{u}(t)$$

$$= \left[\sum_{l=0}^{\infty} \sum_{v=1}^{m} \frac{z^{\alpha l}}{\Gamma(\alpha l+1)} \left(p_{iv;2}(x) \frac{R_{iv;l+1}(x)}{(\det(P_2(x)))^{l+1}} e^{-a|x|^{k'd}} \right) (A) \right]_{1 \le i,j \le m} \vec{x'}$$

and

$$\overline{P_1(A)}\vec{u}(t) = \left[\sum_{l=0}^{\infty} \sum_{v=1}^{m} \frac{z^{\alpha l}}{\Gamma(\alpha l+1)} \left(p_{iv;1}(x) \frac{R_{iv;l}(x)}{(\det(P_2(x)))^l} e^{-a|x|^{k'd}}\right)(A)\right]_{1 \le i,j \le m} \vec{x'},$$

for any $t \ge 0$. Since

$$P_2(x)\frac{[R_{ij;l+1}(x)]_{1\leqslant i,j\leqslant m}}{(\det(P_2(x)))^{l+1}} = P_1(x)\frac{[R_{ij;l}(x)]_{1\leqslant i,j\leqslant m}}{(\det(P_2(x)))^l}, \quad l\in\mathbb{N}_0, \ x\in\mathbb{R}^n,$$

a simple matricial calculation shows that the function $t \mapsto \vec{u}(t), t \ge 0$ is a solution of problem (DFP)'. Now we will prove the uniqueness of solutions of problem (DFP)'. Let $t \mapsto \vec{u}(t), t \ge 0$ be a solution of (DFP)' with $\vec{x} = \vec{0}$. Integrating α -times (DFP)', we get that $\overline{P_2(A)}\vec{u}(t) = \int_0^t g_\alpha(t-s)\overline{P_1(A)}\vec{u}(s)ds, t \ge 0$. Using this equality, as well as the fact that $\overline{P_2(A)}W_{\alpha}(t) - \overline{P_2(A)}C_m = \overline{P_1(A)}(g_{\alpha} *$ $W_{\alpha}(\cdot)(t) \in L(E^m), t \ge 0$, and the proof of [459, Proposition 1.1], we obtain that $0 = (W_{\alpha} * 0)(t) = (\overline{P_2(A)}C_m * \vec{u})(t), t \ge 0$, so that it suffices to prove that the operator $\overline{P_2(A)}$ is injective. Suppose that $\overline{P_2(A)}\vec{x} = \vec{0}$ for some $\vec{x} \in E^m$. By [561, Lemma 1.1(a)], we may assume without loss of generality that $\vec{x} \in E_0^m$ (cf. (b)). It is clear that there exist polynomials $q_{ij}(x)$ $(1 \leq i, j \leq m)$ such that $P_2(x)^{-1} = (\det(P_2(x)))^{-1} [q_{ij}(x)]_{1 \le i,j \le m}$. Using (>), one can prove that $[(\det(P_2(x)))(A)I_m][\phi(A)I_m]\vec{x} = [(\phi(x)q_{ij}(x))(A)]_{1 \le i,j \le m} P_2(A)\vec{x} = \vec{0}, \ \phi \in \mathcal{S}(\mathbb{R}^n).$ By [306, Remark 4.4(i)], the operator $(\det(P_2(x)))(A)$ is injective, whence we may conclude that $[\phi(A)I_m]\vec{x} = \vec{0}, \phi \in \mathcal{S}(\mathbb{R}^n)$. This, in turn, implies $\vec{x} = \vec{0}$ and completes the proof of theorem. \square

REMARK 2.3.21. (i) Suppose that m = 1, $P_1(x) = \sum_{|\alpha| \leq d} a_{\alpha} x^{\alpha}$, $P_2(x) = \sum_{|\alpha| \leq d} b_{\alpha} x^{\alpha}$, $x \in \mathbb{R}^n$ $(a_{\alpha}, b_{\alpha} \in \mathbb{C})$, $P_2(x) \neq 0$, $x \in \mathbb{R}^n$ and E is a function space on which translations are uniformly bounded and strongly continuous (for example, $L^p(\mathbb{R}^n)$ with $p \in [1, \infty)$). Then the obvious choice for A_j is $i\partial/\partial x_j$ $(1 \leq j \leq n)$ and then we have that $\overline{P_1(A)}$ and $\overline{P_2(A)}$ are just the operators $\sum_{|\alpha| \leq d} a_{\alpha} i^{|\alpha|} (\partial/\partial x)^{\alpha}$ and $\sum_{|\alpha| \leq d} b_{\alpha} i^{|\alpha|} (\partial/\partial x)^{\alpha}$, respectively, acting with their maximal distributional domains. Making use of Theorem 2.3.20 and a slight modification of the formula appearing on 1. 1, p. 49 of [**292**], we can prove that for each $\alpha > 0$ there exists a dense subset $E_{\alpha,1}$ of $L^p(\mathbb{R}^n)$ such that the abstract Cauchy problem:

$$\sum_{|\alpha|\leqslant d} b_{\alpha} i^{|\alpha|} (\partial/\partial x)^{\alpha} \mathbf{D}_{t}^{\alpha} u(t,x) = \mathbf{D}_{t}^{\alpha} \sum_{|\alpha|\leqslant d} b_{\alpha} i^{|\alpha|} (\partial/\partial x)^{\alpha} u(t,x)$$
$$= \sum_{|\alpha|\leqslant d} a_{\alpha} i^{|\alpha|} (\partial/\partial x)^{\alpha} u(t,x), \quad t > 0, \ x \in \mathbb{R}^{n};$$
$$\frac{\partial^{l}}{\partial t^{l}} u(t,x)_{|t=0} = f_{l}(x), \quad x \in \mathbb{R}^{n}, \ l = 0, 1, \dots, \lceil \alpha \rceil - 1,$$

has a unique solution provided $f_l(\cdot) \in E_{\alpha,1}$, $l = 0, 1, \ldots, \lceil \alpha \rceil - 1$. A similar assertion can be formulated in E_l -type spaces [541]; we can also move to the spaces $L^{\infty}(\mathbb{R}^n)$, $C_b(\mathbb{R}^n)$ or $C^{\sigma}(\mathbb{R}^n)$ ($0 < \sigma < 1$) by using distributional techniques.

(ii) Denote $\Omega(\omega) = \{\lambda^2 : \operatorname{Re} \lambda > \omega\}$, if $\omega > 0$, and $\Omega(\omega) = \mathbb{C} \smallsetminus (-\infty, -\omega^2]$, if $\omega \leq 0$. In the previous part of Section 2.1 and Section 2.2, we have considered the *C*-wellposedness of the abstract degenerate Cauchy problem

$$(\mathrm{DFP})_2: \begin{cases} \mathbf{D}_t^{\alpha} \overline{P_2(A)} u(t) = \overline{P_2(A)} \mathbf{D}_t^{\alpha} u(t) = \overline{P_1(A)} u(t), & t \ge 0, \\ u(0) = Cx; \ u^{(j)}(0) = 0, & 1 \le j \le \lceil \alpha \rceil - 1, \end{cases}$$

where $0 < \alpha \leq 2$, $P_1(x)$ and $P_2(x)$ are complex polynomials, $P_2(x) \neq 0$, $x \in \mathbb{R}^n$, iA_j , $1 \leq j \leq n$ are commuting generators of bounded C_0 -groups on a Banach space E, and $A = (A_1, \ldots, A_n)$. The basic assumption was that (70) holds for $0 < \alpha < 2$, and that $P_1(x)/P_2(x) \notin \Omega(\omega)$, $x \in \mathbb{R}^n$, provided $\alpha = 2$. Observe that our results from the part (i) of this remark can be applied in the analysis of problem (DFP) in the general case $\alpha > 0$, and also in the case that $0 < \alpha \leq 2$ and the above-mentioned requirements are not satisfied.

(iii) The assertion of [292, Theorem 2.3.5] can be reformulated in the degenerate case, as well. We leave the precise details as an exercise for the interested reader.

EXAMPLE 2.3.22. Unfortunately, Theorem 2.3.20 and the conclusions from the parts of (i) and (ii) of former remark cannot be applied in the analysis of L^{p} -wellposedness of a great number of very important degenerate equations of mathematical physics, like (cf. the monograph by G. V. Demidenko–S. V. Uspenskii [140] for more details):

(a) (The Rossby wave equation, 1939)

$$\Delta u_t + \beta u_u = 0 \quad (n = 2), \qquad u(0, x, y) = u_0(x, y);$$

(b) (The Sobolev equation, 1940)

$$\begin{aligned} \Delta u_{tt} + \omega^2 u_{zz} &= 0 \quad (n = 3), \\ u(0, x, y, z) &= u_0(x, y, z), \quad u_t(0, x, y, z) = u_1(x, y, z), \end{aligned}$$

here $\omega/2$ is the angular velocity;

(c) (The internal wave equation in the Boussinesq approximation, 1903)

$$\Delta u_{tt} + N^2(u_{xx} + u_{yy}) = 0 \quad (n = 3),$$

$$u(0, x, y, z) = u_0(x, y, z), \quad u_t(0, x, y, z) = u_1(x, y, z);$$

(d) (The gravity-gyroscopic wave equation, cf. [222] and [535])

$$\begin{aligned} \Delta u_{tt} + N^2(u_{xx} + u_{yy}) + \omega^2 u_{zz} &= 0 \quad (n = 3), \\ u(0, x, y, z) &= u_0(x, y, z), \quad u_t(0, x, y, z) = u_1(x, y, z); \end{aligned}$$

(e) (Small amplitude oscillations of a rotating viscous fluid)

$$\Delta u_{tt} - 2\nu \Delta^2 u_t + v^2 \Delta^3 u + \omega^2 u_{zz} = 0 \quad (n = 3),$$

$$u(0, x, y, z) = u_0(x, y, z), \quad u_t(0, x, y, z) = u_1(x, y, z).$$

Here $\omega/2$ is the angular velocity and $\nu > 0$ is the viscosity coefficient.

Before including some details on the wellposedness of equations (a)–(e) in L^p spaces, we need to explain how one can reformulate the assertion of Theorem 2.3.20 in the case that there exist a vector $x_0 \in \mathbb{R}^n$ and a non-empty compact subset K of \mathbb{R}^n such that the matrix $P_2(x_0)$ is singular and $\{x \in \mathbb{R}^n : \det(P_2(x)) = 0\} \subset K$; some fractional analogues of (a)–(e) can be analyzed similarly. Denote by \mathcal{G} the class consisting of those $C^{\infty}(\mathbb{R}^n)$ -functions $\phi(\cdot)$ satisfying that there exist two open relatively compact neighborhoods Ω and Ω' of K in \mathbb{R}^n such that $\phi(x) = 0$ for all $x \in \Omega$ and $\phi(x) = 1$ for all $x \in \mathbb{R}^n \setminus \Omega'$. Since the estimate (109) holds for all $x \in \mathbb{R}^n \setminus \Omega$, for each $z \in \mathbb{C}$ we can define the matricial operator $W_{\alpha}(z)$ (cf. the proof of Theorem 2.3.20) by replacing the function $e^{-a|x|^{k'\hat{d}}}$ in (111) with the function $\phi(x)e^{-a|x|^{k'd}}$. Setting $C_{\phi} := (\phi(x)e^{-a|x|^{k'd}})(A)$ for $\phi \in \mathcal{G}$ (then we do not know any longer whether the set $\bigcup_{\phi \in \mathcal{A}} R(C_{\phi})$ is dense in E, and we cannot clarify whether the operator C_{ϕ} is injective or not) and $E'_{\alpha,m} := (\bigcup_{\phi \in \mathcal{A}} R(C_{\phi}))^m$, and assuming additionally the injectivity of matricial operator $\overline{P_2(A)}$ on E^m , then for each $\vec{x} \in E'_{\alpha,m}$ there exists a unique solution $t \mapsto \vec{u}(t), t \ge 0$ of the abstract Cauchy problem (DFP)', which can be extended to the whole complex plane, and (i)-(ii) from the formulation of Theorem 2.3.20 continues to hold. Rewriting any of the equations (a)–(e) in the matricial form, and using the following

LEMMA 2.3.23. Suppose that $1 \leq p < \infty$, $n \in \mathbb{N}$ and $E := L^p(\mathbb{R}^n)$. Denote by $\Delta_{p,n}$ the operator Δ acting on E with its maximal distributional domain. Then $\Delta_{p,n}$ is injective.

PROOF. If 1 , then the statement immediately follows from the fact $that the operator <math>-\Delta_{p,n}$ is non-negative, with dense domain and range (cf. [412, pp. 256, 266]). Suppose now that p = 1 and $\Delta_{p,n}f = 0$ for some $f \in E$. Then [412, Lemma 3.2] implies that, for every $\varphi \in \mathcal{D}(\mathbb{R}^n)$ and for every multi-index $\eta \in \mathbb{N}_0^n$, the function $\varphi * f$ belongs to the space \mathcal{T} consisting of those $C^{\infty}(\mathbb{R}^n)$ -functions whose any partial derivative belongs to $L^1(\mathbb{R}^n) \cap L^{\infty}(\mathbb{R}^n)$. Since $\Delta_{p,n}(\varphi * f) =$ $\varphi * \Delta_{p,n}f = 0, \varphi \in \mathcal{D}(\mathbb{R}^n)$ and the operator $\Delta_{\mathcal{T}}$ is injective by [412, Remark 3.3], we have that $\varphi * f = 0, \varphi \in \mathcal{D}(\mathbb{R}^n)$. Hence, f = 0.

We obtain that there exists a non-trivial subspace $E'_{1,1}$ of $L^p(\mathbb{R}^2)$, resp. $E'_{1,2}$ of $L^p(\mathbb{R}^3) \times L^p(\mathbb{R}^3)$, such that the equation (a), resp. any of the equations (b)–(e), has a unique entire solution provided $u_0(x,y) \in E'_{1,1}$, resp. $(u_0(x,y,z), u_1(x,y,z)) \in E'_{1,2}$. In the present situation, we do not know whether the subspace $E'_{1,1}$, resp. $E'_{1,2}$, can be chosen to be dense in $L^p(\mathbb{R}^2)$, resp. $L^p(\mathbb{R}^3) \times L^p(\mathbb{R}^3)$.

2.3.4. Degenerate k-regularized (C_1, C_2) -existence and uniqueness propagation families for (112). The main purpose of this subsection is to report how the techniques established in a joint paper of C.-G. Li, M. Li and the

author [346] can be successfully applied in the analysis of a wide class of abstract degenerate multi-term fractional differential equations with Caputo derivatives. We propose an important theoretical novelty method capable of seeking of solutions of some very atypical degenerate differential equations in L^p -spaces. Furthermore, we provide generalizations of [347, Theorem 2.3, Theorem 3.1] for degenerate multi-term problems.

We assume that $n \in \mathbb{N} \setminus \{1\}$, $0 \leq \alpha_1 < \cdots < \alpha_n$, $m = \lceil \alpha \rceil$, $\alpha_0 = \alpha$ and $m_i = \lceil \alpha_i \rceil$, $i \in \mathbb{N}_n^0$. Set $D_i := \{j \in \mathbb{N}_{n-1} : m_j - 1 \geq i\}$ $(i \in \mathbb{N}_{m_n-1}^0)$. The state space X is an SCLCS and $A = A_0, A_1, \ldots, A_{n-1}, A_n = B$ denote closed linear operators acting on X. Let T > 0 and $f \in C([0,T] : X)$. Consider the following degenerate multi-term problem:

(112)
$$B\mathbf{D}_{t}^{\alpha_{n}}u(t) + \sum_{i=1}^{n-1} A_{i}\mathbf{D}_{t}^{\alpha_{i}}u(t) = A\mathbf{D}_{t}^{\alpha}u(t) + f(t), \quad t \ge 0;$$
$$u^{(j)}(0) = u_{j}, \quad j = 0, \dots, \lceil \alpha_{n} \rceil - 1,$$

where $0 \leq \alpha < \alpha_n$. By a strong solution of problem (112) on the interval [0,T]we mean any continuous function $t \mapsto u(t), t \in [0,T]$ satisfying that the term $A_i \mathbf{D}_t^{\alpha_i} u(t)$ is well-defined and continuous on [0,T] $(i \in \mathbb{N}_n^0)$, as well as that (112) holds identically on [0,T]. Convoluting both sides of (112) with $g_{\alpha_n}(t)$, we get that:

(113)
$$B\left[u(\cdot) - \sum_{k=0}^{m_n - 1} u_k g_{k+1}(\cdot)\right] + \sum_{j=1}^{n-1} g_{\alpha_n - \alpha_j} * A_j \left[u(\cdot) - \sum_{k=0}^{m_j - 1} u_k g_{k+1}(\cdot)\right]$$
$$= g_{\alpha_n - \alpha} * A\left[u(\cdot) - \sum_{k=0}^{m-1} u_k g_{k+1}(\cdot)\right] + (g_{\alpha_n} * f)(\cdot), \quad t \in [0, T].$$

By a mild solution of (112) on [0,T] we mean any continuous X-valued function $t \mapsto u(t), t \in [0,T]$ satisfying

$$B\left[u(\cdot) - \sum_{k=0}^{m_n - 1} u_k g_{k+1}(\cdot)\right] + \sum_{j=1}^{n-1} A_j \left(g_{\alpha_n - \alpha_j} * \left[u(\cdot) - \sum_{k=0}^{m_j - 1} u_k g_{k+1}(\cdot)\right]\right)$$
$$= A\left(g_{\alpha_n - \alpha} * \left[u(\cdot) - \sum_{k=0}^{m-1} u_k g_{k+1}(\cdot)\right]\right) + (g_{\alpha_n} * f)(\cdot), \quad t \in [0, T].$$

Consider the following inhomogeneous equation:

(114)
$$Bu(t) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * A_j u)(t) = f(t) + (g_{\alpha_n - \alpha} * A u)(t), \quad t \in [0, T].$$

It is said that a function $u \in C([0,T]:X)$ is:

(i) a strong solution of (114) iff $A_j u \in C([0,T] : X), j \in \mathbb{N}_{n-1}^0$ and (114) holds for every $t \in [0,T]$.

(ii) a mild solution of (114) iff $(g_{\alpha_n-\alpha_j}*u)(t) \in D(A_j), t \in [0,T], j \in \mathbb{N}_{n-1}^0$ and

$$Bu(t) + \sum_{j=1}^{n-1} A_j(g_{\alpha_n - \alpha_j} * u)(t) = f(t) + A(g_{\alpha_n - \alpha} * u)(t), \quad t \in [0, T].$$

A mild (strong) solution of problem (112), resp. (114), on $[0,\infty)$ is defined analogously.

We will be interested in the following notions.

DEFINITION 2.3.24. (cf. [346, Definition 2.2] for the case B = I) Suppose $0 < \tau \leq \infty, k \in C([0, \tau)), C, C_1, C_2 \in L(X), C$ and C_2 are injective.

(i) A sequence $((R_0(t))_{t \in [0,\tau)}, \ldots, (R_{m_n-1}(t))_{t \in [0,\tau)})$ of strongly continuous operator families in L(X, [D(B)]) is called a (local, if $\tau < \infty$) k-regularized C_1 -existence propagation family for (112) iff the following holds:

$$\begin{split} B[R_i(\cdot)x - (k*g_i)(\cdot)C_1x] + \sum_{j\in D_i} A_j[g_{\alpha_n-\alpha_j}*(R_i(\cdot)x - (k*g_i)(\cdot)C_1x)] \\ + \sum_{j\in\mathbb{N}_{n-1}\smallsetminus D_i} A_j(g_{\alpha_n-\alpha_j}*R_i)(\cdot)x \\ = \begin{cases} A(g_{\alpha_n-\alpha}*R_i)(\cdot)x, & m-1 < i, \ x\in X, \\ A[g_{\alpha_n-\alpha}*(R_i(\cdot)x - (k*g_i)(\cdot)C_1x)](\cdot), & m-1 \geqslant i, \ x\in X, \end{cases} \end{split}$$

for any $i = 0, ..., m_n - 1$.

(ii) A sequence $((R_0(t))_{t\in[0,\tau)}, \ldots, (R_{m_n-1}(t))_{t\in[0,\tau)})$ of strongly continuous operator families in L(X) is called a (local, if $\tau < \infty$) k-regularized C_2 -uniqueness propagation family for (112) iff

$$\begin{split} [R_i(\cdot)Bx - (k*g_i)(\cdot)C_2Bx] + \sum_{j\in D_i} g_{\alpha_n-\alpha_j}*[R_i(\cdot)A_jx - (k*g_i)(\cdot)C_2A_jx] \\ &+ \sum_{j\in\mathbb{N}_{n-1}\smallsetminus D_i} (g_{\alpha_n-\alpha_j}*R_i(\cdot)A_jx)(\cdot) \\ &= \begin{cases} (g_{\alpha_n-\alpha}*R_i(\cdot)Ax)(\cdot), & m-1 < i, \\ g_{\alpha_n-\alpha}*[R_i(\cdot)Ax - (k*g_i)(\cdot)C_2Ax](\cdot), & m-1 \geqslant i, \end{cases} \end{split}$$

for any $x \in \bigcap_{0 \le j \le n} D(A_j)$ and $i \in \mathbb{N}_{m_n-1}^0$.

(iii) A sequence $((R_0(t))_{t\in[0,\tau)},\ldots,(R_{m_n-1}(t))_{t\in[0,\tau)})$ of strongly continuous operator families in L(X) is called a (local, if $\tau < \infty$) k-regularized Cresolvent propagation family for (112), in short k-regularized C-propagation family for (112), iff $((R_0(t))_{t\in[0,\tau)},\ldots,(R_{m_n-1}(t))_{t\in[0,\tau)})$ is a k-regularized C-uniqueness propagation family for (112), and for every $t \in [0, \tau)$, $i \in \mathbb{N}_{m_n-1}^0$ and $j \in \mathbb{N}_n^0$, one has $R_i(t)A_j \subseteq A_jR_i(t), R_i(t)C = CR_i(t)$ and $CA_j \subseteq A_jC$.

In case $k(t) = g_{\zeta+1}(t)$, where $\zeta \ge 0$, it is also said that $((R_0(t))_{t \in [0,\tau)}, \ldots, (R_{m_n-1}(t))_{t \in [0,\tau)})$ is a ζ -times integrated C_1 -existence propagation family for (112);

0-times integrated C_1 -existence propagation family for (112) is simply called C_1 existence propagation family for (112). For a k-regularized C_1 -existence propagation family $((R_0(t))_{t \in [0,\tau)}, \ldots, (R_{m_n-1}(t))_{t \in [0,\tau)})$, it is said that is locally equicontinuous (exponentially equicontinuous) iff each single operator family

 $(R_0(t))_{t \in [0,\tau)} \subseteq L(X, [D(B)]), \ldots, (R_{m_n-1}(t))_{t \in [0,\tau)} \subseteq L(X, [D(B)])$ is;

 $((R_0(t))_{t\geq 0}, \ldots, (R_{m_n-1}(t))_{t\geq 0})$ is said to be an exponentially equicontinuous, analytic k-regularized C_1 -existence propagation family for problem (112), of angle $\alpha \in (0, \pi/2]$, iff the following holds:

- (a) For every $x \in X$ and $i \in \mathbb{N}_{m_n-1}^0$, the mappings $t \mapsto R_i(t)x$, t > 0 and $t \mapsto BR_i(t)x$, t > 0 can be analytically extended to the sector Σ_{α} ; the extensions will be denoted by the same symbols.
- (b) For every $x \in X$, $\beta \in (0, \alpha)$ and $i \in \mathbb{N}_{m_n-1}^0$, we have $\lim_{z \to 0, z \in \Sigma_\beta} R_i(z)x = R_i(0)x$ and $\lim_{z \to 0, z \in \Sigma_\beta} BR_i(z)x = BR_i(0)x$.
- (c) For every $\beta \in (0, \alpha)$ and $i \in \mathbb{N}_{m_n-1}^0$, there exists $\omega_\beta \ge \max(0, \operatorname{abs}(k))$ $(\omega_\beta = 0)$ such that the family $\{e^{-\omega_\beta z}R_i(z) : z \in \Sigma_\beta\} \subseteq L(X, [D(B)])$ is equicontinuous.

The above terminological agreements and abbreviations are also introduced for the classes of k-regularized C_2 -uniqueness propagation families for (112) and k-regularized C-propagation families for (112).

Immediately from definition of a k-regularized C_1 -existence propagation family for (112) (k-regularized C_2 -uniqueness propagation family for (112)), we can state some obvious facts about the existence and uniqueness of mild (strong) solutions of problem (114); details can be skipped.

The proof of following extension of [346, Proposition 2.3] is omitted, too.

PROPOSITION 2.3.25. Let $i \in \mathbb{N}_{m_n-1}^0$ and $((R_0(t))_{t\in[0,\tau)}, \ldots, (R_{m_n-1}(t))_{t\in[0,\tau)})$ be a locally equicontinuous k-regularized C_1 -existence propagation family for (112). If $R_i(t)A_j \subseteq A_jR_i(t)$ $(j \in \mathbb{N}_n^0, t \in [0,\tau))$, $R_i(t)C_1 = C_1R_i(t)$ $(t \in [0,\tau))$, C_1 is injective, k(t) is a kernel on $[0,\tau)$ and $C_1A_j \subseteq A_jC_1$ $(j \in \mathbb{N}_n^0)$, then the following holds:

(i) The equality

(115)
$$R_i(t)R_i(s) = R_i(s)R_i(t), \quad 0 \le t, \ s < \tau$$

holds, provided that m - 1 < i and the condition

- (*) The assumption $Bf(t) + \sum_{j \in D_i} A_j(g_{\alpha_n \alpha_j} * f)(t) = 0, t \in [0, \tau)$ for some $f \in C([0, \tau) : X)$, implies $f(t) = 0, t \in [0, \tau)$,
- holds.
- (ii) The equality (115) holds provided that $m-1 \ge i$, $\mathbb{N}_{n-1} \setminus D_i \ne \emptyset$, and the condition

holds.

The assertions of [**346**, Proposition 2.5, Proposition 2.6] can be reformulated for degenerate multi-term problems; the same holds for the generalized variation of parameters formula [**346**, Proposition 2.8]:

THEOREM 2.3.26. Let $C_2 \in L(X)$ be injective. Suppose $((R_0(t))_{t \in [0,\tau)}, \ldots, (R_{m_n-1}(t))_{t \in [0,\tau)})$ is a locally equicontinuous k-regularized C_2 -uniqueness propagation family for (112), $T \in (0,\tau)$ and $f \in C([0,T]:X)$. Then the following holds:

(i) If m-1 < i, then any strong solution u(t) of (114) satisfies the equality:

$$(R_i * f)(t) = (k * g_i * C_2 B u)(t) + \sum_{j \in D_i} (g_{\alpha_n - \alpha_j + i} * k * C_2 A_j u)(t)$$

for any $t \in [0,T]$. Therefore, there is at most one strong (mild) solution for (114), provided that k(t) is a kernel on $[0,\tau)$ and (\diamond) holds.

(ii) If $m-1 \ge i$, then any strong solution u(t) of (114) satisfies the equality:

$$(R_i * f)(t) = -\sum_{j \in \mathbb{N}_{n-1} \smallsetminus D_i} (g_{\alpha_n - \alpha_j + i} * k * C_2 A_j u)(t), \quad t \in [0, T].$$

Therefore, there is at most one strong (mild) solution for (114), provided that k(t) is a kernel on $[0, \tau)$, $\mathbb{N}_{n-1} \setminus D_i \neq \emptyset$ and (∞) holds.

As explained in [292, Section 2.10], the notion of a k-regularized C_1 -existence propagation family is probably the best theoretical concept for the investigation of integral solutions of non-degenerate abstract time-fractional equation (112) with $A_j \in L(E), 1 \leq j \leq n-1$. If $A_j \notin L(E)$ for some $j \in \mathbb{N}_{n-1}$, then the vector-valued Laplace transform cannot be so easily applied, which certainly implies that there exist some limitations to this class of propagation families. A similar problem appears in the analysis od degenerate multi-term fractional differential equation (112); because of that, we will only leave as an interesting problem to the reader to try to reconsider the assertions of [346, Theorem 2.9(i), Theorem 2.10–Theorem 2.12] in our new framework. In contrast to the above, it is very simple to reformulate the assertion of [346, Theorem 2.9(ii)] to degenerate equations, without imposing any additional barriers at:

THEOREM 2.3.27. Suppose k(t) satisfies (P1), $\omega \ge \max(0, \operatorname{abs}(k))$, $(R_i(t))_{t\ge 0}$ is strongly continuous, and the family $\{e^{-\omega t}R_i(t):t\ge 0\} \subseteq L(X)$ is equicontinuous, provided $0 \le i \le m_n - 1$. Let $C_2 \in L(X)$ be injective. Then $((R_0(t))_{t\ge 0}, \ldots, (R_{m_n-1}(t))_{t\ge 0})$ is a global k-regularized C_2 -uniqueness propagation family for (112) iff, for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$, and for every $x \in \bigcap_{0 \le j \le n} D(A_j)$, the following equality holds:

$$\begin{split} \int_0^\infty e^{-\lambda t} [R_i(t)Bx - (k*g_i)(t)C_2Bx]dt \\ &+ \sum_{j \in D_i} \lambda^{\alpha_j - \alpha_n} \int_0^\infty e^{-\lambda t} [R_i(t)x - (k*g_i)(t)C_2A_jx]dt \\ &+ \sum_{j \in \mathbb{N}_{n-1} \smallsetminus D_i} \lambda^{\alpha_j - \alpha_n} \int_0^\infty e^{-\lambda t} R_i(t)A_jx\,dt \end{split}$$

$$= \begin{cases} \lambda^{\alpha - \alpha_n} \int_0^\infty e^{-\lambda t} R_i(t) Ax \, dt, & m - 1 < i, \\ \lambda^{\alpha - \alpha_n} \int_0^\infty e^{-\lambda t} [R_i(t) Ax - (k * g_i)(t) C_2 Ax] dt, & m - 1 \ge i. \end{cases}$$

Now we would like to present an instructive example of a local k-regularized I-resolvent propagation family for (112):

EXAMPLE 2.3.28. (cf. [346, Example 5.2] for non-degenerate case) Suppose $1 \leq p \leq \infty, E := L^p(\mathbb{R}), m \colon \mathbb{R} \to \mathbb{C}$ is measurable, $a_j \in L^\infty(\mathbb{R}), (A_j f)(x) := a_j(x)f(x), x \in \mathbb{R}, f \in E \ (1 \leq j \leq n), (Af)(x) := m(x)f(x), x \in \mathbb{R}, with maximal domain, and <math>\alpha = 0$. Assume $s \in (1, 2), \delta = 1/s, M_p = p!^s$ and $k_{\delta}(t) = \mathcal{L}^{-1}(\exp(-\lambda^{\delta}))(t), t \geq 0$. Denote by M(t) the associated function of sequence (M_p) and put $\Lambda'_{\alpha',\beta',\gamma'} := \{\lambda \in \mathbb{C} : \operatorname{Re}\lambda \geq \gamma'^{-1}M(\alpha'\lambda) + \beta'\}, \alpha', \beta', \gamma' > 0$. Clearly, there exists a constant $C_s > 0$ such that $M(\lambda) \leq C_s |\lambda|^{1/s}, \lambda \in \mathbb{C}$. Assume that the following condition holds:

(CH): For every $\tau > 0$, there exist $\alpha' > 0$, $\beta' > 0$ and d > 0 such that $\tau \leq \frac{\cos(\frac{\delta \pi}{2})}{C_{\sigma}(\alpha')^{1/s}}$ and

$$\left|\sum_{j=1}^{n} \lambda^{\alpha_j - \alpha} a_j(x) - m(x)\right| \ge d, \quad x \in \mathbb{R}, \ \lambda \in \Lambda_{\alpha', \beta', 1}.$$

Notice that the above condition holds provided n = 2, $\alpha_2 = 2$, $\alpha_1 = 1$, $c_1 \in L^{\infty}(\mathbb{R})$, $|c_1(x)| \ge d_1 > 0$ for a.e. $x \in \mathbb{R}$, $a_2(x) \in L^{\infty}(\mathbb{R})$, $a_2(x) = 0$, $x \in (-1, 1)$, $a_1(x) = a_2(x)c_1(x)$ and $m(x) = \frac{1}{4}c_1^2(x)a_2(x) - \frac{1}{16}c_1^4(x)a_2(x) - a_2(x)$, $x \in \mathbb{R}$ (cf. [346, (5.7)]), and that the validity of condition (CH) does not imply, in general, the essential boundedness of function $m(\cdot)$ or the injectivity of the operator B. We will prove that there exists a global (not exponentially bounded, in general) k_{δ} -regularized Iresolvent propagation family $((R_0(t))_{t\ge 0}, \ldots, (R_{m_n-1}(t))_{t\ge 0})$ for (112). Clearly, it suffices to show that, for every $\tau > 0$, there exists a local k_{δ} -regularized I-resolvent propagation family for (112) on $[0, \tau)$. Suppose $\tau > 0$ is given in advance, and $\alpha' > 0$, $\beta' > 0$ and d > 0 satisfy (CH), with this τ . Let Γ denote the upwards oriented boundary of ultra-logarithmic region $\Lambda_{\alpha',\beta',1}$. Put, for every $t \in [0, \tau)$, $f \in E$ and $x \in \mathbb{R}$,

$$(R_i(t)f)(x) := \frac{1}{2\pi i} \int_{\Gamma} e^{\lambda t - \lambda^{\delta}} \frac{\left[\lambda^{\alpha_n - \alpha - i}a_n(x) + \sum_{j \in D_i} \lambda^{\alpha_j - \alpha - i}a_j(x)\right]f(x)}{\lambda^{\alpha_n - \alpha}a_n(x) + \sum_{j=1}^{n-1} \lambda^{\alpha_j - \alpha}a_j(x) - m(x)} d\lambda.$$

Then the analysis contained in [346, Example 5.2] shows that $((R_0(t))_{t \in [0,\tau)}, \ldots, (R_{m_n-1}(t))_{t \in [0,\tau)})$ is a local k_{δ} -regularized *I*-resolvent propagation family for (112), as well as that, for every compact set $K \subseteq [0,\infty)$, there exists $h_K > 0$ such that

$$\sup_{t\in K, p\in\mathbb{N}_0, i\in\mathbb{N}_{m_p-1}^0} \frac{\left\|h_k^p \frac{d^\nu}{dt^p} R_i(t)\right\|}{p!^s} < \infty.$$

We can similarly consider the existence of local $k_{1/2}$ -regularized *I*-resolvent propagation families for (112) which obey slight modifications of the properties stated above with s = 2, and with the operators A_j not belonging to the space L(E) for some indexes $j \in \mathbb{N}_n$. Furthermore, we can similarly construct some examples of

(local) k-regularized I-resolvent propagation family for (112) in certain classes of Fréchet function spaces.

2.3.5. Degenerate k-regularized (C_1, C_2) -existence and uniqueness families. In this subsection, we assume that X and Y are two Hausdorff sequentially complete locally convex spaces over the field of complex numbers. By \circledast_X (\circledast , if there is no risk for confusion), we denote the fundamental system of seminorms which defines the topology of X. The fundamental system of seminorms which defines the topology on Y is denoted by \circledast_Y . The symbol I denotes the identity operator on X. Let $0 < \tau \leq \infty$. A strongly continuous operator family $(W(t))_{t \in [0,\tau)} \subseteq L(Y,X)$ is said to be locally equicontinuous iff, for every $T \in (0,\tau)$ and for every $p \in \circledast_X$, there exist $q_p \in \circledast_Y$ and $c_p > 0$ such that $p(W(t)y) \leq c_p q_p(y)$, $y \in Y, t \in [0,T]$; the notion of equicontinuity of $(W(t))_{t \in [0,\tau)}$ is defined similarly. Notice that $(W(t))_{t \in [0,\tau)}$ is always locally equicontinuous in the case that the space Y is barreled.

In the following definition, we will generalize the notion introduced in [346, Definition 3.1] (cf. also R. deLaubenfels [136] and T.-J. Xiao–J. Liang [545] for some other known concepts in the case B = I).

DEFINITION 2.3.29. Suppose $0 < \tau \leq \infty$, $k \in C([0, \tau))$, $C_1 \in L(Y, X)$, and $C_2 \in L(X)$ is injective.

(i) A strongly continuous operator family $(E(t))_{t \in [0,\tau)} \subseteq L(Y,X)$ is said to be a (local, if $\tau < \infty$) k-regularized C_1 -existence family for (112) iff, for every $y \in Y$, the following holds: $E(\cdot)y \in C^{m_n-1}([0,\tau):[D(B)]), E^{(i)}(0)y = 0$ for every $i \in \mathbb{N}_0$ with $i < m_n - 1, A_j(g_{\alpha_n - \alpha_j} * E^{(m_n - 1)})(\cdot)y \in C([0,\tau):X)$ for $0 \leq j \leq n$, and

(116)
$$BE^{(m_n-1)}(t)y + \sum_{j=1}^{n-1} A_j(g_{\alpha_n-\alpha_j} * E^{(m_n-1)})(t)y - A(g_{\alpha_n-\alpha} * E^{(m_n-1)})(t)y = k(t)C_1y,$$

for any $t \in [0, \tau)$.

(ii) A strongly continuous operator family $(U(t))_{t \in [0,\tau)} \subseteq L(X)$ is said to be a (local, if $\tau < \infty$) k-regularized C₂-uniqueness family for (112) iff, for every $\tau \in [0, \tau)$ and $x \in \bigcap_{0 \le i \le n} D(A_j)$, the following holds:

(117)
$$U(t)Bx + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * U(\cdot)A_j x)(t) - (g_{\alpha_n - \alpha} * U(\cdot)Ax)(t)y = (k * g_{m_n - 1})(t)C_2 x.$$

(iii) A strongly continuous family $((E(t))_{t\in[0,\tau)}, (U(t))_{t\in[0,\tau)}) \subseteq L(Y,X) \times L(X)$ is said to be a (local, if $\tau < \infty$) k-regularized (C_1, C_2) -existence and uniqueness family for (112) iff $(E(t))_{t\in[0,\tau)}$ is a k-regularized C_1 -existence family for (112), and $(U(t))_{t\in[0,\tau)}$ is a k-regularized C_2 -uniqueness family for (112).

(iv) Suppose Y = X and $C = C_1 = C_2$. Then a strongly continuous operator family $(R(t))_{t \in [0,\tau)} \subseteq L(X)$ is said to be a (local, if $\tau < \infty$) kregularized C-resolvent family for (112) iff $(R(t))_{t \in [0,\tau)}$ is a k-regularized C-uniqueness family for (112), $R(t)A_j \subseteq A_jR(t)$, for $0 \leq j \leq n$ and $t \in [0,\tau)$, as well as R(t)C = CR(t), $t \in [0,\tau)$, and $CA_j \subseteq A_jC$, for $0 \leq j \leq n$.

If $k(t) = g_{\zeta+1}(t)$, where $\zeta \ge 0$, then it is also said that $(E(t))_{t \in [0,\tau)}$ is a ζ -times integrated C_1 -existence family for (112); 0-times integrated C_1 -existence family for (112) is also said to be a C_1 -existence family for (112). The above terminological agreement can be simply understood for all other classes of uniqueness and resolvent families introduced in Definition 2.3.29.

Albeit the choice of an SCLCS space Y different from X can produce a larger set of initial data for which the abstract Cauchy problem (112) has a strong solution (see e.g. [545, Example 2.5]), in our further work the most important case will be that in which Y = X. Keeping in mind that the operators $A, B, A_1, \ldots, A_{n-1}$ are closed, we can integrate the both sides of (116) sufficiently many times in order to see that:

(118)

$$BE^{(l)}(t)y + \sum_{j=1}^{n-1} A_j (g_{\alpha_n - \alpha_j} * E^{(l)})(t)y - A(g_{\alpha_n - \alpha} * E^{(l)})(t)y = (k * g_{m_n - 1 - l})(t)C_1y,$$

for any $t \in [0, \tau)$, $y \in Y$ and $l \in \mathbb{N}^{0}_{m_n-1}$.

PROPOSITION 2.3.30. Suppose that $((E(t))_{t\in[0,\tau)}, (U(t))_{t\in[0,\tau)})$ is a k-regularized (C_1, C_2) -existence and uniqueness family for (112), and $(U(t))_{t\in[0,\tau)}$ is locally equicontinuous. Then $C_2E(t)y = U(t)C_1y$, $t \in [0, \tau)$, $y \in Y$.

PROOF. The proof of proposition is almost the same as the corresponding proof of [**346**, Proposition 3.2]. Observe only that we can always assume, without loss of generality, that the number α is less than or equal to α_1 .

DEFINITION 2.3.31. (cf. [346, Definition 3.3]) Suppose $0 \le i \le m_n - 1$. Then we define $D'_i := \{j \in \mathbb{N}^0_{n-1} : m_j - 1 \ge i\}, D''_i := \mathbb{N}^0_{n-1} \smallsetminus D'_i$ and

$$\mathbf{D}_i := \left\{ x \in \bigcap_{j \in D_i''} D(A_j) : A_j u_i \in R(C_1), \ j \in D_i'' \right\}.$$

It is not so predictable that [346, Theorem 3.4] continues to hold in the degenerate case without any terminological changes, and that the operator B does not appear in the definition of set \mathbf{D}_i , for which it is well known that represents, in non-degenerate case, the set which consists of all initial values for which the homogeneous counterpart of abstract Cauchy problem (112), with B = I and $u_j = 0$, $j \in \mathbb{N}_{m_n-1}^0 \setminus \{i\}$, has a strong solution (provided that there exists a C_1 -existence family for (112)). It is also worth nothing that we do not use the injectiveness of the operator B in (ii): THEOREM 2.3.32. (i) Suppose $(E(t))_{t \in [0,\tau)}$ is a C_1 -existence family for (112), $T \in (0,\tau)$, and $u_i \in \mathbf{D}_i$ for $0 \leq i \leq m_n - 1$. Then the function

(119)
$$u(t) = \sum_{i=0}^{m_n-1} u_i g_{i+1}(t) - \sum_{i=0}^{m_n-1} \sum_{j \in \mathbb{N}_{n-1} \smallsetminus D_i} (g_{\alpha_n - \alpha_j} * E^{(m_n - 1 - i)})(t) v_{i,j} + \sum_{i=m}^{m_n-1} (g_{\alpha_n - \alpha} * E^{(m_n - 1 - i)})(t) v_{i,0}, \quad 0 \leqslant t \leqslant T,$$

is a strong solution of the problem (112) on [0,T], with $f(t) \equiv 0$, where $v_{i,j} \in Y$ satisfy $A_j u_i = C_1 v_{i,j}$ for $0 \leq j \leq n-1$.

(ii) Suppose (U(t))_{t∈[0,τ)} is a locally equicontinuous k-regularized C₂-uniqueness family for (112), and T ∈ (0,τ). Then there exists at most one strong (mild) solution of (112) on [0,T], with u_i = 0, i ∈ N⁰_{m_n-1}.

PROOF. We will provide all the relevant details for the sake of completeness. Making use of (118), it can be easily verified that:

$$\begin{split} B\bigg[u(\cdot) &- \sum_{i=0}^{m_n-1} u_i g_{i+1}(\cdot)\bigg] + \sum_{j=1}^{n-1} A_j \bigg(g_{\alpha_n - \alpha_j} * \bigg[u(\cdot) - \sum_{i=0}^{m_j-1} u_i g_{i+1}(\cdot)\bigg]\bigg) \\ &= -\sum_{i=0}^{m_n-1} \sum_{j \in \mathbb{N}_{n-1} \smallsetminus D_i} (g_{\alpha_n - \alpha_j} * BE^{(m_n-1-i)})(\cdot)v_{i,j} \\ &+ \sum_{i=m}^{m_n-1} (g_{\alpha_n - \alpha} * BE^{(m_n-1-i)})(\cdot)v_{i,0} \\ &+ \sum_{i=m}^{n-1} A_j \bigg(g_{\alpha_n - \alpha_j} * \bigg\{\sum_{i=m_j}^{m_n-1} g_{i+1}(\cdot)u_i - \sum_{i=0}^{m_n-1} \sum_{l \in \mathbb{N}_{n-1} \smallsetminus D_i} (g_{\alpha_n - \alpha_l} * E^{(m_n-1-i)})(\cdot)v_{i,l} \\ &+ \sum_{i=m}^{m_n-1} (g_{\alpha_n - \alpha} * E^{(m_n-1-i)})(\cdot)v_{i,0}\bigg\}\bigg) \\ &= -\sum_{i=0}^{m_n-1} \sum_{j \in \mathbb{N}_{n-1} \smallsetminus D_i} (g_{\alpha_n - \alpha_j} * BE^{(m_n-1-i)})(\cdot)v_{i,j} \\ &+ \sum_{i=m}^{m_n-1} (g_{\alpha_n - \alpha} * BE^{(m_n-1-i)})(\cdot)v_{i,0} \\ &+ \sum_{i=m}^{n-1} \sum_{i=m_j} C_1 v_{i,j} g_{\alpha_n - \alpha_j} + i+1(\cdot) - \sum_{i=0}^{m_n-1} \sum_{l \in \mathbb{N}_{n-1} \smallsetminus D_i} g_{\alpha_n - \alpha_l} * [-BE^{(m_n-1-i)}(\cdot)v_{i,l} \\ &+ A(g_{\alpha_n - \alpha} * E^{(m_n-1-i)})(\cdot)v_{i,l} + g_{i+1}(\cdot)C_1v_{i,l}] \\ &+ \sum_{i=m}^{m_n-1} g_{\alpha_n - \alpha} * [-BE^{(m_n-1-i)}(\cdot)v_{i,0}] \end{split}$$

$$+ A(g_{\alpha_n - \alpha} * R^{(m_n - 1 - i)})(\cdot)v_{i,0} + g_{i+1}(\cdot)C_1v_{i,0}]$$

= $g_{\alpha_n - \alpha} * A\left[u(\cdot) - \sum_{i=0}^{m-1} u_i g_{i+1}(\cdot)\right],$

since

$$\sum_{j=1}^{n-1} \sum_{i=m_j}^{m_n-1} C_1 v_{i,j} g_{\alpha_n - \alpha_j + i+1}(\cdot) = \sum_{i=0}^{m_n-1} \sum_{j \in \mathbb{N}_{n-1} \smallsetminus D_i} C_1 v_{i,j} g_{\alpha_n - \alpha_j + i+1}(\cdot).$$

This implies that u(t) is a mild solution of (112) on [0,T]. In order to complete the proof of (i), it suffices to show that $\mathbf{D}_t^{\alpha_n} u(t) \in C([0,T]:X)$ and $A_i \mathbf{D}_t^{\alpha_i} u \in C([0,T]:X)$ for all $i \in \mathbb{N}_n^0$. Towards this end, notice that the partial integration implies that, for every $t \in [0,T]$,

$$g_{m_n-\alpha_n} * \left[u(\cdot) - \sum_{i=0}^{m_n-1} u_i g_{i+1}(\cdot) \right](t) = \sum_{i=m}^{m_n-1} (g_{m_n-\alpha+i} * E^{(m_n-1)})(t) v_{i,0} - \sum_{i=0}^{m_n-1} \sum_{j \in \mathbb{N}_{n-1} \smallsetminus D_i} (g_{m_n-\alpha_j+i} * E^{(m_n-1)})(t) v_{i,j}.$$

Therefore, $\mathbf{D}_t^{\alpha_n} u \in C([0,T]:X)$ and, for every $t \in [0,T]$,

(120)
$$\mathbf{D}_{t}^{\alpha_{n}}u(t) = \frac{d^{m_{n}}}{dt^{m_{n}}} \left\{ g_{m_{n}-\alpha_{n}} * \left[u(\cdot) - \sum_{i=0}^{m_{n}-1} u_{i}g_{i+1}(\cdot) \right](t) \right\} \\ = \sum_{i=m}^{m_{n}-1} (g_{i-\alpha} * E^{(m_{n}-1)})(t)v_{i,0} - \sum_{i=0}^{m_{n}-1} \sum_{j\in\mathbb{N}_{n-1}\smallsetminus D_{i}} (g_{i-\alpha_{j}} * E^{(m_{n}-1)})(t)v_{i,j},$$

whence we may directly conclude that $B\mathbf{D}_{t}^{\alpha_{n}}u \in C([0,T]:X)$. Suppose, for the time being, $i \in \mathbb{N}_{n-1}^{0}$. Then $A_{i}u_{j} \in R(C_{1})$ for $j \ge m_{i}$. Moreover, the inequality $l \ge \alpha_{j}$ holds provided $0 \le l \le m_{n} - 1$ and $j \in \mathbb{N}_{n-1} \setminus D_{l}$, and $A_{j}(g_{\alpha_{n}-\alpha_{j}} * E^{(m_{n}-1)})(\cdot)y \in C([0,T]:X)$ for $0 \le j \le n-1$ and $y \in Y$. Using (120), we have:

$$A_{i}\mathbf{D}_{t}^{\alpha_{i}}u(\cdot) = \sum_{j=m_{i}}^{m_{n}-1} g_{j+1-\alpha_{i}}(\cdot)A_{i}u_{j}$$

$$-\sum_{l=0}^{m_{n}-1} \sum_{j\in\mathbb{N}_{n-1}\smallsetminus D_{l}} [g_{l-\alpha_{j}} * A_{i}(g_{\alpha_{n}-\alpha_{i}} * E^{(m_{n}-1)})](\cdot)v_{l,j}$$

$$+\sum_{l=m}^{m_{n}-1} [g_{l-\alpha} * A_{i}(g_{\alpha_{n}-\alpha_{i}} * E^{(m_{n}-1)})](\cdot)v_{l,0} \in C([0,T]:X),$$

finishing the proof of (i). The second part of theorem can be proved as follows. Suppose u(t) is a strong solution of (112) on [0, T], with $u_i = 0, i \in \mathbb{N}^0_{m_n-1}$. Making use of (117) and the equality

$$\int_{0}^{t} \int_{0}^{t-s} g_{\alpha_{n}-\alpha_{j}}(r) U(t-s-r) A_{j} u(s) dr \, ds = \int_{0}^{t} \int_{0}^{s} g_{\alpha_{n}-\alpha_{j}}(r) U(t-s) A_{j} u(s-r) dr \, ds,$$

holding for any $t \in [0,T]$ and $j \in \mathbb{N}_{n-1}^{0}$, imply that

$$\begin{aligned} (UB * u)(t) &= (k * g_{m_n - 1}C_2 * u)(t) \\ &+ \int_0^t \int_0^{t-s} [g_{\alpha_n - \alpha_j}(r)U(t - s - r)A_ju(s) - g_{\alpha_n - \alpha}(r)U(t - s - r)Au(s)]dr \, ds \\ &= (k * g_{m_n - 1}C_2 * u)(t) + (U * Bu)(t), \quad t \in [0, T]. \end{aligned}$$

Therefore, $(k * g_{m_n-1}C_2 * u)(t) = 0, t \in [0,T]$ and $u(t) = 0, t \in [0,T]$.

The standard proof of following theorem is omitted.

THEOREM 2.3.33. Suppose k(t) satisfies (P1), $(E(t))_{t\geq 0} \subseteq L(Y,X)$, $(U(t))_{t\geq 0} \subseteq L(X)$, $\omega \geq \max(0, \operatorname{abs}(k))$, $C_1 \in L(Y,X)$ and $C_2 \in L(X)$ is injective. Set $\mathbf{P}_{\lambda} := B + \sum_{j=1}^{n-1} \lambda^{\alpha_j - \alpha_n} A_j - \lambda^{\alpha - \alpha_n} A$, $\lambda \in \mathbb{C} \smallsetminus \{0\}$.

(i) (a) Let $(E(t))_{t\geq 0}$ be a k-regularized C_1 -existence family for (112), let the family $\{e^{-\omega t}E(t):t\geq 0\}$ be equicontinuous, and let the family $\{e^{-\omega t}A_j(g_{\alpha_n-\alpha_j}*E)(t):t\geq 0\}$ be equicontinuous $(0\leq j\leq n)$. Then the following holds:

$$\mathbf{P}_{\lambda} \int_{0}^{\infty} e^{-\lambda t} E(t) y \, dt = \tilde{k}(\lambda) \lambda^{1-m_n} C_1 y, \quad y \in Y, \ \mathrm{Re}\,\lambda > \omega.$$

(b) Let the operator P_λ be injective for every λ > ω with k̃(λ) ≠ 0. Suppose, additionally, that there exist strongly continuous operator families (W(t))_{t≥0} ⊆ L(Y, X) and (W_j(t))_{t≥0} ⊆ L(Y, X) such that {e^{-ωt}W(t) : t ≥ 0} and {e^{-ωt}W_j(t) : t ≥ 0} are equicontinuous (0 ≤ j ≤ n) as well as that:

$$\int_0^\infty e^{-\lambda t} W(t) y \, dt = \tilde{k}(\lambda) \mathbf{P}_\lambda^{-1} C_1 y$$

and

$$\int_0^\infty e^{-\lambda t} W_j(t) y \, dt = \tilde{k}(\lambda) \lambda^{\alpha_j - \alpha_n} A_j \mathbf{P}_\lambda^{-1} C_1 y,$$

for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\tilde{k}(\lambda) \neq 0$, $y \in Y$ and $j \in \mathbb{N}_n^0$. Then there exists a k-regularized C_1 -existence family for (112), denoted by $(E(t))_{t \geq 0}$. Furthermore, $E^{(m_n-1)}(t)y = W(t)y, t \geq 0$, $y \in Y$ and $A_j(g_{\alpha_n-\alpha_j} * E^{(m_n-1)})(t)y = W_j(t)y, t \geq 0, y \in Y$, $j \in \mathbb{N}_{n-1}^0$.

(ii) Suppose $(U(t))_{t\geq 0}$ is strongly continuous and the operator family $\{e^{-\omega t}U(t):t\geq 0\}$ is equicontinuous. Then $(U(t))_{t\geq 0}$ is a k-regularized C_2 -uniqueness family for (112) iff, for every $x\in \bigcap_{j=0}^n D(A_j)$, the following holds:

$$\int_0^\infty e^{-\lambda t} U(t) \mathbf{P}_\lambda x \, dt = \tilde{k}(\lambda) \lambda^{1-m_n} C_2 x, \quad \operatorname{Re} \lambda > \omega.$$

The assertion of [**346**, Theorem 3.7], concerning the inhomogeneous Cauchy problem (112), can be stated for degenerate multi-term problems without any terminological changes, as well:

THEOREM 2.3.34. Suppose $(E(t))_{t\in[0,\tau)}$ is a locally equicontinuous C_1 -existence family for (112), $T \in (0,\tau)$, and $u_i \in \mathbf{D}_i$ for $0 \leq i \leq m_n - 1$. Let $f \in C([0,T]:X)$, let $g \in C([0,T]:Y)$ satisfy $C_1g(t) = f(t)$, $t \in [0,T]$, and let $G \in C([0,T]:Y)$ satisfy $(g_{\alpha_n-m_n+1}*g)(t) = (g_1*G)(t)$, $t \in [0,T]$. Then the function

(121)
$$u(t) = \sum_{i=0}^{m_n - 1} u_i g_{i+1}(t) - \sum_{i=0}^{m_n - 1} \sum_{j \in \mathbb{N}_{n-1} \smallsetminus D_i} (g_{\alpha_n - \alpha_j} * E^{(m_n - 1 - i)})(t) v_{i,j} + \sum_{i=m}^{m_n - 1} (g_{\alpha_n - \alpha} * E^{(m_n - 1 - i)})(t) v_{i,0} + \int_0^t E(t - s)G(s)ds, \quad 0 \le t \le T,$$

is a mild solution of the problem (113) on [0,T], where $v_{i,j} \in Y$ satisfy $A_j u_i = C_1 v_{i,j}$ for $0 \leq j \leq n-1$. If, additionally, $g \in C^1([0,T]:Y)$ and $(E^{(m_n-1)}(t))_{t \in [0,\tau]} \subseteq L(Y,X)$ is locally equicontinuous, then the solution u(t), given by (121), is a strong solution of (112) on [0,T].

Contrary to the assertion of [**346**, Theorem 3.7], the final conclusions of [**346**, Remark 3.8] cannot be proved for degenerate equations without imposing some additional conditions.

Concerning the action of subordination principles, we can state the following analogue of [**346**, Theorem 4.1] for degenerate multi-term problems (the final conclusions of [**346**, Remark 4.2] can be restated in our new setting, too).

THEOREM 2.3.35. Suppose $C_1 \in L(Y,X)$, $C_2 \in L(X)$ is injective and $\gamma \in (0,1)$.

 (i) Let ω ≥ max(0, abs(k)), and let the assumptions of Theorem 2.3.33(i)–(b) hold. Put

(122)
$$W_{\gamma}(t) := \int_0^\infty t^{-\gamma} \Phi_{\gamma}(t^{-\gamma}s) W(s) y \, ds, \quad t > 0, \ y \in Y \text{ and } W_{\gamma}(0) := W(0).$$

Define, for every $j \in \mathbb{N}_n^0$ and $t \ge 0$, $W_{j,\gamma}(t)$ by replacing W(t) in (122) with $W_j(t)$. Suppose that there exist a number $\nu > 0$ and a continuous kernel $k_{\gamma}(t)$ on $[0,\infty)$ satisfying (P1) and $\widetilde{k_{\gamma}}(\lambda) = \lambda^{\gamma-1} \tilde{k}(\lambda^{\gamma}), \lambda > \nu$. Then there exists an exponentially equicontinuous k_{γ} -regularized C_1 -existence family $(E_{\gamma}(t))_{t\ge 0}$ for (112), with α_j replaced by $\alpha_j\gamma$ therein $(0 \le j \le n)$. Furthermore, the family $\{(1+t^{\lceil \alpha_n \gamma \rceil - 2})^{-1}e^{-\omega^{1/\gamma}t}E_{\gamma}(t): t\ge 0\}$ is equicontinuous.

(ii) Suppose (U(t))_{t≥0} is a k-regularized C₂-uniqueness family for (112), and the family {e^{-ωt}U(t) : t≥0} is equicontinuous. Define, for every t≥ 0, U_γ(t) by replacing W(t) in (122) with U(t). Suppose that there exist a number ν > 0 and a continuous kernel k_γ(t) on [0,∞) satisfying (P1) and k_γ(λ) = λ<sup>γ(2-m_n)-2+[α_nγ] k̃(λ^γ), λ > ν. Then there exists a
</sup>

 k_{γ} -regularized C_2 -uniqueness family for (112), with α_j replaced by $\alpha_j\gamma$ therein $(0 \leq j \leq n)$. Furthermore, the family $\{e^{-\omega^{1/\gamma}t}U_{\gamma}(t) : t \geq 0\}$ is equicontinuous.

Of importance is the following abstract degenerate Volterra equation:

(123)
$$Bu(t) = f(t) + \sum_{j=0}^{n-1} (a_j * A_j u)(t), \quad t \in [0, \tau).$$

where $0 < \tau \leq \infty$, $f \in C([0,\tau) : X)$, $a_0, \ldots, a_{n-1} \in L^1_{loc}([0,\tau))$, and $A = A_0, \ldots, A_{n-1}, B$ are closed linear operators on X. By a mild solution, resp. strong solution, of (123), we mean any function $u \in C([0,\tau) : [D(B)])$ such that $A_j(a_j * u)(t) \in C([0,\tau) : X)$, $j \in \mathbb{N}^0_{n-1}$ and

$$Bu(t) = f(t) + \sum_{j=0}^{n-1} A_j(a_j * u)(t), \quad t \in [0, \tau)$$

resp. any function $u \in C([0,\tau) : [D(B)])$ such that $u(t) \in \bigcap_{j=0}^{n-1} D(A_j), t \in [0,\tau)$, the mapping $t \mapsto A_j u(t), t \in [0,\tau)$ is continuous for $j \in \mathbb{N}_{n-1}^0$, and (123) holds.

The following definition plays a crucial role in our investigation of problem (123).

DEFINITION 2.3.36. (cf. [346, Definition 4.3] for the case B = I) Suppose $0 < \tau \leq \infty, k \in C([0, \tau)), C_1 \in L(Y, X)$, and $C_2 \in L(X)$ is injective.

(i) A strongly continuous operator family $(E(t))_{t \in [0,\tau)} \subseteq L(Y, [D(B)])$ is said to be a (local, if $\tau < \infty$) k-regularized C₁-existence family for (123) iff

$$BE(t)y = k(t)C_1y + \sum_{j=0}^{n-1} A_j(a_j * E)(t)y, \quad t \in [0, \tau), \ y \in Y.$$

(ii) A strongly continuous operator family $(U(t))_{t \in [0,\tau)} \subseteq L(X)$ is said to be a (local, if $\tau < \infty$) k-regularized C₂-uniqueness family for (123) iff

$$U(t)Bx = k(t)C_2x + \sum_{j=0}^{n-1} (a_j * A_j U)(t)x, \quad t \in [0,\tau), \ x \in \bigcap_{j=0}^n D(A_j).$$

As in non-degenerate case, we have the following:

- (i) Suppose $(E(t))_{t\in[0,\tau)}$ is a k-regularized C_1 -existence family for (123). Then, for every $y \in Y$, the function u(t) = E(t)y, $t \in [0,\tau)$ is a mild solution of (123) with $f(t) = k(t)C_1y$, $t \in [0,\tau)$.
- (ii) Let $(U(t))_{t \in [0,\tau)}$ be a locally equicontinuous k-regularized C_2 -uniqueness family for (123). Then there exists at most one mild (strong) solution of (123).

The most important structural properties of k-regularized C_1 -existence families for (123) and k-regularized C_2 -uniqueness families for (123) are stated in the following analogue of Theorem 2.3.33. THEOREM 2.3.37. Suppose k(t) and $|a_0|(t), \ldots, |a_{n-1}|(t)$ satisfy (P1), $(E(t))_{t \ge 0} \subseteq L(Y, X), \quad (U(t))_{t \ge 0} \subseteq L(X), \quad \omega \ge \max(0, \operatorname{abs}(k), \operatorname{abs}(|a_0|), \ldots, \operatorname{abs}(|a_{n-1}|)), \quad C_1 \in L(Y, X) \text{ and } C_2 \in L(X) \text{ is injective. Set}$

$$\mathcal{P}_{\lambda} := B - \sum_{j=0}^{n-1} \widetilde{a_j}(\lambda) A_j, \quad \operatorname{Re} \lambda > \omega.$$

(i) (a) Let $(E(t))_{t \ge 0}$ be a k-regularized C_1 -existence family for (123), let the family $\{e^{-\omega t}E(t):t\ge 0\} \subseteq L(Y,[D(B)])$ be equicontinuous, and let the family $\{e^{-\omega t}A_j(a_j * E)(t):t\ge 0\} \subseteq L(Y,X)$ be equicontinuous $(0 \le j \le n-1)$. Then the following holds:

$$\mathcal{P}_{\lambda} \int_{0}^{\infty} e^{-\lambda t} E(t) y \, dt = \tilde{k}(\lambda) C_{1} y, \quad y \in Y, \ \operatorname{Re} \lambda > \omega.$$

. .

(b) Let the operator P_λ be injective for every λ > ω with k̃(λ) ≠ 0. Suppose, additionally, that there exist strongly continuous operator families (E(t))_{t≥0} ⊆ L(Y, X), (E_B(t))_{t≥0} ⊆ L(Y, X), and (E_j(t))_{t≥0} ⊆ L(Y, X) such that the operator families {e^{-ωt}E(t) : t ≥ 0}, {e^{-ωt}E_B(t) : t ≥ 0}, and {e^{-ωt}E_j(t) : t ≥ 0} are equicontinuous (0 ≤ j ≤ n − 1) as well as that:

$$\int_0^\infty e^{-\lambda t} E(t) y \, dt = \tilde{k}(\lambda) \mathcal{P}_{\lambda}^{-1} C_1 y, \quad \int_0^\infty e^{-\lambda t} E_B(t) y \, dt = \tilde{k}(\lambda) B \mathcal{P}_{\lambda}^{-1} C_1 y$$

and

$$\int_0^\infty e^{-\lambda t} E_j(t) y \, dt = \tilde{k}(\lambda) \tilde{a_j}(\lambda) A_j \mathcal{P}_{\lambda}^{-1} C_1 y,$$

for every $\lambda > \omega$ with $\tilde{k}(\lambda) \neq 0$, $y \in Y$ and $j \in \mathbb{N}_{n-1}^{0}$. Then $(E(t))_{t\geq 0}$ is a k-regularized C_1 -existence family for (123). Furthermore, $BE(t)y = E_B(t)y$, $t \geq 0$, $y \in Y$ and $A_j(a_j * E)(t)y = E_j(t)y$, $t \geq 0$, $y \in Y$, $j \in \mathbb{N}_{n-1}^{0}$.

(ii) Suppose $(U(t))_{t\geq 0}$ is strongly continuous and the operator family $\{e^{-\omega t}U(t):t\geq 0\}\subseteq L(X)$ is equicontinuous. Then $(U(t))_{t\geq 0}$ is a k-regularized C_2 -uniqueness family for (123) iff, for every $x\in \bigcap_{j=0}^n D(A_j)$, the following holds:

$$\int_0^\infty e^{-\lambda t} U(t) \mathcal{P}_\lambda x \, dt = \tilde{k}(\lambda) C_2 x, \quad \operatorname{Re} \lambda > \omega.$$

The proof of following subordination principle is standard and therefore omitted; observe that we correct here some inconsistencies made in the formulation of [**346**, Theorem 4.4].

THEOREM 2.3.38. (i) Suppose that the requirements of Theorem 2.3.37 (i)–(b) hold. Let c(t) be completely positive, let c(t), k(t), $k_1(t)$, $|a_0|(t), \ldots$, $|a_{n-1}|(t)$ and $|b_0|(t), \ldots, |b_{n-1}|(t)$ satisfy (P1), and let $\omega_0 > 0$ be such that, for every $\lambda > \omega_0$ with $\tilde{c}(\lambda) \neq 0$ and $\tilde{k}(1/\tilde{c}(\lambda)) \neq 0$, the following holds:

(124)
$$\widetilde{a}_{j}(1/\tilde{c}(\lambda)) = \widetilde{b}_{j}(\lambda), \quad j \in \mathbb{N}_{n-1}^{0} \text{ and } \widetilde{k}_{1}(\lambda) = \frac{1}{\lambda \tilde{c}(\lambda)} \widetilde{k}(1/\tilde{c}(\lambda)).$$

Then for each $r \in (0,1]$ there exists a locally Hölder continuous (with exponent r), exponentially equicontinuous (k_1*g_r) -regularized C_1 -existence family for

(125)
$$Bu(t) = f(t) + \sum_{j=0}^{n-1} (b_j * A_j u)(t), \quad t \in [0, \tau).$$

(ii) Suppose that the requirements of Theorem 2.3.37(ii) hold. Let c(t) be completely positive, let c(t), k(t), $k_1(t)$, $|a_0|(t), \ldots, |a_{n-1}|(t)$ and $|b_0|(t), \ldots, |b_{n-1}|(t)$ satisfy (P1), and let $\omega_0 > 0$ be such that, for every $\lambda > \omega_0$ with $\tilde{c}(\lambda) \neq 0$ and $\tilde{k}(1/\tilde{c}(\lambda)) \neq 0$, (124) holds. Then for each $r \in (0, 1]$ there exists a locally Hölder continuous (with exponent r), exponentially equicontinuous ($k_1 * g_r$)-regularized C_2 -uniqueness family for (125).

The interested reader may try to transfer the final conclusions of [**347**, Theorem 2.1, Theorem 2.2, Remark 2.1, Proposition 2.1] to degenerate multi-term fractional differential equations. Concerning [**347**, Theorem 2.3], we first need to introduce the following notion.

DEFINITION 2.3.39. A strongly continuous operator family $(U(t))_{t \in [0,\tau)} \subseteq L(X)$ is said to be a (local, if $\tau < \infty$) (k, C_2) -uniqueness family for (112) iff, for every $t \in [0, \tau)$ and $x \in \bigcap_{0 \le j \le n} D(A_j)$, the following holds:

$$U(t)Bx + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * U(\cdot)A_j x)(t) - (g_{\alpha_n - \alpha} * U(\cdot)Ax)(t)x = k(t)C_2 x.$$

Then it is clear that for any strongly continuous operator family $(U(t))_{t \in [0,\tau)}$ the following equivalence relation holds: $(U(t))_{t \in [0,\tau)}$ is a (local) $(k * g_{m_n-1}, C_2)$ uniqueness family for (112) iff $(U(t))_{t \in [0,\tau)}$ is a (local) k-regularized C_2 -uniqueness family for (112).

Consider now the perturbed equation:

(126)
$$B\mathbf{D}_{t}^{\alpha_{n}}u(t) + \sum_{i=1}^{n-1} (A_{i} + F_{i})\mathbf{D}_{t}^{\alpha_{i}}u(t) = (A + F)\mathbf{D}_{t}^{\alpha}u(t) + f(t), \quad t \ge 0;$$
$$u^{(j)}(0) = u_{j}, \quad j = 0, \dots, \lceil \alpha_{n} \rceil - 1,$$

where $F_i \in L(X)$ for $0 \leq i \leq n-1$ and $F_0 \equiv F$. A similar line of reasoning as in the proof of [**347**, Theorem 2.3] shows that the following result about the *C*-wellposedness of problem (126) holds good (observe that the employed method is based on the arguments contained in the proof of [**459**, Theorem 6.1], which does not work any longer if we replace the term $B\mathbf{D}_t^{\alpha_n}u(t)$, in (126), with $(B + F_n)\mathbf{D}_t^{\alpha_n}u(t)$):

- THEOREM 2.3.40. (i) Suppose Y = X, $(E(t))_{t \in [0,\tau)} \subseteq L(X)$ is a (local) C_1 -existence family for (112), $E_j \in L(X)$ and $F_j = C_1 E_j$ $(j \in \mathbb{N}_{n-1}^0)$. Suppose that the following conditions hold:
 - (a) For every $p \in \circledast_X$ and for every $T \in (0, \tau)$, there exists $c_{p,T} > 0$ such that

$$p(E^{(m_n-1)}(t)x) \leq c_{p,T}p(x), \quad x \in X, \ t \in [0,T].$$

(b) For every $p \in \circledast_X$, there exists $c_p > 0$ such that

$$p(E_j x) \leqslant c_p p(x), \quad j \in \mathbb{N}_{n-1}^0, \ x \in X.$$

(c) $\alpha_n - \alpha_{n-1} \ge 1$ and $\alpha_n - \alpha \ge 1$.

Then there exists a (local) C_1 -existence propagation family $(R(t))_{t\in[0,\tau)}$ for (126). If $\tau = \infty$ and if, for every $p \in \circledast_X$, there exist $M \ge 1$ and $\omega \ge 0$ such that

(127)
$$p(E^{(m_n-1)}(t)x) \leqslant M e^{\omega t} p(x), \quad t \ge 0, \ x \in X,$$

respectively (127) and

(128)
$$p(BE^{(m_n-1)}(t)x) \leq Me^{\omega t}p(x), \quad t \ge 0, \ x \in X,$$

then $(R(t))_{t\geq 0}$ is exponentially equicontinuous, and moreover, $(R(t))_{t\geq 0}$ also satisfies the condition (127), repectively (127) and (128), with possibly different numbers $M \geq 1$ and $\omega > 0$.

(ii) Suppose Y = X, (U(t))_{t∈[0,τ)} ⊆ L(X) is a (local) (1, C₂)-uniqueness family for (112), E_j ∈ L(E) and F_j = E_jC₂ (j ∈ N⁰_{n-1}). Suppose that (b)-(c) hold, and that (a) holds with (E^(m_n-1)(t))_{t∈[0,τ}) replaced by (U(t))_{t∈[0,τ}) therein. Then there exists a (local) (1, C₂)-uniqueness family (W(t))_{t∈[0,τ}) for (126). If τ = ∞ and if, for every p ∈ ⊛_X, there exist M ≥ 1 and ω ≥ 0 such that (127) holds, then (W(t))_{t≥0} is exponentially equicontinuous, and moreover, (W(t))_{t≥0} also satisfies the condition (127), with possibly different numbers M ≥ 1 and ω > 0.

Concerning the existence of strong solutions of (112), we can prove the following slight extension of [347, Theorem 3.1]; this result can be viewed of some independent interest and details of proof will be given later, when we will be analyzing abstract degenerate multi-term fractional differential equations with Riemann– Liouville derivatives.

THEOREM 2.3.41. (cf. alsoTheorem 2.4.6) Suppose A, B, A_1, \ldots, A_{n-1} are closed linear operators on X, $\omega > 0$, $L(X) \ni C$ is injective and $u_0, \ldots, u_{m_n-1} \in X$. Set $P_{\lambda} := \lambda^{\alpha_n - \alpha} B + \sum_{j=1}^{n-1} \lambda^{\alpha_j - \alpha} A_j - A$, $\lambda \in \mathbb{C} \setminus \{0\}$. Let the following conditions hold:

- (i) The operator P_{λ} is injective for $\lambda > \omega$ and $D(P_{\lambda}^{-1}C) = X$, $\lambda > \omega$.
- (ii) If $0 \leq j \leq n$, $0 \leq k \leq m_n 1$, m 1 < k, $1 \leq l \leq n$, $m_l 1 \geq k$ and $\lambda > \omega$, then $Cu_k \in D(P_{\lambda}^{-1}A_l)$,

(129)
$$A_{j}\left\{\lambda^{\alpha_{j}}\left[\lambda^{\alpha_{n}-\alpha-k-1}P_{\lambda}^{-1}BCu_{k}+\sum_{l\in D_{k}}\lambda^{\alpha_{l}-\alpha-k-1}P_{\lambda}^{-1}A_{l}Cu_{k}\right]-\sum_{l=0}^{m_{j}-1}\delta_{kl}\lambda^{\alpha_{j}-1-l}Cu_{k}\right\}\in LT-X$$

and

(130)
$$\lambda^{\alpha_n} \left[\lambda^{\alpha_n - \alpha - k - 1} P_{\lambda}^{-1} B C u_k + \sum_{l \in D_k} \lambda^{\alpha_l - \alpha - k - 1} P_{\lambda}^{-1} A_l C u_k \right] - \lambda^{\alpha_n - 1 - k} C u_k \in LT - X.$$

(iii) If $0 \leq j \leq n, 0 \leq k \leq m_n - 1, m - 1 \geq k, \mathbb{N}_{n-1} \setminus D_k \neq \emptyset, s \in \mathbb{N}_{n-1} \setminus D_k$ and $\lambda > \omega$, then $Cu_k \in D(A_s), \sum_{l \in \mathbb{N}_{n-1} \setminus D_k} \lambda^{\alpha_l - \alpha - k - 1} A_l Cu_k \in D(P_\lambda^{-1}),$

(131)
$$A_{j}\left\{\lambda^{\alpha_{j}}\left[\lambda^{-k-1}Cu_{k}-P_{\lambda}^{-1}\sum_{l\in\mathbb{N}_{n-1}\smallsetminus D_{k}}\lambda^{\alpha_{l}-\alpha-k-1}A_{l}Cu_{k}\right]-\sum_{l=0}^{m_{j}-1}\delta_{kl}\lambda^{\alpha_{j}-1-l}Cu_{k}\right\}\in LT-X$$

and

(132)
$$\lambda^{\alpha_n} \left[\lambda^{-k-1} C u_k - P_{\lambda}^{-1} \sum_{l \in \mathbb{N}_{n-1} \smallsetminus D_k} \lambda^{\alpha_l - \alpha - k - 1} A_l C u_k \right] - \lambda^{\alpha_n - 1 - k} C u_k \in LT - X.$$

Then the abstract Cauchy problem (112) has a strong solution, with u_k replaced by $Cu_k \ (0 \leq k \leq m_n - 1)$.

REMARK 2.3.42. Let $0 \leq k \leq m_n - 1$ and m - 1 < k. Then Theorem 2.3.41 continues to hold if we replace the term

$$\lambda^{\alpha_n - \alpha - k - 1} P_{\lambda}^{-1} B C u_k + \sum_{l \in D_k} \lambda^{\alpha_l - \alpha - k - 1} P_{\lambda}^{-1} A_l C u_k$$

i.e., the Laplace transform of $u_k(t)$, in (129)–(130) by

$$\lambda^{-k-1}Cu_k - \sum_{l \in \mathbb{N}_{n-1} \smallsetminus D_k} \lambda^{\alpha_l - \alpha - k - 1} P_{\lambda}^{-1} A_l Cu_k + \lambda^{-k-1} P_{\lambda}^{-1} A Cu_k;$$

in this case, it is indispensable to assume that $Cu_k \in D(P_{\lambda}^{-1}A_l)$, provided $0 \leq l \leq n-1$, $k > m_l - 1$ and $\lambda > \omega$. Let us also observe that a similar modification can be made in the case $0 \leq k \leq m_n - 1$ and $m-1 \geq k$. Strictly speaking, one can replace the term

$$\lambda^{-k-1}Cu_k - P_{\lambda}^{-1} \sum_{l \in \mathbb{N}_{n-1} \smallsetminus D_k} \lambda^{\alpha_l - \alpha - k - 1} A_l Cu_k$$

i.e., the Laplace transform of $u_k(t)$, in (131)–(132) by

$$\lambda^{\alpha_n - \alpha - k - 1} P_{\lambda}^{-1} B C u_k + \sum_{l \in D_k} \lambda^{\alpha_l - \alpha - k - 1} P_{\lambda}^{-1} A_l C u_k - \lambda^{-k - 1} P_{\lambda}^{-1} A C u_k;$$

in this case, one has to assume that $Cu_k \in D(P_{\lambda}^{-1}A_l)$, provided $0 \leq l \leq n$, $m_l - 1 \geq k$ and $\lambda > \omega$.

This is the right time to illustrate our results with some examples.

EXAMPLE 2.3.43. Suppose $1 \leq p < \infty, \emptyset \neq \Omega \subseteq \mathbb{R}^n$ is an open bounded domain with smooth boundary, and $X := L^p(\Omega)$. Consider the equation

(133)
$$(\alpha - \Delta)u_{tt} = \beta \Delta u_t + \Delta u + \int_0^t g(t - s)\Delta u(s, x)ds, \quad t > 0, \ x \in \Omega;$$
$$u(0, x) = \phi(x), \quad u_t(0, x) = \psi(x),$$

where $g \in L^1_{loc}([0,\infty))$, $\alpha \in \mathbb{R}$ and $\beta \in \mathbb{R} \setminus \{0\}$. As explained by M. V. Falaleev and S. S. Orlov in [174], the equation (133) appears in the models of nonlinear viscoelasticity provided n = 3. Integrating (133) twice with the respect to the time-variable t, we obtain the associated integral equation

(134)
$$(\alpha - \Delta)u(t) = (\alpha + (\beta - 1)\Delta)\phi(x) + t(\alpha - \Delta)\psi + \beta\Delta(g_1 * u)(t) + \Delta(g_2 * u) + \Delta(g_2 * g * u)(t),$$

which is of the form (123) with $B := \alpha - \Delta$, $A_2 := \beta \Delta$, $A_1 = A_0 := \Delta$ (acting with the Dirichlet boundary conditions) and $a_2(t) := g_1(t)$, $a_1(t) := g_2(t)$, $a_0(t) := (g_2 * g)(t)$. Then

$$\mathcal{P}_{\lambda} = \frac{\lambda^2 + \beta\lambda + \tilde{g}(\lambda) + 1}{\lambda^2} \bigg[\frac{\alpha\lambda^2}{\lambda^2 + \beta\lambda + \tilde{g}(\lambda) + 1} - \Delta \bigg].$$

(i) In the first part of our analysis, we assume that $\alpha > 0$ and g(t) is of the following form:

$$g(t) = \sum_{j=0}^{l} c_j g_{\beta_j}(t) + f(t), \quad t > 0,$$

where $l \in \mathbb{N}$, $c_j \in \mathbb{C}$ $(0 \leq j \leq l)$, $0 < \beta_1 < \cdots < \beta_l < 1$ and the function f(t) satisfies the requirements of [**296**, Theorem 3.4(i)-(a)] with $\alpha = \pi/2$ and $\omega > 0$ sufficiently large. Using the fact that the operator Δ generates a bounded analytic C_0 -semigroup of angle $\pi/2$, and the resolvent equation, it can be simply verified that

$$\frac{1}{\lambda}\mathcal{P}_{\lambda}^{-1} \in LT - L(X), \ \frac{1}{\lambda}B\mathcal{P}_{\lambda}^{-1} \in LT - L(X) \text{ and } \frac{\widetilde{a_j}(\lambda)}{\lambda}\mathcal{P}_{\lambda}^{-1} \in LT - L(X), \ j = 0, 1, 2.$$

This implies by Theorem 2.3.37 that there exists an exponentially bounded I-existence family $(E(t))_{t\geq 0}$ for (134), satisfying additionally that for each $f \in X$ the mappings $t \mapsto E(t)f$, t > 0, $t \mapsto BE(t)f$, t > 0 and $t \mapsto A_j(a_j * E)(t)f$, t > 0 can be analytically extended to the sector $\Sigma_{\pi/2}$; furthermore, $(E(t))_{t\geq 0}$ is an exponentially bounded I-uniqueness family

for (134). Therefore, for every ϕ , $\psi \in W^{2,p}(\Omega) \cap W_0^{1,p}(\Omega)$, there exists a unique strong solution

$$u(t) = E(t)(\alpha + (\beta - 1)\Delta)\phi + \int_0^t E(s)(\alpha - \Delta)\psi \, ds, \quad t \ge 0,$$

of the integral equation (134), and u(t) can be analytically extended to the sector $\Sigma_{\pi/2}$.

(ii) Denote by $\{\lambda_k\} = \sigma(\Delta)$ the eigenvalues of the Dirichlet Laplacian Δ in $L^2(\Omega)$ (recall that $0 < -\lambda_1 \leq -\lambda_2 \ldots \leq -\lambda_k \leq \cdots \to +\infty$ as $k \to \infty$; cf. Example 2.3.48-Example 2.3.53 for more details); by $\{\phi_k\} \subseteq C^{\infty}(\Omega)$ we denote the corresponding set of mutually orthogonal [in the sense of $L^2(\Omega)$] eigenfunctions. In this part, we will delve into the case in which $\alpha = \lambda_{k_0} \in \sigma(\Delta)$ for some $k_0 \in \mathbb{N}$ and the function g(t) satisfies (P1). Then there exist finite constants $M \ge 1$ and $\omega \ge 0$ such that $|\int_0^t g(s)ds| \le Me^{\omega t}, t \ge 0$ and $\lambda \int_0^\infty e^{-\lambda t} \int_0^t g(s)ds dt = \int_0^\infty e^{-\lambda t} g(t)dt, \lambda > \omega$ [27], which simply implies that the set $\{\tilde{g}(\lambda) : \lambda > \omega + 1\}$ is bounded. Define $D: L^2(\Omega) \to L^2(\Omega)$ by $Df := (-1)\beta^{-1}\sum_{k=1}^\infty \langle \phi_k, f \rangle \phi_k, f \in L^2(\Omega)$. Using Parseval's equality, it can be simply verified that $D, BD \in L(L^2(\Omega))$; furthermore, $||R(\lambda : \Delta)|| = O(|\alpha - \lambda|^{-1})$ as $\lambda \to \alpha$ (see e.g. Example 2.3.48). Using the resolvent equation and these facts, we obtain the existence of a sufficiently large real number R > 0 such that $\mathbf{P}_{\lambda}^{-1} \in L(L^2(\Omega))$ for $|\lambda| \ge R$, as well as that

(135)
$$|\lambda|^{-2} \left[\|\mathbf{P}_{\lambda}^{-1}\| + \|B\mathbf{P}_{\lambda}^{-1}\| + \sum_{j=0}^{2} \|\widetilde{a}_{j}(\lambda)A_{j}\mathbf{P}_{\lambda}^{-1}\| \right] \leqslant M, \quad |\lambda| \ge R,$$

$$\begin{split} \lim_{|\lambda|\to\infty} \lambda^{-1} \mathbf{P}_{\lambda}^{-1} f &= Df, \quad \lim_{|\lambda|\to\infty} \lambda^{-1} B \mathbf{P}_{\lambda}^{-1} f &= BDf \text{ and } \\ \lim_{|\lambda|\to\infty} \lambda^{-1} \tilde{a}_{j}(\lambda) \mathbf{P}_{\lambda}^{-1} f &= 0, \ 0 \leq j \leq 2 \ (f \in L^{2}(\Omega)). \end{split}$$
 Making use of Theorem 2.3.37, we obtain that there exists an exponentially bounded once integrated *I*-existence family $(E_{1}(t))_{t\geq0}$ for (134), satisfying additionally that for each $f \in L^{2}(\Omega)$ the mappings $t \mapsto E_{1}(t)f, \ t > 0$, $t \mapsto BE_{1}(t)f, \ t > 0$ and $t \mapsto A_{j}(a_{j} * E_{1})(t)f, \ t > 0$ can be analytically extended to the sector $\Sigma_{\pi/2}$; furthermore, $(E_{1}(t))_{t\geq0}$ is an exponentially bounded once integrated *I*-uniqueness family for (134). Therefore, for every $\phi, \psi \in H^{2}(\Omega) \cap H_{0}^{1}(\Omega)$, there exists a unique strong solution of the associated once integrated problem

(136)
$$(\alpha - \Delta)u(t) = t(\alpha + (\beta - 1)\Delta)\phi(x) + \frac{t^2}{2}(\alpha - \Delta)\psi + \beta\Delta(g_1 * u)(t) + \Delta(g_2 * u) + \Delta(g_2 * g * u)(t), \quad t \ge 0$$

given by $u(t) = E_1(t)(\alpha + (\beta - 1)\Delta)\phi + \int_0^t E_1(s)(\alpha - \Delta)\psi \, ds, t \ge 0$. It should be noticed that the mappings $t \mapsto u(t), t > 0$ and $t \mapsto Bu(t), t > 0$ can be analytically extended to the sector $\Sigma_{\pi/2}$ and that it is not clear whether there exist analytical extensions of these mappings to the sector Σ_{π} in our general choice of function g(t) (in [174, Theorem 5], the authors have investigated the existence and uniqueness of generalized $(C^2([0,\infty) : L^2(\Omega)))$ solutions of problem (134)). On the other hand, (135), the equality $\lim_{|\lambda|\to\infty} \lambda^{-1} B \mathcal{P}_{\lambda}^{-1} f = BDf$, Theorem 3.2.19 and Remark 3.1.2(v) taken together imply that for each $\theta \in (-\pi,\pi]$ the MLO $e^{i\theta}AB^{-1}$ generates an exponentially bounded, analytic once integrated $(b+g_2(t)+(g_2*g)(t),I)$ -regularized resolvent family $(E_{1,B}(t) \equiv BE_1(t))_{t\geq 0}$ of angle $\pi/2$; cf. Section 3.2 for the notion. Since [**292**, Theorem 2.1.29(ii)] holds in the MLO framework, this immediately yields some results on the existence and uniqueness of analytical (entire) solutions of the problem (136) with the term $t(\alpha + (\beta - 1)\Delta)\phi + \frac{t^2}{2}(\alpha - \Delta)\psi$ replaced by a general inhomogeneity f(t).

(iii) Consider the equation

(137)
$$(\nu - \Delta)u_t = \Delta u + \int_0^t g(t - s)\Delta u(s, x)ds, \quad t > 0, \ x \in \Omega;$$
$$u(0, x) = \phi(x), \quad x \in \Omega,$$

in $L^2(\Omega)$, where $g \in L^1_{loc}([0,\infty))$ and $\nu = \lambda_{k_0} \in \sigma(\Delta)$. This equation appears in the study of non-linear dynamics of hereditarily elastic bodies (cf. [174, Theorem 4]). Clearly, in our new setting, we have that B = $v - \Delta$, n = 1, $a_0(t) = g_1(t)$, $a_1(t) = (g_1 * g)(t)$ and $A_0 = A_1 = \Delta$. Concerning the function g(t), we assume that $\operatorname{abs}(|g|) < \infty$. Arguing as in (ii), we get that there exists an exponentially bounded once integrated *I*-existence family $(E_1(t))_{t\geq 0}$ for (137), satisfying additionally that for each $f \in L^2(\Omega)$ the mappings $t \mapsto E_1(t)f$, t > 0, $t \mapsto BE_1(t)f$, t > 0 and $t \mapsto A_j(a_j * E_1)(t)f$, t > 0 can be analytically extended to the sector $\Sigma_{\pi/2}$, as well as that $(E_1(t))_{t\geq 0}$ is an exponentially bounded once integrated *I*uniqueness family for (137). This, in turn, implies that there exists a unique strong solution $t \mapsto u(t)$, $t \geq 0$ of the associated once integrated equation:

$$(v - \Delta)u(t) = t(v - \Delta)\phi + \Delta(g_1 * u)(t) + \Delta(g_1 * g * u), \quad t \ge 0.$$

Furthermore, the mappings $t \mapsto u(t), t > 0$ and $t \mapsto Bu(t), t > 0$ can be analytically extended to the sector $\Sigma_{\pi/2}$.

EXAMPLE 2.3.44. Suppose $1 , <math>X := L^p(\mathbb{R}^n)$, $1/2 < \gamma \leq 1$, $Q \in \mathbb{N} \setminus \{1\}$, $P_1(x) = \sum_{|\eta| \leq Q} a_{\eta} x^{\eta}$, $P_2(x) = -1 - |x|^2$ $(x \in \mathbb{R}^n)$, $P_1(x)$ is positive, $\sigma \geq 0$, the estimate

$$\left| D^{\eta} \left(\frac{P_1(x)}{P_2(x)} \right) \right| \leqslant c_{\eta} (1+|x|)^{|\eta|(\sigma-1)}, \quad x \in \mathbb{R}^n$$

holds for each multi-index $\eta \in \mathbb{N}_0^n$ with $|\eta| > 0$, $V_2 \ge 0$ and for each $\eta \in \mathbb{N}_0^n$ there exists $M_\eta > 0$ such that $|D^\eta(P_2(x)^{-1})| \le M_\eta(1+|x|)^{|\eta|(V_2-1)}$, $x \in \mathbb{R}^n$. Set $A_2 := \Delta - I$, $A_0 f := \sum_{|\eta| \le Q} a_\eta D^\eta f$ with maximal distributional domain, where $D^\eta \equiv (-i)^{|\eta|} f^\eta$, and $C_1 := (I-\Delta)^{-\frac{n}{2}|\frac{1}{p}-\frac{1}{2}|\max(\sigma,V_2)}$. Let $E_i \in L(X)$ and $F_i = C_1 E_i$ (i = 0, 1). Then we know (cf. [306, 307]) that $\lambda(\lambda^2 A_2 - A_0)^{-1} C_1 \in LT - L(X)$ and $\lambda A_2(\lambda^2 A_2 - A_0)^{-1}C_1 \in LT - L(X)$, which implies by Theorem 2.3.37(i)–(b) that there exists an exponentially bounded C_1 -existence family $(E(t))_{t\geq 0}$ for the following degenerate second order Cauchy problem:

$$\begin{cases} (\Delta - I)u_{tt}(t, x) = \sum_{|\eta| \leqslant Q} a_{\eta} D^{\eta} u(t, x), & t \ge 0, \ x \in \mathbb{R}^n, \\ u(0, x) = u_0(x) = \phi(x), & u_t(0, x) = u_1(x) = \psi(x), \end{cases}$$

obeying the properties (127)–(128) stated in the formulation of Theorem 2.3.40. Applying Theorem 2.3.40, we get there exists an exponentially bounded C_1 -existence family $(R(t))_{t\geq 0}$ for the following degenerate second order Cauchy problem:

$$(P)': \begin{cases} (\Delta - I)u_{tt}(t, x) + F_1 u_t(t, x) = (\sum_{|\eta| \le Q} a_\eta D^\eta + F_0)u(t, x), \ t \ge 0, \ x \in \mathbb{R}^n, \\ u(0, x) = u_0(x) = \phi(x), \quad u_t(0, x) = u_1(x) = \psi(x). \end{cases}$$

Then Theorem 2.3.32(i) shows that there exists a strong solution of problem (P)' provided that

$$\phi, \psi \in \mathcal{S}^{Q+n|\frac{1}{p}-\frac{1}{2}|\max(\sigma, V_2), p}(\mathbb{R}^n), \quad (A_0+F)\phi \in \mathcal{S}^{n|\frac{1}{p}-\frac{1}{2}|\max(\sigma, V_2), p}(\mathbb{R}^n),$$

$$F_1\psi \in \mathcal{S}^{n|\frac{1}{p}-\frac{1}{2}|\max(\sigma, V_2), p}(\mathbb{R}^n) \text{ and } (A_0+F)\psi \in \mathcal{S}^{n|\frac{1}{p}-\frac{1}{2}|\max(\sigma, V_2), p}(\mathbb{R}^n).$$

If we denote by U(t, x), resp. V(t, x), the corresponding strong solution of problem (P)' with the initial values $\phi(x)$ and $\psi(x) \equiv 0$, resp. $\phi(x) \equiv 0$ and $\psi(x)$, then one can simply verify that the function

$$u(t,x) := \int_0^\infty t^{-\gamma} \Phi_\gamma(st^{-\gamma}) U(s,x) ds + \int_0^t g_{1-\gamma}(t-s) \int_0^\infty s^{-\gamma} \Phi_\gamma(rs^{-\gamma}) V(r,x) dr ds,$$

is a strong solution of the following integral equation

(138)
$$A_{2}[u(t,x) - \phi(x) - t\psi(x)] + F_{1} \int_{0}^{t} g_{\gamma}(t-s)[u(s,x) - \phi(x)]ds$$
$$= \int_{0}^{t} g_{2\gamma}(t-s)(A_{0} + F)u(s,x)ds, \quad t \ge 0, \ x \in \mathbb{R}^{n};$$

furthermore, the function $t \mapsto u(t, \cdot) \in X$ can be analytically extended to the sector $\Sigma_{(\frac{1}{\gamma}-1)\frac{\pi}{2}}$. In the present situation, we can only prove that there is at most one strong solution of the integral equation (138) provided that p = 2. Strictly speaking, suppose that u(t, x) is a strong solution of (138) with $\phi(x) \equiv \psi(x) \equiv 0$. Then $A_2^{-1} \in L(X), C_1 = I$ and the function $v(t, x) := A_2 u(t, x)$ is a strong solution of the following non-degenerate integral equation

(139)
$$u(t,x) + \int_0^t g_{\gamma}(t-s)F_1A_2^{-1}u(s,x)ds$$

= $\int_0^t g_{2\gamma}(t-s)(A_0A_2^{-1} + FA_2^{-1})u(s,x)ds, \quad t \ge 0, \ x \in \mathbb{R}^n.$

Since $\lambda(\lambda^2 - A_0A_2^{-1})^{-1} = \lambda A_2(\lambda^2 A_2 - A_0)^{-1} \in LT - L(X)$, the operator $A_0A_2^{-1}$ generates a cosine operator function and we can apply Theorem 2.3.40(ii) in order to see that there exists an exponentially bounded (1, I)-uniqueness family for (139),

with the meaning clear. This proves the claimed assertion on the uniqueness of strong solutions of problem (138). In general case $p \neq 2$, it is not clear how we can prove that there is at most one strong solution of the integral equation (138) without assuming that F_1 and F take some specific forms.

In the remainder of this section, we analyze entire and analytical properties of degenerate resolvent operator families introduced above and apply obtained results in the study of abstract Boussinesq-Love equation and the abstract Barenblatt–Zheltov–Kochina equation (cf. also Example 2.11.19 and the paragraph directly before Subsection 2.2.4; for some other references, one may refer e.g. to [96] and [98]). We reconsider some results obtained by G. A. Sviridyuk and A. A. Zamyshlyaeva in [515, Section 5], and slightly improve the assertion of [509, Theorem 5.1.3(ii)] in L^2 type spaces. We divide our investigation into two parts. In the first part, our standing hypothesis is that the orders α , $\alpha_1, \ldots, \alpha_n$ of Caputo derivatives $\mathbf{D}_t^{\alpha} u(t)$, $\mathbf{D}_t^{\alpha_n} u(t)$, appearing in (112), are non-negative integers; in the second part, where we analyze some fractional analogues of the abstract Boussinesq–Love equation and the abstract Barenblatt–Zheltov–Kochina equation, some of orders α , $\alpha_1, \ldots, \alpha_n$ can be purely fractional numbers.

DEFINITION 2.3.45. (cf. also Definition 2.3.50 and Definition 2.6.1 below) Let $\alpha_i \in \mathbb{N}_0$ for all $i \in \mathbb{N}_n^0$, and let the function $u \in C^{\alpha_n - 1}([0, \infty) : X)$ be a strong solution of problem (112). Then we say that $u(\cdot)$ is an entire solution of problem (112) iff the functions $u(\cdot)$ and $Bu^{(\alpha_n)}(\cdot)$, $A_1u^{(\alpha_1)}(\cdot)$, ..., $A_{n-1}u^{(\alpha_{n-1})}(\cdot)$, $Au^{(\alpha)}(\cdot)$ can be analytically extended from the interval $[0, \infty)$ to the whole complex plane.

Now we are ready to formulate the following theorem.

THEOREM 2.3.46. Suppose k(t) satisfies (P1), $C_1 \in L(Y, X)$ and $C_2 \in L(X)$ is injective. Let $\alpha_i \in \mathbb{N}_0$ for all $i \in \mathbb{N}_n^0$, and let there exist a locally equicontinuous k-regularized C_2 -uniqueness family for (112). Let the following hold:

- Suppose that there exists a sufficiently large number R > 0 such that the operator \mathbf{P}_z is injective for $|z| \ge R$, as well as that the operator families $\{\mathbf{P}_z^{-1}C_1 : |z| \ge R\} \subseteq L(Y, X)$ and $\{z^{\alpha_j - \alpha_n}A_j\mathbf{P}_z^{-1}C_1 : |z| \ge R,$ $j \in \mathbb{N}_n^n\} \subseteq L(Y, X)$ are equicontinuous.
- Let the mappings $z \mapsto \mathbf{P}_z^{-1}C_1 y \in X$, |z| > R and $z \mapsto z^{\alpha_j \alpha_n} A_j \mathbf{P}_z^{-1}C_1 y \in X$, |z| > R be analytic for any $y \in Y$, $j \in \mathbb{N}_n^0$, and let there exist operators $D, D_0, D_1, \ldots, D_n \in L(Y, X)$ such that $\lim_{z \to \infty} \mathbf{P}_z^{-1}C_1 y = Dy$ and $\lim_{z \to \infty} z^{\alpha_j \alpha_n} A_j \mathbf{P}_z^{-1}C_1 y = D_j y$ for any $y \in Y$, $j \in \mathbb{N}_n^0$.
- Suppose that $u_i \in \tilde{\mathbf{D}}_i$ for $0 \leq i \leq m_n 1$, $v_{i,j} \in Y$ satisfy $A_j u_i = C_1 v_{i,j}$ for $0 \leq j \leq n - 1$, as well as that $g \in C^1([0,\infty):Y)$ and $C_1g(t) = f(t)$, $t \geq 0$.

Then there exists a unique strong solution of (112).

 Assume, additionally, that the function t → g(t), t ≥ 0 can be analytically extended to the whole complex plane, resp., to a continuously differentiable function ℝ → Y.

Then there exists a unique entire solution $u(\cdot)$ of (112), resp., the function $t \mapsto u(t)$, $t \ge 0$ can be extended to an α_n -times continuously differentiable function

 $\mathbb{R} \mapsto X \text{ and the functions } Bu^{(\alpha_n)}(\cdot), A_1u^{(\alpha_1)}(\cdot), \ldots, A_{n-1}u^{(\alpha_{n-1})}(\cdot), Au^{(\alpha)}(\cdot) \text{ can}$ be extended to continuously differentiable functions $\mathbb{R} \mapsto X$. Furthermore, in any case set out above, the existence of a positive real number $\omega' > 0$ such that the set $\{e^{-\omega's}g(s): s \ge 0\}$, resp., $\{e^{-\omega'z}g(z): z \in \mathbb{C}\}$ ($\{e^{-\omega'|s|}g(s): s \in \mathbb{R}\}$) is bounded in Y, implies the existence of a positive real number $\omega'' > 0$ such that the set $\{e^{-\omega''s}u(s): s \ge 0\}$, resp., $\{e^{-\omega''z}u(z): z \in \mathbb{C}\}$ ($\{e^{-\omega''|s|}u(s): s \in \mathbb{R}\}$) is bounded in X.

PROOF. Let $\beta \in (-\pi, \pi]$. Then, for every $\theta \in (0, \pi/2)$, there exists a sufficiently large number $\omega_{\beta,\theta} > 0$ satisfying that the function $q_{\beta,\theta}(z) := z^{-1} \mathbf{P}_{ze^{-i\beta}}^{-1} C_1 \in$ $L(Y,X), z \in \omega_{\beta,\theta} + \Sigma_{\theta+(\pi/2)}$ is well-defined, strongly analytic and that for each $\theta' \in (0,\theta)$ the operator family $\{z^{-1}(z-\omega_{\beta,\theta})\mathbf{P}_{ze^{-i\beta}}^{-1}C_1 : z \in \omega_{\beta,\theta} + \Sigma_{\theta'+(\pi/2)}\} \subseteq L(Y,X)$ is equicontinuous. By Theorem 1.4.10(i), we obtain that for each $y \in Y$ there exists an X-valued analytic mapping $z \mapsto W_{\beta,y}(z), z \in \Sigma_{\pi/2}$ satisfying that, for every $\theta \in (0, \pi/2)$, we have that $\int_0^\infty e^{-zt} W_{\beta,y}(t) dt = z^{-1} \mathbf{P}_{ze^{-i\beta}}^{-1} C_1 y$, Re $z > \omega_{\beta,\theta}$ and the set $\{e^{-\omega_{\beta,\theta} z} W_{\beta,y}(z) : z \in \Sigma_{\theta'}\}$ is bounded in X $(y \in Y, \theta' \in (0,\theta))$. Define $W_{\beta}(z)y := W_{\beta,y}(z), z \in \Sigma_{\pi/2}, y \in Y$. By the uniqueness theorem for Laplace transform, it readily follows that $W_{\beta}(z): Y \to X$ is a linear mapping $(z \in \Sigma_{\pi/2})$; furthermore, we can argue as in the proofs of [292, Theorem 2.2.5] and [27, Theorem 2.6.1] so as to conclude that, for every $\theta \in (0, \pi/2)$, $\{e^{-\omega_{\beta,\theta}z}W_{\beta}(z):$ $z \in \Sigma_{\theta'} \subset L(Y, X)$ is an equicontinuous operator family $(\theta' \in (0, \theta))$. Since $\lim_{z\to\infty} \mathbf{P}_z^{-1} C_1 y = Dy \ (y \in Y)$, we can apply Theorem 1.4.10(ii)/(iii) in order to see that, for every $y \in Y$ and $\theta \in (0, \pi/2)$, we have $\lim_{z \to 0, z \in \Sigma_{\theta}} W_{\beta}(z)y = Dy$. Now we will prove that, for every $z \in \Sigma_{\pi/2} \cap e^{i\pi/2} \Sigma_{\pi/2}$, we have $W_0(z) = W_{\pi/2}(ze^{-i\pi/2})$. Let $y \in Y$ be fixed, and let $\arg(z) = \theta$. Set $\Gamma_{\theta} := \{e^{i\theta}t : t \ge 0\}$. Using Cauchy's formula, it is not difficult to see that, for all sufficiently large values of positive real parameter s > 0, we have

$$\int_0^\infty e^{-st} W_0(e^{i\theta}t) y \, dt = e^{-i\theta} \int_{\Gamma_\theta} e^{-se^{-i\theta}v} W_0(v) y \, dv$$
$$= e^{-i\theta} \int_0^\infty e^{-se^{-i\theta}v} W_0(v) y \, dv = s^{-1} \mathbf{P}_{se^{-i\theta}}^{-1} C_1 y, \quad y \in Y.$$

Similarly, $\int_0^\infty e^{-st} W_{\pi/2}(e^{i(\theta-\pi/2)}t)y \, dt = s^{-1} \mathbf{P}_{se^{i\theta}}^{-1} C_1 y, \ y \in Y$ so that the uniqueness theorem for Laplace transform implies that $W_0(e^{i\theta}t) = W_{\pi/2}(e^{i(\theta-\pi/2)}t)$ for all $t \ge 0$. Plugging t = |z|, we get that $W_0(z) = W_{\pi/2}(ze^{-i\pi/2})$, as claimed. A similar line of reasoning shows that the operator family $(W(z))_{z\in\mathbb{C}}$, where

$$W(z) := \begin{cases} W_0(z), & z \in \Sigma_{\pi/2}, \\ W_{\pi/2}(ze^{-i\pi/2}), & \text{if } z \in e^{i\pi/2}\Sigma_{\pi/2}, \\ W_{-\pi/2}(ze^{i\pi/2}), & \text{if } z \in e^{-i\pi/2}\Sigma_{\pi/2}, \\ W_{\pi}(ze^{-i\pi}), & \text{if } z \in e^{i\pi}\Sigma_{\pi/2}, \\ D, & \text{if } z = 0, \end{cases}$$

is well-defined. By the foregoing, we obtain that there exists $\omega > 0$ such that the operator family $\{e^{-\omega z}W(z) : z \in \mathbb{C}\} \subseteq L(Y, X)$ is equicontinuous as well as that, for

every $u \in Y$, the mapping $z \mapsto W(z)u$, $z \in \mathbb{C}$ is entire (because it is weakly entire: this follows from the fact that for each $x^* \in X^*$ the mapping $z \mapsto \langle x^*, W(z) y \rangle$. $z \in \mathbb{C} \setminus \{0\}$ is analytic and has the limit $\langle x^*, Dy \rangle$ as $z \to 0$). Replacing the function $z \mapsto q_{\beta,\theta}(z) = z^{-1} \mathbf{P}_{ze^{-i\beta}}^{-1} C_1 \in L(Y,X), \ z \in \omega_{\beta,\theta} + \Sigma_{\theta+(\pi/2)} \text{ with the function}$ $z \mapsto q_{\beta,\theta,j}(z) := z^{-1} (ze^{-i\beta})^{\alpha_j - \alpha_n} A_j \mathbf{P}_{ze^{-i\beta}}^{-1} C_1 \in L(Y,X), \ z \in \omega_{\beta,\theta} + \Sigma_{\theta+(\pi/2)} \text{ in the}$ first part of proof $(\theta \in (0, \pi/2), \ j \in \mathbb{N}_n^0)$, for each $y \in Y$ we can define an X-valued analytic mapping $z \mapsto W_{\beta,j,y}(z), z \in \Sigma_{\pi/2}$ satisfying that, for every $\theta \in (0, \pi/2)$, we have that $\int_0^\infty e^{-zt} W_{\beta,j,y}(t) dt = z^{-1} (ze^{-i\beta})^{\alpha_j - \alpha_n} A_j \mathbf{P}_{ze^{-i\beta}}^{-1} C_1 y$, $\operatorname{Re} z > \omega_{\beta,\theta}$ and the set $\{e^{-\omega_{\beta,\theta}z}W_{\beta,i,y}(z): z \in \Sigma_{\theta'}\}$ is bounded in X $(y \in Y, \theta' \in (0,\theta))$. Define now $W_{\beta,j}(z)y := W_{\beta,j,y}(z), z \in \Sigma_{\pi/2}, y \in Y$, and $W^j(\cdot)$ by replacing $W_0(\cdot)$, $W_{\pi/2}(\cdot), W_{-\pi/2}(\cdot), W_{\pi}(\cdot)$ and D in the definition of $W(\cdot)$ with $W_{0,j}(\cdot), W_{\pi/2,j}(\cdot),$ $W_{-\pi/2,j}(\cdot), W_{\pi,j}(\cdot)$ and D_j , respectively $(j \in \mathbb{N}_n^0)$. Then there exists $\omega_j > 0$ such that the operator family $\{e^{-\omega_j z} W^j(z) : z \in \mathbb{C}\} \subseteq L(Y, X)$ is equicontinuous and, for every $y \in Y$, the mapping $z \mapsto W^j(z)y, z \in \mathbb{C}$ is entire $(j \in \mathbb{N}^0_r)$. By Theorem 2.3.33(i)/(b), we get that there exists an exponentially equicontinuous C_1 -existence family for (112), denoted by $(E(t))_{t\geq 0}$. Furthermore, for every $y \in Y$, the mapping $t \mapsto E(t)y, t \ge 0$ can be analytically extended to the whole complex plane so that $E^{(\alpha_n-1)}(z)y = W(z)y, \ z \in \mathbb{C}, \ y \in Y \text{ and } A_j(g_{\alpha_n-\alpha_j} * E^{(\alpha_n-1)})(z)y = W^j(z)y,$ $z \in \mathbb{C}, y \in Y, j \in \mathbb{N}_n^0$. Making use of the closedness of operators A_j for $j \in \mathbb{N}_n^0$, the above implies that the functions $z \mapsto A_i E^{(\alpha_n - 1)}(z) y, z \in \mathbb{C}$ are well-defined and entire $(y \in Y, j \in \mathbb{N}_n^0)$. By Theorem 2.3.34 and Theorem 2.3.33(ii), we get that the function $t \mapsto u(t), t \ge 0$, given by (121), with $g(t) = G(t), t \ge 0$, is a unique strong solution of problem (112). Define $v(t) := u(t) - \int_0^t E(t-s)g(s)ds, t \ge 0$. By the proof of Theorem 2.3.32(i), we have:

$$v^{(\alpha_n)}(\cdot) = \sum_{i=m}^{m_n-1} (g_{i-\alpha} * E^{(m_n-1)})(\cdot) v_{i,0} - \sum_{i=0}^{m_n-1} \sum_{j \in \mathbb{N}_{n-1} \smallsetminus D_i} (g_{i-\alpha_j} * E^{(m_n-1)})(\cdot) v_{i,j} \in C([0,\infty):X),$$

 $Bv^{(\alpha_n)}(\cdot) \in C([0,\infty):X)$ and

$$A_{i}v^{(\alpha_{i})}(\cdot) = \sum_{j=m_{i}}^{m_{n}-1} g_{j+1-\alpha_{i}}(\cdot)A_{i}u_{j}$$

$$-\sum_{l=0}^{m_{n}-1} \sum_{j\in\mathbb{N}_{n-1}\smallsetminus D_{l}} \left[g_{l-\alpha_{j}} * A_{i}(g_{\alpha_{n}-\alpha_{i}} * E^{(m_{n}-1)})\right](\cdot)v_{l,j}$$

$$+\sum_{l=m}^{m_{n}-1} \left[g_{l-\alpha} * A_{i}(g_{\alpha_{n}-\alpha_{i}} * E^{(m_{n}-1)})\right](\cdot)v_{l,0} \in C([0,\infty):X),$$

for all $i \in \mathbb{N}_{n-1}^{0}$. These representation formulae imply that the functions $v(\cdot)$ and $Bv^{(\alpha_n)}(\cdot), A_1v^{(\alpha_1)}(\cdot), \ldots, A_{n-1}v^{(\alpha_{n-1})}(\cdot), Av^{(\alpha)}(\cdot)$ can be analytically extended from the interval $[0,\infty)$ to the whole complex plane. Furthermore, $(u - v^{(\alpha_n)})$
$$\begin{split} v)^{(\alpha_n-1)}(t) &= \int_0^t E^{(\alpha_n-1)}(t-s)g(s)ds, \ t \ge 0 \ \text{and} \ (u-v)^{(\alpha_n)}(t) = \int_0^t E^{(\alpha_n-1)}(t-s)g'(s)ds + E^{(\alpha_n-1)}(t)g(0), \ t \ge 0. \\ \text{Now it is quite simple to prove that if the function } t\mapsto g(t), \ t \ge 0 \ \text{can be analytically extended to the whole complex plane, resp., } \\ \text{to a continuously differentiable function } \mathbb{R}\mapsto Y, \ \text{then } u(\cdot) \ \text{is an entire solution of } \\ \text{problem (112), resp., the function } t\mapsto u(t), \ t\ge 0 \ \text{can be extended to an } \alpha_n\text{-times } \\ \text{continuously differentiable function } \mathbb{R}\mapsto X \ \text{and the functions } Bu^{(\alpha_n)}(\cdot), \ A_1u^{(\alpha_1)}(\cdot), \\ \dots, \ A_{n-1}u^{(\alpha_{n-1})}(\cdot), \ Au^{(\alpha)}(\cdot) \ \text{can be extended to continuously differentiable function } \\ \mathbb{R}\mapsto X. \ \text{The rest of the proof can be left to the reader.} \end{split}$$

- REMARK 2.3.47. (i) Suppose that Y = X, $C_1 \in L(X)$ is injective, $C_1A_j \subseteq A_jC_1$, $j \in \mathbb{N}_n^0$, as well as the operator families $\{\mathbf{P}_z^{-1}C_1 : |z| \ge R\} \subseteq L(X)$ and $\{z^{\alpha_j - \alpha_n}A_j\mathbf{P}_z^{-1}C_1 : |z| \ge R, j \in \mathbb{N}_n^0\} \subseteq L(X)$ are equicontinuous and strongly continuous. Then the analyticity of mappings $z \mapsto \mathbf{P}_z^{-1}C_1x \in X$, |z| > R and $z \mapsto z^{\alpha_j - \alpha_n}A_j\mathbf{P}_z^{-1}C_1x \in X$, |z| > R automatically follows for any $x \in X$, $j \in \mathbb{N}_n^0$ (cf. the proof of Lemma 2.6.3 below).
- (ii) Suppose that $g \in C^{\infty}([0,\infty) : Y)$, resp., $g(\cdot)$ can be extended to an infinitely differentiable function $\mathbb{R} \to Y$. Then $u \in C^{\infty}([0,\infty) : X)$ and $Bu^{(\alpha_n)}, A_1u^{(\alpha_1)}, \ldots, A_{n-1}u^{(\alpha_{n-1})}, Au^{(\alpha)} \in C^{\infty}([0,\infty) : X)$, resp., the functions $u(\cdot)$ and $Bu^{(\alpha_n)}(\cdot), A_1u^{(\alpha_1)}(\cdot), \ldots, A_{n-1}u^{(\alpha_{n-1})}(\cdot), Au^{(\alpha)}(\cdot)$ can be extended to infinitely differentiable functions $\mathbb{R} \to X$.
- (iii) Let $0 \leq i \leq m_n 1$, $0 \leq j \leq n 1$ and $i \geq \alpha_j$. If a strong solution $u(\cdot)$ of problem (112) has the property that $u \in C^{\infty}([0, \infty) : X)$ and $Bu^{(\alpha_n)}$, $A_1u^{(\alpha_1)}, \ldots, A_{n-1}u^{(\alpha_{n-1})}, Au^{(\alpha)} \in C^{\infty}([0, \infty) : X)$, then it can be easily seen that the mapping $t \mapsto A_j u^{(\alpha'_j)}(t)$, $t \geq 0$ is well-defined and infinitely differentiable for $\alpha'_j \geq \alpha_j$; hence, $u_i \in D(A_j)$ for $0 \leq i \leq m_n 1$, $j \in D''_i$ and our result on the well-posedness of (112) is optimal provided that $R(C_1) = X$.

Now we would like to present how Theorem 2.3.46 can be applied in the analysis of abstract Boussinesq–Love equation in finite domains.

EXAMPLE 2.3.48. Suppose that $\emptyset \neq \Omega \subseteq \mathbb{R}^n$ is a bounded domain with smooth boundary $\partial \Omega$. In the cylinder $\mathbb{R} \times \Omega$, we consider the following Cauchy–Dirichlet problem for linearized Boussinesq–Love equation:

$$(140) \ (\lambda - \Delta)u_{tt}(t, x) - \alpha(\Delta - \lambda')u_t(t, x) = \beta(\Delta - \lambda'')u(t, x) + f(t, x), \ t \in \mathbb{R}, \ x \in \Omega,$$

$$(141) \ u(0,x) = u_0(x), \ u_t(0,x) = u_1(x), \ (t,x) \in \mathbb{R} \times \Omega; \ u(t,x) = 0, \ (t,x) \in \mathbb{R} \times \partial \Omega,$$

where $\lambda, \lambda', \lambda'' \in \mathbb{R}$, $\alpha, \beta \in \mathbb{R}$ and $\beta \neq 0$ (in [515], the standing hypothesis was that $\alpha \neq 0$; as explained later in [556], the case $\alpha = 0$ is meaningful and has a certain physical meaning). By $\{\lambda_k\} [= \sigma(\Delta)]$ we denote the eigenvalues of the Dirichlet Laplacian Δ in $L^2(\Omega)$ (recall that $0 < -\lambda_1 \leq -\lambda_2 \dots \leq -\lambda_k \leq \dots \rightarrow +\infty$ as $k \to \infty$; cf. [521, Section 5.6], [27, Section 6] and [509, Section 1.3] for more details) numbered in nonascending order with regard to multiplicities. By $\{\phi_k\} \subseteq C^{\infty}(\Omega)$ we denote the corresponding set of mutually orthogonal [in the sense of $L^2(\Omega)$] eigenfunctions. In [515], G. A. Sviridyuk and A. A. Zamyshlyaeva have considered

the well-posedness of problem (140)–(141) in the Sobolev space $W^{p,l}(\Omega)$, where $1 and <math>l \in \mathbb{N}_0$, and the Hölder space $C^{l+\gamma}(\Omega)$, where $0 < \gamma < 1$ and $l \in \mathbb{N}_0$. In order to apply [**515**, Theorem 4.1], the authors imposed the following condition:

- (i) $\lambda \in \rho(\Delta)$, or
- (iii) $\lambda \in \sigma(\Delta) \land \lambda = \lambda' \land \lambda \neq \lambda''.$

Although our results on the well-posedness of problem (140)-(141) in cases (i) or (iii) give some new information about qualitative properties of strong solutions of (140)-(141), in the remainder of this example we will completely focus our attention on the following case:

(ii)
$$\lambda \in \sigma(\Delta) \land \lambda \neq \lambda' \land (\alpha = 0 \Rightarrow \lambda \neq \lambda'').$$

If (ii) holds with $\alpha \neq 0$, then we cannot apply [515, Theorem 4.1] (despite of the validity of requirement stated in the formulation of [515, Lemma 5.1]) because of the violation of condition [515, (A), p. 271]. Here it is also worth noting that the existence and uniqueness of two times continuously differentiable solutions of problem (140)–(141) on the non-negative real axis (understood in a broader sense of [509, Definition 5.6.2]) have been studied in [509, Example 5.7.1, Lemma 5.7.1(ii), Theorem 5.7.3] provided that $Y = H^2(\Omega) \cap H_0^1(\Omega)$, $X = L^2(\Omega)$, as well as that (ii) holds and that, additionally, $\lambda'' \neq 0$; even in this case, we obtain from Theorem 2.3.46 and a simple analysis, with $X = Y = L^2(\Omega)$ and C_1 being the identity operator on X, that a strong solution $t \mapsto u(t), t \ge 0$ of this problem has the property that the mapping $t \mapsto \Delta u(t), t \ge 0$ belongs to the space $C^2([0, \infty) : L^2(\Omega))$ provided that $f \in C^1([0, \infty) : L^2(\Omega))$ needless to say that we obtain the existence and uniqueness of entire solutions of problem (140)–141 provided that the function $f(\cdot)$ can be extended analytically to the whole complex plane.

With a little abuse of notation, we have that n = 2, $B = \lambda - \Delta$, $A_1 = -\alpha(\Delta - \lambda')$, $A = \beta(\Delta - \lambda'')$, $\alpha_2 = 2$, $\alpha_1 = 1$ and $\alpha = 0$ (the use of symbols α and β will be clear from the context). Hence,

$$\mathbf{P}_{z} = z^{-2}[(z^{2}\lambda + \alpha z\lambda' + \beta\lambda'') + (-z^{2} - \alpha z - \beta)\Delta], \quad z \in \mathbb{C} \smallsetminus \{0\}.$$

It is clear that (ii) implies that

$$\lambda \neq \frac{z^2 \lambda + \alpha z \lambda' + \beta \lambda''}{z^2 + \alpha z + \beta} \to \lambda \text{ as } |z| \to \infty.$$

We assume that $X = Y = L^p(\Omega)$ for some $p \in (1, \infty)$, C_1 is the identity operator on X, Δ is the Dirichlet Laplacian on $L^p(\Omega)$ acting with domain $D(\Delta) := W^{p,2}(\Omega) \cap W_0^{p,1}(\Omega)$, as well as that the following condition holds:

HP. There exist a sufficiently large real number R > 0 and a positive real number number number l < 4, resp., l < 2, provided that (ii) holds with $\alpha \neq 0$, resp., $\alpha = 0$, such that

(142)
$$||R(z:\Delta)|| = O(|\lambda - z|^{-l}) \text{ as } z \to \lambda.$$

Before proceeding further, it should be observed that the condition HP. holds in the case that p = 2, with l = 1: Suppose that $\lambda = \lambda_{k_0}$ for some $k_0 \in \mathbb{N}$. Then $g = R(z : \Delta)f = \sum_{k=1}^{\infty} \frac{\langle \phi_k, f \rangle}{z - \lambda_k} \phi_k$ as $z \to \lambda_{k_0}$, so that Parseval's equality implies $\begin{aligned} |z - \lambda_{k_0}|^2 \|g\|^2 &= \sum_{k=1}^{\infty} \frac{|z - \lambda_{k_0}|^2 |\langle \phi_k, f \rangle|^2}{|z - \lambda_k|^2} \leqslant \text{Const.} \sum_{k=1}^{\infty} |\langle \phi_k, f \rangle|^2 = \|f\|^2 \text{ as } z \to \lambda_{k_0} \\ \text{(let us recall that } \lambda_k \to -\infty \text{ as } k \to \infty\text{). Using now the condition HP., the expression} \end{aligned}$

$$\mathbf{P}_{z}^{-1} = z^{-2}(z^{2} + \alpha z + \beta)^{-1} \left[\frac{z^{2}\lambda + \alpha z\lambda' + \beta\lambda''}{z^{2} + \alpha z + \beta} - \Delta \right]^{-1}, \quad |z| \ge R,$$

and the resolvent equation, it readily follows that there exists a positive real number $\zeta > 0$ such that the operator families $\{(1 + |z|)^{\zeta} \mathbf{P}_{z}^{-1} : |z| \ge R\} \subseteq L(X)$ and $\{(1 + |z|)^{\zeta} z^{\alpha_{j}-\alpha_{n}} A_{j} \mathbf{P}_{z}^{-1} : |z| \ge R, \ j \in \mathbb{N}_{2}^{0}\} \subseteq L(X)$ are equicontinuous, as well as that $\lim_{z\to\infty} \mathbf{P}_{z}^{-1}x = 0$ and $\lim_{z\to\infty} z^{\alpha_{j}-\alpha_{n}}A_{j}\mathbf{P}_{z}^{-1}x = 0$ for any $x \in X, \ j \in \mathbb{N}_{n}^{0}$. The strong analyticity of mappings $z \mapsto \mathbf{P}_{z}^{-1}, |z| > R$ and $z \mapsto z^{\alpha_{j}-\alpha_{n}}A_{j}\mathbf{P}_{z}^{-1}, |z| > R$ follows from Remark 2.3.47(i), while the existence of an exponentially bounded I-uniqueness family for the corresponding problem (112) simply follows from Theorem 2.3.33(ii) and the above argumentation; here I stands for the identity operator on X. Hence, there exists a unique entire solution $z \mapsto u(z), \ z \in \mathbb{C}$ of problem (140)–(141), provided that $u_{0}(x) \in W^{p,2}(\Omega) \cap W_{0}^{p,1}(\Omega), \ u_{1}(x) \in W^{p,2}(\Omega) \cap W_{0}^{p,1}(\Omega)$ and the function $f(\cdot)$ can be analytically extended to the whole complex plane; moreover, we have the existence of a positive real number $\omega' > 0$ such that the set $\{e^{-\omega' z}g(z): z \in \mathbb{C}\}$ is bounded in $L^{p}(\Omega)$. Since C_{1} is the identity operator on X, this is an optimal result as long as the condition HP. holds (cf. Remark 2.3.47(ii)).

We continue our analysis by enquiring into the existence and uniqueness of entire solutions to the abstract Barenblatt–Zheltov–Kochina equation in finite domains. We use the argumentation contained in the proof of Theorem 2.3.46 and the approach of N. H. Abdelaziz, F. Neubrander (cf. [6] and Subsection 2.3.3); for the sake of simplicity, we will deal only with the homogenous case.

EXAMPLE 2.3.49. Let $\emptyset \neq \Omega \subseteq \mathbb{R}^n$, $\{\lambda_k\}$, $\{\phi_k\}$ and Δ possess the same meanings as in the previous example, let $X = Y = L^2(\Omega)$, and let C_1 be the identity operator on X. As mentioned above, we analyze entire solutions of the Barenblatt–Zheltov–Kochina equation

(143)
$$(\lambda - \Delta)u_t(t, x) = \zeta \Delta u(t, x), \quad t \in \mathbb{R}, \ x \in \Omega;$$

(144)
$$u(0,x) = u_0(x), \quad x \in \Omega, \ u(t,x) = 0, \ (t,x) \in \mathbb{R} \times \partial\Omega,$$

where $\zeta \in \mathbb{R} \setminus \{0\}$ and $\lambda = \lambda_{k_0} \in \sigma(\Delta)$ (cf. the equation (112) with n = 2, $B = \lambda - \Delta$, $A_1 = 0$, $A = \zeta \Delta$, $\alpha_2 = 1$ and $\alpha_1 = \alpha = 0$; then we have $\mathbf{P}_z = \lambda - (1 + \zeta z^{-1})\Delta)$. Using Parseval's equality, it can be easily seen that the operator $D: f \mapsto (-1)(\zeta \lambda)^{-1} \sum_{\lambda = \lambda_k} \langle \phi_k, f \rangle \phi_k$ ($f \in L^2(\Omega)$) belongs to the space $L(L^2(\Omega))$. Let $\beta \in (-\pi, \pi]$. Then the equation (142) holds with l = 1, which enables us to verify that, for every $\theta \in (0, \pi/2)$, there exists a sufficiently large number $\omega_{\beta,\theta} > 0$ satisfying that the function $q_{\beta,\theta}(z) := z^{-2}\mathbf{P}_{ze^{-i\beta}}^{-1} \in L(X), z \in \omega_{\beta,\theta} + \Sigma_{\theta+(\pi/2)}$ is well-defined, strongly analytic and that for each $\theta' \in (0, \theta)$ the operator family $\{z^{-2}(z - \omega_{\beta,\theta})\mathbf{P}_{ze^{-i\beta}}^{-1} : z \in \omega_{\beta,\theta} + \Sigma_{\theta'+(\pi/2)}\} \subseteq L(X)$ is equicontinuous. As in the proof of Theorem 2.3.46, we obtain that for each $f \in X$ there exists an X-valued analytic mapping $z \mapsto W_{\beta,f}^1(z), z \in \Sigma_{\pi/2}$ satisfying that, for every $\theta \in (0, \pi/2)$, one has $\int_0^\infty e^{-zt} W_{\beta,f}^1(t) dt = z^{-2} \mathbf{P}_{ze^{-i\beta}}^{-1} f$, $\operatorname{Re} z > \omega_{\beta,\theta}$ and the set $\{e^{-\omega_{\beta,\theta} z} W_{\beta,f}^1(z) : z \in \Sigma_{\theta'}\}$ is bounded in X $(f \in X, \theta' \in (0,\theta))$. Define $W_{\beta}^1(z)f := W_{\beta,f}^1(z)$, $z \in \Sigma_{\pi/2}, f \in X$. Then, for every $\theta \in (0, \pi/2), \{e^{-\omega_{\beta,\theta} z} W_{\beta}^1(z) : z \in \Sigma_{\theta}\} \subseteq L(X)$ is an equicontinuous operator family. On the other hand, there exist finite constants R > 0 and c > 0 such that the set $\{|z|^{-1}|(\lambda - \lambda_k)z - \zeta\lambda_k| : |z| \ge R, k \in \mathbb{N} \setminus \{k_0\}\}$ is bounded from below by c, so that we can apply Parseval's equality once more in order to see that:

$$z^{-1}\mathbf{P}_{z}^{-1}f = \frac{1}{\zeta+z} \Big[\frac{\lambda z}{\zeta+z} - \Delta\Big]^{-1}f = \sum_{k=1, k\neq k_{0}}^{\infty} \frac{\langle \phi_{k}, f \rangle}{(\lambda-\lambda_{k})z - \zeta\lambda_{k}} \phi_{k} + Df \to Df,$$

as $|z| \to \infty$ $(f \in X)$; similarly, we have that the operator family $\{z^{-2}B\mathbf{P}_{ze^{-i\beta}}^{-1} : |z| \ge R\} \in L(X)$ is equicontinuous and $z^{-1}B\mathbf{P}_z^{-1}f \to 0, |z| \to \infty$ $(f \in X)$, so that we can define a strongly analytic operator family $(W_{\beta,B}^1(z))_{z\in\Sigma_{\pi/2}} \subseteq L(X)$ satisfying that, for every $\theta \in (0, \pi/2)$, the operator family $\{e^{-\omega'_{\beta,\theta}z}W_{\beta,B}^1(z) : z \in \Sigma_{\theta}\} \subseteq L(X)$ is equicontinuous for some number $\omega'_{\beta,\theta} > 0$. Since $\lim_{|z|\to\infty} z^{-1}\mathbf{P}_z^{-1}f = Df$ $(f \in X)$, an application of Theorem 1.4.10(ii)/(iii) yields that, for every $f \in X$ and $\theta \in (0, \pi/2)$, we have $\lim_{z\to 0, z\in\Sigma_{\theta}} W_{\beta}^1(z)f = Df$. Define

$$W^{1}(z) := \begin{cases} W_{0}^{1}(z), & z \in \Sigma_{\pi/2}, \\ e^{i\pi/2}W_{\pi/2}^{1}(ze^{-i\pi/2}), & \text{if } z \in e^{i\pi/2}\Sigma_{\pi/2}, \\ e^{-i\pi/2}W_{-\pi/2}^{1}(ze^{i\pi/2}), & \text{if } z \in e^{-i\pi/2}\Sigma_{\pi/2}, \\ e^{i\pi}W_{\pi}^{1}(ze^{-i\pi}), & \text{if } z \in e^{i\pi}\Sigma_{\pi/2}, \\ D, & \text{if } z = 0, \end{cases}$$

and $W_B^1(z)$ by replacing the operators $W_0^1(z)$, $W_{\pi/2}^1(ze^{-i\pi/2})$, $W_{-\pi/2}^1(ze^{i\pi/2})$, $W^1_{\pi}(ze^{-i\pi})$ and D in the above definition by the operators $W^1_{0,B}(z), W^1_{\pi/2,B}(ze^{-i\pi/2}),$ $W^1_{-\pi/2,B}(ze^{i\pi/2}), W^1_{\pi,B}(ze^{-i\pi})$ and 0, respectively $(z \in \mathbb{C})$. Then there exists a finite constant $\omega > 0$ such that the operator families $\{e^{-\omega z}W(z) : z \in \mathbb{C}\} \subset L(X)$ and $\{e^{-\omega z}BW(z): z \in \mathbb{C}\} \subset L(X)$ are equicontinuous as well as that, for every $f \in X$, the mappings $z \mapsto W(z)f$, $z \in \mathbb{C}$ and $z \mapsto BW(z)f$, $z \in \mathbb{C}$ are entire; cf. also [291, Proposition 2.4.2, Corollary 2.4.3]. Furthermore, it is not difficult to see that $(W^1(t))_{t\geq 0} \subseteq L(X, [D(B)])$ is a once integrated evolution family generated by A, B in the sense of considerations from [6, Section 2]. By [6, Theorem 2.3] and an elementary analysis, we may conclude that for each function $u_0 \in H^2(\Omega) \cap H^1_0(\Omega)$, orthogonal to the eigenfunction(s) ϕ_k for $k = k_0$, there exists a unique strong solution $t \mapsto u(t), t \ge 0$ of problem (143)–(144) satisfying that there exists a finite constant $\omega' > 0$ such that the mappings $t \mapsto u(t), t \ge 0$ and $t \mapsto Bu(t), t \ge 0$ can be analytically extended to the whole complex plane, as well as that the sets $\{e^{-\omega'z}u(z): z \in \mathbb{C}\}$ and $\{e^{-\omega'z}Bu(z): z \in \mathbb{C}\}$ are bounded. This result slightly improves the assertion of [509, Theorem 5.1.3(ii)] in L^2 spaces.

Now we will investigate the existence and uniqueness of analytical solutions of abstract degenerate Cauchy problem (112), provided that there exists an index

 $i \in \mathbb{N}_n^0$ such that $\alpha_i \notin \mathbb{N}_0$; for the sake of brevity, we will consider only homogeneous equations.

DEFINITION 2.3.50. A function $u \in C([0,\infty) : X)$ is said to be an analytical solution of problem (112) on the region $\mathbb{C} \smallsetminus (-\infty, 0]$ iff $u(\cdot)$ is a strong solution of (112) and it can be extended to the whole complex plane, analytically on the region $\mathbb{C} \smallsetminus (-\infty, 0]$ and continuously on the region $\mathbb{C} \smallsetminus (-\infty, 0)$, as well as any of the terms $A_i \mathbf{D}_t^{\alpha_i} u(t)$ $(0 \le i \le n, t \ge 0)$ is well defined and can be extended to the whole complex plane, analytically on the region $\mathbb{C} \smallsetminus (-\infty, 0]$ and continuously on the region $\mathbb{C} \smallsetminus (-\infty, 0]$ and continuously on the region $\mathbb{C} \backsim (-\infty, 0]$ and continuously on the region $\mathbb{C} \backsim (-\infty, 0]$ and continuously on the region $\mathbb{C} \backsim (-\infty, 0]$.

Set, for every $\gamma \in (-\pi, \pi]$ and $z \in \mathbb{C} \setminus \{0\}$,

$$\mathbf{P}_{z,\gamma} := B + \sum_{j=1}^{n-1} z^{\alpha_j - \alpha_n} e^{i\gamma(\alpha_n - \alpha_j)} A_j - z^{\alpha - \alpha_n} e^{i\gamma(\alpha_n - \alpha)} A_j$$

Before stating the following theorem, it should be observed that $\mathbf{P}_{z,\gamma}$ need not be equal to $\mathbf{P}_{ze^{-i\gamma}}$ for some particular values of γ and z.

THEOREM 2.3.51. Suppose k(t) satisfies (P1), $C_1 \in L(Y, X)$, $C_2 \in L(X)$ is injective, and there exists a locally equicontinuous k-regularized C_2 -uniqueness family for problem (112), with $f(t) \equiv 0$. Let the following hold:

- Suppose that for each number $\gamma \in (-\pi, \pi]$ there exists a sufficiently large number R > 0 such that the operator $\mathbf{P}_{z,\gamma}$ is injective for $|z| \ge R$, as well as that the operator families $\{\mathbf{P}_{z,\gamma}^{-1}C_1 : |z| \ge R, \ z \notin (-\infty, 0]\} \subseteq L(Y, X)$ and $\{z^{\alpha_j \alpha_n}A_j\mathbf{P}_{z,\gamma}^{-1}C_1 : |z| \ge R, \ j \in \mathbb{N}_n^0, \ z \notin (-\infty, 0]\} \subseteq L(Y, X)$ are equicontinuous.
- Let the mappings $z \mapsto \mathbf{P}_{z,\gamma}^{-1}C_1 y \in X$, |z| > R, $z \notin (-\infty, 0]$ and $z \mapsto z^{\alpha_j \alpha_n} A_j \mathbf{P}_{z,\gamma}^{-1}C_1 y \in X$, |z| > R, $z \notin (-\infty, 0]$ be analytic for any $y \in Y$, $j \in \mathbb{N}_n^0$, $\gamma \in (-\pi, \pi]$.
- Let there exist operators $D, D_0, D_1, \ldots, D_n \in L(Y, X)$ such that $\lim_{z \to +\infty} \mathbf{P}_{z,\gamma}^{-1}C_1 y = Dy$ and $\lim_{z \to +\infty} z^{\alpha_j \alpha_n} e^{i\gamma(\alpha_n \alpha_j)} A_j \mathbf{P}_{z,\gamma}^{-1}C_1 y = D_j y$ for any $y \in Y, j \in \mathbb{N}_n^0, \gamma \in (-\pi, \pi]$.
- Suppose that $u_i \in \mathbf{D}_i$ for $0 \leq i \leq m_n 1$.

Then there exists a unique analytical solution of problem (112) on the region $\mathbb{C} \setminus (-\infty, 0]$, with $f(t) \equiv 0$. Denote by $A_i \mathbf{D}_z^{\alpha_i} u(z)$ the analytical extensions of terms $A_i \mathbf{D}_t^{\alpha_i} u(t)$ $(0 \leq i \leq n, t \geq 0)$ on region $\mathbb{C} \setminus (-\infty, 0]$. Then, for every $\zeta \in (0, \pi)$, we have the existence of a positive real number $\omega' > 0$ such that the sets $\{e^{-\omega' z} u(z) : z \in \overline{\Sigma_{\zeta}}\}$ and $\{e^{-\omega' z} A_i \mathbf{D}_z^{\alpha_i} u(z) : z \in \overline{\Sigma_{\zeta}}\}$ $(0 \leq i \leq n)$ are bounded in X.

PROOF. The proof of theorem is very similar to that of Theorem 2.3.46 and, because of that, we will only outline the most relevant details. Fix numbers $\gamma \in$ $(-\pi,\pi]$ and $\zeta \in (0,\pi)$. Then, for every $\theta \in (0,\pi/2)$, there exists a sufficiently large number $\omega_{\gamma,\theta} > 0$ satisfying that the function $r_{\gamma,\theta}(z) := z^{-1}\mathbf{P}_{z,\gamma}^{-1}C_1 \in L(Y,X)$, $z \in \omega_{\gamma,\theta} + \Sigma_{\theta+(\pi/2)}$ is well-defined, strongly analytic and that for each $\theta' \in (0,\theta)$ the operator family $\{z^{-1}(z - \omega_{\gamma,\theta})\mathbf{P}_{z,\gamma}^{-1}C_1 : z \in \omega_{\gamma,\theta} + \Sigma_{\theta'+(\pi/2)}\} \subseteq L(Y,X)$ is equicontinuous. Arguing as in the proof of Theorem 2.3.46, we obtain the existence of a strongly analytic operator family $(W_{\gamma}(z))_{z \in \Sigma_{\pi/2}} \subseteq L(Y,X)$ satisfying that $\lim_{z \to 0, z \in \overline{\Sigma_{\theta}}} W_{\gamma}(z)y = Dy \ (\theta \in (0, \pi/2), \ y \in Y), \ \int_{0}^{\infty} e^{-zt} W_{\gamma}(t)y \ dt = z^{-1} \mathbf{P}_{z,\gamma}^{-1} C_{1} y,$ $\operatorname{Re} z > \omega_{\gamma,\theta}, \ y \in Y$ and that, for every $\theta \in (0, \pi/2), \ \{e^{-\omega_{\gamma,\theta} z} W_{\gamma}(z) : z \in \Sigma_{\theta'}\} \subseteq L(Y,X)$ is an equicontinuous operator family $(\theta' \in (0,\theta))$. Define

$$W(z) := \begin{cases} W_0(z), & z \in \Sigma_{\pi/2}, \\ W_{\pi/2}(ze^{-i\pi/2}), & \text{if } z \in e^{i\pi/2}\Sigma_{\pi/2}, \\ W_{-\pi/2}(ze^{i\pi/2}), & \text{if } z \in e^{-i\pi/2}\Sigma_{\pi/2}, \\ 0, & \text{if } z < 0, \\ D, & \text{if } z = 0. \end{cases}$$

Then it can be simply verified that $W(\cdot)$ is well-defined and there exists $\omega > 0$ such that the operator family $\{e^{-\omega z}W(z): z \in \overline{\Sigma_{\zeta}}\} \subseteq L(Y, X)$ is equicontinuous as well as that, for every $y \in Y$, the mapping $z \mapsto W(z)y, z \in \mathbb{C} \setminus (-\infty, 0]$ is analytic. Replacing the function $z \mapsto q_{\gamma,\theta}(z), z \in \omega_{\gamma,\theta} + \Sigma_{\theta+(\pi/2)}$ with the function $z \mapsto q_{\gamma,\theta,j}(z) := z^{-1} z^{\alpha_j - \alpha_n} e^{i\gamma(\alpha_n - \alpha_j)} A_j \mathbf{P}_{z,\gamma}^{-1} C_1 \in L(Y,X), \ z \in \omega_{\gamma,\theta} + \Sigma_{\theta + (\pi/2)}$ in the first part of proof $(\theta \in (0, \pi/2), j \in \mathbb{N}_n^0)$, we can define a strongly analytic operator family $(W_{\gamma,j}(z))_{z\in\Sigma_{\pi/2}}\subseteq L(Y,X)$ satisfying that, for every $\theta\in(0,\pi/2)$ and $y \in Y$, we have that $\int_0^{\infty} e^{-zt} W_{\gamma,j}(t) y \, dt = z^{-1} z^{\alpha_j - \alpha_n} e^{i\gamma(\alpha_n - \alpha_j)} A_j \mathbf{P}_{z,\gamma}^{-1} C_1 y$, $\operatorname{Re} z > \omega_{\beta,\theta}$ and the operator family $\{e^{-\omega_{\beta,\theta} z} W_{\gamma,i}(z) : z \in \Sigma_{\theta'}\} \subseteq L(Y,X)$ is equicontinuous $(\theta' \in (0, \theta), j \in \mathbb{N}_n^0)$. Define now $W^j(\cdot)$ by replacing $W_0(\cdot), W_{\pi/2}(\cdot)$, $W_{-\pi/2}(\cdot)$ and D in the definition of $W(\cdot)$ with $W_{0,j}(\cdot), W_{\pi/2,j}(\cdot), W_{-\pi/2,j}(\cdot)$ and D_j , respectively $(j \in \mathbb{N}_n^0)$. Then there exists $\omega_j > 0$ such that the operator family $\{e^{-\omega_j z} W^j(z) : z \in \overline{\Sigma_\ell}\} \subseteq L(Y, X)$ is equicontinuous as well as that, for every $y \in Y$, the mapping $z \mapsto W^j(z)y, z \in \mathbb{C} \setminus (-\infty, 0]$ is analytic and the mapping $z \mapsto W^j(z)y, z \in \mathbb{C} \setminus (-\infty, 0)$ is continuous $(j \in \mathbb{N}_n^0)$. By Theorem 2.3.33(i)/(b), we get that there exists an exponentially equicontinuous C_1 -existence family for (112), denoted by $(E(t))_{t\geq 0}$. Furthermore, for every $y \in Y$, the mapping $t \mapsto E(t)y, t \geq 0$ can be analytically extended to the region $z \in \mathbb{C} \setminus (-\infty, 0]$ so that the mapping $z \mapsto E(z)y, z \in \mathbb{C} \setminus (-\infty, 0)$ is continuous as well as that $E^{(m_n-1)}(z)y = W(z)y$, $z \in \mathbb{C} \setminus (-\infty, 0], y \in Y \text{ and } A_j(g_{\alpha_n - \alpha_j} * E^{(m_n - 1)})(z)y = W^j(z)y, z \in \mathbb{C} \setminus (-\infty, 0],$ $y \in Y, j \in \mathbb{N}_n^0$ (this equality can be proved by using the closedness of operators A_j for $j \in \mathbb{N}_n^0$, Proposition 1.4.7 and the arguments already used in the proofs of Kato's analyticity criteria from [291, Section 2.4]). By Theorem 2.3.34 and Theorem 2.3.33(ii), we get that the function $t \mapsto u(t), t \ge 0$, given by (121), with $g(t) = G(t) = 0, t \ge 0$, is a unique strong solution of problem (112). The rest of the proof can be obtained by using a slight modification of the corresponding parts of the proof of Theorem 2.3.46. \square

REMARK 2.3.52. Suppose that Y = X, $C_1 \in L(X)$ is injective, $C_1A_j \subseteq A_jC_1$, $j \in \mathbb{N}_n^0$, as well as that the operator families $\{\mathbf{P}_{z,\gamma}^{-1}C_1 : |z| \ge R, \ z \notin (-\infty, 0]\} \subseteq L(X)$ and $\{z^{\alpha_j - \alpha_n}A_j\mathbf{P}_{z,\gamma}^{-1}C_1 : |z| \ge R, \ z \notin (-\infty, 0], \ j \in \mathbb{N}_n^0\} \subseteq L(X)$ are equicontinuous and strongly continuous for any $\gamma \in (-\pi, \pi]$. Then the mappings

 $z \mapsto \mathbf{P}_{z,\gamma}^{-1}C_1 x \in X, |z| > R, z \notin (-\infty, 0] \text{ and } z \mapsto z^{\alpha_j - \alpha_n} A_j \mathbf{P}_{z,\gamma}^{-1}C_1 x \in X, |z| > R, z \notin (-\infty, 0] \text{ are analytic for any } x \in X, j \in \mathbb{N}_n^0, \gamma \in (-\pi, \pi].$

Now we will continue our analysis of abstract Boussinesq–Love equation and abstract Barenblatt–Zheltov–Kochina equation in finite domains (fractional-order case).

EXAMPLE 2.3.53. (i) Suppose, as in Example 2.3.48, that $\emptyset \neq \Omega \subseteq \mathbb{R}^n$ is a bounded domain with smooth boundary $\partial \Omega$. Let $0 < \alpha_1 < \alpha_2 \leq 2$. We analyze the following fractional degenerate Cauchy-Dirichlet problem:

(145)
$$(\lambda - \Delta)\mathbf{D}_t^{\alpha_2}(t, x) - \alpha(\Delta - \lambda')\mathbf{D}_t^{\alpha_1}(t, x) = \beta(\Delta - \lambda'')u(t, x), \quad t \ge 0, \ x \in \Omega,$$

(146)
$$u(0,x) = u_0(x); \quad u_t(0,x) = u_1(x), \ (t,x) \in [0,\infty) \times \Omega, \text{ if } \alpha_2 > 1;$$

(147)
$$u(t,x) = 0, \quad (t,x) \in [0,\infty) \times \partial\Omega,$$

where Δ denotes the Dirichlet Laplacian on $L^2(\Omega)$, $\lambda, \lambda', \lambda'' \in \mathbb{R}$, $\alpha, \beta \in \mathbb{R}$ and $\beta \neq 0$. Assume that the following condition holds (cf. Example 2.3.48(ii)):

$$\lambda \in \sigma(\Delta) \land \lambda \neq \lambda' \land (\alpha = 0 \Rightarrow \lambda \neq \lambda'')$$

Arguing similarly as in the afore-mentioned example, we can prove that the requirements of Theorem 2.3.51 holds with $D = D_0 = D_1 = D_2 = 0$, so that the abstract Cauchy problem (145)–(147) has a unique analytical solution on the region $\mathbb{C} \setminus (-\infty, 0]$ for any $u_0(x) \in H^2(\Omega) \cap H_0^1(\Omega)$, and $u_1(x) \in H^2(\Omega) \cap H_0^1(\Omega)$, if $\alpha_2 > 1$.

(ii) Let $\emptyset \neq \Omega \subseteq \mathbb{R}^n$ be a bounded domain with smooth boundary $\partial\Omega$, let $\alpha \in (0, 2) \setminus \{1\}$, and let $X = Y = L^2(\Omega)$. Suppose that Δ denotes the Dirichlet Laplacian on X and the operator D has the same meaning as in Example 2.3.49 (we will use the same notation henceforth). Consider the following fractional Barenblatt–Zheltov–Kochina equation:

(148)
$$(\lambda - \Delta)\mathbf{D}_t^{\alpha}u(t, x) = \zeta \Delta u(t, x), \quad t \in \mathbb{R}, \ x \in \Omega;$$

(149)
$$u(0,x) = u_0(x), x \in \Omega; u_t(0,x) = u_1(x), (t,x) \in [0,\infty) \times \Omega, \text{ if } \alpha > 1;$$

(150)
$$u(t,x) = 0, \quad (t,x) \in \mathbb{R} \times \partial \Omega,$$

where $\zeta \in \mathbb{R} \setminus \{0\}$ and $\lambda = \lambda_{k_0} \in \sigma(\Delta)$ (let us recall that the problem (148)–(150) is a special case of problem (112) with $n = 2, B = \lambda - \Delta$, $A_1 = 0, A = \zeta \Delta, \alpha_2 = \alpha$ and $\alpha_1 = \alpha = 0$; now we have $\mathbf{P}_z = \lambda - (1 + \zeta z^{-\alpha})\Delta)$. Let $\gamma \in (-\pi, \pi]$. Then it is not difficult to verify that, for every $\theta \in (0, \pi/2)$, there exists a sufficiently large number $\omega_{\gamma,\theta} > 0$ satisfying that the function $q_{\gamma,\theta}(z) := z^{-\alpha-1}\mathbf{P}_{z,\gamma}^{-1} \in L(X), z \in \omega_{\gamma,\theta} + \Sigma_{\theta+(\pi/2)}$ is well-defined, strongly analytic and that for each $\theta' \in (0, \theta)$ the operator family $\{z^{-\alpha-1}(z-\omega_{\gamma,\theta})\mathbf{P}_{z,\gamma}^{-1}: z \in \omega_{\gamma,\theta} + \Sigma_{\theta'+(\pi/2)}\} \subseteq L(X)$ is equicontinuous. Furthermore, for every $f \in X$, there exists an X-valued analytic mapping $z \mapsto W_{\gamma,f}^{\alpha}(z), z \in \Sigma_{\pi/2}$ satisfying that, for every $\theta \in (0, \pi/2)$, one has $\int_0^{\infty} e^{-zt} W_{\gamma,f}^{\alpha}(t) dt = z^{-\alpha-1} \mathbf{P}_{z,\gamma}^{-1} f$, Re $z > \omega_{\gamma,\theta}$ and the set $\{e^{-\omega_{\gamma,\theta}z}W_{\gamma,f}^{\alpha}(z), z \in \Sigma_{\theta'}\}$ is bounded in X ($f \in X, \theta' \in (0, \theta)$). Define $W_{\gamma}^{\alpha}(z)f := W_{\gamma,f}^{\alpha}(z), z \in \Sigma_{\pi/2}, f \in X$. Then, for every $\theta \in (0, \pi/2)$ for $\theta \in (0, \pi/2)$, $z \in \Sigma_{\pi/2}$, $f \in X$.

 $(0, \pi/2), \{e^{-\omega_{\gamma,\theta}z}W_{\gamma}^{\alpha}(z): z \in \Sigma_{\theta}\} \subseteq L(X)$ is an equicontinuous operator family. By the foregoing, we have $z^{-\alpha}\mathbf{P}_{z}^{-1}f \to Df$, as $|z| \to \infty, z \notin (-\infty, 0] \ (f \in X)$; similarly, the operator family $\{z^{-\alpha-1}B\mathbf{P}_{z,\gamma}^{-1}: |z| \ge R, z \notin (-\infty, 0]\} \in L(X)$ is equicontinuous and $z^{-\alpha}B\mathbf{P}_{z}^{-1}f \to 0, |z| \to \infty, z \notin (-\infty, 0] \ (f \in X)$, so that we can define a strongly analytic operator family $(W_{\gamma,B}^{\alpha}(z))_{z\in\Sigma_{\pi/2}} \subseteq L(X)$ satisfying that, for every $\theta \in (0, \pi/2)$, the operator family $\{e^{-\omega'_{\gamma,\theta}z}W_{\beta,B}^{\alpha}(z): z \in \Sigma_{\theta}\} \subseteq L(X)$ is equicontinuous for some number $\omega'_{\gamma,\theta} > 0$. We have that, for every $f \in X$ and $\theta \in (0, \pi/2)$, $\lim_{z\to 0, z\in\Sigma_{\theta}} W_{\gamma}^{\alpha}(z)f = Df$. Define

$$W^{\alpha}(z) := \begin{cases} W_{0}^{\alpha}(z), & z \in \Sigma_{\pi/2}, \\ e^{i\alpha\pi/2}W_{\pi/2}^{\alpha}(ze^{-i\pi/2}), & \text{if } z \in e^{i\pi/2}\Sigma_{\pi/2}, \\ e^{-i\alpha\pi/2}W_{-\pi/2}^{\alpha}(ze^{i\pi/2}), & \text{if } z \in e^{-i\pi/2}\Sigma_{\pi/2}, \\ 0, & \text{if } z < 0, \\ D, & \text{if } z = 0, \end{cases}$$

and $W^{\alpha}_{B}(z)$ by replacing the operators $W^{\alpha}_{0}(z)$, $W^{\alpha}_{\pi/2}(ze^{-i\pi/2})$, $W^{\alpha}_{-\pi/2}(ze^{i\pi/2})$ and D in the above definition by the operators $W^{\alpha}_{0,B}(z)$, $W^{\alpha}_{\pi/2,B}(ze^{-i\pi/2})$, $W^{\alpha}_{-\pi/2,B}(ze^{i\pi/2})$ and 0, respectively $(z \in \mathbb{C})$. Then, for every $\nu \in (0,\pi)$, there exists a finite constant $\omega_{\nu} > 0$ such that the operator families $\{e^{-\omega_{\nu}z}W^{\alpha}(z):z\in\overline{\Sigma_{\nu}}\}\subseteq L(X)$ and $\{e^{-\omega_{\nu}z}BW^{\alpha}(z):z\in\overline{\Sigma_{\nu}}\}$ $=\{e^{-\omega_{\nu}z}W^{\alpha,B}(z):z\in\overline{\Sigma_{\nu}}\}\subseteq L(X)$ are equicontinuous as well as that, for every $f \in X$, the mappings $z \mapsto W^{\alpha}(z)f$ and $z \mapsto BW^{\alpha}(z)f$ are analytical on $\mathbb{C} \smallsetminus (-\infty, 0]$ and continuous on $\mathbb{C} \smallsetminus (-\infty, 0)$. Now it can be easily seen that $(W^{\alpha}(t))_{t\geq 0} \subseteq L(X, [D(B)])$ is an exponentially bounded $(g_{\alpha}, g_{\alpha+1})$ -regularized resolvent family generated by A, B. Furthermore, there exists a sufficiently large number R' > 0 such that $(zB - A)^{-1} \in L(X)$ for all $z \in S_{R'} := \{z \in \mathbb{C} \smallsetminus (-\infty, 0] : |z| = R'\}$. Denote, for every $f \in H^{2}(\Omega) \cap H^{1}_{0}(\Omega)$ and $z \in S_{R'}$,

$$u(t; f, z) := -W^{\alpha}(t)Bf - \int_{0}^{t} (t-s)^{\alpha-1} E_{\alpha,\alpha}((t-s)^{\alpha-1}z)W^{\alpha}(s)x \, ds + E_{\alpha}(t^{\alpha}z)(zB-A)^{-1}Bf, \quad t \ge 0.$$

Fix a number $z_0 \in S_{R'}$ such that $z_0 e^{\pm i\alpha/2} \in S_{R'}$. Using Theorem 2.3.18, it readily follows that the function $t \mapsto u(t; u_0, z_0) + \int_0^t u(s; u_1, z_0) ds$, $t \ge 0$ is a unique solution of problem (148)–(150) with the initial values u_0 and u_1 replaced by $(z_0 B - A)^{-1} B u_0$ and $(z_0 B - A)^{-1} B u_1$, respectively. On the other hand, using analytical properties of vector-valued Laplace transform we can prove that the function $t \mapsto \int_0^t (t - s)^{\alpha - 1} E_{\alpha,\alpha}((t - s)^{\alpha - 1} z) W^{\alpha}(s) x \, ds, t \ge 0$ can be continuously extended on the region $\mathbb{C} \setminus (-\infty, 0)$, analytically on the region $\mathbb{C} \setminus (-\infty, 0]$, by the formula

$$\begin{cases} \mathcal{L}^{-1} \Big(\frac{1}{z^{\alpha} - z_{0}} \frac{1}{z^{\alpha+1}} \mathbf{P}_{z}^{-1} \Big), & \text{if } z \in \Sigma_{\pi/2}, \\ \mathcal{L}^{-1} \Big(\frac{1}{z^{\alpha} - e^{i\pi\alpha/2} z_{0}} \frac{1}{z^{\alpha+1}} \mathbf{P}_{z,\pi/2}^{-1} \Big), & \text{if } z \in e^{i\pi/2} \Sigma_{\pi/2}, \\ \mathcal{L}^{-1} \Big(\frac{1}{z^{\alpha} - e^{-i\pi\alpha/2} z_{0}} \frac{1}{z^{\alpha+1}} \mathbf{P}_{z,-\pi/2}^{-1} \Big), & \text{if } z \in e^{-i\pi/2} \Sigma_{\pi/2}, \\ 0, & \text{if } z \leqslant 0. \end{cases}$$

This simply implies that, for any two functions $u_{0,1} \in H^2(\Omega) \cap H^1_0(\Omega)$, orthogonal to the eigenfunction(s) ϕ_k for $k = k_0$, there exists a unique analytical solution $t \mapsto u(t), t \ge 0$ of problem (148)–(150) on the region $\mathbb{C} \setminus (-\infty, 0]$ with the property that for each $\nu \in (0, \pi)$ there exists a finite constant $\omega_{\nu} > 0$ such that the sets $\{e^{-\omega_{\nu} z}u(z) : z \in \overline{\Sigma_{\nu}}\}$ and $\{e^{-\omega_{\nu} z}Bu(z) : z \in \overline{\Sigma_{\nu}}\}$ are bounded.

2.4. Abstract degenerate multi-term fractional differential equations with Riemann–Liouville derivatives

Let $n \in \mathbb{N} \setminus \{1\}$, let A, B and A_1, \ldots, A_{n-1} be closed linear operators on a complex Banach space E. Further on, assume that $0 \leq \alpha_1 < \cdots < \alpha_n$, $0 \leq \alpha < \alpha_n$, $0 < \tau \leq \infty$, f(t) is an E-valued function, and D_t^{α} denotes the Riemann–Liouville fractional derivative of order α . In this section, we investigate the abstract multi-term fractional differential equation

(151)
$$BD_t^{\alpha_n}u(t) + \sum_{j=1}^{n-1} A_j D_t^{\alpha_j}u(t) = AD_t^{\alpha}u(t) + f(t), \quad t \in (0,\tau).$$

We introduce and systematically analyze some new types of degenerate k-regularized (C_1, C_2) -existence and uniqueness (propagation) families for (151). Recall that $\alpha_0 = \alpha, m = \lceil \alpha \rceil, A_0 = A, A_n = B$ and $m_i = \lceil \alpha_i \rceil$ for $1 \leq i \leq n$. The following vector-valued modification of the condition (P1) will be used henceforth:

(P1)': $h(t): [0, \infty) \to E$ is Laplace transformable, i.e., $h \in L^1_{loc}([0, \infty) : E)$ and there exists $\beta \in \mathbb{R}$ such that $\tilde{h}(\lambda) := \mathcal{L}(h)(\lambda) := \lim_{b \to \infty} \int_0^b e^{-\lambda t} h(t) dt := \int_0^\infty e^{-\lambda t} h(t) dt$ exists for all $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \beta$. Put $\operatorname{abs}(h) := \inf\{\operatorname{Re} \lambda : \tilde{h}(\lambda) \text{ exists}\}.$

The inclusion $H(\lambda) \in LT_E$ means that there exist a function $h(t): [0, \infty) \to E$ satisfying (P1) and a number a > abs(h) so that $\tilde{h}(\lambda) = H(\lambda), \lambda > a$.

DEFINITION 2.4.1. Suppose $0 < \tau \leq \infty$ and $f \in L^1((0,\tau) : E)$. By a strong solution of (151) we mean any function $u \in L^1((0,\tau) : E)$ for which $g_{m_j-\alpha_j} * u \in W^{m_j,1}((0,\tau) : E)$ $(0 \leq j \leq n)$, $A_j D_t^{\alpha_j} u(t) \in L^1((0,\tau) : E)$ $(0 \leq j \leq n)$, and (151) holds for a.e. $t \in (0,\tau)$.

Now we would like to introduce the concept mild solution of (151), and to endow the equation (151) with corresponding initial conditions. In order to do that, let us assume that u(t) is a strong solution of (151). Then we can integrate the equation (151) α_n -times by using the formula (37) and the closedness of the operators A_j for $j \in \mathbb{N}_{n-1}^0$. In such a way, we get

(152)
$$B\left[u(t) - \sum_{i=0}^{m_n - 1} (g_{m_n - \alpha_n} * u)^{(i)}(0)g_{\alpha_n + i + 1 - m_n}(t)\right] + \sum_{j=1}^{n-1} g_{\alpha_n - \alpha_j} * A_j \left[u(t) - \sum_{i=0}^{m_j - 1} (g_{m_j - \alpha_j} * u)^{(i)}(0)g_{\alpha_j + i + 1 - m_j}(t)\right] = g_{\alpha_n - \alpha} * A\left[u(t) - \sum_{i=0}^{m-1} (g_{m-\alpha} * u)^{(i)}(0)g_{\alpha + i + 1 - m}(t)\right] + (g_{\alpha_n} * f)(t),$$

for all $t \in [0, \tau)$. This immediately implies that any strong solution of (151) satisfies

(153)
$$B\left[u(t) - \sum_{i=0}^{m_n-1} (g_{m_n-\alpha_n} * u)^{(i)}(0)g_{\alpha_n+i+1-m_n}(t)\right] \in C([0,\tau):E),$$

whence we may conclude that $u \in C([0, \tau) : E)$ provided that $B^{-1} \in L(E)$.

DEFINITION 2.4.2. Suppose $0 < \tau \leq \infty$ and $f \in L^1((0,\tau) :)$. By a mild solution of (151) we mean any function $u \in L^1((0,\tau) : E)$ for which $g_{m_j-\alpha_j} * u \in W^{m_j,1}((0,\tau) : E)$ $(0 \leq j \leq n)$, $A_j(g_{\alpha_n} * D_t^{\alpha_j} u(\cdot))(\cdot) \in C([0,\tau) : E)$ $(0 \leq j \leq n-1)$, (153) holds, and

(154)
$$B\left[u(t) - \sum_{i=0}^{m_n - 1} (g_{m_n - \alpha_n} * u)^{(i)}(0)g_{\alpha_n + i + 1 - m_n}(t)\right] + \sum_{j=1}^{n-1} A_j(g_{\alpha_n} * D_t^{\alpha_j} u(\cdot))(t)$$
$$= A(g_{\alpha_n} * D_t^{\alpha} u(\cdot))(t) + (g_{\alpha_n} * f)(t), \quad t \in [0, \tau).$$

By the foregoing, any strong solution of (151) is also a mild solution of the same problem; the converse statement is not true, in general. Observe that the equation (154) can be written in the following way:

(155)
$$B\left[u(t) - \sum_{i=0}^{m_n - 1} (g_{m_n - \alpha_n} * u)^{(i)}(0)g_{\alpha_n + i + 1 - m_n}(t)\right] + \sum_{j=1}^{n-1} A_j \left(g_{\alpha_n - \alpha_j} * \left[u(\cdot) - \sum_{i=0}^{m_j - 1} (g_{m_j - \alpha_j} * u)^{(i)}(0)g_{\alpha_j + i + 1 - m_j}(\cdot)\right]\right)(t) = A \left(g_{\alpha_n - \alpha} * \left[u(\cdot) - \sum_{i=0}^{m-1} (g_{m-\alpha} * u)^{(i)}(0)g_{\alpha + i + 1 - m}(\cdot)\right]\right)(t) + (g_{\alpha_n} * f)(t),$$

for all $t \in [0, \tau)$. Put

$$\mathcal{T}_{(151)} := \begin{cases} 1, & \text{if there exists } j \in \mathbb{N}_n^0 \text{ such that } \alpha_j \in \mathbb{N} \\ 0, & \text{otherwise,} \end{cases}$$

and $S := \{j \in \mathbb{N}_n^0 : \alpha_j \in \mathbb{N}\}$. For the sequel, it will be of crucial importance to recognise the following three subcases of (151):

(SC1) $\alpha_n > 1$: Then for each number $i \in \mathbb{N}_{m_n-1}$ we define the set \mathcal{D}_i by $\mathcal{D}_i := \{j \in \mathbb{N}_n^0 : m_j - 1 \ge i\}$. Observe that $n \in \mathcal{D}_i$ $(i \in \mathbb{N}_{m_n-1})$ and $\mathcal{D}_{m_n-1} \subseteq \cdots \subseteq \mathcal{D}_1$. Set $S_i := \{m_j - \alpha_j : j \in \mathcal{D}_i\}$ and, after that, $s_i := \operatorname{card}(S_i)$. Then $S_i \subseteq [0, 1)$ and S_i can be written in the following way

$$S_i = \{a_{i,1}, \ldots, a_{i,s_i}\},\$$

where $0 \leq a_{i,1} < \cdots < a_{i,s_i} \leq 1$ $(i \in \mathbb{N}_{m_n-1})$. Define $\mathcal{D}_i^l := \{j \in \mathcal{D}_i : m_j - \alpha_j = a_{i,l}\}$ $(i \in \mathbb{N}_{m_n-1}, 1 \leq l \leq s_i)$. Then for each number $i \in \mathbb{N}_{m_n-1}$ we introduce s_i initial values $x_{i,1}, \ldots, x_{i,s_i}$ for terms $(g_{m_j-\alpha_j} * u)^{(i)}(0)$, where $j \in \mathcal{D}_i$. In addition, if there exists $j \in \mathbb{N}_n^0$ such that $\alpha_j \in \mathbb{N}$, i.e., if $S \neq \emptyset$, then one has to introduce a new initial value x_0 for term $(g_0 * u)(0) \equiv u(0)$.

(SC2) $\alpha_n = 1$: Then we introduce only one initial value for term $(g_0 * u)(0) \equiv u(0)$.

(SC3) $\alpha_n < 1$: Then we consider the equation (151) without initial conditions. Define

$$\mathcal{B}_{(151)} := \begin{cases} s_1 + \dots + s_{m_n - 1} + \mathcal{T}_{(151)}, & \text{if } \alpha_n > 1, \\ 1, & \text{if } \alpha_n = 1, \\ 0, & \text{if } \alpha_n < 1. \end{cases}$$

Summa summarum, there will be exactly $\mathcal{B}_{(151)}$ initial conditions for (151).

The subcase (SC3) is very specific and therefore, not discussed henceforth. Consider now, for the sake of illustration and better understanding, the subcase (SC1). Plugging $x_{i,l} = (g_{m_j-\alpha_j} * u)^{(i)}(0)$ in (152), where $j \in \mathcal{D}_i^l$, and choosing other initial values to be zeroes, we obtain

(156)
$$B[u(t) - \chi_{\mathcal{D}_{i}^{l}}(n)g_{\alpha_{n}+i+1-m_{n}}(t)x_{i,l}] + \sum_{j=1}^{n-1} g_{\alpha_{n}-\alpha_{j}} * A_{j}[u(t) - \chi_{\mathcal{D}_{i}^{l}}(j)g_{\alpha_{j}+i+1-m_{j}}(t)x_{i,l}] = g_{\alpha_{n}-\alpha} * A[u(t) - \chi_{\mathcal{D}_{i}^{l}}(0)g_{\alpha+i+1-m}(t)x_{i,l}] \quad \text{for all } t \in [0,\tau).$$

If $S \neq \emptyset$, then inserting the initial value x_0 for u(0) in (152), and choosing $x_{i,l}$ to be zero for $i \in \mathbb{N}_{m_n-1}$ and $1 \leq l \leq s_i$, we obtain similarly that

(157)
$$B[u(t) - \chi_S(n)x_0] + \sum_{j=1}^{n-1} g_{\alpha_n - \alpha_j} * A_j[u(t) - \chi_S(j)x_0]$$
$$= g_{\alpha_n - \alpha} * A[u(t) - \chi_S(0)x_0] \quad \text{for all } t \in [0, \tau).$$

Suppose now, only for the purpose of further analysis, that $0 < \tau \leq \infty$, $K(t) \neq 0$ in $L^1_{loc}([0,\tau))$ and $k(t) = \int_0^t K(s) ds$, $t \in [0,\tau)$. Convoluting the above equations with K(t) and using the procedure similar to that already employed for abstract multi-term problems with Caputo fractional derivatives, we come to the following definition.

DEFINITION 2.4.3. Let $0 < \tau \leq \infty$, $k \in C([0, \tau))$, $C, C_1, C_2 \in L(E)$, and let C and C_2 be injective.

(i) (SC1) Suppose that, for every $i \in \mathbb{N}_{m_n-1}$ and $l \in \mathbb{N}_{s_i}$, $(R_{i,l}(t))_{t \in [0,\tau)} \subseteq L(E, [D(B)])$ is strongly continuous, as well as that, for every $t \in [0, \tau)$, $x \in E, i \in \mathbb{N}_{m_n-1}$ and $l \in \mathbb{N}_{s_i}$, the following functional equation

$$B[R_{i,l}(t)x - \chi_{\mathcal{D}_{i}^{l}}(n)(k * g_{\alpha_{n}+i-m_{n}})(t)C_{1}x] + \sum_{j=1}^{n-1} A_{j}[g_{\alpha_{n}-\alpha_{j}} * (R_{i,l}(\cdot)x - \chi_{\mathcal{D}_{i}^{l}}(j)(k * g_{\alpha_{j}+i-m_{j}})(\cdot)C_{1}x)](t) \\ = A[g_{\alpha_{n}-\alpha} * (R_{i,l}(\cdot)x - \chi_{\mathcal{D}_{i}^{l}}(0)(k * g_{\alpha+i-m})(\cdot)C_{1}x)](t)$$

holds. If $S \neq \emptyset$, then we also introduce a strongly continuous family $(R_{0,1}(t))_{t\in[0,\tau)} \subseteq L(E,[D(B)])$ satisfying that, for every $t \in [0,\tau)$ and $x \in E$,

$$B[R_{0,1}(t)x - \chi_S(n)k(t)C_1x] + \sum_{j=1}^{n-1} A_j [g_{\alpha_n - \alpha_j} * (R_{0,1}(\cdot)x - \chi_S(j)k(\cdot)C_1x)](t)$$

= $A[g_{\alpha_n - \alpha} * (R_{0,1}(\cdot)x - \chi_S(0)k(\cdot)C_1x)](t).$

Then the sequence $((R_{i,l}(t))_{t\in[0,\tau)})_{1\leqslant i\leqslant m_n-1,1\leqslant l\leqslant s_i}$ if $S = \emptyset$, resp., $((R_{i,l}(t))_{t\in[0,\tau)}, (R_{0,1}(t))_{t\in[0,\tau)})_{1\leqslant i\leqslant m_n-1,1\leqslant l\leqslant s_i}$ if $S \neq \emptyset$, is said to be a (local, if $\tau < \infty$) k-regularized C_1 -existence propagation family for (151). (SC2) A strongly continuous family $(R(t))_{t\in[0,\tau)} \subseteq L(E, [D(B)])$ satisfying that, for every $t \in [0, \tau)$ and $x \in E$,

$$B[R(t)x - k(t)C_1x] + \sum_{j=1}^{n-1} A_j(g_{\alpha_n - \alpha_j} * R(\cdot)x)(t) = A(g_{\alpha_n - \alpha} * R(\cdot)x)(t),$$

is said to be a (local, if $\tau < \infty$) k-regularized C₁-existence propagation family for (151).

(ii) (SC1) Suppose that, for every $i \in \mathbb{N}_{m_n-1}$ and $l \in \mathbb{N}_{s_i}$, $(W_{i,l}(t))_{t \in [0,\tau)} \subseteq L(E)$ is strongly continuous, as well as

$$[W_{i,l}(t)Bx - \chi_{\mathcal{D}_{i}^{l}}(n)(k * g_{\alpha_{n}+i-m_{n}})(t)C_{2}Bx] + \sum_{j=1}^{n-1} g_{\alpha_{n}-\alpha_{j}} * [W_{i,l}(\cdot)A_{j}x - \chi_{\mathcal{D}_{i}^{l}}(j)(k * g_{\alpha_{j}+i-m_{j}})(\cdot)C_{2}A_{j}x] = g_{\alpha_{n}-\alpha} * [W_{i,l}(\cdot)Ax - \chi_{\mathcal{D}_{i}^{l}}(0)(k * g_{\alpha+i-m})(\cdot)C_{2}Ax],$$

for every $i \in \mathbb{N}_{m_n-1}$, $l \in \mathbb{N}_{s_i}$, $t \in [0, \tau)$ and $x \in \bigcap_{0 \leq j \leq n} D(A_j)$. If $S \neq \emptyset$, then we also introduce a strongly continuous family $(W_{0,1}(t))_{t \in [0,\tau)} \subseteq L(E)$ satisfying that, for every $x \in \bigcap_{0 \leq j \leq n} D(A_j)$,

 $[W_{0,1}(\cdot)Bx - \chi_S(n)k(\cdot)C_2Bx]$

$$+ \sum_{j=1}^{n-1} g_{\alpha_n - \alpha_j} * [W_{0,1}(\cdot)A_j x - \chi_S(j)k(\cdot)C_2A_j x]$$

= $g_{\alpha_n - \alpha} * [W_{0,1}(\cdot)Ax - \chi_S(0)k(\cdot)C_2Ax].$

Then the sequence $((W_{i,l}(t))_{t\in[0,\tau)})_{1\leqslant i\leqslant m_n-1,1\leqslant l\leqslant s_i}$ if $S = \emptyset$, resp., $((W_{i,l}(t))_{t\in[0,\tau)}, (W_{0,1}(t))_{t\in[0,\tau)})_{1\leqslant i\leqslant m_n-1,1\leqslant l\leqslant s_i}$ if $S \neq \emptyset$, is said to be a (local, if $\tau < \infty$) k-regularized C_2 -uniqueness propagation family for (151). (SC2) A strongly continuous family $(W(t))_{t\in[0,\tau)} \subseteq L(E)$ satisfying that, for every $x \in \bigcap_{0\leqslant j\leqslant n} D(A_j)$ and $t \in [0,\tau)$,

$$[W(t)Bx - k(t)C_2Bx] + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * W(\cdot)C_2A_jx)(t) = (g_{\alpha_n - \alpha} * W(\cdot)C_2Ax)(t),$$

is said to be a (local, if $\tau < \infty$) k-regularized C₂-uniqueness propagation family for (151).

If $k(t) = g_{\zeta+1}(t)$, where $\zeta \ge 0$, then a k-regularized C_1 -existence propagation family for (151) is also said to be ζ -times integrated C_1 -existence propagation family for (151); 0-times integrated C_1 -existence propagation family for (151) is simply called C_1 -existence propagation family for (151); similar noions will be used for the classes of C_2 -uniqueness propagation families for (151) and C-resolvent propagation families for (151).

A k-regularized C_1 -existence propagation family for (151) is said to be locally equicontinuous (exponentially equicontinuous) iff each single operator family of it, considered as an element of the space L(E, [D(B)]), is locally equicontinuous (exponentially equicontinuous). The notion of an exponential equicontinuity of a k-regularized C_1 -existence propagation family for (151), of angle $\alpha \in (0, \pi/2]$, will be understood in the sense of Definition 2.3.24.

We define the notion of a mild (strong) solution of problem (156), resp. (157), as before: Let $\tau \in (0, \infty]$. By a mild solution of problem (156) on $[0, \tau)$ we mean any continuous function $t \mapsto u(t), t \in [0, \tau)$ satisfying that

$$B[u(t) - \chi_{\mathcal{D}_{i}^{l}}(n)g_{\alpha_{n}+i+1-m_{n}}(t)x_{i,l}] + \sum_{j=1}^{n-1} A_{j}(g_{\alpha_{n}-\alpha_{j}} * [u(\cdot) - \chi_{\mathcal{D}_{i}^{l}}(j)g_{\alpha_{j}+i+1-m_{j}}(\cdot)x_{i,l}])(t) = A(g_{\alpha_{n}-\alpha} * [u(\cdot) - \chi_{\mathcal{D}_{i}^{l}}(0)g_{\alpha+i+1-m}(\cdot)x_{i,l}])(t), \quad t \in [0,\tau).$$

By a strong solution of problem (156) on $[0, \tau)$ we mean any continuous function $t \mapsto u(t), t \in [0, \tau)$ satisfying that $t \mapsto A_j[u(t) - \chi_{\mathcal{D}_i^l}(j)g_{\alpha_j+i+1-m_j}(t)x_{i,l}], t \in [0, \tau)$ is continuous, as well as

$$B[u(t) - \chi_{\mathcal{D}_{i}^{l}}(n)g_{\alpha_{n}+i+1-m_{n}}(t)x_{i,l}] + \sum_{j=1}^{n-1} (g_{\alpha_{n}-\alpha_{j}} * A_{j}[u(\cdot) - \chi_{\mathcal{D}_{i}^{l}}(j)g_{\alpha_{j}+i+1-m_{j}}(\cdot)x_{i,l}])(t)$$

$$= (g_{\alpha_n - \alpha} * A[u(\cdot) - \chi_{\mathcal{D}^l}(0)g_{\alpha + i + 1 - m}(\cdot)x_{i,l}])(t), \quad t \in [0, \tau)$$

The notion of a mild (strong) solution of problem (157) on $[0, \tau)$ is defined similarly.

Then the following holds (for the sake of brevity, we shall consider only the subcase (SC1)):

- (A) If $S = \emptyset$ and $((R_{i,l}(t))_{t \in [0,\tau)})_{1 \leq i \leq m_n 1, 1 \leq l \leq s_i}$ is a C_1 -existence propagation family for (151), then the function $u_{i,l}(t) := R_{i,l}(t)C_1x, t \in [0,\tau)$ is a mild solution of (156) with $x_{i,l} = C_1x$ for $1 \leq i \leq m_n - 1, 1 \leq l \leq s_i$. If $S \neq \emptyset$ and $((R_{i,l}(t))_{t \in [0,\tau)}, (R_{0,1}(t))_{t \in [0,\tau)})_{1 \leq i \leq m_n - 1, 1 \leq l \leq s_i}$ is a C_1 -existence propagation family for (151), then the function $u_{0,1}(t) := R_{0,1}(t)C_1x, t \in [0,\tau)$ is a mild solution of (157) with $x_0 = C_1x$.
- (B) If $S = \emptyset$ and $((W_{i,l}(t))_{t \in [0,\tau)})_{1 \leq i \leq m_n 1, 1 \leq l \leq s_i}$ is a C_2 -uniqueness propagation family for (151), as well as $A_j W_{i,l}(t)x = W_{i,l}(t)A_jx$, $t \in [0,\tau)$, $x \in \bigcap_{0 \leq j \leq n} D(A_j)$ and $C_2A_j \subseteq A_jC_2$, $j \in \mathbb{N}_n^0$, then the function $u_{i,l}(t) := W_{i,l}(t)C_2^{-1}x_{i,l}$, $t \in [0,\tau)$ is a strong solution of (156) for $1 \leq i \leq m_n - 1, 1 \leq l \leq s_i$ and $x_{i,l} \in C_2(\bigcap_{0 \leq j \leq n} D(A_j))$. If $S \neq \emptyset$ and $((W_{i,l}(t))_{t \in [0,\tau)}, (W_{0,1}(t))_{t \in [0,\tau)})_{1 \leq i \leq m_n - 1, 1 \leq l \leq s_i}$ is a C₂-uniqueness propagation family for (151), as well as (in addition to the assumptions employed in the case that $S = \emptyset$) $A_j W_{0,1}(t)x = W_{0,1}(t)A_jx$, $t \in [0,\tau)$, $x \in \bigcap_{0 \leq j \leq n} D(A_j)$, then the function $u_0(t) := W_{0,1}(t)C_2^{-1}x_0$, $t \in [0,\tau)$ is a strong solution of (156) for $1 \leq i \leq m_n - 1, 1 \leq l \leq s_i$ and $x_0 \in C_2(\bigcap_{0 \leq j \leq n} D(A_j))$.

The assertions of [**346**, Proposition 2.3, Proposition 2.5, Proposition 2.6] admit reformulations in our new framework (cf. also Theorem 2.3.26). This is also the case with [**346**, Theorem 2.8], as the following theorem shows:

THEOREM 2.4.4. (SC1) Suppose that $S = \emptyset$, $C_2 \in L(E)$ is injective, $f \in C([0,\tau): E)$, and $((W_{i,l}(t))_{t \in [0,\tau)})_{1 \leq i \leq m_n - 1, 1 \leq l \leq s_i}$ is a k-regularized C_2 -uniqueness propagation family for (151). If a function u(t) is a strong solution of problem (114), then

$$\begin{split} & [W_{i,l}(\cdot) - \chi_{\mathcal{D}_{i}^{l}}(n)(k \ast g_{\alpha_{n}+i-m_{n}})(\cdot)C_{2}] \ast f \\ &= \sum_{j=1}^{n-1} [\chi_{\mathcal{D}_{i}^{l}}(j)(g_{\alpha_{n}+i-m_{j}} \ast kC_{2} \ast A_{j}u)(\cdot) - \chi_{\mathcal{D}_{i}^{l}}(n)(g_{2\alpha_{n}-\alpha_{j}+i-m_{n}} \ast kC_{2} \ast A_{j}u)(\cdot)] \\ & - [\chi_{\mathcal{D}_{i}^{l}}(0)(g_{\alpha_{n}+i-m} \ast kC_{2} \ast Au)(\cdot) - \chi_{\mathcal{D}_{i}^{l}}(n)(g_{2\alpha_{n}-\alpha+i-m_{n}} \ast kC_{2} \ast Au)(\cdot)]. \end{split}$$

If $S \neq \emptyset$ and $((W_{i,l}(t))_{t \in [0,\tau)}, (W_{0,1}(t))_{t \in [0,\tau)})_{1 \leq i \leq m_n - 1, 1 \leq l \leq s_i}$ is a k-regularized C_2 -uniqueness propagation family for (151), then we also have the following equality on $[0,\tau)$:

$$[W_{0,1}(\cdot)x - \chi_S(n)k(\cdot)C_2] * f = \sum_{j=1}^{n-1} (\chi_S(j) - \chi_S(n))(g_{\alpha_n - \alpha_j} * kC_2 * A_j u)(\cdot) - (\chi_S(0) - \chi_S(n))(g_{\alpha_n - \alpha} * kC_2 * A_j u)(\cdot).$$

(SC2) Suppose that $C_2 \in L(E)$ is injective, $(W(t))_{t \in [0,\tau)}$ is a k-regularized C_2 uniqueness propagation family for (151), and $f \in C([0,\tau) : E)$. If a function u(t) is a strong solution of problem (114), then the following equality holds on $[0,\tau)$:

$$[W(\cdot)x - k(\cdot)C_2] * f = (g_{\alpha_n - \alpha} * kC_2 * Au)(\cdot) - \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_2 * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_2 * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_j * A_ju)(\cdot) + \sum_{j=1}^{n-1} (g_{\alpha_n - \alpha_j} * kC_ju)(\cdot) + \sum_{j=1}^{n$$

As explained before for the equations with Caputo fractional derivatives, the notion of a k-regularized C_1 -existence propagation family is probably the best theoretical concept for the investigation of existence of integral solutions of nondegenerate abstract time-fractional equation (151) with $A_j \in L(E)$, $1 \leq j \leq n-1$; the Laplace transform cannot be so simply applied in the case that there is an index $j \in \mathbb{N}_{n-1}$ such that $A_j \notin L(E)$. In contrast to the above, it is very simple to reword the assertions of [**346**, Theorem 2.9(ii)] and [**314**, Theorem 2.4], concerning the Laplace transform of k-regularized C_2 -uniqueness propagation families, to degenerate differential equations with Riemann-Liouville fractional derivatives. The assertions of [**346**, Theorem 2.10-Theorem 2.12] can be rephrased for abstract degenerate multi-term problems with Riemann-Liouville fractional derivatives, as well. Having these done, it is not difficult to reconsider [**346**, Example 5.1(i)] in our new setting. Furthermore, it is not so difficult to construct some examples of local k-regularized I-resolvent propagation families for (151); see e.g. Example 2.3.28.

Before proceeding further, we would like to present three interesting examples of non-degenerate fractional differential equations that can be analyzed with the help of k-regularized C-resolvent propagation families:

EXAMPLE 2.4.5. [345]

(i) Based on the so-called CTRW method, T. A. M. Langlands, B. I. Henry and S. L. Wearne have suggested in [372] using Riemann–Liouville fractional derivatives for the approximation of passive electrotonic properties of spiny dendrites. In the abstract form (see e.g. [372, (1.18)]), the equation suggested by them takes the following form

(158)
$$D_t^1 u(t,x) + \mu^2 D_t^{1-\kappa} u(t,x) = D_t^{1-\gamma} \Delta u(t,x),$$

where $0 < \gamma < \kappa < 1$ and $\mu \in \mathbb{R} \setminus \{0\}$. The equation (158) is of type (SC2) and repeating verbatim the corresponding parts the analysis of equation [**346**, (5.4)] we can simply prove that there exists an exponentially equicontinuous, analytic resolvent propagation family for (158), of angle $\theta = \pi/2$ (recall that it is not clear whether the angle of analyticity can be improved by allowing that θ takes the value min $(\pi, \pi/\gamma - \pi/2)$), on a large class of Banach function spaces (consisted of functions acting on finite or infinite domain).

(ii) **[63]** The fractional order differential equation

$$u'(t) = Au(t) + \gamma D_t^{\alpha} Au(t) + f(t), \quad t > 0; \quad u(0) = x_0 \in E$$

has recently been studied in [63], provided that the operator A generates a strongly continuous semigroup on E, $\gamma > 0$, $\alpha \in (0,1)$ and f(t) is an E-valued function. Equations of this type appear in the modelling of unidirectional viscoelastic flows; in particular, they can be of some importance in the analysis of the Rayleigh–Stokes problem for generalized second-grade fluids.

(iii) Suppose that $n \in \mathbb{N} \setminus \{1\}, c_1, \ldots, c_{n-1} \in \mathbb{C}, 0 \leq \alpha_1 < \cdots < \alpha_n \leq 2, 0 < \beta < 1, 1 < \gamma \leq 2, k_\beta, k_\gamma > 0$ and L > 0. The following scalar multi-term time-space Caputo–Riesz fractional advection diffusion equation, (MT-TSCR-FADE) for short,

(159)
$$\mathbf{D}_{t}^{\alpha_{n}}u(t,x) + c_{n-1}\mathbf{D}_{t}^{\alpha_{n-1}}u(t,x) + \cdots + c_{1}\mathbf{D}_{t}^{\alpha_{1}}u(t,x) = k_{\beta}\frac{\partial^{\beta}u(t,x)}{\partial|x|^{\beta}} + k_{\gamma}\frac{\partial^{\gamma}u(t,x)}{\partial|x|^{\gamma}} \\ \left(\frac{\partial^{k}}{\partial t^{k}}u(t,x)\right)_{t=0} = u_{k}(x), \quad k = 0, \dots, \lceil \alpha_{n} \rceil - 1, \ 0 \leqslant x \leqslant L,$$

where $\frac{\partial^{\beta} u(t,x)}{\partial |x|^{\beta}}$ denotes the Riesz fractional operator of order β , has recently been analyzed by H. Jiang, F. Liu, I. Turner and K. Burrage in [260]. In [292, Example 2.9.53], we have considered two different evolution modelings of problem (159). In the first of these modelings, the equation (159) has been rewritten in the form of the following multi-term fractional differential equation:

(160)
$$\mathbf{D}_{t}^{\alpha_{n}}u(t) + c_{n-1}\mathbf{D}_{t}^{\alpha_{n-1}}u(t) + \dots + c_{1}\mathbf{D}_{t}^{\alpha_{1}}u(t) = -k_{\beta}A_{\beta/2} - k_{\gamma}A_{\gamma/2}$$
$$u^{(k)}(0) = u_{k}, \quad k = 0, \dots, \lceil \alpha_{n} \rceil - 1,$$

where the operator A belongs to the class $\mathcal{M}_{C,m}$ for some $m \in \mathbb{R}$, and acts on an appropriately chosen space of functions defined on [0, L] (cf. Section 1.1 and [**292**, Section 2.9] for more details about almost C-sectorial operators). As announced in [**292**], the analysis of equation (160) is quite complicated in the case that $C \neq I$ or $m \neq -1$; suppose, because of that and for the sake of simplicity, that A is a sectorial operator of angle $\omega \in [0, \pi)$, with D(A) and R(A) being dense in E. Let the so-called parabolicity condition $2\pi > (\beta + \gamma)\omega$ hold, let $\alpha_n^{-1}(\pi - (\gamma \omega/2)) - (\pi/2) > 0$, and let $1 < \alpha_n < 2$. Then we can prove with the help of Da Prato– Grisvard theorem (see e.g. [**245**, Theorem 9.3.1, Corollary 9.3.2]) that the operator $k_{\beta}A_{\beta/2} + k_{\gamma}A_{\gamma/2}$ is sectorial of angle $\gamma \omega/2$. Using this fact, it can be easily verified that for each $x_{1,1} \in E$ there exists a unique mild solution $t \mapsto u(t), t > 0$ of the corresponding integral equation (156) and that this solution can be analytically extended to the sector Σ_{δ} , with $\delta = \min(\pi/2, \alpha_n^{-1}(\pi - (\gamma \omega/2)) - (\pi/2))$. In the second modeling, we have analyzed the C-wellposedness of the backwards equation (MT-TSCR-FADE):

(161)
$$\mathbf{D}_{t}^{\alpha_{n}}u(t) + c_{n-1}\mathbf{D}_{t}^{\alpha_{n-1}}u(t) + \dots + c_{1}\mathbf{D}_{t}^{\alpha_{1}}u(t) = -k_{\beta}A_{\beta} - k_{\gamma}A_{\gamma}$$
$$u(0) = u_{0},$$

where $0 < \beta < 1$, $1 < \gamma < 2$, $k_{\beta}, k_{\gamma} > 0$, $0 \leq \alpha_1 < \cdots < \alpha_n < 1$, $\alpha_n^{-1}(\pi - (\gamma \pi/2)) > \pi/2$, $E := \{f \in C^{\infty}[0,1] ; \|f\| := \sup_{p \in \mathbb{N}_0} \frac{\|f^{(p)}\|_{\infty}}{p!^{\zeta}} < \infty \}$ ($\zeta > 1$) and A := -d/ds with $D(A) := \{f \in E : f' \in E, f(0) = 0\}$. Regrettably, in this approach, (161) is a modified abstract timerelaxation equation and almost nothing interesting can be said about the corresponding equation with Riemann–Liouville fractional derivatives (the subcase (SC3)).

Now we would like to state the following result on the existence of strong solutions of equation (151) (cf. [347, Theorem 3.1] and [314, Theorem 3.13] for the corresponding statement in the case of equations with Caputo fractional derivatives).

THEOREM 2.4.6. Suppose $A, B, A_1, \ldots, A_{n-1}$ are closed linear operators on E, $\omega > 0, 0 < \tau < \infty, C \in L(E)$ is injective, $f(t) \equiv 0$, the operator P_{λ} is injective for $\lambda > \omega$ and $D(P_{\lambda}^{-1}C) = E, \lambda > \omega$.

(SC1) Suppose $1 \leq i \leq m_n - 1$, $1 \leq l \leq s_i$, $n \in \mathcal{D}_i^l$, $Cx_{i,l} \in D(P_{\lambda}^{-1}A_j)$, provided $\lambda > \omega$ and $j \in \mathbb{N}_n^0 \cap \mathcal{D}_i^l$, as well as $\alpha_j - \alpha_n + m_n - 1 - i < 0$, provided $j \in \mathbb{N}_{n-1}^0 \smallsetminus \mathcal{D}_i^l$, and the following holds:

(162)
$$\lambda^{\alpha_{n}} P_{\lambda}^{-1} \bigg[\lambda^{m_{n}-i-1-\alpha} BCx_{i,l} + \sum_{j=1}^{n-1} A_{j}(\chi_{\mathcal{D}_{i}^{l}}(j)\lambda^{m_{j}-i-1-\alpha}Cx_{i,l}) - A(\chi_{\mathcal{D}_{i}^{l}}(0)\lambda^{m-i-1-\alpha}Cx_{i,l}) \bigg] - \lambda^{m_{n}-1-i}Cx_{i,l} \in LT_{E}$$

and

(163)
$$A_{j} \bigg\{ \lambda^{\alpha_{j}} P_{\lambda}^{-1} \bigg[\lambda^{m_{n}-i-1-\alpha} BCx_{i,l} + \sum_{j=1}^{n-1} A_{j} (\chi_{\mathcal{D}_{i}^{l}}(j)\lambda^{m_{j}-i-1-\alpha} Cx_{i,l}) - A(\chi_{\mathcal{D}_{i}^{l}}(0)\lambda^{m-i-1-\alpha} Cx_{i,l}) \bigg] - \chi_{\mathcal{D}_{i}^{l}}(j)\lambda^{m_{j}-i-1} Cx_{i,l} \bigg\} \in LT_{E}$$

Then there exists a strong solution of (151) on $(0, \tau)$, with initial value $x_{i,l}$ replaced by $Cx_{i,l}$ and other initial values chosen to be zeroes. If $S \neq \emptyset$ and the above conditions hold for the initial value $x_{0,1} = u(0)$ with set \mathcal{D}_i^l replaced by S (i = 0, l = 1), then there exists a strong solution of (151) on $(0, \tau)$, with initial value $x_{0,1}$ replaced by $Cx_{0,1}$ and other initial values chosen to be zeroes.

(SC2) Suppose $Cx_{0,1} \in D(B)$, $\lambda^{1-\alpha}P_{\lambda}^{-1}BCx_{0,1} - Cx_{0,1} \in LT_E$ and $\lambda^{\alpha_j-\alpha}A_jP_{\lambda}^{-1}BCx_{0,1} \in LT_E$.

Then there exists a strong solution of (151) on $(0, \tau)$, with initial value $x_{0,1}$ replaced by $Cx_{0,1}$ and other initial values chosen to be zeroes.

PROOF. We will prove the assertion of theorem only in the case (SC1) with $1 \leq i \leq m_n - 1$ and $1 \leq l \leq s_i$. Let $u_{i,l} \in L^1_{loc}([0,\infty) : E)$ and $F_{i,l,n} \in L^1_{loc}([0,\infty) : E)$ satisfy:

$$\begin{split} \int_0^\infty e^{-\lambda t} u_{i,l}(t) dt &= P_\lambda^{-1} \bigg[\lambda^{m_n - i - 1 - \alpha} BC x_{i,l} \\ &+ \sum_{j=1}^{n-1} A_j(\chi_{\mathcal{D}_i^l}(j) \lambda^{m_j - i - 1 - \alpha} C x_{i,l}) - A(\chi_{\mathcal{D}_i^l}(0) \lambda^{m - i - 1 - \alpha} C x_{i,l}) \bigg] \end{split}$$

and

$$\int_0^\infty e^{-\lambda t} F_{i,l,n}(t) dt = \lambda^{\alpha_n} P_\lambda^{-1} \left[\lambda^{m_n - i - 1 - \alpha} B C x_{i,l} - A(\chi_{\mathcal{D}_i^l}(0) \lambda^{m_n - i - 1 - \alpha} C x_{i,l}) \right]$$
$$+ \sum_{j=1}^{n-1} A_j(\chi_{\mathcal{D}_i^l}(j) \lambda^{m_j - i - 1 - \alpha} C x_{i,l}) - \lambda^{m_n - 1 - i} C x_{i,l},$$

for $\lambda > \omega$ suff. large; cf. (162). By performing the Laplace transform, it can be easily checked that:

$$(g_{m_n} * F_{i,l,n})(t) = (g_{m_n - \alpha_n} * u_{i,l})(t) - g_{i+1}(t)Cu_{i,l}, \quad t > 0.$$

This implies that $D_t^{\alpha_n} u_{i,l}(t)$ is well defined for t > 0 (more precisely, on any finite subinterval of $(0, \infty)$) and $F_{i,l,n}(t) = D_t^{\alpha_n} u_{i,l}(t), t > 0$. Keeping in mind that $n \in \mathcal{D}_i^l$ and $\alpha_j - \alpha_n + m_n - 1 - i < 0$ for $j \in \mathbb{N}_{n-1}^0 \setminus \mathcal{D}_i^l$, we can conclude from (162) that $\lambda^{\alpha_j} \widetilde{u_{i,l}}(\lambda) - \chi_{\mathcal{D}_i^l}(j) \lambda^{m_j - i - 1} C x_{i,l} \in LT_E$ for all $j \in \mathbb{N}_{n-1}^0$, as well as that $D_t^{\alpha_j} u_{i,l}(t)$ is well defined for t > 0, with

$$\int_0^\infty e^{-\lambda t} D_t^{\alpha_j} u_{i,l}(t) dt = \lambda^{\alpha_j} \widetilde{u_{i,l}}(\lambda) - \chi_{\mathcal{D}_i^l}(j) \lambda^{m_j - i - 1} C x_{i,l} \in LT_E,$$

for all $j \in \mathbb{N}_{n-1}^0$. Using (163) and [27, Proposition 1.7.6], it readily follows that $A_j D_t^{\alpha_j} u_{i,l}(t)$ is well defined for t > 0, and

$$\int_0^\infty e^{-\lambda t} A_j D_t^{\alpha_j} u_{i,l}(t) dt = A_j [\lambda^{\alpha_j} \widetilde{u_{i,l}}(\lambda) - \chi_{\mathcal{D}_i^l}(j) \lambda^{m_j - i - 1} C x_{i,l}] \in LT_E,$$

for all $j \in \mathbb{N}_{n-1}^0$. Finally, a simple calculation yields that:

$$\int_0^\infty e^{-\lambda t} [BD_t^{\alpha_n} u_{i,l}(t) + A_{n-1}D_t^{\alpha_{n-1}} u_{i,l}(t) + \dots + A_1D_t^{\alpha_1} u_{i,l}(t) - AD_t^{\alpha} u_{i,l}(t)] dt = 0,$$

which implies by the uniqueness theorem for the Laplace transform that $u_{i,l}(\cdot)$ is a strong solution of the problem (151) with initial value $x_{i,l}$ replaced by $Cx_{i,l}$ and other initial values chosen to be zeroes.

REMARK 2.4.7. Consider the subcase (SC1) and suppose first that $1 \leq i \leq m_n - 1$, $1 \leq l \leq s_i$, $n \in \mathcal{D}_i^l$, as well as $\alpha_j - \alpha_n + m_n - 1 - i < 0$, provided

 $j \in \mathbb{N}_{n-1}^0 \setminus \mathcal{D}_i^l$. Then a straightforward calculation shows that the assertion of Theorem 2.4.6 continues to hold if we replace the term

$$P_{\lambda}^{-1} \bigg[\lambda^{m_n - i - 1 - \alpha} BCx_{i,l} + \sum_{j=1}^{n-1} A_j(\chi_{\mathcal{D}_i^l}(j) \lambda^{m_j - i - 1 - \alpha} Cx_{i,l}) - A(\chi_{\mathcal{D}_i^l}(0) \lambda^{m - i - 1 - \alpha} Cx_{i,l}) \bigg],$$

i.e., the Laplace transform of solution $t \mapsto u_{i,l}(t), t > 0$ with the term

$$\lambda^{m_n-i-1-\alpha_n} C x_{i,l} - P_{\lambda}^{-1} \bigg[\sum_{j \in \mathbb{N}_{n-1} \smallsetminus \mathcal{D}_i^l} A_j (\lambda^{m_n-i-1-\alpha_n+\alpha_j-\alpha} C x_{i,l}) + A(\chi_{\mathbb{N}_{n-1} \smallsetminus \mathcal{D}_i^l}(0) \lambda^{m_n-i-1-\alpha_n} C x_{i,l}) \bigg],$$

and suppose that $Cx_{i,l} \in D(P_{\lambda}^{-1}A_j)$ for $\lambda > \omega$ and $j \in \mathbb{N}_{n-1}^0 \setminus \mathcal{D}_i^l$, instead of $Cx_{i,l} \in D(P_{\lambda}^{-1}A_j)$ for $\lambda > \omega$ and $j \in \mathbb{N}_{n-1}^0 \cap \mathcal{D}_i^l$. If $S \neq \emptyset$, i = 0 and l = 1, then one has to replace the set \mathcal{D}_i^l with S. The corresponding analysis of the subcase (SC2) is left to the interested reader.

Before proceeding further, we would like to present an illustrative example in which the existence of strong solutions of problem (151) can be proved trivially and which also shows that there exists a large class of (degenerate) multi-term problems which do have strong solutions that are completely independent of the choice of operators A_j .

EXAMPLE 2.4.8. Consider the subcase (SC1) with $1 \leq i \leq m_n - 1$, l = 1, $m_j \geq i+1$, $j \in \mathbb{N}_0$ and $m_j - \alpha_j = a_{i,1}$, $j \in \mathbb{N}_0$. Let $0 < \tau < \infty$ and $f(t) \equiv 0$. Then $D_t^{\alpha_j} g_{i+1-a_{i,1}}(t) = 0$, $j \in \mathbb{N}_0$ so that the function $u(t) = g_{i+1-a_{i,1}}(t)x_{i,1}$, $t \in [0, \tau)$ is a strong solution of problem (151) with B = I, $x_{i,1} = (g_{m_j - \alpha_j} * u)^{(i)}(0) \in X$ and other initial values chosen to be zeroes.

In the remainder of this section, we assume that X (the state space) and Y are complex Banach spaces; the norm of an element $y \in Y$ will be denoted by $||y||_Y$. Now the closed linear operators A, B, A_1, \ldots, A_{n-1} are acting on X. By a (local) C_1 -existence family, resp. k-regularized C_2 -uniqueness family, we mean (local) C_1 existence family for (112), resp. k-regularized C_2 -uniqueness family for (112).

The first part of subsequent theorem can be proved following the analysis carried out in [345, Section 3]; the second part of this theorem can be proved by using Theorem 2.3.32(ii) and the fact that $D_t^{\zeta}u(t) = \mathbf{D}_t^{\zeta}u(t)$, t > 0, provided that $\zeta > 0$, $\mathbf{D}_t^{\zeta}u(t)$ is defined and $u^{(i)}(0) = 0$ for all $i \in \mathbb{N}_{\lceil \zeta \rceil - 1}^0$. Observe also that we can reformulate the final conclusions from Theorem 2.3.6 for degenerate multi-term problems with Riemann–Liouville fractional derivatives and that it is very difficult to state, in contrast to the equations with Caputo fractional derivatives, some satisfactory results on the existence of strong solutions of (151), provided that there exists a (local) C_1 -existence family. THEOREM 2.4.9. Let $0 < \tau \leq \infty$, let $C_1 \in L(Y, X)$, and let $C_2 \in L(X)$ be injective.

(i) Suppose that (E(t))_{t∈[0,τ)} is a (local) C₁-existence family, 1 ≤ i ≤ m_n − 1 and 1 ≤ l ≤ s_i. Assume further that, for every j ∈ N⁰_{n-1} ∩ D^l_i, there exists an element y_{i,l,j} ∈ Y such that A_jx_{i,l} = C₁y_{i,l,j}, as well as that the condition n ∈ D^l_i implies the existence of an element y_{i,l} ∈ Y such that Bx_{i,l} = C₁y_{i,l}. Define

$$u(t) := \int_0^t E^{(m_n-1)}(t-s) \bigg[\chi_{\mathcal{D}_i^l}(n) g_{\alpha_n+i-m_n}(s) y_{i,l} + \sum_{j \in \mathbb{N}_{n-1} \cap \mathcal{D}_i^l} g_{\alpha_n+i-m_j}(s) y_{i,l,j} - \chi_{\mathcal{D}_i^l}(0) g_{\alpha_n+i-m}(s) y_{i,l,0} \bigg] ds, \quad t \in [0,\tau).$$

Then the function $t \mapsto u(t), t \in [0, \tau)$ is a mild solution of the problem (156). If $S \neq \emptyset$, i = 0 and l = 1, then the above holds with the set \mathcal{D}_i^l replaced by S.

(ii) Suppose that (U(t))_{t∈[0,τ)} is a k-regularized C₂-uniqueness family and k(t) is a kernel on [0,τ). Then every two strong (mild) solutions of the equation (151) possessing the same initial conditions (cf. (152) and (155)) are identically equal on [0,τ).

In Theorem 2.3.34, we have analyzed inhomogeneous multi-term problems with Caputo fractional derivatives. Similarly we can analyze the inhomogeneous multiterm problems with Riemann–Liouville fractional derivatives.

2.5. The existence and uniqueness of solutions of abstract degenerate fractional differential equations: ultradistribution theory

Our first task in this section will be to extend the assertions of [303, Theorem 2.1, Corollary 2.1] to abstract degenerate fractional differential equations (the Gevrey case). In order to do that, fix the numbers $\zeta \in (0, 1]$, $\alpha > 0$, $\beta > 0$, $l \ge 1$, $\xi \ge 0$, $b \in (0, 1)$, and denote by $M_v(\cdot)$ the associated function of the sequence $(p^{\frac{p}{v}})$ $(v \in (0, 1))$. Then we know that $M_v(t) \sim (ve)^{-1}t^v$ as $t \to +\infty$. If (N_p) and (R_p) are two sequences of positive real numbers, then we write $N_p \prec R_p$ iff for each number $\sigma > 0$ we have

$$\sup_{p\in\mathbb{N}_0}\frac{N_p\sigma^p}{R_p}<\infty.$$

Henceforth we shall always assume that (M_p) is a sequence of positive real numbers such that $M_0 = 1$ and the condition (M.1) holds. Suppose that

(164)
$$p^{\frac{p}{b}} \prec M_p$$

Then, for every $\mu > 0$, there exist positive real constants $c_{\mu} > 0$ and $C_{\mu} > 0$ such that $\lim_{\mu \to 0} c_{\mu} = 0$ and

(165)
$$M(l\lambda) \leqslant M_b(\mu l\lambda) + C_\mu \leqslant c_\mu |\lambda|^b + C_\mu, \quad \lambda \ge 0.$$

As mentioned earlier, the (M_n) -ultralogarithmic region of type l

$$\Lambda_{\alpha,\beta,l} := \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda \ge \alpha M(l | \operatorname{Im} \lambda|) + \beta\},\$$

where $\alpha > 0, \beta \in \mathbb{R}$ and $l \ge 1$, was defined for the first time by J. Chazarain in 1971 [99]. We assume that the boundary of the ultra-logarithmic region $\Lambda_{\alpha,\beta,l}$, denoted by Γ_l , is upwards oriented. Set

$$\Lambda^{\zeta}_{\alpha,\beta,l} := \{\lambda^{\zeta} : \lambda \in \Lambda_{\alpha,\beta,l}\} \text{ and } \Omega := \mathbb{C} \smallsetminus \Lambda^{\zeta}_{\alpha,\beta,l}$$

By \mathcal{A} we denote the class consisting of all continuous functions $f: \overline{\Omega} \to \mathbb{C}$ that are analytic in Ω and satisfy that there exist constants $a_1 > 0$ and $a_2 > \xi$ such that:

(166)
$$|f(\lambda)| \leq a_1 e^{-a_2|\lambda|^{b/\zeta}}, \quad \lambda \in \bar{\Omega}.$$

Suppose that $f \in \mathcal{A}, f \neq 0$. Then we define $F(\cdot)$ by

$$F(\lambda) := f(-\lambda^{\frac{\pi - (\zeta \pi/2)}{\pi/2}}), \quad \lambda \in \overline{\Sigma_{\pi/2}}.$$

The function $F(\cdot)$ can be analytically extended to an open neighborhood of the region $\overline{\Sigma_{\pi/2}}$ and satisfies that:

$$|F(\lambda)|\leqslant a_1e^{-a_2|\lambda|^{\frac{\pi-(\zeta\pi/2)}{\pi/2}\frac{b}{\zeta}}},\quad\lambda\in\overline{\Sigma_{\pi/2}}.$$

Now we can apply the Phragmén-Lindelöf type theorems (see e.g. [**376**, p. 40]) in order to see that the inequality $\frac{\pi - (\zeta \pi/2)}{\pi/2} \frac{b}{\zeta} \ge 1$ implies f = 0 identically. Hence, one has to assume that $\frac{\pi - (\zeta \pi/2)}{\pi/2} \frac{b}{\zeta} < 1$, i.e., that

(167)
$$\frac{1}{2-\zeta} > \frac{b}{\zeta}$$

in order to ensure the non-triviality of class \mathcal{A} (observe that $1/(2-\zeta) \in (1/2, 1]$ for $\zeta \in (0, 1]$, so that (167) automatically implies $b < \zeta$). Suppose now that (167) holds. Then the class \mathcal{A} is non-trivial, indeed, and we can simply prove this fact as follows. Put $\theta := \arctan(\cos(\frac{b}{\zeta}(\pi - \frac{\pi\zeta}{2})))$. Then the function

$$f(\lambda) = f_t(\lambda) := e^{-t(-\lambda+\omega)^{b/\zeta}}, \quad \lambda \in \overline{\Omega}$$

belongs to \mathcal{A} provided $t = t_1 + it_2 \in \Sigma_{\theta}$, $\omega > \beta^{\zeta}$ and $t_1 \tan \theta - |t_2| > \xi$, because $\arg(-\lambda^{\zeta} + \omega) \to \pi - \frac{\pi}{2}\zeta$ as $|\lambda| \to \infty$, $\lambda \in \Gamma_l$ and there exists R > 0 such that, for every $t = t_1 + it_2 \in \Sigma_{\theta}$,

$$\begin{aligned} |e^{-t(-\lambda+\omega)^{b/\zeta}}| &= e^{-t_1|-\lambda+\omega|^{b/\zeta}\cos(\frac{b}{\zeta}\arg(-\lambda+\omega))+t_2|-\lambda+\omega|^{b/\zeta}\sin(\frac{b}{\zeta}\arg(-\lambda+\omega))} \\ &\leqslant e^{-(t_1\cos(\frac{b}{\zeta}\arg(-\lambda+\omega))-|t_2|)|-\lambda+\omega|^{b/\zeta}} \leqslant e^{-(t_1\tan\theta-|t_2|)|-\lambda+\omega|^{b/\zeta}}, \ \lambda\in\bar{\Omega}, \ |\lambda|\geqslant R. \end{aligned}$$

It is clear that $f \cdot g, f + g, zf \in \mathcal{A}$, provided $f, g \in \mathcal{A}$ and $z \in \mathbb{C}$. In order to stay consistent with the notation used in our joint paper with V. Fedorov [213], the time-variable in this section will be denoted by s. Further on, let $n \in \mathbb{N}$, let p_0, p_1, \ldots, p_n and q_0, q_1, \ldots, q_n be given non-negative integers satisfying $p_0 = q_0 = 0$ and $0 < p_1 + q_1 \leq p_2 + q_2 \leq \ldots \leq p_n + q_n$. Let $A_0 = A, A_1, \ldots, A_{n-1}, A_n = B$ be closed linear operators acting on an SCLCS E. Set, with a little abuse of notation, $T_i u(s) := (\mathbf{D}_s^{\zeta})^{p_i} A_i(\mathbf{D}_s^{\zeta})^{q_i} u(s), \ s \ge 0, \ i \in \mathbb{N}_n^0, \ S_l := \{i \in \mathbb{N}_n : q_i \ge 1\},$ $S_r := \{i \in \mathbb{N}_n : p_i \ge 1\},$

$$P_{\lambda} := \lambda^{(p_n + q_n)\zeta} B + \sum_{i=0}^{n-1} \lambda^{(p_i + q_i)\zeta} A_i, \quad \lambda \in \mathbb{C} \smallsetminus \{0\},$$

and conventionally, $\max(\emptyset) := \emptyset$, $\mathbb{N}^0_{\emptyset} := \emptyset$. We analyze the abstract degenerate Cauchy problem [(168)-(169)], where

(168)
$$\sum_{i=0}^{n} T_i u(s) = 0, \quad s \ge 0$$

and

(169)
$$((\mathbf{D}_{s}^{\zeta})^{j}u(s))_{s=0} = u_{j}, \quad j \in \mathbb{N}_{\max\{q_{i}-1:i \in S_{l}\}}^{0}, \text{ and} \\ ((\mathbf{D}_{s}^{\zeta})^{j}A_{i}(\mathbf{D}_{s}^{\zeta})^{q_{i}}u(s))_{s=0} = u_{i,j} \quad (i \in S_{r}, \ j \in \mathbb{N}_{p_{i}-1}^{0}).$$

Denote by $(\text{DFP})'_R$ $((\text{DFP})'_L)$ the homogeneous abstract fractional Cauchy problem obtained by replacing the number α and the operator A in problem $(\text{DFP})_R$ $((\text{DFP})_L)$ with the number $\zeta \in (0, 1]$ and the operator -A, respectively:

$$(DFP)'_{R}: \begin{cases} \mathbf{D}_{s}^{\zeta}Bu(s) + Au(s) = 0, & s \ge 0, \\ Bu(0) = Bx, \end{cases}$$
$$(DFP)'_{L}: \begin{cases} B\mathbf{D}_{s}^{\zeta}u(s) + Au(s) = 0, & s \ge 0, \\ u(0) = x. \end{cases}$$

The problems $(\text{DFP})'_R$ and $(\text{DFP})'_L$ are special cases of the problem [(168)-(169)], with n = 1, $q_1 = 0$, $p_1 = 1$ and $u_{1,0} = Bx$, in the case of problem $(\text{DFP})'_R$, and n = 1, $q_1 = 1$, $p_1 = 0$, $u_1 = x$, in the case of problem $(\text{DFP})'_L$. A function $u \in C([0,\infty) : E)$ is said to be a strong solution of problem [(168)-(169)] iff the term $T_i u(s)$ is well defined and continuous for any $s \ge 0$, $i \in \mathbb{N}^0_n$, and [(168)-(169)]holds identically on $[0,\infty)$. If the function $u \in C([0,\infty) : E)$ is a strong solution of problem [(168)-(169)] with all initial values chosen to be zeroes, then we can integrate the equation $(168) ((p_n + q_n)\zeta)$ -times; taking into account the equality (38) and our choice of initial values in (169), we get that

(170)
$$Bu(s) + \sum_{i=0}^{n-1} A_i(g_{((p_n+q_n)-(p_i+q_i))\zeta} * u)(s) = 0, \quad s \ge 0.$$

Now we are ready to formulate the following extension of [303, Theorem 2.1]:

THEOREM 2.5.1. Suppose that (M_p) satisfies (M.1), $b \in (0,1)$, $\zeta \in (0,1]$ and (164) holds. Let $\nu > -1$, $\xi \ge 0$, $\alpha > 0$, $\beta > 0$, $l \ge 1$, and let (167) hold. Suppose, further, that the operator P_{λ} is injective for all $\lambda \in \Lambda_{\alpha,\beta,l}$, as well as that $P_{\lambda}^{-1}C \in L(E)$, $\lambda \in \Lambda_{\alpha,\beta,l}$, the mapping $\lambda \mapsto P_{\lambda}^{-1}Cx$, $\lambda \in \Lambda_{\alpha,\beta,l}$ is continuous for every fixed element $x \in E$, and the operator family

$$\{(1+|\lambda|)^{-\nu}e^{-\xi|\lambda|^o}P_{\lambda}^{-1}C:\lambda\in\Lambda_{\alpha,\beta,l}\}\subseteq L(E)$$

is equicontinuous. Set, for every function $f \in \mathcal{A}$,

(171)
$$S_f(s)x := \frac{\zeta}{2\pi i} \int_{\Gamma_l} f(\lambda^{\zeta}) \lambda^{\zeta-1} E_{\zeta}(s^{\zeta} \lambda^{\zeta}) P_{\lambda}^{-1} Cx \, d\lambda, \quad s \ge 0, \ x \in E.$$

Then $(S_f(s))_{s\geq 0} \subseteq L(E)$ is strongly continuous, the mapping $s \mapsto S_f(s) \in L(E)$, $s \geq 0$ ($s \mapsto S_f(s) \in L(E)$, s > 0) is infinitely differentiable provided $\zeta = 1$, $f \in \mathcal{A}$ ($\zeta \in (0,1)$, $f \in \mathcal{A}$) and, for every $p \in \mathbb{N}_0$ and $f \in \mathcal{A}$, the mapping $s \mapsto (\mathbf{D}_{\varsigma}^{\zeta})^p S_f(s) \in L(E)$, $s \geq 0$ is well-defined, with

(172)
$$(\mathbf{D}_{s}^{\zeta})^{p}S_{f}(s)x := \frac{\zeta}{2\pi i} \int_{\Gamma_{l}} f(\lambda^{\zeta})\lambda^{\zeta-1}\lambda^{p\zeta}E_{\zeta}(s^{\zeta}\lambda^{\zeta})P_{\lambda}^{-1}Cx\,d\lambda, \quad s \ge 0, \ x \in E.$$

Furthermore, the following holds:

(i) Suppose that there exists $i \in \mathbb{N}_n^0$ such that the mappings $\lambda \mapsto A_j P_{\lambda}^{-1} Cx$, $\lambda \in \Lambda_{\alpha,\beta,l}$ are continuous for some $x \in E$ $(j \in \mathbb{N}_n^0 \setminus \{i\})$ and for each seminorm $\mathbf{p} \in \circledast$ the set

$$\{(1+|\lambda|)^{-\nu}e^{-\xi|\lambda|^{b}}\mathbf{p}(A_{j}P_{\lambda}^{-1}Cx):\lambda\in\Lambda_{\alpha,\beta,l},\ j\in\mathbb{N}_{n}^{0}\smallsetminus\{i\}\}$$

is bounded. Then we have

(173)
$$(\mathbf{D}_{s}^{\zeta})^{p} A_{i}(\mathbf{D}_{s}^{\zeta})^{q} S_{f}(s) x = \frac{\zeta}{2\pi i} \int_{\Gamma_{l}} f(\lambda^{\zeta}) \lambda^{\zeta-1} \lambda^{(p+q)\zeta} E_{\zeta}(s^{\zeta}\lambda^{\zeta}) A_{i} P_{\lambda}^{-1} Cx \, d\lambda,$$

for any $x \in E$, $s \ge 0$, $i \in \mathbb{N}_n^0$ and $p, q \in \mathbb{N}_0$. Moreover, the mapping $s \mapsto u(s) := S_f(s)x$, $s \ge 0$ is a strong solution of the problem [(168)–(169)], with the initial value u_j obtained by plugging p = j and s = 0 into the right-hand side of (172), for $j \in \mathbb{N}_{\max\{q_i-1:i\in S_l\}}^0$, and the initial value $u_{i,j}$ obtained by plugging p = j, $q = q_i$ and s = 0 into the right-hand side of (173), for $i \in S_r$ and $j \in \mathbb{N}_{p_i-1}^0$ ($f \in \mathcal{A}$). If $CA_i \subseteq A_iC$ for all $i \in \mathbb{N}_n^0$, then there exists at most one strong solution of problem [(168)–(169)].

(ii) Suppose that f ∈ A, q ∈ ⊛, B is a bounded subset of E and K is a compact subset of [0,∞). Then there exists h₀ > 0 such that

(174)
$$\sup_{p \in \mathbb{N}_0, s \in K, x \in \mathbb{B}} \frac{(h_0)^p q((\mathbf{D}_s^{\zeta})^p S_f(s)x)}{p^{p\zeta/b}} < \infty.$$

PROOF. We will basically follow the proof of [303, Theorem 2.1]. Let $f \in \mathcal{A}$ be such that (166) holds with some numbers $a_1 > 0$ and $a_2 > \xi$. In order to prove that $S_f(s) \in L(E)$ for all $s \ge 0$, observe that Theorem 1.5.1 in combination with (165) and the equicontinuity of operator family $\{(1 + |\lambda|)^{-\nu}e^{-\xi|\lambda|^b}P_{\lambda}^{-1}C : \lambda \in \Lambda_{\alpha,\beta,l}\}$ implies that for each $\mathbf{p} \in \circledast$ there exist $c_{\mathbf{p}} > 0$ and $\mathbf{q} \in \circledast$ such that, for any sufficiently small number $\mu > 0$, the following holds with an appropriate constant $M_{\mu} > 0$:

(175)
$$|f(\lambda^{\zeta})\lambda^{\zeta-1}E_{\zeta}(s^{\zeta}\lambda^{\zeta})\mathbf{p}(P_{\lambda}^{-1}Cx)|$$
$$\leqslant a_{1}M_{\mu}c_{\mathbf{p}}e^{-(a_{2}-\xi)|\lambda|^{b}}e^{s(\beta+c_{\mu}|\lambda|^{b})}(1+|\lambda|)^{\nu+\zeta}\mathbf{q}(x), \quad \lambda\in\Gamma_{l}, \ |\lambda|\geqslant R, \ x\in E.$$

Keeping in mind that $\lim_{\mu\to 0} c_{\mu} = 0$, we obtain from (175) that $S_f(s) \in L(E)$ for all $s \ge 0$, and that the operator family $(S_f(s))_{s\ge 0} \subseteq L(E)$ is strongly continuous. The

infinite differentiability of mapping $s \mapsto S_f(s) \in L(E)$, $s \ge 0$ for $\zeta = 1$ and $f \in \mathcal{A}$ can be simply proved. In order to prove that the mapping $s \mapsto S_f(s) \in L(E)$, s > 0 is infinitely differentiable for $\zeta < 1$ and $f \in \mathcal{A}$, we need to recall the well known fact that, for every $l \in \mathbb{N}$, there exist real numbers $(c_{j,\zeta})_{1 \le j \le l}$ and $(c_{j,l,\zeta})_{1 \le j \le l}$ such that

$$\frac{d^l}{ds^l}E_{\zeta}(zs^{\zeta}) = \sum_{j=1}^l c_{j,\zeta}s^{j\zeta-l}E_{\zeta}^{(j)}(zs^{\zeta}), \quad s > 0, \ z \in \mathbb{C}$$

and

$$\frac{d^l}{dz^l}E_{\zeta}(z) = \sum_{j=1}^l c_{j,l,\zeta}E_{\zeta,\zeta l-(l-j)}(z), \quad z \in \mathbb{C}$$

(cf. Section 1.2 and [292, Section 1.3]). This implies that, for every $l \in \mathbb{N}$, and for every h > 0 suff. small, we have:

$$(176) \quad \frac{E_{\zeta}^{(l)}((s+h)^{\zeta}\lambda^{\zeta}) - E_{\zeta}^{(l)}(s^{\zeta}\lambda^{\zeta})}{h} - \frac{d^{l+1}}{ds^{l+1}}E_{\zeta}(s^{\zeta}\lambda^{\zeta}) = \frac{1}{h}\sum_{j=1}^{l+2}\sum_{i=1}^{j}\int_{s}^{s+h}\int_{s}^{r}c_{j,\zeta}c_{i,j,\zeta}\tau^{j\zeta-(l+2)}E_{\zeta,\zeta j-(i-j)}(\tau^{\zeta}\lambda^{\zeta})d\tau\,dr, \quad s > 0, \ \lambda \in \Gamma_{l}.$$

An application of Theorem 1.5.1 yields that, for every $l \in \mathbb{N}$, there exists a constant $\delta > 0$ satisfying that, for every $j \in \mathbb{N}$ with $j \leq l+2$, and for every $i \in \mathbb{N}$ with $i \leq j$, we have

$$|E_{\zeta,\zeta j-(i-j)}(\tau^{\zeta}\lambda^{\zeta})| \leq \delta[1+(\tau\lambda)^{(1+(i-j)-\zeta j)/\zeta}e^{\tau\operatorname{Re}\lambda}], \quad \tau > 0, \ \lambda \in \Gamma_l.$$

Combining this estimate with (176), it readily follows that the mapping $s \mapsto S_f(s) \in L(E)$, s > 0 is *l*-times continuously differentiable, with

(177)
$$\frac{d^l}{ds^l}S_f(s)x = \frac{\zeta}{2\pi i} \int_{\Gamma_l} f(\lambda^{\zeta})\lambda^{\zeta-1} \frac{d^l}{ds^l} [E_{\zeta}(s^{\zeta}\lambda^{\zeta})] P_{\lambda}^{-1} Cx \, d\lambda, \quad s > 0, \ x \in E.$$

Using the identity $\lambda^{\zeta}(g_{\lceil \zeta \rceil} * E_{\zeta}(\cdot^{\zeta} \lambda^{\zeta}))(s) = (g_{\lceil \zeta \rceil - \zeta} * [E_{\zeta}(\cdot^{\zeta} \lambda^{\zeta}) - 1])(s), s \ge 0, \lambda \in \Gamma_l$ (see e.g. [61, (1.25)] and the proof of [292, Lemma 3.3.1]) and a straightforward integral computation, it is checked at once that for each $x \in E$ and $s \ge 0$ we have:

$$[g_{\lceil \zeta \rceil - \zeta} * (S_f(\cdot)x - S_f(0)x)](s) = \left[g_{\lceil \zeta \rceil} * \frac{\zeta}{2\pi i} \int_{\Gamma_l} f(\lambda^{\zeta}) \lambda^{\zeta - 1} \lambda^{\zeta} E_{\zeta}(\cdot^{\zeta} \lambda^{\zeta}) P_{\lambda}^{-1} Cx \, d\lambda\right](s).$$

This implies the validity of (172) with p = 1. Inductively, we obtain that (172) holds for any integer $p \in \mathbb{N}$ by repeating verbatim the above arguments.

Suppose now that the requirements of (i) hold for some element $x \in E$. Using the resolvent equation, we obtain that the mappings $\lambda \mapsto A_i P_{\lambda}^{-1}Cx$, $\lambda \in \Lambda_{\alpha,\beta,l}$ are continuous for all $i \in \mathbb{N}_n^0$ and there exists a number $\nu' > 0$ such that for each seminorm $\mathbf{p} \in \circledast$ the set $\{(1 + |\lambda|)^{-\nu'} e^{-\xi|\lambda|^b} \mathbf{p}(A_i P_{\lambda}^{-1}Cx) : \lambda \in \Lambda_{\alpha,\beta,l}, i \in \mathbb{N}_n^0\}$ is bounded, which clearly implies that the mapping $s \mapsto A_i S_f(s) x, s \ge 0$ is well defined for any $x \in E$ and $i \in \mathbb{N}_n^0$. By the foregoing, we have that (173) holds for any $x \in E$, $s \ge 0$, $i \in \mathbb{N}_n^0$ and $p, q \in \mathbb{N}_0$. Using the substitution $z = \lambda^{\zeta}$, Theorem 1.5.1 and the Cauchy formula, we get that

(178)
$$\int_{\Gamma_l} f(\lambda^{\zeta}) \lambda^{\zeta-1} E_{\zeta}(s^{\zeta} \lambda^{\zeta}) d\lambda = 0, \quad s \ge 0.$$

By (173)–(178), it readily follows that the mapping $s \mapsto u(s) = S_f(s)x$, $s \ge 0$ is a strong solution of the problem [(168)–(169)] with the prescribed set of initial values. If $CA_i \subseteq A_iC$ for all $i \in \mathbb{N}_n^0$, then the uniqueness of strong solutions of associated integral equation (170) is an immediate consequence of [**307**, Theorem 2.2], finishing in a routine manner the proof of (i). The existence of number $h_0 > 0$ in (ii) and the proof of inequality (174) follows from (165), (172) and a simple calculus.

For the sequel, we need some preliminaries concerning abstract Beurling spaces. We define the abstract Beurling space of (M_p) class associated to A, $E^{(M_p)}(A)$ for short, as in the Banach space case (cf. [109,291] for more details): Put $D_{\infty}(A) =: \bigcap_{n \in \mathbb{N}} D(A^n)$,

$$E^{(M_p)}(A) := \operatorname{projlim}_{h \to +\infty} E_h^{(M_p)}(A),$$

where for each h > 0,

$$E_h^{(M_p)}(A) := \left\{ x \in D_{\infty}(A) : \|x\|_{h,q}^{(M_p)} = \sup_{p \in \mathbb{N}_0} \frac{h^p q(A^p x)}{M_p} < \infty \text{ for all } q \in \mathfrak{B} \right\}.$$

In this place, it is worth noting that for each h > 0 the calibration $(\|\cdot\|_{h,q}^{(M_p)})_{q\in \circledast}$ induces a Hausdorff locally convex space topology on $E_h^{(M_p)}(A)$, as well as that $E_{h'}^{(M_p)}(A) \subseteq E_h^{(M_p)}(A)$ provided $0 < h < h' < \infty$, and that the spaces $E_h^{(M_p)}(A)$ and $E^{(M_p)}(A)$ are continuously embedded in E; cf. [291]. Following the ideas of R. Beals [65], we define the space $E^{\langle M_p \rangle}(A)$ as the inductive limit of spaces $E_h^{(M_p)}(A)$ as $h \to 0+$; that is

$$E^{\langle M_p \rangle}(A) := \operatorname{indlim}_{h \to 0+} E_h^{(M_p)}(A).$$

In the following rather long remark, we will collect a great number of various thoughts and insights about Theorem 2.5.1.

REMARK 2.5.2. (i) In the case that $\zeta < 1$, Theorem 2.5.1 seems to be new and not considered elsewhere (even supposing that B = I). If $\zeta = 1$, then there exist two possibilities: n = 1 and n > 1. If n = 1 and $\zeta = 1$, then the assertion of Theorem 2.5.1 seems to be new in the case that E is not a Banach space and $B \neq I$, or that $B \neq I$ and $C \neq I$ (cf. [296, Theorem 3.16, Example 4.5] for some results in locally convex spaces, with B = I). If n = 1, $\zeta = 1$ and E is a Banach space, then it is worth noting that A. Favini [185] was the first who considered R. Beals's type regularization process [64, 65] for seeking of solutions of degenerate equations of first order, provided in addition that C = I (cf. also [199, Section 5.4] for the case $B \neq I$, as well as [132, 243, 244, Section XXIII], [291, Section 1.4], [292, 303, 364, 502, Section 2.9] and [541, Section 4.4] for more details concerning the case B = I). If n > 1 and $\zeta = 1$, then the assertion of Theorem 2.5.1 seems to be considered only in the case that C = I, $\xi = 0$, $p_i = 0$ for all $i \in \mathbb{N}_n^0$, and E is a Banach space (cf. [185, Application 2, Assumption H.10] and compare with our assumptions made in (i) of Theorem 2.5.1). There is no need to say that the usual converting of higher-order (degenerate) differential equations into first order matrix (degenerate) differential equations, used in numerous papers on higher-order abstract differential equations and, in particular, in the above-mentioned Application 2 of [185], cannot offer significant help in the analysis of problem [(168)–(169)], in general.

(ii) Let $v \in \mathbb{Z}$, let $f \in \mathcal{A}$, and let an element $x \in E$ satisfy the requirements of (i). Define

$$S_{f,v}(s)x := \frac{\zeta}{2\pi i} \int_{\Gamma_l} f(\lambda^{\zeta}) \lambda^{\zeta-1} \lambda^{v\zeta} E_{\zeta}(s^{\zeta}\lambda^{\zeta}) P_{\lambda}^{-1} Cx \, d\lambda, \quad s \ge 0, \ x \in E.$$

Then the mapping $s \mapsto S_{f,v}(s)x$, $s \ge 0$ is likewise a strong solution of problem (168), with the initial values (169) endowed similarly as in the formulation of (i).

- (iii) In the formulation of [303, Theorem 2.1], it has been additionally assumed that the sequence (M_p) satisfies the condition (M.2). The proof of Theorem 2.5.1 shows that we can completely neglect this condition from our analysis.
- (iv) (cf. also Remark 2.10.25(iii)) The worth noticing is that the term $\mathbf{D}_{s}^{\nu_{1}+\nu_{2}}u(s)$ need not be defined for some functions $s \mapsto u(s), s \ge 0$ for which the term $\mathbf{D}_{s}^{\nu_{1}}\mathbf{D}_{s}^{\nu_{2}}u(s)$ is defined. Consider, by way of illustration, the case $\nu_{1} = \nu_{2} = 1/2, \lambda > 0$ and $u(s) = E_{1/2}(\lambda^{1/2}s^{1/2}), s \ge 0$; then [**61**, (1.25)] implies that $\mathbf{D}_{s}^{\nu_{1}}\mathbf{D}_{s}^{\nu_{2}}u(s) = \lambda E_{1/2}(\lambda^{1/2}s^{1/2}), s \ge 0$. On the other hand, $\mathbf{D}_{s}^{1}u(s)$ is not defined for $s \ge 0$ because the function $s \mapsto u(s), s \ge 0$ is not continuously differentiable for $s \ge 0$; even if we accept a slightly weaker definition of Caputo fractional derivatives from [**61**], when $\mathbf{D}_{s}^{1}E_{1/2}(\lambda^{1/2}s^{1/2})$ exists and equals to $\sum_{k=1}^{\infty} \frac{\lambda^{k/2}s^{(k/2)-1}}{\Gamma(k/2)}$ for s > 0, the equality

$$\mathbf{D}_{s}^{1/2}\mathbf{D}_{s}^{1/2}E_{1/2}(\lambda^{1/2}s^{1/2}) = \mathbf{D}_{s}^{1}E_{1/2}(\lambda^{1/2}s^{1/2}), \quad s > 0$$

does not hold for any $\lambda > 0$ because $\lambda E_{1/2}(\lambda^{1/2}s^{1/2}) \sim \lambda$ as $s \to 0+$ while $\sum_{k=1}^{\infty} \frac{\lambda^{k/2}s^{(k/2)-1}}{\Gamma(k/2)} \sim (\frac{\lambda}{\pi s})^{1/2}$ as $s \to 0+$ (cf. [**302**], Remark 2.10.25(iv) and the equation (241) below). Hence, we will have to make a strict distinction between the operator $(\mathbf{D}_s^{\zeta})^p$ and the operator $\mathbf{D}_s^{\zeta p}$. As explained in Remark 2.10.25(iii), the method proposed in Theorem 2.5.1 cannot be used for proving the existence of strong solutions of (non-degenerate) problem

$$B\mathbf{D}_s^{\alpha_n}u(s) + \sum_{i=0}^{n-1} A_i \mathbf{D}_s^{\alpha_i}u(s) = 0, \quad s \ge 0,$$

provided that n > 1 and there exists an index $i \in \mathbb{N}_n^0$ such that the order α_i of the Caputo fractional derivative $\mathbf{D}_s^{\alpha_i} u(s)$ does not belong to \mathbb{N}_0 . Here, $0 = \alpha_0 < \alpha_1 < \cdots < \alpha_n$.

- (v) It can be simply verified that $\int_0^s g_{\zeta}(s-r)(-A)S_f(r)x\,dr = BS_f(s)x BS_f(0)x$, provided that $n = 1, s \ge 0, f \in \mathcal{A}$ and $x \in E$ satisfies the requirements of Theorem 2.5.1(i).
- (vi) It is well known that the notion of an abstract Beurling space plays an important role in the theory of ultradistribution semigroups in Banach spaces (cf. Theorem 2.5.1(i) with n = 1 and B = I). Unfortunately, it is very problematic to introduce a similar concept for degenerate differential equations of first order, especially in the case that the operator B is not injective. For the purpose of illustration of Theorem 2.5.1(i), we will present two examples in which we use the abstract Beurling spaces:
 - (vi.1) Suppose that $n = 1, x \in D_{\infty}(B)$, the element $B^{p}x$ satisfies the requirements of Theorem 2.5.1(i) for all $p \in \mathbb{N}_{0}$, and $B(zB+A)^{-1}CB^{p}x = (zB+A)^{-1}CB^{p+1}x, p \in \mathbb{N}_{0}, z \in \Lambda_{\alpha,\beta,l}^{\zeta}$. Then $A^{p}S_{f}(s)x = \frac{(-1)^{p}\zeta}{2\pi i}\int_{\Gamma_{l}} f(\lambda^{\zeta})\lambda^{\zeta-1}\lambda^{p\zeta}E_{\zeta}(s^{\zeta}\lambda^{\zeta})P_{\lambda}^{-1}CB^{p}x\,d\lambda, s \ge 0,$ $p \in \mathbb{N}, f \in \mathcal{A}$. This, in turn, implies $\bigcup_{s \ge 0, f \in \mathcal{A}} \{S_{f}(s)x\} \subseteq E^{\langle p^{p\zeta/b} \rangle}(A),$ provided that the orbit $\{B^{p}x : p \in \mathbb{N}_{0}\}$ is bounded, and $\bigcup_{s \ge 0, f \in \mathcal{A}} \{S_{f}(s)x\} \subseteq E^{\langle p^{2p\zeta/b} \rangle}(A),$ provided that $Bx \in E^{\langle p^{p\zeta/b} \rangle}(A).$
 - (vi.2) (cf. also Remark 2.5.5) Suppose that n = 1, B is injective and an
 - element $x \in E$ satisfies the requirements of Theorem 2.5.1(i). Then B^{-1} is closed and we can inductively prove that $(B^{-1}A)^p S_f(s)x =$ $(-1)^p (\mathbf{D}_s^{\zeta})^p S_f(s)x, s \ge 0, p \in \mathbb{N}, f \in \mathcal{A}$. Taking into account (174), the above implies $\bigcup_{s \ge 0, f \in \mathcal{A}} \{S_f(s)x\} \subseteq E^{\langle p^{p\zeta/b} \rangle}(B^{-1}A)$.
- (vii) Let $f \in \mathcal{A}$, let $\varepsilon > 0$, and let $g: \mathbb{C} \smallsetminus \Lambda_{\alpha,\beta,l+\varepsilon}^{\zeta} \to \mathbb{C}$ be continuous in D(g)and analytic in $\operatorname{int}(D(g))$. Suppose, further, that there exist constants $a'_1 > 0$ and $a'_2 > \xi$ such that (166) holds with f = g, $a_1 = a'_1$, $a_2 = a'_2$, $\lambda \in D(g)$, as well as that n = 1 and the family $\{(1+|\lambda|)^{-\nu}e^{-\xi|\lambda|^b}BP_{\lambda}^{-1}C:$ $\lambda \in \Lambda_{\alpha,\beta,l}\} \subseteq L(E)$ is both equicontinuous and strongly continuous. Let

(179)
$$CB(zB+A)^{-1}C = B(zB+A)^{-1}C^2, \quad z \in \Lambda_{\alpha,\beta,l}^{\zeta},$$

and let $\Gamma_l^{\zeta}(\Gamma_{l,\varepsilon}^{\zeta})$ denote the upwards oriented boundary of $\Lambda_{\alpha,\beta,l}^{\zeta}(\Lambda_{\alpha,\beta,l+\varepsilon}^{\zeta})$. Then, for every $z, z' \in \Lambda_{\alpha,\beta,l}^{\zeta}$ and $x \in E$, the resolvent equation

$$(zB+A)^{-1}C^{2}x - (z'B+A)^{-1}C^{2}x = (z'-z)(zB+A)^{-1}CB(z'B+A)^{-1}Cx,$$

holds, which clearly implies that the mapping $z \mapsto B(zB + A)^{-1}C^2 x$, $z \in int(\Lambda_{\alpha,\beta,l}^{\zeta})$ is analytic $(x \in E)$. Using (179) and the consideration from [138, Remark 2.7], the above implies that the mapping $z \mapsto B(zB + A)^{-1}Cx$, $z \in int(\Lambda_{\alpha,\beta,l}^{\zeta})$ is analytic, as well $(x \in E)$. Applying the substitution $z = \lambda^{\zeta}$ and the Cauchy formula, we then get that $BS_g(0)x = (2\pi i)^{-1} \int_{\Gamma_{l,\varepsilon}^{\zeta}} g(z)B(zB + A)^{-1}Cx \, dz$, $x \in E$. Proceeding as in the proof

179

of [65, Lemma 4.2], it readily follows that

$$S_f(0)BS_g(0)x = S_{fg}(0)Cx, \quad x \in E.$$

If $\xi = 0$, $f(\lambda) = f_t(\lambda)$ and $g(\lambda) = f_s(\lambda)$, with $t, s \in \Sigma_{\theta}$, the above means that T(t)BT(s) = T(t+s)C. In the notion of Section 2.7 (cf. Definition 2.7.1), we have that $(T(t))_{t \in \Sigma_{\theta}}$ is an analytic (B, C)-regularized semigroup of growth order $(\nu + 1)\zeta/b$.

- (viii) If B = I, n = 1, $P_{\lambda}^{-1}C$ exists and is polynomially bounded on the region $\Lambda_{\alpha,\beta,l}$ (with the meaning clear), then it might be surprising that we must impose the condition (167) in order to ensure the existence of strong solutions of problem (DFP)'_R with the initial value $x \neq 0$. If we replace the condition (167) with the condition that $\frac{1}{2-\zeta} \leq \frac{b}{\zeta}$ (which clearly implies $\zeta < 1$ and the triviality of class \mathcal{A}), and accept all the remaining assumptions from the formulation of this theorem, with B = I, n = 1 and $\xi = 0$, then it is not clear whether there exist a Hilbert space (Banach space, sequentially complete locally convex space) E and a closed linear operator A acting on E such that the problem (DFP)'_R has no local strong solutions unless x = 0 (cf. [64, Theorem 2, Theorem 2'] for more details concerning the case $\zeta = 1$). This is an open problem we would like to address to our readers.
 - (ix) If the assumptions of Theorem 2.5.1 hold with the region $\Lambda_{\alpha,\beta,l}$ replaced by the right half-plane $RHP_{\bar{\omega}} \equiv \{z \in \mathbb{C} : \operatorname{Re} z > \bar{\omega}\}$ (and with the set Ω replaced by $\mathbb{C} \setminus (RHP_{\bar{\omega}})^{\zeta}$, then for each $p \in \mathbb{N}_0$ and $f \in \mathcal{A}$ the operator family $\{e^{-\bar{\omega}s}(\mathbf{D}_s^{\zeta})^p S_f(s) : s \ge 0\}$ is equicontinuous $(\bar{\omega} > 0);$ cf. [29] for some interesting examples given in the case that $\zeta = 1, \xi = 0$ and C = I. It is also worth noting that we can consider, instead of the region $\Lambda_{\alpha,\beta,l}$ considered above, a region of the form $\Omega(\omega) = \{\lambda \in$ \mathbb{C} : Re $\lambda \ge \max(x_0, \omega(|\operatorname{Im} \lambda|))$, where $x_0 > 0, \omega \colon [0, \infty) \to [0, \infty)$ is a continuous, concave, increasing function satisfying $\lim_{t\to\infty} \omega(t) = \infty$, $\lim_{t\to\infty}(\omega(t)/t) = 0$ and $\int_1^{\infty}(\omega(t)/t^2)dt < \infty$ (cf. [64, 65, 111, 291, 303] and [296, Example 4.5]), or the exponential region $E(a,b) = \{\lambda \in \mathbb{C} :$ $\operatorname{Re} \lambda \ge b$, $|\operatorname{Im} \lambda| \le e^{a \operatorname{Re} \lambda}$ (a, b > 0), introduced for the first time by W. Arendt, O. El-Mennaoui and V. Keyantuo in [28] (cf. also C. Foiaş [220] for a very similar notion of the logarithmic region $\Lambda(\alpha, \beta, \omega)$ which can also be used here). We will not go into further details concerning these questions.
 - (x) Suppose $x \in \bigcap_{v=0}^{n} D(A_v), i \in \mathbb{N}_n^0, j \in \mathbb{N}_0, (f_{\varepsilon}(\lambda))_{\varepsilon > 0}$ is a net of functions from \mathcal{A} and $CA_v \subseteq A_vC, v \in \mathbb{N}_n^0$. Denote $u_{i,\varepsilon}^j := ((\mathbf{D}_s^{\zeta})^j S_{f_{\varepsilon}}(s)A_ix)_{s=0}$ $(\varepsilon > 0)$. Then the following equality holds: $P_{\lambda}^{-1}CA_ix = \lambda^{-(p_i+q_i)\zeta}[Cx - \sum_{v \in \mathbb{N}_n^0 \smallsetminus \{i\}} \lambda^{(p_v+q_v)\zeta} P_{\lambda}^{-1}CA_vx], \lambda \in \Lambda_{\alpha,\beta,l}$, which implies that

$$u_{i,\varepsilon}^{j} = \frac{1}{2\pi i} \int_{\Gamma_{l}^{\zeta}} f_{\varepsilon}(\lambda) \lambda^{j-(p_{i}+q_{i})} \bigg[Cx - \sum_{v \in \mathbb{N}_{n}^{0} \smallsetminus \{i\}} \lambda^{(p_{v}+q_{v})} P_{\lambda^{1/\zeta}}^{-1} CA_{v} x \bigg] d\lambda.$$

If we impose some additional conditions on the net $(f_{\varepsilon}(\lambda))_{\varepsilon>0}$ (for example, the condition that $f_{\varepsilon}(0) \neq 0$, $\varepsilon > 0$, as well as $f_{\varepsilon}^{(p_i+q_i-j-1)}(0) \rightarrow z_0^{i,j}$ as $\varepsilon \to 0$, provided $p_i + q_i - j - 1 \ge 0$, and the limit equality $f_{\varepsilon}(\lambda) \to 1$ as $\varepsilon \to 0$ ($\lambda \in \Gamma_l^{\zeta}$), uniformly on compacts of Γ_l^{ζ} , at least) and if suppose that the operator family $\{(1 + |\lambda|)^{-\nu'} P_{\lambda}^{-1}C : \lambda \in \Lambda_{\alpha,\beta,l}\} \subseteq L(E)$ is both equicontinuous and strongly continuous for a sufficiently large negative number $\nu' < 0$ (cf. also [541, Theorem 4.2, p. 168]), then we may apply the dominated convergence theorem and the residue theorem in order to see that $\lim_{\varepsilon \to 0} u_{i,\varepsilon}^{j}$ equals0, if $j \ge p_i + q_i$, and $[(p_i + q_i - j - 1)!]^{-1} z_0^{i,j} Cx$, otherwise. If we use the net of functions of form $f_{\varepsilon}(\lambda) = e^{-\varepsilon(-\lambda+\omega)^{b/\zeta}}$ ($\varepsilon > 0$), then we have that $z_0^{i,j} = 1$ if $p_i + q_i - j - 1 = 0$, and $z_0^{i,j} = 0$ if $p_i + q_i - j - 1 > 0$ [502,541]. Using this idea, we can prove the following (cf. also Theorem 2.6.2 below): Suppose that $x_v \in \bigcap_{p=0}^n D(A_p)$ for all $v \in \mathbb{N}_q^0$, and the super the set of the proventies of the

(180)
$$-\infty < \nu' < \min_{v \in \mathbb{N}^0_{q_n-1}} \left[-(q_n - 1 - v + \max\{q_j : j \in \mathbb{N}^0_n \smallsetminus S_v\}) \right]$$

and the operator family $\{(1 + |\lambda|)^{-\nu'} P_{\lambda}^{-1}C : \lambda \in \Lambda_{\alpha,\beta,l}\} \subseteq L(E)$ is both equicontinuous and strongly continuous. Assume, further, that for each $x \in E$ and $i \in \mathbb{N}_{n-1}^{0}$ the mapping $\lambda \mapsto A_i P_{\lambda^{1/\zeta}}^{-1}Cx$, $\lambda \in \Lambda_{\alpha,\beta,l}^{\zeta}$ is continuous and there exists $v_i \in \mathbb{N}$ such that the operator family $\{(1 + |\lambda|)^{-v_i}A_iP_{\lambda^{1/\zeta}}^{-1}C : \lambda \in \Lambda_{\alpha,\beta,l}^{\zeta}\} \subseteq L(E)$ is equicontinuous. Let the function $f_{\varepsilon}(\lambda)$ be chosen as above, and let $p_v = 0, v \in \mathbb{N}_n^0$ (with the exception of problem $(DFP)_R$, the analysis becomes very difficult in the case that there exists $v_0 \in \mathbb{N}_n^0$ such that $p_{v_0} > 0$). Then the function

$$s \mapsto u_{\varepsilon}(s) := \sum_{v=0}^{q_n-1} \sum_{j \in S_v} \frac{1}{2\pi i} \int_{\Gamma_{\ell}^{\zeta}} e^{-\varepsilon(-\lambda+\omega)^{b/\zeta}} \\ \times E_{\zeta}(s^{\zeta}\lambda) \lambda^{q_j-1-v} P_{\lambda^{1/\zeta}}^{-1} CA_j x_v \, d\lambda, \quad s \ge 0$$

is a strong solution of problem (168) with the initial values $(u_0^{\varepsilon}, \ldots, u_{q_n-1}^{\varepsilon})$, converging to $(Cx_0, \ldots, Cx_{q_n-1})$ as $\varepsilon \to 0+$. Hence, the set \mathfrak{W} consisting of all initial values $(y_0, \ldots, y_{q_n-1}) \in E^{q_n}$ subjected to some strong solution $s \mapsto u(s), s \ge 0$ of problem (168) is dense in $(C(\bigcap_{v=0}^n D(A_v)))^{q_n}$ (cf. Example 2.5.8 below for an interesting application of this result, with C not being the identity operator). Generally, it is very difficult to say anything else about the set \mathfrak{W} in the case that n > 1.

(xi) Following the method employed in the proof of Theorem 2.5.1, one can extend the assertions of [298, Theorem 2.1, Theorem 2.2] to abstract degenerate (multi-term) fractional differential equations, thus proving some results on the *C*-wellposedness of problem [(168)–(169)] in the case that $\zeta > 2$ [298, Theorem 2.1] and $2 \ge \zeta > 1$ [298, Theorem 2.2]. Consider, for example, the case $2 \ge \zeta > 1$. Let $\vartheta \in (\pi(2 - \zeta)/2, \pi/2)$, let $b \in (1/\zeta, \pi/(2(\pi - \vartheta)))$ and let $z \in \Sigma_{\vartheta'}$, where $\vartheta' := \arctan(\cos(b(\pi - \vartheta)))$. If there exist $d \in (0, 1]$ and $\nu > -1$ such that the operator family $\{(1 + |\lambda|)^{-\nu}P_{\lambda}^{-1}C : \lambda \in \Sigma_{\vartheta/\zeta} \cup B_d\} \subseteq L(E)$ is both equicontinuous and strongly continuous (for the sake of simplicity, we shall only deal with the case in which $\xi = 0$), then for each number $s \ge 0$ we can define the bounded linear operator S(s) by

$$S(s)x := \frac{1}{2\pi i} \int_{\Gamma_{\zeta,d}} e^{-z(-\lambda)^b} E_{\zeta}(s^{\zeta}\lambda) P_{\lambda^{1/\zeta}}^{-1} Cx \, d\lambda, \quad x \in E, \ s \ge 0,$$

where $c \in (0, \vartheta)$ is chosen so that $b \in (1/\zeta, \pi/(2(\pi - c)))$ and $\vartheta < \arctan(\cos(b(\pi - c)))$ (cf. (171) and apply the substitution $\lambda \mapsto \lambda^{\zeta}$). Suppose, further, that there exists $i \in \mathbb{N}_n^0$ such that the mappings $\lambda \mapsto$ $A_j P_{\lambda}^{-1} Cx, \ \lambda \in \Sigma_{\vartheta/\zeta} \cup B_d$ are continuous for some $x \in E \ (j \in \mathbb{N}_n^0 \setminus \{i\})$ and for each seminorm $p \in \circledast$ the set $\{(1 + |\lambda|)^{-\nu} p(A_j P_{\lambda}^{-1} Cx) : \lambda \in \Sigma_{\vartheta/\zeta} \cup B_d, j \in \mathbb{N}_n^0 \setminus \{i\}\}$ is bounded. Then the final conclusions stated in Theorem 2.5.1 continue to hold after some obvious modifications. In the situation of [298, Theorem 2.1] (the case $\zeta > 2$), which is very specific, we can assume that the operators $P_{\lambda}^{-1}C$ exist on a certain region of the complex plane which does not contain any acute angle. The interested reader should carry out details concerning the transmitting our previous results and comments from the items (i)-(x) of this remark to the case in which $\zeta > 1$. The method proposed in [298, 541, Section 4.4, pp. 167–175] as well as in parts (x)-(xi) of this remark can serve one to prove some results on the existence of entire and analytical solutions of degenerate (multiterm) fractional differential equations and their systems. The reader may consult [205, 311], Theorem 2.3.20 and Theorem 2.6.2 below for similar results in this direction.

The proof of following extension of [303, Corollary 2.1] is omitted.

THEOREM 2.5.3. Suppose that $0 < c < b < \zeta \leq 1$, $\sigma > 0$, $\nu > -1$, $\xi \ge 0$, $\varsigma > 0$ and (167) holds. Denote

$$\Pi_{c,\sigma,\varsigma} := \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda \ge \sigma | \operatorname{Im} \lambda|^c + \varsigma\},\$$
$$\Pi_{c,\sigma,\varsigma}^{\zeta} := \{\lambda^{\zeta} : \lambda \in \Pi_{c,\sigma,\varsigma}\} \text{ and } \Omega' := \mathbb{C} \smallsetminus \Pi_{c,\sigma,\varsigma}^{\zeta}\}$$

Let $f: \overline{\Omega'} \to \mathbb{C}$ be a continuous function that is analytic in Ω' and satisfies that there exist constants $a_1 > 0$ and $a_2 > \xi$ such that $|f(\lambda)| \leq a_1 e^{-a_2|\lambda|^{b/\zeta}}$, $\lambda \in \overline{\Omega'}$. Suppose, further, that the operator P_{λ} is injective for all $\lambda \in \Pi_{c,\sigma,\varsigma}$, as well as that $P_{\lambda}^{-1}C \in L(E)$, $\lambda \in \Pi_{c,\sigma,\varsigma}$, the mapping $\lambda \mapsto P_{\lambda}^{-1}Cx$, $\lambda \in \Pi_{c,\sigma,\varsigma}$ is continuous for every fixed element $x \in E$, and the operator family

$$\{(1+|\lambda|)^{-\nu}e^{-\xi|\lambda|^b}P_{\lambda}^{-1}C:\lambda\in\Pi_{c,\sigma,\varsigma}\}\subseteq L(E)$$

is equicontinuous. Set

$$T_f(s)x := \frac{\zeta}{2\pi i} \int_{\Gamma_c} f(\lambda^{\zeta}) \lambda^{\zeta-1} E_{\zeta}(s^{\zeta}\lambda^{\zeta}) P_{\lambda}^{-1} Cx \, d\lambda, \quad s \ge 0, \ x \in E,$$

where Γ_c denotes the upwards oriented boundary of $\Pi_{c,\sigma,\varsigma}$. Then $(T_f(s))_{s \ge 0} \subseteq L(E)$ is strongly continuous, the mapping $s \mapsto T_f(s) \in L(E)$, $s \ge 0$ ($s \mapsto T_f(s) \in L(E)$, s > 0) is infinitely differentiable provided $\zeta = 1$ ($\zeta \in (0,1)$) and, for every $p \in \mathbb{N}_0$, the mapping $s \mapsto (\mathbf{D}_s^{\zeta})^p T_f(s) \in L(E)$, $s \ge 0$ is well-defined. Furthermore, (172) holds with $(S_f(s))_{s \ge 0}$ and Γ_l replaced respectively by $(T_f(s))_{s \ge 0}$ and Γ_c , and we have the following:

- (i) Suppose that there exists i ∈ N⁰_n such that the mappings λ → A_jP⁻¹_λCx, λ ∈ Π_{c,σ,ζ} are continuous for some x ∈ E (j ∈ N⁰_n \ {i}) and for each seminorm **p** ∈ ⊛ the set {(1 + |λ|)^{-ν}e^{-ξ|λ|^b}**p**(A_jP⁻¹_λCx) : λ ∈ Π_{c,σ,ζ}, j ∈ N⁰_n \ {i}} is bounded. Then (173) holds with (S_f(s))_{s≥0} and Γ_l replaced respectively by (T_f(s))_{s≥0} and Γ_c, the mapping s → u(s) := T_f(s)x, s ≥ 0 is a strong solution of the problem [(168)–169], with the initial value u_j obtained by plugging p = j and s = 0 into the right-hand side of (172), for j ∈ N⁰<sub>max{q_i-1:i∈S_l}, and the initial value u_{i,j} obtained by plugging p = j, q = q_i and s = 0 into the right-hand side of (173), for i ∈ S_r and j ∈ N⁰_{p_i-1} (with the obvious replacements described above). If CA_i ⊆ A_iC for all i ∈ N⁰_n, then there exists at most one strong solution of problem [(168)–(169)].
 </sub>
- (ii) Suppose that $q \in \circledast$, \mathbb{B} is a bounded subset of E and K is a compact subset of $[0, \infty)$. Then there exists $h_0 > 0$ such that (174) holds.

Assume now that $\alpha > 0$, $\beta > 0$, $l \ge 1$, $0 < \zeta \le 1$ and, as before, that (M_p) is a sequence of positive real numbers such that $M_0 = 1$ and the condition (M.1) is fulfilled for (M_p) . Recall that $\Omega = \mathbb{C} \smallsetminus \Lambda_{\alpha,\beta,l}^{\zeta}$. If $g: [0,\infty) \to [0,\infty)$ is a monotonically increasing, continuous function satisfying

(181)
$$\lim_{t \to +\infty} (1+t)^v e^{\sigma M(st) - g(t)} = 0, \quad v \in \mathbb{N}, \ s \ge 0, \ \sigma > 0,$$

then we denote by \mathcal{A}_g the class consisting of all continuous functions $f: \overline{\Omega} \to \mathbb{C}$ that are analytic in Ω and satisfy that $|f(z)| \leq \text{Const} \cdot e^{-g(|z|^{1/\zeta})}, z \in \overline{\Omega}$. The main purpose of following theorem is to consider a non-Gevrey analogue of Theorem 2.5.1 and Theorem 2.5.3 (the proof is very similar to that of Theorem 2.5.1 and we shall skip details for the sake of brevity):

THEOREM 2.5.4. Suppose that (M_p) satisfies (M.1), as well as that there exists a monotonically increasing, continuous function $g: [0, \infty) \to [0, \infty)$ satisfying that (181) holds and the class \mathcal{A}_g is non-trivial. Let $0 < \zeta \leq 1$, $\nu > -1$, $\xi \geq 0$, $\alpha > 0$, $\beta > 0$ and $l \geq 1$. Suppose, further, that the operator P_λ is injective for all $\lambda \in \Lambda_{\alpha,\beta,l}$, as well as that $P_\lambda^{-1}C \in L(E)$, $\lambda \in \Lambda_{\alpha,\beta,l}$, the mapping $\lambda \mapsto P_\lambda^{-1}Cx$, $\lambda \in \Lambda_{\alpha,\beta,l}$ is continuous for every fixed element $x \in E$, and the operator family

$$\{(1+|\lambda|)^{-\nu}e^{-M(\xi\lambda)}P_{\lambda}^{-1}C:\lambda\in\Lambda_{\alpha,\beta,l}\}\subseteq L(E)$$

is equicontinuous. Set, for every function $f \in \mathcal{A}_g$, the bounded linear operator $S_f(s)$ $(s \ge 0)$ through (171). Then $(S_f(s))_{s\ge 0} \subseteq L(E)$ is strongly continuous, the mapping $s \mapsto S_f(s) \in L(E)$, $s \ge 0$ $(s \mapsto S_f(s) \in L(E)$, s > 0) is infinitely differentiable provided $\zeta = 1$, $f \in \mathcal{A}_g$ $(\zeta \in (0, 1), f \in \mathcal{A}_g)$ and, for every $p \in \mathbb{N}_0$ and

183

 $f \in \mathcal{A}_g$, the mapping $s \mapsto (\mathbf{D}_s^{\zeta})^p S_f(s) \in L(E)$, $s \ge 0$ is well-defined. Furthermore, (172) and (i)–(ii) hold, where:

- (i) Suppose that there exists i ∈ N_n⁰ such that the mappings λ → A_jP_λ⁻¹Cx, λ ∈ Λ_{α,β,l} are continuous for some x ∈ E (j ∈ N_n⁰ \ {i}) and for each seminorm **p** ∈ ⊕ the set {(1+|λ|)^{-ν}e^{-M(ξλ)}**p**(A_jP_λ⁻¹Cx) : λ ∈ Λ_{α,β,l}, j ∈ N_n⁰ \ {i}} is bounded. Then (173) holds for any x ∈ E, s ≥ 0, i ∈ N_n⁰ and p, q ∈ N₀. Moreover, the mapping s → u(s) := S_f(s)x, s ≥ 0 is a strong solution of the problem [(168)–(169)], with the initial value u_j obtained by plugging p = j and s = 0 into the right-hand side of (172), for j ∈ N<sub>max{q_i-1:i∈S_l}, and the initial value u_{i,j} obtained by plugging p = j, q = q_i and s = 0 into the right-hand side of (173), for i ∈ S_r and j ∈ N_{p_i-1}⁰ (f ∈ A). If CA_i ⊆ A_iC for all i ∈ N_n⁰, then there exists at most one strong solution of problem [(168)–(169)].
 </sub>
- (ii) Let (N_p)_{p∈N₀} be a sequence of positive real numbers satisfying N₀ = 1, (M.1) and the property that for each v ∈ N, s ≥ 0 and σ > 0 there exists h > 0 such that lim_{t→+∞}(1 + t)^ve^{σM(st)+N(ht^ζ)-g(t)} = 0. Suppose that f ∈ A, q ∈ ⊛, B is a bounded subset of E and K is a compact subset of [0,∞). Then there exists h₀ > 0 such that (174) holds with the sequence (p^{pζ/b}) replaced by (N_p) therein.

It is worth noting that we have analyzed a slightly different growth rate of $P_{\lambda}^{-1}C$ in Theorem 2.5.1 (Theorem 2.5.3), and that one has to assume that for each $v \in \mathbb{N}$, $s \ge 0$ and $\sigma > 0$ there exists h > 0 such that $\lim_{t \to +\infty} (1 + t)^v e^{\sigma M(st) + \xi |\lambda|^b + N(ht^{\zeta}) - g(t)} = 0$ ($\lim_{t \to +\infty} (1 + t)^v e^{st^c + \xi |\lambda|^b + N(ht^{\zeta}) - g(t)} = 0$) in order to deduce Theorem 2.5.1 (Theorem 2.5.3) from Theorem 2.5.4. Also, it is worth noting that Theorem 2.5.4 is closely linked with the assertions of [303, Theorem 3.3] and [354, Theorem 7.6], where we have studied the regularization of ultradistribution semigroups in Banach spaces $(B = I, n = 1, \zeta = 1, \xi \ge 0, -A$ generates an ultradistribution semigroup of (M_p) -class; cf. [291] and [354] for the notion, as well as [109,111,289,291,330,349,355,363] and [424] for more details concerning ultradistribution semigroups). If the corresponding sequence (M_p) satisfies the conditions (M.1), (M.2) and (M.3), then we have proved in the afore-mentioned theorems that there exist two functions, $q(\cdot)$ and $f \in \mathcal{A}_q$, such that the operator -A generates a global locally equicontinuous C-regularized semigroup $(S_f(s))_{s\geq 0}$, with $C = S_f(0)$ being injective, satisfying additionally that the mapping $s \mapsto S_f(s) \in L(E)$, $s \ge 0$ is infinitely differentiable and $E^{(M_p)}(A) \subset C(D_{\infty}(A))$. The proof of this fact is based upon the existence of a sequence (N_p) of positive real numbers satisfying $N_0 = 1$, (M.1), (M.2), (M.3) and $N_p \prec M_p$ (cf. [303, Lemma 3.2]), and by putting $f(\cdot) = 1/\omega_{l',(N_n)}(-\cdot)$ $(l' \in \mathbb{N}$ sufficiently large) after that, where $\omega_{l',(N_p)}(\lambda) := \prod_{p=1}^{\infty} (1 + \frac{l' \lambda N_{p-1}}{N_p}), \ \lambda \in \mathbb{C} \ (l' > 0).$

REMARK 2.5.5. The comments from Remark 2.5.2 can be reformulated in the case that the assumptions of Theorem 2.5.3 or Theorem 2.5.4 below hold. Here we would like to point out some facts closely linked with Remark 2.5.2[(ii),(vi.1)] and [65, Lemma 1, Lemma 4]. Consider the situation of Theorem 2.5.3 with $\xi = 0$ and

n = 1 (the final conclusions continue to hold in the case of examination of Theorem 2.5.1; after the replacement of region $\Pi_{c,\sigma,\varsigma}^{\zeta}$ with $\Lambda_{\alpha,\beta,l}^{\zeta}$, one just has to make some obvious terminological changes). Suppose, additionally, that $b/\zeta < 1/(2-\zeta)$, $\omega > \varsigma^{\zeta}$, $CA \subseteq AC$, $CB \subseteq BC$, B is injective,

$$(182) \ B^{-1}A(zB+A)^{-1}Cx = (zB+A)^{-1}CB^{-1}Ax, \quad x \in D_{\infty}(B^{-1}A), \ z \in \Pi_{c,\sigma,\varsigma}^{\zeta},$$

the family $\{(1+|z|)^{-\nu}B(zB+A)^{-1}C: z \in \Pi_{c,\sigma,\varsigma}^{\zeta}\} \subseteq L(E)$ is both equicontinuous and strongly continuous, $y \in D(B)$ satisfies that $Cy \in D(A)$, BACy = ACBy and there exists $h_0 > 0$ such that the set $\{h_0^p p^{(-p\zeta)/b}(B^{-1}A)^p By: p \in \mathbb{N}_0\}$ is bounded. Then it is readily seen that $C(zB+A)^{-1}C = (zB+A)^{-1}C^2, z \in \Pi_{c,\sigma,\varsigma}^{\zeta}$ and the mapping $z \mapsto (zB+A)^{-1}Cx, z \in \operatorname{int}(\Pi_{c,\sigma,\varsigma}^{\zeta})$ is analytic $(x \in E)$; cf. Remark 2.5.2(vii). Using the Cauchy formula and the foregoing arguments, we have that $S_{f_t}(0)x = (2\pi i)^{-1}\int_{\Gamma_c^{\zeta}-\omega} e^{-t(-\lambda)^{b/\zeta}}((\lambda+\omega)B+A)^{-1}Cx d\lambda \ (x \in E, t > 0)$, where Γ_c^{ζ} denotes the upwards oriented boundary of $\Pi_{c,\sigma,\varsigma}^{\zeta}$. Let the curve Γ' be sufficiently close to Γ_c^{ζ} , on the right of Γ_c^{ζ} , and let the curve $\Gamma'_{\omega} := \Gamma' - \omega$ be upwards oriented. Modifying slightly the second part of proof of [65, Lemma 4] (the proof of this lemma contains some typographical mistakes but the essence and final conclusions are true; we can apply Stirling's formula), and keeping in mind the boundedness of set $\{h_0^p p^{(-p\zeta)/b}(B^{-1}A)^p By: p \in \mathbb{N}_0\}$, we get that there exists a number $\delta > 0$ such that for each integer $p \in \mathbb{N}$ there exists an integer $n(p) \in \mathbb{N} \cap (\frac{bp}{\zeta} + \nu + 2, \frac{bp}{\zeta} + \nu + 3]$ such that the series $\sum_{p=0}^{\infty} x_p$ and $\sum_{p=0}^{\infty} Bx_p$ are convergent, where

$$x_p := \frac{\delta^p}{2\pi i p!} \int_{\Gamma'_{\omega}} (-\lambda)^{bp/\zeta} (\lambda+\omega)^{-n(p)} ((\lambda+\omega)B + A)^{-1} C (B^{-1}A)^{n(p)} By \, d\lambda.$$

Let $x = \sum_{p=0}^{\infty} x_p$ and $Bx = \sum_{p=0}^{\infty} Bx_p$; arguing as in Remark 2.5.2(vii), we obtain with the help of equation (182), the Cauchy formula and the resolvent equation that

(183)
$$S_{f_t}(0)Bx = \frac{1}{2\pi i} \int_{\Gamma'_{\omega}} e^{-(t-\delta)(-\lambda)^{b/\zeta}} ((\lambda+\omega)B + A)^{-1}C^2 By \, d\lambda, \quad t > \delta.$$

Let $\lambda_0 \in \mathbb{C}$ be on the right of Γ_c^{ζ} , and simultaneously, on the left of Γ' . Making use of the identity [296, (3.16)], with the operator A replaced by $-B^{-1}A$ therein (we only need the linearity of operator $B^{-1}A$, not its closedness), we get that

(184)
$$CBy = \sum_{j=0}^{|\nu|+2} \frac{(-1)^j}{((\lambda+\omega)-\lambda_0)^{j+1}} ((\lambda+\omega)+B^{-1}A)CBy + (-1)^{\lceil\nu\rceil+1} \frac{C(\lambda_0I+B^{-1}A)^{\lceil\nu\rceil+3}By}{((\lambda+\omega)-\lambda_0)^{\lceil\nu\rceil+3}}, \quad \lambda \in \Gamma'_{\omega}.$$

Since $Cy \in D(A)$, BACy = ACBy and $CB \subseteq BC$, we have that $((\lambda + \omega) + B^{-1}A)CBy = ((\lambda + \omega)B + A)Cy, \lambda \in \Gamma'_{\omega}$. Applying the operator $((\lambda + \omega)B + A)^{-1}C$ on the both sides of (184), the above implies

$$\begin{split} ((\lambda+\omega)B+A)^{-1}C^2By &= \sum_{j=0}^{\lceil\nu\rceil+2} \frac{(-1)^j}{((\lambda+\omega)-\lambda_0)^{j+1}}C^2y \\ &+ (-1)^{\lceil\nu\rceil+1}\frac{((\lambda+\omega)B+A)^{-1}C}{((\lambda+\omega)-\lambda_0)^{\lceil\nu\rceil+3}}(\lambda_0I+B^{-1}A)^{\lceil\nu\rceil+3}CBy, \quad \lambda\in\Gamma'_\omega. \end{split}$$

Inserting this expression in (183), and using after that the limit equality [502, Lemma 2.7; p. 543, l. -7], as well as the residue theorem and the dominated convergence theorem, we obtain that $S_{f_{\delta}}(0)Bx = \lim_{t\to\delta+} S_{f_t}(0)Bx = C^2y$. Keeping in mind Remark 2.5.2(vi.2), the above implies

(185)
$$C^2(E(A;B)) \subseteq \bigcup_{t>0} R(S_{f_t}(0)B) \subseteq \bigcup_{t>0} R(S_{f_t}(0)) \subseteq E^{\langle p^{p\zeta/b} \rangle}(B^{-1}A),$$

where

$$E(A;B) := \{ y \in B^{-1}(E^{\langle p^{p\zeta/b} \rangle}(B^{-1}A)) : Cy \in D(A), \ BACy = ACBy \}.$$

In the present situation, we do not know whether the equation (185) continues to hold if we replace the term $C^2(E(A;B))$ with C(E(A;B)).

REMARK 2.5.6. Consider the case B = I, n = 1 and $\zeta \in (0, 1)$ (cf. [354] for the case $\zeta = 1$). As before, we assume that (M_p) is a sequence of positive real numbers satisfying $M_0 = 1$ and (M.1), as well as that there exist constants $l \ge 1$, $\alpha > 0$, $\beta > 0$, $\nu > -1$ and $\xi \ge 0$ such that $\Lambda_{\alpha,\beta,l}^{\zeta} \subseteq \rho_C(-A)$. Suppose that the operator family $\{(1 + |\lambda|)^{-\nu}e^{-M(\xi|\lambda|)}(\lambda^{\zeta} + A)^{-1}C : \lambda \in \Lambda_{\alpha,\beta,l}^{\zeta}\}$ is both equicontinuous and strongly continuous. Then we have found that the problem

(PR) In which cases does there exist an injective operator $C' \in L(E)$ such that the operator -A generates a global locally equicontinuous (g_{ζ}, C') -regularized resolvent family $(S(t))_{t\geq 0}$ on E?

is very difficult to answer in general. Here we will prove that the problem (PR) can be answered in the affirmative provided that (164) holds with some $b \in (0, 1)$, as well as that $\xi = 0$ and $1/(2 - \zeta) > b/\zeta$ (cf. Theorem 2.5.1, Theorem 2.5.3 and Remark 2.5.2(v)). Then $(S_{f_t}(0))_{t \in \Sigma_{\theta}}$ is an analytic *C*-regularized semigroup of growth order $(\nu + 1)\zeta/b$, consisting of injective operators, with the closed linear operator $-(-A - \omega)_{b/\zeta}$ being its integral generator $(\omega > 0$ is a sufficiently large real number; cf. [103, Theorem 3.5, Theorem 3.7]), and the following holds:

- (a) $(S_{f_t}(s))_{s \ge 0}$ is a locally equicontinuous $(g_{\zeta}, S_{f_t}(0))$ -regularized resolvent family generated by -A $(t \in \Sigma_{\theta})$. If $q \in \mathfrak{B}$, \mathbb{B} is a bounded subset of Eand K is a compact subset of $[0, \infty)$, then there exists $h_0 > 0$ such that (174) holds with $f = f_t$ $(t \in \Sigma_{\theta})$.
- (b) Suppose that 0 < c < b, $\sigma > 0$, $\nu > -1$, $\varsigma > 0$, $\Pi_{c,\sigma,\varsigma}^{\zeta} \subseteq \rho_C(-A)$, and the operator family $\{(1+|\lambda|)^{-\nu}(\lambda^{\zeta}+A)^{-1}C : \lambda \in \Pi_{c,\sigma,\varsigma}\} \subseteq L(E)$ is both equicontinuous and strongly continuous. Then the conclusions stated in (a) continue to hold.

Therefore, a great number of multiplication and (pseudo-)differential operators in L^p -spaces can serve as examples of the integral generators of fractional Cregularized resolvent families. Although the applications of theoretical results from statements (a)–(b) and Remark 2.5.5 can be also made to (pseudo-)differential operators with empty resolvent set, and to the operators considered in certain classes of Fréchet function spaces (cf. also Example 2.7.6 below), we shall present only one illustrative example of application of the results from (b) and Remark 2.5.5 to abstract non-degenerate fractional differential equations. Assume 0 < c < b < 1, $1/(2-\zeta) > b/\zeta$, $\sigma > 0$, $\varsigma > 0$, $p \in [1, \infty)$, m > 0, $\rho \in [0, 1]$, r > 0, $a \in S^m_{\rho,0}$ satisfies (H_r) , the inequality

(186)
$$n \left| \frac{1}{2} - \frac{1}{p} \right| \left(\frac{m - r - \rho + 1}{r} \right) < 1$$

holds, $E = L^p(\mathbb{R}^n)$ or $E = C_0(\mathbb{R}^n)$ (in the last case, we assume that (186) holds with $p = \infty$), and $A := -\operatorname{Op}_E(a)$ (cf. [27, Chapter 8] for the notion and terminology). If $\operatorname{dist}(a(\mathbb{R}^n), \prod_{c,\sigma,\varsigma}^{\zeta}) > 0$, then there exists a number $\nu > -1$ such that the operator family $\{(1 + |\lambda|)^{-\nu}(\lambda^{\zeta} + A)^{-1} : \lambda \in \prod_{c,\sigma,\varsigma}\} \subseteq L(E)$ is both equicontinuous and strongly continuous (C = I), so that $(S_{f_t}(s))_{s \ge 0}$ is a global $(g_{\zeta}, S_{f_t}(0))$ -regularized resolvent family generated by -A $(t \in \Sigma_{\theta})$; furthermore, if K is a compact subset of $[0, \infty)$ and $t \in \Sigma_{\theta}$, then there exists $h_0 > 0$ such that

$$\sup_{p \in \mathbb{N}_0, s \in K} \frac{(h_0)^p \| (\mathbf{D}_s^{\zeta})^p S_{f_t}(s) \|}{p^{p\zeta/b}} < \infty.$$

The proof of (185) implies that $\bigcup_{t>0} S_{f_t}(0)(D(A)) = E^{\langle p^{p\zeta/b} \rangle}(A)$, so that the problem $(\text{DFP})'_R$, with B = I, has a unique strong solution for all $x \in E^{\langle p^{p\zeta/b} \rangle}(A)$, given by $u(s) := S_{f_t}(s)S_{f_t}(0)^{-1}x, s \ge 0$, where t > 0 satisfies $x \in S_{f_t}(0)(D(A))$. A concrete example can be simply constructed.

EXAMPLE 2.5.7. Suppose that $\zeta = 1 - c > c(1 + c)$. This, in turn, implies $1/(2 - \zeta) > b/\zeta > c/\zeta$ as $1 > b \rightarrow c+$. Since $(x + ix^{1/c})^{\zeta} = (x^2 + x^{2/c})^{\zeta/2} [\cos(\zeta \arctan x^{(1/c)-1}) + i\sin(\zeta \arctan x^{(1/c)-1})]$, x > 0, an elementary calculus shows that

$$\operatorname{Re}((x+ix^{1/c})^{\zeta})/\operatorname{Im}((x+ix^{1/c})^{\zeta}) \sim 1/\tan(\zeta\pi/2) \text{ as } x \to +\infty,$$

and

$$\operatorname{Re}((x+ix^{1/c})^{\zeta}) - (\operatorname{tan}(\zeta \pi/2))^{-1} \operatorname{Im}((x+ix^{1/c})^{\zeta}) \sim \zeta(\sin(\zeta \pi/2))^{-1} x^{((\zeta-1)/c)+1} = \zeta(\sin(\zeta \pi/2))^{-1} \text{ as } x \to +\infty$$

(similar formulae hold if we look into the term $(x - ix^{1/c})^{\zeta}$ in place of $(x + ix^{1/c})^{\zeta}$). Using these asymptotic expansions, it readily follows that for each number $d \in (0, \zeta/\sin(\zeta \pi/2))$ there exists a sufficiently large number $r_d > 0$ such that $\operatorname{dist}(a(\mathbb{R}^n), \prod_{c,\sigma_d,\varsigma_d}^{\zeta}) > 0$ for suitable chosen numbers $\sigma_d > 0$ and $\varsigma_d > 0$, provided that $a(x) = d + (r_d + P(x))e^{\pm i\pi\zeta/2}$, where P(x) is a positive real elliptic polynomial in n variables, of order m (then (186) holds with m = r and $\rho = 1$).

The polynomials of the operator A := -d/ds, $D(A) := \{f \in E ; f' \in E, f(0) = 0\}$, acting on the Banach space

$$E := \left\{ f \in C^{\infty}[0,1] \; ; \; \|f\| := \sup_{p \ge 0} \frac{\|f^{(p)}\|_{\infty}}{p!^s} < \infty \right\} \quad (s > 1),$$

have been considered multiple times by now. The interested reader may throws oneself into the problem of proving some upper bounds on the growth rate of the term

$$\left\| \left(\lambda^{(p_n+q_n)\zeta} P_n(A) + \sum_{i=0}^{n-1} \lambda^{(p_i+q_i)\zeta} P_i(A) \right)^{-1} \right\|,$$

where $P_i(z)$ is a complex non-zero polynomial $(1 \le i \le n)$, thus providing certain applications of Theorem 2.5.1 and Theorem 2.5.3.

EXAMPLE 2.5.8. Suppose that E is a general SCLCS, $b \in (0, 1)$, (M_p) satisfies (M.1) and (164), $\zeta = 1$, $p_i = 0$ for all $i \in \mathbb{N}_n^0$, $q_n > q_{n-1}$, $\alpha > 0$, $\beta > 0$, $l \ge 1$, $\emptyset \ne \Omega' \subseteq \mathbb{C}$, $N \in \mathbb{N}$, A is a densely defined closed linear operator in E satisfying that $\Omega' \subseteq \rho(A)$ and the operator family $\{(1 + |\lambda|)^{-N}(\lambda - A)^{-1} : \lambda \in \Omega'\} \subseteq L(E)$ is equicontinuous. Suppose, further, that $P_i(z)$ is a complex polynomial $(i \in \mathbb{N}_n^0)$, $P_n(z) \not\equiv 0$, $\lambda_0 \in \rho(A) \smallsetminus \{z \in \mathbb{C} : P_n(z) = 0\}$, dist $(\lambda_0, \Omega') > 0$, as well as that for each $\lambda \in \Lambda_{\alpha,\beta,l}$ all roots of the polynomial

$$z \mapsto \lambda^{q_n} P_n(z) + \sum_{i=0}^{n-1} \lambda^{q_i} P_i(z), \quad z \in \mathbb{C}$$

belong to Ω' . Set $B := P_n(A)$ and $A_i := P_i(A)$ $(i \in \mathbb{N}_{n-1}^0)$. Then there exist $M \in \mathbb{N}$, λ -polynomials $F_0(\lambda), \ldots, F_M(\lambda)$ and not necessarily distinct numbers $f_1(\lambda) \in \Omega', \ldots, f_M(\lambda) \in \Omega'$, continuously depending on λ , such that $\lambda^{q_n} P_n(z) + \sum_{i=0}^{n-1} \lambda^{q_i} P_i(z) = F_M(\lambda) z^M + \cdots + F_1(\lambda) z + F_0(\lambda) = (-1)^M F_M(\lambda) (f_M(\lambda) - z) \ldots$ $(f_1(\lambda) - z)$ for all $\lambda \in \Lambda_{\alpha,\beta,l} \smallsetminus \mathcal{P}$ and $z \in \mathbb{C}$, where $\mathcal{P} \equiv \{\lambda \in \mathbb{C} : F_M(\lambda) = 0\}$; furthermore, for each $\lambda \in \Lambda_{\alpha,\beta,l} \smallsetminus \mathcal{P}$ the following holds:

$$\left(\lambda^{q_n} P_n(A) + \sum_{i=0}^{n-1} \lambda^{q_i} P_i(A)\right)^{-1} = (-1)^M (F_M(\lambda))^{-1} (f_M(\lambda) - A)^{-1} \dots (f_1(\lambda) - A)^{-1}.$$

Using the generalized resolvent equation, it readily follows that for any integer $Q \ge N+2$ the operator family $\{(f_i(\lambda) - \lambda_0)(f_i(\lambda) - A)^{-1}(\lambda_0 - A)^{-Q} : \lambda \in \Lambda_{\alpha,\beta,l} \smallsetminus \mathcal{P}\} \subseteq L(E)$ is equicontinuous $(1 \le i \le M)$. This implies that there exists a sufficiently large integer $Q' \ge N+2$ such that for each seminorm $p \in \mathfrak{B}$ there exist $c_p > 0$ and $q \in \mathfrak{B}$ such that, for every $j \in \mathbb{N}_{n-1}^0$, $\lambda \in \Lambda_{\alpha,\beta,l} \smallsetminus \mathcal{P}$ and $x \in E$,

$$p\left(\left(\lambda^{q_n}P_n(A) + \sum_{i=0}^{n-1} \lambda^{q_i}P_i(A)\right)^{-1} (\lambda_0 - A)^{-Q'}x\right) + p\left(P_j(A)\left(\lambda^{q_n}P_n(A) + \sum_{i=0}^{n-1} \lambda^{q_i}P_i(A)\right)^{-1} (\lambda_0 - A)^{-Q'}x\right) = p\left((F_M(\lambda))^{-1} (f_M(\lambda) - A)^{-1} \dots (f_1(\lambda) - A)^{-1} (\lambda_0 - A)^{-Q'}x\right)$$

$$+ p(P_{j}(A)(F_{M}(\lambda))^{-1}(f_{M}(\lambda) - A)^{-1} \dots (f_{1}(\lambda) - A)^{-1}(\lambda_{0} - A)^{-Q'}x)$$

$$\leq c_{p}|F_{M}(\lambda)|^{-1}|(f_{M}(\lambda) - \lambda_{0}) \dots (f_{1}(\lambda) - \lambda_{0})|^{-1}q(x)$$

$$= c_{p}|F_{M}(\lambda)|^{-1}|F_{M}(\lambda)||F_{M}(\lambda)\lambda_{0}^{m} + \dots + F_{1}(\lambda)\lambda_{0} + F_{0}(\lambda)|^{-1}q(x)$$

$$= c_{p}\left|\lambda^{q_{n}}P_{n}(\lambda_{0}) + \sum_{i=0}^{n-1}\lambda^{q_{i}}P_{i}(\lambda_{0})\right|^{-1}q(x) \sim c_{p}|P_{n}(\lambda_{0})|^{-1}|\lambda|^{-q_{n}}q(x) \text{ as } |\lambda| \to \infty.$$

Therefore, there exists a sufficiently large number $\beta' > \beta$ such that the operator families $\{(1 + |\lambda|)^{q_n}(\lambda^{q_n}P_n(A) + \sum_{i=0}^{n-1}\lambda^{q_i}P_i(A))^{-1}(\lambda_0 - A)^{-Q'} : \lambda \in \Lambda_{\alpha,\beta',l}\} \subseteq L(E)$ and $\{(1 + |\lambda|)^{q_n}P_j(A)(\lambda^{q_n}P_n(A) + \sum_{i=0}^{n-1}\lambda^{q_i}P_i(A))^{-1}(\lambda_0 - A)^{-Q'} : \lambda \in \Lambda_{\alpha,\beta',l}\} \subseteq L(E)$ are equicontinuous $(j \in \mathbb{N}_{n-1}^0)$. Since $q_n > q_{n-1}$ and P(A) is dense in E for any complex polynomial $P(z) \in \mathbb{C}[z]$, the analysis contained in Remark 2.5.2(x), with $C \equiv (\lambda_0 - A)^{-Q'}$, shows that for each $(x_0, \ldots, x_{q_n-1}) \in E^{q_n}$ there exists a net $(u_{\varepsilon}(t))_{\varepsilon>0}$ of strong solutions of problem (168) with the subjected initial values $(u_0^{\varepsilon}, \ldots, u_{q_n-1}^{\varepsilon})$, converging to (x_0, \ldots, x_{q_n-1}) as $\varepsilon \to 0+$ (for the topology of E^{q_n}). This example can be also used to provide the possible applications of Theorem 2.6.2 below, with C not being the identity operator on E.

2.6. Entire and analytical solutions of abstract degenerate Cauchy problem (PN)

Before starting our work in this section, we need to repeat some notations and preliminaries from the previous one; cf. also Introduction. We assume that $n \in \mathbb{N}, 0 < \zeta \leq 2, q_0, q_1, \ldots, q_n$ are given non-negative integers satisfying $q_0 =$ 0 and $0 < q_1 \leq q_2 \leq \ldots \leq q_n$, as well as that $p_i = 0$ for all $i \in \mathbb{N}_n^0$, and $A_0, A_1, \ldots, A_{n-1}, A_n$ are closed linear operators acting on E; we also write B for A_n . Hence, $T_i u(s) = A_i (\mathbf{D}_s^{\zeta})^{q_i} u(s), s \geq 0, i \in \mathbb{N}_n^0$ and $P_\lambda = \lambda^{q_n \zeta} B + \sum_{i=0}^{n-1} \lambda^{q_i \zeta} A_i$, $\lambda \in \mathbb{C} \setminus \{0\}$. Set $S_\omega := \{j \in \mathbb{N}_n^0 : q_j - 1 \geq \omega\}, \omega \in \mathbb{N}_{q_n-1}^0$; then $n \in S_\omega$ for all $\omega \in \mathbb{N}_{q_n-1}^0$. Suppose that the equation (180) holds with the number v replaced with ω therein.

In this section, we will take up the study of entire and analytical solutions of the abstract Cauchy problem (168) subjected with the initial conditions of the form

(187)

$$((\mathbf{D}_{s}^{\zeta})^{j}u(s))_{s=0} = u_{j}, \ j \in \mathbb{N}_{q_{n}-1}^{0}, \ \text{if } \zeta \in (0,1], \text{ resp.},$$

$$((\mathbf{D}_{s}^{\zeta})^{j}u(s))_{s=0} = u_{j}, \ j \in \mathbb{N}_{q_{n}-1}^{0};$$

$$\left(\frac{d}{ds}(\mathbf{D}_{s}^{\zeta})^{j}u(s)\right)_{s=0} = v_{j}, \ j \in \mathbb{N}_{q_{n}-1}^{0}, \text{ if } \zeta \in (1,2].$$

To simplify the notation, this initial value problem will be also called (PN) problem.

DEFINITION 2.6.1. (ii) A function $u \in C([0,\infty) : E)$ is said to be an entire solution of problem (PN) iff $u(\cdot)$ is a strong solution of (PN) and it can be analytically extended to the whole complex plane, as well as any of the terms $A_i u^{(p)}(\cdot)$ $(0 \leq i \leq n, p \in \mathbb{N}_0)$ can be analytically extended to the whole complex plane.

(iii) A function $u \in C([0,\infty) : E)$ is said to be an analytical solution of problem (PN) on the region $\mathbb{C} \smallsetminus (-\infty, 0]$ iff $u(\cdot)$ is a strong solution of (PN) and it can be extended to the whole complex plane, analytically on the region $\mathbb{C} \smallsetminus (-\infty, 0]$ and continuously on the region $\mathbb{C} \smallsetminus (-\infty, 0)$, as well as any of the terms $A_i(\mathbf{D}_s^{\zeta})^p u(s)$ $(0 \le i \le n, p \in \mathbb{N}_0, s \ge 0)$ is well defined and can be extended to the whole complex plane, analytically on the region $\mathbb{C} \smallsetminus (-\infty, 0]$ and continuously on the region $\mathbb{C} \smallsetminus (-\infty, 0)$.

Now we are ready to formulate the following theorem.

THEOREM 2.6.2. Suppose that the operator $C \in L(E)$ is injective, $CA_i \subseteq A_iC$, $i \in \mathbb{N}_n^0$, $0 < \zeta \leq 2$, $\phi \in (-\pi, \pi]$, $\theta \in (\pi - \pi\zeta, \pi - (\pi\zeta)/2)$, a > r > 0 and ν' satisfies (180). Assume, further, that the following holds:

- (i) The operator family $\{(1+|\lambda|)^{-\nu'}P_{\lambda^{1/\zeta}}^{-1}C : \lambda \in e^{i\phi}\Sigma_{(\zeta\pi/2)+\theta}, |\lambda| \ge r\} \subseteq L(E)$ is both strongly continuous and equicontinuous.
- (ii) For every $x \in E$ and $i \in \mathbb{N}_{n-1}^{0}$, the mapping $\lambda \mapsto A_{i}P_{\lambda^{1/\zeta}}^{-1}Cx$, $\lambda \in e^{i\phi}\Sigma_{(\zeta\pi/2)+\theta}$, $|\lambda| \ge r$ is continuous and there exists $v_{i} \in \mathbb{N}$ such that the operator family $\{(1+|\lambda|)^{-v_{i}}A_{i}P_{\lambda^{1/\zeta}}^{-1}C : \lambda \in e^{i\phi}\Sigma_{(\zeta\pi/2)+\theta}, |\lambda| \ge r\} \subseteq L(E)$ is equicontinuous.

Denote by $\mathfrak{W}(\mathfrak{W}_e)$ the subspace of E^{q_n} , resp. E^{2q_n} , consisting of all initial values $(u_0, \ldots, u_{q_n-1}) \in E^{q_n}$, resp. $(u_0, \ldots, u_{q_n-1}; v_0, \ldots, v_{q_n-1}) \in E^{2q_n}$, subjected to some analytical solution $u(\cdot)$ of problem (168) on the region $\mathbb{C} \smallsetminus (-\infty, 0]$ (entire solution $u(\cdot)$ of problem (168)). Then \mathfrak{W} is dense in $(C(\bigcap_{j=0}^n D(A_j)))^{q_n}$ for the topology of E^{q_n} , provided that $0 < \zeta < 1$, resp. $(C(\bigcap_{j=0}^n D(A_j)))^{2q_n}$ for the topology of E^{2q_n} , provided that $1 < \zeta < 2$. Furthermore, if $\zeta = 1$, resp. $\zeta = 2$, then the set \mathfrak{W}_e is dense in $(C(\bigcap_{j=0}^n D(A_j)))^{q_n}$ for the topology of E^{q_n} , resp. $(C(\bigcap_{j=0}^n D(A_j)))^{2q_n}$ for the topology of E^{2q_n} .

To prove Theorem 2.6.2, we need the following lemma (cf. also [515, Lemma 1.1, Theorem 1.1]).

LEMMA 2.6.3. Let $x \in E$. Then the mapping

$$\lambda \mapsto P_{(\lambda e^{i\phi})^{1/\zeta}}^{-1} Cx, \quad \lambda \in \Sigma_{(\zeta \pi/2) + \theta}, \ |\lambda| > r$$

is analytic.

PROOF. Without loss of generality, we may assume that $q_i = i$ $(i \in \mathbb{N}_n^0)$, $\zeta = 1$ and $\phi = 0$. Clearly, (ii) holds for every $x \in E$ and $i \in \mathbb{N}_n^0$. Furthermore, the following analogue of the Hilbert resolvent equation holds:

$$P_{\lambda}^{-1}C^{2}x - P_{z}^{-1}C^{2}x = (z-\lambda)P_{\lambda}^{-1}C$$

$$\times \left[\sum_{k=1}^{n-1} \binom{n}{k}(z-\lambda)^{k-1}\lambda^{n-k}B + \sum_{k=1}^{n-2} \binom{n-1}{k}(z-\lambda)^{k-1}\lambda^{n-1-k}A_{n-1} + \dots + A_{1}\right]$$

$$\times P_{z}^{-1}Cx, \text{ provided } \lambda, \ z \in \Sigma_{(\zeta\pi/2)+\theta} \text{ and } |\lambda|, \ |z| > r.$$

This implies that the mapping $\lambda \mapsto P_{\lambda}^{-1}C^2x$, $\lambda \in \Sigma_{(\zeta \pi/2)+\theta}$, $|\lambda| > r$ is weakly analytic and therefore analytic, as well as that

$$\frac{d}{d\lambda}\langle x^*, P_{\lambda}^{-1}C^2x\rangle = -\langle x^*, P_{\lambda}^{-1}[n\lambda^{n-1}B + (n-1)\lambda^{n-2}A_{n-1} + \dots + A_1]P_{\lambda}^{-1}Cx\rangle,$$

provided $x^* \in E^*$, $\lambda \in \Sigma_{(\zeta \pi/2)+\theta}$ and $|\lambda| > r$. Using the Morera theorem and the observation from [138, Remark 2.7], the above implies that the mapping $\lambda \mapsto P_{\lambda}^{-1}Cx$, $\lambda \in \Sigma_{(\zeta \pi/2)+\theta}$, $|\lambda| > r$ is analytic, as claimed.

Now we can proceed to the proof of Theorem 2.6.2.

PROOF. Clearly, $(\zeta \pi/2) + \theta < \pi$, $\pi \zeta/2 > \pi - (\zeta \pi/2) - \theta$ and we can find a number $b \in \mathbb{R}$ satisfying

$$1 < b < \frac{\pi\zeta/2}{\pi - (\zeta\pi/2) - \theta}$$

Denote by Γ the upwards oriented boundary of the region $\{\lambda \in \Sigma_{(\zeta \pi/2)+\theta} : |\lambda| \ge r\}$. Let Ω' be the open region on the left of Γ . Then there exists a sufficiently large number R > 0 such that $a - \lambda \in \Sigma_{\pi - (\zeta \pi/2) - \theta}$ for all $\lambda \in \Omega' \cup \Gamma$ with $|\lambda| \ge R$. This implies $|e^{-\varepsilon(a-\lambda)^{b/\zeta}}| = e^{-\varepsilon \operatorname{Re}((a-\lambda)^{b/\zeta})} \le e^{-\varepsilon|a-\lambda|^{b/\zeta} \cos(b\zeta^{-1}(\pi - (\pi\zeta/2) - \theta))}$, provided $\varepsilon > 0, \lambda \in \Omega' \cup \Gamma$ and $|\lambda| \ge R$. Keeping in mind Theorem 1.5.1, we obtain the existence of a constant $c_{\zeta} > 0$ such that $|E_{\zeta}(z^{\zeta}\lambda e^{i\phi})| \le E_{\zeta}(|z|^{\zeta}|\lambda|) \le c_{\zeta}' e^{|z||\lambda|^{1/\zeta}}$ for all $z \in \mathbb{C}$ and $\lambda \in \mathbb{C}$. Hence, there exists a constant $c_{\zeta} > 0$ such that

(188)
$$|e^{-\varepsilon(a-\lambda)^{b/\zeta}}E_{\zeta}(z^{\zeta}\lambda e^{i\phi})| \leqslant c_{\zeta}e^{-\varepsilon|a-\lambda|^{b/\zeta}\cos(b\zeta^{-1}(\pi-(\pi\zeta/2)-\theta))+|z||\lambda|^{1/\zeta}},$$

for any $z \in \mathbb{C}$, $\varepsilon > 0$ and $\lambda \in \Omega' \cup \Gamma$. Suppose now that $x_w \in \bigcap_{j=0}^n D(A_j)$ for all $w \in \mathbb{N}^0_{q_n-1}$. Then (i) and the estimate (188) enable us to define the function $z \mapsto u_{\varepsilon}(z), z \in \mathbb{C}$, for any $\varepsilon > 0$, by

$$u_{\varepsilon}(z) := \sum_{w=0}^{q_n-1} \sum_{j \in S_{\omega}} \frac{1}{2\pi i} \int_{\Gamma} e^{-\varepsilon(a-\lambda)^{b/\zeta}} E_{\zeta}(z^{\zeta} \lambda e^{i\phi}) (\lambda e^{i\phi})^{q_j-1-w} P_{(\lambda e^{i\phi})^{1/\zeta}}^{-1} CA_j x_w d\lambda.$$

It can be simply verified that the mapping $z \mapsto u_{\varepsilon}(z), z \in \mathbb{C} \setminus (-\infty, 0)$ is continuous $(\varepsilon > 0)$. Using Theorem 1.5.1 and the proof of Theorem 2.5.1, it readily follows that the mapping $z \mapsto u_{\varepsilon}(z), z \in \mathbb{C} \setminus (-\infty, 0]$ is analytic $(\varepsilon > 0)$, with

(189)
$$u_{\varepsilon}'(z) = \sum_{w=0}^{q_n-1} \sum_{j \in S_{\omega}} \frac{1}{2\pi i} \int_{\Gamma} e^{-\varepsilon(a-\lambda)^{b/\zeta}} z^{\zeta-1} E_{\zeta,\zeta}(z^{\zeta} \lambda e^{i\phi}) (\lambda e^{i\phi})^{q_j-w} P_{(\lambda e^{i\phi})^{1/\zeta}}^{-1} CA_j x_w d\lambda,$$

for any $\varepsilon > 0$ and $z \in \mathbb{C} \setminus (-\infty, 0]$; furthermore, the mapping $z \mapsto u_{\varepsilon}(z), z \in \mathbb{C}$ is entire provided $\varepsilon > 0, \zeta = 1$ and, in this case, (189) holds for any $\varepsilon > 0$ and $z \in \mathbb{C}$. The proof of Theorem 2.5.1 also shows that the term $(\mathbf{D}_s^{\zeta})^p u_{\varepsilon}(s), s \ge 0$ is well defined for any $p \in \mathbb{N}_0$ and $\varepsilon > 0$, with

(190)
$$(\mathbf{D}_{s}^{\zeta})^{p} u_{\varepsilon}(s)$$
$$= \sum_{w=0}^{q_{n}-1} \sum_{j \in S_{\omega}} \frac{1}{2\pi i} \int_{\Gamma} e^{-\varepsilon (a-\lambda)^{b/\zeta}} E_{\zeta}(s^{\zeta} \lambda e^{i\phi}) (\lambda e^{i\phi})^{p+q_{j}-1-w} P_{(\lambda e^{i\phi})^{1/\zeta}}^{-1} CA_{j} x_{w} d\lambda;$$

cf. also the formula [61, (1.25)]. Combined with the Cauchy theorem, (ii) and Theorem 1.5.1, the above implies that the term $A_i(\mathbf{D}_s^{\zeta})^p u_{\varepsilon}(s)$ is well defined for $s \ge 0, i \in \mathbb{N}_n^0, p \in \mathbb{N}_0$ and $\varepsilon > 0$, with

$$A_{i}(\mathbf{D}_{s}^{\zeta})^{p}u_{\varepsilon}(s)$$

$$=\sum_{w=0}^{q_{n}-1}\sum_{j\in S_{\omega}}\frac{1}{2\pi i}\int_{\Gamma}e^{-\varepsilon(a-\lambda)^{b/\zeta}}E_{\zeta}(s^{\zeta}\lambda e^{i\phi})(\lambda e^{i\phi})^{p+q_{j}-1-w}A_{i}P_{(\lambda e^{i\phi})^{1/\zeta}}^{-1}CA_{j}x_{w}d\lambda.$$

This implies that, for every $\varepsilon > 0$, any of the terms $A_i(\mathbf{D}_s^{\zeta})^p u_{\varepsilon}(\cdot)$ $(0 \leq i \leq n, p \in \mathbb{N}_0)$ can be extended to the whole complex plane, analytically on the region $\mathbb{C} \setminus (-\infty, 0]$ and continuously on the region $\mathbb{C} \setminus (-\infty, 0)$; we only need to replace the variable $s \geq 0$, appearing in the above formula, with the variable $z \in \mathbb{C}$. Furthermore,

$$\sum_{i=0}^{n} A_{i}(\mathbf{D}_{s}^{\zeta})^{q_{i}} u_{\varepsilon}(s)$$

$$= \sum_{w=0}^{q_{n}-1} \sum_{j \in S_{\omega}} \sum_{i=0}^{n} \frac{1}{2\pi i} \int_{\Gamma} e^{-\varepsilon(a-\lambda)^{b/\zeta}} E_{\zeta}(s^{\zeta}\lambda e^{i\phi}) (\lambda e^{i\phi})^{q_{i}+q_{j}-1-w} A_{i} P_{(\lambda e^{i\phi})^{1/\zeta}}^{-1} CA_{j} x_{w} d\lambda$$

$$= \sum_{w=0}^{q_{n}-1} \sum_{j \in S_{\omega}} \frac{1}{2\pi i} \int_{\Gamma} e^{-\varepsilon(a-\lambda)^{b/\zeta}} E_{\zeta}(s^{\zeta}\lambda e^{i\phi}) (\lambda e^{i\phi})^{q_{j}-1-w} CA_{j} x_{\omega} d\lambda = 0, \ s \ge 0, \ \varepsilon > 0,$$

so that for each $\varepsilon > 0$ the mapping $s \mapsto u_{\varepsilon}(s), s \ge 0$ is an analytical solution of problem (168) on the region $\mathbb{C} \smallsetminus (-\infty, 0]$ (entire solution of problem (168), provided that $\zeta = 1$), with the initial values $(u_0^{\varepsilon}, \ldots, u_{q_n-1}^{\varepsilon})$ subjected. Let $u_l^{\varepsilon} = ((\mathbf{D}_s^{\varepsilon})^l u_{\varepsilon}(s))_{s=0}, l \in \mathbb{N}_{q_n-1}^0$ ($\varepsilon > 0$). Now we will prove that $(u_0^{\varepsilon}, \ldots, u_{q_n-1}^{\varepsilon})$ converges to $e^{-i\phi}(Cx_0, \ldots, Cx_{q_n-1})$ as $\varepsilon \to 0+$, for the topology of E^{q_n} . Let $\omega \in \mathbb{N}_{q_n-1}^0$ and $l \in \mathbb{N}_{q_n-1}^0$ be fixed. Keeping in mind (190), it suffices to prove that the following holds:

$$\lim_{\varepsilon \to 0+} \sum_{j \in S_{\omega}} \frac{1}{2\pi i} \int_{\Gamma} e^{-\varepsilon (a-\lambda)^{b/\zeta}} (\lambda e^{i\phi})^{l+q_j-1-w} P^{-1}_{(\lambda e^{i\phi})^{1/\zeta}} CA_j x_w d\lambda = e^{-i\phi} \delta_{\omega,l} Cx_{\omega},$$

i.e., that

(191)
$$\lim_{\varepsilon \to 0+} \frac{1}{2\pi i} \int_{\Gamma} e^{-\varepsilon (a-\lambda)^{b/\zeta}} (\lambda e^{i\phi})^{l-1-w} \\ \times \left[Cx_{\omega} - \sum_{j \in \mathbb{N}_{n}^{0} \smallsetminus S_{\omega}} (\lambda e^{i\phi})^{q_{j}} P_{(\lambda e^{i\phi})^{1/\zeta}}^{-1} CA_{j} x_{w} \right] d\lambda = e^{-i\phi} \delta_{\omega,l} Cx_{\omega},$$

where $\delta_{\omega,l}$ denotes the Kronecker delta. Since $|e^{-\varepsilon(a-\lambda)^{b/\zeta}}| \leq 1, \lambda \in \Gamma, \varepsilon > 0$, (180) and (i) holds, we have that there exists $\sigma > 0$ such that

$$|e^{-\varepsilon(a-\lambda)^{b/\zeta}}(\lambda e^{i\phi})^{l-1-w}(\lambda e^{i\phi})^{q_j}P^{-1}_{(\lambda e^{i\phi})^{1/\zeta}}CA_jx_w| \leq \text{Const.} \ |\lambda|^{-1-\sigma},$$

for any $\lambda \in \Gamma$, $\varepsilon > 0$ and $j \in \mathbb{N}_n^0 \setminus S_\omega$. Applying the dominated convergence theorem, Lemma 2.6.3 and the Cauchy theorem, we get that

$$\lim_{\varepsilon \to 0+} \frac{1}{2\pi i} \int_{\Gamma} e^{-\varepsilon (a-\lambda)^{b/\zeta}} (\lambda e^{i\phi})^{l-1-w} \sum_{j \in \mathbb{N}_n^0 \smallsetminus S_\omega} (\lambda e^{i\phi})^{q_j} P_{(\lambda e^{i\phi})^{1/\zeta}}^{-1} CA_j x_w d\lambda$$
$$= \frac{1}{2\pi i} \int_{\Gamma} (\lambda e^{i\phi})^{l-1-w} \sum_{j \in \mathbb{N}_n^0 \smallsetminus S_\omega} (\lambda e^{i\phi})^{q_j} P_{(\lambda e^{i\phi})^{1/\zeta}}^{-1} CA_j x_w d\lambda = 0.$$

Taking into account the last formula on p. 170 of [541], it readily follows that (191) golds good. The proof of theorem is thereby complete in the case that $0 < \zeta \leq 1$. Suppose now $1 < \zeta \leq 2$. Then it is not difficult to show that there exists a finite constant $d_{\zeta} > 0$ such that the function $F_{\lambda}(z) := zE_{\zeta,2}(z^{\zeta}\lambda e^{i\phi}), \ z \in \mathbb{C} \ (\lambda \in \mathbb{C})$ satisfies $F'_{\lambda}(z) = E_{\zeta}(z^{\zeta}\lambda e^{i\phi}), \ z \in \mathbb{C} \setminus (-\infty, 0]$ and $|F_{\lambda}(z)| \leq d_{\zeta}(1+|z|)e^{|z||\lambda|^{1/\zeta}}, \ z \in \mathbb{C} \ (\lambda \in \mathbb{C})$. Since for any function $u \in C^1([0,\infty): X)$ with u'(0) = 0 we have $\mathbf{D}_s^{\zeta}(g_1 * u)(s) = (g_1 * \mathbf{D}_{\cdot}^{\zeta}u)(s), \ s \geq 0$, provided in addition that the term $\mathbf{D}_s^{\zeta}u(s)$ is defined for $s \geq 0$, it readily follows that $\mathbf{D}_s^{\zeta}F_{\lambda}(s) = (g_1 * \mathbf{D}_s^{\zeta}E_{\zeta}(\cdot^{\zeta}\lambda e^{i\phi}))(s) = \lambda e^{i\phi}F_{\lambda}(s), \ s \geq 0 \ (\lambda \in \mathbb{C})$. Let $x_w, y_w \in \bigcap_{j=0}^n D(A_j)$ for all $w \in \mathbb{N}_{q_n-1}^0$. Define now the solution $u_{\varepsilon}(\cdot)$ by

$$u_{\varepsilon}(z) := \sum_{w=0}^{q_n-1} \sum_{j \in S_{\omega}} \frac{1}{2\pi i} \int_{\Gamma} e^{-\varepsilon (a-\lambda)^{b/\zeta}} E_{\zeta}(z^{\zeta} \lambda e^{i\phi}) (\lambda e^{i\phi})^{q_j-1-w} P_{(\lambda e^{i\phi})^{1/\zeta}}^{-1-w} CA_j x_w d\lambda$$
$$+ \sum_{w=0}^{q_n-1} \sum_{j \in S_{\omega}} \frac{1}{2\pi i} \int_{\Gamma} e^{-\varepsilon (a-\lambda)^{b/\zeta}} F_{\lambda}(z) (\lambda e^{i\phi})^{q_j-1-w} P_{(\lambda e^{i\phi})^{1/\zeta}}^{-1-w} CA_j y_w d\lambda,$$

for any $z \in \mathbb{C}$ and $\varepsilon > 0$. It is not difficult to prove that, for every $p \in \mathbb{N}_0$, $s \ge 0$ and $\varepsilon > 0$, the following holds:

$$\begin{aligned} (\mathbf{D}_{s}^{\zeta})^{p} u_{\varepsilon}(s) \\ &= \sum_{w=0}^{q_{n}-1} \sum_{j \in S_{\omega}} \frac{1}{2\pi i} \int_{\Gamma} e^{-\varepsilon(a-\lambda)^{b/\zeta}} E_{\zeta}(s^{\zeta} \lambda e^{i\phi}) (\lambda e^{i\phi})^{p+q_{j}-1-w} P_{(\lambda e^{i\phi})^{1/\zeta}}^{-1-w} CA_{j} x_{w} d\lambda \\ &+ \sum_{w=0}^{q_{n}-1} \sum_{j \in S_{\omega}} \frac{1}{2\pi i} \int_{\Gamma} e^{-\varepsilon(a-\lambda)^{b/\zeta}} F_{\lambda}(s) (\lambda e^{i\phi})^{p+q_{j}-1-w} P_{(\lambda e^{i\phi})^{1/\zeta}}^{-1-w} CA_{j} y_{w} d\lambda \end{aligned}$$

and

$$\begin{aligned} \frac{d}{ds} (\mathbf{D}_{s}^{\zeta})^{p} u_{\varepsilon}(s) \\ &= \sum_{w=0}^{q_{n}-1} \frac{1}{2\pi i} \int_{\Gamma} e^{-\varepsilon (a-\lambda)^{b/\zeta}} s^{\zeta-1} E_{\zeta,\zeta}(s^{\zeta} \lambda e^{i\phi}) (\lambda e^{i\phi})^{p+q_{j}-w} P_{(\lambda e^{i\phi})^{1/\zeta}}^{-1} CA_{j} x_{w} d\lambda \\ &+ \sum_{w=0}^{q_{n}-1} \frac{1}{2\pi i} \int_{\Gamma} e^{-\varepsilon (a-\lambda)^{b/\zeta}} E_{\zeta}(s^{\zeta} \lambda e^{i\phi}) (\lambda e^{i\phi})^{p+q_{j}-w-1} P_{(\lambda e^{i\phi})^{1/\zeta}}^{-1} CA_{j} y_{w} d\lambda. \end{aligned}$$

The remainder of the proof can be deduced by repeating almost verbatim the corresponding parts of the first part of proof (the case $0 < \zeta \leq 1$).

- REMARK 2.6.4. (i) Observe that Theorem 2.6.2 seems to be new and not considered elsewhere provided that $B \neq I$ or $\zeta \neq 1$. It is also worth noting that the assertion of Theorem 2.6.2 can be reformulated in the case that $\zeta > 2$ (cf. also [298, Theorem 2.1, Theorem 2.2]) and the condition (180) holds provided that $-\infty < \nu' < 1 q_n$. We can prove a similar result on the existence and uniqueness of entire and analytical solutions of problem (DFP)_R; we leave the reader to make this precise.
- (ii) The notion of an entire solution of the abstract Cauchy problem (ACP_n) , introduced in [544, Definition 1.1], is slightly different from the corresponding notion introduced in Definition 2.6.1(ii). Strictly speaking, if $u(\cdot)$ is an entire solution of the abstract Cauchy problem (ACP_n) in the sense of Definition 2.6.1(ii), then $u(\cdot)$ is an entire solution of problem (ACP_n) in the sense of [544, Definition 1.1]. The converse statement holds provided that for each index $i \in \mathbb{N}_{n-1}$ the initial values u_0, \ldots, u_{i-1} belong to $D(A_i)$.
- (ii) The uniqueness of analytical solutions of problem (PN) on the region C \ (-∞, 0] can be proved as follows. Let u(·) be an analytical solution of problem (PN) on the region C \ (-∞, 0], with the initial values u_j, resp. u_j, v_j, being zeroes (0 ≤ j ≤ q_n − 1). Then the choice of initial values in (187) enables us to integrate the equation (168) (q_nζ)-times by using the formula (38). Keeping in mind the analyticity of u(·), we easily infer that for each i ∈ N⁰_{n-1} the mappings z → Bu(z), z ∈ C \ (-∞, 0) (z → Bu(z), z ∈ C \ (-∞, 0)] and z → (g_{(qn-qi)ζ} * A_iu)(z), z ∈ C \ (-∞, 0) (z → (g_{(qn-qi)ζ} * A_iu)(z)), z ∈ C \ (-∞, 0)) (z → (g_{(qn-qi)ζ} * A_iu)(z)), z ∈ C \ (-∞, 0)) are well defined and continuous (analytical), as well as that

$$Bu(se^{i\gamma}) + \sum_{i=0}^{n-1} \int_0^{se^{i\gamma}} g_{(q_n - q_i)\zeta}(v) A_i u(se^{i\gamma} - v) dv = 0, \quad s \ge 0, \ \gamma \in (-\pi, \pi),$$

i.e., that

$$Bu(se^{i\gamma}) + \sum_{i=0}^{n-1} (e^{i\gamma})^{(q_n - q_i)\zeta} \int_0^s g_{(q_n - q_i)\zeta}(s - v) A_i u(ve^{i\gamma}) dv = 0, \quad t \ge 0, \ \gamma \in (-\pi, \pi).$$

It is clear that there exists $\gamma \in (-\pi, \pi)$ such that $2re^{-i\gamma\zeta} \in \{\lambda \in e^{i\phi}\Sigma_{(\zeta\pi/2)+\theta} : |\lambda| \ge r\}$. Setting $\phi' := \gamma\zeta$, $u_{\gamma}(s) := u(se^{i\gamma})$, $s \ge 0$, we obtain that $u_{\gamma} \in C([0,\infty):E)$ and

$$e^{-iq_n\phi'}Bu_{\gamma}(s) + \sum_{i=0}^{n-1} e^{-iq_i\phi'}A_i(g_{(q_n-q_i)\zeta} * u_{\gamma})(s) = 0, \quad s \ge 0.$$

On the other hand,

$$P_{(\lambda e^{-i\phi'})^{1/\zeta}} = \lambda^{q_n} e^{-iq_n\phi'} B + \sum_{i=0}^{n-1} \lambda^{q_i} e^{-iq_i\phi'} A_i, \quad \lambda \in \mathbb{C} \smallsetminus \{0\}.$$

Using the previous two equalities and Theorem 2.3.6 (applied to the operators $e^{-iq_i\phi'}A_i$ in place of the operators A_i appearing in the formulation of this theorem), we get that $u_{\gamma}(s) = 0$, $s \ge 0$, which clearly implies that $u(z) = 0, z \in \mathbb{C} \setminus (-\infty, 0)$.

It is also worth noting that Theorem 2.6.2 is an extension of [544, Theorem 2.1] (cf. also [541, Theorem 4.2, p. 168]), where it has been assumed that B = C = I, $\zeta = 1, E$ is a Banach space and $\bigcap_{i=0}^{n} D(A_i)$ is dense in E (in our opinion, the strong continuity in (ii) is very important for the validity of Theorem 2.6.2 and cannot be so simply neglected here (cf. [541, (4.8), p. 169]); also, it ought to be observed that Lemma 2.6.3 is very important for filling some absences in the proof of [541. Theorem 4.2], appearing on the lines 1–6, p. 171 in [541], where the Cauchy formula has been used by assuming the analyticity of mapping $\lambda \mapsto R_{e^{i\phi}\lambda}, \lambda \in \Sigma_{(\pi/2)+\theta}$ $|\lambda| > r$ a priori); observe also that, in the concrete situation of abstract Cauchy problem (ACP_n) , our estimate on the growth rate of $P_{e^{i\phi}}^{-1}$ (cf. the equation (180) with $\nu' < -(n-1)$ is slightly better than the corresponding estimate [541, (4.2)], where it has been required that $\nu' \leq -n$. If B = C = I and $\zeta = 2$, then Theorem 2.6.2 strengthens [544, Theorem 2.1] in a drastic manner. In actual fact, our basic requirement in (i) is that the operator $P_{\lambda}^{-1} = (\lambda^{2n} + \lambda^{2n-2}A_{n-1} + \cdots + A_0)^{-1}$ exists on the region $\{\lambda^{1/2} : \lambda \in e^{i\phi} \Sigma_{(\zeta \pi/2)+\theta} : |\lambda| \ge r\}$, which can be contained in an arbitrary acute angle at vertex (0, 0); on the other hand, in the formulation of [544, Theorem 2.1], T.-J. Xiao and J. Liang require the existence of operator P_{λ}^{-1} for any complex number λ having the modulus greater than or equal to r and belonging to the obtuse angle $e^{i\phi} \Sigma_{(\pi/2)+\theta}$.

Before proceeding further, we would like to note that an illustrative example of application of Theorem 2.5.3 and the conclusions from Remark 2.5.5 to (multi-term) degenerate Cauchy problems will be presented within Example 2.7.6.

2.7. Abstract incomplete degenerate differential equations

Let E be an SCLCS. We start this section by introducing the following definition.

DEFINITION 2.7.1. Suppose that B is a closed linear operator on E and $C \in L(E)$ is an injective operator.

(i) An operator family $(T(t))_{t>0} \subseteq L(E)$ is said to be a pre-(B, C)-regularized semigroup of growth order r > 0 iff $R(T(t)) \subseteq D(B)$, t > 0 and the following holds:

(a) T(t+s)C = T(t)BT(s) for all t, s > 0,

(b) for every $x \in E$, the mapping $t \mapsto T(t)x$, t > 0 is continuous, and (c) the family $\{t^{r}T(t) : t \in (0, 1]\} \subseteq L(E)$ is equicontinuous.

 $(T(t))_{t>0} \subseteq L(E)$ is said to be a (B, C)-regularized semigroup of growth order r > 0 iff, in addition to (a)–(c), we have that:

(d) for every $x \in E$, the mapping $t \mapsto BT(t)x$, t > 0 is continuous, and (e) the family $\{t^r BT(t) : t \in (0, 1]\} \subseteq L(E)$ is equicontinuous.

(ii) Suppose $0 < \gamma \leq \pi/2$, $(T(t))_{t>0}$ is a pre-(B, C)-regularized semigroup of growth order r > 0, and the mapping $t \mapsto T(t)x$, t > 0 has an analytic extension to the sector Σ_{γ} , denoted by the same symbol $(x \in E)$. If

there exists $\omega \in \mathbb{R}$ ($\omega = 0$) such that, for every $\delta \in (0, \gamma)$, the family $\{|z|^r e^{-\omega \operatorname{Re} z} T(z) : z \in \Sigma_{\delta}\} \subset L(E)$ is equicontinuous, then $(T(z))_{z \in \Sigma_{\alpha}} \subset L(E)$ L(E) is said to be an (equicontinuous) analytic pre-(B, C)-regularized semigroup of growth order r.

- (iii) If $(T(t))_{t>0}$ is a (B, C)-regularized semigroup of growth order r > 0, the mappings $t \mapsto T(t)x$, t > 0 and $t \mapsto BT(t)x$, t > 0 admit analytic extensions to the sector Σ_{γ} , denoted by the same symbols $(x \in E)$, and if there exists $\omega \in \mathbb{R}$ ($\omega = 0$) such that, for every $\delta \in (0, \gamma)$, the families $\{|z|^r e^{-\omega \operatorname{Re} z} T(z) : z \in \Sigma_{\delta}\} \subseteq L(E)$ and $\{|z|^r e^{-\omega \operatorname{Re} z} B T(z) : z \in \Sigma_{\delta}\}$ $\Sigma_{\delta} \subseteq L(E)$ are equicontinuous, then $(T(z))_{z \in \Sigma_{\alpha}} \subseteq L(E)$ is said to be an (equicontinuous) analytic (B, C)-regularized semigroup of growth order r.
- (i) Our assumption $R(T(t)) \subset D(B), t > 0$ immediately Remark 2.7.2. implies that BT(t) is a closed linear operator for all t > 0. In the case that E is a webbed bornological space, then the above implies by the closed graph theorem that $BT(t) \in L(E)$ for all t > 0.
- (ii) If $(T(t))_{t>0}$ is a (B, C)-regularized semigroup of growth order r > 0 satis fying additionally that the preassumption BT(t)x = 0, t > 0 implies x = 0, then $(BT(t))_{t \ge 0} \subseteq L(E)$ is a C-regularized semigroup of growth order r > 0 (in the sense of [103, Definition 3.4]). A similar statement can be formulated for the class of analytic (B, C)-regularized semigroups.

If $(T(t))_{t>0} \subseteq L(E)$ is a pre-(B, C)-regularized semigroup of growth order r > 0, then we define the integral generator of $(T(t))_{t>0}$ by

$$\mathcal{A}_B := \left\{ (x, y) \in E \times E : BT(t)x - BT(s)x \\ = B \int_s^t T(r)y \, dr \text{ for all } t, s > 0 \text{ with } t \ge s \right\}.$$

Then it is clear that \mathcal{A}_B is a multivalued linear operator in E and that the local equicontinuity of operator family $(T(t))_{t>0}$ combined with the assumption $B \in$ L(E) implies that \mathcal{A}_B is closed; in the case that there exists a linear manifold Y of E such that $E = \mathcal{A}_B 0 \oplus Y$ (cf. [424, Proposition 1.6.4]), we can single out a single-valued branch of \mathcal{A}_B . It is very difficult to say what will be the complete infinitesimal generator of $(T(t))_{t>0}$ (cf. [103] for the case B=I) in the degenerate case.

The reader may consult [291, Section 1.2] for further information about nondegenerate semigroups of growth order r > 0, i.e., (I, I)-regularized semigroups of growth order r > 0. It is an open problem to find a necessary and sufficient condition for generation of degenerate semigroups of growth order r > 0 in terms of spectral properties of their integral generators.

In the sequel of this section, we shall always assume that A and B are two closed linear operators acting on E, as well as that $CA \subseteq AC$ and $CB \subseteq BC$. Sometimes we use the following condition.

196

(H): A and B are closed linear operators on $E, C \in L(E)$ is injective, $0 \leq \omega < \pi, q \in \mathbb{R}, \mathbb{C} \setminus \overline{\Sigma_{\omega}} \subseteq \rho_C(B, A)$, the families

$$\{(|\lambda|^{-1} + |\lambda|^q)^{-1}(\lambda B - A)^{-1}C : \lambda \notin \Sigma_{\omega'} \cup \{0\}\} \subseteq L(E) \text{ and} \\\{(|\lambda|^{-1} + |\lambda|^q)^{-1}B(\lambda B - A)^{-1}C : \lambda \notin \Sigma_{\omega'} \cup \{0\}\} \subseteq L(E)$$

are equicontinuous for every $\omega < \omega' < \pi$, as well as the mappings $\lambda \mapsto (\lambda B - A)^{-1}Cx, \ \lambda \in \mathbb{C} \setminus \overline{\Sigma_{\omega}}$ and $\lambda \mapsto B(\lambda B - A)^{-1}Cx, \ \lambda \in \mathbb{C} \setminus \overline{\Sigma_{\omega}}$ are continuous for every fixed element $x \in E$.

Since we have assumed that $CA \subseteq AC$ and $CB \subseteq BC$, the analysis contained in Remark 2.5.2(vii) shows that the validity of condition (H) implies that the mappings $\lambda \mapsto (\lambda B - A)^{-1}Cx, \ \lambda \in \mathbb{C} \setminus \overline{\Sigma_{\omega}}$ and $\lambda \mapsto B(\lambda B - A)^{-1}Cx, \ \lambda \in \mathbb{C} \setminus \overline{\Sigma_{\omega}}$ are analytic, indeed, for every fixed element $x \in E$. By $\mathcal{M}_{B,C,q}$ we denote the class consisting of all closed linear operators A' on E, satisfying that the families $\{(|\lambda|^{-1} + |\lambda|^q)^{-1}(\lambda B - A')^{-1}C : \lambda \in (-\infty, 0)\} \subseteq L(E)$ and $\{(|\lambda|^{-1} + |\lambda|^q)^{-1}B(\lambda B - A')^{-1}C : \lambda \in (-\infty, 0)\} \subseteq L(E)$ are equicontinuous, as well as that the mappings $\lambda \mapsto (\lambda B - A')^{-1}Cx, \lambda \in (-\infty, 0)$ and $\lambda \mapsto B(\lambda B - A')^{-1}Cx, \lambda \in (-\infty, 0)$ are continuous for every fixed element $x \in E$. Following A. V. Balakrishnan [44], we introduce the function $f_{t,\gamma}(\lambda)$ by

$$f_{t,\gamma}(\lambda) := \frac{1}{\pi} e^{-t\lambda^{\gamma} \cos \pi\gamma} \sin(t\lambda^{\gamma} \sin \pi\gamma)$$

= $\frac{1}{2\pi i} (e^{-t\lambda^{\gamma} e^{-i\pi\gamma}} - e^{-t\lambda^{\gamma} e^{i\pi\gamma}}), \quad \lambda > 0 \ (t > 0, \ \gamma \in (0, 1/2)).$

This function enjoys the following properties [413]:

1. $|f_{t,\gamma}(\lambda)| \leq \pi^{-1} e^{-\lambda^{\gamma} \varepsilon_{t,\gamma}}, \lambda > 0$, where $\varepsilon_{t,\gamma} := t \cos \pi \gamma > 0$. 2. $|f_{t,\gamma}(\lambda)| \leq \gamma t \lambda^{\gamma} e^{-t\lambda^{\gamma} \sin \varepsilon_{t,\gamma}}, \lambda > 0$. 3. $\int_{0}^{\infty} \lambda^{n} f_{t,\gamma}(\lambda) d\lambda = 0, n \in \mathbb{N}_{0}$.

Set

$$H_n(\omega, z) := \frac{d^n}{dz^n} \exp(-\omega z^{\gamma}), \quad \omega \in \mathbb{C}, \ z \in \mathbb{C} \smallsetminus (-\infty, 0],$$

and, if that makes any sense,

$$W_{\gamma}(t)x := \int_0^\infty f_{t,\gamma}(\lambda)(\lambda B + A)^{-1}Cx\,d\lambda, \quad t > 0, \ x \in E \ (\gamma \in (0, 1/2)).$$

Then the function $H_n(\omega, z)$ is analytic in $\mathbb{C} \setminus (-\infty, 0]$ for every fixed number ω , and entire in \mathbb{C} for every fixed number z (cf. also the proof of [413, Proposition 3.5]).

The main purpose of following theorem is to transmit the assertion of [292, Theorem 2.9.48] to abstract degenerate differential equations (some parts of this theorem cannot be so simply formulated for degenerate differential equations because we do not know what would be the fractional power A_z ($z \in \mathbb{C}_+$) in the newly arisen situation; cf. also [181]).

THEOREM 2.7.3. Let $0 < \gamma < 1/2$, and let $A \in \mathcal{M}_{B,C,q}$ where $q + \gamma > -1$. Suppose that $CA \subseteq AC$ and $CB \subseteq BC$. Then the families $\{(1 + t^{-(q+1)/\gamma})^{-1}W_{\gamma}(t) : t > 0\} \subseteq L(E)$ and $\{(1 + t^{-(q+1)/\gamma})^{-1}BW_{\gamma}(t) : t > 0\} \subseteq L(E)$ are equicontinuous, and there exists an operator family $(\mathbf{W}_{\gamma}(z))_{z \in \Sigma(\pi/2) - \gamma\pi} \subseteq L(E)$, resp.,

 $(\mathbf{W}_{\gamma,B}(z))_{z\in\Sigma_{(\pi/2)-\gamma\pi}} \subseteq L(E)$, such that, for every $x \in E$, the mapping $z \mapsto \mathbf{W}_{\gamma}(z)x$, $z \in \Sigma_{(\pi/2)-\gamma\pi}$, resp., $z \mapsto \mathbf{W}_{\gamma,B}(z)x$, $z \in \Sigma_{(\pi/2)-\gamma\pi}$, is analytic as well as that $\mathbf{W}_{\gamma}(t) = W_{\gamma}(t)$, t > 0, resp., $\mathbf{W}_{\gamma,B}(t) = BW_{\gamma}(t)$, t > 0 (in the sequel, we will not make any difference between $\mathbf{W}_{\gamma}(\cdot)$ and $W_{\gamma}(\cdot)$, $\mathbf{W}_{\gamma,B}(\cdot)$ and $W_{\gamma,B}(\cdot)$). Furthermore, the following holds:

(i) $W_{\gamma}(z_1)BW_{\gamma}(z_2) = W_{\gamma}(z_1 + z_2)C$ for all $z_1, z_2 \in \Sigma_{(\pi/2) - \gamma \pi}$.

(192)

(ii) Let $-1 - \gamma < q \leq -1$. If $(\lambda B - A)^{-1}CA \subseteq A(\lambda B - A)^{-1}C$, $\lambda \in \rho_C(A, B)$ and $x \in \overline{D(A)}$, then

$$\lim_{z \to 0, z \in \Sigma_{(\pi/2) - \gamma \pi - \varepsilon}} BW_{\gamma}(z)x = Cx, \quad \varepsilon \in (0, (\pi/2) - \gamma \pi);$$

if, in addition to this, the condition (H) holds, then we can extend the operator family $W_{\gamma}(\cdot)$ to the sector $\sum_{\frac{\pi}{2}-\omega\gamma}$ and the limit equality (192) remains true for each $x \in \overline{D(A)}$, with the number $(\pi/2) - \gamma\pi$ replaced by $(\pi/2) - \omega\gamma$.

- (iii) Let q > -1. Then (W_γ(z))_{z∈Σ(π/2)-γπ} is an equicontinuous analytic (B, C)-regularized semigroup of growth order (q + 1)/γ. If, additionally, the condition (H) holds, then (W_γ(t))_{t>0} can be extended to an equicontinuous analytic (B, C)-regularized semigroup (W_γ(z))_{z∈Σ(π/2)-γω} of growth order (q+1)/γ. Suppose that the operator B is injective, x ∈ D((B⁻¹A)^[q+2]) ∩ D(B) and (λB A)⁻¹CB ⊆ B(λB A)⁻¹C (λ ∈ ρ_C(A, B)). Then lim_{z→0,z∈Σ(π/2)-γπ-ε} BW_γ(z)x = Cx, ε ∈ (0, (π/2) γπ); if, in addition to this, the condition (H) holds, then the above limit equality remains true with the number (π/2) γπ replaced by (π/2) ωγ.
- (iv) Suppose that q < 0 and $x \in D(A) \cap D(B)$. Then $\lim_{z \to 0, z \in \Sigma_{(\pi/2) - \gamma \pi - \varepsilon}} W_{\gamma}(z) Bx = Cx, \ \varepsilon \in (0, (\pi/2) - \gamma \pi); \ if, \ in \ addition$ to this, the condition (H) holds, then the above limit equality remains true
 with the number $(\pi/2) - \gamma \pi$ replaced by $(\pi/2) - \omega \gamma$.
- (v)(v.1) Let $q + \gamma \ge 0$, $z_0 \in \mathbb{C}_+$, let B be injective, $x \in D((B^{-1}A)^{\lfloor q + \gamma \rfloor + 2}) \cap D(B)$, and let $(\lambda B A)^{-1}CB \subseteq B(\lambda B A)^{-1}C$ $(\lambda \in \rho_C(A, B))$. Then

(193)
$$\lim_{z \to 0, z \in \Sigma_{(\pi/2) - \gamma \pi - \varepsilon}} \frac{BW_{\gamma}(z)x - Cx}{z}$$
$$= -z_0^{\gamma} Cx + \sum_{k=2}^{\lfloor q+\gamma \rfloor + 2} \frac{(-1)^{k-1}}{(k-1)!} H_{k-1}(0, z_0) (z_0 - B^{-1}A)^{k-1} Cx$$
$$-\sin(\pi\gamma) \int_0^\infty \lambda^{\gamma} \frac{B(\lambda + A)^{-1} C(z_0 - B^{-1}A)^{\lfloor q+\gamma \rfloor + 2}x}{(\lambda + z_0)^{\lfloor q+\gamma \rfloor + 2}} d\lambda, \quad \varepsilon \in (0, (\pi/2) - \gamma\pi);$$

if the condition (H) holds, then the formula (193) remains true with the number $(\pi/2) - \gamma \pi$ replaced by $(\pi/2) - \omega \gamma$.

(v.2) Let
$$q + \gamma < 0$$
, $z_0 \in \mathbb{C}_+$ and $x \in D(A) \cap D(B)$. Then

(194)
$$\lim_{z \to 0, z \in \Sigma_{(\pi/2) - \gamma \pi - \varepsilon}} \frac{W_{\gamma}(z)Bx - Cx}{z} = -z_0^{\gamma}Cx$$

$$-\sin(\pi\gamma)\int_0^\infty \lambda^\gamma \frac{(\lambda+A)^{-1}C(z_0B-A)x}{(\lambda+z_0)}d\lambda, \quad \varepsilon \in (0, (\pi/2) - \gamma\pi);$$

if the condition (H) holds, then the formula (194) remains true with the number $(\pi/2) - \gamma \pi$ replaced by $(\pi/2) - \omega \gamma$.

PROOF. The existence of operator families $(\mathbf{W}_{\gamma}(z))_{z \in \Sigma_{(\pi/2)-\gamma\pi}}$ and $(\mathbf{W}_{\gamma,B}(z))_{z \in \Sigma_{(\pi/2)-\gamma\pi}}$, satisfying the properties stated in the first part of formulation of theorem, before the assertion (i), follows similarly as in the case that B = I. Suppose $\delta \in (0, (\pi/2) - \pi\gamma)$ and $r \in \mathfrak{B}$. Arguing as in the proof of [**292**, Theorem 2.9.48], we obtain that there exist $r' \in \mathfrak{B}$, m > 0 and c > 0 such that, for every $x \in E$ and $z \in \Sigma_{\delta}$, we have

(195)
$$r(W_{\gamma}(z)x) \leqslant \left(1+|z|^{-\frac{q+1}{\gamma}}\right)r'(x) \text{ and } r(BW_{\gamma}(z)x) \leqslant \left(1+|z|^{-\frac{q+1}{\gamma}}\right)r'(x).$$

The semigroup property stated in (i) can be proved by means of the resolvent equation from Remark 2.5.2(vii) and direct computations, similar to those appearing in [44, Section 3]. Therefore, $(W_{\gamma}(z))_{z \in \Sigma_{(\pi/2)-\gamma\pi}}$ is an equicontinuous analytic (B, C)regularized semigroup of growth order $(q + 1)/\gamma$, provided that q > -1. Suppose temporarily that the operator B is injective and $B(\lambda B - A)^{-1}C \subseteq (\lambda B - A)^{-1}CB$ $(\lambda \in \rho_C(A, B))$. Then the following equality holds

(196)
$$(\lambda + B^{-1}A)^{-1}Cx = B(\lambda B + A)^{-1}Cx, \quad \lambda > 0, \ x \in D(B).$$

Combined with the identity [413, p. 212, l. 8], (196) implies that, for every $\lambda > 0$ and $z_0 \in \mathbb{C}_+$, we have

$$B(\lambda B + A)^{-1}Cx = \sum_{j=1}^{\lfloor q+2 \rfloor} \frac{(z_0 - B^{-1}A)^{j-1}Cx}{(\lambda + z_0)^j} + \frac{B(\lambda B + A)^{-1}C(z_0 - B^{-1}A)^{\lfloor q+2 \rfloor}Cx}{(\lambda + z_0)^{\lfloor q+2 \rfloor}}$$

so that the proof of limit equality in (iii), on proper subsectors of $\Sigma_{(\pi/2)-\gamma\pi}$, follows in almost the same way as in the proof of [413, Proposition 3.5]. Using the proof of [292, Theorem 2.9.48(v)], we obtain similarly that the limit equality in (v.1) holds on proper subsectors of $\Sigma_{(\pi/2)-\gamma\pi}$. We continue the proof of (iii). Suppose that $\varepsilon > 0$ is sufficiently small. Let q > -1, and let the condition (H) hold. Then one can take numbers $\theta_1 \in (0, \pi - \omega)$ and $\theta_2 \in (\omega - \pi, 0)$ such that $(\pi/2) - \gamma\omega >$ $(\pi/2) - \gamma\pi + \gamma\theta_1 > (\pi/2) - \gamma\omega - \varepsilon$ and $\omega\gamma - (\pi/2) + \varepsilon > \gamma\pi - (\pi/2) + \gamma\theta_2 > \omega\gamma - (\pi/2)$. Set, for every $\theta \in (\omega - \pi, \pi - \omega)$,

$$W_{\theta,\gamma}(z)x := \int_0^\infty f_{z,\gamma}(\lambda)(\lambda B + e^{i\theta}A)^{-1}Cx\,d\lambda, \quad x \in E, \ z \in \Sigma_{(\pi/2) - \gamma\pi}$$

Then it can be simply verified that $(W_{\theta,\gamma}(z))_{z\in\Sigma_{(\pi/2)-\gamma\pi}}$ is an equicontinuous analytic (B,C)-regularized semigroup of growth order $(q+1)/\gamma$. Define, for every $x \in E$,

$$W_{\gamma}(z)x := \begin{cases} W_{\gamma}(z)x, & \text{if } z \in \Sigma_{(\pi/2) - \gamma \pi}, \\ W_{\theta_1, \gamma}(ze^{-i\theta_1 \gamma})x, & \text{if } z \in e^{i\theta_1 \gamma} \Sigma_{(\pi/2) - \gamma \pi}, \\ W_{\theta_2, \gamma}(ze^{-i\theta_2 \gamma})x, & \text{if } z \in e^{i\theta_2 \gamma} \Sigma_{(\pi/2) - \gamma \pi}. \end{cases}$$

By Cauchy formula, we have that $W_{\gamma}(z)x = W_{\theta_1,\gamma}(ze^{-i\theta_1\gamma})$, if $z \in \Sigma_{(\pi/2)-\gamma\pi} \cap$ $e^{i\theta_1\gamma}\Sigma_{(\pi/2)-\gamma\pi}$, and $W_{\gamma}(z)x = W_{\theta_2,\gamma}(ze^{-i\theta_2\gamma})$, if $z \in \Sigma_{(\pi/2)-\gamma\pi} \cap e^{i\theta_2\gamma}\Sigma_{(\pi/2)-\gamma\pi}$, whence we may conclude that the operator family $(W_{\gamma}(z))_{z \in \Sigma_{\frac{\pi}{2}-\omega_{\gamma}}} \subseteq L(E)$ is well defined. We define the operator family $(BW_{\gamma}(z))_{z \in \Sigma_{\frac{\pi}{4}-\omega\gamma}} \subseteq L(E)$ similarly. Then it is checked at once that $(W_{\gamma}(z))_{z \in \Sigma_{\frac{\pi}{2}-\omega\gamma}}$ is an equicontinuous analytic (B,C)-regularized semigroup of growth order $(q+1)/\gamma$. The way of construction of $(W_{\gamma}(z))_{z \in \Sigma_{\frac{\pi}{4} - \omega_{\gamma}}} \subseteq L(E)$ shows that the limit equality stated in (iii) continues to hold for each $x \in D((B^{-1}A)^{\lfloor q+2 \rfloor}) \cap D(B)$, with the number $(\pi/2) - \gamma \pi$ replaced by $(\pi/2) - \gamma \omega$ (if $q \leq -1$, then we define $(W_{\gamma}(z))_{z \in \Sigma_{\frac{\pi}{2} - \omega \gamma}} \subseteq L(E)$ and $(BW_{\gamma}(z))_{z\in\Sigma_{\frac{\pi}{2}-\omega_{\gamma}}}\subseteq L(E)$ in the same way as above, showing also that any considered limit equality from (ii)–(v) continues to hold with the number $(\pi/2) - \gamma \pi$ replaced by $(\pi/2) - \gamma \omega$). The proof of (iii) is therefore completed. Suppose now that $-1 - \gamma < q \leq -1$. Then an insignificant modification of the proof of [410. Theorem 5.5.1(iv)] shows that $\lim_{t\to 0+} BW_{\gamma}(t)x = Cx$, provided that $x \in E$ satisfies $\lim_{\lambda \to +\infty} \lambda B(\lambda B + A)^{-1}Cx = Cx$. Since we have assumed that A commutes with $(\lambda B+A)^{-1}C$ $(\lambda \in \rho_C(A, B))$, and that $\lambda B(\lambda B+A)^{-1}Cx - Cx = A(\lambda B+A)^{-1}Cx = A(\lambda B+A)^{-1}Cx$ $(\lambda B + A)^{-1}CAx, x \in D(A), \lambda > 0$, one has $\lim_{t\to 0+} BW_{\gamma}(t)x = Cx, x \in D(A)$. Using (195), it readily follows that $\lim_{t\to 0+} BW_{\gamma}(t)x = Cx, x \in \overline{D(A)}$, so that the limit equality (192) follows from an application of [296, Theorem 3.4(ii)]. The limit equalities stated in (iv) and (v.2), on proper subsectors of $\Sigma_{(\pi/2)-\gamma\pi}$, can be proved by using the equality

$$(\lambda B + A)^{-1}CBx = \frac{Cx}{\lambda + z_0} + \frac{(\lambda B + A)^{-1}C}{\lambda + z_0}(z_0Bx - Ax),$$

which holds for $z_0 \in \mathbb{C}_+$, $\lambda > 0$, $x \in D(A) \cap D(B)$, and a slight modification of the proofs of [413, Lemma 3.4, Proposition 3.5]. The proof of the theorem is thereby complete.

In Theorem 2.7.4, we use the same terminology as in the formulation and proof of Theorem 2.7.3. We continue our previous analysis by investigating the existence of solutions of abstract incomplete degenerate Cauchy problems

$$(FP_{\alpha_1,\beta_1,\theta}): \begin{cases} u \in C^{\infty}((0,\infty):E), \\ D_{-}^{\alpha_1}BD_{-}^{\beta_1}u(s) = e^{i\theta/\gamma}Au(s), \ s > 0, \\ \lim_{s \to 0+} Bu(s) = Cx, \\ \text{the set } \{Bu(s):s > 0\} \text{ is bounded in } E \end{cases}$$

and

$$(FP_{\alpha_1,\beta_1,\theta})': \begin{cases} u \in C^{\infty}((0,\infty):E), \\ D_{-}^{\alpha_1} B D_{-}^{\beta_1} u(s) = e^{i\theta/\gamma} A u(s), \ s > 0, \\ \lim_{s \to 0+} B u(s) = C x, \\ \text{the sets } \{(1 + s^{-(q+1)/\gamma})^{-1} u(s): s > 0\} \\ \text{and } \{(1 + s^{-(q+1)/\gamma})^{-1} B u(s): s > 0\} \text{ are bounded in } E. \end{cases}$$

THEOREM 2.7.4. Let $0 < \gamma < 1/2$, and let $A \in \mathcal{M}_{B,C,q}$ where $q + \gamma > -1$. Suppose that $CA \subseteq AC$, $CB \subseteq BC$, $\alpha_1 \ge 0$, $\beta_1 \ge 0$, $\alpha_1 + \beta_1 = 1/\gamma$ and $\theta \in (\gamma \pi - (\pi/2), (\pi/2) - \gamma \pi)$. Then the following holds:

(i) Let q > -1. Denote by Ω_{θ,γ}, resp. Ψ_γ, the set consisting of those elements x ∈ E for which lim_{s→0+} BW_γ(se^{iθ})x = Cx, resp., lim_{z→0,z∈Σ(π/2)-γπ-ε} BW_γ(z)x = Cx for all ε ∈ (0, (π/2) - γπ). Then, for every x ∈ Ω_{θ,γ}, the incomplete abstract degenerate Cauchy problem (FP_{α1,β1,θ}) has a solution u(s) = W_γ(se^{iθ})x, s > 0, which can be analytically extended to the sector Σ_{(π/2)-γπ-|θ|}. If, additionally, x ∈ Ψ_γ, then for every δ ∈ (0, (π/2) - γπ - |θ|) and j ∈ N₀, we have that the set {z^jBu^(j)(z) : z ∈ Σ_δ} is bounded in E. Assume that the condition (H) holds. Then the solution s ↦ u(s), s > 0 can be analytically continued to the sector Σ_{(π/2)-γω}; if, in addition to this,

$$\lim_{z \to 0, z \in \Sigma_{(\pi/2) - \gamma \omega - \varepsilon}} BW_{\gamma}(z) x = Cx \text{ for all } \varepsilon \in (0, (\pi/2) - \gamma \omega),$$

then for every $\delta \in (0, (\pi/2) - \gamma \omega - |\theta|)$ and $j \in \mathbb{N}_0$, we have that the set $\{z^j Bu^{(j)}(z) : z \in \Sigma_{\delta}\}$ is bounded in E.

(ii) Let -1 - γ ≤ q ≤ -1, (λB - A)⁻¹CA ⊆ A(λB - A)⁻¹C, λ ∈ ρ_C(A, B), and let x ∈ D(A). Then the incomplete abstract degenerate Cauchy problem (FP_{α1,β1,θ})' has a solution u(s) = W_γ(se^{iθ})x, s > 0, which can be analytically extended to the sector Σ_{(π/2)-γπ-|θ|}. Moreover, for every δ ∈ (0, (π/2)-γπ-|θ|) and j ∈ N₀, the sets {|z|^j(1+|z|^{-(q+1)/γ})⁻¹u^(j)(z) : z ∈ Σ_δ} and {|z|^j(1+|z|^{-(q+1)/γ})⁻¹Bu^(j)(z) : z ∈ Σ_δ} are bounded in E. If, additionally, the condition (H) holds, then the above statements remain true with the number (π/2) - γπ replaced by (π/2) - ωγ.

PROOF. The proof of theorem almost completely follows from the arguments used in those of [292, Theorem 2.9.48] and [103, Theorem 3.5(i)/b'], and the only non-trivial thing that should be explained here is the way how we can prove that, for every $\delta \in (0, (\pi/2) - \gamma \pi - |\theta|)$ and $j \in \mathbb{N}_0$, the solution $s \mapsto u(s)$, s > 0 has the property that the sets $\{|z|^j(1 + |z|^{-(q+1)/\gamma})^{-1}u^{(j)}(z) : z \in \Sigma_{\delta}\}$ and $\{|z|^j(1 + |z|^{-(q+1)/\gamma})^{-1}Bu^{(j)}(z) : z \in \Sigma_{\delta}\}$ are bounded in E (cf. (ii)). In order to show this, observe first that for each $x \in E$, $z \in \Sigma_{\frac{\pi}{2} - \gamma \pi}$ and $j \in \mathbb{N}$ we have

(197)
$$\frac{d^j}{dz^j}W_{\gamma}(z)x = \frac{1}{2\pi i}\int_0^\infty (-\lambda^{\gamma}e^{-i\pi\gamma})^j e^{-z\lambda^{\gamma}e^{-i\pi\gamma}}(\lambda B + A)^{-1}Cx\,d\lambda$$

$$-\frac{1}{2\pi i}\int_0^\infty (-\lambda^\gamma e^{i\pi\gamma})^j e^{-z\lambda^\gamma e^{i\pi\gamma}} (\lambda B+A)^{-1} Cx\,d\lambda$$

Hence, $B\frac{d^j}{dz^j}W_{\gamma}(z)x = \frac{d^j}{dz^j}BW_{\gamma}(z)x$ $(x \in E, z \in \Sigma_{\frac{\pi}{2}-\gamma\pi}, j \in \mathbb{N})$, whence it easily follows that $Bu^{(j)}(z) = (Bu)^{(j)}(z)$ $(z \in \Sigma_{\frac{\pi}{2}-\gamma\pi}, j \in \mathbb{N})$. Having in mind this equality, the boundedness of sets $\{|z|^j(1+|z|^{-(q+1)/\gamma})^{-1}u^{(j)}(z) : z \in \Sigma_{\delta}\}$ and $\{|z|^j(1+|z|^{-(q+1)/\gamma})^{-1}Bu^{(j)}(z) : z \in \Sigma_{\delta}\}$ can be proved by using (197) and a direct computation involving the estimates used in the proof of [**291**, Theorem 1.4.15].

In this monograph, we will not consider again the assertions of [**292**, Theorem 2.9.39, Theorem 2.9.40, Theorem 2.9.58, Theorem 2.9.60] for degenerate differential equations. In the following theorem, we will focus entirely on the assertion of [**292**, Theorem 2.9.51(iii)].

THEOREM 2.7.5. (cf. [292, Theorem 2.9.51(iii)] for non-degenerate case) Suppose that the condition (H) holds, as well as that $CA \subseteq AC$ and $CB \subseteq BC$.

(i) Let -1 < q < (-1)/2, and let $(\lambda B - A)^{-1}CA \subseteq A(\lambda B - A)^{-1}C$ $(\lambda \in \rho_C(A, B))$. Then, for every $x \in D(A)$, the incomplete abstract degenerate Cauchy problem

$$(P_{2,q,B}): \begin{cases} u \in \mathcal{A}(\Sigma_{(\pi/2)-(\omega/2)}:E), \quad Bu \in \mathcal{A}(\Sigma_{(\pi/2)-(\omega/2)}:E), \\ Bu''(z) = \frac{d^2}{dz^2}Bu(z) = Au(z), \quad z \in \Sigma_{(\pi/2)-(\omega/2)}, \\ \lim_{z \to 0, z \in \Sigma_{\delta}}Bu(z) = Cx, \quad for \; every \; \delta \in (0, (\pi/2) - (\omega/2)), \\ the \; sets \; \{(1+|z|^{-(2q+2)})^{-1}u(z):z \in \Sigma_{\delta}\} \quad and \\ \{(1+|z|^{-(2q+2)})^{-1}Bu(z):z \in \Sigma_{\delta}\} \quad are \; bounded \; in \; E \\ for \; every \; \delta \in (0, (\pi/2) - (\omega/2)), \end{cases}$$

has a solution u(z) $(z \in \Sigma_{(\pi/2)-(\omega/2)})$. Moreover, for every $\delta \in (0, (\pi/2) - (\omega/2))$ and $j \in \mathbb{N}_0$, we have that the sets $\{|z|^j(1+|z|^{-(2q+2)})^{-1}u^{(j)}(z) : z \in \Sigma_{\delta}\}$ and $\{|z|^j(1+|z|^{-(2q+2)})^{-1}(Bu)^{(j)}(z) : z \in \Sigma_{\delta}\}$ are bounded in E. (ii) Let $-1 \ge q > (-3)/2$, and let $(\lambda B - A)^{-1}CA \subseteq A(\lambda B - A)^{-1}C$ $(\lambda \in \rho_C(A, B))$. Then, for every $x \in \overline{D(A)}$, the problem $(P_{2,q,B})$ has a solution u(z) $(z \in \Sigma_{(\pi/2)-(\omega/2)})$. Moreover, for every $\delta \in (0, (\pi/2) - (\omega/2))$ and $j \in \mathbb{N}_0$, we have that the sets $\{|z|^j(1+|z|^{-(2q+2)})^{-1}u^{(j)}(z) : z \in \Sigma_{\delta}\}$ are bounded in E.

PROOF. Suppose first that q = -1 and $(\lambda B - A)^{-1}CA \subseteq A(\lambda B - A)^{-1}C$ $(\lambda \in \rho_C(A, B))$; cf. (ii). Let $0 < \delta' < \delta < (\pi - \omega)/2$, $1/2 > \gamma_0 > \delta/(\pi - \omega)$ and $\theta \in (\omega - \pi, (-\delta)/\gamma_0)$. Then, for every $\gamma \in (\gamma_0, 1/2)$, we have $\theta \in (\omega - \pi, (-\delta)/\gamma)$ and $\gamma > \delta/(\pi - \omega)$. Suppose that $\varepsilon \in (0, \omega/2)$ is sufficiently small. Define, for every $\gamma \in (\gamma_0, 1/2)$ and $x \in E$,

(198)
$$F_{\gamma}(\lambda)x := \begin{cases} \int_{0}^{\infty} \frac{e^{i\theta\gamma} \sin(\gamma\pi)v^{\gamma}(vB+e^{i\theta}A)^{-1}Cx\,dv}{\pi(\lambda e^{i\theta\gamma}+v^{\gamma}\cos\pi\gamma)^{2}+v^{2\gamma}\sin^{2}\gamma\pi}, & \text{if } \arg(\lambda) \in (-\varepsilon, (\pi/2)+\delta), \\ \int_{0}^{\infty} \frac{e^{-i\theta\gamma} \sin(\gamma\pi)v^{\gamma}(vB+e^{-i\theta}A)^{-1}Cx\,dv}{\pi(\lambda e^{-i\theta\gamma}+v^{\gamma}\cos\pi\gamma)^{2}+v^{2\gamma}\sin^{2}\gamma\pi}, & \text{if } \arg(\lambda) \in (-(\pi/2)-\delta,\varepsilon). \end{cases}$$

If $x \in E$ and $\arg(\lambda) \in (-\varepsilon, (\pi/2) + \delta)$, resp., $\arg(\lambda) \in (-(\pi/2) - \delta, \varepsilon)$, then it is very simple to prove that

$$\int_0^\infty e^{-\lambda e^{i\theta\gamma}t} W_{\theta,\gamma}(t) x \, dt = \frac{\sin\gamma\pi}{\pi} \int_0^\infty \frac{v^\gamma (vB + e^{i\theta}A)^{-1} Cx}{(\lambda e^{i\theta\gamma} + v^\gamma \cos\pi\gamma)^2 + v^{2\gamma} \sin^2\gamma\pi} dv,$$

resp.,

$$\int_0^\infty e^{-\lambda e^{-i\theta\gamma}t} W_{-\theta,\gamma}(t) x \, dt = \frac{\sin\gamma\pi}{\pi} \int_0^\infty \frac{v^\gamma (vB + e^{-i\theta}A)^{-1} Cx}{(\lambda e^{-i\theta\gamma} + v^\gamma \cos\pi\gamma)^2 + v^{2\gamma} \sin^2\gamma\pi} dv,$$

where we use the notation from the proof of Theorem 2.7.3. Denote, with a little abuse of notation, $\Gamma_{\vartheta} := \{te^{i\vartheta} : t \ge 0\} \ (\vartheta \in (-\pi, \pi])$. Since, for every $x \in E$,

$$e^{i\theta\gamma} \int_0^\infty e^{-\lambda e^{i\theta\gamma}t} W_{\theta,\gamma}(t) x \, dt = \int_{\Gamma_{\theta\gamma}} e^{-\lambda v} W_{\theta,\gamma}(v e^{-i\theta\gamma}) x \, dv = \int_{\Gamma_{\theta\gamma}} e^{-\lambda v} W_{\gamma}(v) x \, dv$$

and

$$e^{-i\theta\gamma} \int_0^\infty e^{-\lambda e^{-i\theta\gamma}t} W_{-\theta,\gamma}(t) x \, dt = \int_{\Gamma_{-\theta\gamma}} e^{-\lambda v} W_{-\theta,\gamma}(v e^{i\theta\gamma}) x \, dv$$
$$= \int_{\Gamma_{-\theta\gamma}} e^{-\lambda v} W_{\gamma}(v) x \, dv$$

(cf. the construction of $(W_{\gamma}(z))_{z \in \Sigma_{(\pi/2)-\omega\gamma}}$, and observe that $|\theta\gamma| \in (\delta, \gamma(\pi - \omega))$), the Cauchy formula yields that

(199)
$$e^{i\theta\gamma} \int_0^\infty e^{-\lambda e^{i\theta\gamma}t} W_{\theta,\gamma}(t) x \, dt = e^{-i\theta\gamma} \int_0^\infty e^{-\lambda e^{-i\theta\gamma}t} W_{-\theta,\gamma}(t) x \, dt, \quad \lambda \in \Sigma_{\varepsilon}.$$

This, in turn, implies that the function $\lambda \mapsto F_{\gamma}(\lambda)x$, $\lambda \in \Sigma_{(\pi/2)+\delta}$ is well defined, analytic and bounded by $\operatorname{Const}_{\delta'} |\lambda|^{-1}$ on $\Sigma_{(\pi/2)+\delta'}$ ($x \in E$). Further on, with the help of [**296**, Theorem 3.4] (cf. also [**27**, Theorem 2.6.1] for the Banach space case) and the uniqueness theorem for the Laplace transform we can prove that

$$W_{\gamma}(z)x = \frac{1}{2\pi i} \int_{\Gamma_{\delta',z}} e^{\lambda z} F_{\gamma}(\lambda) x \, d\lambda, \quad x \in E, \ z \in \Sigma_{\delta'}, \ \gamma \in (\gamma_0, 1/2),$$

where $\Gamma_{\delta',z} := \Gamma_{\delta',z,1} \cup \Gamma_{\delta',z,2}$, $\Gamma_{\delta',z,1} := \{re^{i((\pi/2)+\delta')} : r \ge |z|^{-1}\} \cup \{|z|^{-1}e^{i\vartheta} : \vartheta \in [0, (\pi/2) + \delta']\}$ and $\Gamma_{\delta',z,2} := \{re^{-i((\pi/2)+\delta')} : r \ge |z|^{-1}\} \cup \{|z|^{-1}e^{i\vartheta} : \vartheta \in [-(\pi/2) - \delta', 0]\}$ are oriented counterclockwise. Applying the dominated convergence theorem, we get that

$$(200) \qquad \lim_{\gamma \to \frac{1}{2} -} W_{\gamma}(z)x = \frac{e^{i\theta/2}}{2\pi^{2}i} \int_{\Gamma_{\delta',z,1}} e^{\lambda z} \int_{0}^{\infty} \frac{v^{1/2}(vB + e^{i\theta}A)^{-1}Cx}{\lambda^{2}e^{i\theta} + v} dv \, d\lambda$$
$$+ \frac{e^{-i\theta/2}}{2\pi^{2}i} \int_{\Gamma_{\delta',z,2}} e^{\lambda z} \int_{0}^{\infty} \frac{v^{1/2}(vB + e^{-i\theta}A)^{-1}Cx}{\lambda^{2}e^{-i\theta} + v} dv \, d\lambda$$
$$:= W_{1/2}(z)x, \quad x \in E, \ z \in \Sigma_{\delta'}.$$

Define $F_{1/2}(\lambda)$ by replacing the number γ with the number 1/2 in definition of $W_{\gamma}(\lambda)$. Then, for every $x \in E$, the function $\lambda \mapsto F_{1/2}(\lambda)x$, $\lambda \in \Sigma_{(\pi/2)+\delta}$ is well defined and analytic on $\Sigma_{(\pi/2)+\delta}$ because $F_{1/2}(\lambda)x = \lim_{\gamma \to 1/2^-} F_{\gamma}(\lambda)x$, $\lambda \in \Sigma_{(\pi/2)+\delta}$

and the convergence is uniform on compacts of $\Sigma_{(\pi/2)+\delta}$ (cf. [296, Lemma 3.3]). Furthermore, we can argue as in the proof of estimate [292, (334)] so as to conclude that for each $q \in \circledast$ there exists $r_q \in \circledast$ such that $q(F_{1/2}(\lambda)x) \leq r_q(x) \operatorname{Const}_{\delta'} |\lambda|^{-1}$, $\lambda \in \Sigma_{(\pi/2)+\delta'}, x \in E$. Now it becomes apparent that we can define the operator family $(W_{1/2}(z))_{z \in \Sigma_{(\pi-\omega)/2}} \subseteq L(E)$, equicontinuous on any proper subsector of $\Sigma_{(\pi-\omega)/2}$, satisfying additionally that $\lim_{\gamma \to \frac{1}{2}} W_{\gamma}(z) x = W_{1/2}(z) x, z \in \Sigma_{(\pi-\omega)/2}$, $x \in E$, and that the mapping $z \mapsto W_{1/2}(z)x, z \in \Sigma_{(\pi-\omega)/2}$ is analytic for all $x \in E$. Let us prove that for each $x \in \overline{D(A)}$ the function $u(z) := W_{1/2}(z)x, z \in \Sigma_{(\pi-\omega)/2}$ is a solution of problem $(P_{2,q,B})$ with q = -1. Suppose first that $x \in D(A)$. Then the condition $(\lambda B - A)^{-1}CA \subseteq A(\lambda B - A)^{-1}C$ $(\lambda \in \rho_C(A, B))$ in combination with the closedness of A shows that $W_{1/2}(z)Ax = AW_{1/2}(z)x, z \in \Sigma_{(\pi-\omega)/2}$. By the foregoing, we also have that the operator family $(BW_{1/2}(z))_{z\in\Sigma_{(\pi-\omega)/2}}\subseteq L(E)$ is equicontinuous and the mapping $z \mapsto BW_{1/2}(z)x, z \in \Sigma_{(\pi-\omega)/2}$ is analytic $(x \in E)$, as well as that $(d^2/dz^2)BW_{1/2}(z)x = B(d^2/dz^2)W_{1/2}(z)x, z \in \Sigma_{(\pi-\omega)/2}, x \in E$ and $BW_{1/2}(z) \in L(E), (d^2/dz^2)BW_{1/2}(z) \in L(E), B(d^2/dz^2)W_{1/2}(z) \in L(E)$ for $z \in \Sigma_{(\pi-\omega)/2}$. Furthermore, the dominated convergence theorem yields that

$$\lim_{\gamma \to 1/2-} BW_{\gamma}^{(n)}(z) = BW_{1/2}^{(n)}(z)x, \quad z \in \Sigma_{(\pi-\omega)/2}, \ x \in E, \ n \in \mathbb{N}_0.$$

By Theorem 2.7.3(vi.2) and definition of modified Liouville right-sided fractional derivatives, we get that

(201)
$$\frac{d^2}{ds^2} \int_0^\infty g_{3-\frac{1}{\gamma}}(t) BW'_{\gamma}(t+s) x \, dt = -W_{\gamma}(s) Ax, \quad s > 0, \ \gamma \in (\gamma_0, 1/2),$$

i.e.,

$$\int_0^\infty g_{3-\frac{1}{\gamma}}(t) BW_{\gamma}'''(t+s) x \, dt = -W_{\gamma}(s) Ax, \quad s > 0, \ \gamma \in (\gamma_0, 1/2).$$

The integration by parts leads us to the following

$$-\int_{0}^{\infty}g_{4-\frac{1}{\gamma}}(t)BW_{\gamma}^{(iv)}(t+s)x\,dt = -W_{\gamma}(s)Ax, \quad s > 0, \ \gamma \in (\gamma_{0}, 1/2).$$

Using again the dominated convergence theorem, we obtain by letting $\gamma \to 1/2-$ that

$$\int_0^\infty t B W_{1/2}^{(iv)}(t+s) x \, ds = W_{1/2}(s) A x, \quad s > 0.$$

which clearly implies after an application of the partial integration that $BW''_{1/2}(s) = AW_{1/2}(s)x$, s > 0. By the uniqueness theorem for analytic functions, this equality continues to hold for all $z \in \Sigma_{(\pi-\omega)/2}$. Further on, we can compute $F_{\gamma}(\lambda)x$ for $\lambda > 0$ ($\gamma \in (\gamma_0, 1/2)$) by plugging $\theta = 0$ in either of two terms appearing in (198). Taking the limit as $\gamma \to 1/2-$, we get that $F_{1/2}(\lambda)x = \pi^{-1} \int_0^\infty v^{1/2} (\lambda^2 + \nu)^{-1} (vB + A)^{-1} Cx \, dv$, $\lambda > 0$. Now we will prove that $BW_{1/2}(z)x - Cx \to 0$ as $z \to 0$, $z \in \Sigma_{(\pi/2)+\delta'}$. Due to Theorem 1.4.10(iii), it suffices to show that $\lim_{\lambda\to+\infty} \lambda BF_{1/2}(\lambda)x = Cx$. This follows by applying the dominated convergence

theorem on the integral appearing on the right-hand side of the equality

$$\lambda BF_{1/2}(\lambda)x - Cx = \frac{1}{\pi} \int_0^\infty \frac{v^{1/2}\lambda}{\lambda^2 + v} \Big[B(vB + A)^{-1}Cx - \frac{Cx}{v} \Big] dv,$$

and by observing that

(202)
$$q\left(B(vB+A)^{-1}Cx - \frac{Cx}{v}\right) = \frac{1}{v}q((vB+A)^{-1}CAx) = O(v^{-2}), \quad q \in \mathbb{B}, v > 1$$

and

(203)
$$q\left(B(vB+A)^{-1}Cx - \frac{Cx}{v}\right) = O(v^{-1}), \quad q \in \circledast, \ v \in (0,1).$$

Keeping in mind the Cauchy integral formula, the proof of (ii) follows immediately in the case that q = -1 and $x \in D(A)$. The proof of (ii) in the case that q = -1and $x \in \overline{D(A)}$ follows from the standard limit procedure. If (-3)/2 < q < -1, then for each $\varepsilon > 0$ the condition (H), with the operators A and B replaced respectively by $A + \varepsilon B$ and B, holds with q = -1 and the same spectral angle ω ; in this case, the proof of (ii) can be deduced by slightly modifying the corresponding part of the proof of [**292**, Theorem 2.9.51(iii.2)]. The proof of (i) is very similar to that of (ii); for the sake of completeness, we will include almost all relevant details. As in the proof of (ii), it will be assumed that $0 < \delta' < \delta < (\pi - \omega)/2, 1/2 > \gamma_0 > \delta/(\pi - \omega),$ $\theta \in (\omega - \pi, (-\delta)/\gamma_0)$ and that $\varepsilon \in (0, \omega/2)$ is sufficiently small. Due to the proof of [**292**, Theorem 2.9.48] (cf. also the estimate (195)), we have that, for every $\theta' \in (\omega - \pi, \pi - \omega)$ and $\gamma \in (q + 1, 1/2)$, the mapping $t \mapsto W_{\theta',\gamma}(t)x, t \ge 0$ ($x \in E$) satisfies the condition (P1), as well as that

$$\mathcal{L}(W_{\theta',\gamma}(t)x)(\mu) = \frac{\sin\gamma\pi}{\pi} \int_0^\infty \frac{v^{\gamma}(vB + e^{i\theta'}A)^{-1}Cx}{(\mu + v^{\gamma}\cos\pi\gamma)^2 + v^{2\gamma}\sin^2\gamma\pi} dv, \quad \mu \in \mathbb{C}_+, \ x \in E,$$

and that for each $q \in \circledast$ there exists $r_q \in \circledast$ such that

$$q(\mathcal{L}(W_{\theta',\gamma}(t)x)(\mu)) = r_q(x)O(|\mu|^{-1} + |\mu|^{\frac{q+1}{\gamma}-1}), \ \mu \in \Sigma_{\frac{\pi}{2}-\varepsilon}, \ x \in E \ (\varepsilon \in (0,\pi/2)).$$

This implies that, for every $\gamma \in (\max(\gamma_0, q+1), 1/2)$ and $x \in E$, we can define $F_{\gamma}(\lambda)x$ through (198). If $x \in E$ and $\arg(\lambda) \in (-\varepsilon, (\pi/2) + \delta)$, resp., $\arg(\lambda) \in (-(\pi/2) - \delta, \varepsilon)$, then the equality (410), resp., (411), continues to hold for those values of parameter γ . Furthermore, (199) holds for any $\lambda \in \Sigma_{\varepsilon}$; hence, the function $\lambda \mapsto F_{\gamma}(\lambda)x, \lambda \in \Sigma_{(\pi/2)+\delta}$ is well defined, analytic and bounded by $\operatorname{Const}_{\delta'}(|\lambda|^{-1} + |\lambda|^{((q+1)/\gamma)-1})$ on $\Sigma_{(\pi/2)+\delta'}$ ($x \in E, \gamma \in (\max(\gamma_0, q+1), 1/2)$). If $x \in E, z \in \Sigma_{\delta'}$, $\gamma \in (\max(\gamma_0, q+1), 1/2)$ and $\zeta \ge 0$, then we define

$$W_{\gamma}^{(-\zeta)}(z)x := \frac{1}{2\pi i} \int_{\Gamma_{\delta',z}^{\omega'}} e^{\lambda z} \lambda^{-\zeta} F_{\gamma}(\lambda) x \, d\lambda,$$

where $\omega' > 0$ is taken arbitrarily, $\Gamma_{\delta',z}^{\omega'} := \Gamma_{\delta',z,1}^{\omega'} \cup \Gamma_{\delta',z,2}^{\omega'}$, $\Gamma_{\delta',z,1}^{\omega'} := \{\omega' + re^{i((\pi/2) + \delta')} : r \ge |z|^{-1}\} \cup \{\omega' + |z|^{-1}e^{i\vartheta} : \vartheta \in [0, (\pi/2) + \delta']\}$ and $\Gamma_{\delta',z,2}^{\omega'} := \{\omega' + re^{-i((\pi/2) + \delta')} : r \ge |z|^{-1}\} \cup \{\omega' + |z|^{-1}e^{i\vartheta} : \vartheta \in [-(\pi/2) - \delta', 0]\}$ are oriented counterclockwise. By Theorem 1.4.10, we get that $\mathcal{L}(W_{\gamma}^{(-1)}(t)x)(\lambda) = \lambda^{-1}F_{\gamma}(\lambda)x, \lambda > 0, x \in E$; using this fact, as well as the equality $\mathcal{L}(W_{\gamma}(t)x)(\lambda) = F_{\gamma}(\lambda)x, \lambda > 0, x \in E$, the uniqueness

205

theorem for Laplace transform and the uniqueness theorem for analytic functions. we obtain that $(d/dz)W_{\gamma}^{(-1)}(z)x = W_{\gamma}(z)x, x \in E, z \in \Sigma_{\delta'}$ ($\gamma \in (\max(\gamma_0, q + \omega))$ (1, 1/2)). On the other hand, the dominated convergence theorem yields that $(d/dz)W_{\gamma}^{(-1)}(z)x = W_{\gamma}^{(0)}(z)x$, so that $W_{\gamma}^{(0)}(z)x = W_{\gamma}(z)x$, $x \in E, z \in \Sigma_{\delta'}$ ($\gamma \in C_{\delta'}$ $(\max(\gamma_0, q+1), 1/2))$. Define $F_{1/2}(\lambda)$ $(\lambda \in \Sigma_{(\pi/2)+\delta})$ and $(W_{1/2}(z))_{z \in \Sigma_{s'}} \subseteq L(E)$ in exactly the same vein as in the proof of (ii). As before, we have that, for every $x \in E$, the function $\lambda \mapsto F_{1/2}(\lambda)x, \lambda \in \Sigma_{(\pi/2)+\delta}$ is well defined and analytic on $\Sigma_{(\pi/2)+\delta}$ because $F_{1/2}(\lambda)x = \lim_{\gamma \to 1/2-} F_{\gamma}(\lambda)x, \lambda \in \Sigma_{(\pi/2)+\delta}$ and the convergence is uniform on compacts of $\Sigma_{(\pi/2)+\delta}$; furthermore, for each $q \in \circledast$ there exists $r_q \in \circledast$ such that $q(F_{1/2}(\lambda)x) \leqslant r_q(x) \operatorname{Const}_{\delta'}(|\lambda|^{-1} + |\lambda|^{2q+1}), \lambda \in \Sigma_{(\pi/2)+\delta'}, x \in E.$ Then the limit equality (200) continues to hold (with the replacements of contours $\Gamma_{\delta',z,1}$, $\Gamma_{\delta',z,2}$ with $\Gamma_{\delta',z,1}^{\omega'}$, $\Gamma_{\delta',z,2}^{\omega'}$, respectively), the operator families $(W_{1/2}(z))_{z\in\Sigma(\pi-\omega)/2}\subseteq L(E)$ and $(BW_{1/2}(z))_{z \in \Sigma_{(\pi-\omega)/2}} \subseteq L(E)$ can be defined in the very obvious way, and we have that $\lim_{\gamma \to 1/2^-} BW_{\gamma}^{(n)}(z)x = BW_{1/2}^{(n)}(z)x$ for all $x \in E, z \in \Sigma_{(\pi-\omega)/2}$ and $n \in \mathbb{N}_0$. Arguing as in the proof of [292, Theorem 2.9.51(iii.1)], we can show that, for every $j \in \mathbb{N}_0$, the families $\{|z|^j (1+|z|^{-(2q+2)})^{-1} W_{1/2}^{(j)}(z) : z \in \Sigma_{\delta'}\} \subseteq L(E)$ and $\{|z|^{j}(1+|z|^{-(2q+2)})^{-1}(BW_{1/2})^{(j)}(z): z \in \Sigma_{\delta'}\} \subseteq L(E)$ are equicontinuous. Suppose now that $x \in D(A)$. Then $AW_{1/2}(z)x = W_{1/2}(z)Ax$, $z \in \Sigma_{(\pi-\omega)/2}$ and for each $q \in \circledast$ there exist $c_{q,\delta'} > 0$ and $r_q \in \circledast$ such that (cf. (202)–(203) with $\theta = 0$):

$$(204) \quad q\left(\frac{e^{i\theta/2}}{\pi}\int_0^\infty \frac{v^{1/2}B(vB+e^{i\theta}A)^{-1}Cx}{\lambda^2 e^{i\theta}+v}dv - \frac{Cx}{\lambda}\right)$$

$$\leqslant c_{q,\delta'}[r_q(x)+r_q(Ax)] \left[\int_0^1 \frac{v^{1/2}}{|\lambda|^2+v} \left(\frac{1}{v}+v^q\right)dv + \int_1^\infty \frac{v^{1/2}}{|\lambda|^2+v} \frac{1}{v} \left(\frac{1}{v}+v^q\right)dv\right]$$

$$\leqslant c_{q,\delta'}[r_q(x)+r_q(Ax)]|\lambda|^{-1}, \quad \arg(\lambda) \in (-\varepsilon, (\pi/2)+\delta').$$

If $\arg(\lambda) \in (-((\pi/2) + \delta'), -\varepsilon)$, then we can estimate the term

$$q\left(\frac{e^{-i\theta/2}}{\pi}\int_0^\infty \frac{v^{1/2}B(vB+e^{-i\theta}A)^{-1}Cx}{\lambda^2 e^{-i\theta}+v}dv-\frac{Cx}{\lambda}\right)$$

in the same way as above, from which we may conclude that $q(BF_{1/2}(\lambda)Ax) = (r_q(x) + r_q(Ax))O(|\lambda|^{-1}), \ \lambda \in \Sigma_{(\pi/2)+\delta'}$. A similar line of reasoning as performed in the case that q = -1 and $x \in D(A)$ enables us to deduce that

(205)
$$\lambda \left[\frac{e^{i\theta/2}}{\pi} \int_0^\infty \frac{v^{1/2} B (vB + e^{i\theta} A)^{-1} C x}{\lambda^2 e^{i\theta} + v} dv - \frac{C x}{\lambda} \right] \to 0, \quad \lambda \to +\infty.$$

Applying [296, Theorem 3.4] and (205), we get that $\mathcal{L}(BW_{1/2}(t)x)(\lambda) = BF_{1/2}(\lambda)x$, $\lambda > 0$ and $\lim_{z\to 0, z\in \Sigma_{\delta'}} BW_{1/2}(z)x = Cx$. In the final part of proof of Theorem 2.7.3, we have proved that $B\frac{d^j}{dz^j}W_{\gamma}(z)x = \frac{d^j}{dz^j}BW_{\gamma}(z)x$ ($z \in \Sigma_{\frac{\pi}{2}-\gamma\pi}, j \in \mathbb{N}$). As a consequence of this equality and the Cauchy integral formula, we have that $q(B\frac{d^j}{dt^j}W_{\gamma}(t)x) = O(t^{-j}), t > 0$. On the other hand, the proof of [292, Theorem 2.9.48] shows that the equation (201) continues to hold, which simply implies

that $BW_{1/2}''(z)x = AW_{1/2}(z)x$, $z \in \Sigma_{(\pi-\omega)/2}$. The proof of theorem is thereby completed.

The main problem in application of Theorem 2.7.3–Theorem 2.7.5 to incomplete abstract degenerate differential equations lies in the fact that, in many concrete situations, any of conditions $(\lambda B - A)^{-1}CA \subseteq A(\lambda B - A)^{-1}C$ $(\lambda \in \rho_C(A, B))$ and $(\lambda B - A)^{-1}CB \subseteq B(\lambda B - A)^{-1}C$ ($\lambda \in \rho_C(A, B)$) is not satisfied. Suppose, for example, that $\emptyset \neq \Omega \subset \mathbb{R}^n$ is an open bounded set with C^{∞} -boundary, $E := L^2(\Omega)$. $A := \Delta$ with the Dirichlet boundary conditions, $a(x) \in L^{\infty}(\Omega), a(x) \leq 0$ on $\overline{\Omega}$. a(x) < 0 almost everywhere in Ω , and $Bf(x) := a(x)^{-1}f(x)$ with maximal domain (cf. [199, Example 3.8, pp. 81-83]). Then $B^{-1} \in L(E)$ and the operator AB^{-1} is closed. Suppose, in addition, that $a^{-1} \in L^r(\Omega)$ for some $r \ge 2$ (resp., r > 2, r > n), if n = 1 (resp., $n = 2, n \ge 3$). Then it has been proved in the above-mentioned example that the condition [292, (HQ), p. 207] holds with the number $\omega = 0$, $C = I, m = -1 + (n/2r) \in (-1, (-1/2)),$ and with the operator A replaced by AB^{-1} therein. This implies by [292, Theorem 2.9.51(i.3)] that the operator $-(AB^{-1})_{1/2}$ is the integral generator of an exponentially bounded, analytic (n/r)-times integrated semigroup of angle $\pi/2$ on E, and that the abstract incomplete Cauchy problem [292, $(P_{2,m})$], which corresponds to the equation $u_{tt}(t,x) = \Delta\{a(x)u(t,x)\}, t > 0$, has a unique solution that is analytically extensible to the right half plane. It is clear that Theorem 2.7.5 cannot be applied here directly, by regarding the problem $u_{tt}(t,x) = \Delta\{a(x)u(t,x)\}, t > 0$ as a problem of the form $(P_{2,q,B})$ dealt with above.

EXAMPLE 2.7.6. Assume that $n \in \mathbb{N}$ and $iA_j, 1 \leq j \leq n$ are commuting generators of bounded C_0 -groups on a Banach space E; set $\mathbb{A} := (A_1, \ldots, A_n)$. Assume, further, that $0 < \delta < 2$, $P_1(x)$ and $P_2(x)$ are non-zero complex polynomials, $N_1 = dg(P_1(x)), N_2 = dg(P_2(x)), \beta > \frac{n}{2} \frac{(N_1+N_2)}{\min(1,\delta)}$ (resp. $\beta \geq n | \frac{1}{p} - \frac{1}{2} | \frac{(N_1+N_2)}{\min(1,\delta)}$, if $E = L^p(\mathbb{R}^n)$ for some $1), <math>P_2(x) \neq 0, x \in \mathbb{R}^n$ and

(206)
$$\sup_{x \in \mathbb{R}^n} \operatorname{Re}\left(\left(\frac{P_1(x)}{P_2(x)}\right)^{1/\delta}\right) \leqslant 0.$$

Define

$$R_{\delta}(t) := \left(E_{\delta} \left(t^{\delta} \frac{P_{1}(x)}{P_{2}(x)} \right) (1 + |x|^{2})^{-\beta/2} \right) (\mathbb{A}), \ t \ge 0, \ G_{\delta}(t) := \overline{P_{2}(\mathbb{A})}^{-1} R_{\delta}(t), \ t \ge 0,$$

 $A' := \overline{P_1(\mathbb{A})}, B' := \overline{P_2(\mathbb{A})}$ and $C := R_{\delta}(0)$. Then we know that $(G_{\delta}(t))_{t \ge 0}$ is an exponentially equicontinuous (g_{δ}, C) -regularized resolvent family generated by A', B', as well as that $\|R_{\delta}(t)\| + \|G_{\delta}(t)\| = O(1 + t^{\max(1,\delta)n/2}), t \ge 0 \ (\|R_{\delta}(t)\| + \|G_{\delta}(t)\| = O(1 + t^{\max(1,\delta)n|\frac{1}{p} - \frac{1}{2}|}), t \ge 0$, if $E = L^p(\mathbb{R}^n)$ for some 1),

$$\lambda^{\delta-1}(\lambda^{\delta}B'-A')^{-1}Cx = \int_0^\infty e^{-\lambda t} G_{\delta}(t) x \, dt, \quad \operatorname{Re} \lambda > 0, \ x \in E,$$

and

$$\lambda^{\delta^{-1}}B'(\lambda^{\delta}B'-A')^{-1}Cx = \int_0^\infty e^{-\lambda t}B'G_{\delta}(t)x\,dt = \int_0^\infty e^{-\lambda t}R_{\delta}(t)x\,dt, \quad \operatorname{Re}\lambda > 0, \ x \in E.$$

This implies

 $\|(\lambda B' - A')^{-1}C\| + \|B'(\lambda B' - A')^{-1}C\| = O\left(|\lambda|^{-1} + |\lambda|^{-1 - \frac{\max(1,\delta)n}{2\delta}}\right), \ \lambda \in \Sigma_{\delta\pi/2},$ in the case of a general space E, and

 $\|(\lambda B' - A')^{-1}C\| + \|B'(\lambda B' - A')^{-1}C\| = O\left(|\lambda|^{-1} + |\lambda|^{-1 - \frac{\max(1,\delta)n|\frac{1}{p} - \frac{1}{2}|}{\delta}}\right), \ \lambda \in \Sigma_{\delta\pi/2},$

in the case that $E = L^p(\mathbb{R}^n)$ for some 1 . Setting <math>A := -A' and B := B', we have that the condition (H) holds wih $\omega = \pi - (\delta \pi/2)$ as well as that $C^{-1}AC = A, C^{-1}BC = B, (\lambda B - A)^{-1}CA \subseteq A(\lambda B - A)^{-1}C$ and $(\lambda B - A)^{-1}CB \subseteq B(\lambda B - A)^{-1}C$ ($\lambda \in \rho_C(A, B)$), which implies that Theorem 2.7.3(vi.2), resp., Theorem 2.5.4(ii) is susceptible to applications in the case that $E = L^p(\mathbb{R}^n)$ for some 1 satisfying

$$\frac{\max(1,\delta)n\left|\frac{1}{p} - \frac{1}{2}\right|}{\delta} < \gamma < 1/2, \text{ resp., } \frac{\max(1,\delta)n\left|\frac{1}{p} - \frac{1}{2}\right|}{\delta} < 1/2.$$

Before proceeding further, we would like to mention in passing that the operator B is injective and that the operator $B^{-1}A$ is closable because $\rho_{C'}(B^{-1}A) \neq \emptyset$ ($C' := C(B+A)^{-1}C$), which follows from the equality $(\lambda + B^{-1}A)^{-1}C(B+A)^{-1}Cx = B(\lambda B + A)^{-1}C(B + A)^{-1}Cx$, $\lambda > 0$, $x \in E$ (cf. (196)). It is also worth noting that the operator AB^{-1} is closable because $\rho_C(AB^{-1}) \neq \emptyset$; this is a consequence of the equality $(\lambda + AB^{-1})^{-1}Cx = B(\lambda B + A)^{-1}Cx$, $\lambda > 0$, $x \in E$. Now we will prove the uniqueness of solution of problem $(P_{2,q,B})$ in our particular case; recall that $E = L^p(\mathbb{R}^n)$ for some $1 satisfying <math>\zeta := \frac{\max(1,\delta)n|\frac{1}{p}-\frac{1}{2}|}{\delta} < 1/2$. Denote by $\mathcal{M}_{C,q,\omega}$ the class which consists of all closed linear operators D acting on E such that $\mathbb{C} \setminus \overline{\Sigma_{\omega}} \subseteq \rho_C(D)$, $DC \subseteq CD$ and the family

$$\left\{ (|\lambda|^{-1} + |\lambda|^q)^{-1} (\lambda - D)^{-1} C : \lambda \notin \Sigma_{\omega'} \right\}$$

is equicontinuous for every $\omega < \omega' < \pi$; here $q \in \mathbb{R}$ and $\omega \in [0, \pi)$. Then we have $\zeta \in [0, 1/2)$ and our previous examinations show that $\overline{AB^{-1}} \in \mathcal{M}_{C, -1-\zeta, \pi-(\delta\pi/2)}$. If $z \mapsto u(z), z \in \Sigma_{(\pi-\omega)/2}$ is a solution of problem $(P_{2,q,B})$ with x = 0, then $v(z) := Bu(z), z \in \Sigma_{(\pi-\omega)/2}$ is a solution of problem

$$(P_{2,q}): \begin{cases} v \in \mathcal{A}(\Sigma_{(\pi/2)-(\omega/2)}:E), \\ \frac{d^2}{dz^2}v(z) = \overline{AB^{-1}}v(z), \ z \in \Sigma_{(\pi/2)-(\omega/2)}, \\ \lim_{z \to 0, z \in \Sigma_{\delta}} v(z) = 0, \ \text{for every } \delta \in (0, (\pi/2) - (\omega/2)), \\ \text{the set } \{(1+|z|^{-(2q+2)})^{-1}v(z): z \in \Sigma_{\delta}\} \text{ is bounded in } E \\ \text{for every } \delta \in (0, (\pi/2) - (\omega/2)). \end{cases}$$

An application of **[292**, Theorem 2.9.51(iii.2)] yields v(z) = 0, $z \in \Sigma_{(\pi-\omega)/2}$, so that u(z) = 0, $z \in \Sigma_{(\pi-\omega)/2}$ by the injectiveness of B. In the remainder of this example, we will provide certain applications of Theorem 2.5.3. It will be assumed that 0 < c < b < 1, $0 < \zeta \leq 1$, $1/(2 - \zeta) > c/\zeta$, $\sigma > 0$ and $\varsigma > 0$; we define the operators A and B in the same way as above but, instead of estimate (206), we assume that

(207)
$$\operatorname{dist}\left(\{-P_1(x)P_2(x)^{-1} : x \in \mathbb{R}^n\}, \Pi_{c,\sigma,\varsigma}^{\zeta}\right) > 0.$$

Then there exist sufficiently large numbers $\beta' \ge 0$ and $\nu \ge 0$ (the proofs of [292, Theorem 2.5.2] and Theorem 2.2.20 can give more detailed and accurate information about β' and ν ; we leave the reader to make this precise) such that

$$\left(\frac{1}{\lambda^{\zeta} P_2(x) + P_1(x)} (1 + |x|^2)^{-\beta'/2} \right) (\mathbb{A}) = (\lambda^{\zeta} B + A)^{-1} C, \quad \lambda \in \Pi_{c,\sigma,\varsigma},$$
$$\left(\frac{P_2(x)}{\lambda^{\zeta} P_2(x) + P_1(x)} (1 + |x|^2)^{-\beta'/2} \right) (\mathbb{A}) = B(\lambda^{\zeta} B + A)^{-1} C, \quad \lambda \in \Pi_{c,\sigma,\varsigma},$$

and the operator families $\{(1+|\lambda|)^{-\nu}(\lambda^{\zeta}B+A)^{-1}C:\lambda\in\Pi_{c,\sigma,\varsigma}\}\subseteq L(E)$ and $\{(1+|\lambda|)^{-\nu}B(\lambda^{\zeta}B+A)^{-1}C:\lambda\in\Pi_{c,\sigma,\varsigma}\}\subseteq L(E)$ are both equicontinuous and strongly continuous $(C:=((1+|x|^2)^{-\beta'/2})(\mathbb{A}))$, so that Theorem 2.5.3 can be applied. Although the equation (185) of Remark 2.5.5 holds in our concrete situation, it is our duty to say that Theorem 2.5.3–Remark 2.5.5 certainly have some disadvantages in the degenerate case because it is very difficult to say whether an element $x \in$ E belongs to the space $E^{\langle p^{p\zeta/b}\rangle}(B^{-1}A)$ or not, with the exception of some very special cases. Suppose now that the operators A_k and B_k are defined by $A_k :=$ $-\overline{P_{1,k}}(\mathbb{A}), B_k := \overline{P_{2,k}}(\mathbb{A})$, and that the estimate (207) holds with the polynomials $P_1(x)$ and $P_2(x)$ replaced respectively with the polynomials $P_{1,k}(x)$ and $P_{2,k}(x)$. Then Theorem 2.5.3 can be applied to a large class of multi-term (non-)degenerate differential equations of the form (168), whose P_{λ} looks like

$$P_{\lambda} = (\lambda^{\zeta} B_1 + A_1)(\lambda^{\zeta} B_2 + A_2)\dots(\lambda^{\zeta} B_k + A_k).$$

The choice of regularizing operator C is essentially the same as above but we must eventually increase the value of β' .

The analysis contained in Example 2.7.6 shows that there exists at most one solution of problem $(P_{2,q,B})$ stated in Theorem 2.5.4(i), resp., Theorem 2.5.4(ii), provided that the operator B is injective, the operator AB^{-1} is closable and $\overline{AB^{-1}} \in \mathcal{M}_{C,q,\omega}$ for some $q \in (-1, (-1)/2)$, resp., $q \in ((-3/2), -1]$, and $\omega \in [0, \pi)$. It is not clear, however, in which other cases the uniqueness of solutions of problem $(P_{2,q,B})$ can be proved $(B \neq I)$.

2.8. Abstract degenerate fractional differential equations in locally convex spaces with a σ -regular pair of operators

The results presented in this section are obtained jointly with V. E. Fedorov [214]. Suppose that X is an SCLCS and A is a closed linear operator acting on X. Let us recall that the regular resolvent set of A, $\rho^r(A)$ shortly, is defined as the union of those complex numbers $\lambda \in \rho(A)$ for which $(\lambda - A)^{-1} \in R(X)$, where R(X) denotes the set of all regular bounded linear operators $D \in L(X)$.

If $D \in R(X)$, $\alpha > 0$, $m = \lceil \alpha \rceil$, $f \in C^m([0, \tau) : X)$ and $x_k \in X$ $(0 \leq k \leq m-1)$, then we have proved in [214, Theorem 1] that the unique solution of the non-degenerate differential equation

$$\mathbf{D}_t^{\alpha} u(t) = Du(t) + f(t), \quad t \in [0, \tau); \ u^{(k)}(0) = x_k \ (0 \le k \le m - 1)$$

is given by

(208)
$$u(t) = \sum_{k=0}^{m-1} t^k E_{\alpha,k+1}(t^{\alpha}D) x_k + \int_0^t (t-s)^{\alpha-1} E_{\alpha,\alpha}(D(t-s)^{\alpha}) f(s) ds, \ t \in [0,\tau),$$

where the operator $E_{\alpha,\nu}(t^{\alpha}D)$ is defined in the very obvious way $(0 < \tau \leq \infty, \nu \in \mathbb{R})$.

In the remainder of this section, we assume that Y is an SCLCS as well as that A and B are two closed linear operators acting between the spaces X and Y. Set, as before, $R^B_{\lambda}(A) := (\lambda B - A)^{-1}B$ and $L^B_{\lambda}(A) := B(\lambda B - A)^{-1}$. Then the regular B-resolvent set of the operator A, $\rho^B_r(A)$ for short, is defined to be

$$\rho_r^B(A) := \left\{ \lambda \in \mathbb{C} : (\lambda B - A)^{-1} \in L(Y, X), \ R_\lambda^B(A) \in L(X), \ L_\lambda^B(A) \in L(Y) \right\}.$$

The following proposition holds true:

PROPOSITION 2.8.1. Suppose that $D(A) \cap D(B)$ is sequentially dense in X. Then we have:

- (i) $\rho_r^B(A)$ is an open subset of \mathbb{C} .
- (ii) If $\rho_r^B(A) \neq \emptyset$, then the mappings $\lambda \mapsto (\lambda B A)^{-1}$, $\lambda \in \rho_r^B(A)$, $\lambda \mapsto L_{\lambda}^B(A)$, $\lambda \in \rho_r^B(A)$ and $\lambda \mapsto L_{\lambda}^B(A)$, $\lambda \in \rho_r^B(A)$ are strongly analytic.
- (iii) If $\rho_r^B(A) \neq \emptyset$, then the constants of regularity of operators $L_{\lambda}^B(A)$ and $R_{\lambda}^B(A)$ are continuously dependent of parameter $\lambda \in \rho_r^B(A)$.

It is said that the operator A is (B, σ) -regular iff there exists a finite number a > 0 such that, for every $\lambda \in \mathbb{C}$ with $|\lambda| > a$, we have $\lambda \in \rho_r^B(A)$. If so, then we also say that a pair of operators (A, B) is σ -regular.

Suppose that the pair of operators (A, B) is σ -regular and R > a. Denote by γ the positively oriented contour $\gamma = \{\lambda \in \mathbb{C} : |\lambda| = R\}$. Set

$$Px := \frac{1}{2\pi i} \int_{\gamma} R^B_{\lambda}(A) x \, d\lambda \text{ and } Qy := \frac{1}{2\pi i} \int_{\gamma} L^B_{\lambda}(A) y \, d\lambda \quad (x \in X, \ y \in Y).$$

Then it is very simple to prove that $P \in L(X)$ and $Q \in L(Y)$ are projections. Define $X_0 := N(P), Y_0 := N(Q), X_1 := R(P)$ and $Y_1 := R(Q)$. Clearly, $X = X_0 \oplus X_1$ and $Y = Y_0 \oplus Y_1$. By A_k (B_k) we denote the restriction of operator A (B) to the space $X_k, k = 0, 1$. By $\rho_{r,k}^B(A)$ we denote the regular B_k -resolvent set of the operator $A_k, k = 0, 1$.

We have the following:

PROPOSITION 2.8.2. [214] Suppose that the pair of operators (A, B) is σ -regular. Then the following holds:

- (i) For every $x \in X$, we have $Px \in D(A)$.
- (ii) $B_k \in L(X_k, Y_k), k = 0, 1.$
- (iii) A_0 is a closed linear operator acting between the spaces X_0 and Y_0 ; $A_1 \in L(X_1, Y_1)$.
- (iv) There exists operator $B_1^{-1} \in L(Y_1, X_1)$.
- (v) $\rho_{r,0}^B(A) = \mathbb{C}$ and, in particular, there exists operator $A_0^{-1} \in L(Y_0, X_0)$.
- (vi) The operators $S_1 := B_1^{-1}A_1$ and $T_1 := A_1B_1^{-1}$ are regular.

Set $H := A_0^{-1}B_0$. If there exists $p \in \mathbb{N}_0$ such that $H^p \neq \mathbb{O}$ and $H^{p+1} = \mathbb{O}$, then we say that the operator A is (B, p)-regular or, equivalently, that the pair of operators (A, B) is p-regular (the symbol \mathbb{O} denotes the null operator in X_0). It is well known (see e.g. [509, p. 92]) that the existence or non-existence of number $p \in \mathbb{N}_0$ with the above property characterizes the behaviour of relative resolvent $(\lambda B - A)^{-1}$ at the point ∞ . Assume that $Bx_0 = 0$ for some $x_0 \in X \setminus \{0\}$ (then we simply say that x is an eigenvector of B). An ordered set of vectors $\{x_0, x_1, \ldots\}$ is called a chain of A-adjoint vectors of x_0 iff $x_k \notin N(B)$ for $k \in \mathbb{N}$, and

$$Bx_{k+1} = Ax_k, \quad k = 0, 1, \dots$$

Call the index of a vector in the chain (starting from 0) its height, and the eigenvectors the A-adjoint vectors of height 0. It is well known that a vector $x \neq 0$ is an A-adjoint vector of height at most l of B iff $(R_{\lambda}^{B}(A))^{l+1}x = 0$ for some (all) $\lambda \in \rho_{r}^{B}(A)$. Furthermore, the operator A is (B, 0)-regular iff $N(B) = X_{0}$; if this is the case, then $R(B) = Y_{1}$ and, for every $x_{0} \in N(B) \setminus \{0\}$, we have that $x_{0} \notin D(A)$ or $Ax_{0} \notin R(B)$. The (B, p)-regularity of operator A has been profiled in [214, Theorem 3(i)].

Suppose that the pair of operators (A, B) is σ -regular, $\alpha > 0$ and $\beta > 0$. Set

$$U_{\alpha,\beta}(t)x := \frac{1}{2\pi i} \int_{\gamma} E_{\alpha,\beta}(\lambda t^{\alpha}) R^B_{\lambda}(A) x \, d\lambda, \quad x \in X,$$

and

$$V_{\alpha,\beta}(t)y := \frac{1}{2\pi i} \int_{\gamma} E_{\alpha,\beta}(\lambda t^{\alpha}) R^B_{\lambda}(A) y \, d\lambda, \quad y \in Y.$$

Then the following theorem holds:

THEOREM 2.8.3. (i) $(U_{\alpha,\beta}(t))_{t\geq 0} \subseteq L(X)$ and $(V_{\alpha,\beta}(t))_{t\geq 0} \subseteq L(Y)$ are strongly continuous.

- (ii) $U_{\alpha,\beta}(t)P = PU_{\alpha,\beta}(t) = U_{\alpha,\beta}(t)$ and $V_{\alpha,\beta}(t)Q = QV_{\alpha,\beta}(t) = V_{\alpha,\beta}(t)$ for all $t \ge 0$.
- (iii) $X_0 \subseteq N(U_{\alpha,\beta}(t)), R(U_{\alpha,\beta}(t)) \subseteq X_1, Y_0 \subseteq N(V_{\alpha,\beta}(t)) \text{ and } R(V_{\alpha,\beta}(t)) \subseteq Y_1 \text{ for all } t \ge 0.$
- (iv) $U_{\alpha,\beta}(t) = E_{\alpha,\beta}(B_1^{-1}A_1t^{\alpha})$ and $V_{\alpha,\beta}(t) = E_{\alpha,\beta}(A_1B_1^{-1}t^{\alpha})$ for all $t \ge 0$.

PROOF. The proof of (i) is trivial and therefore omitted. Let $t \ge 0$ and $x \in X$. By the Fubini theorem, we have that $U_{\alpha,\beta}(t)Px = PU_{\alpha,\beta}(t)x$. In order to see that $PU_{\alpha,\beta}(t)x = U_{\alpha,\beta}(t)x$, let $\gamma_1 = \{z \in \mathbb{C} : |z| = R + 1\}$ be positively oriented, where R > a. Then the residue theorem and the resolvent equation together imply

$$\begin{aligned} PU_{\alpha,\beta}(t)x &= \frac{1}{(2\pi i)^2} \int_{\gamma} \int_{\gamma_1} R^B_{\lambda}(A) R^B_{\mu}(A) E_{\alpha,\beta}(t^{\alpha}\mu) x \, d\mu \, d\lambda \\ &= \frac{1}{(2\pi i)^2} \int_{\gamma} R^B_{\mu}(A) E_{\alpha,\beta}(t^{\alpha}\mu) \bigg[\int_{\gamma_1} \frac{x \, d\lambda}{\lambda - \mu} \bigg] d\mu \\ &- \frac{1}{(2\pi i)^2} \int_{\gamma_1} R^B_{\lambda}(A) \bigg[\int_{\gamma} \frac{E_{\alpha,\beta}(t^{\alpha}\mu) x}{\lambda - \mu} d\mu \bigg] d\lambda \\ &= U_{\alpha,\beta}(t) x - 0 = U_{\alpha,\beta}(t) x. \end{aligned}$$

The proof of second equality in (ii) can be deduced analogically; (iii) is an immediate consequence of (ii). The proof of (iv) essentially follows from the following calculus $(S \equiv A_1^{-1}B_1)$:

$$\begin{split} U_{\alpha,\beta}(t)x &= PU_{\alpha,\beta}(t)x \\ &= \frac{1}{2\pi i} \int_{\gamma} R^{B_1}_{\mu}(A_1) PE_{\alpha,\beta}(t^{\alpha}\mu)x \, d\mu \\ &= \frac{1}{2\pi i} \int_{\gamma} (\mu I - S)^{-1} PE_{\alpha,\beta}(t^{\alpha}\mu)x \, d\mu \\ &= \frac{1}{2\pi i} \int_{\gamma} \sum_{k=0}^{\infty} \mu^{-k-1} S^k P \sum_{n=0}^{\infty} \frac{t^{\alpha n} \mu^n}{\Gamma(\alpha n + \beta)} x \, d\mu \\ &= \sum_{k=0}^{\infty} \frac{t^{\alpha k} S^k}{\Gamma(\alpha k + \beta)} Px = E_{\alpha,\beta}(t^{\alpha}S) Px. \end{split}$$

Along with the problem

(DF):
$$\begin{cases} \mathbf{D}_t^{\alpha} Bu(t) = Au(t), & t \ge 0, \\ u(0) = x_0, & u^{(k)}(0) = 0, & 1 \le k \le m - 1, \end{cases}$$

we examine the following one:

$$(\mathrm{DF})_1 : \begin{cases} \mathbf{D}_t^{\alpha} L_r^B(A) v(t) = L_r^B(A) v(t), & t \ge 0, \\ v(0) = (\zeta B - A)^{-1} x_0, \ v^{(k)}(0) = 0, & 0 \le k \le m - 1, \end{cases}$$

where $\zeta \in \rho_r^B(A)$. The solution u(t) of (DF) and solution v(t) of (DF)₁ are linked by the relation $u(t) = (\zeta B - A)^{-1}v(t), t \ge 0$. Suppose that $x \in X$ and $y \in Y$. Set $u(t) := U_{\alpha,1}(t)x, t \ge 0$ and $v(t) := V_{\alpha,1}(t)y, t \ge 0$. Then it can be easily seen that

$$Bu(t) = \frac{1}{2\pi i} \int_{\gamma} \mu A R^B_{\mu}(A) E_{\alpha}(t^{\alpha}\mu) x \, d\mu,$$

which is a continuous function for $t \ge 0$. Evidently, the function $t \mapsto Au(t), t \ge 0$ is (m-1)-times continuously differentiable for $t \ge 0$. Making use of the Cauchy theorem and the closedness of operator B, we have $(m = \lceil \alpha \rceil)$:

$$\begin{split} \mathbf{D}_{t}^{\alpha}Au(t) &= \frac{1}{2\pi i}\int_{\gamma}AR_{\mu}^{B}(A)J_{t}^{m-\alpha}\sum_{n=1}^{\infty}\frac{\mu^{n}\alpha^{\alpha n-m}}{\Gamma(\alpha n-m+1)}x\,d\mu\\ &= \frac{1}{2\pi i}\int_{\gamma}AR_{\mu}^{B}(A)\sum_{n=1}^{\infty}\frac{\mu^{n}\alpha^{\alpha(n-1)}}{\Gamma(\alpha(n-1)+1)}x\,d\mu\\ &= Bu(t), \quad t \geqslant 0. \end{split}$$

This shows that the mapping $t \mapsto u(t)$, $t \ge 0$ is a solution of problem (DF) considered without initial conditions; we can similarly prove that the mapping $t \mapsto v(t)$, $t \ge 0$, is a solution of problem (DF)₁ considered without initial conditions; cf. [214, Lemma 4]. Since no confusion seems likely, the above problems will be denoted by the same symbols (DF) and (DF)₁.

DEFINITION 2.8.4. A set $\mathcal{P} \subseteq X$ is said to be the phase space of problem (DF) iff the following holds:

- (i) for any strong solution u(t) of (DF), we have that $u(t) \in \mathcal{P}$ for all $t \ge 0$,
- (ii) for any $x_0 \in \mathcal{P}$, there exists a unique strong solution of (DF).

In order to profile the phase space of problem (DF), we need the following technical result.

LEMMA 2.8.5. Suppose that the operator G is nilpotent of order $p \in \mathbb{N}_0$. Let the function $t \mapsto g(t), t \ge 0$ satisfy that $(\mathbf{D}_t^{\alpha}G)^k g \in C([0,\infty):X)$ for all $k \in \mathbb{N}_p^0$. Then there exists a unique solution of the following time-fractional initial value problem

(209)
$$\mathbf{D}_t^{\alpha} G z(t) = z(t) + g(t), \quad t \ge 0,$$

and the solution z(t) is given by the formula

(210)
$$z(t) = -\sum_{k=0}^{p} (\mathbf{D}_{t}^{\alpha} G)^{k} g(t), \quad t \ge 0.$$

PROOF. Suppose that z(t) is a solution of (209). Applying the operator G on the both sides of this equation, we get that $G\mathbf{D}_t^{\alpha}Gz(t) = Gz(t) + Gg(t), t \ge 0$. Since the Caputo fractional derivative of the right hand side of the last equality exists, it is clear that the same holds for the left hand side. Performing now the operator \mathbf{D}_t^{α} , we obtain $(\mathbf{D}_t^{\alpha}G)^2z(t) = \mathbf{D}_t^{\alpha}Gz(t) + \mathbf{D}_t^{\alpha}Gg(t) = z(t) + g(t) + \mathbf{D}_t^{\alpha}Gg(t), t \ge 0$. Repeating this procedure, it readily follows that $(\mathbf{D}_t^{\alpha}G)^{p+1}z(t) = z(t) + \sum_{k=0}^{p}(\mathbf{D}_t^{\alpha}G)^kg(t), t \ge 0$. Taking into account the nilpotency of the operator G, it is checked at once that $(\mathbf{D}_t^{\alpha}G)^{p+1}z(t) = (\mathbf{D}_t^{\alpha})^{p+1}G^{p+1}z(t) = 0, t \ge 0$. This implies that any solution of (209) has the form (210). Direct computation shows that the function z(t), given by (210), is a solution of (209), finishing the proof of lemma. \Box

In [214, Theorem 5], we have proved that the σ -regularity of pair (A, B) for some non-negative integer p implies that the phase space of problem (DF), resp., (DF)₁, coincides with X_1 (Y_1). We will include all relevant details of proof for problem (DF): Suppose that u(t) is a solution of (DF). Set $u_0(t) := (I - P)u(t)$, $t \ge 0$ and $u_1(t) := Pu(t)$, $t \ge 0$. Then

$$\begin{aligned} \mathbf{D}_{t}^{\alpha}Hu_{0}(t) &= u_{0}(t), \quad t \ge 0 \; \; ; \; \; H_{0} \equiv B_{0}^{-1}A_{0}, \\ \mathbf{D}_{t}^{\alpha}u_{1}(t) &= Su_{1}(t), \quad t \ge 0 \; \; ; \; \; S \equiv B_{1}^{-1}A_{1}. \end{aligned}$$

By Lemma 2.8.5, we get that $u_0(t) = 0$, $t \ge 0$ and $u_1(t) \in X_1$ for all $t \ge 0$. Since the operator S is regular in X_1 , by the foregoing we have that for each $Pu_0 = u_0^1 \in X_1$ a unique solution of Cauchy problem (DF) is given by $u(t) = E_{\alpha}(t^{\alpha}S)u_0^1 = U_{\alpha,1}(t)u_0$, $t \ge 0$. The main results of [214] are given in the following theorem:

THEOREM 2.8.6. (i) Suppose that the pair (A, B) is p-regular, $0 < \tau \leq \infty$, $Qf \in C^m([0, \tau) : Y)$, there exist fractional operators $(\mathbf{D}_t^{\alpha} H)^n A_0^{-1}(I - Q)f \in C([0, \tau) : X)$ for $0 \leq n \leq p$, $x_k \in X$ for $0 \leq k \leq m - 1$, and the

compatibility condition

$$-\mathbf{D}_{t}^{k} \sum_{n=0}^{p} (\mathbf{D}_{t}^{\alpha} H)^{n} A_{0}^{-1} (I-Q) f(t)_{|t=0} = (I-P) x_{k}, \quad 0 \leq k \leq m-1$$

holds. Then there exists a unique strong solution u(t) of problem

$$(DF)_{f}: \begin{cases} \mathbf{D}_{t}^{\alpha}Bu(t) = Au(t) + f(t), & t \ge 0\\ u^{(k)}(0) = x_{k}, & 0 \le k \le m - 1. \end{cases}$$

Furthermore, u(t) is given by the following formula:

$$u(t) = \sum_{k=0}^{m-1} t^k U_{\alpha,k+1}(t) x_k + \int_0^t (t-s)^{\alpha-1} U_{\alpha,\alpha}(t-s) B_1^{-1} Q f(s) ds$$
$$- \sum_{n=0}^p (\mathbf{D}_t^{\alpha} H)^n A_0^{-1} (I-Q) f(t), \quad t \in [0,\tau).$$

(ii) Suppose that the pair (A, B) is p-regular, $0 < \tau \leq \infty$, $Qf \in C^m([0, \tau) : Y)$, there exist fractional operators $(\mathbf{D}_t^{\alpha}H)^n A_0^{-1}(I-Q)f \in C([0, \tau) : X)$ for $0 \leq n \leq p$, and $x_k \in X$ for $0 \leq k \leq m-1$. Then there exists a unique strong solution of problem

$$(DF)_{f,P}: \begin{cases} \mathbf{D}_t^{\alpha} Bu(t) = Au(t) + f(t), & t \ge 0, \\ (Pu)^{(k)}(0) = x_k, & 0 \le k \le m - 1. \end{cases}$$

Furthermore, the form of solution u(t) is the same as in (i).

PROOF. We will prove only (i). It is not difficult to show that

(211)
$$\mathbf{D}_t^{\alpha} H u_0(t) = u_0(t) + B_0^{-1} (I - Q) f(t), \quad t \ge 0,$$

(212)
$$\mathbf{D}_{t}^{\alpha}u_{1}(t) = Su_{1}(t) + h(t), \quad t \ge 0 \quad ; \quad h(t) \equiv A_{1}^{-1}Qf(t).$$

In view of Lemma 2.8.5, the unique solution of problem (211) is given by $u_0(t) = -\sum_{n=0}^{p} (\mathbf{D}_t^{\alpha} H)^n B_0^{-1} (I-Q) f(t), t \ge 0$. Since the operator S is regular in X_1 , the result now simply follows from the equality (208), which gives the form of solution of equation (212).

Now we would like to illustrate Theorem 2.8.6 with the following example.

EXAMPLE 2.8.7. Let $(E, \|\cdot\|)$ be a complex Banach space, and let A be a closed linear operator acting on E. For any $\tau > 0$, we put

$$D_{\infty,\tau,0}(A) := \Big\{ x \in D_{\infty}(A) : \limsup_{k \to \infty} \|A^k x\|^{1/k} \leqslant \tau \Big\}.$$

Let $D_{\infty,\tau}(A)$ be the biggest closed subspace of $D_{\infty,\tau,0}(A)$ with respect to the topology inherited from $D_{\infty}(A)$. Then it is well known that $D_{\infty,\tau}(A)$ is a Fréchet space and that $A_{|D_{\infty,\tau}(A)}$ is a regular operator on $D_{\infty,\tau}(A)$. Set $\mathbb{D}_{\infty}(A)$ be the inductive limit of spaces $D_{\infty,n}(A)$ as $n \to \infty$. Then, for every two entire functions G(z) and $J_1(z)$, we have that G(A) and $J_1(A)$ are continuous linear operators on $\mathbb{D}_{\infty}(A)$, which is known to be a separable and sequentially complete locally convex space [226]. In [214, Theorem 8], we have proved that the assumptions E is a Hilbert space and A is a self-adjoint operator on E imply that the operator $J_1(A)$ is (G(A), 0)-radial, provided in addition to the above that the functions G(z) and $J_1(z)$ do not have common roots on $\sigma(A)$ as well as that there exists a finite constant a > 0 such that $|J_1(\lambda)/G(\lambda)| \leq a$ for all $\lambda \in \sigma(A)$ with $G(\lambda) \neq 0$. The main feature of [214, Theorem 9] is to analyze the following special case: $E =: L^2[0, 1]$ and

$$A := id/dx, \quad D(A) := \{ u \in L^2[0,1] : u' \in L^2[0,1], u(0) = u(1) \}.$$

Applying [214, Theorem 8], we have proved that the following initial boundary value problem

$$\mathbf{D}_t^{\alpha} G(A)u(x,t) = J_1(A)u(x+h,t) + f(x,t), \quad x \in \mathbb{R}, \ t \ge 0,$$
$$u(x,t) = u(x+1,t), \quad x \in \mathbb{R}, \ t \ge 0,$$
$$\frac{\partial^n u}{\partial x^n}u(x,0) = u_n(x), \quad n = 0, 1, \dots, m-1, \ x \in [0,1]$$

has a unique solution provided that $|G(2\pi k)| + |J_1(2\pi k)| \neq 0, k \in \mathbb{Z}$, the set $\{|J_1(2\pi k)|/|G(2\pi k)| : k \in \mathbb{Z}, G(2\pi k) \neq 0\}$ is bounded, $f(x,t) = f(x+1,t), x \in \mathbb{R}, t \geq 0, f \in C^m([0,\infty) : \mathbb{D}_{\infty}(A)), u_n \in \mathbb{D}_{\infty}(A) \ (n=0,1,\ldots,m-1),$ and

$$\int_{G_0} d\mathcal{E}_{\lambda} \Big(J_1(\lambda) e^{-ih\lambda} u_n + f^{(n)}(\cdot, 0) \Big) = 0, \quad n = 0, 1, \dots, m-1,$$

where \mathcal{E}_{λ} ($\lambda \in \mathbb{R}$) is the resolution of the identity for A and $G_0 := \{\lambda \in \mathbb{R} : G(\lambda) \neq 0\}$. Furthermore, the solution $u(\cdot, t)$ is given by the following formula:

$$\begin{split} u(\cdot,t) &= \sum_{n=0}^{m-1} t^n \int_{\sigma(A)\smallsetminus G_0} E_{\alpha,n+1} \Big(t^\alpha \frac{J_1(\lambda)e^{-ih\lambda}}{G(\lambda)} \Big) d\mathcal{E}_\lambda u_n \\ &+ \int_0^t (t-s)^{\alpha-1} \int_{\sigma(A)\smallsetminus G_0} E_{\alpha,\alpha} \Big(t^\alpha \frac{J_1(\lambda)e^{-ih\lambda}}{G(\lambda)} \Big) \frac{d\mathcal{E}_\lambda f(\cdot,s)}{G(\lambda)} ds \\ &- \int_{G_0} \frac{e^{ih\lambda} d\mathcal{E}_\lambda f(\cdot,t)}{J_1(\lambda)}, \quad t \ge 0. \end{split}$$

The solution of this equation is analytic with repect to the variable t, in $\mathbb{C} \setminus (-\infty, 0]$, if $f(\cdot, \cdot)$ has some expected properties, as well.

We end this section with the observation that Fedorov and Gordievskikh [211] have analyzed the following abstract degenerate fractional Cauchy problem:

$$\mathbf{D}_{t}^{\alpha}P_{n}(\Delta) = Q_{r}(\Delta)u(x,t), \quad (x,t) \in \Omega \times [0,\infty),$$

$$(1-\theta)\Delta^{k} + \theta \frac{\partial\Delta^{k}u}{\partial n}u(x,t) = 0, \quad k = 0, 1, \dots, n-1, \ (x,t) \in \partial\Omega \times [0,\infty),$$

$$\frac{\partial^{k}u}{\partial t^{k}}(x,0) = 0, \quad k = 0, 1, \dots, m-1;$$

here, Ω is a smooth domain in \mathbb{R}^n , $\alpha > 0$, $\theta \in \mathbb{R}$, Δ is the Dirichlet Laplacian in $L^2(\Omega)$ and P_n , Q_r are non-zero complex polynomials of degrees n, r. In [211, Theorem 8] and [211, Theorem 9], the authors explore separately the cases $r \leq n$ and r > n, respectively.

2.9. Abstract degenerate non-scalar Volterra equations

In this section, we delve into the details regarding abstract degenerate Volterra equations of non-scalar type. Let X and Y be two complex Banach spaces satisfying that Y is continuously embedded in X, let the operator $C \in L(X)$ be injective, and let $\tau \in (0, \infty]$. The norm in X, resp. Y, will be denoted by $\|\cdot\|_X$, resp. $\|\cdot\|_Y$; as before, [R(C)] denotes the Banach space R(C) equipped with the norm $\|x\|_{R(C)} = \|C^{-1}x\|_X$, $x \in R(C)$. By B we denote a closed linear operator with domain and range contained in X. If Z is a general topological space and $Z_0 \subseteq Z$, then by $\overline{Z_0}^Z$ we denote the adherence of Z_0 in Z (we will use the abbreviation $\overline{Z_0}$, if there is no risk for confusion).

Let A(t) be a locally integrable function from $[0, \tau)$ into L(Y, X). Unless stated otherwise, we assume that A(t) is not of scalar type, i.e., that there does not exist $a \in L^1_{loc}([0, \tau)), a \neq 0$, and a closed linear operator A in X such that Y = [D(A)]and A(t) = a(t)A for a.e. $t \in [0, \tau)$. In the sequel, we will basically follow the notation employed in the monograph of J. Prüss [459] and our previous paper [299] on abstract non-degenerate equations of non-scalar type (cf. also [262] and [269] for some other references in this direction); the meaning of symbol A will be clear from the context.

DEFINITION 2.9.1. Let $k \in C([0,\tau))$ and $k \neq 0$, let $\tau \in (0,\infty]$, $f \in C([0,\tau) : X)$, and let $A \in L^1_{loc}([0,\tau) : L(Y,X))$. Consider the linear degenerate Volterra equation

(213)
$$Bu(t) = f(t) + \int_0^t A(t-s)u(s)ds, \quad t \in [0,\tau).$$

Then a function $u \in C([0, \tau) : [D(B)])$ is said to be:

- (i) a strong solution of (213) iff $u \in L^{\infty}_{loc}([0,\tau):Y)$ and (213) holds on $[0,\tau)$,
- (ii) a mild solution of (213) iff there exist a sequence (f_n) in $C([0, \tau) : X)$ and a sequence (u_n) in $C([0, \tau) : [D(B)])$ such that $u_n(t)$ is a strong solution of (213) with f(t) replaced by $f_n(t)$ and that $\lim_{n\to\infty} f_n(t) = f(t)$ as well as $\lim_{n\to\infty} u_n(t) = u(t)$, uniformly on compact subsets of $[0, \tau)$.

The abstract Cauchy problem (213) is said to be (kC)-well posed (*C*-well posed, if $k(t) \equiv 1$) iff for every $y \in Y$, there exists a unique strong solution of

(214)
$$Bu(t;y) = k(t)Cy + \int_0^t A(t-s)u(s;y)ds, \quad t \in [0,\tau)$$

and if $u(t; y_n) \to 0$ in [D(B)], uniformly on compact subsets of $[0, \tau)$, whenever (y_n) is a zero sequence in Y; (213) is said to be *a*-regularly (kC)-well posed (*a*-regularly C-well posed, if $k(t) \equiv 1$), where $a \in L^1_{loc}([0, \tau))$, iff (213) is (kC)-well posed and if the equation

$$Bu(t) = (a * k)(t)Cx + \int_0^t A(t - s)u(s)ds, \quad t \in [0, \tau)$$

admits a unique strong solution for every $x \in X$.

We would like to point out that every strong solution of (213) is also a mild solution of (213) as well as that the notion introduced in Definition 2.9.1 generalizes the corresponding one from [299, Definition 1], given in the case that B = I. It is also clear that the concept of a strong (mild) solution of (213) and the concept of a (kC)-well posedness of (213) can be defined in some other ways; we will skip all related details.

The following definition will be crucial for our further work.

DEFINITION 2.9.2. Let $\tau \in (0, \infty]$, $k \in C([0, \tau))$, $k \neq 0$ and $A \in L^1_{loc}([0, \tau) : L(Y, X))$. A family $(S(t))_{t \in [0, \tau)}$ in L(X, [D(B)]) is called an (A, k, B)-regularized C-pseudoresolvent family iff the following holds:

- (S1) The mappings $t \mapsto S(t)x$, $t \in [0, \tau)$ and $t \mapsto BS(t)x$, $t \in [0, \tau)$ are continuous in X for every fixed $x \in X$, BS(0) = k(0)C and S(t)C = CS(t), $t \in [0, \tau)$.
- (S2) Put $U(t)x := \int_0^t S(s)x \, ds, x \in X, t \in [0, \tau)$. Then (S2) means $U(t)Y \subseteq Y$, $U(t)_{|Y} \in L(Y), t \in [0, \tau)$ and $(U(t)_{|Y})_{t \in [0, \tau)}$ is locally Lipschitz continuous in L(Y).
- (S3) The resolvent equations

(215)
$$BS(t)y = k(t)Cy + \int_0^t A(t-s)dU(s)y, \quad t \in [0,\tau), \ y \in Y,$$

(216)
$$BS(t)y = k(t)Cy + \int_0^t S(t-s)A(s)y\,ds, \quad t \in [0,\tau), \ y \in Y,$$

hold; (215), resp. (216), is called the first resolvent equation, resp. the second resolvent equation.

An (A, k, B)-regularized C-pseudoresolvent family $(S(t))_{t \in [0,\tau)}$ is said to be an (A, k, B)-regularized C-resolvent family if additionally:

(S4) For every $y \in Y$, $S(\cdot)y \in L^{\infty}_{loc}([0, \tau) : Y)$.

An operator family $(S(t))_{t\in[0,\tau)}$ in L(X, [D(B)]) is called a weak (A, k, B)-regularized C-pseudoresolvent family iff (S1) and (216) hold. Finally, a weak (A, k, B)-regularized C-pseudoresolvent family $(S(t))_{t\in[0,\tau)}$ is said to be a-regular $(a \in L^1_{loc}([0,\tau)))$ iff $a * S(\cdot)x \in C([0,\tau) : Y), x \in \overline{Y}^X$.

Let us agree on the following: A (weak) (A, k, B)-regularized C-(pseudo)resolvent family with $k(t) \equiv g_{\alpha+1}(t)$, where $\alpha \ge 0$, is also called a (weak) α -times integrated (A, B)-regularized C-(pseudo)resolvent family, whereas a (weak) 0-times integrated (A, B)-regularized C-(pseudo)resolvent family is also said to be a (weak) (A, B)-regularized C-(pseudo)resolvent family. A (weak) (A, k, B)-regularized C-(pseudo)resolvent family is also said to be a (weak) (A, B)-regularized C-(pseudo)resolvent family. A (weak) (A, k, B)-regularized C-(pseudo)resolvent family is also called a (weak) (A, k, B)-regularized (pseudo)resolvent family (for C = I (if C = I and $k(t) \equiv 1$).

As in non-degenerate case, the integral appearing in the first resolvent equation (215) is understood in the sense of discussion following [459, Definition 6.2, p. 153].

Observe also that the condition (S3) can be rewritten in the following equivalent form:

(S3)'
$$BU(t)y = \Theta(t)Cy + \int_0^t A(t-s)U(s)y \, ds, \quad t \in [0,\tau), \ y \in Y,$$
$$BU(t)y = \Theta(t)Cy + \int_0^t U(t-s)A(s)y \, ds, \quad t \in [0,\tau), \ y \in Y.$$

By the norm continuity we mean the continuity in L(X) and, in many places, we do not distinguish $S(\cdot)$ $(U(\cdot))$ and its restriction to Y.

The notion of an (A, k, B)-regularized C-uniqueness family plays a crucial role in proving the uniqueness of solutions of abstract degenerate Cauchy problem (213).

DEFINITION 2.9.3. Let $\tau \in (0, \infty]$, $k \in C([0, \tau))$, $k \neq 0$ and $A \in L^1_{loc}([0, \tau) : L(Y, X))$. A strongly continuous operator family $(V(t))_{t \in [0, \tau)} \subseteq L(X)$ is said to be an (A, k, B)-regularized C-uniqueness family iff

$$V(t)By = k(t)Cy + \int_0^t V(t-s)A(s)y \, ds, \quad t \in [0,\tau), \ y \in Y \cap D(B).$$

Before stating the following propositions, whose proofs can be deduced as in non-degenerate case (cf. [459] and [299]), we want to observe that the notion of an (A, k, I)-regularized *C*-uniqueness family is a special case of the notion of a weak (A, k, I)-regularized *C*-pseudoresolvent family and the assertion of [299, Proposition 2(i)] holds even if the condition $S(t)C = CS(t), t \in [0, \tau)$ is disregarded (cf. also Proposition 2.9.5(i) below).

- PROPOSITION 2.9.4. (i) Suppose that $(S_1(t))_{t \in [0,\tau)}$ is an (A, k_1, B) -regularized C_1 -pseudoresolvent family and $(S_2(t))_{t \in [0,\tau)}$ is an (A, k_2, B) -regularized C_2 -uniqueness family. Then $C_2(k_2 * S_1)(t)x = (k_1 * S_2)(t)C_1x$, $t \in [0,\tau)$, $x \in \bar{Y}^X$.
- (ii) Let $(S(t))_{t \in [0,\tau)}$ be an (A, k, B)-regularized C-pseudoresolvent family. Assume that Y has the Radon-Nikodym property. Then $(S(t))_{t \in [0,\tau)}$ is an (A, k, B)-regularized C-resolvent family. Furthermore, if Y is reflexive, then $S(t)(Y) \subseteq Y$, $t \in [0, \tau)$ and the mapping $t \mapsto S(t)y$, $t \in [0, \tau)$ is weakly continuous in Y for all $y \in Y$.
- PROPOSITION 2.9.5. (i) Assume that $(V(t))_{t \in [0,\tau)}$ is an (A, k, B)-regularized C-uniqueness family, $f \in C([0,\tau) : X)$ and u(t) is a mild solution of (213). Then $(kC * u)(t) = (V * f)(t), t \in [0,\tau)$ and mild solutions of (213) are unique provided in addition that k(t) is a kernel on $[0,\tau)$.
- (ii) Assume $n \in \mathbb{N}$, $(S(t))_{t \in [0,\tau)}$ is an (n-1)-times integrated (A, B)-regularized C-pseudoresolvent family, $C^{-1}f \in C^{n-1}([0,\tau):X)$ and $f^{(i)}(0) = 0$, $0 \leq i \leq n-1$. Then the following assertions hold:
 - (a) Let $(C^{-1}f)^{(n-1)} \in AC_{loc}([0,\tau):Y)$ and $(C^{-1}f)^{(n)} \in L^{1}_{loc}([0,\tau):Y)$. Then the function $t \mapsto u(t), t \in [0,\tau)$ given by

$$u(t) = \int_0^t S(t-s)(C^{-1}f)^{(n)}(s)ds = \int_0^t dU(s)(C^{-1}f)^{(n)}(t-s)$$

is a strong solution of (213). Moreover, $u \in C([0, \tau) : Y)$.

- (b) Let $(C^{-1}f)^{(n)} \in L^{1}_{loc}([0,\tau):X)$ and $\bar{Y}^{X} = X$. Then the function $u(t) = \int_{0}^{t} S(t-s)(C^{-1}f)^{(n)}(s)ds, t \in [0,\tau)$ is a mild solution of (213).
- (c) Let $C^{-1}g \in W^{n,1}_{loc}([0,\tau) : \bar{Y}^X)$, $a \in L^1_{loc}([0,\tau))$, $f(t) = (g_n * a * g^{(n)})(t)$, $t \in [0,\tau)$ and let $(S(t))_{t \in [0,\tau)}$ be a-regular. Then the function $u(t) = \int_0^t S(t-s)(a * (C^{-1}g)^{(n)})(s)ds$, $t \in [0,\tau)$ is a strong solution of (213).

The uniqueness of solutions in (a), (b) or (c) holds provided that for each $y \in Y \cap D(B)$ we have $S(t)By = BS(t)y, t \in [0, \tau)$.

(iii) Let $(S(t))_{t\in[0,\tau)}$ be an (A,k,B)-regularized C-resolvent family. Put $u(t;y) := S(t)y, t \in [0,\tau), y \in Y$. Then u(t;y) is a strong solution of (214), and (214) is (kC)-well posed if k(t) is a kernel on $[0,\tau)$ and $S(t)By = BS(t)y, t \in [0,\tau), y \in Y \cap D(B)$.

Before we clarify a Hille–Yosida type theorem for (A, k, B)-regularized C-pseudoresolvent families, it should be observed that there exists a great number of statements from [299] which can be reconsidered in the degenerate case; without going into details, we only want to capture our readers' attention to the assertions of [299, Proposition 1(iii)(b)/(c), Proposition 3(ii)–(iii), Proposition 4, Remark 1, Theorem 2, Remark 2, Proposition 5]. It is also worth noting that the class of (A, k, B)-regularized C-uniqueness families can be characterized through the vector-valued Laplace transform and we need the condition $S(t)By = BS(t)y, t \ge 0, y \in Y \cap D(B)$ (see the formulation of Theorem 2.9.6) in order to show the injectiveness of operator $B - \tilde{A}(\lambda)$ for $\operatorname{Re} \lambda > \omega_0$, $\tilde{k}(\lambda) \neq 0$ (in Theorem 2.9.10 below, this condition will not be used).

THEOREM 2.9.6. Assume $A \in L^1_{loc}([0,\tau) : L(Y,X))$, $a \in L^1_{loc}([0,\tau))$, $a \neq 0$, |a|(t) and k(t) satisfy (P1), $\varepsilon_0 \ge 0$ and

(217)
$$\int_0^\infty e^{-\varepsilon t} \|A(t)\|_{L(Y,X)} dt < \infty, \quad \varepsilon > \varepsilon_0.$$

(i) Let $(S(t))_{t\geq 0}$ be an (A, k, B)-regularized C-pseudoresolvent family such that $S(t)By = BS(t)y, t \geq 0, y \in Y \cap D(B)$ and there exists $\omega \geq 0$ with

(218)
$$\sup_{t>0} e^{-\omega t} \Big(\|S(t)\|_{L(X)} + \|BS(t)\|_{L(X)} + \sup_{0 < s < t} (t-s)^{-1} \|U(t) - U(s)\|_{L(Y)} \Big) <$$

Put $\omega_0 := \max(\omega, \operatorname{abs}(k), \varepsilon_0)$ and $H(\lambda)x := \int_0^\infty e^{-\lambda t} S(t) x \, dt, \ x \in X$, Re $\lambda > \omega_0$. Then the following holds:

- (N1) $(\tilde{A}(\lambda))_{\operatorname{Re}\lambda>\varepsilon_0}$ is analytic in L(Y,X), $R(C_{|Y}) \subseteq R(B-\tilde{A}(\lambda))$, $\operatorname{Re}\lambda > \omega_0$, $\tilde{k}(\lambda) \neq 0$, and $B \tilde{A}(\lambda)$ is injective, $\operatorname{Re}\lambda > \omega_0$, $\tilde{k}(\lambda) \neq 0$.
- (N2) $H(\lambda)y = \lambda \tilde{U}(\lambda)y, y \in Y$, $\operatorname{Re} \lambda > \omega_0, (B \tilde{A}(\lambda))^{-1}C_{|Y} \in L(Y),$ $\operatorname{Re} \lambda > \omega_0, \ \tilde{k}(\lambda) \neq 0, \ (H(\lambda))_{\operatorname{Re} \lambda > \omega_0}$ is analytic in both spaces, L(X)

 ∞ .

and L(Y), $H(\lambda)C = CH(\lambda)$, $\operatorname{Re} \lambda > \omega_0$, $H(\lambda)By = BH(\lambda)y$, $\operatorname{Re} \lambda > \omega_0$, $y \in Y \cap D(B)$, and for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega_0$ and $\tilde{k}(\lambda) \neq 0$, the following holds:

$$(B - \tilde{A}(\lambda))H(\lambda)y = \tilde{k}(\lambda)Cy, \ y \in Y \ and \ H(\lambda)(B - \tilde{A}(\lambda))y = \tilde{k}(\lambda)Cy, \ y \in Y \cap D(B).$$

(N3)
$$\sup_{n \in \mathbb{N}_{0}} \sup_{\lambda > \omega_{0}, \tilde{k}(\lambda) \neq 0} \frac{(\lambda - \omega)^{n+1}}{n!} \left(\left\| \frac{d^{n}}{d\lambda^{n}} H(\lambda) \right\|_{L(X)} + \left\| \frac{d^{n}}{d\lambda^{n}} BH(\lambda) \right\|_{L(X)} + \left\| \frac{d^{n}}{d\lambda^{n}} H(\lambda) \right\|_{L(Y)} \right) < \infty$$

- (ii) Assume that (N1)-(N3) hold. Then there exists an exponentially bounded (A, Θ, B) -regularized C-resolvent family $(S_1(t))_{t \ge 0}$ satisfying $S_1(t)By = BS_1(t)y, t \ge 0, y \in Y \cap D(B)$.
- (iii) Assume that (N1)-(N3) hold, $B^{-1} \in L(X)$ and $\bar{Y}^X = X$. Then there exists an exponentially bounded (A, k, B)-regularized C-pseudoresolvent family $(S(t))_{t\geq 0}$ such that (218) holds and $S(t)By = BS(t)y, t \geq 0, y \in Y \cap D(B)$.
- (iv) Assume $(S(t))_{t \ge 0}$ is an (A, k, B)-regularized C-pseudoresolvent family satisfying (218) with some $\omega \ge 0$. Let $\omega' \ge \max(\omega, \operatorname{abs}(|a|), \operatorname{abs}(k), \varepsilon_0)$. Then $(S(t))_{t \ge 0}$ is a-regular and $\sup_{t \ge 0} e^{-\omega' t} ||a * S(t)||_{L(\bar{Y}^X, Y)} < \infty$ iff there exists a number $\omega_1 \ge \omega'$ such that

$$\sup_{n \in \mathbb{N}_0} \sup_{\lambda > \omega_1, \tilde{k}(\lambda) \neq 0} \frac{(\lambda - \omega')^{n+1}}{n!} \left\| \frac{d^n}{d\lambda^n} (\tilde{a}(\lambda) H(\lambda)) \right\|_{L(\bar{Y}^X, Y)} < \infty.$$

The hyperbolic perturbation results for non-scalar Volterra equations have been studied in [459, Theorem 6.1, p. 159] and [299, Theorem 3]. It is worth noting that the above-mentioned results can be generalized to degenerate non-scalar Volterra equations. More precisely, the following theorem holds true (the proof can be deduced by slightly modifying the corresponding proof of [459, Theorem 6.1], with $K_0 = S * C^{-1}BD_0$ and $K_1 = S * C^{-1}BD_1$):

THEOREM 2.9.7. Assume $L^1_{loc}([0,\tau)) \ni a$ is a kernel on $[0,\tau)$, $C(Y) \subseteq Y$, $\bar{Y}^X = X$, $CB \subseteq BC$,

$$D(t)y = D_0(t)y + (a * D_1)(t)y, \quad t \in [0, \tau), \ y \in Y,$$

where $(D_0(t))_{t\in[0,\tau)} \subseteq L(Y)$, $(BD_0(t))_{t\in[0,\tau)} \subseteq L(X, [R(C)])$, $(BD_1(t))_{t\in[0,\tau)} \subseteq L(Y, [R(C)])$,

- (i) $C^{-1}BD_0(\cdot)y \in BV_{loc}([0,\tau):Y)$ for all $y \in Y$, $C^{-1}BD_0(\cdot)x \in BV_{loc}([0,\tau):X)$ for all $x \in X$,
- (ii) $C^{-1}BD_1(\cdot)y \in BV_{loc}([0,\tau):X)$ for all $y \in Y$, and
- (iii) $CBD(t)y = BD(t)Cy, y \in Y, t \in [0, \tau).$

Then the existence of an a-regular (A, B)-regularized C-(pseudo)resolvent family $(S(t))_{t \in [0,\tau)}$ is equivalent with the existence of an a-regular (A+BD, B)-regularized C-(pseudo)resolvent family $(R(t))_{t \in [0,\tau)}$.

Theorem 2.9.7 can be applied to abstract degenerate non-scalar Volterra equations involving abstract differential operators. For example, let 1 .let $X := L^p(\mathbb{R}^n)$, and let $0 < \alpha < 2$. Then it is clear that the operators $\partial/\partial x_i$, acting with their maximal distributional domains, are commuting generators of bounded C_0 -groups on X; set $A := (-i\partial/\partial x_1, \ldots, -i\partial/\partial x_n)$. Suppose that $P_1(x)$ and $P_2(x)$ are non-zero complex polynomials satisfying $P_2(x) \neq 0$. $x \in \mathbb{R}^n$ and the estimate (70). Define the strongly continuous operator family $(G_{\alpha}(t))_{t\geq 0} \subset L(X)$ in the same way as it has been done in Remark 2.3.16(ii), and the operators $\overline{P_i(A)}$, i = 1, 2 in the usual way. Set $Y := D(\overline{P_1(A)}) \cap D(\overline{P_2(A)})$, $\|f\| := \|f\|_X + \|\overline{P_1(A)}f\|_X + \|\overline{P_2(A)}f\|_X \ (f \in Y), \ A(t) := g_\alpha(t)\overline{P_1(A)}_{|_V} \ (t > 0)$ and $C := \overline{P_2(A)}G_{\alpha}(0)$. Then $(Y, \|\cdot\|_Y)$ is a Banach space continuously embedded in X, $\overline{Y}^X = X$, $C\overline{P_2(A)} \subseteq \overline{P_2(A)}C$, $C(Y) \subseteq Y$, and a simple analysis shows that $(G_{\alpha}(t))_{t\geq 0}$ is an $(A, \overline{P_2(A)})$ -resolvent C-regularized resolvent family obeying the property that for each $f \in Y$ the mapping $t \mapsto \int_0^t G_\alpha(s) f \, ds$, $t \ge 0$ is continuously differentiable in Y. Therefore, Theorem 2.9.7 can be applied with the regularizing operator C being in general the non-identity operator on X. We continue by introducing the following definition (cf. [299, Definition 3(i)] for non-degenerate case).

DEFINITION 2.9.8. Let $k \in C([0,\infty)), k \neq 0, A \in L^1_{loc}([0,\infty) : L(Y,X)), \alpha \in (0,\pi]$, and let $(S(t))_{t\geq 0} \subseteq L(X, [D(B)])$ be a (weak) (A, k, B)-regularized C-(pseudo)resolvent family. Then it is said that $(S(t))_{t\geq 0}$ is an analytic (weak) (A, k, B)-regularized C-(pseudo)resolvent family of angle α , if there exists an analytic function $\mathbf{S} \colon \Sigma_{\alpha} \to L(X, [D(B)])$ satisfying $\mathbf{S}(t) = S(t), t > 0, \lim_{z\to 0, z\in \Sigma_{\gamma}} \mathbf{S}(z)x = S(0)x$ and $\lim_{z\to 0, z\in \Sigma_{\gamma}} \mathbf{S}(z)x = BS(0)x$ for all $\gamma \in (0, \alpha)$ and $x \in X$. We say that $(S(t))_{t\geq 0}$ is an exponentially bounded, analytic (weak) (A, k, B)-regularized C-(pseudo)resolvent family, resp. bounded analytic (weak) (A, k, B)-regularized C-(pseudo)resolvent family, of angle α , if $(S(t))_{t\geq 0}$ is an analytic (weak) (A, k, B)-regularized C-(pseudo)resolvent family, of angle α , if $(S(t))_{t\geq 0}$ is an analytic (weak) (A, k, B)-regularized C-(pseudo)resolvent family, of angle α , if $(S(t))_{t\geq 0}$ is an analytic (weak) (A, k, B)-regularized C-(pseudo)resolvent family, of angle α , if $(S(t))_{t\geq 0}$ is an analytic (weak) (A, k, B)-regularized C-(pseudo)resolvent family, of angle α , if $(S(t))_{t\geq 0}$ is an analytic (weak) (A, k, B)-regularized C-(pseudo)resolvent family, of angle α , if $(S(t))_{t\geq 0}$ is an analytic (weak) (A, k, B)-regularized C-(pseudo)resolvent family, of angle α , if $(S(t))_{t\geq 0}$ is an analytic (weak) (A, k, B)-regularized C-(pseudo)resolvent family, of angle α , if $(S(t))_{t\geq 0}$ is an analytic (weak) (A, k, B)-regularized C-(pseudo)resolvent family, of angle α , if $(S(t))_{t\geq 0}$ is an analytic (weak) (A, k, B)-regularized C-(pseudo)resolvent family of angle α and for each $\gamma \in (0, \alpha)$ there exist $M_{\gamma} > 0$ and $\omega_{\gamma} \ge 0$, resp. $\omega_{\gamma} = 0$, such that $||\mathbf{S}(z)||_{L(X)} + ||B\mathbf{S}(z)||_{L(X)} \le M_{\gamma}e^{\omega_{\gamma}|z|}, z \in \Sigma_{\gamma}$. Since no confusion seems likely, we shall identify $S(\cdot)$ and $\mathbf{S}(\cdot)$ in the sequel.

The most important properties of exponentially bounded, analytic (weak) (A, k, B)-regularized C-(pseudo)resolvent families are collected in the subsequent theorems.

- THEOREM 2.9.9. (i) Assume $\varepsilon_0 \ge 0$, k(t) satisfies (P1), $\omega \ge \max(\operatorname{abs}(k), \varepsilon_0)$, (217) holds, $(S(t))_{t\ge 0}$ is a weak analytic (A, k, B)regularized C-pseudoresolvent family of angle $\alpha \in (0, \pi/2]$ and
- (219) $\sup_{z\in\Sigma_{\gamma}} \left[\|e^{-\omega z}S(z)\|_{L(X)} + \|e^{-\omega z}BS(z)\|_{L(X)} \right] < \infty \quad \text{for all } \gamma \in (0,\alpha).$

Then there exists an analytic mapping $H : \omega + \sum_{\frac{\pi}{2} + \alpha} \to L(X, [D(B)])$ such that:

(a) $BH(\lambda)y - H(\lambda)\tilde{A}(\lambda)y = \tilde{k}(\lambda)Cy, \ y \in Y, \ \operatorname{Re}\lambda > \omega, \ \tilde{k}(\lambda) \neq 0;$ $H(\lambda)C = CH(\lambda), \ \operatorname{Re}\lambda > \omega,$

- (b) $\sup_{\lambda \in \omega + \Sigma_{\frac{\pi}{2} + \gamma}} \left[\| (\lambda \omega) H(\lambda) \|_{L(X)} + \| (\lambda \omega) BH(\lambda) \|_{L(X)} \right] < \infty$ for all $\gamma \in (0, \alpha)$.
- (c) there exists an operator F ∈ L(X, [D(B)]) such that BFx = k(0)Cx, x ∈ X and lim_{λ→+∞, k̃(λ)≠0} λH(λ)x = Fx, x ∈ X, and
 (d) lim_{λ→+∞} k̃(λ)≠0 λBH(λ)x = k(0)Cx, x ∈ X.
- (ii) Assume $\varepsilon_0 \ge 0$, k(t) satisfies (P1), (217) holds, $\omega \ge \max(\operatorname{abs}(k), \varepsilon_0)$, $\alpha \in$
- (ii) Assume $c_0 \ge 0$, $\kappa(t)$ satisfies (1.1), (211) holds, $\omega \ge \max(\operatorname{abs}(\kappa), c_0)$, $\alpha \in (0, \pi/2]$, there exists an analytic mapping $H : \omega + \sum_{\frac{\pi}{2} + \alpha} \to L(X, [D(B)])$ such that (a), (b) and (c) of the item (i) hold and that, in the case $\bar{Y}^X \ne X$, (d) also holds. Then there exists a weak analytic (A, k, B)-regularized C-pseudoresolvent family $(S(t))_{t\ge 0}$ of angle α such that (219) holds.

THEOREM 2.9.10. (i) Assume $\varepsilon_0 \ge 0$, k(t) satisfies (P1), $\omega_0 \ge \max(\operatorname{abs}(k), \varepsilon_0)$, (217) holds, $\alpha \in (0, \pi/2]$, $(S(t))_{t\ge 0}$ is an analytic (A, k, B)-regularized C-resolvent family of angle α , the mapping $t \mapsto U(t) \in L(Y)$, t > 0 can be analytically extended to the sector Σ_{α} (we shall denote the analytical extensions of $U(\cdot)$ and $S(\cdot)$ in the space L(Y) by the same symbols), and for each $\gamma \in (0, \alpha)$ one has:

(220)
$$\sup_{z \in \Sigma_{\gamma}} \left[\|e^{-\omega_0 z} S(z)\|_{L(X)} + \|e^{-\omega_0 z} BS(z)\|_{L(X)} + \sup_{z \in \Sigma_{\gamma}} \|e^{-\omega_0 z} S(z)\|_{L(Y)} \right] < \infty.$$

Denote $H(\lambda)x = \int_0^\infty e^{-\lambda t} S(t)x \, dt$, $x \in X$, $\operatorname{Re} \lambda > \omega_0$. Then $H: \omega + \sum_{\frac{\pi}{2} + \alpha} \to L(X, [D(B)])$ is an analytic mapping, $H_{|Y}: \omega + \sum_{\frac{\pi}{2} + \alpha} \to L(Y)$ is likewise an analytic mapping, and the following holds:

- (a) $\sup_{\lambda \in \omega_0 + \Sigma_{\frac{\pi}{2} + \gamma}} \left[\| (\lambda \omega_0) H(\lambda) \|_{L(X)} + \| (\lambda \omega_0) B H(\lambda) \|_{L(X)} + \| (\lambda \omega_0) H(\lambda) \|_{L(Y)} \right] < \infty \text{ for all } \gamma \in (0, \alpha),$
- (b) $BH(\lambda)y H(\lambda)\tilde{A}(\lambda)y = \tilde{k}(\lambda)Cy, \ y \in Y, \ \operatorname{Re}\lambda > \omega, \ \tilde{k}(\lambda) \neq 0;$ $BH(\lambda)y - \tilde{A}(\lambda)H(\lambda)y = \tilde{k}(\lambda)Cy, \ y \in Y, \ \operatorname{Re}\lambda > \omega, \ \tilde{k}(\lambda) \neq 0;$ $H(\lambda)C = CH(\lambda), \ \operatorname{Re}\lambda > \omega_0,$
- (c) there exists an operator $F \in L(X, [D(B)])$ such that BFx = k(0)Cx, $x \in X$, $\lim_{\lambda \to +\infty, \tilde{k}(\lambda) \neq 0} \lambda H(\lambda)x = Fx$, $x \in X$, and
- (d) $\lim_{\lambda \to +\infty, \tilde{k}(\lambda) \neq 0} \lambda B H(\lambda) x = k(0) C x, x \in X.$
- (ii) Assume α ∈ (0,π/2], ε₀ ≥ 0, k(t) satisfies (P1) and (217) holds. Let ω₀ ≥ max(abs(k), ε₀), and let there exist an analytic mapping H: ω + Σ_{π/2}+α → L(X, [D(B)]) such that H_{|Y}: ω + Σ_{π/2}+α → L(Y) is an analytic mapping, as well as that (a)-(c) of the item (i) of this theorem hold and, in the case Ȳ^X ≠ X, (d) also holds. Then there exists an analytic (A, k, B)-regularized C-resolvent family (S(t))_{t≥0} of angle α such that (220) holds and the mapping t ↦ U(t) ∈ L(Y), t > 0 can be analytically extended to the sector Σ_α.

REMARK 2.9.11. If $B^{-1} \in L(X)$, then the condition (i)/(c) in Theorem 2.9.9, i.e., the condition (i)/(c) in Theorem 2.9.10, automatically holds. Therefore, Theorem 2.9.9 and Theorem 2.9.10 taken together provide extensions of [**299**, Theorem 4-Theorem 5].

Based on the evidence contained in the proofs of [299, Theorem 6-Theorem 7], we can simply clarify the basic results about differentiability of (A, k, B)-regularized C-pseudoresolvent families in Banach spaces. Although interesting, this theme will not be considered here; let us only mention that the considerations from [299, Example 3] and [314, Example 2.5] enable one to construct some important examples of (local) (A, k, B)-regularized resolvent families with a certain hypoanalytic behaviour.

2.10. Hypercyclic and topologically mixing properties of abstract degenerate fractional differential equations

In this section, which is broken down into four subsections, we analyze hypercyclic and topologically mixing properties of various classes of abstract (multi-term) degenerate fractional differential equations with Caputo derivatives. In Subsection 2.10.1–Subsection 2.10.3, by $(E, \|\cdot\|)$ we denote a non-trivial separable Banach space over the field \mathbb{C} (some of our results presented in these subsections continue to hold in the setting of separable, infinite-dimensional, Hausdorff, sequentially complete locally convex spaces over the field $\mathbb{K} \in \{\mathbb{R}, \mathbb{C}\}$; cf. [112, 292, Section 3.1] and [309] for more details).

2.10.1. Hypercyclic and topologically mixing properties of solutions of the equations $\mathbf{D}_t^{\alpha} Bu(t) = Au(t)$ and $B\mathbf{D}_t^{\alpha} u(t) = Au(t)$ ($\alpha > 0$). Let A and Bbe closed linear operators acting on E, and let $C \in L(E)$ be an injective operator. If $\emptyset \neq \Omega \subseteq \mathbb{C} \setminus (-\infty, 0]$ and $\alpha > 0$, set $\Omega^{\alpha} := \{\lambda^{\alpha} : \lambda \in \Omega\}$. By \mathcal{D} we denote the set which consists of all infinitely differentiable, compactly supported functions from \mathbb{R} into \mathbb{C} ; if $\emptyset \neq K \subseteq \mathbb{R}$, then we define $\mathcal{D}_K := \{\varphi \in \mathcal{D} : \operatorname{supp}(\varphi) \subseteq K\}$. We enquire into the hypercyclic and topologically mixing solutions of homogeneous counterparts of fractional Sobolev equations

$$(\mathrm{DFP})_R : \begin{cases} \mathbf{D}_t^{\alpha} Bu(t) = Au(t) + f(t), & t \ge 0, \\ Bu(0) = Bx; & (Bu)^{(j)}(0) = 0, \ 1 \le j \le \lceil \alpha \rceil - 1 \end{cases}$$

and

$$(\mathrm{DFP})_L : \begin{cases} B\mathbf{D}_t^{\alpha}u(t) = Au(t) + f(t), & t \ge 0, \\ u(0) = x; & u^{(j)}(0) = 0, \ 1 \le j \le \lceil \alpha \rceil - 1 \end{cases}$$

Throughout the whole section, we will use the same symbols to denote such problems; this little abuse of notation will not create any confusion henceforward. Suppose that the function u(t) is a strong solution of problem $(DFP)_R$ $((DFP)_L)$; cf. Definition 2.3.1 (in the existing literature on non-degenerate first order equations, this is equivalent to say that u(t) is a classical solution of problem $(DFP)_R$; cf. [445]). Convoluting both sides of $(DFP)_R$ $((DFP)_L)$ with $g_\alpha(t)$, and using the equality (38), it readily follows that every solution of $(DFP)_R$ $((DFP)_L)$ satisfies the equality

$$Bu(t) - Bx = (g_{\alpha} * Au(\cdot))(t), \quad t \ge 0 \qquad (B(u(t) - x) = (g_{\alpha} * Au(\cdot))(t), \quad t \ge 0).$$

If a continuous function $t \mapsto u(t), t \ge 0$ satisfies the first of these equations with $Au, Bu \in C([0,\infty) : E)$, then it can be easily checked that u(t) is a strong solution of problem $(DFP)_{R}$; observe, however, that we cannot prove the corresponding statement for the problem $(DFP)_L$. Let us recall that a mild solution of the problem $(DFP)_R$ is any continuous function $t \mapsto u(t), t \ge 0$ such that the mapping $t \mapsto Bu(t), t \ge 0$ is continuous and $A(g_{\alpha} * u)(t) = Bu(t) - Bx$, $t \ge 0$. Denote by $Z_{\alpha,R}(A,B)$ $(Z_{\alpha,L}(A,B))$ the set which consists of those elements $x \in D(B)$ ($x \in D(A)$) for which there exists a strong solution of $(DFP)_R$ $((\text{DFP})_L)$. By $Z_{\alpha,R}^{uniq}(A,B)$ $(Z_{\alpha,L}^{uniq}(A,B))$ we denote the set which consists of those elements $x \in D(B)$ ($x \in D(A)$) for which there exists a unique strong solution of $(DFP)_R$ $((DFP)_L)$. Then $Z_{\alpha,R}(A,B)$ $(Z_{\alpha,L}(A,B))$ is a linear subspace of E, and $Z_{\alpha,R}^{uniq}(A,B)$ $(Z_{\alpha,L}^{uniq}(A,B))$ is a linear subspace of E provided that the zero function is a unique strong solution of the problem $(DFP)_R$ $((DFP)_L)$ with x = 0. In this case, we define for each $t \ge 0$ the linear operator $T_R(t; \cdot) : Z_{\alpha, R}^{uniq}(A, B) \rightarrow$ $E (T_L(t; \cdot) : Z^{uniq}_{\alpha,L}(A, B) \rightarrow E)$ by $T_R(t; x) := u(t; x), x \in Z^{uniq}_{\alpha,R}(A, B)$ $(x \in C^{uniq}_{\alpha,R}(A, B))$ $Z^{uniq}_{\alpha L}(A,B)$, where $u(\cdot;x)$ denotes the strong solution of $(DFP)_R$ ((DFP)_L).

In this subsection, we will restrict ourselves to the analysis of (subspace) hypercyclicity, topological transitivity and topological mixing property of problem $(DFP)_R$ ($(DFP)_L$); in Subsection 2.10.3, we will also look into the (subspace) chaoticity and weakly mixing property of first and second order degenerate equations (cf. [**292**, Chapter 3] and references cited there, as well as Subsection 2.10.4, for some other dynamical properties of abstract degenerate differential equations that are not considered in Subsection 2.10.1–Subsection 2.10.3).

DEFINITION 2.10.1. Let $\alpha > 0$, and let \tilde{E} be a closed linear subspace of E. Then it is said that:

(i) an element $x \in Z_{\alpha,R}(A,B) \cap \tilde{E}$ $(x \in Z_{\alpha,L}(A,B) \cap \tilde{E})$ is a \tilde{E} -hypercyclic vector for $(DFP)_R$ $((DFP)_L)$ iff there exists a strong solution $t \mapsto u(t;x)$, $t \ge 0$ of problem $(DFP)_R$ $((DFP)_L)$ with the property that the set $\{u(t;x): t \ge 0\}$ is a dense subset of \tilde{E} .

Furthermore, we say that the abstract Cauchy problem $(DFP)_R$ $((DFP)_L)$ is:

- (ii) \tilde{E} -topologically transitive iff for every $y, z \in \tilde{E}$ and for every $\varepsilon > 0$, there exist $x \in Z_{\alpha,R}(A,B) \cap \tilde{E}$ $(x \in Z_{\alpha,L}(A,B) \cap \tilde{E})$, a strong solution $t \mapsto u(t;x), t \ge 0$ of problem $(DFP)_R$ $((DFP)_L)$ and $t \ge 0$ such that $||y-x|| < \varepsilon$ and $||z-u(t;x)|| < \varepsilon$;
- (iii) \tilde{E} -topologically mixing iff for every $y, z \in \tilde{E}$ and for every $\varepsilon > 0$, there exists $t_0 \ge 0$ such that, for every $t \ge t_0$, there exist $x_t \in Z_{\alpha,R}(A,B) \cap \tilde{E}$ $(x_t \in Z_{\alpha,L}(A,B) \cap \tilde{E})$ and a strong solution $t \mapsto u(t;x_t), t \ge 0$ of problem $(\text{DFP})_R$ $((\text{DFP})_L)$ such that $||y x_t|| < \varepsilon$ and $||z u(t;x_t)|| < \varepsilon$.

In the case $\tilde{E} = E$, it is also said that a \tilde{E} -hypercyclic vector for $(DFP)_R$ $((DFP)_L)$ is a hypercyclic vector for $(DFP)_R$ $((DFP)_L)$, and that $(DFP)_R$ $((DFP)_L)$ is topologically transitive, resp. topologically mixing.

Before proceeding further, it should be noted that the study we have taken up shows that there is no substantial difference in the analysis of hypercyclic properties of problems $(DFP)_R$ and $(DFP)_L$, and there is no ideal option that would provide us a general method for studying hypercyclic properties of abstract degenerate equations. In Definition 2.10.1, which is, in our opinion, the best theorethical

provide us a general method for studying hypercyclic properties of abstract degenerate equations. In Definition 2.10.1, which is, in our opinion, the best theorethical concept for the analysis of hypercyclicity of abstract degenerate equations, we work only with strong solutions. This is primarily caused by the fact that it is very difficult to define the notion of a mild solution of the problem $(DFP)_L$ in a satisfactory way, as well as that the solutions appearing in Theorem 2.10.3 below, which is the main result of this subsection, are strong in fact (in the case of problem $(DFP)_B$, the notions from Definition 2.10.1 can be introduced with mild solutions instead of strong ones, in the very obvious way: this has been the usual way for introducing \tilde{E} -hypercyclic vectors, \tilde{E} -topological transitivity and \tilde{E} -topologically mixing property of non-degenerate equations so far). As a large class of important examples shows (see e.g. Example 2.10.5–Example 2.10.6 below), it is much better to introduce the notions from Definition 2.10.1 with the set $Z_{\alpha,R}(A,B)$ $(Z_{\alpha,L}(A,B))$ than with $Z_{\alpha,R}^{uniq}(A,B)$ $(Z_{\alpha,L}^{uniq}(A,B))$; cf. [6,95,424] and [210] for some results ensuring the uniqueness of solutions of problem $Z_{\alpha,R}(A,B)$ $(Z_{\alpha,L}(A,B))$. Having things set up in such a way, we will face some very strange pecularities of hypercyclic degenerate equations later (cf. Example 2.10.15). Even in the case that B = I, we do not assume in Definition 2.10.1 the existence of a global (q_{α}, q_{β}) -regularized C-resolvent family for $(DFP)_R$ ($(DFP)_L$), which has been the general framework for the analysis of hypercyclic abstract PDEs so far $(\beta \ge 1, C \in L(E))$ injective).

We continue by observing that, for any arbitrarily large number $\alpha > 0$, we can simply construct a Banach space E and closed linear operators A and B on E such that the problems $(DFP)_R$ and $(DFP)_L$ are both topologically mixing. For example, let E be the weighted l^1 -space $l_r^1 := \{(x_k)_{k\in\mathbb{N}} : x_k \in \mathbb{C}, \sum_{k=1}^{\infty} r_k |x_k| < \infty\}$, normed by $||(x_k)_{k\in\mathbb{N}}|| := \sum_{k=1}^{\infty} r_k |x_k|, (x_k)_{k\in\mathbb{N}} \in l_r^1$; here $(r_k)_{k\in\mathbb{N}}$ denotes any sequence of positive real numbers satisfying that there exists M > 0 such that $r_k r_{k+1}^{-1} \leq M$ for all $k \in \mathbb{N}$. Take any $B \in L(E)$ and define after that $A \in L(E)$ by $A(x_k)_{k\in\mathbb{N}} :=$ $(Bx_{k+1})_{k\in\mathbb{N}}, (x_k)_{k\in\mathbb{N}} \in l_r^1$; then, by [**301**, Theorem 2.3], the problems $(DFP)_R$ and $(DFP)_L$ are both topologically mixing. We leave to the interested reader details about construction of hypercyclic and topologically mixing degenerate (multi-term) fractional differential equations on weighted l^1 -spaces, involving operators that are certain functions of unilateral backward shift operators (cf. [**143**, **470**, Section 5] and [**301**] for further information).

Let E = E, let HC_R (HC_L) designate the set of all hypercyclic vectors for $(DFP)_R$ $((DFP)_L)$, and let $\mathcal{O} = (O_n)_{n \in \mathbb{N}}$ be an open base of the topology of E $(O_n \neq \emptyset, n \in \mathbb{N})$. Assuming that the strong solutions of problems $(DFP)_R$ and $(DFP)_L$ are unique, it can be simply proved that the equality $HC_{R,L} = \bigcap_{n \in \mathbb{N}} \bigcup_{t \geq 0} T_{R,L}(t)^{-1}(O_n)$ holds and $\bigcup_{t \geq 0} T_{R,L}(t)^{-1}(O_n)$ is a dense subset of E for all $n \in \mathbb{N}$. Regrettably, we cannot conclude from the above that the topological transitivity of $(DFP)_R$ $((DFP)_L)$ implies the existence of hypercyclic vectors for $(DFP)_R$ $((DFP)_L)$.

We need the following extension of [**300**, Lemma 2.1].

LEMMA 2.10.2. Suppose $\alpha > 0$, $\lambda \in \mathbb{C} \setminus \{0\}$, $x \in D(A) \cap D(B)$ and $Ax = \lambda Bx$. Then the function $u(t;x) := E_{\alpha}(\lambda t^{\alpha})x, t \ge 0$ satisfies $u, Au, Bu \in C([0,\infty):E)$, as well as the equalities

$$(g_{\alpha} * Au(\cdot; x))(t) = B(u(t; x) - x) = Bu(t; x) - Bx, \quad t \ge 0$$

Furthermore, the function $u(\cdot; x)$ is a strong solution of problem $(DFP)_R$ ($(DFP)_L$).

PROOF. The proof is very similar to that of [**300**, Lemma 2.1]; it is only worth noticing here that a direct computation combined with the equality $\mathbf{D}^{\alpha}_{\iota} E_{\alpha}(\lambda t^{\alpha}) =$ $\lambda E_{\alpha}(\lambda t^{\alpha}), t \ge 0$ (see e.g. [61, (1.25)]) implies that the function $t \mapsto u(t;x), t \ge 0$ is a strong solution of problem $(DFP)_L$.

In the following theorem, we rethink the Desch–Schappacher–Webb and Banasiak–Moszyński criteria for chaos of strongly continuous semigroups.

THEOREM 2.10.3. Suppose that $\alpha \in (0,2)$ and Ω is an open connected subset of \mathbb{C} which satisfies $\Omega \cap (-\infty, 0] = \emptyset$ and $\Omega \cap i\mathbb{R} \neq \emptyset$. Let $f: \Omega^{\alpha} \to E$ be an analytic mapping such that $f(\lambda^{\alpha}) \in \operatorname{Kern}(A - \lambda^{\alpha}B) \setminus \{0\}, \lambda \in \Omega$, and let $\check{E} :=$ $\overline{span\{f(\lambda^{\alpha}):\lambda\in\Omega\}}$. Then the problems $(DFP)_R$ and $(DFP)_L$ are \tilde{E} -topologically mixing; furthermore, if $Af(\lambda^{\alpha}) \in \tilde{E}$ for all $\lambda \in \Omega$, then the problems $(DFP)_{R}^{\tilde{E}}$ and $(DFP)_L^{\tilde{E}}$, obtained by replacing the operators A and B in $(DFP)_R$ and $(DFP)_L$ with the operators $A_{1\tilde{E}}$ and $B_{1\tilde{E}}$, respectively, are topologically mixing in the Banach space \tilde{E} .

PROOF. Without loss of generality, we may assume that $\Omega \cap i(0,\infty) \neq \emptyset$. Observe the following: If Ω_0 is an arbitrary non-empty subset of Ω which admits a cluster point in Ω , then the (weak) analyticity of mapping $\lambda \mapsto f(\lambda^{\alpha}) \in E$, $\lambda \in \Omega$ implies that $\Psi(\Omega_0) := span\{f(\lambda^\alpha) : \lambda \in \Omega_0\}$ is dense in the Banach space \vec{E} . Further on, it is clear that there exist numbers $\lambda_0 \in \Omega \cap i(0,\infty)$ and $\delta > 0$ such that any of the sets $\Omega_{0,+} := \{\lambda \in \Omega : |\lambda - \lambda_0| < \delta, \arg(\lambda) \in (\frac{\pi}{2} - \delta, \frac{\pi}{2})\}$ and $\Omega_{0,-} := \{\lambda \in \Omega : |\lambda - \lambda_0| < \delta, \arg(\lambda) \in (\frac{\pi}{2}, \frac{\pi}{2} + \delta)\}$ admits a cluster point in Ω as well as that $\arg(\lambda^{\alpha}t^{\alpha}) < \alpha\pi/2, \ \lambda \in \Omega_{0,+}$ and $\arg(-\lambda^{\alpha}t^{\alpha}) \in \pi - \alpha\pi/2, \ \lambda \in \Omega_{0,-}$. Due to Theorem 1.5.1 (cf. also (40)-42), we have:

(221)
$$E_{\alpha}(\lambda^{\alpha}t^{\alpha}) \to \infty, t \to \infty, \lambda \in \Omega_{0,+}$$
 and $E_{\alpha}(\lambda^{\alpha}t^{\alpha}) \to 0, t \to \infty, \lambda \in \Omega_{0,-}$.

Suppose that $y \in \Psi(\Omega_{0,-}), z \in \Psi(\Omega_{0,+}), \varepsilon > 0, y = \sum_{i=1}^{n} \beta_i f(\lambda_i^{\alpha}),$ $z = \sum_{j=1}^{m} \gamma_j f(\tilde{\lambda_j}^{\alpha}), \alpha_i, \beta_j \in \mathbb{C}, \lambda_i \in \Omega_{0,-} \text{ and } \tilde{\lambda_j} \in \Omega_{0,+} \text{ for } 1 \leqslant i \leqslant n \text{ and } 1 \leqslant j \leqslant m.$ Using (221), we get that there exists a number $t_0(z) > 0$ such that $E_{\alpha}(\tilde{\lambda_j}^{\alpha} t^{\alpha}) \neq 0, t \geqslant t_0(z).$ Put $z_t := \sum_{j=1}^{m} \frac{\gamma_j}{E_{\alpha}(\tilde{\lambda_j}^{\alpha} t^{\alpha})} f(\tilde{\lambda_j}^{\alpha})$ and $x_t := y + z_t,$ $t \ge t_0(z)$. Owing to Lemma 2.10.2, we have $\{x_t, y, z_t\} \subseteq Z_\alpha(A, B) \cap \tilde{E}, t \ge t_0(z);$ denote by $u(\cdot; x_t)$, $u(\cdot; y)$ and $u(\cdot; z_t)$ the corresponding strong solutions $(t \ge t_0(z))$. Keeping in mind (221) and Lemma 2.10.2, we obtain that there exists a number $t(y, z, \varepsilon) > t_0(z)$ such that: $||u(t; y)|| < \varepsilon$, $u(t; z_t) = z$ and $||z_t|| < \varepsilon$, $t \ge t(y, z, \varepsilon)$. Hence, $||x_t - y|| = ||z_t|| < \varepsilon$ and $||u(t; x_t) - z|| = ||u(t; y)|| < \varepsilon, t \ge t(y, z, \varepsilon)$, whence

FERTIES OF ...

it easily follows without any substantial difficulties that the problems $(DFP)_R$ and $(DFP)_L$ are \tilde{E} -topologically mixing. The rest of the proof is simple and therefore omitted.

REMARK 2.10.4. It is worth noting that Theorem 2.10.3 does not continue to hold in the case that $\alpha = 2$. Then we can pass to the theory of degenerate equations of first order on product spaces and derive, in such a way, some conclusions on the subspace topologically mixing (chaotic) properties of problems $(DFP)_R$ and $(DFP)_{I}$. In order to better explain these facts, suppose that $\emptyset \neq \Omega$ is an open connected subset of \mathbb{C} which intersects the imaginary axis, and that $f: \Omega^2 \to E$ is an analytic mapping satisfying $Af(\lambda^2) = \lambda^2 Bf(\lambda^2), \lambda \in \Omega$. Define the operators $\mathcal{A}, \mathcal{A}', \mathcal{B}$ and \mathcal{B}' on $E \times E$ by $\mathcal{A}(x, y) := (y, Ax), (x, y) \in E \times D(A)$ $\mathcal{A}'(x,y) := (By, Ax), (x,y) \in D(A) \times D(B), \ \mathcal{B}(x,y) := (x, By), (x,y) \in E \times D(B)$ and $\mathcal{B}'(x,y) := (Bx, By), (x,y) \in D(B) \times D(B)$. Then $\mathcal{A}(f(\lambda^2), \lambda f(\lambda^2)) =$ $\lambda \mathcal{B}(f(\lambda^2), \lambda f(\lambda^2)), \ \lambda \in \Omega \ \text{and} \ \mathcal{A}'(f(\lambda^2), \lambda f(\lambda^2)) = \lambda \mathcal{B}'(f(\lambda^2), \lambda f(\lambda^2)), \ \lambda \in \Omega.$ Making use of Theorem 2.10.3 with $\alpha = 1$, and the first of these equalities (resp., the second one), we can prove the validity of some results about (subspace) topologically mixing (chaotic) properties of the problem $(DFP)_L$, with the initial condition u'(0) = 0 replaced by u'(0) = y (resp., the problem (DFP)_B, with the initial condition (Bu)'(0) = 0 replaced by (Bu)'(0) = By; for more details, see [292, Lemma 3.2.33, Theorem 3.2.34, Remark 3.2.35].

Observe that the conclusions stated in [300, Remark 1(i),(iii)] can be reformulated for degenerate fractional differential equations. Now we want to illustrate Theorem 2.10.3 with some examples.

EXAMPLE 2.10.5. Suppose that $0 < \alpha < 2$, $n \in \mathbb{N} \setminus \{1\}$ and $E := L^p([0,\infty))$ $(1 \leq p < \infty)$. Let $P(z) = \sum_{j=0}^n a_j z^j$, $z \in \mathbb{C}$ be a complex polynomial of degree n $(a_j \in \mathbb{C}, 0 \leq j \leq n)$, and let $Q(z) = \sum_{j=0}^m b_j z^j$, $z \in \mathbb{C}$ be a non-zero complex polynomial of degree $m \leq n-1$ $(b_j \in \mathbb{C}, 0 \leq j \leq m)$. Suppose, further, that there exists $z_0 \in \mathbb{C}_- \equiv \{z \in \mathbb{C} : \operatorname{Re} z < 0\}$ such that $P(z_0) = 0$ and $P'(z_0)Q(z_0) \neq 0$. Set $Af := \sum_{j=0}^n a_j f^{(j)}$ and $Bf := \sum_{j=0}^m b_j f^{(j)}$ with maximal distributional domain; then A and B are closed linear operators on E. The mapping $z \mapsto (P/Q)(z)$ is analytic in a neighborhood of point z_0 so that there exists an open connected subset U of \mathbb{C} containing zero, satisfying additionally that the inverse mapping $(P/Q)^{-1}: U \to (P/Q)^{-1}(U)$ is well defined, analytic and that $(P/Q)^{-1}(U) \subseteq \mathbb{C}_-$. Let Ω be any open connected subset of $\mathbb{C} \setminus (-\infty, 0]$ such that $\Omega^{\alpha} \subseteq U$. Set $\tilde{E} := span\{\exp(\cdot(P/Q)^{-1}(\lambda^{\alpha})): \lambda \in \Omega\}$. Then it readily follows from Theorem 2.10.3 that the corresponding problems (DFP)_R and (DFP)_L, with A and B replaced respectively by $A_{|\tilde{E}}$ and $B_{|\tilde{E}}$, are topologically mixing in the Banach space \tilde{E} .

EXAMPLE 2.10.6. A great number of differential operators generating strongly continuous semigroups whose (subspace) chaoticity has been proved by applying the Desch–Schappacher–Webb or Banasiak–Moszyński criterion (cf. [238], [292, Chapter 3] and references cited there for corresponding examples) can be used for construction of degenerate topologically mixing fractional PDEs, in many different

ways. Generally speaking, let \mathcal{A} be a closed linear operator on E satisfying that there exist an open connected subset $\emptyset \neq \Lambda$ of \mathbb{C} and an analytic mapping $g: \Lambda \rightarrow E \setminus \{0\}$ such that $\mathcal{A}g(\nu) = \nu g(\nu), \nu \in \Lambda$. Suppose that P(z) and Q(z) are non-zero complex polynomials, and $0 < \alpha < 2$. Define $S := \{z \in \mathbb{C} : P(z) = 0\}, \Lambda' := \Lambda \setminus S,$ $A := Q(\mathcal{A}), B := P(\mathcal{A})$ (with the exception of Example 2.10.15 below, this will be a kind of prototype for construction of topologically mixing degenerate fractional differential equations in the sequel of this section) and suppose that

(222)
$$\frac{Q}{P}(\Lambda') \cap \{te^{\pm i\alpha\pi/2} : t \ge 0\} \neq \emptyset.$$

t

Then Theorem 2.10.3 implies that the problems $(DFP)_R$ and $(DFP)_L$ are both \tilde{E} topologically mixing, with $\tilde{E} := \overline{span\{g(\nu) : \nu \in \Lambda\}}$. Now we shall briefly describe
how one can employ this result in the study of topologically mixing properties of
the linearized Benney–Luke equation

(223)
$$(z_0 - \Delta) \mathbf{D}_t^{\alpha} u = \alpha_0 \Delta u - \beta_0 \Delta^2 u \quad (0 < \alpha < 2, \ \alpha_0 > 0, \ \beta_0 > 0, \ z_0 \in \mathbb{C}),$$

on symmetric spaces of non-compact type, Damek–Ricci, Riemannian symmetric or Heckman–Opdam root spaces (cf. [35, 259, 457, 474] and Example 2.10.11 below). In the integer case $\alpha = 1$, the equation (223) is also chaotic (cf. Subsection 2.3) and, as already mentioned, this equation is important in evolution modelling of some problems appearing in the theory of liquid filtration [199, 485, p. 6]. Here $\mathcal{A} = \Delta$, $P(z) = z_0 - z$ and $Q(z) = \alpha_0 z - \beta_0 z^2$ ($z \in \mathbb{C}$). Observing that the equality $Ag(\nu) = e^{i\alpha\pi/2}tBg(\nu)$ holds provided that $t \in \mathbb{R}$ and $\Lambda \ni \nu = (2\beta_0)^{-1}(\alpha_0 + e^{i\alpha\pi/2}t - \sqrt{(\alpha_0 + e^{i\alpha\pi/2}t)^2 - 4\beta_0 e^{i\alpha\pi/2}tz_0})$, as well as that

$$\lim_{\to +\infty} (2\beta_0)^{-1} \left(\alpha_0 + e^{i\alpha\pi/2} t - \sqrt{(\alpha_0 + e^{i\alpha\pi/2} t)^2 - 4\beta_0 e^{i\alpha\pi/2} t z_0} \right) = z_0,$$

it readily follows that the condition (222) holds provided that $z_0 \in \Lambda$. Notice that, in this case, $z_0 \in \sigma_p(\mathcal{A})$ so that the operator B is not injective.

We close this subsection by stating the following extension of [143, Theorem 2.3] and [85, Theorem 1.2].

THEOREM 2.10.7. Suppose that $F \in \{R, L\}$, $\alpha > 0$ and $(t_n)_{n \in \mathbb{N}}$ is a sequence of positive reals tending to $+\infty$. If the set $E_{0,F}$, which consists of those elements $y \in Z_F(A, B) \cap \tilde{E}$ for which there exists a strong solution $t \mapsto u(t; y), t \ge 0$ of problem (DFP)_F such that $\lim_{n\to\infty} u(t_n; y) = 0$, is dense in \tilde{E} , and if the set $E_{\infty,F}$, which consists of those elements $z \in Z_F(A, B) \cap \tilde{E}$ for which there exist a null sequence $(\omega_n)_{n\in\mathbb{N}} \in Z_F(A, B) \cap \tilde{E}$ and a sequence $(u_n(\cdot; \omega_n))_{n\in\mathbb{N}}$ of strong solutions of problem (DFP)_F such that $\lim_{n\to\infty} u(t_n; \omega_n) = z$, is also dense in \tilde{E} , then the problem (DFP)_F is \tilde{E} -topologically transitive.

PROOF. The proof of theorem is quite similar to the proofs of above-mentioned theorems. Let $y, z \in \tilde{E}$ and $\varepsilon > 0$ be given in advance. Since $E_{0,F}$ and $E_{\infty,F}$ are both dense in \tilde{E} , there exist $y_0 \in E_{0,F} \cap \tilde{E}$ and $z_0 \in E_{\infty,F} \cap \tilde{E}$ such that $\|y - y_0\| < \varepsilon/2$ and $\|z - z_0\| < \varepsilon/3$. By definition of sets $E_{0,F}$ and $E_{\infty,F}$, we have that there exists a strong solution $u(\cdot; y_0)$ of problem (DFP)_F such that
$$\begin{split} \lim_{n \to +\infty} u(t_n; y_0) &= 0 \text{ and there exist a null sequence } (\omega_n)_{n \in \mathbb{N}} \in Z_F(A, B) \cap \tilde{E} \\ \text{and a sequence } (u_n(\cdot; \omega_n))_{n \in \mathbb{N}} \text{ of strong solutions of problem } (\text{DFP})_F \text{ such that} \\ \lim_{n \to \infty} u(t_n; \omega_n) &= z_0. \text{ Let } n_0 \in \mathbb{N} \text{ satisfy that } \|\omega_{n_0}\| < \varepsilon/2, \|z_0 - u(t_{n_0}; \omega_{n_0})\| < \varepsilon/3 \text{ and } \|u(t_{n_0}; y_0)\| < \varepsilon/3. \text{ Put } x := y_0 + \omega_{n_0} \text{ and } t := t_{n_0}. \text{ Then} \end{split}$$

$$\|y-x\| \leq \|y-y_0\| + \|\omega_{n_0}\| \leq \frac{\varepsilon}{2} + \|\omega_{n_0}\| < \varepsilon,$$

and

$$||z - u(t;x)|| = ||z - u(t;y_0) - u(t;\omega_{n_0})||$$

$$\leq ||z - z_0|| + ||z_0 - u(t;\omega_{n_0})|| + ||u(t;y_0)|| < 3\frac{\varepsilon}{3} = \varepsilon.$$

This completes the proof.

2.10.2. Hypercyclic and topologically mixing properties of certain classes of degenerate abstract multi-term fractional differential equations. In this subsection, we shall assume that $n \in \mathbb{N} \setminus \{1\}$, $A_0 = A, A_1, \ldots, A_{n-1}, A_n = B$ are closed linear operators on $E, 0 \leq \alpha_1 < \cdots < \alpha_{n-1} < \alpha_n$ and $0 \leq \alpha < \alpha_n$. Fix a number $i \in \mathbb{N}_{mn-1}^0$. Let us remind ourselves that $\alpha_0 = \alpha$, $m_j = \lceil \alpha_j \rceil \ (j \in \mathbb{N}_0^n), D_i = \{j \in \mathbb{N}_{n-1} : m_j - 1 \geq i\}$ and $\mathcal{D}_i = \{j \in \mathbb{N}_{n-1}^0 : m_j - 1 \geq i\}$. Recall that $T_{j,L}u(t) = A_j \mathbf{D}_t^{\alpha_j}u(t), T_{j,R}u(t) = \mathbf{D}_t^{\alpha_j}A_ju(t)$ and $T_ju(t)$ denotes exactly one of terms $T_{j,L}u(t)$ or $T_{j,R}u(t)$ $(t \geq 0, j \in \mathbb{N}_{n-1}^0)$. In order to avoid the confusion with the existence of solutions of problem [(90)-(91)], already described in Section 2.2, we shall consider henceforth (the only exception is Remark 2.10.10(i) below) the hypercyclic and topologically mixing properties of the following degenerate abstract multi-term fractional problem:

(224)
$$B\mathbf{D}_{t}^{\alpha_{n}}u(t) + \sum_{j=1}^{n-1}T_{j}u(t) = T_{0}u(t), \quad t \ge 0;$$
$$u^{(i)}(0) = x, \quad u^{(k)}(0) = 0, \ k \in \mathbb{N}_{m_{n}-1}^{0} \smallsetminus \{i\}.$$

This is, clearly, a problem of kind [(90)-(91)] because we can always get the term $T_0u(t)$ on the left hand side of (224).

DEFINITION 2.10.8. Let \tilde{E} be a closed linear subspace of E. Then it is said that the equation (224) is:

- (i) \tilde{E} -hypercyclic iff there exist an element $x \in \tilde{E}$ and a strong solution $t \mapsto u_i(t;x), t \ge 0$ of (224) such that $\{u_i(t;x) : t \ge 0\}$ is a dense subset of \tilde{E} ; such an element is called a \tilde{E} -hypercyclic vector of (224);
- (ii) \tilde{E} -topologically transitive iff for every $y, z \in \tilde{E}$ and for every $\varepsilon > 0$, there exist an element $x \in \tilde{E}$, a strong solution $t \mapsto u_i(t; x), t \ge 0$ of (224) and a number $t \ge 0$ such that $||y x|| < \varepsilon$ and $||z u_i(t; x)|| < \varepsilon$;
- (iii) \tilde{E} -topologically mixing iff for every $y, z \in \tilde{E}$ and for every $\varepsilon > 0$, there exists $t_0 \ge 0$ such that, for every $t \ge t_0$, there exist an element $x_t \in \tilde{E}$ and a strong solution $t \mapsto u_i(t; x_t), t \ge 0$ of (224), with x replaced by x_t , such that $||y x_t|| < \varepsilon$ and $||z u_i(t; x_t)|| < \varepsilon$.

229

 \square

In the case $\tilde{E} = E$, it is also said that a \tilde{E} -hypercyclic vector of (224) is a hypercyclic vector of (224) and (eq224 is topologically transitive, resp. topologically mixing.

The following extension of [**302**, Theorem 2.3] is crucially important in the analysis of topologically mixing degenerate multi-term fractional problems.

THEOREM 2.10.9. Suppose that $\emptyset \neq \Omega$ is an open connected subset of $\mathbb{C} \setminus \{0\}$, $f: \Omega \to E \setminus \{0\}$ is an analytic function, and $f_j: \Omega \to \mathbb{C} \setminus \{0\}$ is a scalar-valued function $(1 \leq j \leq n)$ so that

(225)
$$Af(\lambda) = f_n(\lambda)Bf(\lambda) = f_j(\lambda)A_jf(\lambda), \quad \lambda \in \Omega, \ 1 \le j \le n.$$

Suppose, further, that Ω_+ and Ω_- are two non-empty subsets of Ω , and each of them admits a cluster point in Ω . Define $\tilde{E} := \overline{span\{f(\lambda) : \lambda \in \Omega\}}$,

$$(226) \quad H_i(\lambda,t) := \mathcal{L}^{-1} \left(\frac{z^{\alpha_n - i - 1} + \sum_{j \in D_i} \frac{f_n(\lambda)}{f_j(\lambda)} z^{\alpha_j - i - 1} - \chi_{\mathcal{D}_i}(0) f_n(\lambda) z^{\alpha - i - 1}}{z^{\alpha_n} + \sum_{j = 1}^{n - 1} \frac{f_n(\lambda)}{f_j(\lambda)} z^{\alpha_j} - f_n(\lambda) z^{\alpha}} \right)(t),$$

and

$$F_i(\lambda, t) := H_i(\lambda, t) f(\lambda),$$

for any $t \ge 0$ and $\lambda \in \Omega$. If

(227)
$$\lim_{t \to +\infty} |H_i(\lambda, t)| = +\infty, \quad \lambda \in \Omega_+ \quad and \quad \lim_{t \to +\infty} H_i(\lambda, t) = 0, \quad \lambda \in \Omega_-,$$

then (224) is \tilde{E} -topologically mixing; furthermore, if $Af(\lambda)$, $Bf(\lambda)$ and $A_jf(\lambda)$ belongs to \tilde{E} for all $\lambda \in \Omega$, then (224) is \tilde{E} -topologically mixing with the operators A_1, \ldots, A_{n-1} , A and B replaced respectively by $A_{1|\tilde{E}}, \ldots, A_{n-1|\tilde{E}}, A_{|\tilde{E}}$ and $B_{|\tilde{E}}$.

PROOF. First of all, let us observe that the (weak) analyticity of mapping $\lambda \mapsto f(\lambda) \in \tilde{E}, \lambda \in \Omega$ implies that the linear span of set $\{f(\lambda) : \lambda \in \Omega'\}$ is dense in \tilde{E} for each non-empty subset Ω' of Ω which admits a cluster point in Ω . Using the analytical properties of Laplace transform stated in Theorem 1.4.10, it can be easily proved that for each $\lambda \in \Omega$ and for each real number $\nu \leq \alpha_n - 1$, the function

$$t \mapsto \mathcal{L}^{-1} \bigg(\frac{z^{\nu}}{z^{\alpha_n} + \sum_{j=1}^{n-1} \frac{f_n(\lambda)}{f_j(\lambda)} z^{\alpha_j} - f_n(\lambda) z^{\alpha}} \bigg)(t), \quad t \ge 0,$$

is well defined, continuous for $t \ge 0$ and exponentially bounded (furthermore, this function can be analytically extended to the sector $\Sigma_{\pi/2}$). Hence, the function $t \mapsto H_i(\lambda, t), t \ge 0$ has the above-mentioned properties ($\lambda \in \Omega$). Keeping in mind the proof of [**300**, Theorem 2.3] and (225)–(227), the only non-trivial thing that should be explained is the fact that the function $u_i(t; f(\lambda)) := F_i(\lambda, t), t \ge 0$ is a strong solution of (224) with $x = f(\lambda)$ ($\lambda \in \Omega$). By performing the Laplace transform, it can be simply verified with the help of formula (39) that the above holds if we prove that the Caputo fractional derivative $\mathbf{D}_i^{\zeta} F_i(\lambda, t)$ is well defined for any $t \ge 0$, $\lambda \in \Omega$ and $\zeta \in [0, \alpha_n]$. For this, observe first that there exists a sufficiently large number $\omega > 0$ such that the function $z \mapsto z^{\zeta} \widetilde{F_i(\lambda; t)}(z) - \chi_{[i+1,\infty)}(\lceil \zeta \rceil) z^{\zeta-i-1} f(\lambda)$ is identically equal to

$$\frac{z^{\alpha_n+\zeta-i-1}+\sum_{j\in D_i}\frac{f_n(\lambda)}{f_j(\lambda)}z^{\alpha_j+\zeta-i-1}-\chi_{\mathcal{D}_i}(0)f_n(\lambda)z^{\alpha+\zeta-i-1}}{z^{\alpha_n}+\sum_{j=1}^{n-1}\frac{f_n(\lambda)}{f_j(\lambda)}z^{\alpha_j}-f_n(\lambda)z^{\alpha}}f(\lambda),$$

for $\operatorname{Re} z > \omega$, $\lceil \zeta \rceil < i - 1$, $\lambda \in \Omega$, and

$$\frac{-\sum_{j\in\mathbb{N}_{n-1}\smallsetminus D_i}\frac{f_n(\lambda)}{f_j(\lambda)}z^{\alpha_j+\zeta-i-1}-f_n(\lambda)z^{\alpha+\zeta-i-1}(\chi_{\mathcal{D}_i}(0)-1)}{z^{\alpha_n}+\sum_{j=1}^{n-1}\frac{f_n(\lambda)}{f_j(\lambda)}z^{\alpha_j}-f_n(\lambda)z^{\alpha_j}}f(\lambda)$$

for $\operatorname{Re} z > \omega$, $\lceil \zeta \rceil \ge i - 1$ and $\lambda \in \Omega$. By the foregoing, for every $\lambda \in \Omega$ and $\zeta \in [0, \alpha_n]$, there exists a continuous exponentially bounded *E*-valued mapping $t \mapsto G_i(\lambda; t; \zeta), t \ge 0$ such that

$$\widetilde{G_i(\lambda;t;\zeta)}(z) = z^{\zeta} \widetilde{F_i(\lambda;t)}(z) - \chi_{[i+1,\infty)}(\lceil \zeta \rceil) z^{\zeta-i-1} f(\lambda), \quad \operatorname{Re} z > \omega.$$

Applying the Laplace transform, it can be easily checked that, for every $\lambda \in \Omega$ and $\zeta \in [0, \alpha_n]$, one has:

$$(228) \qquad (g_{m_n} * G_i(\lambda; \cdot, \alpha_n))(t) = (g_{m_n - \alpha_n} * [F_i(\lambda, \cdot) - g_{i+1}(\cdot)f(\lambda)])(t), \quad t \ge 0,$$

(229)
$$G_i(\lambda; t; \zeta) = (g_{\alpha_n - \zeta} * G_i(\lambda; \cdot; \alpha_n))(t), \quad t \ge 0, \text{ if } \lceil \zeta \rceil \ge i - 1,$$

and

$$(230) \ (g_{\alpha_n-\zeta}*G_i(\lambda;\cdot;\alpha_n))(t)+g_{i+1-\lceil\zeta\rceil}(t)f(\lambda)=G_i(\lambda;t;\zeta), \ t\ge 0, \text{ if } \lceil\zeta\rceil< i-1.$$

Making use of (228), we get that $F_i(\lambda, \cdot) \in C^{m_n-1}([0,\infty) : E)$ for all $\lambda \in \Omega$. Consider first the case $\lceil \zeta \rceil < i - 1$. Convoluting (228) with $g_{1+\lceil \zeta \rceil - \zeta + \alpha_n - m_n}(t)$, and using after that the definition of Caputo fractional derivatives and (230), we get that $\mathbf{D}_t^{\zeta} F_i(\lambda, t) = G_i(\lambda; t; \zeta), t \geq 0, \lambda \in \Omega$. Suppose finally that $\lceil \zeta \rceil \geq i - 1$. Then the previous equality continues to hold, which can be seen by using the same arguments, with the equation (230) replaced by (229).

REMARK 2.10.10. (i) Consider the equation (224) with term $B\mathbf{D}_{t}^{\alpha_{n}}u(t)$ replaced by $\mathbf{D}_{t}^{\alpha_{n}}Bu(t)$, and endowed with the same initial conditions (therefore, the choice of initial conditions is not the same as in Section 2.2). Denote by (224)' the above described Cauchy problem. A function $u \in C^{m_{n}-1}([0,\infty): E)$ is said to be a strong solution of problem (224)' iff $u^{(i)}(0) = x, u^{(k)}(0) = 0, k \in \mathbb{N}_{m_{n}-1}^{0} \setminus \{i\}$, the terms $\mathbf{D}_{t}^{\alpha_{n}}Bu(t)$ and $T_{j}u(t)$ are well defined and continuous for any $t \ge 0, j \in \mathbb{N}_{n-1}^{0}$, and $\mathbf{D}_{t}^{\alpha_{n}}Bu(t) + \sum_{j=1}^{n-1}T_{j}u(t) = T_{0}u(t), t \ge 0$. As in Definition 2.10.8, it will be said that the problem (224)' is \tilde{E} -topologically mixing iff for every $y, z \in \tilde{E}$ and for every $\varepsilon > 0$, there exists $t_{0} \ge 0$ such that, for every $t \ge t_{0}$, there exist an element $x_{t} \in \tilde{E}$ and a strong solution $t \mapsto u_{i}(t; x_{t}),$ $t \ge 0$ of (224)', with x replaced by x_{t} , such that $||y - x_{t}|| < \varepsilon$ and $||z - u_{i}(t; x_{t})|| < \varepsilon$. Then the assertion of Theorem 2.10.9 continues to

231

hold for problem (224)' because $B\mathbf{D}_t^{\alpha_n}F_i(\lambda;t) = \mathbf{D}_t^{\alpha_n}BF_i(\lambda;t), t \ge 0, \lambda \in \Omega.$

(ii) In the present state of our knowledge, we do not know how the equalities appearing in (227) can be proved without a direct calculation of function $H_i(\lambda, t)$.

Although formulated for degenerate equations, it should be noted that Theorem 2.10.9 can be also used for construction of new important examples of nondegenerate multi-term fractional differential equations:

EXAMPLE 2.10.11. Let $0 < \gamma \leq 1$, a > 0, p > 2, and let X be a symmetric space of non-compact type and rank one. Our intention is to prove some results on topologically mixing properties of the following fractional analog of strongly damped Klein–Gordon equation (see e.g. [433, p. 276])

(231)
$$\mathbf{D}_t^{2\gamma}u(t) + 2a\Delta_{X,p}^{\natural}\mathbf{D}_t^{\gamma}u(t) + (\Delta_{X,p}^{\natural} - b)u(t) = 0, \quad t > 0.$$

Let P_p be the parabolic domain defined in [259]; then we know that $int(P_p) \subseteq \sigma_p(\Delta_{X,p}^{\natural})$. Consider first the equation (231) with $1/2 < \gamma \leq 1$ and i = 1. Let b be any positive real number such that $b - P_p$ intersects the imaginary axis. Then we have B = I, $f_2(\lambda) = \lambda$, $f_1(\lambda) = \lambda/(2a(b-\lambda))$, $\lambda \in \Omega := b - int(P_p)$ and

$$\widetilde{H_1(\lambda,t)}(z) = \frac{z^{2\gamma-2} - 2a(b-\lambda)z^{\gamma-2}}{z^{2\gamma} - 2a(b-\lambda)z^{\gamma} - \lambda}, \quad \lambda \in \Omega, \ t \ge 0.$$

Observing that $z^{2\gamma} - 2a(b-\lambda)z^{\gamma} - \lambda = (z^{\gamma} - r_1(\lambda))(z^{\gamma} - r_2(\lambda)), \ z \in \mathbb{C}, \ \lambda \in \Omega$, where

$$r_{1,2}(\lambda) = a(\lambda - b) \pm \sqrt{a^2(\lambda - b)^2 + \lambda}, \quad \lambda \in \Omega,$$

it can be simply verified with the help of formula (43) that, for every $t \ge 0$ and for every $\lambda \in \Omega$ such that $a^2(\lambda - b)^2 + \lambda \ne 0$,

$$H_1(\lambda, t) = \frac{t^{1-\gamma}}{2\sqrt{a^2(\lambda-b)^2 + \lambda}} \left[E_{\gamma,2-\gamma}(r_1(\lambda)t^{\gamma}) - E_{\gamma,2-\gamma}(r_2(\lambda)t^{\gamma}) \right] \\ - \frac{ta(\lambda-b)}{\sqrt{a^2(\lambda-b)^2 + \lambda}} \left[E_{\gamma,2}(r_1(\lambda)t^{\gamma}) - E_{\gamma,2}(r_2(\lambda)t^{\gamma}) \right].$$

This implies by Theorem 1.5.1 that there exist sufficiently small numbers $\varepsilon > 0$, $x_+ > 0$ and a sufficiently large negative number $x_- < 0$ such that (227) holds with $\Omega_+ := \{\lambda \in \Omega : |\lambda - x_+| < \varepsilon\}$ and $\Omega_- := \{\lambda \in \Omega : |\lambda - x_-| < \varepsilon\}$. By Theorem 2.10.9, it readily follows that the equation (231) admits a topologically mixing solution; using the consideration given in the next example, one can simply prove that the same holds if $0 < \gamma \leq 1$ and i = 0. Observe, finally, that for certain values of complex parameters $b, d \in \mathbb{C}$, the method described above proves the existence of topologically mixing solutions of the following fractional analog of vibrating beam type equation

$$\mathbf{D}_t^{2\gamma}u(t) + 2(aD+bI)\mathbf{D}_t^{\gamma}u(t) + (D+dI)u(t) = 0, \quad t > 0,$$

where D denotes the square of operator $\Delta_{X,p}^{\natural}$.

At the end of this subsection, we shall present an application of Theorem 2.10.9 in the study of hypercyclic and topologically mixing properties of degenerate multi-term fractional problems involving the bounded perturbations of one-dimensional Ornstein–Uhlenbeck operators [115,427]. Observe that we can similarly construct corresponding examples of degenerate equations with multi-dimensional Ornstein–Uhlenbeck operators or with n = 3 (cf. [302, Example 1, 3.-4.]).

EXAMPLE 2.10.12. Suppose $\alpha \in (0, 1]$, $E := L^2(\mathbb{R})$, $c_1 > c > \frac{b}{2} > 0$, P(z) is a non-zero complex polynomial, S is the set which consists of all roots of P(z), $\Omega := \{\lambda \in \mathbb{C} : \lambda \notin S, \lambda \neq c - c_1, \text{ Re } \lambda < c - \frac{b}{2}\}$ and $\mathcal{A}_c u := u'' + bxu' + cu$ is the bounded perturbation of the one-dimensional Ornstein–Uhlenbeck operator acting with domain $D(\mathcal{A}_c) := \{u \in L^2(\mathbb{R}) \cap W^{2,2}_{loc}(\mathbb{R}) : \mathcal{A}_c u \in L^2(\mathbb{R})\}$. Define $g_1 : \Omega \to E$ and $g_2 : \Omega \to E$ by

$$g_1(\lambda) := \mathcal{F}^{-1}\left(e^{-\frac{\xi^2}{2b}}\xi|\xi|^{-(2+\frac{\lambda-c}{b})}\right)(\cdot), \quad \lambda \in \Omega$$

and

$$g_2(\lambda) := \mathcal{F}^{-1}\left(e^{-\frac{\xi^2}{2b}}|\xi|^{-(1+\frac{\lambda-c}{b})}\right)(\cdot), \quad \lambda \in \Omega.$$

Then the mapping $g_i: \Omega \to E$ is analytic for i = 1, 2, and for every open connected subset Ω' of Ω which admits a cluster point in Ω , one has

 $E = span\{g_i(\lambda) : \lambda \in \Omega', i = 1, 2\}$; cf. [300, Example 2.5(iii)]. We analyze the topologically mixing properties of the equation

(232)
$$\left(\frac{\partial^2}{\partial x^2} + bx\frac{\partial}{\partial x} + c_1\right)\mathbf{D}_t^{2\alpha}u(t,x) + P\left(\frac{\partial^2}{\partial x^2} + bx\frac{\partial}{\partial x}\right)\mathbf{D}_t^{\alpha}u(t,x) = (c-c_1)u(t),$$

endowed with the initial conditions described in (224); here n = 2, $\alpha_j = j\alpha$ (j = 0, 1, 2), $A = (c - c_1)I$, $B = \mathcal{A}_c + A$, $A_1 = P(\mathcal{A}_c - cI)$, $f_1(\lambda) = (c - c_1)/P(\lambda - c)$ and $f_2(\lambda) = (c - c_1)/(\lambda - (c - c_1))$ $(\lambda \in \Omega)$. Put

$$r_{1,2}(\lambda) := \frac{1}{2} \left[-\frac{P(\lambda - c)}{\lambda - (c - c_1)} \pm \sqrt{\left(\frac{P(\lambda - c)}{\lambda - (c - c_1)}\right)^2 + 4\frac{c - c_1}{\lambda - (c - c_1)}} \right], \quad \lambda \in \Omega,$$

and denote by S' the set which consists of all roots of polynomial $z \mapsto (P(z-c))^2 + 4(c-c_1)(z-(c-c_1)), z \in \mathbb{C}$. Then

$$\widetilde{H_0(\lambda,t)}(z) = \frac{z^{2\alpha-1} + \frac{P(\lambda-c)}{\lambda-(c-c_1)} z^{\alpha-1}}{z^{2\alpha} + \frac{P(\lambda-c)}{\lambda-(c-c_1)} z^{\alpha} - \frac{c-c_1}{\lambda-(c-c_1)}}, \quad \lambda \in \Omega, \ t \ge 0,$$

and, in the case that $1/2 < \alpha \leq 1$,

$$\widetilde{H_1(\lambda,t)}(z) = \frac{z^{2\alpha-2}}{z^{2\alpha} + \frac{P(\lambda-c)}{\lambda - (c-c_1)} z^{\alpha} - \frac{c-c_1}{\lambda - (c-c_1)}}, \quad \lambda \in \Omega, \ t \ge 0,$$

which simply implies by the formula (43) that, for every $\lambda \in \Omega \setminus S'$ and $t \ge 0$,

$$H_0(\lambda,t) = \frac{\left(r_1(\lambda) + \frac{P(\lambda-c)}{\lambda-(c-c_1)}\right)e^{r_1(\lambda)t}}{\sqrt{\left(\frac{P(\lambda-c)}{\lambda-(c-c_1)}\right)^2 + 4\frac{c-c_1}{\lambda-(c-c_1)}}} - \frac{\left(r_2(\lambda) + \frac{P(\lambda-c)}{\lambda-(c-c_1)}\right)e^{r_2(\lambda)t}}{\sqrt{\left(\frac{P(\lambda-c)}{\lambda-(c-c_1)}\right)^2 + 4\frac{c-c_1}{\lambda-(c-c_1)}}},$$

233

234

if $\alpha = 1$,

$$H_0(\lambda, t) = \frac{t^{-\alpha}}{\sqrt{\left(\frac{P(\lambda-c)}{\lambda-(c-c_1)}\right)^2 + 4\frac{c-c_1}{\lambda-(c-c_1)}}} \left[E_{\alpha,1-\alpha}(r_1(\lambda)t^{\alpha}) - E_{\alpha,1-\alpha}(r_2(\lambda)t^{\alpha}) \right] + \frac{P(\lambda-c)}{\lambda-(c-c_1)} \left[E_{\alpha}(r_1(\lambda)t^{\alpha}) - E_{\alpha}(r_2(\lambda)t^{\alpha}) \right],$$

if $0 < \alpha < 1$, and

$$H_1(\lambda, t) = \frac{t^{1-\alpha}}{\sqrt{\left(\frac{P(\lambda-c)}{\lambda-(c-c_1)}\right)^2 + 4\frac{c-c_1}{\lambda-(c-c_1)}}} \left[E_{\alpha,2-\alpha}(r_1(\lambda)t^{\alpha}) - E_{\alpha,2-\alpha}(r_2(\lambda)t^{\alpha})\right],$$

if $1/2 < \alpha \leq 1$. Consider now the case in which $P(x) \in \mathbb{R}[x]$ and $P(\xi) > 0$ for some $\xi \in (-c_1, (-b)/2)$; albeit the last condition might seem strange, it is worth noting here that the second eqaulity in (227) does not hold provided that $\alpha = 1$ and $P(x) = -x - c_1, x \in \mathbb{R}$. Then there exist a sufficiently small number $\varepsilon > 0$ and a sufficiently large negative number $x_+ < 0$ such that, for every $\lambda' \in \Omega_+ \equiv$ $\{\lambda \in \Omega : |\lambda - x_+| < \varepsilon\}$, we have $r_1(\lambda') > 0$ and $r_2(\lambda') < 0$. On the other hand, it can be simply verified that there exists $x_- \in (c - c_1, c - (b/2))$ such that, for every $\lambda' \in \Omega_- \equiv \{\lambda \in \Omega : |\lambda - x_-| < \varepsilon\}$, we have $r_1(\lambda') < 0$ and $r_2(\lambda') < 0$. Keeping in mind Theorem 1.5.1, Theorem 2.10.9 and the fact that the conclusions stated in [**302**, Remark 1;3.] can be extended to degenerate (multi-term) fractional differential equations, we have that (227) holds with sets Ω_+ and Ω_- . An endevour should be made for finding some other classes of complex polynomials P(z) for which the problem (232) admits a topologically mixing solution.

2.10.3. Hypercyclic and topologically mixing properties of abstract degenerate first and second order equations. The main purpose of this subsection is to provide the basic information about hypercyclic and topologically mixing properties of abstract degenerate first and second order equations. We start by stating the following proposition, the proof is a consequence of a straightforward computation and therefore omitted (cf. [292, Proposition 3.2.19] for the case $\alpha = 2$).

- PROPOSITION 2.10.13. (i) Suppose that $\alpha = 1$, $x \in Z_{1,R}(A, B)$ ($x \in Z_{1,L}(A, B)$) and the function $t \mapsto u(t; x)$, $t \ge 0$ is a strong solution of $(DFP)_R$ ($(DFP)_L$). Then, for every $s \ge 0$, $u(s; x) \in Z_{1,R}(A, B)$ ($u(s; x) \in Z_{1,L}(A, B)$) and a strong solution of $(DFP)_R$ ($(DFP)_L$), with initial condition x replaced by u(s; x), is given by u(t; u(s; x)) := u(t+s; x), $t \ge 0$.
 - (ii) Suppose that $\alpha = 2, x \in Z_{2,R}(A, B)$ $(x \in Z_{2,L}(A, B))$ and the function $t \mapsto u(t;x), t \ge 0$ is a strong solution of $(DFP)_R$ $((DFP)_L)$. Then, for every $s \ge 0, u(s;x) \in Z_{2,R}(A, B)$ $(u(s;x) \in Z_{2,L}(A, B))$ and a strong solution of $(DFP)_R$ $((DFP)_L)$, with initial condition x replaced by u(s;x), is given by $u(t;u(s;x)) := \frac{1}{2}[u(t+s;x)+u(|t-s|;x)], t \ge 0$.

If $\alpha \in \{1, 2\}$, $F \in \{R, L\}$, $x \in Z_{\alpha, F}(A, B)$, the strong solutions of $(DFP)_F$ are unique, t > 0 and u(t; x) = x, then Proposition 2.10.13 implies that u(nt; x) = x,

235

 $n \in \mathbb{N}$, so that the notion of a periodic point of problem $(DFP)_F$ is meaningful for such class of degenerate first and second order equations. In contrast to this, if the strong solutions of problem $(DFP)_F$ are not unique, the supposition u(t;x) = xfor some t > 0 does not imply the validity of equality u(nt;x) = x, $n \in \mathbb{N}$, in general (cf. the final part of Example 2.10.15), so that Definition 2.10.14(i) below can be viewed only as an attempt to define satisfactorily the notion of (subspace) chaoticity of problem $(DFP)_F$. Compared with the classical Devaney definition of chaos, we assume here the (subspace) topological transitivity of problem $(DFP)_F$ in place of its hypercyclicity.

Recall that the symbol $E \oplus E$ $(\tilde{E} \oplus \tilde{E})$ denotes the direct sum of Banach spaces E and E $(\tilde{E} \text{ and } \tilde{E})$. Define $D(A \oplus A) := D(A) \times D(A)$ $(D(B \oplus B) := D(B) \times D(B))$ and $A \oplus A(x, y) := (Ax, Ay), x, y \in D(A)$ $(B \oplus B(x, y) := (Bx, By), x, y \in D(B))$. Then $A \oplus A$ and $B \oplus B$ are closed linear operators on $E \oplus E$.

DEFINITION 2.10.14. Let $\alpha \in \{1, 2\}$, and let \tilde{E} be a closed linear subspace of E. Then it is said that the problem $(DFP)_R$ $((DFP)_L)$ is:

- (i) \tilde{E} -chaotic, if $(DFP)_R$ $((DFP)_L)$ is \tilde{E} -topologically transitive and the set of \tilde{E} -periodic points of $(DFP)_R$ $((DFP)_L)$, i.e., the set of those elements $x \in Z_{\alpha,R}(A,B) \cap \tilde{E}$ $(x \in Z_{\alpha,L}(A,B) \cap \tilde{E})$ for which there exist a strong solution $t \mapsto u(t;x), t \ge 0$ of problem $(DFP)_R$ $((DFP)_L)$, and a positive real number t > 0 such that u(t;x) = x, is dense in \tilde{E} ,
- (ii) \tilde{E} -weakly mixing, if the problem $(DFP)^{\oplus}_{R}((DFP)^{\oplus}_{L})$, obtained by replacing the operators A and B in problem $(DFP)_{R}((DFP)_{L})$ with the operators $A \oplus A$ and $B \oplus B$, respectively, is $(\tilde{E} \oplus \tilde{E})$ -hypercyclic in $E \oplus E$.

Recall that we have already introduced the notions of (subspace) hypercyclicity, topological transitivity and topological mixing property for problem $(DFP)_R$ $((DFP)_L)$ in Definition 2.10.8. If $\alpha = 1$ and the assumptions of Theorem 2.10.3 hold, then it can be easily seen (see e.g. the proof of [143, Theorem 3.1]) that the set of those elements $x \in Z_{\alpha,R}(A,B) \cap \tilde{E}$ ($x \in Z_{\alpha,L}(A,B) \cap \tilde{E}$) for which there exist a strong solution $t \mapsto u(t;x), t \ge 0$ of problem $(DFP)_R$ ($(DFP)_L$), and a positive real number t > 0 such that u(nt;x) = x for all $n \in \mathbb{N}$, is dense in \tilde{E} (no matter whether the strong solutions of problem $(DFP)_R$ ($(DFP)_L$) are unique or not); in particular, the set of \tilde{E} -periodic points of problem $(DFP)_R$ ($(DFP)_L$) is dense in \tilde{E} , and $(DFP)_R$ ($(DFP)_L$) is \tilde{E} -chaotic.

In the following illustrative example, we shall present some new principal features of hypercyclic degenerate equations in comparison with non-degenerate ones (cf. also the paragraph directly after Definition 2.10.1).

EXAMPLE 2.10.15. Let $I := [0, \infty)$, $1 \leq p < \infty$, let $\rho : I \to (0, \infty)$ be an admissible weight function on I, and let $E := L^p_{\rho}(I, \mathbb{C})$ (cf. [143, Definition 4.1, Definition 4.3]). Then the set $\mathcal{D}_{(0,\infty)}$ is dense in E. Consider the problems $(\text{DFP})_R$ and $(\text{DFP})_L$ with $\alpha = 1$ and the closed linear operators Bu(x) := u'(x) + u(x), $x \geq 0$ and Au(x) := u''(x) - u(x), $x \geq 0$, acting with their maximal distributional domains. We will prove that the problems $(\text{DFP})_R$ and $(\text{DFP})_L$ are topologically mixing, independently of the choice of $\rho(\cdot)$. In order to do that, suppose that $y, z \in \mathcal{D}_{(0,\infty)}$ and $\varepsilon > 0$ are given in advance; let $y, z \in \mathcal{D}_{(0,l)}$ for some l > 0. Then there exists a C^{∞} strictly decaying function $H: (-\infty, l] \to (0, \infty)$ such that

(233)
$$H(l-x) \ge \rho(x+l), \quad x \ge 0.$$

Let $G: \mathbb{R} \to (0, \infty)$ by any C^{∞} -function such that

(234)
$$G(x) \leq 1, \quad x \geq l+1$$
 and $G(x) = e^{-2x}(1+H(x))^{1/p}, \quad x \leq 0.$

Then a simple computation shows that, for every $\varphi \in \mathcal{D}_{(0,\infty)}$, the function

(235)
$$u(t;G(x)\varphi(x)) := e^{-t}G(x-t)\varphi(x+t), \quad x \ge 0, \ t \ge 0,$$

is a strong solution of problems $(DFP)_R$ and $(DFP)_L$, with $u(0)(x) = G(x)\varphi(x)$, $x \ge 0$. Define, for every $t \ge 0$, the function $z_t \colon I \to \mathbb{C}$ by

$$z_t(x) := 0$$
 for $x < t$ and $z_t(x) := e^t \frac{u(x-t)}{G(x-2t)}$ for $x \ge t$.

Then $z_t \in \mathcal{D}_{(0,\infty)}$ and $u(t; G(x)z_t(x)) = z(x), x \ge 0$ $(t \ge 0)$. The existence of positive real number $t_0 \ge l+1$ such that the requirements of part (iii) of Definition 2.10.1 hold simply follows if we prove that $\lim_{t\to+\infty} G(x)z_t(x) = 0$ in E (then, for every $t \ge t_0$, we can put $x_t(\cdot) := y(\cdot) + G(\cdot)z_t(\cdot)$, observe only that u(t; G(x)(y(x)/G(x))) = 0 for $t \ge l$). This can be proved by using (233)– (234), [143, Lemma 4.2] and the following calculus (with a suitable constant $M_1 > 0$ depending only on l and $\rho(\cdot)$):

$$\begin{split} \|G(x)z_{t}(x)\|^{p} &= e^{tp} \int_{t}^{t+l} \left| \frac{z(x-t)G(x)}{G(x-2t)} \right|^{p} \rho(x) dx \\ &\leqslant M_{1}e^{tp} \rho(t+l) \int_{0}^{l} \left| \frac{z(x)G(x+t)}{G(x-t)} \right|^{p} dx \\ &\leqslant M_{1}e^{-tp} \rho(t+l) \int_{0}^{l} \frac{|z(x)|^{p}e^{2xp}}{1+H(x-t)} dx \\ &\leqslant M_{1}e^{-tp} \int_{0}^{l} |z(x)|^{p}e^{2xp} dx, \quad t \ge l+1. \end{split}$$

The above conclusions continue to hold, with almost the same proof, if $I = \mathbb{R}$, $\rho: I \to (0, \infty)$ is an admissible weight function on I, and $E := L^p_{\rho}(I, \mathbb{C})$. Strictly speaking, for any $\varepsilon > 0$ and $y, z \in \mathcal{D}_{(-l,l)}$ given in advance, we can simply construct a C^{∞} function $G: \mathbb{R} \to (0, \infty)$ such that $G(x) \sim e^{ax}$, $x \to -\infty$ for some $a > \max(0, (\omega - p)/2p)$, and

$$G(x+t) \leqslant G(t-l) \leqslant \rho(t+l)^{(-1)/p} e^{-at-t^2}, \quad x \ge -l, \ t \ge 0;$$

here we assume that $\omega \ge 0$ and there exists $M \ge 1$ such that $\rho(x) \le M e^{\omega t} \rho(x+t)$, $x, t \in \mathbb{R}$. Arguing as in the case $I = [0, \infty)$, it can be easily seen that there exists a finite constant $M_2 \ge 1$ such that

$$\int_{-\infty}^{\infty} \left| e^{-t} G(x-t) \frac{y}{G}(x+t) \right|^p \rho(x) dx$$

$$\leq M e^{(\omega-p)t} \left\| \frac{y}{G}(x) \right\|^p \sup_{x \in [-l-2t, l-2t]} G(x)^p \to 0 \text{ as } t \to +\infty$$

and

$$\begin{split} \left\| G(x)e^{t} \frac{u(x-t)}{G(x-2t)} \right\|^{p} &= \int_{t-l}^{t+l} \left| G(x)e^{t} \frac{u(x-t)}{G(x-2t)} \right|^{p} \rho(x) dx \\ &\leq M_{2}e^{tp} \rho(t+l) \int_{-l}^{l} |u(x)|^{p} \left| \frac{G(x+t)}{G(x-t)} \right|^{p} dx \\ &\leq M_{2}e^{tp-t^{2}} \int_{-l}^{l} |u(x)|^{p} e^{-ax} dx \to 0 \text{ as } t \to +\infty. \end{split}$$

Then we can prove in a routine manner that the corresponding problems $(DFP)_R$ and $(DFP)_L$ are topologically mixing. It is also worth noting that the problems $(DFP)_R$ and $(DFP)_L$ are chaotic in the sense of Definition 2.10.14(i), we shall only outline the main details needed for the proof of this fact in the case that $I = \mathbb{R}$. Clearly, it suffices to show that, for every l > 0, $\varepsilon > 0$, $\varphi \in \mathcal{D}_{(-l,l)}$ and P > 2lgiven in advance, we can find a function $v_P \in E$ such that $||v_P - \varphi|| < \varepsilon$ and $u(P; v_P(x)) = v_P(x), x \in \mathbb{R}$; cf. (235). Let $G \colon \mathbb{R} \to (0, \infty)$ be any C^{∞} function such that $G(x) = 1, x \in [-l - P, l]$. Using the calculation similar to that already employed in the proof of [**416**, Theorem 2], we have that there exists a finite constant $M_3 \ge 1$ such that the function

$$v_{P,G}(x) := \varphi(x) + \sum_{n \in \mathbb{Z} \setminus \{0\}} e^{nP} \frac{G(x)G(x-P)}{G(x-nP)G(x-(n+1)P)} \varphi(x-nP), \quad x \ge 0,$$

satisfies $u(P; v_{P,G}(x)) = v_{P,G}(x), x \in \mathbb{R}$ and

$$||v_{P,G} - \varphi||^p \leq M_3 \sum_{n \in \mathbb{Z} \setminus \{0\}} \rho(nP+l)e^{nP} \int_{-l}^{l} |G(x+nP)G(x+(n-1)P)|^p \varphi(x) dx.$$

The term appearing on the right hand side of the above inequality will not exceed ε if we choose $G(\cdot)$ sufficiently small on intervals [-l + nP, l + nP] $(n \in \mathbb{N})$ and [-l - nP, l - nP] $(n \in \mathbb{N} \setminus \{1\})$. The problems $(DFP)_R$ and $(DFP)_L$ are also weakly mixing; we leave the proof to the reader.

The assertion of [163, Theorem 2.1] in which S. El Mourchid has investigated the connection between the imaginary point spectrum and hypercyclicity of strongly continuous semigroups cannot be reformulated for degenerate fractional equations. The following theorem shows that the above-mentioned result can be extended to linear Sobolev equations of first order:

THEOREM 2.10.16. Assume that $\alpha = 1$, $\omega_1, \omega_2 \in \mathbb{R} \cup \{-\infty, \infty\}$, $\omega_1 < \omega_2$, $t_0 > 0$ and $k \in \mathbb{N}$. Let $f_j: (\omega_1, \omega_2) \to E$ be a Bochner integrable function with the property that, for every $j = 1, \ldots, k$, we have $Af_j(s) = isBf_j(s)$ for a.e. $s \in (\omega_1, \omega_2)$. Put $\psi_{r,j} := \int_{\omega_1}^{\omega_2} e^{irs} f_j(s) ds$, $r \in \mathbb{R}$, $1 \leq j \leq k$.

(i) Assume that span{f_j(s) : s ∈ (ω₁, ω₂) \ Ω, 1 ≤ j ≤ k} is dense in E for every subset Ω of (ω₁, ω₂) with zero measure. Let the following condition hold

237

- (PS): The functions $s \mapsto Bf_j(s)$, $s \in (\omega_1, \omega_2)$ and $s \mapsto sBf_j(s)$, $s \in (\omega_1, \omega_2)$ $(s \mapsto sf_j(s), s \in (\omega_1, \omega_2)$ and $s \mapsto sBf_j(s)$, $s \in (\omega_1, \omega_2)$) are Bochner integrable for $1 \leq j \leq k$.
- Then the problem $(DFP)_R$ ($(DFP)_L$) is topologically mixing.
- (ii) Put $\tilde{E} := \overline{span\{\psi_{r,j} : r \in \mathbb{R}, 1 \leq j \leq k\}}$. If (PS) holds, then the problem $(DFP)_R$ ((DFP)_L) is \tilde{E} -topologically mixing.

SKETCH OF PROOF. The proof of theorem is quite similar to those of [163, Theorem 2.1] and [291, Theorem 3.1.42(i)], and we shall only outline here that the validity of condition (PS) implies by direct calculations that $span\{\psi_{r,j} : r \in \mathbb{R}, 1 \leq j \leq k\}$ is contained in $Z_{1,R}(A,B)$ $(Z_{1,L}(A,B))$, as well as that $u(t;\psi_{r,j}) = \psi_{r+t,j}, t \geq 0$ provided $r \in \mathbb{R}$ and $1 \leq j \leq k$.

For more details about Theorem 2.10.16, we refer the reader to [163, Remark 2.2–Remark 2.4] and [291, Remark 3.1.43]; observe only that the uniqueness of strong solutions of $(DFP)_R$ ($(DFP)_L$) implies that the operator $T_R(t_0; \cdot)$ ($T_L(t_0; \cdot)$) is topologically mixing, provided that (i) holds, and that the part of operator $T_R(t_0; \cdot)$ ($T_L(t_0; \cdot)$) in \tilde{E} is topologically mixing in the Banach space \tilde{E} , provided that (ii) holds. We shall illustrate Theorem 2.10.16 with the following simple modification of Example 2.10.5.

EXAMPLE 2.10.17. Let $E := BUC(\mathbb{R})$ or $E := C_b(\mathbb{R})$, and let $\alpha = 1, \omega_1, \omega_2 \in \mathbb{R} \cup \{-\infty, \infty\}, \omega_1 < \omega_2$. Suppose that P(z) and Q(z) are non-zero complex polynomials such that, for every $s \in (\omega_1, \omega_2)$, all zeroes of polynomial $z \mapsto R_s(z) \equiv P(z) - isQ(z), z \in \mathbb{C}$ lie on the imaginary axis. Denote by $ig_1(s), \ldots, ig_k(s)$ the zeroes of $R_s(z)$. Then there exists a positive function $s \mapsto h(s), s \in \mathbb{R}$ such that the condition (PS) stated in the formulation of Theorem 2.10.16 holds with the operators A := P(d/dx) and B := Q(d/dx), acting with maximal distributional domains, and with functions $f_j(s)x = h(s)^{-1}e^{ig_j(s)x}$ ($s \in (\omega_1, \omega_2), 1 \leq j \leq k, x \in \mathbb{R}$). Define \tilde{E} as in the formulation of part (ii) of Theorem 2.10.16. Then the corresponding problems (DFP)_R and (DFP)_L are both \tilde{E} -topologically mixing (it should be observed that the uniqueness of strong solutions of problems (DFP)_R and (DFP)_L holds in a great number of particular choices of polynomials P(z) and Q(z) satisfying the above requirements, see e.g. [6] and [306, Theorem 4.2, Remark 4.4]).

In order to formulate The Hypercyclicity Criterion for degenerate abstract differential equations of first order, we need to introduce some additional notions. If $x \in Z_{1,R}(A, B)$ $(x \in Z_{1,L}(A, B))$, then we denote by $S_{R,x}$ $(S_{L,x})$ the set which consists of all strong solutions of problem $(DFP)_R$ $((DFP)_L)$. Then a linear mapping $\mathcal{T}_R: Z_{1,R}(A, B) \to C([0, \infty) : E)$ $(\mathcal{T}_L: Z_{1,R}(A, B) \to C([0, \infty) : E))$ is said to be regular if $\mathcal{T}_R(x) \in S_{R,x}$, $x \in Z_{1,R}(A, B)$ $(\mathcal{T}_L(x) \in S_{L,x}, x \in Z_{1,L}(A, B))$ and (cf. Proposition 2.10.13(i))

 $\mathcal{T}_F(\mathcal{T}_F(x)(t))(s) = \mathcal{T}_F(x)(t+s), \quad x \in Z_{1,F}(A,B), \ t, s \ge 0, \ F \in \{R, L\}.$

If $\mathcal{T}_R: Z_{1,R}(A, B) \to C([0, \infty): E)$ $(\mathcal{T}_L: Z_{1,R}(A, B) \to C([0, \infty): E))$ is a regular mapping and $t \ge 0$, then we define the linear mapping $T_{R,t,\mathcal{T}}: Z_{1,R}(A, B) \to$

 $Z_{1,R}(A,B)$ $(T_{L,t,\mathcal{T}}: Z_{1,L}(A,B) \to Z_{1,L}(A,B))$ by $T_{R,t,\mathcal{T}}(x) := (\mathcal{T}_R(x))(t), x \in Z_{1,R}(A,B)$ $(T_{L,t,\mathcal{T}}(x) := (\mathcal{T}_L(x))(t), x \in Z_{1,L}(A,B))$; notice that in the case of uniqueness of strong solutions of $(\text{DFP})_R$ $((\text{DFP})_L)$, one has $T_{F,t,\mathcal{T}}(x) = T_F(t;x), t \ge 0, x \in Z_{1,F}(A,B), F \in \{R,L\}$. A closed linear subspace \tilde{E} of E is said to be \mathcal{T}_R -admissible $(\mathcal{T}_L$ -admissible) if

$$T_{F,t,\mathcal{T}}(Z_{1,F}(A,B)\cap \tilde{E}) \subseteq Z_{1,F}(A,B)\cap \tilde{E}$$
 for $F=R$ $(F=L)$.

Observing that for any regular mapping $\mathcal{T}_F(\cdot)$ we have

$$T_{F,t,\mathcal{T}_F}(T_{F,s,\mathcal{T}_F}(x)) = T_{F,t+s,\mathcal{T}_F}(x), \quad x \in Z_{1,F}(A,B), \ t,s \ge 0, \ F \in \{R,L\},$$

one can repeat verbatim the proof of [291, Theorem 3.1.34] so as to verify that the following theorem holds true.

THEOREM 2.10.18 (The Hypercyclicity Criterion for Degenerate First Order Equations). Let $F \in \{R, L\}$, let $\mathcal{T}_F \colon Z_{1,F}(A, B) \to C([0, \infty) : E)$ be regular, and let \tilde{E} be a \mathcal{T}_F -admissible closed linear subspace of E. Assume that there exist subsets $\overline{Y_1}, \overline{Y_2} \subseteq Z_{1,F}(A, B) \cap \tilde{E}$, both dense in \tilde{E} , a mapping $\overline{S} \colon \overline{Y_1} \to \overline{Y_1}$ and a bounded linear operator D in \tilde{E} such that:

(i) $T_{F,s,\mathcal{T}_F}(\bar{S}y) = y, y \in \overline{Y_1},$

(ii)
$$\lim_{n\to\infty} \bar{S}^n y = 0, y \in \overline{Y_1}$$

- (iii) $\lim_{n\to\infty} T_{F,n,\mathcal{T}_F}(\omega) = 0, \ \omega \in \overline{Y_2},$
- (iv) R(D) is dense in \tilde{E} ,
- (v) $R(D) \subseteq Z_{1,F}(A,B) \cap \tilde{E}, T_{F,n,\mathcal{T}_F}D \in L(\tilde{E}), n \in \mathbb{N}$ and
- (vi) $DT_{F,1,\mathcal{T}_F}(x) = T_{F,1,\mathcal{T}_F}(Dx), x \in Z_{1,F}(A,B) \cap \tilde{E}.$

Then the abstract Cauchy problem $(DFP)_F$ is both $(\tilde{E} \oplus \tilde{E})$ -hypercyclic and $(\tilde{E} \oplus \tilde{E})$ topologically transitive; in particular, $(DFP)_F$ is \tilde{E} -weakly mixing.

Before proceeding further, it should be noticed that Theorem 2.10.18 clarifies only sufficient conditions for the $(\tilde{E} \oplus \tilde{E})$ -hypercyclicity and $(\tilde{E} \oplus \tilde{E})$ -topological transitivity of the abstract Cauchy problem $(DFP)_F$; because of that, we can hardly name this theorem as criterion. Observe, however, that there exist much more important results named as criteria, like the Desch–Schappacher–Webb or Banasiak– Moszyński criterion, which clarify only sufficient conditions, here concretly, for the (subspace) chaoticity of strongly continuous semigroups.

In some concrete situations, the assumptions of Theorem 2.10.18 can be relaxed; for example, if B = I, $\tilde{E} = E$ and if A generates a global C-regularized semigroup, with R(C) being dense in E, then the conditions (iv)–(vi) automatically hold (cf. [**291**, Subsection 3.1.4] and [**137**, Theorem 3.4]). Our impossibility to define a mild solution of problem (DFP)_L in a proper way, or to conclude that there exists a mild solution of problem (DFP)_R for any $x \in E$, has a series of obvious unpleasant consequences concerning extending [**292**, Theorems 3.1.13-3.1.14, 3.1.16; Proposition 3.1.17] to weakly mixing degenerate Cauchy problems of first order. In contrast to the above, [**85**, Corollary 1.3, Theorem 1.4] (cf. also [**292**, Theorem 3.2.26(ii)-(iii)]) can be rephrased for degenerate second order problems: THEOREM 2.10.19. Let \tilde{E} be a closed linear subspace of E, let $\alpha = 2$, and let $F \in \{R, L\}$.

- (i) Suppose that (t_n)_{n∈N} is a sequence of positive reals tending to +∞. Denote by X_{1,Ē} the set which consists of those elements x ∈ Z_F(A, B) ∩ Ē for which there exists a strong solution t → u(t; x), t ≥ 0 of problem (DFP)_F such that u(0; x) = x and lim_{n→∞} u(t_n; x) = lim_{n→∞} u(2t_n; x) = 0. If X_{1,Ē} is dense in Ē, then the problem (DFP)_F is Ē-topologically transitive.
- (ii) Denote by X'_{1,Ē} the set which consists of those elements x ∈ Z_F(A, B) ∩ Ẽ for which there exists a strong solution t → u(t; x), t ≥ 0 of problem (DFP)_F such that u(0; x) = x and lim_{t→+∞} u(t; x) = 0. If X'_{1,Ē} is dense in Ẽ, then the problem (DFP)_F is Ẽ-topologically mixing.

PROOF. Denote by $E_{0,F}$ the set which consists of those elements $y \in Z_F(A, B) \cap \tilde{E}$ for which there exists a strong solution $t \mapsto u(t; y), t \ge 0$ of problem $(DFP)_F$ such that $\lim_{n\to\infty} u(t_n; y) = 0$, and by $E_{\infty,F}$ the set which consists of those elements $z \in Z_F(A, B) \cap \tilde{E}$ for which there exist a null sequence $(\omega_n)_{n\in\mathbb{N}} \in Z_F(A, B) \cap \tilde{E}$ and a sequence $(u_n(\cdot; \omega_n))_{n\in\mathbb{N}}$ of strong solutions of problem $(DFP)_F$ such that $\lim_{n\to\infty} u(t_n; \omega_n) = z$. Then we have the obvious inclusion $X_{1,\tilde{E}} \subseteq E_{0,F}$, so that $E_{0,F}$ is dense in \tilde{E} . Now we will prove that $X_{1,\tilde{E}} \subseteq E_{\infty,F}$. In order to do that, suppose $z \in X_{1,\tilde{E}}$. Set $\omega_n := 2u(t_n; z), n \in \mathbb{N}$. Then $\lim_{n\to+\infty} \omega_n = 0$, and Proposition 2.10.13(ii) implies that $\omega_n \in Z_F(A, B), n \in \mathbb{N}$; the corresponding strong solution is given by $u_n(s; \omega_n) := u(s + t_n; z) + u(|t_n - s|; z), s \ge 0, n \in \mathbb{N}$. The prescribed assumptions imply that $u_n(t_n; \omega_n) = u(2t_n; z) + z, n \in \mathbb{N}$, so that $\lim_{n\to+\infty} u(t_n: \omega_n) = z$. Therefore, $E_{\infty,F}$ is also dense in \tilde{E} and the proof of (i) follows from an application of Theorem 2.10.7. The proof of (ii) is quite similar and therefore omitted.

In order to illustrate Theorem 2.10.19(ii), suppose that $I := \mathbb{R}$, $\rho: I \to (0, \infty)$ is an admissible weight function on I, and $E := L^p_{\rho}(I, \mathbb{C})$. Let the operators $(Bu)(x) := u''(x) + 2u'(x) + u(x), x \ge 0$ and $(Au)(x) := u^{(iv)}(x) - 2u''(x) + u(x),$ $x \ge 0$ act with their maximal distributional domains. Then, for every C^{∞} function $G: \mathbb{R} \to (0, \infty)$ and for every $\varphi \in \mathcal{D}$, the function

$$u(t;G(x)\varphi(x)) := \frac{1}{2}[e^{-t}G(x-t)\varphi(x+t) + e^{t}G(x+t)\varphi(x-t)], \quad x \in \mathbb{R}, \ t \ge 0,$$

is a strong solution of problem $(\text{DFP})_R$ $((\text{DFP})_L)$ with $\alpha = 2$, and $u(0; G(x)\varphi(x)) = G(x)\varphi(x), x \ge 0$. Arguing as in Example 2.10.15 we can prove that $\mathcal{D} \subseteq X'_{1,\tilde{E}}$. Hence, the problem $(\text{DFP})_R$ $((\text{DFP})_L)$ is topologically mixing by Theorem 2.10.19(ii).

For further information on hypercyclicity and topologically mixing property of non-degenerate differential equations of first and second order, we refer the reader to the work of T. Kalmes [264, 266]. Li–Yorke chaotic properties of abstract non-degenerate differential equations of first order has recently been analyzed in [328]. As pointed out there, we are not in a position to apply the method from our previous research of distributionally chaotic properties of linear operators [112] (cf. also

240

Section 2.11 and [68]) in the analysis of Li–Yorke chaotic properties of abstract nondegenerate fractional differential equations. For more more details about Li–Yorke chaotic and distributionally chaotic properties of abstract degenerate differential equations, we refer the reader to [294].

2.10.4. \mathcal{D} -Hypercyclic and \mathcal{D} -topologically mixing properties of degenerate fractional differential equations. In this subsection, we shall work in the setting of separable infinite-dimensional Fréchet spaces over the field \mathbb{C} . Let E be such a space; the use of symbol \mathcal{D} is clear from the context.

In [237], K.-G. Grosse-Erdmann and S. G. Kim have proposed the way of computing the orbit of a pair (x, y) under the action of a bilinear mapping $B: E \times E \to E$ E, with E being a separable Banach space. After that, the notion of bihypercyclicity of mapping B has been introduced. Several interesting examples of bihypercyclic bilinear mappings have been presented in [237], showing also that every separable Banach space supports a bihypercyclic bilinear mapping as well as that every separable Banach space E supports a bihypercyclic symmetric bilinear mapping whenever E supports a non-injective hypercyclic operator. In the setting of infinite-dimensional separable Fréchet spaces, a slightly different way of computing the orbit of a pair (x, y) under the action of bilinear mapping B has been proposed by J. Bès and J. A. Conejero in [70, Definition 1]. The notion of orbit of a tuple $(x_1, x_2, \ldots, x_n) \in E^N$ under the action of an N-linear operator $M: E^n \to E$ and the notion of supercyclicity of the operator M have been introduced in the same definition, while the notion of N-linear Devaney chaos of M has been introduced in [70, Definition 18] $(N \ge 2)$. In [70, Theorem 5, Theorem 8], it has been proved that every separable infinite-dimensional Fréchet space E supports, for any integer $N \ge 2$, an N-linear operator having a residual set of supercyclic vectors as well as that, for any integer $N \ge 2$, there exists an N-linear operator on the space $\omega = \mathbb{K}^{\mathbb{N}}$ (endowed with the product topology) that supports a dense N-linear orbit. The existence of hypercyclic N-linear operators on the Fréchet space $H(\mathbb{C})$ has been investigated in [70, Section 4].

Following the approaches used in [237] and [70], we define the orbits $\operatorname{Orb}(S; (B_i)_{1 \leq i \leq l})$ and $\operatorname{Orb}(S; (M_i)_{1 \leq i \leq l})$ for any non-empty subset S of E^N and any mappings $B_i : (E^N)^{b_i} \to E^N$, $M_i : E^N \to E$ $(1 \leq i \leq l)$; before going any further, it would be worthwhile to note that, in our analysis, these mappings need not be (separately) linear or continuous. Having this done, we have an open door (after a necessary patching up with some technicalities concerning the well-posedness of problem [(236)–(237)] below) to introduce the notions of \mathfrak{D} -hypercyclicity and \mathfrak{D} -topologically mixing property of degenerate abstract multi-term fractional problems (for more details, cf. Definition 2.10.20). In Theorem 2.10.22 and Theorem 2.10.23, we reformulate [237, Theorem 2] in our context, and prove the conjugacy lemma for abstract degenerate multi-term fractional differential equations. The main objective in Theorem 2.10.24 is to clarify the kind of Desch–Schappacher–Webb and Banasiak–Moszyński criteria ([48,122,143]) for \mathfrak{D} -topologically mixing of certain classes of abstract degenerate higher-order differential equations with integer order

derivatives. As explained in Remark 2.10.25(iii), Theorem 2.10.24 cannot be so easily transmitted to abstract degenerate differential equations with Caputo fractional derivatives.

Let $n \in \mathbb{N} \setminus \{1\}$, $0 = \alpha_0 < \alpha_1 < \cdots < \alpha_n$, $f \in C([0,\infty) : E)$, and let $A_0, A_1, \ldots, A_{n-1}, B$ be closed linear operators on E. Denote $m_i = \lceil \alpha_i \rceil$, $i \in \mathbb{N}_n^0$, $A_0 = A$, $A_n = B$, $T_{i,L}u(t) = A_i \mathbf{D}_t^{\alpha_i} u(t)$, if $t \ge 0$, $i \in \mathbb{N}_n^0$ and $\alpha_i > 0$, $T_{i,R}u(t) = \mathbf{D}_t^{\alpha_i} A_i u(t)$, if $t \ge 0$ and $i \in \mathbb{N}_n^0$. Henceforth we shall always assume that, for every $t \ge 0$ and $i \in \mathbb{N}_n^0$, $T_i u(t)$ denotes either $T_{i,L}u(t)$ or $T_{i,R}u(t)$. Denote, with a little abuse of notation,

$$P_{\lambda} = \lambda^{\alpha_n} B + \sum_{i=0}^{n-1} \lambda^{\alpha_i} A_i, \quad \lambda \in \mathbb{C} \smallsetminus \{0\},\$$

 $\mathcal{I} = \{i \in \mathbb{N}_n^0 : \alpha_i > 0 \text{ and } T_{i,L}u(t) \text{ appears on the left hand side of (236)}\}, Q = \max \mathcal{I}, \text{ if } \mathcal{I} \neq \emptyset \text{ and } Q = m_Q = 0, \text{ if } \mathcal{I} = \emptyset.$ In this subsection, we introduce and further analyze the notions of \mathcal{D} -hypercyclicity and \mathcal{D} -topologically mixing property of the following homogeneous degenerate abstract multi-term problem:

(236)
$$\sum_{i=0}^{n} T_{i}u(t) = 0, \quad t \ge 0,$$

equipped with the following initial conditions (cf. also Section 2.2; the problem is the same but the summation of $T_i u(t)$ is taken over all *i* going from 1 to n-1):

(237)
$$u^{(j)}(0) = u_j, \quad 0 \le j \le m_Q - 1 \text{ and } (A_i u)^{(j)}(0) = u_{i,j} \text{ if } m_i - 1 \ge j \ge m_Q.$$

If $T_n u(t) = T_{n,L} u(t)$, then (237) reads as follows:

$$u^{(j)}(0) = u_j, \quad 0 \leq j \leq m_n - 1.$$

Although the introduced notions seem to be new even for non-degenerate abstract differential equations of first order [445], we shall focus our attention almost completely on degenerate multi-term problems.

In Subsection 2.10.1, we have investigated the hypercyclicity and topologically mixing property of the equations $(DFP)_R$ and $(DFP)_L$ with $x_0 = x$ and $x_1 = \cdots = x_{m-1} = 0$, as well as the problem

(238)
$$B\mathbf{D}_t^{\alpha_n}u(t) + \sum_{i=0}^{n-1} T_iu(t) = 0, \quad t \ge 0; \ u^{(j)}(0) = u_j, \ j = 0, \dots, m_n - 1,$$

provided that there exists an index $i \in \mathbb{N}_{m_n-1}^0$ such that $u_j = 0, j \in \mathbb{N}_{m_n-1}^0 \setminus \{i\}$. Here we continue the study of hypercyclicity and topologically mixing property of problems (238) and $(\text{DFP})_R$ by allowing that there exist two or more nonzero components of the tuple (u_0, \ldots, u_{m_n-1}) (i.e., the tuple (Bu_0, \ldots, Bu_{m-1}) in the case of consideration of problem $(\text{DFP})_R$). The analysis of \mathcal{D} -hypercyclicity and \mathcal{D} -topologically mixing property of problem [(236)-(237)] is very intricate in general case and, with the exception of some minor facts and results concerning the existence and uniqueness of solutions, the most general abstract form of problem [(236)-(237)] will not be considered in this subsection any longer (cf. Subsection 2.11.2 for some results on distributionally chaotic properties of this problem). The notion of a strong solution of problem [(236)-(237)] will be understood in the sense of Definition 2.3.1.

For any $p \in \mathbb{N}$ and $r \in \mathbb{N}_p$, we define $\operatorname{Proj}_{r,p} : E^p \to E$ by $\operatorname{Proj}_{r,p}(x_1, \ldots, x_p) :=$ $x_r, \vec{x} = (x_1, \ldots, x_p) \in E^p$. Denote by \mathfrak{T} the exact number of initial values subjected to the equation [(236)-(237)]; in other words, \mathfrak{T} is the sum of number m_O and the cardinality of set consisting of those pairs $(i, j) \in \mathbb{N}_n \times \mathbb{N}_{m_n-1}^0$ for which $m_i - 1 \ge m_i$ $j \ge m_Q$. To make this thing more precise, suppose that $\{i_1, \ldots, i_s\} = \{i \in \mathbb{N}_n :$ $m_i - 1 \ge m_Q$ and $i_1 < \cdots < i_s$. Then the set of all initial values appearing in (237) is given by $\{u_0, \ldots, u_{m_Q-1}; u_{i_1, m_Q}, \ldots, u_{i_1, m_{i_1}-1}; \ldots; u_{i_s, m_Q}, \ldots, u_{i_s, m_{i_s}-1}\}$ s)m_Q. Denote by $\mathfrak{Z}(\mathfrak{Z}_{uniq})$ the set of all tuples of initial values $\vec{x} = ((u_i)_{0 \le i \le m_Q - 1};$ $(u_{i_{s'},j})_{1 \leq s' \leq s, m_Q \leq j \leq m_i, -1} \in E^{\mathfrak{T}}$ for which there exists a (unique) strong solution of problem [(236)-(237)]. Then \mathfrak{Z} is a linear subspace of $E^{\mathfrak{T}}$ and $\mathfrak{Z}_{unig} \subseteq \mathfrak{Z}$. The equality $\mathfrak{Z} = \mathfrak{Z}_{unig}$ holds iff the zero function is a unique strong solution of the problem [(236)–(237)] with the initial value $\vec{x} = \vec{0}$. For any $\vec{x} \in 3$, we denote by $\mathfrak{S}(\vec{x})$ the set consisting of all strong solutions of problem [(236)–(237)] with the initial value \vec{x} . In the remainder of this section, we shall only explore the problems (238) and $(DFP)_R$; observe that the problem $(DFP)_L$ is a very special case of problem (238) and that $\mathfrak{T} = m_n$ for problem (238), and $\mathfrak{T} = m$ for problem (DFP)_R. By (PQ) we denote either (238) or $(DFP)_R$. We shall always assume henceforth that $\emptyset \neq W \subseteq \mathbb{N}_{\mathfrak{T}}, \hat{E}_i$ is a linear subspace of E $(i \in W), \tilde{E}, \check{E}$ are linear subspaces of $E^{\mathfrak{T}}$, as well as that $\vec{\beta} := (\beta_0, \beta_1, \dots, \beta_{\mathfrak{T}-1}) \in [0, \alpha_n]^{\mathfrak{T}}, \ l \in \mathbb{N}, \ \emptyset \neq S \subseteq E^{\mathfrak{T}}, B_i : (E^{\mathfrak{T}})^{b_i} \to E^{\mathfrak{T}}$ and $M_i : E^{\mathfrak{T}} \to E$ are given mappings $(b_i \in \mathbb{N} \text{ for } 1 \leq i \leq l).$ Set $\mathfrak{B} := (\tilde{E}, \check{E}, S, (B_i)_{1 \leq i \leq l}, \{\hat{E}_i : i \in W\}, \vec{\beta})$ and $\mathfrak{M} := (\tilde{E}, \check{E}, S, (M_i)_{1 \leq i \leq l}, \{\hat{E}_i : i \in W\}, \vec{\beta})$ $i \in W$, $\vec{\beta}$). Let $\mathfrak{P}: \mathfrak{Z} \to P(\bigcup_{\vec{x} \in \mathfrak{Z}} \mathfrak{S}(\vec{x}))$ be a fixed mapping satisfying $\emptyset \neq \mathfrak{P}(\vec{x}) \subseteq$ $\mathfrak{S}(\vec{x}), \vec{x} \in \mathfrak{Z}$. Following K.-G. Grosse-Erdmann-S. G. Kim [237, pp. 701-702], we introduce the set $\mathbb{U}_p(S)$ $(p \in \mathbb{N}_0)$ recursively by $\mathbb{U}_0(S) := S$, $\mathbb{U}_{p+1}(S) := \mathbb{U}_p(S) \cup$ $\{B_i(\vec{x_1},\ldots,\vec{x_{b_i}}):1\leqslant i\leqslant l,\ \vec{x_1},\ldots,\vec{x_{b_i}}\in\mathbb{U}_p(S)\}$. If $\mathfrak{T}\geqslant 2$, then we introduce the set $\mathbf{U}_p(S)$ $(p \in \mathbb{N}_0)$ following the approach of J. Bès-J. A. Conejero [70, pp. 2-3]: $l, (x_1, x_2, \ldots, x_{\mathfrak{T}}) \in \mathbf{U}_p(S)$; if $\mathfrak{T} = 1$, then $\mathbf{U}_p(S) := S, p \in \mathbb{N}_0$. Define

$$\operatorname{Orb}(S; (B_i)_{1 \leq i \leq l}) := \bigcup_{p \in \mathbb{N}_0} \mathbb{U}_p(S), \quad \operatorname{Orb}(S; (M_i)_{1 \leq i \leq l}) := \bigcup_{p \in \mathbb{N}_0} \mathbf{U}_p(S)$$

and denote by $\mathcal{M}_{\mathfrak{B}}(\mathcal{M}_{\mathfrak{M}})$ the set consisting of those tuples $\vec{x} \in \operatorname{Orb}(S; (B_i)_{1 \leq i \leq l}) \cap \mathfrak{Z}$ $\mathfrak{Z}(\vec{x} \in \operatorname{Orb}(S; (M_i)_{1 \leq i \leq l}) \cap \mathfrak{Z})$ for which $\operatorname{Proj}_{i,\mathfrak{T}}(\vec{x}) \in \hat{E}_i, i \in W$. In the sequel, we shall denote by $D_i(\mathfrak{D})$ either B_i or $M_i(\mathfrak{B} \text{ or } \mathfrak{M})$ and, in the case that l = 1, we shall also write $\operatorname{Orb}(S; B_1)$, $\operatorname{Orb}(S; M_1)$ and $\operatorname{Orb}(S; D_1)$ in place of $\operatorname{Orb}(S; (B_i)_{1 \leq i \leq l})$, $\operatorname{Orb}(S; (M_i)_{1 \leq i \leq l})$ and $\operatorname{Orb}(S; (D_i)_{1 \leq i \leq l})$, respectively. A similar terminological agreement will be used in the case that the set W is a singleton.

Motivated by some results from the theory of abstract higher-order differential equations with integer order derivatives, obtained by the usual converting of higherorder equations into first order matrix differential equations by introducing the derivative, the second derivative, ..., the (n-1)th derivative of the unknown *E*-valued function as a part of a new enlarged unknown E^n -valued function (cf. [541, pp. 79-83], [199, Section 5.7], [210, Theorem 5.6.3] and Theorem 2.10.24 below for further information), we would like to propose the following definition (concerning the abstract multi-term differential equations with Caputo fractional derivatives, we do not yet recognize the ideal option for work).

DEFINITION 2.10.20. The abstract Cauchy problem (238) is said to be:

- (i) $(\mathfrak{D}, \mathfrak{P})$ -hypercyclic iff there exist a tuple $\vec{x} \in \mathcal{M}_{\mathfrak{D}} \cap \tilde{E}$ and a function $u(\cdot; \vec{x}) \in \mathfrak{P}(\vec{x})$ such that $\{((\mathbf{D}_{s}^{\beta_{0}}u(s; \vec{x}))_{s=t}, (\mathbf{D}_{s}^{\beta_{1}}u(s; \vec{x}))_{s=t}, \ldots, (\mathbf{D}_{s}^{\beta_{s-1}}u(s; \vec{x}))_{s=t}) : t \geq 0\}$ is a dense subset of \check{E} ; such a vector is called a $(\mathfrak{D}, \mathfrak{P})$ -hypercyclic vector of problem (238).
- (ii) D-hypercyclic iff it is (D, G)-hypercyclic; any (D, G)-hypercyclic vector of problem (238) will be also called a D-hypercyclic vector of problem (238).
- (iii) $\mathfrak{D}_{\mathfrak{P}}$ -topologically transitive iff for every open non-empty subsets U and Vof $E^{\mathfrak{T}}$ satisfying that $U \cap \tilde{E} \neq \emptyset$ and $V \cap \check{E} \neq \emptyset$, there exist a tuple $\vec{x} \in \mathcal{M}_{\mathfrak{D}}$, a function $u(\cdot; \vec{x}) \in \mathfrak{P}(\vec{x})$ and a number $t \ge 0$ such that $\vec{x} \in U \cap \tilde{E}$ and $((\mathbf{D}_{s}^{\beta_{0}}u(s; \vec{x}))_{s=t}, (\mathbf{D}_{s}^{\beta_{1}}u(s; \vec{x}))_{s=t}, \ldots, (\mathbf{D}_{s}^{\beta_{\mathfrak{T}-1}}u(s; \vec{x}))_{s=t}) \in V \cap \check{E}.$
- (iv) \mathfrak{D} -topologically transitive iff it is $\mathfrak{D}_{\mathfrak{S}}$ -topologically transitive.
- (v) $\mathfrak{D}_{\mathfrak{P}}$ -topologically mixing iff for every open non-empty subsets U and Vof $E^{\mathfrak{T}}$ satisfying that $U \cap \tilde{E} \neq \emptyset$ and $V \cap \check{E} \neq \emptyset$, there exists a number $t_0 \geq 0$ such that, for every number $t \geq t_0$, there exist a tuple $\vec{x_t} \in \mathcal{M}_{\mathfrak{D}}$ and a function $u(\cdot; \vec{x_t}) \in \mathfrak{P}(\vec{x_t})$ such that $\vec{x_t} \in U \cap \tilde{E}$ and $((\mathbf{D}_s^{\beta_0} u(s; \vec{x_t}))_{s=t}, (\mathbf{D}_s^{\beta_1} u(s; \vec{x_t}))_{s=t}, \ldots, (\mathbf{D}_s^{\beta_{\mathfrak{T}}-1} u(s; \vec{x_t}))_{s=t}) \in V \cap \check{E}.$
- (vi) \mathfrak{D} -topologically mixing iff it is $\mathfrak{D}_{\mathfrak{S}}$ -topologically mixing.

If $\mathfrak{Q}(\vec{x})$ is any non-empty subset consisting of solutions of problem $(DFP)_R$ with the initial value \vec{x} , then we denote by $\mathfrak{Q}_s(\vec{x})$ the set $\mathfrak{Q}(\vec{x}) \cap C^{m-1}([0,\infty) : E)$ $(\vec{x} \in \mathfrak{Z})$. We introduce the notions of $(\mathfrak{D}, \mathfrak{P}_s)$ -hypercyclicity, $\mathfrak{D}_{\mathfrak{P}_s}$ -topological transitivity and $\mathfrak{D}_{\mathfrak{P}_s}$ -topologically mixing property of problem $(DFP)_R$ in the same way as in Definition 2.10.20, with the sets $\mathfrak{P}(\vec{x})$ and $\mathfrak{P}(\vec{x}_t)$ replaced respectively by $\mathfrak{P}_s(\vec{x})$ and $\mathfrak{P}_s(\vec{x}_t)$. Finally, we say that the problem $(DFP)_R$ is \mathfrak{D} -hypercyclic $(\mathfrak{D}$ -topologically transitive, \mathfrak{D} -topologically mixing) iff it is $\mathfrak{D}_{\mathfrak{S}_s}$ -hypercyclic $(\mathfrak{D}_{\mathfrak{S}_s}$ topologically transitive, $\mathfrak{D}_{\mathfrak{S}_s}$ -topologically mixing).

REMARK 2.10.21. (i) We have presented only one way for computing the orbit $\operatorname{Orb}(S;(D_i)_{1 \leq i \leq l})$. In the case that $\mathfrak{T} \geq 2$, the orbit $\operatorname{Orb}(S;(D_i)_{1 \leq i \leq l})$ can have a very unpleasant form and it is very difficult to say, in general, whether there exists an element of $\operatorname{Orb}(S;(D_i)_{1 \leq i \leq l})$ that is a $(\mathfrak{D},\mathfrak{P})$ -hypercyclic vector of problem (238). On the other hand, in the definition of $\mathcal{M}_{\mathfrak{D}}$ we can take any non-empty subset S' of $E^{\mathfrak{T}}$ instead of $\operatorname{Orb}(S;(D_i)_{1 \leq i \leq l})$; but, this is only a very special case of our definition with $\mathfrak{D} = \mathfrak{B}, \ l = 1, \ b_1 = 1 \ \text{and} \ B_1 : E^{\mathfrak{T}} \to E^{\mathfrak{T}}$ being the identity mapping. It is also worth noting that the continuous version of Herrero-Bourdon theorem [238, Theorem 7.17, pp. 190-191] suggests us to define the set $\mathcal{M}_{\mathfrak{D}}$ as the union of those vectors $\vec{x} \in span\{\operatorname{Orb}(S;(D_i)_{1 \leq i \leq l})\} \cap \mathfrak{Z}$

244

for which $\operatorname{Proj}_{i,\mathfrak{T}}(\vec{x}) \in \hat{E}_i$, $i \in W$. If we define $\mathcal{M}_{\mathfrak{D}}$ in such a way, then the assertion of Theorem 2.10.23 below continues to hold, the assertion of Theorem 2.10.22 continues to hold with the mapping $\mathfrak{P}'(\cdot) = c\mathfrak{P}(\cdot/c)$ replaced by $\mathfrak{P}(\cdot)$, while the assertion of Theorem 2.10.24 continues to hold if we assume that $\{\vec{x}_{\lambda} : \lambda \in \Omega\} \subseteq \operatorname{Orb}(S; (D_i)_{1 \leq i \leq l})$. Observe also that the various notions of hypercyclicity and topologically mixing of abstract degenerate equations introduced in Definition 2.10.1 and Definition 2.10.8 are special cases of the notion introduced in Definition 2.10.20, and that [**301**. Theorem 2.4] can be restated in our context.

(ii) Let $0 \leq \beta \leq \alpha < 2$, and let the requirements of Theorem 2.10.3 hold (in (iii) and in the sequel of (ii) of this remark, we will use the almost same terminology as in Subsection 2.10.1; the only exception will be the notation used to denote the space \tilde{E}). Applying Theorem 1.5.1, we get that

$$\lim_{\lambda \to +\infty} \frac{t^{\alpha-\beta} E_{\alpha,\alpha-\beta+1}(\lambda^{\alpha} t^{\alpha})}{E_{\alpha}(\lambda^{\alpha} t^{\alpha})} = \lambda^{\beta-\alpha}, \quad \lambda \in \mathbb{C}_{+}$$

and

$$\lim_{t \to +\infty} t^{\alpha - \beta} E_{\alpha, \alpha - \beta + 1}(\lambda^{\alpha} t^{\alpha}) = 0, \quad \lambda \in \Omega_{0, -}$$

Using the identity

t

$$\mathbf{D}_{t}^{\beta}E_{\alpha}(\lambda^{\alpha}t^{\alpha}) = \lambda^{\alpha}t^{\alpha-\beta}E_{\alpha,\alpha-\beta+1}(\lambda^{\alpha}t^{\alpha}), \quad t \ge 0, \ \lambda \in \mathbb{C} \smallsetminus (-\infty, 0],$$

which can be shown directly, we may conclude by a careful inspection of the proof of Theorem 2.10.3 that the problem $(DFP)_L$ is $\mathfrak{D}_{\mathfrak{P}}$ -topologically mixing, provided that $\vec{\beta} = (\beta, \beta), W = \{1\}, \hat{E}_1 = \overline{span\{f(\lambda^{\alpha}) : \lambda \in \Omega\}},$ $\tilde{E} = \hat{E}_1 \times \{0\}, \ \check{E} = \{(z,z) : z \in \hat{E}_1\}, \ \hat{E}_1 \times \{0\} \subseteq \operatorname{Orb}(S; (D_i)_{1 \leqslant i \leqslant l})$ and $\mathfrak{P}((\sum_{i=1}^m \alpha_i f(\lambda_i^{\alpha}), 0)) = \{\sum_{i=1}^m \alpha_i E_\alpha(\cdot^{\alpha} \lambda_i^{\alpha}) f(\lambda_i^{\alpha})\} \ (m \in \mathbb{N}, \ \alpha_i \in \mathbb{C}, \mathbb{C}\}$ $\lambda_i \in \Omega$ for $1 \leq i \leq m$; this, in turn, implies that the problem $(DFP)_R$ is $\mathfrak{D}_{\mathfrak{P}_s}$ -topologicially mixing (the only thing worth noticing here is that, given y and z as in the proof of Theorem 2.10.3, the vector $\vec{x_t}$ can be chosen to be $(y + \sum_{j=1}^{m} \frac{\gamma_j}{\tilde{\lambda_j}^{\beta} E_{\alpha}(\tilde{\lambda_j}^{\alpha} t^{\alpha})} f(\tilde{\lambda_j}^{\alpha}), 0), t > 0$ sufficiently large). The above is a slight improvement of above-mentioned result, which can be applied in the analysis of fractional analogues of the linearized Boussinesq equation $(\sigma^2 \Delta - 1)u_{tt} + \gamma^2 \Delta u = 0$ on symmetric spaces of non-compact type, as well. Observe, finally, that Definition 2.10.20, Definition 2.10.1 and Definition 2.10.8 have some advantages over [300, Definition 2.2] and [**302**, Definition 2.2]. For example, an application of Theorem 2.10.3 shows that the abstract Cauchy problems $(DFP)_R$ and $(DFP)_L$, with $1 < \alpha < 2$, $E = L^2(\mathbb{R}), B = I$ and $A = \mathcal{A}_c$ being the bounded perturbation of the one-dimensional Ornstein–Uhlenbeck operator from [300, Example 2.5(iii)], are both topologically mixing in the sense of Definition 2.10.1; the topologically mixing property of corresponding problems in the sense of [300, Definition 2.2] can be proved only in the case that $0 < \alpha \leq 1$, cf. [115] and [300].

(iii) Consider the situation of Theorem 2.10.9 with the second equality in (227) replaced by

$$\lim_{t \to +\infty} \mathbf{D}_t^{\beta_j} H_i(\lambda, t) = 0, \quad \lambda \in \Omega_-, \ 0 \leq j \leq \mathfrak{T} - 1,$$

and with the first equality in (227) replaced by

$$\lim_{\lambda \to +\infty} |F(\lambda, t)| = +\infty, \quad \lambda \in \Omega_+, \ 0 \le j \le \mathfrak{T} - 1,$$

where a > 0 and $F : \Omega_+ \times (a, +\infty) \to \mathbb{C}$ is a certain function. Set $E_0 := \overline{span\{f(\lambda) : \lambda \in \Omega\}}$. If we suppose additionally that there exist complex numbers $G_{\beta_0}, \ldots, G_{\beta_{\mathfrak{T}-1}}$ such that

$$\lim_{t \to +\infty} \frac{\mathbf{D}_t^{\beta_j} H_i(\lambda, t)}{F(\lambda, t)} = G_{\beta_j}, \quad \lambda \in \Omega_+, \ 0 \leqslant j \leqslant \mathfrak{T} - 1.$$

then the proof of Theorem 2.10.9 shows that the abstract Cauchy problem (238) is $\mathfrak{D}_{\mathfrak{P}}$ -topologically mixing, provided that $W = \{i\}, \hat{E}_i = E_0, \tilde{E} = \{\vec{x} \in E^{\mathfrak{T}} : \operatorname{Proj}_{i,\mathfrak{T}}(\vec{x}) \in E_0 \text{ and } \operatorname{Proj}_{j,\mathfrak{T}}(\vec{x}) = 0 \text{ for } j \in \mathbb{N}_{\mathfrak{T}} \setminus \{i\}\}, \tilde{E} = \{(G_{\beta_0}z, \ldots, G_{\beta_{\mathfrak{T}-1}}z) : z \in \hat{E}_1\}, \tilde{E} \subseteq \operatorname{Orb}(S; (D_j)_{1 \leq j \leq l}) \text{ and } \mathfrak{P}((\sum_{j=1}^m \alpha_j f(0, \ldots, \lambda_j, \ldots, 0)) = \{\sum_{j=1}^m \alpha_j H_i(\lambda_j, \cdot)f(\lambda_j)\} \ (m \in \mathbb{N}, \alpha_j \in \mathbb{C}, \lambda_j \in \Omega \text{ for } 1 \leq j \leq m), \text{ where } \lambda_j \text{ appears on } i\text{-th place starting from zero (there exists a great number of concrete examples in which the above conditions hold with <math>\vec{\beta}$ being the constant multiple of $(1, 1, \ldots, 1)$, see e.g. our analysis of topologically mixing properties of strongly damped Klein–Gordon equation in Example 2.10.11; we refer the reader to Theorem 2.10.24 and Example 2.10.26 for the case in which $\vec{\beta}$ is not of the form described above). Also, it should be noticed that the comments from (ii) and (iii) can be formulated in the light of [**300**, Remark 1(iii)] and [**302**, Remark 1;3.], as well as that the proof of Theorem 2.10.9 implies that the term $\mathbf{D}_t^{\beta_i} H_i(\lambda, t)$ is equal to

$$\mathcal{L}^{-1}\left(\frac{z^{\alpha_n+\beta_l-i-1}+\sum_{j\in D_i}\frac{f_n(\lambda)}{f_j(\lambda)}z^{\alpha_j+\beta_l-i-1}-\chi_{\mathcal{D}_i}(0)f_n(\lambda)z^{\alpha+\beta_l-i-1}}{z^{\alpha_n}+\sum_{j=1}^{n-1}\frac{f_n(\lambda)}{f_j(\lambda)}z^{\alpha_j}-f_n(\lambda)z^{\alpha}}\right)(t),$$

for $t \ge 0, \lambda \in \Omega, \ l \in \mathbb{N}^0_{\mathfrak{T}-1}, \ \lceil \beta_l \rceil < i-1, \text{ and}$
$$\mathcal{L}^{-1}\left(\frac{-\sum_{j\in \mathbb{N}_{n-1}\smallsetminus D_i}\frac{f_n(\lambda)}{f_j(\lambda)}z^{\alpha_j+\beta_l-i-1}-f_n(\lambda)z^{\alpha+\beta_l-i-1}(\chi_{\mathcal{D}_i}(0)-1)}{z^{\alpha_n}+\sum_{j=1}^{n-1}\frac{f_n(\lambda)}{f_j(\lambda)}z^{\alpha_j}-f_n(\lambda)z^{\alpha}}\right),$$

for $t \ge 0, \lambda \in \Omega, l \in \mathbb{N}^0_{\mathfrak{T}-1}, \lceil \beta_l \rceil \ge i-1.$

(iv) As indicated in Subsection 2.10.1, it is much better to introduce the notions of \mathfrak{D} -hypercyclicity, \mathfrak{D} -topological transitivity and \mathfrak{D} -topologically mixing property of problem (PQ) with the set \mathfrak{Z} than with \mathfrak{Z}_{uniq} (the choice of strong solutions in Definition 2.10.20 is almost inevitable). Consider now, for the sake of brevity, the abstract Cauchy problem (238). If $\mathfrak{Z} = \mathfrak{Z}_{uniq}$, then we define the operator $T(t) : \mathcal{M}_{\mathfrak{D}} \to E^{\mathfrak{T}}$ by $T(t)\vec{x} :=$

246

 $((\mathbf{D}_{s}^{\beta_{0}}u(s;\vec{x}))_{s=t}, (\mathbf{D}_{s}^{\beta_{1}}u(s;\vec{x}))_{s=t}, \dots, (\mathbf{D}_{s}^{\beta_{T}-1}u(s;\vec{x}))_{s=t}) \ (t \ge 0)$, where $u(\cdot;\vec{x})$ denotes the unique strong solution of problem (238) with the initial value \vec{x} . Let $(O_{n})_{n\in\mathbb{N}}$ be an open base of the topology of $E^{\mathfrak{T}}$ $(O_{n} \ne \emptyset, n \in \mathbb{N})$. If $\tilde{E} = \check{E} = E^{\mathfrak{T}}$ and if we denote by $HC_{\mathfrak{D}}$ the set which consists of all \mathfrak{D} -hypercyclic vectors of problem (238), then we have the obvious equality $HC_{\mathfrak{D}} = \bigcap_{n\in\mathbb{N}} \bigcup_{t\ge 0} T(t)^{-1}(O_{n})$; cf. also [237, Theorem 1], [70, Proposition 4] and Subsection 2.10.1. Unfortunately, we cannot conclude from the above that \mathfrak{D} -topological transitivity of problem (238) implies its \mathfrak{D} -hypercyclicity (in this place, it is worth noting that there exists a continuous linear operator on the space $\varphi = \bigoplus_{n\in\mathbb{N}} \mathbb{K}$ that is topologically transitive but not hypercyclic [84, Theorem 2.2], so that the connections between \mathfrak{D} -hypercyclicity and \mathfrak{D} -topological transitivity seem to be more intricate in non-metrizable locally convex spaces).

It is worth noticing that the assertion of [237, Theorem 2] admits an adequate reformulation in our context. Before we state the corresponding theorem, it would be very helpful to introduce the sets $[(B_i)_{1 \leq i \leq l}]_p(S)$ and $[(M_i)_{1 \leq i \leq l}]_p(S)$ ($p \in \mathbb{N}_0$) recursively by $[(B_i)_{1 \leq i \leq l}]_0(S) := [(M_i)_{1 \leq i \leq l}]_0(S) := S$, $[(B_i)_{1 \leq i \leq l}]_{p+1}(S) :=$ $\{B_i(\vec{x_1}, \ldots, \vec{x_{b_i}}) : 1 \leq i \leq l, \vec{x_1} \in [(B_i)_{1 \leq i \leq l}]_{j_1}(S), \ldots, \vec{x_{b_i}} \in [(B_i)_{1 \leq i \leq l}]_{j_{b_i}}(S)$ for some numbers $j_1, \ldots, j_{b_i} \in \mathbb{N}_p$ with $j_1 + \cdots + j_{b_i} = p\}$, $[(M_i)_{1 \leq i \leq l}]_{p+1}(S) :=$ $\mathbf{U}_{p+1}(S) \setminus \mathbf{U}_p(S), p \in \mathbb{N}_0$. Then the set $[(B_i)_{1 \leq i \leq l}]_p(S)$ ($[(M_i)_{1 \leq i \leq l}]_p(S)$) contains all the elements from $\operatorname{Orb}(S; (B_i)_{1 \leq i \leq l})$ ($\operatorname{Orb}(S; (M_i)_{1 \leq i \leq l})$) obtained by napplications of operators B_1, \ldots, B_l (M_1, \ldots, M_l), totally counted, and the following holds:

$$\operatorname{Orb}(S; (B_i)_{1 \leq i \leq l}) = \bigcup_{p \in \mathbb{N}_0} [(B_i)_{1 \leq i \leq l}]_p(S),$$

(239)
$$\operatorname{Orb}(S; (M_i)_{1 \leq i \leq l}) = \bigcup_{p \in \mathbb{N}_0} [(M_i)_{1 \leq i \leq l}]_p(S).$$

Suppose now that $c \in \mathbb{K} \setminus \{0\}$ as well as $B'_i : (E^{\mathfrak{T}})^{b_i} \to E^{\mathfrak{T}}$ and $M'_i : E^{\mathfrak{T}} \to E$ satisfy

$$B'_i(c\vec{x_1},\ldots,c\vec{x_{b_i}}) = cB_i(\vec{x_1},\ldots,\vec{x_{b_i}}),$$

provided $\vec{x_1}, \ldots, \vec{x_{b_i}} \in E^{\mathfrak{T}}, \ 1 \leq i \leq l$, and

$$(cx_2,\ldots,cx_{\mathfrak{T}},M_i'(cx_1,cx_2,\ldots,cx_{\mathfrak{T}}))=c(x_2,\ldots,x_{\mathfrak{T}},M_i(x_1,x_2,\ldots,x_{\mathfrak{T}})),$$

provided that $(x_1, x_2, \ldots, x_{\mathfrak{T}}) \in E^{\mathfrak{T}}$, $1 \leq i \leq l$. Define $S_c := \{c\vec{x} : \vec{x} \in S\}$. Then we can inductively prove that $[(B'_i)_{1 \leq i \leq l}]_p(S_c) = \{c\vec{x} : \vec{x} \in [(B_i)_{1 \leq i \leq l}]_p(S)\}$ and $[(M'_i)_{1 \leq i \leq l}]_p(S_c) = \{c\vec{x} : \vec{x} \in [(M_i)_{1 \leq i \leq l}]_p(S)\}$ for all $p \in \mathbb{N}_0$, so that (239) implies

$$\{c\vec{x}: \vec{x} \in \operatorname{Orb}(S; (B_i)_{1 \leqslant i \leqslant l})\} = \operatorname{Orb}(S_c; (B'_i)_{1 \leqslant i \leqslant l})$$

and

$$\{c\vec{x}: \vec{x} \in \operatorname{Orb}(S; (M_i)_{1 \leqslant i \leqslant l})\} = \operatorname{Orb}(S_c; (M'_i)_{1 \leqslant i \leqslant l})$$

Now it is very simple to prove the following

THEOREM 2.10.22. Set $\mathfrak{D}' := (\tilde{E}, \check{E}, S_c, (D'_i)_{1 \leq i \leq l}, \{\hat{E}_i : i \in W\}, \vec{\beta})$ and $\mathfrak{P}' : \mathfrak{Z} \to P(\cup_{\vec{x} \in \mathfrak{Z}} \mathfrak{S}(\vec{x})), by \mathfrak{P}'(\vec{x}) := c \mathfrak{P}(\vec{x}/c), \vec{x} \in \mathfrak{Z}.$ Then the abstract Cauchy problem (238), resp. (DFP)_R, is \mathfrak{D} -hypercyclic $((\mathfrak{D}, \mathfrak{P})$ -hypercyclic, $\mathfrak{D}_{\mathfrak{P}}$ -topologically transitive, resp. $\mathfrak{D}_{\mathfrak{P}_s}$ -topologically transitive, \mathfrak{D} -topologically mixing, \mathfrak{D} -topologically mixing) iff the abstract Cauchy problem (238), resp. $(\mathfrak{DFP})_R$, is \mathfrak{D}' -hypercyclic $((\mathfrak{D}', \mathfrak{P}')$ -hypercyclic, $\mathfrak{D}'_{\mathfrak{P}'}$ -topologically mixing, \mathfrak{D} -topologically mixing) iff the abstract Cauchy problem (238), resp. $(\mathfrak{DFP})_R$, is \mathfrak{D}' -hypercyclic $((\mathfrak{D}', \mathfrak{P}')$ -hypercyclic, $\mathfrak{D}'_{\mathfrak{P}'}$ -topologically transitive, resp. $\mathfrak{D}_{\mathfrak{P}'_s}$ -topologically transitive, \mathfrak{D}' -topologically transitive, $\mathfrak{D}'_{\mathfrak{P}'}$ -topologically mixing, resp. $\mathfrak{D}_{\mathfrak{P}'_s}$ -topologically mixing, \mathfrak{D}' -topologically mixing, \mathfrak{D}' -topologically mixing, \mathfrak{D}' -topologically mixing).

Suppose now that X is another Fréchet space over the field of \mathbb{C} and $\phi: X \to E$ is a linear topological homeomorphism. Then the mapping $\phi^{\mathfrak{T}}: X^{\mathfrak{T}} \to E^{\mathfrak{T}}$, defined in the very obvious way, is a linear topological homeomorphism between the spaces $X^{\mathfrak{T}}$ and $E^{\mathfrak{T}}$. Define $S_{\phi} := (\phi^{\mathfrak{T}})^{-1}(S)$ and the closed linear operators A_i^X on X by $D(A_i^X) := \phi^{-1}(D(A_i))$ and $A_i^X x = y$ iff $A_i(\phi x) = \phi y$ $(0 \le i \le n)$. For any E-valued function $t \mapsto u(t), t \ge 0$ we define the X-valued function $t \mapsto u_{\phi}(t)$. $t \ge 0$ by $u_{\phi}(t) := \phi^{-1}(u(t)), t \ge 0$. Then it is readily seen that the Caputo fractional derivative $\mathbf{D}_{t}^{\alpha}u(t)$ is defined for $t \ge 0$ iff the Caputo fractional derivative $\mathbf{D}_t^{\alpha} u_{\phi}(t)$ is defined for $t \ge 0$; if this is the case, we have $\mathbf{D}_t^{\alpha} u_{\phi}(t) = \phi^{-1}(\mathbf{D}_t^{\alpha} u(t))$, $t \ge 0$. Using this fact, we can simply prove that the function $t \mapsto u(t), t \ge 0$ is a strong solution of problem (PQ) with the initial value $\vec{x} = (x_1, \ldots, x_{\mathfrak{T}}) \in E^{\mathfrak{T}}$ iff the function $t \mapsto u_{\phi}(t), t \ge 0$ is a strong solution of problem $(PQ)_{\phi}$ with the initial value $\vec{x}^{\phi} := (\phi^{-1}(x_1), \dots, \phi^{-1}(x_{\mathfrak{T}})) \in X^{\mathfrak{T}}$, where the abstract Cauchy problem $(PQ)_{\phi}$ is defined by replacing all the operators A_i in the problem (PQ) with the operators A_i^X $(0 \leq i \leq n)$. If we denote by \mathfrak{Z}^{ϕ} $(\mathfrak{Z}_{unig}^{\phi})$ the set consisting of those tuples $\vec{x}^{\phi} \in X^{\mathfrak{T}}$ for which there exists a (unique) strong solution of the problem (PQ)_{ϕ}, then the above implies $\mathfrak{Z}^{\phi} = (\phi^{\mathfrak{T}})^{-1}\mathfrak{Z}$ $(\mathfrak{Z}^{\phi}_{uniq} = (\phi^{\mathfrak{T}})^{-1}\mathfrak{Z}_{uniq}).$

Define the mappings $B_{i,\phi}: (X^{\mathfrak{T}})^{b_i} \to X^{\mathfrak{T}}$ and $M_{i,\phi}: X^{\mathfrak{T}} \to X$ by

$$B_{i,\phi}(\vec{x_1}, \dots, \vec{x_{b_i}}) := (\phi^{\mathfrak{T}})^{-1} (B_i(\phi^{\mathfrak{T}} \vec{x_1}, \dots, \phi^{\mathfrak{T}} \vec{x_{b_i}})) \text{ and } M_{i,\phi}(\vec{x}) := \phi^{-1} (M_i(\phi^{\mathfrak{T}} \vec{x})),$$

for any $\vec{x_1}, \ldots, \vec{x_{b_i}}, \vec{x} \in X^{\mathfrak{T}}, 1 \leq i \leq l$, as well as the mappings $\mathfrak{P}_{\phi} : \mathfrak{Z}^{\phi} \to P(\{\mathfrak{S}(\vec{x}^{\phi}) : \vec{x}^{\phi} \in \mathfrak{Z}^{\phi}\})$ and $(\mathfrak{P}_{\phi})_s : \mathfrak{Z}^{\phi} \to P(\{\mathfrak{S}(\vec{x}^{\phi}) : \vec{x}^{\phi} \in \mathfrak{Z}^{\phi}\})$ by $\mathfrak{P}_{\phi}((\phi^{\mathfrak{T}})^{-1}\vec{x}) := \{u_{\phi}(\cdot) : u(\cdot) \in \mathfrak{P}_s(\vec{x})\}$ and $(\mathfrak{P}_{\phi})_s((\phi^{\mathfrak{T}})^{-1}\vec{x}) := \{u_{\phi}(\cdot) : u(\cdot) \in \mathfrak{P}_s(\vec{x})\}$ ($\vec{x} \in \mathfrak{Z}$), respectively. Set

$$\mathfrak{D}_{\phi} := ((\phi^{\mathfrak{T}})^{-1}(\tilde{E}), (\phi^{\mathfrak{T}})^{-1}(\check{E}), S_{\phi}, (D_{i,\phi})_{1 \leq i \leq l}, \{\phi^{-1}(\hat{E}_i) : i \in W\}, \vec{\beta}).$$

Having in mind the proof of [237, Theorem 3], we can show that

$$\phi^{\mathfrak{T}}(\operatorname{Orb}(S_{\phi}; (D_{i,\phi})_{1 \leq i \leq l})) = \operatorname{Orb}(S; (D_i)_{1 \leq i \leq l}).$$

Now it is quite simple to prove the following conjugacy lemma for abstract degenerate multi-term fractional differential equations (cf. [263, Lemma 1.4] for a pioneering result in this direction):

THEOREM 2.10.23. The abstract Cauchy problem (238), resp. $(DFP)_R$, is \mathfrak{D} -hypercyclic $((\mathfrak{D}, \mathfrak{P})$ -hypercyclic, $\mathfrak{D}_{\mathfrak{P}}$ -topologically transitive, resp. $\mathfrak{D}_{\mathfrak{P}_s}$ -topologically transitive, $\mathfrak{D}_{\mathfrak{P}}$ -topologically mixing, resp. $\mathfrak{D}_{\mathfrak{P}_s}$ -topologi-

mathcally mixing, \mathfrak{D} -topologically mixing) iff the abstract Cauchy problem $(238)_{\phi}$, resp. $(DFP)_{R,\phi}$, is \mathfrak{D}_{ϕ} -hypercyclic $((\mathfrak{D}_{\phi},\mathfrak{P}_{\phi})$ -hypercyclic, $\mathfrak{D}_{\phi\mathfrak{P}_{\phi}}$ -topologically transitive, resp. $\mathfrak{D}_{\phi(\mathfrak{P}_{\phi})_s}$ -topologically transitive, \mathfrak{D}_{ϕ} -topologically transitive, $\mathfrak{D}_{\phi\mathfrak{P}_{\phi}}$ topologically mixing, resp. $\mathfrak{D}_{\phi(\mathfrak{P}_{\phi})_s}$ -topologically mixing, \mathfrak{D}_{ϕ} -topologically mixing).

We continue by stating the following theorem.

THEOREM 2.10.24. Let $\alpha_i = i$ for all $i \in \mathbb{N}_n$, let Ω be an open non-empty subset of \mathbb{C} intersecting the imaginary axis, and let $f: \Omega \to E$ be an analytic mapping satisfying that

(240)
$$P_{\lambda}f(\lambda) = \left(\lambda^{\alpha_n}B + \sum_{i=0}^{n-1}\lambda^{\alpha_i}A_i\right)f(\lambda) = 0, \quad \lambda \in \Omega.$$

Set $\vec{x_{\lambda}} := [f(\lambda) \ \lambda f(\lambda) \ \dots \ \lambda^{n-1} f(\lambda)]^T \ (\lambda \in \Omega), \ E_0 := span\{\vec{x_{\lambda}} : \lambda \in \Omega\}, \ \tilde{E} := \check{E}_0, \ \vec{\beta} := (0, 1, \dots, n-1), \ W := \mathbb{N}_n \ and \ \hat{E}_i := span\{f(\lambda) : \lambda \in \Omega\}, \ i \in W.$ Let $\emptyset \neq S \subseteq E^n$ be such that $E_0 \subseteq \operatorname{Orb}(S; (D_i)_{1 \leq i \leq l}).$ Then $\vec{x_{\lambda}} \in \mathfrak{M}_{\mathfrak{D}}, \ \lambda \in \Omega$ and the abstract Cauchy problem (238) is $\mathfrak{D}_{\mathfrak{P}}$ -topologically mixing provided that $\sum_{j=1}^q e^{\lambda_j} f(\lambda_j) \in \mathfrak{P}(\sum_{j=1}^q x_{\lambda_j}) \ for \ any \sum_{j=1}^q x_{\lambda_j} \in E_0 \ (q \in \mathbb{N}; \lambda_j \in \Omega, \ 1 \leq j \leq q).$

PROOF. We shall content ourselves with sketching it. Consider the operator matrices

$$A := \begin{bmatrix} 0 & I & 0 & \dots & 0 \\ 0 & 0 & I & \dots & 0 \\ \vdots & \vdots & \ddots & \ddots & \vdots \\ 0 & 0 & 0 & \dots & I \\ -A_0 & -A_1 & -A_2 & \dots & -A_{n-1} \end{bmatrix}$$

and

$$\mathcal{B} := \begin{bmatrix} I & 0 & 0 & \dots & 0 \\ 0 & I & 0 & \dots & 0 \\ \cdot & \cdot & \cdot & \dots & \cdot \\ 0 & 0 & 0 & \cdots I & 0 \\ 0 & 0 & 0 & \dots & B \end{bmatrix},$$

acting on E^n with their maximal domains. Then the operator matrix \mathcal{A} is closable, the operator matrix \mathcal{B} is closed and, due to (240), $\overline{\mathcal{A}}\vec{x_{\lambda}} = \lambda \mathcal{B}\vec{x_{\lambda}}, \lambda \in \Omega$. Furthermore, if we suppose that Ω_0 is an arbitrary open connected subset of Ω which admits a cluster point in Ω , then the linear span of the set $\{\vec{x_{\lambda}} : \lambda \in \Omega_0\}$ is dense in \tilde{E} . Now the statement follows similarly as in the proof of Theorem 2.10.1. \Box

REMARK 2.10.25. (i) The assertions of Theorem 2.10.1 and Theorem 2.10.9 continue to hold, with appropriate modifications, in the setting of separable sequentially complete locally convex spaces. The conclusion stated in Theorem 2.10.24 remains true if we consider the equation (238) with the same initial conditions and with the term $B \frac{d^n}{dt^n} u(t)$ replaced by $\frac{d^n}{dt^n} Bu(t)$ (cf. [308, Remark 12(i)]).

(ii) Suppose $\alpha_1 \in (0, 1)$ and $\alpha_i = i\alpha_1, i \in \mathbb{N}_n$. Keeping in mind the proofs of Theorem 2.10.24 and [308, Theorem 5], we can deduce some results about \mathfrak{D} -topologically mixing properties of the problem

(241)
$$B(\mathbf{D}_{t}^{\alpha_{1}})^{n}u(t) + \sum_{i=0}^{n-1} A_{i}(\mathbf{D}_{t}^{\alpha_{1}})^{i}u(t) = 0, \quad t \ge 0,$$
$$((\mathbf{D}_{t}^{\alpha_{1}})^{j}u(t))_{t=0} = u_{j}, \quad j = 0, \dots, n-1,$$

and its analogues obtained by replacing, optionally, some of the terms $B(\mathbf{D}_t^{\alpha_1})^n u(t)$ and $A_i(\mathbf{D}_t^{\alpha_1})^i u(t)$ by $(\mathbf{D}_t^{\alpha_1})^n Bu(t)$ and $(\mathbf{D}_t^{\alpha_1})^i A_i u(t)$, respectively $(0 \leq i \leq n-1)$. The case $\alpha_1 \in (1,2)$ can be considered quite similarly.

(iii) It should be emphasized that Theorem 2.10.24 cannot be so simply reformulated in the case that there exists an index $i \in \mathbb{N}_n$ such that $\alpha_i \notin \mathbb{N}$. In actual fact, probably the only way to exploit (240) is to find analytic functions $F_i: \Omega \to \mathbb{C}$ $(0 \leq i \leq m_n - 1)$ such that the equation (238), equipped with the initial conditions $u^{(i)}(0) = F_i(\lambda)f(\lambda), 0 \leq i \leq m_n - 1$, has a strong solution of the form $u(t; \lambda) = G(\lambda, t)f(\lambda), t \geq 0$ $(\lambda \in \Omega)$, where

(242)
$$\lambda^{-\alpha_n} \mathbf{D}_t^{\alpha_n} G(\lambda, t) = \dots = \lambda^{-\alpha_1} \mathbf{D}_t^{\alpha_1} G(\lambda, t) = G(\lambda, t), \quad t \ge 0 \quad (\lambda \in \Omega).$$

By [153, Theorem 7.2], the validity of (242) would imply that for each $t \ge 0, \lambda \in \Omega$ and $i \in \mathbb{N}_n$, we have:

$$G(\lambda,t) = F_0(\lambda)E_{\alpha_i}(\lambda^{\alpha_i}t^{\alpha_i}) + \sum_{k=1}^{m_i-1}F_k(\lambda)\int_0^t \frac{(t-s)^{k-1}}{(k-1)!}E_{\alpha_i}(\lambda^{\alpha_i}s^{\alpha_i})ds,$$

i.e., that for each $t \ge 0$, $\lambda \in \Omega$ and $i \in \mathbb{N}_n$, we have:

$$G(\lambda,t) = \sum_{l=0}^{\infty} \sum_{k=0}^{m_i-1} \lambda^{\alpha_i l} F_k(\lambda) \frac{t^{\alpha_i l+k}}{\Gamma(\alpha_i l+k+1)}.$$

The function $t \mapsto G(\lambda, t) - F_0(\lambda)$, $t \ge 0$ behaves asymptotically like $\lambda^{\alpha_1}F_0(\lambda)g_{\alpha_1+1}(t)$ as $t \to 0+$, so that the number α_1 cannot be an element of the interval (0,1) (to see this, consider the asymptotic behaviour of function $t \mapsto G(\lambda, t) - F_0(\lambda)$, $t \ge 0$ as $t \to 0+$, with the number α_1 replaced by α_2). Considering the asymptotic behaviour of function $t \mapsto G(\lambda, t) - F_0(\lambda)$, $t \ge 0$ ($t \mapsto G(\lambda, t) - F_0(\lambda) - tF_1(\lambda)$, $t \ge 0$), $t \ge 0$, $t \ge$

(iv) Hypercyclic and topologically mixing properties of higher-order non-degenerate differential equations with integer order derivatives have been studied in a series of recent papers by using the usual reduction into first order matrix differential equations (cf. [114, 118], [292, Section 3.2] and the references cited there). It should be observed that we can prove a slight extension of this theorem by using the analyses from [**300**, Remark 1(iii)] and [**302**, Remark 1;3.] (cf. also Remark 2.10.21(iii) and Example 2.10.26(i) below).

(v) In Theorem 2.10.16, we have reconsidered the well known assertion of S. El Mourchid [163, Theorem 2.1] concerning the connection between the imaginary point spectrum and hypercyclicity of strongly continuous semigroups. On the basis of this result, we can state some new facts about \mathcal{D} -topologically mixing properties of problem (238) considered in Theorem 2.10.24, provided that the equation (240) holds for all values of complex parameter λ belonging to some subinterval of imaginary axis.

We close this section by providing some illustrative examples.

(i) Consider the equation (238) with $\alpha_i = i, i \in \mathbb{N}_n$ EXAMPLE 2.10.26. and with the operator $A_0 = A$ replaced by -A. Although it may seem contrary, Theorem 2.10.24 is not so easily comparable to Theorem 2.10.9 in this case. For example, in the situation of Example 2.10.12 with P(z) =-z and $\alpha = 1$, the assumptions of Theorem 2.10.9 are satisfied with E := $L^{2}(\mathbb{R}), c_{1} > c > \frac{b}{2} > 0, A := (c-c_{1})I, B := \mathcal{A}_{c} - A, A_{1} := -\mathcal{A}_{c} + cI, \text{ where }$ the operator \mathcal{A}_c is defined by $D(\mathcal{A}_c) := \{u \in L^2(\mathbb{R}) \cap W_{loc}^{2,2}(\mathbb{R}) : \mathcal{A}_c u \in L^2(\mathbb{R})\}$ and $\mathcal{A}_c u := u'' + bxu' + cu, u \in D(\mathcal{A}_c), \Omega := \{\lambda \in \mathbb{C} : \lambda \neq 0, \lambda \neq c-c_1, \operatorname{Re} \lambda < c - \frac{b}{2}\}, f(\lambda) := g_1(\lambda) := \mathcal{F}^{-1}(e^{-\frac{\xi^2}{2b}}\xi|\xi|^{-(2+\frac{\lambda-c}{b})})(\cdot), \lambda \in \Omega$ or $f(\lambda) := g_2(\lambda) := \mathcal{F}^{-1}(e^{-\frac{\xi^2}{2b}}|\xi|^{-(1+\frac{\lambda-c}{b})})(\cdot), \, \lambda \in \Omega, \, f_1(\lambda) := (c-c_1)/(c-\lambda)$ and $f_2(\lambda) := (c - c_1)/(\lambda - (c - c_1))$ ($\lambda \in \Omega$). In particular, there is no open connected subset Ω' of Ω satisfying $\Omega' \cap i\mathbb{R} \neq \emptyset$ and $(\lambda^2/(f_2(\lambda)) +$ $\lambda/(f_1(\lambda)) - 1)Af(\lambda) = 0, \ \lambda \in \Omega'$, i.e., the equation (240) does not hold with this choice of $f(\lambda)$. This is quite predictable because the equation (240), with the set Ω and the function $f(\cdot)$ replaced respectively by ' Ω and $f(\cdot)$ therein (and in our further analysis, for the sake of consistency of notation), is equivalent to say that $(\lambda^2 - \lambda)\mathcal{A}_c' f(\lambda) = (\lambda^2 (c - c_1) - \lambda c + (c - c_1))\mathcal{A}_c' f(\lambda)$ (c_1))' $f(\lambda), \lambda \in \Omega$. Denote by Λ the set of all complex numbers $z \in i\mathbb{R} \setminus \{0\}$ for which there exists $\delta(z) > 0$ such that $\{0,1\} \cap L(z,\delta(z)) = \emptyset$, as well as that for each $\lambda \in L(z, \delta(z))$ we have $\operatorname{Re}(c - c_1 - \frac{c_1}{\lambda - 1} + \frac{c - c_1}{\lambda^2 - \lambda}) < c - \frac{b}{2}$. Recalling that $\{z \in \mathbb{C} : \operatorname{Re} z < c - \frac{b}{2}\} \subseteq \sigma_p(\mathcal{A}_c)$, it readily follows that Theorem 2.10.24 can be also applied here with $\Omega := \bigcup_{z \in \Lambda} L(z, \delta(z))$ and $f_i(\lambda) := g_i(c - c_1 - \frac{c_1}{\lambda - 1} + \frac{c - c_1}{\lambda^2 - \lambda}), \lambda \in \Omega$ (i = 1, 2), producing slightly different results from those obtained by applying Theorem 2.10.9 (with $\hat{E} = \check{E} = \overline{span\{[f_i(\lambda) \ \lambda' f_i(\lambda)]^T : \lambda \in '\Omega, \ i = 1, 2\}}, \text{ the subspace of } E^2$ whose first and second projection equals to E; cf. [115] and [300]). On the other hand, there exists a great number of very simple (non-)degenerate equations where we can apply Theorem 2.10.24 but not Theorem 2.10.9. Consider, for example, the equation $u'''(t) + (c_2 - A_c)u'(t) + c_1u(t) = 0$, $t \ge 0$, where $c_1 \in \mathbb{C} \setminus \{0\}$ and $c_2 \in \mathbb{C}$. The analysis taken up in [302, Remark 1(vi)] (cf. also [292, Theorem 3.3.9, Remark 3.3.10(v)]), with $c_3 = 0$, shows that there do not exist an open connected subset Ω_- of \mathbb{C}

and an index $i \in \{0, 1, 2\}$ such that the second equality in (227) holds. Contrary to this, there exist t > 0 and $\varepsilon > 0$ such that the equation (240) holds with $\Omega = L(it, \varepsilon)$.

(ii) Suppose Ω is an open non-empty subset of \mathbb{C} intersecting the imaginary axis, $f: \Omega \to E$ is an analytic mapping, $g: \Omega \to \mathbb{C} \setminus \{0\}$ is a scalar-valued mapping and $Af(\lambda) = g(\lambda)f(\lambda), \lambda \in \Omega$. Let $P_j(z)$ be non-zero complex polynomials $(j \in \mathbb{N}_n^0)$, and let

$$\lambda^n P_n(g(\lambda)) + \sum_{j=0}^{n-1} \lambda^j P_j(g(\lambda)) = 0, \quad \lambda \in \Omega.$$

Then the equation (240) holds with $B := P_n(A)$ and $A_j := P_j(A)$, $j \in \mathbb{N}_{n-1}^0$. If, additionally, the preasumption $\sum_{j=0}^{n-1} \langle x_j^*, \lambda^j f(\lambda) \rangle = 0$, $\lambda \in \Omega$ for some continuous linear functionals $x_j^* \in E^*$ given in advance $(j \in \mathbb{N}_{n-1}^0)$ implies $x_j^* = 0$ for all $j \in \mathbb{N}_{n-1}^0$, then the space E_0 from the formulation of Theorem 2.10.24 equals to E^n (cf. [114, Theorem 3.1] for a concrete example of this type with n = 3). Some applications of Theorem 2.10.24 in Fréchet function spaces can be given following the analysis from [292, Example 3.1.29].

2.11. The existence of distributional chaos in abstract degenerate fractional differential equations

The main aim of this section is to investigate a class of distributionally chaotic abstract degenerate (multi-term) fractional differential equations. We assume that X is an infinite-dimensional Fréchet space over the field \mathbb{C} and the topology of X is induced by the fundamental system $(p_n)_{n \in \mathbb{N}}$ of increasing seminorms. Let us recall that the translation invariant metric $d: X \times X \to [0, \infty)$, defined by

(243)
$$d(x,y) := \sum_{n=1}^{\infty} \frac{1}{2^n} \frac{p_n(x-y)}{1+p_n(x-y)}, \quad x,y \in X,$$

satisfies the following properties:

$$d(x+u, y+v) \leqslant d(x, y) + d(u, v), \quad x, y, u, v \in X,$$

$$(244) d(cx,cy) \leqslant (|c|+1)d(x,y), \quad c \in \mathbb{C}, x, y \in X,$$

and

(245)
$$d(\alpha x, \beta x) \ge \frac{|\alpha - \beta|}{1 + |\alpha - \beta|} d(0, x), \quad x \in X, \ \alpha, \beta \in \mathbb{C}.$$

Given $\varepsilon > 0$ in advance, set $L(0, \varepsilon) := \{x \in X : d(x, 0) < \varepsilon\}$. By Y we denote another Fréchet space over the field \mathbb{C} ; we assume that the topology of Y is induced by the fundamental system $(p_n^Y)_{n \in \mathbb{N}}$ of increasing seminorms. We define the translation invariant metric $d_Y : Y \times Y \to [0, \infty)$ by replacing $p_n(\cdot)$ with $p_n^Y(\cdot)$ in (243). In the case that $(X, \|\cdot\|)$ or $(Y, \|\cdot\|_Y)$ is a Banach space, then it will be assumed that the distance of two elements $x, y \in X$ $(x, y \in Y)$ is given by $d(x, y) := \|x - y\|$ $(d_Y(x,y) := ||x - y||_Y)$. With this terminological change, our structural results clarified in this section continue to hold in the case that X or Y is a Banach space.

We will split our exposition into two parts. In Subsection 2.11.1, we consider distributionally chaotic properties of linear (not necessarily continuous) operators and their sequences. Subsection 2.11.2 is devoted to te study of distributionally chaotic properties of abstract degenerate (multi-term) fractional differential equations.

2.11.1. Distributional chaos for single operators. The main purpose of this section is to investigate the basic distributionally chaotic properties of a sequence $(T_k)_{k \in \mathbb{N}}$ of linear mappings between the spaces X and Y. We start with the following definition.

DEFINITION 2.11.1. Suppose that, for every $k \in \mathbb{N}$, $T_k \colon D(T_k) \to Y$ is a linear (not necessarily continuous) operator and \tilde{X} is a closed linear subspace of X. Then we say that the sequence $(T_k)_{k\in\mathbb{N}}$ is \tilde{X} -distributionally chaotic iff there exist an uncountable set $S \subseteq \bigcap_{k=1}^{\infty} D(T_k) \cap \tilde{X}$ and $\sigma > 0$ such that for each $\varepsilon > 0$ and for each pair $x, y \in S$ of distinct points we have that

$$\overline{dens}(\{k \in \mathbb{N} : d_Y(T_k x, T_k y) \ge \sigma\}) = 1 \text{ and}$$
$$\overline{dens}(\{k \in \mathbb{N} : d_Y(T_k x, T_k y) < \varepsilon\}) = 1.$$

The sequence $(T_k)_{k\in\mathbb{N}}$ is said to be densely \tilde{X} -distributionally chaotic iff S can be chosen to be dense in \tilde{X} . A linear operator $T: D(T) \to Y$ is said to be (densely) \tilde{X} -distributionally chaotic iff the sequence $(T_k \equiv T^k)_{k\in\mathbb{N}}$ is. The set S is said to be $\sigma_{\tilde{X}}$ -scrambled set (σ -scrambled set in the case that $\tilde{X} = X$) of the sequence $(T_k)_{k\in\mathbb{N}}$ (the operator T).

The above notions are clearly equivalent in the case that $\tilde{X} = X$ and then we also say that the sequence $(T_k)_{k \in \mathbb{N}}$ (the operator T) is distributionally chaotic. Of course, it is of interest to know the minimal linear subspace \tilde{X} for which the sequence $(T_k)_{k \in \mathbb{N}}$ is \tilde{X} -distributionally chaotic because it is then \hat{X} -distributionally chaotic for any other linear subspace \hat{X} of X such that $\tilde{X} \subseteq \hat{X}$. Observe that there exist some important cases in which the sequence of linear mappings $(T_{k|\tilde{X}})$, acting between the spaces \tilde{X} and $Y = \tilde{X}$, is distributionally chaotic; see e.g. the proof of implication (i) \Rightarrow (ii) of [68, Theorem 12]. On the other hand, we are always trying to find the maximal possible subspace \tilde{X} of X for which the sequence $(T_k)_{k \in \mathbb{N}}$ (the operator T) is densely \tilde{X} -distributionally chaotic.

DEFINITION 2.11.2. Suppose that, for every $k \in \mathbb{N}$, $T_k \colon D(T_k) \to Y$ is a linear (not necessarily continuous) operator, \tilde{X} is a closed linear subspace of $X, x \in \bigcap_{k=1}^{\infty} D(T_k)$ and $m \in \mathbb{N}$. Then we say that:

(i) the orbit of x under $(T_k)_{k \in \mathbb{N}}$, i.e. the set $\{T_k x : k \in \mathbb{N}\}$, is distributionally near to 0 iff there exists $A \subseteq \mathbb{N}$ such that $\overline{dens}(A) = 1$ and $\lim_{k \in A, k \to \infty} T_k x = 0$,

- (ii) the orbit of x under $(T_k)_{k\in\mathbb{N}}$ is distributionally m-unbounded iff there exists $B \subseteq \mathbb{N}$ such that $\overline{dens}(B) = 1$ and $\lim_{k\in B, k\to\infty} p_m^Y(T_kx) = \infty$; the orbit of x under $(T_k)_{k\in\mathbb{N}}$ is said to be distributionally unbounded iff there exists $q \in \mathbb{N}$ such that this orbit is distributionally q-unbounded (if Y is a Banach space, this simply means that $\lim_{k\in B, k\to\infty} ||T_kx||_Y = \infty$),
- (iii) x is a \tilde{X} -distributionally irregular vector for the sequence $(T_k)_{k\in\mathbb{N}}$ (distributionally irregular vector for the sequence $(T_k)_{k\in\mathbb{N}}$, in the case that $\tilde{X} = X$) iff $x \in \bigcap_{k=1}^{\infty} D(T_k) \cap \tilde{X}$ and the orbit of x under $(T_k)_{k\in\mathbb{N}}$ is both distributionally near to 0 and distributionally unbounded.

If $T: D(T) \to Y$ is a linear operator and $x \in D_{\infty}(T)$, then we say that the orbit $\{T^k x : k \in \mathbb{N}\}$ is distributionally near to 0 (distributionally unbounded) iff the sequence $(T_k \equiv T^k)_{k \in \mathbb{N}}$ is distributionally near to 0 (distributionally unbounded); x is said to be a \tilde{X} -distributionally irregular vector for T (distributionally irregular vector for T, in the case that $\tilde{X} = X$) iff x is \tilde{X} -distributionally irregular vector for the sequence $(T_k \equiv T^k)_{k \in \mathbb{N}}$ (distributionally irregular vector for the sequence $(T_k \equiv T^k)_{k \in \mathbb{N}}$ (distributionally irregular vector for the sequence $(T_k \equiv T^k)_{k \in \mathbb{N}}$, in the case that $\tilde{X} = X$).

Suppose that $X' \subseteq \tilde{X}$ is a linear manifold. Then we say that X' is \tilde{X} -distributionally irregular manifold for the sequence $(T_k)_{k\in\mathbb{N}}$ (distributionally irregular manifold in the case that $\tilde{X} = X$) iff any element $x \in (X' \cap \bigcap_{k=1}^{\infty} D(T_k)) \setminus \{0\}$ is a \tilde{X} -distributionally irregular vector for the sequence $(T_k)_{k\in\mathbb{N}}$; the notion of a (\tilde{X}) -distributionally irregular manifold for a linear operator $T: D(T) \to Y$ is defined similarly. Following [68, Definition 14], it will be said that X' is a uniformly \tilde{X} -distributionally irregular manifold for the sequence $(T_k)_{k\in\mathbb{N}}$ (uniformly distributionally irregular manifold for the sequence $(T_k)_{k\in\mathbb{N}}$ is both that the orbit of each vector $x \in (X' \cap \bigcap_{k=1}^{\infty} D(T_k)) \setminus \{0\}$ under $(T_k)_{k\in\mathbb{N}}$ is both distributionally *m*-unbounded and distributionally near to 0. If so, then (243)–245 implies that X' is 2^{-m} -scrambled set for the sequence $(T_k)_{k\in\mathbb{N}}$. It can be simply verified that the following holds: If $x \in \tilde{X} \cap \bigcap_{k=1}^{\infty} D(T_k)$ is a \tilde{X} -distributionally irregular manifold for the sequence $(T_k)_{k\in\mathbb{N}}$. If X' is dense in \tilde{X} , then we deal with the notions of dense (\tilde{X}) -distributionally irregular manifolds, etc.

If $(T_k)_{k\in\mathbb{N}}$ and \tilde{X} are given in advance, then we define the linear mappings $\mathcal{T}_k \colon D(\mathcal{T}_k) \to Y$ by $D(\mathcal{T}_k) := D(T_k) \cap \tilde{X}$ and $\mathcal{T}_k x := T_k x$, $x \in D(\mathcal{T}_k)$ $(k \in \mathbb{N})$. Then $(\mathcal{T}_k)_{k\in\mathbb{N}}$ is a sequence of linear mappings between the Fréchet spaces \tilde{X} and Y. In some way, the next proposition enables us to reduce our further research in this subsection to the case in which $\tilde{X} = X$. An uncomplicated proof is left to the reader.

PROPOSITION 2.11.3. (i) The sequence $(T_k)_{k\in\mathbb{N}}$ is \tilde{X} -distributionally chaotic iff the sequence $(\mathcal{T}_k)_{k\in\mathbb{N}}$ is distributionally chaotic.

(ii) A vector x is a X-distributionally irregular vector for the sequence $(T_k)_{k\in\mathbb{N}}$ iff x is a distributionally irregular vector for the sequence $(\mathcal{T}_k)_{k\in\mathbb{N}}$. (iii) A linear manifold X' is a (uniformly) \tilde{X} -distributionally irregular manifold for the sequence $(T_k)_{k\in\mathbb{N}}$ iff X' is a (uniformly) distributionally irregular manifold for the sequence $(\mathcal{T}_k)_{k\in\mathbb{N}}$.

We continue by stating the following theorem.

THEOREM 2.11.4. (i) Suppose that $(T_k)_{k\in\mathbb{N}}$ is a sequence in L(X,Y) and X_0 is a dense linear subspace of X, satisfying that for each $x \in X_0$ there exists a set $A_x \subseteq \mathbb{N}$ such that $\overline{dens}(A_x) = 1$ and $\lim_{k\in A_x} T_k x = 0$. If there exist a zero sequence (y_k) in X, a number $\varepsilon > 0$ and a strictly increasing sequence (N_k) in \mathbb{N} such that, for every $k \in \mathbb{N}$ and some $m \in \mathbb{N}$,

$$\operatorname{card}(\{1 \leq j \leq N_k : p_m(T_j y_k) > \varepsilon\}) \ge N_k(1 - k^{-1}),$$

(for every $k \in \mathbb{N}$, card($\{1 \leq j \leq N_k : ||T_j y_k||_Y > \varepsilon\}$) $\geq N_k(1 - k^{-1})$, in the case that Y is a Banach space), then there exists a distributionally irregular vector for the sequence $(T_k)_{k \in \mathbb{N}}$, and particularly, $(T_k)_{k \in \mathbb{N}}$ is distributionally chaotic.

- (ii) Suppose that X is separable, (T_k)_{k∈N} is a sequence in L(X,Y), X₀ is a dense linear subspace of X, as well as:
 - (a) $\lim_{k\to\infty} T_k x = 0, x \in X_0, and$
 - (b) there exists $x \in X$ such that its orbit under $(T_k)_{k \in \mathbb{N}}$ is distributionally unbounded.

Then there exists a dense uniformly distributionally irregular manifold for $(T_k)_{k\in\mathbb{N}}$, and particularly, $(T_k)_{k\in\mathbb{N}}$ is densely distributionally chaotic.

PROOF. We will only outline the most relevant details of the proof. In the case that Y is a Fréchet space, the assertion (i) can be simply proved by replacing the operator T^j with the operator T_i $(j \in \mathbb{N})$ throughout the proofs of [68, Propositions 7 and 9]. Assuming that Y is a Banach space, the required assertion follows from the above by endowing Y with the following increasing family of seminorms $p_n^Y(y) :=$ $n\|y\|_Y$ $(n \in \mathbb{N}, y \in Y)$, which turns the space Y into a linearly and topologically homeomorphic Fréchet space. A careful inspection of the proof of [68, Theorem 15] shows that the assertion (ii) holds provided that X and Y are Fréchet spaces, and $p_m^Y(T_i x) \leq p_{i+m}(x), x \in X, i, m \in \mathbb{N}$; the only thing we need to do is replace any operator T^i appearing in the proof of afore-mentioned theorem with the operator T_i $(i \in \mathbb{N})$. Observe further that we can always construct a fundamental system $(p'_n(\cdot))_{n\in\mathbb{N}}$ of increasing seminorms on the space X, inducing the same topology, so that $p_m^Y(T_ix) \leq p'_{i+m}(x), x \in X, i, m \in \mathbb{N}$. In such a way, we may conclude that the assertion [68, Theorem 15] holds provided that X and Y are Fréchet spaces. In the case that X or Y is a Banach space, then we can 'renorm' it, as described above, and use after that the obtained result for Fréchet spaces. \square

Suppose now that $T: D(T) \subseteq X \to X$ is a linear mapping, $C \in L(X)$ is an injective mapping, as well as

(246)
$$R(C) \subseteq D_{\infty}(T) \text{ and } T^n C \in L(X) \text{ for all } n \in \mathbb{N}.$$

256

Then (246) implies that, for every $n \in \mathbb{N}$, the mapping $T_n \colon R(C) \to X$ defined by $T_n(Cx) := T^n Cx, x \in X, n \in \mathbb{N}$ is an element of the space L([R(C)], X). The next theorem follows almost immediately from Theorem 2.11.4.

THEOREM 2.11.5. Let the above conditions hold.

(i) Suppose that X_0 is a dense linear subspace of X, satisfying that for each $x \in X_0$ there exists a set $A_x \subseteq \mathbb{N}$ such that $\overline{dens}(A_x) = 1$ and $\lim_{k \in A_r, k \to \infty} T^k C x = 0$. If there exist a zero sequence (z_k) in X, a number $\varepsilon > 0$ and a strictly increasing sequence (N_k) in \mathbb{N} such that, for every $k \in \mathbb{N}$ and some $m \in \mathbb{N}$.

$$\operatorname{card}(\{1 \leq j \leq N_k : p_m(T^j C z_k) > \varepsilon\}) \ge N_k(1 - k^{-1}),$$

(for every $k \in \mathbb{N}$, card($\{1 \leq j \leq N_k : ||T^j C z_k||_Y > \varepsilon\}$) $\geq N_k(1 - k^{-1})$, in the case that Y is a Banach space), then there exists a distributionally vector $x \in R(C)$ for the operator T. In particular, T is distributionally chaotic and σ -scrambled set S of T can be chosen to be a linear submanifold of R(C).

- (ii) Suppose that X is separable, X_0 is a dense linear subspace of X, as well as:
 - (a) $\lim_{k\to\infty} T^k C x = 0, x \in X_0, and$
 - (b) there exist $x \in X$, $m \in \mathbb{N}$ and a set $B \subset \mathbb{N}$ such that $\overline{dens}(B) = 1$. and $\lim_{k\to\infty,k\in B} p_m(T^kCx) = \infty$, resp. $\lim_{k\to\infty,k\in B} ||T^kCx|| = \infty$ if X is a Banach space.

Then there exists a uniformly distributionally irregular manifold W for the operator T, and particularly, T is distributionally chaotic. Furthermore, if R(C) is dense in X, then W can be chosen to be dense in X and T is densely distributionally chaotic.

EXAMPLE 2.11.6. Suppose that X is separable, D(A) and R(C) are dense in $X, CA \subseteq AC, z_0 \in \mathbb{C} \setminus \{0\}, \beta \ge -1, d \in (0, 1], m \in (0, 1), \varepsilon \in (0, 1], \gamma > -1 \text{ and}$ the following conditions hold:

- (§) $P_{z_0,\beta,\varepsilon,m} := e^{i \arg(z_0)} (|z_0| + (P_{\beta,\varepsilon,m} \cup B_d)) \subseteq \rho_C(A), (\varepsilon, m(1+\varepsilon)^{-\beta}) \in \partial B_d,$ (§§) the family $\{(1+|\lambda|)^{-\gamma}(\lambda-A)^{-1}C : \lambda \in P_{z_0,\beta,\varepsilon,m}\} \subseteq L(X)$ is equicontinuous, and
- (§§§) the mapping $\lambda \mapsto (\lambda A)^{-1}Cx, \lambda \in P_{z_0,\beta,\varepsilon,m}$ is continuous for every fixed element $x \in E$.

Let $\Gamma_1(z_0,\beta,\varepsilon,m) = \{e^{i \arg(z_0)}(|z_0|+\xi+i\eta): \xi \ge \varepsilon, \eta = -m(1+\xi)^{-\beta}\},\$

 $\Gamma_2(z_0,\beta,\varepsilon,m) = \{e^{i\arg(z_0)}(|z_0|+\xi+i\eta):\xi^2+\eta^2 = d^2, \ \xi \leqslant \varepsilon\} \text{ and } \Gamma_3(z_0,\beta,\varepsilon,m) = 0$ $\{e^{i \arg(z_0)}(|z_0| + \xi + i\eta) : \xi \ge \varepsilon, \ \eta = m(1 + \xi)^{-\beta}\}.$ We assume that the curve $\Gamma(z_0, \beta, \varepsilon, m) = \bigcup_{i=1}^3 (e^{-i \arg(z_0)} \Gamma_i(z_0, \beta, \varepsilon, m) - |z_0|)$ is oriented so that $\operatorname{Im} \lambda$ decreases along $(e^{-i \arg(z_0)} \Gamma_1(z_0, \beta, \varepsilon, m) - |z_0|)$; since there is no risk for confusion, we also write Γ for $\Gamma(z_0, \beta, \varepsilon, m)$. Let $b \in (0, 1/2)$ be fixed, set $\delta_b := \arctan(\cos \pi b)$ and $A_0 := e^{-i \arg(z_0)} A - |z_0|$. Define, for every $z \in \Sigma_{\delta_b}$,

$$T_b(z)x := \frac{1}{2\pi i} \int_{\Gamma} e^{-z(-\lambda)^b} (\lambda - A_0)^{-1} Cx \, d\lambda, \quad x \in X.$$

Then $T_b(z) \in L(X)$ is injective and has dense range in X for any $z \in \Sigma_{\delta_b}$ (see e.g. the proofs of [296, Theorem 3.15-Theorem 3.16]). Furthermore, $T_b(z)A \subseteq AT_b(z)$, $z \in \Sigma_{\delta_b}$ and the condition (246) holds with T and C replaced respectively by A and $T_b(z)$ ($z \in \Sigma_{\delta_b}$). If there exist a dense subset X_0 of X and a number $\lambda \in \sigma_p(A)$ such that $\lim_{k\to\infty} A^k x = 0$, $x \in X_0$ and $|\lambda| > 1$, then Theorem 2.11.5(ii) implies that the operator zA^n is densely distributionally chaotic for any $n \in \mathbb{N}$ and $z \in \mathbb{C}$ with |z| = 1. Now we will illustrate this result by some examples.

(i) **[143]** Let $a, b, c > 0, c < \frac{b^2}{2a} < 1, X := L^2([0, \infty))$ and

$$\Lambda := \Big\{ \lambda \in \mathbb{C} : \Big| \lambda - \Big(c - \frac{b^2}{4a} \Big) \Big| \leqslant \frac{b^2}{4a}, \text{ Im } \lambda \neq 0 \text{ if } \operatorname{Re} \lambda \leqslant c - \frac{b^2}{4a} \Big\}.$$

Consider the operator -A defined by $D(-A) := \{f \in W^{2,2}([0,\infty)) : f(0) = 0\}$ and $-Au := au_{xx} + bu_x + cu$, $u \in D(A)$. Let $P(z) = \sum_{j=0}^n a_j z^j$ be a non-constant complex polynomial such that $a_n > 0$ and

$$P(-\Lambda) \cap \{z \in \mathbb{C} : |z| = 1\} \neq \emptyset.$$

Then -P(A) generates an analytic C_0 -semigroup of angle $\pi/2$, $P(-\Lambda) \subseteq \sigma_p(P(A))$ and it is not difficult to prove that the operator P(A) is densely distributionally chaotic.

- (ii) [115, 292, 427] This example has already appeared in our previous examinations from Section 2.10. Suppose $X := L^2(\mathbb{R}), c > b/2 > 0$, $\Omega := \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda < c - b/2\}$ and $\mathcal{A}_c u := u'' + 2bxu' + cu$ is the bounded perturbation of the one-dimensional Ornstein–Uhlenbeck operator acting with domain $D(\mathcal{A}_c) := \{u \in L^2(\mathbb{R}) \cap W^{2,2}_{loc}(\mathbb{R}) : \mathcal{A}_c u \in L^2(\mathbb{R})\}$. Then \mathcal{A}_c generates a strongly continuous semigroup, $\Omega \subseteq \sigma_p(\mathcal{A}_c)$, and for any open connected subset Ω' of Ω which admits a cluster point in Ω , we have $E = \overline{span\{g_i(\lambda) : \lambda \in \Omega', i = 1, 2\}}$, where $g_1(\lambda)$ and $g_2(\lambda)$ are defined as before. This simply implies that the operator \mathcal{A}_c is densely distributionally chaotic, this is also the property which holds for the multi-dimensional Ornstein–Uhlenbeck operators from [115, Section 4].
- (iii) [168] Suppose r > 0, $\sigma > 0$, $\nu = \sigma/\sqrt{2}$, $\gamma = r/\mu \mu$, s > 1, $s\nu > 1$ and $\tau \ge 0$. Set

$$Y^{s,\tau} := \Big\{ u \in C((0,\infty)) : \lim_{x \to 0} \frac{u(x)}{1 + x^{-\tau}} = \lim_{x \to \infty} \frac{u(x)}{1 + x^s} = 0 \Big\}.$$

Then $Y^{s,\tau}$, equipped with the norm

$$\|u\|_{s,\tau} := \sup_{x>0} \left| \frac{u(x)}{(1+x^{-\tau})(1+x^s)} \right|, \quad u \in Y^{s,\tau}.$$

becomes a separable Banach space. Let $D_{\mu} := \nu x d/dx$, with maximal domain in $Y^{s,\tau}$, and let the Black–Scholes operator \mathcal{B} be defined by $\mathcal{B} := D_{\nu}^2 + \gamma D_{\mu} - r$. H. Emamirad, G. R. Goldstein and J. A. Goldstein have proved in [168] that the operator \mathcal{B} generates a chaotic strongly continuous semigroup (it can be easily seen that the Black–Scholes semigroup is densely distributionally chaotic, as well; see the next subsection for the

notion). By [168, Lemma 3.3], the proof of [168, Lemma 3.5] (cf. especially the Figure 1 in the abovementioned paper, in the Ox'y' coordinate system, with $x' = x/\nu$ and $y' = y/\nu$) and the previous examination, it readily follows that the operator \mathcal{B} is densely distributionally chaotic.

(iv) [137] Assume that ω_1 , ω_2 , V_{ω_2,ω_1} , Q(z), Q(B), N and X possess the same meaning as in [137, Section 5], and

$$Q(\operatorname{int}(V_{\omega_2,\omega_1})) \cap \{z \in \mathbb{C} : |z| = 1\} \neq \emptyset.$$

Then it can be proved with the help of Theorem 2.11.5(ii) that the unbounded operator Q(B) is densely distributionally chaotic; the Devaney chaoticity of Q(B) can be proved in a similar fashion.

(v) Making use of Theorem 2.11.5(ii), we can also prove that certain polynomials of the Laplace–Beltrami operator $\Delta_{X,p}^{\natural}$, acting on the symmetric spaces of non-compact type, Damek–Ricci or Heckman–Opdam root spaces ([35, 259, 474]), are densely distributionally chaotic.

It is not difficult to see that the operators considered in the previous example are also chaotic (in the sense of [137, Definition 2.1]). Here we would like to mention in passing that Q. Menet [425] has recently solved [68, Problem 37] by constructing a linear continuous operator T acting on a classical Banach function space that is both chaotic and not distributionally chaotic. Motivated by the research of J. Bes, C. K. Chan and S. M. Seubert [69], where the chaotic behaviour of the abstract Laplace operator Δ has been analyzed, we would like to propose the following problem:

PROBLEM DC. Suppose $1 \leq p < \infty$, $\emptyset \neq \Omega \subseteq \mathbb{R}^n$ is an open (possibly unbounded) set, and the operator Δ acts on $L^p(\Omega)$ with maximal distributional domain and without any boundary conditions. Is it true that Δ is densely distributionally chaotic?

The proof of following slight extension of [430, Theorem 3] relies on use of K. Ball's planck theorems [430, Theorems 1 and 2] (cf. also [419, Proposition 9.1(a)]):

PROPOSITION 2.11.7. Suppose that X and Y are Banach spaces, and $T_k \in L(X,Y)$ for each $k \in \mathbb{N}$. If either

(i) $\sum_{k=1}^{\infty} \frac{1}{\|T_k\|} < \infty$

or

(ii) X is a complex Hilbert space and $\sum_{k=1}^{\infty} \frac{1}{\|T_k\|^2} < \infty$, then there exists $x \in X$ such that $\lim_{k \to \infty} \|T_k x\|_Y = \infty$.

Taking into account the assertion of Proposition 2.11.7 and the fact that [R(C)] is a Banach (complex Hilbert space) provided that X is, we may conclude that the following holds (cf. [68, Theorem 16] and [66, Corollary 30]): If $B \subseteq \mathbb{N}$, $\overline{dens}(B) = 1$, the assumptions of Theorem 2.11.5(i) and (ii)(a) are valid, and exactly one of the following two conditions holds:

(i) X is a Banach space and $\sum_{k \in B} \frac{1}{\|T^k C\|} < \infty$,

or

(ii) X is a complex Hilbert space and $\sum_{k \in B} \frac{1}{\|T^k C\|^2} < \infty$,

then there exists $x \in X$ such that $\lim_{k\to\infty,k\in B} ||T^k Cx|| = \infty$, i.e., the condition stated in Theorem 2.11.5(ii)(b) automatically holds.

Before we move ourselves to the next subsection, we want to mention distributionally chaotic properties of multivalued linear operators as an interesting theme for our researchers.

2.11.2. Distributionally chaotic properties of abstract degenerate fractional differential equations. The following continuous version of Theorem 2.11.5(ii) will be essentially utilized in this subsection.

THEOREM 2.11.8. Suppose that X is separable, X_0 is a dense linear subspace of X, $(T(t))_{t\geq 0} \subseteq L(X,Y)$ is a strongly continuous operator family, as well as:

- (a) $\lim_{t\to\infty} T(t)x = 0, x \in X_0$,
- (b) there exist $x \in X$, $m \in \mathbb{N}$ and a set $B \subseteq [0, \infty)$ such that $\overline{Dens}(B) = 1$, and $\lim_{t \to \infty, t \in B} p_m(T(t)x) = \infty$, resp. $\lim_{t \to \infty, t \in B} ||T(t)x|| = \infty$ if X is a Banach space.

Then there exist a dense linear subspace S of X and a number $\sigma > 0$ such that for each $\varepsilon > 0$ and for each pair $x, y \in S$ of distinct points we have

$$\overline{Dens}(\{t \ge 0 : d_Y(T(t)x, T(t)y) \ge \sigma\}) = 1$$

and

$$\overline{Dens}(\{t \ge 0 : d_Y(T(t)x, T(t)y) < \varepsilon\}) = 1$$

PROOF. The proof is very similar to those of [68, Theorem 15] and Theorem 2.11.5(ii). Consider first the case in which X and Y are Frechét spaces. If so, the family $(T(t))_{t\geq 0} \subseteq L(X,Y)$ is locally equicontinuous because it is strongly continuous and X is barreled ([296]). Hence, for every $l, n \in \mathbb{N}$, there exist $c_{l,n} > 0$ and $a_{l,n} \in \mathbb{N}$ such that $p_l^Y(T(t)x) \leq c_{l,n}p_{a_{l,n}}(x), x \in X, t \in [0, n]$. Suppose, for the time being, that:

(247)
$$p_k^Y(T(t)x) \leqslant p_{k+\lceil t\rceil}(x), \quad x \in X, \ t \ge 0, \ k \in \mathbb{N}.$$

Without loss of generality, we may assume that m = 1. Then one can find a sequence $(x_k)_{k \in \mathbb{N}}$ in X_0 such that $p_k(x_k) \leq 1$, $k \in \mathbb{N}$ and a strictly increasing sequence of positive real numbers $(t_k)_{k \in \mathbb{N}}$ tending to infinity such that:

$$Dens(\{1 \leq t \leq t_k : p_1(T(t)x_k) > k2^k\}) \ge t_k(1-k^{-2})$$

and

$$\overline{Dens}(\{1 \le t \le t_k : p_k(T(t)x_l) < k^{-1}\}) \ge t_k(1-k^{-2}), \quad l = 1, \dots, k-1.$$

Let $(r_k)_{k\in\mathbb{N}}$ be a strictly increasing sequence in \mathbb{N} so that:

$$r_{j+1} \ge 1 + r_j + \lceil t_{r_j+1} \rceil, \quad j \in \mathbb{N}.$$

Due to the proof of [68, Theorem 15], we obtain the existence of a dense linear subspace S of X such that, for every $x \in S$, there exist two sets A_x , $B_x \subseteq [0, \infty)$ such that $\overline{Dens}(A) = \overline{Dens}(B) = 1$, $\lim_{t\to\infty,t\in A_x} T(t)x = 0$ and $\lim_{t\to\infty,t\in B_x} p_1(T(t)x)$ $=\infty$. Now the final conclusion of theorem follows as in the discrete case. Introducing recursively the following fundamental system of increasing seminorms $p'_n(\cdot)$ $(n \in \mathbb{N})$ on X:

$$p'_{1}(x) \equiv p_{1}(x), \quad x \in X,$$

$$p'_{2}(x) \equiv p'_{1}(x) + c_{1,1}p_{a_{1,1}}(x) + p_{2}(x), \quad x \in X,$$

...

$$p'_{n+1}(x) \equiv p'_{n}(x) + c_{1,n}p_{a_{1,n}}(x) + \dots + c_{n,1}p_{a_{n,1}}(x) + p_{n+1}(x), \quad x \in X,$$

...

we may assume without loss of generality that (247) holds, hence the assertion is proved in the case that X and Y are Frechét spaces. If X or Y is a Banach space, then we can 'renorm' it, as it has been explained in the proof of Theorem 2.11.5(i), and use after that the above result to finish the proof of theorem.

Suppose that $n \in \mathbb{N} \setminus \{1\}, A_1, \ldots, A_{n-1}, A_n = B$ are closed linear operators on X and $0 \leq \alpha_1 < \cdots < \alpha_n$. In this section, we scrutinize distributionally chaotic properties of abstract degenerate fractional Cauchy problem [(90)-(91)], which will be simply denoted by (ACP) here. We focus special attention on distributionally chaotic solutions of the fractional Sobolev equations (DFP)_R and (DFP)_L, as well. Along with the problems (DFP)_R and (DFP)_L, we consider the associated abstract integral equation (this is, clearly, a special case of problem (52) with $a(t) = g_{\alpha}(t)$):

(248)
$$Bu(t) = f(t) + \int_0^t g_\alpha(t-s)Au(s)ds, \quad t \ge 0,$$

where $f \in C([0,\infty): X)$. Henceforth (DFP) denotes either (DFP)_R or (DFP)_L. By a mild solution of the problem (DFP)_R we mean any continuous function $t \mapsto u(t), t \ge 0$ such that the mapping $t \mapsto Bu(t), t \ge 0$ is continuous and $A(g_{\alpha} * u)(t) = Bu(t) - \sum_{k=0}^{\lceil \alpha \rceil - 1} g_{k+1}(t)Bx_k, t \ge 0$. The set of all vectors $\vec{x} = (Bx_0, Bx_1, \ldots, Bx_{\lceil \alpha \rceil - 1})$ for which there exists a mild solution of problem (DFP)_R will be denoted by $Z_{\alpha,R}^{mild}(A, B)$. Denote by \mathfrak{T} the exact number of initial values subjected to the problem (ACP); that is, \mathfrak{T} is the sum of number m_Q and the cardinality of set consisting of those pairs $(i, j) \in \mathbb{N}_n \times \mathbb{N}_{m_n-1}^0$ for which $m_i - 1 \ge j \ge m_Q$. By $\mathfrak{Z}(\mathfrak{Z}_{uniq})$ we denote the set of all tuples of initial values $\vec{x} = ((u_j)_{0 \le j \le m_Q-1}; (u_{i_s', j})_{1 \le s' \le s, m_Q \le j \le m_{i_{s'}} - 1}) \in X^{\mathfrak{T}}$ for which there exists a (unique) strong solution of problem (ACP).

The notion of (subspace) distributional chaoticity of problem (ACP) is introduced in the following definition.

DEFINITION 2.11.9. Let \tilde{X} be a closed linear subspace of $X^{\mathfrak{T}}$. Then it is said that the abstract Cauchy problem (ACP) is \tilde{X} -distributionally chaotic iff there are an uncountable set $S \subseteq \tilde{X} \cap \mathfrak{Z}$ and $\sigma > 0$ such that for each $\varepsilon > 0$ and for each pair $\vec{x}, \ \vec{y} \in S$ of distinct tuples we have that there exist strong solutions $t \mapsto u(t; \vec{x}),$ $t \ge 0$ and $t \mapsto u(t; \vec{y}), t \ge 0$ of problem (ACP) with the property that

$$Dens(\{t \ge 0 : d(u(t; \vec{x}), u(t; \vec{y})) \ge \sigma\}) = 1$$
 and

$$\overline{Dens}(\{t \ge 0 : d(u(t; \vec{x}), u(t; \vec{y})) < \varepsilon\}) = 1.$$

If we can choose S to be dense in \tilde{X} , then we also say that the problem (ACP) is densely \tilde{X} -distributionally chaotic (S is called a $\sigma_{\tilde{X}}$ -scrambled set). In the case that $\tilde{X} = X$, it is also said that the problem (ACP) is (densely) distributionally chaotic; S is then called a σ -scrambled set.

The notion introduced in Definition 2.11.9 can be slightly modified for the problem $(\text{DFP})_R$ by requiring that, for every two distinct tuples vectors of $\sigma_{\tilde{X}}$ -scrambled set $S \subseteq \tilde{X} \cap Z_{\alpha,R}^{mild}(A,B)$, there exist mild solutions $t \mapsto u(t; \vec{x}), t \ge 0$ and $t \mapsto u(t; \vec{y}), t \ge 0$ of problem $(\text{DFP})_R$ obeying the properties prescribed. We will not follow this approach henceforth.

DEFINITION 2.11.10. Let $n \in \mathbb{N}$, let \tilde{X} be a closed linear subspace of $X^{\mathfrak{T}}$, and let $\vec{x} \in \tilde{X} \cap \mathfrak{Z}$. Then it is said that the vector \vec{x} is:

(i) \tilde{X} -(ACP)-distributionally near to 0 iff there exist a set $Z \subseteq [0, \infty)$ and a strong solution $t \mapsto u(t; \vec{x}), t \ge 0$ of problem (ACP) such that

$$\overline{Dens}(Z) = 1$$
 and $\lim_{t \in Z, t \to +\infty} u(t; \vec{x}) = 0;$

(ii) \tilde{X} -(ACP)-distributionally *n*-unbounded iff there exist a set $Z \subseteq [0, \infty)$ and a strong solution $t \mapsto u(t; \vec{x}), t \ge 0$ of problem (ACP) such that

$$\overline{Dens}(Z) = 1$$
 and $\lim_{t \in Z, t \to +\infty} p_n(u(t; \vec{x})) = +\infty;$

 \vec{x} is said to be \tilde{X} -(ACP)-distributionally unbounded iff there exists $q \in \mathbb{N}$ such that \vec{x} is \tilde{X} -(ACP)-distributionally q-unbounded (if $(X, \|\cdot\|)$) is a Banach space, this simply means that there exist a set $Z \subseteq [0, \infty)$ and a strong solution $t \mapsto u(t; \vec{x}), t \ge 0$ of problem (ACP) such that $\overline{Dens}(Z) =$ 1 and $\lim_{t \in Z, t \to +\infty} \|u(t; \vec{x})\| = +\infty$);

(iii) a \tilde{X} -(ACP)-distributionally irregular vector iff there exist an integer $q \in \mathbb{N}$, two subsets B_0 , B_∞ of $[0,\infty)$ with $\overline{Dens}(B_0) = \overline{Dens}(B_\infty) = 1$ and a strong solution $t \mapsto u(t; \vec{x}), t \ge 0$ of problem (ACP) such that

(249)
$$\lim_{t \in B_0, t \to +\infty} u(t; \vec{x}) = 0 \text{ and } \lim_{t \in B_\infty, t \to +\infty} p_q(u(t; \vec{x})) = +\infty.$$

In the case that $\tilde{X} = X^{\mathfrak{T}}$, then we also say that \vec{x} is (ACP)-distributionally near to 0, resp., (ACP)-distributionally *n*-unbounded, (ACP)-distributionally unbounded; a \tilde{X} -distributionally irregular vector for (ACP) is then called a distributionally irregular vector for (ACP).

Suppose that $X' \subseteq \tilde{X} \cap \mathfrak{Z}$ is a linear manifold. Then we say that X' is a \tilde{X} -distributionally irregular manifold for (ACP) (distributionally irregular manifold for (ACP), in the case that $\tilde{X} = X^{\mathfrak{T}}$) iff any element $x \in X' \setminus \{0\}$ is \tilde{X} -distributionally irregular vector for (ACP). Further on, we say that X' is a uniformly \tilde{X} -distributionally irregular manifold for (ACP) (uniformly distributionally irregular manifold for (ACP), in the case that $\tilde{X} = X^{\mathfrak{T}}$) iff there exists $q \in \mathbb{N}$ such that, for every $\vec{x} \in X' \setminus \{0\}$, there exist two subsets B_0 , B_{∞} of $[0, \infty)$ with

 $\overline{Dens}(B_0) = \overline{Dens}(B_\infty) = 1$ and a strong solution $t \mapsto u(t; \vec{x}), t \ge 0$ of problem (ACP) such that (249) holds. It can be simply verified with the help of translation invariance of metric $d(\cdot, \cdot)$ and inequalities (244)–(245) that the following holds: If $0 \ne \vec{x} \in \tilde{X} \cap \mathfrak{Z}$ is a \tilde{X} -distributionally irregular vector for (ACP), then $X' \equiv span\{\vec{x}\}$ is a uniformly \tilde{X} -distributionally irregular manifold for (ACP).

REMARK 2.11.11. (i) If \vec{x} is a \tilde{X} -distributionally irregular vector for (ACP), then \vec{x} is both \tilde{X} -(ACP)-distributionally near to 0 and \tilde{X} -(ACP)-distributionally unbounded. The converse statement holds provided that strong solutions of problem (ACP) are unique. If this is not the case and $\vec{x} \neq 0$ is both \tilde{X} -(ACP)-distributionally near to 0 and \tilde{X} -(ACP)-distributionally unbounded, then we can prove the following (cf.Definition 2.11.9): There are an uncountable set $S \subseteq \tilde{X} \cap \mathfrak{Z}$ (S is, in fact, equal to $span\{\vec{x}\}$) and $\sigma > 0$ such that for each $\varepsilon > 0$ and for each pair $\vec{x}, \vec{y} \in S$ of distinct vectors we have that there exist strong solutions $t \mapsto u_i(t; \vec{x})$, $t \ge 0$ and $t \mapsto u_i(t; \vec{y})$, $t \ge 0$ (i = 1, 2) of problem (ACP) with the property that

$$\overline{Dens}(\{t \ge 0 : d(u_1(t; \vec{x}), u_1(t; \vec{y})) \ge \sigma\}) = 1 \text{ and}$$
$$\overline{Dens}(\{t \ge 0 : d(u_2(t; \vec{x}), u_2(t; \vec{y})) < \varepsilon\}) = 1.$$

If this is the case, we say that the problem (ACP) is quasi \tilde{X} -distributionally chaotic (quasi distributionally chaotic, provided that $\tilde{X} = X^{\mathfrak{T}}$). The set S is called quasi $\sigma_{\tilde{X}}$ -scrambled set (quasi σ -scrambled set, provided that $\tilde{X} = X^{\mathfrak{T}}$).

- (ii) Suppose that (ACP) is \tilde{X} -distributionally chaotic and S is the corresponding $\sigma_{\tilde{X}}$ -scrambled set. Then, for every two distinct vectors $\vec{x}, \ \vec{y} \in S, \ \vec{x} \vec{y}$ is a \tilde{X} -distributionally vector for (ACP).
- (iii) Suppose that (ACP) is quasi \tilde{X} -distributionally chaotic and S is the corresponding quasi $\sigma_{\tilde{X}}$ -scrambled set. Then, for every two distinct vectors $\vec{x}, \vec{y} \in S, \vec{x} \vec{y}$ is a quasi \tilde{X} -distributionally vector for (ACP), i.e., $\vec{x} \vec{y}$ is both \tilde{X} -(ACP)-distributionally near to 0 and \tilde{X} -(ACP)-distributionally unbounded.

It is worth noting that the non-triviality of subspace $\bigcap_{i=1}^{n} N(A_i)$ in X immediately implies that the problem (ACP) is distributionally chaotic:

EXAMPLE 2.11.12. Suppose that $0 \neq x \in \bigcap_{i=1}^{n} N(A_i)$. We can always find a sequence $(a_n)_{n \in \mathbb{N}_0}$ of non-negative real numbers and a scalar-valued function $f \in C^{\infty}([0,\infty))$ such that $a_0 = 0$, $a_n > a_{n-1}+2$, $n \in \mathbb{N}$, $\lim_{n \to +\infty} (a_{n-1}-1)(a_n-1)^{-1} = 0$, f(t) = 0 for $t \in \bigcup_{n \in \mathbb{N}} [a_{2n-1}, a_{2n}]$ and f(t) = 2n for $t \in \bigcup_{n \in \mathbb{N}_0} [a_{2n} + 1, a_{2n+1} - 1]$. Since the sets $B_0 := \bigcup_{n \in \mathbb{N}} [a_{2n-1}, a_{2n}]$ and $B_{\infty} := \bigcup_{n \in \mathbb{N}_0} [a_{2n} + 1, a_{2n+1} - 1]$ have the upper densities equal to 1, it is very simple to verify that the function $u(t; \vec{x}) := f(t)x, t \ge 0$ is a strong solution of problem (ACP) with the initial value $\vec{x} = ((u_j \equiv f^{(j)}(0)x)_{0 \le j \le m_Q-1}; (u_{i_{s'},j} \equiv 0)_{1 \le s' \le s, m_Q \le j \le m_{s'}-1}) \in X^{\mathfrak{T}}$, as well as

that \vec{x} is a distributionally irregular vector for (ACP). In particular, $\vec{x} = \vec{0}$ can be a distributionally irregular vector for (ACP).

It is not clear whether there exists a quasi distributionally chaotic problem (ACP) that is not distributionally chaotic. In the sequel, we will consider only the classical notion of (subspace) distributional chaoticity of problem (ACP).

The notions of exponentially equicontinuous (q_{α}, C) -regularized resolvent family for (248) and exponentially equicontinuous (q_{α}, C) -regularized resolvent family generated by A, B will be crucially important in our work. We know the following facts about exponentially equicontinuous (q_{α}, C) -regularized resolvent families (cf. Section 2.1–Section 2.2 for more details).

- Lemma 2.11.13. (i) Let $(R(t))_{t\geq 0}$ be an exponentially equicontinuous (q_{α}, C) -regularized resolvent family for (248). Suppose that the following condition holds:
 - (P)" There exists a number $\omega_1 > \omega$ such that, for every $x \in X$, there exists a function $h(\lambda; x) \in LT - X$ such that $h(\lambda; x) = \lambda^{\alpha - 1} (\lambda^{\alpha} B - A)^{-1} Cx$. provided $\operatorname{Re} \lambda > \omega_1$.

Let $x_0, \ldots, x_{\lceil \alpha \rceil - 1} \in D(A) \cap D(B)$. Then the function $u(t; (BCx_0, \ldots,$ $BCx_{\lceil \alpha \rceil - 1}) := \sum_{j=0}^{\lceil \alpha \rceil - 1} \int_0^t g_j(t-s) R(s) x_j \, ds, \ t \ge 0 \ is \ a \ unique \ strong$ solution of $(DFP)_R$, with the initial values Bx_i replaced by BCx_i $(0 \leq$ $j \leq \lceil \alpha \rceil - 1$).

(ii) Let $(R(t))_{t\geq 0}$ be an exponentially equicontinuous (g_{α}, C) -regularized resolvent family generated by A, B. Then for each $x_0, \ldots, x_{\lceil \alpha \rceil - 1} \in D(A) \cap$ $D(B) \text{ the function } u(t; (Cx_0, \dots, Cx_{\lceil \alpha \rceil - 1})) := \sum_{j=0}^{\lceil \alpha \rceil - 1} \int_0^t g_j(t-s)R(s)$ $Bx_i ds, t \ge 0$ is a unique strong solution of problem $(DFP)_L$, with the initial values x_j replaced by Cx_j $(0 \leq j \leq \lceil \alpha \rceil - 1)$.

In the following theorem, we will consider the subspace distributionally chaotic properties of problem $(DFP)_R$.

THEOREM 2.11.14. Suppose that $\alpha > 0, C \in L(X)$ is injective, $CA \subseteq AC$, $CB \subseteq BC$, $(R(t))_{t\geq 0}$ is an exponentially equicontinuous (g_{α}, C) -regularized resolvent family for (248), (P)" holds and $\emptyset \neq \mathcal{V} \subseteq \mathbb{N}^0_{\lceil \alpha \rceil - 1}$. Suppose, further, that the following conditions hold:

- Let F_i be a separable complex Fréchet space, let $F_i \subseteq D(A) \cap D(B)$, and let F_i be continuously embedded in X $(i \in \mathcal{V})$.
- Suppose that for each $n \in \mathbb{N}$ and $i \in \mathcal{V}$ there exist a number $c_{n,i} > 0$ and a continuous seminorm $q_{n,i}(\cdot)$ on F_i so that $p_n(CBf_i) \leq c_{n,i}q_{n,i}(f_i)$, $f_i \in F_i$. Set $G_i := F_i$, if $i \in \mathcal{V}$, $G_i := \{0\}$, if $i \in \mathbb{N}^0_{\lceil \alpha \rceil - 1} \smallsetminus \mathcal{V}$, and
- $\lim_{t \to +\infty} (g_i * R(\cdot)f_i)(t) = 0, \ f_i \in F_i^0.$
- Let there exist $\vec{f_{\infty}} = (f_{0,\infty}, \dots, f_{\lceil \alpha \rceil 1,\infty}) \in F, m \in \mathbb{N}$ and a set $D \subseteq$ $[0,\infty)$ such that $\overline{Dens}(D) = 1$, and

 $\lim_{t \to +\infty, t \in D} p_m(\sum_{i \in \mathcal{V}} (g_i * R(\cdot) f_{i,\infty})(t)) = +\infty, \text{ resp. } \lim_{t \to +\infty, t \in D} \|\sum_{i \in \mathcal{V}} (g_i * R(\cdot) f_{i,\infty})(t)\| = +\infty \text{ if } (X, \|\cdot\|) \text{ is a Banach space.}$

Then we have that the problem $(DFP)_R$ is densely

$$\overline{\{(CBf_0,\ldots,CBf_{\lceil \alpha\rceil-1}):\vec{f}=(f_0,\ldots,f_{\lceil \alpha\rceil-1})\in F\}}^{X^{\lceil \alpha\rceil}}$$
-distributionally chaotic.

PROOF. It is clear that F is an infinite-dimensional separable complex Fréchet space. Define $V(t)\vec{f} := \sum_{i=0}^{\lceil \alpha \rceil - 1} (g_i * R(\cdot)f_i)(t), t \ge 0$ $(\vec{f} = (f_0, \dots, f_{\lceil \alpha \rceil - 1}) \in F)$ and $F_0 := \prod_{i=0}^{\lceil \alpha \rceil - 1} G_i^0$, where $G_i^0 := F_i^0$, if $i \in \mathcal{V}$, and $G_i^0 := \{0\}$, if $i \in \mathbb{N}_{\lceil \alpha \rceil - 1}^0 \setminus \mathcal{V}$. Then F_0 is dense in F and $(V(t))_{t\ge 0} \subseteq L(F, X)$ is a strongly continuous operator family. An application of Theorem 2.11.8 yields that there exist a dense linear subspace S of F and a number $\sigma > 0$ such that for each $\varepsilon > 0$ and for each pair $\vec{f'}, \vec{f''} \in S$ of distinct vectors we have

$$\overline{Dens}(\{t \ge 0 : d(V(t)\vec{f'}, V(t)\vec{f''}) \ge \sigma\}) = 1 \text{ and }$$

$$\overline{Dens}(\{t \ge 0 : d(V(t)\vec{f'}, V(t)\vec{f''}) < \varepsilon\}) = 1.$$

Suppose that $CBf_i = 0$ for all $i \in \mathcal{V}$ and $f_i \in F_i$. Then (54) and the uniqueness theorem for the Laplace transform together imply that $R(t)f_i = 0$ for all $i \in \mathcal{V}$ and $f_i \in F_i$, which contradicts the existence of *m*-distributionally unbounded vector $\vec{f_{\infty}}$ from *F*. Hence, there exist $i \in \mathcal{V}$ and $f_i \in F_i$ such that $CBf_i \neq 0$. Using this fact and the continuity of mapping $CB \colon F_i \to X$ for each $i \in \mathcal{V}$, we can simply verify that $\{(CBf_0, \ldots, CBf_{\lceil \alpha \rceil - 1}) : \vec{f} = (f_0, \ldots, f_{\lceil \alpha \rceil - 1}) \in S\}$ is a non-trivial subspace of $X^{\lceil \alpha \rceil}$. Now the final conclusion simply follows by using the continuity of mappings $CB \colon F_i \to X$ ($i \in \mathcal{V}$) once more, and Lemma 2.11.13(i). \Box

Similarly, by using Theorem 2.11.8 and Lemma 2.11.13(ii), we can prove the following theorem on subspace distributional chaoticity of problem $(DFP)_L$.

THEOREM 2.11.15. Suppose that $\alpha > 0$, $C \in L(X)$ is injective, $CA \subseteq AC$, $CB \subseteq BC$, $(R(t))_{t \ge 0}$ is an exponentially equicontinuous (g_{α}, C) -regularized resolvent family generated by A, B, and $\emptyset \neq \mathcal{V} \subseteq \mathbb{N}^{0}_{\lceil \alpha \rceil - 1}$. Suppose, further, that the following conditions hold:

- Let F_i be a separable complex Fréchet space, and let $F_i \subseteq D(A) \cap D(B)$ $(i \in \mathcal{V}).$
- Suppose that for each $n \in \mathbb{N}$ and $i \in \mathcal{V}$ there exist a number $c_{n,i} > 0$ and a continuous seminorm $q_{n,i}(\cdot)$ on F_i so that $p_n(Bf_i) + p_n(Cf_i) \leq c_{n,i}q_{n,i}(f_i), f_i \in F_i$. Set $G_i := F_i$, if $i \in \mathcal{V}, G_i := \{0\}$, if $i \in \mathbb{N}_{\lceil \alpha \rceil - 1}^0 \setminus \mathcal{V}$, and $F := \prod_{i=0}^{\lceil \alpha \rceil - 1} G_i$.
- Suppose, further, that for each $i \in \mathcal{V}$ there exists a dense subset F_i^0 of F_i satisfying that $\lim_{t \to +\infty} (g_i * R(\cdot)Bf_i)(t) = 0, f_i \in F_i^0$.
- Let there exist $\vec{f} = (f_0, \ldots, f_{\lceil \alpha \rceil 1}) \in F$, $m \in \mathbb{N}$ and a set $D \subseteq [0, \infty)$ such that $\overline{Dens}(D) = 1$, and $\lim_{t \to +\infty, t \in D} p_m(\sum_{i \in \mathcal{V}} (g_i * R(\cdot)Bf_i)(t)) = +\infty$, resp. $\lim_{t \to +\infty, t \in D} \|\sum_{i \in \mathcal{V}} (g_i * R(\cdot)Bf_i)(t)\| = +\infty$ if $(X, \|\cdot\|)$ is a Banach space.

 $\frac{\text{Then we have that the problem } (\text{DFP})_L \text{ is densely}}{\{(Cf_0, \dots, Cf_{\lceil \alpha \rceil - 1}) : \vec{f} = (f_0, \dots, f_{\lceil \alpha \rceil - 1}) \in F\}} \overset{X^{\lceil \alpha \rceil}}{\rightarrow} \text{-distributionally chaotic.}}$

- REMARK 2.11.16. (i) Suppose that $l \in \mathbb{N}_0$. Then the increasing family of seminorms $p_{n,l}^{A,B,C}(\cdot) := p_n(C^{-l}\cdot) + p_n(C^{-l}A\cdot) + p_n(C^{-l}B\cdot)$ $(n \in \mathbb{N})$ turns $C^{l}(D(A) \cap D(B))$ into a Fréchet space, which will be denoted by $[D(A) \cap D(B)]_C^l$ in the sequel. In the concrete situation of Theorem 2.11.14 or Theorem 2.11.15, F_i can be chosen to be some of closed linear subspaces of $[D(A) \cap D(B)]_C^l$ that is separable for the topology induced from $[D(A) \cap D(B)]_C^l$. If X is separable and C = I, then for each number $\lambda > \omega$ the mapping $(\lambda^{\alpha} B - A)^{-1}$: $X \to [D(A) \cap D(B)] (\equiv [D(A) \cap D(B)]_I^0)$ is a linear topological isomorphism and, in this case, $[D(A) \cap D(B)]$ and F_i will be separable $(i \in \mathcal{V})$.
 - (ii) If we suppose additionally that $R(t)B \subseteq BR(t), t \ge 0$ in the formulation of Theorem 2.11.15, then we do not need to assume that for each $n \in \mathbb{N}$ and $i \in \mathcal{V}$ there exist a number $c_{n,i} > 0$ and a continuous seminorm $q_{n,i}(\cdot)$ on F_i so that $p_n(Bf_i) \leq cq_{n,i}(f_i), f_i \in F_i$ (because, in this case, the operator $V_L(t) \vec{\cdot} = \sum_{i=0}^{\lceil \alpha \rceil - 1} (g_i * R(\cdot)B \cdot i)(t), \ t \ge 0 \ (\vec{\cdot} = (\cdot_0, \dots, \cdot_{\lceil \alpha \rceil - 1}) \in F) \text{ belongs}$ to the space L(F, X) and $(V_L(t))_{t \ge 0}$ is a strongly continuous operator family in L(F, X)).

Keeping in mind Theorem 2.10.3, it is very natural to raise the following issue: **PROBLEM 1.** Suppose that $\alpha \in (0,2)$ and Ω is an open connected subset of \mathbb{C} which satisfies $\Omega \cap (-\infty, 0] = \emptyset$ and $\Omega \cap i\mathbb{R} \neq \emptyset$. Let $f: \Omega^{\alpha} \to X$ be an analytic mapping such that $f(\lambda^{\alpha}) \in N(A - \lambda^{\alpha}B) \setminus \{0\}, \lambda \in \Omega$. Does there exist a closed linear subspace X' of $X^{\lceil \alpha \rceil}$ such that the problems $(DFP)_R$ and $(DFP)_L$ are (densely) X'-distributionally chaotic?

The method proposed in the proofs of [112, Theorem 4.1] and its discrete precursor [68, Theorem 15] cannot be applied here and, because of that, one has to follow new paths capable of moving us towards a solution of this problem. Unfortunately, we will present only some partial answers to Problem 1 by assuming that the strong solutions of problem (DFP) are governed by an exponentially equicontinuous (q_{α}, C) -regularized resolvent family for (248) (in the case of problem (DFP)_R) or an exponentially equicontinuous (g_{α}, C) -regularized resolvent family generated by A, B (in the case of problem $(DFP)_L$).

We start by stating the following result.

- Theorem 2.11.17. (i) Suppose that $0 < \alpha < 2$, $C \in L(X)$ is injective, $CA \subseteq AC$, $CB \subseteq BC$, $(R(t))_{t \ge 0}$ is an exponentially equicontinuous (g_{α}, C) -regularized resolvent family for (248), (P)" holds and $\emptyset \neq \mathcal{V} \subseteq$ $\mathbb{N}^{0}_{\lceil \alpha \rceil - 1}$. Suppose, further, that the following conditions hold:
 - Let F_i be a separable complex Fréchet space, let $F_i \subseteq D(A) \cap D(B)$, and let F_i be continuously embedded in X $(i \in \mathcal{V})$.
 - Suppose that for each $n \in \mathbb{N}$ and $i \in \mathcal{V}$ there exist a number $c_{n,i} >$ 0 and a continuous seminorm $q_{n,i}(\cdot)$ on F_i so that $p_n(CBf_i) \leq$

 $c_{n,i}q_{n,i}(f_i), f_i \in F_i.$ Set $G_i := F_i, if i \in \mathcal{V}, G_i := \{0\}, if i \in \mathbb{N}_{[\alpha]-1}^0 \setminus \mathcal{V}, and F := \prod_{i=0}^{[\alpha]-1} G_i.$

 $-Let H_i: \Omega^{\alpha} \to F_i$ be an analytic mapping such that $H_i(\lambda^{\alpha}) \in N(A - A)$ $\lambda^{\alpha}B) \smallsetminus \{0\}, \ \lambda \in \Omega \ (i \in \mathcal{V}).$

 $Set F'_{i} := \overline{span\{H_{i}(\lambda^{\alpha}): \lambda \in \Omega\}}^{F_{i}} \ (i \in \mathcal{V}), \ F'_{i} := \{0\} \ (i \in \mathbb{N}^{0}_{\lceil \alpha \rceil - 1} \smallsetminus \mathcal{V})$ and $F' := \prod_{i=0}^{\lceil \alpha \rceil - 1} F'_i$. Then the problem $(DFP)_R$ is densely $\frac{1}{\{(CBf'_0,\ldots,CBf'_{\lceil \alpha\rceil-1}):\vec{f'}=(f'_0,\ldots,f'_{\lceil \alpha\rceil-1})\in F'\}} \overset{X^{\lceil \alpha\rceil}}{-distributionally}$ chaotic.

- (ii) Suppose that $0 < \alpha < 2$, $C \in L(X)$ is injective, $CA \subseteq AC$, $CB \subseteq BC$, $(R(t))_{t\geq 0}$ is an exponentially equicontinuous (g_{α}, C) -regularized resolvent family generated by A, B, and $\emptyset \neq \mathcal{V} \subseteq \mathbb{N}^0_{\lceil \alpha \rceil - 1}$. Suppose, further, that the following conditions hold:
 - Let F_i be a separable complex Fréchet space, and let $F_i \subseteq D(A) \cap D(B)$ $(i \in \mathcal{V}).$
 - Suppose that for each $n \in \mathbb{N}$ and $i \in \mathcal{V}$ there exist a number $c_{n,i} > 0$ and a continuous seminorm $q_{n,i}(\cdot)$ on F_i so that $p_n(Bf_i) + p_n(Cf_i) \leq$ $c_{n,i}q_{n,i}(f_i), f_i \in F_i. \quad Set \ G_i := F_i, \text{ if } i \in \mathcal{V}, \ G_i := \{0\}, \text{ if } i \in \mathbb{N}_{[\alpha]-1} \setminus \mathcal{V}, \text{ and } F := \prod_{i=0}^{\lceil \alpha \rceil - 1} G_i.$
 - Let $H_i: \Omega^{\alpha} \to F_i$ be an analytic mapping such that $H_i(\lambda^{\alpha}) \in N(A I)$

 $\lambda^{\alpha}B) \smallsetminus \{0\}, \ \lambda \in \Omega \ (i \in \mathcal{V}).$ Set $F'_i := \overline{span\{H_i(\lambda^{\alpha}) : \lambda \in \Omega\}}^{F_i} \ (i \in \mathcal{V}), \ F'_i := \{0\} \ (i \in \mathbb{N}^0_{\lceil \alpha \rceil - 1} \smallsetminus \mathcal{V})$ and $F' := \prod_{i=0}^{\lceil \alpha \rceil - 1} F'_i$. Then the problem $(DFP)_L$ is densely $\overline{\{(Cf'_0,\ldots,Cf'_{\lceil \alpha\rceil-1}):\vec{f'}=(f'_0,\ldots,f'_{\lceil \alpha\rceil-1})\in F'\}}^{X^{\lceil \alpha\rceil}} \text{-} distributionally}$ chaotic

PROOF. Suppose that Ω_0 is an arbitrary open connected subset of Ω which admits a cluster point in Ω . Then the (weak) analyticity of mapping $\lambda \mapsto H_i(\lambda^{\alpha}) \in$ $F_i, \lambda \in \Omega$ implies that $\Psi(\Omega_0, i) := \operatorname{span}\{H_i(\lambda^\alpha) : \lambda \in \Omega_0\}$ is dense in the Fréchet space F'_i ; in particular, $(F'_i)_0 := \Psi(\Omega \cap \mathbb{C}_{-}, i)$ is dense in F'_i $(i \in \mathcal{V})$. The rest of proof is almost the same in cases (i) and (ii), so that we will consider only (i). Since $H_i(\lambda^{\alpha}) \in N(A - \lambda^{\alpha}B) \setminus \{0\}, \lambda \in \Omega$, we can apply the uniqueness theorem for Laplace transform, (43) and (54) in order to see that $R(t)H_i(\lambda^{\alpha}) =$ $E_{\alpha}(t^{\alpha}\lambda^{\alpha})CH_i(\lambda^{\alpha}), t \ge 0, \lambda \in \Omega$ and that $(g_i * E_{\alpha}(\cdot^{\alpha}\lambda^{\alpha}))(t) = t^i E_{\alpha,i+1}(t^{\alpha}\lambda^{\alpha}),$ $t \ge 0, i \in \mathbb{N}_0$ $(i \in \mathcal{V})$. Now the claimed assertion follows from an application of Theorem 2.11.14 and the asymptotic expansion formulae (40)-(42).

REMARK 2.11.18. Suppose that the requirements of Problem 1 hold, $C \in L(X)$ is injective, $CA \subseteq AC$, $CB \subseteq BC$, X is separable, $l \in \mathbb{N}_0$ and $\emptyset \neq \mathcal{V} \subseteq \mathbb{N}^0_{\lceil \alpha \rceil - 1}$.

(i) Let $(R(t))_{t\geq 0}$ be an exponentially equicontinuous (g_{α}, C) -regularized resolvent family for (248), and let (P)" hold. Then the closed graph theorem implies that $(\lambda^{\alpha}B - A)^{-1}C \in L(X)$ for all $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega_1$. Suppose that $\operatorname{Re} \lambda_0 > \omega_1$, $(\lambda^{\alpha} B - A)^{-1} CA \subseteq A(\lambda^{\alpha} B - A)^{-1} C$ and $(\lambda^{\alpha}B - A)^{-1}CB \subseteq B(\lambda^{\alpha}B - A)^{-1}C$ for all $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega_1$. Set

$$X_l := \overline{span\{C^l(\lambda_0^{\alpha}B - A)^{-1}Cf(\lambda^{\alpha}) : \lambda \in \Omega\}}^{[D(A) \cap D(B)]_C^l}$$

Then the mapping $G: X \to [D(A) \cap D(B)]_C^l$, given by $G(x) := C^l (\lambda_0^{\alpha} B - A)^{-1}Cx, x \in X$, is continuous and the space $\overline{G(X)}^{[D(A) \cap D(B)]_C^l}$ is separable. Set $F_i := [D(A) \cap D(B)]_C^l$ $(i \in \mathcal{V})$. Define G_i and the space F as in the formulation of Theorem 2.11.17(i). Then, for every $i \in \mathcal{V}$, the mapping $H_i: \Omega^{\alpha} \to F_i$, given by $H_i(\lambda^{\alpha}) := C^l (\lambda_0^{\alpha} B - A)^{-1} Cf(\lambda^{\alpha}), \lambda \in \Omega$, is analytic $(i \in \mathcal{V})$. Define now F'_i and F' as in the formulation of Theorem 2.11.17(i). Applying Theorem 2.11.17(i), we get that $(DFP)_R$ is densely $\overline{\{(CBf'_0, \ldots, CBf'_{\lceil \alpha \rceil - 1}) : \vec{f'} = (f'_0, \ldots, f'_{\lceil \alpha \rceil - 1}) \in F'\}}^{X^{\lceil \alpha \rceil}}$ -distributionally chaotic (observe that $C^l(X_m) = X_{m+l}$ and $C(X_l) \subseteq X_l$ for all $l, m \in \mathbb{N}_0$, as well as that $X_0 \supseteq X_1 \supseteq \cdots \supseteq X_l \supseteq \ldots$ and $\{(CBf'_0, \ldots, CBf'_{\lceil \alpha \rceil - 1}) \in F'\} = \{(x_0, \ldots, x_{\lceil \alpha \rceil - 1}) \in X^{\lceil \alpha \rceil} : x_i = 0 \text{ for } i \in \mathbb{N}^0_{\lceil \alpha \rceil - 1} \smallsetminus \mathcal{V}, \text{ and } x_i \in B(X_{l+1}) \text{ for } i \in \mathcal{V}\};$ in the sequel, we will use the abbreviation $X_{B_l, \mathcal{V}}^{\lceil \alpha \rceil}$ to denote the above set).

(ii) Let $(R(t))_{t\geq 0}$ be an exponentially equicontinuous (g_{α}, C) -regularized resolvent family generated by A, B, obeying additionally that $(\lambda^{\alpha}B-A)^{-1}C$ commutes with A and B for all $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$. Then Theorem 2.11.17(ii) and the above consideration imply that the problem $(DFP)_L$ is densely

$$\overline{\{(x_0,\ldots,x_{\lceil\alpha\rceil-1})\in X^{\lceil\alpha\rceil}: x_i=0, i\in\mathbb{N}^0_{\lceil\alpha\rceil-1}\smallsetminus\mathcal{V}; x_i\in X_{l+1}, i\in\mathcal{V}\}}^{X^{\lceil\alpha\rceil}}$$
-distributionally chaotic. We will denote the above set simply by $X_{l,\mathcal{V}}^{\lceil\alpha\rceil}$.

Now we will present an illustrative example of application of obtained results.

EXAMPLE 2.11.19. Suppose that $0 < \alpha < 2$, $\cos(\pi/\alpha) \leq 0$, $l \in \mathbb{N}_0$, $\emptyset \neq \mathcal{V} \subseteq \mathbb{N}^0_{\lceil \alpha \rceil - 1}$, $1 \leq p < \infty$, $\omega \geq 0$, $P_1(z)$ and $P_2(z)$ are non-zero complex polynomials, $N_1 = dg(P_1(z))$, $N_2 = dg(P_2(z))$, $P_2(x) \neq 0$ for all $x \in \mathbb{R}$, $\beta > \frac{1}{2} \frac{(N_1 + N_2)}{\min(1, \alpha)}$ and (70) holds. Then there exist numbers $z_0 \in \mathbb{C}$ and $r \geq 0$ such that:

$$P_1(-iz_0) = re^{\pm i\alpha\pi/2}P_2(-iz_0), \quad P_2(z_0) \neq 0$$

and

$$P_1(-i \cdot)'(z_0) P_2(-i z_0) - P_1(z_0) P_2(-i \cdot)'(z_0) \neq 0.$$

Let a > 0 be such that $|\operatorname{Re}(z_0)| < a/p$. Set $\rho(x) := e^{-a|x|}, x \in \mathbb{R}$,

$$L^p_{\rho}(\mathbb{R}) := \left\{ f \colon \mathbb{R} \to \mathbb{C} \mid f(\cdot) \text{ is measurable, } \int_{\mathbb{R}} |f(x)|^p \rho(x) dx < \infty \right\}$$

and $||f|| := (\int_{\mathbb{R}} |f(x)|^p \rho(x) dx)^{1/p}$; equipped with this norm, $X := L^p_{\rho}(\mathbb{R})$ becomes an infinite-dimensional separable complex Banach space. It is well known that the operator $-iA_0$, defined by

$$D(-iA_0) := \{ f \in X \mid f(\cdot) \text{ is loc. abs. continuous, } f' \in X \}, \ (-iA_0)f := f',$$

is the generator of a C_0 -group on X (cf. [143, Theorem 4.9]). Therefore, we can define the closed linear operators $A := \overline{P_1(A_0)}$ and $B := \overline{P_2(A_0)}$ on X by using the functional calculus for bounded commuting C_0 -groups; recall that these operators are densely defined, and B is injective. Arguing as in Example 2.10.5, we obtain that there exist an open connected subset Ω of $\mathbb{C} \setminus (-\infty, 0]$ intersecting the imaginary axis and an open connected neighborhood W of point z_0 , contained in the vertical strip $\{z \in \mathbb{C} : |\operatorname{Re}(z)| < a/p\}$, such that the mapping $(P_1(-i \cdot)/P_2(-i \cdot))^{-1} \colon \Omega^{\alpha} \to$ W is well defined, analytic and bijective. Set

$$f(\lambda^{\alpha}) := e^{(P_1(-i\cdot)/P_2(-i\cdot))^{-1}(\lambda^{\alpha})\cdot}, \quad \lambda \in \Omega$$

and, for every $t \ge 0$,

$$R_{\alpha}(t) := \left(E_{\alpha} \left(t^{\alpha} \frac{P_1(x)}{P_2(x)} \right) (1 + |x|^2)^{-\beta/2} \right) (A_0), \quad G_{\alpha}(t) := \overline{P_2(A_0)}^{-1} R_{\alpha}(t).$$

Then $(R_{\alpha}(t))_{t\geq 0} \subseteq L(X)$ is a global exponentially bounded $(q_{\alpha}, R_{\alpha}(0))$ -regularized resolvent family for (248), (P)" holds, $(G_{\alpha}(t))_{t\geq 0} \subseteq L(X, [D(B)])$ is a global exponentially bounded $(q_{\alpha}, R_{\alpha}(0))$ -regularized resolvent family generated by A, B, the mapping $f: \Omega^{\alpha} \to X$ is analytic and $Af(\lambda^{\alpha}) = \lambda^{\alpha} Bf(\lambda^{\alpha}), \lambda \in \Omega$. Furthermore, there exists $\omega_1 > \omega$ such that $(\lambda^{\alpha} B - A)^{-1}C$ commutes with A and B for all $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega_1$; here $C = R_{\alpha}(0)$. Now it readily follows that the problem $(DFP)_R$ $((DFP)_L)$ is densely $X_{B,l,\mathcal{V}}^{\lceil \alpha \rceil}$ -distributionally chaotic (densely $X_{IV}^{[\alpha]}$ -distributionally chaotic); unfortunately, in the present situation, we do not know to say anything about the optimality of this result. Before passing on to some concrete examples where the established conclusions can be applied, it should be noticed that we can prove a similar result provided that the state space is chosen to be the Banach space $C_{0,\rho}(\mathbb{R})$ (cf. [143, Definition 4.3]) or the Fréchet space $X' := \{ f \in C^{\infty}(\mathbb{R}) : f^{(n)} \in L^p_{\rho}(\mathbb{R}) \text{ for all } n \in \mathbb{N}_0 \}, \text{ equipped with the following family of seminorms } p_n(f) := \sum_{j=0}^n \|f^{(j)}\|_{L^p_{\rho}(\mathbb{R})}, n \in \mathbb{N}_0; \text{ in this case, the opera$ tors $A_{|X'}$ and $B_{|X'}$ are linear and continuous on X', $(C_{|X'})^{-1} \in L(X')$, as well as $((C_{|X'})^{-1}R_{\alpha}(t)_{|X'})_{t\geq 0} \subseteq L(X')$ is a global exponentially equicontinuous $(g_{\alpha}, I_{X'})$ regularized resolvent family for (248), (P)" holds in our concrete situation, and $((C_{|X'})^{-1}G_{\alpha}(t)_{|X'})_{t\geq 0} \subseteq L(X', [D(B_{|X'})])$ is a global exponentially equicontinuous $(g_{\alpha}, I_{X'})$ -regularized resolvent family generated by $A_{|X'}, B_{|X'}$:

(i) Assuming that $P_1(z) = -\alpha_0 z^2 - \beta_0 z^4$ and $P_2(z) = \gamma_0 + z^2$, where $\alpha_0, \beta_0, \gamma_0$ are positive real numbers, we are in a position to clarify some results on subspace distributionally chaoticity of fractional linearized Benney–Luke equation

$$(\gamma_0 - \Delta) \mathbf{D}_t^{\alpha} u = \alpha_0 \Delta u - \beta_0 \Delta^2 u.$$

(ii) Assuming that $P_1(z) = z^2$ and $P_2(z) = -\eta z^2 - 1$, where $\eta > 0$, we are in a position to clarify some results on subspace distributionally chaoticity

of the fractional Barenblatt–Zheltov–Kochina equation

$$(\eta \Delta - 1) \mathbf{D}_t^{\alpha} u(t) + \Delta u = 0 \quad (\eta > 0)$$

In the remainder of subsection, we will always assume that

$$T_n u(t) = B \mathbf{D}_t^{\alpha_n} u(t) = T_{n,L} u(t)$$

Then it is evident that the abstract degenerate Cauchy problem

(250)
$$B\mathbf{D}_t^{\alpha_n} u(t) + \sum_{i=1}^{n-1} T_i u(t) = 0; \quad u^{(k)}(0) = u_k, \ 0 \le k \le m_n - 1$$

is a special subcase of problem (ACP). The Caputo fractional derivative $\mathbf{D}_t^{\alpha_n} u(t)$ is defined for any strong solution $t \mapsto u(t)$, $t \ge 0$ of problem (250) and this, in turn, implies that we can define the Caputo fractional derivative $\mathbf{D}_t^{\zeta} u(t)$ for any number $\zeta \in [0, \alpha_n]$. Motivated by our results from Subsection 2.10.4, we introduce the following notion:

DEFINITION 2.11.20. Let \tilde{X} be a closed linear subspace of X^{m_n} , let $k \in \mathbb{N}$, and let $\vec{\beta} = (\beta_1, \beta_2, \ldots, \beta_k) \in [0, \alpha_n]^k$. Then it is said that the abstract Cauchy problem (250) is $(\tilde{X}, \vec{\beta})$ -distributionally chaotic iff there are an uncountable set $S \subseteq \tilde{X} \cap \mathfrak{Z}$ and $\sigma > 0$ such that for each $\varepsilon > 0$ and for each pair $\vec{x}, \vec{y} \in S$ of distinct tuples we have that there exist strong solutions $t \mapsto u(t; \vec{x}), t \ge 0$ and $t \mapsto u(t; \vec{y}), t \ge 0$ of problem (250) with the property that

$$\overline{Dens}\left(\left\{t \ge 0 : \sum_{i=1}^{k} d(\mathbf{D}_{t}^{\beta_{i}}u(t;\vec{x}), \mathbf{D}_{t}^{\beta_{i}}u(t;\vec{y})) \ge \sigma\right\}\right) = 1 \text{ and}$$
$$\overline{Dens}\left(\left\{t \ge 0 : \sum_{i=1}^{k} d(\mathbf{D}_{t}^{\beta_{i}}u(t;\vec{x}), \mathbf{D}_{t}^{\beta_{i}}u(t;\vec{y})) < \varepsilon\right\}\right) = 1.$$

As before, if we can choose S to be dense in \tilde{X} , then we say that the problem (250) is densely $(\tilde{X}, \vec{\beta})$ -distributionally chaotic (S is called a $(\sigma_{\tilde{X}}, \vec{\beta})$ -scrambled set). In the case $\tilde{X} = X^{m_n}$, it is also said that the problem (250) is (densely) $\vec{\beta}$ -distributionally chaotic; S is then called a $(\sigma, \vec{\beta})$ -scrambled set.

The classical definitions of \tilde{X} -distributional chaos of problem 250 can be obtained by plugging $\vec{\beta} = (0, 0, ..., 0) \in [0, \alpha_n]^k$ in Definition 2.11.20; we can also define some other notions of \tilde{X} -distributional chaos of problem 250 by replacing, optionally, some of terms $\mathbf{D}_t^{\beta_i} u(t; \vec{x})$ and $\mathbf{D}_t^{\beta_i} u(t; \vec{y})$ in Definition 2.11.20 with $\mathbf{D}_t^{\beta_i} A'_i u(t; \vec{x})$ or $A''_i \mathbf{D}_t^{\beta_i} u(t; \vec{x})$, and $\mathbf{D}_t^{\beta_i} B'_i u(t; \vec{y})$ or $B''_i \mathbf{D}_t^{\beta_i} u(t; \vec{y})$, respectively, where A'_i, A''_i, B'_i, B''_i are closed linear operators on X $(1 \leq i \leq k)$.

Now we would like to present an illustrative example from [113].

EXAMPLE 2.11.21. The study of various hypercyclic, chaotic and topologically mixing properties of the viscous van Wijngaarden–Eringen equation:

(251)
$$(1 - a_0^2 u_{xx})u_{tt} = (\operatorname{Re}_b)^{-1} u_{xxt} + u_{xx},$$

which corresponds to the linearized version of the equation that models the acoustic planar propagation in bubbly liquids, has recently been carried out by J. A. Conejero, C. Lizama and M. Murillo-Arcila in [113]; here, $a_0 > 0$ denotes the dimensionless bubble radius and $\text{Re}_b > 0$ is a Reynolds number. The state space in their analysis is X_{ρ} , which consists of real analytic functions of Herzog type

$$X_{\rho} := \bigg\{ f \colon \mathbb{R} \to \mathbb{C} \ ; \ f(x) = \sum_{n=0}^{\infty} \frac{a_n \rho^n}{n!} x^n, \ x \in \mathbb{R} \text{ for some } (a_n)_{n \ge 0} \in c_0(\mathbb{N}_0) \bigg\},$$

which is an isomorphic copy of the sequence space $c_0(\mathbb{N}_0)$. More precisely, it has been proved that the bounded matricial operator

$$A := \begin{bmatrix} O & I \\ -(1 - a_0^2 u_{xx})^{-1} u_{xx} & (\operatorname{Re}_b)^{-1} (1 - a_0^2 u_{xx})^{-1} u_{xx} \end{bmatrix}$$

generates a strongly continuous semigroup on X_{ρ}^2 satisfying the assumptions of Desch–Schappacher–Webb criterion, provided $a_0 < 1$, $\sqrt{5}/6 < a_0 \operatorname{Re}_b < 1/2$ and $\rho > r_0 a_0^{-1}/(2^{-1}a_0^{-2}(\operatorname{Re}_b)^{-1} - 3r_0)$ ($r_0 := 4^{-1}a_0^{-2}(\operatorname{Re}_b)^{-1}(1 - 4a_0^2\operatorname{Re}_b^2)^{1/2}$). This immediately implies that the abstract degenerate second order Cauchy problem (251) is densely (0, 1)-distributionally chaotic (cf. Definition 2.11.20 and Theorem 2.10.24).

The following problem is in a close connection with the last mentioned theorem:

PROBLEM 2. Let $\alpha_i = i$ for all $i \in \mathbb{N}_n$, let Ω be an open non-empty subset of \mathbb{C} intersecting the imaginary axis, and let $f: \Omega \to E$ be an analytic mapping satisfying (240). Does there exist a tuple $\vec{\beta} \in [0, \alpha_n]^k$ and a closed linear subspace X' of X^n such that the problem (ACP)_{B,n} is (densely) $(X', \vec{\beta})$ -distributionally chaotic?

Now we are going to enquire into the basic distributionally chaotic properties of the following special subcase of problem (250):

(252)
$$B\mathbf{D}_t^{\alpha_n} u(t) + \sum_{i=1}^{n-1} A_i \mathbf{D}_t^{\alpha_i} u(t) = 0; \quad u^{(k)}(0) = u_k, \ 0 \le k \le m_n - 1.$$

In our analysis, we are primarily concerned with exploiting Theorem 2.11.8 and, because of that, we need to assume that strong solutions of (252) are governed by some known degenerate resolvent families for (252); besides of that, it is very important to know whether strong solutions of (252) are unique or not. Here we will focus our attention on the use of (C_1, C_2) -existence and uniqueness families for (252), with both operators $C_1 \in L(X)$ and $C_2 \in L(X)$ being injective. Let us recall that for $0 \leq i \leq m_n - 1$ we define the sets $D_i := \{j \in \mathbb{N}_{n-1} : m_j - 1 \geq i\},$ $D'_i := \mathbb{N}_{n-1} \setminus D_i$ and

$$\mathbf{D}_i := \bigg\{ u_i \in \bigcap_{j \in D'_i} D(A_j) : A_j u_i \in R(C_1), \ j \in D'_i \bigg\}.$$

The existence of a C_2 -uniqueness family for (252) implies the uniqueness of strong solutions of this problem, while the existence of a C_1 -existence family for (252)

implies the following assertion, already known from our previous examinations (cf. Subsection 2.3.5 for more details):

LEMMA 2.11.22. Suppose that $(E(t))_{t\geq 0}$ is a C_1 -existence family for (252), and $u_i \in \mathbf{D}_i$ for $0 \leq i \leq m_n - 1$. Define, for every $t \geq 0$,

(253)
$$u(t) := \sum_{i=0}^{m_n-1} u_i g_{i+1}(t) - \sum_{i=0}^{m_n-1} \sum_{j \in \mathbb{N}_{n-1} \smallsetminus D_i} (g_{\alpha_n - \alpha_j} * E^{(m_n-1-i)})(t) C_1^{-1} A_j u_i.$$

Then the function $t \mapsto u(t), t \ge 0$ is a strong solution of (252).

The following theorem can be proved by using Theorem 2.11.8 and Lemma 2.11.22.

THEOREM 2.11.23. Suppose that $C_1 \in L(X)$ is injective and $(E(t))_{t\geq 0}$ is a C_1 -existence family for (252). Let F be a separable complex Fréchet space, let $F \subseteq \mathbf{D}_0 \times \mathbf{D}_1 \times \cdots \times \mathbf{D}_{m_n-1}$, and let F be continuously embedded in X^{m_n} . Define $V(t) : F \to X$ by $V(t)\vec{u} := \sum_{i=0}^{m_n-1} u_i g_{i+1}(t) - \sum_{i=0}^{m_n-1} \sum_{j \in \mathbb{N}_{n-1} \setminus D_i} (g_{\alpha_n - \alpha_j} * E^{(m_n-1-i)})(t)C_1^{-1}A_ju_i$ $(t \ge 0, \vec{u} = (u_0, u_1, \dots, u_{m_n-1}) \in F; cf.$ (253)). Suppose that $V(t) \in L(F, X)$ for all $t \ge 0$, $\vec{x} \in F_0$. Let there exist $\vec{x} \in F$, $m \in \mathbb{N}$ and a set $B \subseteq [0, \infty)$ such that $\overline{Dens}(B) = 1$, and $\lim_{t\to +\infty, t\in B} p_m(V(t)\vec{x}) = +\infty$, resp. $\lim_{t\to +\infty, t\in B} \|V(t)\vec{x}\|_X = +\infty$ if $(X, \|\cdot\|_X)$ is a Banach space. Then the problem (252) is densely $F^{X^{m_n}}$ -distributionally chaotic.

Now we will apply Theorem 2.11.23 in the study of distributionally chaotic properties of fractional analogues of the viscous van Wijngaarden-Eringen equation.

EXAMPLE 2.11.24. Suppose $1/2 < \alpha \leq 1$ and p > 2. Consider the following fractional degenerate multi-term problem:

(254)
$$(1 + a_0^2 \Delta_{X,p}^{\natural}) \mathbf{D}_t^{2\alpha} u(t,x) + (\operatorname{Re}_b)^{-1} \Delta_{X,p}^{\natural} \mathbf{D}_t^{\alpha} u(t,x) + \Delta_{X,p}^{\natural} u(t,x) = 0, \ t \ge 0;$$

 $u(0,x) = u_0(x), \ u_t(0,x) = u_1(x),$

on a symmetric space X of non-compact type and rank one. Let P_p be the parabolic domain defined in [259]; then we know that $int(P_p) \subseteq \sigma_p(\Delta_{X,p}^{\natural})$. In our concrete situation, we have that n = 3, $\alpha_3 = 2\alpha$, $\alpha_2 = \alpha$, $\alpha_1 = 0$, $B = (1 + a_0^2 \Delta_{X,p}^{\natural})$, $A_2 = (\operatorname{Re}_b)^{-1} \Delta_{X,p}^{\natural}$, $A_1 = \Delta_{X,p}^{\natural}$ and $\mathbf{P}_{\lambda} = 1 + (a_0^2 + \lambda^{-\alpha} + \lambda^{-2\alpha}) \Delta_{X,p}^{\natural}$ for $\operatorname{Re} \lambda > 0$. Then it is clear that $z(\lambda) := (a_0^2 + \lambda^{-\alpha} + \lambda^{-2\alpha})^{-1} \to a_0^{-2}$ for $|\lambda| \to \infty$, as well as that

$$\lambda^{-1}\mathbf{P}_{\lambda}^{-1} = \lambda^{-1} z(\lambda)(z(\lambda) + \Delta_{X,p}^{\natural})^{-1}, \quad \operatorname{Re} \lambda > 0 \text{ suff. large.}$$

Taking into account Theorem 1.4.10 and the fact that the operator $-\Delta_{X,p}^{\natural}$ generates an analytic strongly continuous semigroup on X, we may conclude from the above that $\lambda^{-1}\mathbf{P}_{\lambda}^{-1} \in LT - L(X)$. Since $\lambda^{-1}z(\lambda)I \in LT - L(X)$ (cf. the proof of Theorem 2.10.9), we can apply the resolvent equation and Theorem 1.4.10 in order to see that

$$\lambda^{-2\alpha-1}\Delta_{X,p}^{\natural} z(\lambda)(z(\lambda) + \Delta_{X,p}^{\natural})^{-1} \in LT - L(X),$$

$$\lambda^{-\alpha-1}(\operatorname{Re}_b)^{-1}\Delta_{X,p}^{\natural} z(\lambda)(z(\lambda) + \Delta_{X,p}^{\natural})^{-1} \in LT - L(X)$$

and

$$\lambda^{-1}(1+a_0^2\Delta_{X,p}^{\natural})z(\lambda)(z(\lambda)+\Delta_{X,p}^{\natural})^{-1} \in LT - L(X).$$

By Theorem 2.3.33(i)/(b), we have that there exists an exponentially bounded I-existence family $(E(t))_{t\geq 0}$ for (254). It is not difficult to see with the help of Theorem 2.3.33(ii) that $(E(t))_{t\geq 0}$ is likewise an exponentially bounded I-uniqueness family for (254), so that the strong solutions of (254) are unique. Furthermore, we have that $\mathbf{D}_i = D(\Delta_{X,p}^{\natural})$ for i = 0, 1. Let $f: int(P_p) \to X \setminus \{0\}$ be an analytic mapping satisfying that $\Delta_{X,p}^{\natural}f(\lambda) = \lambda f(\lambda), \lambda \in int(P_p)$. Using the proof of Theorem 2.10.9, we get that the function $t \mapsto u(t; (f(\lambda), f(\lambda'))), t \geq 0$, given by

$$u(t; (f(\lambda), f(\lambda'))) := H_0(\lambda, t)f(\lambda) + H_1(\lambda', t)f(\lambda'), \quad t \ge 0 \quad (\lambda, \lambda' \in int(P_p)),$$

where

$$H_0(\lambda, t) := \mathcal{L}^{-1} \Big(\frac{z^{2\alpha - 1} - (\operatorname{Re}_b)^{-1} (\lambda - a_0^2)^{-1} z^{\alpha - 1}}{z^{2\alpha} - (\operatorname{Re}_b)^{-1} (\lambda - a_0^2)^{-1} z^{\alpha} - (\lambda - a_0^2)^{-1}} \Big)(t), \quad t \ge 0,$$

and

$$H_1(\lambda',t) := \mathcal{L}^{-1} \Big(\frac{z^{2\alpha-1}}{z^{2\alpha} - (\operatorname{Re}_b)^{-1} (\lambda' - a_0^2)^{-1} z^{\alpha} - (\lambda' - a_0^2)^{-1}} \Big)(t), \quad t \ge 0,$$

is a unique strong solution of (254) with $u(0, \cdot) = f(\lambda)$ and $u_t(0, \cdot) = f(\lambda')$. Direct computations show that

$$H_0(\lambda, t) = \frac{r_1(\lambda) - (\operatorname{Re}_b)^{-1}(\lambda - a_0^2)^{-1}}{\sqrt{D_\lambda}} e^{r_1(\lambda)t} - \frac{r_2(\lambda) - (\operatorname{Re}_b)^{-1}(\lambda - a_0^2)^{-1}}{\sqrt{D_\lambda}} e^{r_2(\lambda)t}$$

if $\alpha = 1$,

$$\begin{split} H_0(\lambda,t) &= \frac{t^{-\alpha}}{\sqrt{D_\lambda}} [E_{\alpha,1-\alpha}(r_1(\lambda)t^{\alpha}) - E_{\alpha,1-\alpha}(r_2(\lambda)t^{\alpha})] \\ &- \frac{(\operatorname{Re}_b)^{-1}(\lambda - a_0^2)^{-1}}{\sqrt{D_\lambda}} [E_\alpha(r_1(\lambda)t^{\alpha}) - E_\alpha(r_2(\lambda)t^{\alpha})], \quad t > 0, \end{split}$$

if $0 < \alpha < 1$, and

$$H_1(\lambda',t) = \frac{t^{1-\alpha}}{\sqrt{D_\lambda}} [E_{\alpha,2-\alpha}(r_1(\lambda')t^\alpha) - E_{\alpha,2-\alpha}(r_2(\lambda')t^\alpha)], \quad t > 0,$$

where

$$r_{1,2}(\lambda) := \frac{(\operatorname{Re}_b)^{-1}(\lambda - a_0^2)^{-1} \pm \sqrt{(\operatorname{Re}_b)^{-2}(\lambda - a_0^2)^{-2} + 4(\lambda - a_0^2)^{-1}}}{2}$$

and

$$D_{\lambda} := (\operatorname{Re}_{b})^{-2} (\lambda - a_{0}^{2})^{-2} + 4(\lambda - a_{0}^{2})^{-1}.$$

Set $F := [D(\Delta_{X,p}^{\natural})] \times [D(\Delta_{X,p}^{\natural})]$. Then F is a separable infinite-dimensional complex Banach space. Define V(t) as in the formulation of Theorem 2.11.23, with $C_1 = I$; then it is clear that $V(t) \in L(F, X)$ for all $t \ge 0$. From the uniqueness of strong solutions of (254), it readily follows that $V(t)(f(\lambda), f(\lambda')) = u(t; (f(\lambda), f(\lambda')))$, $t \ge 0$ $(\lambda, \lambda' \in int(P_p))$. Using the asymptotic expansion formulae (40)–(42) and a

simple analysis, we obtain that there exist a sufficiently small number $\varepsilon > 0$ and two sufficiently large negative numbers $x_{-} < 0$ and $x'_{-} < 0$ such that the first requirement in Theorem 2.11.23 holds with $F_0 = span\{(1 + \Delta_{X,p}^{\natural})^{-1}f(\lambda) : \lambda \in L(x_{-},\varepsilon)\} \times span\{(1 + \Delta_{X,p}^{\natural})^{-1}f(\lambda') : \lambda' \in L(x'_{-},\varepsilon)\}.$

It is clear that there exists a great number of concrete situations (consider, for example, the case in which $a_0 \to 0+$ and $\operatorname{Re}_b \to +\infty$) in which there exists a number $\lambda_0 \in int(P_p)$ such that

$$r_1(\lambda_0) \in \Sigma_{\gamma \pi/2}.$$

If this is the case, the vector $(f(\lambda_0), 0)$ is distributionally unbounded and the problem (254) is densely distributionally chaotic; observe also that the problem (254) is densely $(X \times \{0\})$ -distributionally chaotic and $(\{0\} \times X)$ -distributionally chaotic by Theorem 2.11.23. The same holds for the problem (254)' obtained by interchanging the terms $(\operatorname{Re}_b)^{-1}\Delta_{X,p}^{\natural}\mathbf{D}_t^{\alpha}u(t,x)$ and $(\operatorname{Re}_b)^{-1}\mathbf{D}_t^{\alpha}\Delta_{X,p}^{\natural}u(t,x)$ in (254); this follows directly from Definition 2.11.9 and the fact that the mapping $t \mapsto u(t; (f(\lambda), f(\lambda')))$, $t \ge 0$, defined above, is still a strong solution of (254)' for $\lambda, \lambda' \in int(P_p)$. Observe, finally, that all established conclusions for the problems (254) and (254)' continue to hold if we replace the operator $\Delta_{X,p}^{\natural}$ and the state space X in our analysis with the operator $(\Delta_{X,p}^{\natural})_{\infty}$ and the Fréchet space $[D_{\infty}(\Delta_{X,p}^{\natural})]$, respectively.

In the previous example, we have employed some ideas contained in the proof of Theorem 2.10.9. Assuming that the requirements of this theorem hold, we can pose the problem of existence of a closed linear subspace X' of X^{m_n} , an integer $k \in \mathbb{N}$ and a tuple $\vec{\beta} \in [0, \alpha_n]^k$ such that the problem (224) is (densely) $(X', \vec{\beta})$ distributionally chaotic; a similar question can be posed for the problem (250). We close this section with the observation that, in the present situation, we can give only some partial answers to the problems addressed above by using Theorem 2.11.8 as an essential tool in the consideration. For more details about various generalizations of distributional chaos and applications to the abstract PDEs, the reader may consult [**294**].

2.12. Appendix and notes

In his fundamental paper [507] (1994), G. A. Sviridyuk succeeded in developing the phase space method for the Sobolev equations of first order and some classes of incomplete higher-order Sobolev equations. The notions of a relatively σ -bounded operator, a relatively *p*-sectorial operator, a chain of associated vectors and a group of solution operators have been introduced in the Banach space setting. Concerning applications, mention should be made of the author's analysis of abstract linearized Oskolkov system [442, 443], which models the dynamics of an incompressible viscoelastic fluid of Kelvin-Voigt type. We refer the reader to the monograph [509] for a comprehensive survey of results of G. A. Sviridyuk and his students obtained in the period 1994–2003. Here, we will only insribe the most important results established in the papers [515] by G. A. Sviridyuk–A. A. Zamyshlyaeva and [204] by V. E. Fedorov, and say a few words about optimal control problems of abstract degenerate differential equations.

In [515], the authors have investigated the phase space of homogeneous linear higher-order equation

(255)
$$Au^{(n)}(t) = B_{n-1}u^{(n-1)}(t) + \dots + B_0u(t), \quad t \in \mathbb{R},$$

where $n \in \mathbb{N} \setminus \{1\}$, A and B_{n-1}, \ldots, B_0 are closed linear operators acting between the Banach spaces X and Y. Let us recall that a set $\mathcal{P} \subseteq X$ is said to be the phase space of equation (255) iff the following two conditions are satisfied:

- (i) If u(t) is a strong solution of (255), then $u(t) \in \mathcal{P}$ for all $t \ge 0$;
- (ii) For any $x_k \in \mathcal{P}$ $(0 \leq k \leq n-1)$, there exists a unique strong solution of equation (255) with the initial conditions $u^{(k)}(0) = x_k$ $(0 \leq k \leq n-1)$.

Set $\vec{B} := (B_{n-1}, \ldots, B_0)$. Then we say that the operator pencil \vec{B} is polynomially A-bounded iff there exists a finite number a > 0 such that $(\lambda^n A - \lambda^{n-1} B_{n-1} - \cdots - B_0)^{-1} \in L(Y, X)$ for all $\lambda \in \mathbb{C}$ with $|\lambda| > a$ (cf. [161, 225, 248, 408, 426] and [463] for further information concerning operator pencils). If the condition [515, (A), p. 271] holds and the operator pencil \vec{B} is polynomially A-bounded, then the point ∞ can be a removable singular point of the A-resolvent of the pencil \vec{B} , a pole of order $p \in \mathbb{N}$ of the A-resolvent of the pencil \vec{B} and, finally, an essentially singular point of the A-resolvent of the pencil \vec{B} . In the first two cases, the phase space of equation (255) coincides with the range of projection

$$P := \frac{1}{2\pi i} \int_{\gamma} \lambda^{n-1} (\lambda^n A - \lambda^{n-1} B_{n-1} - \dots - B_0)^{-1} A \, d\lambda,$$

where $\gamma = \{\lambda \in \mathbb{C} : |\lambda| = R > a\}$; cf. [515, Definition 3.2, Theorem 3.1]. If the point ∞ is an essentially singular point of the *A*-resolvent of the pencil \vec{B} , then the strong solutions of equation (255) need not be unique and the phase space \mathcal{P} of equation (255) is very difficult to be profiled. The main result of paper is [515, Theorem 4.1], in which the authors proved the existence and uniqueness of strong solutions of equation (255) provided that the point ∞ is not an essentially singular point of the *A*-resolvent of the pencil \vec{B} , and the above-mentioned condition (A) holds. Applications have been made to the abstract Boussinesq-Love equation in finite domains, which has been already considered in Example 2.3.48 and Example 2.3.53 in the case that the condition (A) is violated.

In [204], the problem of existence of an exponentially bounded solution semigroup strongly holomorphic in a sector is studied for a Sobolev-type linear equation Lu'(t) = Mu(t), where $L \in L(X, Y)$ is non-injective operator and M is a closed densely defined operator acting between the sequentially complete locally convex spaces X and Y. The author has introduced the notion of a (L, p)-sectoriality of the operator M as follows: The operator M is said to be (L, p)-sectorial iff the following holds:

(i) there exist $\omega \in \mathbb{R}$ and $\theta \in (0, \pi)$ such that $\omega + \Sigma_{\theta} \subseteq \rho^{L}(M)$;

(ii) the operator families

$$\left\{R_{(\lambda,p)}^{L}(M)\prod_{k=0}^{p}(\lambda_{k}-a):\lambda=(\lambda_{0},\ldots,\lambda_{p})\in(\omega+\Sigma_{\theta})^{p+1}\right\}\subseteq L(X)$$

and

$$\left\{L_{(\lambda,p)}^{L}(M)\prod_{k=0}^{p}(\lambda_{k}-a):\lambda=(\lambda_{0},\ldots,\lambda_{p})\in(\omega+\Sigma_{\theta})^{p+1}\right\}\subseteq L(Y)$$

are equicontinuous.

Here, for any $p \in \mathbb{N}_0$ and $\lambda = (\lambda_0, \dots, \lambda_p) \in (\rho^L(M))^{p+1}$,

$$R_{(\lambda,p)}^{L}(M) = \prod_{k=0}^{p} R_{\lambda}^{L}(M) \text{ and } L_{(\lambda,p)}^{L}(M) = \prod_{k=0}^{p} L_{\lambda}^{L}(M).$$

After that, the author introduces the notion of a strong (L, p)-sectoriality of the operator M from the right (left) and proves the existence of limit $\lim_{t\to 0+} U(t)$ $(\lim_{t\to 0+} V(t))$ in L(X) (L(Y)) for a strongly holomorphic solution semigroup $(U(t))_{t>0} \subseteq L(X)$ $((V(t))_{t>0} \subseteq L(Y))$ in this case; cf. [204, Theorem 2.1]. Under the general assumption on (L, p)-sectoriality of operator M, semireflexivity of spaces X and Y plays an important role since it allows one to decompose the spaces X and Y in a certain way; observe, however, that we do not need this assumption in the case that the operator M is a strongly (L, p)-sectorial from the right (left); cf. [204, Theorem 5.1, Proposition 5.1] for more details. An interesting application has been obtained in the analysis of periodic solutions of certain degenerate Cauchy problems of first order [204, Theorem 9.2].

Concerning control of abstract degenerate differential equations with integer order derivatives, the reader may consult the recent monograph [451] by M. V. Plekhanova and V. E. Fedorov. Applications have been given in the analysis of optimal control of Benny–Luke equation, equation of transition processes in semiconductors

$$\begin{aligned} (\lambda - \Delta)\omega_t(x, t) &= \alpha \omega(x, t) + y(x, t), \quad (x, t) \in \Omega \times (0, T), \\ \omega(x, t) &= 0, \quad (x, t) \in \partial\Omega \times (0, T), \\ \omega(x, 0) &= u(x), \quad x \in \Omega, \end{aligned}$$

the systems of linearized quasi-stationary phase-field equations

$$\begin{aligned} z(x,0) &= z_0(x), \quad x \in \Omega, \\ \nu \frac{\partial}{\partial n} z(x,t) + (1-\nu)z(x,t) &= 0, \quad (x,t) \in \partial\Omega \times (0,T), \\ \nu \frac{\partial}{\partial n} \Theta(x,t) + (1-\nu)\Theta(x,t) &= 0, \quad (x,t) \in \partial\Omega \times (0,T), \\ \frac{\partial}{\partial t} z(x,t) &= \Delta z(x,t) - \Delta\Theta(x,t) + f(x,t), \quad (x,t) \in \Omega \times (0,T), \\ \Delta\Theta(x,t) - \beta\Theta(x,t) + z(x,t) + g(x,t), \quad (x,t) \in \Omega \times (0,T), \end{aligned}$$

the systems of linearized systems of Boussinesq equations

$$\begin{split} z(s,0) &= z_0(s), \quad r(s,0) = r_0(s), \quad \Theta(s,0) = \Theta_0(s), \qquad s \in \Omega, \\ z(s,t) &= 0, \quad \Theta(s,t) = 0, \qquad (s,t) \in \partial\Omega \times (0,T), \\ \frac{\partial z(s,t)}{\partial t} &= \nu \Delta z(s,t) - r(s,t) - \alpha \Theta(s,t) e_3 + u(s,t), \quad (s,t) \in \Omega \times (0,T), \\ \nabla \cdot z &= 0, \quad (s,t) \in \Omega \times (0,T), \\ \frac{\partial \Theta(s,t)}{\partial t} &= \beta \Delta \Theta(s,t) + z_3(s,t) + \omega(s,t), \quad (s,t) \in \Omega \times (0,T), \end{split}$$

as well as the systems of linearized Oskolkov and Sobolev equations. Concerning control of abstract fractional degenerate equations, we have been able to locate only one research paper [450], written by M. V. Plekhanova.

Now we would like to introduce the notion of an (a, k)-regularized *C*-resolvent family of solving operators and the phase space of certain classes of abstract degenerate Volterra integral equations (cf. also Section 2.8). We assume that the closed graph theorem holds for mappings from Y into Y, and for mappings from Y into X.

Consider the following abstract degenerate Cauchy problem (cf. (52)):

(256)
$$Lu(t) = f(t) + \int_0^t a(t-s)Mu(s)ds, \quad t \in [0,\tau)$$

where $t \mapsto f(t) \in Y$, $t \in [0, \tau)$ is a continuous mapping and $a \in L^1_{loc}([0, \tau))$, $a \neq 0$. Let $\lambda \in \rho^L(M)$. Then, along with the equation (256), we consider the equation

(257)
$$(\lambda L - M)^{-1} Lu(t) = (\lambda L - M)^{-1} f(t) + \int_0^t a(t-s)(\lambda L - M)^{-1} Mu(s) ds, \quad t \in [0,\tau),$$

and, after the substitution $v(t) = (\lambda L - M)u(t), t \in [0, \tau)$, the equation

(258)
$$L(\lambda L - M)^{-1}v(t) = f(t) + \int_0^t a(t-s)M(\lambda L - M)^{-1}v(s)ds, \quad t \in [0,\tau).$$

Both of these equations can be rewritten in the form of abstract degenerate Cauchy problem (52):

- (257): Take $A = (\lambda L M)^{-1}M$, $B = (\lambda L M)^{-1}L$ and $g(t) = (\lambda L M)^{-1}f(t)$. Then $B \in L(X)$, A is a linear operator acting on X but A is not closed, in general. This obviously hinders the analysis of problem (257).
- (258): Take $A = M(\lambda L M)^{-1}$, $B = L(\lambda L M)^{-1}$ and g(t) = f(t). Then the situation is much simpler because $A \in L(Y)$ and $B \in L(Y)$.

For ease the presentation, we need to introduce the following technical definition: Let X and Y be SCLCSs, let $0 < \tau \leq \infty$, and let $A: D(A) \subseteq X \to Y$ and $B: D(B) \subseteq X \to Y$ be linear (not necessarily closed) operators. Consider the following abstract degenerate Cauchy problem (52). Then a function $u \in C([0,\tau): X)$ is said to be a (mild) solution of (52) iff $(a * u)(t) \in D(A)$, $t \in [0, \tau), \ A(a * u)(t) = Bu(t) - f(t), \ t \in [0, \tau)$ and the mapping $t \mapsto Bu(t), t \in [0, \tau)$ is continuous.

Put Z := X, in the case of examination of problem (257), and Z := Y, in the case of examination of problem (258). In order to explain our next steps, suppose that any of these two problems is reduced to the problem (52), and that the single-valued operator $B^{-1}A$ generates a (local) (a, k)-regularized *C*-resolvent family $(R(t))_{t \in [0,\tau)} \subseteq L(Z)$, obeying the property [**292**, (22)] with *A* and *x* replaced therein with $B^{-1}A$ and *u*, respectively. Then we have

$$R(t)u = k(t)Cu + B^{-1}A \int_0^t a(t-s)R(s)u\,ds, \quad t \in [0,\tau), \ u \in Z, \text{ i.e.}$$
$$BR(t)u = k(t)BCu + A \int_0^t a(t-s)R(s)u\,ds, \quad t \in [0,\tau), \ u \in Z,$$

so that for each $u \in Z$ the function $t \mapsto R(t)u$, $t \in [0, \tau)$ is a mild solution of (52) with f(t) = k(t)BCu, $t \in [0, \tau)$. For the sake of simplicity and better understanding, in the remainder of this analysis we will consider only the global case $\tau = \infty$. Our results from [**292**, Section 2.8] motivate us to introduce the following definition (cf. [**509**, Definitions 2.3.1, 2.8.2, 3.2.1 and 4.4.1] for the semigroup case $(a(t) \equiv 1))$:

DEFINITION 2.12.1. Let Z be an SCLCS, let $B \in L(Z)$, and let $A: D(A) \subseteq Z \to Z$ be a linear operator acting on Z. Suppose that $C \in L(Z)$ is injective, $a \in L^1_{loc}([0,\infty)), a \neq 0$ and $k \in C([0,\infty))$. Set $\mathcal{G} := \{f \in C([0,\infty): Z) : (\exists w \in Z) f(t) = k(t)Bw \text{ for all } t \geq 0\}.$

- (i) By a (global) (a, k)-regularized C-resolvent family of solving operators for the problem (52), with $f \in \mathcal{G}$, we mean any strongly continuous operator family $(R(t))_{t \ge 0} \subseteq L(Z)$ satisfying the following two conditions:
 - (a) $R(s)(a*R)(t)u-(a*R)(s)R(t)u = k(s)(a*R)(t)Cu-k(t)(a*R)(s)Cu, t, s \ge 0, u \in \mathbb{Z}.$
 - (b) For each $u \in Z$ the function $t \mapsto R(t)u, t \ge 0$ is a mild solution of problem (52) with $f(t) = k(t)BCu, t \ge 0$.

We say that $(R(t))_{t\geq 0}$ is exponentially equicontinuous (locally equicontinuous) iff there exists $\omega \geq 0$ such that the operator family $\{e^{-\omega t}R(t): t\geq 0\} \subseteq L(Z)$ is equicontinuous (iff for each T>0 the operator family $\{R(t): t\in [0,T]\} \subseteq L(Z)$ is equicontinuous).

(ii) It is said that a global (a, k)-regularized C-resolvent family (R(t))_{t≥0} of solving operators for the problem (52), with f ∈ G, is analytic of angle θ ∈ (0, π], iff there exists a function R: Σ_θ → L(Z) which satisfies that, for every u ∈ Z, the mapping z → R(z)u, z ∈ Σ_θ is analytic as well as that R(t) = R(t), t > 0. It is said that (R(t))_{t≥0} is an exponentially equicontinuous, analytic (a, k)-regularized C-resolvent family of solving operators for the problem (52), with f ∈ G, of angle θ, resp. equicontinuous analytic (a, k)-regularized C-resolvent family of solving operators for the problem (52), with f ∈ G, of angle θ, resp. equicontinuous analytic (2, k)-regularized C-resolvent family of solving operators for the problem (52), with f ∈ G, of angle θ, iff for every γ ∈ (0, θ), there exists ω_γ ≥ 0,

resp. $\omega_{\gamma} = 0$, such that the set $\{e^{-\omega_{\gamma} \operatorname{Re} z} \mathbf{R}(z) : z \in \Sigma_{\gamma}\} \subseteq L(Z)$ is equicontinuous.

(iii) It is said that a global (a, k)-regularized *C*-resolvent family $(R(t))_{t\geq 0}$ of solving operators for the problem (52), with $f \in \mathcal{G}$, is entire iff there exists a function $\mathbf{R} \colon \mathbb{C} \to L(Z)$ which satisfies that, for every $u \in Z$, the mapping $z \mapsto \mathbf{R}(z)u, z \in \mathbb{C}$ is entire (since there is no risk for confusion, we will identify $R(\cdot)$ and $\mathbf{R}(\cdot)$ in the sequel).

Observe that the notion of an analytic (a, k)-regularized *C*-resolvent family of solving operators for the problem (52), with $f \in \mathcal{G}$, is introduced in [**204**, Definition 2.1] in a slightly different manner, provided that $a(t) \equiv 1$, $k(t) \equiv 1$ and C = I. It is also worth noting that we do not require the validity of limit equality $\lim_{z\to 0, z\in\Sigma_{\gamma}} \mathbf{R}(z)u = R(0)u \ (\gamma \in (0, \theta), u \in Z)$ in (b).

The notion of phase space of problem (52) is introduced in the following definition.

DEFINITION 2.12.2. Let $B \in L(Z)$, let A be a linear operator on Z, and let $\tau = \infty$. Then a set $\mathcal{P} \subseteq Z$ is called the phase space of equation (52), with $f \in \mathcal{G}$, iff the following holds:

- (i) any mild solution $t \mapsto u(t), t \ge 0$ of (52), with f(t) = k(t)Bw $(t \ge 0)$ belongs to \mathcal{P} , that is, $u(t) \in \mathcal{P}, t \ge 0$;
- (ii) for any $w \in \mathcal{P}$ there exists a unique mild solution $t \mapsto u(t), t \ge 0$ of (52), with $f(t) = k(t)Bw, t \ge 0$.

It is very interesting and unpleasant question to profile the phase space of Eq. (52). In connection with this problem, we would like to mention that the method proposed in [515] does not work for abstract degenerate multi-term fractional problems with Caputo derivatives. We have found the problem of characterizing the phase space of such equations (defined similarly as for higher-order Cauchy problems with integer order derivatives) very serious and difficult to be answered, as well.

We continue by stating the following generalization of [204, Theorem 2.1].

THEOREM 2.12.3. Suppose that the operator M is (L, p)-sectorial and $0 < \alpha < 2\theta\pi^{-1}$. Let the space Z and the operators A, B be defined as in (257), resp. (258), and let the class \mathcal{G} be defined as in Definition 2.12.1, with $k(t) \equiv 1$. Then there exists an exponentially equicontinuous, analytic (g_{α}, I) -regularized resolvent family $(R(t))_{t\geq 0}$ for (52), with $a(t) = g_{\alpha}(t)$ and $f \in \mathcal{G}$, of angle $\min(\theta\alpha^{-1} - \pi 2^{-1}, \pi 2^{-1})$.

PROOF. Without loss of generality, we may assume that the operator M is (L, p)-sectorial with the constant $\omega \ge 0$. Let $0 < \varepsilon < \gamma' < \gamma$ and $\delta > 0$. Then there exists a sufficiently large number $\omega' > \omega$ such that the operator families $\{(\lambda - \omega')\lambda^{\alpha-1}R_{\lambda\alpha}^L(M) : \lambda \in \omega' + \Sigma_{\frac{\pi}{2}+\gamma'}\} \subseteq L(Z)$ and $\{(\lambda - \omega')\lambda^{\alpha-1}L_{\lambda\alpha}^L(M) : \lambda \in \omega' + \Sigma_{\frac{\pi}{2}+\gamma'}\} \subseteq L(Z)$ and $\{(\lambda - \omega')\lambda^{\alpha-1}L_{\lambda\alpha}^L(M) : \lambda \in \omega' + \Sigma_{\frac{\pi}{2}+\gamma'}\} \subseteq L(Z)$ are equicontinuous. Let the oriented contour Γ be as in the second part of proof of [27, Theorem 2.6.1], with the numbers ω and γ replaced therein with the numbers ω' and γ' , respectively. Denote by Ω the open region on the right of Γ . Then the mappings $\lambda \mapsto \lambda^{\alpha-1}R_{\lambda\alpha}^L(M)$, $\lambda \in \Omega$ and

 $\lambda \mapsto \lambda^{\alpha-1} L^L_{\lambda^{\alpha}}(M), \ \lambda \in \Omega$ are strongly analytic (and strongly continuous on $\overline{\Omega}$). Define, for every $u \in Z$ and $z \in \Sigma_{\gamma'-\varepsilon}$,

$$R(z)u:=\frac{1}{2\pi i}\int_{\Gamma}e^{\lambda z}\lambda^{\alpha-1}R^L_{\lambda^\alpha}(M)u\,d\lambda,$$

in the case of consideration of problem (257), and

$$R(z)u := \frac{1}{2\pi i} \int_{\Gamma} e^{\lambda z} \lambda^{\alpha - 1} L_{\lambda^{\alpha}}^{L}(M) u \, d\lambda$$

in the case of consideration of problem (258). Then the proof of [27, Theorem 2.6.1] shows that for each $p \in \circledast_Z$ there exist $q \in \circledast_Z$ and c > 0 such that $p(R(z)u) \leq cq(u), u \in Z, z \in \Sigma_{\gamma'-\varepsilon}$, as well as that $\mathcal{L}(R(t)u)(\lambda) = \lambda^{\alpha-1}R_{\lambda^{\alpha}}^L(M)u, u \in Z$, in the case of consideration of problem (257), and $\mathcal{L}(R(t)x)(\lambda) = \lambda^{\alpha-1}L_{\lambda^{\alpha}}^L(M)u, u \in Z$, in the case of consideration of problem (258). The proof of (b) follows similarly as in the proof of [204, Theorem 2.1], while the equality (a) from Definition 2.12.1 follows by taking the Laplace transform twice, with respect to variables t and s separately, and using the resolvent equations. This completes the proof of theorem.

If the conclusions of Theorem 2.12.3 hold, then for each $\varepsilon \in (0, \gamma)$ there exists $\omega_{\varepsilon} \geq \omega$ such that the set $\{z^k e^{-\omega z} R^{(k)}(z) : z \in \Sigma_{\gamma-\varepsilon, k \in \mathbb{N}_0}\}$ is bounded; cf. [27, p. 86] and [296, Theorem 3.4(i)]. Unfortunately, Theorem 2.12.3 is only a partial result and it is an open problem to reconsider the remaining results from [204] in fractional case.

Stochastic degenerate Cauchy problems. In this monograph, we analyze only degenerate deterministic Volterra integro-differential equations. Stochastic Cauchy problems and stochastic differential equations occupy a great deal of mathematical territory nowdays; for further information concerning degenerate case, we recommend for the reader [98, 194, 511, 552] and [554].

Semilinear degenerate Cauchy problems. In Subsection 2.2.5, we have considered semilinear degenerate fractional Cauchy problems with abstract differential operators. For more details about abstract semilinear degenerate Cauchy problems with integer order derivatives, we refer the reader to [59,83,89,94,192, 208, 209, 267, 268, 319, 468, 469, 505] and [508]. Semilinear degenerate Cauchy inclusions will be investigated in Subsection 3.5.1-Subsection 3.5.2.

Nonlinear degenerate Cauchy problems. In this monograph, we are focused entirely on the linear theory of abstract degenerate Volterra integro-differential equations. Concerning quasilinear and purely nonlinear degenerate equations of Sobolev type, mention should be made of monographs A. Favini, G. Marinoschi [190, 491]by R. Showalter, [503] by A. G. Sveshnikov, A. B. Al'shin, M. O. Korpusov, Yu. D. Pletner, [19] by A. B. Al'shin, M. O. Korpusov, A. G. Sveshnikov, [216] by V. E. Fedorov, A. V. Nagumanova, as well as the papers [88, 193, 201, 368, 369, 400, 443, 486, 492, 493] and [525]. Nonlinear Sobolev equations arise naturally in many problems in various fields of mathematical physics (cf. [19] for further information); here are some examples: (a) The Camassa–Holm equation (1993)

$$u_t - u_{xxt} + 3uu_x = 2u_x u_{xx} + uu_{xxx}$$

is important in modeling of undirectional propagation of shalow-water waves over a flat bottom.

(b) The Benjamin–Bona–Mahony equation (or BBM equation)

$$u_t + u_x + uu_x - u_{xxt} = 0.$$

is also known as the regularized long-wave equation (RLWE). This equation was investigated by T. B. Benjamin, J. L. Bona and J. J. Mahony (1972) as an improvement of the Korteweg–de Vries equation (KdV equation) for modeling long surface gravity waves of small amplitude. Strictly speaking, BBM equation was introduced by D. H. Peregrine, in 1966, for the purpose of study of undular bores.

(c) The Rosenau–Burgers equation

$$u_t + u_{xxxxt} - \alpha u_{xx} + (u^{p+1}/p + 1)_x = 0 \quad (p > 0)$$

is important in modeling of bore propagation. In the case that $\alpha = 0$, the equation was introduced by P. Rosenau (1988) for treating the dynamics of dense discrete systems in order to overcome the shortenings by the KdV equation.

(d) The semiconductor equation (2005)

$$\frac{\partial}{\partial t}(\Delta u - u) + \Delta u + \alpha u^3 = 0, \quad \alpha \in \mathbb{R}$$

models nonstationary processes in crystalline semiconductors.

(e) The Longern wave equation (1975)

- 0

$$\frac{\partial^2}{\partial t^2}(u_{xx} - \alpha u + \beta u^2) + u_{xx} = 0 \quad (\alpha > 0, \ \beta > 0)$$

describes electric signals in telegraph lines.

We also want to outline that the pioneering works of L. A. Lyusternik, S. G. Krein, N. A. Sidorov and B. V. Loginov (cf. [496, 497] for further information), in which the authors had analyzed bifurcating solutions of nonlinear parameter-dependent singular and degenerate differential equations, influenced many authors to investigate mathematical modelings based on the Showalter-Sidorov problem (cf. [206, 214, 272, 511, 514, 551] and references cited therein for further information).

Abstract degenerate integro-differential equations with delay. The analysis of abstract degenerate integro-differential equations with delay has received much attention recently. For the basic references in this direction, we refer the reader to L. V. Borel, V. E. Fedorov [86], C. Bu [91], A. Favini, L. Vlasenko [198], V. E. Fedorov, E. A. Omel'chenko [217, 218], C. Lizama, R. Ponce [397], and to the doctoral dissertation of R. A. A. Cuello [123]. Concerning non-degenerate case, the monographs by S. Abbas, M. Benchohra [4], whose twelfth chapter also contains some information on functional evolution inclusions, N. V. Azbelev, V. P.

Maksimov, L. F. Rakhmatullina [39], J. K. Hale, S. M. Verduyn Lunel [246], Y. Kuang [362] and J. Wu [540] are of crucial importance.

In his doctoral dissertation [123], R. A. A. Cuello has investigated the abstract, degenerate, functional second order differential equations in the vectorvalued Lebesgue, Besov and Triebel–Lizorkin function spaces. Special focus is placed on following degenerate functional differential equations

$$\begin{split} (Mu')'(t) &-\Lambda u'(t) - \frac{d}{dt} \int_{-\infty}^t c(t-s)u(s)ds \\ &= \gamma_\infty u(t) + Au(t) + \frac{d}{dt} \int_{-\infty}^t a(t-s)Au(s)ds \\ &+ b_\infty Bu(t) + \int_{-\infty}^t b(t-s)Bu(s)ds + f(t), \quad t \in [0, 2\pi], \end{split}$$

equipped with periodic boundary conditions $u(0) = u(2\pi)$, $(Mu')(0) = (Mu')(2\pi)$, and

$$(Mu)''(t) - \Lambda u'(t) - \frac{d}{dt} \int_{-\infty}^{t} c(t-s)u(s)ds$$

= $\gamma_{\infty}u(t) + Au(t) + \frac{d}{dt} \int_{-\infty}^{t} a(t-s)Au(s)ds$
+ $b_{\infty}Bu(t) + \int_{-\infty}^{t} b(t-s)Bu(s)ds + f(t), \quad t \in [0, 2\pi],$

equipped with periodic boundary conditions $\Lambda u(0) = \Lambda u(2\pi)$, $(Mu)(0) = (Mu)(2\pi)$ and $(Mu')(0) = (Mu')(2\pi)$. Here, A, B, Λ and M are closed linear operators on a complex Banach space E, satisfying that $D(A) \cap D(B) \subseteq D(\Lambda) \cap D(M)$, $a, b, c \in L^1([0, \infty))$, $f(\cdot)$ is an E-valued function defined on $[0, 2\pi]$, and $\gamma_{\infty}, b_{\infty}$ are certain complex constants. The author has applied abstract results in the analysis of the following abstract degenerate functional Cauchy problem

$$\begin{split} \frac{\partial}{\partial t}(m(x)u_t(x,t)) &- \Delta u_t(t,x) \\ &= \Delta u(t,x) + \int_{-\infty}^t a(t-s)\Delta u(s,x)ds + f(t,x), \quad (t,x) \in [0,2\pi] \times \Omega, \\ &u(t,x) = \left(\frac{\partial}{\partial t}u(t,x)\right)_{t=0} = 0, \quad (t,x) \in [0,2\pi] \times \partial\Omega, \\ &u(0,x) = u(2\pi,x), \ m(x) \left(\frac{\partial}{\partial t}u(t,x)\right)_{t=0} = m(x) \left(\frac{\partial}{\partial t}u(t,x)\right)_{t=2\pi}, \ x \in \Omega, \ t \in [0,2\pi] \end{split}$$

where Ω is an open domain in \mathbb{R}^n and Δ is the Dirichlet Laplacian (see also the paper [24] by R. A. A. Cuello and V. Keyantuo). The fractional analogues of above equations have not been analyzed by now.

Abstract degenerate Cauchy problems of first order: semigroup theory approach. In [517], B. Thaller and S. Thaller have continued the investigations of R. Showalter [494, 495] concerning the well-posedness of abstract degenerate Cauchy problems (DFP)_R and (DFP)_L of first order. The results of [517], formulated in the setting of Hilbert spaces, have been reconsidered and substantially improved by the authors in the paper [518], whose main ideas are described as follows.

Suppose that A and B are two linear operators acting between the Banach spaces X and Y. In [518, Section 2], the authors have assumed that there is a linear subspace $D_X \subseteq D(A) \cap D(B)$ such that for all $\lambda > \omega$, where $\omega \in \mathbb{R}$, the restriction of operator $A - \lambda B$ to D_X is invertible and with values in R(B). Assuming certain Hille–Yosida type conditions on the operators $(\lambda B - A)^{-1}B$ and $B(\lambda B - A)^{-1}$ ($\lambda > \omega$), the authors have applied the results from the theory of non-degenerate strongly continuous semigroups for proving the existence and uniqueness of strict solutions of problems (DFP)_R and (DFP)_L (cf. [518, Definition 1.1, Theorem 3.1–Theorem 3.2]). Using the notion of joint closedness of operators A and B [475], the authors have considered the so-called invariant subspaces for the time evolution in Section 4. In Section 5, it has been assumed that the operator B is closed. Then N(B)is a closed subspace in X and the authors have enquired into the possibility to consider the inital degenerate Cauchy problem by passing to the corresponding non-degenerate Cauchy problem in the Banach space X/N(B). Section 6 considers the case in which X and Y are Hilbert spaces.

Singular abstract Cauchy problems. In their fundamental monograph [95], R. Caroll and R. Showalter have primarily dealt with singular or degenerate equations of the following two forms:

(259)
$$A(t)u_{tt} + B(t)u_t + C(t)u = g, \quad t > 0$$

and

(260)
$$(A(\cdot)u)_{tt} + (B(\cdot)u)_t + C(\cdot)u = g, \quad t > 0,$$

where A(t), B(t) and C(t) are three families of possibly nonlinear differential operators acting on an SCLCS E. According to R. Caroll and R. Showalter, the equation (259) is called singular if at least one of the operator coefficients tends to infinity as $t \to 0+$, while the equation (259) is called degenerate if some of the operator coefficients tends to zero as $t \to 0+$; the same notions are accepted for the equation (260))=. Roughly speaking, if these equations can be solved with respect to the highest order derivative, then we call them of parabolic or hyperbolic type; if this is not the case, then they are of Sobolev type (as the reader can easily observe, in this monograph we do not distinguish, in some heuristical manner, Sobolev and degenerate type equations). There is an enormous literature on singular abstract Cauchy problems with integer order derivatives, especially on the famous Euler-Poisson-Darboux equation. Concerning purely fractional case, there exists only a few research papers containing very limited material about this important class of equations (cf.B. Baeumer, M. M. Meerschaert, E. Nane [41], V. Keyantuo, C. Lizama [275], M. M. Meerschaert, E. Nane, P. Vellaisamy [418] and E. Nane [432]; even for the above-mentioned Euler–Poisson–Darboux equation, there is no significant reference covering the fractional evolution case).

Abstract degenerate differential equations and empathy theory. In [476], N. Sauer has investigated the abstract Cauchy problem

$$\frac{d}{dt}(Bu(t)) = Au(t), \quad t > 0; \quad \lim_{t \to 0+} Bu(t) = y \in Y,$$

where $A: D \to Y$ and $B: D \to Y$ are two linear operators defined on a common domain **D** contained in a Banach space X and Y is another Banach space. For this purpose, he defines the notion of an empathy, that is any double family of evolution operators $\langle (S(t))_{t>0}, (E(t))_{t>0} \rangle$ satisfying the following properties:

- (i) $E(t) \in L(Y)$ and $S(t) \in L(Y, X)$ for all t > 0;
- (ii) S(t+s) = S(t)E(s) for arbitrary t, s > 0 (the empathy relation);
- (iii) There exists $\xi > 0$ such that $P(\xi)$ is invertible;
- (iv) For every $y \in Y$ and $\lambda > 0$, we have $e^{-\lambda \cdot} E(\cdot)y \in L^1((0,\infty) : Y)$ and $e^{-\lambda \cdot} S(\cdot)y \in L^1((0,\infty) : X)$.

It is well known that, for any empathy $\langle (S(t))_{t>0}, (E(t))_{t>0} \rangle$, the following holds:

- (a) $(E(t))_{t>0}$ is a semigroup;
- (b) $(S(t))_{t>0}$ and $(E(t))_{t>0}$ are strongly continuous and the norms ||S(t)||, ||E(t)|| are locally uniformly bounded on $(0, \infty)$.
- (c) For any number $\zeta > 0$, the operator $P(\zeta)$ is invertible.

Define $R(\lambda)y := \int_0^\infty e^{-\lambda t} E(t)y \, dt$ and $P(\lambda)y := \int_0^\infty e^{-\lambda t} E(t)y \, dt$ $(\lambda > 0, y \in Y)$. Then we have:

- (d) $R(\lambda) R(\mu) = (\mu \lambda)R(\lambda)R(\mu) = (\mu \lambda)R(\mu)R(\lambda)$ and $P(\lambda) P(\mu) = (\mu \lambda)P(\lambda)R(\mu) = (\mu \lambda)P(\mu)R(\lambda)$ for $\lambda, \mu > 0$.
- (e) $N(R(\lambda)) = N(R(\mu)) := N_E$, $R(R(\lambda)) = R(R(\mu)) := \mathbf{D}_E \subseteq Y$ and $R(P(\lambda)) = R(P(\mu)) := \mathbf{D} \subseteq Y$ for any $\lambda, \mu > 0$;
- (f) $S(t)[\mathbf{D}_E] \subseteq \mathbf{D}$ and $E(t)[\mathbf{D}_E] \subseteq \mathbf{D}$ for t > 0;
- (g) For any $y \in \mathbf{D}_E$, we have $\lim_{t\to 0} E(t)y = y$;
- (h) There exists a linear operator $C_0: \mathbf{D}_E \to \mathbf{D}$ such that $\lim_{t\to 0} S(t)y = C_0 y$ for any $y \in \mathbf{D}_E$;
- (i) For any $\lambda, \mu > 0$, we have that $A_{\lambda} = A_{\mu}$ and $B_{\lambda} = B_{\mu}$, where $A_{\lambda} := [\lambda R(\lambda) I]P^{-1}(\lambda) : \mathbf{D} \to Y$ and $B_{\lambda} := R(\lambda)P^{-1}(\lambda) : \mathbf{D} \to \mathbf{D}_{E}$.

Set $A := A_1$ and $B := B_1$. Then the pair $\langle A, B \rangle$ is said to be the generator of the empathy $\langle (S(t))_{t>0}, (E(t))_{t>0} \rangle$.

It is clear that applications of empathy theory to abstract degenerate Volterra integro-differential equations are very limited. Mention should be made of an intriguing application of [476, Theorem 8.2] to the following system of PDEs of first order

(261)
$$\rho(x)v_t(x,t) = f_x(x,t), \quad \sigma f_t(x,t) = v_x(x,t),$$

for which it is well-known that plays an important role in modeling of logitudinal vibrations in an elastic bar of length l and linear density $\rho = \rho(x), 0 \leq x \leq l$.

In (261), v denotes the velocity, f the internal force and $\sigma > 0$ the reciprocal of Hooke's constant.

For further information on empathy theory and its applications to abstract differential equations, we refer the reader to the recent doctoral dissertation of W.-S. Lee [374]. Up to now, we do not have any reference dealing with some fractional analogues of empathy. Before we move ourselves to the next chapter, we want also to recommend the monograph [473] by A. M. Samoilenko, M. I. Shkil' and V. P. Yakovets' for the basic results on linear degenerate systems of ordinary differential equations.

CHAPTER 3

MULTIVALUED LINEAR OPERATORS APPROACH

The main purpose of this chapter is to study various classes of abstract degenerate Volterra integro-differential equations by using the multivalued linear operators approach. Applying suitable substitutions, we put these equations into the form of abstract degenerate Volterra integro-differential inclusions with multivalued linear operators and examine after that the possibility of getting well-posedness results by making use of some known results and methods from the corresponding theory of non-degenerate equations (observe, however, that this approach has some drawbacks because we inevitably lose some valuable information on the well-posedness of initial problems). Despite of this, abstract degenerate Volterra integro-differential inclusions with multivalued linear operators deserve special attention and analysis.

We use the standard terminology throughout this chapter. By E, X, Y, \ldots we denote Hausdorff sequentially complete locally convex spaces over the field of complex numbers. By \circledast_X (\circledast , if there is no risk for confusion), we denote the fundamental system of seminorms which defines the topology of X. Any place where we need the pivot space to be a Banach (Fréchet) space will be especially emphasized.

3.1. Abstract degenerate Volterra integro-differential inclusions

Let $0 < \tau \leq \infty$, $\alpha > 0$, $a \in L^1_{loc}([0,\tau))$, $a \neq 0$, $\mathcal{F}: [0,\tau) \to P(Y)$, and let $\mathcal{A}: X \to P(Y)$, $\mathcal{B}: X \to P(Y)$ be two given mappings (possibly non-linear). Suppose that *B* is a single-valued linear operator. As mentioned in the introductory part, the main aim of this section is to analyze the following abstract degenerate Volterra inclusion:

(262)
$$\mathcal{B}u(t) \subseteq \mathcal{A} \int_0^t a(t-s)u(s)ds + \mathcal{F}(t), \quad t \in [0,\tau),$$

as well as the following fractional Sobolev inclusions:

$$(DFP)_{\mathbf{R}}: \begin{cases} \mathbf{D}_t^{\alpha} Bu(t) \in \mathcal{A}u(t) + \mathcal{F}(t), & t \ge 0, \\ (Bu)^{(j)}(0) = Bx_j, & 0 \le j \le \lceil \alpha \rceil - 1, \end{cases}$$

and

$$(\text{DFP})_{\mathbf{L}} : \begin{cases} \mathcal{B}\mathbf{D}_t^{\alpha}u(t) \subseteq \mathcal{A}u(t) + \mathcal{F}(t), & t \ge 0, \\ u^{(j)}(0) = x_j, & 0 \le j \le \lceil \alpha \rceil - 1. \end{cases}$$

In the following general definition, we introduce various types of solutions to the abstract degenerate inclusions (262), $(DFP)_{\mathbf{R}}$ and $(DFP)_{\mathbf{L}}$.

DEFINITION 3.1.1. (i) A function $u \in C([0,\tau) : X)$ is said to be a presolution of (262) iff $(a*u)(t) \in D(\mathcal{A})$ and $u(t) \in D(\mathcal{B})$ for $t \in [0,\tau)$, as well as (262) holds. By a solution of (262), we mean any pre-solution $u(\cdot)$ of (262) satisfying additionally that there exist functions $u_{\mathcal{B}} \in C([0,\tau) : Y)$ and $u_{a,\mathcal{A}} \in C([0,\tau) : Y)$ such that $u_{\mathcal{B}}(t) \in \mathcal{B}u(t)$ and $u_{a,\mathcal{A}}(t) \in \mathcal{A} \int_0^t a(t-s)u(s)ds$ for $t \in [0,\tau)$, as well as

$$u_{\mathcal{B}}(t) \in u_{a,\mathcal{A}}(t) + \mathcal{F}(t), \quad t \in [0,\tau).$$

Strong solution of (262) is any function $u \in C([0, \tau) : X)$ satisfying that there exist two continuous functions $u_{\mathcal{B}} \in C([0, \tau) : Y)$ and $u_{\mathcal{A}} \in C([0, \tau) : Y)$ such that $u_{\mathcal{B}}(t) \in \mathcal{B}u(t), u_{\mathcal{A}}(t) \in \mathcal{A}u(t)$ for all $t \in [0, \tau)$, and

$$u_{\mathcal{B}}(t) \in (a * u_{\mathcal{A}})(t) + \mathcal{F}(t), \quad t \in [0, \tau).$$

- (ii) Let $B = \mathcal{B}$ be single-valued. By a *p*-solution of $(DFP)_{\mathbf{R}}$, we mean any *X*-valued function $t \mapsto u(t), t \ge 0$ such that the term $t \mapsto \mathbf{D}_t^{\alpha} Bu(t), t \ge 0$ is well-defined, $u(t) \in D(\mathcal{A})$ for $t \ge 0$, and the requirements of $(DFP)_{\mathbf{R}}$ hold; a pre-solution of $(DFP)_{\mathbf{R}}$ is any *p*-solution of $(DFP)_{\mathbf{R}}$ that is continuous for $t \ge 0$. Finally, a solution of $(DFP)_{\mathbf{R}}$ is any pre-solution $u(\cdot)$ of $(DFP)_{\mathbf{R}}$ satisfying additionally that there exists a function $u_{\mathcal{A}} \in C([0,\infty) : Y)$ such that $u_{\mathcal{A}}(t) \in \mathcal{A}u(t)$ for $t \ge 0$, and $\mathbf{D}_t^{\alpha} Bu(t) \in u_{\mathcal{A}}(t) + \mathcal{F}(t), t \ge 0$.
- (iii) By a pre-solution of $(DFP)_{\mathbf{L}}$, we mean any continuous X-valued function $t \mapsto u(t), t \ge 0$ such that the term $t \mapsto \mathbf{D}_{t}^{\alpha}u(t), t \ge 0$ is well defined and continuous, as well as that $\mathbf{D}_{t}^{\alpha}u(t) \in D(\mathcal{B})$ and $u(t) \in D(\mathcal{A})$ for $t \ge 0$, and that the requirements of $(DFP)_{\mathbf{L}}$ hold; a solution of $(DFP)_{\mathbf{L}}$ is any presolution $u(\cdot)$ of $(DFP)_{\mathbf{L}}$ satisfying additionally that there exist functions $u_{\alpha,\mathcal{B}} \in C([0,\infty):Y)$ and $u_{\mathcal{A}} \in C([0,\infty):Y)$ such that $u_{\alpha,\mathcal{B}}(t) \in \mathcal{B}\mathbf{D}_{t}^{\alpha}u(t)$ and $u_{\mathcal{A}}(t) \in \mathcal{A}u(t)$ for $t \ge 0$, as well as that $u_{\alpha,\mathcal{B}}(t) \in u_{\mathcal{A}}(t) + \mathcal{F}(t), t \ge 0$.

Before proceeding further, we want to observe that the existence of solutions to (262), (DFP)_R or (DFP)_L immediately implies that $\sec_c(\mathcal{F}) \neq \emptyset$, as well as that any strong solution of (262) is already a solution of (262), provided that \mathcal{A} and \mathcal{B} are MLOs with \mathcal{A} being closed; this can be simply verified with the help of Theorem 1.2.3. The notion of a (pre-)solution of problems (DFP)_R and (DFP)_L can be similarly defined on any finite interval $[0, \tau)$ or $[0, \tau]$, where $0 < \tau < \infty$, and extends so the notion of a strict solution of problem (E) given on pp. 33–34 of [**199**] ($\mathcal{B} = I$, $\alpha = 1$, $\mathcal{F}(t) = f(t)$ is continuous single-valued). We refer the reader to [**211**] and [**306**] for some results about the wellposedness of some special cases of problem (DFP)_R. In our further work, it will be assumed that \mathcal{A} and \mathcal{B} are multivalued linear operators. Observe that we cannot consider the qualitative properties of solutions of problems (262), (DFP)_R or (DFP)_L in full generality by a simple passing to the multivalued linear operators $\mathcal{B}^{-1}\mathcal{A}$ or \mathcal{AB}^{-1} (see e.g. the definition of a solution of (262)). Concerning this question, we have the following remark. REMARK 3.1.2. (cf. also Section 2.1) Suppose that $0 < \tau \leq \infty$, $\alpha > 0$, as well as that $A: D(A) \subseteq X \to Y$ and $B: D(B) \subseteq X \to Y$ are two single-valued linear operators. Then $B^{-1}A$ is an MLO in X, and AB^{-1} is an MLO in Y.

- (i) Suppose that $u(\cdot)$ is a pre-solution (or, equivalently, solution) of problem (262) with $\mathcal{B} = I_X$, $\mathcal{A} = B^{-1}A$ and $\mathcal{F} = f : [0, \tau) \to D(B)$ being single-valued. Then $u \in C([0, \tau) : X)$ and Bu(t) = A(a * u)(t) + Bf(t), $t \in [0, \tau)$. If, in addition to this, $B \in L(X, Y)$ and $u(\cdot)$ is a strong solution of problem (262) with the above requirements being satisfied, then the mappings $t \mapsto Au(t), t \in [0, \tau)$ and $t \mapsto Bu(t), t \in [0, \tau)$ are continuous, and $(a * Au)(t) = Bu(t) - Bf(t), t \in [0, \tau)$.
- (ii) Suppose that $v(\cdot)$ is a pre-solution (solution) of problem (DFP)_{**R**} with $\mathcal{B} = I_Y, \mathcal{A} = AB^{-1}, \mathcal{F} = f : [0, \tau) \to Y$ being single-valued, and $v_j = Bx_j$ ($0 \leq j \leq \lceil \alpha \rceil - 1$). Let $B^{-1} \in L(Y, X)$. Then the function $u(t) := B^{-1}v(t), t \geq 0$ is a pre-solution (solution) of problem (DFP)_{**R**} with $\mathcal{B} = B$ and $\mathcal{A} = A$.
- (iii) Suppose that $\mathcal{F} = f : [0, \tau) \to D(B)$ is single-valued and $u(\cdot)$ is a presolution of problem $(DFP)_{\mathbf{L}}$ with $\mathcal{B} = I_X$ and $\mathcal{A} = B^{-1}A$. Then $u(\cdot)$ is a pre-solution of problem $(DFP)_{\mathbf{L}}$ with $\mathcal{B} = B$, $\mathcal{A} = A$ and $\mathcal{F}(t) = Bf(t)$, $t \in [0, \tau)$. If, in addition to this, $B \in L(X, Y)$ and $u(\cdot)$ is a solution of problem $(DFP)_{\mathbf{L}}$ with the above requirements being satisfied, then $u(\cdot)$ is a solution of problem $(DFP)_{\mathbf{L}}$ with $\mathcal{B} = B$, $\mathcal{A} = A$ and $\mathcal{F}(t) = Bf(t)$, $t \in [0, \tau)$.
- (iv) Suppose that $u: [0, \infty) \to D(A) \cap D(B)$. Then $u(\cdot)$ is a *p*-solution of problem (DFP)_{**R**} with $\mathcal{B} = B$ and $\mathcal{A} = A$ iff $v = Bu(\cdot)$ is a pre-solution of problem

$$\begin{cases} \mathbf{D}_t^{\alpha} v(t) \in AB^{-1} v(t) + \mathcal{F}(t), & t \ge 0, \\ v^{(j)}(0) = Bx_j, & 0 \le j \le \lceil \alpha \rceil - 1. \end{cases}$$

(v) Suppose that $C_Y \in L(Y)$ is injective and the closed graph theorem holds for the mappings from Y into Y. Then we define the set $\rho_{C_Y}^B(A) := \{\lambda \in \mathbb{C} : \lambda B - A \text{ is injective and } (\lambda B - A)^{-1}C_Y \in L(Y)\}$. It can be simply checked that $\rho_{C_Y}^B(A) \subseteq \rho_{C_Y}(AB^{-1})$, as well as that

(263)
$$(\lambda - AB^{-1})^{-1}C_Y = B(\lambda B - A)^{-1}C_Y, \quad \lambda \in \rho^B_{C_Y}(A).$$

This is an extension of [199, Theorem 1.14] and holds even in the case that the operator C_Y does not commute with AB^{-1} , when we define the C_Y -resolvent set of the operator $\lambda - AB^{-1}$ in the same way as before. Observe also that the assumption $D(A) \subseteq D(B)$, which has been used in [199, Section 1.6], is not necessary for the validity of (263).

(vi) Suppose that X = Y, $C \in L(X)$ is injective, $B \in L(X)$, $CA \subseteq AC$ and $CB \subseteq BC$. Define the set $\rho_C^B(A)$ as above. Then we have $\rho_C^B(A) \subseteq \rho_C(B^{-1}A)$ and

$$(\lambda - B^{-1}A)^{-1}Cx = (\lambda B - A)^{-1}CBx, \quad x \in X.$$

Furthermore, if C = I, $X \neq Y$ and $B \in L(X, Y)$, then $\rho^B(A) \subseteq \rho(B^{-1}A)$ and the previous equality holds.

Consider now the case in which the operator \mathcal{A} is closed, the operator $\mathcal{B} = B$ is single-valued and the function $\mathcal{F}(t) = f(t)$ is Y-continuous at each point $t \ge 0$. Then any pre-solution $u(\cdot)$ of problem (DFP)_{**R**} is already a solution of this problem, and Theorem 1.2.3 in combination with the identity (38) implies that

$$Bu(t) - \sum_{k=0}^{\lceil \alpha \rceil - 1} g_{k+1}(t) Bx_k - (g_\alpha * f)(t) \in \mathcal{A}(g_\alpha * u)(t), \quad t \ge 0.$$

Suppose, conversely, that there exists a function $u_{\mathcal{A}} \in C([0,\infty) : Y)$ such that $u_{\mathcal{A}}(t) \in \mathcal{A}u(t), t \ge 0$ and

$$Bu(t) - \sum_{k=0}^{\lceil \alpha \rceil - 1} g_{k+1}(t) Bx_k - (g_\alpha * f)(t) = (g_\alpha * u_\mathcal{A})(t), \quad t \ge 0.$$

Then it can be simply verified that $u(\cdot)$ is a solution of problem $(DFP)_{\mathbf{R}}$; it is noteworthy that we do not need the assumption on closedness of \mathcal{A} in this direction. Even in the case that $\mathcal{A} = \mathcal{A}$ is a closed single-valued linear operator, a corresponding statement for the problem $(DFP)_{\mathbf{L}}$ cannot be proved. Suppose, finally, that the operators \mathcal{A} and \mathcal{B} are closed, $u(\cdot)$ is a solution of problem $(DFP)_{\mathbf{R}}$, the function $\mathcal{F}(t) = f(t)$ is Y-continuous at each point $t \ge 0$, as well as the functions $u_{\alpha,\mathcal{B}} \in C([0,\infty):Y)$ and $u_{\mathcal{A}} \in C([0,\infty):Y)$ satisfy the requirements stated in Definition 3.1.1(iii). Using again Theorem 1.2.3 and the identity (38), it readily follows that

$$\mathcal{B}\left[u(t) - \sum_{k=0}^{\lceil \alpha \rceil - 1} g_{k+1}(t) x_k\right] \ni (g_\alpha * u_{\alpha, \mathcal{B}})(t)$$
$$= (g_\alpha * u_\mathcal{A})(t) + (g_\alpha * f)(t) \in \mathcal{A}(g_\alpha * u)(t) + (g_\alpha * f)(t), \quad t \ge 0.$$

The proof of following important theorem can be deduced by using Theorem 1.2.3, Theorem 1.4.2[(iv),(vi)], Theorem 1.4.4 and the argumentation already seen in the proof of [**280**, Theorem 3.1] (cf. also [**285**, Fundamental Lemma 3.1]); observe that we do not use the assumption on the exponential boundedness of function u(t) here. After formulation, we will only include the most relevant details needed for the proof of implication (iii) \Rightarrow (iv).

THEOREM 3.1.3. Suppose that $\mathcal{A}: X \to P(Y)$ and $\mathcal{B}: X \to P(Y)$ are MLOs, as well as that \mathcal{A} is $X_{\mathcal{A}} \times Y_{\mathcal{A}}$ -closed. Assume, further, that $a \in L^{1}_{loc}([0,\infty))$, $a \neq 0$, $\operatorname{abs}(|a|) < \infty$, $u \in C([0,\infty): X)$, $u \in (P1) - X$, as well as that $u(t) \in D(\mathcal{B})$, $t \ge 0$, $a * u \in C([0,\infty): X_{\mathcal{A}})$, $a * u \in (P1) - X_{\mathcal{A}}$, $\operatorname{abs}_{Y_{\mathcal{A}}}(\mathcal{B}u) < \infty$, $\operatorname{abs}_{Y_{\mathcal{A}}}(\mathcal{F}) < \infty$, and $\omega > \max(0, \omega_{X}(u), \operatorname{abs}_{Y_{\mathcal{A}}}(\mathcal{B}u), \operatorname{abs}_{Y_{\mathcal{A}}}(\mathcal{F}), \operatorname{abs}_{X_{\mathcal{A}}}(a * u))$. Consider the following assertions:

- (i) $u(\cdot)$ is a solution of (262) with $\tau = \infty$.
- (ii) $u(\cdot)$ is a pre-solution of (262) with $\tau = \infty$.

(iii) For any section $u_{\mathcal{B}} \in \sec(\mathcal{B}u)$ there is a section $f \in \sec(\mathcal{F})$ such that

$$\widetilde{u}_{\mathcal{B}}(\lambda) - \widetilde{f}(\lambda) \in \widetilde{a}(\lambda)\mathcal{A}\widetilde{u}(\lambda), \quad \operatorname{Re}\lambda > \omega, \ \widetilde{a}(\lambda) \neq 0.$$

(iv) For any section $u_{\mathcal{B}} \in \sec(\mathcal{B}u)$ there is a section $f \in \sec(\mathcal{F})$ such that

(264)
$$\widetilde{u}_{\mathcal{B}}(\lambda) - \tilde{f}(\lambda) \in \tilde{a}(\lambda) \mathcal{A}\tilde{u}(\lambda), \quad \lambda \in \mathbb{N}, \ \lambda > \omega, \ \tilde{a}(\lambda) \neq 0.$$

(v) For any section $u_{\mathcal{B}} \in \sec(\mathcal{B}u)$ there is a section $f \in \sec(\mathcal{F})$ such that

(265)
$$(1 * u_{\mathcal{B}})(t) - (1 * f)(t) \in \mathcal{A}(1 * a * u)(t), \quad t \ge 0.$$

Then we have (i) \Rightarrow (ii) \Rightarrow (iii) \Rightarrow (iv) \Rightarrow (v). Furthermore, if $\mathcal{B} = B$ is single-valued, $Bu \in C([0,\infty): Y_{\mathcal{A}})$ and $\mathcal{F} = f \in C([0,\infty): Y_{\mathcal{A}})$ is single-valued, then the above is equivalent.

SKETCH OF PROOF FOR (iv) \Rightarrow (v). Suppose that for any section $u_{\mathcal{B}} \in \operatorname{sec}(\mathcal{B}u)$ there is a section $f \in \operatorname{sec}(\mathcal{F})$ such that (264) holds. Let a number $\lambda \in \mathbb{N}$ with $\lambda > \omega$ and $\tilde{a}(\lambda) = 0$ be temporarily fixed. Then there exists a sequence $(\lambda_n)_{n \in \mathbb{N}}$ in (λ, ∞) such that $\tilde{a}(\lambda_n) \neq 0$ and $\lim_{n \to +\infty} \lambda_n = \lambda$. Since $(\tilde{a}(\lambda_n)\tilde{u}(\lambda_n), \tilde{u}_{\mathcal{B}}(\lambda_n) - \tilde{f}(\lambda_n)) \in \mathcal{A}$, $n \in \mathbb{N}$, i.e., $(\widetilde{a * u}(\lambda_n), \widetilde{u}_{\mathcal{B}}(\lambda_n) - \tilde{f}(\lambda_n)) \in \mathcal{A}$, $n \in \mathbb{N}$, and \mathcal{A} is $X_{\mathcal{A}} \times Y_{\mathcal{A}}$ -closed, it readily folows that $(\widetilde{a * u}(\lambda), \widetilde{u}_{\mathcal{B}}(\lambda) - \tilde{f}(\lambda)) \in \mathcal{A}$; in other words, $(0, \widetilde{u}_{\mathcal{B}}(\lambda) - \tilde{f}(\lambda)) \in \mathcal{A}$. By the foregoing, we have that $(\widetilde{a * u}(\lambda), \widetilde{u}_{\mathcal{B}}(\lambda) - \tilde{f}(\lambda)) \in \mathcal{A}$ for all $\lambda \in \mathbb{N}$ with $\lambda > \omega$. Using Theorem 1.2.3, we get that $\int_0^{\infty} e^{-\lambda t} (u_{\mathcal{B}} - f)^{[2]}(t) dt \in \mathcal{A} \int_0^{\infty} e^{-\lambda t} (a * u)^{[2]}(t) dt$ ($\lambda \in \mathbb{N}, \lambda > \omega$) and now we can apply Theorem 1.4.4, along with the $X_{\mathcal{A}} \times Y_{\mathcal{A}}$ -closedness of \mathcal{A} , in order to see that $u_{\mathcal{B}}^{[2]}(t) - f^{[2]}(t) \in \mathcal{A}(a * u)^{[2]}(t), t \geq 0$. This simply implies (265).

REMARK 3.1.4. Observe that we do not require any type of closedness of the operator \mathcal{B} in the formulation of Theorem 3.1.3. Even in the case that X = Y and $\mathcal{B} = B = I$, we cannot differentiate the equation (265) once more without making an additional assumption that $\mathcal{F} = f \in C([0, \infty) : Y_A)$ is single-valued (cf. [280, 1. -1, p. 173; l. 1–03, p. 174], where the author has made a small mistake in the consideration; in actual fact, the equation [280, (3.1)] has to be valid for some $f \in \sec_c(\mathcal{F})$ in order for the proof of implication (iii) \Rightarrow (i) of [280, Theorem 3.1] to work).

If Ω is a non-empty open subset of \mathbb{C} and $G: \Omega \to X$ is an analytic mapping that it is not identically equal to the zero function, then we can simply prove that for each zero λ_0 of $G(\cdot)$ there exists a uniquely determined natural number $n \in \mathbb{N}$ such that $G^{(j)}(\lambda_0) = 0$ for $0 \leq j \leq n-1$ and $G^{(n)}(\lambda_0) \neq 0$. Owing to this fact, we can repeat almost verbatim the arguments given in the proof of [**280**, Theorem 3.2] to verify the validity of the following Ljubich uniqueness type theorem:

THEOREM 3.1.5. Suppose $\mathcal{A}: X \to P(Y)$ is an MLO, $\mathcal{B} = B: D(B) \subseteq X \to Y$ is a single-valued linear operator, \mathcal{A} is $X_{\mathcal{A}} \times Y_{\mathcal{A}}$ -closed and B is $X_B \times Y_B$ -closed, where $Y_{\mathcal{A}} \hookrightarrow Y_B$. Assume, further, that $a \in L^1_{loc}([0,\infty))$, $a \neq 0$, $abs(|a|) < \infty$, $\mathcal{F} = f \in C([0,\infty): Y_{\mathcal{A}})$ is single-valued, $abs_{Y_{\mathcal{A}}}(f) < \infty$, and there exist a sequence $(\lambda_k)_{k\in\mathbb{N}}$ of complex numbers and a number $\omega > \operatorname{abs}(|a|)$ such that $\lim_{k\to\infty} \operatorname{Re} \lambda_k = +\infty$, $\tilde{a}(\lambda_k) \neq 0$, $k \in \mathbb{N}$, and

$$\frac{1}{\tilde{a}(\lambda_k)}Bx \notin \mathcal{A}x, \quad k \in \mathbb{N}, \ 0 \neq x \in D(\mathcal{A}) \cap D(B)$$

Then there exists a unique pre-solution of (262), with $\tau = \infty$, satisfying that $u \in (P1) - X_B$, $u(t) \in D(B)$, $t \ge 0$, $Bu \in C([0,\infty) : Y_A)$, $a * u \in C([0,\infty) : X_A)$, $a * u \in (P1) - X_A$ and $\operatorname{abs}_{Y_A}(Bu) < \infty$.

In the following extension of [**292**, Theorem 2.1.34], we will prove one more Ljubich's uniqueness criterium for abstract Cauchy problems with multivalued linear operators (cf. also [**285**, Theorem 3.5] and [**292**, Theorem 2.10.44]).

THEOREM 3.1.6. Suppose $\alpha > 0$, $\lambda > 0$, \mathcal{A} is an MLO in X, $\{(n\lambda)^{\alpha} : n \in \mathbb{N}\} \subseteq \rho_C(\mathcal{A})$ and, for every $\sigma > 0$ and $x \in X$, we have

$$\lim_{n \to \infty} \frac{((n\lambda)^{\alpha} - \mathcal{A})^{-1} C x}{e^{n\lambda\sigma}} = 0$$

Then, for every $x_0, \ldots, x_{\lceil \alpha \rceil - 1} \in X$, there exists at most one pre-solution of the initial value problem (DFP)_{**R**} with $\mathcal{B} = I$.

PROOF. It suffices to show that the zero function is the only pre-solution of the problem $(DFP)_{\mathbf{R}}$ with $\mathcal{B} = I$ and the initial values $x_0, \ldots, x_{\lceil \alpha \rceil - 1}$ chosen to be zeroes. Let $u(\cdot)$ be a pre-solution of such a problem. Set $z_n(t) := ((n\lambda)^{\alpha} - \mathcal{A})^{-1}Cu(t), t \ge 0, n \in \mathbb{N}$. Then it can be easily checked with the help of Theorem 1.2.4(i) that $z_n(\cdot)$ is a solution of the initial value problem:

$$\begin{cases} z_n \in C^{\lceil \alpha \rceil}((0,\infty):X) \cap C^{\lceil \alpha \rceil - 1}([0,\infty):X), \\ \mathbf{D}_t^{\alpha} z_n(t) = (n\lambda)^{\alpha} z_n(t) - Cu(t), \quad t > 0, \\ z_n^{(j)}(0) = 0, \quad 0 \le j \le \lceil \alpha \rceil - 1. \end{cases}$$

This implies $z_n(t) = -(u * \cdot^{\alpha-1} E_{\alpha,\alpha}((n\lambda)^{\alpha} \cdot^{\alpha-1}))(t), t \ge 0, n \in \mathbb{N}$ and

$$\lim_{n \to \infty} e^{-n\lambda\sigma} \int_0^t s^{\alpha-1} E_{\alpha,\alpha}((n\lambda)^\alpha s^\alpha) Cu(t-s) ds = 0 \quad (t > 0, \ \sigma > 0).$$

Now we can argue as in the second part of proof of [292, Theorem 2.1.34] so as to conclude that $u(t) = 0, t \ge 0$ (in the case that $\alpha \in \mathbb{N}$, the assertion can be proved by passing to the theory of abstract Cauchy problems of first order since [292, Lemma 2.1.33(i)] admits an extension to multivalued linear operators).

REMARK 3.1.7. Observe that, in the formulation of Theorem 3.1.6, we do not require any type of closedness of the operator \mathcal{A} .

The following theorem is very similar to [61, Theorem 3.1, Theorem 3.3] and [292, Theorem 2.4.2]. Because of its importance, we will include the most relevant details of proof.

THEOREM 3.1.8 (Subordination principle for abstract time-fractional inclusions). Suppose that $0 < \alpha < \beta$, $\gamma = \alpha/\beta$, $\mathcal{A}: X \to P(Y)$ is an MLO, $\mathcal{B} = B: D(B) \subseteq X \to Y$ is a single-valued linear operator, \mathcal{A} is $X_{\mathcal{A}} \times Y_{\mathcal{A}}$ -closed and B is $X_B \times Y_B$ -closed, where $X_A \hookrightarrow X_B$ and $Y_A \hookrightarrow Y_B$. Assume, further, that $f_\beta \in LT_{or} - Y_A$ is single-valued and there exists a pre-solution (or, equivalently, solution) $u(t) := u_\beta(t)$ of (262), with $\tau = \infty$, $a(t) = g_\beta(t)$ and $\mathcal{F} = f_\beta$, satisfying that $u_\beta \in LT_{or} - X_B$, $Bu_\beta \in LT_{or} - Y_A$, $g_\beta * u_\beta \in LT_{or} - X_A$ and that for each seminorm $p \in \circledast_{X_B}$ there exists $\omega_p \ge 0$ such that $p(u_\beta(t)) = O(e^{\omega_p t})$, $t \ge 0$, $p \in \circledast_{X_B}$. Define

$$u_{\alpha}(t) := \int_0^{\infty} t^{-\gamma} \Phi_{\gamma}(st^{-\gamma}) u_{\beta}(s) ds, \ t > 0 \ and \ u_{\alpha}(0) := u_{\beta}(0)$$

and

$$f_{\alpha}(t)x := \int_{0}^{\infty} t^{-\gamma} \Phi_{\gamma}(st^{-\gamma}) f_{\beta}(s) ds, \ t > 0 \ and \ f_{\alpha}(0) := f_{\beta}(0).$$

Then $u_{\alpha}(t)$ is a solution of (262), with $\tau = \infty$, $a(t) = g_{\alpha}(t)$ and $\mathcal{F}(t) = f_{\alpha}(t) \in LT_{or} - Y_{\mathcal{A}}$, satisfying additionally that $u_{\alpha} \in LT_{or} - X_{B}$, $Bu_{\alpha} \in LT_{or} - Y_{\mathcal{A}}$, $g_{\alpha} * u_{\alpha} \in LT_{or} - X_{\mathcal{A}}$ and

(266)
$$p(u_{\alpha}(t)) = O(\exp(\omega_p^{1/\gamma}t)), \quad p \in \circledast_{X_B}, \ t \ge 0$$

Let $p \in \circledast_{X_B}$ be fixed. Then the condition

(267)
$$p(u_{\beta}(t)) = O((1+t^{\xi_p})e^{\omega_p t}) \text{ for some } \xi_p \ge 0,$$

resp.,

(268)
$$p(u_{\beta}(t)) = O(t^{\xi_p} e^{\omega_p t}), \quad t \ge 0$$

implies that

(269)
$$p(u_{\alpha}(t)) = O((1 + t^{\xi_p \gamma})(1 + \omega_p t^{\xi_p(1-\gamma)}) \exp(\omega_p^{1/\gamma} t)), \quad t \ge 0,$$

resp.,

(270)
$$p(u_{\alpha}(t)) = O(t^{\xi_p \gamma} (1 + \omega_p t^{\xi_p (1-\gamma)}) \exp(\omega_p^{1/\gamma} t)), \quad t \ge 0.$$

Furthermore, the following holds:

- (i) The mapping $t \mapsto u_{\alpha}(t), t > 0$ admits an extension to $\sum_{\min((\frac{1}{\gamma}-1)\frac{\pi}{2},\pi)}$ and the mapping $z \mapsto u_{\alpha}(z), z \in \sum_{\min((\frac{1}{\gamma}-1)\frac{\pi}{2},\pi)}$ is analytic.
- (ii) Let $\varepsilon \in (0, \min((\frac{1}{\gamma} 1)\frac{\pi}{2}, \pi))$. If, for every $p \in \circledast$, one has $\omega_p = 0$, then for each $\theta \in (0, \min((\frac{1}{\gamma} 1)\frac{\pi}{2}, \pi))$ the following holds: $\lim_{z \to 0, z \in \overline{\Sigma_{\theta}}} u_{\alpha}(z) = u_{\alpha}(0)$.
- (iii) If $\omega_p > 0$ for some $p \in \circledast$, then for each $\theta \in (0, \min((\frac{1}{\gamma} 1)\frac{\pi}{2}, \frac{\pi}{2}))$ the following holds: $\lim_{z \to 0, z \in \overline{\Sigma_{\theta}}} u_{\alpha}(z) = u_{\alpha}(0)$.

PROOF. The proofs of (i)–(iii) follows similarly as in that of [61, Theorem 3.3], while the proof that the condition (267), resp. (268), implies (269), resp. (270), follows similarly as in that of [292, Theorem 2.4.2]. Furthermore, it can be easily seen that the estimate (266) holds for solution $u_{\alpha}(\cdot)$. By Theorem 3.1.3, we should only show that $u_{\alpha} \in LT_{or} - X_B$, $Bu_{\alpha} \in LT_{or} - Y_A$, $f_{\alpha} \in LT_{or} - Y_A$, $g_{\alpha} * u_{\alpha} \in LT_{or} - X_A$ and

(271)
$$\widetilde{Bu_{\alpha}}(\lambda) - \widetilde{f_{\alpha}}(\lambda) \in \lambda^{-\alpha} \mathcal{A}\widetilde{u_{\alpha}}(\lambda), \quad \lambda > \omega \text{ suff. large.}$$

Since $u_{\beta} \in LT_{or} - X_B$, the proof of [61, Theorem 3.1] immediately implies that $u_{\alpha} \in LT_{or} - X_B$, as well as that $\widetilde{u_{\alpha}}(\lambda) = \lambda^{\gamma-1}\widetilde{u_{\beta}}(\lambda^{\beta}), \lambda > \omega$ suff. large. Similarly, we have that $f_{\alpha} \in LT_{or} - Y_{\mathcal{A}}$ and $\widetilde{f_{\alpha}}(\lambda) = \lambda^{\gamma-1}\widetilde{f_{\beta}}(\lambda^{\beta}), \lambda > \omega$ suff. large. Keeping in mind that $X_{\mathcal{A}} \hookrightarrow X_B$ and $g_{\beta} * u_{\beta} \in LT_{or} - X_{\mathcal{A}}$, we can prove that $(g_{\alpha} * u_{\alpha})(t) = \int_0^{\infty} t^{-\gamma} \Phi_{\gamma}(st^{-\gamma})(g_{\beta} * u_{\beta})(s)ds, t > 0$ by performing the Laplace transform (the convergence of last integral is taken for the topology of $X_{\mathcal{A}}$). This simply implies that $g_{\alpha} * u_{\alpha} \in LT_{or} - X_{\mathcal{A}}$ and $(\mathcal{L}(g_{\alpha} * u_{\alpha}))(\lambda) = \lambda^{\gamma-1}(\mathcal{L}(g_{\beta} * u_{\beta}))(\lambda^{\gamma}), \lambda > \omega$ suff. large. Since $Y_{\mathcal{A}} \leftrightarrow Y_B$, a similar line of reasoning shows that $Bu_{\alpha}(t) = \int_0^{\infty} t^{-\gamma} \Phi_{\gamma}(st^{-\gamma})(Bu_{\beta})(s)ds, t > 0$ (the convergence of this integral is taken for the topology of $Y_{\mathcal{A}}$) and $\widetilde{Bu_{\alpha}}(\lambda) = \widetilde{Bu_{\alpha}}(\lambda), \lambda > \omega$ suff. large. The proof of (271) now follows from a simple computation.

We can similarly prove the following subordination principles for abstract degenerate Volterra inclusions in locally convex spaces (cf. [459, Section 4] and [292, Theorem 2.1.8, Theorem 2.8.7] for more details concerning non-degenerate case and, especially, the case in which $b(t) = g_1(t)$ or $b(t) = g_2(t)$).

THEOREM 3.1.9. Let b(t) and c(t) satisfy (P1), let $\int_0^\infty e^{-\beta t} |b(t)| dt < \infty$ for some $\beta \ge 0$, and let

$$\alpha = \tilde{c}^{-1} \left(\frac{1}{\beta}\right) if \int_0^\infty c(t) dt > \frac{1}{\beta}, \quad \alpha = 0 \text{ otherwise.}$$

Suppose that $\operatorname{abs}(|a|) < \infty$, $\tilde{a}(\lambda) = \tilde{b}(\frac{1}{\tilde{c}(\lambda)})$, $\lambda > \alpha$, $\mathcal{A} \colon X \to P(Y)$ is an MLO, $\mathcal{B} = B \colon D(B) \subseteq X \to Y$ is a single-valued linear operator, \mathcal{A} is $X_{\mathcal{A}} \times Y_{\mathcal{A}}$ -closed and B is $X_B \times Y_B$ -closed, where $X_{\mathcal{A}} \hookrightarrow X_B$ and $Y_{\mathcal{A}} \hookrightarrow Y_B$. Assume, further, that $f_{\beta} \in LT_{or} - Y_{\mathcal{A}}$ is single-valued and there exists a pre-solution (or, equivalently, solution) $u(t) := u_b(t)$ of (262), with $\tau = \infty$, a(t) replaced with b(t) therein, and $\mathcal{F} = f_b$, satisfying that $u_b \in LT_{or} - X_B$, $Bu_b \in LT_{or} - Y_{\mathcal{A}}$, $b * u_b \in LT_{or} - X_{\mathcal{A}}$ and the family $\{e^{-\omega_b t}u_b(t) : t \ge 0\}$ is bounded in X_B ($\omega_b \ge 0$). Assume, further, that c(t) is completely positive and there exists a function $f_a \in LT_{or} - Y_{\mathcal{A}}$ satisfying

$$\widetilde{f}_a(\lambda) = \frac{1}{\lambda \widetilde{c}(\lambda)} \widetilde{f}_b\left(\frac{1}{\widetilde{c}(\lambda)}\right), \quad \lambda > \omega_0, \ \widetilde{f}_b\left(\frac{1}{\widetilde{c}(\lambda)}\right) \neq 0, \ \text{for some } \omega_0 > 0.$$

Let

$$\omega_a = \tilde{c}^{-1} \left(\frac{1}{\omega_b}\right) \text{ if } \int_0^\infty c(t) dt > \frac{1}{\omega_b}, \quad \omega_a = 0 \text{ otherwise.}$$

Then, for every $r \in (0,1]$, there exists a solution $u(t) := u_{a,r}(t)$ of (262), with $\tau = \infty$, a(t) and $\mathcal{F} = f_r := g_r * f_a$, satisfying that $u_{a,r} \in LT_{or} - X_B$, $Bu_{a,r} \in LT_{or} - Y_A$, $a * u_{a,r} \in LT_{or} - X_A$ and the set $\{e^{-\omega_a t}u_{a,r}(t) : t \ge 0\}$ is bounded in X_B , if $\omega_b = 0$ or $\omega_b \tilde{c}(0) \ne 1$, resp., the set $\{e^{-\varepsilon t}u_{a,r}(t) : t \ge 0\}$ is bounded in X_B for any $\varepsilon > 0$, if $\omega_b > 0$ and $\omega_b \tilde{c}(0) = 1$. Furthermore, the function $t \mapsto u_{a,r}(t) \in X_B$, $t \ge 0$ is locally Hölder continuous with the exponent $r \in (0, 1]$.

REMARK 3.1.10. (i) In Theorem 3.1.8 and Theorem 3.1.9, we have only proved the existence of a solution of the subordinated inclusion. The uniqueness of solutions can be proved, for example, by using Theorem 3.1.5, Theorem 3.1.6 or Theorem 2.3.6.

(ii) In Theorem 3.1.9, we have faced ourselves with a loss of regularity for solutions of the subordinated problem. Even in the case that X = Y and B = I, it is not so simple to prove the existence of a solution of problem (262), with $\tau = \infty$, a(t) and $\mathcal{F} = f_a$, without imposing some additional unfavorable conditions. In the next section, we will introduce various types of solution operator families for the abstract Volterra inclusion (262) and there we will reconsider the problem of loss of regularity for solutions of subordinated problem once more (cf. Theorem 3.2.7).

3.2. Multivalued linear operators as subgenerators of various types of (a, k)-regularized C-resolvent operator families

In [292, Section 2.8], the class of (a, k)-regularized (C_1, C_2) -existence and uniqueness families has been introduced and analyzed within the theory of abstract non-degenerate Volterra equations. The main aim of this section is to consider multivalued linear operators in locally convex spaces as subgenerators of (a, k)regularized (C_1, C_2) -existence and uniqueness families, as well as to consider in more detail the class of (a, k)-regularized *C*-resolvent families. Unless specified otherwise, we assume that $0 < \tau \leq \infty$, $k \in C([0, \tau))$, $k \neq 0$, $a \in L^1_{loc}([0, \tau))$, $a \neq 0$, $\mathcal{A}: X \to P(X)$ is an MLO, $C_1 \in L(Y, X)$, $C_2 \in L(X)$ is injective, $C \in L(X)$ is injective and $C\mathcal{A} \subseteq \mathcal{A}C$. The following definition is an extension of [292, Definition 2.8.2] $(X = Y, A \text{ is a closed single-valued linear operator on <math>X$) and [482, Definition 3.5] $(X = Y, C = C_1, a(t) = k(t) = 1)$.

DEFINITION 3.2.1. Suppose $0 < \tau \leq \infty$, $k \in C([0, \tau))$, $k \neq 0$, $a \in L^1_{loc}([0, \tau))$, $a \neq 0$, $\mathcal{A} \colon X \to P(X)$ is an MLO, $C_1 \in L(Y, X)$, and $C_2 \in L(X)$ is injective.

(i) Then it is said that \mathcal{A} is a subgenerator of a (local, if $\tau < \infty$) mild (a, k)regularized (C_1, C_2) -existence and uniqueness family $(R_1(t), R_2(t))_{t \in [0, \tau)}$ $\subseteq L(Y, X) \times L(X)$ iff the mappings $t \mapsto R_1(t)y, t \ge 0$ and $t \mapsto R_2(t)x,$ $t \in [0, \tau)$ are continuous for every fixed $x \in X$ and $y \in Y$, as well as the following conditions hold:

(272)
$$\left(\int_{0}^{t} a(t-s)R_{1}(s)y\,ds, R_{1}(t)y-k(t)C_{1}y\right) \in \mathcal{A}, \quad t \in [0,\tau), \ y \in Y \text{ and}$$

(273)
$$\int_0^t a(t-s)R_2(s)y\,ds = R_2(t)x - k(t)C_2x$$
, whenever $t \in [0,\tau)$ and $(x,y) \in \mathcal{A}$.

- (ii) Let $(R_1(t))_{t \in [0,\tau)} \subseteq L(Y, X)$ be strongly continuous. Then it is said that \mathcal{A} is a subgenerator of a (local, if $\tau < \infty$) mild (a, k)-regularized C_1 -existence family $(R_1(t))_{t \in [0,\tau)}$ iff (272) holds.
- (iii) Let $(R_2(t))_{t \in [0,\tau)} \subseteq L(X)$ be strongly continuous. Then it is said that \mathcal{A} is a subgenerator of a (local, if $\tau < \infty$) mild (a, k)-regularized C_2 -uniqueness family $(R_2(t))_{t \in [0,\tau)}$ iff (273) holds.

As an immediate consequence of definition, we have that $R(R_1(0) - k(0)C_1) \subseteq \mathcal{A}0$. Furthermore, if a(t) is a kernel on $[0, \tau)$, then $R_2(t)\mathcal{A}$ is single-valued for any $t \in [0, \tau)$ and $R_2(t)y = 0$ for any $y \in \mathcal{A}0$ and $t \in [0, \tau)$. Now we will extend

the definition of an (a, k)-regularized C-resolvent family subgenerated by a singlevalued linear operator (cf. [292, Definition 2.1.1]).

DEFINITION 3.2.2. Suppose that $0 < \tau \leq \infty$, $k \in C([0,\tau))$, $k \neq 0$, $a \in L_{loc}^1([0,\tau))$, $a \neq 0$, $A: X \to P(X)$ is an MLO, $C \in L(X)$ is injective and $CA \subseteq AC$. Then it is said that a strongly continuous operator family $(R(t))_{t \in [0,\tau)} \subseteq L(X)$ is an (a, k)-regularized C-resolvent family with a subgenerator A iff $(R(t))_{t \in [0,\tau)}$ is a mild (a, k)-regularized C-uniqueness family having A as subgenerator, R(t)C = CR(t) and $R(t)A \subseteq AR(t)$ $(t \in [0, \tau))$.

An (a,k)-regularized C-resolvent family $(R(t))_{t\in[0,\tau)}$ is said to be locally equicontinuous iff, for every $t \in (0, \tau)$, the family $\{R(s) : s \in [0, t]\}$ is equicontinuous. In the case $\tau = \infty$, $(R(t))_{t \ge 0}$ is said to be exponentially equicontinuous (equicontinuous) if there exists $\omega \in \mathbb{R}$ ($\omega = 0$) such that the family $\{e^{-\omega t}R(t):$ $t \ge 0$ is equicontinuous; the infimum of such numbers is said to be the exponential type of $(R(t))_{t\geq 0}$. The above notion can be simply understood for the classes of mild (a, k)-regularized C_1 -existence families and mild (a, k)-regularized C_2 -uniqueness families; a mild (a, k)-regularized (C_1, C_2) -existence and uniqueness family $(R_1(t), R_2(t))_{t \in [0,\tau)} \subseteq L(Y, X) \times L(X)$ is said to be locally equicontinuous (exponentially equicontinuous, provided that $\tau = \infty$) iff both operator families $(R_1(t))_{t\geq 0}$ and $(R_2(t))_{t\geq 0}$ are. It would take too long to consider the notion of q-exponential equicontinuity for the classes of mild (a, k)-regularized C_1 -existence families and mild (a, k)-regularized C_2 -uniqueness families (cf. [292, Section 2.4] for more details about non-degenerate case). If $k(t) = g_{\alpha+1}(t)$, where $\alpha \ge 0$, then it is also said that $(R(t))_{t \in [0,\tau)}$ is an α -times integrated (a, C)-resolvent family; 0-times integrated (a, C)-resolvent family is further abbreviated to (a, C)-resolvent family. We will accept a similar terminology for the classes of mild (a, k)-regularized C_1 existence families and mild (a, k)-regularized C_2 -uniqueness families; in the case of consideration of convoluted C-semigroups, it will be always assumed that the condition (272) holds with a(t) = 1 and the operator C_1 replaced by C. Let us mention in passing that the operator semigroups generated by multivalued linear operators have been analyzed by A. G. Baskakov in [57].

The following proposition can be simply proved with the help of Theorem 1.2.3 and Theorem 1.1.4(ii).

PROPOSITION 3.2.3. Suppose that $(R_1(t), R_2(t))_{t \in [0,\tau)} \subseteq L(Y, X) \times L(X)$ is a mild (a, k)-regularized (C_1, C_2) -existence and uniqueness family with a subgenerator \mathcal{A} and $(R(t))_{t \in [0,\tau)} \subseteq L(X)$ is an (a, k)-regularized C-resolvent family with a subgenerator \mathcal{A} . Let $b \in L^1_{loc}([0,\tau))$ be such that $a * b \neq 0$ in $L^1([0,\tau))$ and $k * b \neq 0$ in $C([0,\tau))$. Then $((b * R_2)(t))_{t \geq 0}$ is a mild (a, k * b)-regularized C_2 -uniqueness family with a subgenerator \mathcal{A} . Furthermore, the following holds:

(i) Let \mathcal{A} be $X^1_{\mathcal{A}} \times X^2_{\mathcal{A}}$ -closed. Suppose that, for every $y \in Y$, the mapping $t \mapsto (a * R_1)(t)y, t \in [0, \tau)$ is continuous in $X^1_{\mathcal{A}}$ and the mapping $t \mapsto R_1(t)y, t \in [0, \tau)$ is continuous in $X^2_{\mathcal{A}}$. Then $((b * R_1)(t))_{t \geq 0}$ is a mild (a, k * b)-regularized C_1 -existence family with a subgenerator \mathcal{A} .

(ii) Let \mathcal{A} be $X^1_{\mathcal{A}} \times X^2_{\mathcal{A}}$ -closed. Suppose that, for every $x \in D(\mathcal{A})$ and $y \in R(\mathcal{A})$, the mapping $t \mapsto R(t)x$, $t \in [0, \tau)$ is continuous in $X^1_{\mathcal{A}}$ and the mapping $t \mapsto R(t)y$, $t \in [0, \tau)$ is continuous in $X^2_{\mathcal{A}}$. Then $((b * R)(t))_{t \geq 0}$ is an (a, k * b)-regularized C-regularized family with a subgenerator \mathcal{A} .

Although the parts (i) and (ii) of the above proposition have been stated for $X_{\mathcal{A}}^1 \times X_{\mathcal{A}}^2$ -closed subgenerators, the most important case in our further study will be that in which $X_{\mathcal{A}}^1 = X_{\mathcal{A}}^2 = X$. This is primarily caused by the following fact: Let \mathcal{A} be a subgenerator of a mild (a, k)-regularized C_1 -existence family (mild (a, k)-regularized C_2 -uniqueness family; mild (a, k)-regularized C-resolvent family) $(R_1(t))_{t \in [0,\tau)}$ ($(R_2(t))_{t \in [0,\tau)}$; $(R(t))_{t \in [0,\tau)}$, provided in addition that $(R_2(t))_{t \in [0,\tau)}$; $(R(t))_{t \in [0,\tau)}$; $(R(t))_{t \in [0,\tau)}$ is locally equicontinuous).

Suppose that $(R_1(t), R_2(t))_{t \in [0,\tau)}$ is a mild (a, k)-regularized (C_1, C_2) -existence and uniqueness family with a subgenerator \mathcal{A} . Arguing as in non-degenerate case (cf. the paragraph directly preceding [292, Definition 2.8.3]), we may conclude that

(274)
$$(a * R_2)(s)R_1(t)y - R_2(s)(a * R_1)(t)y$$

= $k(t)(a * R_2)(s)C_1y - k(s)C_2(a * R_1)(t)y, \quad t \in [0, \tau), y \in Y.$

The integral generator of a mild (a, k)-regularized C_2 -uniqueness family $(R_2(t))_{t \in [0,\tau)}$ (mild (a, k)-regularized (C_1, C_2) -existence and uniqueness family $(R_1(t), R_2(t))_{t \in [0,\tau)}$) is defined by

$$\mathcal{A}_{int} := \left\{ (x, y) \in X \times X : R_2(t)x - k(t)C_2x = \int_0^t a(t-s)R_2(s)y \, ds, \ t \in [0, \tau) \right\};$$

we define the integral generator of an (a, k)-regularized *C*-regularized family $(R(t))_{t \in [0,\tau)}$ in the same way as above. The integral generator \mathcal{A}_{int} is an MLO in *X* which is, in fact, the maximal subgenerator of $(R_2(t))_{t \in [0,\tau)}$ $((R(t))_{t \in [0,\tau)})$ with respect to the set inclusion; furthermore, the assumption $R_2(t)C_2 = C_2R_2(t)$, $t \in [0,\tau)$ implies that $C_2^{-1}\mathcal{A}_{int}C_2 = \mathcal{A}_{int}$ so that $C^{-1}\mathcal{A}_{int}C = \mathcal{A}_{int}$ for resolvent families. The local equicontinuity of $(R_2(t))_{t \in [0,\tau)}$ $((R(t))_{t \in [0,\tau)})$ immediately implies that \mathcal{A}_{int} is closed. Observe that, in the above definition of integral generator, we do not require that the function a(t) is a kernel on $[0,\tau)$, as in non-degenerate case. In the case of resolvent families, the following holds:

(i) Suppose that $(R(t))_{t \in [0,\tau)}$ is locally equicontinuous and \mathcal{A} is a closed subgenerator of $(R(t))_{t \in [0,\tau)}$. Then

(275)
$$\left(\int_0^t a(t-s)R(s)x\,ds, R(t)x-k(t)Cx\right) \in \mathcal{A}, \quad t \in [0,\tau), \ x \in \overline{D(\mathcal{A})}.$$

- (ii) If \mathcal{A} is a subgenerator of $(R(t))_{t \in [0,\tau)}$, then $C^{-1}\mathcal{A}C$ is a subgenerator of $(R(t))_{t \in [0,\tau)}$, too.
- (iii) Suppose that a(t) is a kernel on $[0, \tau)$, \mathcal{A} and \mathcal{B} are two subgenerators of $(R(t))_{t\in[0,\tau)}$, and $x \in D(\mathcal{A}) \cap D(\mathcal{B})$. Then R(t)(y-z) = 0, $t \in [0,\tau)$ for each $y \in \mathcal{A}x$ and $z \in \mathcal{B}x$.

(iv) Let \mathcal{A} be a subgenerator of $(R(t))_{t\in[0,\tau)}$, and let $\lambda \in \rho_C(\mathcal{A})$ $(\lambda \in \rho(\mathcal{A}))$. Suppose that $x \in X$, $y = (\lambda - \mathcal{A})^{-1}Cx$ $(y = (\lambda - \mathcal{A})^{-1}x)$ and $z \in \mathcal{A}y$. Then Theorem 1.2.4(i) implies that $\lambda(\lambda - \mathcal{A})^{-1}Cx - Cx \in \mathcal{A}(\lambda - \mathcal{A})^{-1}Cx = \mathcal{A}y$ $(\lambda(\lambda - \mathcal{A})^{-1}x - x \in \mathcal{A}(\lambda - \mathcal{A})^{-1}x = \mathcal{A}y)$, so that $R(t)y - k(t)Cy \in \mathcal{A}\int_0^t a(t - s)R(s)[\lambda(\lambda - \mathcal{A})^{-1}Cx - Cx]ds = \mathcal{A}\{\lambda(\lambda - \mathcal{A})^{-1}C\int_0^t a(t - s)R(s)Cx\,ds \in D(\mathcal{A}), t \in [0, \tau); \text{ from this, we may conclude that } R(t)Cx - k(t)C^2x \in (\lambda - \mathcal{A})\mathcal{A}(\lambda - \mathcal{A})^{-1}C\int_0^t a(t - s)R(s)x\,ds, t \in [0, \tau); \text{ similarly, we have that } \int_0^t a(t - s)R(s)x\,ds \in D(\mathcal{A}) \text{ and } R(t)x - k(t)Cx \in (\lambda - \mathcal{A})\mathcal{A}(\lambda - \mathcal{A})^{-1}C\int_0^t a(t - s)R(s)x\,ds, t \in [0, \tau); \text{ similarly, we have that } \int_0^t a(t - s)R(s)x\,ds, t \in [0, \tau), \text{ provided that } \lambda \in \rho(\mathcal{A}).$

The following extensions of [292, Theorem 2.8.5, Theorem 2.1.5] are stated without proofs.

THEOREM 3.2.4. Suppose \mathcal{A} is a closed MLO in $X, C_1 \in L(Y, X), C_2 \in L(X), C_2$ is injective, $\omega_0 \ge 0$ and $\omega \ge \max(\omega_0, \operatorname{abs}(|a|), \operatorname{abs}(k))$.

- (i) Let $(R_1(t), R_2(t))_{t \ge 0} \subseteq L(Y, X) \times L(X)$ be strongly continuous, and let the family $\{e^{-\omega t}R_i(t) : t \ge 0\}$ be equicontinuous for i = 1, 2.
 - (a) Suppose $(R_1(t), R_2(t))_{t \ge 0}$ is a mild (a, k)-regularized (C_1, C_2) -existence and uniqueness family with a subgenerator \mathcal{A} . Then, for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0$, the operator $I \tilde{a}(\lambda)\mathcal{A}$ is injective, $R(C_1) \subseteq \operatorname{R}(I \tilde{a}(\lambda)\mathcal{A})$,

(276)
$$\tilde{k}(\lambda)(I - \tilde{a}(\lambda)\mathcal{A})^{-1}C_1 y = \int_0^\infty e^{-\lambda t} R_1(t) y \, dt, \quad y \in Y,$$

(277)
$$\left\{\frac{1}{\tilde{a}(z)} : \operatorname{Re} z > \omega, \ \tilde{k}(z)\tilde{a}(z) \neq 0\right\} \subseteq \rho_{C_1}(\mathcal{A})$$

and

(278)
$$\tilde{k}(\lambda)C_2x = \int_0^\infty e^{-\lambda t} [R_2(t)x - (a * R_2)(t)y] dt, \quad whenever \ (x,y) \in \mathcal{A}.$$

Here, $\rho_{C_1}(\mathcal{A})$ is defined in the obvious way.

- (b) Let (277) hold, and let (276) and (278) hold for any $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0$. Then $(R_1(t), R_2(t))_{t \geq 0}$ is a mild (a, k)-regularized (C_1, C_2) -existence and uniqueness family with a subgenerator \mathcal{A} .
- (ii) Let $(R_1(t))_{t \ge 0}$ be strongly continuous, and let the family $\{e^{-\omega t}R_1(t) : t \ge 0\}$ be equicontinuous. Then $(R_1(t))_{t \ge 0}$ is a mild (a,k)-regularized C_1 existence family with a subgenerator \mathcal{A} iff for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0$, one has $R(C_1) \subseteq R(I \tilde{a}(\lambda)\mathcal{A})$ and

$$\tilde{k}(\lambda)C_1y \in (I - \tilde{a}(\lambda)\mathcal{A}) \int_0^\infty e^{-\lambda t} R_1(t)y \, dt, \quad y \in Y.$$

(iii) Let $(R_2(t))_{t\geq 0}$ be strongly continuous, and let the family $\{e^{-\omega t}R_2(t) : t \geq 0\}$ be equicontinuous. Then $(R_2(t))_{t\geq 0}$ is a mild (a,k)-regularized C_2 uniqueness family with a subgenerator \mathcal{A} iff for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0$, the operator $I - \tilde{a}(\lambda)\mathcal{A}$ is injective and (278) holds.

THEOREM 3.2.5. Let $(R(t))_{t\geq 0} \subseteq L(X)$ be a strongly continuous operator family such that there exists $\omega \geq 0$ satisfying that the family $\{e^{-\omega t}R(t) : t \geq 0\}$ is equicontinuous, and let $\omega_0 > \max(\omega, \operatorname{abs}(|a|), \operatorname{abs}(k))$. Suppose that \mathcal{A} is a closed MLO in X and $C\mathcal{A} \subset \mathcal{A}C$.

(i) Assume that A is a subgenerator of the global (a, k)-regularized C-resolvent family (R(t))_{t≥0} satisfying (272) for all x = y ∈ X. Then, for every λ ∈ C with Re λ > ω₀ and ã(λ)k̃(λ) ≠ 0, the operator I − ã(λ)A is injective, R(C) ⊆ R(I − ã(λ)A), as well as

(279)
$$\tilde{k}(\lambda)(I-\tilde{a}(\lambda)\mathcal{A})^{-1}Cx = \int_0^\infty e^{-\lambda t} R(t)x \, dt, \ x \in X, \ \operatorname{Re} \lambda > \omega_0, \ \tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0,$$

(280)
$$\left\{\frac{1}{\tilde{a}(\lambda)} : \operatorname{Re} \lambda > \omega_0, \ \tilde{k}(\lambda)\tilde{a}(\lambda) \neq 0\right\} \subseteq \rho_C(\mathcal{A})$$

and $R(s)R(t) = R(t)R(s), t, s \ge 0.$

(ii) Assume (279)–(280). Then \mathcal{A} is a subgenerator of the global (a, k)-regularized C-resolvent family $(R(t))_{t \ge 0}$ satisfying (272) for all $x = y \in X$ and $R(s)R(t) = R(t)R(s), t, s \ge 0.$

Before proceeding further, we would like to observe that the proof of [27, Proposition 4.1.3] implies that, for every strongly continuous, exponentially equicontinuous operator family $(S(t))_{t\geq 0} \subseteq L(E)$, the following Abel ergodicity property

$$\lim_{\lambda \to +\infty} \lambda \int_0^\infty e^{-\lambda t} S(t) x \, dt = S(0) x, \quad x \in E$$

holds true.

- REMARK 3.2.6. (i) Suppose that $(R(t))_{t \ge 0}$ is a degenerate exponentially equicontinuous (a, k)-regularized *C*-resolvent family in the sense of [**306**, Definition 2.2], and $B \in L(X)$. Using Remark 1.2.1(iv)/(a), Remark 3.1.2(iv) and Theorem 3.2.5(ii), it can be easily seen that $(R(t))_{t\ge 0}$ is an exponentially equicontinuous (a, k)-regularized *C*-resolvent family with a closed subgenerator $B^{-1}A$.
- (ii) Suppose that $n \in \mathbb{N}$, X and Y are Banach spaces, $A: D(A) \subseteq X \to Y$ is closed, $B \in L(X,Y)$ and $(V(t))_{t \ge 0} \subseteq L(X)$ is a degenerate exponentially bounded *n*-times integrated semigroup generated by linear operators A, B, in the sense of [424, Definition 1.5.3]. Then the arguments mentioned above show that $(V(t))_{t \ge 0}$ is an exponentially bounded *n*-times integrated (g_1, I) -regularized family (semigroup) with a closed subgenerator $B^{-1}A$.

- (iii) Let $n \in \mathbb{N}_0$. Due to Theorem 3.2.5(ii), the notion of an exponentially bounded (a, k)-regularized *C*-resolvent family extends the notion of a degenerate exponentially bounded *n*-times integrated semigroup generated by an MLO, introduced in [424, Definition 1.6.6, Definition 1.6.8].
- (iv) Suppose that $(R(t))_{t \ge 0} \subseteq L(X, [D(B)])$ is an exponentially equicontinuous (a, k)-regularized *C*-resolvent family generated by *A*, *B*, in the sense of [**307**, Definition 2.5]. Then [**307**, Theorem 2.3(i)] in combination with Remark 3.1.2(v) and Theorem 3.2.5(ii) implies that $(BR(t))_{t \ge 0}$ is an exponentially equicontinuous (a, k)-regularized *C*-resolvent family generated by $\overline{B^{-1}A}$ (recall that $B^{-1}A$ is closed provided that C = I).

The proof of following extension of [292, Theorem 2.1.8(i), Theorem 2.8.7(i)] is standard and therefore omitted; we can similarly reformulate Theorem 3.1.8 and [292, Proposition 2.1.16] for the class of mild (a, k)-regularized (C_1, C_2) -existence and uniqueness families ((a, k)-regularized *C*-resolvent families). Here it is only worth noting that the existence of a mild (a, k_1) -regularized C_1 -existence family $(R_{0,1}(t))_{t\geq 0}$ in the second part of theorem is not automatically guaranteed by the denseness of \mathcal{A} (even in the case that the operator $\mathcal{A} = \mathcal{A}$ is single-valued, it seems that the condition $C_1\mathcal{A} \subseteq \mathcal{A}C_1$ is indispensable for such a mild existence family to exist).

THEOREM 3.2.7. Suppose $C_1 \in L(Y, X)$, $C_2 \in L(X)$ is injective, \mathcal{A} is a closed MLO in $X, C \in L(X)$ is injective and $C\mathcal{A} \subseteq \mathcal{A}C$. Let b(t) and c(t) satisfy (P1), let $\int_0^\infty e^{-\beta t} |b(t)| dt < \infty$ for some $\beta \ge 0$, and let

$$\alpha = \tilde{c}^{-1} \Big(\frac{1}{\beta} \Big) \ if \ \int_0^\infty c(t) dt > \frac{1}{\beta}, \ \alpha = 0 \ otherwise.$$

Suppose that $\operatorname{abs}(|a|) < \infty$ and $\tilde{a}(\lambda) = \tilde{b}(\frac{1}{\tilde{c}(\lambda)}), \lambda \geq \alpha$. Let \mathcal{A} be a subgenerator of a (b,k)-regularized C_1 -existence family $(R_1(t))_{t\geq 0}$ ((b,k)-regularized C_2 -uniqueness family $(R_2(t))_{t\geq 0}$; (b,k)-regularized C-resolvent family $(R_0(t))_{t\geq 0}$ with the property that (272) holds for $R_1(\cdot)$ replaced with $R_0(\cdot)$ and each $x = y \in X$) satisfying that the family $\{e^{-\omega_b t}R_1(t) : t \geq 0\}$ ($\{e^{-\omega_b t}R_2(t) : t \geq 0\}$; $\{e^{-\omega_b t}R(t) : t \geq 0\}$) is equicontinuous for some $\omega_b \geq 0$. Assume, further, that c(t) is completely positive and there exists a scalar-valued continuous kernel $k_1(t)$ satisfying (P1) and

$$\widetilde{k_1}(\lambda) = \frac{1}{\lambda \widetilde{c}(\lambda)} \widetilde{k}\Big(\frac{1}{\widetilde{c}(\lambda)}\Big), \quad \lambda > \omega_0, \ \widetilde{k}\Big(\frac{1}{\widetilde{c}(\lambda)}\Big) \neq 0, \ \text{for some } \omega_0 > 0.$$

Let

$$\omega_a = \tilde{c}^{-1} \left(\frac{1}{\omega_b} \right) if \int_0^\infty c(t) dt > \frac{1}{\omega_b}, \quad \omega_a = 0 \text{ otherwise.}$$

Then, for every $r \in (0,1]$, \mathcal{A} is a subgenerator of a global $(a, k_1 * g_r)$ -regularized C_1 existence family $(R_{r,1}(t))_{t\geq 0}$ $((a, k_1 * g_r)$ -regularized C_2 -uniqueness family $(R_{r,2}(t))_{t\geq 0}$; $(a, k_1 * g_r)$ -regularized C-resolvent family $(R_{r,0}(t))_{t\geq 0}$ with the property that (272) holds for $R_1(\cdot)$ replaced with $R_{r,0}(\cdot)$ and each $x = y \in X$) such that the family $\{e^{-\omega_a t} R_{r,i}(t) : t \geq 0\}$ is equicontinuous and that the mapping $t \mapsto R_{r,i}(t), t \geq 0$ is locally Hölder continuous with exponent r, if $\omega_b = 0$ or $\omega_b \tilde{c}(0) \neq 1$ (i = 0, 1, 2), resp., for every $\varepsilon > 0$, there exists $M_{\varepsilon} \ge 1$ such that the family $\{e^{-\varepsilon t}R_{r,i}(t): t \ge 0\}$ is equicontinuous and that the mapping $t \mapsto R_{r,i}(t)$, $t \ge 0$ is locally Hölder continuous with exponent r, if $\omega_b > 0$ and $\omega_b \tilde{c}(0) = 1$ (i = 0, 1, 2). Furthermore, if \mathcal{A} is densely defined, then \mathcal{A} is a subgenerator of a global (a, k_1) -regularized C_2 -uniqueness family $(R_{0,2}(t))_{t\ge 0}$ $((a, k_1)$ -regularized C-resolvent family $(R_{0,0}(t))_{t\ge 0}$ with the property that (272) holds for $R_1(\cdot)$ replaced with $R_{0,0}(\cdot)$ and each $x = y \in X$ such that the family $\{e^{-\varepsilon t}R_i(t): t \ge 0\}$ is equicontinuous, resp., for every $\varepsilon > 0$, the family $\{e^{-\varepsilon t}R_i(t): t \ge 0\}$ is equicontinuous (i = 1, 2).

Let $(R_1(t), R_2(t))_{t \in [0, \tau)}$ be a mild (a, k)-regularized (C_1, C_2) -existence and uniqueness family with a subgenerator \mathcal{A} . Then it is not complicated to see that the function $t \mapsto R_1(t)y, t \in [0, \tau)$ $(y \in Y)$, resp. $t \mapsto R_2(t)x, t \in [0, \tau)$ $(x \in D(\mathcal{A}))$, is a solution of problem (262) with $\mathcal{B} = I$ and $f(t) = k(t)C_1y, t \in [0, \tau)$, resp. a strong solution of (262) with $\mathcal{B} = I$ and $f(t) = k(t)C_2x, t \in [0, \tau)$, provided additionally in the last case that $R_2(t)x \in D(\mathcal{A}), t \in [0, \tau)$ and $R_2(t)\mathcal{A}x \subseteq \mathcal{A}R_2(t)x, t \in [0, \tau)$. Furthermore, it is very simple to transmit the assertions of [**292**, Proposition 2.8.8, Proposition 2.8.9] to mild (a, k)-regularized (C_1, C_2) -existence and uniqueness families subgenerated by multivalued linear operators:

- PROPOSITION 3.2.8. (i) Suppose that $(R_1(t), R_2(t))_{t \in [0,\tau)}$ is a mild (a, k)regularized (C_1, C_2) -existence and uniqueness family with a subgenerator \mathcal{A} , as well as that $(R_2(t))_{t \in [0,\tau)}$ is locally equicontinuous and the functions a(t) and k(t) are kernels on $[0, \tau)$. Then $C_2R_1(t) = R_2(t)C_1$, $t \in [0, \tau)$.
- (ii) Suppose that $(R_2(t))_{t\in[0,\tau)}$ is a locally equicontinuous mild (a,k)-regularized C_2 -uniqueness family with a subgenerator \mathcal{A} . Then every strong solution u(t) of (262) with $\mathcal{B} = I$ and $\mathcal{F} = f \in C([0,\tau) : X)$ satisfies

$$(R_2 * f)(t) = (kC_2 * u)(t), \quad 0 \le t < \tau.$$

Furthermore, the problem (262) has at most one pre-solution provided, in addition, that the functions a(t) and k(t) are kernels on $[0, \tau)$ and the function $\mathcal{F}(t)$ is single-valued.

The first part of following theorem is an extension of [292, Theorem 2.1.28(ii)] and its validity can be verified with the help of proof of [395, Theorem 2.7], Lemma 1.2.2 and Theorem 1.2.3; the second part of theorem is an extension of [292, Proposition 2.1.31] and can be shown by the arguments contained in the proof of [438, Theorem 2.5], along with Lemma 1.2.2.

- THEOREM 3.2.9. (i) Suppose that $(R(t))_{t \in [0,\tau)}$ is a locally equicontinuous (a, k)-regularized C-resolvent family generated by \mathcal{A} , the equation (272) holds for each $y = x \in X$, with $R_1(\cdot)$ and C_1 replaced therein with $R(\cdot)$ and C, respectively, k(t) is a kernel on $[0, \tau)$, u, $f \in C([0, \tau) : X)$, and (281) holds with $R_2(\cdot)$ and C_2 replaced therein with $R(\cdot)$ and C, respectively. Then u(t) is a solution of the abstract Volterra inclusion (262) with $\mathcal{B} = I$ and $\mathcal{F} = f$.
- (ii) Suppose that the functions a(t) and k(t) are kernels on $[0, \tau)$, and \mathcal{A} is a closed MLO in X. Consider the following assertions:

- (a) \mathcal{A} is a subgenerator of a locally equicontinuous (a, k)-regularized C-resolvent family $(R(t))_{t \in [0,\tau)}$ satisfying the equation (272) for each $y = x \in X$, with $R_1(\cdot)$ and C_1 replaced therein by $R(\cdot)$ and C, respectively.
- (b) For every $x \in X$, there exists a unique solution of (262) with $\mathcal{B} = I$ and $\mathcal{F}(t) = f(t) = k(t)Cx, t \in [0, \tau)$.

Then (a) \Rightarrow (b). If, in addition, X is a Fréchet space, then the above are equivalent.

It is noteworthy that some additional conditions ensure the validity of implication (b) \Rightarrow (a) in complete locally convex spaces. We will explain this fact for the problem (DFP)_L, where after integration we have $a(t) = g_{\alpha}(t)$. Assume that there exists a unique solution of problem (DFP)_L with $\mathcal{B} = I$, $\mathcal{F}(t) \equiv 0$, $x_0 \in C(D(\mathcal{A}))$ and $x_j = 0$, $1 \leq j \leq \lceil \alpha \rceil - 1$. If, in addition to this, X is complete, \mathcal{A} is closed, $C\mathcal{A} \subseteq \mathcal{A}C$ and for each seminorm $p \in \circledast$ and T > 0 there exist $q \in \circledast$ and c > 0such that $p(u(t; Cx)) \leq cq(x), x \in D(\mathcal{A}), t \in [0, T]$, then the arguments used in non-degenerate case show that \mathcal{A} is a subgenerator of a locally equicontinuous (g_{α}, C) -resolvent family $(R_{\alpha}(t))_{t\geq 0}$. The proof of following complex characterization theorem for (a, k)-regularized C-resolvent families is left to the reader as an easy exercise.

THEOREM 3.2.10. Let $\omega_0 > \max(0, \operatorname{abs}(|a|), \operatorname{abs}(k))$, and let \mathcal{A} be a closed MLO in X. Assume that, for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega_0$ and $\tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0$, the operator $I - \tilde{a}(\lambda)\mathcal{A}$ is injective and $\operatorname{R}(C) \subseteq \operatorname{R}(I - \tilde{a}(\lambda)\mathcal{A})$. If there exists a function $\Upsilon: \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda > \omega_0\} \to L(X)$ which satisfies:

- (i) $\Upsilon(\lambda) = \tilde{k}(\lambda)(I \tilde{a}(\lambda)\mathcal{A})^{-1}C$, $\operatorname{Re} \lambda > \omega_0$, $\tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0$,
- (ii) the mapping $\lambda \mapsto \Upsilon(\lambda)x$, $\operatorname{Re} \lambda > \omega_0$ is analytic for every fixed $x \in X$,
- (iii) there exists $r \ge -1$ such that the family $\{\lambda^{-r}\Upsilon(\lambda) : \operatorname{Re} \lambda > \omega_0\} \subseteq L(X)$ is equicontinuous,

then, for every $\alpha > 1$, \mathcal{A} is a subgenerator of a global $(a, k * g_{\alpha+r})$ -regularized C-resolvent family $(R_{\alpha}(t))_{t \ge 0}$ which satisfies that the family $\{e^{-\omega_0 t}R_{\alpha}(t) : t \ge 0\} \subseteq L(X)$ is equicontinuous. Furthermore, $(R_{\alpha}(t))_{t \ge 0}$ is a mild $(a, k * g_{\alpha+r})$ -regularized C-existence family having \mathcal{A} as subgenerator.

In the first part of following example, we will briefly explain how one can use multiplication operators for construction of local integrated semigroups generated by multivalued operators; in the second part of example, we will apply the complex characterization theorem for proving the existence of a very specific exponentially equicontinuous, convoluted fractional resolvent family.

EXAMPLE 3.2.11. (i) (cf. also [28, Example 4.4(c)]) Suppose that $1 \leq p \leq \infty$, $X := L^p(1,\infty)$, $1 < a < b < \infty$, J := [a,b], $m_b(x) := \chi_J(x)$ and $m_a(x) := x + ie^x$ (x > 1). Consider the multiplication operators $A: D(A) \to X$ and $B \in L(X)$, where $D(A) := \{f(x) \in X : (x+ie^x)f(x) \in X\}$, $Af(x) := (x + ie^x)f(x)$ and $Bf(x) := m_b(x)f(x)$ $(x > 1, f \in X)$. Then it is very simple to prove that, for every $\alpha \in (0, 1)$, the resolvent set of the multivalued linear operator $\mathcal{A} := B^{-1}A$ contains the exponential region $E(\alpha, 1) := \{x + iy : x \ge 1, |y| \le e^{\alpha x}\}$, as well as that $(\lambda - \mathcal{A})^{-1}f(x) = (\lambda B - A)^{-1}Bf(x) = m_b(x)f(x)/\lambda m_b(x) - m_a(x)$ for x > 1, $f \in X$. Furthermore, the operator \mathcal{A} generates a local once integrated semigroup $(S_1(t))_{t\in[0,1]}$, given by

$$(S_1(t)f)(x) = \begin{cases} (x+ie^x)^{-1}[e^{t(x+ie^x)}-1]f(x), & t \in [0,1], \ x \notin J, \ f \in X, \\ 0, & t \in [0,1], \ x \in J, \ f \in X. \end{cases}$$

(ii) Put $X := \{f \in C^{\infty}([0,\infty)) : \lim_{x \to +\infty} f^{(k)}(x) = 0 \text{ for all } k \in \mathbb{N}_0\}$ and $||f||_k := \sum_{j=0}^k \sup_{x \ge 0} |f^{(j)}(x)|, f \in X, k \in \mathbb{N}_0.$ Then the topology induced by these norms turns X into a Fréchet space (cf. also [292, Example 2.4.6(ii)]). Let $\alpha \in (0,1)$ and $J = [a,b] \subseteq [0,\infty)$ be such that $\overline{\Sigma_{\alpha\pi/2}} \cap \{x + ie^x : x \in J\} = \emptyset$, and let $m_b \in C^{\infty}([0,\infty))$ satisfy $0 \leq m_b(x) \leq 1, x \geq 0, m_b(x) = 1, x \notin J \text{ and } m_b(x) = 0, x \in [a + \varepsilon, b - \varepsilon]$ for some $\varepsilon > 0$. As in the first part of this example, we use the multiplication operators $A: D(A) \to X$ and $B \in L(X)$, where $D(A) = \{f(x) \in A\}$ $E: (x + ie^x) f(x) \in X$, $Af(x) := (x + ie^x) f(x)$ and $Bf(x) := m_b(x) f(x)$ $(x \ge 0, f \in X)$. In a recent research study with S. Pilipović and D. Velinov [354], we have shown that A cannot be the generator of any local integrated semigroup in X, as well as that A generates an ultradistribution semigroup of Beurling class. Set $\mathcal{A} := B^{-1}A$. We will prove that there exists a sufficiently large number $\omega > 0$ such that for each s > 1 and d > 0 the operator family $\{e^{-d|\lambda|^{1/s}}(\lambda - \mathcal{A})^{-1} : \operatorname{Re} \lambda > \omega, \ \lambda \in \Sigma_{\alpha \pi/2}\} \subseteq$ L(X) is equicontinuous, which immediately implies by Theorem 3.2.10 that \mathcal{A} generates an exponentially equicontinuous $(q_{\alpha}, \mathcal{L}^{-1}(e^{-d|\lambda|^{\alpha/s}}))$ regularized resolvent family. It is clear that the resolvent of \mathcal{A} will be given by $(\lambda - A)^{-1} f(x) = (\lambda B - A)^{-1} B f(x) = m_b(x) f(x) / \lambda m_b(x) - m_a(x)$ for $x \ge 0, f \in X$. Since $m_b(x)f(x)/\lambda m_b(x) - m_a(x) = 1/\lambda - (x + ie^x)$ for $x \notin J$, our first task will be to estimate the derivatives of function $1/\lambda - (\cdot + ie^{\cdot})$ outside the interval J. In order to do that, observe first that any complex number $\lambda \in \mathbb{C} \setminus S$, where $S := \{x + ie^x : x \ge 0\}$, belongs to the resolvent set of A and

$$(\lambda - A)^{-1}f(x) = \frac{f(x)}{\lambda - (x + ie^x)}, \quad \lambda \in \mathbb{C} \setminus S, \ x \ge 0.$$

Fix, after that, numbers s > 1, d > 0, a > 0, b > 1 satisfying that $x - \ln(((x-b)/a)^s + 1) \ge 1$, $x \ge b$. Set $\Omega := \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda \ge a | \operatorname{Im} \lambda|^{1/s} + b\}$ and denote by Γ the upwards oriented boundary of the region Ω . Inductively, we can prove that for each number $n \in \mathbb{N}$ there exist complex polynomials $P_j(z) = \sum_{l=0}^j a_{j,l} z^l$ $(1 \le j \le n)$ such that $\operatorname{dg}(P_j) = j$, $|a_{j,l}| \le (n+1)!$ $(1 \le j \le n, 0 \le l \le j)$ and

(282)
$$\frac{d^n}{dx^n} (\lambda - (x + ie^x))^{-1} = \sum_{j=1}^n (\lambda - (x + ie^x))^{-j-1} P_j(e^x), \quad x \ge 0, \ \lambda \in \mathbb{C} \smallsetminus S.$$

Suppose $\lambda \in \Omega$ and $x \ge 0$. If $|\operatorname{Im} \lambda - e^x| \ge 1$, then we have the following estimate

(283)
$$\frac{e^{2jx}}{(\operatorname{Re}\lambda - x)^{2k} + (\operatorname{Im}\lambda - e^x)^{2k}} \leqslant \frac{e^{2jx}}{(\operatorname{Im}\lambda - e^x)^{2k}} \\ \leqslant 2^{2j}(1 + |\operatorname{Im}\lambda|)^{2j}, \quad k \in \mathbb{N}_0, \ 0 \leqslant j < k.$$

If $|\operatorname{Im} \lambda - e^x| < 1$, then $\operatorname{Im} \lambda > 0$, $0 \leq x < \ln(\operatorname{Im} \lambda + 1)$, and

(284)
$$\frac{e^{2jx}}{(\operatorname{Re}\lambda - x)^{2k} + (\operatorname{Im}\lambda - e^x)^{2k}} \leqslant \frac{e^{2jx}}{(\operatorname{Re}\lambda - x)^{2k}}$$
$$\leqslant \frac{(\operatorname{Im}\lambda + 1)^{2j}}{\operatorname{Re}\lambda - \ln(((\operatorname{Re}\lambda - b)/a)^s + 1)} \leqslant (\operatorname{Im}\lambda + 1)^{2j}, \quad k \in \mathbb{N}_0, \ 0 \leqslant j < k.$$

Let $\omega' > 0$ be such that $\{\lambda \in \Sigma_{\alpha \pi/2} : \operatorname{Re} \lambda > \omega'\} \subseteq \Omega$. Combining (282)–(284), it can be simply proved that for each number $n \in \mathbb{N}$ there exists a finite constant $c_n > 0$ such that

(285)
$$\sum_{k=0}^{n} \sup_{x \ge 0, x \notin J} \left| \frac{d^n}{dx^n} (\lambda - (x + ie^x))^{-1} \right| \le c_n e^{d|\lambda|^{1/s}}, \quad \lambda \in \Sigma_{\alpha \pi/2}, \text{ Re } \lambda > \omega'.$$

We can similarly prove an estimate of type (285) for the derivatives of function $(\lambda m_b(x) - (x + ie^x))^{-1}$ on the interval J, which is well-defined for $\lambda \in \Sigma_{\alpha \pi/2}$ due to our assumption $0 \leq m_b(x) \leq 1, x \geq 0$ and the condition $\overline{\Sigma_{\alpha \pi/2}} \cap \{x + ie^x : x \in J\} = \emptyset$. In actual fact, induction shows that for each number $n \in \mathbb{N}$ there exist numbers a_{j,l_1,\ldots,l_s} such that $|a_{j,l_1,\ldots,l_s}| \leq (n+1)!$ $(1 \leq j \leq n, 0 \leq l \leq j)$ and

(286)
$$\frac{d^{n}}{dx^{n}} (\lambda m_{b}(x) - (x + ie^{x}))^{-1} = \sum_{j=1}^{n} (\lambda m_{b}(x) - (x + ie^{x}))^{-j-1}$$
$$\times \sum_{l=0}^{j} a_{j,l_{1},...,l_{s}} \prod_{l_{1}m_{1}+\dots+l_{s}m_{s}=n} (\lambda m_{b}^{(l_{j})}(x) - m_{a}^{(l_{j})}(x))^{m_{j}}, \quad x \in J, \ \lambda \in \Sigma_{\alpha\pi/2}.$$
Since $d := \operatorname{dist}(\overline{\Sigma_{\alpha\pi/2}}, \{x + ie^{x} : x \in J\})$ is a positive real number as

Since $d := \operatorname{dist}(\Sigma_{\alpha\pi/2}, \{x + ie^x : x \in J\})$ is a positive real number and $|(\lambda m_b^{(l_j)}(x) - m_a^{(l_j)}(x))^{m_j}| \leq c^{m_j} |\lambda|^{m_j}$ for all $\lambda \in \Sigma_{\alpha\pi/2}$ with $\operatorname{Re} \lambda > \omega$, where the number $\omega > \omega'$ is sufficiently large, (286) shows that for each number $n \in \mathbb{N}$ there exists a finite number $c'_n > 0$ such that

(287)
$$\sum_{k=0}^{n} \sup_{x \ge 0, x \in J} \left| \frac{d^n}{dx^n} (\lambda m_b(x) - (x + ie^x))^{-1} \right| \le c'_n e^{d|\lambda|^{1/s}}, \ \lambda \in \Sigma_{\alpha \pi/2}, \ \operatorname{Re} \lambda > \omega.$$

By (285) and (287), we have that the operator family $\{e^{-d|\lambda|^{1/s}}(\lambda-\mathcal{A})^{-1}: \lambda \in \Sigma_{\alpha\pi/2}, \text{ Re } \lambda > \omega\} \subseteq L(X)$ is equicontinuous, as claimed.

Now we would like to tell something more about the importance of condition $k(0) \neq 0$ in part (ii) of subsequent theorem. If all the necessary requirements hold, the arguments contained in the proof of [285, Theorem 3.6] imply the existence

of a global $(a, k * q_1)$ -regularized C-resolvent family $(R_1(t))_{t>0}$ subgenerated by \mathcal{A} , which additionally satisfies that for each $t \ge 0$ the operator $R_1(t)\mathcal{A}$ is singlevalued on $D(\mathcal{A})$. Then it is necessary to differentiate the equality $R_1(t)x - (k *$ $(q_1)(t)Cx = \int_0^t a(t-s)R_1(s)\mathcal{A}x\,ds, t \ge 0, x \in D(\mathcal{A})$ and employ the fact that $(\frac{d}{dt}R_1(t)x)_{t=0} = k(0)Cx \ (x \in \overline{D(\mathcal{A})})$ (cf. the proof of [285, Theorem 3.6], as well as the proofs of [199, Proposition 2.1] and [292, Proposition 2.1.7]) in order to see that the function $R: D(R) \equiv \{\tilde{a}(\lambda)^{-1} : \lambda > b, \ \tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0\} \to L(\overline{D(\mathcal{A})}),$ given by $R(\tilde{a}(\lambda)^{-1}) := (\tilde{a}(\lambda)^{-1} - \mathcal{A})^{-1}C, \lambda \in D(R)$, is a C-pseudoresolvent in the sense of [384, Definition 3.1], satisfying additionally that $N(R(\lambda)) = \{0\}$, $\lambda \in D(R)$. Only after that, we can use [384, Theorem 3.4] with a view to prove the existence of a single-valued linear operator A, with domain and range contained in $D(\mathcal{A})$, which satisfies the properties required in (ii): this examination shows the full importance of concepts introduced in Definition 3.2.1 and Definition 3.2.2 in integrated and convoluted case k(0) = 0. Keeping in mind Theorem 1.2.4(i) as well as the proofs of [285, Theorem 3.6] and [292, Theorem 1.2.6], the remaining parts of following theorem can be deduced, more or less, as in non-degenerate case.

THEOREM 3.2.12. Suppose $\omega \in \mathbb{R}$, $\operatorname{abs}(k) < \infty$, $\operatorname{abs}(|a|) < \infty$, \mathcal{A} is a closed MLO in X, $\lambda_0 \in \rho_C(\mathcal{A})$, $b \ge \max(0, \omega, \operatorname{abs}(|a|), \operatorname{abs}(k))$,

$$\left\{\frac{1}{\tilde{a}(\lambda)} : \lambda > b, \ \tilde{k}(\lambda)\tilde{a}(\lambda) \neq 0\right\} \subseteq \rho_C(\mathcal{A}),$$

the function $H: D(H) \equiv \{\lambda > b : \tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0\} \rightarrow L(X)$, given by $H(\lambda)x = \tilde{k}(\lambda)(I - \tilde{a}(\lambda)A)^{-1}Cx, x \in X, \lambda \in D(H)$, satisfies that the mapping $\lambda \mapsto H(\lambda)x$, $\lambda \in D(H)$ is infinitely differentiable for every fixed $x \in X$ and, for every $p \in \circledast$, there exist $c_p > 0$ and $r_p \in \circledast$ such that:

(288)
$$p\left(l!^{-1}(\lambda-\omega)^{l+1}\frac{d^l}{d\lambda^l}H(\lambda)x\right) \leqslant c_p r_p(x), \quad x \in X, \ \lambda \in D(H), \ l \in \mathbb{N}_0.$$

Then, for every $r \in (0,1]$, the operator \mathcal{A} is a subgenerator of a global $(a, k * g_r)$ regularized C-resolvent family $(R_r(t))_{t \ge 0}$ satisfying that, for every $p \in \circledast$,

$$p(R_r(t+h)x - R_r(t)x) \leq \frac{2c_p r_p(x)}{r\Gamma(r)} \max(e^{\omega(t+h)}, 1)h^r, \quad t \ge 0, \ h > 0, \ x \in X,$$

and that, for every $p \in \circledast$ and $B \in \mathcal{B}$, the mapping $t \mapsto p_B(R_r(t))$, $t \ge 0$ is locally Hölder continuous with exponent r; furthermore, $(R_r(t))_{t\ge 0}$ is a mild $(a, k * g_r)$ regularized C-existence family having \mathcal{A} as subgenerator, and the following holds:

- (i) Suppose that A is densely defined. Then A is a subgenerator of a global (a, k)-regularized C-resolvent family (R(t))_{t≥0} ⊆ L(X) satisfying that the family {e^{-ωt}R(t) : t ≥ 0} ⊆ L(X) is equicontinuous. Furthermore, (R(t))_{t≥0} is a mild (a, k)-regularized C-existence family having A as subgenerator.
- (ii) Suppose that $k(0) \neq 0$. Then the operator $C' := C_{|\overline{D(A)}|} \in L(\overline{D(A)})$ is injective, A0 is a closed subspace of X, $\overline{D(A)} \cap A0 = \{0\}$, and we have

the following: Define the operator $A: D(A) \subseteq \overline{D(A)} \to \overline{D(A)}$ by $D(A) := \{x \in \overline{D(A)} : Cx = (\lambda_0 - A)^{-1}Cy \text{ for some } y \in \overline{D(A)}\}$ and

$$Ax := C^{-1}\mathcal{A}Cx, \quad x \in D(A).$$

Then A is a well-defined single-valued closed linear operator in $\overline{D(\mathcal{A})}$, and moreover, A is the integral generator of a global (a, k)-regularized C'resolvent family $(S(t))_{t\geq 0} \subseteq L(\overline{D(\mathcal{A})})$ satisfying that the family $\{e^{-\omega t}S(t):$ $t \geq 0\} \subseteq L(\overline{D(\mathcal{A})})$ is equicontinuous, $A \int_0^t a(t-s)S(s)x \, ds = S(t)x$ $k(t)Cx, t \in [0, \tau), x \in \overline{D(\mathcal{A})}$ and $R_1(t)x = \int_0^t S(s)x \, ds, t \geq 0, x \in \overline{D(\mathcal{A})}.$

In the following extension of [395, Proposition 2.5] and [292, Proposition 2.1.4(ii)], we will revisit the condition $k(0) \neq 0$ from Theorem 3.2.12 once more. A straightforward proof is omitted.

PROPOSITION 3.2.13. Let \mathcal{A} be a closed subgenerator of a mild (a, k)-regularized C_1 -resolvent family $(R_1(t))_{t \in [0,\tau)}$ (mild (a, k)-regularized C_2 -uniqueness family $(R_2(t))_{t \in [0,\tau)}$; (a, k)-regularized C-resolvent family $(R(t))_{t \in [0,\tau)}$). If k(t) is absolutely continuous and $k(0) \neq 0$, then \mathcal{A} is a subgenerator of a mild (a, g_1) -regularized C_1 -resolvent family $(R_1(t))_{t \in [0,\tau)}$ (mild (a, g_1) -regularized C_2 -uniqueness family $(R_2(t))_{t \in [0,\tau)}$; (a, g_1) -regularized C-resolvent family $(R(t))_{t \in [0,\tau)}$).

Now we would like to present some illustrative applications of results obtained so far.

EXAMPLE 3.2.14. Let $\alpha \in (0, 1)$.

(i) [199] Consider the following time-fractional analogue of homogeneous counterpart of problem [199, Example 2.1, (2.18)]:

$$(P)_{m,\alpha}: \begin{cases} \mathbf{D}_t^{\alpha}[m(x)v_{\alpha}(t,x)] = -\frac{\partial}{\partial x}v_{\alpha}(t,x), & t \ge 0, \ x \in \mathbb{R}; \\ m(x)v_{\alpha}(0,x) = u_0(x), & x \in \mathbb{R}. \end{cases}$$

Let $X = Y := L^2(\mathbb{R})$, and let the operator A := -d/dx act on X with its maximal distributional domain $H^1(\mathbb{R})$.

(a) Suppose first that $(Bf)(x) := \chi_{(-\infty,a)\cap(b,\infty)}(x)f(x), x \in \mathbb{R} \ (f \in X),$ where $-\infty < a < b < \infty$. Then $B \in L(X), B = B^*, B^2 = B$ and $(P)_{m,\alpha}$ is formulated in X in the following abstract form

$$(P)'_{m,\alpha}:\begin{cases} B^*\mathbf{D}_t^{\alpha}Bv_{\alpha}(t)=\mathbf{D}_t^{\alpha}Bv_{\alpha}(t)=Av_{\alpha}(t), & t \ge 0;\\ Bv_{\alpha}(0)=u_0. \end{cases}$$

Further on, the multivalued linear operator $\mathcal{A} := (B^*)^{-1}AB^{-1}$ is maximal dissipative in the sense of [199, Definition, p. 35] and $\|(\lambda - \mathcal{A})^{-1}\| \leq \lambda^{-1}, \lambda > 0$. By the foregoing, we know that the operator \mathcal{A} is single-valued on $\overline{D(\mathcal{A})}$; with a little abuse of notation, we will denote by $T \subseteq \mathcal{A}$ the single-valued linear operator which generates a bounded strongly continuous semigroup $(T(t))_{t\geq 0}$ on $\overline{D(\mathcal{A})}$ (cf. Theorem 3.2.12(ii), where we have denoted this operator by A). Using [285, Theorem 3.6(a)] and the consideration from the paragraph directly preceding the formulation of [199, Theorem 2.8], it readily follows that D(T) = D(A). Suppose now that $u_0 = Bv_0$, where $v_0 \in D(A)$ and $Av_0 \in R(B^*)$, i.e., that $u_0 \in D(A) = D(T)$ (cf. the proof of [199, Theorem 2.10]). Due to [199, Theorem 2.8, Theorem 2.10], the problem $(P)'_{m,1}$, with $\alpha = 1$, has a unique solution $v_1(t)$ satisfying $Bv_1(t) = T(t)u_0$; moreover,

$$(d/dt)Bv_1(t) = B^*(d/dt)Bv_1(t) = Av_1(t) = (d/dt)T(t)u_0 = T(t)Tu_0, \quad t \ge 0.$$

Since the condition [199, (2.14)] holds, we get that there exists $\lambda_0 > 0$ such that $(\lambda_0 B - A)^{-1} \in L(X)$; hence, $v_1(\cdot) = (\lambda_0 B - A)^{-1}(\lambda_0 B - A)v_1(\cdot) \in C([0,\infty):X)$ is bounded, as well as $(d/dt)Bv_1(t)$, $Bv_1(t)$ and $Av_1(t)$ are continuous and bounded for $t \ge 0$. Define $v_\alpha(t) := \int_0^\infty t^{-\alpha} \Phi_\alpha(st^{-\alpha})v_1(s)ds$, t > 0 and $v_\alpha(0) := v_1(0)$. Using Theorem 3.1.8 and the arguments contained in its proof, it readily follows that the function $v_\alpha(\cdot)$ is a bounded solution of problem $(P)'_{m,\alpha}$, satisfying in addition that the functions $t \mapsto v_\alpha(\cdot)$, t > 0 and $t \mapsto Av_\alpha(\cdot)$, t > 0 can be analytically extended to the sector $\sum_{\min((\frac{1}{\alpha}-1)\frac{\pi}{2},\pi)}$. The uniqueness of solutions of problem $(P)'_{m,\alpha}$ can be proved with the help of Theorem 3.1.6.

- (b) Suppose now that $(Bf)(x) := \chi_{(a,\infty)}(x)f(x), x \in \mathbb{R} \ (f \in X)$, where $-\infty < a < \infty$. Then $B \in L(X)$, $B = B^*$, $B^2 = B$ and the conclusions established in part (a) of this example, ending with the equation (289), continue to hold. In our concrete situation, we have the validity of condition [199, (2.11)] but not the condition [199, (2.14)], in general. Define $f_{\alpha}(t) := \int_{0}^{\infty} t^{-\alpha} \Phi_{\alpha}(st^{-\alpha}) Bv_{1}(s) ds, t > 0$, $f_{\alpha}(0) := Bv_1(0) = u_0, h_{\alpha}(t) := \int_0^\infty t^{-\alpha} \Phi_{\alpha}(st^{-\alpha}) Av_1(s) ds, t > 0$ and $h_{\alpha}(0) := Av_1(0)$. By the foregoing, we have that $f_{\alpha}, h_{\alpha} \in C([0, \infty))$: X) are bounded and $\mathbf{D}_t^{\alpha} f_{\alpha}(t) = h_{\alpha}(t), t \ge 0$, which simply implies $Bh_{\alpha}(t) = h_{\alpha}(t), t \ge 0$. By (289), we have that $Av_1(t) = T(t)Tu_0 \in$ $B^{-1}[Av_1(t)]$ and $BAv_1(t) = Av_1(t)$ $(t \ge 0)$, whence we may conclude that $Av_1(t) \in \mathcal{A}[Bv_1(t)]$ $(t \ge 0)$. Since \mathcal{A} is closed, an application of Theorem 1.2.3 yields that $h_{\alpha}(t) = Bh_{\alpha}(t) \in AB^{-1}f_{\alpha}(t) \ (t \ge 0)$; consequently, the function $t \mapsto f_{\alpha}(t), t \ge 0$ is a pre-solution of problem $(DFP)_R$ with $B \equiv I$, $\mathcal{F}(t) \equiv 0$ and, by Remark 3.1.2(iv), the problem $(P)'_{m,\alpha}$ has a bounded p-solution $v_{\alpha}(\cdot)$ satisfying, in addition, that the functions $t \mapsto Bv_{\alpha}(\cdot), t > 0$ and $t \mapsto Av_{\alpha}(\cdot), t > 0$ can be analytically extended to the sector $\sum_{\min((\frac{1}{\alpha}-1)\frac{\pi}{2},\pi)}$. The uniqueness follows again from an essential application of Theorem 3.1.6.
- (ii) [280,285] Here we would like to observe, without going into full details, that we can similarly prove some results on the existence and uniqueness

of analytical solutions of the abstract Volterra equation

$$\frac{\partial}{\partial r}v_{\alpha}(t,r) = a(r)\int_{0}^{t}g_{\alpha}(t-s)v_{\alpha}(s,r)ds + f(t,r), \quad t \ge 0, \ r \in [0,1],$$

on the sector $\Sigma_{\min((\frac{1}{\alpha}-1)\frac{\pi}{2},\pi)}$, where $a \in C^1[0,1]$ and the mapping $t \mapsto f(t,\cdot)$, $t \ge 0$ is continuous and exponentially bounded with the values in the Banach space C[0,1] (cf. [280, Example 1] and Theorem 3.1.8); using Theorem 3.1.9 instead of Theorem 3.1.8, we can consider the well-posedness in C[0,1] for the equation

$$\frac{\partial}{\partial r}v_c(t,r) = a(r)\int_0^t c(t-s)v_c(s,r)ds + f(t,r), \quad t \ge 0, \ r \in [0,1],$$

where $c(\cdot)$ is a completely positive function.

(iii) Fractional Maxwell's equations have gained much attention in recent years (see e.g. [128, 253, 404, 516, 562] and references cited therein for more details on the subject). Here we want to briefly explain how we can use the analysis of A. Favini and A. Yagi [199, Example 2.2] for proving the existence and uniqueness of analytical solutions of certain classes of inhomogeneous abstract time-fractional Maxwell's equations in ℝ³; the timefractional analogues of Poisson-wave equations (see e.g. [199, Example 2.3, Example 6.23]) will be considered somewhere else.

Consider the following abstract time-fractional Maxwell's equations:

$$\operatorname{rot} E = -\mathbf{D}_t^{\alpha} B, \quad \operatorname{rot} H = \mathbf{D}_t^{\alpha} D + J$$

in \mathbb{R}^3 , where E (resp. H) denotes the electric (resp. magnetic) field intensity, B (resp. D) denotes the electric (resp. magnetic) flux density, and where J is the current density. It is assumed that the medium which fills the space \mathbb{R}^3 is linear but possibly anisotropic and nonhomogeneous, which means that $D = \varepsilon E$, $B = \mu H$ and $J = \sigma E + J'$ with some 3×3 real matrices $\varepsilon(x)$, $\mu(x)$, $\sigma(x)$ ($x \in \mathbb{R}^3$) and J' being a given forced current density. Let any component of $\varepsilon(x)$, $\mu(x)$, $\sigma(x)$ be a bounded, measurable function in \mathbb{R}^3 , let the conditions [199, (2.23)–(2.25)] hold, and let $f(t) = -(J'(\cdot, t) \ 0)^T$. Then we can formulate the problem (290) in the following abstract form

$$(P)_1: \begin{cases} B^* \mathbf{D}_t^{\alpha} B v_1(t) = A v_1(t) + f(t), & t \ge 0; \\ B v_1(0) = u_0, \end{cases}$$

in the space $X := \{L^2(\mathbb{R}^3)\}^6$, using the bounded self-adjoint operator B of multiplication by $\sqrt{c(x)}$ acting in X, and A being the closed linear operator in X given by [199, (2.27)]. In our concrete situation, the conditions [199, (2.10) and (2.14)] hold, so that the assumptions $f \in C^2([0,\infty) : X)$ and $u_0 = Bv_0$ for some $v_0 \in D(A)$ satisfying $Av_0 + f(0) \in R(B^*)$ ensure by [199, Corollary 2.11] that the problem $(P)_1$ has a unique strict solution $v_1(\cdot)$ in the sense of equation [199, (2.13)]. Suppose, additionally, that the function f''(t) is exponentially bounded.

Then we can use [199, Theorem 2.5], the proof of [199, Corollary 2.11] and the arguments from part (i)/(a) of this example in order to see that the solution $v_1 \in C([0,\infty): X)$ is exponentially bounded, as well as that $H(t) := (d/dt)Bv_1(t), Bv_1(t)$ and $Av_1(t)$ are continuous and exponentially bounded for $t \ge 0$. Define $v_\alpha(t)$ and $f_\alpha(t)$ as before, $H_\alpha(t) :=$ $\int_0^\infty t^{-\alpha} \Phi_\alpha(st^{-\alpha})H(s)ds, t > 0$ and $H_\alpha(0) := H(0)$. Performing the Laplace transform, it can be simply verifed that $(g_{1-\alpha} * (Bv_\alpha - u_0))(t) =$ $\int_0^t H_\alpha(s)ds, t \ge 0$, so that $\mathbf{D}_t^\alpha Bv_\alpha(t)$ exists and equals to $H_\alpha(t)$. On the other hand, we have $B^*Bv_1(t) = A(g_1 * v_1)(t) + B^*u_0 + \int_0^t f(s)ds, t \ge 0$, so that $B^*Bv_\alpha(t) = A(g_\alpha * v_\alpha)(t) + B^*u_0 + \int_0^t f_\alpha(s)ds, t \ge 0$ by Theorem 3.1.8. This implies $\mathbf{D}_t^\alpha B^*Bv_\alpha(t) = Av_\alpha(t) + f_\alpha(t)$ and, since $\mathbf{D}_t^\alpha Bv_\alpha(t)$ exists, $B^*\mathbf{D}_t^\alpha Bv_\alpha(t) = Av_\alpha(t) + f_\alpha(t), t \ge 0$. Clearly, $Bv_\alpha(0) = u_0$ so that $v_\alpha \in C([0,\infty): X)$ is an exponentially bounded solution of problem

$$(P)_{\alpha}:\begin{cases} B^* \mathbf{D}_t^{\alpha} B v_{\alpha}(t) = A v_{\alpha}(t) + f_{\alpha}(t), & t \ge 0; \\ B v_{\alpha}(0) = u_0, \end{cases}$$

that is analytically extensible on the sector $\sum_{\min((\frac{1}{\alpha}-1)\frac{\pi}{2},\pi)}$ and satisfies, in addition, that the mapping $Av_{\alpha} \in C([0,\infty):X)$ is exponentially bounded and analytically extensible on the same sector, as well. The uniqueness of solutions of problem (P_{α}) follows from Theorem 3.1.6.

We end this example with the observation that Theorem 3.1.8 and Theorem 3.1.9 can be successfully applied in the analysis of a large class of abstract degenerate Volterra integro-differential equations that are sub-ordinated, in a certain sense, to degenerate differential equations of first and second order for which we know that are well posed ([199,205,210, 475,509,517,518]).

Concerning the adjoint type theorems, it should be noticed that the assertions of [292, Theorem 2.1.12(i)/(ii); Theorem 2.1.13] continue to hold for (a, k)regularized *C*-resolvent families subgenerated by closed multivalued linear operators. Furthermore, it is not necessary to assume that the operator \mathcal{A} is densely defined if analyzing [292, Theorem 2.1.12(i)].

Suppose now that \mathcal{A} is a subgenerator of an (a, k)-regularized C-resolvent family $(R(t))_{t \in [0,\tau)}$, $n \in \mathbb{N}$ and $x_j \in \mathcal{A}x_{j-1}$ for $1 \leq j \leq n$. Then we can prove inductively that, for every $t \in [0, \tau)$,

(291)
$$R(t)x = k(t)Cx_0 + \sum_{j=1}^{n-1} (a^{*,j} * k)(t)Cx_j + (a^{*,n} * R(\cdot)x_n)(t).$$

Keeping in mind the identity (291), Theorem 1.2.3 and Proposition 3.2.8(ii), it is almost straightforward to transfer the assertion of [**292**, Proposition 2.1.32] to degenerate case:

PROPOSITION 3.2.15. (i) Suppose $\alpha \in (0, \infty) \setminus \mathbb{N}$, $x \in D(\mathcal{A})$ as well as $C^{-1}f$, $f_{\mathcal{A}} \in C([0, \tau) : X)$, $f_{\mathcal{A}}(t) \in \mathcal{A}C^{-1}f(t)$, $t \in [0, \tau)$ and \mathcal{A} is a closed subgenerator of a (g_{α}, C) -regularized resolvent family $(R(t))_{t \in [0, \tau)}$. Set

 $v(t) := (g_{\lceil \alpha \rceil - \alpha} * f)(t), t \in [0, \tau).$ If $v \in C^{\lceil \alpha \rceil - 1}([0, \tau) : X)$ and $v^{(k)}(0) = 0$ for $1 \leq k \leq \lceil \alpha \rceil - 2$, then the function $u(t) := R(t)x + (R * C^{-1}f)(t),$ $t \in [0, \tau)$ is a unique solution of the following abstract time-fractional inclusion:

$$\begin{cases} u \in C^{\lceil \alpha \rceil}((0,\tau):X) \cap C^{\lceil \alpha \rceil - 1}([0,\tau):X), \\ \mathbf{D}_t^{\alpha} u(t) \in \mathcal{A}u(t) + \frac{d^{\lceil \alpha \rceil - 1}}{dt^{\lceil \alpha \rceil - 1}}(g_{\lceil \alpha \rceil - \alpha} * f)(t), \quad t \in [0,\tau), \\ u(0) = Cx, \ u^{(k)}(0) = 0, \quad 1 \leqslant k \leqslant \lceil \alpha \rceil - 1. \end{cases}$$

(ii) Suppose $r \ge 0$, $n \in \mathbb{N} \setminus \{1\}$, $x_0 = x$, $f_0(\cdot) = f(\cdot)$, $x_j \in Ax_{j-1}$ for $1 \le j \le n$, $f_j(t) \in \mathcal{A}f_{j-1}(t)$ for $t \in [0, \tau)$ and $1 \le j \le n$, $f_j \in C([0, \tau) : X)$ for $0 \le j \le n$, and \mathcal{A} is a closed subgenerator of a $(g_{1/n}, g_{r+1})$ -regularized C-resolvent family $(R(t))_{t \in [0, \tau)}$. Then the function $v(t) := R(t)x + (R * C^{-1}f)(t)x$, $t \in [0, \tau)$ is a unique solution of the following abstract time-fractional inclusion:

$$\begin{cases} v \in C^{1}((0,\tau):X) \cap C([0,\tau):X), \\ v'(t) \in \mathcal{A}v(t) + \sum_{j=1}^{n-1} g_{(j/n)+r}(t)Cx_{j} + \sum_{j=0}^{n-1} (g_{(j/n)+r} * f_{j})(t) \\ + \frac{d}{dt}g_{r+1}(t)Cx, \quad t \in (0,\tau), \end{cases}$$

$$v(0) = g_{r+1}(0)Cx.$$

Furthermore, $v \in C^1([0, \tau) : X)$ provided that $r \ge 1$ or x = 0 and $r \ge 0$.

3.2.1. Differential and analytical properties of (a, k)-regularized *C*-resolvent families. The main structural characterizations of differential and analytical (a, k)-regularized *C*-resolvent families generated by single-valued linear operators continue to hold in our framework (cf. [75, 76, 199, Chapter III] and [182] for some references on infinitely differentiable semigroups generated by MLOs).

The notions of various types of analyticity of degenerate (a, k)-regularized C-resolvent families are introduced in the following definition.

DEFINITION 3.2.16. (cf. [292, Definition 2.2.1] for non-degenerate case)

(i) Suppose that \mathcal{A} is an MLO in X. Let $\alpha \in (0, \pi]$, and let $(R(t))_{t \ge 0}$ be an (a, k)-regularized C-resolvent family which do have \mathcal{A} as a subgenerator. Then it is said that $(R(t))_{t \ge 0}$ is an analytic (a, k)-regularized C-resolvent family of angle α , if there exists a function $\mathbf{R} : \Sigma_{\alpha} \to L(X)$ which satisfies that, for every $x \in X$, the mapping $z \mapsto \mathbf{R}(z)x, z \in \Sigma_{\alpha}$ is analytic as well as that:

(a) $\mathbf{R}(t) = R(t), t > 0$ and

- (b) $\lim_{z\to 0, z\in\Sigma_{\gamma}} \mathbf{R}(z)x = R(0)x$ for all $\gamma \in (0, \alpha)$ and $x \in X$.
- (ii) Let $(R(t))_{t \ge 0}$ be an analytic (a, k)-regularized *C*-resolvent family of angle $\alpha \in (0, \pi]$. Then it is said that $(R(t))_{t \ge 0}$ is an exponentially equicontinuous, analytic (a, k)-regularized *C*-resolvent family of angle α , resp. equicontinuous analytic (a, k)-regularized *C*-resolvent family of angle α , if for every $\gamma \in (0, \alpha)$, there exists $\omega_{\gamma} \ge 0$, resp. $\omega_{\gamma} = 0$, such that the

family $\{e^{-\omega_{\gamma} \operatorname{Re} z} \mathbf{R}(z) : z \in \Sigma_{\gamma}\} \subseteq L(X)$ is equicontinuous. Since there is no risk for confusion, we will identify in the sequel $R(\cdot)$ and $\mathbf{R}(\cdot)$.

In the following example, we will treat a time-fractional analogue of the linearized fractional Benney–Luke equation in L^2 -spaces and there we will meet some interesting examples of exponentially bounded, analytic fractional resolvent families of bounded operators whose angle of analyticity can be strictly greater than $\pi/2$; in our approach, we do not use neither multivalued linear operators nor relatively *p*-radial operators [**199**, **509**]. The method employed by G. A. Sviridyuk and V. E. Fedorov [**509**] for the usually considered Benney–Luke equation of first order can be very hepful for achieving the final conclusions stated in (i)–(ii), as well as for the concrete choice of the state space X_0 below (cf. also Example 2.3.49 and Example 2.3.53).

EXAMPLE 3.2.17. Suppose that $\emptyset \neq \Omega \subseteq \mathbb{R}^n$ is a bounded domain with smooth boundary, and Δ is the Dirichlet Laplacian in $X := L^2(\Omega)$, acting with domain $H^2(\Omega) \cap H_0^1(\Omega)$. By $\{\lambda_k\} [= \sigma(\Delta)]$ we denote the eigenvalues of Δ in $L^2(\Omega)$ (recall that $0 < -\lambda_1 \leq -\lambda_2 \ldots \leq -\lambda_k \leq \cdots \to +\infty$ as $k \to \infty$) numbered in nonascending order with regard to multiplicities. By $\{\phi_k\} \subseteq C^{\infty}(\Omega)$ we denote the corresponding set of mutually orthogonal eigenfunctions. Then, for every $\zeta > 0$, we define the spectral fractional power $C_{\zeta} \in L(X)$ of $-\Delta$ by $C_{\zeta} := (-\Delta)^{-(\zeta)/2} :=$ $\sum_{k \geq 1} \langle \cdot, \phi_k \rangle (-\lambda_k)^{-(\zeta/2)} \phi_k$ (cf. [484] for more details). Then C_{ζ} is injective and $R(C_{\zeta}) =: D((-\Delta)^{\zeta/2}) = \{f \in L^2(\Omega) : \sum_{k \geq 1} |\langle f, \phi_k \rangle|^2 (-\lambda_k)^{\zeta} < \infty\}$. Let $\lambda \in \sigma(\Delta)$, let $0 < \eta \leq 2$, and let $\alpha, \beta > 0$. Consider the following time-fractional analogue of the linearized Benney–Luke equation:

$$(P)_{\eta,f}:\begin{cases} (\lambda-\Delta)\mathbf{D}_t^{\eta}u(t,x) = (\alpha\Delta-\beta\Delta^2)u(t,x) + f(t,x), & t \ge 0, \ x \in \Omega, \\ \left(\frac{\partial^k}{\partial t^k}u(t,x)\right)_{t=0} = u_k(x), & x \in \Omega, \ 0 \le k \le \lceil \eta \rceil - 1, \\ u(t,x) = \Delta u(t,x) = 0, \quad t \ge 0, \ x \in \partial\Omega. \end{cases}$$

Denote by X_0 the vector space of those functions from X that are orthogonal to the eigenfunctions $\phi_k(\cdot)$ for $\lambda_k = \lambda$. Then X_0 is a closed subspace of X, and therefore, becomes the Banach space equipped with the topology inherited by the X-norm (cf. [509, Example 5.3.1, Theorem 5.3.2] for the case $\eta = 1$). On the other hand, the operators $A := \alpha \Delta - \beta \Delta^2$ and $B := \lambda - \Delta$, acting with maximal domains, are closed in $L^2(\Omega)$. Set $\theta := \min((\pi/\eta) - (\pi/2), \pi)$. Using the Parseval equality and the asymptotic expansion formulae for Mittag–Leffler functions, we can simply prove that the operator family $(T_\eta(z))_{z \in \Sigma_\theta \cup \{0\}} \subseteq L(X_0)$, given by

$$t \mapsto T_{\eta}(z) \cdot := \sum_{k \mid \lambda_k \neq \lambda} E_{\eta} \Big(\frac{\alpha \lambda_k - \beta \lambda_k^2}{\lambda - \lambda_k} z^{\eta} \Big) \langle \cdot, \phi_k \rangle \phi_k, \quad z \in \Sigma_{\theta} \cup \{0\},$$

is well-defined, provided $\eta \in (0, 2)$. If $\eta = 2$, then we define $(T_2(t))_{t \ge 0} \subseteq L(X_0)$ in the same way as above; since $E_2(z^2) = \cosh(z)$, we have that, for every $t \ge 0$,

$$T_2(t) \cdot = \frac{1}{2} \sum_{k|\lambda_k \neq \lambda} \left[e^{it((\beta\lambda^2 - \alpha\lambda_k)/(\lambda - \lambda_k))^{1/2}} + e^{-it((\beta\lambda^2 - \alpha\lambda_k)/(\lambda - \lambda_k))^{1/2}} \right] \langle \cdot, \phi_k \rangle \phi_k$$

and that $(T_2(t))_{t\geq 0}$ is bounded in the uniform operator norm. Differentiating $T_2(t)$ term by term, it can be easily seen that the mapping $t \mapsto T_2(t) f$, $t \ge 0$ is continuously differentiable for any $f \in D((-\Delta)^{1/2}) \cap X_0$, and therefore, continuous. Since $D((-\Delta)^{1/2}) \cap X_0$ is dense in X_0 and $(T_2(t))_{t\geq 0}$ is bounded, we have that $(T_2(t))_{t\geq 0}$ is strongly continuous. Now we can proceed as in the proof of Theorem 3.1.8 in order to see that, for every $\eta \in (0,2)$, $(T_n(t))_{t\geq 0}$ is an exponentially bounded, analytic (q_n, I) -regularized resolvent family of angle θ . A straightforward computation shows that, for every $\eta \in (0,2]$, the integral generator \mathcal{A} of $(T_n(t))_{t\geq 0}$ is a closed single-valued operator in X_0 , given by $\mathcal{A} = \{(f,g) \in X_0 \times X_0 : (\lambda - \lambda_k) \langle g, \phi_k \rangle =$ $(\alpha \lambda_k - \beta \lambda_k^2) \langle f, \phi_k \rangle$ for all $k \in \mathbb{N}$ with $\lambda_k \neq \lambda$; in particular, \mathcal{A} is an extension of the operator $B^{-1}A_{|X_0}$. It is also clear that $(T_\eta(t))_{t\geq 0}$ is a mild (g_η, I) -existence family generated by \mathcal{A} . Keeping in mind the identity [61, (1.25)], we can directly compute that the homogeneous counterpart of problem $(P)_{n,f} \equiv (P)_{n,0}$, with $x_i = 0$ for $1 \leq j \leq [\zeta] - 1$, has an exponentially bounded pre-solution $u_{h,0}(t) = T_n(t)x_0$, $t \ge 0$ for any $x_k \in D(A) \cap X_0$ $(0 \le k \le \lceil \eta \rceil - 1)$, which seems to be an optimal result in the case that $\eta \leq 1$. Concerning the homogeneous counterpart of problem $(P)_{\eta,0}$ with $x_0 = 0$, its solution $u_{h,1}(t)$ has to be found in the form $u_{h,1}(t) = \int_0^t T_\eta(s) x_1 ds$, $t \ge 0$. Consider first the case $\eta \in (1, 2)$. Then for each $k \in \mathbb{N}$ with $\lambda_k \ne \lambda$, we have

$$\frac{d^2}{dt^2} \Big[g_{2-\eta} * \left(E_\eta \left(\frac{\alpha \lambda_k - \beta \lambda_k^2}{\lambda - \lambda_k} \cdot^\eta \right) - 1 \right) \Big] (t) \\ = \mathbf{D}_t^\eta E_\eta \left(\frac{\alpha \lambda_k - \beta \lambda_k^2}{\lambda - \lambda_k} t^\eta \right) = \frac{\alpha \lambda_k - \beta \lambda_k^2}{\lambda - \lambda_k} E_\eta \left(\frac{\alpha \lambda_k - \beta \lambda_k^2}{\lambda - \lambda_k} t^\eta \right), \quad t \ge 0.$$

On the other hand, expanding the function $E_{\eta}(\frac{\alpha\lambda_k-\beta\lambda_k^2}{\lambda-\lambda_k}\cdot^{\eta})-1$ in a power series we get that

$$\frac{d}{dt} \Big[g_{2-\eta} * \Big(E_\eta \Big(\frac{\alpha \lambda_k - \beta \lambda_k^2}{\lambda - \lambda_k} \cdot^\eta \Big) - 1 \Big) \Big](t) = t \sum_{n=0}^{\infty} \frac{(\frac{\alpha \lambda_k - \beta \lambda_k^2}{\lambda - \lambda_k} t^\eta)^{n+1} t^{n\eta}}{\Gamma(n\eta + 2)}, \quad t \ge 0.$$

The previous two equalities together imply that $\frac{d}{dt}[g_{2-\eta}*(E_{\eta}(\frac{\alpha\lambda_{k}-\beta\lambda_{k}^{2}}{\lambda-\lambda_{k}}\cdot^{\eta})-1)](t) = \frac{\alpha\lambda_{k}-\beta\lambda_{k}^{2}}{\lambda-\lambda_{k}}\int_{0}^{t}E_{\eta}(\frac{\alpha\lambda_{k}-\beta\lambda_{k}^{2}}{\lambda-\lambda_{k}}s^{\eta})ds, t \ge 0$ and

$$\mathbf{D}_{t}^{\eta}[g_{1}*T_{\eta}(\cdot)x_{1}](t) = \sum_{k|\lambda_{k}\neq\lambda} \frac{\alpha\lambda_{k} - \beta\lambda_{k}^{2}}{\lambda - \lambda_{k}} \int_{0}^{t} E_{\eta}\Big(\frac{\alpha\lambda_{k} - \beta\lambda_{k}^{2}}{\lambda - \lambda_{k}}s^{\eta}\Big) ds\langle x_{1}, \phi_{k}\rangle\phi_{k}$$
$$= \sum_{k|\lambda_{k}\neq\lambda} \frac{\alpha\lambda_{k} - \beta\lambda_{k}^{2}}{\lambda - \lambda_{k}} tE_{\eta,2}\Big(\frac{\alpha\lambda_{k} - \beta\lambda_{k}^{2}}{\lambda - \lambda_{k}}t^{\eta}\Big)\langle x_{1}, \phi_{k}\rangle\phi_{k}, \quad t \ge 0.$$

Using again the asymptotic expansion formulae for Mittag-Leffler functions, we obtain that the above series converges for any $x_1 \in X_0$ and belongs to D(B) provided, in addition, that $x_1 \in D(B) \cap X_0$. In this case, the equality $B\mathbf{D}_t^{\eta}u_{h,1}(t) = Au_{h,1}(t), t \ge 0$ readily follows, so that the function $u_h(t) := u_{h,0}(t) + u_{h,1}(t), t \ge 0$ is a pre-solution of problem $(DFP)_L$ provided that $x_0 \in D(A) \cap X_0$ and $x_1 \in D(B) \cap X_0$ (with $X = Y = L^2(\Omega)$ in Definition 3.1.1(iii)); furthermore, the

mappings $t \mapsto u_h(t) \in L^2(\Omega)$, t > 0 and $t \mapsto Bu_h(t) \in L^2(\Omega)$, t > 0 can be analytically extended to the sector Σ_{θ} . The situation is slightly different in the case that $\eta = 2$; then a simple calculus shows that, formally,

$$Bu_{h,1}''(t) = Au_{h,1}(t) = \frac{1}{2} \sum_{k|\lambda_k \neq \lambda} [i((\beta\lambda^2 - \alpha\lambda_k)/(\lambda - \lambda_k))^{1/2} e^{it((\beta\lambda^2 - \alpha\lambda_k)/(\lambda - \lambda_k))^{1/2}} - i((\beta\lambda^2 - \alpha\lambda_k)/(\lambda - \lambda_k))^{1/2} e^{-it((\beta\lambda^2 - \alpha\lambda_k)/(\lambda - \lambda_k))^{1/2}}]\langle x_1, \phi_k \rangle \phi_k, \quad t \ge 0$$

Hence, the function $u_h(t) := u_{h,0}(t) + u_{h,1}(t), t \ge 0$ is a pre-solution of problem $(DFP)_L$ with $x_0 \in D(A) \cap X_0$ and $x_1 \in D((-\Delta)^{3/2}) \cap X_0$. The range of any pre-solution of problem $(P)_{\eta,0}$ must be contained in X_0 , so that the uniqueness of solutions of problem $(P)_{\eta,f}$ follows from its linearity and Proposition 3.2.8(ii).

Before dealing with the inhomogeneous problem $(P)_{\eta,f}$, we would like to observe that the assumptions $(x, y) \in \mathcal{A}$ and $x \in D(\mathcal{A})$ imply $(x, y) \in B^{-1}A_{|X_0}$. Keeping in mind this remark, Theorem 1.2.3, as well as the fact that the assertion of [459, Proposition 2.1(iii)] admits a reformulation in our framework, we can simply prove that for any function $h \in W_{loc}^{1,1}([0,\infty):X_0)$ satisfying that

(292)
$$t \mapsto \sum_{k|\lambda_k \neq \lambda} (\alpha \lambda_k - \beta \lambda_k^2) \Big\langle \frac{d}{dt} (g_\eta * h)(t), \phi_k \Big\rangle \phi_k \in L^1_{loc}([0,\infty) : X_0),$$

the function $u_{Bh}(t) := \int_0^t T_\eta(t-s) \frac{d}{ds} (g_\eta * h) ds$, $t \ge 0$ is a solution of problem $(P)_{\eta,Bh}$. On the other hand, the operator B annihilates any function from $span\{\phi_k: k|\lambda = \lambda_k\}$ so that the function $t \mapsto \sum_{k|\lambda_k=\lambda} \frac{\langle f(t),\phi_k \rangle}{\beta \lambda_k^2 - \alpha \lambda_k} \phi_k$, $t \ge 0$ is a presolution of problem $(P)_{\eta,\sum_{k|\lambda_k=\lambda} \langle f(\cdot),\phi_k \rangle \phi_k}$, provided that the following condition holds:

(Q) : $\mathbf{D}_{t}^{\eta}\langle f(t), \phi_{k}\rangle$ exists in $L^{2}(\Omega)$ for $k|\lambda = \lambda_{k}, \langle x_{0}, \phi_{k}\rangle = 0$ for $k|\lambda \neq \lambda_{k}, \langle x_{1}, \phi_{k}\rangle = 0$ for $k|\lambda \neq \lambda_{k}, 1 < \eta \leq 2, \langle x_{0}, \phi_{k}\rangle = \frac{\langle f(0), \phi_{k}\rangle}{\beta\lambda_{k}^{2} - \alpha\lambda_{k}}$ for $k|\lambda = \lambda_{k}, \lambda_{k}, \lambda_{k}, \lambda_{k} = \frac{\langle f'(0), \phi_{k}\rangle}{\beta\lambda_{k}^{2} - \alpha\lambda_{k}}$ for $k|\lambda = \lambda_{k}, \lambda_{k}, \lambda_{k} = \lambda_{k}, \lambda_{k}$ and $\langle x_{1}, \phi_{k}\rangle = \frac{\langle f'(0), \phi_{k}\rangle}{\beta\lambda_{k}^{2} - \alpha\lambda_{k}}$ for $k|\lambda = \lambda_{k}, \lambda_{k}, \lambda_{k} = \lambda_{k}$.

Summa summarum, we have the following:

- (i) $0 < \eta < 2$: Suppose that $x_0 \in D(A) \cap X_0$, $x_1 \in D(B) \cap X_0$, if $\eta > 1$, $\sum_{k|\lambda_k \neq \lambda} \frac{\langle f(\cdot), \phi_k \rangle}{\lambda \lambda_k} \phi_k = h \in W^{1,1}_{loc}([0,\infty) : X_0)$ satisfies (292), and the condition (Q) holds. Then there exists a unique pre-solution of problem $(P)_{\eta,f}$.
- (ii) $\eta = 2$: Suppose $x_1 \in D((-\Delta)^{3/2}) \cap X_0$ and the remaining assumptions from (i) hold. Then there exists a unique pre-solution of problem $(P)_{\eta,f}$.

Observe also that our results on the well-posedness of fractional analogue of the Benney–Luke equation, based on a very simple approach, are completely new provided that $\eta > 1$, as well as that we have obtained some new results on the well-posedness of the inhomogeneous Cauchy problem $P_{\eta,f}$ in the case that $\eta < 1$ (cf. [210, Theorem 4.2] for the first result in this direction).

The following theorem can be deduced by making use of the argumentation contained in the proof of [295, Theorem 2.16]. Here we would like to observe that

the equality $R_{\lambda,\mu} = 0$, stated on [295, p. 12, l. 4], can be proved by taking the Laplace transform of term appearing on [295, p. 12, l. 1-2] in variable μ , and by using the strong analyticity of mapping $\lambda \mapsto F(\lambda) \in L(X), \lambda \in N$, along with the equality $R_{\lambda,\mu} = 0$ for $\operatorname{Re} \lambda > \omega$, $\tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0$ (the repeated use of identity [295, (2.30)] on [295, p. 12, l. 4] is wrong and makes a circulus vitiosus):

THEOREM 3.2.18 (cf. [292, Theorem 2.2.4] for non-degenerate case). Suppose that $\alpha \in (0, \pi/2]$, $\operatorname{abs}(k) < \infty$, $\operatorname{abs}(|a|) < \infty$, and $\tilde{k}(\lambda)$ can be analytically continued to a function $g: \omega + \Sigma_{\frac{\pi}{2}+\alpha} \to \mathbb{C}$, where $\omega \ge \max(0, \operatorname{abs}(k), \operatorname{abs}(|a|))$. Suppose, further, that \mathcal{A} is a closed subgenerator of an analytic (a, k)-regularized C-resolvent family $(R(t))_{t\ge 0}$ of angle α satisfying that the family $\{e^{-\omega z}R(z): z \in \Sigma_{\gamma}\} \subseteq L(X)$ is equicontinuous for all angles $\gamma \in (0, \alpha)$, as well as that the equation (272) holds for each $y = x \in X$, with $R_1(\cdot)$ and C_1 replaced therein by $R(\cdot)$ and C, respectively. Set

$$N := \{\lambda \in \omega + \sum_{\frac{\pi}{2} + \alpha} : g(\lambda) \neq 0\}$$

Then N is an open connected subset of \mathbb{C} . Assume that there exists an analytic function $\hat{a}: N \to \mathbb{C}$ such that $\hat{a}(\lambda) = \tilde{a}(\lambda)$, $\operatorname{Re} \lambda > \omega$. Then the operator $I - \hat{a}(\lambda)\mathcal{A}$ is injective for every $\lambda \in N$, $R(C) \subseteq R(I - \hat{a}(\lambda)C^{-1}\mathcal{A}C)$ for every $\lambda \in N_1 := \{\lambda \in N : \hat{a}(\lambda) \neq 0\}$, the operator $(I - \hat{a}(\lambda)C^{-1}\mathcal{A}C)^{-1}C \in L(X)$ is single-valued $(\lambda \in N_1)$, the family

$$\{(\lambda - \omega)g(\lambda)(I - \hat{a}(\lambda)C^{-1}\mathcal{A}C)^{-1}C : \lambda \in N_1 \cap (\omega + \Sigma_{\frac{\pi}{2} + \gamma_1})\} \subseteq L(X)$$

is equicontinuous for every angle $\gamma_1 \in (0, \alpha)$, the mapping

$$\lambda \mapsto (I - \hat{a}(\lambda)C^{-1}\mathcal{A}C)^{-1}Cx, \quad \lambda \in N_1 \text{ is analytic for every } x \in X,$$

and

$$\lim_{\lambda \to +\infty, \tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0} \lambda \tilde{k}(\lambda) (I - \tilde{a}(\lambda)\mathcal{A})^{-1} C x = R(0)x, \quad x \in X$$

Keeping in mind Lemma 1.2.2, Theorem 1.2.3 and Theorem 3.2.5, we can repeat almost verbatim the proof of [**292**, Theorem 2.2.5] in order to see that the following result holds.

THEOREM 3.2.19. Assume that \mathcal{A} is a closed MLO in $X, C\mathcal{A} \subseteq \mathcal{A}C, \alpha \in (0, \pi/2]$, $\operatorname{abs}(k) < \infty$, $\operatorname{abs}(|a|) < \infty$ and $\omega \ge \max(0, \operatorname{abs}(k), \operatorname{abs}(|a|))$. Assume, further, that for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\tilde{k}(\lambda) \neq 0$, the operator $I - \tilde{a}(\lambda)\mathcal{A}$ is injective with $R(C) \subseteq R(I - \tilde{a}(\lambda)\mathcal{A})$. If there exist a function $q : \omega + \Sigma_{\frac{\pi}{2} + \alpha} \to L(X)$ and an operator $D \in L(X)$ such that, for every $x \in X$, the mapping $\lambda \mapsto q(\lambda)x$, $\lambda \in \omega + \Sigma_{\frac{\pi}{2} + \alpha}$ is analytic as well as that:

$$q(\lambda)x = \tilde{k}(\lambda)(I - \tilde{a}(\lambda)\mathcal{A})^{-1}Cx, \quad \operatorname{Re}\lambda > \omega, \ \tilde{k}(\lambda) \neq 0, \ x \in X,$$

the family $\{(\lambda - \omega)q(\lambda) : \lambda \in \omega + \Sigma_{\frac{\pi}{2} + \gamma}\} \subseteq L(X)$ is equicontinuous for $\gamma \in (0, \alpha)$, and

$$\lim_{\lambda \to +\infty} \lambda q(\lambda) x = Dx, \quad x \in X, \text{ if } \overline{D(\mathcal{A})} \neq X.$$

then \mathcal{A} is a subgenerator of an exponentially equicontinuous, analytic (a, k)-regularized C-resolvent family $(R(t))_{t \ge 0}$ of angle α satisfying that $R(z)\mathcal{A} \subseteq \mathcal{A}R(z), z \in$ Σ_{α} , the family $\{e^{-\omega z}R(z) : z \in \Sigma_{\gamma}\} \subseteq L(X)$ is equicontinuous for all angles $\gamma \in (0, \alpha)$, as well as that the equation (272) holds for each $y = x \in X$, with $R_1(\cdot)$ and C_1 replaced therein by $R(\cdot)$ and C, respectively.

The classes of exponentially equicontinuous, analytic (a, k)-regularized C_1 existence families and (a, k)-regularized C_2 -uniqueness families can be introduced and analyzed, as well. For the sequel, we need the following notion.

DEFINITION 3.2.20. Let X = Y, and let \mathcal{A} be a subgenerator of a C_1 -existence family $(R_1(t))_{t \ge 0}$ (cf. Definition 3.2.1(i) with $a(t) \equiv 1$ and $k(t) \equiv 1$). Then $(R_1(t))_{t \ge 0}$ is said to be entire iff, for every $x \in X$, the mapping $t \mapsto R_1(t)x, t \ge 0$ can be analytically extended to the whole complex plane.

Using the arguments contained in the proof of [296, Theorem 3.15], we can deduce the following result.

THEOREM 3.2.21. Suppose $r \ge 0$, $\theta \in (0, \pi/2)$, \mathcal{A} is a closed MLO and $-\mathcal{A}$ is a subgenerator of an exponentially equicontinuous, analytic r-times integrated C-semigroup $(S_r(t))_{t\ge 0}$ of angle θ . Then there exists an operator $C_1 \in L(X)$ such that \mathcal{A} is a subgenerator of an entire C_1 -existence family in X.

- REMARK 3.2.22. (i) It should be observed that we do not require the injectivity of operator C_1 here. The operators $T_{\alpha}(z)$ and $S_{\alpha,z_0}(z)$, appearing in the proof of [296, Theorem 3.15], annulates on the subspace $\mathcal{A}0$.
- (ii) Theorem 3.2.21 is closely linked with the assertions of [298, Theorem 2.1, Theorem 2.2]. These results can be extended to abstract degenerate fractional differential inclusions, as well.

EXAMPLE 3.2.23. In a great number of research papers, many authors have investigated infinitely differentiable semigroups generated by multivalued linear operators of form AB^{-1} or $B^{-1}A$, where the operators A and B satisfy the condition [199, (3.14)], or its slight modification, with certain real constants $0 < \beta \leq \alpha \leq 1$, $\gamma \in \mathbb{R}$ and c, C > 0 (in our notation, we have A = L and B = M). The validity of this condition with $\alpha = 1$ (see e.g. [199, Example 3.3, 3.6]) immediately implies by Theorem 3.2.19 and Remark 3.1.2(v) that the operator AB^{-1} generates an exponentially bounded, analytic σ -times integrated semigroup of angle $\Sigma_{\arctan(1/c)}$, provided that $\sigma > 1 - \beta$; in the concrete situation of [199, Example 3.4, 3.5], the above holds with the operator AB^{-1} replaced by $B^{-1}A$. Unfortunately, this fact is not sufficiently enough for taking up a fairly complete study of the abstract degenerate Cauchy problems that are subordinated to those appearing in the above-mentioned examples; later on, we will construct the corresponding subordination fractional operator families with removable singularities at zero and analyze their basic structural properties (cf. the proof of [61, Theorem 3.1]). On the other hand, from the point of view of possible applications of Theorem 3.2.21, it is very important to know that the operators AB^{-1} or $B^{-1}A$ generate exponentially bounded, analytic integrated semigroups. This enables us to consider the abstract degenerate Cauchy problems that are backward to those appearing in [199, Example 3.3-Example 3.6].

For example, we can treat the following modification of the backward Poisson heat equation in the space $L^{p}(\Omega)$:

$$(P)_b: \begin{cases} \frac{\partial}{\partial t}[m(x)v(t,x)] = -\Delta v + bv, & t \ge 0, \ x \in \Omega; \\ v(t,x) = 0, & (t,x) \in [0,\infty) \times \partial\Omega, \\ m(x)v(0,x) = u_0(x), & x \in \Omega, \end{cases}$$

where Ω is a bounded domain in \mathbb{R}^n , b > 0, $m(x) \ge 0$ a.e. $x \in \Omega$, $m \in L^{\infty}(\Omega)$ and 1 . Let*B* $be the multiplication in <math>L^p(\Omega)$ with m(x), and let $A = \Delta - b$ act with the Dirichlet boundary conditions. Then Theorem 3.2.21 implies that there exists an operator $C_1 \in L(L^p(\Omega))$ such that $\mathcal{A} = -AB^{-1}$ is a subgenerator of an entire C_1 -existence family; hence, for every $u_0 \in R(C_1)$, the problem (*P*) has a unique solution $t \mapsto u(t), t \ge 0$ which can be extended entirely to the whole complex plane. Furthermore, the set of all initial values u_0 for which there exists a solution of problem (*P*)_b is dense in $L^p(\Omega)$ provided that there exists a constant d > 0 such that $|m(x)| \ge d$ a.e. $x \in \Omega$.

EXAMPLE 3.2.24. It is clear that the examples presented in [199, Chapter III] can serve one for examination of a wide class of abstract degenerate relaxation equations which are not subordinated to the problems of first order (cf. Subsection 3.5.2 for the continuation): Suppose that the condition [199, (3.1)] holds with certain real constants $0 < \beta \leq \alpha \leq 1$, c, M > 0, as well as that $\theta \in (\pi/2, 0)$, $\zeta \in (0, 1)$ and $\frac{\pi}{2} > \pi - \arctan \frac{1}{c} + \theta > \frac{1}{2}\pi\zeta$. Then $\sum_{\pi-\arctan \frac{1}{c}+\theta} \subseteq \rho(e^{i\theta}\mathcal{A})$ and, in general, $\rho(e^{i\theta}\mathcal{A})$ does not contain any right half plane. An application of Theorem 3.2.19 shows that the operator $e^{i\theta}\mathcal{A}$ generates an exponentially bounded, analytic (g_{ζ}, g_{r+1}) -regularized resolvent family of angle $\theta' := \min((\pi - \arctan(1/c) + \theta - (\pi\zeta/2))/\zeta, \pi/2)$, where $r > \zeta(1 - \beta)$, if \mathcal{A} is not densely defined, and $r = \zeta(1 - \beta)$, otherwise.

Suppose now that $x \in E$, $1-\zeta > \eta > 1-\zeta\beta$, $\delta > 0$, $0 < \gamma < \theta'$, t > 0 is fixed temporarily, $\Gamma_1 := \{re^{i((\pi/2)+\gamma)} : r \ge t^{-1}\} \cup \{t^{-1}e^{i\theta} : \theta \in [0, (\pi/2)+\gamma]\},$ $\Gamma_2 := \{re^{-i((\pi/2)+\gamma)} : r \ge t^{-1}\} \cup \{t^{-1}e^{i\theta} : \theta \in [-(\pi/2)-\gamma, 0]\}$ and $\Gamma := \Gamma_1 \cup \Gamma_2$ is oriented counterclockwise. Define u(0) := 0 and

$$u(t) := \frac{1}{2\pi i} \int_{\Gamma} e^{\lambda t} \lambda^{-\eta} (\lambda^{\zeta} - e^{i\theta} \mathcal{A})^{-1} x \, d\lambda.$$

Arguing as in [27, Theorem 2.6.1, Theorem 2.6.4], it readily follows that $u \in C([0,\infty): E)$, $||u(t)|| = O(t^{\eta+\zeta\beta-1})$, $t \ge 0$ and the mapping $t \mapsto u(t)$, t > 0 can be analytically extended to the sector $\Sigma_{\theta'}$. Keeping in mind Theorem 1.2.3 and Theorem 1.2.4(i), we obtain that there exists a continuous section $t \mapsto u_{\mathcal{A},\theta,\zeta}(t)$, t > 0 of the multivalued mapping $t \mapsto e^{i\theta}\mathcal{A}(g_{\zeta} * u)(t)$, t > 0, with the meaning clear, such that

$$u(t) = u_{\mathcal{A},\theta,\zeta}(t) + g_{\eta+\zeta}(t)x, \quad t > 0.$$

Observe, finally, that the Riemann–Liouville fractional derivative $D_t^{\zeta} u(t)$ need not be defined here.

Let us recall that for any sequence $(M_n)_{n \in \mathbb{N}_0}$ of positive real numbers satisfying $M_0 = 1, (M.1), (M.2)$ and (M.3)', we define the function

$$\omega_L(t) := \sum_{n=0}^{\infty} \frac{t^n}{M_n}, \quad t \ge 0.$$

The most important results concerning differential properties of non-degenerate (a, k)-regularized *C*-resolvent families remain true, with almost minimal reformulations, in our new setting. The proofs of following extensions of [**292**, Theorem 2.2.15, Theorem 2.2.17] are omitted.

THEOREM 3.2.25. Suppose that \mathcal{A} is a closed MLO in X, $\operatorname{abs}(k) < \infty$, $\operatorname{abs}(|a|) < \infty$, $r \ge -1$ and there exists $\omega \ge \max(0, \operatorname{abs}(k), \operatorname{abs}(|a|))$ such that, for every $z \in \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda > \omega, \ \tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0\}$, we have that the operator $I - \tilde{a}(z)\mathcal{A}$ is injective and $R(C) \subseteq R(I - \tilde{a}(z)\mathcal{A})$. If, additionally, for every $\sigma > 0$, there exist $C_{\sigma} > 0$ and an open neighborhood $\Omega_{\sigma,\omega}$ of the region

$$\Lambda_{\sigma,\omega} := \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda \leqslant \omega, \ \operatorname{Re} \lambda \geqslant -\sigma \ln |\operatorname{Im} \lambda| + C_{\sigma} \} \cup \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda \geqslant \omega \},\$$

and a function $h_{\sigma}: \Omega_{\sigma,\omega} \to L(X)$ such that, for every $x \in X$, the mapping $\lambda \mapsto h_{\sigma}(\lambda)x, \lambda \in \Omega_{\sigma,\omega}$ is analytic as well as that $h_{\sigma}(\lambda) = \tilde{k}(\lambda)(I - \tilde{a}(\lambda)\mathcal{A})^{-1}C$, $\operatorname{Re} \lambda > \omega$, $\tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0$, and the family $\{|\lambda|^{-r}h_{\sigma}(\lambda): \lambda \in \Lambda_{\sigma,\omega}\}$ is equicontinuous, then, for every $\zeta > 1$, \mathcal{A} is a subgenerator of an exponentially equicontinuous $(a, k * g_{\zeta+r})$ -regularized C-resolvent family $(R_{\zeta}(t))_{t\geq 0}$ satisfying that the mapping $t \mapsto R_{\zeta}(t)$, t > 0 is infinitely differentiable in L(X).

THEOREM 3.2.26. Let $(M_n)_{n \in \mathbb{N}_0}$ satisfy (M.1), (M.2) and (M.3)'.

(i) Suppose that abs(k) < ∞, abs(|a|) < ∞, A is a closed subgenerator of a (local) (a, k)-regularized C-resolvent family (R(t))_{t∈[0,τ)}, ω > max(0,abs(k), abs(|a|)) and m ∈ N. Denote, for every ε ∈ (0,1) and a corresponding K_ε > 0,

$$F_{\varepsilon,\omega} := \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda \ge -\ln \omega_L(K_\varepsilon |\operatorname{Im} \lambda|) + \omega\}.$$

Assume that, for every $\varepsilon \in (0,1)$, there exist $K_{\varepsilon} > 0$, an open neighborhood $O_{\varepsilon,\omega}$ of the region $G_{\varepsilon,\omega} := \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda \ge \omega, \ \tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0\} \cup \{\lambda \in F_{\varepsilon,\omega} : \operatorname{Re} \lambda \le \omega\}$, a mapping $h_{\varepsilon} : O_{\varepsilon,\omega} \to L(E)$ and analytic mappings $f_{\varepsilon} : O_{\varepsilon,\omega} \to \mathbb{C}, g_{\varepsilon} : O_{\varepsilon,\omega} \to \mathbb{C}$ such that:

- (a) $f_{\varepsilon}(\lambda) = \tilde{k}(\lambda), \operatorname{Re} \lambda > \omega; g_{\varepsilon}(\lambda) = \tilde{a}(\lambda), \operatorname{Re} \lambda > \omega,$
- (b) for every λ ∈ F_{ε,ω}, the operator I − g_ε(λ)A is injective and R(C) ⊆ R(I − g_ε(λ)A),
- (c) for every $x \in X$, the mapping $\lambda \mapsto h_{\varepsilon}(\lambda)x$, $\lambda \in G_{\varepsilon,\omega}$ is analytic, $h_{\varepsilon}(\lambda) = f_{\varepsilon}(\lambda)(I - g_{\varepsilon}(\lambda)\mathcal{A})^{-1}C$, $\lambda \in G_{\varepsilon,\omega}$,
- (d) the family $\{(1+|\lambda|)^{-m}e^{-\varepsilon|\operatorname{Re}\lambda|}h_{\varepsilon}(\lambda):\lambda\in F_{\varepsilon,\omega}, \operatorname{Re}\lambda\leqslant\omega\}\subseteq L(X)$ is equicontinuous and the family $\{(1+|\lambda|)^{-m}h_{\varepsilon}(\lambda):\lambda\in\mathbb{C}, \operatorname{Re}\lambda\geqslant\omega\}\subseteq L(X)$ is equicontinuous.

Then the mapping $t \mapsto R(t), t \in (0, \tau)$ is infinitely differentiable in L(X)and, for every compact set $K \subseteq (0, \tau)$, there exists $h_K > 0$ such that the set $\{\frac{h_K^n \frac{d^n}{dt^n} R(t)}{M_n} : t \in K, n \in \mathbb{N}_0\}$ is equicontinuous. (ii) Suppose that $\operatorname{abs}(k) < \infty$, $\operatorname{abs}(|a|) < \infty$, \mathcal{A} is a closed subgenerator of a

(ii) Suppose that abs(k) < ∞, abs(|a|) < ∞, A is a closed subgenerator of a (local) (a,k)-regularized C-resolvent family (R(t))_{t∈[0,τ)}, ω>max(0,abs(k), abs(|a|)) and m ∈ N. Denote, for every ε ∈ (0,1), ρ ∈ [1,∞) and a corresponding K_ε > 0,

$$F_{\varepsilon,\omega,\rho} := \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda \ge -K_{\varepsilon} | \operatorname{Im} \lambda |^{1/\rho} + \omega \}.$$

Assume that, for every $\varepsilon \in (0,1)$, there exist $K_{\varepsilon} > 0$, an open neighborhood $O_{\varepsilon,\omega}$ of the region $G_{\varepsilon,\omega,\rho} := \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda \ge \omega, \ \tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0\} \cup \{\lambda \in F_{\varepsilon,\omega,\rho} : \operatorname{Re} \lambda \le \omega\}$, a mapping $h_{\varepsilon} : O_{\varepsilon,\omega} \to L(X)$ and analytic mappings $f_{\varepsilon} : O_{\varepsilon,\omega} \to \mathbb{C}$ and $g_{\varepsilon} : O_{\varepsilon,\omega} \to \mathbb{C}$ such that the conditions (i)(a)–(d) of this theorem hold with $F_{\varepsilon,\omega}$, resp. $G_{\varepsilon,\omega}$, replaced by $F_{\varepsilon,\omega,\rho}$, resp. $G_{\varepsilon,\omega,\rho}$. Then the mapping $t \mapsto R(t)$, $t \in (0, \tau)$ is infinitely differentiable in L(X) and, for every compact set $K \subseteq (0, \tau)$, there exists $h_K > 0$ such that the set $\{\frac{h_K^n \frac{d^n}{dt^n} R(t)}{n!^{\rho}} : t \in K, n \in \mathbb{N}_0\}$ is equicontinuous.

Let us recall that the case $\rho = 1$ in Theorem 3.2.26 is very important because it gives a sufficient condition for an (a, k)-regularized C-resolvent family to be real analytic.

Following J. Prüss [459, Definition 3.1, p. 68] and the author [292, Definition 2.1.23], it will be said that the abstract Volterra inclusion (262) with $\mathcal{B} = I$ (denoted henceforth by (VI)) is (kC)-parabolic iff the following holds:

- (i) |a|(t) and k(t) satisfy (P1) and there exist meromorphic extensions of the functions ã(λ) and k(λ) on C₊, denoted by â(λ) and k(λ). Let N be the subset of C₊ which consists of all zeroes and possible poles of â(λ) and k(λ).
- (ii) There exists $M \ge 1$ such that, for every $\lambda \in \mathbb{C}_+ \setminus N$, $1/\hat{a}(\lambda) \in \rho_C(\mathcal{A})$ and $||\hat{k}(\lambda)(I - \hat{a}(\lambda)\mathcal{A})^{-1}C|| \le M/|\lambda|$.

If $k(t) \equiv 1$, resp. C = I, then it is also said that (VI) is C-parabolic, resp. k-parabolic.

For the sequel, we need to repeat the following well-known notion. Suppose that $n \in \mathbb{N}$, |a|(t) satisfies (P1) and abs(a) = 0. Following [459, Definition 3.3, p. 69], we say that a(t) is *n*-regular iff there exists c > 0 such that

$$|\lambda^m \hat{a}^{(m)}(\lambda)| \leq c |\hat{a}(\lambda)|, \quad \lambda \in \mathbb{C}_+, \ 1 \leq m \leq n.$$

Set $a^{(-1)}(t) := \int_0^t a(s)ds$, $t \ge 0$ and suppose that a(t) and b(t) are *n*-regular for some $n \in \mathbb{N}$. Then we know that $\hat{a}(\lambda) \ne 0$, $\lambda \in \mathbb{C}_+$, as well as that (a * b)(t) and $a^{(-1)}(t)$ are *n*-regular, and a'(t) is *n*-regular provided that abs(a') = 0. Furthermore, the *n*-regularity of a(t) is equivalent to say that there exists c' > 0 such that

$$|(\lambda^m \hat{a}(\lambda))^{(m)}| \leq c' |\hat{a}(\lambda)|, \quad \lambda \in \mathbb{C}_+, \ 1 \leq m \leq n.$$

In the case that $\arg(\hat{a}(\lambda)) \neq \pi$, $\lambda \in \mathbb{C}_+$, the *n*-regularity of a(t) is also equivalent with the existence of a constant c'' > 0 such that

$$|\lambda^m (\ln \hat{a}(\lambda))^{(m)}| \leq c'', \quad \lambda \in \mathbb{C}_+, \ 1 \leq m \leq n.$$

The assertion of [292, Theorem 2.1.24] continues to hold in the degenerate case:

THEOREM 3.2.27. Assume $n \in \mathbb{N}$, a(t) is n-regular, $(E, \|\cdot\|)$ is a Banach space, \mathcal{A} is a closed MLO in E, (VI) is C-parabolic, and the mapping $\lambda \mapsto (I - \tilde{a}(\lambda)\mathcal{A})^{-1}C$, $\lambda \in \mathbb{C}_+$ is continuous. Denote by D_t^{ζ} the Riemann-Liouville fractional derivative of order $\zeta > 0$. Then, for every $\alpha \in (0,1]$, \mathcal{A} is a subgenerator of an $(a, g_{\alpha+1})$ regularized C^2 -resolvent family $(S_{\alpha}(t))_{t\geq 0}$ which satisfies $\sup_{h>0,t\geq 0} h^{-\alpha}||S_{\alpha}(t + h) - S_{\alpha}(t)|| < \infty$, $D_t^{\alpha}S_{\alpha}(t)C^{k-1} \in C^{k-1}((0,\infty): L(E))$, $1 \leq k \leq n$ as well as:

(293)
$$||t^j D_t^j D_t^\alpha S_\alpha(t) C^{k-1}|| \leq M, \quad t \ge 0, \ 1 \le k \le n, \ 0 \le j \le k-1,$$

$$(294) \quad \|t^k D_t^{k-1} D_t^{\alpha} S_{\alpha}(t) C^{k-1} - s^k D_s^{k-1} D_s^{\alpha} S_{\alpha}(s) C^{k-1}\| \\ \leqslant M |t-s| \Big(1 + \ln \frac{t}{t-s} \Big), \quad 0 \leqslant s < t < \infty, \ 1 \leqslant k \leqslant n,$$

and, for every T > 0, $\varepsilon > 0$ and $k \in \mathbb{N}_n$, there exists $M^{\varepsilon}_{T,k} > 0$ such that

(295)
$$\|t^k D_t^{k-1} D_t^{\alpha} S_{\alpha}(t) C^{k-1} - s^k D_s^{k-1} D_s^{\alpha} S_{\alpha}(s) C^{k-1} \|$$
$$\leq M_{T,k}^{\varepsilon} (t-s)^{1-\varepsilon}, \quad 0 \leq s < t \leq T, \ 1 \leq k \leq n.$$

Furthermore, if \mathcal{A} is densely defined, then \mathcal{A} is a subgenerator of a bounded (a, C^2) -regularized resolvent family $(S(t))_{t\geq 0}$ which satisfies $S(t)C^{k-1} \in C^{k-1}((0,\infty) : L(E)), 1 \leq k \leq n$ and (293)–(295) with $D_t^{\alpha}S_{\alpha}(t)C^{k-1}$ replaced by $S(t)C^{k-1}$ $(1 \leq k \leq n)$ therein.

The representation formula [459, (3.41), p. 81] and the assertions of [459, Corollary 3.2-Corollary 3.3, pp. 74–75] can be extended to exponentially bounded (a, C)-regularized resolvent families subgenerated by multivalued linear operators, as well. For more details about parabolicity of problem (VI) in non-degenerate case, we refer the reader to [459, Chapter I, Section 3].

In the remainder of this section, we consider the abstract degenerate Volterra inclusion

(296)
$$0 \in \mathcal{B}u(t) + \sum_{j=0}^{n-1} \mathcal{A}_j(a_j * u)(t) + \mathcal{F}(t), \quad t \in [0, \tau).$$

Here, $n \in \mathbb{N}$, $0 < \tau \leq \infty$, $\mathcal{F}: [0, \tau) \to P(Y)$, $a_0, \ldots, a_{n-1} \in L^1_{loc}([0, \tau))$, and $\mathcal{A} \equiv \mathcal{A}_0, \ldots, \mathcal{A}_{n-1}, \mathcal{B} \equiv \mathcal{A}_n$ are multivalued linear operators acting between the sequentially complete locally convex spaces X and Y. Set $a_n(t) := \delta$ -distribution and $\tilde{\delta} := 1$. In the following definition, we introduce the notion of a \mathcal{V} -(pre)-solution of inclusion (296); see Introduction.

DEFINITION 3.2.28. Suppose $\mathcal{V} \subseteq \mathbb{N}_n^0$.

(i) A function $u \in C([0,\tau) : X)$ is said to be a \mathcal{V} -pre-solution of problem (296) iff there exist functions $u_j \in C([0,\tau) : Y)$ $(j \in \mathbb{N}_n^0 \smallsetminus \mathcal{V})$ such that $u_j(t) \in \mathcal{A}_j u(t), t \in [0,\tau), j \in \mathbb{N}_n^0 \smallsetminus \mathcal{V}, (a_j * u)(t) \in D(\mathcal{A}_j), t \in [0,\tau), j \in \mathcal{V}$ and

$$0 \in \sum_{j \in \mathbb{N}_n^0 \smallsetminus \mathcal{V}} (a_j * u_j)(t) + \sum_{j \in \mathcal{V}} \mathcal{A}_j(a_j * u)(t) + \mathcal{F}(t), \quad t \in [0, \tau).$$

(ii) A function $u \in C([0,\tau) : X)$ is said to be a \mathcal{V} -solution of problem (296) iff there exist functions $u_j \in C([0,\tau) : Y)$ $(j \in \mathbb{N}_n^0 \smallsetminus \mathcal{V})$ and $u_{j,\mathcal{A}_j} \in C([0,\tau) : Y)$ $(j \in \mathcal{V})$ such that $u_j(t) \in \mathcal{A}_j u(t), t \in [0,\tau), j \in \mathbb{N}_n^0 \smallsetminus \mathcal{V},$ $u_{j,\mathcal{A}_j}(t) \in \mathcal{A}_j(a_j * u)(t), t \in [0,\tau), j \in \mathcal{V}$ and

(297)
$$0 \in \sum_{j \in \mathbb{N}_n^0 \smallsetminus \mathcal{V}} (a_j * u_j)(t) + \sum_{j \in \mathcal{V}} u_{j,\mathcal{A}_j}(t) + \mathcal{F}(t), \quad t \in [0,\tau).$$

Suppose that $\mathcal{V} \subseteq \mathbb{N}_n^0$ and u(t) is a \mathcal{V} -solution of problem (296). Then it is clear that $\sec_c(\mathcal{F}) \neq \emptyset$. Furthermore, if $\mathcal{V}_1 \subseteq \mathcal{V}_2 \subseteq \mathbb{N}_n^0$, then we can simply prove with the help of Theorem 1.2.3 that any \mathcal{V}_1 -(pre)-solution of problem (296) is a \mathcal{V}_2 -(pre)-solution of (296), provided that the operators \mathcal{A}_j are closed for $j \in \mathcal{V}_2 \smallsetminus \mathcal{V}_1$. If $\mathcal{V} = \emptyset$, then the notion of a pre-solution of (296) coincides with that of solution of (296); if this is the case, then any solution of (296) is also said to be a strong solution of (296). Here it is worth noting that the assumption $u \in C([0, \tau) : X)$ in Definition 3.2.28(i) is slightly redundant in the case that $\mathcal{V} = \emptyset$ because then we do not have a problem with defining the convolutions $(a_j * u)(t)$ for $j \in \mathcal{V}$. Assuming $\mathcal{V} = \emptyset$ and all the remaining assumptions from Definition 3.2.28(i) hold, u(t) will be called a *p*-strong solution of problem (296). In the case that n = 1, the most important examples of *p*-strong solutions of problem (296) with $a_0(t) = g_\alpha(t)$, where $\alpha > 0$, are obtained by integrating α -times *p*-solutions of abstract degenerate fractional problem (DFP)_R.

REMARK 3.2.29. Suppose that A_0, \ldots, A_{n-1}, B are single-valued linear operators between the spaces X and Y, and the mapping $f: [0, \tau) \to D(B)$ is given in advance. Consider the following degenerate Volterra integral equation:

(298)
$$0 = Bu(t) + \sum_{j=0}^{n-1} A_j(a_j * u)(t) + Bf(t), \quad t \in [0, \tau),$$

which is, unquestionably, the most important subcase of (296). Along with this equation, we examine the accompanied problem of type (296):

(299)
$$0 \in u(t) + \sum_{j=0}^{n-1} B^{-1} A_j(a_j * u)(t) + f(t), \quad t \in [0, \tau),$$

with the multivalued linear operators $\mathcal{A}_0 := B^{-1}A_0, \ldots, \mathcal{A}_{n-1} := B^{-1}A_{n-1}, \mathcal{B} := I$ in X. Let $\mathcal{V} \subseteq \mathbb{N}_n^0$. Then the following holds:

(i) If $\mathcal{V} = \mathbb{N}_n^0$, then any \mathcal{V} -pre-solution of problem (299) is a \mathcal{V} -pre-solution of problem (298).

(ii) If $B \in L(X, Y)$, then any \mathcal{V} -(pre)-solution of problem (299) is a \mathcal{V} -(pre)-solution of problem (298).

REMARK 3.2.30. Suppose that $\tau = \infty$, $|a_j|(t)$ $(0 \leq j \leq n-1)$ satisfy (P1), and the MLOS \mathcal{A}_j are closed for $0 \leq j \leq n$. Let $u \in C([0,\tau) : X)$, $u_j \in C([0,\tau) : Y)$ $(j \in \mathbb{N}_n^0 \smallsetminus \mathcal{V})$, and $u_{j,\mathcal{A}_j} \in C([0,\tau) : Y)$ $(j \in \mathcal{V})$ be Laplace transformable functions. Suppose that there exists a real number $\omega \geq \max(0, \operatorname{abs}(|a_j|))$ $(0 \leq j \leq n-1)$ such that $\widetilde{u}_j(\lambda) \in \mathcal{A}_j \widetilde{u}(\lambda)$, $\lambda > \omega$, $j \in \mathbb{N}_n^0 \smallsetminus \mathcal{V}$ and $\widetilde{u}_{j,\mathcal{A}_j}(\lambda) \in \mathcal{A}_j(\widetilde{a}_j(\lambda)\widetilde{u}(\lambda))$, $\lambda > \omega$, $j \in \mathcal{V}$. Then an application of Proposition 1.4.7 yields that $u_j(t) \in \mathcal{A}_j u(t)$, $t \geq 0$, $j \in \mathbb{N}_n^0 \smallsetminus \mathcal{V}$ and $u_{j,\mathcal{A}_j}(t) \in \mathcal{A}_j(a_j * u)(t)$, $t \geq 0$, $j \in \mathcal{V}$. If, in addition to this, there exists a continuous section $f \in \operatorname{sec}_c(\mathcal{F})$ that is Laplace transformable (not necessarily exponentially bounded) and

$$\sum_{j \in \mathbb{N}_n^0 \smallsetminus \mathcal{V}} \widetilde{a_j}(\lambda) \widetilde{u_j}(\lambda) + \sum_{j \in \mathcal{V}} \widetilde{u_{j,\mathcal{A}_j}}(\lambda) + \widetilde{f}(\lambda) = 0, \quad \lambda > \omega,$$

then the uniqueness theorem for Laplace transform implies that u(t) is a \mathcal{V} -solution of problem (296).

The proofs of following subordination principles for \mathcal{V} -solutions of problem (296) are standard and therefore omitted; it is worth noting that we can formulate similar results provided that the multivalued linear operators employed in our analysis are relatively closed.

THEOREM 3.2.31. Suppose that $\tau = \infty$, $\gamma \in (0,1)$, $\omega > 0$, $|a_j|(t)$ and $|b_j|(t)$ $(0 \leq j \leq n-1)$ satisfy (P1), $b_n(t) := \delta$ -distribution and

$$\widetilde{a_j}(\lambda^{\gamma}) := \widetilde{b_j}(\lambda), \quad \lambda > \omega.$$

Suppose that the MLOs \mathcal{A}_j are closed for $0 \leq j \leq n$. Let $u \in LT_{or} - X$ be a \mathcal{V} -solution of problem (296), with $\mathcal{F} = f \in LT_{or} - Y$ being single-valued, and the functions $u_j \in C([0,\tau):Y)$ $(j \in \mathbb{N}_n^0 \setminus \mathcal{V})$, $u_{j,\mathcal{A}_j} \in C([0,\tau):Y)$ $(j \in \mathcal{V})$, appearing in (297), being Laplace transformable and exponentially bounded. Define

$$\begin{split} u_{\gamma}(t) &:= \int_{0}^{\infty} t^{-\gamma} \Phi_{\gamma}(st^{-\gamma}) u(s) ds, \quad t > 0, \ u_{\gamma}(0) := u(0), \\ u_{j,\gamma}(t) &:= \int_{0}^{\infty} t^{-\gamma} \Phi_{\gamma}(st^{-\gamma}) u_{j}(s) ds, \quad t > 0, \ u_{j,\gamma}(0) := u_{j}(0), \ j \in \mathbb{N}_{n}^{0} \smallsetminus \mathcal{V}, \\ u_{j,\mathcal{A}_{j},\gamma}(t) &:= \int_{0}^{\infty} t^{-\gamma} \Phi_{\gamma}(st^{-\gamma}) u_{j,\mathcal{A}_{j}}(s) ds, \quad t > 0, \ u_{j,\mathcal{A}_{j},\gamma}(0) := u_{j,\mathcal{A}_{j}}(0), \ j \in \mathcal{V}, \\ f_{\gamma}(t) &:= \int_{0}^{\infty} t^{-\gamma} \Phi_{\gamma}(st^{-\gamma}) f(s) ds, \quad t > 0, \ f_{\gamma}(0) := f(0). \end{split}$$

Then the function $u_{\gamma}(t)$ is an exponentially bounded \mathcal{V} -solution of problem (296), with the functions $a_j(t)$ replaced by $b_j(t)$ $(0 \leq j \leq n-1)$ and the inhomogeneity $\mathcal{F}(t)$ replaced by $f_{\gamma}(t)$. Furthermore, the functions $u_{j,\gamma} \in C([0,\tau):Y)$ $(j \in \mathbb{N}_n^0 \smallsetminus \mathcal{V})$ and $u_{j,\mathcal{A}_j,\gamma} \in C([0,\tau):Y)$ $(j \in \mathcal{V})$ satisfy the requirements from Definition 3.2.28(ii). THEOREM 3.2.32. Let $|a_j|(t)$, $|b_j|(t)$ and c(t) satisfy (P1), let $\beta \ge 0$ be such that $\int_0^\infty e^{-\beta t} |b_j(t)| dt < \infty$, $j \in \mathbb{N}_{n-1}^0$, and let \mathcal{A}_j be closed for $0 \le j \le n$; $b_n(t) := \delta$ -distribution. Set

$$\alpha := \tilde{c}^{-1} \Big(\frac{1}{\beta} \Big) \ if \ \int_0^\infty c(t) dt > \frac{1}{\beta}, \ \alpha := 0 \ otherwise.$$

Let $u^b \in LT_{or} - X$ be a \mathcal{V} -solution of problem (296), with the function $\mathcal{F} = f^b \in LT_{or} - Y$ being single-valued, and let the set $\{e^{-\omega_b t}u_b(t) : t \ge 0\}$ be bounded in X $(\omega_b \ge 0)$. Let $u^b_j \in LT_{or} - Y$ $(j \in \mathbb{N}^0_n \smallsetminus \mathcal{V})$ and $u^b_{j,\mathcal{A}_j} \in LT_{or} - Y$ $(j \in \mathcal{V})$ be the functions fulfilling the requirements from Definition 3.2.28(ii). Suppose that c(t) is completely positive and

(300)
$$\widetilde{b_j}(1/\tilde{c}(\lambda)) = \widetilde{a_j}(\lambda), \quad j \in \mathbb{N}_{n-1}^0, \ \lambda > \alpha.$$

Let

$$\omega_a = \tilde{c}^{-1} \left(\frac{1}{\omega_b}\right) if \int_0^\infty c(t) dt > \frac{1}{\omega_b}, \ \omega_a = 0 \ otherwise$$

Then, for every $r \in (0,1]$, there exist locally Hölder continuous (with the exponent $r \in (0,1]$) functions $u^{a,r} \in LT_{or} - X$, $u_j^{a,r} \in LT_{or} - Y$ $(j \in \mathbb{N}_n^0 \setminus \mathcal{V})$, $u_{j,\mathcal{A}_j}^{a,r} \in LT_{or} - Y$ $(j \in \mathcal{V})$ and $f^a \in LT_{or} - Y$ such that

$$\begin{split} \widetilde{u^{a,r}}(\lambda) &= \frac{1}{\lambda^{1+r} \widetilde{c}(\lambda)} \widetilde{u^b}(1/\widetilde{c}(\lambda)), \quad \lambda > \alpha, \\ \widetilde{u^{a,r}_j}(\lambda) &= \frac{1}{\lambda^{1+r} \widetilde{c}(\lambda)} \widetilde{u^b_j}(1/\widetilde{c}(\lambda)), \quad j \in \mathbb{N}_n^0 \smallsetminus \mathcal{V}, \ \lambda > \alpha, \\ \widetilde{u^{a,r}_{j,\mathcal{A}_j}}(\lambda) &= \frac{1}{\lambda^{1+r} \widetilde{c}(\lambda)} \widetilde{u^b_{j,\mathcal{A}_j}}(1/\widetilde{c}(\lambda)), \quad j \in \mathbb{N}_n^0 \smallsetminus \mathcal{V}, \ \lambda > \alpha \end{split}$$

and

$$\widetilde{f^{a,r}}(\lambda) = \frac{1}{\lambda^{1+r} \widetilde{c}(\lambda)} \widetilde{f^b}(1/\widetilde{c}(\lambda)) \text{ for } \lambda > \alpha.$$

Furthermore, for every $r \in (0,1]$, $u^{a,r}(t)$ is a \mathcal{V} -solution of problem (296), with the function $\mathcal{F} = f^{a,r}$, as well as the functions $u_j^{a,r}(t)$ $(j \in \mathbb{N}_n^0 \setminus \mathcal{V})$ and $u_{j,\mathcal{A}_j}^{a,r}(t)$ $(j \in \mathcal{V})$ satisfy the requirements from Definition 3.2.28(ii), and the set $\{e^{-\omega_a t} u^{a,r}(t) : t \ge 0\}$ is bounded in X, if $\omega_b = 0$ or $\omega_b \tilde{c}(0) \neq 1$, resp., the set $\{e^{-\varepsilon t} u^{a,r}(t) : t \ge 0\}$ is bounded in X for any $\varepsilon > 0$, if $\omega_b > 0$ and $\omega_b \tilde{c}(0) = 1$.

It is worth noting that the mapping $t \mapsto u_{\gamma}(t), t > 0$ (cf. Theorem 3.2.31) admits an extension to $\sum_{\min((\frac{1}{\gamma}-1)\frac{\pi}{2},\pi)}$ and the mapping $z \mapsto u_{\gamma}(z), z \in \sum_{\min((\frac{1}{\gamma}-1)\frac{\pi}{2},\pi)}$ is analytic. Furthermore, the solution $u_{\gamma}(t)$ possesses some extra properties clarified so many times before.

Assume that Z is a sequentially complete locally convex space. We would like to propose the following general definition.

DEFINITION 3.2.33. Suppose that $k \in C([0, \tau)), C_1 \in L(Z, Y)$ and $(E(t))_{t \in [0, \tau)} \subseteq L(Z, X)$ is a strongly continuous operator family.

(i) It is said that $(E(t))_{t\in[0,\tau)}$ is a pre- (k, C_1, \mathcal{V}) -existence family for (296) iff for each $z \in Z$ there exist continuous sections $u_{j,z} \in \sec_c(\mathcal{A}_j E(\cdot)z)$ $(j \in \mathbb{N}_n^0 \smallsetminus \mathcal{V})$ such that $(a_j * E(\cdot)z)(t) \in D(\mathcal{A}_j), t \in [0,\tau), j \in \mathcal{V}, z \in Z$ and

$$0 \in \sum_{j \in \mathbb{N}_n^0 \smallsetminus \mathcal{V}} (a_j * u_{j,z})(t) + \sum_{j \in \mathcal{V}} \mathcal{A}_j(a_j * E(\cdot)z)(t) + k(t)C_1z, \quad t \in [0,\tau), \ z \in Z.$$

(ii) It is said that $(E(t))_{t\in[0,\tau)}$ is a (k, C_1, \mathcal{V}) -existence family for (296) iff for each $z \in Z$ there exist continuous sections $u_{j,z} \in \sec_c(\mathcal{A}_j E(\cdot)z)$ $(j \in \mathbb{N}_n^0 \setminus \mathcal{V})$ and $u_{j,\mathcal{A}_i,z} \in \sec_c(\mathcal{A}_j(a_j * E(\cdot)z))$ such that

(301)
$$\sum_{j \in \mathbb{N}_n^0 \smallsetminus \mathcal{V}} (a_j * u_{j,z})(t) + \sum_{j \in \mathcal{V}} u_{j,\mathcal{A}_j,z}(t) + k(t)C_1 z = 0, \quad t \in [0,\tau), \ z \in Z.$$

If $\operatorname{card}(\mathcal{V}) \leq 1$, then any pre- (k, C_1, \mathcal{V}) -existence family for (296) is automatically a (k, C_1, \mathcal{V}) -existence family for (296). Moreover, if $(E(t))_{t \in [0,\tau)}$ is a (pre-) (k, C_1, \mathcal{V}) -existence family for (296), then it is clear that, for every $z \in Z$, the mapping $u_z(t) := E(t)z, t \in [0, \tau)$ is a $\mathcal{V}(\text{-pre})$ -solution of problem (296) with $\mathcal{F}(t) := k(t)C_1z, t \in [0, \tau)$. In our analyses of uniqueness of solutions of problem (296), it will be crucial for us to assume that X = Y. We will use the following definition.

DEFINITION 3.2.34. Suppose that $k \in C([0, \tau)), C_2 \in L(X)$ is injective, $(U(t))_{t \in [0,\tau)} \subseteq L(X)$ is strongly continuous and X = Y. Then we say that $(U(t))_{t \in [0,\tau)}$ is a k-regularized C_2 -uniqueness family for (296) iff

$$U(t)x_n + \sum_{j=0}^{n-1} (a_j * U(\cdot)x_j) + k(t)C_2x = 0, \quad t \in [0,\tau),$$

whenever $x \in \bigcap_{j=0}^{n} D(\mathcal{A}_j)$ and $x_j \in \mathcal{A}_j x \ (0 \leq j \leq n)$.

The validity of following proposition can be simply verified.

PROPOSITION 3.2.35. Suppose that $k \in C([0, \tau))$, k(t) is a kernel on $[0, \tau)$, $C_2 \in L(X)$ is injective, $(U(t))_{t \in [0, \tau)}$ is a locally equicontinuous k-regularized C_2 -uniqueness family for (296), X = Y and the function $\mathcal{F} = f : [0, \tau) \to Y$ is single-valued. Then the problem (296) has at most one strong solution.

Observe that, in the formulation of Proposition 3.2.35, we do not require the closedness of multivalued linear operators $\mathcal{A}_0, \ldots, \mathcal{A}_{n-1}, \mathcal{B}$. Using Proposition 3.2.35, we can simply prove that there exists a unique \mathcal{V} -solution of problem (296), provided the closedness of all operators $\mathcal{A}_0, \ldots, \mathcal{A}_{n-1}, \mathcal{B}$ and the validity of conditions from Proposition 3.2.35.

The class of exponentially equicontinuous k-regularized C_2 -uniqueness families for (123) can be simply characterized by using the vector-valued Laplace transform:

THEOREM 3.2.36. Let k(t) and $|a_j|(t)$ $(0 \leq j \leq n-1)$ satisfy (P1). Suppose that $\tau = \infty, k \in C([0,\infty)), C_2 \in L(X)$ is injective, $(U(t))_{t\geq 0}$ is strongly continuous and the operator family $\{e^{-\omega t}U(t) : t \geq 0\} \subseteq L(X)$ is equicontinuous for some

real number $\omega \ge \max(0, \operatorname{abs}(k), \operatorname{abs}(|a_j|))$ $(0 \le j \le n-1)$. Then $(U(t))_{t\ge 0}$ is a k-regularized C_2 -uniqueness family for (296) iff the following holds:

$$\int_0^\infty e^{-\lambda t} U(t) \left[x_n + \sum_{j=0}^{n-1} \tilde{a}_j(\lambda) x_j \right] dt = -\tilde{k}(\lambda) C_2 x, \quad \operatorname{Re} \lambda > \omega,$$

whenever $x \in \bigcap_{j=0}^{n} D(\mathcal{A}_j)$ and $x_j \in \mathcal{A}_j x \ (0 \leq j \leq n)$.

The proof of following subordination principle for k-regularized C_2 -uniqueness families is omitted.

THEOREM 3.2.37. Suppose that $\tau = \infty$, $k, k_1 \in C([0, \infty))$, $C_2 \in L(X)$ is injective, c(t) is completely positive, c(t), k(t), $k_1(t)$, $|a_0|(t), \ldots, |a_{n-1}|(t), |b_0|(t), \ldots, |b_{n-1}|(t)$ satisfy (P1), and $(U(t))_{t \ge 0}$ is an exponentially equicontinuous k-regularized C_2 -uniqueness family for problem (296), with the functions $a_j(t)$ replaced by functions $b_j(t)$ therein $(0 \le j \le n)$. Let $\omega_0 > 0$ be such that, for every $\lambda > \omega_0$ with $\tilde{c}(\lambda) \ne 0$ and $\tilde{k}(1/\tilde{c}(\lambda)) \ne 0$, (300) holds. Then for each $r \in (0,1]$ there exists a locally Hölder continuous (with exponent r), exponentially equicontinuous $(k_1 * g_r)$ -regularized C_2 -uniqueness family for (296).

On the other hand, the class of exponentially equicontinuous pre- (k, C_1, \mathcal{V}) -existence families for (296) cannot be so simply characterized with the help of vectorvalued Laplace transform. As the next theorem shows, this is not the case with the class of exponentially equicontinuous (k, C_1, \mathcal{V}) -existence families for (296):

THEOREM 3.2.38. Suppose that $\tau = \infty$, $k \in C([0,\infty))$, k(t) and $|a_j|(t)$ $(0 \leq j \leq n-1)$ satisfy (P1) and $(E(t))_{t\geq 0}$ is a (k, C_1, \mathcal{V}) -existence family for (296). Let \mathcal{A}_j be closed for $0 \leq j \leq n$. Suppose that there exists a real number $\omega \geq \max(0, \operatorname{abs}(k), \operatorname{abs}(|a_j|))$ $(0 \leq j \leq n-1)$ such that the operator family $\{e^{-\omega t}E(t): t \geq 0\} \subseteq L(Z, X)$ is equicontinuous, as well as that the continuous sections $u_{j,z} \in \operatorname{sec}_c(\mathcal{A}_j E(\cdot) z)$ $(j \in \mathbb{N}_n^0 \setminus \mathcal{V})$ and $u_{j,\mathcal{A}_j,z} \in \operatorname{sec}_c(\mathcal{A}_j(a_j * E(\cdot) z))$, appearing in (301), are exponentially bounded with the exponential growth bound less than or equal to ω (and the meaning clear). Set

$$\mathcal{P}_{\lambda} := \sum_{j=0}^{n} \widetilde{a_j}(\lambda) \mathcal{A}_j, \quad \operatorname{Re} \lambda > \omega$$

Then, for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\widetilde{a}_{j}(\lambda) \neq 0$ $(j \in \mathcal{V})$, we have

$$\tilde{k}(\lambda)C_1z \in \mathcal{P}_{\lambda}\tilde{E}(\lambda)z, \quad z \in Z.$$

PROOF. Performing the Laplace transform, we get:

(302)
$$\sum_{j \in \mathbb{N}_n^0 \smallsetminus \mathcal{V}} \widetilde{a_j}(\lambda) \widetilde{u_{j,z}}(\lambda) + \sum_{j \in \mathcal{V}} \widetilde{u_{j,\mathcal{A}_j,z}}(\lambda) + \widetilde{k}(\lambda) C_1 z = 0, \quad \operatorname{Re} \lambda > \omega, \ z \in Z.$$

Since the operators \mathcal{A}_j are closed for $0 \leq j \leq n$, we can apply Theorem 1.2.3 in order to see that

$$\widetilde{u_{j,z}}(\lambda) \in \mathcal{A}_j \widetilde{E}(\lambda) z \quad (j \in \mathbb{N}_n^0 \smallsetminus \mathcal{V}, \operatorname{Re} \lambda > \omega)$$

and

$$\widetilde{u_{j,\mathcal{A}_j,z}}(\lambda) \in \mathcal{A}_j(a_j * \widetilde{E}(\cdot)z)(\lambda) = \mathcal{A}_j[\widetilde{a_j}(\lambda)\widetilde{E}(\lambda)z] \quad (j \in \mathcal{V}, \ \operatorname{Re} \lambda > \omega).$$

Substituting this in (302) and using the fact that $\mathcal{A}_j[\tilde{a}_j(\lambda)\tilde{E}(\lambda)z] = \tilde{a}_j(\lambda)\mathcal{A}_j\tilde{E}(\lambda)z$ $(j \in \mathcal{V}, \text{ Re } \lambda > \omega, \tilde{a}_j(\lambda) \neq 0)$, the result immediately follows.

The assertion of Theorem 2.3.33(i)/(b) can be extended to multivalued linear operators by assuming some very restrictive additional conditions. Contrary to this, the assertion of Theorem 2.3.32(i) admits a very simple reformulation in our context:

THEOREM 3.2.39. Suppose that $0 \leq \alpha_0 < \cdots < \alpha_n$ and $\mathcal{V} = \mathbb{N}_n^0$. Let $(E(t))_{t \in [0,\tau)}$ be a (g_1, C_1, \mathcal{V}) -existence family for (296), with $a_j(t) := g_{\alpha_n - \alpha_j}(t)$ $(0 \leq j \leq n-1)$, and let $0 < T < \tau$. Suppose, further, that \mathcal{A}_j is closed for $0 \leq j \leq n$. Set $m_i := \lceil \alpha_i \rceil$, $i \in \mathbb{N}_n^0$ and, for every $i \in \mathbb{N}_{m_n-1}^0$, $D_i := \{j \in \mathbb{N}_{n-1} : m_j - 1 \geq i\}$, $D'_i := \{j \in \mathbb{N}_{n-1}^0 : m_j - 1 \geq i\}$, $D''_i := \mathbb{N}_{n-1}^0 \setminus D'_i$ and

$$\mathbf{D}_i := \bigg\{ u_i \in \bigcap_{j \in D_i''} D(\mathcal{A}_j) : \mathcal{A}_j u_i \cap R(C_1) \neq \emptyset, \ j \in D_i'' \bigg\}.$$

If $u_i \in \mathbf{D}_i$ for $0 \leq i \leq m_n - 1$, then we define u(t) by (119) with the term $E^{(m_n 1 - i)}(\cdot)$ replaced by $E(\cdot)$ therein, and with elements $v_{i,j} \in Z$ satisfying $\mathcal{A}_j u_i \cap C_1 v_{i,j} \neq \emptyset$ for $0 \leq j \leq n - 1$. Then the Caputo fractional derivative $\mathbf{D}_t^{\zeta} u(t)$ is defined for any number $\zeta \in [0, \alpha_n]$ and, for every $j \in \mathbb{N}_n^0$, there exists a continuous section $u_{j,\alpha_j}(t)$ of $\mathcal{A}_j \mathbf{D}_t^{\alpha_j}(t)$ such that

$$\sum_{j=0}^{n} u_{j,\alpha_j}(t) = 0, \quad 0 \leqslant t \leqslant T.$$

Before we briefly describe how we can provide some illustrative examples of results obtained so far, it is worth noting that the assertion of Theorem 2.3.34, concerning the inhomogeneous problem (296) with single-valued linear operators $\mathcal{A}_j = \mathcal{A}_j$ ($0 \leq j \leq n$), cannot be extended to the multivalued linear operators case. Possible applications of Theorem 3.2.39 to matrix differential equations with Caputo fractional derivatives will be analyzed somewhere else.

1. In [199, Example 6.1], the authors have considered a damped Poissonwave equation in the space $X := L^2(\Omega)$, where $\Omega \subseteq \mathbb{R}^n$ is a bounded open domain with smooth boundary. Let $m(x) \in L^{\infty}(\Omega)$, $m(x) \ge 0$ a.e. $x \in \Omega$, let Δ be the Dirichlet–Laplacian in $L^2(\Omega)$, acting with domain $H_0^1(\Omega) \cap$ $H^2(\Omega)$, and let A(x; D) be a second order linear differential operator on Ω with coefficients continuous on $\overline{\Omega}$. Using the analysis from the abovementioned example (cf. also the proof of [199, Theorem 6.1]), it readily follows from Theorem 3.2.36 that there exists $\zeta > 0$ such that there exists an exponentially bounded g_{ζ} -regularized $\Delta(1-\Delta)^{-1}$ -uniqueness family for problem (296), with n = 2, $\mathcal{B} = B \in L(X)$ being the scalar multiplication with m(x), $\mathcal{A}_0 := -\Delta$, $\mathcal{A}_1 := A(x; D) := A$, $a_0(t) := g_1(t)$ and $a_1(t) :=$ $g_2(t)$. Applying Theorem 3.2.32 and Theorem 3.2.37, we obtain that for a large class of inhomogeneities f(t, x) there exists a unique solution of the abstract degenerate Volterra integral equation:

$$\begin{split} m(x)u(t,x) &- \int_0^t c(t-s)\Delta u(s,x)ds \\ &+ A\int_0^t (c*c)(t-s)u(s,x)ds = f(t,x), \quad t \geqslant 0, \ x \in \Omega, \end{split}$$

where c(t) is a completely positive function. Observe, finally, that the subordination principles clarified in Theorem 3.2.31-Theorem 3.2.32 and Theorem 3.2.37 can be applied to many other problems from [199, Chapter VI].

2. Suppose that \mathcal{A} is a closed subgenerator of an exponentially equicontinuous (g_{δ}, g_{σ}) -regularized *C*-resolvent family $(R_{\delta}(t))_{t \geq 0}$ $(0 < \delta \leq 2, \sigma \geq 1)$. Then the analyses contained in [**292**, Example 2.10.30(i)] and Remark 3.2.29, along with Theorem 3.2.36, can be used for proving some results on the existence and uniqueness of solutions of the abstract degenerate Volterra inclusion:

$$0 \in u(t) + \sum_{j=1}^{n-1} c_j (g_{\alpha_n - \alpha_j} * u)(t) - \mathcal{A}(g_{\alpha_n} * u)(t) + f(t), \quad t \ge 0$$

where $0 \leq \alpha_1 < \alpha_2 < \cdots < \alpha_n < \delta$ and $c_j \in \mathbb{C}$ $(1 \leq j \leq n-1)$.

3.2.2. Non-injectivity of regularizing operators C_2 and C. In this subsection, we consider multivalued linear operators as subgenerators of mild (a, k)-regularized (C_1, C_2) -resolvent operator families and (a, k)-regularized C-resolvent operator families. We use the same notion and notation as before but now we allow that the operators C_2 and C are possibly non-injective (see Definition 3.2.1–Definition 3.2.2 and Definition 3.2.34). Without any doubt, this choice has some obvious displeasing consequences on the uniqueness of corresponding abstract Volterra integro-differential inclusions (see e.g. Proposition 3.2.8(ii), Theorem 3.2.9(ii) and Proposition 3.2.35).

We assume that X and Y are two SCLCSs, $0 < \tau \leq \infty$, $k \in C([0,\tau))$, $k \neq 0$, $a \in L^1_{loc}([0,\tau))$, $a \neq 0$, $\mathcal{A}: X \to P(X)$ is an MLO, $C_1 \in L(Y,X)$, $C, C_2 \in L(X)$ and $C\mathcal{A} \subseteq \mathcal{A}C$. We define the integral generator \mathcal{A}_{int} of a mild (a,k)-regularized C_2 -uniqueness family $(R_2(t))_{t\in[0,\tau)}$ (mild (a,k)-regularized (C_1, C_2) -existence and uniqueness family $(R_1(t), R_2(t))_{t\in[0,\tau)}$; (a,k)-regularized C-regularized family $(R(t))_{t\in[0,\tau)}$) in the same way as before. Then we have that $\mathcal{A}_{int} \subseteq C_2^{-1}\mathcal{A}_{int}C_2$ ($\mathcal{A}_{int} \subseteq C^{-1}\mathcal{A}_{int}C$) is still the maximal subgenerator of $(R_2(t))_{t\in[0,\tau)}$ ($(R(t))_{t\in[0,\tau)}$) with respect to the set inclusion and the local equicontinuity of $(R_2(t))_{t\in[0,\tau)}$ ($(R(t))_{t\in[0,\tau)}$) implies that \mathcal{A}_{int} is closed; as the next illustrative example shows, $C^{-1}\mathcal{A}_{int}C$ need not be a subgenerator of $(R(t))_{t\in[0,\tau)}$ and the inclusion $C^{-1}\mathcal{A}_{int}C \subseteq \mathcal{A}_{int}$ is not true for resolvent operator families, in general:

EXAMPLE 3.2.40. Let $T(t) := C \in L(X)$ for all $t \ge 0$. Then $(T(t))_{t\ge 0}$ is a global *C*-regularized semigroup (a(t) = k(t) = 1) with the integral generator $\mathcal{A}_{int} = X \times N(C)$ and any MLO \mathcal{A} satisfying $R(C) \times \{0\} \subseteq \mathcal{A}$ and $R(\mathcal{A}) \subseteq N(C)$ is a subgenerator of $(T(t))_{t\ge 0}$. In particular, $\mathcal{A} = R(C) \times N(C)$ is a subgenerator of $(T(t))_{t\ge 0}$, $C^{-1}\mathcal{A}C = C^{-1}\mathcal{A}_{int}C = X \times N(C^2)$ which is, in general, a proper extension of \mathcal{A}_{int} , and not a subgenerator of $(T(t))_{t\ge 0}$ provided that there exists an element $x \in X$ such that $C^2x = 0$ and $Cx \neq 0$. Observe, finally, that $(T(t))_{t\ge 0}$ is exponentially equicontinuous and $N(C) \subseteq (\lambda - \mathcal{A}_{int})^{-1}C0$ for all $\lambda > 0$, so that the operator $(\lambda - \mathcal{A}_{int})^{-1}C$ need not be single-valued in the case that C is not injective.

Suppose that a(t) is a kernel on $[0, \tau)$, \mathcal{A} and \mathcal{B} are two subgenerators of an (a, k)-regularized C-resolvent family $(R(t))_{t \in [0, \tau)}$, and $x \in D(\mathcal{A}) \cap D(\mathcal{B})$. Then $R(t)(y-z) = 0, t \in [0, \tau)$ for each $y \in \mathcal{A}x$ and $z \in \mathcal{B}x$. Furthermore, the local equicontinuity of $(R(t))_{t \in [0, \tau)}$ and the closedness of \mathcal{A} imply that the inclusion (275) continues to hold without injectivity of C being assumed.

In the following definition, we introduce the notion of an (a, k, C)-subgenerator of any strongly continuous operator family $(Z(t))_{t \in [0,\tau)} \subseteq L(X)$. This definition extends the corresponding ones introduced by C.-C. Kuo [366, 367, Definition 2.4] in the setting of Banach spaces, where it has also been assumed that the operator $\mathcal{A} = A$ is linear and single-valued.

DEFINITION 3.2.41. Let $0 < \tau \leq \infty$, $C \in L(X)$, $a \in L^1_{loc}([0,\tau))$, $a \neq 0$, $k \in C([0,\tau))$ and $k \neq 0$. Suppose that $(Z(t))_{t \in [0,\tau)} \subseteq L(X)$ is a strongly continuous operator family. By an (a, k, C)-subgenerator of $(Z(t))_{t \in [0,\tau)}$ we mean any MLO \mathcal{A} in X satisfying the following two conditions:

- (i) $Z(t)x k(t)Cx = \int_0^t a(t-s)Z(s)y\,ds$, whenever $t \in [0,\tau)$ and $y \in \mathcal{A}x$.
- (ii) For all $x \in X$ and $t \in [0, \tau)$, we have $\int_0^t a(t-s)Z(s)x\,ds \in D(\mathcal{A})$ and $Z(t)x k(t)Cx \in \mathcal{A} \int_0^t a(t-s)Z(s)x\,ds$.

The (a, k, C)-integral generator \mathcal{A}_{int} of $(Z(t))_{t \in [0,\tau)}$ (integral generator, if there is no risk for confusion) is defined by

$$\mathcal{A}_{int} := \left\{ (x,y) \in X \times X : Z(t)x - k(t)Cx = \int_0^t a(t-s)Z(s)y \, ds \text{ for all } t \in [0,\tau) \right\}.$$

If \mathcal{A} is a subgenerator of $(Z(t))_{t\in[0,\tau)}$, then it is clear that $(Z(t))_{t\in[0,\tau)}$ is a mild (a, k)-regularized (C, C)-existence and uniqueness family which do have \mathcal{A} as subgenerator. Since we have not assumed that \mathcal{A} commutes with C or $(Z(t))_{t\in[0,\tau)}$, it does not follow automatically from Definition 3.2.41 that $(Z(t))_{t\in[0,\tau)}$ is an (a, k)regularized C-resolvent family with subgenerator \mathcal{A} . By $\chi(Z)$ we denote the set consisting of all subgenerators of $(Z(t))_{t\in[0,\tau)}$. The local equicontinuity of $(Z(t))_{t\in[0,\tau)}$ yields that for each subgenerator $\mathcal{A} \in \chi(Z)$ we have $\overline{\mathcal{A}} \in \chi(Z)$. The set $\chi(Z)$ can have infinitely many elements; if $\mathcal{A} \in \chi(Z)$, then $\mathcal{A} \subseteq \mathcal{A}_{int}$ (cf. [**386**, Example 4.10, 4.11]; in these examples, the partially ordered set $(\chi_{sv}(Z), \subseteq)$, where $\chi_{sv}(Z)$ denotes the set consisting of all single-valued linear subgenerators of $(Z(t))_{t\in[0,\tau)}$, does not have the greatest element) and, if $\chi(Z)$ is finite, then it need not be a singleton [291]. In general, the set $\chi(Z)$ can be empty and the integral generator of $(Z(t))_{t \in [0,\tau)}$ need not be a subgenerator of $(Z(t))_{t \in [0,\tau)}$ in the case that $\tau < \infty$:

EXAMPLE 3.2.42. [385] Let $0 < \tau < \infty$ and let $U: [\tau/2, \tau) \to L(X)$ be a strongly continuous function such that $U(\tau/2) = 0$ and U(t) is injective for all $t \in (\tau/2, \tau)$. Define $T: [0, \tau) \to L(X \times X)$ by $T(t)(x \ y)^T := (0 \ y)^T$ for $t \in [0, \tau/2)$, $x, y \in X$ and $T(t)(x \ y)^T := (U(t)x \ y)^T$ for $t \in [\tau/2, \tau)$, $x, y \in X$. Set C := T(0). Then C is not injective, $(T(t))_{t \in [0, \tau)} \subseteq L(X \times X)$ is a non-degenerate local Cregularized semigroup (cf. the next subsection for the notion) and it can be easily seen that the violation of condition

$$U(t) \int_{\tau/2}^{s} U(r)x \, dr = \int_{\tau/2}^{t} U(r)U(s)x \, dr, \quad x \in E, \ t, \ s \in [\tau/2, \tau)$$

implies that the set $\chi(T)$ is empty (in the MLO sense) as well as that the integral generator \mathcal{A}_{int} of $(T(t))_{t\in[0,\tau)}$ is not a subgenerator of $(T(t))_{t\in[0,\tau)}$. We can similarly construct an example of a non-degenerate local *C*-regularized cosine function $(C(t))_{t\in[0,\tau)}$, with *C* being not injective, so that $(C(t))_{t\in[0,\tau)}$ does not have any subgenerator in the MLO sense [487]. Observe, finally, that the notion introduced in [386, Definition 4.3], [531, Definition 2.1] and [291, Remark 2.1.8(i)] cannot be used for proving the nonemptiness of set $\chi(Z)$ in the degenerate case.

If \mathcal{A} and \mathcal{B} are subgenerators of $(Z(t))_{t\in[0,\tau)}$, then for any complex numbers α, β such that $\alpha + \beta = 1$ we have that $\alpha \mathcal{A} + \beta \mathcal{B}$ is a subgenerator of $(Z(t))_{t\in[0,\tau)}$. Set $\mathcal{A} \wedge B := (1/2)\mathcal{A} + (1/2)\mathcal{B}$. We define the operator $\mathcal{A} \vee_0 \mathcal{B}$ by $D(\mathcal{A} \vee_0 \mathcal{B}) := \operatorname{span}[D(\mathcal{A}) \cup D(\mathcal{B})]$ and

$$\mathcal{A} \vee_0 \mathcal{B}(ax+by) := aAx+bBy, \quad x \in D(\mathcal{A}), \ y \in D(\mathcal{B}), \ a, b \in \mathbb{C};$$

 $\mathcal{A} \vee \mathcal{B} := \overline{\mathcal{A} \vee_0 \mathcal{B}}$. Then $\mathcal{A} \vee_0 \mathcal{B}$ is a subgenerator of $(Z(t))_{t \in [0,\tau)}$, and $\mathcal{A} \vee \mathcal{B}$ is a subgenerator of $(Z(t))_{t \in [0,\tau)}$, provided that $(Z(t))_{t \in [0,\tau)}$ is locally equicontinuous. In the case of non-degenerate K-convoluted C-semigroups, C injective, it is well known that the set $\chi(Z)$, equipped with the operations \wedge and \vee , forms a complete Boolean lattice [531], [291, Remark 2.1.8(ii)-(iii)]. We will not discuss the properties of $(\chi(Z), \wedge, \vee)$ in general case.

If \mathcal{A} is a closed, $\mathcal{A} \in \chi(Z)$, $0 \in \operatorname{supp}(a)$ and $y \in \mathcal{A}x$, then we have $(\int_0^t a(t-s)Z(s)x\,ds, Z(t)x-k(t)Cx) \in \mathcal{A}, t \in [0,\tau)$, i.e.,

(303)
$$\left(\int_0^t a(t-s)Z(s)x\,ds,\int_0^t a(t-s)Z(s)y\,ds\right) \in \mathcal{A}, \quad t \in [0,\tau).$$

Suppose now that $\tau_0 \in (0, \tau)$. By [291, Theorem 3.4.40], there exists a sequence $(f_n)_{n \in \mathbb{N}}$ in $L^1[0, \tau_0]$ such that $(a * f_n)(t) \to g_1(t)$ in $L^1[0, \tau_0]$. Then the closedness of \mathcal{A} along with (303) shows that $(\int_0^t g_1(t-s)Z(s)x\,ds, \int_0^t g_1(t-s)Z(s)y\,ds) \in \mathcal{A}, t \in [0, \tau_0]$. After differentiation, we get that $(Z(t)x, Z(t)y) \in \mathcal{A}, t \in [0, \tau_0]$ and since τ_0 was arbitrary, we have that $Z(t)\mathcal{A} \subseteq \mathcal{A}Z(t), t \in [0, \tau)$ for any closed subgenerator \mathcal{A} of $(Z(t))_{t\in[0,\tau)}$. If this is the case and $Z(t)C = CZ(t), t \in [0, \tau)$, then $C^{-1}\mathcal{A}C$ also commutes with Z(t): Suppose that $(x, y) \in C^{-1}\mathcal{A}C$. Then $Cy \in \mathcal{A}Cx, CZ(t)x = Z(t)Cx \in D(\mathcal{A}), t \in [0, \tau)$ and $CZ(t)y = Z(t)Cy \in CY(t)$.

 $Z(t)\mathcal{A}Cx \subseteq \mathcal{A}CZ(t)x = \mathcal{A}Z(t)Cx, t \in [0,\tau)$ so that $Z(t)y \in C^{-1}\mathcal{A}CZ(t)x$. $t \in [0,\tau)$ and $Z(t)[C^{-1}\mathcal{A}C] \subset [C^{-1}\mathcal{A}C]Z(t), t \in [0,\tau)$. Suppose again that \mathcal{A} is a closed subgenerator of $(Z(t))_{t\in[0,\tau)}, 0 \in \operatorname{supp}(a)$ and $y \in \mathcal{A}x$. Then $(\int_0^t a(t-s)Z(s)y\,ds, Z(t)y - k(t)Cy) = (Z(t)x - k(t)Cx, Z(t)y - k(t)Cy) \in \mathcal{A}, \\ t \in [0,\tau).$ Since $(Z(t)x, Z(t)y) \in \mathcal{A}, t \in [0,\tau),$ the above easily implies that $(Cx, Cy) \in \mathcal{A}$ so that $C\mathcal{A} \subseteq \mathcal{A}C$, i.e., $\mathcal{A} \subseteq C^{-1}\mathcal{A}C$. Now we proceed by repeating some parts of the proof of [291, Proposition 2.1.6(i)]. Let $(x, y) \in \mathcal{A}_{int}$. As above, we have $(\int_0^t a(t-s)Z(s)x\,ds, \int_0^t a(t-s)Z(s)y\,ds) \in \mathcal{A}, t \in [0,\tau)$ and $(Z(t)x, Z(t)y) \in \overline{\mathcal{A}} = \mathcal{A}, t \in [0,\tau)$. This implies $Z(t)y \in \mathcal{A}Z(t)x = \mathcal{A}[\Theta(t)Cx + \int_0^t a(t-s)Z(s)y\,ds]$, $t \in [0, \tau)$ and, since $\int_0^t a(t-s)Z(s)y \, ds \in D(\mathcal{A})$ for $t \in [0, \tau)$, $Cx \in D(\mathcal{A})$ as well as $0 \in \mathcal{A}[\Theta(t)Cx + \int_0^t a(t-s)Z(s)y \, ds - \int_0^t a(t-s)Z(s)y \, ds] - \Theta(t)Cy, t \in [0, \tau)$. Hence, $Cy \in \mathcal{A}Cx$ and $\mathcal{A}_{int} \subseteq C^{-1}\mathcal{A}C$. If, additionally, the operator C is injective and $Z(t)C = CZ(t), t \in [0, \tau)$, then we can simply verify that $C^{-1}AC$ is likewise a closed subgenerator of $(W(t))_{t \in [0,\tau)}$, so that $\mathcal{A}_{int} = C^{-1}\mathcal{A}C$ by previously proved inclusion $\hat{\mathcal{A}} \subseteq C^{-1}\mathcal{A}C$ and the fact that \mathcal{A}_{int} extends any subgenerator from $\chi(W)$. Let \mathcal{A} and \mathcal{B} be two subgenerators of $(Z(t))_{t \in [0,\tau)}$, let \mathcal{B} be closed, and let a(t) kernel on $[0,\tau)$. Suppose that $y \in \mathcal{A}x$. Then $\left(\int_0^t a(t-s)Z(s)y\,ds, Z(t)y - k(t)Cy\right) =$ $(Z(t)x - k(t)Cx, Z(t)y - k(t)Cy) \in \mathcal{B}, t \in [0, \tau)$, which implies by Theorem 1.2.3 that $((a * Z)(t)x - (a * k)(t)Cx, (a * Z)(t)y - (a * k)(t)Cy) \in \mathcal{B}, t \in [0, \tau)$. Since $(a * Z)(t)x \in D(\mathcal{B}), t \in [0, \tau)$, the above implies that $Cx \in D(\mathcal{B})$. Hence, $C(D(\mathcal{A})) \subset D(\mathcal{B})$. We continue by observing that Proposition 3.2.3, Proposition 3.2.8, Proposition 3.2.13, the equation (274) and assertions clarified in the paragraph directly after Theorem 3.2.7 continue to hold without any terminological changes. If $(R_1(t), R_2(t))_{t \in [0,\tau)}$ is strongly continuous and (274) holds, then it can be easily seen that the integral generator \mathcal{A}_{int} of $(R_2(t))_{t\in[0,\tau)}$ is a subgenerator of a mild (a, k)-regularized (C_1, C_2) -existence and uniqueness family $(R_1(t), R_2(t))_{t \in [0,\tau]}$. As Example 3.2.42 shows, this is no longer true if (274) holds only for $0 \leq t, s, t + s < \tau$. Now we will state the following proposition.

PROPOSITION 3.2.43. Suppose that \mathcal{A} is a closed MLO, $0 < \tau \leq \infty$, $a \in L^1_{loc}([0,\tau))$, $a * a \neq 0$ in $L^1_{loc}([0,\tau))$, $k \in C([0,\tau))$ and $k \neq 0$. If $\pm \mathcal{A}$ are subgenerators of mild (a, k)-regularized C_1 -existence families $(R_{1,\pm}(t))_{t\in[0,\tau)}$ (mild (a, k)-regularized C_2 -uniqueness families $(R_{2,\pm}(t))_{t\in[0,\tau)}$; (a, k)-regularized C-resolvent families $(R_{\pm}(t))_{t\in[0,\tau)}$), then \mathcal{A}^2 is a subgenerator of a mild (a * a, k)-regularized C_1 -existence family $(R_1(t) \equiv (1/2)R_1(t) + (1/2)R_{1,-}(t))_{t\in[0,\tau)}$ (mild (a * a, k)-regularized C_2 -uniqueness family $(R_2(t) \equiv (1/2)R_2(t) + (1/2)R_{2,-}(t))_{t\in[0,\tau)}$; mild (a * a, k)-regularized C-resolvent family $(R(t) \equiv (1/2)R_+(t) + (1/2)R_-(t))_{t\in[0,\tau)}$).

PROOF. We will prove the proposition only for mild (a, k)-regularized C_1 existence families. Let $x \in E$ and $t \in [0, \tau)$ be fixed. Then $\frac{1}{2}[R_{1,+}(t)x - R_{1,-}(t)x] = \frac{1}{2}[R_{1,+}(t)x - k(t)C_1x] - [R_{1,-}(t)x - k(t)C_1x] \in \frac{1}{2}\mathcal{A}(a * R_{1,+}(\cdot)x)(t) + \frac{1}{2}\mathcal{A}(a * R_{1,-}(\cdot)x)(t) = \mathcal{A}(a * R_1(\cdot)x)(t)$. Applying Theorem 1.2.3, we get that $\frac{1}{2}(a * [R_{1,+}(\cdot)x - R_{1,-}(\cdot)x])(t) \in \mathcal{A}(a * a * R_1(\cdot)x)(t)$. Since $\pm \mathcal{A}$ are subgenerators of mild (a, k)-regularized C_1 -existence families $(R_{1,\pm}(t))_{t\in[0,\tau)}$, the above inclusion implies $(a * a * R_1(\cdot)x)(t) \in D(\mathcal{A}^2) \text{ and } \frac{1}{2}([R_{1,+}(t)x - k(t)C_1x] + [R_{1,-}(t)x - k(t)C_1x]) = R_1(t)x - k(t)C_1x \in \mathcal{A}^2(a * a * R_1(\cdot)x)(t), \text{ as required.}$

The following analogues of Theorem 3.2.4[(i),(iii)] and Theorem 3.2.5 hold true.

THEOREM 3.2.44. Suppose that \mathcal{A} is a closed MLO in $X, C_1 \in L(Y, X), C_2 \in L(X), |a(t)|$ and k(t) satisfy (P1), as well as that $(R_1(t), R_2(t))_{t \ge 0} \subseteq L(Y, X) \times L(X)$ is strongly continuous. Let $\omega \ge \max(0, \operatorname{abs}(|a|), \operatorname{abs}(k))$ be such that the operator family $\{e^{-\omega t}R_i(t) : t \ge 0\}$ is equicontinuous for i = 1, 2. Then the following holds:

(i) $(R_1(t), R_2(t))_{t \ge 0}$ is a mild (a, k)-regularized (C_1, C_2) -existence and uniqueness family with a subgenerator \mathcal{A} iff for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0$, we have $R(C_1) \subseteq \mathbb{R}(I - \tilde{a}(\lambda)\mathcal{A})$,

(304)
$$\int_0^\infty e^{-\lambda t} R_1(t) y \, dt \in \tilde{k}(\lambda) (I - \tilde{a}(\lambda)\mathcal{A})^{-1} C_1 y, \quad y \in Y,$$

and

(305)
$$\tilde{k}(\lambda)C_2x = \int_0^\infty e^{-\lambda t} [R_2(t)x - (a * R_2)(t)y] dt, \text{ whenever } (x, y) \in \mathcal{A}.$$

(ii) $(R_2(t))_{t\geq 0}$ is a mild (a,k)-regularized C_2 -uniqueness family with a subgenerator \mathcal{A} iff (305) holds for $\operatorname{Re} \lambda > \omega$.

THEOREM 3.2.45. Suppose that \mathcal{A} is a closed MLO in $X, C \in L(X), C\mathcal{A} \subseteq \mathcal{A}C, |a(t)|$ and k(t) satisfy (P1), as well as that $(R(t))_{t \ge 0} \subseteq L(X)$ is strongly continuous and commutes with C on X. Let $\omega \ge \max(0, \operatorname{abs}(|a|), \operatorname{abs}(k))$ be such that the operator family $\{e^{-\omega t}R(t) : t \ge 0\}$ is equicontinuous. Then $(R(t))_{t\ge 0}$ is an (a, k)-regularized C-resolvent family with a subgenerator \mathcal{A} iff for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0$, we have $R(C) \subseteq \operatorname{R}(I - \tilde{a}(\lambda)\mathcal{A})$, (304) holds with $R_1(\cdot), C_1$ and Y, y replaced with $R(\cdot), C$ and X, x therein, as well as (305) holds with $R_2(\cdot)$ and C_2 replaced with $R(\cdot)$ and C therein.

Keeping in mind Theorem 3.2.45 and [292, Theorem 1.2.2], it is very simple to prove the following complex characterization theorem (cf. Theorem 3.2.10):

THEOREM 3.2.46. Suppose that \mathcal{A} is a closed MLO in $X, C \in L(X), C\mathcal{A} \subseteq \mathcal{A}C, |a(t)| and k(t) satisfy (P1), <math>\omega_0 > \max(0, \operatorname{abs}(|a|), \operatorname{abs}(k))$ and, for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega_0$ and $\tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0$, we have $R(C) \subseteq R(I - \tilde{a}(\lambda)\mathcal{A})$. If there exists a function $\Upsilon : \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda > \omega_0\} \to L(X)$ which satisfies:

- (a) $\Upsilon(\lambda)x \in \tilde{k}(\lambda)(I \tilde{a}(\lambda)\mathcal{A})^{-1}Cx$ for $\operatorname{Re} \lambda > \omega_0$, $\tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0$, $x \in X$,
- (b) the mapping $\lambda \mapsto \Upsilon(\lambda)x$, $\operatorname{Re} \lambda > \omega_0$ is analytic for every fixed $x \in X$,
- (c) there exists $r \ge -1$ such that the family $\{\lambda^{-r}\Upsilon(\lambda) : \operatorname{Re} \lambda > \omega_0\} \subseteq L(X)$ is equicontinuous,
- (d) $\Upsilon(\lambda)x \tilde{a}(\lambda)\Upsilon(\lambda)y = k(\lambda)Cx$ for $\operatorname{Re} \lambda > \omega_0$, $(x, y) \in \mathcal{A}$, and
- (e) $\Upsilon(\lambda)Cx = C\Upsilon(\lambda)x$ for $\operatorname{Re} \lambda > \omega_0, x \in X$,

then, for every $\alpha > 1$, \mathcal{A} is a subgenerator of a global $(a, k * g_{\alpha+r})$ -regularized C-resolvent family $(R_{\alpha}(t))_{t \ge 0}$ which satisfies that the family $\{e^{-\omega_0 t}R_{\alpha}(t) : t \ge 0\} \subseteq L(X)$ is equicontinuous.

The real representation theorem for generation of degenerate (a, k)-regularized C-resolvent families can be also formulated but the assertion of Theorem 3.2.12(ii) is not attainable in the case that the operator C is not injective. The assertions of Theorem 3.2.7, Proposition 3.2.36 and Proposition 3.2.37 continue to hold with minimal terminological changes. Since the identity (291) holds for degenerate (a, k)-regularized C-resolvent families, with C being not injective, Proposition 3.2.15 can be reformulated without substantial difficulties, as well, but we cannot prove the uniqueness of solutions of corresponding abstract time-fractional inclusions.

As already mentioned, the adjoint type theorems [**292**, Theorem 2.1.12(i)/(ii); Theorem 2.1.13] continue to hold for (a, k)-regularized *C*-regularized families subgenerated by closed multivalued linear operators and it is not necessary to assume that the operator \mathcal{A} is densely defined in the case of consideration of [**292**, Theorem 2.1.12(i)]. All this remains true if the operator *C* is not injective, when we also do not need to assume that R(C) is dense in *X*.

If C is not injective, then we introduce the notion of (exponential equicontinuous) analyticity of degenerate (a, k)-regularized C-resolvent families in the same way as in Definition 3.2.16. Then Theorem 3.2.18 does not admit a satisfactory reformulation in our new frame. On the other hand, the assertion of Theorem 3.2.19 can be rephrased by taking into consideration the conditions (d)–(e) from Theorem 3.2.46. Differential properties of degenerate (a, k)-regularized C-resolvent families clarified in Theorem 3.2.25–Theorem 3.2.26 continue to hold after a reformulation of the same type.

3.2.3. Degenerate K-convoluted C-semigroups and degenerate K-convoluted C-cosine functions in locally convex spaces. It is well known that the notions of a degenerate (local) K-convoluted C-semigroup and a degenerate (local) K-convoluted C-cosine function in locally convex space can be introduced in a slightly different manner, by using the convoluted versions of semigroup property and d'Alambert functional equation. The main aim of this subsection is to consider the classes of degenerate K-convoluted C-semigroups and degenerate K-convoluted C-cosine functions in locally convex spaces by following this approach (cf. also [291, 296, 366, 367, 386] for similar concepts).

The underlying sequentially complete locally convex space will be denoted by E. In this subsection, we will not require the injectiveness of regularized operator $C \in L(E)$. If $0 < \tau \leq \infty$ and $(W(t))_{t \in [0,\tau)} \subseteq L(E)$ is strongly continuous, then we denote, as before, $W^{[n]}(t)x = \int_0^t g_n(t-s)W(s)x\,ds, x \in E, t,s \in [0,\tau), n \in \mathbb{N}$.

DEFINITION 3.2.47. Let $0 \neq K \in L^1_{loc}([0, \tau))$. A strongly continuous operator family $(S_K(t))_{t \in [0,\tau)} \subseteq L(E)$ is called a (local, if $\tau < \infty$) K-convoluted C-semigroup iff the following holds:

(i) $S_K(t)C = CS_K(t), t \in [0, \tau)$, and

(ii) For all $x \in E$ and $t, s \in [0, \tau)$ with $t + s \in [0, \tau)$, we have

$$S_K(t)S_K(s)x = \left[\int_0^{t+s} - \int_0^t - \int_0^s\right]K(t+s-r)S_K(r)Cx\,dr.$$

DEFINITION 3.2.48. Let $0 \neq K \in L^1_{loc}([0,\tau))$. A strongly continuous operator family $(C_K(t))_{t \in [0,\tau)} \subseteq L(E)$ is called a (local, if $\tau < \infty$) K-convoluted C-cosine function iff the following holds:

- (i) $C_K(t)C = CC_K(t), t \in [0, \tau)$, and
- (ii) For all $x \in E$ and $t, s \in [0, \tau)$ with $t + s \in [0, \tau)$, we have

$$2C_{K}(t)C_{K}(s)x = \left(\int_{t}^{t+s} - \int_{0}^{s}\right)K(t+s-r)C_{K}(r)Cx\,dr + \int_{t-s}^{t}K(r-t+s)C_{K}(r)Cx\,dr + \int_{0}^{s}K(r+t-s)C_{K}(r)Cx\,dr, \quad t \ge s;$$

$$2C_{K}(t)C_{K}(s)x = \left(\int_{s}^{t+s} - \int_{0}^{t}\right)K(t+s-r)C_{K}(r)Cx\,dr + \int_{s-t}^{s}K(r+t-s)C_{K}(r)Cx\,dr + \int_{0}^{t}K(r-t+s)C_{K}(r)Cx\,dr, \quad t < s.$$

By a (local) *C*-regularized semigroup, resp., (local) *C*-regularized cosine function, we mean any strongly continuous operator family $(S(t))_{t\in[0,\tau)} \subseteq L(E)$, resp., $(C(t))_{t\in[0,\tau)} \subseteq L(E)$, satisfying that S(t)C = CS(t), $t \in [0,\tau)$ and S(t+s)C =S(t)S(s) for all $t, s \in [0,\tau)$ with $t+s \in [0,\tau)$, resp., C(t)C = CC(t), $t \in [0,\tau)$ and 2C(t)C(s) = C(t+s)C + C(|t-s|)C for all $t, s \in [0,\tau)$ with $t+s \in [0,\tau)$.

The notions of local equicontinuity and exponential equicontinuity will be taken in the usual way. If $k(t) = g_{\alpha+1}(t)$, where $\alpha \ge 0$, then it is also said that $(S_K(t))_{t \in [0,\tau)}$ is an α -times integrated *C*-semigroup; 0-times integrated semigroup is nothing else but *C*-regularized semigroup. The above notions can be simply understood for the class of *K*-convoluted *C*-cosine functions.

Set $\Theta(t) := \int_0^t K(s) ds$, $t \in [0, \tau)$. For a K-convoluted C-semigroup $(S_K(t))_{t \in [0, \tau)}$, resp., K-convoluted C-cosine function $(C_K(t))_{t \in [0, \tau)}$, we define its (integral) generator $\hat{\mathcal{A}}$ by graph

$$\hat{\mathcal{A}} := \left\{ (x,y) \in E \times E : S_K(t)x - \Theta(t)Cx = \int_0^t S_K(s)y \, ds, \ t \in [0,\tau) \right\}, \text{ resp.},$$
$$\hat{\mathcal{A}} := \left\{ (x,y) \in E \times E : C_K(t)x - \Theta(t)Cx = \int_0^t (t-s)C_K(s)y \, ds, \ t \in [0,\tau) \right\};$$

with $\Theta(t) \equiv 1$, we obtain the definition of integral generator of a C-semigroup (C-cosine function).

Denote by $(W(t))_{t\in[0,\tau)}$ any of the above considered operator families, and $a(t) \equiv 1$ $(a(t) \equiv t)$ in the case of consideration semigroups (cosine functions). In what follows, we will refer to $(W(t))_{t\in[0,\tau)}$ as a (local) (a, Θ) -regularized *C*-resolvent family. It is worth noting that the functional equality of $(W(t))_{t\in[0,\tau)}$ and its strong continuity together imply that W(t)W(s) = W(s)W(t) for all $t, s \in [0, \tau)$ with $t + s \in [0, \tau]$; in general case, it is not true that W(t)W(s) = W(s)W(t) for $\tau < t + s < 2\tau$ (cf. [385] and Example 3.2.42). We will accept the following notion of non-degeneracy: $(W(t))_{t\in[0,\tau)}$ is said to be non-degenerate iff the assumption

W(t)x = 0 for all $t \in [0, \tau)$ implies x = 0 (cf. [386, p.2] for more details on the subject).

It is clear that the integral generator $\hat{\mathcal{A}}$ of $(W(t))_{t \in [0,\tau)}$ is an MLO in E and that the local equicontinuity of $(W(t))_{t \in [0,\tau)}$ implies that $\hat{\mathcal{A}}$ is closed. Furthermore, we have that $\hat{\mathcal{A}} \subseteq C^{-1}\hat{\mathcal{A}}C$ in the MLO sense, and that $\hat{\mathcal{A}} = C^{-1}\hat{\mathcal{A}}C$ provided additionally that the operator C is injective. A subgenerator of $(W(t))_{t \in [0,\tau)}$ is nothing else but an (a, Θ, C) -subgenerator of $(W(t))_{t \in [0,\tau)}$ (see Definition 3.2.41).

In [386, Example 4.10], for each $\alpha > 0$ it has been constructed an example of a global degenerate α -times integrated *C*-semigroup with infinitely many singlevalued bounded subgenerators. This example shows that the equivalence relation $A \subseteq B \Leftrightarrow D(A) \subseteq D(B)$, as well as any of the equalities $Ax = Bx, x \in D(A) \cap D(B)$ and $\rho(A) = \emptyset, A \neq \hat{A}$ does not hold for subgenerators in the degenerate case (cf. [291, Proposition 2.1.6(ii)–(iii),(viii)], and [291, Proposition 2.1.16] for cosine operator functions case). Furthermore, the subgenerators from this example do not have the same eigenvalues, in general, so that the assertion of [291, Proposition 2.1.6(v)] does not hold in the degenerate case, as well; the same example shows that the equality $C^{-1}\mathcal{A}C = C^{-1}\mathcal{B}C$ (in the MLO sense) is not generally true for subgenerators of degenerate integrated *C*-semigroups (cf. [291, Proposition 2.1.6(ii)]).

For the sequel, we need the following useful extension of [366,367, Lemma 2.1]:

LEMMA 3.2.49. Let $0 < \tau \leq \infty$, and let $(W(t))_{t \in [0,\tau)}$ be a strongly continuous operator family which commutes with C. Then the following is equivalent:

- (i) $(W(t))_{t \in [0,\tau)}$ is an (a, Θ) -regularized C-resolvent family.
- (ii) For any complex non-zero polynomial P(z) and for any a₀ ∈ C, (W^{P,a₀}(t) ≡ (P * W)(t) + a₀W(t))_{t∈[0,τ)} is an (a, P * Θ + a₀Θ)-regularized C-resolvent family.
- (iii) There exist a complex non-zero polynomial P(z) and a number $a_0 \in \mathbb{C}$ such that $(W^{P,a_0}(t))_{t\in[0,\tau)}$ is an $(a, P * \Theta + a_0\Theta)$ -regularized C-resolvent family.

PROOF. We will prove the lemma in the case that $(W(t))_{t\in[0,\tau)} = (S_K(t))_{t\in[0,\tau)}$ or $(W(t))_{t\in[0,\tau)} = (C_K(t))_{t\in[0,\tau)}$ for some $K \in L^1_{loc}([0,\tau)), K \neq 0$. We will prove the implication (i) \Rightarrow (ii) by induction on degree of P(z). Consider first the semigroup case. If dg(P) = 0, then there exists a number $a_1 \in \mathbb{C}$ such that $P(z) \equiv a_1$ and, since $(S_{\Theta}(t) \equiv S_K^{[1]}(t))_{t\in[0,\tau)}$ is a Θ -convoluted C-semigroup by the proof of [**366**, Lemma 2.1], it suffices to show that, for every $x \in E$ and for every $t, s \in [0, \tau)$ with $t + s \in [0, \tau)$, we have:

$$S_{\Theta}(t)S_{K}(s) + S_{K}(t)S_{\Theta}(s) = \left[\int_{0}^{t+s} -\int_{0}^{t} -\int_{0}^{s}\right] \{\Theta(t+s-r)S_{K}(r)Cx + K(t+s-r)S_{\Theta}(r)Cx\} dr.$$

Let such x, t, s be fixed. Since $S_K(s)S_{\Theta}(t)x = S_K(t)S_{\Theta}(s)x$, the identities [366, (2.2) and (2.4)] yield

(306)
$$\left[\int_0^{t+s} -\int_0^t -\int_0^s \right] K(t+s-r)S_{\Theta}(r)Cx \, dr$$
$$= S_K(t)S_{\Theta}(s)x + \Theta(s)S_{\Theta}(t)Cx = S_K(s)S_{\Theta}(t)x + \Theta(t)S_{\Theta}(s)Cx$$

and

$$(307) \qquad \left[\int_{0}^{t+s} -\int_{0}^{t} -\int_{0}^{s}\right] \Theta(t+s-r)S_{K}(r)Cx\,dr$$
$$= \left[\int_{0}^{t+s} -\int_{0}^{t} -\int_{0}^{s}\right]K(t+s-r)S_{\Theta}(r)Cx\,dr - \Theta(s)S_{\Theta}(t)Cx - \Theta(t)S_{\Theta}(s)Cx$$
$$= S_{K}(s)S_{\Theta}(t)x + \Theta(t)S_{\Theta}(s)Cx - \Theta(s)S_{\Theta}(t)Cx - \Theta(t)S_{\Theta}(s)Cx$$
$$= S_{K}(s)S_{\Theta}(t)x - \Theta(s)S_{\Theta}(t)Cx.$$

The claimed assertion follows by adding (306) and (307). Suppose now that (ii) holds for any complex non-zero polynomial P(z) of degree strictly less than $n \in \mathbb{N}$. any number $a_0 \in \mathbb{C}$ and any K-convoluted C-semigroup $(S_K(t))_{t \in [0,\tau)}$. Let us prove that (ii) holds for arbitrary complex non-zero polynomial $P(z) = a_{n+1}q_{n+1}(z) + dz$ $a_n g_n(z) + \cdots + a_1 g_1(z)$ and arbitrary number $a_0 \in \mathbb{C}$. Then we can always find complex numbers A_0 , A_1 , B_1 ,..., B_n such that: $a_{n+1} = A_1 B_n$, $a_i = A_1 B_{i-1} + A_1 B_n$ $A_0B_j \text{ for } 1 \leq j \leq n \text{ and } a_0 = A_0B_0, \text{ so that } W^{P,a_0}(\cdot) = S_K^{P,a_0}(\cdot) = A_1(g_1 * S_{K_1})(\cdot) + A_0S_{K_1}(\cdot) = S_{K_1}^{A_1,A_0}(\cdot), \text{ where } S_{K_1}(\cdot) = S_K^{P_1,B_0}(\cdot) \text{ with } P_1(z) = B_ng_n(z) + B_ng$ $B_{n-1}q_{n-1}(z) + \cdots + B_1q_1(z)$. Hence, (ii) follows from its validity for constant polynomials and induction hypothesis. The implication (ii) \Rightarrow (iii) is trivial and the implication (ii) \Rightarrow (iii) holds on account of the proof of [366, Lemma 2.1], with P(z) = 1 and $a_0 = 0$. The proof for K-convoluted C-cosine functions is almost the same and here it is only worth pointing out how one can prove that part (ii) holds for constant complex polynomials. Let $(C_K(t))_{t \in [0,\tau)}$ be a K-convoluted C-cosine function, let $P(z) \equiv a_1 \in \mathbb{C}$, and let $a_0 \in \mathbb{C}$. It is very simple to prove that (ii) holds provided $a_0 = 0$, so that $(C_{\Theta}(t) = C_K^{[1]}(t))_{t \in [0,\tau)}$ is a Θ -convoluted C-cosine function. In the remnant of proof, we assume that $t, s \in [0, \tau)$, $t + s < \tau$ and $t \ge s$ (the case t < s is analogous). It suffices to show that

$$(308) \quad 2[C_{\Theta}(t)C_{K}(s)x + C_{K}(t)C_{\Theta}(s)x]$$

$$= \left(\int_{t}^{t+s} - \int_{0}^{s}\right) \{\Theta(t+s-r)C_{K}(r)Cx + K(t+s-r)C_{\Theta}(r)Cx\}dr$$

$$+ \int_{t-s}^{t} \{\Theta(r-t+s)C_{K}(r)Cx + K(r-t+s)C_{\Theta}(r)Cx\}dr$$

$$+ \int_{0}^{s} \{\Theta(r+t-s)C_{K}(r)Cx + K(r+t-s)C_{\Theta}(r)Cx\}dr$$

Since $(C_{\Theta}(t))_{t \in [0,\tau)}$ is a Θ -convoluted *C*-cosine function, the final part of proof of [**291**, Theorem 2.1.13] shows that (see e.g. the equations [**291**, (36)–(41)])

(309) $2C_K(t)C_{\Theta}(s)x = 2\Theta(t)C_{\Theta}(s)Cx$

$$+\left(\int_{t}^{t+s}-\int_{0}^{s}\right)\Theta(t+s-r)C_{K}(r)Cx\,dr$$
$$+\int_{t-s}^{t}\Theta(r-t+s)C_{K}(r)Cx\,dr-\int_{0}^{s}\Theta(r+t-s)C_{K}(r)Cx\,dr.$$

Using the equality $2C_{\Theta}(t)C_K(s)x = 2(d/ds)C_{\Theta}(t)C_{\Theta}(s)x$ and the composition property for $(C_{\Theta}(t))_{t\in[0,\tau)}$, a straightforward computation shows that the value of term $2C_{\Theta}(t)C_K(s)x$, appearing on the left hand side of (308), is equal to $R_1 - R$, where R_1 , resp., R, is the term on the right hand side of (308), resp., (309). The proof of the lemma is thereby complete.

Inspecting the proofs of [366, 367, Theorem 2.2] yields the following:

THEOREM 3.2.50. Let $0 < \tau \leq \infty$, and let $(W(t))_{t \in [0,\tau)}$ be a strongly continuous operator family which commutes with C.

(i) If $(W(t))_{t \in [0,\tau)}$ is an (a, Θ) -regularized C-resolvent family, then

 $(310) \ (a*W)(t)[W(s) - \Theta(s)C] = [W(t) - \Theta(t)C](a*W)(s) \ for \ 0 \leq t, s, \ t+s < \tau.$

(ii) Suppose that $(W(t))_{t \in [0,\tau)}$ is locally equicontinuous and (310) holds. Then $(W(t))_{t \in [0,\tau)}$ is an (a, Θ) -regularized C-resolvent family.

By Theorem 3.2.50(i), we have that the integral generator $\hat{\mathcal{A}}$ of a global (a, Θ) regularized *C*-resolvent family $(W(t))_{t\geq 0}$ is always its subgenerator (as already
seen, this statement is not true in local case).

The following theorem is a slight extension of [366, 367, Theorem 2.5].

THEOREM 3.2.51. Let $0 < \tau \leq \infty$ and $C \in L(E)$. Suppose that $a(t) \equiv 1$ or $a(t) \equiv t$, \mathcal{A} is an MLO and $(Z(t))_{t \in [0,\tau)} \subseteq L(E)$ is a strongly continuous operator family which commutes with C. Let $(Z(t))_{t \in [0,\tau)}$ be locally equicontinuous. Then the following holds:

- (i) If \mathcal{A} is a subgenerator of $(Z(t))_{t\in[0,\tau)}$, then $(Z(t))_{t\in[0,\tau)}$ is an (a,Θ) -regularized C-resolvent family.
- (ii) Suppose that (i) holds with $\mathcal{A} = A$ being single-valued and linear, as well as that C is injective. Then $(Z(t))_{t \in [0,\tau)}$ is non-degenerate.

PROOF. Let $x \in E$ be fixed. Then (ii) of Definition 3.2.41 yields that $\int_0^s a(s-r)Z(r)x \, dr \in D(\mathcal{A})$ and $Z(s)x - \Theta(s)Cx \in \mathcal{A} \int_0^s a(s-r)Z(r)x \, dr$. By (i) of Definition 3.2.41, we have

$$[Z(t) - \Theta(t)C] \int_0^s a(s-r)Z(r)x \, dr = \int_0^t a(t-r)Z(r)[Z(s)x - \Theta(s)Cx]dr$$
$$= (a*Z)(t)[Z(s)x - \Theta(s)Cx],$$

for $0 \leq t, s < \tau$. Now we can apply Theorem 3.2.50(ii) in order to see that (i) holds true. The proof of (ii) is simple and therefore omitted.

Now it is quite simple to construct an example of a global degenerate strongly continuous semigroup (i.e., *I*-regularized semigroup) which do not have any linear subgenerator:

EXAMPLE 3.2.52. [199] Suppose that \mathcal{A} is a non single-valued MLO which satisfies the Hille–Yosida condition [199, (H-Y), p. 28] on a Banach space E. By [199, Theorem 2.4], \mathcal{A} is the integral generator (a subgenerator) of a global strongly continuous semigroup $(T(t))_{t\geq 0}$ which vanishes on the closed subspace $\mathcal{A}0$ of E. An application of Theorem 3.2.51(ii) shows that $(T(t))_{t\geq 0}$ has no linear, single-valued subgenerator.

It should also be observed that a global degenerate strongly continuous semigroup can have infinitely many subgenerators: Let $P \in L(E)$, $P^2 = P$ and $T(t) := P, t \ge 0$. Then it can be simply verified that, for every linear subspace V of N(P), $\mathcal{A} = N(I - P) \times V$ is a subgenerator of $(T(t))_{t\ge 0}$. The arguments used in the proofs of [**366**, Lemma 2.8] and [**367**, Lemma 2.9] enable one to deduce the following lemma:

LEMMA 3.2.53. Let $0 < \tau \leq \infty$, $x \in E$, $0 \in \text{supp}(\Theta)$ and $(W(t))_{t \in [0,\tau)}$ be an (a, Θ) -regularized C-resolvent family. Then the existence of a number $\tau_0 \in (0, \tau)$ such that W(t)x = 0, $t \in [0, \tau_0)$ implies CW(t)x = 0, $t \in [0, \tau)$.

Keeping in mind Lemma 3.2.53, it is quite simple to prove the following extension of [**366**, Theorem 2.9] and [**367**, Theorem 2.10]:

THEOREM 3.2.54. Suppose that $0 < \tau \leq \infty$, $C \in L(E)$ is injective, $0 \in \text{supp}(\Theta)$ and $(W(t))_{t \in [0,\tau)}$ is an (a, Θ) -regularized C-resolvent family. Then $(W(t))_{t \in [0,\tau)}$ is non-degenerate iff the integral generator $\hat{\mathcal{A}}$ of $(W(t))_{t \in [0,\tau)}$ is its subgenerator.

Before proceeding further, we would like to mention that the examination from Example 3.2.52 shows that the existence of a subgenerator of a (local) C-regularized semigroup in the MLO sense does not imply its non-degeneracy, even supposing that C = I. The reader with a little experience will succeed in transferring [**366**, Theorem 2.13] and [**367**, Theorem 2.14] to locally equicontinuous K-convoluted C-semigroups and K-convoluted C-cosine functions in locally convex spaces.

The following proposition extends the assertions of [291, Proposition 2.1.3] and [366, 367, Proposition 2.3] (for locally equicontinuous operator families, this proposition follows almost immediately from Theorem 3.2.49).

PROPOSITION 3.2.55. Let $0 < \tau \leq \infty$, and let $(W(t))_{t \in [0,\tau)}$ be an (a, Θ) -regularized C-resolvent family. Suppose that $H \in L^1_{loc}([0,\tau))$ and $H *_0 K \neq 0$ in $L^1_{loc}([0,\tau))$. Set $W_H(t)x := \int_0^t H(t-s)W(s)x \, ds$, $x \in E$, $t \in [0,\tau)$. Then $(W_H(t))_{t \in [0,\tau)}$ is an $(a, H *_0 \Theta)$ -regularized C-resolvent family.

PROOF. We will include all details of proof for semigroups, in purely convoluted case; the proof in all other cases can be given by applying the same trick. It is clear that $(W_H(t) = S_{K,H}(t))_{t \in [0,\tau)} \subseteq L(E)$ is a strongly continuous operator family which commutes with C. Therefore, it suffices to show that, for every $x \in E$ and $t, s \in [0, \tau)$ with $t + s \in [0, \tau)$, the following holds:

$$S_{H,K}(t)S_{H,K}(s)x = \left[\int_0^{t+s} -\int_0^t -\int_0^s\right](H*K)(t+s-r)S_{H,K}(r)Cx\,dr,$$

i.e., that for each functional $x^* \in E^*$ we have:

(311)
$$\int_0^t \int_0^s H(t-r)H(s-\sigma) \left[\int_0^{r+\sigma} -\int_0^r -\int_0^\sigma \right] K(r+\sigma-v) \langle x^*, S_K(v)Cx \rangle \, dv \, d\sigma \, dr$$
$$= \left[\int_0^{t+s} -\int_0^t -\int_0^s \right] \int_0^r \int_0^{t+s-r} K(t+s-r-v) \\\times H(v)H(r-\sigma) \langle x^*, S_K(\sigma)Cx \rangle \, dv \, d\sigma \, dr.$$

By Lemma 3.2.49, the above equality holds for all non-zero complex polynomials $H(\cdot)$ so that the final conclusion follows by applying Stone–Weierstrass theorem and the dominated convergence theorem in (311).

- REMARK 3.2.56. (i) Suppose that \mathcal{A} is a closed subgenerator of $(W(t))_{t \in [0,\tau)}$. Then we can simply prove with the help of Theorem 1.2.3 that \mathcal{A} is a subgenerator of $(W_H(t))_{t \in [0,\tau)}$.
- (ii) Suppose that $\mathcal{A}(\mathcal{A}_H)$ is a closed subgenerator of $(W(t))_{t\in[0,\tau)}$ $((W_H(t))_{t\in[0,\tau)})$. Then $\mathcal{A} \subseteq \mathcal{A}_H$, with the equality if H(t) is a kernel on $[0,\tau)$.

Now we will extend the assertion of [291, Theorem 2.1.11] to degenerate operator families.

THEOREM 3.2.57. Suppose that \mathcal{A} is a closed MLO in E, $0 < \tau \leq \infty$, $K \in L^1_{loc}([0,\tau))$, $K \neq 0$ and $(C_K(t))_{t \in [0,\tau)}$ is a strongly continuous operator family which commutes with C. Set

$$S_{\Theta}(t) := \begin{pmatrix} \int_0^t C_K(s) ds & \int_0^t (t-s) C_K(s) ds \\ C_K(t) - \Theta(t) C & \int_0^t C_K(s) ds \end{pmatrix}, \quad 0 \leqslant t < \tau$$

and $\mathcal{C}(x \ y)^T := (Cx \ Cy)^T \ (x, y \in E)$. Then we have:

- (i) The following assertions are equivalent:
 - (a) $(C_K(t))_{t \in [0,\tau)}$ is a K-convoluted C-cosine function on E.
 - (b) $(S_{\Theta}(t))_{t \in [0,\tau)}$ is a Θ -convoluted C-semigroup $(S_{\Theta}(t))_{t \in [0,\tau)}$ on $E \times E$.

Suppose that the equivalence relation (a) \Leftrightarrow (b) in (i) holds. Then we have:

- (ii) \mathcal{A} is a subgenerator of $(C_K(t))_{t \in [0,\tau)}$ iff $\mathcal{B} := \begin{pmatrix} 0 & I \\ \mathcal{A} & 0 \end{pmatrix}$ is a subgenerator of $(S_{\Theta}(t))_{t \in [0,\tau)}$.
- (iii) Let $\hat{\mathcal{A}}$ and $\hat{\mathcal{B}}$ be the integral generators of $(C_K(t))_{t\in[0,\tau)}$ and $(S_{\Theta}(t))_{t\in[0,\tau)}$, respectively. Then the inclusion $\begin{pmatrix} 0 & I \\ \hat{\mathcal{A}} & 0 \end{pmatrix} \subseteq \hat{\mathcal{B}}$ holds true. Furthermore, if $(C_K(t))_{t\in[0,\tau)}$ is non-degenerate, then $\begin{pmatrix} 0 & I \\ \hat{\mathcal{A}} & 0 \end{pmatrix} = \hat{\mathcal{B}}$.

PROOF. Suppose that (a) holds. Then it is clear that $(S_{\Theta}(t))_{t \in [0,\tau)}$ is a strongly continuous operator family in $L(E \times E)$ which commutes with \mathcal{C} . Therefore, it suffices to show that the semigroup property holds for $(S_{\Theta}(t))_{t \in [0,\tau)}$, i.e., that the following holds for $0 \leq t, s, t + s < \tau$ and $x \in E$:

(312)
$$C_{\Theta}(t)C_{\Theta}(s)x + C_{\Theta}^{[1]}(t)[C_K(s) - \Theta(s)C]x$$

$$= \left[\int_0^{t+s} - \int_0^t - \int_0^s\right] \Theta(t+s-\sigma) C_{\Theta}(\sigma) Cx \, d\sigma,$$

(313)
$$C_{\Theta}(t)C_{\Theta}^{[1]}(s)x + C_{\Theta}^{[1]}(t)C_{\Theta}(s)x$$

$$= \left[\int_{0}^{t+s} -\int_{0}^{t} -\int_{0}^{s}\right]\Theta(t+s-\sigma)C_{\Theta}^{[1]}(\sigma)Cx\,d\sigma,$$
(314) $[C_{K}(t) - \Theta(t)C]C_{\Theta}(s)x + C_{\Theta}(t)[C_{K}(s) - \Theta(s)C]x$

$$= \left[\int_{0}^{t+s} -\int_{0}^{t} -\int_{0}^{s}\right]\Theta(t+s-\sigma)\{C_{K}(\sigma)Cx - \Theta(s)C^{2}x\}\,d\sigma$$

and

(315)
$$[C_K(t) - \Theta(t)C]C_{\Theta}^{[1]}(s)x + C_{\Theta}(t)C_{\Theta}(s)x$$
$$= \left[\int_0^{t+s} -\int_0^t -\int_0^s \right]\Theta(t+s-\sigma)C_{\Theta}(\sigma)Cx\,d\sigma.$$

The proofs of equations (312)–(315) and implication (b) \Rightarrow (a) below will be given only in the case that $s \leq t$; the case s > t can be considered similarly. First of all, we will prove (314). By [291, Lemma 2.1.12], we have that

$$\left[\int_0^{t+s} - \int_0^t - \int_0^s\right] \Theta(t+s-\sigma)\Theta(s) \, d\sigma = 0.$$

Therefore, we need to prove that

(316)
$$[C_K(t) - \Theta(t)C]C_{\Theta}(s)x + C_{\Theta}(t)[C_K(s) - \Theta(s)C]x$$
$$= \left[\int_0^{t+s} - \int_0^t - \int_0^s \right]\Theta(t+s-\sigma)C_K(\sigma)Cx\,d\sigma.$$

In order to do that, observe that the partial integration and Newton-Leibniz formula imply:

(317)
$$\left[\int_0^{t+s} - \int_0^t - \int_0^s \right] K(t+s-\sigma) C_{\Theta}(\sigma) Cx \, d\sigma$$
$$= \Theta(s) C_{\Theta}(t) Cx + \Theta(t) C_{\Theta}(s) Cx$$
$$+ \left[\int_0^{t+s} - \int_0^t - \int_0^s \right] \Theta(t+s-\sigma) C_{\Theta}(\sigma) Cx \, d\sigma,$$
$$f^t$$

(318)
$$\int_{t-s}^{s} \{\Theta(r-t+s)C_K(r)Cx + K(r-t+s)C_\Theta(r)Cx\}dr = \Theta(s)C_\Theta(t)Cx$$

and

(319)
$$\int_0^s \{\Theta(r+t-s)C_K(r)Cx + K(r+t-s)C_\Theta(r)Cx\}dr = \Theta(t)C_\Theta(s)Cx.$$

Inserting (317)–(319) in (308) and dividing after that both sides of obtained equation with two, we immediately get (316). Consider now the equation (312). The

both sides equal zero for t = 0 and it suffices therefore to show that their derivatives in variable t are equal, i.e., that:

$$C_K(t)C_{\Theta}(s)x + C_{\Theta}(t)[C_K(s) - \Theta(s)C]x$$

= $\frac{d}{dt} \left[\int_0^{t+s} - \int_0^t - \int_0^s \right] \Theta(t+s-\sigma)C_{\Theta}(\sigma)Cx \, d\sigma.$

This follows from (314) and the usual limit procedure. The equation (315) is a consequence of (312) and Theorem 3.2.50(i), so that it remains to be proved (312). Here we can clearly deal with the function $\Theta(t)$ replaced in all places with the function K(t): then (312) follows immediately from (314) by applying the partial intgration. Hence, (b) holds. The implication (b) \Rightarrow (a) can be proved as follows. By Lemma 3.2.49, we may assume without loss of generality that K(t) is locally absolutely continuous on $[0, \tau)$. Since (314) holds, [**291**, Lemma 2.1.12] implies that the equation [**291**, (26)] holds true. Due to the proof of [**291**, Theorem 2.1.13] (cf. [**291**, (27)-(35)]) shows that the composition property holds for $(C_K(t))_{t \in [0, \tau)}$. This completes the proof of (i). The proofs of (ii) and (iii) follow from simple calculations and therefore omitted.

- REMARK 3.2.58. (i) Let \mathcal{A} be an MLO in E, and let \mathcal{B} be defined as in (i). Then \mathcal{A} is closed iff \mathcal{B} is closed. Furthermore, $\begin{pmatrix} 0 & I \\ C^{-1}\mathcal{A}C & 0 \end{pmatrix} \subseteq C^{-1}\mathcal{B}C$, with the equality in the case that the operator C is injective.
- (ii) Theorem 3.2.57 can be simply reformulated for *C*-regularized cosine functions on *E* and induced once integrated *C*-semigroups on the product space $E \times E$. In non-degenerate case, a similar assertion is also known for (a * a, k)-regularized *C*-resolvent families on *E* and induced (a, a * k)regularized *C*-resolvent families on the product space $E \times E$ (cf. [291, Theorem 2.1.27(xiv)]). We leave to the interested reader as an interesting problem to find an appropriate analogue of the last mentioned result for degenerate regularized resolvent families.

The extension type theorems for non-degenerate integrated semigroups has been considered for the first time by W. Arendt, O. El-Mennaoui and V. Keyantuo [28], who have proved that a closed linear operator A generates a local (2n)times integrated semigroup on the interval $[0, 2\tau)$, provided that A generates a local n-times integrated semigroup on the interval $[0, \tau)$ ($n \in \mathbb{N}$, $0 < \tau < \infty$). Immediately after that, I. Ciorănescu and G. Lumer [110] have extended their result to the class of local K-convoluted C-semigroups (see e.g. [291, Theorem 2.1.9] for a precise formulation). On the other hand, S. W. Wang and M. C. Gao [532] have investigated automatic extensions of non-degenerate local C-regularized semigroups and non-degenerate local C-regularized cosine functions. By all means, the method established in [532] cannot be used for proving extension type theorems for degenerate C-regularized semigroups and degenerate C-regularized cosine functions which do not have subgenerators. The situation is much more simpler if degenerate operator families under our consideration have subgenerators. Keeping in mind Theorem 1.2.3 and elementary properties of multivalued linear operators, the following two theorems can be proved in almost the same way as in the single-valued linear case (cf. V. Keyantuo, P. J. Miana, L. Sánchez-Lajusticia [278, Theorem 4.4], P. J. Miana, V. Poblete [429, Theorem 3.3], and [291, Theorem 2.1.9, Corollary 2.1.10, Theorem 2.1.14, Corollary 2.1.15] for some special cases of two last mentioned results):

THEOREM 3.2.59. Suppose that \mathcal{A} is a closed MLO. Let $n \in \mathbb{N}$, $0 < \tau < \infty$, $0 < \tau_0 < \tau$, $K \in L^1_{loc}([0, (n+1)\tau))$, $K \neq 0$, and let $(S_K(t))_{t \in [0,\tau)}$ be a local $K_{|[0,\tau)}$ -convoluted C-semigroup with a subgenerator \mathcal{A} . Define recursively the family of operators $(S_{K,n+1}(t))_{t \in [0,(n+1)\tau_0]}$ by $S_{K,n+1}(t)x := \int_0^t K(t-s)S_{K,n}(s)Cx \, ds, \, x \in E$, for $t \in [0, n\tau_0]$ and

$$S_{K,n+1}(t)x := S_{K,n}(n\tau_0)S_K(t-n\tau_0)x + \int_0^{n\tau_0} K(t-s)S_{K,n}(s)Cx\,ds + \int_0^{t-n\tau_0} K^{*,n}(t-s)S_K(s)Cx\,ds$$

for $x \in E$ and $t \in (n\tau_0, (n+1)\tau_0]$. Then $(S_{K,n+1}(t))_{t \in [0,(n+1)\tau_0)}$ is a local $(K^{*,n+1})_{|[0,(n+1)\tau_0]}$ -convoluted C^{n+1} -semigroup with a subgenerator \mathcal{A} .

THEOREM 3.2.60. Suppose that \mathcal{A} is a closed MLO. Let $n \in \mathbb{N}$, $0 < \tau < \infty$, $0 < \tau_0 < \tau$, $K \in L^1_{loc}([0, (n+1)\tau))$, $K \neq 0$, and let $(C_K(t))_{t \in [0,\tau)}$ be a local $K_{|[0,\tau)}$ -convoluted C-cosine function with a subgenerator \mathcal{A} . Define recursively the family of operators $(C_{K,n+1}(t))_{t \in [0,(n+1)\tau_0]}$ by $C_{K,n+1}(t)x := \int_0^t K(t-s)C_{K,n}(s)Cx \, ds$, $x \in E$, for $t \in [0, n\tau_0]$ and

$$C_{K,n+1}(t)x := 2C_{K,n}(n\tau_0)C_K(t-n\tau_0)x + \int_0^{n\tau_0} K(t-s)C_{K,n}(s)Cx\,ds + \int_0^{t-n\tau_0} K^{*,n}(t-s)C_K(s)Cx\,ds - \int_{2n\tau_0-t}^{n\tau_0} K(t+s-2n\tau_0)C_{K,n}(s)Cx\,ds - \int_0^{t-n\tau_0} K(s-t+2n\tau_0)C_K(s)x\,ds$$

for $x \in E$ and $t \in (n\tau_0, (n+1)\tau_0]$. Then $(C_{K,n+1}(t))_{t \in [0,(n+1)\tau_0)}$ is a local $(K^{*,n+1})_{|[0,(n+1)\tau_0)}$ -convoluted C^{n+1} -cosine function with a subgenerator \mathcal{A} .

REMARK 3.2.61. Consider the situation of Theorem 3.2.59 (Theorem 3.2.60). Let $\hat{\mathcal{A}}$ be the integral generator of $(S_K(t))_{t\in[0,\tau)}$ $((C_K(t))_{t\in[0,\tau)})$. Then it is a very undesirable question to precisely profile the integral generator $\hat{\mathcal{A}}_{n+1}$ of $(S_{K,n+1}(t))_{t\in[0,(n+1)\tau_0)}$ $((C_{K,n+1}(t))_{t\in[0,(n+1)\tau_0)})$ in general case. We will prove that the integral generator of $(S_{K,n+1}(t))_{t\in[0,(n+1)\tau_0)}$ $((C_{K,n+1}(t))_{t\in[0,(n+1)\tau_0)})$ is $\hat{\mathcal{A}}$, provided that the operator C is injective (observe that we do not require the condition $0 \in \supp(K)$ here). The analysis is quite similar in both cases and we will consider only K-convoluted C-semigroups. Since \mathcal{A} is a subgenerator of $(S_{K,n+1}(t))_{t\in[0,(n+1)\tau_0)}$, and $\hat{\mathcal{A}}$ is the integral generator of $(S_K(t))_{t\in[0,\tau)}$, the foregoing arguments yield that $\hat{\mathcal{A}}_{n+1} = C^{-(n+1)}\mathcal{A}C^{n+1}$ and $\hat{\mathcal{A}} = C^{-1}\mathcal{A}C = C^{-1}\hat{\mathcal{A}}C$. Inductively, $C^{-l}\hat{\mathcal{A}}C^{l} = \hat{\mathcal{A}}, \ l \in \mathbb{N}$ so that

$$\hat{\mathcal{A}}_{n+1} = C^{-(n+1)} \mathcal{A} C^{n+1} = C^{-n} [C^{-1} \mathcal{A} C] C^n = C^{-n} \hat{\mathcal{A}} C^n = \hat{\mathcal{A}}.$$

The statements clarifed in Theorem 3.2.59–Theorem 3.2.60 and Remark 3.2.61 can be simply reformulated for the classes of local *C*-regularized semigroups and local *C*-regularized cosine functions. It is also worth noting that the assertions of [**291**, Proposition 2.3.3, Proposition 2.3.4] [**291**, Proposition 2.3.8(i)–(ii)] can be reword for abstract degenerate inclusions of first order (second order) by replacing the sequence $x_0 = x$, $Ax = x_1, \ldots, A^k x = x_k$ in their formulations with an arbitrary sequence $(x_j)_{0 \leq j \leq k}$ satisfying $x_j \in \mathcal{A}x_{j-1}$ ($1 \leq j \leq k$). Involving into consideration the conditions (d)–(e) from Theorem 3.2.46, one can simply prove the analogues of [**291**, Proposition 2.4.2. Corollary 2.4.3, Theorem 2.4.5, Theorem 2.4.8, Corollary 2.4.9] for degenerate analytic *K*-convoluted *C*-semigroups in locally convex spaces.

We close this subsection by rethinking our structural results from [291, Section 2.7] for degenerate K-convoluted C-semigroups subgenerated by multivalued linear operators. The subsequent theorem is very similar to [291, Theorem 2.7.1].

THEOREM 3.2.62. (i) Suppose that E is Banach space, M > 0, $\beta \ge 0$, $|K(t)| \le Me^{\beta t}$, $t \ge 0$, $(S_K(t))_{t \in [0,\tau)}$ is a (local) K-convoluted semigroup with a closed subgenerator \mathcal{A} and, for every $\varepsilon > 0$, there exist $\varepsilon_0 \in (0, \tau \varepsilon)$ and $T_{\varepsilon} > 0$ such that $1/|\tilde{K}(\lambda)| \le T_{\varepsilon}e^{\varepsilon_0|\lambda|}$, $\operatorname{Re} \lambda > \beta$, $\tilde{K}(\lambda) \ne 0$. Then, for every $\varepsilon > 0$, there exist $\bar{C}_{\varepsilon} > 0$ and $\bar{K}_{\varepsilon} > 0$ such that, for every λ which belongs to the following set

$$\Omega^1_{\varepsilon} := \{ \lambda \in \mathbb{C} : \tilde{K}(\lambda) \neq 0, \quad \operatorname{Re} \lambda > \beta, \, \operatorname{Re} \lambda \geqslant \varepsilon |\lambda| + \bar{C}_{\varepsilon} \},\$$

there exists an operator $F(\lambda) \in L(E)$ so that $F(\lambda)\mathcal{A} \subseteq \mathcal{A}F(\lambda), \ \lambda \in \Omega^1_{\varepsilon}$, $F(\lambda)x \in (\lambda - \mathcal{A})^{-1}x, \ \lambda \in \Omega^1_{\varepsilon}, \ x \in E, \ F(\lambda)x - x = F(\lambda)y$, whenever $\lambda \in \Omega^1_{\varepsilon}$ and $(x, y) \in \mathcal{A}$,

$$\|F(\lambda)\| \leqslant \bar{K}_{\varepsilon} e^{\varepsilon_0 |\lambda|}, \quad \lambda \in \Omega^1_{\varepsilon}, \ \tilde{K}(\lambda) \neq 0,$$

and that the mapping $\lambda \mapsto F(\lambda) \in L(E)$, $\lambda \in \Omega^1_{\varepsilon}$ is analytic.

(ii) Suppose that E is Banach space, $\alpha > 0$, M > 0, $\beta \ge 0$, $\Phi : \mathbb{C} \to [0, \infty)$, $|K(t)| \le Me^{\beta t}$, $t \ge 0$, $(S_K(t))_{t \in [0, \tau)}$ is a local K-convoluted semigroup with a closed subgenerator \mathcal{A} and $1/|\tilde{K}(\lambda)| \le e^{\Phi(\alpha\lambda)}$, $\operatorname{Re} \lambda > \beta$, $\tilde{K}(\lambda) \ne 0$. Then, for every $t \in (0, \tau)$, there exist $\beta(t) > 0$ and M(t) > 0 such that, for every λ which belongs to the following set

$$\Lambda_{t,\alpha,\beta(t)} := \Big\{ \lambda \in \mathbb{C} : \tilde{K}(\lambda) \neq 0, \text{ Re } \lambda \geqslant \frac{\Phi(\alpha\lambda)}{t} + \beta(t) \Big\},\$$

there exists an operator $F(\lambda) \in L(E)$ so that $F(\lambda)\mathcal{A} \subseteq \mathcal{A}F(\lambda), \lambda \in \Lambda_{t,\alpha,\beta(t)}, F(\lambda)x \in (\lambda - \mathcal{A})^{-1}x, \lambda \in \Lambda_{t,\alpha,\beta(t)}, x \in E, F(\lambda)x - x = F(\lambda)y,$ whenever $\lambda \in \Lambda_{t,\alpha,\beta(t)}$ and $(x, y) \in \mathcal{A}$,

$$\|F(\lambda)\| \leqslant M(t)e^{\Phi(\alpha\lambda)}, \quad \lambda \in \Lambda_{t,\alpha,\beta(t)}, \ \tilde{K}(\lambda) \neq 0$$

and that the mapping $\lambda \mapsto F(\lambda) \in L(E)$, $\lambda \in \Lambda_{t,\alpha,\beta(t)}$ is analytic. Furthermore, the existence of a sequence (t_n) in $[0,\tau)$ satisfying $\lim_{n\to\infty} t_n =$

 τ and $\sup_{n\in\mathbb{N}} \ln \|S_K(t_n)\| < \infty$ implies that there exist $\beta' > 0$ and M' > 0such that the above holds with the region $\Lambda_{t,\alpha,\beta(t)}$ replaced by $\Lambda_{\tau,\alpha,\beta'}$ and the number M(t) replaced by M'.

THEOREM 3.2.63. Suppose that K(t) satisfies (P1), \mathcal{A} is a closed MLO, $r_0 \ge \max(0, \operatorname{abs}(K)), \Phi: [r_0, \infty) \to [0, \infty)$ is a continuously differentiable, strictly increasing mapping, $\lim_{t\to\infty} \Phi(t) = +\infty, \Phi'(\cdot)$ is bounded on $[r_0, \infty)$ and there exist $\alpha > 0, \gamma > 0$ and $\beta > r_0$ such that, for every λ which belongs to the following set

$$\Psi_{\alpha,\beta,\gamma} := \left\{ \lambda \in \mathbb{C} : \operatorname{Re} \lambda \geqslant \frac{\Phi(\alpha |\operatorname{Im} \lambda|)}{\gamma} + \beta \right\}$$

there exists an operator $F(\lambda) \in L(E)$ so that $F(\lambda)A \subseteq AF(\lambda)$, $\lambda \in \Psi_{\alpha,\beta,\gamma}$, $F(\lambda)x \in (\lambda - A)^{-1}Cx$, $\lambda \in \Psi_{\alpha,\beta,\gamma}$, $x \in E$, $F(\lambda)C = CF(\lambda)$, $\lambda \in \Psi_{\alpha,\beta,\gamma}$, $F(\lambda)x - Cx = F(\lambda)y$, whenever $\lambda \in \Psi_{\alpha,\beta,\gamma}$ and $(x,y) \in A$, and that the mapping $\lambda \mapsto F(\lambda)x$ is analytic on $\Omega_{\alpha,\beta,\gamma}$ and continuous on $\Gamma_{\alpha,\beta,\gamma}$, where $\Gamma_{\alpha,\beta,\gamma}$ denotes the upwards oriented boundary of $\Psi_{\alpha,\beta,\gamma}$ and $\Omega_{\alpha,\beta,\gamma}$ the open region which lies to the right of $\Gamma_{\alpha,\beta,\gamma}$. Let the following conditions hold.

- (i) There exists $\sigma > 0$ such that the operator family $\{e^{\Phi(\sigma|\lambda|)}F(\lambda) : \lambda \in \overline{\Omega_{\alpha,\beta,\gamma}}\} \subseteq L(E)$ is equicontinuous.
- (ii) There exists a function $m: [0, \infty) \to (0, \infty)$ such that m(s) = 1, $s \in [0, 1]$ and that, for every s > 1, there exists an $r_s > r_0$ so that $\frac{\Phi(t)}{\Phi(st)} \ge m(s)$, $t \ge r_s$.
- (iii) $\lim_{t\to\infty} te^{-\Phi(\sigma t)} = 0.$
- (iv) $(\exists a \ge 0)(\exists r'_a > r_0)(\forall t > r'_a)\frac{\ln t}{\Phi(t)} \ge a.$

Then \mathcal{A} is a subgenerator of a local K-convoluted C-semigroup on $[0, a + m(\frac{\alpha}{\sigma\gamma}))$.

We can similarly formulate analogues of [291, Theorem 2.7.2(iii)–(iv)] for local integrated (C-)semigroups and [291, Theorem 2.7.3] for local K-convoluted C-cosine functions.

The proof of [296, Theorem 3.15] (cf. also Theorem 3.2.21 and Example 3.2.23, where we have studied the entire solutions of backward heat Poisson equation) essentially shows that the existence of numbers $r \ge 0$ and $\theta \in (0, \pi/2)$ and an injective operator $C \in L(E)$ such that $-\mathcal{A}$ is a closed subgenerator of an exponentially equicontinuous, analytic *r*-times integrated *C*-semigroup $(S_r(t))_{t\ge 0}$ of angle θ implies that there exists an operator $C_1 \in L(E)$ such that \mathcal{A} is a subgenerator of an entire C_1 -regularized semigroup in *E*. Using the usual matrix reduction, we can apply this result in the analysis of problem

$$\frac{d}{dt}(Cu') - Bu' + Au(t) = F(t), \quad 0 < t \le T,$$

$$u(0) = u_0, \quad Cu'(0) = Cu_1,$$

provided that the conditions from [199, Section 6.1] hold. Multivalued matricial operators on product spaces can also serve one for construction of exponentially bounded degenerate integrated semigroups; see e.g. [292, Example 3.2.24].

Now we will provide two more illustrative examples of application of our abstract results. EXAMPLE 3.2.64. Let $E := l_2(\mathbb{C})$ be the Hilbert space consisted of all squaresummable complex sequences equipped with the norm $||x|| := ||\langle x_1, x_2, \ldots, x_n, \ldots \rangle||$ $:= (\sum_{n=1}^{\infty} |x_n|^2)^{1/2}, x \in E$. Let $(b_n)_{n \in \mathbb{N}}$ be any real sequence with the property that $0 \leq b_n \leq 1, n \in \mathbb{N}$ and let $a_n = n + (n^{-2}e^{-2n} - n^2)^{1/2}, n \in \mathbb{N}$. Define, for every $\langle x_1, x_2, \ldots, x_n, \ldots \rangle \in E$ and $t \in [0, 1)$,

$$T(t)\langle x_1, x_2, \dots, x_n, \dots \rangle := \langle b_1 e^{ta_1} x_1, b_2 e^{ta_2} x_2, \dots, b_n e^{ta_n} x_n, \dots \rangle,$$

as well as $C := T(0), I := \{j \in \mathbb{N} : b_j \neq 0\}$ and $D \in L(E)$ by $D\langle x_1, x_2, \ldots, x_n, \ldots \rangle$ $:= \langle y_1, y_2, \ldots, y_n, \ldots \rangle$, where $y_j = a_j x_j$ for $j \in I$ and $y_j = 0$ for $j \notin I$. Then $(T(t))_{t \in [0,1)}$ is a local *C*-regularized semigroup with the integral generator

$$\hat{\mathcal{A}} := \{ (\langle x_1, x_2, \dots, x_n, \dots \rangle, \langle y_1, y_2, \dots, y_n, \dots \rangle) : y_j = a_j x_j \text{ for all } j \in I \}$$

and any linear subspace \mathcal{A} of $E \times E$ satisfying $E \times R(D) \subseteq \mathcal{A} \subseteq \hat{\mathcal{A}}$ is a subgenerator of $(T(t))_{t \in [0,1)}$.

EXAMPLE 3.2.65. Put $E := \{f \in C^{\infty}([0,\infty)) : \lim_{x \to +\infty} f^{(k)}(x) = 0 \text{ for all } k \in \mathbb{N}_0\}$ and $||f||_k := \sum_{j=0}^k \sup_{x \ge 0} |f^{(j)}(x)|, f \in X, k \in \mathbb{N}_0$. This calibration induces a Fréchet topology on E. Let $J = [a,b] \subseteq (0,\infty)$, and let $m_b \in C^{\infty}([0,\infty))$ satisfy $0 \le m_b(x) \le 1, x \ge 0, m_b(x) = 1, x \notin J$ and $m_b(x) = 0, x \in [a + \varepsilon, b - \varepsilon]$ for some $\varepsilon > 0$. Consider the multiplication operators $A : D(A) \to E$ and $B \in L(E)$, where $D(A) = \{f(x) \in E : (-1 - x + ie^x)f(x) \in E\}$, $Af(x) := (-1 - x + ie^x)f(x)$ and $Bf(x) := m_b(x)f(x)$ ($x \ge 0, f \in E$). Set $\mathcal{A} := B^{-1}A$. A similar line of reasoning as in Example 3.2.11(ii) shows that, for every s > 1, d > 0 and $\omega > 0$, the operator family $\{e^{-d|\lambda|^{1/s}}(\lambda - \mathcal{A})^{-1} : \operatorname{Re} \lambda > \omega\} \subseteq L(E)$ is equicontinuous. Now we can apply Theorem 3.2.46 in order to see that \mathcal{A} is the integral generator of a local n-times integrated semigroup $(S_n(t))_{t\in[0,\tau)}$ on E. In actual fact, $(S_n(t))_{t\in[0,\tau)}$ must be given by the following formula

$$(S_n(t)f)(x) := \left[\frac{e^{t(-1-x+ie^x)}}{(-1-x+ie^x)^n} - \frac{t^{n-1}}{(n-1)!} \frac{1}{-1-x+ie^x} - \dots - \frac{t}{(-1-x+ie^x)^{n-1}} - \frac{1}{(-1-x+ie^x)^n}\right]f(x),$$

for any $f \in E$, $x \ge 0$ and $t \in [0, \tau)$. This immediately implies that for each $t \in (0, 1)$ there exists $f_t \in E$ such that $||S_n(t)f_t||_{n+1} = +\infty$, which is a contradiction.

We close this subsection by enquiring into the basic properties of stationary dense multivalued linear operators, subgenerators of integrated semigroups and subgenerators of integrated cosine functions. The following definition has been introduced by P. C. Kunstmann [363] in single-valued case.

DEFINITION 3.2.66. A multivalued linear operator \mathcal{A} is said to be stationary dense iff

$$n(\mathcal{A}) := \inf\{k \in \mathbb{N}_0 : D(\mathcal{A}^m) \subseteq D(\mathcal{A}^{m+1}) \text{ for all } m \ge k\} < \infty.$$

The number $n(\mathcal{A})$ is called the stationarity of \mathcal{A} .

- REMARK 3.2.67. (i) Since the translations of a multivalued linear operator \mathcal{A} have the same domains of non-negative integer powers as \mathcal{A} as, we have that for each complex number λ the stationarity of $\lambda - \mathcal{A}$ is the same as that of \mathcal{A} .
- (ii) Let us recall that a densely defined single-valued linear operator A need not be stationary dense (see e.g. [363, Example 1.3]) and that the validity of additional condition ρ(A) ≠ Ø implies that A must be stationary dense with n(A) = 0. Let us prove that the last statement continues to hold in multivalued linear setting. More precisely, let A be a densely defined MLO and ρ(A) ≠ Ø. Then A is stationary dense and n(A) = 0. In order to see this, it suffices to assume that 0 ∈ ρ(A); cf.
 (i). Observe that, for every x ∈ E, there exists a sequence (x_k)_{k∈ℕ} in D(A) such that lim_{k→+∞} x_k = x. This implies that lim_{k→+∞} A⁻ⁿx_k = lim_{k→+∞} (A⁻¹)ⁿx_k = (A⁻¹)ⁿx = A⁻ⁿx in E, so that D(Aⁿ⁺¹) is dense in R(A⁻ⁿ) = D(Aⁿ) for all n ∈ ℕ₀.
- (iii) Let \mathcal{A} be an MLO with $\rho(\mathcal{A}) \neq \emptyset$. Then part (i) and an elementary argumentation show that $n(\mathcal{A}) = \inf\{k \in \mathbb{N}_0 : D(\mathcal{A}^k) \subseteq \overline{D(\mathcal{A}^{k+1})}\}$.

Suppose that $0 < \tau \leq \infty$, $n \in \mathbb{N}_0$ and \mathcal{A} is a closed MLO satisfying that the abstract Cauchy inclusion

$$\left(\text{ACI}\right)_1: \begin{cases} u'(t) \in \mathcal{A}u(t), & t \in [0, \tau), \\ u(0) = x \end{cases}$$

has a mild solution (that is any continuous function $u(\cdot; x) \in C([0, \tau) : E)$ such that u(0; x) = x and $u(t; x) - x \in \mathcal{A} \int_0^t u(s; x) ds$, $t \in [0, \tau)$) for all initial values $x \in D(\mathcal{A}^n)$. Let an element $x \in D(\mathcal{A}^n)$ be fixed, and let $u(\cdot; x)$ be a mild solution of (ACI)₁ (observe that we do not require the uniqueness of mild solutions here). Using Theorem 1.2.3 and partial integration, we can inductively prove that

$$(-1)^k \int_0^\tau \varphi^{(k)}(t) u(t;x) dt \in \mathcal{A}^k \int_0^\tau \varphi(t) u(t;x) dt, \quad \varphi \in \mathcal{D}_{[0,\tau)}, \ k \in \mathbb{N}$$

This, in particular, implies $u(t;x) \in D_{\infty}(\mathcal{A}), t \in [0,\tau)$, so that $D(\mathcal{A}^n) \subseteq D_{\infty}(\mathcal{A})$ and $n(\mathcal{A}) \leq n$. This is an extension of [363, Lemma 1.7] to multivalued linear case.

The following proposition is well-known in non-degenerate case [291, 363].

PROPOSITION 3.2.68. Let $n \in \mathbb{N}_0$ and $0 < \tau \leq \infty$.

- (i) Suppose that \mathcal{A} is a closed subgenerator of a (local) n-times integrated existence family $(S_n(t))_{t\in[0,\tau)}$. Then $n(\mathcal{A}) \leq n+1$, and the validity of condition $S_n(0) = \delta_{0,n}I$ implies $n(\mathcal{A}) \leq n$.
- (ii) Suppose that \mathcal{A} is a closed subgenerator of a (local) n-times integrated cosine existence family $(C_n(t))_{t \in [0,\tau)}$. Then $n(\mathcal{A}) \leq \lfloor \frac{n+2}{2} \rfloor$, and the validity of condition $C_n(0) = \delta_{0,n}I$ implies $n(\mathcal{A}) \leq \lfloor \frac{n+1}{2} \rfloor$.

PROOF. (i): Suppose first that $S_n(0) = \delta_{0,n}I$. Then it suffices to show that the abstract Cauchy inclusion (ACI)₁ has a unique mild solution for all initial values $x \in D(\mathcal{A}^n)$. This follows from a simple computation showing that, for every $x \in D(\mathcal{A}^n)$, the function

$$u(t;x) = S_n(t)y_n + \frac{t^{n-1}}{(n-1)!}y_{n-1} + \dots + ty_1 + x, \quad t \in [0,\tau)$$

is a mild solution of $(ACI)_1$, where the sequence $(y_j)_{1 \leq j \leq n}$ is chosen so that $(x, y_1) \in \mathcal{A}$, $(y_1, y_2) \in \mathcal{A}, \ldots, (y_{n-1}, y_n) \in \mathcal{A}$. The rest of (i) is a consequence of the fact that $(S_n^{[1]}(t))_{t \in [0,\tau)}$ is an (n+1)-times integrated semigroup with a subgenerator \mathcal{A} and $S_n^{[1]}(0) = 0$. (ii): The proof of (ii) can be given by using the fact that

$$S_{n+1}(t) \equiv \begin{pmatrix} \int_0^t C_n(s)ds & \int_0^t (t-s)C_n(s)ds \\ C_n(t) - g_{n+1}(t) & \int_0^t C_n(s)ds \end{pmatrix}, \quad 0 \leqslant t < \tau$$

is an (n+1)-times integrated cosine existence family with a subgenerator $\mathcal{B} = \begin{pmatrix} 0 & I \\ \mathcal{A} & 0 \end{pmatrix}$ (cf. Theorem 3.2.57 and the proof of [**291**, Theorem 2.1.11(i)]) and the fact that

$$n(\mathcal{A}) = 2n(\mathcal{B}),$$

which can be verified by reasoning similarly as in the proof of [291, Lemma 2.1.22].

REMARK 3.2.69. The estimates $n(\mathcal{A}) \leq n+1$ and $n(\mathcal{A}) \leq \lfloor \frac{n+2}{2} \rfloor$ obtained above cannot be refined in the degenerate case. We will explain this in the case that n = 0. Suppose $P \in L(E)$ and $P^2 = P$. Then $(T(t) \equiv P)_{t \geq 0}$ $((C(t) \equiv P)_{t \geq 0})$ is a strongly continuous semigroup (strongly continuous cosine operator function) with a closed subgenerator $\mathcal{A} = N(I - P) \times \{0\}$, which satisfies $n(\mathcal{A}) = 1$ provided that N(I - P) is not densely defined in E (consider, for example, the matricial operator $\begin{pmatrix} 0 & 1 \\ 0 & 1 \end{pmatrix}$).

It is also worth noting that the following extension of [363, Lemma 1.5] holds in our framework.

PROPOSITION 3.2.70. Suppose that \mathcal{A} is a closed MLO, $\alpha \ge -1$ and $(\lambda_n)_{n\in\mathbb{N}}$ is a sequence of complex numbers satisfying that $\lim_{n\to+\infty} |\lambda_n| = +\infty$ and the family $\{(1+|\lambda_n|)^{-\alpha}R(\lambda_n:\mathcal{A}) ; n\in\mathbb{N}\}\subseteq L(E)$ is equicontinuous. Then \mathcal{A} is stationary dense and $n(\mathcal{A}) \le \lfloor \alpha \rfloor + 2$.

PROOF. Without loss of generality, we may assume that $|\lambda_n| \ge 1$, $n \in \mathbb{N}$ and $(|\lambda_n|)_{n \in \mathbb{N}}$ is strictly increasing. Let $k \ge \lfloor \alpha \rfloor + 2$. We will have to prove that $D(\mathcal{A}^k) \subseteq \overline{D(\mathcal{A}^{k+1})}$. So, let $x \in D(\mathcal{A}^k)$. Then there exists a sequence $(y_j)_{1 \le j \le k}$ such that $(x, y_1) \in \mathcal{A}, (y_1, y_2) \in \mathcal{A}, \ldots, (y_{k-1}, y_k) \in \mathcal{A}$. Then $\lambda_n R(\lambda_n : \mathcal{A}) x \in D(\mathcal{A}^{k+1})$, $n \in \mathbb{N}$ and it can be easily seen with the help of Theorem 1.2.4(i) and Theorem 1.2.8(ii) that, for every $n \in \mathbb{N}$, we have

$$\lambda_n R(\lambda_n : \mathcal{A})x - x = R(\lambda_n : \mathcal{A})y_1 \to 0, \quad n \to +\infty.$$

Hence, $D(\mathcal{A}^k) \subseteq \overline{D(\mathcal{A}^{k+1})}$ as claimed.

3.3. Degenerate *C*-distribution semigroups and degenerate *C*-ultradistribution semigroups in locally convex spaces

In our recent joint research study with S. Pilipović and D. Velinov [354], we have introduced and systematically analyzed the classes of C-distribution semigroups and C-ultradistribution semigroups in locally convex spaces (cf. [65,99,109, 177, 282, 289, 291, 351, 364, 365, 391, 424, 488, 530 and references cited therein for the current state of theory). The main aim of this section is to present the most important things about the classes of degenerate C-distribution semigroups and degenerate C-ultradistribution semigroups in barreled sequentially complete locally convex spaces. We consider multivalued linear operators as infinitesimal generators of such semigroups and allow the regularizing operator C to be noninjective (cf. [58, 282, 406] and [423, 424] for the primary source of information on degenerate distribution semigroups in Banach spaces). In contrast to the analyses from [424, Section 2.2] and [58, Section 3], we do not use any decomposition of the state space E. Throughout this section, we assume that $C \in L(E)$ is not necessarily injective operator as well as that $(M_n)_{n \in \mathbb{N}_0}$ is a sequence of positive real numbers satisfying (M.1), (M.2) and (M.3)'. Any employment of the condition (M.3) will be explicitly emphasized. Since E is barreled, the uniform boundedness principle [419, p. 273] implies that each $\mathcal{G} \in \mathcal{D}'(L(E))$ ($\mathcal{G} \in \mathcal{D}'^*(L(E))$) is boundedly equicontinuous, i.e., that for every $p \in \mathbb{R}$ and for every bounded subset B of \mathcal{D} (\mathcal{D}^*) , there exist c > 0 and $q \in \mathfrak{B}$ such that $p(\mathcal{G}(\varphi)x) \leq cq(x), \varphi \in B, x \in E$.

DEFINITION 3.3.1. Let $\mathcal{G} \in \mathcal{D}'_0(L(E))$ ($\mathcal{G} \in \mathcal{D}'^*_0(L(E))$) satisfy $C\mathcal{G} = \mathcal{G}C$. Then it is said that \mathcal{G} is a pre-(C-DS) (pre-(C-UDS) of *-class) iff the following holds:

(C.S.1)
$$\mathcal{G}(\varphi *_0 \psi)C = \mathcal{G}(\varphi)\mathcal{G}(\psi), \quad \varphi, \ \psi \in \mathcal{D} \ (\varphi, \ \psi \in \mathcal{D}^*).$$

If, additionally,

(C.S.2)
$$\mathcal{N}(\mathcal{G}) := \bigcap_{\varphi \in \mathcal{D}_0} \mathcal{N}(\mathcal{G}(\varphi)) = \{0\} \quad \left(\mathcal{N}(\mathcal{G}) := \bigcap_{\varphi \in \mathcal{D}_0^*} \mathcal{N}(\mathcal{G}(\varphi)) = \{0\}\right),$$

then \mathcal{G} is called a *C*-distribution semigroup (*C*-ultradistribution semigroup of *class), (C-DS) ((C-UDS) of *-class) in short. A pre-(C-DS) \mathcal{G} is called dense iff

(C.S.3)
$$\mathcal{R}(\mathcal{G}) := \bigcup_{\varphi \in \mathcal{D}_0} R(\mathcal{G}(\varphi))$$
 is dense in $E\left(\mathcal{R}(\mathcal{G}) := \bigcup_{\varphi \in \mathcal{D}_0^*} R(\mathcal{G}(\varphi))$ is dense in $E\right).$

The notion of a dense pre-(C-UDS) \mathcal{G} of *-class (and the set $\mathcal{R}(\mathcal{G})$) is defined similarly.

If C = I, then we also write pre-(DS), pre-(UDS), (DS), (UDS), respectively, instead of pre-(C-DS), pre-(C-UDS), (C-DS), (C-UDS).

Suppose that \mathcal{G} is a pre-(C-DS) (pre-(C-UDS) of *-class). Then $\mathcal{G}(\varphi)\mathcal{G}(\psi) = \mathcal{G}(\psi)\mathcal{G}(\varphi)$ for all $\varphi, \psi \in \mathcal{D} (\varphi, \psi \in \mathcal{D}^*)$, and $\mathcal{N}(\mathcal{G})$ is a closed subspace of E.

The structural characterization of a pre-(C-DS) \mathcal{G} (pre-(C-UDS) \mathcal{G} of *-class) on its kernel space $\mathcal{N}(\mathcal{G})$ is described in the following theorem (cf. [291, Proposition 3.1.1] and the proofs of [364, Lemma 2.2], [291, Proposition 3.5.4]).

- THEOREM 3.3.2. (i) Let \mathcal{G} be a pre-(C-DS), and let the space $L(\mathcal{N}(\mathcal{G}))$ be admissible. Then, with $N = \mathcal{N}(\mathcal{G})$ and G_1 being the restriction of \mathcal{G} to N ($G_1 = \mathcal{G}_{|N}$), we have: There exists an integer $m \in \mathbb{N}$ for which there exist unique operators $T_0, T_1, \ldots, T_m \in L(\mathcal{N}(\mathcal{G}))$ commuting with C so that $G_1 = \sum_{j=0}^m \delta^{(j)} \otimes T_j, T_i C^i = (-1)^i T_0^{i+1}, 0 \leq i \leq m-1$ and $T_0 T_m = T_0^{m+2} = 0.$
- (ii) Let (M_p) satisfy (M.3), let G be a pre-(C-UDS) of *-class, and let the space N(G) be barreled. Then, with N = N(G) and G₁ being the restriction of G to N (G₁ = G_{|N}), we have: There exists a unique set of operators (T_j)_{j∈N₀} in L(N(G)) commuting with C so that G₁ = ∑_{j=0}[∞] δ^(j) ⊗ T_j, T_jC^j = (-1)^jT₀^{j+1}, j ∈ N and the set {M_jT_jL^j : j ∈ N₀} is bounded in L(N(G)), for some L > 0 in the Beurling case, resp. for every L > 0 in the Roumieu case.

Let $\mathcal{G} \in \mathcal{D}'_0(L(E))$ ($\mathcal{G} \in \mathcal{D}'_0(L(E))$), and let $T \in \mathcal{E}'_0(T \in \mathcal{E}'_0)$, i.e., T is a scalarvalued distribution (ultradistribution of *-class) with compact support contained in $[0, \infty)$. Define

$$G(T) := \{ (x, y) \in E \times E : \mathcal{G}(T * \varphi) x = \mathcal{G}(\varphi) y \text{ for all } \varphi \in \mathcal{D}_0 \ (\varphi \in \mathcal{D}_0^*) \}.$$

Then it can be easily seen that G(T) is a closed MLO; furthermore, if $\mathcal{G} \in \mathcal{D}'_0(L(E))$ $(\mathcal{G} \in \mathcal{D}_0^{\prime*}(L(E)))$ satisfy (C.S.2), then G(T) is a closed linear operator. Assuming that the regularizing operator C is injective, definition of G(T) can be equivalently introduced by replacing the set \mathcal{D}_0 (\mathcal{D}_0^*) with the set $\mathcal{D}_{[0,\varepsilon)}$ ($\mathcal{D}_{[0,\varepsilon)}^*$) for any $\varepsilon >$ 0. In general case, for every $\psi \in \mathcal{D}$ ($\psi \in \mathcal{D}^*$), we have $\psi_+ := \psi \mathbf{1}_{[0,\infty)} \in \mathcal{E}'_0$ (\mathcal{E}'_0^*) , where $\mathbf{1}_{[0,\infty)}$ stands for the characteristic function of $[0,\infty)$, so that the definition of $G(\psi_{+})$ is clear. We define the (infinitesimal) generator of a pre-(C-DS) \mathcal{G} by $\mathcal{A} := G(-\delta')$ (cf. [354] for more details about non-degenerate case, and [58, Definition 3.4] and [282] for some other approaches used in the degenerate case). Then $\mathcal{N}(\mathcal{G}) \times \mathcal{N}(\mathcal{G}) \subset \mathcal{A}$ and $\mathcal{N}(\mathcal{G}) = \mathcal{A}0$, which simply implies that \mathcal{A} is single-valued iff (C.S.2) holds. If this is the case, then we also have that the operator C must be injective: Suppose that Cx = 0 for some $x \in E$. By (C.S.1), we get that $\mathcal{G}(\varphi)\mathcal{G}(\psi)x = 0, \ \varphi, \ \psi \in \mathcal{D}$. In particular, $\mathcal{G}(\psi)x \in \mathcal{N}(\mathcal{G}) = \{0\}$ so that $\mathcal{G}(\psi)x = 0, \ \psi \in \mathcal{D}$. Hence, $x \in \mathcal{N}(\mathcal{G}) = \{0\}$ and therefore x = 0. Further on, if \mathcal{G} is a pre-(C-DS) (pre-(C-UDS) of *-class), $T \in \mathcal{E}'_0$ ($T \in \mathcal{E}'_0$) and $\varphi \in \mathcal{D}$ ($\varphi \in \mathcal{D}^*$), then $\mathcal{G}(\varphi)G(T) \subseteq G(T)\mathcal{G}(\varphi), \ CG(T) \subseteq G(T)C \ \text{and} \ \mathcal{R}(\mathcal{G}) \subseteq D(G(T)).$ If \mathcal{G} is a pre-(C-DS) (pre-(C-UDS) of *-class) and $\varphi, \psi \in \mathcal{D}(\varphi, \psi \in \mathcal{D}^*)$, then the assumption $\varphi(t) = \psi(t), t \ge 0$, implies $\mathcal{G}(\varphi) = \mathcal{G}(\psi)$. As in the Banach space case, we can prove the following (cf. [291, Proposition 3.1.3, Lemma 3.1.6]): Suppose that \mathcal{G} is a pre-(C-DS) (pre-(C-UDS) of *-class). Then $(Cx, \mathcal{G}(\psi)x) \in G(\psi_+), \psi \in \mathcal{D}, x \in E$ $(\psi \in \mathcal{D}^*, x \in E)$ and $\mathcal{A} \subseteq C^{-1}\mathcal{A}C$, while $C^{-1}\mathcal{A}C = \mathcal{A}$ provided that C is injective. Furthermore, the following holds:

PROPOSITION 3.3.3. Let \mathcal{G} be a pre-(C-DS) (pre-(C-UDS) of *-class), $S, T \in \mathcal{E}'_0$ ($S, T \in \mathcal{E}'^*_0$), $\varphi \in \mathcal{D}_0$ ($\varphi \in \mathcal{D}^*_0$), $\psi \in \mathcal{D}$ ($\psi \in \mathcal{D}^*$) and $x \in E$. Then we have:

- (i) $(\mathcal{G}(\varphi)x, \mathcal{G}(\overline{T*\cdots*T}*\varphi)x) \in G(T)^m, m \in \mathbb{N}.$
- (ii) $G(S)G(T) \subseteq G(S*T)$ with $D(G(S)G(T)) = D(G(S*T)) \cap D(G(T))$, and $G(S) + G(T) \subseteq G(S+T)$.
- (iii) $(\mathcal{G}(\psi)x, \mathcal{G}(-\psi')x \psi(0)Cx) \in G(-\delta').$

m

(iv) If \mathcal{G} is dense, then its generator is densely defined.

The assertions (ii)-(vi) of [291, Proposition 3.1.2] can be reformulated for pre-(C-DS)'s (pre-(C-UDS)'s of *-class) in locally convex spaces; here it is only worth noting that the reflexivity of state space E implies that the spaces E^* and $E^{**} = E$ are both barreled and sequentially complete:

PROPOSITION 3.3.4. Let \mathcal{G} be a pre-(C-DS) (pre-(C-UDS) of *-class). Then the following holds:

- (i) $C(\overline{\langle \mathcal{R}(\mathcal{G}) \rangle}) \subseteq \overline{\mathcal{R}(\mathcal{G})}$, where $\langle \mathcal{R}(\mathcal{G}) \rangle$ denotes the linear span of $\mathcal{R}(\mathcal{G})$.
- (ii) Assume \mathcal{G} is not dense and $\overline{C\mathcal{R}(\mathcal{G})} = \overline{\mathcal{R}(\mathcal{G})}$. Put $R := \overline{\mathcal{R}(\mathcal{G})}$ and $H := \mathcal{G}_{|R}$. Then H is a dense pre-(C_1 -DS) (pre-(C_1 -UDS) of *-class) on R with $C_1 = C_{|R}$.
- (iii) The dual $\mathcal{G}(\cdot)^*$ is a pre-(C*-DS) (pre-(C*-UDS) of *-class) on E^* and $\mathcal{N}(\mathcal{G}^*) = \overline{\mathcal{R}(\mathcal{G})}^\circ$.
- (iv) If E is reflexive, then $\mathcal{N}(\mathcal{G}) = \overline{\mathcal{R}(\mathcal{G}^*)}^{\circ}$.
- (v) The \$\mathcal{G}^*\$ is a (C*-DS) ((C*-UDS) of *-class) in \$E^*\$ iff \$\mathcal{G}\$ is a dense pre-(C-DS) (pre-(C-UDS) of *-class). If \$E\$ is reflexive, then \$\mathcal{G}^*\$ is a dense pre-(C*-DS) (pre-(C*-UDS) of *-class) in \$E^*\$ iff \$\mathcal{G}\$ is a (C-DS) ((C-UDS) of *-class).

The following proposition has been considered for the first time by J. Kisyński in [282, Proposition 2] (E Banach space, C = I).

PROPOSITION 3.3.5. Suppose that $\mathcal{G} \in \mathcal{D}'_0(L(E))$ $(\mathcal{G} \in \mathcal{D}'_0(L(E)))$ and $\mathcal{G}(\varphi)C = C\mathcal{G}(\varphi), \varphi \in \mathcal{D}$ $(\varphi \in \mathcal{D}^*)$. Then \mathcal{G} is a pre-(C-DS) (pre-(C-UDS) of *-class) iff

$$(320) \ \mathcal{G}(\varphi')\mathcal{G}(\psi) - \mathcal{G}(\varphi)\mathcal{G}(\psi') = \psi(0)\mathcal{G}(\varphi)C - \varphi(0)\mathcal{G}(\psi)C, \quad \varphi, \psi \in \mathcal{D} \ (\varphi, \psi \in \mathcal{D}^*).$$

PROOF. If \mathcal{G} satisfies (C.S.1), then (320) follows immediately from (C.S.1) and the equality $\varphi' *_0 \psi - \varphi *_0 \psi' = \psi(0)\varphi - \varphi(0)\psi$, $\varphi, \psi \in \mathcal{D}$ ($\varphi, \psi \in \mathcal{D}^*$). Suppose now that (320) holds, $\varphi, \psi \in \mathcal{D}$ ($\varphi, \psi \in \mathcal{D}^*$), a > 0 and $\operatorname{supp}(\psi) \subseteq (-\infty, a]$. Since $\mathcal{G} \in \mathcal{D}'_0(L(E))$ ($\mathcal{G} \in \mathcal{D}_0^*(L(E))$), and the function $t \mapsto \int_0^a [\varphi(t-s)\psi(s) - \varphi(-s)\psi(t+s)] ds$, $t \in \mathbb{R}$ belongs to \mathcal{D} (\mathcal{D}^*) with ($\varphi *_0 \psi$)(t) = $\int_0^a [\varphi(t-s)\psi(s) - \varphi(-s)\psi(t+s)] ds$, $t \ge 0$, we have

$$\mathcal{G}(\varphi *_{0} \psi)Cx = \mathcal{G} \int_{0}^{a} [\varphi(\cdot - s)\psi(s) - \varphi(-s)\psi(\cdot + s)]Cx \, ds$$

$$= \int_{0}^{a} [\psi(s)\mathcal{G}(\varphi(\cdot - s))Cx - \varphi(-s)\mathcal{G}(\psi(\cdot + s))Cx]ds$$

(321)
$$= \int_{0}^{a} [\mathcal{G}(\varphi'(\cdot - s))\mathcal{G}(\psi(\cdot + s))x - \mathcal{G}(\varphi(\cdot - s))\mathcal{G}(\psi'(\cdot + s))x]ds$$

(322)
$$= -\int_{0}^{a} \frac{d}{ds} [\mathcal{G}(\varphi(\cdot - s))\mathcal{G}(\psi(\cdot + s))x] ds$$
$$= \mathcal{G}(\varphi)\mathcal{G}(\psi)x - \mathcal{G}(\varphi(\cdot - a))\mathcal{G}(\psi(\cdot + a))x$$
$$= \mathcal{G}(\varphi)\mathcal{G}(\psi)x - \mathcal{G}(\varphi(\cdot - a))0x = \mathcal{G}(\varphi)\mathcal{G}(\psi)x$$

for any $x \in E$ and $\varphi, \psi \in \mathcal{D}$ ($\varphi, \psi \in \mathcal{D}^*$), where (321) follows from an application of (320), and (322) from an elementary argumentation involving the continuity of \mathcal{G} as well as the facts that for each function $\zeta \in \mathcal{D}$ ($\zeta \in \mathcal{D}^*$) we have that $\lim_{h\to 0} (\tau_h \zeta) = \zeta$ in \mathcal{D} (\mathcal{D}^*), $\lim_{h\to 0} \frac{1}{h} (\tau_h \zeta - \zeta) = \zeta'$ in \mathcal{D} (\mathcal{D}^*) and the set { $\tau_h \zeta : |h| \leq 1$ } is bounded in \mathcal{D} (\mathcal{D}^*). The proof of proposition is complete.

In [351], we have recently proved that every (C-DS) ((C-UDS) of *-class) in locally convex space is uniquely determined by its generator. Contrary to the single-valued case, different pre-(C-DS)'s (pre-(C-UDS)'s of *-class) can have the same generator. To see this, we can employ [364, Example 2.3]: Let C = I, Eis a Banach space and $T \in L(E)$ is nilpotent of order $n \ge 2$. Then the pre-(C-DS)'s $\mathcal{G}_1(\cdot) \equiv \sum_{i=0}^{n-2} \cdot^{(i)}(0)T^{i+1}$ and $\mathcal{G}_2(\cdot) \equiv 0$ have the same generator $\mathcal{A} \equiv E \times E$. In Theorem 3.3.6 and Theorem 3.3.8, we clarify connections between degenerate C-distribution semigroups and degenerate local integrated C-semigroups. For the proof of first theorem, we need some preliminaries from our previous research study of distribution cosine functions (see e.g. [291, Section 3.4]): Let $\eta \in \mathcal{D}_{[-2,-1]}$ be a fixed test function satisfying $\int_{-\infty}^{\infty} \eta(t)dt = 1$. Then, for every fixed $\varphi \in \mathcal{D}$, we define $I(\varphi)$ as follows

(323)
$$I(\varphi)(x) := \int_{-\infty}^{x} \left[\varphi(t) - \eta(t) \int_{-\infty}^{\infty} \varphi(u) du\right] dt, \quad x \in \mathbb{R}$$

It can be simply verified that, for every $\varphi \in \mathcal{D}$ and $n \in \mathbb{N}$, we have $I(\varphi) \in \mathcal{D}$, $I^n(\varphi^{(n)}) = \varphi, \frac{d}{dx}I(\varphi)(x) = \varphi(x) - \eta(x) \int_{-\infty}^{\infty} \varphi(u) du, x \in \mathbb{R}$ as well as that, for every $\varphi \in \mathcal{D}_{[a,b]}$ $(-\infty < a < b < \infty)$, we have: $\operatorname{supp}(I(\varphi)) \subseteq [\min(-2, a), \max(-1, b)]$. This simply implies that, for every $\tau > 2, -1 < b < \tau$ and for every $m, n \in \mathbb{N}$ with $m \leq n$, we have:

(324)
$$I^{n}(\mathcal{D}_{(-\tau,b]}) \subseteq \mathcal{D}_{(-\tau,b]} \text{ and } \frac{d^{m}}{dx^{m}} I^{n}(\varphi)(x) = I^{m-n}\varphi(x), \quad \varphi \in \mathcal{D}, \ x \ge 0,$$

.....

where $I^0 \varphi := \varphi, \varphi \in \mathcal{D}$. Now we are ready to show the following extension of [**364**, Proposition 4.3 a)], given here with a different proof.

THEOREM 3.3.6. Let \mathcal{G} be a pre-(C-DS) generated by \mathcal{A} , and let \mathcal{G} be of finite order. Then, for every $\tau > 0$, there exist a number $n_{\tau} \in \mathbb{N}$ and a local n_{τ} -times integrated C-semigroup $(S_{n_{\tau}}(t))_{t \in [0,\tau)}$ such that

(325)
$$\mathcal{G}(\varphi)x = (-1)^{n_{\tau}} \int_0^\infty \varphi^{(n_{\tau})}(s) S_{n_{\tau}}(s) x \, ds, \quad \varphi \in \mathcal{D}_{(-\tau,\tau)}, \ x \in E.$$

Furthermore, $(S_{n_{\tau}}(t))_{t \in [0,\tau)}$ is an n_{τ} -times integrated C-existence family with a subgenerator \mathcal{A} , and the admissibility of space $L(\mathcal{N}(\mathcal{G}))$ implies that $S_{n_{\tau}}(t)x = 0$, $t \in [0,\tau)$ for some $x \in \mathcal{N}(\mathcal{G})$ iff $T_i x = 0$ for $0 \leq i \leq n_{\tau} - 1$; see Theorem 3.3.2(i) with $m \geq n_{\tau} - 1$.

PROOF. Let $\tau > 2$ and $\rho \in \mathcal{D}_{[0,1]}$ with $\int \rho \, dm = 1$ be fixed. Set $\rho_n(\cdot) := n\rho(n\cdot)$, $n \in \mathbb{N}$. Then, for every $t \in [0, \tau)$, the sequence $\rho_n^t(\cdot) := \rho_n(\cdot - t)$ converges to δ_t as $n \to +\infty$ (in the space of scalar-valued distributions). Since \mathcal{G} is of finite order, we know that there exist a number $n_{\tau} \in \mathbb{N}$ and a strongly continuous operator family $(S_{n_{\tau}}(t))_{t \in [0, \tau)} \subseteq L(E)$ such that (325) holds good. We will first prove that $(S_{n_{\tau}}(t))_{t \in [0, \tau)}$ is a local n_{τ} -times integrated C-existence family which commutes with C and do have \mathcal{A} as a subgenerator. In order to do that, observe that the commutation of $\mathcal{G}(\cdot)$ and C yields

$$\int_0^\infty \varphi^{(n_\tau)}(s) CS_{n_\tau}(s) x \, ds = \int_0^\infty \varphi^{(n_\tau)}(s) S_{n_\tau}(s) Cx \, ds, \quad \varphi \in \mathcal{D}_{(-\tau,\tau)}, \ x \in E$$

Plugging $\varphi = I^{n_{\tau}}(\rho_n^t)$ in this expression (cf. also (324)), we get that

$$\int_0^\infty \rho_n^t(s) CS_{n_\tau}(s) x \, ds = \int_0^\infty \rho_n^t(s) S_{n_\tau}(s) Cx \, ds, \quad \varphi \in \mathcal{D}_{(-\tau,\tau)}, \ x \in E, \ t \in [0,\tau).$$

Letting $n \to +\infty$ we obtain $CS_{n_{\tau}}(t)x = S_{n_{\tau}}(t)Cx, x \in E, t \in [0, \tau)$. Now we will prove that the condition (B) holds with the number α replaced with the number n_{τ} therein. By Proposition 3.3.3(iii), we have $(\mathcal{G}(\varphi)x, \mathcal{G}(-\varphi')x - \varphi(0)Cx) \in \mathcal{A}, \varphi \in \mathcal{D}, x \in E$. Applying integration by parts and multiplying with $(-1)^{n_{\tau}+1}$ after that, the above implies

$$\left(\int_{0}^{\infty} \varphi^{(n_{\tau}+1)}(s) \int_{0}^{s} S_{n_{\tau}}(r) x \, dr \, ds, \int_{0}^{\infty} \varphi^{(n_{\tau}+1)}(s) S_{n_{\tau}}(s) x \, ds + (-1)^{n_{\tau}} \varphi(0) Cx\right) \in \mathcal{A},$$

for any $\varphi \in \mathcal{D}_{(-\tau,\tau)}$ and $x \in E$. Plugging $\varphi = I^{n_{\tau}+1}(\rho_n^t)$ in this expression, we get that

(326)
$$\left(\int_{0}^{\infty} \rho_{n}^{t}(s) \int_{0}^{s} S_{n\tau}(r) x \, dr \, ds, \int_{0}^{\infty} \rho_{n}^{t}(s) S_{n\tau}(s) x \, ds + (-1)^{n\tau} I^{n\tau+1}(\rho_{n}^{t})(0) Cx\right) \in \mathcal{A},$$

for any $t \in [0, \tau)$ and $x \in E$. Let us prove that

(327)
$$\lim_{n \to +\infty} I^{n_{\tau}+1}(\rho_n^t)(x) = (-1)^{n_{\tau}+1}g_{n_{\tau}+1}(t-x), \quad t \in [0,\tau), \ 0 \le x \le t.$$

Let $t \in [0, \tau)$ and $x \in [0, t]$ be fixed. Then a straightforward integral calculation shows that

$$I^{n_{\tau}+1}(\varphi)(x) = (-1)^{n_{\tau}+1} \int_{x}^{\infty} \int_{x_{n_{\tau}}}^{\infty} \int_{x_{n_{\tau}-1}}^{\infty} \cdots \int_{x_{2}}^{\infty} \varphi(x_{1}) dx_{1} dx_{2} \dots dx_{n_{\tau}+1}$$

for any $\varphi \in \mathcal{D}$. For $\varphi = I^{n_\tau + 1}(\rho_n^t)$, we have

$$I^{n_{\tau}+1}(\rho_{n}^{t})(0) = (-1)^{n_{\tau}+1} \int_{x}^{t+(1/n)} \int_{x_{n_{\tau}}}^{t+(1/n)} \int_{x_{n_{\tau}-1}}^{t+(1/n)} \cdots \int_{x_{2}}^{t+(1/n)} \\ \times \rho_{n}^{t}(x_{1}) dx_{1} dx_{2} \dots dx_{n_{\tau}+1} \\ = (-1)^{n_{\tau}+1} \int_{x}^{t+(1/n)} \int_{x_{n_{\tau}}}^{t+(1/n)} \int_{x_{n_{\tau}-1}}^{t+(1/n)} \cdots \int_{x_{3}}^{t+(1/n)} \\ \times \left[1 - \int_{0}^{nx_{2}-nt} \rho(x_{1}) dx_{1}\right] dx_{2} \dots dx_{n_{\tau}+1}$$

3.3. DEGENERATE C-DISTRIBUTION SEMIGROUPS...

$$= (-1)^{n_{\tau}+1} \int_{x}^{t+(1/n)} \int_{x_{n_{\tau}}}^{t+(1/n)} \int_{x_{n_{\tau}-1}}^{t+(1/n)} \cdots \int_{x_{3}}^{t+(1/n)} \\ \times dx_{2} \dots dx_{n_{\tau}+1} \\ - (-1)^{n_{\tau}+1} \int_{x}^{t+(1/n)} \int_{x_{n_{\tau}}}^{t+(1/n)} \int_{x_{n_{\tau}-1}}^{t+(1/n)} \cdots \int_{t}^{t+(1/n)} \\ \times \int_{0}^{nx_{2}-nt} \rho(x_{1}) dx_{1} dx_{2} \dots dx_{n_{\tau}+1} \\ := (-1)^{n_{\tau}+1} [I_{1}(t,x,n) - I_{2}(t,x,n)], \quad t \in [0,\tau).$$

Since

$$\int_{t}^{t+(1/n)} \int_{0}^{nx_{2}-nt} \rho(x_{1}) dx_{1} dx_{2} \leq 1/n, \quad t \in [0,\tau), \ n \in \mathbb{N},$$

we have that $\lim_{n\to+\infty} I_2(t,x,n) = 0, t \in [0,\tau)$. Clearly,

$$\lim_{n \to +\infty} I_1(t, x, n) = \int_x^t \int_{x_{n_\tau}}^t \int_{x_{n_\tau-1}}^t \cdots \int_{x_3}^t dx_2 \dots dx_{n_\tau+1} = g_{n_\tau+1}(t-x).$$

This gives (327). Keeping in mind this equality and letting $n \to +\infty$ in (326), we obtain (B). It remains to be proved the semigroup property of $(S_{n_{\tau}}(t))_{t \in [0,\tau)}$. Toward this end, let us recall that

(328)
$$(\varphi *_0 \psi)^{(n_\tau)}(u) = (\varphi^{(n_\tau)} *_0 \psi)(u) + \sum_{j=0}^{n_\tau - 1} \varphi^{(j)}(0) \psi^{(n_\tau - 1 - j)}(u), \ \varphi, \psi \in \mathcal{D}, \ u \in \mathbb{R}.$$

Fix $x \in E$ and $t, s \in [0, \tau)$ with $t + s \in [0, \tau)$. Using (328), (C.S.1) and the foregoing arguments, we get that, for every $m, n \in \mathbb{N}$ sufficiently large:

$$\int_{0}^{t} \int_{0}^{s} \rho_{n}^{t}(u) \rho_{m}^{s}(v) S_{n\tau}(u) S_{n\tau}(v) x \, du \, dv$$

= $(-1)^{n\tau} \int_{0}^{t+s} \left[(\rho_{n}^{t} *_{0} I^{n\tau}(\rho_{m}^{s}))(u) + \sum_{j=0}^{n\tau-1} I^{n\tau-j}(\rho_{n}^{t})(0) I^{j+1}(\rho_{m}^{s})(u) \right] S_{n\tau}(u) Cx \, du.$

Letting $n \to +\infty$, we obtain with the help of (327) that

$$\begin{split} \int_{0}^{s} \rho_{m}^{s}(v) S_{n_{\tau}}(t) S_{n_{\tau}}(v) x \, dv \\ &= (-1)^{n_{\tau}} \lim_{n \to +\infty} \int_{0}^{t+s} \left[\left(\rho_{n}^{t} *_{0} I^{n_{\tau}}(\rho_{m}^{s}) \right)(u) \right. \\ &+ \sum_{j=0}^{n_{\tau}-1} I^{n_{\tau}-j}(\rho_{n}^{t})(0) I^{j+1}(\rho_{m}^{s})(u) \right] S_{n_{\tau}}(u) Cx \, du \\ &= (-1)^{n_{\tau}} \int_{0}^{t} \left[\sum_{j=0}^{n_{\tau}-1} (-1)^{n_{\tau}-j} g_{n_{\tau}-j}(t) I^{j+1}(\rho_{m}^{s})(u) \right] S_{n_{\tau}}(u) Cx \, du \end{split}$$

$$+ (-1)^{n_{\tau}} \int_{t}^{t+s} \left[I^{n_{\tau}}(\rho_{m}^{s})(u-t) + \sum_{j=0}^{n_{\tau-1}} (-1)^{n_{\tau}-j} g_{n_{\tau}-j}(t) I^{j+1}(\rho_{m}^{s})(u) \right] S_{n_{\tau}}(u) Cx \, du$$

$$= \sum_{j=0}^{n_{\tau-1}} (-1)^{j} g_{n_{\tau}-j}(t) \int_{0}^{s} I^{j+1}(\rho_{m}^{s})(u) S_{n_{\tau}}(u) Cx \, du$$

$$+ (-1)^{n_{\tau}} \int_{t}^{t+s} I^{n_{\tau}}(\rho_{m}^{s})(u-t) S_{n_{\tau}}(u) Cx \, du.$$

The semigroup property now easily follows by letting $m \to +\infty$ in the above expression, with the help of (327) and the identity

$$\sum_{j=0}^{n_{\tau-1}} g_{n_{\tau}-j}(t)g_{j+1}(s-u) = g_{n_{\tau}}(t+s-u), \quad u>0.$$

The rest essentially follows as in the proof of [364, Proposition 4.3 a)].

- REMARK 3.3.7. (i) We have already seen that $\mathcal{G}(\cdot) \equiv 0$ is a degenerate pre-distribution semigroup with the generator $\mathcal{A} \equiv E \times E$. Then, for every $\tau > 0$ and for every number $n_{\tau} \in \mathbb{N}$, there exists only one local n_{τ} -times integrated semigroup $(S_{n_{\tau}}(t) \equiv 0)_{t \in [0,\tau)}$ so that (325) holds. It is clear that the condition (B) holds and that condition (A) does not hold here. Denote by \mathcal{A}_{τ} the integral generator of $(S_{n_{\tau}}(t) \equiv 0)_{t \in [0,\tau)}$. Then $\mathcal{A}_{\tau} = \{0\} \times E$ is strictly contained in the integral generator \mathcal{A} of \mathcal{G} . Furthermore, if $C \neq 0$, then there do not exist $\tau > 0$ and $n_{\tau} \in \mathbb{N}$ such that \mathcal{A} is the integral generator (subgenerator) of a local n_{τ} -times integrated C-semigroup.
- (ii) A similar line of reasoning as in the final part of the proof of [**291**, Theorem 3.1.9] shows that for each $(x, y) \in \mathcal{A}$ there exist elements $x_0, x_1, \ldots, x_{n_{\tau}}$ in E such that

$$S_{n_{\tau}}(t)x - g_{n_{\tau}+1}(t)Cx - \int_{0}^{t} S_{n_{\tau}}(s)y \, ds = \sum_{j=0}^{n_{\tau}} g_{j+1}(t)x_{j}, \quad t \in [0,\tau)$$

and $x_j \in \mathcal{A}x_{j-1}$ for $1 \leq j \leq n_{\tau}$. In purely mutivalued case, it is not clear how we can prove that $x_j = 0$ for $0 \leq j \leq n_{\tau}$ without imposing some additional displeasing conditions; if \mathcal{A} is single-valued, then it can be easily seen that $x_j = 0$ for $0 \leq j \leq n_{\tau}$ so that $(S_{n_{\tau}}(t))_{t \in [0,\tau)}$ is an n_{τ} -times integrated C-semigroup with a subgenerator \mathcal{A} .

(iii) Using dualization, we can simply reformulate the second equality appearing on the second line after the equation [364, (11)] in our context.

The proof of subsequent theorem is very similar to that of [291, Theorem 3.1.8].

THEOREM 3.3.8. Suppose that there exists a sequence $((p_k, \tau_k))_{k \in \mathbb{N}_0}$ in $\mathbb{N}_0 \times (0, \infty)$ such that $\lim_{k \to \infty} \tau_k = \infty$, $(p_k)_{k \in \mathbb{N}_0}$ and $(\tau_k)_{k \in \mathbb{N}_0}$ are strictly increasing, as

 \square

well as that for each $k \in \mathbb{N}_0$ there exists a local p_k -times integrated C-semigroup $(S_{p_k}(t))_{t \in [0,\tau_k)}$ on E so that

(329)
$$S_{p_m}(t)x = (g_{p_m-p_k} *_0 S_{p_k}(\cdot)x)(t), \quad x \in E, \ t \in [0, \tau_k),$$

provided k < m. Define

$$\mathcal{G}(\varphi)x := (-1)^{p_k} \int_0^\infty \varphi^{(p_k)}(t) S_{p_k}(t) x \, dt, \quad \varphi \in \mathcal{D}_{(-\infty,\tau_k)}, \ x \in E, \ k \in \mathbb{N}_0$$

Then \mathcal{G} is well-defined and \mathcal{G} is a pre-(C-DS).

- REMARK 3.3.9. (i) Denote by \mathcal{A}_k the integral generator of $(S_{p_k}(t))_{t \in [0,\tau_k)}$ $(k \in \mathbb{N}_0)$. Then $\mathcal{A}_k \subseteq \mathcal{A}_m$ for k > m and $\bigcap_{k \in \mathbb{N}_0} \mathcal{A}_k \subseteq \mathcal{A}$, where \mathcal{A} is the integral generator of \mathcal{G} . Even in the case that C = I, $\bigcup_{k \in \mathbb{N}_0} \mathcal{A}_k$ can be a proper subset of \mathcal{A} .
- (ii) Suppose that \mathcal{A} is a subgenerator of $(S_{p_k}(t))_{t \in [0,\tau_k)}$ for all $k \in \mathbb{N}_0$. Then (329) automatically holds.
- (iii) In the case that C = I, then it suffices to suppose that there exists an MLO \mathcal{A} such that \mathcal{A} is a subgenerator of a local *p*-times integrated semigroup $(S_p(t))_{t \in [0,\tau)}$ for some $p \in \mathbb{N}$ and $\tau > 0$ [**329**].

Suppose that $\alpha \in (0,\infty) \setminus \mathbb{N}$ and $f \in C([0,\infty) : E)$. Set $f_{n-\alpha}(t) := (g_{n-\alpha} * f)(t), t \ge 0$. Making use of the dominated convergence theorem, and the change of variables $s \mapsto s - t$, we get that

$$\frac{1}{\Gamma(n-\alpha)}\frac{d^n}{dt^n}\int_t^\infty (s-t)^{n-\alpha-1}\varphi(s)ds = \int_0^\infty g_{n-\alpha}(s)\varphi^{(n)}(t+s)ds, \quad t \ge 0, \ \varphi \in \mathcal{D}.$$
 Hence

Hence,

$$\int_0^\infty W_+^\alpha \varphi(t) f(t) dt = (-1)^n \int_0^\infty g_{n-\alpha}(s) \varphi^{(n)}(t+s) f(t) ds \, dt$$
$$= (-1)^n \int_0^\infty \int_0^t \varphi^{(n)}(t) g_{n-\alpha}(s) f(t-s) ds \, dt$$
$$= (-1)^n \int_0^\infty \varphi^{(n)}(t) f_{n-\alpha}(t) dt, \quad \varphi \in \mathcal{D},$$

where W^{α}_{+} denotes the Weyl fractional derivative. Therefore, if \mathcal{A} is the integral generator of a global α -times integrated C-semigroup $(S_{\alpha}(t))_{t\geq 0}$ on E, then we have that:

$$\int_0^\infty W^{\alpha}_+ \varphi(t) S_{\alpha}(t) x \, dt = (-1)^n \int_0^\infty \varphi^{(n)}(t) S_n(t) x \, dt, \quad x \in E, \ \varphi \in \mathcal{D},$$

with $(S_n(t))_{t\geq 0}$ being the global *n*-times integrated *C*-semigroup generated by \mathcal{A} . Keeping in mind this equality and the proof of [**291**, Theorem 3.1.8], we can deduce the following:

THEOREM 3.3.10. Assume that $\alpha \ge 0$ and \mathcal{A} is the integral generator of a global α -times integrated C-semigroup $(S_{\alpha}(t))_{t\ge 0}$ on E. Set

$$\mathcal{G}_{\alpha}(\varphi)x := \int_{0}^{\infty} W_{+}^{\alpha}\varphi(t)S_{\alpha}(t)x\,dt, \quad x \in E, \ \varphi \in \mathcal{D}.$$

Then \mathcal{G} is a pre-(C-DS) whose integral generator contains \mathcal{A} .

We will accept the following definition of an exponential pre-(C-DS) (pre-(C-UDS) of *-class).

DEFINITION 3.3.11. Let \mathcal{G} be a pre-(C-DS) (pre-(C-UDS) of *-class). Then \mathcal{G} is said to be an exponential pre-(C-DS) (pre-(C-UDS) of *-class) iff there exists $\omega \in \mathbb{R}$ such that $e^{-\omega t} \mathcal{G} \in \mathcal{S}'(L(E))$ ($e^{-\omega t} \mathcal{G} \in \mathcal{S}'^*(L(E))$). We use the shorthand pre-(C-EDS) (pre-(C-EUDS) of *-class) to denote an exponential pre-(C-DS) (pre-(C-UDS) of *-class).

We have the following fundamental result:

THEOREM 3.3.12. Assume that $\alpha \ge 0$ and \mathcal{A} generates an exponentially equicontinuous α -times integrated C-semigroup $(S_{\alpha}(t))_{t\ge 0}$. Define \mathcal{G} through $\mathcal{G}_{\alpha}(\varphi)x := \int_{0}^{\infty} W_{+}^{\alpha}\varphi(t)S_{\alpha}(t)x \, dt, \ x \in E, \ \varphi \in \mathcal{D}$. Then \mathcal{G} is a pre-(C-EDS) whose integral generator contains \mathcal{A} .

REMARK 3.3.13. (i) Suppose that \mathcal{G} is a pre-(C-EDS) generated by \mathcal{A} , $\omega \in \mathbb{R}$ and $e^{-\omega t} \mathcal{G} \in \mathcal{S}'(L(E))$. Suppose, further, that there exist a nonnegative integer n and a continuous function $V : \mathbb{R} \to L(E)$ satisfying that

$$\langle e^{-\omega t}\mathcal{G}, \varphi \rangle = (-1)^n \int_{-\infty}^{\infty} \varphi^{(n)}(t) V(t) dt, \quad \varphi \in \mathcal{D},$$

and that there exists a number $r \ge 0$ such that the operator family $\{(1 + t^r)^{-1}V(t) : t \ge 0\} \subseteq L(E)$ is equicontinuous. Since $e^{-\omega \cdot \mathcal{G}}$ is a pre-(C-EDS) generated by $\mathcal{A}-\omega$, the proof of Theorem 3.3.6 shows that $(V(t))_{t\ge 0}$ is an exponentially equicontinuous *n*-times integrated *C*-semigroup; by Theorem 3.3.12, the integral generator $\hat{\mathcal{A}}^{\omega}$ of $(V(t))_{t\ge 0}$ is contained in $\mathcal{A}-\omega$. Define

$$S_n(t)x := e^{\omega t}V(t)x + \int_0^t \sum_{k=1}^\infty \binom{n}{k} \frac{(-1)^k \omega^k (t-s)^{k-1}}{(k-1)!} e^{\omega s}V(s)x \, ds$$

Arguing as in the proof of [291, Theorem 2.5.1, Theorem 2.5.3], we can prove that the MLO $\hat{\mathcal{A}}^{\omega} + \omega \ (\subseteq \mathcal{A})$ is the integral generator of an exponentially equicontinuous *n*-times integrated *C*-semigroup $(S_n(t))_{t\geq 0}$.

(ii) The conclusions from Theorem 3.3.12 and the first part of this remark can be reword for the classes of q-exponentially equicontinuous integrated C-semigroups and q-exponentially equicontinuous pre-(C-DS)'s; cf. [354] for the notion.

REMARK 3.3.14. Suppose that $\mathcal{G} \in \mathcal{D}'_0(L(E))$ ($\mathcal{G} \in \mathcal{D}'_0(L(E))$), $\mathcal{G}(\varphi)C = C\mathcal{G}(\varphi), \ \varphi \in \mathcal{D} \ (\varphi \in \mathcal{D}^*)$ and \mathcal{A} is a closed MLO on E satisfying that $\mathcal{G}(\varphi)\mathcal{A} \subseteq \mathcal{A}\mathcal{G}(\varphi), \ \varphi \in \mathcal{D} \ (\varphi \in \mathcal{D}^*)$ and

(330)
$$\mathcal{G}(-\varphi')x - \varphi(0)Cx \in \mathcal{AG}(\varphi)x, \quad x \in E, \ \varphi \in \mathcal{D} \ (\varphi \in \mathcal{D}^*).$$

In [354], we have proved the following:

- (i) If $\mathcal{A} = A$ is single-valued, then \mathcal{G} satisfies (C.S.1).
- (ii) If \mathcal{G} satisfies (C.S.2) holds, C is injective and $\mathcal{A} = A$ is single-valued, then \mathcal{G} is a (C-DS) ((C-UDS) of *-class) generated by $C^{-1}AC$.
- (iii) Consider the distribution case. If E is admissible and $\mathcal{A} = A$ is single-valued, then the condition (C.S.2) automatically holds for \mathcal{G} .

As we have already seen, the conclusion from (ii) immediately implies that $\mathcal{A} = A$ must be single-valued and that the operator C must be injective.

Concerning the assertion (i), its validity is not true in multivalued case: Let C = I, let $\mathcal{A} \equiv E \times E$, and let $\mathcal{G} \in \mathcal{D}'_0(L(E))$ ($\mathcal{G} \in \mathcal{D}'^*_0(L(E))$) be arbitrarily chosen. Then \mathcal{G} commutes with \mathcal{A} and (330) holds but \mathcal{G} need not satisfy (C.S.1).

Concerning the assertion (iii) in multivalued case, we can prove that the admissibility of state space E implies that for each $x \in \mathcal{N}(\mathcal{G})$ there exist an integer $k \in \mathbb{N}$ and a finite sequence $(y_i)_{0 \leq i \leq k-1}$ in $D(\mathcal{A})$ such that $y_i \in \mathcal{A}y_{i+1}$ $(0 \leq i \leq k-1)$ and $Cx \in \mathcal{A}y_0 \subseteq \mathcal{A}^{k+2}0$.

Now we will revisit some conditions introduced by J. L. Lions [391] in our new framework. Suppose that $\mathcal{G} \in \mathcal{D}'_0(L(E))$ ($\mathcal{G} \in \mathcal{D}'^*(L(E))$) and \mathcal{G} commutes with C. We analyze the following conditions for \mathcal{G} :

- $(d_1) \ \mathcal{G}(\varphi * \psi)C = \mathcal{G}(\varphi)\mathcal{G}(\psi), \ \varphi, \ \psi \in \mathcal{D}_0 \ (\varphi, \ \psi \in \mathcal{D}_0^*),$
- $(d_3) \ \mathcal{R}(\mathcal{G})$ is dense in E,
- (d₄) for every $x \in \mathcal{R}(\mathcal{G})$, there exists a function $u_x \in C([0,\infty) : E)$ so that $u_x(0) = Cx$ and $\mathcal{G}(\varphi)x = \int_0^\infty \varphi(t)u_x(t)dt, \ \varphi \in \mathcal{D} \ (\varphi \in \mathcal{D}^*),$
- $(d_5) \ (Cx, \mathcal{G}(\psi)x) \in G(\psi_+), \ \psi \in \mathcal{D}, \ x \in E \ (\psi \in \mathcal{D}^*, \ x \in E).$

Suppose that $\mathcal{G} \in \mathcal{D}'_0(L(E))$ ($\mathcal{G} \in \mathcal{D}'^*_0(L(E))$) is a pre-(C-DS) (pre-(C-UDS) of *-class). Then it is clear that \mathcal{G} satisfies (d_1) , our previous examinations shows that \mathcal{G} satisfies (d_5) ; by the proof of [**291**, Proposition 3.1.24], we have that \mathcal{G} also satisfies (d_4) . On the other hand, it is well known that (d_1) , (d_4) and (C.S.2) taken together do not imply (C.S.1), even if C = I; see e.g. [**291**, Remark 3.1.20]. Furthermore, let (d_1) , (d_3) and (d_4) hold. Then (d_5) holds, as well. In order to see this, fix $x \in \mathcal{R}(\mathcal{G})$ and $\varphi \in \mathcal{D}$; then it suffices to show that $(Cx, \mathcal{G}(\varphi)x) \in G(\varphi_+)$. Suppose that (ρ_n) is a regularizing sequence and $u_x(t)$ is a function appearing in the formulation of the property (d_4) . Due to the proof of [**291**, Proposition 3.1.19], we have that, for every $\eta \in \mathcal{D}_0$,

$$\begin{split} \mathcal{G}(\rho_n)\mathcal{G}(\varphi_+*\eta)x &= \mathcal{G}((\varphi_+*\rho_n)*\eta)Cx = \mathcal{G}(\eta)\mathcal{G}(\varphi_+*\rho_n)x\\ &= \mathcal{G}(\eta)\int_0^\infty (\varphi_+*\rho_n)(t)u_x(t)dt\\ &\to \mathcal{G}(\eta)\int_0^\infty \varphi(t)u_x(t)dt = \mathcal{G}(\eta)\mathcal{G}(\varphi)x, \ n \to \infty;\\ \mathcal{G}(\rho_n)\mathcal{G}(\varphi_+*\eta)x &= \mathcal{G}(\varphi_+*\eta*\rho_n)Cx \to \mathcal{G}(\varphi_+*\eta)Cx, \ n \to \infty. \end{split}$$

Hence, $\mathcal{G}(\varphi_+ * \eta)Cx = \mathcal{G}(\eta)\mathcal{G}(\varphi)x$ and (d_5) holds, as claimed. On the other hand, (d₁) is a very simple consequence of (d₅); to verify this, observe that for each $\varphi \in \mathcal{D}_0$ and $\psi \in \mathcal{D}$ we have $\psi_+ * \varphi = \psi *_0 \varphi = \varphi *_0 \psi$, so that (d₅) is equivalent to say that $\mathcal{G}(\varphi *_0 \psi)C = \mathcal{G}(\varphi)\mathcal{G}(\psi) \ (\varphi \in \mathcal{D}_0, \psi \in \mathcal{D}).$ In particular,

(331)
$$\mathcal{G}(\varphi)\mathcal{G}(\psi) = \mathcal{G}(\psi)\mathcal{G}(\varphi), \quad \varphi \in \mathcal{D}_0, \ \psi \in \mathcal{D}.$$

Suppose now that (d_5) holds. Let $\varphi \in \mathcal{D}_0$ and $\psi, \eta \in \mathcal{D}$. Observing that $\psi_+ *\eta_+ *\varphi = (\psi *_0 \eta)_+ *\varphi$, we have (cf. also [364, Remark 3.13]):

(332)
$$\mathcal{G}(\varphi)\mathcal{G}(\eta)\mathcal{G}(\psi) = C\mathcal{G}(\eta_{+}*\varphi)\mathcal{G}(\psi)$$
$$= C\mathcal{G}(\psi_{+}*\eta_{+}*\varphi) = C\mathcal{G}((\psi*_{0}\eta)_{+}*\varphi)C$$
$$= C\mathcal{G}(\varphi)\mathcal{G}(\psi*_{0}\eta) = \mathcal{G}(\varphi)\mathcal{G}(\psi*_{0}\eta)C.$$

By (331)-(332), we get

(333)
$$\mathcal{G}(\eta)\mathcal{G}(\psi)\mathcal{G}(\varphi) = \mathcal{G}(\psi *_0 \eta)C\mathcal{G}(\varphi).$$

Due to (331)-(333), we have the following:

- (i) (d_5) and (d_3) together imply (C.S.1); in particular, (d_1) , (d_3) and (d_4) together imply (C.S.1). This is an extension of [**291**, Proposition 3.1.19].
- (ii) (d_5) and (d_2) together imply that \mathcal{G} is a (C-DS) ((C-UDS) of *-class); in particular, $\mathcal{A} = A$ must be single-valued and C must be injective.

On the other hand, (d_5) does not imply (C.S.1) even supposing that C = I. A simple counterexample is $\mathcal{G} \in \mathcal{D}'_0(L(E))$ ($\mathcal{G} \in \mathcal{D}'^*_0(L(E))$) given by $\mathcal{G}(\varphi)x := \varphi(0)x$, $x \in E, \varphi \in \mathcal{D} \ (\varphi \in \mathcal{D}^*)$.

We have already mentioned that the exponential region E(a, b) has been defined for the first time by W. Arendt, O. El-Mennaoui and V. Keyantuo in [28]:

$$E(a,b) := \{ \lambda \in \mathbb{C} : \operatorname{Re} \lambda \ge b, |\operatorname{Im} \lambda| \le e^{a \operatorname{Re} \lambda} \} \ (a,b>0).$$

Set

$$\hat{\varphi}(\lambda) := \frac{1}{2\pi} \int_{-\infty}^{\infty} e^{\lambda t} \varphi(t) dt, \quad \lambda \in \mathbb{C}, \ \varphi \in \mathcal{D} \ (\varphi \in \mathcal{D}^*).$$

Now we are able to state the following theorem:

THEOREM 3.3.15. Let a > 0, b > 0 and $\alpha > 0$. Suppose that \mathcal{A} is a closed MLO and, for every λ which belongs to the set E(a, b), there exists an operator $F(\lambda) \in L(E)$ so that $F(\lambda)\mathcal{A} \subseteq \mathcal{A}F(\lambda)$, $\lambda \in E(a,b)$, $F(\lambda)x \in (\lambda - \mathcal{A})^{-1}Cx$, $\lambda \in E(a,b)$, $x \in E$, $F(\lambda)C = CF(\lambda)$, $\lambda \in E(a,b)$, $F(\lambda)x - Cx = F(\lambda)y$, whenever $\lambda \in E(a,b)$ and $(x,y) \in \mathcal{A}$, and that the mapping $\lambda \mapsto F(\lambda)x$ is analytic on $\Omega_{a,b}$ and continuous on $\Gamma_{a,b}$, where $\Gamma_{a,b}$ denotes the upwards oriented boundary of E(a,b)and $\Omega_{a,b}$ the open region which lies to the right of $\Gamma_{a,b}$. Let the operator family $\{(1 + |\lambda|)^{-\alpha}F(\lambda) : \lambda \in E(a,b)\} \subseteq L(E)$ be equicontinuous. Define

$$\mathcal{G}(\varphi)x := (-i) \int_{\Gamma_{a,b}} \hat{\varphi}(\lambda) F(\lambda) x \, d\lambda, \quad x \in E, \ \varphi \in \mathcal{D}.$$

Then \mathcal{G} is a pre-(C-DS) generated by an extension of \mathcal{A} .

PROOF. Arguing as in non-degenerate case [354], we can prove that $\mathcal{G} \in \mathcal{D}'_0(L(E))$, as well as that \mathcal{G} commutes with C and \mathcal{A} . The prescribed assumptions imply by [329, Theorem 3.23] (cf. also [291, Theorem 2.7.2(iv)]) that for

each $n \in \mathbb{N}$ with $n > \alpha + 1$ the MLO \mathcal{A} subgenerates a local *n*-times integrated *C*-semigroup $(S_n(t))_{t \in [0, a(n-\alpha-1))}$. It is straightforward to prove [354] that

$$\mathcal{G}(\varphi)x = (-1)^n \int_{-\infty}^{\tau} \varphi^{(n)}(t) S_n(t) x \, dt, \quad x \in E, \ \varphi \in \mathcal{D}_{(-\infty, a(n-\alpha-1))}.$$

Now the conclusion directly follows from Theorem 3.3.8 and Remark 3.3.9 (i)–(ii). $\hfill\square$

- REMARK 3.3.16. (i) If C is injective, $\mathcal{A} = A$ is single-valued, $\rho_C(A) \subseteq E(a, b)$ and $F(\lambda) = (\lambda \mathcal{A})^{-1}C$, $\lambda \in E(a, b)$, then \mathcal{G} is a (C-DS) generated by $C^{-1}AC$ [354]. Even in the case that C = I, the integral generator \mathcal{A} of \mathcal{G} , in multivalued case, can strictly contain $C^{-1}\mathcal{A}C$; see Remark 3.3.7(i).
 - (ii) Let us briefly consider the ultradistribution case. Suppose that there exist $l > 0, \beta > 0$ and k > 0, in the Beurling case, resp., for every l > 0 there exists $\beta_l > 0$, in the Roumieu case, such that the assumptions of Theorem 3.3.8 hold with the exponential region E(a, b) replaced with the region $\Omega_{l,\beta}^{(M_p)} := \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda \ge M(l|\lambda|) + \beta\}$, resp. $\Omega_{l,\beta_l}^{\{M_p\}} := \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda \ge M(l|\lambda|) + \beta\}$, resp. $\Omega_{l,\beta_l}^{\{M_p\}} := \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda \ge M(l|\lambda|) + \beta_l\}$. Define \mathcal{G} similarly as above. Then $\mathcal{G} \in \mathcal{D}_0^{\prime*}(L(E)), \mathcal{G}$ commutes with C and \mathcal{A} , and (330) holds. But, in the present situation, we do not know whether \mathcal{G} has to satisfy (C.S.1) in the degenerate case. This is a new open problem we would like to address to our readers.

Concerning smoothing properties of C-(ultra)distribution semigroups in locally convex spaces, we will only present, in Theorem 3.3.18, how one can solve the second part of the fourth question raised at the end of paper [**330**] in the affirmative. For this purpose, we need the following definition (C will be injective and $\mathcal{A} = A$ will be single-valued).

DEFINITION 3.3.17 (cf. [330, Definition 2.3] for the Banach space case). Suppose that \mathcal{G} is a *C*-distribution semigroup (*C*-ultradistribution semigroup of *class), $j \in \mathbb{N}$ and $\alpha \in (0, \frac{\pi}{2}]$. Then it is said that \mathcal{G} is an (infinitely, *j*-times) differentiable *C*-distribution semigroup of c-ultradistribution semigroup of *-class), resp. an analytic *C*-distribution semigroup of angle α (analytic *C*-ultradistribution semigroup of *-class and angle α), iff $G(\delta_t)C \in L(E)$, t > 0 and the mapping $t \mapsto G(\delta_t)C \in L(E)$, t > 0 is (infinitely, *j*-times) differentiable, resp. iff $G(\delta_t)C \in L(E)$, t > 0 and the mapping $t \mapsto G(\delta_t)C \in L(E)$, t > 0 and the mapping $t \mapsto G(\delta_t)C \in L(E)$, t > 0 and the mapping $t \mapsto G(\delta_t)C \in L(E)$, t > 0 and the mapping $t \mapsto G(\delta_t)C \in L(E)$, t > 0 and the mapping $t \mapsto G(\delta_t)C \in L(E)$, t > 0 and the mapping $t \mapsto G(\delta_t)C \in L(E)$, t > 0 and the mapping $t \mapsto G(\delta_t)C \in L(E)$, t > 0 and the mapping $t \mapsto G(\delta_t)C \in L(E)$, t > 0 and the mapping $t \mapsto G(\delta_t)C \in L(E)$, t > 0 and the mapping $t \mapsto G(\delta_t)C \in L(E)$, t > 0 and the mapping $t \mapsto G(\delta_t)C \in L(E)$, t > 0 and $t \models sector \Sigma_\alpha$ (since no confusion seems likely, we shall denote the extension to the sector Σ_α by the same symbol); \mathcal{G} is called real analytic iff $G(\delta_t)C \in L(E)$, t > 0 and if for every t > 0 there exist a number $c_t > 0$ and an analytic function $f : \{z \in \mathbb{C} : |z - t| < c_t\} \to L(E)$ such that $f(s) = G(\delta_s)C$, $s \in (t - c_t, t + c_t)$.

Suppose that \mathcal{G} is a differentiable (C-DS) ((C-UDS) of *-class) generated by A. Let $x \in E$ and $\varphi \in \mathcal{D}_{(0,\infty)}$ ($\varphi \in \mathcal{D}^*_{(0,\infty)}$) be fixed. Then

(334)
$$\mathcal{G}(\psi)G(\delta_t)Cx = G(\delta_t)\mathcal{G}(\psi)Cx = \mathcal{G}(\tau_t\psi)Cx, \quad t > 0, \ \psi \in \mathcal{D}_0 \ (\psi \in \mathcal{D}_0^*).$$

By the proof of [291, Proposition 3.1.24], we have

(335)
$$\mathcal{G}(\varphi)\mathcal{G}(\psi)x = \int_0^\infty \varphi(t)\mathcal{G}(\tau_t\psi)Cx\,dt, \quad \psi \in \mathcal{D}_0 \quad (\psi \in \mathcal{D}_0^*).$$

Owing to (334)–(335), we get that $\mathcal{G}(\psi)\mathcal{G}(\varphi)x = \mathcal{G}(\psi)\int_0^\infty \varphi(t)G(\delta_t)Cx\,dt$. Due to (C.S.2), it follows that

(336)
$$\mathcal{G}(\varphi)x = \int_0^\infty \varphi(t)G(\delta_t)Cx\,dt, \quad x \in E, \ \varphi \in \mathcal{D}_{(0,\infty)} \ (\varphi \in \mathcal{D}^*_{(0,\infty)}).$$

Since $A\mathcal{G}(\varphi)x = \mathcal{G}(-\varphi')x$, $x \in E$, $\varphi \in \mathcal{D}_{(0,\infty)}$ ($\varphi \in \mathcal{D}^*_{(0,\infty)}$), we obtain from (336) that

$$A\int_0^\infty \varphi(t)G(\delta_t)Cx\,dt = \int_0^\infty \varphi(t)\frac{d}{dt}G(\delta_t)Cx\,dt, \quad x \in E, \ \varphi \in \mathcal{D}_{(0,\infty)} \ (\varphi \in \mathcal{D}^*_{(0,\infty)}).$$

Choosing a sequence $(\varphi_n)_{n \in \mathbb{N}}$ in $\mathcal{D}_{(0,\infty)}$ $(\mathcal{D}^*_{(0,\infty)})$ converging to δ_{t_0} in the sense of (ultra-)distributions, and using the closedness of A, we obtain from the previous equality that

(337)
$$AG(\delta_{t_0})Cx = \left(\frac{d}{dt}G(\delta_t)Cx\right)_{t=t_0}, \quad x \in E, \ t_0 > 0.$$

Now we are able to formulate the following theorem.

THEOREM 3.3.18. Suppose that \mathcal{G} is a differentiable distribution semigroup (ultradistribution semigroup of *-class) generated by A. Then \mathcal{G} is infinitely differentiable.

PROOF. Put $T(t) := G(\delta_t), t > 0$. Combining Proposition 3.3.3(ii) and the equality $\delta_{t+s} = \delta_t * \delta_s$ $(t, s \ge 0)$, it readily follows that $(T(t))_{t>0} \subseteq L(E)$ is a semigroup, i.e., that T(t+s) = T(t)T(s), t, s > 0. It suffices to show that, for every $n \in \mathbb{N}$, the following holds:

(a) The mapping $t \mapsto T(t) \in L(E)$, t > 0 is *n*-times differentiable and $T^{(n)}(t) = A^n T(t) \in L(E), t > 0.$

We will prove the validity of assertion (a) by induction on n (cf. also [445, Lemma 4.2, p. 52]). Clearly, the differentiability of \mathcal{G} taken together with (337) implies that (a) holds for n = 1. Suppose now that (a) holds for some $n \in \mathbb{N}$; let us prove that (a) holds with the number n replaced by n+1. Let the number t > 0 be fixed, and let 0 < s < t. Then the induction hypothesis implies

$$T^{(n)}(t') = A^n T(t') = A^n T(t'-s)T(s) = T(t'-s)A^n T(s) \in L(E), \quad t' > s.$$

Hence,

$$T^{(n+1)}(t) = \lim_{h \to 0} \frac{T^{(n)}(t+h) - T^{(n)}(t)}{h}$$

=
$$\lim_{h \to 0} \frac{T(t+h-s) - T(t-s)}{h} A^n T(s) = T'(t-s) A^n T(s) \in L(E).$$

This shows that the mapping $t \mapsto T(t) \in L(E), t > 0$ is (n+1)-times differentiable and $T^{(n+1)}(t) = T'(t-s)A^nT(s) \in L(E)$. It remains to be proved that $T^{(n+1)}(t) = A^{n+1}T(t)$. But, this simply follows from the following equalities involving (337): $T'(t-s)A^nT(s) = [AT(t-s)]A^nT(s) = A[T(t-s)A^nT(s)] = A[A^nT(t)] = A^{n+1}T(t)$.

In [330], we have essentially used some results from the theory of distribution semigroups for the purpose of characterizing differentiability of non-degenerate integrated semigroups. Unfortunately, these methods are inapplicable in the degenerate case.

3.4. Degenerate *C*-distribution cosine functions and degenerate *C*-ultradistribution cosine functions in locally convex spaces

In this section, we assume that (M_n) is a sequence of positive real numbers satisfying $M_0 = 1$, (M.1), (M.2) and (M.3)'. We need some preliminaries concerning the first antiderivative of a vector-valued (ultra)distribution; cf. also the previous section for distributional case. Let $\eta \in \mathcal{D}_{[-2,-1]}$ $(\eta \in \mathcal{D}^*_{[-2,-1]})$ be a fixed test function satisfying $\int_{-\infty}^{\infty} \eta(t) dt = 1$. Then, for every fixed $\varphi \in \mathcal{D}$ ($\varphi \in \mathcal{D}^*$), we define $I(\varphi)$ by (323). It can be simply verified that, for every $\varphi \in \mathcal{D}$ ($\varphi \in \mathcal{D}^*$) and $n \in \mathbb{N}$, we have $I(\varphi) \in \mathcal{D}(I(\varphi) \in \mathcal{D}^*), I^n(\varphi^{(n)}) = \varphi, \frac{d}{dx}I(\varphi)(x) = \varphi(x) - \eta(x)\int_{-\infty}^{\infty}\varphi(u)du,$ $x \in \mathbb{R}$ as well as that, for every $\varphi \in \mathcal{D}_{[a,b]}$ ($\varphi \in \mathcal{D}^*_{[a,b]}$), where $-\infty < a < b < \infty$, we have: $\operatorname{supp}(I(\varphi)) \subset [\min(-2, a), \max(-1, b)]$. This simply implies that, for every $\tau > 2, -1 < b < \tau$ and for every $m, n \in \mathbb{N}$ with $m \leq n$, (324) holds in both cases, distributional and ultradistributional. Define now G^{-1} by $G^{-1}(\varphi) := -G(I(\varphi))$, $\varphi \in \mathcal{D} \ (\varphi \in \mathcal{D}^*)$. In distributional case, it is well known that $G^{-1} \in \mathcal{D}'(L(E))$ and $(G^{-1})' = G$; more precisely, $-G^{-1}(\varphi') = G(I(\varphi')) = G(\varphi), \varphi \in \mathcal{D}$. Since, for every h > 0 and for every non-empty compact subset K of \mathbb{R} , we have that the convergence $\varphi_n \to \varphi, n \to \infty$ in $\mathcal{D}_K^{M_p,h}$ implies the convergence $I(\varphi_n) \to I(\varphi), n \to \infty$ in $\mathcal{D}_{K'}^{M_p,h}$, where $K' = [\min(-2, \inf(K)), \max(-1, \sup(K))]$, the same holds in ultradistributional case. It is not problematic to verify that, in both cases, distributional and ultradistributional, the implication $\operatorname{supp}(G) \subseteq [0,\infty) \Rightarrow \operatorname{supp}(G^{-1}) \subseteq [0,\infty)$ holds true.

Throughout this section, we assume that E is a barreled SCLCS and $C \in L(E)$ is not necessarily injective operator. We introduce the notions of pre-(C-DCF) and (C-DCF) (pre-(C-UDCF) of *-class and (C-UDCF) of *-class) as follows:

DEFINITION 3.4.1. An element $\mathbf{G} \in \mathcal{D}'_0(L(E))$ ($\mathbf{G} \in \mathcal{D}'_0^*(L(E))$) is called a pre-(C-DCF) (pre-(C-UDCF) of *-class) iff $\mathbf{G}(\varphi)C = C\mathbf{G}(\varphi), \varphi \in \mathcal{D} \ (\varphi \in \mathcal{D}^*)$ and $(CCF_1): \mathbf{G}^{-1}(\varphi *_0 \psi)C = \mathbf{G}^{-1}(\varphi)\mathbf{G}(\psi) + \mathbf{G}(\varphi)\mathbf{G}^{-1}(\psi), \quad \varphi, \psi \in \mathcal{D} \ (\varphi, \psi \in \mathcal{D}^*);$ if, additionally,

$$(CCF_2):$$
 $x = y = 0$ iff $\mathbf{G}(\varphi)x + \mathbf{G}^{-1}(\varphi)y = 0, \quad \varphi \in \mathcal{D}_0 \quad (\varphi \in \mathcal{D}_0^*)$

then **G** is called a *C*-distribution cosine function (*C*-ultradistribution cosine function of *-class), inshort (C-DCF) ((C-UDCF) of *-class). A pre-(C-DCF) (pre-(C-UDCF) of *-class) **G** is called dense iff the set $\mathcal{R}(\mathbf{G}) := \bigcup_{\varphi \in \mathcal{D}_0} R(\mathbf{G}(\varphi))$ $(\mathcal{R}(\mathbf{G}) := \bigcup_{\varphi \in \mathcal{D}_0^*} R(\mathbf{G}(\varphi)))$ is dense in *E*. It is clear that (CCF_2) implies $\mathcal{N}(\mathbf{G}) := \bigcap_{\varphi \in \mathcal{D}_0} N(\mathbf{G}(\varphi)) = \{0\}$ and $\bigcap_{\varphi \in \mathcal{D}_0} N(\mathbf{G}^{-1}(\varphi)) = \{0\}$, and that the assumption $\mathbf{G} \in \mathcal{D}'_0(L(E))$ implies $\mathbf{G}(\varphi) = 0, \varphi \in \mathcal{D}_{(-\infty,0]}$. For $\psi \in \mathcal{D}$, we set $\psi_+(t) := \psi(t)H(t), t \in \mathbb{R}$, where H(t) denotes the Heaviside function. Then $\psi_+ \in \mathcal{E}'_0, \psi \in \mathcal{D}$ and $\varphi * \psi_+ \in \mathcal{D}_0$ for any $\varphi \in \mathcal{D}_0$. The above holds in ultradistributional case, as well.

The following proposition is essential and can be deduced by making use of the arguments given in the proof of [291, Proposition 3.4.3].

PROPOSITION 3.4.2. Let $\mathbf{G} \in \mathcal{D}'_0(L(E))$ ($\mathbf{G} \in \mathcal{D}'_0^*(L(E))$) and $\mathbf{G}(\cdot)C = C\mathbf{G}(\cdot)$. Then \mathbf{G} is a pre-(C-DCF) in E (pre-(C-UDCF) of *-class in E) iff

$$\mathcal{G} \equiv \begin{pmatrix} \mathbf{G} & \mathbf{G}^{-1} \\ \mathbf{G}' - \delta \otimes C & \mathbf{G} \end{pmatrix}$$

is a pre-(C-DS) in $E \oplus E$ (pre-(C-UDS) of *-class in $E \oplus E$), where

$$\mathcal{C} \equiv \begin{pmatrix} C & 0 \\ 0 & C \end{pmatrix}.$$

Moreover, \mathcal{G} is a (C-DS) ((C-UDS) of *-class) iff **G** is a pre-(C-DCF) (pre-(C-UDCF) of *-class) which satisfies (CCF₂).

Using Proposition 3.4.2, the proof of [291, Proposition 3.4.4], as well as the fact that any pre-(C-DS) (pre-(C-UDS) of *-class) satisfies (d_5) , and the fact that, for every $\mathbf{G} \in \mathcal{D}'_0(L(E))$ ($\mathbf{G} \in \mathcal{D}'_0(L(E))$) commuting with C, the validity of conditions (d_5) and (C.S.2) implies that \mathbf{G} is a (C-DS) ((C-UDS) of *-class), we can prove the following generalization of [292, Proposition 3.2.4(ii)].

PROPOSITION 3.4.3. Let $\mathbf{G} \in \mathcal{D}'_0(L(E))$ ($\mathbf{G} \in \mathcal{D}'_0^*(L(E))$) and $\mathbf{G}(\cdot)C = C\mathbf{G}(\cdot)$. Then the following holds:

(i) If **G** is a pre-(C-DCF) (pre-(C-UDCF) of
$$*$$
-class), then

(338) $\mathbf{G}^{-1}(\varphi * \psi_{+})C = \mathbf{G}^{-1}(\varphi)\mathbf{G}(\psi) + \mathbf{G}(\varphi)\mathbf{G}^{-1}(\psi), \quad \varphi \in \mathcal{D}_{0}, \ \psi \in \mathcal{D}.$

(ii) If (CCF_2) and (338) hold, then **G** is a (C-DCF) ((C-UDCF) of *-class).

If **G** is a pre-(C-DCF) (pre-(C-UDCF) of *-class), then we can almost directly prove that the dual $\mathbf{G}(\cdot)^*$ is a pre-(C*-DCF) (pre-(C*-UDCF) of *-class) on E^* satisfying $\mathcal{N}(\mathbf{G}^*) = \overline{\mathcal{R}(\mathbf{G})}^{\circ}$, and that the reflexivity of E additionally implies that $\mathcal{N}(\mathbf{G}) = \overline{\mathcal{R}(\mathbf{G}^*)}^{\circ}$.

Making use of Proposition 3.4.2 and Proposition 3.3.5, we can achieve the following proposition.

PROPOSITION 3.4.4. Suppose that $\mathbf{G} \in \mathcal{D}'_0(L(E))$ ($\mathbf{G} \in \mathcal{D}'_0^*(L(E))$) and $\mathbf{G}(\cdot)C = C\mathbf{G}(\cdot)$. Then \mathbf{G} is a pre-(C-DCF) (pre-(C-UDCF) of *-class) iff for every $\varphi, \psi \in \mathcal{D}$ ($\varphi, \psi \in \mathcal{D}^*$), we have:

$$\mathbf{G}^{-1}(\varphi)\mathbf{G}'(\psi) - \mathbf{G}'(\varphi)\mathbf{G}^{-1}(\psi) = \psi(0)\mathbf{G}^{-1}(\varphi)C - \varphi(0)\mathbf{G}^{-1}(\psi)C.$$

Assume **G** is a pre-(C-DCF) (pre-(C-UDCF) of *-class). Then we define the (integral) generator **A** of **G** by

$$\mathbf{A} := \{ (x, y) \in E \oplus E : \mathbf{G}^{-1}(\varphi'') x = \mathbf{G}^{-1}(\varphi) y \text{ for all } \varphi \in \mathcal{D}_0 \}$$

Then **A** is a closed MLO and it can be easily seen that $\mathbf{A} \subseteq C^{-1}\mathbf{A}C$, with the equality in the case that the operator C is injective. If (CCF_2) holds, then it is clear that $\mathbf{A} = A$ is a closed single-valued linear operator.

We can almost straightforwardly extend the assertion of [291, Lemma 3.4.7]:

LEMMA 3.4.5. Let **A** be the generator of a pre-(C-DCF) (pre-(C-UDCF) of *class) **G**. Then $\mathcal{A} \subseteq \mathcal{B}$, where $\mathcal{A} \equiv \begin{pmatrix} 0 & I \\ \mathbf{A} & 0 \end{pmatrix}$ and \mathcal{B} is the generator of \mathcal{G} . Furthermore, $(x, y) \in \mathbf{A} \Leftrightarrow \begin{pmatrix} x \\ 0 \end{pmatrix}, \begin{pmatrix} 0 \\ y \end{pmatrix} \in \mathcal{B}$ and \mathcal{B} is single-valued iff **G** is a (C-DCF) ((C-UDCF) of *-class).

It is worth observing that

(339)
$$\left(\begin{pmatrix} 0\\ x \end{pmatrix}, \begin{pmatrix} x\\ 0 \end{pmatrix} \right) \in \mathcal{B}, \quad x \in E.$$

Suppose now that **G** is a (**C**-DCF) ((**C**-UDCF) of *-class) generated by **A**. Then Proposition 3.4.2 yields that \mathcal{G} is a (C-DS) ((C-UDS) of *-class). This implies that the integral generator \mathcal{B} of \mathcal{G} is single-valued and the operator \mathcal{C} is injective. Therefore, the integral generator **A** of **G** is single-valued and the operator \mathcal{C} is injective.

It is not clear whether the assumption that the integral generator \mathbf{A} of a pre-(C-DCF) (pre-(C-UDCF) of *-class) \mathbf{G} is single-valued implies (CCF_2) for \mathbf{G} . Let \mathbf{G} be a pre-(C-DCF) (pre-(C-UDCF) of *-class) generated by \mathbf{A} . Due to the proofs of [291, Proposition 3.4.8] and [291, Proposition 3.4.30], we obtain that the following holds:

(a) Let
$$\psi \in \mathcal{D}$$
 ($\psi \in \mathcal{D}^*$) and $x, y \in E$. Then $(\mathbf{G}(\psi)x, y) \in \mathbf{A}$ iff
 $\mathbf{G}(\psi'')x + \psi'(0)Cx - y \in \bigcap_{\varphi \in \mathcal{D}_0} N(\mathbf{G}^{-1}(\varphi)) \quad \left(\in \bigcap_{\varphi \in \mathcal{D}_0^*} N(\mathbf{G}^{-1}(\varphi)) \right).$

(b)
$$(\mathbf{G}(\psi)x, \mathbf{G}(\psi'')x + \psi'(0)Cx) \in \mathbf{A}, \psi \in \mathcal{D} \ (\psi \in \mathcal{D}^*), x \in E.$$

- (c) $(\mathbf{G}^{-1}(\psi)x, -\mathbf{G}(\psi')x \psi(0)Cx) \in \mathbf{A}, \ \psi \in \mathcal{D} \ (\psi \in \mathcal{D}^*), \ x \in E.$
- (d) $\mathbf{G}(\varphi *_0 \psi)Cx \mathbf{G}(\varphi)\mathbf{G}(\psi)x \in \mathbf{A}\mathbf{G}^{-1}(\varphi)\mathbf{G}^{-1}(\psi)x, \ \varphi, \psi \in \mathcal{D} \ (\varphi, \psi \in \mathcal{D}^*), x \in E.$

If **G** is a (**C**-DCF) ((**C**-UDCF) of *-class) generated by **A**, then the operators \mathcal{B} and **A** are single-valued; then a similar line of reasoning as in the proof of [**291**, Proposition 3.4.8(iii)-(iv)] shows that, for every $\psi \in \mathcal{D}$ ($\psi \in \mathcal{D}^*$), we have $\mathbf{G}(\psi)\mathbf{A} \subseteq \mathbf{AG}(\psi)$ and $\mathbf{G}^{-1}(\psi)\mathbf{A} \subseteq \mathbf{AG}^{-1}(\psi)$. It is not clear whether the above inclusions hold for pre-(**C**-DCF)'s (pre-(**C**-UDCF)'s of *-class) in general case.

THEOREM 3.4.6. Suppose that $\mathbf{G} \in \mathcal{D}'_0(L(E))$ ($\mathbf{G} \in \mathcal{D}'_0(L(E))$), $\mathbf{G}(\cdot)C = C\mathbf{G}(\cdot)$, and \mathbf{A} is a closed MLO on E satisfying that $\mathbf{G}(\cdot)\mathbf{A} \subseteq \mathbf{AG}(\cdot)$ and

(340)
$$\mathbf{G}(\varphi'')x + \varphi'(0)Cx \in \mathbf{AG}(\varphi)x, \quad x \in E, \ \varphi \in \mathcal{D} \ (\varphi \in \mathcal{D}^*).$$

Then the following holds:

(i) If **A** = A is single-valued, then **G** is a pre-(C-DCF) (pre-(C-UDCF) of *-class).

- (ii) If **G** satisfies (CCF₂), C is injective and $\mathbf{A} = A$ is single-valued, then **G** is a (C-DCF) ((C-UDCF) of *-class) generated by $C^{-1}AC$.
- (iii) Consider the distribution case. If E is admissible, then the condition (CCF_2) automatically holds for **G**.

PROOF. We will only outline the most important details of proof. It can be simply proved that $\mathcal{G} \in \mathcal{D}'_0(L(E \oplus E))$ ($\mathcal{G} \in \mathcal{D}'^*_0(L(E \oplus E))$), $\mathcal{G}(\cdot)\mathcal{C} = \mathcal{CG}(\cdot)$, and that \mathcal{A} is a closed MLO in $E \oplus E$. Furthermore, $\mathcal{G}(\cdot)\mathcal{A} \subseteq \mathcal{AG}(\cdot)$ and

$$\mathcal{G}(-\varphi')(x \ y)^T - \varphi(0)\mathcal{C}(x \ y)^T \in \mathcal{A}\mathcal{G}(\varphi)(x \ y)^T, \quad x, y \in E, \ \varphi \in \mathcal{D} \ (\varphi \in \mathcal{D}^*).$$

By Remark 3.3.14, it readily follows that \mathcal{G} is a pre-(\mathcal{C} -DS) in $E \oplus E$ so that (i) follows immediately from Proposition 3.4.2. In order to prove (ii), notice that \mathcal{G} satisfies (C.S.2) and that the analysis from Remark 3.3.14 implies that \mathcal{G} is a pre-(\mathcal{C} -DS) in $E \oplus E$ generated by $\mathcal{C}^{-1}\mathcal{A}\mathcal{C} = \begin{pmatrix} 0 & I \\ C^{-1}\mathcal{A}\mathcal{C} & 0 \end{pmatrix}$. Now the part (ii) simply follows from Proposition 3.4.2 and Lemma 3.4.5. Observing that the admissibility of E implies the admissibility of $E \oplus E$, the proof of (iii) can be deduced similarly. \Box

REMARK 3.4.7. Concerning the assertion (i), its validity is not true in multivalued case: Let C = I, let $\mathbf{A} \equiv E \times E$, and let $\mathbf{G} \in \mathcal{D}'_0(L(E))$ ($\mathcal{G} \in \mathcal{D}'^*_0(L(E))$) be arbitrarily chosen. Then \mathbf{G} commutes with \mathbf{A} and (340) holds but \mathbf{G} need not satisfy (CCF_1).

REMARK 3.4.8. Let $\mathbf{G} \in \mathcal{D}'_0(L(E))$ ($\mathbf{G} \in \mathcal{D}'^*_0(L(E))$) and $\mathbf{G}(\cdot)C = C\mathbf{G}(\cdot)$. Suppose that $\mathcal{A} = A$ is single-valued and C is injective. If \mathbf{G} is a (C-DCF) in E ((C-UDCF) of *-class in E), then \mathcal{B} is single-valued and we can proceed as in the proof of [**291**, Proposition 3.4.8(iii)] so as to conclude that $\mathbf{G}(\cdot)\mathbf{A} \subseteq \mathbf{AG}(\cdot)$. Combining this fact with Proposition 3.4.2, the consideration from Remark 3.3.14 and with the proof of Theorem 3.4.6, we get that \mathbf{G} is a (C-DCF) in E ((C-UDCF) of *-class in E) generated by \mathbf{A} iff \mathcal{G} is a (C-DS) in $E \oplus E$ ((C-UDS) of *-class in $E \oplus E$) generated by \mathcal{A} . This is an extension of [**291**, Theorem 3.2.8(ii)]. In the degenerate case, the integral generator of \mathcal{G} can strictly contain \mathcal{A} . In order to verify this, let E be an arbitrary Banach space, let $P \in L(E)$, and let $P^2 = P$. Define $\mathbf{G}_P(\varphi)x := \int_0^\infty \varphi(t)dt Px, x \in E, \varphi \in \mathcal{D}$. Then $\mathbf{G}_P^{-1}(\varphi)x = \int_0^\infty t\varphi(t)dt Px, x \in E, \varphi \in \mathcal{D}$.

$$\{x, y\} \subseteq N(P) \Leftrightarrow \mathbf{G}_P(\varphi)x + \mathbf{G}_P^{-1}(\varphi)y = 0 \text{ for all } \varphi \in \mathcal{D}_0;$$

see [291, Example 3.4.46]. A simple computation shows that the integral generator of \mathbf{G}_P is the MLO $\mathbf{A} = E \times N(P)$. Furthermore, $([x \ y]^T, [z \ u]^T) \in \mathcal{A}$ iff y = z and $u \in N(P)$, while $([x \ y]^T, [z \ u]^T) \in \mathcal{B}$ iff $y - z \in N(P)$ and $u \in N(P)$. Hence, \mathcal{B} strictly contains \mathcal{A} .

REMARK 3.4.9. Suppose that $\mathcal{A} = A$ is single-valued and C is injective. Since any (C-DS) in $E \oplus E$ ((C-UDS) of *-class in $E \oplus E$) is uniquely determined by its generator, the conclusion established in Remark 3.4.8 shows that there exists at most one (C-DCF) in E ((C-UDCF) of *-class in E) generated by A. Even in the case that E is a Banach space and C = I, this is no longer true in the degenerate case. To see this, let E be an arbitrary Banach space, let $P_1 \in L(E)$, $P_1^2 = P_1$, $P_2 \in L(E), P_2^2 = P_2, N(P_1) = N(P_2)$ and $P_1 \neq P_2$; cf. the previous remark. Then pre-(DCF)'s \mathbf{G}_{P_1} and \mathbf{G}_{P_2} are different but have the same integral generator. We can choose, for example, the matricial operators

$$P_1 = \begin{bmatrix} 0 & 0 \\ 0 & 1 \end{bmatrix}$$
 and $P_2 = \begin{bmatrix} 0 & 1 \\ 0 & 1 \end{bmatrix}$.

We continue by stating the following theorem.

THEOREM 3.4.10. Let a > 0, b > 0 and $\alpha > 0$. Suppose that **A** is a closed MLO and, for every λ which belongs to the set E(a, b), there exists an operator $H(\lambda) \in L(E)$ so that $H(\lambda)\mathbf{A} \subseteq \mathbf{A}H(\lambda), \lambda \in E(a,b), \lambda H(\lambda)x - Cx \in \mathbf{A}[H(\lambda)x/\lambda], \lambda \in E(a,b), x \in E, H(\lambda)C = CH(\lambda), \lambda \in E(a,b), \lambda H(\lambda)x - Cx = H(\lambda)y/\lambda$, whenever $\lambda \in E(a,b)$ and $(x,y) \in \mathbf{A}$, and that the mapping $\lambda \mapsto H(\lambda)$ is strongly analytic on $\Omega_{a,b}$ and strongly continuous on $\Gamma_{a,b}$, where $\Gamma_{a,b}$ denotes the upwards oriented boundary of E(a,b) and $\Omega_{a,b}$ the open region which lies to the right of $\Gamma_{a,b}$. Let the operator family $\{(1 + |\lambda|)^{-\alpha}H(\lambda) : \lambda \in E(a,b)\} \subseteq L(E)$ be equicontinuous. Set

$$\mathbf{G}(\varphi)x := (-i) \int_{\Gamma_{a,b}} \hat{\varphi}(\lambda) H(\lambda) x \, d\lambda, \quad x \in E, \ \varphi \in \mathcal{D}.$$

Then \mathbf{G} is a pre-(C-DCF) generated by an extension of \mathbf{A} .

PROOF. Set

$$F(\lambda) := \begin{bmatrix} H(\lambda) & H(\lambda)/\lambda \\ \lambda H(\lambda) - C & H(\lambda) \end{bmatrix}, \quad \lambda \in E(a,b)$$

and

$$\mathcal{G}(\varphi)(x \ y)^T := (-i) \int_{\Gamma_{a,b}} \hat{\varphi}(\lambda) F(\lambda)(x \ y)^T \ d\lambda, \quad x, y \in E, \ \varphi \in \mathcal{D}.$$

The prescribed assumptions imply that the function $F(\cdot)$ has the properties necessary for applying Theorem 3.3.15. Furthermore, $\operatorname{supp}(\mathbf{G}) \subseteq [0, \infty)$, **G** commutes with C and, by the proof of [**292**, Theorem 3.2.6], we have that

$$\mathcal{G} = \begin{bmatrix} \mathbf{G} & \mathbf{G}^{-1} \\ \mathbf{G}' - \delta \otimes C & \mathbf{G} \end{bmatrix}.$$

By Theorem 3.3.15, we get that \mathcal{G} is a pre-(\mathcal{C} -DS) generated by an extension of \mathcal{A} . Due to Proposition 3.4.2 and Lemma 3.4.5, we obtain that **G** is a pre-(C-DCF) generated by an extension of **A**, as claimed.

- REMARK 3.4.11. (i) Suppose that C is injective, $\mathbf{A} = A$ is single-valued, $\rho_C(A) \subseteq E^2(a,b) \equiv \{\lambda^2 : \lambda \in E(a,b)\}$ and $H(\lambda) = \lambda(\lambda^2 - \mathcal{A})^{-1}C$, $\lambda \in E^2(a,b)$. Then \mathcal{G} is a (C-DCF) generated by $C^{-1}AC$. Even in the case that C = I, the integral generator \mathbf{A} of \mathbf{G} , in multivalued case, can strictly contain $C^{-1}\mathbf{A}C$.
- (ii) Let **A** be a closed MLO, let *C* be injective and commute with **A**, and let $\rho_C(\mathbf{A}) \subseteq E^2(a, b)$. Then the choice $H(\lambda) = \lambda(\lambda^2 \mathbf{A})^{-1}C$, $\lambda \in E(a, b)$ is always possible.

(iii) In ultradistributional case, it is necessary to replace the exponential region E(a, b) from the formulation of Theorem 3.4.10 with a corresponding ultra-logarithmic region. Define the operator $\mathbf{G}(\varphi)$ similarly as above. In non-degenerate case ($\mathbf{A} = A$ single-valued, C injective), it can be proved that $\mathbf{G}(\varphi)$ is a pre-(C-UDCF) generated by an extension of \mathbf{A} ; unfortunately, we do not know then whether \mathbf{G} has to satisfy (CCF_1) in the degenerate case.

Since any pre-(C-DS) (pre-(C-UDS) of *-class) satisfies the condition (d_3) , we want to observe that the assertion of [**292**, Theorem 3.2.11(i)] can be formulated in our context. In non-degenerate case ($\mathbf{A} = A$ single-valued, C injective), the assertion of [**292**, Theorem 3.2.11(ii)] can be simply reformulated for (C-DCF)'s of finite order, when we can also prove the equivalence of the statements (a)–(f) clarified on p. 380 of [**292**].

The analysis of degenerate almost C-(ultra)distribution cosine functions is without the scope of this book. For more details, see [428] and [291, Subsection 3.4.5] and [292, pp. 380–384].

In the remainder of this section, we investigate relations between degenerate C-distribution cosine functions and degenerate integrated C-cosine functions. We start by starting the following result.

THEOREM 3.4.12. Let **G** be a pre-(C-DCF) generated by **A**, and let **G** be of finite order. Then, for every $\tau > 0$, there exist a number $n_{\tau} \in \mathbb{N}$ and a local n_{τ} -times integrated C-cosine function $(C_{n_{\tau}}(t))_{t \in [0,\tau)}$ such that

(341)
$$\mathcal{G}(\varphi) = (-1)^{n_{\tau}} \int_0^{\infty} \varphi^{(n_{\tau})}(t) C_{n_{\tau}}(t) dt, \quad \varphi \in \mathcal{D}_{(-\tau,\tau)}.$$

Furthermore, $(C_{n_{\tau}}(t))_{t \in [0,\tau)}$ is an n_{τ} -times integrated C-cosine existence family with a subgenerator **A**.

PROOF. Let \mathcal{G} and \mathcal{C} be as in the formulation of Proposition 3.4.2, and let \mathcal{A} be the MLO defined in Lemma 3.4.5. Then \mathcal{G} is a pre-(\mathcal{C} -DS) in $E \oplus E$ generated by a closed MLO \mathcal{B} which contains \mathcal{A} . Since **G** is of finite order, we know that, for every $\tau > 0$, there exist a number $n_{\tau} \in \mathbb{N}$ and a continuous mapping $C_{n_{\tau}} : [0, \tau) \to L(E)$ such that (341) holds true. Define

$$S_{n_{\tau}+1}(t) := \begin{pmatrix} \int_0^t C_{n_{\tau}}(s) ds & \int_0^t (t-s) C_{n_{\tau}}(s) ds \\ C_{n_{\tau}}(t) - g_{n_{\tau}+1}(t) C & \int_0^t C_{n_{\tau}}(s) ds \end{pmatrix}, \quad 0 \le t < \tau.$$

Then $S_{n_{\tau}+1}: [0, \tau) \to L(E \oplus E)$ is continuous and

$$\mathcal{G}(\varphi) = (-1)^{n_{\tau}+1} \int_0^\infty \varphi^{(n_{\tau}+1)}(t) S_{n_{\tau}+1}(t) dt, \quad \varphi \in \mathcal{D}_{(-\tau,\tau)}.$$

This immediately implies that $(S_{n_{\tau}+1}(t))_{t\in[0,\tau)}$ is an $(n_{\tau}+1)$ -times integrated Cintegrated semigroup with a subgenerator \mathcal{B} . Due to Theorem 3.2.57, we have that $(C_{n_{\tau}}(t))_{t\in[0,\tau)}$ is an n_{τ} -times integrated C-times integrated cosine function so that it remains to be proved that $(C_{n_{\tau}}(t))_{t\in[0,\tau)}$ is an n_{τ} -times integrated C-cosine existence family with subgenerator **A**, i.e., that $(\int_0^t (t-s)C_{n_\tau}(s)x\,ds, C_{n_\tau}(t)x - g_{n_\tau+1}(t)Cx) \in \mathbf{A}$ for all $t \in [0, \tau)$ and $x \in E$. This is equivalent to say that

$$\left(\begin{pmatrix} \int_0^t (t-s)C_{n_\tau}(s)x\,ds\\ 0 \end{pmatrix}, \begin{pmatrix} 0\\ \int_0^t (t-s)C_{n_\tau}(s)x\,ds \end{pmatrix} \right) \in \mathcal{B}, \quad x \in E, \ t \in [0,\tau).$$

which simply follows from the inclusion (339) and the fact that $(S_{n_{\tau}+1}(t))_{t \in [0,\tau)}$ is an $(n_{\tau}+1)$ -times integrated C-integrated existence family with a subgenerator \mathcal{B} . The proof of the theorem is thereby complete.

- REMARK 3.4.13. (i) It is not clear how we can reconsider Theorem 3.3.6 and Theorem 3.4.12 in ultradistributional case.
 - (ii) If $\mathbf{A} = A$ is single-valued, then \mathcal{A} is single-valued, as well. If so, then $(S_{n_{\tau}+1}(t))_{t\in[0,\tau)}$ is an $(n_{\tau}+1)$ -times integrated \mathcal{C} -integrated semigroup with a subgenerator \mathcal{A} , which implies by Theorem 3.2.57(ii) that $(C_{n_{\tau}}(t))_{t\in[0,\tau)}$ is an n_{τ} -times integrated C-cosine function with a subgenerator \mathcal{A} .
- (iii) If the assumptions of Theorem 3.4.12 hold, then $\mathbf{G}(\varphi)\mathbf{G}(\psi) = \mathbf{G}(\psi)\mathbf{G}(\varphi)$, $\varphi, \psi \in \mathcal{D}$ (in the Banach space setting, this gives the affirmative answer to the question raised on p. 769 of [**331**]). As a simple consequence, we have that, for every $\psi \in \mathcal{D}$, we have $\mathbf{G}(\psi)\mathbf{A} \subseteq \mathbf{AG}(\psi)$ and $\mathbf{G}^{-1}(\psi)\mathbf{A} \subseteq$ $\mathbf{AG}^{-1}(\psi)$.
- (iv) Suppose that C = I and \mathcal{A} (**A**) is the integral generator of a pre-(DS) \mathcal{G} (pre-(DCF) **G**) of finite order. Then Proposition 3.2.68 combined with Theorem 3.3.6 and Theorem 3.4.12 shows that \mathcal{A} (**A**) is stationary dense. Unfortunately, we have already seen in some examples that the resolvent set of \mathcal{A} (**A**) can be empty (compare to [**363**, Theorem 3.5] in semigroup case).

Making use of Theorem 3.2.57 and Proposition 3.4.2, we can simply prove the following analogue of Theorem 3.3.8 for degenerate differential equations of second order. This is an extension of [**291**, Theorem 3.2.5(iii)].

THEOREM 3.4.14. Assume that there exists a sequence $((p_k, \tau_k))_{k \in \mathbb{N}_0}$ in $\mathbb{N}_0 \times (0, \infty)$ such that $\lim_{k\to\infty} \tau_k = \infty$, $(p_k)_{k\in\mathbb{N}_0}$ and $(\tau_k)_{k\in\mathbb{N}_0}$ are strictly increasing, as well as for each $k \in \mathbb{N}_0$ there exists a local p_k -times integrated C-cosine function $(C_{p_k}(t))_{t\in[0,\tau_k)}$ on E satisfying that

(342)
$$C_{p_m}(t)x = (g_{p_m-p_k} * C_{p_k}(\cdot)x)(t), \quad x \in E, \ t \in [0, \tau_k),$$

provided k < m. Define

$$\mathbf{G}(\varphi)x := (-1)^{p_k} \int_0^\infty \varphi^{(p_k)}(t) C_{p_k}(t) x \, dt, \quad \varphi \in \mathcal{D}_{(-\infty,\tau_k)}, \ x \in E, \ k \in \mathbb{N}_0.$$

Then \mathbf{G} is well-defined and \mathbf{G} is a pre-(C-DCF).

As in the case of degenerate C-distribution semigroups, we have the following remarks and comments on Theorem 3.4.14.

- REMARK 3.4.15. (i) Let \mathbf{A}_k be the integral generator of $(C_{p_k}(t))_{t \in [0,\tau_k)}$ $(k \in \mathbb{N}_0)$. Then $\mathbf{A}_k \subseteq \mathbf{A}_m$ for k > m and $\bigcap_{k \in \mathbb{N}_0} \mathbf{A}_k \subseteq \mathbf{A}$, where \mathbf{A} is the integral generator of \mathbf{G} . Even in the case that C = I, $\bigcup_{k \in \mathbb{N}_0} \mathbf{A}_k$ can be a proper subset of \mathbf{A} .
- (ii) Suppose that **A** is a subgenerator of $(C_{p_k}(t))_{t \in [0,\tau_k)}$ for all $k \in \mathbb{N}_0$. Then (342) automatically holds.
- (iii) If C = I, then it suffices to suppose that there exists an MLO **A** subgenerating a local *p*-times integrated cosine function $(C_p(t))_{t \in [0,\tau)}$ for some $p \in \mathbb{N}$ and $\tau > 0$ [**329**].

Proposition 3.4.2 enables us to simply introduce the notion of an exponential pre-(C-DCF) in E (exponential pre-(C-UDCF) of *-class in E):

DEFINITION 3.4.16. Let **G** be a pre-(C-DCF) (pre-(C-UDCF) of *-class). Then **G** is said to be an exponential pre-(C-DCF) (pre-(C-UDCF) of *-class) iff there exists $\omega \in \mathbb{R}$ such that $e^{-\omega t} \mathcal{G} \in \mathcal{S}'(L(E \oplus E))$ ($e^{-\omega t} \mathcal{G} \in \mathcal{S}'^*(L(E \oplus E))$). We use the shorthand pre-(C-EDCF) (pre-(C-EUDCF) of *-class) to denote an exponential pre-(C-DCF) (pre-(C-UDCF) of *-class).

It can be simply verified that a pre-(C-DCF) (pre-(C-UDCF) of *-class) **G** is exponential iff there exists $\omega \in \mathbb{R}$ such that $e^{-\omega t} \mathbf{G}^{-1} \in \mathcal{S}'(L(E))$ ($e^{-\omega t} \mathbf{G}^{-1} \in \mathcal{S}'^*(L(E))$); see e.g. [292, Theorem 3.2.10(i)] and the proof of [291, Proposition 3.4.21].

By the foregoing, we have the following result.

THEOREM 3.4.17. Assume that $\alpha \ge 0$ and **A** is the integral generator of a global α -times integrated C-cosine function $(C_{\alpha}(t))_{t\ge 0}$ on E. Set

$$\mathbf{G}_{\alpha}(\varphi)x := \int_{0}^{\infty} W_{+}^{\alpha}\varphi(t)C_{\alpha}(t)x\,dt, \quad x \in E, \ \varphi \in \mathcal{D}.$$

Then **G** is a pre-(C-DCF) whose integral generator contains **A**. Furthermore, if $(C_{\alpha}(t))_{t\geq 0}$ is exponentially equicontinuous, then **G** is exponential.

REMARK 3.4.18. It is clear that $\mathbf{G}(\cdot) \equiv 0$ is a degenerate pre-distribution cosine function with the generator $\mathcal{A} \equiv E \times E$, as well as that, for every $\tau > 0$ and for every integer $n_{\tau} \in \mathbb{N}$, there exists only one local n_{τ} -times integrated cosine function $(C_{n_{\tau}}(t) \equiv 0)_{t \in [0, \tau)}$ satisfying (341). Then condition (B)' holds and condition (A)' does not hold here. Designate by \mathbf{A}_{τ} the integral generator of $(C_{n_{\tau}}(t) \equiv 0)_{t \in [0, \tau)}$. Then $\mathbf{A}_{\tau} = \{0\} \times E$ is strictly contained in the integral generator \mathbf{A} of \mathbf{G} . Furthermore, if $C \neq 0$, then there do not exist numbers $\tau > 0$ and $n_{\tau} \in \mathbb{N}$ such that \mathbf{A} generates (subgenerates) a local n_{τ} -times integrated Ccosine function.

The notion of a q-exponential pre-(C-DCF) (pre-(C-UDCF) of *-class) can be also introduced and further analyzed. For the sake of brevity, we shall skip all related details concerning this topic here. It is also worth noting that the assertions of [**291**, Theorem 3.6.13, Theorem 3.6.14] can be simply reformulated for non-degenerate ultradistribution sines in locally convex spaces. For more details concerning the semigroup case, the reader may consult [**355**]. We would like to round off this section by drawing the readers' attention on some instructive examples of (pre-)ultradistribution sines in Fréchet and Banach function spaces.

EXAMPLE 3.4.19. (i) Several times before, we have dealt with the Fréchet space $E = \{f \in C^{\infty}([0,\infty)) : \lim_{x \to +\infty} f^{(k)}(x) = 0 \text{ for all } k \in \mathbb{N}_0\},$ equipped with the family of norms $\|f\|_k := \sum_{j=0}^k \sup_{x \ge 0} |f^{(j)}(x)|, f \in E$ $(k \in \mathbb{N}_0).$ Suppose $c_0 > 0, \beta > 0, s > 1, \bar{a} > 0$ and $M_p := p!^s$. Define the operator A by $D(A) := \{u \in E : c_0 u'(0) = \beta u(0)\}$ and $Au := c_0 u''.$ Then, for every two sufficiently small number $\varepsilon > 0, \varepsilon' > 0$ and for every integer $k \in \mathbb{N}_0$, there exist constants $c(\varepsilon, \varepsilon') > 0$ and $c(k, \varepsilon, \varepsilon') > 0$ such that

$$\|(\lambda - A)^{-1}f\|_k \leqslant c(k,\varepsilon,\varepsilon')e^{c(\varepsilon,\varepsilon')|\lambda|^{\varepsilon'}}\|f\|_k, \quad f \in E, \ \lambda \in \Sigma_{\pi-\varepsilon}$$

Suppose now that P(z) is a non-constant complex polynomial of degree $k \in \mathbb{N}$ satisfying that there exists a positive real number a > 0 such that, for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > a$, all zeroes of polynomial $z \mapsto P(z) - \lambda$, $z \in \mathbb{C}$ belong to $\mathbb{C} \setminus (-\infty, 0]$. Let $\bar{a} > a$. Then it can be easily seen with the help of (343) that, for every two sufficiently small number $\varepsilon > 0$, $\varepsilon' > 0$ and for every integer $k \in \mathbb{N}_0$, there exist constants $c(\varepsilon, \varepsilon') > 0$ and $c(k, \varepsilon, \varepsilon') > 0$ such that

$$\|(\lambda - P(A))^{-1}f\|_k \leq c(k,\varepsilon,\varepsilon')e^{c(\varepsilon,\varepsilon')|\lambda|^{\varepsilon'}}\|f\|_k, \quad f \in E, \ \operatorname{Re} \lambda > \bar{a}.$$

Set

$$\mathbf{G}_P(\varphi)f := (-i) \int_{\bar{a}-i\infty}^{\bar{a}+i\infty} \lambda \hat{\varphi}(\lambda) (\lambda^2 - P(A))^{-1} f \, d\lambda, \quad f \in E, \ \varphi \in \mathcal{D}^{(M_p)}.$$

Then \mathbf{G}_P is an exponential pre-(EUDCF) of (M_p) -class generated by P(A), and it is very difficult to answer whether the condition (CCF_2) holds for \mathbf{G}_P , in general.

(ii) In this part, we use the notation from [27, Chapter 8]. Let $p \in [1, \infty)$, $m > 0, \rho \in [0, 1], r > 0$, and let $a \in S^m_{\rho,0}$ satisfies (H_r) . Suppose that $E = L^p(\mathbb{R}^n)$ or $E = C_0(\mathbb{R}^n)$ (in the last case, we assume $p = \infty$), $0 \leq l \leq n$, $A := \operatorname{Op}_E(a)$ and the following inequality

$$n\Big|\frac{1}{2} - \frac{1}{p}\Big|\Big(\frac{m-r-\rho+1}{r}\Big) < 1$$

holds. Let us recall that if $a(\cdot)$ is an elliptic polynomial of order m, then (344) holds with m = r and $\rho = 1$. Suppose that there exists a sequence (M_p) satisfying (M.1), (M.2) and (M.3)', as well as that $a(\mathbb{R}^n) \cap \Lambda^2_{l,\zeta,\eta} = \emptyset$ for some constants $l \ge 1, \zeta > 0$ and $\eta \in \mathbb{R}$. Here,

$$\Lambda_{l,\zeta,\eta} = \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda \geqslant \zeta M(l | \operatorname{Im} \lambda|) + \eta\} \text{ and } \Lambda^2_{l,\zeta,\eta} = \{\lambda^2 : \lambda \in \Lambda_{l,\zeta,\eta}\}.$$

Put $\mathbb{N}_0^l := \{\eta \in \mathbb{N}_0^n : \eta_{l+1} = \cdots = \eta_n = 0\}$ and $E_l := \{f \in E : f^{(\eta)} \in E \text{ for all } \eta \in \mathbb{N}_0^l\}$. Then the calibration $(q_\eta(f) := \|f^{(\eta)}\|_E, f \in E_l; \eta \in \mathbb{N}_0^l)$ induces a Fréchet topology on E_l [**541**]. Define the operator A_l on E_l

by $D(A_l) := \{f \in E_l : \operatorname{Op}_E(a) f \in E_l\}$ and $A_l f := \operatorname{Op}_E(a) f \ (f \in D(A_l))$. Then we know that there exist numbers $\eta' \ge \eta$, $N \in \mathbb{N}$ and $M \ge 1$ such that $\Lambda^2_{l \in \eta'} \subseteq \rho(A_l)$ and for each $\eta \in \mathbb{N}^l_0$ we have

$$q_{\eta}(R(\lambda:A_l)f) \leqslant M(1+|\lambda|)^N q_{\eta}(f), \quad \lambda \in \Lambda^2_{l,\zeta,\eta'}, \ f \in E_l.$$

Keeping in mind Theorem 3.4.10 and Remark 3.4.11, we get that A_l generates an ultradistribution cosine function of (M_p) -class in E_l . This implies (see e.g. [291, Theorem 3.6.14] for the Banach space case) that the abstract Cauchy problem

$$(ACP_2): \begin{cases} u \in C^{\infty}([0,\infty):E_l) \cap C([0,\infty):[D(A_l)]), \\ u_{tt}(t,x) = A_l u(t,x), \quad t \ge 0, \ x \in \mathbb{R}^n, \\ u(0,x) = u_0(x), \quad u_t(0,x) = u_1(x), \ x \in \mathbb{R}^n \end{cases}$$

has a unique solution for any $u_0, u_1 \in E^{(M_p)}(A_l)$, where $E^{(M_p)}(A_l)$ is the abstract Beurling space of operator A_l ; furthermore, for every compact set $K \subseteq [0, \infty)$ and for every $n \in \mathbb{N}$ and h > 0, the solution u of (ACP_2) satisfies

$$\sup_{t\in K,\ p\in\mathbb{N}_0}\frac{h^p}{M_p}\Big(\Big\|\frac{d^p}{dt^p}u(t)\Big\|_n+\Big\|\frac{d^{p+1}}{dt^{p+1}}u(t)\Big\|_n\Big)<\infty.$$

Multiplication operators in L^p -spaces generating degenerate locally integrated cosine functions can be simply constructed following the method proposed in Example 3.2.11 and [291, Example 3.4.44]. These examples can serve as examples of non-exponential pre-(DCF)'s in Banach spaces. If the condition (PW) clarified below holds, then it can be simply proved that there exists a continuous linear operator C such that \mathcal{A} generates a global once integrated C-cosine function that is not exponentially bounded, in general. This example can be used for construction of non-exponential pre-(C-DCF)'s in Banach spaces.

3.5. Subordinated fractional resolvent families with removable singularities at zero

In this section, we assume that $(E, \|\cdot\|)$ is a complex Banach space and the following condition holds (cf. also [199, (P), p. 47] with $\alpha = 1$):

(PW) There exist finite constants c, M > 0 and $\beta \in (0, 1]$ such that

$$\Psi := \Psi_c := \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda \ge -c(|\operatorname{Im} \lambda| + 1)\} \subseteq \rho(\mathcal{A})$$

and

$$||R(\lambda : \mathcal{A})|| \leq M(1+|\lambda|)^{-\beta}, \quad \lambda \in \Psi.$$

Our intention is to analyze the fractional resolvent families with removable singularities at zero, which are subordinated to infinitely differentiable semigroups generated by the multivalued linear operators satisfying the condition (PW). We will prove proper extensions of [**529**, Theorem 3.3(ii), Theorem 3.4(iii), Theorem 4.1], given in single-valued case. Define the Yosida approximation $A_n \in L(E)$ of the operator \mathcal{A} by $A_n := n[nR(n : \mathcal{A}) - I]$ $(n \in \mathbb{N})$. Then there exist numbers $c' \in (0, c)$, M' > 0 and $n_0 \in \mathbb{N}$ such that $\Psi_{c'} \subseteq \rho(A_n)$ for $n \ge n_0$, as well as

$$||R(\lambda:A_n)|| \leqslant M'(1+|\lambda|)^{-\beta}, \quad \lambda \in \Psi_{c'}, \ n \ge n_0.$$

Unfortunately, this fact cannot be essentially employed in the analysis of abstract degenerate fractional differential equations involving the multivalued linear operators satisfying the condition (PW) or condition (QP), and a great number of results presented in [199, Section 3.2-Section 3.5] is not attainable in our framework.

Before proceeding any further, we need to slightly modify the definition of Caputo fractional derivatives of order $\gamma \in (0,1)$. In this section, we will use the following notion: Let $0 < T < \infty$. The Caputo fractional derivative $\mathbf{D}_t^{\gamma} u(t)$ (we will not change the terminology we have used so far) is defined for those functions $u: [0,T] \to E$ for which $u_{|(0,T]}(\cdot) \in C((0,T]:E), u(\cdot) - u(0) \in L^1((0,T):E)$ and $g_{1-\gamma} * (u(\cdot) - u(0)) \in W^{1,1}((0,T):E)$, by

$$\mathbf{D}_{t}^{\gamma}u(t) = \frac{d}{dt}[g_{1-\gamma} * (u(\cdot) - u(0))](t), \quad t \in (0,T].$$

Let the contour $\Gamma := \{\lambda = -c(|\eta| + 1) + i\eta : \eta \in \mathbb{R}\}$ be oriented so that $\operatorname{Im} \lambda$ increases along Γ . Set T(0) := I and

$$T(t)x := \frac{1}{2\pi i} \int_{\Gamma} e^{t\lambda} (\lambda - \mathcal{A})^{-1} x \, d\lambda, \quad t > 0, \ x \in E.$$

Then $(T(t))_{t\geq 0} \subseteq L(E)$ is a semigroup on E, and we have the following estimate

(345)
$$||T(t)|| = O(t^{\beta-1}), \quad t > 0;$$

furthermore, for every $\theta \in (0, 1)$,

(346)
$$||T(t)||_{L(E,E^{\theta}_{\mathcal{A}})} = O(t^{\beta-\theta-1}), \quad t > 0.$$

Concerning the strong continuity of $(T(t))_{t\geq 0}$ at zero, it is necessary to remind ourselves of the fact that [199]:

(CW) $T(t)x \to x, t \to 0+$ for any $x \in E$ belonging to the space $D((-\mathcal{A})^{\theta})$ with $\theta > 1 - \beta \ (x \in E^{\theta}_{\mathcal{A}} \text{ with } 1 > \theta > 1 - \beta).$

It is very simple to prove that

(347)
$$R(\lambda : \mathcal{A})x = \int_0^\infty e^{-\lambda t} T(t)x \, dt, \quad \operatorname{Re} \lambda > 0, \ x \in E.$$

From now on, we assume that $0 < \gamma < 1$. Set, for every $\nu \in (-\beta, \infty)$,

(348)
$$T_{\gamma,\nu}(t)x := t^{-\gamma} \int_0^\infty s^{\nu} \Phi_{\gamma}(st^{-\gamma})T(s)x \, ds, \quad t > 0, \ x \in E \text{ and } T_{\gamma,0}(0) := I.$$

Since

(349)
$$T_{\gamma,\nu}(t)x = t^{\gamma\nu} \int_0^\infty s^\nu \Phi_\gamma(s) T(st^\gamma) x \, ds, \quad t > 0, \ x \in E,$$

the estimates (345)-(346) combined with (a3) imply that the integral which defines the operator $T_{\gamma,\nu}(t)$ is absolutely convergent as well as

$$||T_{\gamma,\nu}(t)|| = O(t^{\gamma(\nu+\beta-1)}), \quad t > 0$$

Moreover, for every $\theta \in (0, 1)$ and $\nu > \theta - \beta$.

$$|T_{\gamma,\nu}(t)||_{L(E,E_{A}^{\theta})} = O(t^{\gamma(\nu+\beta-\theta-1)}), \quad t > 0.$$

Further on, (349) taken together with (a3) implies that, for every $\nu > -\beta$,

(350)
$$\frac{T_{\gamma,\nu}(t)}{t^{\gamma\nu}}x - \frac{\Gamma(1+\nu)}{\Gamma(1+\gamma\nu)}x = \int_0^\infty s^\nu \Phi_\gamma(s)[T(st^\gamma)x - x]ds, \quad t > 0, \ x \in E.$$

Using the dominated convergence theorem, (a3), (345), (350) and (CW), we can deduce the following:

(b1)
$$\frac{T_{\gamma,\nu}(t)}{t^{\gamma\nu}}x \to \frac{\Gamma(1+\nu)}{\Gamma(1+\gamma\nu)}x, t \to 0+$$
 provided that $\theta > 1-\beta$ and $x \in D((-\mathcal{A})^{\theta}),$
or that $1 > \theta > 1-\beta$ and $x \in E^{\theta}_{\mathcal{A}}$ $(\nu > -\beta).$

Taking into account the proof of [61, Theorem 3.1] and (347), we get

(b2) $\int_0^\infty e^{-\lambda t} T_{\gamma,0}(t) x \, dt = \lambda^{\gamma-1} \int_0^\infty e^{-\lambda^{\gamma} t} T(t) x \, dt = \lambda^{\gamma-1} (\lambda^{\gamma} - \mathcal{A})^{-1} x, \operatorname{Re} \lambda > 0, x \in E.$

Owing to [199, Theorem 3.5], (a3) and (350), we have

(b3) $\left\|\frac{T_{\gamma,\nu}(t)}{t^{\gamma\nu}}x - \frac{\Gamma(1+\nu)}{\Gamma(1+\gamma\nu)}x\right\| = O(t^{\gamma(\beta+\theta-1)}\|x\|_{[D((-\mathcal{A})^{\theta})]}), t > 0$, provided 1 > $\begin{aligned} \theta > 1 - \beta, x \in D((-\mathcal{A})^{\theta}) \text{ and } \| \frac{T_{\gamma,\nu}(t)}{t^{\gamma\nu}} x - \frac{\Gamma(1+\nu)}{\Gamma(1+\gamma\nu)} x \| &= O(t^{\gamma(\beta+\theta-1)} \|x\|_{E^{\theta}_{\mathcal{A}}}), \\ t > 0, \text{ provided } 1 > \theta > 1 - \beta, x \in E^{\theta}_{\mathcal{A}} \ (\nu > -\beta). \end{aligned}$

Set $\xi := \min((1/\gamma - 1)\pi/2, \pi)$. It is worth noting that the proof of [61, Theorem 3.3(i)–(ii)] implies that, for every $\nu > -\beta$, the mapping $t \mapsto T_{\gamma,\nu}(t)x, t > 0$ can be analytically extended to the sector Σ_{ξ} (we will denote this extension by the same symbol) and that, for every $\theta \in (0, 1)$, $\varepsilon \in (0, \xi)$ and $\nu > -\beta$,

(b4) $||T_{\gamma,\nu}(z)|| = O(|z|^{\gamma(\nu+\beta-1)}), z \in \Sigma_{\mathcal{E}-\varepsilon}.$

Moreover, for every $\theta \in (0, 1)$, $\varepsilon \in (0, \xi)$ and $\nu > \theta - \beta$,

(b5)
$$||T_{\gamma,\nu}(z)||_{L(E,E^{\theta}_{\mathcal{A}})} = O(|z|^{\gamma(\nu+\beta-1-\theta)}), z \in \Sigma_{\xi-\varepsilon}.$$

Keeping in mind (b4)–(b5) and the Cauchy integral formula, we can prove that, for every $\theta \in (0, 1)$, $\varepsilon \in (0, \xi)$, $\nu > -\beta$ and $n \in \mathbb{N}$,

(b4)' $||(d^n/dz^n)T_{\gamma,\nu}(z)|| = O(|z|^{\gamma(\nu+\beta-1)-n}), z \in \Sigma_{\mathcal{E}-\mathcal{E}},$

as well as that, for every $\theta \in (0,1)$, $\varepsilon \in (0,\xi)$, $\nu > \theta - \beta$ and $n \in \mathbb{N}$, (b5)' $||(d^n/dz^n)T_{\nu\nu}(z)||_{z=-\infty} = O(|z|\gamma(\nu+\beta-1-\theta)-n) = c \Sigma$

$$(b5)^{\gamma} \| (d^n/dz^n) T_{\gamma,\nu}(z) \|_{L(E,E^{\theta}_{\mathcal{A}})} = O(|z|^{\gamma(\nu+\beta-1-\theta)-n}), \ z \in \Sigma_{\xi-\varepsilon}$$

In the case that $\varepsilon \in (0,\xi)$ and $z \in \Sigma_{\xi-\varepsilon}$, then the uniqueness theorem for analytic functions, (a3) and the asymptotic expansion formula for the Wright functions (cf. also the first part of proof of [61, Theorem 3.3]) together imply that $\int_0^\infty z^{-\gamma(1+\nu)} s^{\nu} \Phi_{\gamma}(sz^{-\gamma}) ds = \frac{\Gamma(1+r)}{\Gamma(1+\gamma r)}, r > -1; \text{ hence},$

$$\frac{T_{\gamma,\nu}(z)}{z^{\gamma\nu}}x - \frac{\Gamma(1+\nu)}{\Gamma(1+\gamma\nu)}x = \int_0^\infty s^\nu \Phi_\gamma(se^{i\varphi})[T(s|z|^\gamma)x - x]ds,$$

where $\varphi = -\gamma \arg(z)$. Keeping in mind this identity, (C), [199, Theorem 3.5] and the proof of [61, Theorem 3.3], we can deduce the following extension of [529, Theorem 3.4(i)] and the properties (b1), (b3):

- (b1)' Suppose that $\varepsilon \in (0,\xi)$ and $\delta = \xi \varepsilon$. Then $\lim_{z \to 0, z \in \Sigma_{\delta}} \frac{T_{\gamma,\nu}(z)}{z^{\gamma\nu}} x = \frac{\Gamma(1+\nu)}{\Gamma(1+\gamma\nu)} x$, provided that $\theta > 1-\beta$ and $x \in D((-\mathcal{A})^{\theta})$, or that $1 > \theta > 1-\beta$ and $x \in E_{\mathcal{A}}^{\theta}$ $(\nu > -\beta)$.
- (b3)' Suppose that $\varepsilon \in (0,\xi)$ and $\delta = \xi \varepsilon$. Then $\left\|\frac{T_{\gamma,\nu}(z)}{z^{\gamma\nu}}x \frac{\Gamma(1+\nu)}{\Gamma(1+\gamma\nu)}x\right\| = O(|z|^{\gamma(\beta+\theta-1)}\|x\|_{[D((-\mathcal{A})^{\theta})]}), z \in \Sigma_{\delta}$, provided $1 > \theta > 1 \beta, x \in D((-\mathcal{A})^{\theta})$, and $\left\|\frac{T_{\gamma,\nu}(z)}{z^{\gamma\nu}}x \frac{\Gamma(1+\nu)}{\Gamma(1+\gamma\nu)}x\right\| = O(|z|^{\gamma(\beta+\theta-1)}\|x\|_{E^{\theta}_{\mathcal{A}}}), z \in \Sigma_{\delta}$, provided $1 > \theta > 1 \beta, x \in D((-\mathcal{A})^{\theta})$.

REMARK 3.5.1. As already observed, the angle of analyticity of considered operator families can be increased depending on the concrete value of constant c > 0 from the condition (PW). Here we will not discuss this question in more detail.

Following E. Bazhlekova [61] and R.-N. Wang, D.-H. Chen, T.-J. Xiao [529], we define

$$S_{\gamma}(z) := T_{\gamma,0}(z) \text{ and } P_{\gamma}(z) := \gamma T_{\gamma,1}(z)/z^{\gamma}, \quad z \in \Sigma_{\xi};$$

cf. the proof of [529, Theorem 3.1], where the corresponding operators have been denoted by $S_{\gamma}(z)$ and $\mathcal{P}_{\gamma}(z)$. The analysis contained in the proof of property (b4) enables one to see that the estimate [529, (3.1)] holds on closed subsectors of $\Sigma_{\pi/2-\omega}$ (cf. the formulation of [529, Theorem 3.1]). It is clear that the first statement of [529, Theorem 3.2] holds since the operators $S_{\gamma}(z)$ and $\mathcal{P}_{\gamma}(z)$ depend analytically, in the uniform operator topology, on the complex parameter z belonging to an appropriate sector containing $(0, \infty)$. Furthermore, due to (b4)'-(b5)', we have that for each $\varepsilon \in (0,\xi)$ the following holds: $||(d/dz)T_{\gamma,\nu}(z)|| = O(|z|^{\gamma(\nu+\beta-1)-1}),$ $z \in \Sigma_{\xi-\varepsilon}$ and $||(d/dz)P_{\gamma}(z)|| = O(|z|^{\gamma(\nu+\beta-1)-1}), z \in \Sigma_{\xi-\varepsilon}$. By the Darboux inequality, it readily follows that, for every R > 0, the mappings $z \mapsto S_{\gamma}(z) \in L(E)$, $z \in \Sigma_{\xi-\varepsilon} \smallsetminus B_R$ and $z \mapsto P_{\gamma}(z) \in L(E), z \in \Sigma_{\xi-\varepsilon} \smallsetminus B_R$ are uniformly continuous. Arguing in such a way, we have proved an extension of the second statement in [529, Theorem 3.2] for degenerate fractional differential equations.

It is clear that $T_{\gamma,\nu}(z)x = z^{-\gamma} \int_0^\infty s^{\nu} \Phi_{\gamma}(sz^{-\gamma})T(s)x\,ds, \ z \in \Sigma_{\xi}, \ x \in E$ and $s \mapsto \frac{1}{2\pi i} \int_{\Gamma} (-\lambda)^{\theta} e^{s\lambda} (\lambda - \mathcal{A})^{-1} \cdot d\lambda$ is a bounded linear section of the operator $(-\mathcal{A})^{\theta}T(s)$ for $\theta > 1 - \beta$ and $s \ge 0$ (cf. [199, Proposition 3.2, pp. 48–49]). Along with Theorem 1.2.3, the above implies

(351)
$$P_{\gamma,\theta}(z)x := \frac{\gamma z^{-2\gamma}}{2\pi i} \int_0^\infty s\Phi_\gamma(sz^{-\gamma}) \bigg[\int_\Gamma (-\lambda)^\theta e^{s\lambda} (\lambda - \mathcal{A})^{-1} x \, d\lambda \bigg] ds \in (-\mathcal{A})^\theta P_\gamma(z) x$$

for all $z \in \Sigma_{\xi}$ and $x \in E$, as well as that $(P_{\gamma,\theta}(z))_{z \in \Sigma_{\xi}} \subseteq L(E)$ for $\theta > 1 - \beta$. By the foregoing, we have

(352)
$$||P_{\gamma,\theta}(z)|| = O(|z|^{\gamma(\beta-\theta-1)}), \quad z \in \Sigma_{\xi-\varepsilon} \quad (\varepsilon \in (0,\xi)).$$

Differentiating (349), it is not difficult to prove that

(353)
$$\frac{d}{dz}S_{\gamma}(z)x = \frac{\gamma z^{-\gamma-1}}{2\pi i} \int_0^\infty s\Phi_{\gamma}(sz^{-\gamma})T'(s)x\,ds = -z^{\gamma-1}P_{\gamma,1}(z)x, \quad z \in \Sigma_{\xi}, \ x \in E.$$

Applying (351) with $\theta = 1$, and (353), we get that:

(354)
$$\frac{d}{dz}S_{\gamma}(z)x = -z^{\gamma-1}P_{\gamma,1}(z)x \in z^{\gamma-1}\mathcal{A}P_{\gamma}(z)x, \quad z \in \Sigma_{\xi}, \ x \in E.$$

Further on, Theorem 1.2.3 and Theorem 1.2.4(i) can serve one to prove that the assumption $(x, y) \in \mathcal{A}$ implies $S_{\gamma}(z)y \in \mathcal{A}S_{\gamma}(z)x$ and $P_{\gamma}(z)y \in \mathcal{A}P_{\gamma}(z)x$ $(z \in \Sigma_{\xi})$, so that the mapping $t \mapsto \frac{d}{dt}S_{\gamma}(t)x, t > 0$ is locally integrable for any $x \in D(\mathcal{A})$ by (354). Keeping in mind (b4), we have proved a proper extension of [529, Theorem 3.3] to degenerate fractional differential equations. Before proceeding further, we would like to point out that, for every $x \in D((-\mathcal{A})^{\theta}) \cap E_{\mathcal{A}}^{\theta}$, the mapping $z \mapsto \frac{d}{dz}S_{\gamma}(z)x = \frac{d}{dz}[S_{\gamma}(z)x - x]$ is bounded by $|z|^{\gamma(\beta+\theta-1)-1}$ on subsectors of Σ_{ξ} ($1 > \theta > 1 - \beta$); this follows from the Cauchy integral formula and the property (b3)' with $\nu = 0$. In particular, the mapping $t \mapsto \frac{d}{dt}S_{\gamma}(t)x, t > 0$ is locally integrable for any $x \in D((-\mathcal{A})^{\theta}) \cap E_{\mathcal{A}}^{\theta}$, where $1 > \theta > 1 - \beta$. Suppose that $(x, y) \in \mathcal{A}$. Then an elementary application of Cauchy formula, combined with Theorem 1.2.4(i) and definition of $T(\cdot)$, implies that T(s)y = T'(s)x, s > 0. Having in mind (348) with $\nu = 1$, and definition of $P_{\gamma}(\cdot)$, it readily follows that $P_{\gamma}(z)y = -P_{\gamma,1}(z)x, z \in \Sigma_{\xi}$; therefore, $\frac{d}{dz}S_{\gamma}(z)x = z^{\gamma-1}P_{\gamma}(z)y$, provided $z \in \Sigma_{\xi}$ and $(x, y) \in \mathcal{A}$. Under such a circumstance, we obtain after integration that $S_{\gamma}(z)x - x = \int_{0}^{z} \lambda^{\gamma-1}P_{\gamma}(\lambda)y d\lambda$.

Suppose again that $(x, y) \in \mathcal{A}$. Performing the Laplace transform, we obtain with the help of (b2) and Theorem 1.2.4(i) that $(g_{1-\gamma} * [S_{\gamma}(\cdot)x - x])(t) = \int_{0}^{t} S_{\gamma}(s)y \, ds, t \ge 0$. This immediately implies that $\mathbf{D}_{t}^{\gamma}S_{\gamma}(t)x = S_{\gamma}(t)y \in \mathcal{A}S_{\gamma}(t)x, t > 0$, which extends the assertion of [**529**, Theorem 3.4(iii)]. The original proof of this result, much more complicated than ours, is based on the use of functional calculus for almost sectorial operators established by F. Periago and B. Straub in [**447**] (as announced earlier, it is very difficult to develop a similar calculus for almost sectorial multivalued linear operators). Furthermore, we want to observe that this result is not optimal. In actual fact, let $1 > \theta > 1 - \beta$ and let $x \in D((-\mathcal{A})^{\theta}) \cap E_{\mathcal{A}}^{\theta}$ be fixed. Then the mapping $t \mapsto F(t) := (g_{1-\gamma} * [S_{\gamma}(\cdot)x - x])(t), t \ge 0$ is continuous and its restriction on $(0, \infty)$ can be analytically extended to the sector Σ_{ξ} , with the estimate $||F(z)|| = O(|z|^{\gamma(\beta+\theta-2)+1})$ on any proper subsector of Σ_{ξ} (cf. (b3)'). By the Cauchy integral formula, we obtain that $||F'(z)|| = O(|z|^{\gamma(\beta+\theta-2)})$ on proper subsectors of Σ_{ξ} . In particular, the Caputo fractional derivative $\mathbf{D}_{t}^{\gamma}S_{\gamma}(t)x$ is defined. On the other hand, Theorem 1.2.3 in combination with [199, Proposition 3.2, 3.4] implies that

$$t \mapsto F_{\gamma}(t)x := \frac{1}{2\pi i} \int_0^\infty t^{-\gamma} \Phi_{\gamma}(st^{-\gamma}) \bigg[\int_{\Gamma} \lambda e^{\lambda s} R(\lambda : \mathcal{A}) x \, d\lambda \bigg] ds, \quad t > 0$$

is a continuous section of the multivalued mapping $\mathcal{A}S_{\gamma}(t)x, t > 0$, with the meaning clear. Then $T(t)x - x = \int_0^t T'(s)x \, ds, t \ge 0$ and $T'(t)x = \frac{1}{2\pi i} \int_{\Gamma} \lambda e^{\lambda t} R(\lambda : t) \, ds$

 \mathcal{A})x d λ , t > 0, which simply implies by (347) that

(355)
$$\int_0^\infty e^{-zt} T'(t) x \, dt = zR(z:\mathcal{A})x - x$$
$$= \int_0^\infty e^{-zt} \left[\frac{1}{2\pi i} \int_\Gamma \lambda e^{\lambda t} R(\lambda:\mathcal{A}) x \, d\lambda \right] dz, \quad z > 0.$$

Using Fubini theorem, definition of $F_{\gamma}(\cdot)$ and the identity [61, (3.10)], we get that

$$\int_0^\infty e^{-zt} F_{\gamma}(t) x \, dz = z^{\gamma-1} \int_0^\infty e^{-z^{\gamma}t} \left[\frac{1}{2\pi i} \int_{\Gamma} \lambda e^{\lambda t} R(\lambda : \mathcal{A}) x \, d\lambda \right] dz, \quad z > 0,$$

which clearly implies with the help of (355) that:

$$\int_0^\infty e^{-zt} \int_0^t F_{\gamma}(s) x \, ds \, dz = z^{\gamma-2} \int_0^\infty e^{-z^{\gamma}t} \left[\frac{1}{2\pi i} \int_{\Gamma} \lambda e^{\lambda t} R(\lambda : \mathcal{A}) x \, d\lambda \right] dz$$
$$= z^{\gamma-2} [z^{\gamma} R(z^{\gamma} : \mathcal{A}) x - x], \quad z > 0.$$

Using this equation, (b2) and the uniqueness theorem for Laplace transform, it readily follows that

$$(g_{1-\gamma} * [S_{\gamma}(\cdot)x - x])(t) = \int_0^t F_{\gamma}(s)x \, ds, \quad t \ge 0.$$

Now it becomes clear that:

(356) $\mathbf{D}_t^{\gamma} S_{\gamma}(t) x = F_{\gamma}(t) x \in \mathcal{A}S_{\gamma}(t) x, \quad t > 0, \ x \in D((-\mathcal{A})^{\theta}) \cap E_{\mathcal{A}}^{\theta} \quad (1 > \theta > 1 - \beta).$

The identity [61, (3.10)] almost immediately implies that

$$\int_0^\infty e^{-\lambda t} t^{-\gamma - 1} \Phi_\gamma(st^{-\gamma}) dt = \frac{1}{\gamma s} e^{-\lambda^\gamma s}, \quad s > 0, \ \lambda > 0.$$

Keeping in mind this equality and (347), we get that

$$\int_0^\infty \int_0^\infty \gamma s T(s) [e^{-\lambda t} t^{-\gamma - 1} \Phi_\gamma(st^{-\gamma}) dt] x \, ds = (\lambda^\gamma - \mathcal{A})^{-1} x, \quad \lambda > 0, \ x \in E.$$

Using Fubini theorem and definition of $T_{\gamma,1}(\cdot)$, the above yields

$$\int_0^\infty e^{-\lambda t} t^{\gamma-1} P_{\gamma}(t) x \, dt = (\lambda^\gamma - \mathcal{A})^{-1} x, \quad \lambda > 0, \ x \in E.$$

By (b2) and the uniqueness theorem for Laplace transform, we obtain finally the following generalization of [529, Theorem 3.4(iv)]:

(357)
$$S_{\gamma}(t)x = (g_{1-\gamma} * [\cdot^{\gamma-1} P_{\gamma}(\cdot)x])(t), \quad t > 0, \ x \in E.$$

This identity continues to hold on sector Σ_{ξ} . Arguing as in non-degenerate case (cf. [529, Lemma 3.1, Theorem 3.5]), we can prove that the compactness of $R(\lambda : \mathcal{A})$ for some $\lambda \in \rho(\mathcal{A})$ implies the compactness of operators $S_{\gamma}(t)$ and $P_{\gamma}(t)$ for all t > 0.

The consideration from [529, Lemma 4.1] is completely meaningful for abstract degenerate fractional differential equations and gives rise us to introduce the following definition (cf. [529, Definition 4.1, Definition 4.2] and compare to [199, Definition, p. 53):

DEFINITION 3.5.2. Let $T \in (0, \infty)$ and $f \in L^1((0, T) : E)$. Consider the following abstract degenerate fractional inclusion:

$$\left(\mathrm{DFP}\right)_{f}: \begin{cases} \mathbf{D}_{t}^{\gamma}u(t) \in \mathcal{A}u(t) + f(t), & t \in (0,T], \\ u(0) = u_{0}. \end{cases}$$

(i) By a mild solution of $(DFP)_f$, we mean a function

$$u(t) = S_{\gamma}(t)u_0 + \int_0^t (t-s)^{\gamma-1} P_{\gamma}(t-s)f(s)ds, \quad t \in (0,T].$$

(ii) By a classical solution of $(DFP)_f$, we mean any function $u \in C([0,T]:E)$ satisfying that the function $\mathbf{D}_t^{\gamma} u(t)$ is well-defined and belongs to the space C((0,T]:E), as well as that $u(0) = u_0$ and $\mathbf{D}_t^{\gamma} u(t) - f(t) \in \mathcal{A}u(t)$ for $t \in (0,T]$.

A mild solution u(t) of problem $(DFP)_f$ is automatically continuous on (0, T]. If $x \in D((-\mathcal{A})^{\theta}) \cap E^{\theta}_{\mathcal{A}}$, where $1 > \theta > 1 - \beta$, then (356) implies that the mapping $u(t) = S_{\gamma}(t)x$ is a classical solution of $(DFP)_f$, with $f \equiv 0$. The following theorem is an important extension of [**529**, Theorem 4.1], even for non-degenerate fractional differential equations with almost sectorial operators.

THEOREM 3.5.3. Suppose that $T \in (0, \infty)$, $1 \ge \theta > 1 - \beta$ and $x \in D((-\mathcal{A})^{\theta})$, resp. $1 > \theta > 1 - \beta$ and $x \in E^{\theta}_{\mathcal{A}}$, as well as that there exist constants $\sigma > \gamma(1 - \beta)$ and $M \ge 1$ such that

(358)
$$||f(t) - f(s)|| \leq M|t - s|^{\sigma}, \quad 0 < t, \ s \leq T.$$

Let $1 \ge \theta > 1 - \beta$, resp. $1 > \theta > 1 - \beta$, and let

$$f \in L^{\infty}((0,T) : [D((-\mathcal{A})^{\theta})]), \text{ resp. } f \in L^{\infty}((0,T) : E^{\theta}_{\mathcal{A}}).$$

Then there exists a unique classical solution of problem $(DFP)_{f}$.

PROOF. We will prove the theorem only in the case that $1 > \theta > 1 - \beta$ and $x \in D((-\mathcal{A})^{\theta})$ (cf. also Theorem 3.5.12 below). The uniqueness of classical solutions of problem $(DFP)_f$ is an immediate consequence of Theorem 3.1.6. By the foregoing arguments, it suffices to show that the function

$$\omega(t) := \int_0^t (t-s)^{\gamma-1} P_{\gamma}(t-s) f(s) ds, \quad 0 \leqslant t \leqslant T,$$

enjoys the following properties:

- (i) $\omega(t)$ is continuous at the point t = 0;
- (ii) $\mathbf{D}_t^{\gamma}\omega(t) = \omega_1(t) := \int_0^t S'_{\gamma}(t-s)f(s)ds + f(t), \ 0 < t \leq T \text{ and } \omega_1(t) \text{ is continuous on } (0,T];$
- (iii) $\omega_2(t) := \omega_1(t) f(t) = \int_0^t S'_{\gamma}(t-s)f(s)ds \in \mathcal{A}\omega(t), \ 0 < t \leq T.$

The statement (i) follows from the Hölder continuity of $f(\cdot)$ (cf. (358)) and a simple computation involving the estimate $||P_{\gamma}(t)|| = O(t^{\gamma(\beta-1)}), t > 0$. For the proof of (ii), observe that (353) in combination with contour representation of $T'(\cdot)$

and [199, Proposition 3.2, Proposition 3.4] implies that there exist constants $C_{\theta} > 0$ and $C'_{\theta} > 0$ such that, for every $0 < s \leq T$ and $0 < \omega \leq T$,

$$(359) \|S_{\gamma}'(\omega)f(s)\| = \left\|\frac{\gamma}{2\pi i}\int_{0}^{\infty} v\omega^{\gamma-1}\Phi_{\gamma}(v)T'(v\omega^{\gamma})f(s)dv\right\|$$
$$\leq C_{\theta}\frac{\gamma}{2\pi}\|f(s)\|_{[D((-\mathcal{A})^{\theta})]}\int_{0}^{\infty} v\omega^{\gamma-1}\Phi_{\gamma}(v)(v\omega^{\gamma})^{\beta+\theta-2}dv$$
$$= C_{\theta}\frac{\gamma}{2\pi}\|f(s)\|_{[D((-\mathcal{A})^{\theta})]}\omega^{\gamma(\beta+\theta-1)-1}\int_{0}^{\infty} v^{\beta+\theta-1}\Phi_{\gamma}(v)dv$$
$$= C_{\theta}'\|f(s)\|_{[D((-\mathcal{A})^{\theta})]}\omega^{\gamma(\beta+\theta-1)-1}.$$

Using this estimate with $\omega = t - s$, where $0 < s < t \leq T$, and integrating the obtained estimate along the interval [0,T] in variable s, we get that there exists a constant $C_{\theta}^{\prime\prime} > 0$ such that, for every $0 < t \leq T$,

$$(360) \quad \left\| \int_0^t S_{\gamma}'(t-s)f(s)ds \right\| \leqslant C_{\theta}'' \int_0^t (t-s)^{\gamma(\beta+\theta-1)-1} \|f(s)\|_{[D((-\mathcal{A})^{\theta})]}ds$$
$$\leqslant \frac{C_{\theta}''}{\gamma(\beta+\theta-1)} t^{\gamma(\beta+\theta-1)} \|f(\cdot)\|_{L^{\infty}((0,T):[D((-\mathcal{A})^{\theta})])}.$$

Let h > 0 and let $h \leq T - t$ for some fixed time $t \in (0, T)$. Making use of (359) and dominated convergence theorem, we obtain that

(361)
$$\lim_{h \to 0+} \int_0^t \frac{S_{\gamma}(t+h-s) - S_{\gamma}(t-s)}{h} f(s) ds = \int_0^t S_{\gamma}'(t-s) f(s) ds$$

here it is only worth noting that (359) and the mean value theorem together imply that, for every $s \in (0, t)$,

$$\begin{split} \left\|\frac{S_{\gamma}(t+h-s)-S_{\gamma}(t-s)}{h}f(s)\right\| &\leqslant \frac{1}{h}\int_{t-s}^{t+h-s} \|S_{\gamma}'(r)f(s)\|dr\\ &\leqslant \frac{\|f(\cdot)\|_{L^{\infty}((0,T):[D((-\mathcal{A})^{\theta})])}}{h}\int_{t-s}^{t+h-s}r^{\gamma(\beta+\theta-1)-1}dr\\ &\leqslant \text{Const. }\left[(t-s)^{\gamma(\beta+\theta-1)-1}+(t+1-s)^{\gamma(\beta+\theta-1)-1}\right]. \end{split}$$

Having in mind the estimate $||S_{\gamma}(t)|| = O(t^{\gamma(\beta-1)}), t > 0$, the strong continuity of operator family $S_{\gamma}(\cdot)$ on $D((-\mathcal{A})^{\theta})$ and the Hölder continuity of $f(\cdot)$, we can repeat almost verbatim the arguments from the corresponding part of proof of [529, Theorem 4.1] in order to see that

(362)
$$\lim_{h \to 0+} \frac{1}{h} \int_{t}^{t+h} S_{\gamma}(t+h-s)f(s)ds = f(t).$$

Due to (361)–(362), we have that the mapping $t \mapsto \int_0^t S_{\gamma}(t-s)f(s)ds$, 0 < t < T is differentiable from the right; we can similarly prove the differentiability of this mapping from the left, for $0 < t \leq T$, so that

$$\frac{d}{dt} \int_0^t S_{\gamma}(t-s)f(s)ds = \int_0^t S_{\gamma}'(t-s)f(s)ds + f(t), \quad 0 < t \le T.$$

Now it is not difficult to prove with the help of (357) and (360) that $\mathbf{D}_t^{\gamma}\omega(t)$ exists and equals to $\omega_1(t)$, as claimed. Now we will prove that the mapping $t \mapsto \int_0^t S'_{\gamma}(t-s)f(s)ds$, $0 < t \leq T$ is continuous (observe here that this mapping is continuous for t = 0; cf. (360)). As in the proof of [529, Theorem 4.1], we have $\int_0^t S'_{\gamma}(t-s)f(s)ds = I_1(t) + I_2(t)$, where $I_1(t) := \int_0^t S'_{\gamma}(t-s)[f(s) - f(t)]ds$ and $I_2(t) := \int_0^t S'_{\gamma}(t-s)f(t)ds$. By (b3), we have that $I_2(t+h) \to I_2(t)$ as $h \to 0$, for $0 < t \leq T$ and the meaning clear. To complete the whole proof, it suffices to show that the mapping $I_1(t) := \int_0^t S'_{\gamma}(t-s)[f(s) - f(t)]ds$, $0 < t \leq T$ is continuous. For the sake of brevity, we will only prove that the above mapping is continuous from the right, for 0 < t < T. Suppose, as above, h > 0 and $h \leq T - t$. Then

$$I_1(t+h) - I_1(t) = h_1(t) + h_2(t) + h_3(t),$$

where

$$h_1(t) := \int_0^t (S'_{\gamma}(t+h-s) - S'_{\gamma}(t-s))[f(s) - f(t)]ds$$
$$h_2(t) := \int_0^t S'_{\gamma}(t+h-s)[f(t) - f(t+h)]ds$$

and

$$h_3(t) := \int_t^{t+h} S'_{\gamma}(t+h-s)[f(s) - f(t+h)]ds$$

We can prove that $h_1(t) \to 0$ as $h \to 0+$ by means of the dominated convergence theorem and the following estimates (cf. (359)–(360)):

$$\left\| \int_{0}^{t} S_{\gamma}'(t+h-s)[f(s)-f(t)]ds \right\| \leq \text{Const.} \ (t+h-s)^{\gamma(\beta+\theta-1)-1} \|f(s)-f(t)\|_{[D((-\mathcal{A})^{\theta})]} \leq \text{Const.} \ \|f(\cdot)\|_{L^{\infty}((0,T):[D((-\mathcal{A})^{\theta})])}[(t-s)^{\gamma(\beta+\theta-1)-1} + (t+1-s)^{\gamma(\beta+\theta-1)-1}]$$

and

$$\left\|\int_0^t S_{\gamma}'(t-s)[f(s)-f(t)]ds\right\| \leqslant \frac{2C_{\theta}''}{\gamma(\beta+\theta-1)}t^{\gamma(\beta+\theta-1)}\|f(\cdot)\|_{L^{\infty}((0,T):[D((-\mathcal{A})^{\theta})])} \cdot C_{\theta}(t-s)\|f(s)\|_{L^{\infty}((0,T):[D((-\mathcal{A})^{\theta})])} + C_{\theta}(t-s)\|f(s)\|_{L^{\infty}((0,T):[D((-\mathcal{A})^{\theta})]} + C_{\theta}(t-s)\|f($$

On the other hand, we may conclude that $h_2(t) \to 0$ as $h \to 0+$ by using the estimate $||S'_{\gamma}(t)|| = O(t^{\gamma(\beta-1)-1}), t > 0$, the Hölder continuity of $f(\cdot)$ and our standing assumption $\sigma > \gamma(1-\beta)$:

$$\|h_2(t)\| \leq \text{Const.} \int_0^t (t+h-s)^{\gamma(\beta-1)-1} h^\sigma ds$$
$$\leq \text{Const.} h^\sigma[(t+h)^{\gamma(\beta-1)} - h^{\gamma(\beta-1)}] \to 0 \text{ as } h \to 0 + .$$

Finally, an application of (359) yields:

$$\begin{aligned} \|h_3(t)\| &\leq \text{Const. } \int_t^{t+h} (t+h-s)^{\gamma(\beta+\theta-1)-1} \|f(s) - f(t+h)\|_{[D((-\mathcal{A})^\theta)]} ds \\ &\leq \text{Const. } \|f(\cdot)\|_{L^{\infty}((0,T):[D((-\mathcal{A})^\theta)])} h^{\gamma(\beta+\theta-1)} \to 0 \text{ as } h \to 0+. \end{aligned}$$

This proves (ii). The proof of (iii) follows by applying (354), (360) and Theorem 1.2.3. $\hfill \Box$

REMARK 3.5.4. It is clear that the validity of condition (358) implies that the sequence $(f_n(t))_{n \in \mathbb{N}} \subseteq C([0,T]:E)$, where $f_n(t) := f(t)$ for $t \in [1/n,T]$ and $f_n(t) := f(1/n)$ for $t \in [0, 1/n]$, is a Cauchy sequence in C([0,T]:E) and therefore convergent. Hence, there exists $\lim_{t\to 0+} f(t)$ in E and f(t) can be extended to a Hölder continuous function from the space $C^{\sigma}([0,T]:E)$. This implies that the Caputo fractional derivative $\mathbf{D}_t^{\gamma}\omega(t)$ (cf. (ii)) is defined in the strong sense and that (ii) holds, in fact, for $0 \leq t \leq T$.

It is clear that Theorem 3.5.3 can be applied in the analysis of a large class of relaxation degenerate differential equations that are subordinated to degenerate differential equations of first order considered in [199, Section 3.7]. For example, we can prove some new results on the following inhomogeneous fractional Poisson heat equation in the space $L^p(\Omega)$:

$$(P)_{\gamma}: \begin{cases} \mathbf{D}_{t}^{\gamma}[m(x)v(t,x)] = \Delta v - bv + f(t,x), & t \ge 0, \ x \in \Omega; \\ v(t,x) = 0, & (t,x) \in [0,\infty) \times \partial\Omega, \\ m(x)v(0,x) = u_{0}(x), & x \in \Omega, \end{cases}$$

where Ω is a bounded domain in \mathbb{R}^n , b > 0, $m(x) \ge 0$ a.e. $x \in \Omega$, $m \in L^{\infty}(\Omega)$, $1 and <math>0 < \gamma < 1$; cf. also Example 3.5.13(ii) below.

3.5.1. Semilinear degenerate Cauchy inclusions. Assume that the condition (PW) holds. In this section, we treat the following semilinear degenerate fractional Cauchy inclusion:

$$(\mathrm{DFP})_{f,s,\gamma} : \begin{cases} \mathbf{D}_t^{\gamma} u(t) \in \mathcal{A}u(t) + f(t,u(t)), & t \in (0,T], \\ u(0) = u_0, \end{cases}$$

where $T \in (0, \infty)$ and $0 < \gamma \leq 1$. Suppose first that $0 < \gamma < 1$. By a mild solution u(t) of problem $(DFP)_{f,s,\gamma}$, we mean any function $u \in C((0,T]:E)$ such that

$$u(t) = S_{\gamma}(t)u_0 + \int_0^t (t-s)^{\gamma-1} P_{\gamma}(t-s)f(s,u(s))ds, \quad t \in (0,T].$$

As in linear case, a classical solution of $(\text{DFP})_f$ is any function $u \in C([0,T] : E)$ satisfying that the function $\mathbf{D}_t^{\gamma}u(t)$ is well-defined and belongs to the space C((0,T] : E), as well as that $u(0) = u_0$ and $\mathbf{D}_t^{\gamma}u(t) - f(t,u(t)) \in \mathcal{A}u(t)$ for $t \in (0,T]$. In [529, Theorem 5.1, Theorem 5.3, Corollary 5.1], the authors have applied various types of fixed point theorems in the study of existence and uniqueness of mild solutions of problem $(\text{DFP})_{f,s,\gamma}$, provided that the operator \mathcal{A} is single-valued, linear and almost sectorial. In contrast to the assertions of [529, Theorem 5.2, Theorem 5.4], the above-mentioned results can be immediately extended to semilinear degenerate fractional Cauchy inclusion $(\text{DFP})_{f,s,\gamma}$. This is also the case with the assertion of [310, Theorem 2.1].

In [378, Theorem 3.1], F. Li has proved the existence of mild solutions for a class of delay semilinear fractional differential equations. The extension of this result to degenerate equations is immediate, as well. Following T. Dlotko [156], we can similarly define the notions of mild and classical solutions of semilinear degenerate Cauchy inclusion $(DFP)_{f,s,1}$ of first order $(\gamma = 1)$: any function $u \in C((0, T] : E)$ such that

$$u(t) = T(t)u_0 + \int_0^t T(t-s)f(s, u(s))ds, \quad t \in (0, T]$$

is said to be a mild solution of problem $(DFP)_{f,s,1}$. By a classical solution, we mean any function $u \in C([0,T] : E) \cap C^1((0,T] : E)$ such that $u(t) \in D(\mathcal{A})$, $t \in (0,T]$, $u(0) = u_0$ and $u'(t) \in \mathcal{A}u(t) + f(t,u(t))$, $t \in (0,T]$. The extensions of **[156**, Theorem 1, Proposition 2] for semilinear degenerate Cauchy inclusions of first order can be simply proved.

In the remainder of this section, we will reconsider the assertions of [446, Theorem 3.1, Theorem 3.2] for semilinear degenerate Cauchy inclusions; cf. [445, Theorem 1.4, p. 185] for probably the first result in this direction.

As the next two theorems show, Theorem 3.1 and Theorem 3.2 of [446] can be fully generalized to semilinear degenerate Cauchy inclusions of first order.

THEOREM 3.5.5. Let T > 0, and let $\gamma = 1$. Suppose that the mapping $f: [0,T] \times E \to E$ is continuous in t on [0,T] and for each $t_0 > 0$ and K > 0 there exists a constant $L(t_0, K) > 0$ such that $||f(t, x) - f(t, y)|| \leq L(t_0, K)||x - y||$, provided $0 < t < t_0, x, y \in E$ and $||x||, ||y|| \leq K$. Denote by Ω the domain of continuity of semigroup $(T(t))_{t \geq 0}$; that is, $\Omega = \{x \in E : \lim_{t \to 0^+} T(t)x = x\}$. Then, for every $u_0 \in \Omega$, there exist a number $\tau_{\max} = \tau_{\max}(u_0) > 0$ and a unique mild solution $u \in C([0, \tau_{\max}) : E)$ of problem (DFP)_{f,s,1}. If

- (i) $f(t, x) \in D(\mathcal{A})$ for all t > 0 and $x \in \Omega$;
- (ii) for each $t_0 > 0$ and K > 0 there exists a constant $C = C(t_0, K) > 0$ such that

(363)
$$||f(t,x)||_{[D(\mathcal{A})]} \leq C \text{ for all } x \in \Omega \text{ with } ||x|| \leq C \text{ and } 0 < t < t_0;$$

(iii) there exists a function $g \in L^{\infty}_{loc}((0,\infty):\mathbb{R})$ such that

 $||f(t,x)|| \leq g(t)||x||$ a.e. $t \geq 0$ and $x \in \Omega$,

then $\tau_{\max} = \infty$.

PROOF. The proof is almost the same as that of [446, Theorem 3.1]; here we only want to observe that the term

$$\int_0^t \|[T(\tau_{\max} - s) - T(\tau_{\max} - s)]f(s, u(s))\|ds,$$

appearing on [446, p. 418, l. 11], can be estimated with the help of mean value theorem, (363) and [199, Proposition 3.2, 3.4], as follows:

$$\int_0^t \|[T(\tau_{\max} - s) - T(\tau_{\max} - s)]f(s, u(s))\|ds \leqslant (\tau_{\max} - t)C_1Ct^{\beta}/\beta,$$

where C is the constant from (363) and C_1 is the constant from the formulation of [199, Proposition 3.4].

For the sequel, we need the following equality (see e.g. [181, (3.3)]):

(364)
$$\mathcal{A}^{-\theta}x = \frac{\sin(\theta\pi)}{\pi} \int_0^\infty s^{-\theta} (s+\mathcal{A})^{-1} x \, ds, \quad 1 > \theta > 1 - \beta, \ x \in E.$$

Suppose that $y \in (-\mathcal{A})^{\theta} x$, where $1 > \theta > 1 - \beta$. Then (364) and the obvious equality $(s - \mathcal{A})^{-1}T(t)y = T(t)(s - \mathcal{A})^{-1}y$, t, s > 0 together imply

$$(-\mathcal{A})^{-\theta}T(t)y = \frac{\sin(\theta\pi)}{\pi} \int_0^\infty s^{-\theta}(s-\mathcal{A})^{-1}T(t)y\,ds$$
$$= T(t)\frac{\sin(\theta\pi)}{\pi} \int_0^\infty s^{-\theta}(s-\mathcal{A})^{-1}y\,ds$$
$$= T(t)(-\mathcal{A})^{-\theta}y = T(t)x, \quad t > 0.$$

Hence,

(365)
$$T(t)(-\mathcal{A})^{\theta} \subseteq (-\mathcal{A})^{\theta}T(t), \quad t > 0, \ 1 > \theta > 1 - \beta.$$

Owing to (365), we can estimate the term $||u(t; u_0) - u(t; u_1)||_{[D((-\mathcal{A})^{\theta})]}$ (cf. line 11 of Step 2, p. 420, the proof of [446, Theorem 3.2]) as in non-degenerate case; furthermore, on the same page of proof, we can use [199, Theorem 3.5] ([199, Proposition 3.2]) in place of [447, Theorem 3.9(vii)] [447, Theorem 3.9(iii)]. Keeping in mind these observations, we can formulate the following extension of [446, Theorem 3.2] for abstract degenerate Cauchy inclusions of first order.

THEOREM 3.5.6. Let T > 0, let $\gamma = 1$, and let condition (H) hold. Suppose that $\beta > \theta > 1 - \beta$ and $0 < t < \tau_{\max}(u_0)$. Then there exist r > 0 and K > 0 such that the assumption $u_1 \in B_{\theta,r}(u_0) := \{u \in D((-\mathcal{A})^{\theta}) : ||u - u_0||_{[D((-\mathcal{A})^{\theta})]} \leq r\}$ implies that there exists a unique mild solution $u(t; u_1) \in C([0, \tau_{\max}(u_1)) : E)$ of problem (DFP)_{f,s,1} with $\tau_{\max}(u_1) \geq \tau$. Moreover,

$$||u(t; u_0) - u(t; u_1)|| \leq K ||u_0 - u_1||_{[D((-\mathcal{A})^{\theta})]}, \quad 0 \leq t \leq \tau$$

and, for every $\varepsilon \in (0, \tau)$, there exists a constant $C_{\varepsilon} > 0$ such that

$$\|u(t;u_0) - u(t;u_1)\|_{[D((-\mathcal{A})^{\theta})]} \leqslant C_{\varepsilon} \|u_0 - u_1\|_{[D((-\mathcal{A})^{\theta})]}, \quad \varepsilon \leqslant t \leqslant \tau.$$

The situation is much more complicated if we consider abstract degenerate fractional Cauchy inclusion $(DFP)_{f,s,\gamma}$ of order $\gamma \in (0, 1)$. Concerning [446, Theorem 3.1], we would like to point out that we cannot use, in fractional case, the well-known procedure for construction of a mild solution of problem $(DFP)_{f,s,\gamma}$ defined in a maximal time interval (see e.g. the integral equation [447, (8), p. 417]). The best we can do is prove the local existence and uniqueness of mild solutions of problem $(DFP)_{f,s,\gamma}$, as it has been explained in [529, Remark 4.1].

Concerning [446, Theorem 3.2], we can prove the following:

THEOREM 3.5.7. Let $\gamma \in (0,1)$, and let condition (H) hold. Suppose that $\beta > \theta > 1 - \beta$. Then there exist r > 0, $\tau > 0$ and K > 0 such that, for every $u_1 \in B_{\theta,r}(u_0)$, there exists a unique mild solution $u(t; u_1) \in C([0, \tau] : E)$ of problem (DFP)_{f,s,\gamma}. Moreover,

$$||u(t; u_0) - u(t; u_1)|| \leq K ||u_0 - u_1||_{[D((-\mathcal{A})^{\theta})]}, \quad 0 \leq t \leq \tau$$

and there exists a constant C > 0 such that, for every $\varepsilon \in (0, \tau)$, we have

$$\|u(t;u_0) - u(t;u_1)\|_{[D((-\mathcal{A})^{\theta})]} \leq C\varepsilon^{\gamma(\beta-1)}\|u_0 - u_1\|_{[D((-\mathcal{A})^{\theta})]}, \quad \varepsilon \leq t \leq \tau.$$

PROOF. We will only outline the most relevant details of proof in the degenerate case.

1. Line 6 of Step 1, p. 419, the proof of [446, Theorem 3.2]: Due to (359), we have that

$$\|S_{\gamma}(t)x - x\| = \left\| \int_{0}^{t} S_{\gamma}'(s)x \, ds \right\|$$
$$\leq C_{\theta}' t^{\gamma(\beta+\theta-1)} \|x\|_{[D((-\mathcal{A})^{\theta})]}, \quad t > 0;$$

therefore, $\lim_{t\to 0+} ||S_{\gamma}(t)u_1 - u_1|| = 0$, uniformly on the ball $B_{\theta,r}(u_0)$.

2. Line 1, [446, p. 420]: Here we may apply (b3) in order to get the existence of a constant $c_{\theta} > 0$ such that

$$||S_{\gamma}(t)(u_1 - u_0)|| \leq c_{\theta} t^{\gamma(\beta + \theta - 1)} ||u_1 - u_0||_{[D((-\mathcal{A})^{\theta})]}, \quad t > 0.$$

3. By (365), Theorem 1.2.3 and definition of $S_{\gamma}(\cdot)$, we have that

$$S_{\gamma}(t)(-\mathcal{A})^{\theta} \subseteq (-\mathcal{A})^{\theta}S_{\gamma}(t), \quad t > 0, \ 1 > \theta > 1 - \beta.$$

From this, we may conclude that

(366) $||S_{\gamma}(t)x||_{[D((-\mathcal{A})^{\theta})]} \leq ||S_{\gamma}(t)|| ||x||_{[D((-\mathcal{A})^{\theta})]} = O(t^{\gamma(\beta-1)} ||x||_{[D((-\mathcal{A})^{\theta})]}), t > 0.$ On the other hand, Theorem 1.2.3 and (351) together imply that

$$(367) \quad \frac{\gamma}{2\pi i} \int_0^t (t-s)^{-\gamma-1} \\ \times \int_0^\infty r \Phi_\gamma(r(t-s)^{-\gamma}) \bigg[\int_\Gamma (-\lambda)^\theta e^{r\lambda} (\lambda-\mathcal{A})^{-1} f(s, u(s; u_1)) d\lambda \bigg] dr \, ds \\ \in (-\mathcal{A})^\theta \int_0^t (t-s)^{\gamma-1} P_\gamma(t-s) f(s, u(s; u_1)) ds, \quad t > 0, \ 0 < s \leqslant \tau,$$

since $\beta > \theta$ and the norm of integrand in the first line does not exceed $(t - s)^{\gamma(\beta-\theta-1)} || f(s, u(s; u_1)) ||$ by (352). Hence, $u(t; u_1) \in D((-\mathcal{A})^{\theta})$ for all $t \in [0, \tau]$ and $u_1 \in B_{\theta,r}(u_0)$. For a fixed element $u_1 \in B_{\theta,r}(u_0)$, the continuity of mapping $t \mapsto u(t; u_1) \in [D((-\mathcal{A})^{\theta})], t \in (0, \tau]$ follows from (366), the analyticity of $S_{\gamma}(\cdot)$, the expression (367) and the dominated convergence theorem. To finish the proof, we can repeat verbatim the corresponding part of proof of [**446**, Theorem 3.2]. \Box

Observe, finally, that Theorem 3.5.6 and Theorem 3.5.7 continue to hold if we consider the space $E_{\mathcal{A}}^{\theta}$ in place of $[D((-\mathcal{A})^{\theta})]$. These theorems, as well some other theorems previously considered in this section, require the condition $\beta > 1/2$, which seems to be restrictive in the degenerate case (in numerous examples from [199, Chapter III], the condition (PW) holds with $\beta = 1/2$). For example, in the case of consideration of semilinear analogues of problem $(P)_{\gamma}$, Theorem 3.5.6 and Theorem 3.5.7 can be applied provided the additional condition [199, (3.42)] on the function

m(x), which ensures us to get the better exponent $\beta = 1/(2 - \rho)$ in (PW), with $0 < \rho \leq 1$.

3.5.2. Purely fractional case. In this subsection, we will briefly explain how we can extend a great part of our results established in the previous part of this section by investigating the abstract degenerate fractional Cauchy inclusions involving the multivalued linear operators satisfying the following condition (cf. [325] for more details):

(QP): There exist finite numbers $0<\beta\leqslant 1,\, 0< d\leqslant 1,\, M>0$ and $0<\eta'<\eta''<1$ such that

$$\Psi := \{\lambda \in \mathbb{C} : |\lambda| \leqslant d \text{ or } \lambda \in \overline{\Sigma_{\pi\eta''/2}}\} \subseteq \rho(\mathcal{A})$$

and

$$||R(\lambda : \mathcal{A})|| \leq M(1+|\lambda|)^{-\beta}, \quad \lambda \in \Psi.$$

Set $\delta := \min(\pi/2(\eta'' - \eta')/\eta', \pi/2)$. Let $\delta' \in (0, \delta)$, let $0 < \varepsilon < \delta'$ be arbitrarily chosen, and let

$$T_{\eta',r}(z)x := \frac{1}{2\pi i} \int_{\Gamma_{\omega}} e^{\lambda z} \lambda^r (\lambda^{\eta'} - \mathcal{A})^{-1} x \, d\lambda, \quad x \in E, \ r \in \mathbb{R}, \ z \in \Sigma_{\delta' - \varepsilon},$$

where Γ_{ω} is oriented counterclockwise and consists of $\Gamma_{\pm} := \{te^{i((\pi/2)+\delta')} : t \ge \omega\}$ and $\Gamma_0 := \{\omega e^{i\zeta} : |\zeta| \le (\pi/2) + \delta'\}$. Observe that the Cauchy formula implies that the definition of $T_{\eta',r}(z)$ is independent of $\omega > 0$. Arguing as in the proof of [27, Theorem 2.6.1], with $\omega = 1/|z|$, we get that for each $\theta' \in (0, \theta)$ and $r \in \mathbb{R}$, the following holds:

$$||T_{\eta',r}(z)|| = O(|z|^{\eta'\beta-r-1}), \quad z \in \Sigma_{\theta'}$$

and

(368)
$$\int_0^\infty e^{-\lambda t} T_{\eta',r}(t) x \, dt = \lambda^r (\lambda^{\eta'} - \mathcal{A})^{-1} x, \quad x \in E, \ \lambda > 0, \text{ provided } \eta'\beta > r;$$

 $(T_{\eta',\eta'-1}(t))_{t>0}$ and $(T_{\eta',0}(t))_{t>0}$ will be the most important operator families.

First of all, we need to reconsider the assertions from [199, Section 3.1] in the fractional case. In the following proposition, we will prove a fractional analogue of the second inequality from [199, Proposition 3.2].

PROPOSITION 3.5.8. Suppose that $0 < \theta < 1$ and $\theta \leq \beta$. Then we have $R(T_{\eta',\eta'-1}(t)) \subseteq E^{\theta}_{\mathcal{A}}, t > 0$. Furthermore, for every $\theta \in (0,1)$, there exists a constant $C_{\theta} > 0$ such that

$$\sup_{s>0} s^{\theta} \| sR(s:\mathcal{A})T_{\eta',\eta'-1}(t)x - T_{\eta',\eta'-1}(t)x \| \leq C_{\theta} t^{\eta'(\beta-\theta-1)} \|x\|, \quad t>0, \ x\in E.$$

PROOF. Let t > 0 and s > 0 be fixed, and let $\omega > 0$ be such that $\omega^{\eta} < s$. By Lemma 1.2.4(ii) and a simple computation, we have that, for every $x \in E$,

(369)
$$sR(s:\mathcal{A})T_{\eta',\eta'-1}(t)x - T_{\eta',\eta'-1}(t)x$$
$$= \frac{1}{2\pi i} \int_{\Gamma_{\omega}} e^{\lambda t} \frac{\lambda^{\eta'}s}{\lambda^{\eta'}-s} R(s:\mathcal{A})x \, d\lambda - \frac{1}{2\pi i} \int_{\Gamma_{\omega}} e^{\lambda t} \frac{\lambda^{2\eta'-1}}{\lambda^{\eta'}-s} R(\lambda^{\eta'}:\mathcal{A})x \, d\lambda.$$

The Cauchy theorem yields

$$\frac{1}{2\pi i} \int_{\Gamma_{\omega}} e^{\lambda t} \frac{\lambda^{\eta'} s}{\lambda^{\eta'} - s} R(s:\mathcal{A}) x \, d\lambda$$
$$= \int_0^\infty e^{-vt} v^{\eta'-1} \Big[\frac{e^{-i\pi(\eta'-1)}}{e^{-i\pi\eta'} v^{\eta'} - s} - \frac{e^{i\pi(\eta'-1)}}{e^{i\pi\eta'} v^{\eta'} - s} \Big] s R(s:\mathcal{A}) x \, dv, \quad x \in E.$$

It is clear that there exists a constant a > 0 such that $|e^{\pm i\pi\eta'}v^{\eta'} - s| \ge a(v^{\eta'} + s), v > 0$. Using this fact and the inequality

(370)
$$\theta t + (1-\theta)s \ge t^{\theta}s^{1-\theta}, \quad t, \ s \ge 0, \ \theta \in (0,1),$$

we get that there exists a constant $C_{\theta,1} > 0$, independent of s > 0, such that:

$$\begin{split} \left\| s^{\theta} \int_{0}^{\infty} e^{-vt} v^{\eta'-1} \Big[\frac{e^{-i\pi(\eta'-1)}}{e^{-i\pi\eta'} v^{\eta'} - s} - \frac{e^{i\pi(\eta'-1)}}{e^{i\pi\eta'} v^{\eta'} - s} \Big] sR(s:\mathcal{A}) x \, dv \right\| \\ & \leq 2Ma^{-1} \|x\| \int_{0}^{\infty} e^{-vt} v^{\eta'-1} \frac{s^{1-\beta+\theta}}{v^{\eta'} + s} dv \\ & \leq C_{\theta,1} \|x\| \int_{0}^{\infty} e^{-vt} v^{\eta'-1} \frac{s^{1-\beta+\theta}}{v^{\eta'(\beta-\theta)} s^{1-\beta+\theta}} dv \\ & = C_{\theta,1} \|x\| \Gamma(\eta'(1-\beta-\theta)) t^{\eta'(\beta-\theta-1)}, \quad t > 0. \end{split}$$

Now we will estimate the second term in (369) multiplied with s^{θ} . It suffices to consider the following two cases: $s > t^{-\eta'}$ and $s < t^{-\eta'}$. Suppose first that $s > t^{-\eta'}$. Then there exists a constant b > 0 such that $|\lambda^{\eta'} - s| \ge b(|\lambda|^{\eta'} + s), \lambda \in \Gamma_{1/t}$. Using Cauchy theorem and (370), we get that there exist two constants $C_{\theta,2}, C_{\theta,3} > 0$, independent of s > 0, such that:

$$\begin{split} \left\| \frac{1}{2\pi i} \int_{\Gamma_{\omega}} e^{\lambda t} \frac{\lambda^{2\eta'-1} s^{\theta}}{\lambda^{\eta'}-s} R(\lambda^{\eta'}:\mathcal{A}) x \, d\lambda \right\| &= \left\| \frac{1}{2\pi i} \int_{\Gamma_{1/t}} e^{\lambda t} \frac{\lambda^{2\eta'-1} s^{\theta}}{\lambda^{\eta'}-s} R(\lambda^{\eta'}:\mathcal{A}) x \, d\lambda \right\| \\ &\leqslant \frac{1}{2\pi b} \int_{\Gamma_{1/t}} e^{\operatorname{Re}\lambda t} \frac{|\lambda|^{2\eta'-1-\eta'\beta} s^{\theta}}{|\lambda|^{\eta'}+s} |d\lambda| \\ &\leqslant C_{\theta,2} \int_{\Gamma_{1/t}} e^{\operatorname{Re}\lambda t} \frac{|\lambda|^{2\eta'-1-\eta'\beta} s^{\theta}}{|\lambda|^{\eta'(1-\theta)} s^{\theta}} |d\lambda| \\ &= C_{\theta,2} \int_{\Gamma_{1/t}} e^{\operatorname{Re}\lambda t} |\lambda|^{\eta'(1+\theta-\beta)-1} |d\lambda| \\ &\leqslant C_{\theta,3} t^{\eta'(\beta-\theta-1)}, \quad t > 0, \end{split}$$

where the last estimate follows from the calculation contained in the proof of [27, Theorem 2.6.1]. If $s < t^{-\eta'}$, then the equation (369) continues to hold with the number ω replaced with 1/t therein. In the newly arisen situation, the residue theorem shows that

(371)
$$\frac{s^{\theta}}{2\pi i} \int_{\Gamma_{\omega}} e^{\lambda t} \frac{\lambda^{\eta'} s}{\lambda^{\eta'} - s} R(s:\mathcal{A}) x \, d\lambda$$

$$\begin{split} &= \int_0^\infty e^{-vt} s^\theta v^{\eta'-1} \Big[\frac{e^{-i\pi(\eta'-1)}}{e^{-i\pi\eta'}v^{\eta'}-s} - \frac{e^{i\pi(\eta'-1)}}{e^{i\pi\eta'}v^{\eta'}-s} \Big] sR(s:\mathcal{A}) x \, dv \\ &\quad + 2\pi i s^\theta \operatorname{Res}_{\lambda=s^{1/\eta'}} \Big[\frac{e^{\lambda t} \lambda^{\eta'-1}}{\lambda^{\eta'}-s} sR(s:\mathcal{A}) x \Big] \\ &= \int_0^\infty e^{-vt} s^\theta v^{\eta'-1} \Big[\frac{e^{-i\pi(\eta'-1)}}{e^{-i\pi\eta'}v^{\eta'}-s} - \frac{e^{i\pi(\eta'-1)}}{e^{i\pi\eta'}v^{\eta'}-s} \Big] sR(s:\mathcal{A}) x \, dv \\ &\quad + \frac{s^\theta}{\eta'} e^{ts^{1/\eta'}} sR(s:\mathcal{A}) x, \quad x \in E. \end{split}$$

We can estimate the first summand in (371) and the term

$$\left\|\frac{1}{2\pi i}\int_{\Gamma_{\omega}}e^{\lambda t}\frac{\lambda^{2\eta'-1}s^{\theta}}{\lambda^{\eta'}-s}R(\lambda^{\eta'}:\mathcal{A})x\,d\lambda\right\| = \left\|\frac{1}{2\pi i}\int_{\Gamma_{1/t}}e^{\lambda t}\frac{\lambda^{2\eta'-1}s^{\theta}}{\lambda^{\eta'}-s}R(\lambda^{\eta'}:\mathcal{A})x\,d\lambda\right\|$$

as in the case that $s > t^{-\eta'}$, with the same final estimate. For the second summand in (371), we have the following estimates:

$$\|s^{\theta}e^{ts^{1/\eta'}}sR(s:\mathcal{A})x/\eta'\| \leq M\|x\|s^{1+\theta-\beta}e^{1}/\eta' \leq M\|x\|t^{-\eta'(1+\theta-\beta)}e/\eta', \quad t > 0.$$

he proof of the theorem is thereby complete.

The proof of the theorem is thereby complete.

Let Γ' be the integral contour used in the definition of fractional power $(-\mathcal{A})^{\theta}$. $\theta > 1 - \beta$ (cf. Subsection 1.2.1 for more details and the notation used). Denote by Φ the open region on the right of Γ' .

We need the following useful lemma.

(i) Suppose that $1 - \beta < \theta \leq 1$. Then there exists a Lemma 3.5.9. constant $C_{\theta} > 0$ such that

(372)
$$\|\lambda R(\lambda;\mathcal{A})x - x\| \leq C_{\theta} |\lambda|^{(1-\beta-\theta)} \|x\|_{[D((-\mathcal{A})^{\theta})]}, \ \lambda \in \overline{\Sigma_{\eta''\pi/2}}, \ x \in D((-\mathcal{A})^{\theta}).$$

(ii) Suppose that $1 - \beta < \theta < 1$. Then there exists a constant $C_{\theta} > 0$ such that

(373)
$$\|\lambda R(\lambda : \mathcal{A})x - x\| \leq C_{\theta} |\lambda|^{(1-\beta-\theta)} \|x\|_{E^{\theta}_{\mathcal{A}}}, \quad \lambda \in \overline{\Sigma_{\eta''\pi/2}}, \ x \in E^{\theta}_{\mathcal{A}}.$$

PROOF. Suppose first that $\theta = 1$. Then Lemma 1.2.4(i) implies that for any $(x,y) \in \mathcal{A}$ one has

$$\|\lambda R(\lambda:\mathcal{A})x - x\| = \|R(\lambda:\mathcal{A})y\| \leqslant M(1+|\lambda|)^{-\beta}, \quad \lambda \in \overline{\Sigma_{\eta''\pi/2}}.$$

Taking the infimum, we immediately obtain (372). Let $1 - \beta < \theta < 1$. Then the function

$$\lambda \mapsto H(\lambda) := \lambda^{(\beta+\theta-1)} [\lambda R(\lambda : \mathcal{A})x - x], \quad \lambda \in \overline{\Sigma_{\eta''\pi/2}}$$

is continuous on $\overline{\Sigma_{\eta''\pi/2}}$, holomorphic on $\Sigma_{\eta''\pi/2}$ and

$$\|H(\lambda)\| \leqslant M|\lambda|^{(\beta+\theta-1)}[M|\lambda|^{1-\beta}+1]\|x\|, \quad \lambda \in \overline{\Sigma_{\eta''\pi/2}}, \ x \in E.$$

Let $R_1 > 0$ be sufficiently large, obeying the properties that $|z + \lambda| \ge 1$ for $z \in \Gamma'$, $\lambda = Re^{\pm i\pi\eta''/2}$ and $-\lambda = -Re^{\pm i\pi\eta''/2} \in \Phi$ $(R \ge R_1)$. Put $\Gamma_{R,\pm} := \{Re^{\pm i\pi\eta''/2} :$ $R \ge R_1$. Then it is clear that there exists a constant a > 0 such that $a^{-1}(|z| +$

 $|\lambda| \geq |z + \lambda|, z \in \Gamma', \lambda \in \Gamma_{R,\pm}$. By the Phragmén–Lindelöf type theorem [27, Theorem 3.9.8, p. 179], it suffices to show that the estimates (372) and (373) hold for $\lambda \in \Gamma_{R,\pm}$, with an appropriately chosen constant $C_{\theta} > 0$ independent of $\lambda \in \overline{\Sigma_{\eta''\pi/2}}$ and $x \in D((-\mathcal{A})^{\theta})$) $(x \in E^{\theta}_{\mathcal{A}})$. Suppose first that $x \in D((-\mathcal{A})^{\theta})$ and $y \in (-\mathcal{A})^{\theta}x$ is arbitrarily chosen. Then $x = (-\mathcal{A})^{-\theta}y = \frac{1}{2\pi i}\int_{\Gamma'} z^{-\theta}R(z : -\mathcal{A})y\,dz$ and it is not difficult to prove with the help of Lemma 1.2.4(ii) and the residue theorem that

(374)
$$\lambda^{\eta'} R(\lambda^{\eta'} : \mathcal{A}) x - x = \frac{(-1)}{2\pi i} \int_{\Gamma'} \frac{z^{1-\theta}}{z+\lambda} R(z : -\mathcal{A}) y \, dz + (-\lambda)^{-\theta} y, \quad \lambda \in \Gamma_{R,\pm}.$$

Keeping in mind the parametrization of Γ' , (374) and the arbitrariness of y, we get that for each $\lambda \in \Gamma_{R,\pm}$ the following holds:

$$\begin{aligned} \|\lambda R(\lambda:\mathcal{A})x - x\| - |\lambda|^{-\theta} \|x\|_{[D((-\mathcal{A})^{\theta})]} \\ &\leqslant a \|x\|_{[D((-\mathcal{A})^{\theta})]} \int_{-\infty}^{c} \frac{|v|^{1-\theta-\beta}}{|v| + |\lambda|} (1 + \beta^{2} (4M^{2})^{-1} (c-v)^{2\beta-2})^{1/2} dv. \end{aligned}$$

For the estimation of this integral, we divide the path of integration into three segments: $(-\infty, 0]$, [0, c/2] and [c/2, c]. We have

$$\left(\int_{-\infty}^{0} + \int_{0}^{c/2}\right) \frac{|v|^{1-\theta-\beta}}{|v|+|\lambda|} (1+\beta^{2}(4M^{2})^{-1}(c-v)^{2\beta-2})^{1/2} dv$$

$$\leq 2(1+\beta^{2}(4M^{2})^{-1}(c/2)^{2\beta-2})^{1/2} \int_{0}^{\infty} \frac{v^{1-\theta-\beta}}{v+|\lambda|} dv$$

$$= 2(1+\beta^{2}(4M^{2})^{-1}(c/2)^{2\beta-2})^{1/2} |\lambda|^{1-\theta-\beta} \int_{0}^{\infty} \frac{v^{1-\theta-\beta}}{v+1} dv$$

$$= 2(1+\beta^{2}(4M^{2})^{-1}(c/2)^{2\beta-2})^{1/2} |\lambda|^{1-\theta-\beta} \frac{(-\pi)}{\sin\pi(\theta+\beta)}, \quad \lambda \in \Gamma_{R,\pm}.$$

The integral over segment [c/2, c] can be majorized by using the inequality

$$\frac{|v|^{1-\theta-\beta}}{|v|+|\lambda|} \leqslant \frac{|v|^{1-\theta-\beta}}{|v|^{2-\theta-\beta}|\lambda|^{\theta+\beta-1}}, \quad \lambda \in \Gamma_{R,\pm}, \ v \in [c/2,c],$$

giving the same final estimate. This completes the proof of (i). In order to prove (ii), fix an element $x \in E^{\theta}_{\mathcal{A}}$. Let us observe that there exists a finite constant $c_{\theta} > 0$, independent of $x \in E^{\theta}_{\mathcal{A}}$, such that, for every $\lambda \in \Gamma_{R,\pm}$,

$$c_{\theta}|\lambda|^{1-\beta-\theta}||x||_{E_{\mathcal{A}}^{\theta}} \ge \left\| (|\lambda|-\lambda)(\lambda-\mathcal{A})^{-1}[(|\lambda|-\mathcal{A})^{-1}x-x] \right\|$$
$$= \left\| \left(1 - \frac{|\lambda|}{\lambda}\right)[(|\lambda|-\mathcal{A})^{-1}x-x] + (\lambda-\mathcal{A})^{-1}x\right.$$
$$+ \left[\left(1 - \frac{|\lambda|}{\lambda}\right)x - (|\lambda|-\mathcal{A})^{-1}x \right] \right\|,$$

where the equality follows from Lemma 1.2.4(ii) and a simple computation. This implies that, for every $\lambda \in \Gamma_{R,\pm}$,

$$\left|1-\frac{|\lambda|}{\lambda}\right| \|(|\lambda|-\mathcal{A})^{-1}x-x\|$$

$$\leq c_{\theta} |\lambda|^{1-\beta-\theta} \|x\|_{E^{\theta}_{\mathcal{A}}} + \|(\lambda-\mathcal{A})^{-1}x\| + \left\| \left[\left(1 - \frac{|\lambda|}{\lambda}\right)x - (|\lambda| - \mathcal{A})^{-1}x \right] \right\|$$

$$\leq c_{\theta} |\lambda|^{(1-\beta-\theta)} \|x\|_{E^{\theta}_{\mathcal{A}}} + M|\lambda|^{-\beta} \|x\|$$

$$+ \left\| \left(1 - \frac{|\lambda|}{\lambda}\right)x - (|\lambda| - \mathcal{A})^{-1} \left(1 - \frac{|\lambda|}{\lambda}\right)x \right\| + \|(|\lambda| - \mathcal{A})^{-1}x\|$$

$$\leq c_{\theta} |\lambda|^{1-\beta-\theta} \|x\|_{E^{\theta}_{\mathcal{A}}} + M|\lambda|^{-\beta} \|x\| + 2|\lambda|^{-\theta} \|x\|_{E^{\theta}_{\mathcal{A}}} + M|\lambda|^{-\beta} \|x\|.$$

Taking into account this estimate, the proof of (ii) is completed through a routine argument. $\hfill \Box$

- REMARK 3.5.10 ([181]). (i) The operator $(-\mathcal{A})^n$, defined as fractional power, coincides with the usual power $(-\mathcal{A})^n$ $(n \in \mathbb{N})$.
- (ii) The space $[D(\mathcal{A})]$ is continuously embedded in $[D((-\mathcal{A})^{\theta})]$ provided that $\beta > 1/2$ and $1 \beta < \theta < \beta$.

Now we are ready to prove the following generalization of [199, Theorem 3.5] for degenerate fractional differential equations.

THEOREM 3.5.11. Let $\delta' \in (0, \delta)$.

(i) Suppose that $1 - \beta < \theta \leq 1$. Then there exists a constant $C_{\theta,\delta'} > 0$ such that

$$(375) ||T_{\eta',\eta'-1}(z)x-x|| \leq C_{\theta,\delta'} |z|^{\eta'(\beta+\theta-1)} ||x||_{[D((-\mathcal{A})^{\theta})]}, z \in \Sigma_{\delta'}, x \in D((-\mathcal{A})^{\theta}).$$

(ii) Suppose that $1 - \beta < \theta < 1$. Then there exists a constant $C_{\theta,\delta'} > 0$ such that

$$||T_{\eta',\eta'-1}(z)x - x|| \leq C_{\theta,\delta'} |z|^{\eta'(\beta+\theta-1)} ||x||_{E^{\theta}_{\mathcal{A}}}, \quad z \in \Sigma_{\delta'}, \ x \in E^{\theta}_{\mathcal{A}}.$$

PROOF. Let $\delta'' \in (\delta', \delta)$, and let $0 < \varepsilon < \delta - \delta''$ be arbitrarily chosen. Then it is clear that

$$T_{\eta',\eta'-1}(z)x - x = \frac{1}{2\pi i} \int_{\Gamma_{1/|z|}} \frac{e^{\lambda z}}{\lambda} [\lambda^{\eta'} R(\lambda^{\eta'} : \mathcal{A})x - x] d\lambda, \quad z > 0, \ x \in E.$$

Now the result follows from Lemma 3.5.9 and the calculus contained in the proof of [27, Theorem 2.6.1].

Suppose now that $\theta > 1 - \beta$. It is clear that there exists a sufficiently small number $t_0 > 0$ such that, for every $t \in (0, t_0]$ and $\lambda \in \Gamma_{1/t}$, we have $\lambda^{\eta'} \in \Phi$. Then Lemma 1.2.4(ii) and Fubini theorem together imply that, for every $t \in (0, t_0]$,

$$(-\mathcal{A})^{-\theta}T_{\eta',\eta'-1}(t)x = \frac{(-1)}{(2\pi i)^2} \int_{\Gamma_{1/t}} e^{\lambda t} \lambda^{\eta'-1} (\lambda^{\eta'} - \mathcal{A})^{-1} \left[\int_{\Gamma'} \frac{dz}{z^{\theta}(\lambda^{\eta'} - z)} \right] d\lambda$$
$$+ \frac{1}{(2\pi i)^2} \int_{\Gamma_{1/t}} e^{\lambda t} \lambda^{\eta'-1} (\lambda^{\eta'} - \mathcal{A})^{-1} \left[\int_{\Gamma'} \frac{(z - \mathcal{A})^{-1}x}{z^{\theta}(\lambda^{\eta'} - z)} dz \right] d\lambda.$$

Applying the residue theorem on the first integral and Fubini theorem on the second one, we get from the above and definition of $(-\mathcal{A})^{-\theta}$ that, for every $t \in (0, t_0]$,

$$(-\mathcal{A})^{-\theta}T_{\eta',\eta'-1}(t)x = \frac{1}{2\pi i} \int_{\Gamma_{1/t}} e^{\lambda t} \lambda^{\eta'-1-\eta'\theta} (\lambda^{\eta'}-\mathcal{A})^{-1}x \, d\lambda + (-\mathcal{A})^{-\theta}x$$

$$+ \frac{1}{2\pi i} \int_{\Gamma'} z^{1-\theta} (z-\mathcal{A})^{-1} \left[\int_{\Gamma_{1/t}} \frac{e^{\lambda t}}{\lambda(\lambda^{\eta'}-z)} d\lambda \right] dz$$
$$:= I_1(t) + (-\mathcal{A})^{-\theta} x + I_3(t).$$

Making use of dominated convergence theorem, we immediately obtain that $I_1(t) \rightarrow$ 0 as $t \to 0+$. Let $1 > \zeta > 2 - \theta - \beta$. Then (370) implies

$$(376) \qquad \int_{\Gamma_{1/t}} \frac{|d\lambda|}{|\lambda||\lambda^{\eta'} - z|} \leqslant a \int_{\Gamma_{1/t}} \frac{|d\lambda|}{|\lambda|(|\lambda|^{\eta'} + |z|)} \leqslant a \int_{\Gamma_{1/t}} \frac{|d\lambda|}{|\lambda||\lambda|^{\eta'(1-\zeta)}|z|^{\zeta}},$$

where a > 0 is a constant independent of $t \in (0, t_0], \lambda \in \Gamma_{1/t}$ and $z \in \Gamma'$. With (376) in view, we may apply the dominated convergence theorem in order to see that $I_3(t) \to 0$ as $t \to 0+$. Keeping in mind Theorem 3.5.11(ii) and the commutation of operators $(-\mathcal{A})^{-\theta}$ and $T_{n',n'-1}(t)$, we obtain from the above that the following holds:

(CS) $T_{\eta',\eta'-1}(t)x \to x, t \to 0+$ for any $x \in E$ belonging to the space $D((-\mathcal{A})^{\theta})$ with $\theta > 1 - \beta$ ($x \in E^{\theta}_{\mathcal{A}}$ with $1 > \theta > 1 - \beta$).

In the remainder of this subsection, we assume that $0 < \gamma < 1$. For the simplicity of notation, set $\mathcal{T}_{n'}(t) := T_{n',n'-1}(t), t > 0$. Define $\eta := \gamma \eta'$ and, for every $\nu > -1 - \eta'(\beta - 1)$,

$$\mathcal{T}^{\nu}_{\eta',\gamma}(t)x := t^{-\gamma} \int_0^\infty s^{\nu} \Phi_{\gamma}(st^{-\gamma}) \mathcal{T}_{\eta'}(s) x \, ds, \quad t > 0, \ x \in E \text{ and } \mathcal{T}^0_{\eta',\gamma}(0) := I.$$

Then it is clear that:

$$\mathcal{T}^{\nu}_{\eta',\gamma}(t)x = t^{\gamma\nu} \int_0^\infty s^{\nu} \Phi_{\gamma}(s) \mathcal{T}_{\eta'}(st^{\gamma})x \, ds, \quad t > 0, \ x \in E.$$

Keeping in mind the results proved in this subsection, it is quite simple to deduce the following:

(B0) The operator $\mathcal{T}^{\nu}_{n',\gamma}(t)$ is absolutely convergent and

$$\|\mathcal{T}^{\nu}_{\eta',\gamma}(t)\| = O(t^{\gamma(\nu+\eta'(\beta-1))}), \quad t > 0.$$

For every $\nu > -1 - n'(\beta - 1)$, we have

$$\frac{\mathcal{T}_{\eta',\gamma}^{\nu}(t)}{t^{\gamma\nu}}x - \frac{\Gamma(1+\nu)}{\Gamma(1+\gamma\nu)}x = \int_0^\infty s^{\nu}\Phi_{\gamma}(s)[\mathcal{T}_{\eta'}(st^{\gamma})x - x]ds, \quad t > 0, \ x \in E.$$

- (B1) $\frac{\mathcal{T}_{\eta',\gamma}^{\nu}(t)}{t^{\gamma\nu}}x \to \frac{\Gamma(1+\nu)}{\Gamma(1+\gamma\nu)}x, t \to 0+ \text{ provided that } \theta > 1-\beta \text{ and } x \in D((-\mathcal{A})^{\theta}),$ or that $1 > \theta > 1 - \beta$ and $x \in E^{\theta}_{\mathcal{A}}$ $(\nu > -1 - \eta'(\beta - 1)).$
- (B1)' Suppose that $\varepsilon \in (0,\xi)$ and $\delta = \xi \varepsilon$. Then $\lim_{z\to 0, z\in \Sigma_{\delta}} \frac{\mathcal{T}_{\eta',\gamma}^{\nu}(z)}{z^{\gamma\nu}} x =$ $\frac{\Gamma(1+\nu)}{\Gamma(1+\gamma\nu)}x$, provided that $\theta > 1-\beta$ and $x \in D((-\mathcal{A})^{\theta})$, or that $1 > \theta > 1-\beta$
- (B2) $\int_{0}^{\infty} e^{-\lambda t} \mathcal{T}_{\eta',\gamma}^{0}(t) x \, dt = \lambda^{\gamma-1} \int_{0}^{\infty} e^{-\lambda^{\gamma} t} \mathcal{T}_{\eta'}(t) x \, dt = \lambda^{\eta-1} (\lambda^{\eta} \mathcal{A})^{-1} x, \operatorname{Re} \lambda > 0, x \in E.$

- (B3) $\begin{aligned} \|\frac{\mathcal{T}_{\eta',\gamma}^{\nu}(t)}{t^{\gamma\nu}}x \frac{\Gamma(1+\nu)}{\Gamma(1+\gamma\nu)}x\| &= O(t^{\eta(\beta+\theta-1)}\|x\|_{[D((-\mathcal{A})^{\theta})]}), \ t > 0, \ \text{provided} \ 1 > \\ \theta > 1 \beta, \ x \in D((-\mathcal{A})^{\theta}) \ \text{and} \ \|\frac{\mathcal{T}_{\eta',\gamma}^{\nu}}{t^{\gamma\nu}}x \frac{\Gamma(1+\nu)}{\Gamma(1+\gamma\nu)}x\| &= O(t^{\eta(\beta+\theta-1)}\|x\|_{E_{\mathcal{A}}^{\theta}}), \\ t > 0, \ \text{provided} \ 1 > \theta > 1 \beta, \ x \in E_{\mathcal{A}}^{\theta} \ (\nu > -1 \eta'(\beta-1)). \end{aligned}$
- (B3)' Suppose that $\varepsilon \in (0,\xi)$, $\delta = \xi \varepsilon$ and $\nu > -1 \eta'(\beta 1)$. Then $\|\frac{\mathcal{T}_{\eta',\gamma}(z)}{z^{\gamma\nu}}x \frac{\Gamma(1+\nu)}{\Gamma(1+\gamma\nu)}x\| = O(|z|^{\eta(\beta+\theta-1)}\|x\|_{[D((-\mathcal{A})^{\theta})]}), z \in \Sigma_{\delta}$, provided $1 > \theta > 1 \beta$, $x \in D((-\mathcal{A})^{\theta})$, and $\|\frac{\mathcal{T}_{\eta',\gamma}(z)}{z^{\gamma\nu}}x - \frac{\Gamma(1+\nu)}{\Gamma(1+\gamma\nu)}x\| = O(|z|^{\eta(\beta+\theta-1)}\|x\|_{E^{\theta}_{\mathcal{A}}}), z \in \Sigma_{\delta}$, provided $1 > \theta > 1 - \beta, x \in E^{\theta}_{\mathcal{A}}$.
- (B4) For every $\nu > -1 \eta'(\beta 1)$, the mapping $t \mapsto \mathcal{T}^{\nu}_{\eta',\gamma}(t)x, t > 0$ can be analytically extended to the sector Σ_{ξ} (we will denote this extension by the same symbol) and, for every $\theta \in (0, 1), \varepsilon \in (0, \xi)$ and $\nu > -1 - \eta'(\beta - 1),$ $\|\mathcal{T}^{\nu}_{\eta',\gamma}(z)\| = O(|z|^{\gamma(\nu+\eta'(\beta-1))}), z \in \Sigma_{\xi-\varepsilon}.$
- (B4)' For every $\theta \in (0, 1), \varepsilon \in (0, \xi), \nu > -1 \eta'(\beta 1)$ and $n \in \mathbb{N}$, $\|(d^n/dz^n)\mathcal{T}^{\nu}_{\eta',\gamma}(z)\| = O(|z|^{\gamma(\nu+\eta'(\beta-1))-n}), z \in \Sigma_{\xi-\varepsilon}.$
- (B5) For every $\theta \in (0, 1)$, $\varepsilon \in (0, \xi)$ and $\nu > -1 \eta'(\beta \theta 1)$,

$$\sup_{s>0} s^{\theta} \left\| sR(s:\mathcal{A})\mathcal{T}^{\nu}_{\eta',\gamma}(z)x - \mathcal{T}^{\nu}_{\eta',\gamma}(z)x \right\| = O(|z|^{\gamma(\nu+\eta'(\beta-\theta-1))}), \quad z \in \Sigma_{\xi-\varepsilon}.$$

(B5)' For every $\theta \in (0,1)$, $\varepsilon \in (0,\xi)$, $\nu > -1 - \eta'(\beta - \theta - 1)$ and $n \in \mathbb{N}$,

$$\sup_{s>0} s^{\theta} \| sR(s:\mathcal{A})(d^n/dz^n) \mathcal{T}^{\nu}_{\eta',\gamma}(z)x - (d^n/dz^n) \mathcal{T}^{\nu}_{\eta',\gamma}(z)x \| = O(|z|^{\gamma(\nu+\eta'(\beta-\theta-1))-n}),$$

for $z \in \Sigma_{\xi-\varepsilon}$. The angle of analyticity ξ of operator family $(\mathcal{T}^{\nu}_{\eta',\gamma}(t))_{t>0}$ is not optimal and can be improved by using the examinations from the proof of [**296**, Theorem 3.10]. We will not go into further details concerning this question here.

(B6) We have

$$\frac{\mathcal{T}_{\eta',\gamma}^{\nu}(z)}{z^{\gamma\nu}}x - \frac{\Gamma(1+\nu)}{\Gamma(1+\gamma\nu)}x = \int_{0}^{\infty} s^{\nu}\Phi_{\gamma}(se^{i\varphi})[\mathcal{T}_{\eta'}(s|z|^{\gamma})x - x]ds, \quad x \in E,$$

where $\varphi = -\gamma \arg(z).$

Now it is time to distinguish the following operator families:

$$\mathcal{S}_{\eta}(z) := \mathcal{T}^{0}_{\eta',\gamma}(z) \text{ and } \mathcal{P}_{\eta}(z) := z^{1-\eta} \frac{d}{dz} \int_{0}^{z} g_{\eta}(z-s) \mathcal{S}_{\eta}(s) ds \ (z \in \Sigma_{\xi}).$$

Clearly, $S_{\eta}(z)$ and $\mathcal{P}_{\eta}(z)$ depend analytically on parameter z in the uniform operator topology. It is not difficult to prove that, for every $\varepsilon \in (0, \xi)$, we have

$$\|\mathcal{S}_{\eta}(z)\| + \|\mathcal{P}_{\eta}(z)\| = O(|z|^{\eta(\beta-1)}), \quad z \in \Sigma_{\xi-\varepsilon},$$

as well as that $||(d/dz)\mathcal{P}_{\eta}(z)|| = O(|z|^{\eta(\beta-1)-1}), z \in \Sigma_{\xi-\varepsilon}$. Furthermore, we have the following:

(B7) For every R > 0, the mappings $z \mapsto S_{\eta}(z) \in L(E), z \in \Sigma_{\xi-\varepsilon} \setminus B_R$ and $z \mapsto \mathcal{P}_{\eta}(z) \in L(E), z \in \Sigma_{\xi-\varepsilon} \setminus B_R$ are uniformly continuous.

(B8) We have

$$S_{\eta}(z)x = \left(g_{1-\eta} * \left[\cdot^{\eta-1} \mathcal{P}_{\gamma}(\cdot)x\right]\right)(z), \quad z \in \Sigma_{\xi}, \ x \in E,$$
$$\int_{0}^{\infty} e^{-\lambda t} t^{\eta-1} \mathcal{P}_{\eta}(t)x \, dt = (\lambda^{\eta} - \mathcal{A})^{-1}x, \quad \lambda > 0, \ x \in E$$

and

$$\mathcal{P}_{\gamma}(t)x = t^{-\eta} \int_0^\infty \gamma s \Phi_{\gamma}(st^{-\gamma}) T_{\eta',0}(s) x \, ds, \quad t > 0, \ x \in E.$$

The last equality continues to hold on subsectors of Σ_{ξ} .

(B9) We have

$$\frac{d}{dz}\mathcal{S}_{\eta}(z)x \in z^{\eta-1}\mathcal{AP}_{\eta}(z)x, \quad z \in \Sigma_{\xi}, \ x \in E.$$

(B10) Suppose now that $(x, y) \in \mathcal{A}$. Then $S_{\eta}(z)x - x = \int_{0}^{z} \lambda^{\eta-1} \mathcal{P}_{\eta}(\lambda) y \, d\lambda$, $z \in \Sigma_{\xi}$ and

$$\frac{d}{dz}\mathcal{S}_{\eta}(z)x = z^{\eta-1}\mathcal{P}_{\eta}(z)y, \quad z \in \Sigma_{\xi};$$

the mapping $t \mapsto \frac{d}{dt} S_{\eta}(t) x, t > 0$ is locally integrable.

- (B11) Let $1 > \theta > 1 \beta$, and let $x \in D((-\mathcal{A})^{\theta}) \cap E^{\theta}_{\mathcal{A}}$. Then the mapping $t \mapsto F(t) := (g_{1-\eta} * [\mathcal{S}_{\eta}(\cdot)x x])(t)$ is continuous for $t \ge 0$ and can be analytically extended from the positive real axis to the sector Σ_{ξ} , as well as that $\mathbf{D}^{t}_{t} \mathcal{S}_{\eta}(t)x = F'(t) \in \mathcal{AS}_{\eta}(t)x, t > 0$.
- (B12) The mapping $t \mapsto \frac{d}{dt} S_{\eta}(t) x, t > 0$ is locally integrable for any $x \in D((-\mathcal{A})^{\theta}) \cap E_{\mathcal{A}}^{\theta}$.

The following extension of Theorem 3.5.3 holds true:

THEOREM 3.5.12. Suppose that $T \in (0, \infty)$, $1 \ge \theta > 1 - \beta$ and $x \in D((-\mathcal{A})^{\theta})$, resp. $1 > \theta > 1 - \beta$ and $x \in E^{\theta}_{\mathcal{A}}$, as well as that there exist constants $\sigma > \eta(1 - \beta)$ and $M \ge 1$ such that (358) holds. Let $1 \ge \theta > 1 - \beta$, resp. $1 > \theta > 1 - \beta$, and let

$$f \in L^{\infty}((0,T) : [D((-\mathcal{A})^{\theta})]), \ resp. \ f \in L^{\infty}((0,T) : E^{\theta}_{\mathcal{A}}).$$

Then there exists a unique classical solution of problem $(DFP)_f$.

PROOF. We will only provide the most relevant points of proof provided that $1 \ge \theta > 1 - \beta$ and $x \in D((-\mathcal{A})^{\theta})$. Let $\delta' \in (0, \delta)$ be fixed. Then (375) and the Cauchy integral formula together imply:

$$\begin{aligned} \|(d/dz)\mathcal{T}_{\eta'}(z)x\| &= \|(d/dz)[\mathcal{T}_{\eta'}(z)x - x]\| \\ &\leqslant C_{\theta,\delta'}|z|^{\eta'(\beta+\theta-1)-1}\|x\|_{[D((-\mathcal{A})^{\theta})]}, \quad z \in \Sigma_{\delta'}, \ x \in D((-\mathcal{A})^{\theta}). \end{aligned}$$

Using this estimate, we can prove that there exists a constant $C_{\theta} > 0$ such that, for every $0 < s \leq T$ and $0 < \omega \leq T$,

$$\|S'_{\gamma}(\omega)f(s)\| \leqslant C_{\theta}\|f(s)\|_{[D((-\mathcal{A})^{\theta})]}\omega^{\eta(\beta+\theta-1)-1}$$

The rest of proof is almost the same as that of Theorem 3.5.3.

Therefore, it makes sense to consider the mild solutions of following semilinear degenerate fractional Cauchy inclusion:

$$(\text{DFP})_{f,s,\eta} : \begin{cases} \mathbf{D}_{t}^{\eta} u(t) \in \mathcal{A}u(t) + f(t, u(t)), & t \in (0, T], \\ u(0) = u_{0}, \end{cases}$$

where $T \in (0, \infty)$. As before, a mild solution $u(t) := u(t; u_0)$ of problem $(DFP)_{f,s,\eta}$ is any function $u \in C((0, T] : E)$ such that

$$u(t) = S_{\eta}(t)u_0 + \int_0^t (t-s)^{\eta-1} \mathcal{P}_{\eta}(t-s)f(s,u(s))ds, \quad t \in (0,T].$$

Here we would like to observe that it is not clear how we can prove an extension of Theorem 3.5.7 in the case that the operator family $(\mathcal{S}_n(t))_{t>0}$ is not subordinated to a degenerate semigroup. With the exception of this result, all other results from Subsection 3.5.1 continue to hold in our new setting and we will only say a few words about the assertion of [529, Theorem 5.3], where the authors have investigated the existence of mild solutions of semilinear degenerate fractional Cauchy inclusion $(DFP)_{f,s,\eta}$, provided that the resolvent of \mathcal{A} is compact. The operator family $(\mathcal{S}_{\eta}(t))_{t>0}$ is then subordinated to a semigroup $(T(t))_{t>0}$ which do have a removable singularity at zero, and the compactness of operators $\mathcal{S}_n(t)$ and $\mathcal{P}_n(t)$ for t > 0 (cf. [529, Lemma 3.1, Theorem 3.5]) has been proved by following a method based on the use of semigroup property of $(T(t))_{t>0}$. In purely fractional case, we can argue as follows. Recall that the set consisting of all compact operators on E is a closed linear subspace of L(E) forming a two-sided ideal in L(E). Since (368) and (B2) hold in the uniform operator topology, we can apply Lemma 1.2.4(iii), the Post-Widder inversion formula [27, Theorem 1.7.7] and the formulae [61, (2.16)-(2.17)]in order to see that the operator $T_{\eta',r}(t)$ is compact for $\eta'\beta > r, t > 0$ and the operator $S_n(t)$ is compact for t > 0. Keeping in mind the third equality in (B8), we obtain that the operator $\mathcal{P}_n(t)$ is compact for t > 0, as well. Now we can reformulate [529, Theorem 5.3] by means of the following approximation in Step 3 of its proof:

$$\Gamma^{\eta}_{\varepsilon,\delta}(t) := \mathcal{S}_{\eta}(t) + \int_{0}^{t-\varepsilon} (t-s)^{2\gamma-1} \int_{\delta}^{\infty} \gamma \tau \Phi_{\gamma}(\tau) T_{\eta',0}(\tau(t-s)^{\gamma}) d\tau \, ds,$$

for $t \in (0, T]$, $\delta > 0$, $0 < \varepsilon < t$ and $u \in \Omega_r$; cf. [529] for the notion. Assuming the condition of type [529, (H2)], we can prove the estimate

$$\begin{split} \left\| \mathcal{S}_{\eta}(t)u_{0} + \int_{0}^{t} (t-s)^{\eta-1} \mathcal{P}_{\eta}(t-s)f(s,u(s))ds - \Gamma_{\varepsilon,\delta}^{\eta}(t) \right\| \\ &\leqslant \operatorname{Const.} \left(\int_{0}^{t-\varepsilon} (t-s)^{q[\beta\eta+\gamma-1]}ds \right)^{1/q} \|m_{r}\|_{L^{p}(0,T)} \int_{0}^{\infty} \tau^{\eta'\beta} \Phi_{\gamma}(\tau)d\tau \\ &+ \operatorname{Const.} \left(\int_{t-\varepsilon}^{t} (t-s)^{q[\beta\eta+\gamma-1]}ds \right)^{1/q} \|m_{r}\|_{L^{p}(0,T)} \int_{0}^{\infty} \tau^{\eta'\beta} \Phi_{\gamma}(\tau)d\tau \end{split}$$

for p > 1 and q = p/p - 1. Now it is quite simple to reformulate [529, Theorem 5.3] in our context.

EXAMPLE 3.5.13. (i) [527] Suppose that $\alpha \in (0,1), m \in \mathbb{N}, \Omega$ is a bounded domain in \mathbb{R}^n with boundary of class C^{4m} and $E := C^{\alpha}(\overline{\Omega})$. Let us consider the operator $A: D(A) \subseteq C^{\alpha}(\overline{\Omega}) \to C^{\alpha}(\overline{\Omega})$ given by

$$Au(x) := \sum_{|\beta| \leqslant 2m} a_{\beta}(x) D^{\beta}u(x) \text{ for all } x \in \bar{\Omega}$$

with domain $D(A) := \{ u \in C^{2m+\alpha}(\bar{\Omega}) : D^{\beta}u_{|\partial\Omega} = 0 \text{ for all } |\beta| \leq m-1 \}$. Here, $\beta \in \mathbb{N}_0^n$, $|\beta| = \sum_{i=1}^n \beta_j$, $D^{\beta} = \prod_{i=1}^n (\frac{1}{i} \frac{\partial}{\partial x_i})^{\beta_i}$, and we assume that $a_\beta : \bar{\Omega} \to \mathbb{C}$ satisfy the following:

- (i) $a_{\beta}(x) \in \mathbb{R}$ for all $x \in \overline{\Omega}$ and $|\beta| = 2m$.
- (ii) $a_{\beta} \in C^{\alpha}(\overline{\Omega})$ for all $|\beta| \leq 2m$, and
- (iii) there is a constant M > 0 such that

$$M^{-1}|\xi|^{2m} \leq \sum_{|\beta|=2m} a_{\beta}(x)\xi^{\beta} \leq M|\xi|^{2m}$$
 for all $\xi \in \mathbb{R}^n$ and $x \in \overline{\Omega}$.

Then it is well known that there exists a sufficiently large number $\sigma > 0$ such that the operator $-A_{\sigma} \equiv -(A+\sigma)$ satisfies $\Sigma_{\omega} \cup \{0\} \subseteq \rho(-A_{\sigma})$ with some $\omega \in (\frac{\pi}{2}, \pi)$ and

$$||R(\lambda:-A_{\sigma})|| = O(|\lambda|^{\frac{\alpha}{2m}-1}), \quad \lambda \in \Sigma_{\omega}.$$

(377)

Let us recall that A is not densely defined and the exponent $\frac{\alpha}{2m} - 1$ in (377) is sharp. Define $A_{\sigma,\delta} := e^{i(\pi/2\pm\delta)}A_{\sigma}$. Suppose that $\omega - (\pi/2) < \delta < \omega - \eta(\pi/2), 1 \ge \theta > \alpha/2m, u_0 \in D((-A_{\sigma,\delta})^{\theta}), \sigma > \eta\alpha/2m,$ (358) holds and $f \in L^{\infty}((0,T) : [D((-\mathcal{A})^{\theta})])$. Then the condition (QP) holds for each number $\eta'' \in (\eta, 1)$ such that $\omega - (\pi/2) < \delta < \omega - \eta''(\pi/2)$. Applying Theorem 3.5.12, we obtain that the abstract fractional Cauchy problem

$$\begin{cases} \mathbf{D}_{t}^{\eta} u(t,x) = A_{\sigma,\delta} u(t,x) + f(t,x), & t \in (0,T], \\ u(0) = u_{0}, \end{cases}$$

has a unique classical solution, which is analytically extendable to the sector Σ_{ϑ} provided that $f(t,x) \equiv 0$ ($\vartheta \equiv \min((\frac{(2/\pi)(\omega-\delta)}{n}-1)\pi/2,\pi)$).

(ii) [199] Consider now the following modification of inhomogeneous fractional Poisson heat equation in the space $L^p(\Omega)$:

$$(P)^{\delta}_{\eta}: \begin{cases} \mathbf{D}^{\eta}_{t}[m(x)v(t,x)] = e^{\pm i\delta}(\Delta - b)v(t,x) + f(t,x), & t \ge 0, \ x \in \Omega; \\ v(t,x) = 0, & (t,x) \in [0,\infty) \times \partial\Omega, \\ m(x)v(0,x) = u_{0}(x), & x \in \Omega, \end{cases}$$

where Ω is a bounded domain in \mathbb{R}^n , b > 0, $m(x) \ge 0$ a.e. $x \in \Omega$, $m \in L^{\infty}(\Omega)$, $1 and <math>0 < \eta < 1$. Let the operator $A := \Delta - b$ act on E with the Dirichlet boundary conditions, and let B be the multiplication operator by the function m(x). As it has been proved in [199, Example 3.6], there exist an appropriate angle $\omega \in (\frac{\pi}{2}, \pi)$ and a number d > 0 such that the multivalued linear operator $\mathcal{A} := AB^{-1}$ satisfies $\Psi_{d,\omega} = \{\lambda \in \mathbb{C} : |\lambda| \leq d \text{ or } \lambda \in \overline{\Sigma_{\omega}}\} \subseteq \rho(\mathcal{A})$ and $||R(\lambda : \mathcal{A})|| \leq$ $M(1 + |\lambda|)^{-1/p}, \ \lambda \in \Psi_{d,\omega}$; here it is worth noting that the validity of additional condition [199, (3.42)] on the function m(x) enables us to get the better exponent β in (QP), provided that p > 2. Henceforth we consider the general case. Suppose, as in part (i), that $\omega - (\pi/2) < \delta < \omega - \eta(\pi/2), \ 1 \ge \theta > 1 - 1/p, \ u_0 \in D((-e^{\pm i\delta}\mathcal{A})^{\theta}), \ \sigma > \eta(1 - 1/p), \ (358)$ holds and $f \in L^{\infty}((0,T) : [D((-e^{\pm i\delta}\mathcal{A})^{\theta})])$. Then Theorem 3.5.12 implies that the abstract Cauchy problem $(P)^{\delta}_{\eta}$ has a unique solution $t \mapsto v(t, \cdot), t \in (0,T]$, i.e., any function $v(t, \cdot)$ satisfying that $Bv(t, \cdot) \in C([0,T] : E)$, the Caputo fractional derivative $\mathbf{D}^{\eta}_{t}Bv(t, \cdot)$ is well-defined and belongs to the space $C((0,T] : E), Bv(t, \cdot) \in C((0,T] : E), \ m(x)v(0,x) = u_{0}(x), x \in \Omega$ and $(P)^{\delta}_{\eta}$ holds identically.

Observe that the trick used in previous example can be also applied in the analysis of limit problems of fractional diffusion equations in complex systems on the so-called dumbbell domains ([32, 529]), as well as in the analysis of a large class of abstract degenerate fractional differential inclusions involving the rotations of multivalued linear operators considered in [199, Section 3.7]. By [447, Proposition 3.6] ([411, Corollary 5.6]), fractional powers of almost sectorial operators (sectorial multivalued linear operators) satisfy, under some assumptions, the condition (QP) and can therefore be used for providing certain applications of our abstract results. Suitable translations of generators of fractionally integrated semigroups with corresponding growth order satisfy the condition (QP) with $\eta'' = 1$ and $0 < \beta < 1$, as well (cf. [446, Example 3.3]).

3.6. Hypercyclic and topologically mixing properties of abstract degenerate (multi-term) time-fractional inclusions

The main aim of this section, which is divided into four subsections, is to provide the basic information on hypercyclic and topologically mixing properties of abstract degenerate (multi-term) time-fractional inclusions.

By E we denote a separable infinite-dimensional Fréchet space over the field of complex numbers. We assume that the topology of E is induced by the fundamental system $(p_n)_{n \in \mathbb{N}}$ of increasing seminorms. Let us recall that the translation invariant metric $d: E \times E \to [0, \infty)$ is defined by

$$d(x,y) := \sum_{n=1}^{\infty} \frac{1}{2^n} \frac{p_n(x-y)}{1+p_n(x-y)}, \quad x,y \in E.$$

By \mathcal{A} and $\sigma_p(\mathcal{A})$ we denote a multivalued linear operator in E and its point spectrum, respectively.

3.6.1. Hypercyclic and topologically mixing properties of problem $(DFP)_{\alpha,\mathcal{A}}$. In this subsection, we will consider hypercyclic and topologically mixing properties of the following abstract degenerate time-fractional inclusion:

$$(DFP)_{\alpha,\mathcal{A}}: \begin{cases} \mathbf{D}_t^{\alpha} u(t) \in \mathcal{A}u(t), & t \ge 0, \\ u(0) = x; \ u^{(j)}(0) = 0, & 0 \le j \le \lceil \alpha \rceil - 1. \end{cases}$$

Let us recall that by a (strong) solution of $(DFP)_{\alpha,\mathcal{A}}$ we mean any continuous E-valued function $t \mapsto u(t), t \ge 0$ such that the term $t \mapsto \mathbf{D}_t^{\alpha}u(t), t \ge 0$ is well defined and continuous, as well as that $(DFP)_{\alpha,\mathcal{A}}$ holds. It is worth noting here that we do not require a priori the closedness of the operator \mathcal{A} henceforth. Denote by $Z_{\alpha}(\mathcal{A})$ the set which consists of those vectors $x \in E$ for which there exists a solution of problem $(DFP)_{\alpha,\mathcal{A}}$. Then $Z_{\alpha}(\mathcal{A})$ is a linear subspace of E. The following is an extension of [**300**, Lemma 2.1] to multivalued linear operator case. The proof is almost straightforward after pointing out that $\mathbf{D}_t^{\alpha} E_{\alpha}(\lambda t^{\alpha}) = \lambda E_{\alpha}(\lambda t^{\alpha}), t \ge 0, \lambda \in \mathbb{C}, \alpha > 0.$

LEMMA 3.6.1. Suppose $\alpha > 0$, $\lambda \in \mathbb{C}$, $x \in E$ and $\lambda x \in Ax$. Then $x \in Z_{\alpha}(A)$ and one solution of $(DFP)_{\alpha,A}$ is given by $u(t) \equiv u(t;x) = E_{\alpha}(\lambda t^{\alpha})x$, $t \ge 0$.

The notion of a (subspace-)hypercyclicity, (subspace-)topological transitivity and (subspace-)topologically mixing property of problem $(DFP)_{\alpha,\mathcal{A}}$ are introduced in the following definition.

DEFINITION 3.6.2. Let $\alpha > 0$, and let \tilde{E} be a closed linear subspace of E. Then it is said that:

(i) an element $x \in Z_{\alpha}(\mathcal{A}) \cap \tilde{E}$ is a \tilde{E} -hypercyclic vector for $(\text{DFP})_{\alpha,\mathcal{A}}$ iff there exists a strong solution $t \mapsto u(t;x), t \ge 0$ of problem $(\text{DFP})_{\alpha,\mathcal{A}}$ with the property that the set $\{u(t;x): t \ge 0\}$ is a dense subset of \tilde{E} .

Furthermore, we say that the abstract Cauchy problem $(DFP)_{\alpha,\mathcal{A}}$ is:

- (ii) \tilde{E} -topologically transitive iff for every $y, z \in \tilde{E}$ and for every $\varepsilon > 0$, there exist $x \in Z_{\alpha}(\mathcal{A}) \cap \tilde{E}$, a strong solution $t \mapsto u(t;x), t \ge 0$ of problem $(\text{DFP})_{\alpha,\mathcal{A}}$ and $t \ge 0$ such that $d(x,y) < \varepsilon$ and $d(u(t;x),z) < \varepsilon$;
- (iii) \tilde{E} -topologically mixing iff for every $y, z \in \tilde{E}$ and for every $\varepsilon > 0$, there exists $t_0 \ge 0$ such that, for every $t \ge t_0$, there exist $x_t \in Z_{\alpha}(\mathcal{A}) \cap \tilde{E}$ and a strong solution $t \mapsto u(t; x_t), t \ge 0$ of problem $(\text{DFP})_{\alpha,\mathcal{A}}$ such that $d(x_t, y) < \varepsilon$ and $d(u(t; x_t), z) < \varepsilon$.

In the case $\tilde{E} = E$, it is also said that a \tilde{E} -hypercyclic vector for $(DFP)_{\alpha,\mathcal{A}}$ is a hypercyclic vector for $(DFP)_{\alpha,\mathcal{A}}$, and that $(DFP)_{\alpha,\mathcal{A}}$ is topologically transitive, resp. topologically mixing.

In the following theorem, we will reformulate the Desch–Schappacher–Webb and Banasiak–Moszyński criteria for the abstract time-fractional inclusion $(DFP)_{\alpha,\mathcal{A}}$ (cf. [300, Theorem 2.3] and its proof):

THEOREM 3.6.3. Assume $\alpha \in (0,2)$ and there exists an open connected subset Ω of \mathbb{C} which satisfies $\Omega \cap (-\infty, 0] = \emptyset$, $\Omega^{\alpha} := \{\lambda^{\alpha} : \lambda \in \Omega\} \subseteq \sigma_p(\mathcal{A}) \text{ and } \Omega \cap i\mathbb{R} \neq \emptyset$. Let $f : \Omega^{\alpha} \to E$ be an analytic mapping such that $f(\lambda^{\alpha}) \in N(\mathcal{A} - \lambda^{\alpha}) \setminus \{0\}$, $\lambda \in \Omega$ and let $\tilde{E} := \overline{span\{f(\lambda^{\alpha}) : \lambda \in \Omega\}}$. Then the abstract degenerate inclusion (DFP)_{\alpha,\mathcal{A}} is \tilde{E} -topologically mixing.

The assertion of Theorem 2.10.7 can be restated for multivalued linear operators, as well: THEOREM 3.6.4. Suppose that $\alpha > 0$ and $(t_n)_{n \in \mathbb{N}}$ is a sequence of positive reals tending to $+\infty$. If the set $E_{0,\alpha}$, which consists of those elements $y \in Z_{\alpha}(\mathcal{A}) \cap \tilde{E}$ for which there exists a strong solution $t \mapsto u(t; y), t \ge 0$ of problem $(DFP)_{\alpha,\mathcal{A}}$ such that $\lim_{n\to\infty} u(t_n; y) = 0$, is dense in \tilde{E} , and if the set $E_{\infty,\alpha}$, which consists of those elements $z \in Z_{\alpha}(\mathcal{A}) \cap \tilde{E}$ for which there exist a null sequence $(\omega_n)_{n\in\mathbb{N}} \in$ $Z_{\alpha}(\mathcal{A}) \cap \tilde{E}$ and a sequence $(u_n(\cdot; \omega_n))_{n\in\mathbb{N}}$ of strong solutions of problem $(DFP)_{\alpha,\mathcal{A}}$ such that $\lim_{n\to\infty} u(t_n; \omega_n) = z$, is also dense in \tilde{E} , then the problem $(DFP)_{\alpha,\mathcal{A}}$ is \tilde{E} -topologically transitive.

3.6.2. Hypercyclic and topologically mixing properties of abstract degenerate Cauchy problems of first and second order. Concerning linear dynamical properties, we have already seen that the abstract degenerate Cauchy problems of first and second order have numerous peculiarities compared with the abstract degenerate fractional Cauchy problems. The main aim of this subsection is to investigate some of these peculiarities for abstract degenerate inclusions of first and second order.

We start by stating the following simple proposition, which has been already considered in a slightly different context.

- PROPOSITION 3.6.5. (i) Suppose that $\alpha = 1, x \in Z_1(\mathcal{A})$ and the function $t \mapsto u(t; x), t \ge 0$ is a solution of problem $(DFP)_{1,\mathcal{A}}$. Then, for every $s \ge 0, u(s; x) \in Z_1(\mathcal{A})$ and a solution of $(DFP)_{1,\mathcal{A}}$, with initial condition x replaced by u(s; x), is given by $u(t; u(s; x)) := u(t + s; x), t \ge 0$.
- (ii) Suppose that $\alpha = 2, x \in Z_2(\mathcal{A})$ and the function $t \mapsto u(t;x), t \ge 0$ is a solution of $(DFP)_{2,\mathcal{A}}$. Then, for every $s \ge 0, u(s;x) \in Z_2(\mathcal{A})$ and a solution of $(DFP)_{2,\mathcal{A}}$, with initial condition x replaced by u(s;x), is given by $u(t;u(s;x)) := \frac{1}{2}[u(t+s;x) + u(|t-s|;x)], t \ge 0.$

Now we would like to state a result similar to that of S. El Mourchid [163, Theorem 2.1]:

THEOREM 3.6.6. Assume that $\alpha = 1$, $\omega_1, \omega_2 \in \mathbb{R} \cup \{-\infty, \infty\}$, $\omega_1 < \omega_2$, $t_0 > 0$ and $k \in \mathbb{N}$. Let $f_j: (\omega_1, \omega_2) \to E$ be integrable, and let for each $j = 1, \ldots, k$ we have $f_j(s) \in is\mathcal{A}f_j(s)$ for a.e. $s \in (\omega_1, \omega_2)$. Put $\psi_{r,j} := \int_{\omega_1}^{\omega_2} e^{irs} f_j(s) ds$, $r \in \mathbb{R}$, $1 \leq j \leq k$. Put $\tilde{E} := \overline{span\{\psi_{r,j} : r \in \mathbb{R}, 1 \leq j \leq k\}}$. If the operator \mathcal{A} is closed, then the problem $(\text{DFP})_{1,\mathcal{A}}$ is \tilde{E} -topologically mixing.

The assertion of Theorem 2.10.19 can be formulated for multivalued linear operators, as well:

THEOREM 3.6.7. Let \tilde{E} be a closed linear subspace of E, and let $\alpha = 2$.

(i) Suppose that (t_n)_{n∈N} is a sequence of positive reals tending to +∞. Denote by X_{1,Ē} the set which consists of those elements x ∈ Z₂(A) ∩ Ē for which there exists a solution t → u(t; x), t ≥ 0 of problem (DFP)_{2,A} such that u(0; x) = x and lim_{n→∞} u(t_n; x) = lim_{n→∞} u(2t_n; x) = 0. If X_{1,Ē} is dense in Ē, then the problem (DFP)_{2,A} is Ē-topologically transitive.

(ii) Denote by X'_{1,Ē} the set which consists of those elements x ∈ Z₂(A) ∩ Ē for which there exists a strong solution t → u(t; x), t ≥ 0 of problem (DFP)_{2,A} such that u(0; x) = x and lim_{t→+∞} u(t; x) = 0. If X'_{1,Ē} is dense in Ē, then the problem (DFP)_{2,A} is Ē-topologically mixing.

As commented before, Theorem 3.6.3 is no longer true in the case that $\alpha = 2$. If so, then we can pass to the equation of first order with the multivalued linear operator $\begin{bmatrix} 0 & I \\ A & 0 \end{bmatrix}$ and the vector $(f(\lambda^2) \lambda f(\lambda^2))^T$, for λ belonging to an open connected subset Ω of \mathbb{C} intersecting the imaginary axis, and apply Theorem 3.6.3, with $\alpha = 1$, after that (cf. Theorem 3.6.12 below for more details).

The interested reader is encouraged to formulate a version of The Hypercyclicity Criterion for degenerate first order inclusions (cf. Theorem 2.10.18).

3.6.3. Hypercyclic and topologically mixing properties of certain classes of abstract degenerate multi-term fractional differential inclusions. In this subsection, we assume that $n \in \mathbb{N} \setminus \{1\}, \mathcal{A}_1, \ldots, \mathcal{A}_{n-1}, \mathcal{A}$ and \mathcal{B} are multivalued linear operators on E (not necessarily closed), $0 \leq \alpha_1 < \cdots < \alpha_{n-1} < \alpha_n$ and $0 \leq \alpha < \alpha_n$. Fix a number $i \in \mathbb{N}_{m_n-1}^0$. Denote $\mathcal{A}_0 = \mathcal{A}, \alpha_0 = \alpha, m_j = \lceil \alpha_j \rceil$ $(j \in \mathbb{N}_0^n), D_i = \{j \in \mathbb{N}_{n-1} : m_j - 1 \ge i\}$ and $\mathcal{D}_i = \{j \in \mathbb{N}_{n-1}^0 : m_j - 1 \ge i\}$. In Subsection 3.6.3 and Subsection 3.6.4, we will inquire into the hypercyclic and topologically mixing properties of the following abstract degenerate multi-term fractional inclusion:

(378)
$$0 \in \mathcal{B}\mathbf{D}_t^{\alpha_n} u(t) + \sum_{j=1}^{n-1} \mathcal{A}_j \mathbf{D}_t^{\alpha_j} u(t) - \mathcal{A}\mathbf{D}_t^{\alpha} u(t), \quad t \ge 0;$$
$$u^{(k)}(0) = x_k, \quad k \in \mathbb{N}_{m_n-1}^0.$$

In this subsection, we will consider the case in which:

(379)
$$u^{(i)}(0) = x_i = x \text{ and } u^{(k)}(0) = x_k = 0, \quad k \in \mathbb{N}^0_{m_n - 1} \setminus \{i\}$$

Because no confusion seems likely, we will denote such a degenerate inclusion by the same symbol (378). Recall, a (strong) solution of (378) is any continuous *E*valued function $t \mapsto u(t), t \ge 0$ such that the Caputo fractional derivative $\mathbf{D}_t^{\alpha_n} u(t)$ is well-defined, as well as that the initial conditions in (378) hold and there exist continuous sections $a_j(t) \in \sec_c(\mathcal{A}_j \mathbf{D}_t^{\alpha_j} u(t))$ $(0 \le j \le n, t \ge 0)$ such that

$$0 = \sum_{j=1}^{n} a_j(t) - a_0(t), \quad t \ge 0.$$

If (379) holds, then u(t) will be also denoted by $u_i(t) \equiv u_i(t;x)$. We will use the following definition.

DEFINITION 3.6.8. Let \tilde{E} be a closed linear subspace of E. Then it is said that the equation (378) is:

(i) \tilde{E} -hypercyclic iff there exist an element $x \in \tilde{E}$ and a strong solution $t \mapsto u_i(t; x), t \ge 0$ of (378) such that $\{u_i(t; x) : t \ge 0\}$ is a dense subset of \tilde{E} ; such an element is called a \tilde{E} -hypercyclic vector of (378);

- (ii) \tilde{E} -topologically transitive iff for every $y, z \in \tilde{E}$ and for every $\varepsilon > 0$, there exist an element $x \in \tilde{E}$, a strong solution $t \mapsto u_i(t; x), t \ge 0$ of (378) and a number $t \ge 0$ such that $d(x, y) < \varepsilon$ and $d(u_i(t; x), z) < \varepsilon$;
- (iii) \tilde{E} -topologically mixing iff for every $y, z \in \tilde{E}$ and for every $\varepsilon > 0$, there exists $t_0 \ge 0$ such that, for every $t \ge t_0$, there exist an element $x_t \in \tilde{E}$ and a strong solution $t \mapsto u_i(t; x_t), t \ge 0$ of (378), with x replaced by x_t , such that $d(x_t, y) < \varepsilon$ and $d(u_i(t; x_t), z) < \varepsilon$.

In the case $\tilde{E} = E$, it is also said that a \tilde{E} -hypercyclic vector of (378) is a hypercyclic vector of (378) and that (378) is topologically transitive, resp. topologically mixing.

The assertion of Theorem 2.10.9 can be extended to multivalued linear operators as follows:

THEOREM 3.6.9. Suppose that $\emptyset \neq \Omega$ is an open connected subset of $\mathbb{C} \setminus \{0\}$, $f: \Omega \to E \setminus \{0\}$ is an analytic function, $f_j: \Omega \to \mathbb{C} \setminus \{0\}$ is a scalar-valued function $(1 \leq j \leq n), g: \Omega \to E$ satisfies $g(\lambda) \in \mathcal{A}f(\lambda), \lambda \in \Omega$ and

 $g(\lambda) \in f_n(\lambda)\mathcal{B}f(\lambda); \quad g(\lambda) \in f_j(\lambda)\mathcal{A}_jf(\lambda), \qquad \lambda \in \Omega, \ 1 \leq j \leq n-1.$

Suppose, further, that Ω_+ and Ω_- are two non-empty subsets of Ω , and each of them admits a cluster point in Ω . Define $\tilde{E} := \overline{span\{f(\lambda) : \lambda \in \Omega\}}$, $H_i(\lambda, t)$ by (226), and $F_i(\lambda, t) := H_i(\lambda, t)f(\lambda)$, for any $t \ge 0$ and $\lambda \in \Omega$. If

 $\lim_{t \to +\infty} |H_i(\lambda, t)| = +\infty, \quad \lambda \in \Omega_+ \quad and \quad \lim_{t \to +\infty} H_i(\lambda, t) = 0, \quad \lambda \in \Omega_-,$

then (378) is \tilde{E} -topologically mixing. Furthermore, there exist continuous sections $a_{j,i}(\lambda,t) \in \sec_c(\mathcal{A}_jF_i(\lambda,t))$ such that the terms $\mathbf{D}_t^{\alpha_j}a_{j,i}(\lambda,t)$ are well-defined $(0 \leq j \leq n, t \geq 0, \lambda \in \Omega)$ and

$$0 = \sum_{j=1}^{n} \mathbf{D}_{t}^{\alpha_{j}} a_{j,i}(\lambda, t) - \mathbf{D}_{t}^{\alpha} a_{0,i}(\lambda, t), \quad t \ge 0, \ \lambda \in \Omega.$$

3.6.4. *D*-Hypercyclic and *D*-topologically mixing properties of abstract degenerate multi-term fractional differential inclusions. Now we will briefly explain how we can, following the method proposed in Subsection 2.10.4, slightly generalize the notion introduced in the previous three subsections. For the sake of simplicity, we will not consider here the orbits of multilinear mappings.

Denote by $\mathfrak{Z}(\mathfrak{Z}_{uniq})$ the set of all tuples of initial values $\vec{x} = (x_0, x_1, \ldots, x_{m_n-1}) \in E^{m_n}$ for which there exists a (unique) strong solution of problem (378). Then \mathfrak{Z} is a linear subspace of E^{m_n} and $\mathfrak{Z}_{uniq} \subseteq \mathfrak{Z}$. For any $\vec{x} \in \mathfrak{Z}$, we denote by $\mathfrak{S}(\vec{x})$ the set consisting of all strong solutions of problem (378) with the initial value \vec{x} . Assume that $\mathfrak{P}: \mathfrak{Z} \to P(\bigcup_{\vec{x} \in \mathfrak{Z}} \mathfrak{S}(\vec{x}))$ is a fixed mapping satisfying $\emptyset \neq \mathfrak{P}(\vec{x}) \subseteq \mathfrak{S}(\vec{x}), \vec{x} \in \mathfrak{Z}$. Let $\emptyset \neq W \subseteq \mathbb{N}_{m_n}$, let \hat{E}_i be a linear subspace of $E(i \in W)$, and let \tilde{E}, \check{E} be linear subspaces of E^{m_n} . Suppose that the tuple $\vec{\beta} = (\beta_0, \beta_1, \ldots, \beta_{m_n-1}) \in [0, \alpha_n]^{m_n}$ is fixed. Set, with a little abuse of notation in comparision with Subsection 2.10.4,

$$\mathfrak{D} := (\tilde{E}, \check{E}, \{\hat{E}_i : i \in W\}, \vec{\beta}).$$

Denote by $\mathcal{M}_{\mathfrak{D}}$ the set consisting of those tuples $\vec{x} \in \mathfrak{Z}$ for which $\operatorname{Proj}_{i,m_n}(\vec{x}) \in \hat{E}_i$, $i \in W$.

DEFINITION 3.6.10. (cf. also Definition 2.10.20) The abstract Cauchy problem (378) is said to be:

(i) $(\mathfrak{D}, \mathfrak{P})$ -hypercyclic iff there exist a tuple $\vec{x} \in \mathcal{M}_{\mathfrak{D}} \cap \tilde{E}$ and a function $u(\cdot; \vec{x}) \in \mathfrak{P}(\vec{x})$ such that

$$\mathfrak{G} := \left\{ \left((\mathbf{D}_s^{\beta_0} u(s; \vec{x}))_{s=t}, (\mathbf{D}_s^{\beta_1} u(s; \vec{x}))_{s=t}, \dots, (\mathbf{D}_s^{\beta_{m_n-1}} u(s; \vec{x}))_{s=t} \right) : t \ge 0 \right\}$$

is a dense subset of \check{E} ; such a vector is called a $(\mathfrak{D}, \mathfrak{P})$ -hypercyclic vector of problem (378).

- (ii) D-hypercyclic iff it is (D, G)-hypercyclic; any (D, G)-hypercyclic vector of problem (378) will be also called a D-hypercyclic vector of problem (378).
- (iii) $\mathfrak{D}_{\mathfrak{P}}$ -topologically transitive iff for every pair of open non-empty subsets U and V of E^{m_n} satisfying $U \cap \tilde{E} \neq \emptyset$ and $V \cap \check{E} \neq \emptyset$, there exist a tuple $\vec{x} \in \mathcal{M}_{\mathfrak{D}}$, a function $u(\cdot; \vec{x}) \in \mathfrak{P}(\vec{x})$ and a number $t \ge 0$ such that $\vec{x} \in U \cap \tilde{E}$ and

(380)
$$((\mathbf{D}_{s}^{\beta_{0}}u(s;\vec{x}))_{s=t}, (\mathbf{D}_{s}^{\beta_{1}}u(s;\vec{x}))_{s=t}, \dots, (\mathbf{D}_{s}^{\beta_{m_{n-1}}}u(s;\vec{x}))_{s=t}) \in V \cap \check{E}.$$

- (iv) \mathfrak{D} -topologically transitive iff it is $\mathfrak{D}_{\mathfrak{S}}$ -topologically transitive.
- (v) $\mathfrak{D}_{\mathfrak{P}}$ -topologically mixing iff for every pair of open non-empty subsets Uand V of E^{m_n} satisfying $U \cap \tilde{E} \neq \emptyset$ and $V \cap \check{E} \neq \emptyset$, there exists a number $t_0 \ge 0$ such that, for every number $t \ge t_0$, there exist a tuple $\vec{x_t} \in \mathcal{M}_{\mathfrak{D}}$ and a function $u(\cdot; \vec{x_t}) \in \mathfrak{P}(\vec{x_t})$ such that $\vec{x_t} \in U \cap \tilde{E}$ and (380) holds with vector \vec{x} replaced by $\vec{x_t}$ therein.
- (vi) \mathfrak{D} -topologically mixing iff it is $\mathfrak{D}_{\mathfrak{S}}$ -topologically mixing.

REMARK 3.6.11. Let $0 \leq \beta \leq \alpha < 2$, and let the requirements of Theorem 3.6.3 hold (here the notation used to denote the space \tilde{E} is slightly different from that used in the formulation of above-mentioned theorem). Then the consideration from Remark 2.10.21(ii) shows that the problem (DFP)_{α} is $\mathfrak{D}_{\mathfrak{P}}$ -topologically mixing, provided that $\vec{\beta} = (\beta, \beta)$, $W = \{1\}$, $\hat{E}_1 = \overline{span\{f(\lambda^{\alpha}) : \lambda \in \Omega\}}$, $\tilde{E} = \hat{E}_1 \times \{0\}$, $\check{E} = \{(z, z) : z \in \hat{E}_1\}$ and $\mathfrak{P}((\sum_{i=1}^m \alpha_i f(\lambda_i^{\alpha}), 0)) = \{\sum_{i=1}^m \alpha_i E_{\alpha}(\cdot^{\alpha} \lambda_i^{\alpha}) f(\lambda_i^{\alpha})\} (m \in \mathbb{N}, \alpha_i \in \mathbb{C}, \lambda_i \in \Omega \text{ for } 1 \leq i \leq m)$. By assuming some extra conditions, a similar assertion can be proved for a general problem (378) (cf. Remark 2.10.21(iii)).

The conjugacy lemma stated in Theorem 2.10.23 admits a very simple reformulation in our context. Using the proof of Theorem 2.10.24 and the usual matrix conversion of abstract higher-order differential equations with integer order derivatives into the first order matrix differential equation, we can simply verify the validity of following analogue of Theorem 2.10.24.

THEOREM 3.6.12. Let $\alpha_i = i$ for all $i \in \mathbb{N}_n$, let Ω be an open non-empty subset of \mathbb{C} intersecting the imaginary axis, and let $f: \Omega \to E$ be an analytic mapping satisfying

$$0 \in \left(\lambda^{\alpha_n} \mathcal{B} + \sum_{i=1}^{n-1} \lambda^{\alpha_i} \mathcal{A}_i - \mathcal{A}\right) f(\lambda), \quad \lambda \in \Omega.$$

Set $\vec{x_{\lambda}} := [f(\lambda) \ \lambda f(\lambda) \ \dots \ \lambda^{n-1} f(\lambda)]^T \ (\lambda \in \Omega), \ E_0 := span\{\vec{x_{\lambda}} : \lambda \in \Omega\}, \ \tilde{E} := \tilde{E}_0, \ \vec{\beta} := (0, 1, \dots, n-1), \ W := \mathbb{N}_n \ and \ \hat{E}_i := span\{f(\lambda) : \lambda \in \Omega\}, \ i \in W.$ Then $\vec{x_{\lambda}} \in \mathfrak{M}_{\mathfrak{D}}, \ \lambda \in \Omega$ and the abstract Cauchy problem (378) is $\mathfrak{D}_{\mathfrak{P}}$ -topologically mixing provided that $\sum_{j=1}^q e^{\lambda_j \cdot f}(\lambda_j) \in \mathfrak{P}(\sum_{j=1}^q \vec{x_{\lambda_j}}) \ for \ any \ \sum_{j=1}^q \vec{x_{\lambda_j}} \in E_0 \ (q \in \mathbb{N}; \ \lambda_j \in \Omega, \ 1 \leq j \leq q).$

As observed in Remark 2.10.25(iii), Theorem 3.6.12 cannot be so simply reformulated for the abstract degenerate multi-term inclusion (378), provided that there exists an index $i \in \mathbb{N}_n$ such that $\alpha_i \notin \mathbb{N}$. Examples already given in [292, Chapter 3] and Section 2.10 can serve for illustration of our theoretical results. Now we would like to present some new elaborate examples in support of Theorem 3.6.3, Theorem 3.6.9 and Theorem 3.6.12.

EXAMPLE 3.6.13. (i) Suppose that $E := C^2(\mathbb{R})$ is equipped with the usual Fréchet topology, $0 < \alpha < 2$, $m \in C(\mathbb{R})$ and m(x) > 0, $x \in \mathbb{R}$. For any $\lambda \in \mathbb{C}$, we denote by $\{f_{\lambda}^1(x), f_{\lambda}^2(x)\}$ the fundamental set of solutions of ordinary differential equation $y'' = \lambda m(x)y$. Using the elementary theory of linear ordinary differential equations, and direct computation of matrix exponential

$$e^{x\left[-\lambda m(x)^{0}\right]}, \quad x \in \mathbb{R}, \ \lambda \in \mathbb{C} \smallsetminus (-\infty, 0],$$

we can simply prove that for any arbitrarily chosen open connected subset Ω of $\mathbb{C} \setminus (-\infty, 0]$ satisfying that $\Omega \cap \{e^{\pm it\alpha/2} : t \ge 0\} \ne \emptyset$, the mappings $\lambda \mapsto f_{\lambda}^{1}(x) \in E$ and $\lambda \mapsto f_{\lambda}^{2}(x) \in E$ are analytic. Let Ω be such a set. Denote $\tilde{E} := \overline{span\{f_{\lambda}^{i}(x) : \lambda \in \Omega, i = 1, 2\}}$. Then we can apply Theorem 3.6.3 in order to see that the abstract time-fractional Poisson heat equation:

$$\begin{split} \mathbf{D}_t^{\alpha}[m(x)u(t,x)] &= \Delta u(t,x), \quad t \ge 0, \quad x \in \mathbb{R}; \\ m(x)u(0,x) &= \phi(x); \; \left(\frac{\partial}{\partial t}[m(x)u(t,x)]\right)_{t=0} = 0, \; \text{if} \; \alpha > 1, \end{split}$$

is \tilde{E} -topologically mixing, with the meaning clear.

(ii) Suppose that n = 3, $\frac{1}{3} < a < \frac{1}{2}$, $\alpha_3 = 3a$, $\alpha_2 = 2a$, $\alpha_1 = 0$, $\alpha = a$, $c_1 < 0$, $c_2 > 0$ and i = 1. Then the analysis given in [292, Example 3.3.12(iii)], in combination with Theorem 3.6.9, enables one to deduce some results on topologically mixing properties of the following abstract degenerate multi-term inclusion:

$$0 \in \mathbf{D}_t^{3a} u(t) + c_2 \mathbf{D}_t^{2a} u(t) + c_1 \mathbf{D}_t^a u(t) - \mathcal{A}u(t), \quad t > 0,$$

$$u(0) = 0, \quad u'(0) = x, \quad u''(0) = 0,$$

where \mathcal{A} is an MLO and satisfies certain conditions.

(iii) Suppose that **A** is an MLO, Ω is an open non-empty subset of \mathbb{C} intersecting the imaginary axis, $f: \Omega \to E$ is an analytic mapping, $\lambda f(\lambda) \in \mathbf{A}f(\lambda)$, $\lambda \in \Omega$, $P_i(z)$ is a non-zero complex polynomial $(0 \leq i \leq n)$ and

(381)
$$z^{n}P_{n}(z) + \sum_{i=1}^{n-1} z^{i}P_{i}(z) - P_{0}(z) \equiv 0$$

Set $\mathcal{A}_i := P_i(\mathbf{A}), i \in \mathbb{N}_n^0$. Then for any non-zero complex polynomial P(z) we have $P(\lambda)f(\lambda) \in P(\mathbf{A})f(\lambda), \lambda \in \Omega$ so that (381) implies

$$0 \in \left(\lambda^{n} \mathcal{B} + \sum_{i=1}^{n-1} \lambda^{i} \mathcal{A}_{i} - \mathcal{A}\right) f(\lambda), \quad \lambda \in \Omega.$$

Hence, Theorem 3.6.12 is susceptible to applications.

3.7. Perturbation results for abstract degenerate Volterra integro-differential equations

We start this section by observing that the following simple lemma holds for multivalued linear operators in locally convex spaces.

LEMMA 3.7.1. Let \mathcal{A} be an MLO in E, and let $B \in L(E)$. If $\lambda \in \rho(\mathcal{A})$ and $1 \in \rho(B(\lambda - A)^{-1})$, then $\lambda \in \rho(A + B)$ and

$$(\lambda - (\mathcal{A} + B))^{-1} = (\lambda - \mathcal{A})^{-1}(1 - B(\lambda - \mathcal{A})^{-1})^{-1}.$$

PROOF. Clearly,

$$\begin{aligned} (\lambda - \mathcal{A})^{-1} (1 - B(\lambda - \mathcal{A})^{-1})^{-1} &= ((1 - B(\lambda - \mathcal{A})^{-1})(\lambda - \mathcal{A}))^{-1} \\ &= (\lambda - \mathcal{A} - B(\lambda - \mathcal{A})^{-1}(\lambda - \mathcal{A}))^{-1} \\ &\supseteq (\lambda - \mathcal{A} - B)^{-1}. \end{aligned}$$

Therefore, it suffices to show that

$$x \in (\lambda - \mathcal{A} - B)(\lambda - \mathcal{A})^{-1}(1 - B(\lambda - \mathcal{A})^{-1})^{-1}x, \quad x \in E.$$

But, this is an immediate consequence of the fact that $x = (1 - B(\lambda - A)^{-1})^{-1}x - B(\lambda - A)^{-1}$ $B(\lambda - A)^{-1}(1 - B(\lambda - A)^{-1})^{-1}x, x \in E.$

Keeping in mind Lemma 3.7.1, the identity [199, (1.2)] and the argumentation already given in non-degenerate case, the assertions of [292, Theorem 2.6.18-Theorem 2.6.19] can be reformulated for (a, k)-regularized resolvent families in Banach spaces, more or less, without substantial difficulties. The situation is simpler with the assertions of [349, Theorem 4.1, Corollary 4.5], which can be almost straightforwardly reformulated for certain classes of K-convoluted semigroups generated by mutivalued linear operators.

The main problem in transferring [292, Theorem 2.6.3] to (a, k)-regularized resolvent families subgenerated by mutivalued linear operators lies in the fact that it is not clear how one can prove that the operator $I - (\mathcal{A} + B)/\lambda^{\alpha}$, appearing in the final part of the proof of this theorem, is injective for $\operatorname{Re} \lambda > 0$ suff. large and $\hat{k}(\lambda)\tilde{a}(\lambda) \neq 0$ (cf. also [292, Theorem 2.6.5, Corollary 2.6.6-Corollary 2.6.9] for further information on this type of bounded commuting perturbations). Nevertheless, the following illustrative example shows that there exist some situations

396

when we can directly apply [292, Theorem 2.6.3] (here, concretely, one of its most important consequences, [292, Corollary 2.6.6]) in the study of perturbation properties of some well-known degenerate equations of mathematical physics and their fractional analogues:

EXAMPLE 3.7.2. Assume that $n \in \mathbb{N}$ and $iA_j, 1 \leq j \leq n$ are commuting generators of bounded C_0 -groups on $E = L^p(\mathbb{R}^n)$, for some $1 \leq p < \infty$ (possible applications can be given in $L^p(\mathbb{R}^n)_l$ -type spaces, as well; cf. [541]). Set $\mathbf{A} := (A_1, \ldots, A_n)$. Suppose now that $P_1(x)$ and $P_2(x)$ are two non-zero complex polynomials in nvariables and $1 \leq \alpha < 2$; put $N_1 := dg(P_1(x))$ and $N_2 := dg(P_2(x))$. Let $\omega \geq 0$, $N \in \mathbb{N}, r \in (0, N]$, let Q(x) be an r-coercive complex polynomial of degree N, $a \in \mathbb{C} \setminus Q(\mathbb{R}^n)$ and $\gamma = \frac{n}{r} |\frac{1}{p} - \frac{1}{2}| \max\left(N, \frac{N_1 + N_2}{\min(1, \alpha)}\right)$. Suppose that $P_2(x) \neq 0$, $x \in \mathbb{R}^n, P_2(x)$ is an elliptic polynomial, and (70) holds. Then [27, Corollary 8.3.4] yields that $\overline{P_2(\mathbf{A})}^{-1} \in L(E)$ (the violation of this condition has some obvious unpleasant consequences on the existence and uniqueness of solutions of perturbed problems); hence, $\overline{P_1(\mathbf{A})} \ \overline{P_2(\mathbf{A})}^{-1}$ is a closed linear operator in E. Set

$$R_{\alpha}(t) := \left(E_{\alpha} \left(t^{\alpha} \frac{P_1(x)}{P_2(x)} \right) (a - Q(x))^{-\gamma} \right) (\mathbf{A}), \quad t \ge 0.$$

By the foregoing, we have that $(R_{\alpha}(t))_{t\geq 0} \subseteq L(E)$ is a global exponentially bounded $(g_{\alpha}, R_{\alpha}(0))$ -regularized resolvent family generated by $\overline{P_1(\mathbf{A})} \overline{P_2(\mathbf{A})}^{-1}$. Set $Df(x) := \int_{-\infty}^{\infty} \psi(x-y)f(y)dy, f \in E$, where $\psi \in L^1(\mathbb{R}^n)$. Then $D \in L(E)$ and commutes with $\overline{P_1(\mathbf{A})} \overline{P_2(\mathbf{A})}^{-1}$. Applying [292, Corollary 2.6.6], we get that the operator $\overline{P_1(\mathbf{A})} \overline{P_2(\mathbf{A})}^{-1} + D$ generates an exponentially bounded $(g_{\alpha}, R_{\alpha}(0))$ -regularized resolvent family, which can be applied in the study of the following perturbation of the abstract fractional Barenblatt–Zheltov–Kochina equation

$$(\eta \Delta - 1)\mathbf{D}_t^{\alpha} u(t) + \Delta u = \int_{-\infty}^{\infty} \psi(x - y)(\eta \Delta - 1)u(t, y)dy \quad (\eta > 0),$$

equipped with the usual initial conditions. We can similarly treat the following perturbation of abstract Boussinesq equation of second order

$$(\sigma^2 \Delta - 1)u_{tt} + \gamma^2 \Delta u = \int_{-\infty}^{\infty} \psi(x - y)(\sigma^2 \Delta - 1)u(t, y)dy \quad (\sigma > 0, \ \gamma > 0).$$

We shall present one more example in support of use of perturbation theory for abstract non-degenerate differential equations (a similar approach works in the analysis of analytical solutions of perturbed abstract fractional Barenblatt–Zheltov– Kochina equations in finite domains):

EXAMPLE 3.7.3. In Example 3.2.17, we have already considered the following fractional analogue of Benney–Luke equation:

$$(P)_{\eta,f}: \begin{cases} (\lambda - \Delta) \mathbf{D}_t^{\eta} u(t,x) = (\alpha \Delta - \beta \Delta^2) u(t,x) + f(t,x), & t \ge 0, \ x \in \Omega, \\ \left(\frac{\partial^k}{\partial t^k} u(t,x)\right)_{t=0} = u_k(x), & x \in \Omega, \ 0 \le k \le \lceil \eta \rceil - 1, \\ u(t,x) = \Delta u(t,x) = 0, & t \ge 0, \ x \in \partial\Omega, \end{cases}$$

where $\emptyset \neq \Omega \subseteq \mathbb{R}^n$ is a bounded domain with smooth boundary. Δ is the Dirichlet Laplacian in $E = L^2(\Omega)$, acting with domain $H^2(\Omega) \cap H^1_0(\Omega)$, $\lambda \in \sigma(\Delta)$, $0 < \eta < 2$ and $\alpha, \beta > 0$. Denote by $\{\lambda_k\} = \sigma(\Delta)$ the eigenvalues of Δ in $L^2(\Omega)$ (recall that $0 < -\lambda_1 \leqslant -\lambda_2 \ldots \leqslant -\lambda_k \leqslant \cdots \to +\infty$ as $\tilde{k} \to \infty$) numbered in nonascending order with regard to multiplicities; by $\{\phi_k\} \subseteq C^{\infty}(\Omega)$ we denote the corresponding set of mutually orthogonal eigenfunctions. Let E_0 be the closed subspace of E consisting of those functions from E that are orthogonal to the eigenfunctions $\phi_k(\cdot)$ for $\lambda_k = \lambda$. Define the closed single-valued linear operator **A** in E_0 by its graph: $\mathbf{A} = \{(f, q) \in E_0 \times E_0 : (\lambda - \lambda_k) \langle q, \phi_k \rangle = (\alpha \lambda_k - \beta \lambda_k^2) \langle f, \phi_k \rangle \text{ for all } k \in \mathbb{C} \}$ \mathbb{N} with $\lambda_k \neq \lambda$. Then the operator **A** generates an exponentially bounded, analytic (q_n, I) -regularized resolvent family of angle $\theta \equiv \min((\pi/\eta) - (\pi/2), \pi/2)$. Suppose that B is a closed linear operator in E satisfying that there exists a number a > 0 such that for all sufficiently small numbers b > 0 we have $D(\mathbf{A}) \subset D(B)$ and $||Bf|| \leq a||f|| + b||\mathbf{A}f||$, $f \in D(\mathbf{A})$. Applying [61, Theorem 2.25] and our previous analyses, we get that the problem $(P)_{\eta,B,f}$, obtained by replacing the term f(t,x) on the right-hand side of the first equation of problem $(P)_{n,f}$ by $(\lambda - \Delta)Bu(t, x) + f(t, x)$, has a unique solution provided that $x_0 \in D(\Delta^2) \cap E_0$, $x_1 \in D(\Delta) \cap E_0, \text{ if } \eta > 1, \sum_{k|\lambda_k \neq \lambda} \frac{\langle f(\cdot), \phi_k \rangle}{\lambda - \lambda_k} \phi_k = h \in W^{1,1}_{loc}([0,\infty) : E_0) \text{ satisfies}$

$$t \mapsto \sum_{k|\lambda_k \neq \lambda} (\alpha \lambda_k - \beta \lambda_k^2) \Big\langle \frac{d}{dt} (g_\eta * h)(t), \phi_k \Big\rangle \phi_k \in L^1_{loc}([0,\infty) : E_0),$$

 $B\phi_k = 0$ for $\lambda = \lambda_k$, and the condition (Q) holds. Finally, we would like to observe that V. E. Fedorov and O. A. Ruzakova have analyzed in [219, Section 5], by using a completely different method, perturbations of degenerate differential equations of first order involving polynomials of elliptic selfadjoint operators, as well as that V. E. Fedorov and L. V. Borel have analyzed in [207] a class of loaded degenerate integro-differential equations proving particularly some results on the modifed Benney–Luke equation of the form

$$(\lambda - \Delta)u_t(t, x) = (\alpha \Delta - \beta \Delta^2)u(t, x) + \int_0^T \int_\Omega k(x, y, t, s)u(y, s)dy \,d\mu(s), \quad t \in [0, T],$$

where $\mu: [0,T] \to \mathbb{R}$ is a function of bounded variation. Observe that the analysis contained in Example 3.2.17 together with [291, Theorem A.12] enables one to look into the well-posedness of the following integral equation

$$(\lambda - \Delta)u_t(t, x) = (\alpha \Delta - \beta \Delta^2)(g_\eta * u)(t, x) + (\lambda - \Delta) \int_0^t B(t - s)u(s)ds, \quad t \ge 0,$$

where $B(\cdot)$ satisfies certain properties.

The assertion of [**292**, Theorem 2.6.11] admits an extension in our context. We will give some details of the proof immediately after considering Theorem 3.7.5.

THEOREM 3.7.4. Suppose M > 0, $\omega \ge 0$, the functions |a|(t) and k(t) satisfy (P1), as well as \mathcal{A} is a densely defined, closed subgenerator of an (a, k)-regularized *C*-resolvent family $(R(t))_{t\ge 0}$ which satisfies that, for every seminorm $p \in \mathfrak{B}$, we have $p(R(t)x) \le Me^{\omega t}p(x), x \in E, t \ge 0$. Suppose, further, $C^{-1}B \in L(E)$,

 $BCx = CBx, x \in D(\mathcal{A})$, there exist a locally integrable function b(t) and a number $\omega_0 \ge \omega$ such that |b|(t) satisfies (P1) and $\tilde{b}(\lambda) = \frac{\tilde{a}(\lambda)}{\tilde{b}(\lambda)}, \ \lambda > \omega_0, \ \tilde{k}(\lambda) \neq 0.$ Let $\mu > \omega_0$ and $\gamma \in [0,1)$ be such that

(382)
$$\int_0^\infty e^{-\mu t} p\left(C^{-1}B\int_0^t b(t-s)R(s)x\,ds\right)dt \leqslant \gamma p(x), \quad x \in D(\mathcal{A}), \ p \in \circledast.$$

Then the operator $\mathcal{A}+B$ is a closed subgenerator of an (a, k)-regularized C-resolvent family $(R_B(t))_{t\geq 0}$ which satisfies $p(R_B(t)x) \leq \frac{M}{1-x}e^{\mu t}p(x), x \in E, t \geq 0, p \in \mathbb{R}$ and

$$R_B(t)x = R(t)x + \int_0^t R_B(t-r)C^{-1}B \int_0^r b(r-s)R(s)x \, ds \, dr, \quad t \ge 0, \ x \in D(\mathcal{A}).$$

Furthermore, the equation (272) holds with R(t) replaced by $R_B(t)$ therein.

As observed in [292, Theorem 2.6.12], in many cases we do not have the existence of a function b(t) and a complex number z such that $\tilde{a}(\lambda)/\tilde{k}(\lambda) = \tilde{b}(\lambda) + z$, $\operatorname{Re} \lambda > \omega_1, \ \tilde{k}(\lambda) \neq 0$ (in this case, Theorem 3.7.4 continues to hold with some natural adaptations). The above-mentioned theorem admits an extension in our context, as well. Before we formulate this extension, let us only outline a few relevant details needed for its proof. First of all, suppose that \mathcal{A} is a subgenerator of an (a, k)-regularized C-resolvent family $(R(t))_{t \in [0,\tau)}, l \in \mathbb{N}$ and $x_j \in Ax_{j-1}$ for $1 \leq j \leq l$. Then we have already seen that, for every $t \in [0, \tau)$,

$$R(t)x_0 = k(t)Cx_0 + \sum_{j=1}^{l-1} (a^{*,j} * k)(t)Cx_j + (a^{*,l} * R(\cdot)x_l)(t).$$

In the case that $\tau = \infty$ and the Laplace transform can be applied, the above equation implies that, for certain values of complex parameter λ , we have:

$$\tilde{k}(\lambda)(I - \tilde{a}(\lambda)\mathcal{A})^{-1}x_0 = \tilde{k}(\lambda)Cx_0 + \sum_{j=1}^{l-1}\tilde{a}(\lambda)^j\tilde{k}(\lambda)Cx_j + \tilde{a}(\lambda)^l\tilde{k}(\lambda)(I - \tilde{a}(\lambda)\mathcal{A})^{-1}x_l.$$

If we define the operator family $(S(t))_{t\geq 0}$ as explained below, then the previous equation implies that the identity [292, (180)] continues to hold with the singlevalued operator A replaced by the MLO \mathcal{A} , provided in addition that the number λ in this equation satisfies $\tilde{a}(\lambda) \neq 0$. Furthermore, the identities [292, (181), (183)] also hold, and the assumption $y \in (I - \tilde{a}(\lambda)(\mathcal{A} + B))x$ implies on account of Lemma 1.2.4 and the validity of identity [292, (180)] that $R_B(\lambda)y = (I - \tilde{S}(\lambda))^{-1}\tilde{k}(\lambda)(I - \tilde$ $\tilde{a}(\lambda)\mathcal{A})^{-1}Cy = \tilde{k}(\lambda)Cx$ for $\operatorname{Re}\lambda > 0$ suff. large and $\tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0$. Owing to the condition (i) in Theorem 3.7.5, we have that the operator $\mathcal{A} + B$ is closed and commutes with C. The representation $(I - \tilde{S}(\lambda))^{-1} = \sum_{n=0}^{\infty} [(\frac{1}{\tilde{a}(\lambda)} - \mathcal{A})^{-1} C C^{-1} B]^n$ implies along with the closedness of \mathcal{A} that

$$\tilde{k}(\lambda)Cx \in \left(\frac{1}{\tilde{a}(\lambda)} - (\mathcal{A} + B)\right)(I - \tilde{S}(\lambda))^{-1}\tilde{k}(\lambda)\left(\frac{1}{\tilde{a}(\lambda)} - \mathcal{A}\right)^{-1}Cx, \quad x \in E,$$

and $R(C) \subseteq R(I - \tilde{a}(\lambda)(\mathcal{A} + B))$ for $\operatorname{Re} \lambda > 0$ suff. large and $\tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0$. Now it is clear that the Laplace transform identity [292, (182)] holds with the operator

399

A + B replaced by $\mathcal{A} + B$, provided in addition that the number λ in this equation satisfies $\tilde{a}(\lambda) \neq 0$. After that, we can apply Laplace transform. The following holds:

THEOREM 3.7.5. Suppose M, $M_1 > 0$, $\omega \ge 0$, $l \in \mathbb{N}$ and \mathcal{A} is a closed subgenerator of an (a, k)-regularized C-resolvent family $(R(t))_{t\ge 0}$ such that $p(R(t)x) \le Me^{\omega t}p(x)$, $x \in E$, $t \ge 0$, $p \in \circledast$ and (272) holds. Let |a|(t) and k(t) satisfy (P1), and let the following conditions hold:

(i) $BCx = CBx, x \in D(\mathcal{A}); \text{ if } x = x_0 \in \overline{D(\mathcal{A})}, \text{ then } C^{-1}Bx \in D(\mathcal{A}^l) \text{ and}$ there exists a sequence $(x_j)_{1 \leq j \leq l}$ such that $x_j \in \mathcal{A}x_{j-1}$ for $1 \leq j \leq l$, as well as that:

$$p(Cx_j) \leq M_1 p(x), \quad x \in \overline{D(\mathcal{A})}, \ p \in \mathfrak{B}, \ 0 \leq j \leq l-1, \ and$$

 $p(x_l) \leq M_1 p(x), \quad x \in \overline{D(\mathcal{A})}, \ p \in \mathfrak{B}.$

(ii) There exist a locally integrable function b(t) and a complex number z such that |b|(t) satisfies (P1) and

$$\tilde{a}(\lambda)^{l+1}/\tilde{k}(\lambda) = \tilde{b}(\lambda) + z, \quad \operatorname{Re}\lambda > \max(\omega, \operatorname{abs}(|a|), \operatorname{abs}(k)), \ \tilde{k}(\lambda) \neq 0.$$

(iii)
$$\lim_{\lambda \to +\infty} \int_0^\infty e^{-\lambda t} |a(t)| dt = 0$$
 and $\lim_{\lambda \to +\infty} \int_0^\infty e^{-\lambda t} |b(t)| dt = 0$.

Define, for every $x = x_0 \in \overline{D(\mathcal{A})}$ and $t \ge 0$,

$$S(t)x := \sum_{j=0}^{l-1} a^{*,j+1}(t)Cx_j + \int_0^t b(t-s)R(s)x_l ds + zR(t)x_l,$$

where $(x_j)_{1 \leq j \leq l}$ is an arbitrary sequence satisfying the assumptions prescribed in (i). Then, for every $x \in E$, there exists a unique solution of the integral equation

(383)
$$R_B(t)x = R(t)x + (S * R_B)(t)x, \quad t \ge 0;$$

furthermore, $(R_B(t))_{t\geq 0}$ is an (a,k)-regularized C-resolvent family with a closed subgenerator $\mathcal{A} + B$, there exist $\mu \geq \max(\omega, \operatorname{abs}(|a|), \operatorname{abs}(k))$ and $\gamma \in [0,1)$ such that $p(R_B(t)x) \leq \frac{M}{1-\gamma}e^{\mu t}p(x), x \in E, t \geq 0, p \in \mathfrak{B}$ and (272) holds with R(t)replaced by $R_B(t)$ therein.

REMARK 3.7.6. It is worth noting that Theorem 3.7.4 continues to hold, with appropriate changes, in the case that B is not necessarily bounded operator from $\overline{D(\mathcal{A})}$ into E. More precisely, suppose that E is complete, B is a closed linear operator in E, and the requirements of Theorem 3.7.4 hold with the condition $C^{-1}B \in L(E)$ replaced by that $D(\mathcal{A}) \subseteq D(C^{-1}B)$ and the mapping $t \mapsto C^{-1}B(b*R)(t)x, t \ge 0$ is well-defined, continuous and Laplace transformable for all $x \in D(\mathcal{A})$. Then the final conclusions established in Theorem 3.7.4 remain valid; here, it is only worth noting that the closedness of the operator $\mathcal{A} + B$ can be proved (cf. [292, Remark 2.6.13] for more details, especially, the condition (\natural) therein) by using the inclusion $(I - \tilde{S}(\lambda))^{-1}(I - \tilde{a}(\lambda)(\mathcal{A} + B))^{-1}x \subseteq (I - \tilde{a}(\lambda)\mathcal{A})x, x \in D(\mathcal{A})$. REMARK 3.7.7. The method proposed in the proofs of [459, Theorem 1.2, Theorem 2.3] and [292, Theorem 2.6.13] enables one to deduce some results on the well-posedness of perturbed abstract Volterra inclusion:

(384)
$$u(t) \in f(t) + (a + a * k)(t) * Au(t) + (b * u)(t), \quad t \in [0, \tau),$$

provided that \mathcal{A} is a closed subgenerator of an exponentially equicontinuous (a, k)regularized *C*-resolvent family $(R(t))_{t \ge 0}$, $b, k \in L^1_{loc}([0,\infty))$ and $f \in C([0,\infty))$. The starting point is the observation that the regularized resolvent families for (384) satisfy the integral equations like [**459**, (1.28)] or (383).

Now it is quite easy to formulate the following extension of [292, Corollary 2.6.15].

COROLLARY 3.7.8. Suppose $M, M_1 > 0, \ \alpha \ge 0, \ \alpha > 0, \ \beta \ge 0, \ \mathcal{A}$ is a closed subgenerator of a $(g_{\alpha}, g_{\alpha\beta+1})$ -regularized C-resolvent family $(R(t))_{t\ge 0}$ satisfying $p(R(t)x) \le Me^{\omega t}p(x), \ x \in E, \ t \ge 0, \ p \in \mathfrak{B}$ and (272) holds with $a(t) = g_{\alpha}(t)$ and $k(t) = g_{\alpha\beta+1}(t)$. Assume exactly one of the following conditions:

- (i) $\alpha 1 \alpha \beta \ge 0$, BCx = CBx, $x \in D(\mathcal{A})$, and (a) \lor (b), where:
 - (a) $p(C^{-1}Bx) \leq M_1 p(x), x \in \overline{D(\mathcal{A})}, p \in \circledast$.
 - (b) E is complete, (382) holds, $D(\mathcal{A}) \subseteq D(C^{-1}B)$, as well as the mapping $t \mapsto C^{-1}B(b * R)(t)x$, $t \ge 0$ is well-defined, continuous and Laplace transformable for all $x \in D(\mathcal{A})$.
- (ii) $\alpha 1 \alpha \beta < 0$, BCx = CBx, $x \in D(\mathcal{A})$, $l = \lceil \frac{\alpha \beta + 1 \alpha}{\alpha} \rceil$ and (i) of Theorem 3.7.5 holds.

Then there exist $\mu > \omega$ and $\gamma \in [0,1)$ such that $\mathcal{A} + B$ is a closed subgenerator of a $(g_{\alpha}, g_{\alpha\beta+1})$ -regularized C-resolvent family $(R_B(t))_{t\geq 0}$ satisfying $p(R_B(t)x) \leq \frac{M}{1-\gamma}e^{\mu t}p(x), x \in E, t \geq 0, p \in \mathfrak{B}$, and (272) holds with R(t) replaced by $R_B(t)$ therein, with $a(t) = g_{\alpha}(t)$ and $k(t) = g_{\alpha\beta+1}(t)$.

Observe that the local Hölder continuity is an example of the property that is stable under perturbations described in the previous three assertions (cf. [292, Remark 2.6.14] for more details, and [292, Remark 2.6.16] for inheritance of analytical properties under perturbations described in Corollary 3.7.8).

Now we would like to provide an illustrative application of non-degenerate version of Theorem 3.7.4 (possibilities for work exist even in the case that $C \neq I$ and E is not a Banach space).

EXAMPLE 3.7.9. In a great number of research papers, C. Lizama and his coauthors have analyzed possibilities to apply Theorem 1.4.12(i) in the qualitative analysis of abstract non-degenerate Volterra integro-differential equations. The main goal of this example is to show how the above-mentioned result, combined with Theorem 3.7.4, can be used for proving some sufficient conditions for generation of specific classes of (a, k)-regularized resolvent families. We continue the analysis of R. Ponce and M. Warma [456] here.

Suppose that $(E, \|\cdot\|)$ is a Banach space, $\alpha \in \mathbb{R}$, $\alpha \neq 0$, $\beta \ge 0$, $0 < \zeta \le 1$ and $\omega \in \mathbb{R}$. Let any of the following two conditions hold:

(i) $\alpha > 0$, \mathcal{A} is an MLO satisfying

(385) $\omega + \Sigma_{\zeta \pi/2} \subseteq \rho(\mathcal{A}) \text{ and } ||R(\lambda : \mathcal{A})|| = O(|\lambda - \omega|^{-1}), \quad \lambda \in \omega + \Sigma_{\zeta \pi/2}.$ (ii) $\alpha < 0, \ \alpha + \beta^{\zeta} \ge |\alpha|$ and \mathcal{A} is an MLO satisfying (385).

Then it is well-known that the operator $\mathcal{A}_{|\overline{D(\mathcal{A})}}$ is single-valued, linear and densely defined in the Banach space $\overline{D(\mathcal{A})}$, as well as that (385) holds with the operator \mathcal{A} replaced with the operator $\mathcal{A}_{|\overline{D(\mathcal{A})}}$; see e.g. [444, Lemma 4.1]. Set a(t) := 1 + (1 * k)(t), where $k(t) := \alpha e^{-\beta t} g_{\zeta}(t)$. Owing to the proof of [456, Theorem 2.1], we get that $\mathcal{A}_{|\overline{D(\mathcal{A})}}$ generates an exponentially bounded (a, 1)-regularized resolvent family $(S_{\omega}(t))_{t\geq 0}$ in $\overline{D(\mathcal{A})}$, provided that $\omega = 0$. In the general case $\omega \neq 0$, Theorem 3.7.4 and decomposition $\mathcal{A}_{|\overline{D(\mathcal{A})}} = (\mathcal{A}_{|\overline{D(\mathcal{A})}} - \omega I_{|\overline{D(\mathcal{A})}}) + \omega I_{|\overline{D(\mathcal{A})}}$ show that $\mathcal{A}_{|\overline{D(\mathcal{A})}}$ generates an exponentially bounded (a, 1)-regularized resolvent family $(S(t))_{t\geq 0}$ in $\overline{D(\mathcal{A})}$, as well. This extends the assertion of [456, Corollary 2.2], and can be applied in the analysis of Poisson heat equation with memory, in the space $H^{-1}(\Omega)$; see e.g. [199, Example 3.3]. The proof of [456, Theorem 2.3] works in degenerate case and we may conclude the following: Let $\alpha \neq 0$, $\beta \geq 0$, $0 < \zeta < \tilde{\zeta} \leq 1$, $\omega < 0$ and $\beta + \omega \leq 0$. If (i) holds with the number ζ replaced with the number $\tilde{\zeta}$ therein, then $||S(t)|| = O((1 + \alpha \omega t^{\zeta+1})e^{-(\beta - (\alpha \omega)^{1/(\zeta+1)})t}), t \geq 0$.

The proof of following result is very similar to that of [292, Theorem 2.6.22] and, because of that, we will skip it.

THEOREM 3.7.10. Let k(t) and |a|(t) satisfy (P1). Suppose $\delta \in (0, \pi/2], \omega \ge \max(0, \operatorname{abs}(|a|), \operatorname{abs}(k))$, there exist analytic functions $\hat{k} \colon \omega + \sum_{\frac{\pi}{2}+\delta} \to \mathbb{C}$ and $\hat{a} \colon \omega + \sum_{\frac{\pi}{2}+\delta} \to \mathbb{C}$ such that $\hat{k}(\lambda) = \tilde{k}(\lambda)$, $\operatorname{Re} \lambda > \omega$, $\hat{a}(\lambda) = \tilde{a}(\lambda)$, $\operatorname{Re} \lambda > \omega$ and $\hat{k}(\lambda)\hat{a}(\lambda) \neq 0$, $\lambda \in \omega + \sum_{\frac{\pi}{2}+\delta}$. Let \mathcal{A} be a closed subgenerator of an analytic (a, k)-regularized C-resolvent family $(R(t))_{t\geq 0}$ of angle δ , and let (272) hold. Suppose that, for every $\eta \in (0, \delta)$, there exists $c_{\eta} > 0$ such that

$$p(e^{-\omega \operatorname{Re} z} R(z)x) \leq c_n p(x), \quad x \in E, \ p \in \mathfrak{B}, \ z \in \Sigma_n,$$

as well as $b, c \ge 0$, B is a linear operator satisfying $D(C^{-1}AC) \subseteq D(B)$, BCx = CBx, $x \in D(C^{-1}AC)$ and

$$p(C^{-1}Bx) \leq bp(y) + cp(x), \text{ whenever } (x,y) \in C^{-1}\mathcal{A}C, \quad p \in \circledast.$$

Assume that at least one of the following conditions holds:

- (i) \mathcal{A} is densely defined, the numbers b and c are sufficiently small, there exists $|C|_{\circledast} > 0$ such that $p(Cx) \leq |C|_{\circledast}p(x)$, $x \in E$, $p \in \circledast$ and, for every $\eta \in (0, \delta)$, there exists $\omega_{\eta} \geq \omega$ such that $|\hat{k}(\lambda)^{-1}| = O(|\lambda|)$, $\lambda \in \omega_{\eta} + \Sigma_{\frac{\pi}{2} + \eta}$ and $|\hat{a}(\lambda)/\hat{k}(\lambda)| = O(|\lambda|)$, $\lambda \in \omega_{\eta} + \Sigma_{\frac{\pi}{2} + \eta}$.
- (ii) \mathcal{A} is densely defined, the number b is sufficiently small, there exists $|C|_{\circledast} > 0$ such that $p(Cx) \leq |C|_{\circledast} p(x)$, $x \in E$, $p \in \circledast$ and, for every $\eta \in (0, \delta)$, there exists $\omega_{\eta} \geq \omega$ such that $|\hat{k}(\lambda)^{-1}| = O(|\lambda|)$, $\lambda \in \omega_{\eta} + \Sigma_{\frac{\pi}{2}+\eta}$ and $\hat{a}(\lambda)/(\lambda\hat{k}(\lambda)) \to 0$, $|\lambda| \to \infty$, $\lambda \in \omega_{\eta} + \Sigma_{\frac{\pi}{2}+\eta}$.

- (iii) \mathcal{A} is densely defined, the number c is sufficiently small, b = 0 and, for every $\eta \in (0, \delta)$, there exists $\omega_{\eta} \ge \omega$ such that $|\hat{a}(\lambda)/\hat{k}(\lambda)| = O(|\lambda|)$, $\lambda \in \omega_{\eta} + \Sigma_{\frac{\pi}{2} + \eta}$.
- (iv) b = 0 and, for every $\eta \in (0, \delta)$, there exists $\omega_{\eta} \ge \omega$ such that $\hat{a}(\lambda)/(\lambda \hat{k}(\lambda)) \to 0$, $|\lambda| \to \infty$, $\lambda \in \omega_{\eta} + \Sigma_{\frac{\pi}{2} + \eta}$.

Then $C^{-1}(C^{-1}\mathcal{A}C + B)C$ is the integral generator of an exponentially equicontinuous, analytic (a,k)-regularized C-resolvent family $(R_B(t))_{t\geq 0}$ of angle δ , which satisfies $R_B(z)[C^{-1}(C^{-1}\mathcal{A}C + B)C] \subseteq [C^{-1}(C^{-1}\mathcal{A}C + B)C]R_B(z), z \in \Sigma_{\delta}$ and the following condition:

$$\begin{aligned} \forall \eta \in (0,\delta) \ \exists \omega_{\eta}' > 0 \ \exists M_{\eta} > 0 \ \forall p \in \circledast : \\ p(R_B(z)x) \leqslant M_{\eta} e^{\omega_{\eta}' \operatorname{Re} z} p(x), \quad x \in E, \ z \in \Sigma_{\eta}. \end{aligned}$$

Furthermore, in cases (iii) and (iv), the above remains true with the operator $C^{-1}(C^{-1}\mathcal{A}C+B)C$ replaced by $C^{-1}\mathcal{A}C+B$.

In this book, we will not discuss possibilities to generalize results on rank 1-perturbations [26] and time-dependent perturbations [547] to (a, k)-regularized *C*-resolvent families subgenerated by mutivalued linear operators; for more details about non-degenerate case, we refer the reader to [292, Lemma 2.6.26–Theorem 2.6.33] and [292, Theorem 2.6.34, Corollary 2.6.35–Corollary 2.6.38, Theorem 2.6.40, Corollary 2.6.42–Corollary 2.6.45]. In the following theorem, we will extend the assertions of time-dependent perturbations [61, Theorem 2.26] and [292, Theorem 2.6.46(i)] to multivalued linear operators. The proof is very similar to that of Theorem 3.7.13 below and, because of that, we will skip it.

THEOREM 3.7.11. Suppose $\alpha \ge 1$, $M \ge 1$, $\omega \ge 0$ and \mathcal{A} is a closed subgenerator of a (local) (g_{α}, C) -regularized resolvent family $(S_{\alpha}(t))_{t\in[0,\tau)}$ satisfying $p(S_{\alpha}(t)x) \le$ $Me^{\omega t}p(x), t \in [0,\tau), x \in E, p \in \circledast$ and (272) with R(t) and a(t) replaced by $S_{\alpha}(t)$ and $g_{\alpha}(t)$, respectively. Let $(B(t))_{t\in[0,\tau)} \subseteq L(E)$, $R(B(t)) \subseteq R(C), t \in [0,\tau)$ and $C^{-1}B(\cdot) \in C([0,\tau) : L(E))$. Assume that $t \mapsto C^{-1}f(t), t \in [0,\tau)$ is a locally integrable E-valued mapping such that the mapping $t \mapsto (d/dt)C^{-1}f(t)$ is defined for a.e. $t \in [0,\tau)$ and locally integrable on $[0,\tau)$ (in the sense of [410, Definition 4.4.3]). Then there exists a unique solution of the integral Volterra inclusion:

(386)
$$u(t,f) \in f(t) + \mathcal{A} \int_0^t g_\alpha(t-s)u(s,f)ds + \int_0^t g_\alpha(t-s)B(s)u(s,f)ds, \quad t \in [0,\tau);$$

here, by a solution of (386) we mean any continuous function $u \in C([0,\tau): E)$ such that there exists a continuous section $u_{\mathcal{A},\alpha,f}(t) \in \sec_c(\mathcal{A}\int_0^t g_\alpha(t-s)u(s,f)ds)$ for $t \in [0,\tau)$, with the property that $u(t,f) = f(t)+u_{\mathcal{A},\alpha,f}(t)+\int_0^t g_\alpha(t-s)B(s)u(s,f)ds$, $t \in [0,\tau)$. The solution u(t,f) is given by $u(t,f) := \sum_{n=0}^{\infty} S_{\alpha,n}(t)$, $t \in [0,\tau)$, where we define $S_{\alpha,n}(t)$ recursively by

(387)
$$S_{\alpha,0}(t) := S_{\alpha}(t)C^{-1}f(0) + \int_0^t S_{\alpha}(t-s)(C^{-1}f)'(s)ds, \quad t \in [0,\tau)$$

403

and

(388)
$$S_{\alpha,n}(t) := \int_0^t \int_0^{t-\sigma} g_{\alpha-1}(t-\sigma-s)S_\alpha(s)C^{-1}B(\sigma)S_{\alpha,n-1}(\sigma)ds\,d\sigma, \quad t \in [0,\tau).$$

Denote, for every $T \in (0, \tau)$ and $p \in \circledast$, $K_{T,p} := \max_{t \in [0,T]} p(C^{-1}B(t))$ and $F_{T,p} := p(C^{-1}f(0)) + \int_0^T e^{-\omega s} p((C^{-1}f)'(s)) ds$. Then, for every $p \in \circledast$, we have:

$$p(u(t,f)) \leq M e^{\omega t} E_{\alpha}(M K_{T,p} t^{\alpha}) F_{T,p}, \quad t \in [0,T]$$

and

$$p(u(t,f) - S_{\alpha,0}(t)) \leqslant M e^{\omega t} (E_{\alpha}(MK_{T,p}t^{\alpha}) - 1) F_{T,p}, \quad t \in [0,T].$$

Motivated by the research of A. Favini and A. Yagi [199, Chapter III], we introduce the following definition (for the sake of convenience, we shall work only in Banach spaces).

DEFINITION 3.7.12. Suppose that $(E, \|\cdot\|)$ is a Banach space, $\alpha > 0, \zeta \in (0, 1), 0 < \tau \leq \infty, \mathcal{A}$ is an MLO in $E, C \in L(E)$ is injective and $C\mathcal{A} \subseteq \mathcal{A}C$. Then it is said that a strongly continuous operator family $(R(t))_{t \in (0,\tau)} \subseteq L(E)$ is a (g_{α}, C) -regularized resolvent family of growth order ζ , with a subgenerator \mathcal{A} , iff the family $\{t^{\zeta}R(t): t \in (0,\tau)\} \subseteq L(E)$ is bounded, as well as that $R(t)C = CR(t), R(t)\mathcal{A} \subseteq \mathcal{A}R(t)$ $(t \in (0,\tau))$ and

$$\int_0^t g_\alpha(t-s)R(s)y\,ds = R(t)x - Cx, \text{ whenever } t \in (0,\tau) \text{ and } (x,y) \in \mathcal{A}.$$

It directly follows from definition that, for every $\nu > \zeta$, the operator family $((g_{\nu} * R)(t))_{t \in [0,\tau)}$ is a g_{ν} -times integrated (g_{α}, C) -regularized resolvent family with a subgenerator \mathcal{A} . Consider now the abstract integral Volterra inclusion (386) in which we are finding solutions defined on the finite time interval $(0,\tau)$, with $B(\cdot) \in C([0,\tau) : L(E))$ and $f \in C((0,\tau) : E)$. By a solution of (386) on $(0,\tau)$ we mean any continuous function $u \in C((0,\tau) : E)$ such that the mapping $t \mapsto u(t, f)$, $t \in (0,\tau)$ is locally integrable at the point t = 0 and there exists a continuous section $u_{\mathcal{A},\alpha,f}(t) \in \sec_c(\mathcal{A}\int_0^t g_{\alpha}(t-s)u(s,f)ds)$ for $t \in (0,\tau)$. The subsequent theorem is very similar to [**61**, Theorem 2.26] and [**292**, Theorem 2.6.46(i)]. For the sake of clarity, we will include the proof.

THEOREM 3.7.13. Suppose $\alpha \ge 1$, $M \ge 1$, $\omega \ge 0$ and \mathcal{A} is a closed subgenerator of a (local) (g_{α}, C) -regularized resolvent family $(S_{\alpha}(t))_{t\in(0,\tau)}$ of growth order $\zeta \in (0,1)$, satisfying that $||g_{\zeta+1}(t)S_{\alpha}(t)|| \le Me^{\omega t}$, $t \in (0,\tau)$ and that (272) holds for $t \in (0,\tau)$ with R(t) and a(t) replaced by $S_{\alpha}(t)$ and $g_{\alpha}(t)$, respectively. Let $(B(t))_{t\in[0,\tau)} \subseteq L(E)$, $R(B(t)) \subseteq R(C)$, $t \in [0,\tau)$ and $C^{-1}B(\cdot) \in C([0,\tau) : L(E))$. Assume that $t \mapsto C^{-1}f(t)$, $t \in [0,\tau)$ is a continuous E-valued mapping such that the mapping $t \mapsto S_{\alpha,0}(t)$, defined by (387), is a solution of problem

$$v(t,f) \in f(t) + \mathcal{A} \int_0^t g_\alpha(t-s)v(s,f)ds, \quad t \in (0,\tau)$$

and satisfies $||g_{\zeta+1}(t)S_{\alpha,0}(t)|| \leq Me^{\omega t}$, $t \in (0,\tau)$ (cf. [199, Theorem 3.7-Theorem 3.13] for more details). Then there exists a unique solution u(t, f) of the abstract integral Volterra inclusion (386) on the interval $(0,\tau)$. Moreover, the solution u(t, f) is given by $u(t, f) := \sum_{n=0}^{\infty} S_{\alpha,n}(t)$, $t \in (0,\tau)$, where we define $S_{\alpha,n}(t)$ for $t \in (0,\tau)$ recursively by (388). Denote, for every $T \in (0,\tau)$, $K_T := \max_{t \in [0,T]} ||C^{-1}B(t)||$. Then there exists a constant $c_{\alpha,\gamma} > 0$ such that:

(389)
$$\|u(t,f)\| \leqslant c_{\alpha,\gamma} e^{\omega t} t^{-\eta} E_{\alpha-\zeta,1-\zeta}(MK_T t^{\alpha-\zeta}), \quad t \in (0,T]$$

and

(390)
$$||u(t,f) - S_{\alpha,0}(t)|| \leq c_{\alpha,\gamma} e^{\omega t} t^{-\eta} (E_{\alpha-\zeta,1-\zeta}(MK_T t^{\alpha-\zeta}) - 1), \quad t \in (0,T].$$

PROOF. It is very simple to prove that there exists a constant $c_{\alpha,\gamma} > 0$ such that:

$$\|S_{\alpha,n}(t)\| \leq M^{n+1} K_T^n e^{\omega t} (g_{\eta+1}(t))^{-1} \frac{t^{(\alpha-\eta)n}}{\Gamma((\alpha-\eta)n+1-\eta)}, \quad t \in (0,T], \ n \in \mathbb{N}_0,$$

which implies that the series $\sum_{n=0}^{\infty} S_{\alpha,n}(t)$ converges uniformly on compact subsets of $[\varepsilon, T]$ and (389)–(390) hold $(0 < \varepsilon < T)$. Clearly, $u(t, f) = S_{\alpha,0}(t) + \int_0^t (g_{\alpha-1} * S_{\alpha})(t-s)C^{-1}B(s)u(s, f)ds, t \in [0, T]$. With the help of Theorem 1.2.3, this implies

$$\begin{split} u(t,f) &\in f(t) + \mathcal{A} \int_{0}^{t} g_{\alpha}(t-s)v(s,f)ds + [g_{\alpha-1} * S_{\alpha} * C^{-1}B(\cdot)u(\cdot,f)](t) \\ &\in f(t) + \mathcal{A} \int_{0}^{t} g_{\alpha}(t-s)v(s,f)ds + \mathcal{A} \int_{0}^{t} g_{\alpha}(t-s)(g_{\alpha-1} * S_{\alpha} * C^{-1}B(\cdot)u(\cdot,f))(s)ds \\ &\quad + [g_{\alpha-1} * C * C^{-1}B(\cdot)u(\cdot,f)](t) \\ &= f(t) + \mathcal{A} \int_{0}^{t} g_{\alpha}(t-s)u(s,f)ds + \int_{0}^{t} g_{\alpha}(t-s)B(s)u(s,f)ds, \quad t \in (0,\tau). \end{split}$$

Therefore, u(t, x) is a solution of (386). Since the variation of parameters formula holds in our framework, the uniqueness of solutions follows similarly as in the proof of [61, Theorem 2.26].

Now we will present an illustrative example of application of the above result.

EXAMPLE 3.7.14. (i) It is clear that Theorem 3.7.13 can be applied in the analysis of a great number of the abstract degenerate Cauchy problems of first order appearing in [199, Chapter III] (applications can be also made to some time-oscillation degenerate equations for which the range of possible values of corresponding Caputo fractional derivative depends directly on the value of constant c > 0 in condition [199, (P), p. 47], provided that $\alpha = 1$ in (P)). For example, we can consider the following time-dependent perturbation of the Poisson heat equation in the space $E = L^p(\Omega)$:

$$(P)_{t-d}:\begin{cases} \frac{\partial}{\partial t}[m(x)v(t,x)] = \Delta v(t,x) + bv(t,x) + m(x)B(t)v(t,x), & t \ge 0, \ x \in \Omega; \\ v(t,x) = 0, & (t,x) \in [0,\infty) \times \partial\Omega, \\ m(x)v(0,x) = u_0(x), & x \in \Omega, \end{cases}$$

where Ω is a bounded domain in \mathbb{R}^n with smooth boundary, $b > 0, m(x) \ge 0$ a.e. $x \in \Omega, m \in L^{\infty}(\Omega), 1 and <math>B \in C([0, \infty) : L(E))$.

(ii) Suppose that A, B and C are three closed linear operators in E, $D(B) \subseteq D(A) \cap D(C)$, $B^{-1} \in L(E)$ and the conditions [199, (6.4)–(6.5)] hold with certain numbers c > 0 and $0 < \beta \leq \alpha \leq 1$. In [199, Chapter VI], the second order differential equation

$$\frac{d}{dt}(Cu'(t)) + Bu'(t) + Au(t) = f(t), \quad t > 0,$$

has been considered by the usual converting into the first order matricial system

$$\frac{d}{dt}Mz(t) = Lz(t) + F(t), \quad t > 0,$$

where

$$M = \begin{bmatrix} I & O \\ O & C \end{bmatrix}, \ L = \begin{bmatrix} O & I \\ -A & -B \end{bmatrix} \text{ and } F(t) = \begin{bmatrix} 0 \\ f(t) \end{bmatrix} \quad (t > 0).$$

Owing to the proof of [199, Theorem 6.1], we get that the multivalued linear operator $L_{[D(B)]\times E}(M_{[D(B)]\times E})^{-1}$ generates a (g_1, I) -regularized resolvent family $(S_1(t))_{t>0}$ of growth order $\zeta = ((1 - \beta)/\alpha)$ in the pivot space $[D(B)] \times E$, satisfying additionally that there exists $\omega \ge 0$ with the property that $||g_{\zeta+1}(t)S_1(t)|| \le Me^{\omega t}$, t > 0. Assuming that the mappings $t \mapsto B_{1,3}(t) \in L([D(B)])$, $t \ge 0$ and $t \mapsto B_{2,4}(t) \in L(E)$, $t \ge 0$ are continuous, Theorem 3.7.13 is susceptible to applications so that we are in a position to consider the-wellposedness of the following system of equations:

$$u_1'(t) = u_2(t) + B_1(t)u_1(t) + B_2(t)u_2(t) + f_1(t), \quad t > 0;$$

$$\frac{d}{dt}(Cu_2(t)) = -Au_1(t) - Bu_2(t) + B_3(t)u_1(t) + B_4(t)u_2(t) + f_2(t), \quad t > 0.$$

Many concrete examples of applications can be found in [199, Section 6.2].

Evidently, Theorem 3.7.13 can be applied only in the case that $\alpha \ge 1$. As already seen, the case $0 < \alpha < 1$ is much more sofisticated from the theoretical and practical point of view. We close this section with the observation that H. K. Avad and A. V. Glushak [38] have analyzed the perturbed time-fractional problem

$$D_t^{\alpha} u(t) = A u(t) + F(t, B(t)u(t)), \quad t > 0,$$

$$\lim_{t \to 0+} (g_{1-\alpha} * u)(t) = u_0,$$

where A is a closed linear operator acting on a Banach space E, $F(\cdot, \cdot)$ and $B(\cdot)$ possess certain properties, and $0 < \alpha < 1$. Some of their results, for example [38, Theorem 3.1], can be simply reword for abstract time-relaxation inclusions.

Finally, we would like to note that perturbation results for multivalued linear operators have been also examined by R. Cross, A. Favini and Y. Yakubov in [121].

3.8. Approximation and convergence of degenerate (a, k)-regularized C-resolvent families

The state space in this section will be denoted by E. If $(A_l)_{l \in \mathbb{N}_0} (((R_l(t))_{t \ge 0})_{l \in \mathbb{N}_0})$ is a sequence of closed linear operators on E (strongly continuous operator families in L(E), then we also write $A((R(t))_{t\geq 0})$ in place of $A_0((R_0(t))_{t\geq 0})$.

Making use of Theorem 1.4.11 and Theorem 3.2.5, we can simply prove an extension of [**326**, Theorem 2.3] for (a, k)-regularized C-resolvent families subgenerated by multivalued linear operators:

THEOREM 3.8.1. Assume that, for every $n \in \mathbb{N}_0$, $|a_n|(t)$ and $k_n(t)$ satisfy (P1) and \mathcal{A}_n is a closed subgenerator of an (a_n, k_n) -regularized C_n -resolvent family $(R_n(t))_{t\geq 0}$ which satisfies (272) with a(t), R(t) and k(t) replaced respectively by $a_n(t), R_n(t)$ and $k_n(t)$ $(n \in \mathbb{N}_0)$. Assume further that there exists a real number $\omega \ge \sup_{n \in \mathbb{N}_0} \max(0, \operatorname{abs}(|a_n|), \operatorname{abs}(k_n))$ such that, for every $p \in \mathfrak{B}$, there exist $c_p > 0$ 0 and $r_p \in \circledast$ with

(391)
$$p(e^{-\omega t}R_n(t)x) \leqslant c_p r_p(x), \quad t \ge 0, \ x \in E, \ n \in \mathbb{N}_0$$

Let $\lambda_0 \ge \omega$. Put $\mathfrak{T} := \{\lambda > \lambda_0 : \widetilde{a_n}(\lambda) \widetilde{k_n}(\lambda) \neq 0 \text{ for all } n \in \mathbb{N}_0\}.$ Then the following assertions are equivalent.

- (i) $\lim_{n\to\infty} \widetilde{k_n}(\lambda) (I \widetilde{a_n}(\lambda)\mathcal{A}_n)^{-1} C_n x = \widetilde{k}(\lambda) (I \widetilde{a}(\lambda)\mathcal{A})^{-1} C x, \lambda \in \mathfrak{T}, x \in E$ and the sequence $(R_n(t)x)_n$ is equicontinuous at each point $t \ge 0$ $(x \in E)$.
- (ii) $\lim_{n\to\infty} R_n(t)x = R(t)x, t \ge 0, x \in E$, uniformly on compacts of $[0,\infty)$.

Keeping in mind Lemma 1.2.4(i) and Theorem 3.2.5, it is almost straightforward to formulate an extension of [326, Theorem 2.4] for (a, k)-regularized C-resolvent families subgenerated by multivalued linear operators, as well; the only thing worth noting is that, in the proof of last mentioned theorem, we can replace the sequence $(A_n H_n(\lambda')x)_{n \in \mathbb{N}}$ with $(\widetilde{k_n}(\lambda')\widetilde{a_n}(\lambda')^{-1}[\widetilde{a_n}(\lambda')^{-1}(\widetilde{a_n}(\lambda')^{-1} -$ $\mathcal{A}_n)^{-1}C_nx - C_nx])_{n \in \mathbb{N}}:$

THEOREM 3.8.2. Assume that, for every $n \in \mathbb{N}_0$, $|a_n|(t)$ and $k_n(t)$ satisfy (P1) and \mathcal{A}_n is a closed subgenerator of an (a_n, k_n) -regularized C_n -resolvent family $(R_n(t))_{t\geq 0}$ which satisfies (272) with a(t), R(t) and k(t) replaced respectively by $a_n(t), R_n(t)$ and $k_n(t)$ $(n \in \mathbb{N}_0)$. Assume further that there exists a real number $\omega \ge \sup_{n \in \mathbb{N}_0} \max(0, \operatorname{abs}(|a_n|), \operatorname{abs}(k_n))$ such that, for every seminorm $p \in \mathfrak{B}$, there exist a number $c_p > 0$ and a seminorm $r_p \in \circledast$ such that (391) holds. Let $\lambda_0 \ge \omega$, and let \mathfrak{T} be defined as above. Assume that the following conditions hold:

- (i) The sequence $(k_n(t))_n$ is equicontinuous at each point $t \ge 0$.
- (ii) For every bounded sequence $(x_n)_{n\in\mathbb{N}}$ in E, one has $\sup_{n\in\mathbb{N}} p(C_n x_n) < \infty$. (iii) There exists $\lambda' \in \mathfrak{T}$ such that $\mathrm{R}((\frac{1}{\bar{a}(\lambda')} \mathcal{A})^{-1}C)$ is dense in E as well as that the sequences $(\widetilde{k_n}(\lambda')\widetilde{a_n}(\lambda')^{-1})_{n\in\mathbb{N}}$ and $(\widetilde{a_n}(\lambda')^{-1})_{n\in\mathbb{N}}$ are bounded.
- (iv) For every $\varepsilon > 0$ and $t \ge 0$, there exist $\delta \in (0,1)$ and $n_0 \in \mathbb{N}$ such that

$$\int_{0}^{\min(t,s)} |a_{n}(\max(t,s)-r) - a_{n}(\min(t,s)-r)|dr + \int_{\min(t,s)}^{\max(t,s)} |a_{n}(\max(t,s)-r)|dr < \varepsilon,$$

provided $|t-s| < \delta$, $s \ge 0$ and $n \ge n_0$.

Then

r

$$\lim_{n \to \infty} \widetilde{k_n}(\lambda) (I - \widetilde{a_n}(\lambda)\mathcal{A}_n)^{-1} C_n x = \widetilde{k}(\lambda) (I - \widetilde{a}(\lambda)\mathcal{A})^{-1} C x, \quad \lambda \in \mathfrak{T}_0, \ x \in E$$

is equivalent to say that $\lim_{n\to\infty} R_n(t)x = R(t)x$, $t \ge 0$, $x \in E$, uniformly on compacts of $[0,\infty)$.

The conclusions from [**326**, Remark 2.5(ii)] can be formulated in our context; the same holds for the parts (i) and (iii) of this remark. Since subordination principles established in [**61**] and [**459**] hold in our framework, Theorem 3.8.2 can be used for proving the following extension of [**326**, Theorem 2.6]:

THEOREM 3.8.3. Suppose $\alpha > 0$, $\beta \ge 1$, \mathcal{A} is a closed subgenerator of an exponentially equicontinuous (g_{α}, g_{β}) -regularized C-resolvent family $(R(t))_{t\ge 0}$ satisfying (272) with $a(t) = g_{\alpha}(t)$ and $k(t) = g_{\beta}(t)$, and R(C) as well as $D(\mathcal{A})$ are dense in E. Let $(\alpha_n)_{n\in\mathbb{N}}$ be an increasing sequence of positive real numbers with $\lim_{n\to\infty} \alpha_n = \alpha$, and let $\gamma_n = \alpha_n/\alpha$ $(n \in \mathbb{N})$. Then, for every $n \in \mathbb{N}$, the operator \mathcal{A} is a subgenerator of an exponentially equicontinuous $(g_{\alpha_n}, g_{1+\gamma_n(\beta-1)})$ -regularized C-resolvent family $(R_n(t))_{t\ge 0}$ satisfying (272) with $a(t) = g_{\alpha_n}(t)$, $k(t) = g_{1+\gamma_n(\beta-1)}(t)$ and $(R(t))_{t\ge 0}$ replaced by $(R_n(t))_{t\ge 0}$. Furthermore, $\lim_{n\to\infty} R_n(t)x = R(t)x, t \ge 0$, $x \in E$, uniformly on compacts of $[0, \infty)$.

It is very technical to extend the assertion of [326, Theorem 2.7] to (a, k)-regularized C-resolvent families subgenerated by multivalued linear operators.

THEOREM 3.8.4. Assume that, for every $n \in \mathbb{N}_0$, $|a_n|(t)$ and $k_n(t)$ satisfy (P1) and \mathcal{A} is a closed subgenerator of an (a_n, k_n) -regularized C_n -resolvent family $(R_n(t))_{t\geq 0}$ which satisfies (272) with a(t), R(t) and k(t) replaced respectively by $a_n(t)$, $R_n(t)$ and $k_n(t)$ $(n \in \mathbb{N}_0)$. Denote by $a_{n,k}(t)$ the k-th convolution power of the function $a_n(t)$ $(k \in \mathbb{N})$. Assume further that there exists a real number $\omega \geq \sup_{n \in \mathbb{N}_0} \max(0, \operatorname{abs}(|a_n|), \operatorname{abs}(k_n))$ such that, for every seminorm $p \in \circledast$, there exist a number $c_p > 0$ and a seminorm $r_p \in \circledast$ such that (391) holds. Let $\lambda_0 \geq \omega$. Suppose that $l \in \mathbb{N}$ and the following holds:

- (i) $\lim_{n\to\infty} \widetilde{k_n}(\lambda) (I \widetilde{a_n}(\lambda)\mathcal{A})^{-1} C_n x = \widetilde{k}(\lambda) (I \widetilde{a}(\lambda)\mathcal{A})^{-1} C x, \ \lambda \in \mathfrak{T}, \ x \in D(A^l).$
- (ii) The sequences $(k_n(t))_n$, $((a_n * k_n)(t))_n$,..., and $((a_{n,l-1} * k_n)(t))_n$ are equicontinuous at each point $t \ge 0$.
- (iii) The sequence $(C_n x)_n$ is bounded for any $x \in D(\mathcal{A}^l)$.
- (iv) The condition (iv) of Theorem 3.8.2 holds with the function $a_n(t)$ replaced by $a_{n,l}(t)$.

Then, for every $x \in \overline{D(\mathcal{A}^l)}$, one has $\lim_{n\to\infty} R_n(t)x = R(t)x$, $t \ge 0$, uniformly on compacts of $[0,\infty)$.

Using the Hille–Yosida theorem for degenerate (a, k)-regularized C-resolvent families and arguing as in the proof of [**326**, Theorem 2.8], we can prove the following:

THEOREM 3.8.5. Assume that, for every $n \in \mathbb{N}_0$, $|a_n|(t)$ and $k_n(t)$ satisfy (P1), \mathcal{A}_n is a closed MLO, and $\lambda_0 > \omega \ge \sup_{n \in \mathbb{N}_0} \max(0, \operatorname{abs}(|a_n|), \operatorname{abs}(k_n))$. Assume that $\lim_{n\to\infty} \widetilde{a_n}(\lambda) = \tilde{a}(\lambda), \lambda \in \mathfrak{T}$ and $\lim_{n\to\infty} \widetilde{k_n}(\lambda) = \tilde{k}(\lambda), \lambda \in \mathfrak{T}$. Suppose that $L(E) \ge \tilde{k}(\lambda)(I - \tilde{a}(\lambda)\mathcal{A})^{-1}C, \lambda \in \mathfrak{T}$, and for every $n \in \mathbb{N}$, \mathcal{A}_n is a subgenerator of an (a_n, k_n) -regularized C_n -resolvent family $(R_n(t))_{t\ge 0}$ which satisfies (272) with a(t), R(t) and k(t) replaced respectively by $a_n(t), R_n(t)$ and $k_n(t)$. Let (391) hold for $t \ge 0, x \in E$ and $n \in \mathbb{N}$, and let

$$\lim_{n \to \infty} \widetilde{k_n}(\lambda) (I - \widetilde{a_n}(\lambda)\mathcal{A}_n)^{-1} C_n x = \widetilde{k}(\lambda) (I - \widetilde{a}(\lambda)\mathcal{A})^{-1} C x, \quad x \in E, \ \lambda \in \mathfrak{T}.$$

Suppose, further, that for each $\lambda \in \mathfrak{T}$ there exists an open ball $\Omega_{\lambda} \subseteq \{z \in \mathbb{C} : \operatorname{Re} z > \lambda_0\}$, with center at point λ and radius $2\varepsilon_{\lambda} > 0$, so that $\widetilde{a_n}(z)\widetilde{k_n}(z) \neq 0$, $z \in \Omega_{\lambda}$, $n \in \mathbb{N}_0$. Then the following holds:

 (i) For each r ∈ (0,1], A is a subgenerator of a global (a, k * g_r)-regularized C-resolvent family (R_r(t))_{t≥0} satisfying (272) as well as that, for every seminorm p ∈ ⊛,

$$p(R_r(t+h)x - R_r(t)x) \leq \frac{2c_p r_p(x)}{r\Gamma(r)} \max(e^{\omega(t+h)}, 1)h^r, \quad t \ge 0, \ h > 0, \ x \in E,$$

and that, for every seminorm p and for every bounded set $B \in \mathcal{B}$, the mapping $t \mapsto p_B(R_r(t)), t \ge 0$ is locally Hölder continuous with exponent r.

 (ii) If A is densely defined, then A is a subgenerator of a global (a, k)-regularized C-resolvent family (R(t))_{t≥0} satisfying (272) and that the family {e^{-ωt}R(t): t≥0} is equicontinuous.

Suppose that \mathcal{A} is an MLO, $C\mathcal{A} \subseteq \mathcal{A}C$ and $\rho_C(\mathcal{A}) \neq \emptyset$. Then, for every $\lambda \in \rho_C(\mathcal{A})$, we have $\mathcal{A}0 = N((\lambda I - \mathcal{A})^{-1}C)$, which implies that the operator $(\lambda I - \mathcal{A})^{-1}C$ is injective iff \mathcal{A} is single-valued. Although the resolvent equation holds in our framework, the non-injectivity of operator $(\lambda I - \mathcal{A})^{-1}C$ in multivalued case does not permit us to state a satisfactory extension of [**326**, Corollary 2.10] for degenerate resolvent families. The assertion of [**326**, Proposition 2.11(i)] can be formulated in our context (cf. Lemma 1.2.4), which is not the case with the assertions of [**326**, Proposition 2.11(ii)] and [**326**, Proposition 2.12].

The interested reader will want to prove some results on approximation and convergence of degenerate (A, k)-regularized C-(pseudo)resolvent families introduced in Section 2.9 (cf. [**326**, Theorem 2.16, Theorem 2.17] for non-degenerate case), as well as degenerate (a, k)-regularized C-resolvent families introduced in Sections 2.2–2.3.

Now we would like to mention that the assertions of [61, Theorem 2.21, Theorem 2.23] continue to hold in the degenerate case (if $C \neq I$ or E is not a Banach space, then the consideration is left to the interested reader):

REMARK 3.8.6. Let $\alpha > 0$, $(E, \|\cdot\|)$ be a Banach space, let the numbers $b_{k,n}^{\alpha}$ be defined by (393), and let \mathcal{A} be a closed subgenerator of an exponentially bounded (g_{α}, I) -regularized resolvent family $(S_{\alpha}(t))_{t \geq 0}$ satisfying $S_{\alpha}(t)x - t$

$$g_{\alpha+1}(t)x \in \mathcal{A}\int_0^t S_\alpha(s)x \, ds, t \ge 0, x \in E.$$
 Then

$$S_{\alpha}(t)x = \lim_{n \to \infty} \frac{1}{(n-m)!} \sum_{k=1}^{n-m+1} b_{k,n-m+1}^{\alpha} (I - (t/n)^{\alpha} \mathcal{A})^{-k} x, \quad x \in E, \ t \ge 0,$$

and the convergence is uniform on compacts of $[0, \infty)$. The only non-trivial thing worth explaining here is that the formula [**61**, (2.43)] holds in multivalued linear case, with the equality replaced with the inclusion \ni ; this can be proved by induction, Theorem 1.2.4(i) and elementary properties of multivalued linear operators (although the resolvent $(\zeta^{\alpha} - \mathcal{A})^{-1}$ is not necessarily injective for $\zeta > 0$ suff. large, we can apply it on both sides of the equality [**61**, (2.43)], i.e., the inductive hypothesis, and employ after that the equality $(\zeta^{\alpha} - \mathcal{A})(\zeta^{\alpha} - \mathcal{A})^{-1}\mathcal{A}F^{(j+1)}(\lambda)x =$ $\mathcal{A}F^{(j+1)}(\lambda)x, x \in E$ in order to see that [**61**, (2.43)] holds with j replaced by j + 1 therein). This is an extension of [**61**, Theorem 2.21]. Concerning [**61**, Theorem 2.23], it is only worth pointing out that the equality $S_{\alpha}(t)x - S_{\alpha}(s)x =$ $\int_{0}^{t} P_{\alpha}(\tau)y \, d\tau - \int_{0}^{s} P_{\alpha}(\tau)y \, d\tau$, where $(x, y) \in \mathcal{A}$ and $P_{\alpha}(t)y := (g_{\alpha-1} * S_{\alpha}(\cdot)y)(t),$ t > 0 can be proved by taking the Laplace transform in both variables, t and s, and applying after that Theorem 1.2.4(i): Let $\alpha \ge 1$, let $x \in D(\mathcal{A})$, and let $(x, y) \in \mathcal{A}$. Then

$$\left\| S_{\alpha}(t)x - \lim_{n \to \infty} \frac{1}{(n-m)!} \sum_{k=1}^{n-m+1} b_{k,n-m+1}^{\alpha} \times (I - (t/n)^{\alpha} \mathcal{A})^{-k} x \right\| = O(n^{(-1)/2}), \quad n \to +\infty.$$

Furthermore, if $(S_{\alpha}(t))_{t \ge 0}$ is bounded, then there exists a constant $M_{\alpha} > 0$ such that

$$\left\| S_{\alpha}(t)x - \lim_{n \to \infty} \frac{1}{(n-m)!} \sum_{k=1}^{n-m+1} b_{k,n-m+1}^{\alpha} \times (I - (t/n)^{\alpha} \mathcal{A})^{-k} x \right\| \leq M_{\alpha} n^{(-1)/2} t^{\alpha} \|y\|, \quad n \in \mathbb{N}.$$

3.8.1. Laguerre expansions of degenerate (a, k)-regularized *C*-resolvent families. In this subsection, we shall present the basic results about Laguerre expansions of degenerate (a, k)-regularized *C*-resolvent families in locally convex spaces (cf. the recent paper by L. Abadias and P. J. Miana [2] for C_0 -semigroup case).

We start by recalling that Rodrigues' formula gives the following representation of generalized Laguerre polynomials

$$L_n^{\alpha}(t) \equiv e^t \frac{t^{-\alpha}}{n!} \frac{d^n}{dt^n} (e^{-t} t^{n+\alpha}), \quad t \in \mathbb{R} \quad (n \in \mathbb{N}_0, \ \alpha > -1).$$

If $\alpha \notin -\mathbb{N}$ and $n \in \mathbb{N}_0$, then we define

$$l_n^{\alpha}(t) \equiv \frac{1}{\Gamma(n+\alpha+1)} \frac{d^n}{dt^n} (e^{-t} t^{n+\alpha}), \quad t > 0.$$

The reader may consult [2, Proposition 2.6] for the most important properties of functions $l_n^{\alpha}(t)$ ($\alpha \notin -\mathbb{N}, n \in \mathbb{N}_0$). For example, it is well known that

$$l_n^{\alpha}(t) \sim g_{\alpha+1}(t), \quad t \to 0+; \qquad l_n^{\alpha}(t) \sim (-1)^n e^{-t} g_{n+\alpha+1}(t), \quad t \to +\infty,$$

and

$$\frac{d^k}{dt^k}l_n^{\alpha}(t) = l_{n+k}^{\alpha-k}(t), \quad t > 0, \ k \in \mathbb{N}_0$$

The following theorem can be deduced by using the argumentation contained in the proof of [2, Theorem 3.3], stated in the Banach space case (cf. also [373, Theorem 3, Section 4.23]).

THEOREM 3.8.7. Suppose that $f: (0, \infty) \to E$ is a differentiable mapping, $\alpha > -1$ and for each seminorm $p \in \circledast$ we have $\int_0^\infty e^{-t} t^\alpha p(f(t))^2 dt < \infty$. Then

$$f(t) = \sum_{n=0}^{\infty} \frac{n! L_n^{\alpha}(t)}{\Gamma(n+\alpha+1)} \int_0^{\infty} e^{-s} s^{\alpha} L_n^{\alpha}(s) f(s) ds, \quad t > 0.$$

Since

$$\int_{0}^{\infty} e^{-s} s^{\alpha} L_{n}^{\alpha}(s) f(s) ds = \int_{0}^{\infty} \frac{d^{n}}{ds^{n}} (e^{-s} s^{n+\alpha}) \frac{f(s)}{n!} ds$$
$$= \sum_{k=0}^{n} (-1)^{n-k} \binom{n}{k} (n+\alpha) \dots (n+\alpha-(k-1)) \int_{0}^{\infty} e^{-s} s^{n+\alpha-k} \frac{f(s)}{n!} ds$$

and

$$\frac{(n+\alpha)\dots(n+\alpha-(k-1))}{\Gamma(n+\alpha+1)} = \frac{1}{\Gamma(n+\alpha+1-k)}$$

for any $n, k \in \mathbb{N}_0$ with $k \leq n$ and $\alpha > -1$, we immediately obtain the following corollary of Theorem 3.8.7.

COROLLARY 3.8.8. Suppose that $f: (0, \infty) \to E$ is a differentiable mapping, $\alpha > -1$ and for each seminorm $p \in \circledast$ we have $\int_0^\infty e^{-s} s^\alpha p(f(s))^2 ds < \infty$. Then, for every t > 0, the following equality holds:

(392)
$$f(t) = \sum_{n=0}^{\infty} \sum_{k=0}^{n} \frac{L_n^{\alpha}(t)(-1)^{n-k} \binom{n}{k}}{\Gamma(n+\alpha+1-k)} \int_0^{\infty} e^{-s} s^{n+\alpha-k} f(s) ds.$$

Before proceeding further, let us observe that the last formula can be rewritten in the following equivalent way:

$$f(t) = \sum_{n=0}^{\infty} L_n^{\alpha}(t) \int_0^{\infty} l_n^{\alpha}(s) f(s) ds.$$

Suppose now that $(R(t))_{t\geq 0}$ is an exponentially equicontinuous (a, k)-regularized *C*-resolvent family with a closed subgenerator $-\mathcal{A}$, the functions k(t) and |a|(t)satisfy (P1), the family $\{e^{-\omega t}R(t) : t \geq 0\}$ is equicontinuous for some $\omega \geq 0$, $\omega_0 \equiv \max(\omega, \operatorname{abs}(|a|), \operatorname{abs}(k)) < 1/2 \text{ and } \alpha > -1.$ If, in addition, $\tilde{k}(1)\tilde{a}(1) \neq 0$, then Theorem 3.2.5 implies that, for every $\alpha \in \mathbb{N}_0$ and $x \in E$,

$$\begin{split} \int_0^\infty e^{-s} s^{n+\alpha-k} R(s) x \, ds &= (-1)^{n+\alpha-k} \Big(\frac{d^{n+\alpha-k}}{d\lambda^{n+\alpha-k}} (\mathcal{L}R(\cdot)x)(\lambda) \Big)_{\lambda=1} \\ &= (-1)^{n+\alpha-k} \Big(\frac{d^{n+\alpha-k}}{d\lambda^{n+\alpha-k}} \Big[\frac{\tilde{k}(\lambda)}{\tilde{a}(\lambda)} \Big(\frac{1}{\tilde{a}(\lambda)} + \mathcal{A} \Big)^{-1} Cx \Big] \Big)_{\lambda=1}; \end{split}$$

then one can use the product rule, the identity $(d^n/d\lambda^n)(\lambda+\mathcal{A})^{-1}Cx = (-1)^n n!(\lambda+\mathcal{A})^{-n-1}Cx, n \in \mathbb{N}_0, x \in E$, as well as the well known Faà di Bruno's formula

$$\frac{d^n}{dx^n}f(g(x)) = \sum \frac{n!}{m_1!m_2!\dots m_n!} f^{(m_1+m_2+\dots+m_n)}(g(x)) \prod_{j=1}^n \left(\frac{g^{(j)}(x)}{j!}\right)^{m_j},$$

where the summation is taken over those multi-indices $(m_1, m_2, \ldots, m_n) \in \mathbb{N}_0^n$ for which $m_1 + 2m_2 + \cdots + nm_n = n$, in order to express the right hand side of (392), with f(t) = R(t)x, t > 0 in terms of subgenerator $-\mathcal{A}$ (notice, however, that it is very difficult to express the value of $\int_0^\infty e^{-s} s^{n+\alpha-k} R(s)x \, ds$ in terms of $-\mathcal{A}$, if $\alpha \notin \mathbb{N}_0$ and $n \in \mathbb{N}_0$). At any rate, the obtained representation formula is very complicated, practically almost irrelevant, but can be simplified in some cases; for example, if k(t) = 1 and $a(t) = g_{\vartheta}(t)$ for some $\vartheta > 0$, then we have

$$\frac{d^n}{d\lambda^n} \Big[\frac{\tilde{k}(\lambda)}{\tilde{a}(\lambda)} \Big(\frac{1}{\tilde{a}(\lambda)} + \mathcal{A} \Big)^{-1} Cx \Big] = \frac{d^n}{d\lambda^n} [\lambda^{\vartheta - 1} (\lambda^\vartheta + \mathcal{A})^{-1} Cx]$$

= $(-1)^n \lambda^{-(n+1)} \sum_{k=1}^{n+1} b_{k,n+1}^\vartheta \lambda^{\vartheta k} (\lambda^\vartheta + \mathcal{A})^{-k} Cx, \quad n \in \mathbb{N}_0, \ x \in E, \ \operatorname{Re} \lambda > \omega_0,$

where the numbers $b_{k,n+1}^{\vartheta}$ are given by the following recurrence relations:

$$b_{1,1}^{\vartheta} = 1,$$
(393)
$$b_{k,n}^{\vartheta} = (n-1-k\vartheta)b_{k,n-1}^{\vartheta} + \vartheta b_{k-1,n-1}^{\vartheta}, \quad 1 \leq k \leq n, \ n = 2, 3, \dots$$

$$b_{k,n}^{\vartheta} = 0, \quad k > n, \ n = 1, 2, \dots,$$

cf. the formulae [61, (2.16)-(2.17)].

3.8.2. Laguerre expansions of solutions to abstract non-degenerate differential equations of first order. The main aim of this subsection, which can be viewed of some independent interest, is to show how Laguerre expansions can be elegantly used for proving some representation formulae for solutions of abstract non-degenerate differential equations of first order whose solutions are governed by fractionally integrated semigroups and exponential ultradistribution semigroups of Beurling class. All operators considered in this subsection will be single-valued, and all subgenerators of (fractionally integrated) *C*-semigroups under examination will be closed.

Suppose first that the operator -A subgenerates the global *C*-regularized semigroup $(T(t))_{t\geq 0}$ satisfying that there exists $\omega \geq 0$ such that the family $\{e^{-\omega t}T(t): t\geq 0\}$ is equicontinuous. Let $\omega' > \omega - (1/2)$. Then, for every $z \in \mathbb{C}$, the operator -A+z is a subgenerator of the global *C*-regularized semigroup $(e^{zt}T(t))_{t\geq 0}$ and we can construct the complex powers of the operator $A + \omega' + 1$ following the method proposed in Section 1.1. Any of these powers is closed and injective.

Let $\theta \in (\pi/2, \pi)$, and let $d' \in (0, 1)$ be sufficiently small. Then, for every $\beta \in \mathbb{C}_+$, we have defined the operator $(A + \omega' + 1)_C^{-\beta}$ by

$$(A + \omega' + 1)_C^{-\beta} x := \frac{1}{2\pi i} \int_{\Gamma_{\theta,d'}} z^{-\beta} (z - (A + \omega' + 1))^{-1} C x \, dz, \quad x \in E,$$

where $\Gamma_{\theta,d'} = \partial(\Sigma_{\omega'} \setminus B_{d'})$ denotes the boundary of $\Sigma_{\omega'} \setminus B_{d'}$, oriented in such a way that Im z increases along $\Gamma_{\omega',d'}$. If Re $\beta > 1$, then it can be easily seen with the help of resolvent equation and Cauchy formula that, for every $x \in E$,

$$\begin{split} (A+\omega'+1)(A+\omega'+1)_{C}^{-\beta}x &= \frac{1}{2\pi i}\int_{\Gamma_{\theta,d'}} z^{-\beta}(A+\omega'+1)(z-(A+\omega'+1))^{-1}Cx\,dz\\ &= \frac{1}{2\pi i}\int_{\Gamma_{\theta,d'}} z^{-\beta}\Big[-Cx+z(z-(A+\omega'+1))^{-1}Cx\Big]dz\\ &= \frac{1}{2\pi i}\int_{\Gamma_{\theta,d'}} z^{-\beta+1}(z-(A+\omega'+1))^{-1}Cx\,dz\\ &= (A+\omega'+1)_{C}^{-(\beta-1)}x. \end{split}$$

Inductively, we obtain that

$$(A + \omega' + 1)^{n} (A + \omega' + 1)_{C}^{-\beta} x = (A + \omega' + 1)_{C}^{-(\beta - n)} x, \quad \operatorname{Re} \beta > n, \ x \in E \ (n \in \mathbb{N}_{0}),$$

i.e., by definition of powers $(A + \omega' + 1)_{-\beta}$ and $(A + \omega' + 1)_{-(\beta - n)},$
(394) $(A + \omega' + 1)^{n} (A + \omega' + 1)_{-\beta} Cx = (A + \omega' + 1)_{-(\beta - n)} Cx, \quad \operatorname{Re} \beta > n, \ x \in E \ (n \in \mathbb{N}_{0}).$

By [292, Lemma 2.9.73(iii)], we have

(395)
$$(A + \omega' + 1)_{-\beta} Cx = \int_0^\infty \frac{s^{\beta-1}}{\Gamma(\beta)} e^{-(\omega'+1)s} T(s) x \, ds, \quad x \in E, \ \beta > 0.$$

Furthermore, by (394), we have that for each $n \in \mathbb{N}_0$ and $\alpha > -1$, $(A + \omega' + 1)_{-(n+\alpha+1)}Cx \in D((A + \omega' + 1)^n) = D(A^n)$ $(x \in E)$. Applying the binomial formula and (394), we get

(396)
$$(A + \omega')^{n} (A + \omega' + 1)_{-(n+\alpha+1)} Cx$$
$$= ((A + \omega' + 1) - 1)^{n} (A + \omega' + 1)_{-(n+\alpha+1)} Cx$$
$$= \sum_{k=0}^{n} (-1)^{n-k} \binom{n}{k} (A + \omega' + 1)_{-(n+\alpha-k+1)} Cx, \quad x \in E.$$

Observing finally that for each $x \in D(A)$ the mapping $t \mapsto T(t)x$, $t \ge 0$ is continuously differentiable with (d/dt)T(t)x = T(t)Ax, $t \ge 0$, Corollary 3.8.8 combined with the equations (395)–(396) immediately implies the following:

THEOREM 3.8.9. Suppose that the operator -A is a subgenerator of the global *C*-regularized semigroup $(T(t))_{t\geq 0}$ satisfying that there exists $\omega \geq 0$ such that the family $\{e^{-\omega t}T(t): t \ge 0\}$ is equicontinuous. Let $\omega' > \omega - (1/2)$. Then, for every $x \in D(A)$ and $\alpha > -1$, we have that

$$T(t)x = e^{\omega' t} \sum_{n=0}^{\infty} L_n^{\alpha}(t) (A + \omega')^n (A + \omega' + 1)_{-n-\alpha-1} Cx, \quad t > 0.$$

It is worth noting that Theorem 3.8.9 enables one to consider Laguerre expansions of certain classes of semigroups that are strongly continuous for t > 0, like semigroups of class $(C_{(k)})$ and semigroups of growth order r > 0 (cf. [291, Theorems 1.2.15, 1.2.17 and 1.2.19]), which as a further consequence has that we can approximate by Laguerre polynomials solutions of incomplete abstract Cauchy problems, in general with modified Liouville right-sided time-fractional derivatives (cf. [292, Theorems 2.9.58, 2.9.60]). For the sake of brevity, we will present only two applications of Theorem 3.8.9. The first one is connected with Laguerre expansions of solutions of first order abstract Cauchy problems associated with generators of fractionally integrated semigroups in locally convex spaces (cf. [291, Subsection 2.9.7] for more details). The second one is connected with Laguerre expansions of solutions of first order abstract Cauchy problems whose solutions are governed by exponential ultradistribution semigroups of Beurling class (cf. [291, Theorem 3.6.9]):

(i) Assume that $\zeta \in (0,\infty) \setminus \mathbb{N}, \ \omega \ge 0, \ 1 \ge \sigma' > \sigma > 0$, as well as that -Ais the integral generator of an ζ -times integrated semigroup $(S_{\zeta}(t))_{t\geq 0}$ satisfying that the family $\{e^{-\omega t}S_{\zeta}(t):t\geq 0\}$ is equicontinuous. Suppose $\varepsilon > 0, |\zeta| = |\zeta + \varepsilon|$ and $\alpha > -1$. If $\alpha \notin \mathbb{N}_0$, then the right hand side of (392), with f(t) replaced by $e^{-\omega t}S_{\zeta}(t)x$, cannot be so simply evaluate in terms of A. Nevertheless, for every $\gamma \in (0, \pi/2)$, there exists $d \in (0, 1]$ such that $\Sigma(\gamma, d) \subseteq \rho(-A - \omega - \sigma)$ and the family $\{(1+|\lambda|)^{1-\zeta}(\lambda + A + \omega + \sigma)^{-1}:$ $\lambda \in \Sigma(\gamma, d)$ is equicontinuous. Set $C_{\zeta} := (A + \omega + \sigma)^{-1 - \lfloor \zeta \rfloor}$. Then it is not difficult to prove that the operator $A + \omega + \sigma$ is C_{ζ} -sectorial of angle $\pi/2$ and that the condition [103, (H)] holds with $d = \sigma/2$. Therefore, for every $z \in \mathbb{C}$, we can construct the power $(A + \omega + \sigma)_z$ following the method proposed in [103], with the operator C replaced by C_{ζ} . This power is injective, and belongs to L(E) provided that $\operatorname{Re} z < -\zeta$. Furthermore, we know that -A is the integral generator of a global $(A + \omega + \sigma)_{-(\zeta + \varepsilon)}$ regularized semigroup $(T(t))_{t\geq 0}$ and the family $\{e^{-(\omega+\sigma')t}T(t): t\geq 0\}$ is equicontinuous (see e.g. [291, Theorem 2.3.26]). Applying Theorem 3.8.9, we get that for each t > 0 and $x \in D(A)$,

(397)
$$e^{-(\omega+\sigma)t}T(t)x$$
$$=\sum_{n=0}^{\infty}L_{n}^{\alpha}(t)(A+\omega+\sigma)^{n}(A+\omega+\sigma+1)_{-n-\alpha-1}(A+\omega+\sigma)_{-(\zeta+\varepsilon)}x;$$

here we define the power $(A + \omega + \sigma + 1)_{-n-\alpha-1}$ $(n \in \mathbb{N}_0)$ by using the construction from [103]; note that the operator $A + \omega + \sigma + 1$ is *C*-sectorial of angle $\pi/2$ with $C \equiv (A + \omega + \sigma)_{-(\zeta + \varepsilon)}$. Perhaps the most important consequence of (397) is given as follows. Let $x \in D((A + \omega + \sigma)_{1+\zeta + \varepsilon})$.

Then $x \in D((A + \omega + \sigma)_{\zeta + \varepsilon})$, $x_0 := (A + \omega + \sigma)_{\zeta + \varepsilon} x \in D(A + \omega + \sigma)$ and $x = (A + \omega + \sigma)_{-(\zeta + \varepsilon)} x_0$. Furthermore, the function $t \mapsto u(t) \equiv T(t) x_0$, $t \ge 0$ is the classical solution of the abstract Cauchy problem

$$(ACP_1): u'(t) = Au(t), \quad t \ge 0, \ u(0) = x.$$

By (397), it readily follows that, for every t > 0,

$$u(t) = e^{(\omega+\sigma)t} \sum_{n=0}^{\infty} L_n^{\alpha}(t) (A+\omega+\sigma)^n (A+\omega+\sigma+1)_{-n-\alpha-1} x.$$

(ii) Suppose that the sequence $(M_p)_{p \in \mathbb{N}_0}$ of positive real numbers satisfies the conditions (M.1), (M.2) and (M.3), $M_0 = 1$, $(E, \|\cdot\|)$ is a Banach space and -A generates an exponential ultradistribution semigroup of (M_p) -class. Then the abstract Cauchy problem

$$u \in C^{\infty}([0,\infty):E) \cap C([0,\infty):[D(A)]); \quad u'(t) + Au(t) = 0, \qquad t \ge 0, \ u(0) = x_{0}$$

has a unique solution for all $x \in E^{(M_p)}(A)$, there exist an injective operator $C \in L(E)$ and a number $\omega \in \mathbb{R}$ such that $E^{(M_p)}(A) \subseteq C(D_{\infty}(A))$ and that -A generates an exponentially bounded *C*-regularized semigroup $(T(t))_{t\geq 0}$ with $||T(t)|| = O(e^{\omega t})$, $t \geq 0$ (cf. [291, Theorems 3.6.8, 3.6.9; Lemma 3.6.5]). If $x \in E^{(M_p)}(A)$ and $\omega' > \omega - (1/2)$, then it is clear that $u(t) = T(t)C^{-1}x$, $t \geq 0$ and

$$u(t) = e^{\omega' t} \sum_{n=0}^{\infty} L_n^{\alpha}(t) (A + \omega')^n (A + \omega' + 1)_{-n-\alpha-1} x, \quad t > 0.$$

In [3], L. Abadias and P. J. Miana have recently analyzed the Hermite expansions of non-degenerate C_0 -groups and cosine operator functions in Banach spaces. The interested reader may try to reconsider the results from [3] for some other classes of (non-)degenerate resolvent operator families.

For further information on approximation and convergence of degenerate differential equations, we refer the reader to [50, 199, 370] and references cited in the introductory part.

3.9. The existence and uniqueness of solutions of abstract incomplete differential inclusions

The state space in this section will be denoted by X. Let $C \in L(X)$ be not necessarily injective, and let \mathcal{A} be a multivalued linear operator commuting with C. Then the C-resolvent set of \mathcal{A} , $\rho_C(\mathcal{A})$ for short, is defined as in the case that the operator C is injective; $\rho_C(\mathcal{A})$ is the union of those complex numbers $\lambda \in \mathbb{C}$ for which $R(C) \subseteq R(\lambda - \mathcal{A})$ and $(\lambda - \mathcal{A})^{-1}C$ is a single-valued linear continuous operator on X. The operator $\lambda \mapsto (\lambda - \mathcal{A})^{-1}C$ is called the C-resolvent of \mathcal{A} ($\lambda \in \rho_C(\mathcal{A})$). In our previous work, we have faced ourselves with some situations in which it is indispensable to assume that the operator $(\lambda - \mathcal{A})^{-1}C$ is not single-valued. This will not be the case in this section.

It is incredibly important to observe the following:

REMARK 3.9.1. Theorem 1.2.4, Proposition 1.2.6(i) and Theorem 1.2.8 continue to hold even if the injectivity of C is disregarded.

Concerning the analytical properties of C-resolvents, we have the following:

PROPOSITION 3.9.2. Let $\emptyset \neq \Omega \subseteq \rho_C(\mathcal{A})$ be open, and let $x \in X$.

- (i) Suppose that R(C) is dense in X. Then the local boundedness of the mapping λ → (λ − A)⁻¹Cx, λ ∈ Ω implies its analyticity. Furthermore, if X is barreled, then the local boundedness of the mapping λ → (λ − A)⁻¹C, λ ∈ Ω implies its analyticity.
- (ii) Suppose that R(C) is dense in X and A is closed. Then we have $R(C) \subseteq R((\lambda A)^n), n \in \mathbb{N}$ and

(398)
$$\frac{d^{n-1}}{d\lambda^{n-1}}(\lambda - \mathcal{A})^{-1}Cx = (-1)^{n-1}(n-1)!(\lambda - \mathcal{A})^{-n}Cx, \quad n \in \mathbb{N}.$$

Furthermore, if X is barreled, then $R(C) \subseteq R((\lambda - A)^n)$, $n \in \mathbb{N}$ and

(399)
$$\frac{d^{n-1}}{d\lambda^{n-1}}(\lambda - \mathcal{A})^{-1}C = (-1)^{n-1}(n-1)!(\lambda - \mathcal{A})^{-n}C \in L(X), \quad n \in \mathbb{N}.$$

REMARK 3.9.3. Let $\emptyset \neq \Omega \subseteq \rho_C(\mathcal{A})$ be open, and let $x \in X$. Suppose that \mathcal{A} is closed. Then $\operatorname{card}((\lambda - \mathcal{A})^{-n}Cx) \leq 1, \lambda \in \rho_C(\mathcal{A}), n \in \mathbb{N}, x \in X$. This can be proved by induction, observing that $(\lambda - \mathcal{A})^{-1}0$ is a singleton $(\lambda \in \rho_C(\mathcal{A}))$ as well as that for each $y \in (\lambda - \mathcal{A})^{-(n+1)}Cx$ $(\lambda \in \rho_C(\mathcal{A}), n \in \mathbb{N}, x \in X)$ we have

$$\begin{split} & (\lambda - \mathcal{A})^{-(n+1)} C x = y + (\lambda - \mathcal{A})^{-(n+1)} C 0 \\ & = y + (\lambda - \mathcal{A})^{-1} (\lambda - \mathcal{A})^{-n} C 0 = y + (\lambda - \mathcal{A})^{-1} 0 = \{y\}. \end{split}$$

Taking into account the proof of [138, Corollary 2.8], this is a crucial thing in showing that (398) holds, and that (399) holds, provided in addition that X is barreled.

3.9.1. Complex powers of multivalued linear operators with polynomially bounded *C*-resolvent. In this subsection, we deal with the following conditions:

(H)₀: Let $C \in L(X)$ be not necessarily injective, let \mathcal{A} be closed, and let $C\mathcal{A} \subseteq \mathcal{A}C$. There exist real numbers $d \in (0,1], c \in (0,1), \varepsilon \in (0,1]$ and $\alpha \ge -1$ such that $P_{\alpha,\varepsilon,c} \cup B_d \subseteq \rho_C(\mathcal{A})$, the operator family $\{(1 + |\lambda|)^{-\alpha}(\lambda - \mathcal{A})^{-1}C : \lambda \in P_{\alpha,\varepsilon,c} \cup B_d\} \subseteq L(X)$ is equicontinuous, the mapping $\lambda \mapsto (\lambda - \mathcal{A})^{-1}C$ is strongly analytic on $\operatorname{int}(P_{\alpha,\varepsilon,c} \cup B_d)$ and strongly continuous on $\partial(P_{\alpha,\varepsilon,c} \cup B_d)$

and

(HS)₀: Let $C \in L(X)$ be not necessarily injective, let \mathcal{A} be closed, and let $C\mathcal{A} \subseteq \mathcal{A}C$. There exist real numbers $d \in (0, 1]$, $\vartheta \in (0, \pi/2)$ and $\alpha \ge -1$ such that $\Sigma_{\vartheta} \cup B_d \subseteq \rho_C(\mathcal{A})$, the operator family $\{(1 + |\lambda|)^{-\alpha}(\lambda - \mathcal{A})^{-1}C : \lambda \in \Sigma_{\vartheta} \cup B_d\} \subseteq L(X)$ is equicontinuous, the mapping $\lambda \mapsto (\lambda - \mathcal{A})^{-1}C$ is strongly analytic on $\operatorname{int}(\Sigma_{\vartheta} \cup B_d)$ and strongly continuous on $\partial(\Sigma_{\vartheta} \cup B_d)$.

Here, $B_d = \{z \in \mathbb{C} : |z| \leq d\}$ and $P_{\alpha,\varepsilon,c} = \{\xi + i\eta : \xi \geq \varepsilon, \eta \in \mathbb{R}, |\eta| \leq c(1+\xi)^{-\alpha}\}$. Assume that the condition (H)₀, resp. (HS)₀, holds. Without loss of generality, we may assume that there exists a number $\lambda_0 \in \operatorname{int}(\rho_C(\mathcal{A})) \setminus (P_{\alpha,\varepsilon,c} \cup B_d)$, resp., $\lambda_0 \in \operatorname{int}(\rho_C(\mathcal{A})) \setminus (\Sigma_{\vartheta} \cup B_d)$. Then we can prove inductively (cf. also Theorem 1.2.8(i)) that, for every $z \in \rho_C(\mathcal{A}) \setminus \{\lambda_0\}$:

$$(400) \ (z-\mathcal{A})^{-1}C(\lambda_0-\mathcal{A})^{-k}Cx = \frac{(-1)^k}{(z-\lambda_0)^k}(z-\mathcal{A})^{-1}C^2x + \sum_{i=1}^k \frac{(-1)^{k-i}(\lambda_0-\mathcal{A})^{-i}C^2x}{(z-\lambda_0)^{k+1-i}}.$$

Strictly speaking, for k = 1 this is the usual resolvent equation. Suppose that (400) holds for all natural numbers $\leq k$. Then (398) shows that

$$(z-\mathcal{A})^{-1}C(\lambda_0-\mathcal{A})^{-(k+1)}Cx = (z-\mathcal{A})^{-1}C\frac{(-1)^k}{k!} \left(\frac{d^k}{d\lambda^k}(\lambda-\mathcal{A})^{-1}Cx\right)_{\lambda=\lambda_0}$$
$$= (z-\mathcal{A})^{-1}C\frac{(-1)^k}{k!} \left(\frac{d}{d\lambda} \left[(-1)^{k-1}(k-1)!(\lambda-\mathcal{A})^{-k}Cx\right]\right)_{\lambda=\lambda_0}$$
$$= \frac{(-1)}{k} \left(\frac{d}{d\lambda} \left[(z-\mathcal{A})^{-1}C(\lambda-\mathcal{A})^{-k}Cx\right]\right)_{\lambda=\lambda_0}$$

and we can employ the inductive hypothesis and a simple computation in order to see that (400) holds with k replaced with k + 1 therein. Set

(401)
$$C_1 := C(\lambda_0 - \mathcal{A})^{-\lfloor \alpha + 2 \rfloor} C$$
, if $\alpha > -1$, and $C_1 := C$, if $\alpha = -1$

Then Theorem 1.2.4(i) implies by iteration that C_1 commutes with \mathcal{A} . Furthermore, the validity of (H)₀, resp. (HS)₀, implies by Theorem 1.2.8(i) that the following holds:

(H): There exist real numbers $d \in (0,1]$, $c \in (0,1)$ and $\varepsilon \in (0,1]$ such that $P_{\alpha,\varepsilon,c} \cup B_d \subseteq \rho_{C_1}(\mathcal{A})$, the operator family $\{(1+|\lambda|)^{-1}(\lambda-\mathcal{A})^{-1}C_1 : \lambda \in P_{\alpha,\varepsilon,c} \cup B_d\} \subseteq L(X)$ is equicontinuous, the mapping $\lambda \mapsto (\lambda-\mathcal{A})^{-1}C_1$ is strongly analytic on $\operatorname{int}(P_{\alpha,\varepsilon,c} \cup B_d)$ and strongly continuous on $\partial(P_{\alpha,\varepsilon,c} \cup B_d)$,

resp.,

(HS): There exist real numbers $d \in (0, 1]$ and $\vartheta \in (0, \pi/2)$ such that $\Sigma_{\vartheta} \cup B_d \subseteq \rho_{C_1}(\mathcal{A})$, the operator family $\{(1 + |\lambda|)^{-1}(\lambda - \mathcal{A})^{-1}C_1 : \lambda \in \Sigma_{\vartheta} \cup B_d\} \subseteq L(X)$ is equicontinuous, the mapping $\lambda \mapsto (\lambda - \mathcal{A})^{-1}C_1$ is strongly analytic on $\operatorname{int}(\Sigma_{\vartheta} \cup B_d)$ and strongly continuous on $\partial(\Sigma_{\vartheta} \cup B_d)$.

So, the condition (H), resp. (HS), holds and $C_1\mathcal{A} \subseteq \mathcal{A}C_1$; in particular, $-\mathcal{A}$ is C_1 -positive, resp. $-\mathcal{A}$, is C_1 -sectorial of angle $\pi - \vartheta$ and $B_d \subseteq \rho_{C_1}(-\mathcal{A})$. Once this is done, the further analyses may begin. Put $\Gamma_1(\alpha, \varepsilon, c, d) := \{\xi + i\eta : \xi \leq -\varepsilon, \eta = -c(1 + |\xi|)^{-\alpha}\}$, $\Gamma_2(\alpha, \varepsilon, c, d) := \{\xi + i\eta : \xi^2 + \eta^2 = d^2, \xi \geq -\varepsilon\}$ and $\Gamma_3(\alpha, \varepsilon, c, d) := \{\xi + i\eta : \xi \leq -\varepsilon, \eta = c(1 + |\xi|)^{-\alpha}\}$. The curve $\Gamma(\alpha, \varepsilon, c, d) :=$ $\Gamma_1(\alpha, \varepsilon, c, d) \cup \Gamma_2(\alpha, \varepsilon, c, d) \cup \Gamma_3(\alpha, \varepsilon, c, d)$ is oriented so that Im(λ) increases along $\Gamma_2(\alpha, \varepsilon, c, d)$ and that Im(λ) decreases along $\Gamma_1(\alpha, \varepsilon, c, d)$ and $\Gamma_3(\alpha, \varepsilon, c, d)$. Since there is no risk for confusion, we also write Γ for $\Gamma(\alpha, \varepsilon, c, d)$. We similarly define the curves $\Gamma_{1,S}(\vartheta, d), \Gamma_{2,S}(\vartheta, d), \Gamma_{3,S}(\vartheta, d)$ and $\Gamma_S(\vartheta, d)$ for $\vartheta \in (0, \pi/2)$ and $d \in (0, 1]$. Define

$$f_{C_1}(\mathcal{A})x := \frac{1}{2\pi i} \int_{\Gamma} f(z)(z+\mathcal{A})^{-1} C_1 x \, dz, \quad x \in X,$$

where f(z) is a holomorphic function on an open neighborhood $\Omega_{\alpha,\varepsilon,c,d}$ of $-(P_{\alpha,\varepsilon,c} \cup B_d) \smallsetminus (-\infty, 0]$ and the estimate

$$|f(z)| \leq M|z|^{-s}, \quad z \in \Omega_{\alpha,\varepsilon,c,d}$$

holds for some positive number s > 0. Denote by \mathcal{H} the class consisting of such functions. Then an application of Cauchy's theorem shows that the definition of $f_{C_1}(\mathcal{A})$ does not depend on a particular choice of curve $\Gamma(\alpha, \varepsilon, c, d)$ (with the meaning clear). Furthermore, a standard calculus involving the Cauchy theorem, the Fubini theorem and Theorem 1.2.4(ii) shows that

(402)
$$f_{C_1}(\mathcal{A})g_{C_1}(\mathcal{A}) = (fg)_{C_1}(\mathcal{A})C_1, \quad f, g, fg \in \mathcal{H}.$$

Given $b \in \mathbb{C}$ with $\operatorname{Re} b > 0$, set $(-\mathcal{A})_{C_1}^{-b} := (z^{-b})_{C_1}(\mathcal{A})$ and $(-\mathcal{A})_{C_1}^{-0} := C_1$. By Remark 3.9.3(ii) and the residue theorem, we get $(-\mathcal{A})_{C_1}^{-n} = (-\mathcal{A})^{-n}C_1$ $(n \in \mathbb{N})$; moreover, $(-\mathcal{A})_{C_1}^{-b}C_1 = C_1(-\mathcal{A})_{C_1}^{-b}$ (Re b > 0), the mapping $b \mapsto (-\mathcal{A})_{C_1}^{-b}x$, Re b > 0is analytic for every fixed $x \in X$, and the following holds:

$$\frac{d}{db}(-\mathcal{A})_{C_1}^{-b}x = \frac{(-1)}{2\pi i} \int_{\Gamma} (\ln z) z^{-b} (z+\mathcal{A})^{-1} C_1 x \, dz, \quad x \in X, \text{ Re } b > 0.$$

Applying the equality (402) once more, we get that

$$(-\mathcal{A})_{C_1}^{-b_1}(-\mathcal{A})_{C_1}^{-b_2} = (-\mathcal{A})_{C_1}^{-(b_1+b_2)}C_1, \quad \operatorname{Re} b_1, \operatorname{Re} b_2 > 0.$$

It is very simple to prove that

$$(-\mathcal{A})_{C_1}^{-b}x = -\frac{\sin \pi b}{\pi} \int_0^\infty \lambda^{-b} (\lambda - \mathcal{A})^{-1} C_1 x \, d\lambda, \quad 0 < \operatorname{Re} b < 1, \ x \in X,$$

so that the family $\{(-\mathcal{A})_{C_1}^{-b}: 0 < b < 1\}$ is equicontinuous. Define now the powers with negative imaginary part of exponent by

$$(-\mathcal{A})_{-b} := C_1^{-1} (-\mathcal{A})_{C_1}^{-b}, \quad \text{Re}\, b > 0.$$

Then $(-\mathcal{A})_{-b}$ is a closed MLO and $(-\mathcal{A})_{-n} = C_1^{-1}(-\mathcal{A})^{-n}C_1$ $(n \in \mathbb{N})$. We define the powers with positive imaginary part of exponent by

$$(-\mathcal{A})_b := ((-\mathcal{A})_{-b})^{-1} = ((-\mathcal{A})_{C_1}^{-b})^{-1}C_1, \quad \operatorname{Re} b > 0.$$

Clearly, $(-\mathcal{A})_n = C_1^{-1}(-\mathcal{A})^n C_1$ for every $n \in \mathbb{N}$, and $(-\mathcal{A})_b$ is a closed MLO due to the fact that $(-\mathcal{A})_{-b}$ is a closed MLO $(b \in \mathbb{C}_+)$. Following [410, Definition 7.1.2] and our previous analyses of non-degenerate case [103], we introduce the purely imaginary powers of $-\mathcal{A}$ as follows: Let $\tau \in \mathbb{R} \setminus \{0\}$. Then the power $(-\mathcal{A})_{i\tau}$ is defined by

$$(-\mathcal{A})_{i\tau} := C_1^{-2} (1-\mathcal{A})_2 (-\mathcal{A})_{-1} (-\mathcal{A})_{1+i\tau} (1-\mathcal{A})_{-2} C_1^2,$$

where $(1 - \mathcal{A})_2 = C_1^{-1}(1 - \mathcal{A})^2 C_1$ and $(1 - \mathcal{A})_{-2} = C_1^{-1}(1 - \mathcal{A})^{-2} C_1$. We will later see (cf. (S.4)) that $C_1(D(\mathcal{A}^2)) \subseteq D((-\mathcal{A})_{1+i\tau})$, so that the closedness of $(-\mathcal{A})_{i\tau}$ follows from a simple calculation involving the closedness of $(-\mathcal{A})_{1+i\tau}$. Further on, the Cauchy integral formula and (398) together imply that the operator family $\{\lambda^k(\lambda - \mathcal{A})^{-k}C_1 : \lambda > 0\} \subseteq L(X)$ is equicontinuous for all $k \in \mathbb{N}$. If $y \in \mathcal{A}^k x$ for some $k \in \mathbb{N}$ and $x \in D(\mathcal{A}^k)$, then there exists a sequence $(y_j)_{1 \leq j \leq k}$ in X such that $y_k = y$ and $(x, y_1) \in \mathcal{A}, (y_1, y_2) \in \mathcal{A}, \ldots, (y_{k-1}, y_k) \in \mathcal{A}$. Then we can inductively prove with the help of Theorem 1.2.4(i) that

$$\lambda^k (\lambda - \mathcal{A})^{-k} C_1 x = C_1 x + \sum_{j=1}^k \binom{k}{j} (\lambda - \mathcal{A})^{-j} C_1 y_j, \quad \lambda > 0,$$

which implies that $\lim_{\lambda \to +\infty} \lambda^k (\lambda - \mathcal{A})^{-k} C_1 x = C_1 x, k \in \mathbb{N}, x \in D(\mathcal{A}^k)$; cf. [103, Lemma 2.7]. The assertion of [103, Lemma 2.5] also holds in our framework.

Now we will revisit multivalued analogues of some statements established in [103, Theorem 2.8, Theorem 2.10, Lemma 2.14].

- (S.1) Suppose Re $b \neq 0$. Then it is checked at once that $(-\mathcal{A})_b \subseteq C_1^{-1}(-\mathcal{A})_b C_1$, with the equality in the case that the operator C_1 is injective.
- (S.2) Suppose $\operatorname{Re} b_1 < 0$ and $\operatorname{Re} b_2 < 0$. Then $R(C_1) \subseteq D((-\mathcal{A})_z)$, $\operatorname{Re} z < 0$,

$$(-\mathcal{A})_{C_1}^{b_1+b_2} x \in C_1^{-1}(-\mathcal{A})_{C_1}^{b_1+b_2} C_1 x = C_1^{-1}(-\mathcal{A})_{C_1}^{b_1}(-\mathcal{A})_{C_1}^{b_2} x$$
$$\subseteq C_1^{-1}(-\mathcal{A})_{C_1}^{b_1} C_1^{-1}(-\mathcal{A})_{C_1}^{b_2} C_1 x = (-\mathcal{A})_{b_1}(-\mathcal{A})_{b_2} C_1 x, \quad x \in X.$$

This, in turn, implies $(-\mathcal{A})_{C_1}^{b_1+b_2} \subseteq (-\mathcal{A})_{b_1}(-\mathcal{A})_{b_2}C_1, \ C_1^{-1}(-\mathcal{A})_{C_1}^{b_1+b_2} \subseteq C_1^{-1}(-\mathcal{A})_{b_1}(-\mathcal{A})_{b_2}C_1$ and

$$(-\mathcal{A})_{b_1+b_2} \subseteq C_1^{-1}(-\mathcal{A})_{b_1}(-\mathcal{A})_{b_2}C_1.$$

Let $y \in (-\mathcal{A})_{b_1}(-\mathcal{A})_{b_2}x$. Thus, $C_1y \in (-\mathcal{A})_{C_1}^{b_1}C_1^{-1}(-\mathcal{A})_{C_1}^{b_2}x$. This yields the existence of an element $u \in C_1^{-1}(-\mathcal{A})_{C_1}^{b_2}x$ such that $C_1z = (-\mathcal{A})_{C_1}^{b_2}x$ and $C_1y = (-\mathcal{A})_{C_1}^{b_1}u$. So, $C_1^2y = (-\mathcal{A})_{C_1}^{b_1}C_1u = (-\mathcal{A})_{C_1}^{b_1}(-\mathcal{A})_{C_1}^{b_2}C_1x$, $C_1y \in C_1^{-1}(-\mathcal{A})_{C_1}^{b_1+b_2}C_1x = (-\mathcal{A})_{b_1+b_2}C_1x$ and $y \in C_1^{-1}(-\mathcal{A})_{b_1+b_2}C_1x$. Hence,

(404)
$$(-\mathcal{A})_{b_1}(-\mathcal{A})_{b_2} \subseteq C_1^{-1}(-\mathcal{A})_{b_1+b_2}C_1.$$

- (S.3) Suppose now that $\operatorname{Re} b_1 > 0$ and $\operatorname{Re} b_2 > 0$. Using the equations (403)–(404) with b_1 and b_2 replaced respectively by $-b_1$ and $-b_2$ therein, and taking the inverses after that, it readily follows from (S.2) that (403)–(404) holds in this case.
- (S.4) Repeating almost verbatim the corresponding parts of the proof of [103, Theorem 2.8(ii.2)], we can deduce the following: Suppose that $\operatorname{Re} b > 0$ and $k = \lceil \operatorname{Re} b \rceil$, resp. $k = \lceil \operatorname{Re} b \rceil + 1$, provided that $\operatorname{Re} b \notin \mathbb{N}$, resp. $\operatorname{Re} b \in \mathbb{N}$. Let $x = C_1 y$ for some $y \in D(\mathcal{A}^k)$. Then there exists a sequence $(y_j)_{1 \leq j \leq k}$ in X such that $(y, y_1) \in \mathcal{A}$, $(y_1, y_2) \in \mathcal{A}, \ldots, (y_{k-1}, y_k) \in \mathcal{A}$. Furthermore, $C_1(D(\mathcal{A}^k)) \subseteq D((-\mathcal{A})_b)$ and, for every such a sequence, we have

$$\frac{1}{2\pi i} \int_{\Gamma} z^{b-\lfloor \operatorname{Re} b \rfloor - 1} (z+\mathcal{A})^{-1} C_1 y_k \, dz \in (-\mathcal{A})_b x.$$

- (S.5) The assertion of [103, Theorem 2.8(iii)] is not really interested in multivalued case because $(-\mathcal{A})_{b}x$ is not singleton, in general.
- (S.6) Let $\tau \in \mathbb{R}$. Then a straightforward calculation involving (S.1) shows that $(-\mathcal{A})_{i\tau} \subseteq C_1^{-1}(-\mathcal{A})_{i\tau}C_1$. The equality $(-\mathcal{A})_{i\tau} = C_1^{-1}(-\mathcal{A})_{i\tau}C_1$ can be also trivially verified provided that the operator C_1 is injective.
- (S.7) Let $x = C_1 y$ for some $y \in D(\mathcal{A})$, and let $\tau \in \mathbb{R}$. Keeping in mind (400), (402), (S.4), the residue theorem and Theorem 1.2.3, we can prove as in the single-valued linear case that:

$$\frac{1}{2\pi i} \int_{\Gamma} z^{-1+i\tau} \frac{z}{z+1} (z+\mathcal{A})^{-1} C_1^2 x \, dz \in (1-\mathcal{A})(-\mathcal{A})_{-1} (-\mathcal{A})_{1+i\tau} (1-\mathcal{A})_{-2} C_1^2 x.$$

Let $u \in (1 - A)y$. Using Theorem 1.2.3 and Theorem 1.2.4(i), we get from the above that

$$C_1^{-3}(1-\mathcal{A})C_1\left[\frac{1}{2\pi i}\int_{\Gamma} z^{-1+i\tau}\frac{z}{z+1}(z+\mathcal{A})^{-1}C_1^2 x\,dz\right]$$

= $\frac{1}{2\pi i}\int_{\Gamma} z^{-1+i\tau}\frac{z}{z+1}(z+\mathcal{A})^{-1}C_1 u\,dz \in (-\mathcal{A})_{i\tau}x,$

so that $C_1(D(\mathcal{A})) \subseteq D((-\mathcal{A})_{i\tau})$. Unfortunately, a great number of important properties of purely imaginary powers established in [103, Theorem 2.10] does not continue to hold in multivalued linear case.

b)...(n-b) := 1 for n = 0. Then, for every $x \in X$, we have

$$C_1^n(-\mathcal{A})_{C_1}^{-b}x = \frac{(-1)^n n!}{(1-b)(2-b)\dots(n-b)} \frac{\sin \pi (n-b)}{\pi} \int_0^\infty t^{n-b} (t-\mathcal{A})^{-(n+1)} C_1^{n+1} x \, dt.$$

This can be shown following the lines of the proof of [171, Theorem 5.27, p. 138].

3.9.2. Abstract incomplete differential inclusions. Throughout this subsection, we assume that the condition $(HS)_0$ holds. Define C_1 through (401). Then (HS) holds and we can define the fractional powers of $-\mathcal{A}$ as it has been done above.

In our previous work, we have used the function

$$f_t(\lambda) = \frac{1}{\pi} e^{-t\lambda^{\gamma} \cos \pi \gamma} \sin(t\lambda^{\gamma} \sin \pi \gamma)$$
$$= \frac{1}{2\pi i} (e^{-t\lambda^{\gamma} e^{-i\pi\gamma}} - e^{-t\lambda^{\gamma} e^{i\pi\gamma}}), \quad t > 0, \ \lambda > 0,$$

exploited by A. V. Balakrishnan in [44]. As already seen, this function enjoys the following properties:

- Q2. $\int_0^\infty \lambda^n f_t(\lambda) d\lambda = 0, \ n \in \mathbb{N}_0, \ t > 0.$ Q3. Let $m \ge -1$. Then the improper integral $\int_0^\infty \lambda^n f_t(\lambda) (\lambda \mathcal{A})^{-1} C_1 \cdot d\lambda$ is absolutely convergent and defines a bounded linear operator on X ($n \in$ \mathbb{N}_0).

Put now, for $0 < \gamma < 1/2$,

$$S_{\gamma}(t)x := \int_0^\infty f_t(\lambda)(\lambda - \mathcal{A})^{-1}C_1 x \, d\lambda, \quad t > 0, \ x \in X.$$

Then $S_{\gamma}(t) \in L(X)$, t > 0 and the following holds:

LEMMA 3.9.4. We have

(405)
$$S_{\gamma}(t) = (e^{-tz^{\gamma}})_{C_1}(\mathcal{A}), \quad t > 0, \ 0 < \gamma < 1/2.$$

Furthermore, $S_{\gamma}(t)$ can be defined by (405) for all $t \in \Sigma_{(\pi/2)-\gamma\pi}$, and the mapping $t \mapsto S_{\gamma}(t), t \in \Sigma_{(\pi/2)-\gamma\pi}$ is strongly analytic $(0 < \gamma < 1/2)$.

PROOF. Observe that, for every $t = t_1 + it_2 \in \Sigma_{(\pi/2)-\gamma\pi}$ and $z \in \mathbb{C} \smallsetminus \{0\}$, we have

$$e^{-tz^{\gamma}} | \leqslant e^{-|z|^{\gamma}t_1 \cos(\gamma \arg(z))[1-|\tan(\arg(t))|\tan(\gamma \arg(z))]}$$

Keeping this estimate in mind, it is very simple to deform the path of integration $\Gamma_S(\vartheta, d)$ into the negative real axis, showing that for each $t \in \Sigma_{(\pi/2)-\gamma\pi}$ and $x \in X$ we have:

$$\frac{1}{2\pi i} \int_{\Gamma_S(\vartheta,d)} e^{-t\lambda^{\gamma}} (\lambda + \mathcal{A})^{-1} C_1 x \, d\lambda$$
$$= \frac{1}{2\pi i} \int_0^\infty (e^{-t\lambda^{\gamma} e^{-i\pi\gamma}} - e^{-t\lambda^{\gamma} e^{i\pi\gamma}}) (\lambda - \mathcal{A})^{-1} C_1 x \, d\lambda.$$

The rest of proof is left to the interested reader.

Set

$$\varphi_{\gamma} := (\pi/2) - \gamma(\pi - \vartheta), \text{ for } 0 < \gamma \leq 1/2.$$

THEOREM 3.9.5. Put $S_{\gamma}(0) := C_1$, $S_{\gamma,\zeta}(t)x := \int_0^t g_{\zeta}(t-s)S_{\gamma}(s)x \, ds$, $x \in X$, $t \in \Sigma_{(\pi/2)-\gamma\pi}$ ($\zeta > 0$), and $S_{\gamma,0}(t) := S_{\gamma}(t)$, $t \in \Sigma_{(\pi/2)-\gamma\pi}$. Then the family $\{S_{\gamma}(t) : t > 0\}$ is equicontinuous, and there exist strongly analytic operator families $(\mathbf{S}_{\gamma}(t))_{t\in\Sigma_{\varphi\gamma}}$ and $(\mathbf{S}_{\gamma,\zeta}(t))_{t\in\Sigma_{\varphi\gamma}}$ such that $\mathbf{S}_{\gamma}(t) = S_{\gamma}(t)$, t > 0 and $\mathbf{S}_{\gamma,\zeta}(t) = S_{\gamma,\zeta}(t)$, t > 0. Furthermore, the following holds:

- (i) $\mathbf{S}_{\gamma}(t_1)\mathbf{S}_{\gamma}(t_2) = \mathbf{S}_{\gamma}(t_1 + t_2)C_1 \text{ for all } t_1, t_2 \in \Sigma_{\varphi_{\gamma}}.$
- (ii) We have $\lim_{t\to 0, t\in \Sigma_{\varphi_{\gamma}-\varepsilon}} \mathbf{S}_{\gamma}(t)x = C_1 x, x \in \overline{D(\mathcal{A})}, \varepsilon \in (0, \varphi_{\gamma}).$
- (iii) $\mathbf{S}_{\gamma}(z)(-\mathcal{A})_{\nu} \subseteq (-\mathcal{A})_{\nu} \mathbf{S}_{\gamma}(z), \ z \in \Sigma_{\varphi_{\gamma}}, \ \nu \in \mathbb{C}_{+}.$
- (iv) If D(A) is dense in X, then (S_γ(t))_{t≥0} is an equicontinuous analytic C₁-regularized semigroup of angle φ_γ. Moreover, (S_γ(t))_{t≥0} is a C₁-regularized existence family with a subgenerator -(-A)_γ and the supposition (x, y) ∈ -(-A)_γ implies (C₁x, C₁y) ∈ Â_γ, where Â_γ is the integral generator of (S_γ(t))_{t≥0}; otherwise, for every ζ > 0, (S_{γ,ζ}(t))_{t≥0} is an exponentially equicontinuous, analytic ζ-times integrated C₁-regularized semigroup, (S_{γ,ζ}(t))_{t≥0} is a ζ-times integrated C₁-existence family with a subgenerator -(-A)_γ and the supposition (x, y) ∈ -(-A)_γ implies (C₁x, C₁y) ∈ Â_γ.

 \square

(v) For every $x \in X, t \in \Sigma_{(\pi/2)-\gamma\pi}$ and $n \in \mathbb{N}$, we have

(406)
$$\left(S_{\gamma}(t)x, -\int_{0}^{\infty}\lambda^{n}f_{t}(\lambda)(\lambda-\mathcal{A})^{-1}C_{1}x\,d\lambda\right) \in \mathcal{A}^{n}.$$

(vi) Suppose $\beta > 0$. Denote by $\Omega_{\theta,\gamma}$, resp. Ψ_{γ} , the continuity set of $(S_{\gamma}(te^{i\theta}))_{t>0}$, resp. $(S_{\gamma}(t))_{t\in\Sigma_{\varphi\gamma}}$. Then, for every $x \in \Omega_{\theta,\gamma}$, the incomplete abstract Cauchy inclusion

$$(FP_{\beta}): \begin{cases} u \in C^{\infty}((0,\infty):X), \\ D_{-}^{\beta}u(t) \in e^{i\theta\beta}(-\mathcal{A})_{\gamma\beta}u(t), \quad t > 0, \\ \lim_{t \to 0+} u(t) = C_{1}x, \\ the \ set \ \{u(t):t > 0\} \ is \ bounded \ in \ X, \end{cases}$$

has a solution $u(t) = S_{\gamma}(te^{i\theta})x, t > 0$, which can be analytically extended to the sector $\Sigma_{\varphi_{\gamma}-|\theta|}$. If, additionally, $x \in \Psi_{\gamma}$, then for every $\delta \in (0, \varphi_{\gamma})$ and $j \in \mathbb{N}_0$, we have that the set $\{z^j u^{(j)}(z) : z \in \Sigma_{\delta}\}$ is bounded in X.

PROOF. The proof of (i) for real parameters $t_1, t_2 > 0$ follows almost directly from definition of $S_{\gamma}(\cdot)$, by applying (402); (v) is an easy consequence of Theorem 1.2.3, Theorem 1.2.4(i) and the property Q2. A very simple proof of (iii) is omitted. Set, for $|\theta| < \vartheta$ and $0 < \gamma < 1/2$,

$$S_{\theta,\gamma}(t)x := \int_0^\infty f_{t,\gamma}(\lambda)(\lambda - e^{i\theta}\mathcal{A})^{-1}C_1x\,d\lambda, \quad x \in X, \ t \in \Sigma_{(\pi/2) - \gamma\pi}.$$

Let $\theta_1 \in (0, \vartheta)$ and $\theta_2 \in (-\vartheta, 0)$. Define

$$\mathbf{S}_{\gamma}(t)x := \begin{cases} S_{\gamma}(t)x, \ t \in \Sigma_{(\pi/2) - \gamma\pi}, \\ S_{\theta_1,\gamma}(te^{-i\gamma\theta_1}), \ \text{if} \ t \in e^{i\gamma\theta_1}\Sigma_{(\pi/2) - \gamma\pi}, \\ S_{\theta_2,\gamma}(te^{-i\gamma\theta_2}), \ \text{if} \ t \in e^{i\gamma\theta_2}\Sigma_{(\pi/2) - \gamma\pi}. \end{cases}$$

Then an elementary application of Cauchy formula shows that the operator family $(\mathbf{S}_{\gamma}(t))_{t\in\Sigma_{\varphi_{\gamma}}}$ is well defined; furthermore, $(\mathbf{S}_{\gamma}(t))_{t\in\Sigma_{\varphi_{\gamma}}}$ is strongly analytic and equicontinuous on any proper subsector of $\Sigma_{\varphi_{\gamma}}$ (cf. also the proof of [**291**, Theorem 2.9.48]). Using Theorem 1.2.4(i), we get that $\lim_{\lambda\to+\infty} [\lambda(\lambda-\mathcal{A})^{-1}C_1x - \lambda(\lambda+1)^{-1}C_1x] = 0$ as $\lambda \to +\infty$ ($x \in D(\mathcal{A})$). Taking into account this equality and the proof of [**410**, Theorem 5.5.1(iv), p. 130], we get that $\lim_{t\to 0+} S_{\gamma}(t)x = C_1x$, $x \in D(\mathcal{A})$. Now the proofs of (i)-(iii) can be straightforwardly completed.

We will prove (iv) provided that $D(\mathcal{A})$ is dense in E. It is clear that $(S_{\gamma}(t))_{t\geq 0}$ is an equicontinuous analytic C_1 -regularized semigroup $(S_{\gamma}(t))_{t\geq 0}$ of angle φ_{γ} . Since, for every t > 0 and $x \in X$,

$$C_1(-z^{-\gamma}e^{-tz^{\gamma}}+z^{-\gamma})_{C_1}(\mathcal{A})x = -(z^{-\gamma})_{C_1}(\mathcal{A})[(e^{-tz^{\gamma}})_{C_1}(\mathcal{A})x - C_1x],$$

we have

$$C_1 S_{\gamma,1}(t) x = -(-\mathcal{A})_{C_1}^{-\gamma} [S_{\gamma}(t) x - C_1 x], \quad t \ge 0, \ x \in X.$$

This clearly implies that $(S_{\gamma,1}(t)x, S_{\gamma,1}(t)x - C_1x) \in -(-\mathcal{A})_{\gamma}, t \ge 0, x \in X$, so that $(S_{\gamma,\zeta}(t))_{t\ge 0}$ is a ζ -times integrated C_1 -existence family with a subgenerator $-(-\mathcal{A})_{\gamma}$. The supposition $(x, y) \in -(-\mathcal{A})_{\gamma}$ implies $C_1x = -(-\mathcal{A})_{C_1}^{-\gamma}y$ and we can

similarly prove that $(C_1 x, C_1 y) \in \hat{\mathcal{A}}_{\gamma}$. Arguing as in the proof of [103, Theorem 3.5(i)/(b)'], we get that, for every $x \in X$ and t > 0, the following equality holds, with $z = te^{i\theta} \in \Sigma_{(\pi/2)-\gamma\pi}$,

(407)
$$D^{\beta}_{-}S_{\gamma}(te^{i\theta})x = \frac{e^{i\theta\beta}}{2\pi i} \int_{0}^{\infty} \lambda^{\gamma\beta} [e^{-i\gamma\beta\pi} e^{-z\lambda^{\gamma}e^{-i\pi\gamma}} - e^{i\gamma\beta\pi} e^{-z\lambda^{\gamma}e^{i\pi\gamma}}](\lambda - \mathcal{A})^{-1}C_{1}x \, d\lambda.$$

Deforming the path of integration $\Gamma_S(\vartheta, d)$ into the negative real axis, as it has been done in the proof of Lemma 3.9.4, we get

(408)
$$(\cdot^{\gamma\beta}e^{-z\cdot^{\gamma}})_{C_1}(\mathcal{A})$$

= $\frac{1}{2\pi i}\int_0^\infty \lambda^{\gamma\beta} [e^{-i\gamma\beta\pi}e^{-z\lambda^{\gamma}e^{-i\pi\gamma}} - e^{i\gamma\beta\pi}e^{-z\lambda^{\gamma}e^{i\pi\gamma}}](\lambda - \mathcal{A})^{-1}C_1x d\lambda.$

Since

$$C_1(e^{-z\cdot\gamma})_{C_1}(\mathcal{A}) = (\cdot^{-\gamma\beta})_{C_1}(\mathcal{A})(\cdot^{-\gamma\beta}e^{-z\cdot\gamma})_{C_1}(\mathcal{A$$

(407)-(408) immediately implies that

$$(e^{-i\theta\beta}D^{\beta}_{-}S_{\gamma}(te^{i\theta})x, S_{\gamma}(te^{i\theta})x) \in C_{1}^{-1}(-\mathcal{A})^{\gamma\beta}_{C_{1}}, \quad t > 0, \ x \in X,$$

i.e.,

$$(S_{\gamma}(te^{i\theta})x, e^{-i\theta\beta}D_{-}^{\beta}S_{\gamma}(te^{i\theta})x) \in (-\mathcal{A})_{\gamma\beta}, \quad t > 0, \ x \in X.$$

The proof of (vi) now can be completed routinely.

- REMARK 3.9.6. (i) If $l = \beta \gamma \in \mathbb{N}$, then the operator $(-\mathcal{A})_{\gamma\beta}$ in the formulation of problem $(FP)_{\beta}$ can be replaced with the operator $(-\mathcal{A})^{l}$ therein; cf. (406).
- (ii) Suppose that the operator C_1 is injective. Then we can simply prove that $(S_{\gamma,\zeta}(t))_{t\geq 0}$ is a ζ -times integrated C_1 -semigroup with a subgenerator $-(-\mathcal{A})_{\gamma}$, which implies that the integral generator of $(S_{\gamma,\zeta}(t))_{t\geq 0}$ is $-C_1^{-1}(-\mathcal{A})_{\gamma}C_1 = -(-\mathcal{A})_{\gamma}$. A similar statement holds in the case that $\gamma = 1/2$, which is further discussed in the following theorem.

THEOREM 3.9.7. The limit contained in the expression

(409)
$$S_{1/2}(t)x := \frac{1}{\pi} \lim_{N \to \infty} \int_0^N \sin(t\sqrt{\lambda})(\lambda - \mathcal{A})^{-1} C_1 x \, d\lambda, \quad t > 0,$$

exists in L(X) for every $x \in X$. Put $S_{1/2}(0) := C_1$. Then the family $\{S_{1/2}(t) : t > 0\}$ is equicontinuous, there exists a strongly analytic operator family $(\mathbf{S}_{1/2}(t))_{t \in \Sigma_{\varphi_{1/2}}}$ such that $\mathbf{S}_{1/2}(t) = S_{1/2}(t)$, t > 0 and the following holds:

(i)
$$\mathbf{S}_{1/2}(t)\mathbf{S}_{1/2}(s) = \mathbf{S}_{1/2}(t+s)C_1$$
 for all $t, s \in \Sigma_{\varphi_{1/2}}$.

(ii)
$$\lim_{t\to 0, t\in \Sigma_{\varphi_1/2}-\varepsilon} \mathbf{S}_{1/2}(t)x = C_1 x, x \in D(\mathcal{A}), \varepsilon \in (0, \varphi_{1/2}).$$

- (iii) $\mathbf{S}_{1/2}(t)(-\mathcal{A})_{\nu} \subseteq (-\mathcal{A})_{\nu} \mathbf{S}_{1/2}(t), t \in \Sigma_{\varphi_{1/2}}, \nu \in \mathbb{C}_+.$
- (iv) If $D(\mathcal{A})$ is dense in X, then $(S_{1/2}(t))_{t \ge 0}$ is an equicontinuous analytic C_1 -regularized semigroup of angle φ_{γ} . Furthermore, $(S_{1/2}(t))_{t \ge 0}$ is a C_1 -regularized existence family with a subgenerator $-(-\mathcal{A})_{1/2}$ and the supposition $(x, y) \in -(-\mathcal{A})_{1/2}$ implies $(C_1x, C_1y) \in \hat{\mathcal{A}}_{1/2}$, where $\hat{\mathcal{A}}_{1/2}$ is the integral generator of $(S_{1/2}(t))_{t \ge 0}$; otherwise, for every $\zeta > 0$,

 \square

 $(S_{1/2,\zeta}(t))_{t \ge 0}$ is an exponentially equicontinuous, analytic ζ -times integrated C_1 -regularized semigroup, $(S_{1/2,\zeta}(t))_{t \ge 0}$ is a ζ -times integrated C_1 -existence family with a subgenerator $-(-\mathcal{A})_{1/2}$ and the supposition $(x, y) \in -(-\mathcal{A})_{1/2}$ implies $(C_1x, C_1y) \in \hat{\mathcal{A}}_{1/2}$.

(v) Then $R(S_{1/2}(t)) \subseteq D_{\infty}(\mathcal{A}), t > 0$ and, for every $x \in \overline{D(\mathcal{A})}$, the incomplete abstract Cauchy problem

$$(P_2): \begin{cases} u \in C^{\infty}((0,\infty):X), \\ u''(t) \in -\mathcal{A}u(t), \quad t > 0, \\ \lim_{t \to 0+} u(t) = C_1 x, \\ the \ set \ \{u(t):t > 0\} \ is \ bounded \ in \ X, \end{cases}$$

has a solution $u(t) = S_{1/2}(t)x$, t > 0. Moreover, the mapping $t \mapsto u(t)$, t > 0 can be analytically extended to the sector $\Sigma_{\varphi_{1/2}}$ and, for every $\delta \in (0, \varphi_{1/2})$ and $j \in \mathbb{N}_0$, we have that the set $\{z^j u^{(j)}(z) : z \in \Sigma_\delta\}$ is bounded in X.

PROOF. First of all, observe that $\varphi_{1/2} = \vartheta/2$. Applying the partial integration, (398) and the equicontinuity of family $\{\lambda^2(\lambda - \mathcal{A})^{-2}C_1 : \lambda > 0\}$, we obtain that the limit contained in (409) exists and equals

$$S_{1/2}(t)x = \int_0^\infty f(\lambda, t)(\lambda - \mathcal{A})^{-2}C_1 x \, d\lambda, \quad t > 0, \ x \in X,$$

where $f(\lambda, t) = 2\pi^{-1}t^{-2}[\sin(t\sqrt{\lambda}) - t\sqrt{\lambda}\cos(t\sqrt{\lambda})]$ for $\lambda > 0$ and t > 0. As in the single-valued case, the change of variables $x = t\sqrt{\lambda}$ shows that the operator family $\{S_{1/2}(t) : t > 0\}$ is both equicontinuous and strongly continuous. Let $(x, y) \in \mathcal{A}$. Then Theorem 1.2.4(i) shows that

$$S_{1/2}(t)x - C_1 x = \frac{1}{\pi} \lim_{N \to \infty} \int_0^N \sin(t\sqrt{\lambda})((\lambda + A)^{-1}C_1 x - \lambda^{-1}C_1 x)d\lambda$$
$$= \frac{1}{\pi} \lim_{N \to \infty} \int_0^N \frac{\sin(t\sqrt{\lambda})}{\lambda} (\lambda - A)^{-1}C_1 y d\lambda.$$

Keeping in mind the last equality and the equicontinuity of family $\{S_{1/2}(t) : t \ge 0\}$, we get that $\lim_{t\to 0} S_{1/2}(t)x = C_1x$ for all $x \in \overline{D(\mathcal{A})}$. Now we proceed as in the proof of Theorem 2.7.5. Let $0 < \delta' < \delta < \vartheta/2$, $1/2 > \gamma_0 > \delta/\vartheta$ and $\theta \in (-\vartheta, (-\delta)/\gamma_0)$. Then, for every $\gamma \in (\gamma_0, 1/2)$, we have $\theta \in (-\vartheta, (-\delta)/\gamma)$ and $\gamma > \delta/\vartheta$. Let $\varepsilon \in (0, (\pi - \vartheta)/2)$ be sufficiently small. Define, for every $\gamma \in (\gamma_0, 1/2)$ and $x \in X$,

$$F_{\gamma}(\lambda)x := \begin{cases} \frac{e^{i\theta\gamma}\sin\gamma\pi}{\pi} \int_{0}^{\infty} \frac{v^{\gamma}(v-e^{i\theta}\mathcal{A})^{-1}C_{1}x\,dv}{(\lambda e^{i\theta\gamma}+v^{\gamma}\cos\pi\gamma)^{2}+v^{2\gamma}\sin^{2}\gamma\pi}, \text{ if } \arg(\lambda) \in (-\varepsilon, (\pi/2)+\delta), \\ \frac{e^{-i\theta\gamma}\sin\gamma\pi}{\pi} \int_{0}^{\infty} \frac{v^{\gamma}(v-e^{-i\theta}\mathcal{A})^{-1}C_{1}x\,dv}{(\lambda e^{-i\theta\gamma}+v^{\gamma}\cos\pi\gamma)^{2}+v^{2\gamma}\sin^{2}\gamma\pi}, \text{ if } \arg(\lambda) \in (-(\pi/2)-\delta,\varepsilon). \end{cases}$$

If $x \in X$ and $\arg(\lambda) \in (-\varepsilon, (\pi/2) + \delta)$, resp., $\arg(\lambda) \in (-(\pi/2) - \delta, \varepsilon)$, then

(410)
$$\int_0^\infty e^{-\lambda e^{i\theta\gamma}t} S_{\theta,\gamma}(t) x \, dt = \frac{\sin\gamma\pi}{\pi} \int_0^\infty \frac{v^{\gamma}(v-e^{i\theta}\mathcal{A})^{-1}C_1 x}{(\lambda e^{i\theta\gamma}+v^{\gamma}\cos\pi\gamma)^2+v^{2\gamma}\sin^2\gamma\pi} dv,$$

resp.,

(411)
$$\int_0^\infty e^{-\lambda e^{-i\theta\gamma}t} S_{-\theta,\gamma}(t) x \, dt = \frac{\sin\gamma\pi}{\pi} \int_0^\infty \frac{v^\gamma (v+e^{-i\theta}\mathcal{A})^{-1} C_1 x}{(\lambda e^{-i\theta\gamma} + v^\gamma \cos\pi\gamma)^2 + v^{2\gamma} \sin^2\gamma\pi} dv.$$

Furthermore,

(412)
$$e^{i\theta\gamma} \int_0^\infty e^{-\lambda e^{i\theta\gamma}t} S_{\theta,\gamma}(t) x \, dt = e^{-i\theta\gamma} \int_0^\infty e^{-\lambda e^{-i\theta\gamma}t} S_{-\theta,\gamma}(t) x \, dt, \quad \lambda \in \Sigma_{\varepsilon}.$$

By (410)–(412), we deduce that the function $\lambda \mapsto F_{\gamma}(\lambda)x$, $\lambda \in \Sigma_{(\pi/2)+\delta}$ is well defined, analytic and bounded by $\operatorname{Const}_{\delta'} |\lambda|^{-1}$ on sector $\Sigma_{(\pi/2)+\delta'}$ $(x \in X)$, as well as

(413)
$$S_{\gamma}(z)x = \frac{1}{2\pi i} \int_{\Gamma_{\delta',z}} e^{\lambda z} F_{\gamma}(\lambda) x \, d\lambda, \quad x \in X, \ z \in \Sigma_{\delta'}, \ \gamma \in (\gamma_0, 1/2),$$

where $\Gamma_{\delta',z} := \Gamma_{\delta',z,1} \cup \Gamma_{\delta',z,2}$, $\Gamma_{\delta',z,1} := \{re^{i((\pi/2)+\delta')} : r \ge |z|^{-1}\} \cup \{|z|^{-1}e^{i\vartheta} : \vartheta \in [0, (\pi/2)+\delta']\}$ and $\Gamma_{\delta',z,2} := \{re^{-i((\pi/2)+\delta')} : r \ge |z|^{-1}\} \cup \{|z|^{-1}e^{i\vartheta} : \vartheta \in [-(\pi/2)-\delta', 0]\}$ are oriented counterclockwise. The dominated convergence theorem shows that, for every $x \in X$ and $z \in \Sigma_{\delta'}$,

$$\lim_{\gamma \to \frac{1}{2}-} S_{\gamma}(z)x = \frac{e^{i\theta/2}}{2\pi^2 i} \int_{\Gamma_{\delta',z,1}} e^{\lambda z} \int_0^\infty \frac{v^{1/2} (v - e^{i\theta} \mathcal{A})^{-1} C_1 x}{\lambda^2 e^{i\theta} + v} dv \, d\lambda$$
$$+ \frac{e^{-i\theta/2}}{2\pi^2 i} \int_{\Gamma_{\delta',z,2}} e^{\lambda z} \int_0^\infty \frac{v^{1/2} (v - e^{-i\theta} \mathcal{A})^{-1} C_1 x}{\lambda^2 e^{-i\theta} + v} dv \, d\lambda$$
$$:= \mathbf{S}_{1/2}(z) x.$$

Define $F_{1/2}(\lambda)$ by replacing the number γ with the number 1/2 in definition of $F_{\gamma}(\lambda)$. Then, for every $x \in X$, the function $\lambda \mapsto F_{1/2}(\lambda)x$, $\lambda \in \Sigma_{(\pi/2)+\delta}$ is well defined and analytic on $\Sigma_{(\pi/2)+\delta}$; furthermore, for each $q \in \circledast$ there exists $r_q \in \circledast$ such that $q(F_{1/2}(\lambda)x) \leq r_q(x) \operatorname{Const}_{\delta'} |\lambda|^{-1}$, $\lambda \in \Sigma_{(\pi/2)+\delta'}$, $x \in X$ [**317**]. Define $(\mathbf{S}_{1/2}(z))_{z \in \Sigma_{\vartheta/2}} \subseteq L(X)$ by $\mathbf{S}_{1/2}(z)x := \lim_{\gamma \to \frac{1}{2}^{-}} S_{\gamma}(z)x$, $z \in \Sigma_{\vartheta/2}$, $x \in X$; this operator family is equicontinuous on any proper subsector of $\Sigma_{\vartheta/2}$ and satisfies additionally that the mapping $z \mapsto \mathbf{S}_{1/2}(z)x$, $z \in \Sigma_{\vartheta/2}$ is analytic for all $x \in X$. Letting $\gamma \to \frac{1}{2}$ - in (413), it is not difficult to prove that

$$\mathbf{S}_{\frac{1}{2}}(z)x = \frac{1}{2\pi i} \int_{\Gamma_{\delta',z}} e^{\lambda z} F_{\frac{1}{2}}(\lambda) x \, d\lambda, \quad x \in X, \ z \in \Sigma_{\delta'},$$

so that the proof of [27, Theorem 2.6.1] implies

(414)
$$\int_0^\infty e^{-\lambda t} \mathbf{S}_{\frac{1}{2}}(t) x \, dt = F_{\frac{1}{2}}(\lambda) x, \quad x \in X, \ \lambda > 0.$$

On the other hand, by the proof of [410, Theorem 5.5.2, p. 133] we have

(415)
$$\int_{0}^{\infty} e^{-\lambda t} S_{\frac{1}{2}}(t) x \, dt = \frac{1}{\pi} \int_{0}^{\infty} \frac{\sqrt{\nu}}{\lambda^{2} + \nu} (\nu - \mathcal{A})^{-1} C_{1} x \, d\nu = F_{\frac{1}{2}}(\lambda) x, \quad x \in X, \ \lambda > 0.$$

Using the uniqueness theorem for the Laplace transform, we obtain from (414)–(415) that $\mathbf{S}_{1/2}(t) = S_{1/2}(t), t > 0$. Now the proofs of (i)–(iii) become standard and therefore omitted.

For simplicity, we assume that \mathcal{A} is densely defined in (iv). Then the only non-trivial thing that should be proved is that the supposition $(x, y) \in -(-\mathcal{A})_{1/2}$ implies $(C_1x, C_1y) \in \hat{\mathcal{A}}_{1/2}$. So, let $(x, y) \in -(-\mathcal{A})_{1/2}$, i.e., $C_1x = -(-\mathcal{A})_{C_1}^{-1/2}y$. A similar line of reasoning as in the proof of identity [103, (51), p. 489] shows that

$$C_1 \int_0^\infty e^{-\lambda t} S_{\gamma}(t) y \, dt = C_1 (-\mathcal{A})_{C_1}^{-\gamma} y - \lambda \int_0^\infty e^{-\lambda t} S_{\gamma}(t) y \, dt, \quad \lambda > 0, \ \gamma \in (0, 1/2).$$

Taking the limits of both sides of previous equality when $\gamma \to 1/2-$, we get that

$$C_1 \int_0^\infty e^{-\lambda t} S_{1/2}(t) y \, dt = C_1 (-\mathcal{A})_{C_1}^{-1/2} y - \lambda \int_0^\infty e^{-\lambda t} S_{1/2}(t) y \, dt, \quad \lambda > 0.$$

Then the uniqueness theorem for Laplace transform simply implies that

$$S_{1/2}(t)C_1x - C_1^2x = \int_0^t S_{1/2}(s)C_1y\,ds, \quad t \ge 0,$$

as claimed.

Now we will prove (v) by slightly modifying the corresponding part of proof of Theorem 2.7.5(i). In order to do that, we will first show that for each $x \in X$ we have $S_{1/2}''(t)x \in -\mathcal{A}S_{1/2}(t)x, t > 0$. Fix temporarily an element $x \in X$. Owing to Theorem 3.9.5(v) and (407), cf. also Remark 3.9.6(i), we have that

$$D_{-}^{\frac{1}{\gamma}}S_{\gamma}'(t)x \in -\mathcal{A}S_{\gamma}(t)x, \quad t > 0,$$

i.e.,

$$\frac{d^2}{dt^2} \int_0^\infty g_{3-\frac{1}{\gamma}}(s) S_{\gamma}'(t+s) x \, ds \in \mathcal{A}S_{\gamma}(t) x, \quad t > 0, \ \gamma \in (\gamma_0, 1/2).$$

Therefore,

$$\int_0^\infty g_{3-\frac{1}{\gamma}}(s) S_{\gamma}''(t+s) x \, ds \in \mathcal{A}S_{\gamma}(t) x, \quad t > 0, \ \gamma \in (\gamma_0, 1/2).$$

Applying the partial integration, we get

$$\int_{0}^{\infty} g_{4-\frac{1}{\gamma}}(s) S_{\gamma}^{(iv)}(t+s) x \, ds \in -\mathcal{A}S_{\gamma}(t) x, \quad t > 0, \ \gamma \in (\gamma_{0}, 1/2).$$

The dominated convergence theorem yields by letting $\gamma \to 1/2$ – that

$$\int_0^\infty s S_{1/2}^{(iv)}(t+s) x \, ds \in -\mathcal{A}S_{1/2}(t) x, \quad t > 0,$$

which clearly implies after an application of integration by parts that $S_{1/2}''(t)x \in -\mathcal{A}S_{1/2}(t)x, t > 0$, as claimed. By (ii), the function $u(t) = S_{1/2}(t)x, t > 0$ is a solution of problem (P_2) for $x \in \overline{D(\mathcal{A})}$. Furthermore, we obtain by induction that $S_{1/2}^{(2n)}(t)x \in (-1)^n \mathcal{A}^n S_{1/2}(t)x, t > 0, n \in \mathbb{N}, x \in X$, so that $R(S_{1/2}(t)) \subseteq D_{\infty}(\mathcal{A}), t > 0$. The proof of the theorem is thereby complete.

Examples of exponentially bounded integrated semigroups generated by multivalued linear operators can be found in [199, Section 5.3, Section 5.8] (cf. also Example 3.2.11(i)). These example can serve one to provide possible applications of Theorem 3.9.5 and Theorem 3.9.7.

EXAMPLE 3.9.8. Assume that $M_p = p!^s$ for some s > 1. Set $\omega(z) := \prod_{i=1}^{\infty} (1 + \frac{iz}{p^s}), z \in \mathbb{C}$. Suppose that there exist constants l > 0 and $\omega > 0$ satisfying that $RHP_{\omega} \equiv \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda > \omega\} \subseteq \rho(\mathcal{A})$ and the operator family $\{e^{-M(l|\lambda|)}R(\lambda : \mathcal{A}) \mid \lambda \in RHP_{\omega}\} \subseteq L(X)$ is equicontinuous (cf. [291, 292] for a great number of such examples with \mathcal{A} being single-valued, and Example 3.2.65 for purely multivalued linear case, with X being a Fréchet space); here, $M(\cdot)$ denotes the associated function of sequence (M_p) . Let $\bar{\omega} > \omega$. Then there exists a sufficiently large number $n \in \mathbb{N}$ such that the expression

$$S(t) := \frac{1}{2\pi i} \int_{\bar{\omega} - i\infty}^{\bar{\omega} + i\infty} e^{\lambda t} \frac{R(\lambda : \mathcal{A})}{\omega^n(i\lambda)} d\lambda, \quad t \geqslant 0$$

defines an exponentially equicontinuous $C \equiv S(0)$ -regularized semigroup $(S(t))_{t \ge 0}$ with a subgenerator \mathcal{A} (cf. Theorem 1.2.3, Theorem 1.2.4(i) and the proof of [291, Theorem 3.6.4]). It is not difficult to see that $(\lambda - \mathcal{A})^{-1}Cf = \int_0^\infty e^{-\lambda t}S(t)f dt$, $\operatorname{Re} \lambda > \bar{\omega}, f \in X$, so that Theorem 3.9.5 and Theorem 3.9.7 can be applied with the operator \mathcal{A} replaced with the operator $\mathcal{A} - \bar{\omega}$ therein. Observe that, even in singlevalued linear case, the operator C need not be injective because our assumptions do not imply that $\mathcal{A} = A$ generates an ultradistribution semigroup of Beurling class.

In this book, we will not discuss the generation of degenerate fractional regularized resolvent families by the negatives of constructed fractional powers. For more details, cf. [103, Section 3] and [291, Remark 2.9.49].

At the end, we would like to observe that the assertions of Theorem 2.7.3–Theorem 2.7.5 can be formulated in the multivalued linear operators setting. For applications, the most important is the following case: X is a Banach space, $\Sigma_{\vartheta} \cup$ $B_d \subseteq \rho(\mathcal{A})$, there exist finite numbers $M_1 \ge 1$ and $\nu \in (0, 1]$ such that (13) holds with the operator \mathcal{A} and number β replaced with $-\mathcal{A}$ and ν therein; see [199, Chapter III, Chapter VI] for a great number of concrete examples. Define the operators $S_{\gamma}(\cdot)$ as before. Then we may conclude the following:

- (i) Suppose that $\beta\gamma > 1 \nu$. Then $(S_{\gamma}(t))_{t \in \Sigma_{\varphi\gamma}}$ is an analytic semigroup of growth order $\frac{1-\nu}{\gamma}$. Denote by $\Omega_{\theta,\gamma}$, resp. Ψ_{γ} , the continuity set of $(S_{\gamma}(te^{i\theta}))_{t>0}$, resp. $(S_{\gamma}(t))_{t \in \Sigma_{\varphi\gamma}}$. Then $\overline{D(\mathcal{A})} \subseteq \Psi_{\gamma}$ and, for every $x \in \Omega_{\theta,\gamma}$, the incomplete abstract Cauchy inclusion (FP_{β}) , with $C_1 = I$, has a solution $u(t) = S_{\gamma}(te^{i\theta})x, t > 0$, which can be analytically extended to the sector $\Sigma_{\varphi\gamma-|\theta|}$. If, additionally, $x \in \Psi_{\gamma}$, then for every $\delta \in (0, \varphi_{\gamma})$ and $j \in \mathbb{N}_0$, we have that the set $\{z^j u^{(j)}(z) : z \in \Sigma_{\delta}\}$ is bounded in X.
- (ii) Suppose that $1/2 < \nu < 1$. Then the incomplete abstract Cauchy problem (P_2) , with $C_1 = I$, has a solution u(t), t > 0 for all $x \in D(\mathcal{A})$. Moreover, the mapping $t \mapsto u(t)$, t > 0 can be analytically extended to the sector

 $\Sigma_{\varphi_{1/2}}$ and, for every $\delta \in (0, \varphi_{1/2})$ and $j \in \mathbb{N}_0$, we have that the set $\{z^j(1+|z|^{2\nu-2})^{-1}u^{(j)}(z): z \in \Sigma_{\delta}\}$ is bounded in X.

As announced before, it is very non-trivial to find some necessary and sufficient conditions ensuring the uniqueness of solutions of problems (FP_{β}) and (P_2) .

3.10. Inverse generator problem

The main purpose of this section is to analyze the inverse generator problem for abstract degenerate Volterra integro-differential equations in sequentially complete locally convex spaces. More specifically, we consider the problem of generation of mild (a, k)-regularized (C_1, C_2) -existence and uniqueness families and (a, k)-regularized C-resolvent families by the inverses of closed multivalued linear operators. As before, by X and Y we denote two Hausdorff sequentially complete locally convex spaces over the field of complex numbers. The injectiveness of operators C, C_1 , C_2 , if needed, will be explicitly emphasized.

We start by stating the following useful result:

PROPOSITION 3.10.1. Suppose that $C \in L(X)$, $\lambda \in \mathbb{C} \setminus \{0\}$, \mathcal{A} is an MLO and $\lambda^{-1} \in \rho_C(\mathcal{A})$. Then we have $\lambda \in \rho_C(\mathcal{A}^{-1})$ and

$$(\lambda - \mathcal{A}^{-1})^{-1}C = \lambda^{-1}[C - \lambda^{-1}(\lambda^{-1} - \mathcal{A})^{-1}C].$$

PROOF. Suppose $x \in X$. Then a simple computation involving the definition of inverse of an MLO shows that

$$(Cx,\lambda^{-1}[Cx-\lambda^{-1}(\lambda^{-1}-\mathcal{A})^{-1}Cx]) \in (\lambda-\mathcal{A}^{-1})^{-1}$$

iff

$$-Cx + \lambda^{-1}(\lambda^{-1} - \mathcal{A})^{-1}Cx \in \mathcal{A}(\lambda^{-1} - \mathcal{A})^{-1}Cx$$

which is true due to Theorem 1.2.4(i). It suffices to prove that the operator $(\lambda - \mathcal{A}^{-1})^{-1}C$ is single-valued. If we suppose that $\{y, z\} \subseteq (\lambda - \mathcal{A}^{-1})^{-1}Cx$, then we have $\lambda y - Cx \in \mathcal{A}^{-1}y$ and $\lambda z - Cx \in \mathcal{A}^{-1}z$. Hence, $y \in \mathcal{A}[\lambda y - Cx]$ and $z \in \mathcal{A}[\lambda z - Cx]$. This simply implies $Cx \in (\lambda^{-1} - \mathcal{A})^{-1}C[\lambda Cx - \lambda y]$ and $Cx \in (\lambda^{-1} - \mathcal{A})^{-1}C[\lambda Cx - \lambda z]$. Since the operator $(\lambda^{-1} - \mathcal{A})^{-1}C$ is single-valued, we simply get from the above that $y = z = \lambda^{-1}[Cx - \lambda^{-1}(\lambda^{-1} - \mathcal{A})^{-1}Cx]$.

Now we will state a general result which gives the necessary and sufficient conditions for a multivalued linear operator \mathcal{A}^{-1} to be a subgenerator of an exponentially equicontinuous mild (a, k)-regularized C_1 -existence family or an exponentially equicontinuous mild (a, k)-regularized C_2 -uniqueness family.

PROPOSITION 3.10.2. Suppose that \mathcal{A} is a closed MLO in $X, C_1 \in L(Y, X), C_2 \in L(X), |a(t)| and k(t)$ satisfy (P1), as well as that $(R_1(t), R_2(t))_{t \ge 0} \subseteq L(Y, X) \times L(X)$ is strongly continuous. Let $\omega \ge \max(0, \operatorname{abs}(|a|), \operatorname{abs}(k))$ be such that the operator family $\{e^{-\omega t}R_i(t) : t \ge 0\}$ is equicontinuous for i = 1, 2. Then the following holds:

(i) $(R_1(t), R_2(t))_{t \ge 0}$ is a mild (a, k)-regularized (C_1, C_2) -existence and uniqueness family with a subgenerator \mathcal{A}^{-1} iff for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0$, we have $R(C_1) \subseteq R(\tilde{a}(\lambda) - \mathcal{A})$,

(416)
$$\tilde{a}(\lambda)\tilde{k}(\lambda)C_1y \in (\tilde{a}(\lambda) - \mathcal{A})\int_0^\infty e^{-\lambda t}[k(t)C_1y - R_1(t)y]dt, \quad y \in Y$$

and

(417)
$$\tilde{k}(\lambda)C_2x = \int_0^\infty e^{-\lambda t} [R_2(t)x - (a * R_2)(t)y] dt, \quad \text{whenever } (y, x) \in \mathcal{A}.$$

- (ii) $(R_1(t))_{t\geq 0}$ is a mild (a,k)-regularized C_1 -existence family with a subgenerator \mathcal{A}^{-1} iff for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0$, we have $R(C_1) \subseteq R(\tilde{a}(\lambda) - \mathcal{A})$ and (416).
- (iii) $(R_2(t))_{t\geq 0}$ is a mild (a,k)-regularized C_2 -uniqueness family with a subgenerator \mathcal{A}^{-1} iff (417) holds for $\operatorname{Re} \lambda > \omega$.

PROOF. It is clear that \mathcal{A}^{-1} is a closed MLO. The part (iii) follows immediately from Lemma 3.2.44 and definition of \mathcal{A}^{-1} . For the rest, it suffices to prove (ii). Suppose first that, for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0$, we have $R(C_1) \subseteq R(\tilde{a}(\lambda) - \mathcal{A})$ and (416). Then a simple computation gives that for such values of parameter λ we have

$$-\int_0^\infty e^{-\lambda t} (a * R_1)(t) y \, dt \in \mathcal{A} \int_0^\infty e^{-\lambda t} [k(t)C_1 y - R_1(t)y] dt, \quad y \in Y,$$

 $R(C_1) \subseteq R(I - \tilde{a}(\lambda)\mathcal{A}^{-1})$ and

(418)
$$\tilde{k}(\lambda)C_1y \in (I - \tilde{a}(\lambda)\mathcal{A}^{-1}) \int_0^\infty e^{-\lambda t} R_1(t)y \, dt, \quad y \in Y.$$

By Lemma 3.2.44(ii), we get that $(R_1(t))_{t\geq 0}$ is a mild (a, k)-regularized C_1 -existence family with a subgenerator \mathcal{A}^{-1} . For the converse, we can apply Lemma 3.2.44(ii) again so as to conclude that, for every $y \in Y$ and for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0$, we have $R(C_1) \subseteq R(I - \tilde{a}(\lambda)\mathcal{A}^{-1})$ and (418). As above, this simply implies $R(C_1) \subseteq R(\tilde{a}(\lambda) - \mathcal{A})$ and (416).

Using Lemma 3.2.45 and a similar argumentation, we can prove the following:

PROPOSITION 3.10.3. Suppose that \mathcal{A} is a closed MLO in $X, C \in L(X)$, $C\mathcal{A} \subseteq \mathcal{A}C, |a(t)|$ and k(t) satisfy (P1), as well as that $(R(t))_{t\geq 0} \subseteq L(X)$ is strongly continuous and commutes with C on X. Let $\omega \geq \max(0, \operatorname{abs}(|a|), \operatorname{abs}(k))$ be such that the operator family $\{e^{-\omega t}R(t): t\geq 0\}$ is equicontinuous. Then $(R(t))_{t\geq 0}$ is an (a, k)-regularized C-resolvent family with a subgenerator \mathcal{A}^{-1} iff for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\tilde{a}(\lambda)\tilde{k}(\lambda) \neq 0$, we have $R(C) \subseteq \operatorname{R}(\tilde{a}(\lambda) - \mathcal{A})$, (416) holds with $R_1(\cdot), C_1$ and Y, y replaced with $R(\cdot), C$ and X, x therein, as well as (417) holds with $R_2(\cdot)$ and C_2 replaced with $R(\cdot)$ and C therein.

The complex characterization theorem ensuring the existence of an exponentially equicontinuous degenerate (a, k)-regularized *C*-resolvent families, in combination with Proposition 3.10.3, enables one to formulate the subsequent result, which provides sufficient conditions for the operator \mathcal{A}^{-1} to be a subgenerator of an exponentially equicontinuous, analytic (a, k)-regularized C-resolvent family:

PROPOSITION 3.10.4. Assume that \mathcal{A} is a closed MLO in $X, C\mathcal{A} \subseteq \mathcal{A}C$, $\alpha \in (0, \pi/2]$, $\operatorname{abs}(k) < \infty$, $\operatorname{abs}(|a|) < \infty$ and $\omega \ge \max(0, \operatorname{abs}(k), \operatorname{abs}(|a|))$. Assume, further, that for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\tilde{a}(\lambda) \ne 0$, we have $R(C) \subseteq R(\tilde{a}(\lambda) - \mathcal{A})$ as well as that there exist a function $\Upsilon : \omega + \Sigma_{\frac{\pi}{2} + \alpha} \to L(X)$ and an operator $D' \in L(X)$ such that, for every $x \in X$, the mapping $\lambda \mapsto \Upsilon(\lambda)x$, $\lambda \in \omega + \Sigma_{\frac{\pi}{2} + \alpha}$ is analytic as well as that:

- (i) There exists a function k: Σ_α ∪ {0} → C which is analytic on Σ_α, continuous on any closed subsector Σ_γ (0 < γ < α) and which additionally satisfies that sup_{z∈Σ_γ} |e^{-ωz}k(z)| < ∞ (0 < γ < α) and k(t) = k(t) for all t ≥ 0;
- (ii) $\Upsilon(\lambda)x \in \tilde{a}(\lambda)(\tilde{a}(\lambda) A)^{-1}Cx$ for every $x \in X$ and $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$, $\tilde{a}(\lambda) \neq 0$;
- (iii) $\Upsilon(\lambda)Cx = C\Upsilon(\lambda)x$ for $\operatorname{Re} \lambda > \omega, x \in X$;
- (iv) $\tilde{a}(\lambda)\Upsilon(\lambda)x \Upsilon(\lambda)y = \tilde{a}(\lambda)Cx$, provided $\operatorname{Re} \lambda > \omega$ and $(x, y) \in \mathcal{A}$;
- (v) $\lim_{\lambda \to +\infty} \Upsilon(\lambda) x = D'x, x \in X, \text{ if } \overline{R(\mathcal{A})} \neq X.$

Then the function $\lambda \mapsto \tilde{k}(\lambda)$, $\operatorname{Re} \lambda > \omega$ has an analytic extension $\lambda \mapsto \hat{k}(\lambda)$, $\lambda \in \omega + \Sigma_{(\pi/2)+\alpha}$ satisfying that $\sup_{\omega + \Sigma_{(\pi/2)+\gamma}} |(\lambda - \omega)\hat{k}(\lambda)| < \infty$ for $0 < \gamma < \alpha$ and $\lim_{\lambda \to +\infty} \lambda \hat{k}(\lambda) = k(0)$. If, additionally,

(vi) the family $\{(\lambda - \omega)\hat{k}(\lambda)\Upsilon(\lambda) : \lambda \in \omega + \Sigma_{\frac{\pi}{2} + \gamma}\} \subseteq L(X)$ is equicontinuous for all $\gamma \in (0, \alpha)$,

then \mathcal{A}^{-1} is a subgenerator of an exponentially equicontinuous, analytic (a, k)-regularized C-resolvent family $(R(t))_{t\geq 0}$ of angle α satisfying that $R(z)\mathcal{A}^{-1} \subseteq \mathcal{A}^{-1}R(z)$, $z \in \Sigma_{\alpha}$, the family $\{e^{-\omega z}R(z) : z \in \Sigma_{\gamma}\} \subseteq L(X)$ is equicontinuous for all angles $\gamma \in (0, \alpha)$, as well as that the equation (272) holds for each $y = x \in X$, with \mathcal{A} , $R_1(\cdot)$ and C_1 replaced therein by \mathcal{A}^{-1} , $R(\cdot)$ and C, respectively.

PROOF. It is clear that \mathcal{A}^{-1} is a closed MLO in X and $C\mathcal{A}^{-1} \subseteq \mathcal{A}^{-1}C$. As already seen multiple times before, condition (i) yields that the function $\lambda \mapsto \tilde{k}(\lambda)$, $\operatorname{Re} \lambda > \omega$ has an analytic extension $\lambda \mapsto \hat{k}(\lambda)$, $\lambda \in \omega + \Sigma_{(\pi/2)+\alpha}$ satisfying that $\sup_{\omega + \Sigma_{(\pi/2)+\gamma}} |(\lambda - \omega)\hat{k}(\lambda)| < \infty$ for $0 < \gamma < \alpha$ and $\lim_{\lambda \to +\infty} \lambda \hat{k}(\lambda) = k(0)$. Define $q(\lambda) := \hat{k}(\lambda)C - \hat{k}(\lambda)\Upsilon(\lambda)$, $\lambda \in \omega + \Sigma_{(\pi/2)+\alpha}$ and D := k(0)C - k(0)D'. Then $q(\cdot)$ is analytic and a simple computation involving condition (ii) shows that for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega$ and $\tilde{a}(\lambda) \neq 0$, we have $R(C) \subseteq R(I - \tilde{a}(\lambda)\mathcal{A}^{-1})$ with $Cx \in (I - \tilde{a}(\lambda)\mathcal{A}^{-1})[Cx - \Upsilon(\lambda)x]$, $x \in X$. The remainder of proof follows from an elementary argumentation and corresponding statement for the operator \mathcal{A} . \Box

The most intriguing case in which the assumptions of Proposition 3.10.4 hold is: $\omega = 0, a(t) = g_{\alpha}(t)$ for some number $\alpha \in (0, 2), k(t) = 1, C \in L(X)$ is injective and satisfies that $C^{-1}\mathcal{A}C = \mathcal{A}$ is the integral generator of an equicontinuous analytic (g_{α}, g_1) -regularized C-resolvent family $(S(t))_{t \ge 0}$ of angle $\gamma \in (0, \min(\pi/2, (\pi/\alpha) - (\pi/2))]$. Then we can make a choice in which $\Upsilon(\lambda) = \lambda^{\alpha} (\lambda^{\alpha} - \mathcal{A})^{-1}C$ and D' = S(0), providing thus a proper extension of [383, Theorem 4.1(i)] and [135, Proposition 1]. Since subordination principles can be formulated for (a, k)-regularized C-resolvent families subgenerated by MLOs (C need not be injective, in general), we can simply prove an extension of [383, Theorem 4.1(ii)] for (g_{α}, g_{β}) -regularized C-resolvent families.

In connection with Proposition 3.10.4 and [383, Theorem 4.1], we would like to propose the following:

PROPOSITION 3.10.5. Suppose that \mathcal{A} is a closed MLO, $C \in L(X)$, $C\mathcal{A} \subseteq \mathcal{A}C$, $\beta \ge 0, \alpha \in (0,2), a(t) = g_{\alpha}(t), k(t) = g_{\beta+1}(t) \text{ and } \gamma \in (0, \min(\pi/2, (\pi/\alpha) - (\pi/2))].$ Suppose, further, that $\Sigma_{((\pi/2)+\gamma)\alpha} \subseteq \rho_C(\mathcal{A})$, the mapping $\lambda \mapsto (\lambda - \mathcal{A})^{-1}Cx, \lambda \in \Sigma_{((\pi/2)+\gamma)\alpha}$ is analytic $(x \in X)$ and the following two conditions are satisfied:

- (i) For every $\gamma' \in (0, \gamma)$, there exists a finite constant $M_{\gamma'} > 0$ such that the operator family $\{\lambda^{\alpha+\beta}(\lambda^{\alpha}-\mathcal{A})^{-1}C : \lambda \in \Sigma_{(\pi/2)+\gamma'}, |\lambda| \leq 1\} \subseteq L(X)$ is equicontinuous.
- (ii) If $\overline{R(\mathcal{A})} \neq X$, then there exists $D' \in L(X)$ such that $\lim_{\lambda \to 0+} \lambda(\lambda \mathcal{A})^{-1}Cx = D'x, x \in X$.

Then the operator \mathcal{A}^{-1} is a subgenerator of an exponentially equicontinuous, analytic (a,k)-regularized C-resolvent family $(R(t))_{t\geq 0}$ of angle γ satisfying that $R(z)\mathcal{A}^{-1} \subseteq \mathcal{A}^{-1}R(z), z \in \Sigma_{\gamma}$ and the family $\{e^{-\omega z}R(z) : z \in \Sigma_{\gamma}\} \subseteq L(X)$ is equicontinuous for every real numbers $\omega > 0$ and $\gamma \in (0, \gamma')$. Moreover, the equation (272) holds for each $y = x \in X$, with \mathcal{A} , $R_1(\cdot)$ and C_1 replaced therein by \mathcal{A}^{-1} , $R(\cdot)$ and C, respectively.

PROOF. As above, we have that \mathcal{A}^{-1} is a closed MLO in X and $C\mathcal{A}^{-1} \subseteq \mathcal{A}^{-1}C$. It is clear that the function k(t) satisfies condition (i) from Proposition 3.10.4. If $\omega > 0$, then we can apply Proposition 3.10.4 with the function $\Upsilon : \omega + \Sigma_{\frac{\pi}{2} + \gamma} \to L(X)$ defined by $\Upsilon(\lambda) := \lambda^{-\alpha} (\lambda^{-\alpha} - \mathcal{A})^{-1}C$, $\lambda \in \omega + \Sigma_{\frac{\pi}{2} + \gamma}$. In actual fact, conditions (ii)-(iv) of Proposition 3.10.4 clearly hold; condition (v) of Proposition 3.10.4 holds because of assumption (ii) of this proposition, while condition (vi) of Proposition 3.10.4 follows from condition (i) of this proposition and a simple computation with a new variable $z = 1/\lambda$.

REMARK 3.10.6. It is worth noting that the behaviour of function $\lambda \mapsto (\lambda - \mathcal{A})^{-1}C$, $\lambda \in \Sigma_{((\pi/2)+\gamma)\alpha}$ at the point $\lambda = \infty$ does not play any role for applying Proposition 3.10.5.

It seems that Proposition 3.10.5 is not considered elsewhere, even for the abstract non-degenerate differential equations of first order. We will only present an illustrative application of Proposition 3.10.5 to the abstract degenerate differential equations of fractional order:

EXAMPLE 3.10.7. Let us recall that, for every two linear single-valued operators A and B, we have $(AB^{-1})^{-1} = BA^{-1}$ and $(B^{-1}A)^{-1} = A^{-1}B$ in the MLO sense. Consider now, for the sake of illustration, the fractional Poisson heat equation $(P)_b$ in the space $L^p(\Omega)$, where Ω is a bounded domain in \mathbb{R}^n , 1 , <math>b > 0, $m(x) \ge 0$ a.e. $x \in \Omega$, $m \in L^{\infty}(\Omega)$ and 1 ; then <math>B is the multiplication in $L^p(\Omega)$ with m(x), and $A = \Delta - b$ acts with the Dirichlet boundary conditions (see also [199, Example 3.6]). Let $\mathcal{A} := AB^{-1}$; then for a suitable chosen number b > 0, we have the existence of an angle $\theta \in (\pi/2, \pi)$ and a finite number M > 0such that

(419)
$$\|(\lambda - \mathcal{A})^{-1}\| \leq M |\lambda|^{(-1)/p}, \quad \lambda \in \Sigma_{\theta}.$$

Let $\alpha \in [1, 2\theta/\pi)$. By Proposition 3.10.5, with D' = 0 and $\beta = 0$, we get that \mathcal{A}^{-1} generates an analytic (g_{α}, g_1) -regularized resolvent family $(R(t))_{t \geq 0}$ of angle $\gamma \in (0, \min((\theta/\alpha) - (\pi/2), \pi/2)]$, satisfying that the operator family $\{e^{-\omega z}R(z) : z \in \Sigma_{\gamma'}\} \subseteq L(X)$ is bounded for every $\omega > 0$ and $\gamma' \in (0, \gamma)$. Moreover, let $0 < \varepsilon < \gamma' < \gamma$. Then we have the following integral representation

$$R(z)x = x - \frac{1}{2\pi i} \int_{\Gamma_{\omega}} e^{\lambda z} \frac{(\lambda^{-\alpha} - \mathcal{A})^{-1}x}{\lambda^{\alpha+1}} d\lambda, \quad x \in X, \ z \in \Sigma_{\gamma' - \varepsilon},$$

where the contour Γ is defined in the proof of [27, Theorem 2.6.1] (we only need to replace the number γ with the number γ' therein). Using the estimate (419) and the integral computation contained in the proof of afore-mentioned theorem, after letting $\omega \to 0+$ we get that $(R(t))_{t\geq 0}$ is an equicontinuous analytic (g_{α}, g_1) regularized resolvent family of angle γ . Hence, we can analyze the well-posedness of the reversed fractional Poisson heat equation in the space $L^p(\Omega)$:

$$(P)_{b;r}: \begin{cases} \mathbf{D}_{t}^{\alpha}[(\Delta-b)v(t,x)] = m(x)v(t,x), & t \ge 0, \ x \in \Omega; \\ v(t,x) = 0, & (t,x) \in [0,\infty) \times \partial\Omega, \\ (\Delta-b)v(0,x) = v_{0}(x), \ \left(\frac{d}{dt}[(\Delta-b)v(t,x)]\right)_{t=0} = v_{1}(x), \quad x \in \Omega. \end{cases}$$

For possible applications to abstract degenerate second-order differential equations, we refer the reader to [199, Example 6.1] and [293, Example 3.10.10].

It is worth noticing that the existence and behaviour of *C*-resolvent of a multivalued linear operator \mathcal{A} around zero is most important for the generation of certain classes of (a, k)-regularized *C*-resolvent families, *C*-(ultra)distribution semigroups and *C*-(ultra)distribution cosine functions by the inverse operator \mathcal{A}^{-1} . More to the point, the existence of *C*-resolvent of \mathcal{A} at the point $\lambda = +\infty$ does not play any role for the generation of *C*-(ultra)distribution semigroups and *C*-(ultra)distribution cosine functions by the inverse operator \mathcal{A}^{-1} ; in the following example, we will explain this fact only for *C*-distribution semigroups (a similar statement holds for the generation of locally defined fractional *C*-resolvent families):

EXAMPLE 3.10.8. Suppose that a > 0, b > 0 and recall that $E(a, b) := \{\lambda \in \mathbb{C} \mid \operatorname{Re} \lambda \geq b, |\operatorname{Im} \lambda| \leq e^{a \operatorname{Re} \lambda}\}$. It can be easily seen that the set $1/E(a, b) := \{1/\lambda : \lambda \in E(a, b)\}$ is a relatively compact subset of \mathbb{C} , as well as that 1/E(a, b) is contained in the strip $\{\lambda \in \mathbb{C} : 0 < \operatorname{Re} \lambda < 1/b\}$. Let \mathcal{A} be a closed MLO commuting with the operator $C \in L(X)$, and let there exist $n \in \mathbb{N}$ such that the operator family $\{\lambda^n(\lambda - \mathcal{A})^{-1}C : \lambda \in 1/E(a, b)\} \subseteq L(X)$ is equicontinuous. If we suppose additionally that the mapping $\lambda \mapsto (\lambda - \mathcal{A})^{-1}Cx$ is analytic on $1/\Omega_{a,b}$ and continuous on $1/\Gamma_{a,b}$, with the meaning clear, where $\Gamma_{a,b}$ denotes the upwards oriented boundary of E(a, b) and $\Omega_{a,b}$ the open region which lies to the

right of $\Gamma_{a,b}$, then an extension of the operator \mathcal{A}^{-1} generates a *C*-distribution semigroup \mathcal{G} . Moreover, if *C* is injective and \mathcal{A}^{-1} is single valued, then the operator $C^{-1}\mathcal{A}^{-1}C$ is the integral generator of \mathcal{G} . On the other hand, for the generation of exponential *C*-distribution semigroups by an extension of the operator \mathcal{A}^{-1} one has to assume that the operator family $\{\lambda^n(\lambda - \mathcal{A})^{-1}C : 0 < \operatorname{Re} \lambda < c\} \subseteq L(X)$ is equicontinuous for some real number c > 0 and integer $n \in \mathbb{N}$. Finally, let $\alpha \in (0, 2)$ and $\omega > 0$. Denote by Ω the unbounded region lying between the boundary of sector $\Sigma_{\alpha\pi/2}$ and the curve $\{\lambda^{-\alpha} : \operatorname{Re} \lambda = \omega\}$. If $\Omega \subseteq \rho_C(\mathcal{A})$ and the family $\{\lambda^n(\lambda - \mathcal{A})^{-1}C : \lambda \in \Omega\} \subseteq L(X)$ is equicontinuous for some integer $n \in \mathbb{N}$, then there exists a positive real number $\beta > 0$ such that the operator \mathcal{A} is a subgenerator of a global $(g_{\alpha}, g_{\beta+1})$ -regularized *C*-resolvent operator family $(R(t))_{t\geq 0}$ satisfying that the operator family $\{e^{-\omega t}R(t) : t \geq 0\} \subseteq L(X)$ is equicontinuous.

In the following theorem, we reconsider the statement of [**383**, Theorem 4.1(i)] for subgenerators of degenerate $(g_{\alpha}, g_{\beta+1})$ -regularized *C*-resolvent families, where $\alpha \in (0, 2], \beta \ge 0$ and the operator $C \in L(X)$ is possibly non-injective. More to the point, here the subgenerator \mathcal{A} is not necessarily injective or single-valued (the interested reader may try to extend the statements of [**383**, Theorem 4.1(ii), Corollary 4.1, Corollary 4.2, Corollary 4.3], as well), which will be crucial for applications carried out in Example 3.10.19 below:

THEOREM 3.10.9. Suppose that $\alpha \in (0,2]$, $\beta \ge 0$ and a closed MLO \mathcal{A} is a subgenerator of an exponentially equicontinuous $(g_{\alpha}, g_{\beta+1})$ -regularized C-resolvent family $(S(t))_{t\ge 0}$ such that the operator family $\{t^{-\beta}S(t):t>0\} \subseteq L(X)$ is equicontinuous. Then, for every number $\gamma > \beta + (1/2)$, the operator \mathcal{A}^{-1} is a subgenerator of an $(g_{\alpha}, g_{\gamma+1})$ -regularized C-resolvent family $(R(t))_{t\ge 0}$ satisfying that the operator family $\{t^{-\gamma}R(t):t>0\} \subseteq L(X)$ is equicontinuous.

PROOF. Define

$$R(t)x := g_{\gamma+1}(t)Cx - t^{1+\beta+\gamma} \int_0^\infty J_{1+\beta+\gamma}(2\sqrt{st})s^{-\frac{1+\beta+\gamma}{2}}S(s)x\,ds, \quad t > 0, \ x \in X.$$

Arguing as in the proof of [383, Theorem 4.1(i)], we get that $(R(t))_{t\geq 0} \subseteq L(X)$ is strongly continuous as well as that the operator family $\{t^{-\gamma}R(t): t>0\} \subseteq L(X)$ is equicontinuous and

(420)
$$\int_0^\infty e^{-\lambda t} R(t) x \, dt = \lambda^{-(1+\gamma)} C x - \lambda^{-(2+\beta+\gamma)} \int_0^\infty e^{-s/\lambda} S(s) x \, ds, \quad \operatorname{Re} \lambda > 0, \ x \in X.$$

Further on, \mathcal{A}^{-1} is a closed MLO and, by Lemma 3.2.45, we have

(421)
$$\frac{Cx}{\lambda^{\beta+1}} \in \left(I - \frac{\mathcal{A}}{\lambda^{\alpha}}\right) \int_0^\infty e^{-\lambda t} S(t) x \, dt, \quad \operatorname{Re} \lambda > 0, \ x \in X$$

and

(422)
$$\frac{Cy}{\lambda^{\beta+1}} = \int_0^\infty e^{-\lambda t} S(t) y \, dt - \frac{1}{\lambda^\alpha} \int_0^\infty e^{-\lambda t} S(t) x \, dt,$$

provided $\operatorname{Re} \lambda > 0$ and $(x, y) \in X$. Having in mind Lemma 3.2.45 and (420), it suffices to show that

$$(423) \quad \frac{Cx}{\lambda^{1+\gamma}} \in \lambda^{-(1+\gamma)} Cx - \lambda^{-(2+\beta+\gamma)} \int_0^\infty e^{-s/\lambda} S(s) x \, ds \\ - \lambda^{-\alpha} \mathcal{A}^{-1} \bigg[\lambda^{-(1+\gamma)} Cx - \lambda^{-(2+\beta+\gamma)} \int_0^\infty e^{-s/\lambda} S(s) x \, ds \bigg], \quad \operatorname{Re} \lambda > 0, \ x \in X$$

and

(424)
$$\frac{Cx}{\lambda^{1+\gamma}} = \lambda^{-(1+\gamma)}Cx - \lambda^{-(2+\beta+\gamma)} \int_0^\infty e^{-s/\lambda} S(s)x \, ds \\ -\lambda^{-\alpha} \bigg[\lambda^{-(1+\gamma)}Cx - \lambda^{-(2+\beta+\gamma)} \int_0^\infty e^{-s/\lambda} S(s)y \, ds \bigg],$$

provided $\operatorname{Re} \lambda > 0$ and $(y, x) \in \mathcal{A}$. Keeping in mind the definition of operator \mathcal{A}^{-1} , the equation (423) follows almost immediately from (421), while the equation (424) follows almost immediately from (422), with the number λ replaced therein with the number $1/\lambda$.

- REMARK 3.10.10. (i) Keeping in mind the proof of [**383**, Theorem 4.1(i)], Theorem 3.10.9 and Lemma 3.2.44(ii), the above result can be simply reformulated for the classes of exponentially equicontinuous mild $(g_{\alpha}, g_{\beta+1})$ regularized C_1 -existence families and exponentially equicontinuous mild $(g_{\alpha}, g_{\beta+1})$ -regularized C_2 -uniqueness families. It is also worth noticing that the representation formula obtained in [**383**] with the help of (47) is motivated by earlier results of R. deLaubenfels established in [**134**]. In all these results, the Laplace transform identities for various Bessel type functions play a crucial role.
- (ii) For applications, it will be crucial to reconsider and extend the conclusions obtained in [383, Remark 4.2] for abstract degenerate fractional differential equations. Suppose that the operator family $\{(1+t^{\delta})^{-1}S(t): t > 0\} \subseteq L(X)$ is equicontinuous for some number $\delta \ge 0$ and all remaining assumptions in Theorem 3.10.9 hold. Then, for every non-negative real number $\gamma > 2\delta + (1/2) - \beta$, the operator \mathcal{A}^{-1} is a subgenerator of an $(g_{\alpha}, g_{\gamma+1})$ regularized *C*-resolvent family $(R(t))_{t\ge 0}$. Moreover, a simple calculation shows that the operator family $\{t^{-\gamma}(1+t^{\beta-\delta}+t^{\beta})^{-1}R(t): t>0\} \subseteq L(X)$ is equicontinuous.

It is crucial to formulate the following proper generalization of Theorem 3.10.9 (where $\omega'_0 = 0$, $f(\lambda) = 1/\lambda$, $a(t) = b(t) = g_{\alpha}(t)$, $k(t) = g_{\beta+1}(t)$, $k_1(t) = g_{\gamma+1}(t)$, $g(t) = g_{\gamma-\beta}(t)$ and $S_0(t) = t^{1+\beta+\gamma} \int_0^\infty J_{1+\beta+\gamma}(2\sqrt{st})s^{-\frac{1+\beta+\gamma}{2}}S(s)x\,ds$, t > 0, $x \in X$; see also Remark 3.10.10) for various classes of (a, k)-regularized C-resolvent families:

THEOREM 3.10.11. Suppose that \mathcal{A} is a closed MLO in $X, C, C_2 \in L(X)$, $C_1 \in L(Y, X), C\mathcal{A} \subseteq \mathcal{A}C, \omega'_0 \geq \max(\operatorname{abs}(|a|), \operatorname{abs}(k), 0)$, the functions b(t) and $k_1(t)$ satisfy (P1) with $\omega_0 \geq \max(0, \operatorname{abs}(|b|))$, the function $k_1(t)$ is continuous for $t \geq 0$ and $|k_1(t)| = O(e^{\omega_0 t} P(t))$ for $t \geq 0$, where $P(t) = \sum_{j=0}^l a_j t^{\zeta_j}, t \geq 0$ $(l \in \mathbb{N}, a_j \geq 0 \text{ and } \zeta_j \geq 0 \text{ for } 1 \leq j \leq l)$. Let $f: \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda > \omega_0\} \rightarrow \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda > \omega_0\}$ ω'_0 and $G: \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda > \omega_0\} \to \mathbb{C}$ be two given functions, let $\tilde{a}(\lambda) \neq 0$ for $\operatorname{Re} \lambda > \omega'_0$, and let

(425)
$$\tilde{b}(\lambda) = \frac{1}{\tilde{a}(f(\lambda))}$$
 and $\tilde{k}_1(\lambda) = G(\lambda)\tilde{k}(f(\lambda)), \quad \operatorname{Re} \lambda > \omega_0.$

(i) Suppose, further, A is a subgenerator of a global mild (a,k)-regularized C₁-existence family (R₁(t))_{t≥0} ⊆ L(Y,X) such that the operator family {e^{-ωt}R₁(t) : t ≥ 0} ⊆ L(Y,X) is equicontinuous for each number ω > ω'₀, as well as there exists a strongly continuous operator family (S₀(t))_{t≥0} ⊆ L(Y,X) such that the family {e^{-ω0t}S₀(t) : t ≥ 0} ⊆ L(Y,X) is equicontinuous and

(426)
$$\int_0^\infty e^{-\lambda t} S_0(t) y \, dt = G(\lambda) \int_0^\infty e^{-sf(\lambda)} R_1(s) y \, ds, \quad y \in Y, \text{ Re } \lambda > \omega_0.$$

Then the operator \mathcal{A}^{-1} is a subgenerator of a global mild (b, k_1) -regularized C_1 -existence family $(S_1(t) \equiv k_1(t)C_1 - S_0(t))_{t \ge 0} \subseteq L(Y, X)$ and the operator family $\{e^{-\omega_0 t}(P(t))^{-1}S_1(t) : t > 0\} \subseteq L(Y, X)$ is equicontinuous.

- (ii) Suppose, further, A is a subgenerator of a global mild (a, k)-regularized C₂-uniqueness family (R₂(t))_{t≥0} ⊆ L(X) such that the operator family {e^{-ωt}R₂(t) : t ≥ 0} ⊆ L(X) is equicontinuous for each number ω > ω'₀, as well as there exists a strongly continuous operator family (S₀(t))_{t≥0} ⊆ L(X) such that the family {e^{-ω₀t}S₀(t) : t ≥ 0} ⊆ L(X) is equicontinuous and (426) holds with y = x ∈ X and R₁(·) replaced by R₂(·) therein. Then the operator A⁻¹ is a subgenerator of a global mild (b, k₁)-regularized C₂-uniqueness family (S₂(t) ≡ k₁(t)C₂ S₀(t))_{t≥0} ⊆ L(X) and the operator family {e^{-ω₀t}(P(t))⁻¹S₂(t) : t > 0} ⊆ L(X) is equicontinuous.
- (iii) Suppose, further, A is a subgenerator of a global (a, k)-regularized C-resolvent family (R(t))_{t≥0} ⊆ L(X) such that the operator family {e^{-ωt}R(t): t≥0} ⊆ L(X) is equicontinuous for each number ω > ω'₀, as well as there exists a strongly continuous operator family (S₀(t))_{t≥0} ⊆ L(X) such that the family {e^{-ω₀t}S₀(t) : t≥0} ⊆ L(X) is equicontinuous and (426) holds with y = x ∈ X and R₁(·) replaced by R(·) therein. Then the operator A⁻¹ is a subgenerator of a global mild (b, k₁)-regularized C-resolvent family (S(t) ≡ k₁(t)C S₀(t))_{t≥0} ⊆ L(X) and the operator family {e^{-ω₀t}(P(t))⁻¹S(t) : t > 0} ⊆ L(X) is equicontinuous.

PROOF. We will prove only (i). Let $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > \omega_0$ and $\tilde{b}(\lambda)\tilde{k_1}(\lambda) \neq 0$ be given. Due to Proposition 3.10.2(ii), it suffices to show that $R(C_1) \subseteq R(\tilde{b}(\lambda) - \mathcal{A})$ and

(427)
$$\tilde{b}(\lambda)\tilde{k}_1(\lambda)C_1y \in (\tilde{b}(\lambda) - \mathcal{A}) \int_0^\infty e^{-\lambda t} [k_1(t)C_1y - S_1(t)y] dt, \quad y \in Y.$$

But, our assumption (425) implies $\tilde{a}(f(\lambda))\tilde{k}(f(\lambda)) \neq 0$. Since $\operatorname{Re}(f(\lambda)) > \omega'_0$, Lemma 3.2.44(ii) yields that $R(C_1) \subseteq R([\tilde{a}(f(\lambda))]^{-1} - \mathcal{A})$ and

$$\widetilde{R_1}(f(\lambda))y \in \frac{k(f(\lambda))}{\tilde{a}(f(\lambda))}([\tilde{a}(f(\lambda))]^{-1} - \mathcal{A})^{-1}C_1y, \quad y \in Y.$$

This simply gives (427) after a simple calculation involving the equations (425) and (426). $\hfill \Box$

REMARK 3.10.12. In the existing literature concerning the inverse generator problem, the authors have investigated only the following special case: $\omega'_0 = 0$, $f(\lambda) = 1/\lambda$ and $a(t) = b(t) = g_\alpha(t)$, for some number $\alpha \in (0, 2)$. Even if $f(\lambda) = 1/\lambda$, the equality $a(t) = b(t) = g_\alpha(t)$, where $\alpha \in (0, 2)$, is not necessary for applying Theorem 3.10.11. For example, suppose that $P(\lambda) = \sum_{j=0}^{n} a_j \lambda^{\zeta_j}$ and $Q(\lambda) = \sum_{j=0}^{m} b_j \lambda^{\eta_j}$ for some non-negative real numbers ζ_j $(0 = \zeta_0 \leq \zeta_1 \leq \ldots \leq \zeta_n)$, η_j $(0 = \eta_0 \leq \eta_1 \leq \ldots \leq \eta_m)$ and complex numbers a_j $(0 \leq j \leq n)$, b_j $(0 \leq j \leq m)$ such that $b_0 = 0$, $a_0 a_n b_m \neq 0$, $\eta_m > \zeta_n$ and $P(\lambda)Q(\lambda) \neq 0$ for Re $\lambda > 0$ [we can take, for example, $P(\lambda) = \lambda + 2$ and $Q(\lambda) = \lambda^3$]. If $a(t) = \mathcal{L}^{-1}(P(\lambda)/Q(\lambda))(t)$, $t \geq 0$, then we can prove that abs(|a|) = 0. Furthermore, we can prove that there exists a function b(t) satisfying abs(|b|) = 0 and $\tilde{b}(\lambda) = 1/\tilde{a}(1/\lambda) = Q(1/\lambda)/P(1/\lambda)$, Re $\lambda > 0$. So, if the second equality in (425) holds with the functions k(t) and $k_1(t)$ being continuous for $t \geq 0$, then the most simplest case in which Theorem 3.10.11(iii) is applicable is that case in which $X := \mathbb{C}$, C := I, $\mathcal{A} := 0$ and R(t) := k(t)I, $t \geq 0$, when $S_0(t) = k_1(t)I$, $t \geq 0$.

REMARK 3.10.13. In a great deal of concrete situations, it is almost impossible to represent $S_0(\cdot)$ in terms of $R_1(\cdot)$ directly, so that the use of complex characterization theorem for the Laplace transform is sometimes unavoidable.

We continue by stating the following corollary of Theorem 3.10.11 (the case in which $\sigma = -1$ and $a = b = \beta \ge 0$ has been already considered in Theorem 3.10.9 and remarks following it):

THEOREM 3.10.14. Suppose that $\alpha \in (0,2)$, $\sigma \in (-1,0)$, $\beta \ge 0$, \mathcal{A} is a closed MLO in $X, C \in L(X), C\mathcal{A} \subseteq \mathcal{A}C, \mathcal{A}$ is a subgenerator of an exponentially equicontinuous $(g_{\alpha}, g_{\beta+1})$ -regularized C_1 -resolvent family $(R(t))_{t\ge 0}$ such that the operator family $\{(t^a + t^b)^{-1}R(t) : t > 0\} \subseteq L(X)$ is equicontinuous for two real numbers a, b such that $-1 < a \le b$. Let $\eta > 1 + b$ and $\eta \ge 1 + \beta$. Define

$$F(t) := t^{|\sigma|(\eta-\beta-1)} + (t^{|\sigma|(\eta-b-1)}\chi_{(0,1]}(t) + t^{|\sigma|(\eta-a-1)}\chi_{[1,\infty)}(t)), \quad t > 0.$$

Then the operator \mathcal{A}^{-1} is a subgenerator of a global $(g_{\alpha|\sigma|}, g_{1+|\sigma|(\eta-\beta-1)})$ -regularized *C*-resolvent family $(S(t))_{t\geq 0} \subseteq L(X)$ and the operator family $\{[F(t)]^{-1}S(t) : t > 0\} \subseteq L(X)$ is equicontinuous.

PROOF. We will apply Theorem 3.10.11(iii) with $\omega_0 = \omega'_0 = 0$, $f(\lambda) = \lambda^{\sigma}$, $a(t) = g_{\alpha}(t)$, $b(t) = g_{|\sigma|\alpha}(t)$, $k(t) = g_{\beta+1}(t)$, $k_1(t) = g_{|\sigma|(\eta-\beta-1)\beta+1}(t)$ and $G(\lambda) = \lambda^{-1+\sigma\eta}$. Set

$$S_0(t)x := \int_0^\infty t^{-\sigma\eta} \phi(-\sigma, 1 - \sigma\eta; -st^{-\sigma}) R(s) x \, ds, \quad t \ge 0, \ x \in X$$

Using (46), the assumption $-1 < \sigma < 0$ and the dominated convergence theorem, it readily follows that the mapping $t \mapsto S_0(t)x$, t > 0 is continuous for every fixed element $x \in X$. Since $S_0(0) = 0$ and $[F(\cdot) - \cdot^{|\sigma|(\eta - \beta - 1)}](0) = 0$, it suffices to show that the operator family $\{[F(t)]^{-1}S(t) : t > 0\} \subseteq L(X)$ is equicontinuous as well as that

(428)
$$\int_0^\infty e^{-\lambda t} S_0(t) x \, dt = \lambda^{-1+\sigma\eta} \int_0^\infty e^{-sf(\lambda)} R(s) x \, ds, \quad x \in X, \text{ Re } \lambda > 0.$$

The asymptotic formula (46) and the fact that the operator family $\{(t^a+t^b)^{-1}R(t): t > 0\} \subseteq L(X)$ is equicontinuous together imply that for each seminorm $p \in \circledast$ there exist a finite real constant m > 0 and a seminorm $q \in \circledast$ such that

$$p(S_0(t)x) \leqslant mq(x)t^{-\sigma\eta} \int_0^\infty \exp(-m(st^{-\sigma})^{1/(1+\sigma)})(s^a + s^b)ds$$

= $mq(x)t^{\sigma(1-\eta)} \int_0^\infty \exp(-mr^{1/(1+\sigma)})(t^{\sigma a}r^a + t^{\sigma b}r^b)dr$.

Observe that the above integral converges due to our assumption $-1 < a \leq b$, which also implies $\sigma a \geq \sigma b$ and the equicontinuity of operator family $\{[F(t)]^{-1}S(t) : t > 0\}$. Moreover, by the equation (45) and the Fubini theorem, we have

$$\int_0^\infty e^{-\lambda t} S_0(t) x \, dt = \int_0^\infty R(s) x \cdot \int_0^\infty e^{-\lambda t} t^{-\sigma\eta} \phi(-\sigma, 1 - \sigma\eta; -st^{-\sigma}) dt \, ds$$
$$= \lambda^{-1+\sigma\eta} \int_0^\infty e^{-sf(\lambda)} R(s) x \, ds, \quad x \in X, \text{ Re } \lambda > 0.$$

This implies (428) and completes the proof of theorem.

As in Remark 3.10.10, it is worth noting that we can formulate the above result for the classes of exponentially equicontinuous mild $(g_{\alpha}, g_{\beta+1})$ -regularized C_1 -existence families and exponentially equicontinuous mild $(g_{\alpha}, g_{\beta+1})$ -regularized C_2 -uniqueness families (concerning the inverse generator problem, it is our duty to say that we have not been able to find certain applications with these classes of solution operator families). It is also worth noting the following:

REMARK 3.10.15. Let the requirements of Theorem 3.10.14 hold with a = 0. Then the subordination principle for degenerate (a, k)-regularized *C*-resolvent families shows that the operator \mathcal{A} is a subgenerator of a global $(g_{\alpha|\sigma|}, g_{1+|\sigma|\beta})$ -regularized *C*-resolvent family $(W(t))_{t\geq 0}$ satisfying that the operator family $\{(1 + t^{b|\sigma|})^{-1}W(t) : t \geq 0\} \subseteq L(X)$ is equicontinuous. Arguing as in [383, Remark 4.2], with the function $f(\lambda) = 1/\lambda$, we get that the operator \mathcal{A}^{-1} is a subgenerator of a global $(g_{\alpha|\sigma|}, g_{1+\gamma})$ -regularized *C*-resolvent family $(W(t))_{t\geq 0}$ provided $\gamma \geq 0$ and $\gamma > 2b|\sigma| + (1/2) - \beta|\sigma|$. The integration rate obtained here with the function $f(\lambda) = \lambda^{\sigma}$ is better provided that $|\sigma|(\eta - 2b - 1) < 1/2$.

Before we move ourselves to Subsection 3.10.1, we will analyze the Poisson wave type equation in the space $L^2(\Omega)$, where $\emptyset \neq \Omega \subseteq \mathbb{R}^n$ is an open bounded domain with smooth boundary (see [199, Example 2.3]):

EXAMPLE 3.10.16. Let $X := H_0^1(\Omega) \times L^2(\Omega)$ and $m \in L^{\infty}(\Omega)$. Consider the bounded linear operator

$$M := \begin{pmatrix} 1 & 0\\ 0 & m(x) \end{pmatrix},$$

 \square

in X, and an unbounded linear operator

$$L := \begin{pmatrix} 0 & 1 \\ \Delta & 0 \end{pmatrix},$$

in X, with domain $D(L) := [H^2(\Omega) \cap H_0^1(\Omega)] \times H_0^1(\Omega)$. Then we know that the MLO $\mathcal{A} := M^{-1}LM^{-1} - I$ satisfies $(0, \infty) \subseteq \rho(\mathcal{A})$ and $||R(\lambda : \mathcal{A})|| \leq 1/\lambda, \lambda > 0$. An application of Theorem 3.2.12 gives that for each number r > 0 the MLO $\mathcal{A} - I$ generates a global (g_1, g_{1+r}) -resolvent family $(S_r(t))_{t \ge 0}$ such that $||S_r(t)|| = O(t^r), t \ge 0$. Suppose $\sigma \in (-1, 0), r > 0$ and $\eta > 1 + r$. Then, due to Theorem 3.10.14, the operator $(\mathcal{A} - I)^{-1}$ is the integral generator of an $(g_{|\sigma|}, g_{1+|\sigma|(\eta-r-1)})$ -resolvent family $(R_r(t))_{t \ge 0}$ such that $||R_r(t)|| = O(t^{|\sigma|(\eta-r-1)}), t \ge 0$. Suppose that $(u_0 v_0)^T \in X, (u_1 v_1)^T \in R(\mathcal{A} - I), (f_1(\cdot) f_2(\cdot))^T \in C([0, \infty) : X),$

(429)
$$(g_{|\sigma|} * f_1)(t) + u_0 = g_{1+|\sigma|(\eta-r-1)}(t)u_1 \text{ and} (g_{|\sigma|} * f_2)(t) + v_0 = g_{1+|\sigma|(\eta-r-1)}(t)v_1,$$

for any $t \ge 0$. Since $D_t^{|\sigma|}(u(t) v(t))^T \in (\mathcal{A} - I)^{-1}(u(t) v(t))^T + (f_1(t) f_2(t))^T$, $t \ge 0$ is equivalent with $(u(t) v(t))^T \in (\mathcal{A} - I)[D_t^{|\sigma|}(u(t) v(t))^T - (f_1(t) f_2(t))^T]$, $t \ge 0$, after a simple computation involving the condition (429), we get that the function $t \mapsto (u(t) v(t))^T \equiv R_r(t)(u_1 v_1)^T$, $t \ge 0$ is a unique strong solution of the following system:

$$\begin{split} m(x)[u(t,x) + D_t^{|\sigma|}u(t,x) - f_1(t,x)] &= D_t^{|\sigma|}v(t,x) - f_2(t,x),\\ m(x)[v(t,x) + D_t^{|\sigma|}v(t,x) - f_2(t,x)] &= \Delta[D_t^{|\sigma|}u(t,x) - f_1(t,x)];\\ u(0,x) &= u_0(x), \ v(0,x) = v_0(x), \quad x \in \Omega. \end{split}$$

3.10.1. Applications to degenerate time-fractional equations with abstract differential operators. Assume that $n \in \mathbb{N}$ and $iA_j, 1 \leq j \leq n$ are commuting generators of bounded C_0 -groups on a Banach space X. Assume, further, that $P_1(x)$ and $P_2(x)$ are non-zero complex polynomials in n variables, and $0 < \alpha < 2$; set $N_1 := dg(P_1(x)), N_2 := dg(P_2(x))$ and $m := \lceil \alpha \rceil$. In Section 2.2.3, we have analyzed the generation of some specific classes of (g_α, C) -regularized resolvent families associated with the following fractional degenerate abstract Cauchy problem

$$(DFP): \begin{cases} \mathbf{D}_t^{\alpha} \overline{P_2(A)} u(t) = \overline{P_1(A)} u(t) + f(t), & t \ge 0, \\ u(0) = Cx; & u^{(j)}(0) = 0, & 1 \le j \le \lceil \alpha \rceil - 1, \end{cases}$$

provided that $P_2(x) \neq 0, x \in \mathbb{R}^n$ and there exists a non-negative real number $\omega \geq 0$ such that (70) holds, where $0^{1/\alpha} := 0$. Let us recall that our assumptions imply that the operator $\overline{P_2(A)}$ is injective; here we want to note that the operator $\overline{P_2(A)}$ need not be invertible, in general (see e.g. [27, Remark 8.3.5]).

In the remainder of this subsection, we will focus our attention on the case $\omega = 0$. If $0 \notin P_1(\mathbb{R}^n)$ and $\omega = 0$, then we have $\sup_{x \in \mathbb{R}^n} \operatorname{Re}((P_2(x)/P_1(x))^{1/\alpha}) \leq 0$, so that the well-posedness of the reverse fractional degenerate abstract Cauchy

problem

$$(\mathrm{DFP})_r : \begin{cases} \mathbf{D}_t^{\alpha} \overline{P_1(A)} u(t) = \overline{P_2(A)} u(t) + f(t), & t \ge 0, \\ u(0) = Cx; & u^{(j)}(0) = 0, & 1 \le j \le \lceil \alpha \rceil - 1 \end{cases}$$

can be analyzed as in Section 2.2.3 [306]. But, the real problems occur if $0 \in P_1(\mathbb{R}^n)$, when the methods established in [306] are inapplicable. Our main result concerning the well-posedness of problem $(DFP)_r$ is stated as follows:

THEOREM 3.10.17. Suppose $0 < \alpha < 2$, $\sigma \in (-1,0)$, $P_1(x)$ and $P_2(x)$ are non-zero complex polynomials, $N_1 = dg(P_1(x))$, $N_2 = dg(P_2(x))$, $N \in \mathbb{N}$ and $r \in (0, N]$. Let Q(x) be an r-coercive complex polynomial of degree N, $a \in \mathbb{C} \setminus Q(\mathbb{R}^n)$, $\gamma > \frac{n \max(N, \frac{N_1+N_2}{\min(1,\alpha)})}{2r}$ (resp. $\gamma = \frac{n}{r} |\frac{1}{p} - \frac{1}{2}| \max(N, \frac{N_1+N_2}{\min(1,\alpha)})$, if $E = L^p(\mathbb{R}^n)$ for some $1), <math>P_2(x) \neq 0$, $x \in \mathbb{R}^n$ and (70) holds with $\omega = 0$. Set

$$\mathcal{B} \equiv \overline{\overline{P_2(A)} \cdot \overline{P_1(A)}^{-1}},$$

 $C := ((a - Q(x))^{-\gamma})(A), \ \delta := \max(1, \alpha)n/2, \ if \ E \neq L^p(\mathbb{R}^n) \ for \ all \ p \in (1, \infty),$ and $\delta := \max(1, \alpha)n|(1/p) - (1/2)|, \ if \ E = L^p(\mathbb{R}^n) \ for \ some \ p \in (1, \infty).$ Then $C \in L(X)$ is injective and the following holds:

- (i) For each positive real number $\gamma > 2\delta + (1/2)$ the multivalued linear operator \mathcal{B} is a subgenerator of a global exponentially bounded $(g_{\alpha}, g_{\gamma+1})$ regularized C-resolvent family $(R_{\alpha}(t))_{t\geq 0}$ satisfying that the operator family $\{[t^{\gamma}(1 + t^{\beta-\delta} + t^{\beta})]^{-1}R_{\alpha}(t) : t > 0\} \subseteq L(X)$ is equicontinuous.
- (ii) For each positive real number $\eta > 1 + \delta$, the multivalued linear operator \mathcal{B} is a subgenerator of a global exponentially bounded $(g_{\alpha|\sigma|}, g_{|\sigma|(\eta-1)})$ regularized C-resolvent family $(R_{\alpha}(t))_{t\geq 0}$ satisfying that the operator family $\{[F(t)]^{-1}R_{\alpha}(t): t>0\} \subseteq L(X)$ is equicontinuous, where

$$F(t) = t^{|\sigma|(\eta-1)} + (t^{|\sigma|(\eta-\delta-1)}\chi_{(0,1]}(t) + t^{|\sigma|(\eta-1)}\chi_{[1,\infty)}(t)), \quad t > 0.$$

PROOF. We will prove only (i). Set

$$S_{\alpha}(t) := \left(E_{\alpha} \left(t^{\alpha} \frac{P_1(x)}{P_2(x)} \right) (a - Q(x))^{-\gamma} \right) (A), \quad t \ge 0.$$

Then we have that the operator $C \in L(X)$ is injective as well as

$$\lambda^{\alpha-1}(\lambda^{\alpha} - \overline{P_1(A)} \cdot \overline{P_2(A)}^{-1})^{-1}Cx = \int_0^\infty e^{-\lambda t} S_\alpha(t) x \, dt, \quad \operatorname{Re} \lambda > 0, \ x \in X.$$

On the other hand, for any two closed linear operators A and B in X the following holds:

$$\overline{AB^{-1}}^{-1} = \overline{BA^{-1}}$$

Using (430) and Theorem 1.2.4(i), the above implies that, for every $\lambda \in \mathbb{C}$ with $\operatorname{Re} \lambda > 0$, we have $R(C) \subseteq R(I - \lambda^{-\alpha} \mathcal{B}^{-1})$, (304) holds with $R_1(\cdot), C_1, \mathcal{A}$ and Y, y replaced with $R(\cdot), C, \mathcal{B}^{-1}$ and X, x therein, as well as (305) holds with $R_2(\cdot), \mathcal{A}$ and C_2 replaced with $S_{\alpha}(\cdot), \mathcal{B}^{-1}$ and C therein. Due to Lemma 3.2.45, we get that

 $(S_{\alpha}(t))_{t \ge 0} \subseteq L(X)$ is a global exponentially bounded (g_{α}, C) -regularized resolvent family with a subgenerator \mathcal{B}^{-1} . Moreover, we have

(431)
$$||S_{\alpha}(t)|| \leq M(1 + t^{\max(1,\alpha)n/2}), \quad t \ge 0, \text{resp.}, \\ ||S_{\alpha}(t)|| \leq M(1 + t^{\max(1,\alpha)n|\frac{1}{p} - \frac{1}{2}|}), \quad t \ge 0.$$

Using (431), Theorem 3.10.9 and Remark 3.10.10(iii), the required assertion easily follows.

- REMARK 3.10.18. (i) As before, we can choose the regularizing operator C in a slightly different manner and refine the obtained conclusions by imposing some additional conditions on behaviour of the rational function $P_1(x)/P_2(x)$.
- (ii) The operator $\overline{P_2(A)} \cdot \overline{P_1(A)}^{-1}$ is closed provided that at least one of the operators $\overline{P_1(A)}$ or $\overline{P_2(A)}$ is invertible, when we have $\mathcal{B} = \overline{P_2(A)} \cdot \overline{P_1(A)}^{-1}$.
- (iii) Consider the statement (i). The foregoing arguments also imply

$$\lambda^{\alpha-\gamma-1}\overline{P_1(A)}\left(\lambda^{\alpha}\overline{P_1(A)} - \overline{P_2(A)}\right)^{-1}Cx = \int_0^\infty e^{-\lambda t}R_\alpha(t)x\,dt, \quad \operatorname{Re}\lambda > 0, \ x \in X,$$

so that for each positive real number $\gamma > 2\delta + (1/2)$ we have that $(R_{\alpha}(t))_{t \ge 0}$ is an exponentially bounded $(g_{\alpha}, g_{\gamma+1})$ -regularized *C*-resolvent family for the abstract degenerate Cauchy problem

$$\overline{P_1(A)}u(t) = f(t) + \int_0^t a(t-s)\overline{P_2(A)}u(s)ds, \quad t \ge 0.$$

Due to Theorem 2.2.8, we have that for each $x \in D(\overline{P_1(A)}) \cap D(\overline{P_2(A)})$ the function $u(t) = R_{\alpha}(t)x, t \ge 0$ is a unique strong solution of (52) with $f(t) = g_{1+\gamma}(t)C\overline{P_1(A)}x, t \ge 0$. Moreover, we know that $(\overline{P_2(A)}^{-1}S_{\alpha}(t))_{t\ge 0}$ is an exponentially equicontinuous (g_{α}, C) -regularized resolvent family generated by $\overline{P_2(A)}, \overline{P_1(A)}$. Unfortunately, the operator $\overline{P_1(A)}$ need not be injective and the above fact cannot be used for the construction of $(g_{\alpha}, g_{1+\gamma})$ -regularized *C*-resolvent family generated by $\overline{P_1(A)}, \overline{P_2(A)}$ so that it is not clear how we can consider the well-posedness of problem

$$(\mathrm{DFP})_{L,r}: \begin{cases} \overline{P_1(A)}\mathbf{D}_t^{\alpha}u(t) = \overline{P_2(A)}u(t) + f(t), & t \ge 0, \\ u(0) = Cx; & u^{(j)}(0) = 0, & 1 \le j \le \lceil \alpha \rceil - 1 \end{cases}$$

in general. Similar conclusions can be formulated for the statement (ii).

It is worth noting that Theorem 3.10.17 can be simply reformulated in E_l -type spaces. In the following application of Theorem 3.10.17, we consider the case in which $P_2(x) \equiv 1$:

EXAMPLE 3.10.19. (i) Suppose $0 < \alpha < 2$. Let E be one of the spaces $L^p(\mathbb{R}^n)$ $(1 \leq p \leq \infty)$, $C_0(\mathbb{R}^n)$, $C_b(\mathbb{R}^n)$, $BUC(\mathbb{R}^n)$ and let $0 \leq l \leq n$. Let the operator $\mathbf{T}_1\langle\cdot\rangle$, the Fréchet space $X := E_l$, the set \mathbb{N}_0^l and the seminorms $q_\eta(\cdot)$ (\mathbb{N}_0^l) possess the same meaning as before, let $a_\eta \in \mathbb{C}$, $0 \leq l \leq n$. $|\eta| \leq N$ and let $P(D)f := \sum_{|\eta| \leq N} a_{\eta} D^{\eta} f$, with its maximal distributional domain. Suppose that

$$\sup_{x \in \mathbb{R}^n} \operatorname{Re}(P(x)^{1/\alpha}) \leqslant 0.$$

Define $S_{\alpha}(t) =: \mathbf{T}_{\mathbf{l}} \langle E_{\alpha}(t^{\alpha} P(x))(a - P(x))^{-\gamma} \rangle, t \ge 0$. Then $C := S_{\alpha}(0) \in L(X)$ is injective, $(S_{\alpha}(t))_{t \ge 0}$ is an exponentially equicontinuous (g_{α}, C) -regularized resolvent family with the integral generator P(D), and

$$q_{\eta}(S_{\alpha}(t)f) \leqslant M(1+t^{\max(1,\alpha)n/2})q_{\eta}(f), \quad t \ge 0, \ f \in E_l, \ \eta \in \mathbb{N}_0^l, \ \text{resp.},$$

(432)
$$q_{\eta}(S_{\alpha}(t)f) \leqslant M(1 + t^{\max(1,\alpha)n|\frac{1}{p} - \frac{1}{2}|})q_{\eta}(f), \quad t \ge 0, \ f \in E_l, \ \eta \in \mathbb{N}_0^l,$$

with M being independent of $f \in X$ and $\eta \in \mathbb{N}_0^l$; see also [292, Remark 2.5.5]. Set $\delta := \max(1, \alpha)n/2$, if $E \neq L^p(\mathbb{R}^n)$ for all $p \in (1, \infty)$, and $\delta := \max(1, \alpha)n|(1/p) - (1/2)|$, if $E = L^p(\mathbb{R}^n)$ for some $p \in (1, \infty)$. By Theorem 3.10.9 and Remark 3.10.10(ii), we have that for each positive real number $\gamma > 2\delta + (1/2)$, the multivalued linear operator $P(D)^{-1}$ is a subgenerator of an exponentially equicontinuous $(g_\alpha, g_{\gamma+1})$ -regularized C-resolvent family $(R_\alpha(t))_{t\geq 0}$; moreover, since $C^{-1}P(D)C = P(D)$, we have $C^{-1}P(D)^{-1}C = P(D)^{-1}$ and $P(D)^{-1}$ is therefore the integral generator of $(R_\alpha(t))_{t\geq 0}$. In our concrete situation, we can employ (432) in order to see that

$$q_{\eta}(R_{\alpha}(t)f) \leqslant t^{\gamma}(1+t^{\beta-\delta}+t^{\beta})q_{\eta}(f), \quad t \ge 0, \ f \in E_{l}, \ \eta \in \mathbb{N}_{0}^{l}.$$

The established conclusions can be applied in many different directions and here we will present only an application of Proposition 3.2.15(ii): Suppose $m \in \mathbb{N} \setminus \{1\}, x_0 = x, f_0(\cdot) = f(\cdot), P(D)x_j = x_{j-1} \text{ for } 1 \leq j \leq m,$ $P(D)f_j(t) = f_{j-1}(t) \text{ for } t \geq 0 \text{ and } 1 \leq j \leq m, f_j \in C([0,\infty) : X) \text{ for}$ $0 \leq j \leq m, \text{ and } \alpha = 1/m.$ Then the function $v(t) := R_\alpha(t)x + (R_\alpha * C^{-1}f)(t)x, t \geq 0$ is a unique solution of the following abstract timefractional equation:

$$\begin{aligned} V &\in C^{1}((0,\infty):X) \cap C([0,\infty):X), \\ v(t) &= P(D) \Big[v_{t}(t,x) - \sum_{j=1}^{m-1} g_{(j/m)+r}(t) C x_{j} \\ &- \sum_{j=0}^{m-1} (g_{(j/m)+\gamma} * f_{j})(t) - g_{\gamma}(t) C x \Big], \quad t > 0, \\ v(0) &= 0. \end{aligned}$$

Furthermore, $v \in C^1([0,\infty): X)$ provided that $\gamma \ge 1$ or x = 0.

(ii) It is worth noting that the operator P(D) need not be injective. For example, if l = 0, n = 1 and $P(x) = -x^2 + ix$, $x \in \mathbb{R}$, then the state space is $X = BUC(\mathbb{R})$ but the operator $P(D) = \cdot'' + \cdot'$ is not injective since P(D)f = 0 for all constant functions $f(\cdot)$; in particular, the operator $P(D)^{-1}$ is not single-valued and $P(D)^{-1} \notin L(X)$. On the other hand, the injectiveness of operator P(D) holds in many concrete situations; for example, we have already shown that the Laplacian Δ in $L^p(\mathbb{R}^n)$ is injective as well as that the assumption $0 \notin P(\mathbb{R}^n)$ implies that the operator

P(D) is injective. The Korteweg–De Vries operator $P(D) \cdot = \cdot'' + \cdot'$ in $L^p(\mathbb{R})$ is injective $(1 \leq p < \infty)$, as well.

We close the section with the following example:

EXAMPLE 3.10.20. Suppose that $P_1(x) = -|x|^2$ and $P_2(x) = \sum_{|\eta| \leq Q} a_{\eta} x^{\eta}$ $(x \in \mathbb{R}^n), 0 \notin P_2(\mathbb{R}^n)$ and (70) holds with $\omega = 0$. By Theorem 3.10.17, there exist a non-negative real number $\gamma \geq 0$ and an injective operator $C \in L(X)$ such that the operator \mathcal{B} is a subgenerator of a global polynomially bounded $(g_{\alpha}, g_{1+\gamma})$ -regularized C-resolvent family $(R_{\alpha}(t))_{t\geq 0}$ (cf. also the conclusions from Remark 3.10.18(ii)–(iii)), so that we can analyze the existence and uniqueness of strong (mild) solutions of the following fractional degenerate Cauchy problem of order $\alpha \in (0, 2)$:

$$(PR)_{\alpha}: \begin{cases} \mathbf{D}_t^{\alpha} u_{xx}(t,x) = \sum_{|\eta| \leqslant Q} a_{\eta} D^{\eta} u(t,x) + f(t,x), & t \ge 0, \ x \in \mathbb{R}^n, \\ u(0,x) = C\phi(x); \ u_t(0,x) = C\psi(x) \text{ if } \alpha > 1. \end{cases}$$

Without going into full details, we will only note that in the case $\alpha \in (1, 2)$, the validity of certain conditions on the initial values $\phi(x)$, $\psi(x)$ and the inhomogenity f(t, x) yield that the unique strong solution of $(PR)_{\alpha}$ is given by

$$u(t,x) := \mathbf{D}_t^{1+\gamma} \bigg[R_\alpha(t)\phi(x) + \int_0^t R_\alpha(s)\psi(x)ds + (g_{\alpha-1}*R_\alpha*f)(t,x) \bigg], \ t \ge 0, \ x \in \mathbb{R}^n.$$

For more details, see [387, Theorem 13] and [388, Lemma 4.1].

3.11. Quasi-asymptotically almost periodic functions and applications

Throughout this section, we assume that $(X, \|\cdot\|)$ and $(Y, \|\cdot\|_Y)$ are two complex Banach spaces. By $\|T\|_{L(X,Y)}$ we denote the norm of a continuous linear mapping $T \in L(X,Y)$. As explained in [336], the notion of quasi-asymptotical almost periodicity is very important in the study of qualitative properties of the infinite convolution product

(433)
$$\mathbf{F}(t) := \int_{-\infty}^{t} R(t-s)f(s)ds, \quad t \in \mathbb{R},$$

where $f : \mathbb{R} \to X$ is a Weyl-*p*-almost periodic function satisfying certain extra conditions and $(R(t))_{t>0} \subseteq L(X,Y)$ is a strongly continuous operator family having a certain growth order at zero and infinity.

The class of q-aap. functions is maybe unique in the existing literature with regards to its invariance under the action of infinite convolution product (433), for the functions defined on \mathbb{R} , and its invariance under the action of finite convolution product

(434)
$$F(t) := \int_0^t R(t-s)f(s)ds, \quad t \ge 0,$$

for the functions defined on $[0,\infty)$; all that we need is the uniform integrability of solution operator family $(R(t))_{t>0}$, i.e., the condition $\int_0^\infty ||R(s)||_{L(X,Y)} ds < \infty$ (Proposition 3.11.33, Proposition 3.11.34). Similar statements hold for Stepanov classes of q-aap. functions, where we use a slightly stronger condition

$$\sum_{k=0}^{\infty} \|R(\cdot)\|_{L^q[k,k+1]} < \infty$$

(1/p+1/q = 1; see Proposition 3.11.35 and Proposition 3.11.36). It is clear that the results of this section are susceptible to applications to a wide class of inhomogenous abstract Volterra integro-differential equations and inclusions; basically, application is possible at any place where the variation of parameters formula or some of its generalizations plays a role.

Let $I = \mathbb{R}$ or $I = [0, \infty)$ in the sequel. By $C_b(I : X)$, $C_0(I : X)$ and BUC(I : X)we denote the vector spaces consisting of all bounded continuous functions $f : I \to X$, all bounded continuous functions $f : I \to X$ such that $\lim_{|t|\to+\infty} ||f(t)|| = 0$ and all bounded uniformly continuous functions $f : I \to X$, respectively.

3.11.1. Asymptotically almost periodic type functions, asymptotically almost automorphic type functions and their generalizations. Let $f: I \to X$ be continuous. Given $\varepsilon > 0$, we call $\tau > 0$ an ε -period for $f(\cdot)$ iff $||f(t+\tau) - f(t)|| \leq \varepsilon, t \in I$. By $\vartheta(f, \varepsilon)$ we denote the set consisted of all ε -periods for $f(\cdot)$. It is said that $f(\cdot)$ is almost periodic (ap.) iff for each $\varepsilon > 0$ the set $\vartheta(f, \varepsilon)$ is relatively dense in I, which means that there exists l > 0 such that any subinterval of I of length l meets $\vartheta(f, \varepsilon)$. The space consisted of all almost periodic functions from the interval I into X will be denoted by AP(I:X).

The class of asymptotically almost periodic functions was introduced by M. Fréchet in 1941, for the case that $I = [0, \infty)$ (more details about aap. functions with values in Banach spaces can be found in [100, 144, 240] and references cited therein). If $I = \mathbb{R}$, there are several non-equivalent notions of an aap. function. Here we follow the approach of C. Zhang [557]:

A function $f \in C_b(I:X)$ is said to be asymptotically almost periodic iff for every $\varepsilon > 0$ we can find numbers l > 0 and M > 0 such that every subinterval of I of length l contains, at least, one number τ such that $||f(t+\tau) - f(t)|| \leq \varepsilon$ provided $|t|, |t+\tau| \geq M$. The space consisting of all aap. functions from I into Xis denoted by AAP(I:X). For a function $f \in C_b(I:X)$, the following statements are equivalent (see [465] for the case that $I = [0, \infty)$ and [557, Theorem 2.6] for the case that $I = \mathbb{R}$):

- (i) $f \in AAP(I:X)$.
- (ii) There exist uniquely determined functions $g \in AP(I:X)$ and $\phi \in C_0(I:X)$ such that $f = g + \phi$.

Unless stated otherwise, in the sequel we will always assume that $1 \leq p < \infty$. Let l > 0, and let $f, g \in L_{loc}^{p}(I : X)$. We define the Stepanov 'metric' by

$$D^p_{S_l}[f(\cdot), g(\cdot)] := \sup_{x \in I} \left[\frac{1}{l} \int_x^{x+l} \|f(t) - g(t)\|^p dt \right]^{1/p}$$

The Stepanov 'norm' of $f(\cdot)$ is defined by $||f||_{S_l^p} := D_{S_l}^p[f(\cdot), 0]$. It is said that a function $f \in L_{loc}^p(I:X)$ is Stepanov *p*-bounded, S^p -bounded shortly, iff $||f||_{S^p} := \sup_{t \in I} (\int_t^{t+1} ||f(s)||^p ds)^{1/p} < \infty$. The space $L_S^p(I:X)$ consisted of all S^p -bounded functions becomes a Banach space equipped with the above norm. We say that a function $f \in L_S^p(I:X)$ is Stepanov *p*-almost periodic, S^p -ap. shortly, iff the function $\hat{f}: I \to L^p([0,1]:X)$, defined by $\hat{f}(t)(s) := f(t+s), t \in I, s \in [0,1]$ is ap.. It is said that $f \in L_S^p(I:X)$ is asymptotically Stepanov *p*-almost periodic, S^p -aap. shortly, iff $\hat{f}: I \to L^p([0,1]:X)$ is ap.. By $APS^p(I:X)$ and $AAPS^p(I:X)$ we denote the spaces consisted of all S^p -ap. functions $I \mapsto X$ and S^p -aap. functions $I \mapsto X$, respectively.

Let $1 \leq p < \infty$, l > 0, and $f, g \in L^p_{loc}(I : X)$, where $I = \mathbb{R}$ or $I = [0, \infty)$. Recall that, for every two numbers $l_1, l_2 > 0$, there exist two positive real constants $k_1, k_2 > 0$ independent of f, g, such that

$$k_1 D_{S_{l_1}}^p[f(\cdot), g(\cdot)] \leqslant D_{S_{l_2}}^p[f(\cdot), g(\cdot)] \leqslant k_2 D_{S_{l_1}}^p[f(\cdot), g(\cdot)].$$

The symbol $S_0^p([0,\infty):X)$ stands for the vector space consisting of all functions $q \in L^p_{loc}([0,\infty):X)$ such that $\hat{q} \in C_0([0,\infty):L^p([0,1]:X))$. If $1 \leq p < q < \infty$ and $f(\cdot)$ is (asymptotically) Stepanov q-almost periodic, then $f(\cdot)$ is (asymptotically) Stepanov p-almost periodic. Therefore, the (asymptotic) Stepanov p-almost periodicity of $f(\cdot)$ for some $p \in [1,\infty)$ implies the (asymptotical) Stepanov p-almost periodic (respectively, asymptotically almost periodic) function then $f(\cdot)$ is also S^p -almost periodic (respectively, asymptotically S^p -almost periodic) for $1 \leq p < \infty$. And in general, the converse statement is false.

The notion of an (equi-)Weyl almost periodic function is given as follows (see [22, 293] and references cited therein for more details on the subject):

DEFINITION 3.11.1. Let $1 \leq p < \infty$ and $f \in L^p_{loc}(I:X)$.

(i) We say that the function $f(\cdot)$ is equi-Weyl-*p*-almost periodic, $f \in e - W_{ap}^p(I : X)$ for short, iff for each $\varepsilon > 0$ we can find two real numbers l > 0 and L > 0 such that any interval $I' \subseteq I$ of length L contains a point $\tau \in I'$ such that

$$\sup_{x\in I} \left[\frac{1}{l} \int_{x}^{x+l} \|f(t+\tau) - f(t)\|^{p} dt\right]^{1/p} \leqslant \varepsilon, \text{ i.e., } D_{S_{l}}^{p}[f(\cdot+\tau), f(\cdot)] \leqslant \varepsilon.$$

(ii) We say that the function $f(\cdot)$ is Weyl-*p*-almost periodic, $f \in W^p_{ap}(I:X)$ for short, iff for each $\varepsilon > 0$ we can find a real number L > 0 such that any interval $I' \subseteq I$ of length L contains a point $\tau \in I'$ such that

$$\lim_{l \to \infty} \sup_{x \in I} \left[\frac{1}{l} \int_{x}^{x+l} \|f(t+\tau) - f(t)\|^{p} dt \right]^{1/p} \leqslant \varepsilon, \text{ i.e., } \lim_{l \to \infty} D_{S_{l}}^{p}[f(\cdot+\tau), f(\cdot)] \leqslant \varepsilon.$$

We also need the definition of an asymptotically almost automorphic function defined on the interval I. For beginning, let us recall that a continuous function $f: \mathbb{R} \to X$ is said to be almost automorphic (aa., for short) iff for every real sequence (b_n) there exist a subsequence (a_n) of (b_n) and a map $g: \mathbb{R} \to X$ such that $\lim_{n\to\infty} f(t+a_n) = g(t)$ and $\lim_{n\to\infty} g(t-a_n) = f(t)$, pointwise for $t \in \mathbb{R}$. Any aa. function $f(\cdot)$ needs to be bounded and the following supremum formula holds (see e.g. [293, Lemma 3.9.9]):

$$||f||_{\infty} := \sup_{x \in \mathbb{R}} ||f(x)|| = \sup_{x \ge a} ||f(x)|| \quad \text{for any number} \quad a \in \mathbb{R}.$$

We will use the following notion (see also [155, Definition 2.3]):

- DEFINITION 3.11.2. (i) A bounded continuous function $f: \mathbb{R} \to X$ is said to be asymptotically almost automorphic iff there exist two functions $h \in AA(\mathbb{R}: X)$ and $q \in C_0(\mathbb{R}: X)$ such that f = h + q on \mathbb{R} .
- (ii) A bounded continuous function $f: [0, \infty) \to X$ is said to be asymptotically almost automorphic iff there exist two functions $h \in AA(\mathbb{R} : X)$ and $q \in C_0([0, \infty) : X)$ such that f = h + q on $[0, \infty)$.

It is well known that any (asymptotically) almost periodic function defined on the interval I is (asymptotically) almost automorphic as well as that the converse statement does not hold in general. Following G. M. N'Guérékata and A. Pankov [242], a function $f \in L^p_{loc}(\mathbb{R} : X)$ is called Stepanov *p*-almost automorphic, S^p -aa. for short, iff for asymptotically every real sequence (a_n) , there exist a subsequence (a_{n_k}) and a function $g \in L^p_{loc}(\mathbb{R} : X)$ such that

$$\lim_{k \to \infty} \int_t^{t+1} \|f(a_{n_k} + s) - g(s)\|^p ds = 0 \text{ and } \lim_{k \to \infty} \int_t^{t+1} \|g(s - a_{n_k}) - f(s)\|^p ds = 0$$

for each $t \in \mathbb{R}$. By $AAS^p(\mathbb{R} : X)$ we denote the vector space consisting of all S^p -aa. functions from \mathbb{R} to X.

The following definition seems to be new in case $I = \mathbb{R}$:

- DEFINITION 3.11.3. (i) An S^p -bounded function $f : \mathbb{R} \to X$ is said to be asymptotically Stepanov *p*-almost automorphic, S^p -aaa. for short, iff there exist two functions $h \in AAS^p(\mathbb{R} : X)$ and an S^p -bounded function $q : \mathbb{R} \to X$ such that $\hat{q} \in C_0(\mathbb{R} : L^p([0, 1] : X))$ and f = h + q a.e. on \mathbb{R} .
- (ii) An S^p -bounded function $f: [0, \infty) \to X$ is said to be asymptotically Stepanov *p*-almost automorphic iff there exist two functions $h \in AAS^p(\mathbb{R} : X)$ and an S^p -bounded function $q: [0, \infty) \to X$ such that $\hat{q} \in C_0([0, \infty) : L^p([0, 1] : X))$ and f = h + q a.e. on $[0, \infty)$.

By $AAAS^p(I:X)$ we denote the vector space consisting of all asymptotically S^p -almost automorphic functions $I \to X$.

It can be easily verified that the (asymptotical) S^{p} -almost automorphy of $f(\cdot)$ implies the (asymptotical) almost automorphy of the mapping $\hat{f}: I \to L^{p}([0, 1]: X)$ defined above. Any (asymptotically) S^{p} -almost periodic function $f: I \mapsto X$ has to be (asymptotically) S^{p} -almost automorphic, while the converse statement does not hold in general.

Denote by $C_{\omega}(I:X)$ the space consisting of all continuous ω -periodic functions $g: I \to X$. The following two definitions as well Definition 3.11.6 seem to be new in case $I = \mathbb{R}$, likewise (see H. R. Henríquez et al. [251], H. R. Henríquez [250], W. Dimbour, S. M. Manou-Abi [154] for case $I = [0, \infty)$):

DEFINITION 3.11.4. Let $\omega \in I$. Then we say that a bounded continuous function $f: I \to X$ is S-asymptotically ω -periodic iff $\lim_{|t|\to\infty} ||f(t+\omega) - f(t)|| = 0$. Denote by $SAP_{\omega}(I:X)$ the space consisting of all such functions.

DEFINITION 3.11.5. Let $\omega \in I$. A bounded continuous function $f: I \to X$ is said to be asymptotically ω -almost periodic iff there exist a function $g \in C_{\omega}(I:X)$ and a function $q \in C_0(I:X)$ such that f(t) = g(t) + q(t) for all $t \in I$. Denote by $AP_{\omega}(I:X)$ the vector space consisting of all such functions.

It is straightforward to see that $AP_{\omega}(I:X) \subseteq SAP_{\omega}(I:X)$ and the inclusion is strict. We will also work with the class of Stepanov S^{p} -asymptotically ω -periodic functions.

DEFINITION 3.11.6. Let $\omega \in I$. A Stepanov *p*-bounded function $f(\cdot)$ is said to be Stepanov *p*-asymptotically ω -periodic iff

$$\lim_{t \to \infty} \int_{t}^{t+1} \|f(s+\omega) - f(s)\|^p ds = 0.$$

Denote by $S^p SAP_{\omega}(I:X)$ the space consisting of all such functions.

We have that $SAP_{\omega}(I:X) \subseteq S^p SAP_{\omega}(I:X)$ and the inclusion is strict.

3.11.2. Evolution systems and Green's functions. The following definition is well known in the existing literature:

DEFINITION 3.11.7. A family $\{U(t,s) : t \ge s, t, s \in \mathbb{R}\} \subseteq L(X)$ is said to be an evolution system iff the following statements hold:

- (a) U(s,s) = I, U(t,s) = U(t,r)U(r,s) for $t \ge r \ge s$ and $t, r, s \in \mathbb{R}$,
- (b) $\{(\tau, s) \in \mathbb{R}^2 : \tau > s\} \ni (t, s) \mapsto U(t, s)x$ is continuous for any fixed element $x \in X$.

In the sequel, it will be always assumed that the family $A(\cdot)$ satisfies the following condition introduced by P. Acquistapace and B. Terreni in [5] (with $\omega = 0$):

(H1): There is a number $\omega \ge 0$ such that the family of closed linear operators $A(t), t \in \mathbb{R}$ on X satisfies $\overline{\Sigma_{\phi}} \subseteq \rho(A(t) - \omega)$,

$$\|R(\lambda:A(t)-\omega)\| = O((1+|\lambda|)^{-1}), \quad t \in \mathbb{R}, \ \lambda \in \overline{\Sigma_{\phi}}, \text{ and }$$

 $\|(A(t) - \omega)R(\lambda : A(t) - \omega)[R(\omega : A(t)) - R(\omega : A(s))]\| = O(|t - s|^{\mu}|\lambda|^{-\nu}),$ for any $t, s \in \mathbb{R}, \lambda \in \overline{\Sigma_{\phi}}$, where $\phi \in (\pi/2, \pi), 0 < \mu, \nu \leq 1$ and $\mu + \nu > 1$.

Then we know that there exists an evolution system $U(\cdot, \cdot)$ generated by $A(\cdot)$, satisfying that ||U(t,s)|| = O(1) for $t \ge s$, as well as a great deal of other conditions [5]. Besides (H1), we will also assume that the following condition holds:

(H2): The evolution system $U(\cdot, \cdot)$ generated by $A(\cdot)$ is hyperbolic (or, equivalently, has exponential dichotomy), i.e., there exist a family of projections $(P(t))_{t\in\mathbb{R}} \subseteq L(X)$, being uniformly bounded and strongly continuous in t, and constants $M', \omega > 0$ such that (a)-(c) holds with Q := I - P and $Q(\cdot) := I - P(\cdot)$, where I stands for the identity operator on X and: (a) U(t,s)P(s) = P(t)U(t,s) for all $t \ge s$,

- (b) the restriction $U_Q(t,s): Q(s)X \to Q(t)X$ is invertible for all $t \ge s$ (here we define $U_Q(s,t) = U_Q(t,s)^{-1}$), (c) $||U(t,s)P(s)|| \leq M'e^{-\omega(t-s)}$ and $||U_Q(s,t)Q(t)|| \leq M'e^{-\omega(t-s)}$ for all
- $t \ge s.$

It is said that $U(\cdot, \cdot)$ is exponentially stable iff the choice P(t) = I for all $t \in \mathbb{R}$ can be made; $U(\cdot, \cdot)$ is said to be (bounded) exponentially bounded iff there exist two finite real constants M > 0 and $(\omega = 0) \ \omega \in \mathbb{R}$ such that $||U(t,s)P(s)|| \leq |U(t,s)P(s)|| \leq |U(t,s)P(s)|| \leq |U(t,s)P(s)||$ $Me^{-\omega(t-s)}$ for all $t \ge s$. The associated Green's function $\Gamma(\cdot, \cdot)$ is defined through

$$\Gamma(t,s) := \begin{cases} U(t,s)P(s), & t \ge s, \ t,s \in \mathbb{R}, \\ -U_Q(t,s)Q(s), & t < s, \ t,s \in \mathbb{R}. \end{cases}$$

Let M' be the constant appearing in (H2). Then

(435)
$$\|\Gamma(t,s)\| \leqslant M' e^{-\omega|t-s|}, \quad t, \ s \in \mathbb{R}$$

and the function

(436)
$$u(t) := \int_{-\infty}^{+\infty} \Gamma(t, s) f(s) ds, \quad t \in \mathbb{R}$$

is a unique bounded continuous function on \mathbb{R} satisfying

$$u(t) = U(t,s)u(s) + \int_{s}^{t} U(t,\tau)f(\tau)d\tau, \quad t \ge s;$$

cf. [407]. In the sequel, it will be said that $u(\cdot)$ is a mild solution of the abstract Cauchy problem (449).

Let $f: [0,\infty) \to X$ be continuous. By a mild solution of the abstract Cauchy problem (450) we mean the function

(437)
$$u(t) := U(t,0)x + \int_0^t U(t,s)f(s)ds, \quad t \ge 0.$$

For more details on the subject, we refer the reader to [407, Section 5].

We will also consider the following semilinear Cauchy problems:

(438)
$$u'(t) = A(t)u(t) + F(t, u(t)), \quad t \in \mathbb{R}$$

and

(439)
$$u'(t) = A(t)u(t) + F(t, u(t)), \quad t > 0; \ u(0) = x.$$

DEFINITION 3.11.8. (i) A function $u \in C_b(\mathbb{R} : X)$ is said to be a mild solution of (438) iff

$$u(t) = \int_{-\infty}^{+\infty} \Gamma(t, s) F(s, u(s)) ds, \quad t \in \mathbb{R}.$$

(ii) A function $u \in C_b([0,\infty): X)$ is said to be a mild solution of (439) iff

$$u(t) = U(t,0)x + \int_0^t U(t,s)F(s,u(s))ds, \quad t \ge 0.$$

3.11.3. Quasi-asymptotically almost periodic functions and their generalizations. We start by recalling the following definition [336]:

DEFINITION 3.11.9. Suppose that $I = [0, \infty)$ or $I = \mathbb{R}$. Then we say that a bounded continuous function $f: I \to X$ is quasi-asymptotically almost periodic iff for each $\varepsilon > 0$ there exists a finite number $L(\varepsilon) > 0$ such that any interval $I' \subseteq I$ of length $L(\varepsilon)$ contains at least one number $\tau \in I'$ satisfying that there exists a finite number $M(\varepsilon, \tau) > 0$ such that

(440) $||f(t+\tau) - f(t)|| \leq \varepsilon$, provided $t \in I$ and $|t| \geq M(\varepsilon, \tau)$.

Denote by Q - AAP(I : X) the set consisting of all quasi-asymptotically almost periodic functions from I into X.

In order to avoid unnecessary repeating, we will use the shorthand

(S): "there exists a finite number $L(\varepsilon) > 0$ such that any interval $I' \subseteq I$ of length $L(\varepsilon)$ contains at least one number $\tau \in I'$ satisfying that there exists a finite number".

REMARK 3.11.10. It is not relevant whether we will write (440) or

 $||f(t+\tau) - f(t)|| \leq \varepsilon$, provided $t \in I$, $|t| \geq M(\varepsilon, \tau)$ and $|t+\tau| \geq M(\varepsilon, \tau)$.

Using this observation, it can be easily seen that the class of aap. functions is contained in the class of q-aap. functions (the number M depends only on ε and not on τ for aap. functions). The converse statement is not true, however: Let $I = \mathbb{R}$ and let $f(\cdot)$ be any bounded scalar-valued continuous function such that f(t) = 1 for all $t \ge 0$ and f(t) = 0 for all $t \le -1$. Then $f(\cdot)$ is q-aap., not aap. and not equi-Weyl-*p*-ap. for any $p \in [1, \infty)$ [22,336]. Applying Theorem 3.11.13(i) below we can see that $f(\cdot)$ is not aaa., as well.

We continue by providing an illustrative example and one more remark.

EXAMPLE 3.11.11. Suppose that $f \in C^1(I:X) \cap C_b(I:X)$ and $f' \in C_0(I:X)$. Then $f \in Q - AAP(I:X)$. In order to see this, it suffices to apply the Lagrange mean value theorem as well as to choose, in Definition 3.11.9, $L(\varepsilon) > 0$ arbitrarily and any $\tau \neq 0$ from an arbitrary interval I' of length $L(\varepsilon)$. Then, for this $\varepsilon > 0$ and $\tau \in I'$, there exists a sufficiently large $M(\varepsilon, \tau) > 0$ such that $[t, t + \tau] \subseteq \{s \in$ $I : |s| \ge M_0(\varepsilon, \tau)\}$ for $|t| \ge M(\varepsilon, \tau)$, where $M_0(\varepsilon, \tau)$ is already chosen so that $\|f'(\xi)\| \le \varepsilon/|\tau|$ for $|\xi| \ge M_0(\varepsilon, \tau)$; then we have

$$\|f(t+\tau) - f(t)\| \leq |\tau| \sup_{\xi \in [t,t+\tau]} \|f'(\xi)\| \leq \varepsilon, \quad |t| \ge M(\varepsilon,\tau).$$

It is worth noticing that there exists a function $f(\cdot)$ that is not aap. and satisfies the above properties; a typical example is given by $f(t) := \sin(\ln(1+t)), t \ge 0$ (see also [466, Example 4.1, Theorem 4.2]).

REMARK 3.11.12. In our joint research paper [341] with D. Velinov, we have recently introduced and analyzed the class of (asymptotically) almost anti-periodic functions. The notion of quasi-asymptotically almost anti-periodicity (q-aanp., for short) can be introduced in the following way: A bounded continuous $f: I \to X$ is called quasi-asymptotically almost anti-periodic iff for each $\varepsilon > 0$ (S) holds with a number $M(\varepsilon, \tau) > 0$ such that

(441)
$$||f(t+\tau) + f(t)|| \leq \varepsilon$$
, provided $t \in I$ and $|t| \ge M(\varepsilon, \tau)$.

Suppose that $t, t + \tau \in I$ as well as that $|t| \ge M(\varepsilon, \tau) + |\tau|$. Then $|t + \tau| \ge M(\varepsilon, \tau)$ and applying (441) twice, we get that

$$\begin{aligned} \|f(t+2\tau) - f(t)\| &= \|[f(t+2\tau) + f(t+\tau)] - [f(t+\tau) + f(t)]\| \\ &\leqslant \|f(t+2\tau) + f(t+\tau)\| + \|f(t+\tau) + f(t)\| \leqslant 2\varepsilon. \end{aligned}$$

Hence, any q-aanp. function is automatically q-aap. Further analysis of q-aanp. functions and their Stepanov generalizations are without scope of this book.

The space $C_b(I:X) \smallsetminus Q - AAP(I:X)$ is sufficiently large; it is clearly nonempty because it is very plainly to construct an example of an infinite-differentiable bounded function $f: I \to \mathbb{C}$ such that for each number $\tau \in I$ there exists a sequence $(t_n)_{n \in \mathbb{N}}$ in I with the properties that $\lim_{n \to \infty} |t_n| = \infty$ and $|f(t_n + \tau) - f(t_n)| \ge 1$ for all $n \in \mathbb{N}$. Furthermore, we have the following:

THEOREM 3.11.13. (i)
$$AAA(I:X) \cap Q - AAP(I:X) = AAP(I:X)$$

and $[AAA(I:X) \setminus AAP(I:X)] \cap Q - AAP(I:X) = \emptyset$.
(ii) $AA(\mathbb{R}:X) \cap Q - AAP(\mathbb{R}:X) = AP(\mathbb{R}:X)$.

PROOF. For the sake of brevity, we will consider only the case that $I = \mathbb{R}$. It is clear that $AAP(\mathbb{R}:X) \subseteq AAA(\mathbb{R}:X) \cap Q - AAP(\mathbb{R}:X)$. To prove the converse inclusion, suppose that $f \in AAA(\mathbb{R}:X) \cap Q - AAP(\mathbb{R}:X)$. Then there exist two functions $h \in AA(\mathbb{R}:X)$ and $q \in C_0(\mathbb{R}:X)$ such that f = h + q on \mathbb{R} and for each $\varepsilon > 0$ (S) holds with a number $M(\varepsilon, \tau) > 0$ such that

 $(442) ||[h(t+\tau) - h(t)] + [q(t+\tau) - q(t)]|| \leq \varepsilon, \text{ provided } t \in \mathbb{R} \text{ and } |t| \geq M(\varepsilon, \tau).$

Fix a number $\varepsilon > 0$ and suppose that the real number τ satisfies (442) for $|t| \ge M(\varepsilon, \tau)$. Since $q \in C_0(\mathbb{R} : X)$, we have that there exists a finite number $M_1(\varepsilon, \tau) \ge M(\varepsilon, \tau)$ such that

(443)
$$||h(t+\tau) - h(t)|| \leq 2\varepsilon$$
, provided $t \in \mathbb{R}$ and $|t| \geq M_1(\varepsilon, \tau)$.

Define the function $H \colon \mathbb{R} \to X$ by $H(t) := h(t + \tau) - h(t), t \in \mathbb{R}$. Since the space $AA(\mathbb{R} : X)$ is translation invariant, we have $H \in AA(\mathbb{R} : X)$. Applying supremum formula and (443), we get

$$\sup_{t\in\mathbb{R}} \|H(t)\| = \sup_{t\geqslant M_1(\varepsilon,\tau)} \|H(t)\| = \sup_{t\geqslant M_1(\varepsilon,\tau)} \|h(t+\tau) - h(t)\| \leqslant 2\varepsilon.$$

Hence, $||h(t + \tau) - h(t)|| \leq 2\varepsilon$ for all $t \in \mathbb{R}$ and $h(\cdot)$ is ap. by definition. Hence, $AAP(\mathbb{R} : X) = AAA(\mathbb{R} : X) \cap Q - AAP(\mathbb{R} : X)$, which immediately implies the second equality in (i). The proof of (ii) follows from the above arguments, as well.

It is expected that the range of a function $f \in Q - AAA(I:X) \cap BUC(I:X)$ need not be relatively compact, as in the case of aap. functions. In the following example, we will explain this fact in case $I = [0, \infty)$:

EXAMPLE 3.11.14. Let $X := c_0(\mathbb{C})$. Although the final conclusions presented here holds for the function $f(\cdot)$ considered in [251, Example 3.1], we will prove that the range of function

$$f(t) := \left(\frac{4n^2t^2}{(t^2 + n^2)^2}\right)_{n \in \mathbb{N}}, \quad t \ge 0$$

is not relatively compact in X. Using a similar analysis as in the afore-mentioned example, we get that the function $f(\cdot)$ is bounded and uniformly continuous with the estimate $||f(t+s) - f(t)|| \leq 8s, t, s > 0$ holding true. Further on, for each t > 0and $\tau \ge 0$, we have:

$$\|f(t+\tau) - f(t)\| \leqslant \sup_{n \in \mathbb{N}} \frac{4n^2[(t+\tau)^2 + \tau^2]}{(t^2 + n^2)^2((t+\tau)^2 + n^2)^2} \leqslant t^{-4} + 4\frac{\tau^2}{t^2}, \quad t > 0, \ \tau \geqslant 0.$$

By [251, Remark 3.1, Proposition 3.3], it readily follows that the range of $f(\cdot)$ is not relatively compact as well as that there is no number $\tau > 0$ such that $f(\cdot)$ is τ -normal on compact subsets; see [251, Definition 3.2] for the notion.

Now we will prove that the existence of a number $\omega \in I$ such that $f: I \to X$ is S-asymptotically ω -periodic implies that $f(\cdot)$ is gaap.:

PROPOSITION 3.11.15. We have $SAP_{\omega}(I:X) \subseteq Q - AAP(I:X)$.

PROOF. For $\varepsilon > 0$ given in advance, we can take $L(\varepsilon) = 2\omega$. Then any interval $I' \subset I$ of length $L(\varepsilon)$ contains a number $\tau = n\omega$ for some $n \in \mathbb{N}$. For this n and ε , there exists a finite number $M(\varepsilon, n) > 0$ such that $\|f(t+\omega) - f(t)\| \leq \varepsilon/n\omega$ for $|t| \ge M(\varepsilon, n)$. Then the final conclusion follows from the estimates

$$\|f(t+n\omega) - f(t)\| \leq \sum_{k=0}^{n-1} \|f(t+(k+1)\omega) - f(t+k\omega)\| \leq \sum_{k=0}^{n-1} \frac{\varepsilon}{n\omega} = \varepsilon/\omega,$$

vided $|t| \geq M(\varepsilon, n) + n\omega.$

provided $|t| \ge M(\varepsilon, n) + n\omega$.

In [548, Example 17], R. Xie and C. Zhang have constructed an example of a function $f \in SAP_2([0,\infty):X)$ that is not uniformly continuous. By the above proposition, the function $f(\cdot)$ is q-aap. and not uniformly continuous.

The following simple proposition, already known in the case that $I = [0, \infty)$, can be deduced by using the arguments contained in the proof of [78, Proposition 3.6]. An alternative proof can be given by using Theorem 3.11.13, Proposition 3.11.15 and an easy reformulation of [251, Lemma 3.1] in case $I = \mathbb{R}$:

Proposition 3.11.16. Let $\omega \in I$.

- (i) Suppose that $f \in SAP_{\omega}(I:X) \cap AAA(I:X)$. Then $f \in AP_{\omega}(I:X)$.
- (ii) Suppose that $f \in SAP_{\omega}(I:X) \cap AA(I:X)$. Then $f \in C_{\omega}(I:X)$.

Now we will introduce the Stepanov generalization of q-aap. functions:

DEFINITION 3.11.17. Let $f \in L^p_S(I : X)$. Then it is said $f(\cdot)$ is Stepanov *p*-quasi-asymptotically almost periodic (S^{*p*}-qaap., for short) iff for each $\varepsilon > 0$ (S) holds with a number $M(\varepsilon, \tau) > 0$ such that

(444)
$$\int_{t}^{t+1} \|f(s+\tau) - f(s)\|^{p} ds \leqslant \varepsilon^{p}, \text{ provided } t \in I \text{ and } |t| \ge M(\varepsilon, \tau).$$

Denote by $S^pQ - AAP(I:X)$ the set consisting of all Stepanov *p*-quasi-asymptotically almost periodic functions from *I* into *X*.

Using only definition, it readily follows that $Q - AAP(I:X) \subseteq S^pQ - AAP(I:X)$; it is clear that this inclusion can be strict since the function $f(t) := \chi_{[-1,\infty)}(t)$, $t \in \mathbb{R}$ is in class $S^pQ - AAP(\mathbb{R}:X)$ but not in class $Q - AAP(\mathbb{R}:X)$ because $f(\cdot)$ is not continuous. Furthermore, it follows immediately from definition that any S^p -aap. function is S^p -qaap. so that $AAPS^p(I:X) \subseteq S^pQ - AAP(I:X)$; this inclusion can be also strict since the continuous function $f(\cdot)$ from Remark 3.11.10 is not S^p -aap.. Furthermore, if $1 \leq p < p' < \infty$, then $S^{p'}Q - AAP(I:X) \subseteq S^pQ - AAP(I:X)$ and for any function $f \in L^p_S(I:X)$, we have that $f(\cdot)$ is S^p q-aap. iff the function $\hat{f}: I \to L^p([0,1]:X)$ is q-aap.. Using this fact and Theorem 3.11.13, we can simply verify the validity of following result:

THEOREM 3.11.18. (i)
$$S^pAAA(I:X) \cap S^pQ - AAP(I:X) = S^pAAP(I:X)$$

 X) and $[S^pAAA(I:X) \smallsetminus S^pAAP(I:X)] \cap S^pQ - AAP(I:X) = \emptyset$.
(ii) $S^pAA(\mathbb{R}:X) \cap S^pQ - AAP(\mathbb{R}:X) = S^pAP(\mathbb{R}:X)$.

The proof of following result is very similar to that of Proposition 3.11.15; therefore, it is omitted:

PROPOSITION 3.11.19. We have $S^p SAP_{\omega}(I:X) \subseteq S^p Q - AAP(I:X)$.

After introduction of Definition 3.11.17, it seems reasonable to ask whether we can analyze Weyl and Besicovitch generalizations of q-aap. functions. The following result says that the space $S^pQ - AAP(I:X)$ is contained in $W^p_{ap}(I:X)$ and that the above question is uninteresting (we have already seen that $S^pQ - AAP(I:X) \notin e - W^p_{ap}(I:X)$):

PROPOSITION 3.11.20. We have $S^pQ - AAP(I:X) \subseteq W^p_{ap}(I:X)$.

PROOF. Let $\varepsilon > 0$ be given. Then (S) holds with a number $M(\varepsilon, \tau) > 0$ such that (444) is satisfied. We need to estimate the term

(445)
$$\sup_{x \in I} \left[\frac{1}{l} \int_{x}^{x+l} \|f(t+\tau) - f(t)\|^{p} dt \right]^{1/p}$$

as $l \to +\infty$. There exist four possibilities:

 $\begin{array}{ll} 1. & |x| \geqslant M(\varepsilon,\tau) \text{ and } |x+l| \geqslant M(\varepsilon,\tau). \\ 2. & |x| \geqslant M(\varepsilon,\tau) \text{ and } |x+l| \leqslant M(\varepsilon,\tau). \\ 3. & |x| \leqslant M(\varepsilon,\tau) \text{ and } |x+l| \geqslant M(\varepsilon,\tau). \\ 4. & |x| \leqslant M(\varepsilon,\tau) \text{ and } |x+l| \leqslant M(\varepsilon,\tau). \end{array}$

Let us consider the first case. For $x \ge 0$, we have

$$\frac{1}{l}\int_x^{x+l} \|f(t+\tau) - f(t)\|^p dt$$

$$\leq \frac{1}{l} \left(\int_x^{x+1} \|f(t+\tau) - f(t)\|^p dt + \dots + \int_{x+\lfloor l\rfloor}^{x+l} \|f(t+\tau) - f(t)\|^p dt \right) \leq \frac{1}{l} l\varepsilon^p.$$

If $I = \mathbb{R}$ and $x \leq 0$, then $x \leq -M(\varepsilon, \tau)$, $x + l \ge M(\varepsilon, \tau)$ and arguing as above we get

This implies the existence of a sufficiently large number $l(\varepsilon, \tau) > 0$ such that the term in (445) is not greater than ε for any $l \ge l(\varepsilon, \tau)$. The analysis of cases 2.-4. is analogous and therefore omitted.

Proposition 3.11.20 is motivated by the old results of A. S. Kovanko [357, Théorème I, Théorème II], where the notion of asymptotical almost periodicity has been taken in a slightly different manner. It is also worth noting that the inclusion $S^pQ - AAP(I:X) \subseteq W^p_{ap}(I:X)$ is strict because the space $W^p_{ap}(I:X)$ contains certain Stepanov unbounded functions (see e.g. [22, Example 4.28] with $I = \mathbb{R}$ and p = 1).

Further on, arguing as in the proofs of structural results of [74, pp. 3–4], we may deduce the following:

THEOREM 3.11.21. Let $f: I \to X$ be q-aap. (S^p-qaap.). Then we have:

- (i) $cf(\cdot)$ is q-aap. $(S^p$ -qaap.) for any $c \in \mathbb{C}$.
- (ii) If $X = \mathbb{C}$ and $\inf_{x \in I} |f(x)| = m > 0$, then $1/f(\cdot)$ is q-aap. (S^p-qaap.).
- (iii) If $(g_n \colon I \to X)_{n \in \mathbb{N}}$ is a sequence of q-aap. functions and $(g_n)_{n \in \mathbb{N}}$ converges uniformly to a function $g \colon I \to X$, then $g(\cdot)$ is q-aap..
- (iv) If $(g_n: I \to X)_{n \in \mathbb{N}}$ is a sequence of S^p -qaap. functions and $(g_n)_{n \in \mathbb{N}}$ converges to a function $g: I \to X$ in the space $L^p_S(I:X)$, then $g(\cdot)$ is S^p -qaap..
- (v) The functions $f(\cdot + a)$ and $f(b \cdot)$ are likewise q-aap. (S^p-qaap.), where $a \in I$ and $b \in I \setminus \{0\}$.

Concerning the pointwise products of (Stepanov) scalar-valued q-aap. functions and (Stepanov) vector-valued q-aap. functions, the following classes play an important role:

DEFINITION 3.11.22. By $Q_h - AAP(I:X)$ $(S^pQ_h - AAP(I:X))$ we denote the class consisting of all q-aaa. $(S^p$ -qaap.) $f: I \to X$ satisfying that for each $\varepsilon > 0$ and $\tau \in I$ there exists a finite number $M(\varepsilon, \tau) > 0$ such that (440) ((444)) holds true.

The functions from Example 3.11.11 belong to the class $Q_h - AAP(I:X)$. It is also worth noting that the class $S^pQ_h - AAP(I:X)$ contains all functions that are S-asymptotically ω -periodic in the Stepanov sense for any number $\omega > 0$. Let $f: I \to X$ and $g: I \to \mathbb{C}$ be given. Using the elementary definitions and inequality $\|fg(t+\tau) - fg(t)\| \leq |g(t+\tau)| \|f(t+\tau) - f(t)\| + \|f(t)\| |g(t+\tau) - g(t)|, \quad t, \tau \in I,$ it readily follows the validity of following proposition:

PROPOSITION 3.11.23. Let the functions $f: I \to X$ and $g: I \to \mathbb{C}$ be bounded. If $f \in Q_h - AAP(I:X)$ and $g \in Q - AAP(I:\mathbb{C})$ $(f \in S^pQ_h - AAP(I:X)$ and $g \in S^pQ - AAP(I:\mathbb{C}))$ or $f \in Q - AAP(I:X)$ and $g \in Q_h - AAP(I:\mathbb{C})$ $(f \in S^pQ - AAP(I:X)$ and $g \in S^pQ_h - AAP(I:\mathbb{C}))$, then we have $fg \in Q_h$

 $Q - AAP(I:X) \ (fq \in S^pQ - AAP(I:X)).$

The conclusion established in the above proposition cannot be deduced if the functions $f(\cdot)$ and $g(\cdot)$ belong to the classes of qaap. functions or Stepanov qaap. functions, as the next instructive examples show:

EXAMPLE 3.11.24. We have that $AP(\mathbb{R}:\mathbb{C}) \cdot SAP_2(\mathbb{R}:\mathbb{C})$ is not a subset of $Q - AAP(\mathbb{R}:\mathbb{C})$ and, in particular, $Q - AAP(\mathbb{R}:\mathbb{C}) \cdot Q - AAP(\mathbb{R}:\mathbb{C})$ is not a subset of $Q - AAP(\mathbb{R}:\mathbb{C})$. A typical example of function belonging to the space $[AP(\mathbb{R}:\mathbb{C}) \cdot SAP_2(\mathbb{R}:\mathbb{C})] \cap S^pAAP(\mathbb{R}:\mathbb{C})$ but not to the space $Q - AAP(\mathbb{R}:\mathbb{C})$ is the function $\cos(\sqrt{2\pi} \cdot)f(\cdot)$, which can be verified as for the function $fg(\cdot)$ considered below, but much simpler.

Assume that $\alpha, \beta \in \mathbb{R}$ and $\alpha \beta^{-1}$ is a well-defined irrational number. Then the functions

$$g_0(t) = \sin\left(\frac{1}{2 + \cos \alpha t + \cos \beta t}\right), \quad t \in \mathbb{R}$$

and

$$g(t) = \cos\left(\frac{1}{2 + \cos \alpha t + \cos \beta t}\right), \quad t \in \mathbb{R}$$

are Stepanov *p*-ap. but not ap.. These functions are bounded continuous, not uniformly continuous and cannot be q-aap. because they are also aa. (see Theorem 3.11.13 and **[293]** for more details). Consider now the case that $\alpha = \pi$ and $\beta = \sqrt{2\pi}$ for function $g(\cdot)$. Let the function $f(\cdot)$ be defined on the real line by zero outside the non-negative real axis and by $f(t) := f_{\overline{\{1/n+1\}}}(t), t \ge 0$, where the function $f_{\overline{\{1/n+1\}}}(\cdot)$ has the same meaning in **[548**, Example 17]. That is, we define $f_{\overline{\{1/n+1\}}}(\cdot)$ by $f_{\overline{\{1/n+1\}}}(t) := 0$ for $t \in \{0, 2, 2n + 1 - \frac{1}{n+1}, 2n + 1 + \frac{1}{n+1} : n \in \mathbb{N}\}$,

 $f_{\overline{\{1/n+1\}}}(t) := 1$ for $t \in 2\mathbb{N} + 1$ and linearly outside. Assume that for each $\varepsilon > 0$ (S) holds with $I = [0, \infty)$ and a number $M(\varepsilon, \tau) > 0$ satisfying (440). If $\tau \notin 2\mathbb{N}$, then there exist two integers $k \in \mathbb{N}_0$, $n_0 \in \mathbb{N}$ and a real number $d \in (0, 2)$ such that $\tau = 2k + d$ and d > 2/1 + n for all $n \ge n_0$. Take now any number s = 2n + 1, where $n \in \mathbb{N}$ is chosen so that $2n + 1 \ge M(\varepsilon, \tau)$ and

(446)
$$\left|\cos\left(\frac{1}{1+\cos\sqrt{2}(2n+1)\pi}\right)\right| > \frac{4\varepsilon}{3}.$$

The existence of such a number can be easily shown. Then $f(2n+1) \ge 3/4$ and $2n+1+\tau \in [2(n+k)+\frac{1}{n+k+1}, 2(n+k+1)+1-\frac{1}{n+k+1}]$ due to the inequality d > 2/1+n. In combination with (446), the above yields $fg(2n+1+\tau) = 0$ and $|fg(2n+1)| > \varepsilon$, which is a contradiction since (440) holds with t = 2n+1. If $\tau = 2k$ for some $k \in \mathbb{N}$, then there exists a sufficiently large $s_0(\varepsilon) > 0$ such that

$$\begin{split} |fg(s+2k) - fg(s)| \geqslant |f(s+2k)||g(s+2k) - g(s)| - |f(s+2k) - f(s)| \\ \geqslant |f(s+2k)||g(s+2k) - g(s)| - \frac{\varepsilon}{2}, \ s \geqslant s_0(\varepsilon). \end{split}$$

If $n \in \mathbb{N}$ is arbitrary and $s = 2n + 1 \ge s_0(\varepsilon)$, the above estimate yields

(447)
$$|fg(2n+1+2k) - fg(2n+1)| \ge \frac{3}{4}|g(2n+1+2k) - g(2n+1)| - \frac{\varepsilon}{2}.$$

Further on, it is very elementary to prove that there exists a strictly increasing sequence $(2a_{l,k} + 1)_{l \in \mathbb{N}}$ of odd integers such that the inequality

$$\left|\cos\left(\frac{1}{1+\cos\sqrt{2}(2a_{l,k}+1+2k)\pi}\right)-\cos\left(\frac{1}{1+\cos\sqrt{2}(2a_{l,k}+1)\pi}\right)\right| \ge \frac{8\varepsilon}{3}$$

holds provided $l \in \mathbb{N}$.

In combination with (447), we get that (440) does not hold with s = 2n + 1. Hence, $fg \notin Q - AAP(\mathbb{R} : \mathbb{C})$ while in the meantime $g \in S^pQ - AAP(\mathbb{R} : \mathbb{C}) \setminus Q - AAP(\mathbb{R} : \mathbb{C})$ and $f \in SAP_2(\mathbb{R} : \mathbb{C})$. Since the function $f(\cdot)$ is Stepanov *p*-vanishing, i.e, $\lim_{t \to +\infty} \int_t^{t+1} |f(s)|^p ds = 0$, it can be easily seen that $fg \in S^pAAP([0,\infty) : X)$, as well.

EXAMPLE 3.11.25. It is very simple to illustrate that the pointwise product of two essentially bounded functions from $S^pQ - AAP(I : \mathbb{C})$ need not belong to the same class. We will show this only in the case that $I = [0, \infty)$ by giving an example of a Stepanov *p*-ap. function $g(\cdot)$ and a function $f \in SAP_4([0, \infty) : \mathbb{C})$ such that $fg \notin S^pQ - AAP([0, \infty) : \mathbb{C})$. To see this, put $g(t) := \operatorname{sign}(\sin t)$, $t \ge 0$, where $\operatorname{sign}(0) := 0$. Then it is well known that $g(\cdot)$ is Stepanov *p*-ap. function; see e.g. [**293**]. We construct $f(\cdot)$ in the following way: Define f(t) := 0for $t \in \{0, 43/10, 46/10, 4n + \frac{1}{2} - \frac{1}{4n+1}, 4n + \frac{3}{2} + \frac{1}{4n+1} : n \in \mathbb{N}\}$, f(t) := 1 for $t \in \bigcup_{n \in \mathbb{N}} [4n + 1/2, 4n + 3/2]$ and linearly outside. Then it can be easily seen that $f \in SAP_4([0, \infty) : \mathbb{C})$. Furthermore, the function $fg(\cdot)$ does not belong to the class $S^pQ - AAP([0, \infty) : \mathbb{C})$ because if we suppose the contrary, then we can always take the segment $[4n + 1/2, 4n + 3/2] \subseteq [M(\varepsilon, \tau), \infty)$ sufficiently large, with the meaning clear, and the condition (444) will be always violated with t = 4n + 1/2. This can be seen by considering separately two possible cases: $\tau \in 4\mathbb{N}$ and $\tau \notin 4\mathbb{N}$. In the first case, we have the existence of a number $k \in \mathbb{N}$ such that $\tau = 4k$. Then

$$\begin{split} \int_{4n+\frac{1}{2}}^{4n+\frac{3}{2}} |\operatorname{sign}(\sin(s+\tau))f(s+\tau) - \operatorname{sign}(\sin s)f(s)|^p ds \\ &= \int_{4n+\frac{1}{2}}^{4n+\frac{3}{2}} |\operatorname{sign}(\sin(s+\tau)) - \operatorname{sign}(\sin s)|^p ds \end{split}$$

for all $n \in \mathbb{N}$. Further on, observe that

$$\sin(s+\tau) - \sin s = 2\sin 2k\cos(s+2k), \quad s \in \mathbb{R}.$$

If $\sin 2k > 0$, the terms $\sin(s + \tau)$ and $\sin s$ will have different signs for all $s \in [4n + 1/2, 4n + 3/2]$ provided that there exists a natural number $m \in \mathbb{N}$ such that 4n + 2k + 1/2 and 4n + 2k + 3/2 belong to the set $(\pi/2 + 2m\pi, 3\pi/2 + 2m\pi)$. This could happen for arbitrarily large values of $n \in \mathbb{N}$, so that (444) does not hold. The examination is similar provided that $\sin 2k < 0$. In the second case, let $\tau = 4m + \tau_0$ for some $m \in \mathbb{N}_0$ and $\tau_0 \in (0, 4)$. Since the integer multiples of π get arbitrarily close to the integers, there is a strictly increasing sequence of natural numbers $(n_k)_{k \in \mathbb{N}}$ such that $[4n_k + 1/2, 4n_k + 3/2] \subseteq \cup_{l \in \mathbb{N}} (2l\pi, (2l + 1)\pi)$. For $n = n_k \ge M(\varepsilon, \tau)$, we have sign($\sin s$) = 1 for all $s \in [4n_k + 1/2, 4n_k + 3/2]$ and we can use the estimate

$$\begin{split} \int_{4n_k+\frac{3}{2}}^{4n_k+\frac{3}{2}} |\operatorname{sign}(\sin(s+\tau))f(s+\tau) - \operatorname{sign}(\sin s)|^p ds \\ \geqslant 2^{p-1} \int_{4n_k+\frac{3}{2}}^{4n_k+\frac{3}{2}} [1 - |f(s+\tau)|^p] ds \\ &= 2^{p-1} \bigg[1 - \int_{4n_k+\tau+\frac{1}{2}}^{4n_k+\tau+\frac{3}{2}} |f(s)|^p ds \bigg]. \end{split}$$

to conclude that there exist a zero sequence $(a_k)_{k\in\mathbb{N}}$ and a positive constant $c(\tau_0) \in (0,1)$ such that

$$\int_{4(n_k+m)+\tau_0+1/2}^{4(n_k+m)+\tau_0+3/2} |f(s)|^p ds > c(\tau_0) + a_k$$

for all $k \in \mathbb{N}$ such that $n_k \ge M(\varepsilon, \tau)$, the violation of (444) becomes apparent.

It is clear that the sets $Q_h - AAP(I:X)$ and $S^pQ_h - AAP(I:X)$ equipped with the usual operations of pointwise sums and products with scalars form vector spaces. This is no longer true for the spaces Q - AAP(I:X) and $S^pQ - AAP(I:X)$, as the following example shows:

EXAMPLE 3.11.26. For the sake of simplicity, we will consider only the case that $I = [0, \infty)$. Let $f(\cdot)$ and $g(\cdot)$ be as in the former example. Repeating the same argumentation (in the case that $\tau \notin 4\mathbb{N}$, then we can use the estimate $|\operatorname{sign}(\sin(s + \tau)) + f(s + \tau) - \operatorname{sign}(\sin s) - f(s)| = |\operatorname{sign}(\sin(s + \tau)) + f(s + \tau) - 2| \ge 1 - |f(s + \tau)|$ for all $s \in [4n_k + 1/2, 4n_k + 3/2] \subseteq \bigcup_{l \in \mathbb{N}} (2l\pi, (2l + 1)\pi))$, we get that $f + g \notin 4\mathbb{N}$ $S^pQ - AAP([0,\infty):\mathbb{C})$, so that $S^pQ - AAP([0,\infty):\mathbb{C}) + S^pQ - AAP([0,\infty):\mathbb{C})$ is not contained in $S^pQ - AAP([0,\infty):\mathbb{C})$. Now we will prove by a simple indirect proof that $AP([0,\infty):\mathbb{C}) + SAP_4([0,\infty):\mathbb{C})$ does not make a subclass of the class $S^pQ - AAP([0,\infty):\mathbb{C})$ so that $Q - AAP([0,\infty):X)$ cannot be a vector space, as announced above. There exists a sequence $(g_n)_{n\in\mathbb{N}}$ converging to $g(\cdot)$ in $L_S^p([0,\infty):\mathbb{C})$. This implies that $g_n + f$ converges to g + f in $L_S^p([0,\infty):\mathbb{C})$ as $n \to \infty$. Due to Theorem 3.11.21(iv), there exists a number $n_0 \in \mathbb{N}$ such that $g_{n_0} + f$ does not belong to the class $S^pQ - AAP([0,\infty):\mathbb{C})$; it is almost trivial to construct a sequence $(g_n)_{n\in\mathbb{N}}$ in $C_{2\pi}([0,\infty):\mathbb{C})$ converging to $g(\cdot)$ in $L_S^p([0,\infty):\mathbb{C})$, so that a function from $C_{2\pi}([0,\infty):\mathbb{C}) + SAP_4([0,\infty):\mathbb{C})$ need not belong to the class $S^pQ - AAP([0,\infty):\mathbb{C})$, as well. The interested reader may try to provide some concrete examples here.

The compactness in the spaces of (equi-)Weyl-*p*-almost periodic functions has been analyzed in [**358**, **359**] with the help of Lusternik type theorems. It is without scope this book to analyze similar problems for the space of q-aap. functions and its Stepanov generalizations.

3.11.4. Quasi-asymptotically almost periodic functions depending on two parameters and composition principles. The main aim of this subsection is to investigate q-aap. functions depending on two parameters and compositions of q-aap. functions. We start with the following definition:

DEFINITION 3.11.27. Suppose that $F : I \times X \to Y$ is a continuous function. Then we say that $F(\cdot, \cdot)$ is quasi-asymptotically almost periodic, uniformly on bounded subsets of X, iff for each $\varepsilon > 0$ (S) holds with a number $M(\varepsilon, \tau) > 0$ such that for each bounded subset B of X we have:

$$||F(t+\tau,x) - F(t,x)||_Y \leq \varepsilon$$
, provided $t \in I$, $x \in B$ and $|t| \geq M(\varepsilon,\tau)$

Denote by $Q - AAP(I \times X : Y)$ the set consisting of all quasi-asymptotically almost periodic functions from $I \times X$ into Y.

Arguing as in the proofs of [144, Theorem 3.30, Theorem 3.31], we may deduce the following results about compositions of q-aap. functions:

THEOREM 3.11.28. Suppose that $F \in Q-AAP(I \times X : Y)$ and $f \in Q-AAP(I : X)$. If there exists a finite number L > 0 such that

(448)
$$||F(t,x) - F(t,y)||_Y \leq L||x-y||, \quad x,y \in X, \ t \in I,$$

then the function $t \mapsto F(t, f(t)), t \in I$ belongs to the class Q - AAP(I : Y).

THEOREM 3.11.29. Suppose that $F \in Q - AAP(I \times X : Y)$ and $f \in Q - AAP(I : X)$. If the function $x \mapsto F(t, x)$, $t \in I$ is uniformly continuous on every bounded subset $B \subseteq X$ uniformly for $t \in I$, then the function $t \mapsto F(t, f(t))$, $t \in I$ belongs to the class Q - AAP(I : Y).

The class of S^p -qaap. functions depending on two parameters is introduced in the following definition:

DEFINITION 3.11.30. Suppose that a function $F: I \times X \to Y$ satisfies that for each $x \in X$ the function $t \mapsto F(t, x), t \in I$ is Stepanov *p*-bounded. Then we say that $F(\cdot, \cdot)$ is Stepanov *p*-quasi-asymptotically almost periodic, uniformly on bounded subsets of X, iff for each $\varepsilon > 0$ (S) holds with a number $M(\varepsilon, \tau) > 0$ such that for each bounded subset B of X we have:

$$\int_{t}^{t+1} \|F(s+\tau,x) - F(s,x)\|^{p} \, ds \leqslant \varepsilon^{p}, \text{ provided } t \in I, \ x \in B \text{ and } |t| \ge M(\varepsilon,\tau).$$

Denote by $S^pQ - AAP(I \times X : Y)$ the set consisting of all Stepanov *p*-quasiasymptotically almost periodic functions from $I \times X$ into *Y*.

In [335, Definition 3.1], we have recently introduced the class $e - W_{ap,K}(I \times X, X)$ consisting of equi-Weyl-*p*-ap. functions, uniformly with respect to compact subsets of X. The class $e - W_{ap,K}(I \times X, Y)$ with two different pivot spaces X and Y as well as the class $(e-)W_{ap,B}(I \times X, Y)$ consisting of all (equi-)Weyl-*p*-ap. functions, uniformly with respect to bounded subsets of X, can be introduced in a similar way. Following the method proposed in the proof of Proposition 3.11.20, we can show then $S^pQ - AAP(I \times X : Y) \subseteq W_{ap,B}(I \times X, Y)$. The following composition principles can be deduced in exactly the same way as it has been done in the proofs of [399, Lemma 2.1, Theorem 2.2]:

THEOREM 3.11.31. Suppose that the following conditions hold:

(i) $F \in S^pQ - AAP(I \times X : Y)$ with p > 1, and there exist a number $r \ge \max(p, p/p - 1)$ and a function $L_F \in L_S^r(I)$ such that:

 $||F(t,x) - F(t,y)|| \leq L_F(t)||x - y||_Y, \quad t \in I, \ x, y \in Y;$

(ii) $x \in S^pQ - AAP(I : X)$, and there exists a set $E \subseteq I$ with m(E) = 0 such that $K := \{x(t) : t \in I \setminus E\}$ is relatively compact in X; here, $m(\cdot)$ denotes the Lebesgue measure.

Then $q := pr/p + r \in [1, p)$ and $F(\cdot, x(\cdot)) \in S^qQ - AAP(I:Y)$.

THEOREM 3.11.32. Suppose that the following conditions hold:

(i) $F \in S^pQ - AAP(I \times X : Y)$ with $p \ge 1, L > 0$ and

 $||F(t,x) - F(t,y)|| \le L||x - y||, \quad t \in I, \ x, y \in X.$

(ii) $x \in S^pQ - AAP(I : X)$, and there exists a set $E \subseteq I$ with m(E) = 0 such that $K = \{x(t) : t \in I \setminus E\}$ is relatively compact in Y.

Then $F(\cdot, x(\cdot)) \in S^q Q - AAP(I:Y).$

3.11.5. Invariance of quasi-asymptotical almost periodicity under the action of convolution products. Concerning the invariance of quasi-asymptotical almost periodicity under the action of finite convolution product, we have the following result:

PROPOSITION 3.11.33. Suppose that $(R(t))_{t>0} \subseteq L(X,Y)$ is a strongly continuous operator family and $\int_0^\infty ||R(s)|| ds < \infty$. If $f \in Q - AAP([0,\infty) : X)$, then the function $F(\cdot)$, defined through (434), belongs to the class $Q - AAP([0,\infty) : Y)$. PROOF. Without loss of generality, we may assume that X = Y. It is clear that $||F(t)|| = ||\int_0^t R(s)f(t-s)ds|| \leq ||f||_{\infty} \int_0^\infty ||R(s)||ds, t \geq 0$ so that $F(\cdot)$ is bounded. Since

$$||F(t) - F(t')|| \leq \int_0^\infty ||R(s)|| ||f(t-s) - f(t'-s)|| ds$$

for any $t, t' \ge 0$, the continuity of F(t) for $t \ge 0$ follows from the boundedness of $f(\cdot)$ and the dominated convergence theorem. Let $\varepsilon > 0$ be given. Then (S) holds with a number $M(\varepsilon, \tau) > 0$ such that (440) holds. On the other hand, the condition $\int_0^\infty ||R(s)|| ds < \infty$ implies $\lim_{t \to +\infty} \int_t^\infty ||R(s)|| ds = 0$ so that there exists a finite number $M_0(\varepsilon) > 0$ such that $\int_t^\infty ||R(s)|| ds < \varepsilon$ for any $t \ge M_0(\varepsilon)$. Let $t \ge M(\varepsilon, \tau) + M_0(\varepsilon)$. Then we have

$$\begin{split} \|F(t+\tau) - F(t)\| \\ &= \left\| \int_0^t R(s) [f(t+\tau-s) - f(t-s)] ds + \int_t^{t+\tau} R(s) f(t+\tau-s) ds \right\| \\ &\leqslant \int_0^t \|R(s)\| \|f(t+\tau-s) - f(t-s)\| ds + \|f\|_{\infty} \int_t^{t+\tau} \|R(s)\| ds \\ &\leqslant \int_0^{t-M(\varepsilon,\tau)} \|R(s)\| \|f(t+\tau-s) - f(t-s)\| ds \\ &+ \int_{t-M(\varepsilon,\tau)}^t \|R(s)\| \|f(t+\tau-s) - f(t-s)\| ds + \varepsilon \|f\|_{\infty} \\ &\leqslant \varepsilon \int_0^\infty \|R(s)\| ds + 2\|f\|_{\infty} \varepsilon + \varepsilon \|f\|_{\infty}, \end{split}$$

which completes the proof in a routine manner.

The situation is quite similar for the infinite convolution product:

PROPOSITION 3.11.34. Suppose that $(R(t))_{t>0} \subseteq L(X,Y)$ is a strongly continuous operator family and $\int_0^\infty ||R(s)|| ds < \infty$. If $f \in Q - AAP(\mathbb{R} : X)$, then the function $\mathbf{F}(t)$, defined through (433), belongs to the class $Q - AAP(\mathbb{R} : Y)$.

PROOF. Without loss of generality, we may assume that X = Y. The boundedness and continuity of $\mathbf{F}(\cdot)$ can be proved as in the former proposition. To prove that $\mathbf{F}(\cdot)$ satisfies the remaining requirement from definition of quasi-asymptotical almost periodicity, fix a number $\varepsilon > 0$. By definition, (S) holds with a number $M(\varepsilon, \tau) > 0$ satisfying (440). On the other hand, the condition $\int_0^\infty ||R(s)|| ds < \infty$ implies $\lim_{t\to+\infty} \int_t^\infty ||R(s)|| ds = 0$ so that there exists a finite number $M_0(\varepsilon) > 0$ such that $\int_t^\infty ||R(s)|| ds < \varepsilon$ for any $t \ge M_0(\varepsilon)$. Let $|t| \ge M(\varepsilon, \tau) + M_0(\varepsilon)$. If $t \le -M(\varepsilon, \tau) - M_0(\varepsilon)$, then $t \le -M(\varepsilon, \tau), |t-s| \ge M(\varepsilon, \tau)$ for all $s \ge 0$, and

$$\|\mathbf{F}(t+\tau) - \mathbf{F}(t)\| \leq \int_0^\infty \|R(s)\| \|f(t+\tau-s) - f(t-s)\| ds \leq \varepsilon \int_0^\infty \|R(s)\| ds.$$

If $t \ge M(\varepsilon, \tau) + M_0(\varepsilon)$, then $t - M(\varepsilon, \tau) \ge M_0(\varepsilon)$ so that $\int_{t-M(\varepsilon, \tau)}^{\infty} \|R(s)\| ds < \varepsilon$; furthermore, in this case we have:

$$\begin{split} \|\mathbf{F}(t+\tau) - \mathbf{F}(t)\| &\leqslant \int_{0}^{\infty} \|R(s)\| \|f(t+\tau-s) - f(t-s)\| ds \\ &\leqslant \varepsilon \Big(\int_{0}^{t-M(\varepsilon,\tau)} + \int_{t+M(\varepsilon,\tau)}^{\infty} \Big) \|R(s)\| ds + 2\|f\|_{\infty} \int_{t-M(\varepsilon,\tau)}^{t+M(\varepsilon,\tau)} \|R(s)\| ds \\ &\leqslant 2\varepsilon \int_{0}^{\infty} \|R(s)\| ds + 2\|f\|_{\infty} \int_{t-M(\varepsilon,\tau)}^{\infty} \|R(s)\| ds \\ &\leqslant 2\varepsilon \int_{0}^{\infty} \|R(s)\| ds + 2\|f\|_{\infty} \int_{t-M(\varepsilon,\tau)}^{\infty} \|R(s)\| ds + 2\|f\|_{\infty} \varepsilon, \end{split}$$
hich completes the proof of proposition.

which completes the proof of proposition.

Suppose that 1/p + 1/q = 1 and $\sum_{k=0}^{\infty} \|R(\cdot)\|_{L^q[k,k+1]} < \infty$. This condition implies $\int_0^\infty \|R(s)\| ds < \infty$ and can be used for the examination of q-aap. properties of convolution products with Stepanov S^{p} -qaap. inhomogenities $f(\cdot)$. Keeping in mind the proofs of Proposition 3.11.33 and Proposition 3.11.34, as well as the proofs of Proposition 2.6.11 and Proposition 3.5.3 in [293], the following results can be deduced:

PROPOSITION 3.11.35. Suppose that 1/p + 1/q = 1, $(R(t))_{t>0} \subseteq L(X,Y)$ is a strongly continuous operator family and $\sum_{k=0}^{\infty} ||R(\cdot)||_{L^q[k,k+1]} < \infty$. If $f \in S^pQ$ – $AAP([0,\infty): X)$, then the function $F(\cdot)$, defined by (434), belongs to the class $Q - AAP([0,\infty):Y).$

PROPOSITION 3.11.36. Suppose that 1/p + 1/q = 1, $(R(t))_{t>0} \subseteq L(X,Y)$ is a strongly continuous operator family and $\sum_{k=0}^{\infty} ||R(\cdot)||_{L^q[k,k+1]} < \infty$. If $f \in S^pQ$ – $AAP(\mathbb{R}:X)$, then the function $\mathbf{F}(\cdot)$, defined by (433), belongs to the class Q – $AAP(\mathbb{R}:Y).$

For the sake of completeness, we will include the proofs (the preassumption X = Y can be made):

PROOF OF PROPOSITION 3.11.35. It is easy to see that

$$||F(t)|| \leq \sum_{k=0}^{\infty} ||R(\cdot)||_{L^{q}[k,k+1]} ||f||_{S^{p}}, \quad t \ge 0.$$

The continuity of F(t) for $t \ge 0$ can be proved as follows. Let $t, t' \ge 0$ and $|t - t'| \leq 1$. Then we have

$$\begin{aligned} \|F(t) - F(t')\| &\leqslant \int_0^t \|R(s)\| \|f(t-s) - f(t'-s)\| ds + \int_t^{t'} \|R(s)\| \|f(t'-s)\| ds \\ &\leqslant \sum_{k=0}^{\lfloor t \rfloor} \|R(\cdot)\|_{L^q[k,k+1]} \|f(t-\cdot) - f(t'-\cdot)\|_{L^p[k,k+1]} + \|f\|_{S^p} \|R(\cdot)\|_{L^q[\min(t,t'),\max(t,t')]}. \end{aligned}$$

Since $f \in L^p_{loc}([0,\infty): X)$, we have $\lim_{t'\to t} \|f(t-\cdot) - f(t'-\cdot)\|_{L^p[k,k+1]} = 0$ for $k = 0, \ldots, \lfloor t \rfloor$. Clearly, $\lim_{t'\to t} \|R(\cdot)\|_{L^q[\min(t,t'),\max(t,t')]} = 0$ so that the function $F(\cdot)$ is continuous at point t. Let $\varepsilon > 0$ be given. Then (S) holds with a number $M(\varepsilon, \tau) > 0$ satisfying (444). Furthermore, there exists $k_0(\varepsilon) \in \mathbb{N}$ such that $\sum_{k=k_0(\varepsilon)}^{\infty} \|R(\cdot)\|_{L^q[k,k+1]} < \varepsilon$. For any $t \ge M(\varepsilon, \tau) + k_0(\varepsilon) + 1$, we have

$$\begin{split} \|F(t+\tau) - F(t)\| &\leq \sum_{k=0}^{\lfloor t \rfloor} \|R(\cdot)\|_{L^{q}[k,k+1]} \|f(t+\tau-\cdot) - f(t-\cdot)\|_{L^{p}[k,k+1]} \\ &\leq \varepsilon \sum_{k=0}^{k_{0}(\varepsilon)} \|R(\cdot)\|_{L^{q}[k,k+1]} + 2^{p-1} \|f\|_{S^{p}} \sum_{k=k_{0}(\varepsilon)}^{\infty} \|R(\cdot)\|_{L^{q}[k,k+1]} \\ &\leq \varepsilon \sum_{k=0}^{\infty} \|R(\cdot)\|_{L^{q}[k,k+1]} + 2\|f\|_{S^{p}} \varepsilon, \end{split}$$

finishing the proof.

PROOF OF PROPOSITION 3.11.36. The boundedness and continuity of function $\mathbf{F}(\cdot)$ can be shown as in the proof of [**293**, Proposition 3.5.3]. Let $\varepsilon > 0$ be given. Then (S) holds with a number $M(\varepsilon, \tau) > 0$ satisfying (444). As above, there exists $k_0(\varepsilon) \in \mathbb{N}$ such that $\sum_{k=k_0(\varepsilon)}^{\infty} ||R(\cdot)||_{L^q[k,k+1]} < \varepsilon$. Let $t \in \mathbb{R}$ be such that $|t| \ge M(\varepsilon, \tau) + k_0(\varepsilon) + 1$. Then we have

$$\|F(t+\tau) - F(t)\| \leq \sum_{k=0}^{\infty} \|R(\cdot)\|_{L^{q}[k,k+1]} \|f(\cdot+\tau) - f(\cdot)\|_{L^{p}[t-(k+1),t-k]}.$$

If $t \leq -M(\varepsilon,\tau)$, then $[t - (k+1), t - k] \subseteq (-\infty, -M(\varepsilon,\tau)]$ for any $k \in \mathbb{N}_0$ and the above estimate immediately implies $||F(t+\tau) - F(t)|| \leq \varepsilon \sum_{k=0}^{\infty} ||R(\cdot)||_{L^q[k,k+1]}$. If $t \geq M(\varepsilon,\tau) + k_0(\varepsilon) + 1$, then $\lfloor t - M(\varepsilon,\tau) \rfloor \geq k_0(\varepsilon)$ so that

$$\begin{split} \|F(t+\tau)-F(t)\| &\leqslant \sum_{k=0}^{\lfloor t-M(\varepsilon,\tau) \rfloor} \|R(\cdot)\|_{L^q[k,k+1]} \|f(\cdot+\tau)-f(\cdot)\|_{L^p[t-(k+1),t-k]} \\ &+ \sum_{k=\lfloor t-M(\varepsilon,\tau) \rfloor}^{\lceil t+M(\varepsilon,\tau) \rceil} \|R(\cdot)\|_{L^q[k,k+1]} \|f(\cdot+\tau)-f(\cdot)\|_{L^p[t-(k+1),t-k]} \\ &+ \sum_{k=\lceil t+M(\varepsilon,\tau) \rceil}^{\infty} \|R(\cdot)\|_{L^q[k,k+1]} \|f(\cdot+\tau)-f(\cdot)\|_{L^p[t-(k+1),t-k]} \\ &\leqslant \varepsilon \bigg(\sum_{k=0}^{\lfloor t-M(\varepsilon,\tau) \rfloor} + \sum_{k=\lceil t+M(\varepsilon,\tau) \rceil}^{\infty} \bigg) \|R(\cdot)\|_{L^q[k,k+1]} + 2\|f\|_{S^p} \varepsilon \\ &\leqslant 2\varepsilon \sum_{k=0}^{\infty} \|R(\cdot)\|_{L^q[k,k+1]} + 2\|f\|_{S^p} \varepsilon, \end{split}$$

finishing the proof.

 \square

3.11.6. Applications to abstract nonautonomous differential equations of first order. Throughout this subsection, it will be always assumed that the operator family $A(\cdot)$ satisfies the condition (H1) and the evolution system $U(\cdot, \cdot)$ generated by $A(\cdot)$ is hyperbolic, i.e., the condition (H2) holds true.

We analyze here the following abstract quasi-linear differential equations of first order:

(449)
$$u'(t) = A(t)u(t) + f(t), \quad t \in \mathbb{R},$$

(450)
$$u'(t) = A(t)u(t) + f(t), \quad t > 0; \ u(0) = x$$

and their semilinear analogues. In our recent research studies [338, 339], the author has considered the existence and uniqueness of generalized almost periodic properties of mild solutions of (449)–(450) and their semilinear analogues. In the formulations and proofs of all structural results from [338, Section 2] and [339, Section 3], the essential boundedness of forcing term $f(\cdot)$ has been required as well as certain additional conditions on the generalized almost periodicity of $f(\cdot)$. In contrast to this, in the formulation of the following result, we require the Stepanov p-boundedness of function $f(\cdot)$ for some exponent $p \in [1,\infty)$; by \mathcal{F} we denote a general function space consisted of continuous functions from $[0,\infty)$ into X.

THEOREM 3.11.37. Let $I = [0, \infty)$, 1/p + 1/q = 1 and $f \in S^pQ - AAP(\mathbb{R} : X)$. If $x \in P(0)X \cap \overline{D(A(0))}$, the function $t \mapsto \int_0^t U(t,s)Q(s)f(s)ds$, $t \ge 0$ belongs to the class \mathcal{F} and for each $\varepsilon > 0$ (S) holds with a number $M(\varepsilon, \tau) > 0$ such that

(451)
$$\sum_{k=0}^{\infty} \|\Gamma(t+\tau,t+\tau-\cdot) - \Gamma(t,t-\cdot)\|_{L^q[k,k+1]} \leqslant \varepsilon, \quad provided \ t \ge M(\varepsilon,\tau),$$

then there exists a unique mild solution $u(\cdot)$ of (450) and this solution belongs to the class $Q - AAP([0,\infty): X) + \mathcal{F}$.

PROOF. Since $x \in P(0)X \cap \overline{D(A(0))}$, the mapping $t \mapsto U(t,0)x, t > 0$ is continuous, exponentially decaying and satisfies $\lim_{t\to 0+} U(t,0)x = x$ [407]. The continuity of function $u(t) = U(t,0)x + \int_0^t U(t,s)f(s)ds, t \ge 0$, given by (437), can be deduced as in the proof of [338, Theorem 2.1], since any of the considered terms in the corresponding part of proof of above-mentioned result can be majorized in a similar way, by using the S^p -boundedness of function $f(\cdot)$ and the Hölder inequality. Clearly, $u(t) = \int_0^t \Gamma(t,s)f(s)ds + \int_0^t U(t,s)Q(s)f(s)ds, t \ge 0$ and, by our preassumption made, it suffices to show that the function $t \mapsto \int_0^t \Gamma(t,s)f(s)ds,$ $t \ge 0$ belongs to the class $Q - AAP([0,\infty) : X)$. Applying the Hölder inequality, the estimate (435) and the S^p -boundedness of function $f(\cdot)$, we get that there exists a finite positive constant M'' > 0 such that

$$\|u(t)\| \leq M' \|f\|_{S^p} \sum_{k=0}^{\infty} \|e^{-\omega|\cdot|}\|_{L^q[t-(k+1),t-k]}$$
$$\leq M' \|f\|_{S^p} \sum_{k=0}^{\infty} \|e^{-\omega|\cdot|}\|_{L^{\infty}[t-(k+1),t-k]}$$

$$\leq M' \|f\|_{S^p} \sum_{k=0}^{\infty} [e^{-\omega|t-k|} + e^{-\omega|t-k-1|}]$$

$$\leq M' \|f\|_{S^p} e^{\omega t} \sum_{k=0}^{\infty} [e^{-\omega k} + e^{-\omega(k+1)}] \leq M'' e^{\omega|t|}$$

for any $t \ge 0$. Fix a number $\varepsilon > 0$. By definition, (S) holds with a number $M(\varepsilon, \tau) > 0$ satisfying (444). Keeping in mind the estimate (451), the final conclusion follows from the computation

$$\begin{split} \|u(t+\tau) - u(t)\| &\leq \int_{0}^{t} \|\Gamma(t+\tau, t+\tau-s) - \Gamma(t, t-s)\| \|f(t-s)\| ds \\ &+ \int_{t}^{t+\tau} \|\Gamma(t+\tau, t+\tau-s)\| \|f(t+\tau-s) - f(t-s)\| ds \\ &\leq \|f\|_{S^{p}} \sum_{k=0}^{\lfloor t \rfloor} \|\Gamma(t+\tau, t+\tau-\cdot) - \Gamma(t, t-\cdot)\|_{L^{q}[k,k+1]} + 2\|f\|_{S^{p}} \sum_{k=\lfloor t \rfloor}^{\lceil t+\tau \rceil} \|e^{-\omega|\cdot|}\|_{L^{q}[k,k+1]} \\ &\leq \|f\|_{S^{p}} \sum_{k=0}^{\infty} \|\Gamma(t+\tau, t+\tau-\cdot) - \Gamma(t, t-\cdot)\|_{L^{q}[k,k+1]} + 2\|f\|_{S^{p}} \sum_{k=\lfloor t \rfloor}^{\infty} \|e^{-\omega|\cdot|}\|_{L^{q}[k,k+1]} \end{split}$$

and the obvious equality $\lim_{t\to\infty}\sum_{k=\lfloor t\rfloor}^{\infty} \|e^{-\omega|\cdot|}\|_{L^q[k,k+1]} = 0.$

REMARK 3.11.38. It can be simply shown that (451) implies

(452)
$$\int_{0}^{+\infty} \|\Gamma(t+\tau,t+\tau-s) - \Gamma(t,t-s)\| ds \leqslant \varepsilon, \quad t \ge M(\varepsilon,\tau).$$

If we assume that $f \in L^{\infty}([0,\infty) : X)$ in place of $f \in L^p_S([0,\infty) : X)$, then the validity of (452) in place of (451) implies that $u(\cdot)$ is q-aap..

Concerning the abstract Cauchy problem (449), we have the following result:

THEOREM 3.11.39. Let $I = \mathbb{R}$, 1/p + 1/q = 1 and $f \in S^pQ - AAP(\mathbb{R} : X)$. If for each $\varepsilon > 0$ (S) holds with a number $M(\varepsilon, \tau) > 0$ such that

(453)
$$\sum_{k\in\mathbb{Z}} \|\Gamma(t+\tau,t+\tau-\cdot)-\Gamma(t,t-\cdot)\|_{L^q[k,k+1]} \leqslant \varepsilon, \text{ provided } t\in\mathbb{R} \text{ and } |t| \ge M(\varepsilon,\tau),$$

then there exists a unique mild solution $u(\cdot)$ of (449) and this solution is q-aap...

PROOF. As in the proof of Theorem 3.11.37, we can deduce that the function $u(t) = \int_{-\infty}^{+\infty} \Gamma(t,s) f(s) ds, t \in \mathbb{R}$, defined by (436), is bounded. The continuity of $u(\cdot)$ can be shown following the lines of proof of [**338**, Theorem 2.1]. Assume now that $\varepsilon > 0$ is a given number. Then (S) holds with a number $M(\varepsilon, \tau) > 0$ satisfying (444). It is clear that, for every $t \in \mathbb{R}$, we have:

$$\|u(t+\tau) - u(t)\| \leq \int_{-\infty}^{t} e^{-\omega|s|} \|f(t+\tau-s) - f(t-s)\|ds + \int_{t}^{\infty} e^{-\omega|s|} \|f(t+\tau-s) - f(t-s)\|ds$$

$$\begin{split} &+ \int_{-\infty}^{\infty} \|\Gamma(t+\tau,t+\tau-s) - \Gamma(t,t-s)\| \|f(t-s)\| ds \\ &\leqslant \int_{-\infty}^{t} e^{-\omega|s|} \|f(t+\tau-s) - f(t-s)\| ds \\ &+ \int_{t}^{\infty} e^{-\omega|s|} \|f(t+\tau-s) - f(t-s)\| ds \\ &+ \|f\|_{S^{p}} \sum_{k \in \mathbb{Z}} \|\Gamma(t+\tau,t+\tau-\cdot) - \Gamma(t,t-\cdot)\|_{L^{q}[k,k+1]}. \end{split}$$

Keeping in mind (453), we get that:

$$\begin{aligned} \|u(t+\tau) - u(t)\| &\leq \int_{-\infty}^{t} e^{-\omega|s|} \|f(t+\tau-s) - f(t-s)\| ds \\ &+ \int_{t}^{\infty} e^{-\omega|s|} \|f(t+\tau-s) - f(t-s)\| ds + \|f\|_{S^{p}} \varepsilon, \end{aligned}$$

provided $|t| \ge M(\varepsilon, \tau)$. By the proof of Proposition 3.11.36, it follows the existence of a finite real number $M_1(\varepsilon, \tau) > 0$ such that

$$\int_{-\infty}^{t} e^{-\omega|s|} \|f(t+\tau-s) - f(t-s)\|ds < \varepsilon, \text{ provided } |t| \ge M_1(\varepsilon,\tau).$$

On the other hand, there exists an integer $k_0(\varepsilon) \in \mathbb{N}$ such that $e^{-\omega k} \leq \varepsilon$ for all $k \geq k_0(\varepsilon)$. Let $|t| \geq 2M(\varepsilon, \tau) + 1 + k_0(\varepsilon)$. For the second addend, we can use the following calculus involving the Hölder inequality, after dividing the interval of integration $(-\infty, 0]$ into two subintervals $(-\infty, -M(\varepsilon, \tau))$ and $[-M(\varepsilon, \tau), 0]$:

$$\begin{split} &\int_{t}^{\infty} e^{-\omega|s|} \|f(t+\tau-s) - f(t-s)\| ds \\ &\leqslant \varepsilon \sum_{k=0}^{\infty} \|e^{-\omega|\cdot|}\|_{L^{q}[t+M(\varepsilon,\tau)+k,t+M(\varepsilon,\tau)+k+1]} \\ &\leqslant + \int_{-M(\varepsilon,\tau)}^{0} e^{-\omega|s|} \|f(t+\tau-s) - f(t-s)\| ds \\ &\leqslant \varepsilon e^{-\omega|t+M(\varepsilon,\tau)|} \sum_{k=0}^{\infty} [e^{-\omega k} + e^{-\omega(k+1)}] + 2\|f\|_{S^{p}} \sum_{k=0}^{\lfloor M(\varepsilon,\tau) \rfloor} \left\|e^{-\omega|\cdot|}\right\|_{L^{q}[t+k,t+k+1]} \\ &\leqslant \varepsilon e^{-\omega k_{0}(\varepsilon)} \sum_{k=0}^{\infty} [e^{-\omega k} + e^{-\omega(k+1)}] + 2\|f\|_{S^{p}} \sum_{k=0}^{\lfloor M(\varepsilon,\tau) \rfloor} [e^{-\omega|t-(k+1)|} + e^{-\omega|t-k|}] \\ &\leqslant \varepsilon e^{-\omega k_{0}(\varepsilon)} \sum_{k=0}^{\infty} [e^{-\omega k} + e^{-\omega(k+1)}] + 2\|f\|_{S^{p}} (1+M(\varepsilon,\tau))e^{-\omega(M(\varepsilon,\tau)+k_{0}(\varepsilon))} \\ &\leqslant \varepsilon e^{-\omega k_{0}(\varepsilon)} \sum_{k=0}^{\infty} [e^{-\omega k} + e^{-\omega(k+1)}] + 2\|f\|_{S^{p}} (1+\omega)e^{-\omega k_{0}(\varepsilon)}, \end{split}$$

which completes the proof.

REMARK 3.11.40. The condition (453) implies

(454)
$$\int_{-\infty}^{+\infty} \|\Gamma(t+\tau,t+\tau-s) - \Gamma(t,t-s)\| ds \leqslant \varepsilon, \text{ provided } t \in \mathbb{R} \text{ and } |t| \ge M(\varepsilon,\tau).$$

If we assume that $f \in L^{\infty}(\mathbb{R} : X)$ in place of $f \in L_S^p(\mathbb{R} : X)$, then the validity of (454) in place of (453) implies that $u(\cdot)$ is q-aap..

REMARK 3.11.41. It is worth noting that Theorem 3.11.37 and Remark 3.11.38, as well as Theorem 3.11.39 and Remark 3.11.40, continue to hold in the case that the operator family $(A(t))_{t \in \mathbb{R}}$ generates an exponentially stable evolution family $(U(t,s))_{t \geq s}$ in the sense of [145, Definition 3.1]; in this case, the condition (H1) need not be satisfied and the condition (H2) holds with P(t) = I and Q(t) = 0, $t \in \mathbb{R}$; $\Gamma(t,s) \equiv U(t,s)$.

3.11.7. Semilinear Cauchy problems. In this subsection, we consider the existence and uniqueness of q-aap. solutions of the abstract Cauchy problems (438) and (439). We first state the following result about the abstract Cauchy problem (439):

THEOREM 3.11.42. Let $I = [0, \infty)$, the evolution system $U(\cdot, \cdot)$ be exponentially stable, let $x \in P(0)X \cap \overline{D(A(0))}$ and let $F \in Q - AAP([0,\infty) \times X : X)$. Suppose that for each $\varepsilon > 0$ (S) holds with a number $M(\varepsilon, \tau) > 0$ satisfying (452). If there exists a finite number $L \in (0, \omega/M')$ such that (448) holds, then there exists a unique mild solution $u(\cdot)$ of (439) belonging to the class $Q - AAP([0,\infty) : X)$.

PROOF. As before, the mapping $t \mapsto U(t,0)x$, t > 0 is continuous, exponentially decaying and satisfies $\lim_{t\to 0+} U(t,0)x = x$. Let $\mathcal{P}: Q - AAP([0,\infty):X) \to Q - AAP([0,\infty):X)$ be defined through

$$\mathcal{P}f(t) := U(t,0)x + \int_0^t U(t,s)F(s,f(s))ds, \quad t \ge 0.$$

We will first show that the mapping \mathcal{P} is well defined. Since $C_0([0,\infty):X) + Q - AAP([0,\infty):X) = Q - AAP([0,\infty):X)$ and $Q - AAP([0,\infty):X)$ is a complete metric space by Theorem 3.11.21(iii), it suffices to show that the mapping $t \mapsto \int_0^t U(t,s)F(s,f(s))ds, t \ge 0$ belongs to the class $Q - AAP([0,\infty):X)$. Due to Theorem 3.11.28, the function $F(\cdot,f(\cdot))$ is in class $Q - AAP([0,\infty):X)$; since Q(t) = 0 for all $t \in \mathbb{R}$, the prescribed assumption on the condition (452) yields that Theorem 3.11.37 (see also Remark 3.11.38) can be applied, showing that the mapping $t \mapsto \int_0^t U(t,s)F(s,f(s))ds, t \ge 0$ belongs to the class $Q - AAP([0,\infty):X)$; Furthermore, the condition $L \in (0, \omega/M')$ implies after a simple calculation involving (435) and (448) that $\mathcal{P}(\cdot)$ is a contraction, so that the final conclusion simply follows by applying the Banach contraction principle.

We can similarly prove the following result on the abstract Cauchy problem (438):

THEOREM 3.11.43. Let $I = \mathbb{R}$, the evolution system $U(\cdot, \cdot)$ be exponentially stable and $F \in Q - AAP(\mathbb{R} \times X : X)$. Suppose that for each $\varepsilon > 0$ (S) holds with a number $M(\varepsilon,\tau) > 0$ satisfying (454). If there exists a finite number $L \in (0, \omega/2M')$ such that (448) holds, then there exists a unique mild solution $u(\cdot)$ of (438) belonging to the class $Q - AAP(\mathbb{R} : X)$.

As mentioned before, the semilinear Cauchy problems with S^p -qaap. forcing term $F(\cdot, \cdot)$ cannot be so easily considered because the range of function $x(\cdot)$, appearing in the formulations of Theorem 3.11.31 and Theorem 3.11.32, need not be relatively compact.

We close the section by providing an illustrative example.

EXAMPLE 3.11.44. Let $X := L^2[0,\pi]$ and Δ denote the Dirichlet Laplacian in X, acting with the domain $H^2[0,\pi] \cap H^1_0[0,\pi]$; then we know that Δ generates a strongly continuous semigroup $(T(t))_{t\geq 0}$ on X, satisfying the estimate $||T(t)|| \leq e^{-t}, t \geq 0$. Of concern is the following problem

(455)
$$u_t(t,x) = u_{xx}(t,x) + q(t,x)u(t,x) + f(t,x), \quad t \ge 0, \ x \in [0,\pi];$$

(456)
$$u(0) = u(\pi) = 0, \ u(0, x) = u_0(x) \in X$$

where $q: \mathbb{R} \times [0, \pi] \to \mathbb{R}$ is a jointly continuous function satisfying that $q(t, x) \leq -\gamma_0$, $(t, x) \in \mathbb{R} \times [0, \pi]$, for some number $\gamma_0 > 0$. Define

$$A(t)\varphi := \Delta \varphi + q(t,\cdot)\varphi, \quad \varphi \in D(A(t)) := D(\Delta) = H^2[0,\pi] \cap H^1_0[0,\pi], \qquad t \in \mathbb{R}.$$

Then $(A(t))_{t\in\mathbb{R}}$ generates an exponentially stable evolution family $(U(t,s))_{t\geq s}$ in the sense of [145, Definition 3.1], which is given by

$$U(t,s)\varphi := T(t-s)e^{\int_s^t q(r,\cdot)dr}\varphi, \quad t \ge s.$$

It is clear that we can rewrite the initial value problem (455)-(456) in the following form:

$$u'(t) = A(t)u(t) + f(t), \quad t \ge 0; \ u(0) = u_0.$$

Hence, Theorem 3.11.37, resp. Theorem 3.11.39 (Theorem 3.11.42, resp. Theorem 3.11.43), can be applied provided that for each $\varepsilon > 0$ (S) holds with a number $M(\varepsilon, \tau) > 0$ such that the following condition holds:

$$(457) \quad \sum_{k=0}^{\infty} \left\| e^{-|\cdot|} \sup_{x \in [0,\pi]} \left| e^{\int_{t+\tau-\cdot}^{t+\tau} q(r,x)dr} - e^{\int_{t-\cdot}^{t} q(r,x)dr} \right| \right\|_{L^q[k,k+1]} < \varepsilon, \quad t \ge M(\varepsilon,\tau),$$

resp.

$$(458) \sum_{k\in\mathbb{Z}} \left\| e^{-|\cdot|} \sup_{x\in[0,\pi]} \left| e^{\int_{t+\tau-\cdot}^{t+\tau} q(r,x)dr} - e^{\int_{t-\cdot}^{t} q(r,x)dr} \right| \right\|_{L^q[k,k+1]} < \varepsilon, \quad |t| \ge M(\varepsilon,\tau).$$

The conditions (457) and (458) hold for a wide class of functions $q(\cdot, \cdot)$ and we will prove here that this condition particularly holds for the function $q(t, x) := -\gamma_0 - 3t^2 - f(x), t \ge 0, x \in [0, \pi]$, where $f: [0, \infty) \to [0, \infty)$ is a continuous function (see also [338, Example 3.1], where we have analyzed the same choice). In our concrete situation, we have

$$\sup_{x \in [0,\pi]} \left| e^{\int_{t+\tau-s}^{t+\tau} q(r,x)dr} - e^{\int_{t-s}^{t} q(r,x)dr} \right| \\ \leqslant \text{Const.} \cdot e^{-s^3} |e^{3s(t+\tau)(s-t-\tau)} - e^{3st(s-t)}|, \ t, s, \tau \ge 0.$$

Using this estimate and the Lagrange mean value theorem, it readily follows that:

$$\begin{split} & \left\| e^{-|\cdot|} \sup_{x \in [0,\pi]} \left| e^{\int_{t+\tau-\cdot}^{t+\tau-\cdot} q(r,x)dr} - e^{\int_{t-\cdot}^{t} q(r,x)dr} \right| \right\|_{L^{\infty}[k,k+1]} \\ \leqslant \text{Const.} \cdot |\tau| \sup_{s \in [k,k+1]} \left[e^{3st(s-t)} + e^{3s(t+\tau)(s-t-\tau)} \right] \cdot \left[3(k+1)^2 + 6(k+1)(t+\tau) \right] \\ \leqslant \text{Const.} \cdot |\tau| \left[e^{3kt(k-t)} + e^{3(k+1)t(k+1-t)} + e^{3k(t+\tau)(k-t-\tau)} + e^{3(k+1)(t+\tau)(k+1-t-\tau)} \right] \\ \cdot \left[3(k+1)^2 + 6(k+1)(t+\tau) \right], \quad t,\tau \ge 0, \ k \in \mathbb{N}_0. \end{split}$$

Let 3/4 < c < 1. Then $3ts(s - ct) \leq 3s^3/4c$ for all $t, s \geq 0$ and therefore we can continue the calculation as follows:

$$\begin{split} &\leqslant \operatorname{Const.} \cdot |\tau| [e^{3kt(k-ct)} e^{-3ct^2} + e^{3(k+1)t(k+1-ct)} e^{-3ct^2} \\ &+ e^{3k(t+\tau)(k-c(t+\tau))} e^{-3c(t+\tau)^2} + e^{3(k+1)(t+\tau)(k+1-c(t+\tau))} e^{-3c(t+\tau)^2}] \\ &\cdot [3(k+1)^2 + 6(k+1)(t+\tau)] \\ &\leqslant \operatorname{Const.} \cdot |\tau| [e^{3k^3/4c} e^{-3ct^2} + e^{3(k+1)^3/4c} e^{-3ct^2} \\ &+ e^{3k^3/4c} e^{-3c(t+\tau)^2} + e^{3(k+1)^3/4c} e^{-3c(t+\tau)^2}] \\ &\cdot [3(k+1)^2 + 6(k+1)(t+\tau)] \\ &\leqslant \operatorname{Const.} \cdot |\tau| e^{3(k+1)^3/4c} e^{-3ct^2} [3(k+1)^2 + 6(k+1)(t+\tau)], \quad t,\tau \geqslant 0 \end{split}$$

Since 3/4c < 1, the series in (457) is convergent with $q = \infty$ and has a sum which does not exceed Const. $|\tau|e^{-3ct^2}(1+t+\tau), t, \tau \ge 0$. At the end, it suffices to observe that for each $\varepsilon > 0$ and $\tau \ge 0$ there exists a finite number $M(\varepsilon, \tau) > 0$ such that $|\tau|e^{-3ct^2}(1+t+\tau) < \varepsilon$ for any $t \ge M(\varepsilon, \tau)$. This shows that Theorem 3.11.37 can be applied with any exponent $p \in [1, \infty)$.

3.12. Almost periodic and asymptotically almost periodic type solutions with variable exponents $L^{p(x)}$

As in the previous one, in this section we will assume that $(X, \|\cdot\|)$ and $(Y, \|\cdot\|_Y)$ are two non-trivial complex Banach spaces.

We will use the following notion of Caputo fractional derivatives of order $\gamma \in (0,1)$. If $u: [0,\infty) \to X$ satisfies, for every T > 0, $u \in C((0,T]:X)$, $u(\cdot) - u(0) \in L^1((0,T):X)$ and $g_{1-\gamma}*(u(\cdot)-u(0)) \in W^{1,1}((0,T):X)$, then we define its Caputo fractional derivative by,

$$\mathbf{D}_{t}^{\gamma}u(t) = \frac{d}{dt}[g_{1-\gamma} * (u(\cdot) - u(0))](t), \quad t \in (0,T].$$

The Weyl fractional derivative W^{α}_{+} of order $\alpha > 0$ is sometimes also called the Weyl-Liouville fractional derivative of order α . This fractional derivative can be applied to a wide class of functions containing the class of rapidly decreasing functions. In this section, we will use the following definition of the Weyl–Liouville fractional derivative $D^{\gamma}_{t,+}u(t)$ of order $\gamma \in (0,1)$: The Weyl–Liouville fractional derivative $D^{\gamma}_{t,+}u(t)$ of order γ is defined for those continuous functions $u: \mathbb{R} \to$ X such that $t \mapsto \int_{-\infty}^{t} g_{1-\gamma}(t-s)u(s)ds, t \in \mathbb{R}$ is a well-defined continuously differentiable mapping, by

$$D_{t,+}^{\gamma}u(t) := \frac{d}{dt} \int_{-\infty}^{t} g_{1-\gamma}(t-s)u(s)ds, \quad t \in \mathbb{R}.$$

Set $\mathbf{D}_t^1 u(t) := (d/dt)u(t)$ and $D_{t+}^1 u(t) := -(d/dt)u(t)$.

3.12.1. Lebesgue spaces with variable exponents $L^{p(x)}$. Let $\emptyset \neq \Omega \subseteq \mathbb{R}$ be a nonempty subset and let $M(\Omega : X)$ stand for we the collection of all measurable functions $f: \Omega \to X$; $M(\Omega) := M(\Omega : \mathbb{R})$. Furthermore, $\mathcal{P}(\Omega)$ denotes the vector space of all Lebesgue measurable functions $p: \Omega \to [1, \infty]$. For any $p \in \mathcal{P}(\Omega)$ and $f \in M(\Omega : X)$, set

$$\varphi_{p(x)}(t) := \begin{cases} t^{p(x)}, & t \ge 0, \quad 1 \le p(x) < \infty, \\ 0, & 0 \le t \le 1, \quad p(x) = \infty, \\ \infty, & t > 1, \quad p(x) = \infty \end{cases}$$

and

$$\rho(f) := \int_{\Omega} \varphi_{p(x)}(\|f(x)\|) dx.$$

We define the Lebesgue space $L^{p(x)}(\Omega : X)$ with variable exponent as follows,

$$L^{p(x)}(\Omega:X) := \left\{ f \in M(\Omega:X) : \lim_{\lambda \to 0+} \rho(\lambda f) = 0 \right\}$$

equivalently

 $L^{p(x)}(\Omega:X) = \{f \in M(\Omega:X): \text{ there exists } \lambda > 0 \text{ such that } \rho(\lambda f) < \infty\};$ see, e.g., [152, p. 73].

For every $u \in L^{p(x)}(\Omega : X)$, we introduce the Luxemburg norm of $u(\cdot)$ in the following manner:

$$||u||_{p(x)} := ||u||_{L^{p(x)}(\Omega;X)} := \inf\{\lambda > 0 : \rho(f/\lambda) \leq 1\}.$$

Equipped with the above norm, the space $L^{p(x)}(\Omega : X)$ becomes a Banach space (see e.g. [152, Theorem 3.2.7] for scalar-valued case), coinciding with the usual Lebesgue space $L^{p}(\Omega : X)$ in the case that $p(x) = p \ge 1$ is a constant function. For any $p \in M(\Omega)$, we set

$$p^- := \operatorname{essinf}_{x \in \Omega} p(x)$$
 and $p^+ := \operatorname{esssup}_{x \in \Omega} p(x)$

Define

$$C_{+}(\Omega) := \{ p \in M(\Omega) : 1 < p^{-} \leqslant p(x) \leqslant p^{+} < \infty \text{ for a.e. } x \in \Omega \}$$

and

$$D_+(\Omega) := \{ p \in M(\Omega) : 1 \leqslant p^- \leqslant p(x) \leqslant p^+ < \infty \text{ for a.e. } x \in \Omega \}.$$

For $p \in D_+([0,1])$, the space $L^{p(x)}(\Omega : X)$ behaves nicely, with almost all fundamental properties of the Lesbesgue space with constant exponent $L^p(\Omega : X)$ being retained; in this case, we know that

 $L^{p(x)}(\Omega:X) = \{ f \in M(\Omega:X) : \text{ for all } \lambda > 0 \text{ we have } \rho(\lambda f) < \infty \}.$

Furthermore, if $p \in C_+(\Omega)$, then $L^{p(x)}(\Omega : X)$ is uniformly convex and thus reflexive [176].

We will use the following lemma (see, e.g., [152, Lemma 3.2.20, (3.2.22); Corollary 3.3.4; p. 77] for scalar-valued case):

LEMMA 3.12.1. (i) Let $p, q, r \in \mathcal{P}(\Omega)$ such that $\frac{1}{q(x)} = \frac{1}{p(x)} + \frac{1}{r(x)}, \quad x \in \Omega.$

Then, for every $u \in L^{p(x)}(\Omega : X)$ and $v \in L^{r(x)}(\Omega)$, we have $uv \in L^{q(x)}(\Omega : X)$ and

$$||uv||_{q(x)} \leq 2||u||_{p(x)}||v||_{r(x)}.$$

- (ii) Let Ω be of a finite Lebesgue's measure and let $p, q \in \mathcal{P}(\Omega)$ such $q \leq p$ a.e. on Ω . Then $L^{p(x)}(\Omega : X)$ is continuously embedded in $L^{q(x)}(\Omega : X)$.
- (iii) Let $f \in L^{p(x)}(\Omega:X)$, $g \in M(\Omega:X)$ and $0 \leq ||g|| \leq ||f||$ a.e. on Ω . Then $g \in L^{p(x)}(\Omega:X)$ and $||g||_{p(x)} \leq ||f||_{p(x)}$.

For additional details upon Lebesgue spaces with variable exponents $L^{p(x)}$, we refer the reader to the following sources: [150, 152, 176] and [436].

3.12.2. Generalized almost periodic and generalized asymptotically almost periodic functions in Lebesgue spaces with variable exponents $L^{p(x)}$. The following notion of Stepanov p(x)-boundedness differs from the one introduced by T. Diagana and M. Zitane in [150, Definition 3.10] and [151, Definition 4.5], where the authors have used the condition $p \in C_+(\mathbb{R})$:

DEFINITION 3.12.2. Let $p \in \mathcal{P}([0,1])$ and let $I = \mathbb{R}$ or $I = [0,\infty)$. A function $f \in M(I:X)$ is said to be Stepanov p(x)-bounded (or $S^{p(x)}$ -bounded), if $f(\cdot+t) \in L^{p(x)}([0,1]:X)$ for all $t \in I$, and $\sup_{t \in I} \|f(\cdot+t)\|_{p(x)} < \infty$, that is,

$$\|f\|_{S^{p(x)}} := \sup_{t \in I} \inf \left\{ \lambda > 0 : \int_0^1 \varphi_{p(x)} \Big(\frac{\|f(x+t)\|}{\lambda} \Big) dx \leqslant 1 \right\} < \infty.$$

The collection of such functions will be denoted by $L_S^{p(x)}(I:X)$.

From Definition 3.12.2 it follows that the space $L_S^{p(x)}(I : X)$ is translation invariant in the sense that, for every $f \in L_S^{p(x)}(I : X)$ and $\tau \in I$, we have $f(\cdot + \tau) \in$ $L_S^{p(x)}(I : X)$. This is not the case with the notion introduced by T. Diagana and M. Zitane [150,151], since there the space $L_S^{p(x)}(I : X)$ may or may not be translation invariant depending on p(x). Furthermore, let us note that the notion introduced in these papers is meaningful even in the case that $p \in \mathcal{P}(\mathbb{R})$. We introduce the concept of (asymptotic) $S^{p(x)}$ -almost periodicity as follows:

DEFINITION 3.12.3. (i) Let $p \in \mathcal{P}([0,1])$ and let $I = \mathbb{R}$ or $I = [0,\infty)$. A function $f \in L_S^{p(x)}(I:X)$ is said to be Stepanov p(x)-almost periodic (or Stepanov p(x)-a.p.), if the function $\hat{f}: I \to L^{p(x)}([0,1]:X)$ is almost periodic. The collection of such functions will be denoted by $APS^{p(x)}(I)$: X).

(ii) Let $p \in \mathcal{P}([0,1])$ and let $I = [0,\infty)$. A function $f \in L_S^{p(x)}(I:X)$ is said to be asymptotically Stepanov p(x)-almost periodic (or asymptotically Stepanov p(x)-a.p.), if the function $\hat{f}: I \to L^{p(x)}([0,1]:X)$ is asymptotically almost periodic. The collection of such functions will be denoted by $AAPS^{p(x)}(I:X)$. The abbreviation $S_0^{p(x)}([0,\infty):X)$ will be used to denote the set of all functions $q \in L_S^{p(x)}([0,\infty):X)$ such that $\hat{q} \in C_0([0,\infty): L^{p(x)}([0,1]:X)).$

As in the case of Stepanov p(x)-boundedness, the space $APS^{p(x)}(I : X)$ is translation invariant in the sense that, for every $f \in APS^{p(x)}(I:X)$ and $\tau \in I$. we have $f(\cdot + \tau) \in APS^{p(x)}(I : X)$. A similar statement holds for the space $AAPS^{p(x)}([0,\infty) : X)$. It is clear that the notions of (asymptotic) Stepanov p(x)-boundedness and (asymptotic) Stepanov p(x)-almost periodicity are equivalent with those introduced in the previous section, provided that $p(x) \equiv p \ge 1$ is a constant function.

Equipped with the norm $\|\cdot\|_{S^{p(x)}}$, the space $L_S^{p(x)}(I:X)$ consisting of all S^p bounded functions is a Banach space, which is continuously embedded in $L^1_S(I:X)$, for any $p \in \mathcal{P}([0,1])$. Furthermore, it can be easily shown that $APS^{p(x)}(I:X)$ $(AAPS^{p(x)}(I : X) \text{ with } I = [0, \infty))$ is a closed subspace of $L_S^{p(x)}(I : X)$ and therefore is Banach space itself, for any $p \in \mathcal{P}([0, 1])$.

If $p \in \mathcal{P}([0,1])$, then Lemma 3.12.1(ii) yields $L^{p(x)}([0,1]:X) \hookrightarrow L^1([0,1]:X)$, where the symbol \hookrightarrow stands for a "continuous embedding", so that $L_S^{p(x)}(I:X) \hookrightarrow$ $L^1_{\mathcal{S}}(I:X)$, as well.

We have

PROPOSITION 3.12.4. Suppose $p \in \mathcal{P}([0,1])$. Then the following continuous embedding hold,

(i) $L_S^{p(x)}(I:X) \hookrightarrow L_S^1(I:X)$; and (ii) $APS^{p(x)}(I:X) \hookrightarrow APS^1(I:X)$ and $AAPS^{p(x)}([0,\infty):X) \hookrightarrow AAPS^1([0,\infty):X).$

Similarly,

PROPOSITION 3.12.5. Suppose $p \in D_+([0,1])$ and $1 \leq p^- \leq p(x) \leq p^+ < \infty$ for a.e. $x \in [0,1]$. Then the following continuous embedding hold,

- (i) $L_S^{p^+}(I:X) \hookrightarrow L_S^{p(x)}(I:X) \hookrightarrow L_S^{p^-}(I:X);$ and (ii) $APS^{p^+}(I:X) \hookrightarrow APS^{p(x)}(I:X) \hookrightarrow APS^{p^-}(I:X)$ and $AAPS^{p^+}([0,\infty):$ $X) \hookrightarrow AAPS^{p(x)}([0,\infty):X) \hookrightarrow AAPS^{p^{-}}([0,\infty):X).$

Now we will prove that any almost periodic function is $S^{p(x)}$ -almost periodic, for any $p \in \mathcal{P}([0, 1])$.

PROPOSITION 3.12.6. Let $p \in \mathcal{P}([0,1])$, and let $f: I \to X$ be almost periodic. Then $f(\cdot)$ is $S^{p(x)}$ -almost periodic.

PROOF. To prove that $f(\cdot)$ is $S^{p(x)}$ -bounded and $||f||_{L^{p(x)}_{c}} \leq ||f||_{\infty}$, it suffices to show that, for every $t \in \mathbb{R}$, we have:

(459)
$$[||f||_{\infty}, \infty) \subseteq \left\{ \lambda > 0 : \int_0^1 \varphi_{p(x)} \Big(\frac{||f(x+t)||}{\lambda} \Big) dx \leqslant 1 \right\}.$$

For $\lambda \ge \|f\|_{\infty}$, we have $\|f(x+t)\|/\lambda \le 1$, $t \in I$. It can be simply perceived that, in this case.

$$\varphi_{p(x)}\left(\frac{\|f(x+t)\|}{\lambda}\right) \leqslant 1, \quad t \in I,$$

so that the integrand does not exceed 1: as a matter of fact, by definition of $\varphi_{p(x)}(\cdot)$, we only need to observe that, for every $x \in [0,1]$ with $p(x) < \infty$, we have $(\|f(t+x)\|/\lambda)^{p(x)} \leq 1^{p(x)} = 1, t \in I$. Hence, (459) holds. Using the uniform continuity of $f(\cdot)$ and a similar argumentation, we can show that the function $\hat{f}: I \to L^{p(x)}([0,1]:X)$ is uniform continuous. For direct proof of almost periodicity of function $\hat{f}: I \to L^{p(x)}([0,1]:X)$, we can argue as follows. For $\varepsilon > 0$ given as above, there is a finite number l > 0 such that any subinterval I' of I of length l contains a number $\tau \in I'$ such that $||f(t+\tau) - f(t)|| \leq \varepsilon$. $t \in I$. It suffices to observe that, for this $\varepsilon > 0$, we can choose the same length l > 0 and the same ε -almost period τ from I' ensuring the validity of inequality $\|\hat{f}(t+\tau+\cdot)-\hat{f}(t+\cdot)\|_{L^{p(x)}([0,1];X)} \leq \varepsilon, t \in I$: in order to see that the last inequality holds true, we only need to prove that, for every $t \in I$, we have

$$[\varepsilon,\infty) \subseteq \bigg\{\lambda > 0 : \int_0^1 \varphi_{p(x)} \Big(\frac{\|f(t+\tau+x) - f(t+x)\|}{\lambda} \Big) dx \leqslant 1 \bigg\}.$$

Indeed, if $\lambda \ge \varepsilon$, then $\|f(t+\tau+x) - f(t+x)\|/\lambda \le 1$, $t \in I$ and the integrand cannot exceed 1: this simply follows from definition of $\varphi_{p(x)}(\cdot)$ and observation that, for every $x \in [0,1]$ with $p(x) < \infty$, we have $(\|f(t+\tau+x) - f(t+x)\|/\lambda)^{p(x)} \le 1^{p(x)} = 1$, $t \in I$. The proof of the proposition is thereby complete. \square

We can similarly prove the following proposition:

PROPOSITION 3.12.7. Let $p \in \mathcal{P}([0,1])$, and let $f: [0,\infty) \to X$ be asymptotically almost periodic. Then $f(\cdot)$ is asymptotically $S^{p(x)}$ -almost periodic.

Taking into account Proposition 3.12.4(ii) and the method employed in the proof of Proposition 3.12.6, we can state the following:

PROPOSITION 3.12.8. Assume that $p \in \mathcal{P}([0,1])$ and $f \in L_S^{p(x)}(I:X)$. Then the following holds:

- $\begin{array}{ll} (\mathrm{i}) & L^{\infty}(I:X) \hookrightarrow L^{p(x)}_{S}(I:X) \hookrightarrow L^{1}_{S}(I:X). \\ (\mathrm{i}) & AP(I:X) \hookrightarrow APS^{p(x)}(I:X) \hookrightarrow APS^{1}(I:X) \text{ and } AAP([0,\infty):X) \hookrightarrow \end{array}$ $AAPS^{p(x)}([0,\infty):X) \hookrightarrow AAPS^1([0,\infty):X).$
- (iii) The continuity (uniform continuity) of $f(\cdot)$ implies continuity (uniform continuity) of $\hat{f}(\cdot)$.

In general case, we have the following:

PROPOSITION 3.12.9. Assume that $p, q \in \mathcal{P}([0,1])$ and $p \leq q$ a.e. on [0,1]. Then we have:

(i) $L_S^{q(x)}(I:X) \hookrightarrow L_S^{p(x)}(I:X).$ (ii) $APS^{q(x)}(I:X) \hookrightarrow APS^{p(x)}(I:X)$ and $AAPS^{q(x)}([0,\infty):X) \hookrightarrow AAPS^{p(x)}([0,\infty):X).$ (iii) If $p \in D_+([0,1])$, then $L^{\infty}(I-X) \cong APS^{p(x)}(I-X) = L^{\infty}(I-X) \cong APS^{1}(I-X).$

$$L^{\infty}(I:X) \cap APS^{p(x)}(I:X) = L^{\infty}(I:X) \cap APS^{1}(I:X)$$

and

$$L^{\infty}([0,\infty):X) \cap AAPS^{p(x)}([0,\infty):X) = L^{\infty}([0,\infty):X) \cap AAPS^{1}([0,\infty):X).$$

PROOF. We will prove only (iii) for almost periodicity. Keeping in mind Proposition 3.12.5(ii), it suffices to assume that $p(x) \equiv p > 1$. Then, clearly, $L^{\infty}(I:X) \cap APS^{p}(I:X) \subseteq L^{\infty}(I:X) \cap APS^{1}(I:X)$ and it remains to be proved the opposite inclusion. So, let $f \in L^{\infty}(I:X) \cap APS^{1}(I:X)$. The required conclusion is a consequence of elementary definitions and following simple calculation, which is valid for any $t, \tau \in \mathbb{R}$:

$$\begin{split} \left[\int_{t}^{t+1} \|f(\tau+s) - f(s)\|^{p} ds \right]^{1/p} \\ &\leqslant \left[\int_{t}^{t+1} (2\|f\|_{\infty})^{p-1} \|f(\tau+s) - f(s)\| ds \right]^{1/p} \\ &= (2\|f\|_{\infty})^{(p-1)/p} \left[\int_{t}^{t+1} \|f(\tau+s) - f(s)\| ds \right]^{1/p}. \quad \Box \end{split}$$

REMARK 3.12.10. It is well known that $APS^{p(x)}(I : X)$ can be strictly contained in $APS^1(I : X)$, even in the case that $p(x) \equiv p > 1$ is a constant function. For example, H. Bohr and E. Følner have proved that, for any given number p > 1, we can construct a Stepanov almost periodic function defined on the whole real axis that is not Stepanov *p*-almost periodic (see [**81**, Example, p. 70]). The same example shows that $AAPS^p([0, \infty) : X)$ can be strictly contained in $AAPS^1([0, \infty) : X)$ for p > 1 (see e.g. [**249**, Lemma 1]).

REMARK 3.12.11. Proposition 3.12.6 and Proposition 3.12.7 can be simply deduced by using Proposition 3.12.9(ii) and the equalities $AP(I:X) = APS^{\infty}(I:X) \cap C(I:X)$, $AAP([0,\infty):X) = AAPS^{\infty}([0,\infty):X) \cap C([0,\infty):X)$, which can be proved almost trivially.

Now we would like to present the following illustrative example:

EXAMPLE 3.12.12. Define sign(0) := 0. Then, for every almost periodic function $f : \mathbb{R} \to \mathbb{R}$, we have that the function $F(\cdot) := \operatorname{sign}(f(\cdot))$ is Stepanov 1-almost periodic [377]. Since $F \in L^{\infty}(\mathbb{R})$, Proposition 3.12.9(iii) yields that the function $F(\cdot)$ is Stepanov *p*-almost periodic for any $p \ge 1$, while Proposition 3.12.8(i) yields that the function $F(\cdot)$ is Stepanov p(x)-bounded for any $p \in \mathcal{P}([0,1])$. Due to Proposition 3.12.5(ii), we have $F \in APS^{p(x)}(\mathbb{R} : \mathbb{C})$ for any $p \in D_+([0,1])$. Consider now the case that $f(x) := \sin x + \sin \sqrt{2}x$, $x \in \mathbb{R}$ and $p(x) := 1 - \ln x$, $x \in [0,1]$. We will prove that $F \notin APS^{p(x)}(\mathbb{R} : \mathbb{C})$. Speaking-matter-of-factly, it is sufficient to show that, for every $\lambda \in (0, 2/e)$ and for every l > 0, we can find an interval $I \subseteq \mathbb{R}$ of length l > 0 such that, for every $\tau \in I$, there exists $t \in \mathbb{R}$ such that

$$\int_0^1 \left(\frac{1}{\lambda}\right)^{1-\ln x} \left| \operatorname{sign}[\sin(x+t+\tau) + \sin\sqrt{2}(x+t+\tau)] - \operatorname{sign}[\sin(x+t) + \sin\sqrt{2}(x+t)] \right|^{1-\ln x} dx = \infty.$$

Let $\lambda \in (0, 2/e)$ and l > 0 be given. Take arbitrarily any interval $I \subseteq \mathbb{R} \setminus \{0\}$ of length l and after that take arbitrarily any number $\tau \in I$. Since $(1/\lambda)^{1-\ln x} \ge 1/x$, $x \in [0, 1]$ and $1 - \ln x \ge 1$, $x \in [0, 1]$, a continuity argument shows that it is enough to prove the existence of a number $t \in \mathbb{R}$ such that

(460)
$$[\sin(t+\tau) + \sin\sqrt{2}(t+\tau)] \cdot [\sin t + \sin\sqrt{2}t] < 0.$$

If $\sin \tau + \sin \sqrt{2\tau} > 0$ $(\sin \tau + \sin \sqrt{2\tau} < 0)$, then we can take $t \sim 0 (t \sim 0+)$. Hence, we assume henceforward $\sin \tau + \sin \sqrt{2\tau} = 0$ and $\tau \neq 0$. There exist two possibilities:

$$au \in \frac{2\mathbb{Z}\pi}{1+\sqrt{2}} \smallsetminus \{0\} \quad \text{or} \quad au \in \frac{(2\mathbb{Z}+1)\pi}{\sqrt{2}-1}.$$

In the first case, take $t_0 = \frac{\pi}{\sqrt{2}-1}$. Then an elementary argumentation shows that $\tau + t_0 \notin \frac{2\mathbb{Z}\pi}{1+\sqrt{2}} \cup \frac{(2\mathbb{Z}+1)\pi}{\sqrt{2}-1}$ so that $\sin(t_0 + \tau) + \sin\sqrt{2}(t_0 + \tau) \neq 0$. If $\sin(t_0 + \tau) + \sin\sqrt{2}(t_0 + \tau) > 0$ ($\sin(t_0 + \tau) + \sin\sqrt{2}(t_0 + \tau) < 0$), then for t satisfying (460) we can take any number belonging to a small left/right interval around t_0 for which $\sin t + \sin\sqrt{2}t < 0$ ($\sin t + \sin\sqrt{2}t > 0$). In the second case, there exists an integer $m \in \mathbb{Z}$ such that $\tau = \frac{(2m+1)\pi}{\sqrt{2}-1}$ and we can take $t_0 = \frac{(-2m+1)\pi}{\sqrt{2}-1}$. Then $\tau + t_0 = \frac{2\pi}{\sqrt{2}-1}$ and $\sin(t_0 + \tau) + \sin\sqrt{2}(t_0 + \tau) \neq 0$, so that we can use a trick similar to that used in the first case. Let us only mention in passing that, with the notion introduced in [149], the function $F(\cdot)$ cannot be $S^{p(x)}$ -almost automorphic, as well.

The situation is quite different if we consider the case that $f(x) := \sin x, x \in \mathbb{R}$. Then $F(\cdot)$ is Stepanov p(x)-almost periodic for any $p \in \mathcal{P}([0, 1])$. Speaking-matterof-factly, it can be easily shown that the mapping $\hat{F} : \mathbb{R} \to L^{p(x)}[0, 1]$ is continuous and $\|F(t + \tau + \cdot) - F(t + \cdot)\|_{L^{p(x)}[0, 1]} = 0$ for all $t \in \mathbb{R}$ and $\tau \in 2\pi\mathbb{Z}$. This, in turn, implies the claimed statement.

Keeping in mind the proofs of Proposition 3.12.6, [149, Proposition 3.5] and [249, Lemma 1], we can clarify the following result:

PROPOSITION 3.12.13. Suppose that $p \in \mathcal{P}([0,1])$ and $f: [0,\infty) \to X$ is an asymptotically $S^{p(x)}$ -almost periodic function. Then there are uniquely determined $S^{p(x)}$ -bounded functions $g: \mathbb{R} \to X$ and $q: [0,\infty) \to X$ satisfying the following conditions:

- (i) g is $S^{p(x)}$ -almost periodic,
- (ii) \hat{q} belongs to the class $C_0([0,\infty): L^{p(x)}([0,1]:X))$,

(iii) f(t) = q(t) + q(t) for all $t \ge 0$.

Moreover, there exists an increasing sequence $(t_n)_{n\in\mathbb{N}}$ of positive reals such that $\lim_{n\to\infty} t_n = \infty$ and $g(t) = \lim_{n\to\infty} f(t+t_n)$ a.e. $t \ge 0$.

REMARK 3.12.14. The definition of an (equi-)Weyl p(x)-almost periodic function (see e.g. [293] for the case that $p(x) \equiv p \in [1, \infty)$) can be introduced as follows: Suppose $I = \mathbb{R}$ or $I = [0, \infty)$. Let $p \in \mathcal{P}(I)$ and $f(\cdot + \tau) \in L^{p(x)}(K : X)$ for any $\tau \in I$ and any compact subset K of I.

(i) It is said that the function $f(\cdot)$ is equi-Weyl-p(x)-almost periodic, $f \in e - W_{ap}^{p(x)}(I:X)$ for short, iff for each $\varepsilon > 0$ we can find two real numbers l > 0 and L > 0 such that any interval $I' \subseteq I$ of length L contains a point $\tau \in I'$ such that

$$\sup_{t\in I} [l^{(-1)/p(t)} \|f(\cdot+\tau) - f(\cdot)\|_{L^{p(x)}[t,t+l]}] \leqslant \varepsilon.$$

(ii) It is said that the function $f(\cdot)$ is Weyl-p(x)-almost periodic, $f \in W_{ap}^{p(x)}(I : X)$ for short, iff for each $\varepsilon > 0$ we can find a real number L > 0 such that any interval $I' \subseteq I$ of length L contains a point $\tau \in I'$ such that

$$\lim_{l \to \infty} \sup_{t \in I} [l^{(-1)/p(t)} \| f(\cdot + \tau) - f(\cdot) \|_{L^{p(x)}[t, t+l]}] \leqslant \varepsilon.$$

The notion of (equi-)Weyl p(x)-almost periodicity as well as the corresponding notion for Besicovitch classes of almost periodic functions will not attract our attention here. We will also skip all details concerning asymptotical p(x)-almost periodicity for Weyl and Besicovitch classes.

3.12.3. Generalized two-parameter almost periodic type functions and composition principles. Assume that $(Y, \|\cdot\|_Y)$ is a complex Banach space, as well as that $I = \mathbb{R}$ or $I = [0, \infty)$. By $C_0([0, \infty) \times Y : X)$ we denote the space consisting of all continuous functions $h: [0, \infty) \times Y \to X$ such that $\lim_{t\to\infty} h(t, y) =$ 0 uniformly for y in any compact subset of Y. A continuous function $f: I \times Y \to X$ is said to be uniformly continuous on bounded sets, uniformly for $t \in I$ iff for every $\varepsilon > 0$ and every bounded subset K of Y there exists a number $\delta_{\varepsilon,K} > 0$ such that $\|f(t,x) - f(t,y)\| \leq \varepsilon$ for all $t \in I$ and all $x, y \in K$ satisfying that $\|x - y\| \leq \delta_{\varepsilon,K}$. If $f: I \times Y \to X$, set $\hat{f}(t,y) := f(t + \cdot, y), t \geq 0, y \in Y$. We need to recall the following well-known definition (see e.g. [293] for more details):

Definition 3.12.15. Let $1 \leq p < \infty$.

- (i) A function $f: I \times Y \to X$ is said to be almost periodic iff $f(\cdot, \cdot)$ is bounded, continuous as well as for every $\varepsilon > 0$ and every compact $K \subseteq Y$ there exists $l(\varepsilon, K) > 0$ such that every subinterval $J \subseteq I$ of length $l(\varepsilon, K)$ contains a number τ with the property that $||f(t + \tau, y) - f(t, y)|| \leq \varepsilon$ for all $t \in I, y \in K$. The collection of such functions will be denoted by $AP(I \times Y : X)$.
- (ii) A function $f: [0, \infty) \times Y \to X$ is said to be asymptotically almost periodic iff it is bounded continuous and admits a decomposition f = g + q,

where $g \in AP([0,\infty) \times Y : X)$ and $q \in C_0([0,\infty) \times Y : X)$. Denote by $AAP([0,\infty) \times Y : X)$ the vector space consisting of all such functions.

The notion of (asymptotical) Stepanov p(x)-almost periodicity for the functions depending on two parameters is introduced as follows:

DEFINITION 3.12.16. Let $p \in \mathcal{P}([0, 1])$.

- (i) A function f: I × Y → X is called Stepanov p(x)-almost periodic, S^{p(x)}-almost periodic for short, iff f̂ : I × Y → L^{p(x)}([0,1] : X) is almost periodic. The vector space consisting of all such functions will be denoted by APS^{p(x)}(I × Y : X).
- (ii) A function $f: [0, \infty) \times Y \to X$ is said to be asymptotically $S^{p(x)}$ -almost periodic iff $\hat{f}: [0, \infty) \times Y \to L^{p(x)}([0, 1] : X)$ is asymptotically almost periodic. The vector space consisting of all such functions will be denoted by $AAPS^{p(x)}([0, \infty) \times Y : X)$.

The proof of following proposition is very similar to the proof of [293, Lemma 2.2.6] and therefore omitted.

PROPOSITION 3.12.17. Let $p \in \mathcal{P}([0,1])$. Suppose that $f: [0,\infty) \times Y \to X$ is an asymptotically $S^{p(x)}$ -almost periodic function. Then there are two functions $g: \mathbb{R} \times Y \to X$ and $q: [0,\infty) \times Y \to X$ satisfying that for each $y \in Y$ the functions $g(\cdot, y)$ and $q(\cdot, y)$ are Stepanov p(x)-bounded, as well as that the following holds:

- (i) $\hat{g}: \mathbb{R} \times Y \to L^{p(x)}([0,1]:X)$ is almost periodic,
- (ii) $\hat{q} \in C_0([0,\infty) \times Y : L^{p(x)}([0,1]:X)),$
- (iii) f(t,y) = g(t,y) + q(t,y) for all $t \ge 0$ and $y \in Y$.

Moreover, for every compact set $K \subseteq Y$, there exists an increasing sequence $(t_n)_{n \in \mathbb{N}}$ of positive reals such that $\lim_{n\to\infty} t_n = \infty$ and $g(t, y) = \lim_{n\to\infty} f(t+t_n, y)$ for all $y \in Y$ and a.e. $t \ge 0$.

In [293, Theorem 2.7.1, Theorem 2.7.2], we have slightly improved the important composition principle attributed to W. Long, S.-H. Ding [399, Theorem 2.2]. Further refinements for $S^{p(x)}$ -almost periodicity can be deduced similarly, with appealing to Lemma 3.12.1(i)–(iii) and the arguments employed in the proof of [399, Theorem 2.2]:

THEOREM 3.12.18. Let $I = \mathbb{R}$ or $I = [0, \infty)$, and let $p \in \mathcal{P}([0, 1])$. Suppose that the following conditions hold:

(i) $f \in APS^{p(x)}(I \times Y : X)$ and there exist a function $r \in \mathcal{P}([0, 1])$ such that $r(\cdot) \ge \max(p(\cdot), p(\cdot)/p(\cdot) - 1)$ and a function $L_f \in L_S^{r(x)}(I)$ such that:

(461)
$$||f(t,x) - f(t,y)|| \leq L_f(t) ||x - y||_Y, \quad t \in I, \ x, \ y \in Y;$$

 (ii) u ∈ APS^{p(x)}(I:Y), and there exists a set E ⊆ I with m(E) = 0 such that K := {u(t) : t ∈ I \ E} is relatively compact in Y; here, m(·) denotes the Lebesgue measure. Define $q \in \mathcal{P}([0,1])$ by q(x) := p(x)r(x)/p(x) + r(x), if $x \in [0,1]$ and $r(x) < \infty$, q(x) := p(x), if $x \in [0,1]$ and $r(x) = \infty$. Then $q(x) \in [1, p(x))$ for $x \in [0,1]$, $r(x) < \infty$ and $f(\cdot, u(\cdot)) \in APS^{q(x)}(I : X)$.

Concerning asymptotical two-parameter Stepanov p(x)-almost periodicity, we can deduce the following composition principles with X = Y; the proof is very similar to those of [293, Proposition 2.7.3, Proposition 2.7.4] established in the case of constant functions p, q, r:

PROPOSITION 3.12.19. Let $I = [0, \infty)$, and let $p \in \mathcal{P}([0, 1])$. Suppose that the following conditions hold:

- (i) $g \in APS^{p(x)}(I \times X : X)$, there exist a function $r \in \mathcal{P}([0,1])$ such that $r(\cdot) \ge \max(p(\cdot), p(\cdot)/p(\cdot) 1)$ and a function $L_g \in L_S^{r(x)}(I)$ such that (461) holds with the function $f(\cdot, \cdot)$ replaced by the function $g(\cdot, \cdot)$ therein.
- (ii) $v \in APS^{p(x)}(I : X)$, and there exists a set $E \subseteq I$ with m(E) = 0 such that $K = \{v(t) : t \in I \setminus E\}$ is relatively compact in X.
- (iii) f(t,x) = g(t,x) + q(t,x) for all $t \ge 0$ and $x \in X$, where $\hat{q} \in C_0([0,\infty) \times X : L^{q(x)}([0,1]:X))$ with $q(\cdot)$ defined as above;
- (iv) $u(t) = v(t) + \omega(t)$ for all $t \ge 0$, where $\hat{\omega} \in C_0([0,\infty) : L^{p(x)}([0,1] : X))$.
- (v) There exists a set $E' \subseteq I$ with m(E') = 0 such that $K' = \{u(t) : t \in I \setminus E'\}$ is relatively compact in X.

Then $f(\cdot, u(\cdot)) \in AAPS^{q(x)}(I:X)$.

3.12.4. Generalized (asymptotical) almost periodicity in Lebesgue spaces with variable exponents $L^{p(x)}$: action of convolution products. Throughout this section, we assume that $p \in \mathcal{P}([0, 1])$ and a multivalued linear operator \mathcal{A} fulfills the condition (P). We will first investigate infinite convolution products. The results obtained can be simply incorporated in the study of existence and uniqueness of almost periodic solutions of the following abstract Cauchy differential inclusion of first order

$$u'(t) \in \mathcal{A}u(t) + g(t), \quad t \in \mathbb{R}$$

and the following abstract Cauchy relaxation differential inclusion

(462)
$$D_{t,+}^{\gamma}u(t) \in -\mathcal{A}u(t) + g(t), \quad t \in \mathbb{R},$$

where $D_{t,+}^{\gamma}$ denotes the Weyl–Liouville fractional derivative of order $\gamma \in (0, 1)$ and $g: \mathbb{R} \times X \to X$ satisfies certain assumptions; see [293] for further information in this direction. Keeping in mind composition principles clarified in the previous section, it is almost straightforward to reformulate some known results concerning semilinear analogues of the above inclusions (see e.g. [293, Theorem 2.7.6–Theorem 2.7.9; Theorem 2.9.10–Theorem 2.9.11; Theorem 2.9.17–Theorem 2.9.18]); because of that, this question will not be examined here for the sake of brevity.

We start by stating the following generalization of [337, Proposition 2.11] (the reflexion at zero keeps the spaces of Stepanov *p*-almost periodic functions unchanged, which may or may not be the case with the spaces of Stepanov p(x)-almost periodic functions):

PROPOSITION 3.12.20. Suppose that $q \in \mathcal{P}([0,1])$, 1/p(x) + 1/q(x) = 1 and $(R(t))_{t>0} \subseteq L(X,Y)$ is a strongly continuous operator family satisfying that $M := \sum_{k=0}^{\infty} ||R(\cdot + k)||_{L^{q(x)}[0,1]} < \infty$. If $\check{g} \colon \mathbb{R} \to X$ is $S^{p(x)}$ -almost periodic, then the function $G \colon \mathbb{R} \to Y$, given by (433) with the function $f(\cdot)$ replaced by the function $g(\cdot)$ therein, is well-defined and almost periodic.

PROOF. Without loss of generality, we may assume that X = Y. It is clear that, for every $t \in \mathbb{R}$, we have that $G(t) = \int_0^\infty R(s)g(t-s)ds$ and that the last integral is absolutely convergent due to Lemma 3.12.1(i) and $S^{p(x)}$ -boundedness of function $\check{g}(\cdot)$:

$$\begin{split} \int_0^\infty \|R(s)\| \|g(t-s)\| ds &= \sum_{k=0}^\infty \int_k^{k+1} \|R(s)\| \|g(t-s)\| ds \\ &= \sum_{k=0}^\infty \int_0^1 \|R(s+k)\| \|g(t-s-k)\| ds \\ &\leqslant 2\sum_{k=0}^\infty \|R(\cdot+k)\|_{L^{q(x)}([0,1]:X)} \|g(t-k-\cdot)\|_{L^{p(x)}([0,1]:X)} \\ &\leqslant 2M \sup_{t\in\mathbb{R}} \|\check{g}(\cdot-t)\|_{L^{p(x)}([0,1]:X)}, \end{split}$$

for any $t \in \mathbb{R}$. Let a number $\varepsilon > 0$ be fixed. Then there is a finite number l > 0such that any subinterval I of \mathbb{R} of length l contains a number $\tau \in I$ such that $\|\check{g}(t-\tau+\cdot)-\check{g}(t+\cdot)\|_{L^{p(x)}([0,1]:X)} \leq \varepsilon, t \in \mathbb{R}$. Invoking Lemma 3.12.1(i) and this fact, we get

$$\begin{split} \|G(t+\tau) - G(t)\| &\leqslant \int_0^\infty \|R(r)\| \cdot \|g(t+\tau-r) - g(t-r)\| dr \\ &= \sum_{k=0}^\infty \int_k^{k+1} \|R(r)\| \cdot \|g(t+\tau-r) - g(t-r)\| dr \\ &= \sum_{k=0}^\infty \int_0^1 \|R(r+k)\| \cdot \|g(t+\tau-r-k) - g(t-r-k)\| dr \\ &\leqslant 2 \sum_{k=0}^\infty \|R(\cdot+k)\|_{L^{q(x)}[0,1]} \|g(t+\tau-\cdot-k) - g(t-\cdot-k)\|_{L^{p(x)}[0,1]} \\ &= 2 \sum_{k=0}^\infty \|R(\cdot+k)\|_{L^{q(x)}[0,1]} \|\check{g}(\cdot-t-\tau+k) - \check{g}(\cdot-t+k)\|_{L^{p(x)}[0,1]} \\ &\leqslant 2\varepsilon \sum_{k=0}^\infty \|R(\cdot+k)\|_{L^{q(x)}[0,1]} = 2M\varepsilon, \quad t \in \mathbb{R}, \end{split}$$

which clearly implies that the set of all ε -periods of $G(\cdot)$ is relatively dense in \mathbb{R} . It remains to be proved the uniform continuity of $G(\cdot)$. Since $\hat{g}(\cdot)$ is uniformly continuous, we have the existence of a number $\delta \in (0, 1)$ such that

$$(463) \qquad \|\check{g}(\cdot - t') - \check{g}(\cdot - t)\|_{L^{p(x)}[0,1]} < \varepsilon, \text{ provided } t, \ t' \in \mathbb{R} \text{ and } |t - t'| < \delta.$$

For any $\delta' \in (0, \delta)$, the above computation with $\tau = \delta' = t' - t$ and (463) together imply that, for every $t \in \mathbb{R}$,

$$\begin{split} \|G(t+\delta') - G(t)\| &\leq 2\sum_{k=0}^{\infty} \|R(\cdot+k)\|_{L^{q(x)}[0,1]} \|\check{g}(\cdot-t'+k) - \check{g}(\cdot-t+k)\|_{L^{p(x)}[0,1]} \\ &\leq 2\varepsilon \sum_{k=0}^{\infty} \|R(\cdot+k)\|_{L^{q(x)}[0,1]} = 2M\varepsilon. \end{split}$$

This completes the proof of proposition.

EXAMPLE 3.12.21. (i) Suppose that $\beta \in (0,1)$ and $(R(t))_{t>0} = (T(t))_{t>0}$ is a degenerate semigroup generated by \mathcal{A} . Let us recall that there exists a finite constant M > 0 such that $||T(t)|| \leq Mt^{\beta-1}$, $t \in (0,1]$ and $||T(t)|| \leq Me^{-ct}$, $t \geq 1$. Let $p_0 > 1$ be such that

$$\frac{p_0}{p_0-1}(\beta-1)\leqslant -1,$$

let $p \in \mathcal{P}([0,1])$, and let $||T(\cdot)||_{L^{q(x)}[0,1]} < \infty$. Assume that we have constructed a function $\check{g} \in APS^{p(x)}(\mathbb{R} : X)$ such that $\check{g} \notin APS^{p}(\mathbb{R} : X)$ for all $p \ge p_0$ (Question: Could we manipulate here somehow with the construction established in [81, Example, p. 70]?). Then, in our concrete situation, [337, Proposition 2.11] cannot be applied since

$$\frac{p}{p-1}(\beta-1) \leqslant -1, \quad p \in [1, p_0).$$

Now we will briefly explain that $\sum_{k=0}^{\infty} \|R(\cdot+k)\|_{L^{q(x)}[0,1]} < \infty$, showing that Proposition 3.12.20 is applicable. Strictly speaking, for k = 0, $\|T(\cdot)\|_{L^{q(x)}[0,1]} < \infty$ by our assumption, while, for $k \ge 1$, it can be simply shown that $\|R(\cdot+k)\|_{L^{q(x)}[0,1]} \le Me^{-ck}$ so that $\sum_{k=0}^{\infty} \|R(\cdot+k)\|_{L^{q(x)}[0,1]} < \infty$, as claimed.

(ii) By a mild solution of (462), we mean the function $t \mapsto \int_{-\infty}^{t} R_{\gamma}(t - s)g(s)ds, t \in \mathbb{R}$. Let $p \in \mathcal{P}([0,1])$, and let $\|R_{\gamma}(\cdot)\|_{L^{q(x)}[0,1]} < \infty$. Then, for $k \ge 1$, we have $\|R_{\gamma}(\cdot+k)\|_{L^{q(x)}[0,1]} \le M_2 k^{-1-\gamma}$. Hence, $\sum_{k=0}^{\infty} \|R_{\gamma}(\cdot+k)\|_{L^{q(x)}[0,1]} < \infty$ and we can apply Proposition 3.12.20.

In the following proposition, whose proof is very similar to that of [149, Proposition 3.12], we state some invariance properties of generalized asymptotical almost periodicity in Lebesgue spaces with variable exponents $L^{p(x)}$ under the action of finite convolution products (see also [293, Proposition 2.7.5, Lemma 2.9.3] for similar results). This proposition generalizes [337, Proposition 2.13] provided that p > 1 in its formulation.

PROPOSITION 3.12.22. Suppose that $p \in \mathcal{P}([0,1]), q \in D_+([0,1]), 1/p(x) + 1/q(x) = 1$ and $(R(t))_{t>0} \subseteq L(X)$ is a strongly continuous operator family satisfying that, for every $t \ge 0$, we have that $m_t := \sum_{k=0}^{\infty} ||R(\cdot + t + k)||_{L^{q(x)}[0,1]} < \infty$. Suppose, further, that $\check{g} \colon \mathbb{R} \to X$ is $S^{p(x)}$ -almost periodic, $q \in L_S^{p(x)}([0,\infty) \colon X)$ and $f(t) = g(t) + q(t), t \ge 0$. Let $r_1, r_2 \in \mathcal{P}([0,1])$ and the following holds: (i) For every $t \ge 0$, the mapping $x \mapsto \int_0^{t+x} R(t+x-s)q(s)ds$, $x \in [0,1]$ belongs to the space $L^{r_1(x)}([0,1]:X)$ and we have

$$\lim_{t \to +\infty} \left\| \int_0^{t+x} R(t+x-s)q(s)ds \right\|_{L^{r_1(x)}[0,1]} = 0.$$

(ii) For every $t \ge 0$, the mapping $x \mapsto m_{t+x}$, $x \in [0,1]$ belongs to the space $L^{r_2(x)}[0,1]$ and we have

$$\lim_{t \to +\infty} \|m_{t+x}\|_{L^{r_2(x)}[0,1]} = 0.$$

Then the function $H(\cdot)$, given by

$$H(t) := \int_0^t R(t-s)f(s)ds, \quad t \ge 0,$$

is well defined, bounded and belongs to the class $APS^{p(x)}(\mathbb{R}:X) + S_0^{r_1(x)}([0,\infty):X) + S_0^{r_2(x)}([0,\infty):X)$, with the meaning clear.

REMARK 3.12.23. In [337, Remark 2.14], we have examined the conditions under which the function $H(\cdot)$ defined above is asymptotically almost periodic, provided that the function $g(\cdot)$ is S^p -almost periodic for some $p \in [1, \infty)$. The interested reader may try to analyze similar problems with function $\check{g}(\cdot)$ being $S^{p(x)}$ -almost periodic for some $p \in \mathcal{P}([0, 1])$.

3.12.5. Some applications. Let $\Omega \subseteq \mathbb{R}^n$ be an open bounded subset with smooth boundary $\partial\Omega$ and let $1 . Among other things, one can make use of Proposition 3.12.22 to establish the existence and uniqueness of asymptotically <math>S^{p(x)}$ -almost automorphic solutions to the damped Poisson-wave type equation, in the spaces $X := H^{-1}(\Omega)$ or $X := L^p(\Omega)$, given by

$$\begin{cases} \frac{\partial}{\partial t} \left(m(x) \frac{\partial u}{\partial t} \right) + (2\omega m(x) - \Delta) \frac{\partial u}{\partial t} + (A(x;D) - \omega \Delta \\ &+ \omega^2 m(x)) u(x,t) = f(x,t), \quad t \ge 0, \ x \in \Omega \\ u = \frac{\partial u}{\partial t} = 0, \quad (x,t) \in \partial\Omega \times [0,\infty), \\ u(0,x) = u_0(x), \ m(x) \Big[\Big(\frac{\partial u}{\partial t} \Big)(x,0) + \omega u_0 \Big] = m(x) u_1(x), \quad x \in \Omega, \end{cases}$$

where $m(x) \in L^{\infty}(\Omega)$, $m(x) \ge 0$ a.e. $x \in \Omega$, Δ is the Dirichlet Laplacian in $L^{2}(\Omega)$, acting on its maximal domain, $H_{0}^{1}(\Omega) \cap H^{2}(\Omega)$, and A(x; D) is a second-order linear differential operator on Ω with continuous coefficients on $\overline{\Omega}$, see, e.g., [199, Example 6.1] and [293] for further details.

Notice that we can also consider the existence and uniqueness of asymptotically $S^{p(x)}$ -almost periodic solutions to the following fractional damped Poisson-wave

type equation, in the spaces $X := H^{-1}(\Omega)$ or $X := L^p(\Omega)$, given by

$$\begin{cases} \mathbf{D}_t^{\gamma}(m(x)\mathbf{D}_t^{\gamma}u) + (2\omega m(x) - \Delta)\mathbf{D}_t^{\gamma}u + (A(x;D) \\ & -\omega\Delta + \omega^2 m(x))u(x,t) = f(x,t), \quad t \ge 0, \ x \in \Omega; \\ u = \mathbf{D}_t^{\gamma}u = 0, \quad (x,t) \in \partial\Omega \times [0,\infty), \\ u(0,x) = u_0(x), \ m(x)[\mathbf{D}_t^{\gamma}u(x,0) + \omega u_0] = m(x)u_1(x), \quad x \in \Omega. \end{cases}$$

Additionally, it is also clear that Proposition 3.12.20 can be used to study the existence and uniqueness of almost periodic solutions of the following abstract integral inclusion

$$u(t) \in \mathcal{A} \int_{-\infty}^{t} a(t-s)u(s)ds + g(t), \quad t \in \mathbb{R}$$

where $a \in L^1_{loc}([0,\infty)), a \neq 0, \check{g} \colon \mathbb{R} \to X$ is $S^{p(x)}$ -almost periodic and \mathcal{A} is a closed multivalued linear operator on X, see, e.g., [293].

3.13. Appendix and notes

Without any doubt, the most important monograph which obeys the multivalued linear operators approach to abstract degenerate differential equations is [199], written by A. Favini and A. Yagi. The fundamental part of this monograph is Chapter III, where the authors have considered the generation of infinitely differentiable semigroups by multivalued linear operators, provided that there exist finite constants c, M > 0 and $\beta \in (0, 1]$ such that the condition (PW) holds (cf. Section 3.5 for more details). As already mentioned, a great part of results from [199, Section 3.2-Section 3.5] is not attainable in fractional case.

We start by sketching briefly the most relevant details about fractional powers and interpolation theory for multivalued linear operators satisfying the condition (PW); cf. [181]. Let \mathcal{A} be such an MLO, and let $(T(t))_{t>0}$ be a semigroup generated by \mathcal{A} (the pivot space is one of Banach's and will be denoted by E). Then the fractional power $(-\mathcal{A})^{\theta}$ is defined for $|\theta| > 1 - \beta$ and it is very difficult to tell what would be the fractional power $(-\mathcal{A})^{\theta}$ in the case that $|\theta| \leq 1 - \beta$ (we can provide a substantially larger information base about fractional powers in the classical case $\beta = 1$), without using some regularizing techniques established in the previous section. The spaces $(E; D(\mathcal{A}))_{\gamma,p}$ and $E_{\mathcal{A}}^{\gamma,p}$ are defined as follows. If X is a Banach space, $g: (0, \infty) \to X$ is an X-valued strongly measurable function and $q \in [1, \infty)$, set $||g||_{L_q^*(X)} := (\int_0^\infty ||g(t)||_X^q \frac{dt}{t})^{1/q}$ and $||g||_{L_\infty^*(X)} := \operatorname{ess\,sup}_{t>0} ||g(t)||_X$. Suppose that either $p_0, p_1 \in [1, \infty)$ or $p_0 = p_1 = \infty$; then for any $\gamma \in (0, 1)$, we set $p = ((1 - \gamma)p_0^{-1} + \gamma p_1^{-1})^{-1}$ if $p_0, p_1 \in [1, \infty)$, and $p = \infty$ if $p_0 = p_1 = \infty$. The real interpolation space $(E; D(\mathcal{A}))_{\gamma,p}$ between E and $D(\mathcal{A})$ is defined by

$$(E; D(\mathcal{A}))_{\gamma, p} := \left\{ x \in E : x = v_0(t) + v_1(t) \text{ for some } t > 0, \text{ where} \\ v_0 \in C((0, \infty) : E), v_1 \in C((0, \infty) : [D(\mathcal{A})]), \\ \| \cdot^{\gamma} v_0(\cdot) \|_{L^*_{p_0}(E)} + \| \cdot^{\gamma-1} v_1(\cdot) \|_{L^*_{p_1}([D(\mathcal{A})])} < \infty \right\}.$$

Equipped with the norm

$$\|x\|_{(E;D(\mathcal{A}))_{\gamma,p}} := \inf_{v_0,v_1} \left[\|\cdot^{\gamma} v_0(\cdot)\|_{L^*_{p_0}(E)} + \|\cdot^{\gamma-1} v_1(\cdot)\|_{L^*_{p_1}([D(\mathcal{A})])} \right],$$

where the infimum is taken over all possible representations of x having the above form, $(E; D(\mathcal{A}))_{\gamma,p}$ becomes a Banach space. It is well known that there exists a constant $c_1(\gamma, p) > 0$ such that the interpolation inequality $||x||_{(E;D(\mathcal{A}))_{\gamma,p}} \leq$ $c_1(\gamma, p) \|x\|^{1-\gamma} \|x\|_{[D(\mathcal{A})]}^{\gamma}, x \in D(\mathcal{A}) \text{ holds, as well as that } [D(\mathcal{A})] \hookrightarrow (E; D(\mathcal{A}))_{\gamma, p} \hookrightarrow D(\mathcal{A})$ E. The Banach space $E_{\mathcal{A}}^{\gamma,p}$ ($\gamma \in (0,1), p \in [1,\infty]$) is defined as follows:

$$E_{\mathcal{A}}^{\gamma,p} := \left\{ x \in E \; ; \; [x]_{E_{\mathcal{A}}^{\gamma,p}} := \| t^{\gamma} [t(t-A)^{-1}x - x] \|_{L_{p}^{*}(E)} < \infty, \\ \| x \|_{E_{\mathcal{A}}^{\gamma,p}} := \| x \| + [x]_{E_{\mathcal{A}}^{\gamma,p}} \right\}.$$

Observe that we have already used the space $E_A^{\gamma} = E_A^{\gamma,\infty}$ in our previous work. The following holds:

- (a) $[D((-\mathcal{A})^{\theta_1})] \hookrightarrow [D((-\mathcal{A})^{\theta_2})]$ for $1 \beta < \theta_2 < \theta_1 + \beta 1;$ (b) If $\gamma \in (0, 1)$ and $1 \leq p_1 \leq p \leq p_2 \leq \infty$, then $[D(\mathcal{A})] \hookrightarrow (E; D(\mathcal{A}))_{\gamma, p_1} \hookrightarrow (E; D(\mathcal{A}))_{\gamma, p} \hookrightarrow (E; D(\mathcal{A}))_{\gamma, p_2} \hookrightarrow E;$ furthermore, if $0 < \gamma_2 < \gamma_1 < 1$, then $(E; D(\mathcal{A}))_{\gamma_1,\infty} \hookrightarrow (E; D(\mathcal{A}))_{\gamma_2,1};$

- (c) $E_{\mathcal{A}}^{\gamma,p} \cap \mathcal{A}0 = \{0\}$ for $\gamma \in (0,1), p \in [1,\infty];$ (d) $E_{\mathcal{A}}^{\gamma,p} \hookrightarrow (E; D(\mathcal{A}))_{\gamma,p}$ for $\gamma \in (0,1), p \in [1,\infty];$ (e) $(E; D(\mathcal{A}))_{\gamma,p} \hookrightarrow E_{\mathcal{A}}^{\gamma+\beta-1,p}$ for $\gamma \in (1-\beta,1), p \in [1,\infty]$ and $E_{\mathcal{A}}^{\gamma,1} \hookrightarrow$ $[D((-\mathcal{A})^{\theta})]$ for $\gamma \in (1-\beta, 1);$
- (f) $(T(t))_{t>0}$ is strongly continuous on $(E; D(\mathcal{A}))_{\gamma,p}$ and $E_{\mathcal{A}}^{\gamma,p}$ for $\gamma \in (1 1)^{-1}$ $\beta, 1), p \in [1, \infty];$
- (g) Define T(0) := I and let $X \in \{E_{\mathcal{A}}^{\gamma,p}, (E; D(\mathcal{A}))_{\gamma,p}\} \ (\gamma \in (0,1), p \in [1,\infty]).$ Then for each $\gamma \in (2 - \alpha - \beta, 1)$ there exists a constant c > 0 such that

$$||T(t) - T(s)||_{L(X,E)} \leq c(t-s)^{(\alpha+\beta+\gamma-2)/\alpha} \text{ for } t > s \ge 0.$$

For further information about fractional powers and interpolation spaces of multivalued linear operators, as well as their applications in the qualitative analysis of abstract degenerate differential equations of first order, we refer the reader to the paper [180] by A. Favaron.

Concerning the monograph [199], we only want to say a few new words about the most important results from [199, Chapter VI], which cannot be so simply reconsidered in fractional case, as well. The first section of this chapter is devoted to the study of following initial value problem of parabolic type:

(464)
$$\frac{d}{dt}(Cu') + Bu' + Au = f(t), \quad t \in (0,T]; \ u(0) = u_0, \ Cu'(0) = Cu_1,$$

where A, B, C are three closed linear operators acting on E, $D(B) \subseteq D(A)$ and B has a bounded inverse. By a solution of (464) we mean any function $u \in C^1((0,T])$: E) such that $u(t) \in D(A), t \in (0,T], u'(t) \in D(B) \cap D(C), t \in (0,T], Cu' \in (0,T],$ $C^{1}((0,T]:E), Bu' \in C((0,T]:E), (464)$ holds on (0,T] and $\lim_{t\to 0+} u(t) = u_{0}$, $\lim_{t\to 0+} Cu'(t) = Cu_1$. As already mentioned in Example 3.7.14(ii), the problem (464) can be rewritten in the following matricial form

$$\frac{d}{dt}Mz(t) = Lz(t) + F(t), \quad t \in (0,T],$$

where

$$M = \begin{bmatrix} I & O \\ O & C \end{bmatrix}, \ L = \begin{bmatrix} O & I \\ -A & -B \end{bmatrix} \text{ and } F(t) = \begin{bmatrix} 0 \\ f(t) \end{bmatrix} \ (t \in (0,T]).$$

The underlying Banach space is chosen to be $[D(B)] \times E$. Using this approach, the authors have proved that there exists a unique solution of (464) provided the validity of following conditions (cf. [199, Theorem 6.1]):

(i) There exist constants c > 0, $\alpha \in (0, 1]$ and $\beta \in (0, \alpha]$ such that

$$\Theta_{\alpha,c} := \{\lambda \in \mathbb{C} : \operatorname{Re} \lambda \leqslant c(1 + |\operatorname{Im} \lambda|)^{\alpha}\} \subseteq \rho_C(B)$$

and

$$\|C(\lambda C+B)^{-1}\| = O((1+|\lambda|)^{-\beta}), \quad \lambda \in -\Theta_{\alpha,c};$$

(ii) $u_0, u_1 \in D(B), 2\alpha + \beta > 2, (2 - \alpha - \beta)/\alpha < \sigma \leq 1$, and (iii) $f \in C^{\sigma}([0, T] : E).$

Under some additional conditions, in [199, Theorem 6.2] it has been shown the Hölder continuity of functions Bu' and (Cu')', where u is a solution of (464).

- Set $U := AB^{-1}$ and $\mathcal{A} := BC^{-1}$. The conditions
- $||B(\lambda B + A)^{-1}|| = O((1 + |\lambda|)^{-\gamma}), \quad \lambda \in -\Theta_{\alpha,c},$ (a)
- where $0 < \gamma \leq 1$, and $\|U(\lambda + A)^{-1}\| = O((1 + |\lambda|)^{-\delta}), \quad \lambda \in -\Theta_{\alpha,c}, \ |\lambda|$ suff. large, (b) where $\delta > 0$.

play a crucial role in [199, Theorem 6.3–Theorem 6.4]. The leading feature of [199, Theorem 6.7] is to extend these assertions to the following higher-order initial value problem:

$$\begin{cases} \frac{d}{dt}(A_n u^{(n-1)}) + \sum_{j=0}^{n-1} A_j u^{(j)} = f(t), & t \in (0,T], \\ u^{(j)}(0) = u_j, & 0 \leq j \leq \lceil \alpha \rceil - 2, \\ A_n u^{(n-1)}(0) = A_n u_{n-1}, \end{cases}$$

by using an idea of M. K. Balaev [43]. In [199, Section 6.2], the authors have presented a great number of concrete applications of results from Section 6.1, while in [199, Section 6.3], the complete second order degenerate Cauchy problem

$$\begin{cases} Cu''(t) + Bu'(t) + Au(t) = f(t), & t \in (0,T], \\ u(0) = u_0, & u'(0) = u_1, \end{cases}$$

has been considered in a Hilbert space H. The abstract results have been applied in the analysis of Poisson wave equation

$$\begin{cases} m_1(x)\frac{\partial^2 u}{\partial t^2}(x,t) + m_2(x)\frac{\partial u}{\partial t}(x,t) - \Delta u(x,t) = f(x,t), & (x,t) \in Q, \\ u(x,t) = 0, & (x,t) \in \Sigma, \\ u(x,0) = u_0(x), \frac{\partial u}{\partial t}u(x,0) = u_1(x), & x \in \Omega, \end{cases}$$

in $L^2(\Omega)$, where Ω is an open bounded domain in \mathbb{R}^n with a smooth boundary Γ , $Q = \Omega \times (0,T)$, $\Sigma = \Gamma \times (0,T)$ and $m_1(x)$, $m_2(x)$ are two given nonnegative continuous functions on $\overline{\Omega}$. We would like to address the problem of qualitative analysis of some fractional analogues of Poisson wave equation to our readers.

Section 6.5 is devoted to the study of initial value problem

$$\begin{cases} -\frac{d^2}{dt^2}(Bu) + Au = f(t), & 0 \le t \le 1, \\ Bu(0) = Bu(1) = 0, \end{cases}$$

provided that there exist constants c > 0, $\alpha \in (0,1]$ and $\beta \in (0,\alpha]$ such that $\Theta_{\alpha,c} \subseteq \rho(AB^{-1})$ and $||B(\lambda B - A)^{-1}|| = O((1 + |\lambda|)^{-\beta})$, $\lambda \in \Theta_{\alpha,c}$. Finally, the abstract Cauchy problem

$$\begin{cases} u''(t) = t^m B u(t), & t \ge 0, \\ u(0) = u_0, & \frac{du}{dt}(0) = u_1, \end{cases}$$

where m > 0 and B is a closed densely defined operator in a Banach space E satisfying that there exists a finite constant M > 0 such that $\|(\lambda + B)^{-1}\| \leq M\lambda^{-1}$, $\lambda > 0$ has been investigated in Section 6.6.

Semilinear degenerate Cauchy inclusions. Concerning semilinear degenerate differential inclusions of first order, we can warmly recommend the reading of monograph [268] by M. Kamenskii, V. Obukhovskii and P. Zecca. In what follows, we shall briefly describe the most important results from the paper [437].

Suppose that E is a real reflexive Banach space, $F: [0,T] \times E \to P(E)$ is a given multimap, as well as that A and B are two single-valued linear operators on E satisfying that $D(A) \subseteq D(B)$ and $B(D(A)) \subseteq R(A)$. In [437], V. Obukhovskii and P. Zecca have analyzed the semilinear differential equation

(465)
$$\frac{d}{dt}[Bu(t)] = Au(t) + F(t, Bu(t)), \quad t \in [0, T] ; Bu(0) = y_0 \in \overline{B(D(A))}$$

by using the change of variables v(t) = Bu(t) and passing after that to the corresponding abstract degenerate semilinear Cauchy inclusion with the multivalued linear operator $\mathcal{A} = AB^{-1}$. They have assumed that the operator \mathcal{A} satisfies the Hille–Yosida condition, so that \mathcal{A} generates a strongly continuous semigroup $(U(t))_{t\geq 0}$ on E. A function u(t) is said to be a mild solution of (465) iff there exists a measurable section f(t) of the multivalued mapping $t \mapsto F(t, Bu(t)), t \in [0, T]$ such that:

$$Bu(t) = U(t)Bu(0) + \int_0^t U(t-s)f(s)ds, \quad t \in [0,T].$$

Denote $P_0(E) := \{S \subseteq E : S \neq \emptyset\}$, $K(E) := \{S \in P_0(E) : S \text{ is compact}\}$ and $Kv(E) := \{S \in K(E) : S \text{ is convex}\}$. Following [437, Definition 2.12], we say that a multimap $\mathbf{F} : X \to P_0(Y)$, where X and Y are real Banach spaces, is:

- (i) upper semicontinuous (u.s.c.) iff $\mathbf{F}^{-1}(V) = \{x \in X : \mathbf{F}(x) \subseteq V\}$ is an open subset of X for every open set $V \subseteq Y$;
- (ii) lower semicontinuous (l.s.c.) iff $\mathbf{F}^{-1}(W)$ is a closed subset of X for every closed set $W \subseteq Y$.

Recall that the Hausdorff measure of noncompactness of a non-empty subset Ω of E is defined by

$$\chi(\Omega) = \inf\{\varepsilon > 0 : \Omega \text{ has a finite } \varepsilon \text{-net}\}.$$

In [437, Theorem 3.1], the authors have proved that the set Σ consisting of all mild solutions of problem (465) is nonempty as well as that the set $B\Sigma = \{v \in C([0,T] : E) \mid v(t) = Bu(t), u \in \Sigma\}$ is compact in C([0,T] : E), provided the following conditions:

- (T1) $F(\cdot, \cdot)$ has nonempty, compact, and convex values;
- (T2) For every $x \in E$, the multimapping $F(\cdot, x) \colon [0, T] \to Kv(E)$ has a strongly measurable selection;
- (T3) The multimapping $F(t, \cdot) : E \to Kv(E)$ is u.s.c. for a.e. $t \in [0, T]$;
- (T4) There exists a function $\alpha \in L^1([0,T])$ such that $\alpha(t) \ge 0$ for a.e. $t \in [0,T]$, and

$$||F(t,x)|| := \sup\{||z|| : z \in F(t,x)\} \leq \alpha(t)(1+||x||) \text{ for a.e. } t \in [0,T];$$

- (T5) There exists a function $k \in L^1([0,T])$ such that $k(t) \ge 0$ for a.e. $t \in [0,T]$, and
 - $\chi(F(t,D)) \leq k(t)\chi(D)$ for a.e. $t \in [0,T]$ and every bounded set $D \subseteq E$.

The conditions (T2) and (T3) are so-called the upper Carathéodory conditions. In [437, Theorem 3.2], the condition (T1) has been replaced by the following one:

(T6) There exists a sequence of disjoint compact sets $\{I_n\}, I_n \subseteq [0,T]$, such that $m([0,T] \setminus I) = 0$, where $I = \bigcup_n I_n$, and the restriction of $F(\cdot, \cdot)$ on each set $I_n \times E$ is l.s.c.

If we assume the separability of space E and (T4)-(T6), then there exists at least one mild solution of problem (465). Boundary value problems have been analyzed in [437, Subsection 3.2].

Now we will present the most important results established in the research study [221] by M. Fuhrman.

Sums of generators of analytic semigroups: multivalued linear operators approach. Approximation and interpolation results for multivalued linear operators have been essentially utilized by M. Fuhrman in proving a few remarkable perturbation results for generators of analytic semigroups [221]. Let us recall that the generator of an analytic semigroup is any closed (not necessarily densely defined) single-valued operator A satisfying that there exist numbers $\omega \in \mathbb{R}$, $\alpha \in (0, \pi/2)$ and M > 0 such that $\omega + \Sigma_{\pi-\alpha} \subseteq \rho(A)$ and $||R(\lambda : A)|| \leq M/|\lambda - \omega|$, $\lambda \in \omega + \Sigma_{\pi-\alpha}$. Let B be another linear operator with domain and range contained in E. Looking for conditions under the operator A + B is again a generator of an analytic semigroup is a famous problem in the perturbation theory of linear operators.

The author employs the following two conditions:

(H1.S) A and B are linear operators in E and there exist numbers c > 0 and $\theta \in (0, \pi/2)$ such that

$$||(z-A)^{-1}|| + ||(z-B)^{-1}|| \leq c/|z|, \quad z \in \Sigma_{\pi-\theta}.$$

(H2.DPG.S) For every $v \in \Sigma_{\pi-\theta}$, we have $(A - v)^{-1}(D(B)) \subseteq D(B)$ and there exist numbers c > 0, α , β such that $-1 \leq \alpha < \beta \leq 1$ and

 $||[B; (A-v)^{-1}](B-z)^{-1}|| \leq c|v|^{\alpha-1}|z|^{-\beta}, \quad v, \ z \in \Sigma_{\pi-\theta}$

(here, by [P;Q] we denote the commutator PQ - QP of two linear operators in E).

The condition (H2.DPG.S) has its origins in the pioneering paper by G. da Prato, P. Grisvard [127].

If the conditions (H1.S) and (H2.DPG.S) are satisfied with D(A) and D(B)being dense in E, then the operator A+B is closable and its closure $\overline{A+B}$ generates a strongly continuous analytic semigroup. On the other hand, the operator A+Bneed not be closable if some of domains D(A) or D(B) is not dense in E. Since any multivalued linear operator is closable, in this case we can consider the operator $\overline{A+B}$ in the MLO sense. This is the starting point of analysis conducted in [221].

Using the Yosida approximations of operators A and B, M. Fuhrman has proved the following results:

- (i) Assume that (H1.S) and (H2.DPG.S) hold, as well as that D(B) is dense in E. Then $\overline{A+B}$ generates an analytic semigroup in the MLO sense.
- (ii) Assume that (H1.S) and (H2.DPG.S) hold. Then there exists a linear relation \mathcal{A} which extends the operator A + B and generates an analytic semigroup in the MLO sense. Furthermore, we have

$$D(A) \cap D(B) \subseteq D(A) \subseteq (E, [D(A)])_{\nu,p} \cap (E, [D(B)])_{\nu,p} \quad (\nu \in (0, 1), \ p \in [1, \infty]).$$

If we assume some additional conditions, then there exists $\nu_0 \in (0, 1)$ such that for each $\nu \in (0, \nu_0)$ we have the equality

$$(E, [D(\mathcal{A})])_{\nu, p} = (E, [D(\mathcal{A})])_{\nu, p} \cap (E, [D(\mathcal{B})])_{\nu, p} \quad (p \in [1, \infty]).$$

Maximal time regularity for abstract degenerate Volterra integrodifferential equations. The study of maximal time and space regularity for abstract (degenerate) differential equations is still an active field of research. Here we will present the main results of research study [183] by A. Favaron and A. Favini (cf. also [182]) regarding the question of maximal time regularity for the following abstract degenerate Volterra integro-differential equation

(466)
$$\frac{d}{dt}(Mv(t)) = [\lambda_0 M + L]v(t) + \int_0^t k(t-s)L_1v(s)ds + h(t)y + f(t), \quad t \in [0,T]; \ Mv(0) = Mv_0,$$

where L, M, L_1 are three closed linear operators with domains and ranges contained in a complex Banach space $X, \lambda_0 \in \mathbb{C}, D(L) \subseteq D(M) \cap D(L_1), v_0 \in D(M), y \in X,$ $f: [0,T] \to X$ and $h, k: [0,T] \to \mathbb{C}$. The authors have shown that the Hölder continuity of mappings h, k, f in time, combined with some extra assumptions on the operator $(\lambda_0 M + L)$ and values of f(0), y, implies that there exists a unique global strict solutions of (466) and that the derivative (Mv)' has the same Hölder exponent as the mappings h and k as. The obtained result is in full accordance with that already established in the monograph [199] in the case that $(\lambda_0, h, k) =$ (0, 0, 0), which provides a certain loss of regularity of derivative (Mv)' with respect to f. The authors obey the multivalued linear operators approach to (466), pointing out that the substitution $v_1 = Mv(t)$ cannot be directly applied. An application of abstract results to a concrete degenerate integro-differential equation, arising in modeling direct and inverse problems of heat conduction for materials with memory, has been provided. The time-relaxation analogue of (466), obtained by replacing the term $\frac{d}{dt}(Mv(t))$ with $\mathbf{D}_t^{\alpha}(Mv(t))$ ($0 < \alpha < 1$) is much more complicated for the analysis and we would like to propose the question of time regularity for such an abstract degenerate integro-differential equation.

Abstract fractional differential inclusions with Riemann–Liouville derivatives. The study of abstract fractional inclusions with Riemann–Liouville derivatives has not been received much attention in this chapter. We have already explored, in the second chapter, the abstract multi-term fractional differential equations of type (SC1), (SC2) or (SC3), pointing out that the case (SC3) is very sofisticated. Here it is worth noting that A. V. Glushak has investigated, in a series of his research papers [227, 230], the abstract fractional differential equations with Riemann–Liouville derivatives of orders $\alpha, \beta \in (0, 1)$, accompanied with the initial conditions that is not of type (SC3). For example, the abstract fractional Cauchy problems like

$$\begin{split} D_t^{\alpha}(t^k D_t^{\beta} u(t)) &= t^{\gamma} A u(t) + f(t), \ t > 0;\\ \lim_{t \to 0+} J_t^{1-\beta} u(t) &= u_0, \ \lim_{t \to 0+} J_t^{1-\alpha}(t^k D_t^{\beta} u(t)) = 0,\\ D_t^{\alpha} D_t^{\beta} u(t) &= A u(t) + f(t), \ t > 0; \ \lim_{t \to 0+} J_t^{1-\beta} u(t) = u_0, \ \lim_{t \to 0+} J_t^{1-\alpha} D_t^{\beta} u(t) = 0, \end{split}$$

and

$$D_t^{\alpha}u(t) = Au(t) + f(t), \ t > 0; \ \lim_{t \to 0+} J_t^{1-\alpha}u(t) = u_0$$

have been investigated. The structural results obtained by A. V. Glushak hold, with insignificant modifications, in the multivalued linear operators framework.

And, at the very end of monograph, we shall say a few words about a singular perturbation problem for abstract Volterra integro-differential inclusions.

Singular perturbation theory for abstract Volterra integro-differential inclusions. When talking about abstract Volterra integro-differential inclusions, it is almost inevitable to say some basic things about singular perturbation theory. As already marked in [292], singular perturbations of various kinds of abstract non-degenerate Volterra integro-differential equations have been investigated, among many other research papers, in [80, 130, 141, 142, 170, 178, 179, 223, 235, 273, 283, 389, 392, 393, 398] and [500]. Due primarily to the time limits, we have not been able to edit the fundamental info about this intriguing topic here. We only want to say that it is very simply and straightforwardly, after creating a stable theory of degenerate resolvent operator families in this chapter, to clarify some results about the singular perturbation problem for abstract Volterra integro-differential inclusions. For example, C. Lizama and H. Prado [398] have investigated the singular differential problem

$$\varepsilon^2 u_{\varepsilon}''(t) + u_{\varepsilon}'(t) = A u_{\varepsilon}(t) + (K * u_{\varepsilon})(t) + f_{\varepsilon}(t), \quad t \ge 0, \ \varepsilon > 0$$

for the abstract integro-differential equation

$$\omega'(t) = A\omega(t) + (K * \omega)(t) + f(t), \quad t \ge 0,$$

when $\varepsilon \to 0+$. The main results of this research study, [398, Theorem 3.6, Theorem 3.8], admit very simple reformulations for corresponding abstract integrodifferential inclusions obtained by replacing the closed single-valued linear operator A in the above equations with a closed multivalued linear operator \mathcal{A} . Details can be left to the interested readers.

Index

(A, B)-regularized C-(pseudo)resolvent family, 217 weak, 217 (A, B)-regularized (pseudo)resolvent family, 217weak. 217 (A, k, B)-regularized C-(pseudo)resolvent family. 217 analytic, 221 exponentially bounded analytic, 221 weak, 217 a-regular, 217 (A, k, B)-regularized C-resolvent family, 217(A, k, B)-regularized (pseudo)resolvent family, 217 weak. 217 (C, B)-resolvent, 69 $(\alpha, \alpha, \overline{P_1(A)}, \overline{P_2(A)}, C)$ -resolvent family, 100 $(\sigma, \vec{\beta})$ -scrambled set, 269 $(\tilde{X}, \vec{\beta})$ -distributionally chaoticity, 269 (a, C)-resolvent family, 294, 307 (a, k)-regularized (C_1, C_2) -existence and uniqueness family, 293, 294 (a, k)-regularized C-resolvent family, 294 analytic, 308, 312, 430, 431 approximation, 407 convergence, 407 entire, 313 equicontinuous, 294 equicontinuous, analytic, 308 exponentially equicontinuous, 294 exponentially equicontinuous, analytic, 308 hypoanalytic, 316 infinitely differentiable, 315 of solving operators, 277 real analytic, 316

(a, k)-regularized C_1 -existence family, 293 (a, k)-regularized C_2 -uniqueness family, 293 (a, k, C)-subgenerator, 325 (q_{α}, C) -regularized resolvent family of growth order $\zeta > 0, 404$ (k, C_1, \mathcal{V}) -existence family, 321 (k, C_2) -uniqueness family, 143 (kC)-parabolicity, 316 (kC)-well posedness, 216 $(\mathfrak{D}, \mathfrak{P})$ -hypercyclicity, 244, 394 $(\mathfrak{D}, \mathfrak{P}_s)$ -hypercyclicity, 244 C-distribution cosine function, 357 dense, 357 C-distribution semigroup, 344 C-pseudoresolvent, 71, 303 C-resolvent propagation family, 132 \tilde{E} -hypercyclic, 229, 392 \tilde{E} -topologically mixing, 229, 392 \tilde{E} -topologically transitive, 229, 392 C-resolvent set, 4, 29 C-ultradistribution cosine function, 357 dense, 357 C-ultradistribution semigroup of *-class, 344 C_1 -existence propagation family, 165 E_l -type spaces, 96, 118, 128 K-convoluted C-cosine functions, 330 N-linear Devaney chaos, 241 α -times integrated (A, B)-regularized C-(pseudo)resolvent family, 217 weak, 217 α -times integrated (a, C)-resolvent family, 294 \mathcal{V} -pre-solution, 318 σ -scrambled set, 253 $\sigma_{\tilde{X}}$ -scrambled set quasi, 262 \tilde{X} -distributional chaoticity quasi, 262

longitudinal waves, 10

 \tilde{X} -distributionally chaotic operator, 253 \tilde{X} -distributionally chaotic sequence of operators, 253 denselv. 253 \tilde{X} -distributionally irregular manifold, 254 uniformly, 254 \tilde{X} -distributionally irregular vector, 254 \tilde{E} -hypercyclic vector, 224, 230, 390 \tilde{E} -hypercyclicity, 229 \tilde{E} -topological transitivity, 224 \tilde{E} -topologically mixing property, 224, 229, 300 \tilde{E} -topologically transitivity, 229, 390 \tilde{E} -weakly mixing property, 235 ζ -times integrated C-resolvent propagation family, 131 ζ -times integrated C_1 -existence family, 136 ζ -times integrated C_1 -resolvent propagation family, 165 k-regularized (C_1, C_2) -existence and uniqueness family, 135 k-regularized C-resolvent family, 136 k-regularized C-resolvent propagation family, 131 k-regularized C_1 -existence family, 135, 141 k-regularized C_1 -existence propagation family, 131, 133, 164-167 k-regularized C_2 -uniqueness family, 135, 141, 321 k-regularized C_2 -uniqueness propagation family, 131 *n*-regular kernel, 316 p-solution, 286 q-exponential equicontinuity, 294 \mathcal{T}_L -admissibility, 239 \mathcal{T}_R -admissibility, 239 D-hypercyclicity, 241, 244, 394 D-topological transitivity, 244, 394 D-topologically mixing property, 241, 244, 394D₃₃-topological transitivity, 244 $\mathfrak{D}_{\mathfrak{B}_{\circ}}$ -topologically mixing property, 244 $\mathfrak{D}_{\mathfrak{P}}$ -topological transitivity, 244, 394 $\mathfrak{D}_{\mathfrak{N}}$ -topologically mixing property, 244, 394 (C-DCF), 357 (C-UDCF) of *-class, 357 (A, k, B)-regularized C-uniqueness family, 218 $\sigma_{\tilde{v}}$ -scrambled set, 253 \tilde{E} -chaoticity, 235 bilinear mapping, 241 Hooke's constant, 284 integral generator, 295

Post-Widder inversion formula, 60 Stirling's formula, 185 translation invariant metric, 252 abstract Beurling spaces, 179, 187 abstract Cauchy problem, 6 degenerate. 7 degenerate semilinear, 103 first and second order, 15, 97, 227, 234, 237, 239, 391 fractional degenerate, 91, 438, 439 incomplete degenerate, 200 inhomogeneous, 96, 130, 140, 172 abstract degenerate fractional inclusion, 17 multi-term. 392 abstract degenerate multi-term fractional differential equations, 8 abstract degenerate non-scalar Volterra equations, 13, 216 abstract degenerate Volterra equation, 141 abstract degenerate Volterra inclusion, 17, 285.404 multi-term, 18, 324 abstract differential operators, 90, 100, 121, 128, 207, 221, 397, 438 abstract incomplete differential inclusions. 420 abstract nonautonomous differential equations of first order, 461 abstract Weierstrass formula, 87 acoustic planar propagation in bubbly liquids, 270 almost C-nonnegative, 36 analyticity weak, 36 angular velocity, 129 approximation of Laplace transform, 63 Arendt-Widder theorem, 62 associated functions, 134, 172 asymptotic expansions, 66, 67 backward shift operators, 225 Baire Category, 14 Banach space, 28 barreled space, 28 bidual, 28 bihypercyclicity, 241 binary relations, 46 Borel measurable set, 32 bounded equicontinuity, 344 bounded subset, 28 Cauchy formula, 179, 185, 195, 200

chain of adjoint vectors, 211 class LT - X.58 $LT -_{or} X$, 58 A. 173 \mathcal{A}_q , 183 \mathcal{M}_{Cm} , 36, 168 $\mathcal{M}_{B,C,q}, 197$ $\mathcal{M}_{C,q,\omega}, 208$ (P1)-X, 57 coercivity, 90, 98, 100 comparision principles, 48 complex inversion theorem, 62 condition $(CCF_1), 357$ $(CCF_2), 357$ (M.1), 30, 184(M.2), 30, 178, 184(M.3), 30, 184(M.3)'. 30 (◊). 132 (◊◊), 132 (C.1)-(C.5), 115 (C1). 108 (C2), 109 (C3), 109 (C4), 109 (C5), 109 (CH), 134 (H), 196, 197 (H1), 446 (H1.S), 483 (H2), 446 (H2.DPG.S), 483 (H3), 22 (P)", 263 (P1). 56 (P1)', 161 (P2), 113 (PS), 237 (PW), 366 (Q), 311, 398 (Q1), 60 (QP), 379 (T1), 483 (T2), 483 (T3), 483 (T4), 483 (T5), 483 HP., 154 upper Carathéodory, 483 conjugacy lemma, 241, 248, 394 constant of regularity, 12

continuous linear mapping, 28 convolution products, 29 core, 29 cosine operator function, 149 C-regularized. 330 α -times integrated C-. 330 criterion Banasiak-Moszyński, 14, 226, 241, 390 Desch-Schappacher-Webb, 14, 226, 241, 270.390 cvclicity, 14 degenerate K-convoluted C-semigroups, 83 degenerate K-convoluted semigroups, 396 degenerate fractional equations, 121 degenerate integrated semigroups, 7, 298 delta distribution, 29 Devaney chaos, 16 Dirichlet boundary conditions, 7, 146, 153, 159, 309, 398 discriminant of polynomial, 89 distributional chaos, 15 distributionally irregular manifold, 254 uniformly, 254 distributionally irregular vector, 16, 254 dual space, 28 electric field intensity, 306 electromagnetic waves, 167 elliptic polynomial, 187 elliptic selfadjoint operators, 398 empathy, 283 equation backward Poisson heat, 314 Barenblatt-Zheltov-Kochina, 10, 97, 150, 155, 159, 269, 397 BBM, 280 Benney-Luke, 89, 268, 309, 397 Boussinesg, 121, 245, 397 Boussinesq-Love, 10, 150, 153, 159 Camassa-Holm, 280 Caputo-Riesz fractional advection diffusion, 168 damped Klein-Gordon, 246 damped Klein-Gordon, 232 damped Poisson-wave, 323, 478, 479 fractional Maxwell, 306 fractional Poisson heat, 375 fractional Poisson-wave equation, 306 fractional Sobolev, 8 gravity-gyroscopic wave, 128 inhomogeneous Poisson heat, 388 integral, 106, 109, 133, 146, 148, 168, 260

internal wave equation in the Boussinesq approximation, 128 Lonngern wave, 280 Poisson heat, 24, 405 reversed fractional Poisson heat, 432 Rosenau-Burgers, 280 Rossby wave, 128 semiconductor, 280 small amplitude oscillations of a rotating viscous fluid, 129 Sobolev, 128 vibrating beam type equation, 232 viscous van Wijngaarden-Eringen, 269 viscous van Wijngaarden-Eringen, 271 equicontinuous family of operators, 29 evolution system, 446 exponentially stable, 447 Green's function, 447 hyperbolic, 447 exponential region, 180, 354 exponential type, 294 exponentially equicontinuous $(\sigma - 1)$ -times integrated C-resolvent propagation families, 116 exponentially equicontinuous (a, C)-regularized resolvent family generated by A, B, 119exponentially equicontinuous (a, k)-regularized C-resolvent families generated by A, B, 119exponentially equicontinuous (a, k)-regularized C-resolvent family differentiable, 86 exponentially equicontinuous (k; C)-regularized resolvent (i, j)-propagation family, 9, 124 analytic, 124 exponentially equicontinuous C-regularized resolvent (i, j)-propagation family, 108 exponentially equicontinuous k-regularized C-resolvent (i, j)-propagation family, 9, 107 analytic, 114 exponentially equicontinuous k-regularized C_1 -existence propagation family analytic, 132 exponentially equicontinuous r-times integrated (a, C)-regularized resolvent family generated by A, B, 119exponentially equicontinuous r-times integrated C-regularized resolvent (i, j)-propagation family, 108

exponentially equicontinuous (equicontinuous) (a, k)-regularized C-resolvent family analytic, 84 Faà di Bruno's formula, 412 fluid filtration. 10 Fourier multiplier, 91 Fourier transform, 90 fractional calculus, 63, 64 fractional derivatives Caputo, 8, 64, 244, 367 Liouville right-sided, 65, 200, 201, 422 modified Liouville right-sided, 204 Riemann-Liouville, 10, 15, 64, 317 Wevl. 65 Weyl–Liouville, 466 fractional differential equations, 63, 64 fractional powers, 37, 197, 381, 413 purely imaginary, 418 spectral, 309 with negative imaginary part of exponent, 418 with positive imaginary part of exponent, 418 fractional Sobolev inclusions, 17, 285 fractional Sobolev space, 99 function μ -integrable, 33, 34 asymptoticallyStepanov almost periodic. 111 absolutely continuous, 32 admissible weight, 240 almost automorphic, 444 almost periodic, 443 associated, 31 asymptotically ω -almost periodic, 446 asymptotically almost automorphic, 444 asymptotically almost periodic, 443 asymptotically Stepanov p(x)-almost periodic, 468 Bernstein, 75 Bessel, 68, 433 Bochner integrable, 237 completely monotonic, 67, 75 completely positive, 76, 80, 143, 292, 298, 306, 324 creep, 75 entire of class (M_p) , 31 equi-Weyl-p-almost periodic, 444 Gamma, 29 Hölder continuous, 31, 64, 143, 292, 299, 322 Laplace transformable, 56

Lipschitz continuous, 104, 217 measurable by seminorms, 33 Mittag-Leffler, 66 of bounded variation, 32 quasi-asymptotically almost periodic, 448 S-asymptotically ω -periodic. 446 simple, 32, 33 Stepanov p(x)-almost periodic, 468 Stepanov p(x)-bounded, 468 Stepanov asymptotically ω -almost periodic, 446 Stepanov bounded, 444 Stepanov quasi-asymptotically almost periodic, 450 Stepanov two-parameter quasi-asymptotically almost periodic, 457 strongly measurable, 33 superharmonic, 7 two-parameter almost periodic, 473 two-parameter asymptotically almost periodic, 473 two-parameter asymptotically Stepanov p(x)-almost periodic, 474 two-parameter quasi-asymptotically almost periodic, 456 two-parameter Stepanov p(x)-almost periodic, 474 vector-valued analytic, 35 weakly measurable, 33 Weyl-p-almost periodic, 444 Wright, 66, 67, 437 functional calculus for commuting generators of bounded C_0 -groups, 90, 117, 268 generalized resolvent equation, 188 generator of empathy, 283 Gevrey sequence, 30 graph Cayley, 47 simple, 46 Gronwall inequality, 104 Hölder inequality, 476 heat conductivity, 10 Hermite expansions, 415 hypercyclic N-linear operators, 241 hypercyclic vector, 224, 230, 390 hypercyclicity, 14, 234 Hypercyclicity Criterion, 238, 392 hypoanalytic exponentially equicontinuous

k-regularized I-resolvent

(n, j)-propagation family, 118

identity theorem for analytic functions, 36 incomplete abstract Cauchy inclusion, 422. 424 incomplete abstract degenerate Cauchy problems, 201 integrated C-cosine functions, 362 integrated solution families, 90, 147, 156, 207. 301. 412 integration in locally convex spaces, 32 internal waves, 121 inverse Laplace transform, 56 inverse problems, 20 ioint closedness, 282 Kato's analyticity criteria, 158 kernel, 76, 105 Kronecker delta, 192 Laplace integral abscissa of convergence, 57, 161 Laplace transform, 56 Lebesgue point, 60 linear ordinary differential equation fundamental set of solutions, 395 linear relation, 38 linear topological homeomorphism, 248 liquid filtration, 89, 228 locally convex space complete, 77, 300, 400 non-metrizable, 247 reflexive, 218 logitudinal vibrations, 283 Lusternik type theorems, 456 Luxemburg norm, 467 magnetic field density, 306 magnetic flux density, 306 measure Lebesgue, 33 locally finite Borel, 32, 40 meromorphic extension, 316 moisture transfer, 10 multivalued linear operator, 16 C-resolvent, 40 adjoint. 39 chaotic, 44 closed, 39 complex powers, 416 d-hypercyclic vector, 45 disjoint chaotic, 44 disjoint hypercyclic, 44 disjoint topologically mixing, 44 disjoint topologically transitive, 44 fractional powers, 43

generalized resolvent equations, 42 hypercyclic, 44 hypercyclic vector, 45 integer powers, 38 inverse, 38 kernel, 38 MLO, 37 periodic point, 44 polynomial spectral mapping theorem, 42 product, 38 regular resolvent set, 42 relatively closed, 39, 288-290, 292, 294 resolvent equation, 41 restriction. 38 section. 38 single-valued branch, 196 stationary dense, 341 sum. 38 topologically mixing, 44 topologically transitive, 44 multivalued mappings Laplace transformable, 61 sections, 61, 288, 319 net, 28 non-negative operators, 129 norm. 28 norm continuity, 97, 101. 218 operator (B, σ) -regular, 13, 210 (B, p)-regular, 211 adjoint, 29 almost sectorial, 372 Balakrishnan, 36 Black-Scholes, 257 chaotic, 14 closed, 28 distributionally chaotic, 15 hypercyclic, 14 Laplace-Beltrami, 50, 258 linear, 28 maximal dissipative, 304 multiplication, 7, 300 Ornstein–Uhlenbeck, 233, 245, 257 positively supercyclic, 14 regular, 13, 42, 209 relatively p-radial, 309 Riesz fractional, 168 sectorial, 168 self-adjoint, 215, 306 supercyclic, 14 orbit, 241, 243, 393 distributionally *m*-unbounded, 254

distributionally near to 0, 253 projective, 14 parabolicity condition, 168 Parseval's equality, 147, 155 part of operator, 29 periodic point, 14, 235 perturbations, 20, 143 rank 1-, 403 hyperbolic, 220 time-dependent, 403, 404 Phragmén-Doetsch inversion formula, 60 point spectrum, 389 imaginary, 237, 251 polynomial matrices, 125 polynomially bounded C-resolvent, 416 pre- (k, C_1, \mathcal{V}) -existence family, 321 pre-(C-DCF), 357 exponential, 364 pre-(C-DS), 344 dense, 344 exponential. 352 pre-(C-EDCF), 364 pre-(C-EUDCF) of *-class, 364 pre-(C-UDCF) of *-class, 357 exponential, 364 pre-(C-UDS) of *-class, 344 dense, 344 exponential, 352 pre-solution, 286, 290, 310 problem $(P_{2,a,B}), 202$ $(DFP)_{2}, 97$ $(FP_{\alpha_1,\beta_1,\theta}), 11, 200$ $(FP_{\alpha_1,\beta_1,\theta})', 12$ $(FP_{\beta}), 422$ $(P)_L, 95$ $(P)_R, 121$ $(P)_{\eta,f}, 309$ $(P)_{m,\alpha}, 304$ $(PR)_{\alpha}, 442$ $(P_2), 100$ $(P_{2,q,B}), 12$ (ACP), 260 (DF), 212 $(DF)_1, 212$ $(DF)_p, 214$ $(DF)_{n,P}, 214$ (DFP), 91, 438, 439 $(DFP)_L$, 124, 223 $(DFP)'_L, 174$ $(DFP)_R, 107, 223$ $(DFP)'_{R}, 174$ $(DFP)_{\alpha,\mathcal{A}}, 389$

(DFP)_T, 17, 285 (DFP)_B, 17, 285 $(DFP)_{f,s,\gamma}, 375$ $(DFP)_{sl}, 100$ (DFP)', 125 (P).99 (PR), 186 (238), 2481.265 2, 270 $(DFP)_{F}, 239$ $(DFP)_{\alpha 4}, 390$ DC, 258 inhomogeneous, 308, 311 PN. 189 problem (151) subcase (SC1), 162 subcase (SC2), 163 subcase (SC3), 163 projection, 210 Radon-Nikodym property, 218 range, 28 Ravleigh-Stokes problem for generalized second-grade fluids, 168 regular B-resolvent set, 12, 210 removable singularity at zero, 366 renorming, 255 rescaling, 83, 88 resolution of the identity, 215 resolvent equation, 71 resolvent set, 28 Reynolds number, 270 Riemann-Liouville fractional integral, 64 Rodrigues' formula, 410 semigroup (B, C)-regularized of growth order r > 0, 195analytic, 196 C-regularized, 184, 239, 325, 326, 330, 412, 413 K-convoluted C-, 329 α -times integrated C-, 330 infinitely differentiable, 308 analytic C-regularized of growth order r > 0, 186Black-Scholes, 257 chaotic, 14 infinitely differentiable, 313 of class $(C_{(k)})$, 414 of growth order r > 0, 414pre-(B, C)-regularized of growth order r > 0, 195

analytic, 196 strongly continuous, 14, 168, 237, 257, 270.304 distributionally chaotic, 15 strongly continuous for t > 0, 366 semilinear degenerate fractional Cauchy inclusion, 375, 387 semilinear degenerate relaxation equations, 100 seminorm. 28 sequentially complete locally convex space. 57 Hausdorff. 28 singular perturbation theory, 485 solution, 286 V-, 318 \mathcal{V} -mild. 106 p-strong, 318 analytical, 125, 127, 148, 157, 159, 190, 306, 397 classical, 375, 376 entire, 125, 148, 150, 154, 189, 194, 314 local 134 mild, 76, 80, 130, 161, 165, 216, 375, 376 strong, 76, 81, 105, 130, 145, 165, 169, 174, 216, 226, 231, 286 weak. 76 space $C^{2}(\mathbb{R}), 395$ $D_{\infty}(A), 36$ $L^{p(x)}(\Omega : X), 467$ T, 129Damek-Ricci, 258 Fréchet, 76, 135 Banach. 33 barreled, 135 Fréchet, 14, 29, 44, 300 Hölder, 154, 388 Hardy, 69 Heckman-Opdam root, 228, 258 Hilbert, 33, 258 locally compact, 32, 40 measure, 33 of real analytic functions of Herzog type, 270phase, 213, 276 reflexive, 33 Riemannian symmetric, 228 Schwartz, 29, 90 separable metric, 14, 32, 40 symmetric of non-compact type, 228, 232, 245, 258, 271 webbed bornological, 29, 196

weighted function, 225, 235 space decomposition. 7 spectrum, 28 square-free polynomial, 89 stationarity, 341 Stepanov (asymptotic) almost automorphy. 445 Stepanov metric, 443 Stepanov norm, 444 subgenerator, 77, 293, 294 subordinated fractional resolvent family, 366 subordination principles, 79, 88, 115, 140, 142, 290, 292, 298, 307, 313, 319, 322, 324 symmetric spaces of non-compact type, 50 system of seminorms, 14, 28, 135, 252 systems of abstract degenerate differential equations, 125, 406 the uniform boundedness principle, 57 theorem closed graph, 266 Titchmarsh, 76 adjoint, 307, 329 Bolzano-Weierstrass, 82 Cauchy, 192 closed graph, 29, 276, 287 complex characterization, 300 Da Prato-Grisvard, 168 dominated convergence, 34, 181, 186, 192 extension type, 338 fixed point, 103 Fubini–Tonelli, 60 Hahn-Banach. 40 Herrero-Bourdon, 244 Hille-Yosida, 78, 219, 303, 408 K. Ball's planck, 258 Lagrange mean value, 98 Ljubich uniqueness, 289 Ljubich uniquness, 290 Mackey's, 57 Morera, 191 Phragmén-Lindelöf, 382 Phragmén-Lindelöf, 173 residue, 181, 186 Riesz-Fischer, 33 Seidenberg-Tarski, 90 Stone-Weierstrass, 335 Titchmarsh–Foias, 76 uniqueness for analytic functions, 204 uniqueness for Laplace transform, 61 topological transitivity, 14, 390 topologically mixing property, 13, 234, 390

translation invariant metric, 14 two-sided ideal, 387 ultra-logarithmic region, 134, 173 ultradifferentiable functions, 31 ultradifferential operator of *-class, 31 ultradistribution semigroup of Beurling class. 301, 412 ultradistributions, 30 unidirectional viscoelastic flows, 168 upper density, 16 Väisälä-Brunt frequency, 121 variation of parameters formula, 96, 133 vector \tilde{X} -(ACP)-distributionally irregular, 261 \tilde{X} -(ACP)-distributionally near to 0, 261 \tilde{X} -(ACP)-distributionally *n*-unbounded. 261height of adjoint, 211 vector-valued distributions, 30 Laplace transform, 16, 58, 160, 296, 297 Sobolev space, 32, 64, 161, 311 ultradistributions of *-class, 31

weakly mixing property, 224

Bibliography

- 1. L. Abadias, P.J. Miana, A subordination principle on Wright functions and regularized resolvent families, J. Funct. Spaces, Volume 2015 (2015), Article ID 158145, 9 pp.
- L. Abadias, P. J. Miana, C₀-semigroups and resolvent operators approximated by Laguerre expansions, J. Approx. Theory **213** (2017), 1–22.
- L. Abadias, P.J. Miana, Hermite expansions of C₀-groups and cosine families, J. Math. Anal. Appl. 426 (2015), 288–311.
- S. Abbas, M. Benchohra, Advanced Functional Evolution Equations and Inclusions, Developments in Mathematics, 39. Springer, Cham, 2015.
- P. Acquistapace, B. Terreni, A uniffied approach to abstract linear nonautonomous parabolic equations, Rend. Sem. Mat. Univ. Padova 78 (1987), 47–107.
- N. H. Abdelaziz, F. Neubrander, Degenerate abstract Cauchy problems, In: Seminar Notes in Functional Analysis and PDE, Louisiana State University 1991/1992.
- M. Adival, H. C. Koyuncuoglu, Y. F. Raffoul, Almost automorphic solutions of delayed neutral dynamic systems on hibrid domains, Appl. Anal. Discrete Math. 10 (2016), 128–151.
- K. Aissani, M. Benchohra, Controllability of fractional integrodifferential equations with state-dependent delay, J. Integral Equations Appl. 28 (2016), 149–167.
- R. P. Agarwal, B. de Andrade, C. Cuevas, Weighted pseudo-almost periodic solutions of a class of semilinear fractional differential equations, Nonlinear Anal. 11 (2010), 3532–3554.
- R. P. Agarwal, C. Cuevas, H. Soto, Pseudo-almost periodic solutions of a class of semilinear fractional differential equations, J. Appl. Math. Comput. 37 (2011), 625–634.
- El H. Alaarabiou, Calcul fonctionnel et puissance fractionnaire d' opérateurs linéaries multivoques non négatifs, C. R. Acad. Sci. Paris. Série I 313 (1991), 163–166.
- El H. Alaarabiou, Calcul fonctionnel et puissance fractionnaire d'opérateurs linéaries multivoques non négatifs, Pub. Math. Besançon, An. non linéaire, fasc. 13, 1991.
- A. A. Albanese, X. Barrachina, E. M. Mangino, A. Peris, *Distributional chaos for strongly continuous semigroups of operators*, Commun. Pure Appl. Anal. **12** (2013), 2069–2082.
- M. Al Horani, A. Favini, Degenerate second-order identification problem in Banach spaces, J. Optim. Theory Appl. 120 (2004), 305–326.
- M. Al Horani, A. Favini, Perturbation method for first-and complete second-order differential equations, J. Opt. Theory Appl. 166 (2015), 949–967.
- A. R. Aliev, A. L. Elbably, Well-posedness of a boundary value problem for a class of third-order operator-differential equations, Boundary Value Problems, 2013, 2013:140 DOI:10.1186/1687-2770-2013-140.
- A. R. Aliev, A. L. Elbably, Completeness of derivative chains for polynomial operator pencil of third order with multiple characteristics, Azerbaijan J. Math. 4 (2014), 3–9.
- A.B. Aliev, B.H. Lichaei, Existence and non-existence of global solutions of the Cauchy problem for higher order semilinear pseudo-hyperbolic equations, Nonlinear Anal. 72 (2010), 3275–3288.
- A. B. Al'shin, M. O. Korpusov, A. G. Sveshnikov, Blow Up in Nonlinear Sobolev Type Equations, De Gruyter, Berlin/New York, 2011.
- H. Amann, Operator-valued Fourier multipliers, vector-valued Besov spaces and applications, Math. Nachr. 186 (1997), 5–56.

BIBLIOGRAPHY

- 21. A. Ammar, A. Jeribi, N. Lazrag, Sequence of multivalued linear operators converging in the generalized sense, Bull. Iranian Math. Soc., accepted.
- J. Andres, A. M. Bersani, R. F. Grande, *Hierarchy of almost-periodic function spaces*, Rend. Mat. Appl. 26 (2006), 121-188.
- M. H. Annaby, Z. S. Mansour, *q-Fractional Calculus and Equations*, Lecture Notes Math. 2056, Springer-Verlag, 2012.
- 24. R. Aparicio, V. Keyantuo, Besov Maximal regularity for a class of degenerate integrodifferential equations with infinite delay in Banach spaces, Math. Methods Appl. Sci., accepted.
- 25. W. Arendt, Approximation of degenerate semigroups, Taiwanese J. Math. 5 (2001), 279–295.
- W. Arendt, C.J.K. Batty, Rank-1 perturbations of cosine functions and semigroups, J. Funct. Anal. 238 (2006), 340–352.
- W. Arendt, C. J. K. Batty, M. Hieber, F. Neubrander, Vector-valued Laplace Transforms and Cauchy Problems, Monographs in Mathematics 96, Birkhäuser/Springer Basel AG, Basel, 2001.
- W. Arendt, O. El-Mennaoui, V. Keyantuo, Local integrated semigroups: evolution with jumps of regularity, J. Math. Anal. Appl. 186 (1994), 572–595.
- W. Arendt, A. Favini, Integrated solutions to implicit differential equations, Rend. Sem. Mat. Univ. Poi. Torino 51 (1993), 315–329.
- 30. R. Arens, Operational calculus of linear relations, Pacific J. Math. 11 (1961), 9-23.
- R. M. Aron, J. B. Seoane-Sepúlveda, A. Weber, *Chaos on function spaces*, Bull. Austral. Math. Soc. **71** (2005), 411–415.
- J. M. Arrieta, A. Carvalho, G. Lozada-Cruz, Dynamics in dumbbell domains I. Continuity of the set of equilibria, J. Differential Equations 231 (2006), 551–597.
- A. Ashyralyev, Well-posedness of fractional parabolic equations, Boundary Value Problems 31 (2013), DOI:10.1186/1687-2770-2013-31.
- A. Ashyralyev, F. Dal, Z. Pinar, A note on the fractional hyperbolic differential and difference equations, Applied Math. Comp. 217 (2011), 4642–4664.
- F. Astengo, B. di Blasio, Dynamics of the heat semigroup in Jacobi analysis, J. Math. Anal. Appl. 391 (2012), 48–56.
- T. M. Atanacković, S. Pilipović, B. Stanković, D. Zorica, Fractional Calculus with Applications in Mechanics: Vibrations and Diffusion Processes, Wiley-ISTE, London, 2014.
- T. M. Atanacković, S. Pilipović, B. Stanković, D. Zorica, Fractional Calculus with Applications in Mechanics: Wave Propagation, Impact and Variational Principles, Wiley-ISTE, London, 2014.
- H.K. Avad, A.V. Glushak, On perturbations of abstract fractional differential equations by nonlinear operators, CMFD 35 (2010), 5–21.
- N. V. Azbelev, V. P. Maksimov, L. F. Rakhmatullina, Introduction to the Theory of Functional Differential Equations: Methods and Applications, Contemporary Mathematics and Its Applications, 3. Hindawi Publishing Corporation, New York, 2007.
- B. Baeumer, On the inversion of the convolution and Laplace transform, Trans. Amer. Math. Soc. 355 (2002), 1201–1212.
- B. Baeumer, M. M. Meerschaert, E. Nane, Brownian subordinators and fractional Cauchy problems, Trans. Amer. Math. Soc. 361 (2009), 3615–3630.
- K. Balachandran, S. Kiruthika, Existence of solutions of abstract fractional integrodifferential equations of Sobolev type, Comput. Math. Appl. 64 (2012), 3406–3413.
- M. K. Balaev, Higher order parabolic type evolution equations, Dokl. Akad. Nauk. Azerbaidjan 41 (1988), 7–10 (Russian).
- A. V. Balakrishnan, Fractional powers of closed operators and the semigroups generated by them, Pacific J. Math. 10 (1960), 419–437.
- D. Baleanu, K. Diethelm, E. Scalas, J. J. Trujillo, Fractional Calculus Models and Numerical Methods, Series on Complexity, Nonlinearity and Chaos, World Scientific, Singapore, 2012.
- F. Balibrea, B. Schweizer, A. Sklar, J. Smítal, Generalized specification property and distributional chaos, Internat. J. Bifur. Chaos 13 (2003), 1683–1694.

BIBLIOGRAPHY

- J. M. Ball, Strongly continuous semigroups, weak solutions, and the variation of constants formula, Proc. Amer. Math. Soc. 63 (1997), 370–373.
- J. Banasiak, M. Moszyński, A generalization of Desch-Schappacher-Webb criterion for chaos, Discrete Contin. Dyn. Syst. 12 (2005), 959-972.
- V. Barbu, A. Favini, Periodic problems for degenerate differential equations, Rend. Instit. Mat. Univ. Trieste 28(Supplement) (1997), 29–57.
- V. Barbu, A. Favini, Convergence of solutions of implicit differential equations, Differential Integral Equations 7 (1994), 665–688.
- V. Barbu, A. Favini, S. Romanelli, Degenerate evolution equations and regularity of their associated semigroups, Funkcial. Ekvac. 39 (1996), 421–448.
- G. I. Barenblatt, Yu. P. Zheltov, I. N. Kochina, Basic concepts in the theory of seepage of homogeneous liquids in fissured rocks (strata), J. Appl. Math. Mech. 24 (1961), 1286–1303.
- X. Barrachina, Distributional chaos of C₀-semigroups of operators, PhD. Thesis, Universitat Politèchnica, València, 2013.
- 54. X. Barrachina, J. A. Conejero, Devaney chaos and distributional chaos in the solution of certain partial differential equations, Abstr. Appl. Anal. **2012** (2012), Art. ID 457019, 11 pp.
- X. Barrachina, J.A. Conejero, M. Murillo-Arcila, J.B. Seoane-Sepúlveda, Distributional chaos for the Forward and Backward Control traffic model, Linear Algebra Appl. 479 (2015), 202–215.
- X. Barrachina, A. Peris, Distributionally chaotic translation semigroups, J. Difference Equ. Appl. 18 (2012), 751–761.
- A. G. Baskakov, Linear relations as generators of semigroups of operators, Math. Notes 84 (2008), 166–183.
- A. G. Baskakov, K. I. Chernyshov, On distribution semigroups with a singularity at zero and bounded solutions of differential inclusions, Math. Notes 1 (2006), 19–33.
- A. G. Baskakov, V. Obukhovskii, P. Zecca, On solutions of differential inclusions in homogeneous spaces of functions, J. Math. Anal. Appl. 324 (2006), 1310–1323.
- F. Bayart, E. Matheron, *Dynamics of Linear Operators*, vol. **179** of Cambridge Tracts in Mathematics, Cambridge University Press, Cambridge, UK, 1 edition, 2009.
- 61. E. Bazhlekova, *Fractional evolution equations in Banach spaces*, PhD. Thesis, Eindhoven University of Technology, Eindhoven, 2001.
- 62. E. Bazhlekova, Completely monotone functions and some classes of fractional evolution equations, Integral Transform. Spec. Funct. 26 (2015), 737–752.
- E. Bazhlekova, Subordination principle for a class of fractional order differential equations, Mathematica 2 (2015), 412–427.
- 64. R. Beals, On the abstract Cauchy problem, J. Funct. Anal. 10 (1972), 281-299.
- 65. R. Beals, Semigroups and abstract Gevrey spaces, J. Funct. Anal. 10 (1972), 300–308.
- T. Bermúdez, A. Bonilla, F. Martinez-Gimenez, A. Peris, *Li-Yorke and distributionally chaotic operators*, J. Math. Anal. Appl. **373** (2011), 83–93.
- 67. L. Bernal-González, Disjoint hypercyclic operators, Studia Math. 182 (2007), 113-131.
- N. C. Bernardes Jr., A. Bonilla, V. Müler, A. Peris, Distributional chaos for linear operators, J. Funct. Anal. 265 (2013), 2143–2163.
- J. Bès, C.K. Chan, S.M. Seubert, *Chaotic unbounded differentiation operators*, Integral Equations Operator Theory 40 (2001), 257–267.
- J. Bès, J. A. Conejero, An extension of hypercyclicity for N-linear operators, Abstr. Appl. Anal. 2014 (2014), Article ID 609873, 11 pp.
- J. Bès, Ö. Martin, A. Peris, S. Shkarin, *Disjoint mixing operators*, J. Funct. Anal. 263 (2013), 1283–1322.
- J. Bès, Ö. Martin, R. Sanders, Weighted shifts and disjoint hypercyclicity, J. Operator Theory 72 (2014), 15-40.
- 73. J. Bès, A. Peris, Disjointness in hypercyclicity, J. Math. Anal. Appl. 336 (2007), 297-315.
- 74. A.S. Besicovitch, Almost Periodic Functions, Dover Publications Inc., New York, 1954.

- M. S. Bichegkuev, To the theory of infinitely differentiable semigroups of operators, St. Petersburg Math. J. 22 (2011), 175–182.
- M. S. Bichegkuev, On some classes of infinitely differentiable operator semigroups, Differ. Uravn. 46 (2010), 224–238.
- 77. G. D. Birkhoff, Démonstration d'un théorème élémentaire sur les fonctions entières, C. R. Acad. Sci. Paris 189 (1929), 473–475.
- J. Blot, P. Cieutat, G. M. N'Guérékata, S-Asymptotically ω-periodic functions and applications to evolution equations, African Diaspora J.Math. 12 (2011), 113–121.
- A. Bobrowski, On Hille-type approximation of degenerate semigroups of operators, Linear Algebra Appl. 511 (2016), 31–53.
- A. Bobrowski, Singular perturbations involving fast diffusion, J. Math. Anal. Appl. 427 (2015), 1004–1026.
- H. Bohr, E. Følner, On some types of functional spaces: A contribution to the theory of almost periodic functions, Acta Math. 76 (1944), 31–155.
- Yu. V. Bogacheva, Resolution's problems of initial problems for abstract differential equations with fractional derivatives, PhD. Thesis, Belgorod, 2006.
- T. A. Bokareva, G. A. Sviridyuk, Whithney folds of the phase spaces of some semilinear Sobolev type equations, Math. Notes 55 (1993), 612–617.
- J. Bonet, L. Frerick, A. Peris, J. Wengenroth, Transitive and hypercyclic operators on locally convex spaces, Bull. London Math. Soc. 37 (2005), 254–264.
- A. Bonilla, P.J. Miana, Hypercyclic and topologically mixing cosine functions on Banach spaces, Proc. Amer. Math. Soc. 136 (2008), 519–528.
- L. V. Borel, V. E. Fedorov, On unique solvability of the system of gravitational-gyroscopic waves in the Boussinesg approximation, Chelyabinsk Phy. Math. J. 1 (2016), 16–23.
- M. Bostan, *Periodic Solutions for Evolution Equations*, Electronic J. Diff. Equ., Monograph 03, 41 pp., 2002.
- H. Brezis, On some degenerate nonlinear parabolic equations, Nonlinear Functional Analysis, Proc. Symp. Pure Math. 18 (1970), 28–38.
- H. Brill, A semilinear Sobolev evolution equation in a Banach space, J. Differential Equations 24 (1977), 412–425.
- S. Q. Bu, Well-posedness of degenerate differential equations in Hölder continuous function spaces, Front. Math. China 10 (2015), 239–248.
- S. Q. Bu, L^p-maximal regularity of degenerate delay equations with periodic conditions, Banach J. Math. Anal. 8 (2014), 49–59.
- 92. S.Q. Bu, C. Gang, Solutions of second order degenerate integro-differential equations in vector-valued function spaces, Science China Math. 56 (2013), 1059–1072.
- T. A. Burton, B. Zhang, Periodic solutions of abstract differential equations with infinite delay, J. Differential Equations 90 (1991), 357–396.
- 94. T. Cardinali, L. Santori, Boundary value problems for semilinear evolution inclusions: Carathéodory selections approach, Comment. Math. Univ. Carolin. **52** (2011), 115–125.
- R. W. Carroll, R. E. Showalter, Singular and Degenerate Cauchy Problems, Academic Press, New York, 1976.
- Y.-K. Chang, R. Ponce, Properties of solution sets for Sobolev type fractional differential inclusions via resolvent family of operators, Eur. Phys. J. Special Topics 226 (2017), 3391–3409.
- Y.-K. Chang, A. Pereira, R. Ponce, Approximate controllability for fractional differential equations of sobolev type via properties on resolvent operators, Fract. Calc. Appl. Anal. 20 (2017), 963–978.
- Y.-K. Chang, Y. Pei, R. Ponce, Existence and optimal controls for fractional stochastic evolution equations of Sobolev type via fractional resolvent operators, J. Optim. Theory Appl. (2018). https://doi.org/10.1007/s10957-018-1314-5, 15 pp.
- J. Chazarain, Problémes de Cauchy abstraites et applications à quelques problémes mixtes, J. Funct. Anal. 7 (1971), 386–446.

- D. N. Cheban, Asymptotically Almost Periodic Solutions of Differential Equations, Hindawi Publishing Corporation, 2009.
- C. Chen, M. Kostić, M. Li, Complex powers of almost C-nonnegative operators, Contemp. Anal. Appl. Math. 2 (2014), 1–77.
- C. Chen, M. Kostić, M. Li, Representation of complex powers of C-sectorial operators, Fract. Calc. Appl. Anal. 17 (2014), 827–854.
- 103. C. Chen, M. Kostić, M. Li, M. Žigić, Complex powers of C-sectorial operators. Part I, Taiwanese J. Math. 17 (2013), 465–499.
- 104. C. Chen, M. Li, On fractional resolvent operator functions, Semigroup Forum 80 (2010), 121–142.
- C. Chen, M. Li, F.-B. Li, On boundary values of fractional resolvent families, J. Math. Anal. Appl. 384 (2011), 453–467.
- C.-C. Chen, Disjoint topological transitivity for cosine operator functions on groups, Filomat 31 (2017), 2413–2423.
- 107. C.-C. Chen, J. A. Conejero, M. Kostić, M. Murillo-Arcila, Dynamics of multivalued linear operators, Open Math. 15 (2017),
- C.-C. Chen, J. A. Conejero, M. Kostić, M. Murillo-Arcila, Dynamics on binary relations over topological spaces, Symmetry (2018) 10, 211; doi:10.3390/sym10060211, 12 pp.
- 109. I. Ciorănescu, Beurling spaces of class (M_p) and ultradistribution semi-groups, Bull. Sci. Math. 102 (1978), 167–192.
- 110. I. Ciorănescu, G. Lumer, Problèmes d'évolution régularisés par un noyan général K(t). Formule de Duhamel, prolongements, théorèmes de génération, C. R. Acad. Sci. Paris Sér. I Math. **319** (1995), 1273–1278.
- I. Cioranescu, L. Zsido, ω-Ultradistributions and Their Applications to Operator Theory, in: Spectral Theory, Banach Center Publications 8, Warsawza 1982, 77–220.
- J. A. Conejero, M. Kostić, P. J. Miana, M. Murillo-Arcila, *Distributionally chaotic families* of operators on Fréchet spaces, Comm. Pure Appl. Anal. 15 (2016), 1915–1939.
- 113. J.A. Conejero, C. Lizama, M. Murillo-Arcila, On the existence of chaos for the viscous van Wijngaarden-Eringen equation, Chaos Solit. Fract. 89 (2016), 100–104.
- 114. J.A. Conejero, C. Lizama, F. Rodenas, Chaotic behaviour of the solutions of the Moore-Gibson-Thompson equation, Appl. Math. Inf. Sci. 9 (2015), 1–6.
- J. A. Conejero, E. Mangino, Hypercyclic semigroups generated by Ornstein-Uhlenbeck operators, Mediterr. J. Math. 7 (2010), 101–109.
- 116. J.A. Conejero, F. Martinez-Gimenez, *Chaotic differential operators*, Rev. R. Acad. Cienc. Exactas Fis. Nat. Ser. A Math. RACSAM **105** (2011), 423–431.
- J.A. Conejero, A. Peris, *Chaotic translation semigroups*, Discrete Contin. Dyn. Syst. Supplement (2007), 269–276.
- J. A. Conejero, F. Ródenas, M. Trujillo, *Chaos for the hyperbolic bioheat equation*, Discrete Contin. Dyn. Syst. Ser. A **35** (2015), 653–668.
- 119. C. Corduneanu, Almost Periodic Oscillations and Waves, Springer-Verlag, New York, 2009.
- 120. R. Cross, Multivalued Linear Operators, Marcel Dekker Inc., New York, 1998.
- 121. R. Cross, A. Favini, Y. Yakubov, Perturbation results for multivalued linear operators, in: Parabolic Poblems, Progress in Nonlinear Differential Equations and Their Applications 80 (2011), The Herbert Amann Festschrift, J. Escher et al. eds., 111–130.
- 122. R. A. A. Cuello, *Traslaciones y semigrupos C0 topológicamente mezclantes*, PhD. Thesis, Universidad de Puerto Rico, Recinto Universitario de Mayagëz, 2008.
- 123. R. A. A. Cuello, Well-posedness of degenerate integro-differential equations with infinite delay in Banach spaces, PhD. Thesis, Universidad de Puerto Rico, Río Piedras Campus, 2015.
- 124. C. Cuevas, C. Lizama, Almost automorphic solutions to a class of semilinear fractional differential equations, Appl. Math. Lett. 21 (2008), 1315–1319.
- 125. C. Cuevas, M. Pierri, A. Sepulveda, Weighted S-asymptotically ω-periodic solutions of a class of fractional differential equations, Adv. Diff. Equ. 2011 (2011), Article ID 584874.
- 126. G. Da Prato, Semigruppi regolarizzabilli, Ricerche Mat. 15 (1966), 223-248.

- 127. G. Da Prato, P. Grisvard, Sommes d'opérateurs linéaires et équations différentielles opérationnelles, J. Math. Pures Appl. 54 (1975), 305–387.
- 128. S. Das, Functional Fractional Calculus, Springer-Verlag, Berlin, 2011.
- 129. M. De La Rosa, C. Read, A hypercyclic operator whose direct sum is not hypercyclic, J. Operator Theory 61 (2009), 369–380.
- E. M. de Jager, Singular perturbations of hyperbolic type, Nieuw Arch. Wisk. 23 (1975), 145–172.
- 131. E. C. de Oliveira, J. A. T. Machado, A review of definitions for fractional derivatives and integrals, Mathematical Problems in Engineering, vol. 2014, Article ID 238459, 6 pp.
- R. deLaubenfels, Existence Families, Functional Calculi and Evolution Equations, Springer-Verlag, New York, 1994.
- 133. R. deLaubenfels, Inverses of generators, Proc. Amer. Math. Soc. 104 (1988), 443-448.
- R. deLaubenfels, Inverses of generators of nonanalytic semigroups, Studia Math. 191 (2009), 11–38.
- R. deLaubenfels, Inverses of generators of integrated or regularized semigroups, Semigroup Forum 75 (2007), 457–463.
- R. deLaubenfels, Existence and uniqueness families for the abstract Cauchy problem, J. London Math. Soc. s2-44 (1991), 310–338.
- R. deLaubenfels, H. Emamirad, K.-G. Grosse-Erdmann, Chaos for semigroups of unbounded operators, Math. Nachr. 261/262 (2003), 47–59.
- R. deLaubenfels, F. Yao, S. W. Wang, Fractional powers of operators of regularized type, J. Math. Anal. Appl. 199 (1996), 910–933.
- 139. K. Deimling, Multivalued Differential Equations, W. de Gruyter, Berlin, 1992.
- 140. G. V. Demidenko, S. V. Uspenskii, Partial Differential Equations And Systems Not Solvable With Respect To The Highest-Order Derivative, Pure and Applied Mathematics Series 256, CRC Press, New York, 2003.
- W. Desch, R. Grimmer, Propagation of singularities for integro-differential equations, J. Differential Equations 65 (1986), 411–426.
- W. Desch, R. Grimmer, W. Schappacher, Some considerations for linear integro-differential equations, J. Math. Anal. Appl. 104 (1984), 219–234.
- 143. W. Desch, W. Schappacher, G. F. Webb, Hypercyclic and chaotic semigroups of linear operators, Ergodic Theory Dynam. Systems 17 (1997), 1–27.
- 144. T. Diagana, Almost Automorphic Type and Almost Periodic Type Functions in Abstract Spaces, Springer, New York, 2013.
- 145. T. Diagana, Stepanov-like pseudo-almost periodicity and its applications to some nonautonomous differential equations, Nonlinear Anal. 69 (2008), 4277–4285.
- T. Diagana, G.M. N'Guérékata, Almost automorphic solutions to semilinear evolution equations, Funct. Differ. Equ. 13 (2006), 195–206.
- 147. T. Diagana, G.M. N'Guérékata, Almost automorphic solutions to some classes of partial evolution equations, Appl. Math. Lett. 20 (2007), 462–466.
- 148. T. Diagana, M. Kostić, Almost periodic and asymptotically almost periodic type functions in Lebesgue spaces with variable exponents $L^{p(x)}$, Filomat, in press.
- 149. T. Diagana, M. Kostić, Almost automorphic and asymptotically almost automorphic type functions in Lebesgue spaces with variable exponents $L^{p(x)}$, Chapter in Book: Recent Studies in Differential Equations, Nova Science Publishers, Inc., New York, in press.
- 150. T. Diagana, M. Zitane, Weighted Stepanov-like pseudo-almost periodic functions in Lebesgue space with variable exponents $L^{p(x)}$, Afr. Diaspora J. Math. **15** (2013), 56–75.
- 151. T. Diagana, M. Zitane, Stepanov-like pseudo-almost automorphic functions in Lebesgue spaces with variable exponents $L^{p(x)}$, Electron. J. Differential Equations 188 (2013), 20 pp.
- 152. L. Diening, P. Harjulehto, P. Hästüso, M. Ruzicka, Lebesgue and Sobolev Spaces with Variable Exponents, Lecture Notes in Mathematics, 2011. Springer, Heidelberg, 2011.
- 153. K. Diethelm, *The Analysis of Fractional Differential Equations*, Springer-Verlag, Berlin, 2010.

- 154. W. Dimbour, S. M. Manou-Abi, Asymptotically ω-periodic functions in the Stepanov sense and its application for an advanced differential equation with piecewise constant argument in a Banach space, Mediterranean J. Math. 15:25 (2018), https://doi.org/10.1007/s00009-018-1071-61660-5446/18/010001-18.
- 155. H.-S. Ding, S.-M. Wan, Asymptotically almost automorphic solutions of differential equations with piecewise constant argument, Open Math. 15 (2018), 595–610.
- T. Dlotko, Semilinear Cauchy problems with almost sectorial operators, Bull. Pol. Acad. Sci. Math. 55 (2007), 333–346.
- 157. T. Dlotko, Navier-Stokes equation and its fractional approximations, Appl. Math. Optim., in press: DOI 10.1007/s00245-016-9368-y.
- G. Doetsch, Theorie und Anwendung der Laplace-Transformation, Verlag Julius Springer, Berlin, 1937.
- 159. J. Duan, X.-C. Fu, P.-D. Liu, A. Manning, A linear chaotic quantum harmonic oscillator, Appl. Math. Lett. 12 (1999), 15–19.
- 160. S. Dugowson, Les différentielles métaphysiques: historie et philosophie de la généralisation de l'ordre de dérivation, PhD. Thesis, University of Paris, 1994.
- R. Dzhabarzadeh, Eds., Spectral Theory of Multiparameter Operator Pencils and Its Applications, Pure Appl. Math. J., Special Issues, 2015.
- 162. T. Eisner, H. Zwart, The growth of a C₀-semigroup characterised by its cogenerator, J. Evol. Equ. 8 (2008), 749–764.
- S. El Mourchid, The imaginary point spectrum and hypercyclicity, Semigroup Forum 76 (2006), 313–316.
- 164. S. El Mourchid, G. Metafune, A. Rhandi, J. Voigt, On the chaotic behaviour of size structured cell population, J. Math. Anal. Appl. 339 (2008), 918–924.
- 165. A. M. A. El-Sayed, Fractional-order evolutionary integral equations, Appl. Math. Comput. 98 (1999), 139–146.
- 166. A. M. A. El-Sayed, H. H. G. Hashem, E. A. A. Ziada, Picard and Adomian decomposition methods for a coupled system of quadratic integral equations of fractional order, J. Nonlinear Anal. Optim. 3 (2012), 171–183.
- 167. H. A. Emami-Rad, Les semi-groupes distributions de Beurling, C. R. Acad. Sci. Série A 276 (1973), 117–119.
- H. Emamirad, G. R. Goldstein, J. A. Goldstein, *Chaotic solution for the Black–Scholes equa*tion, Proc. Amer. Math. Soc. 140 (2012), 2043–2052.
- H. Emamirad, A. Rougirel, Solution operators of three time variables for fractional linear problems, Math. Methods Appl. Sci. (2016), DOI: 10.1002/mma.4079.
- 170. K. J. Engel, On singular perturbations of second order Cauchy problems, Pacific J. Math. 152 (1992), 79–91.
- K.-J. Engel, R. Nagel, One-Parameter Semigroups for Linear Evolution Equations, Springer-Verlag, Berlin, 2000.
- 172. S. Fackler, A short counterexample to the inverse generator problem on non-Hilbertian reflexive L^p-spaces, Arch. Math. 106 (2016), 383–389.
- 173. M.V. Falaleev, S.S. Orlov, Degenerate integro-differential operators in Banach spaces and their applications, Russian Math. (Iz. VUZ) 55 (2011), 59–69.
- 174. M.V. Falaleev, S.S. Orlov, Continuous and generalized solutions of singular integrodifferential equations in Banach spaces, IIGU Ser. Matematika 5 (2012), 62–74.
- 175. M. V. Falaleev, S. S. Orlov, Integro-differential equations with degeneration in Banach spaces and it's applications in mathematical theory of elasticity, IIGU Ser. Matematika 4 (2011), 118–134.
- 176. X. L. Fan, D. Zhao, On the spaces L^{p(x)}(O) and W^{m,p(x)}(O), J. Math. Anal. Appl. 263 (2001), 424–446.
- 177. H.O. Fattorini, The Cauchy Problem, Addison-Wesley, 1983. MR84g:34003.
- 178. H.O. Fattorini, Second Order Linear Differential Equations in Banach Spaces, North-Holland Math. Stud. 108, North-Holland, Amsterdam, 1985.

- H. O. Fattorini, The hyperbolic singular perturbation problem: an operator theoretic approach, J. Differential Equations 70 (1987), 1–41.
- A. Favaron, Optimal time and space regularity for solutions of degenerate differential equations, Cent. Eur. J. Math. 7 (2009), 249–271.
- A. Favaron, A. Favini, Fractional powers and interpolation theory for multivalued linear operators and applications to degenerate differential equations, Tsukuba J. Math. 35 (2011), 259–323.
- 182. A. Favaron, A. Favini, On the behaviour of singular semigroups in intermediate and interpolation spaces and its applications to maximal regularity for degenerate integro-differential evolution equations, Abstr. Appl. Anal. 2013 (2013), Art. ID 275494, 37 pp.
- A. Favaron, A. Favini, Maximal time regularity for degenerate evolution integro-differential equations, J. Evol. Equ. 10 (2010), 377-412.
- 184. A. Favaron, A. Favini, H. Tanabe, *Petrubation methods for inverse problems on degenerate differential equations*, preprint.
- A. Favini, An operational method for abstract degenerate evolution equations of hyperbolic type, J. Funct. Anal. 76 (1988), 432–456.
- A. Favini, Perturbative methods for inverse problems on degenerate differential equations, preprint. DOI:10.6092/issn.2240-2829/3422.
- 187. A. Favini, M. Fuhrman, Approximation results for semigroups generated by multivalued linear operators and applications, Differential Integral Equations 5 (1989), 781–805.
- 188. A. Favini, A. Lorenzi, H. Tanabe, Degenerate integrodifferential equations of parabolic type, in: Differential Equations: Inverse and Direct Problems. Chapman and Hall/CRC Press, Boca Raton, 2006, pp. 91-109.
- 189. A. Favini, A. Lorenzi, H. Tanabe, Degenerate integrodifferential equations of parabolic type with Robin boundary conditions: L²-theory, J. Math. Soc. Japan **61** (2009), 133–176.
- A. Favini, G. Marinoschi, Degenerate Nonlinear Diffusion Equations, Lecture Notes in Mathematics 2049, Springer-Verlag, Berlin, 2012.
- A. Favini, G. Marinoschi, Periodic behavior for a degenerate fast diffusion equation, J. Math. Anal. Appl. 351 (2009), 509–521.
- 192. A. Favini, F. Plazzi, Some results concerning the abstract degenerate nonlinear equation (d/dt)Mu(t) + Lu(t) = f(t, Ku(t)), Circuits Systems Signal Process. 5 (1986), 261–274.
- 193. A. Favini, A. Rutkas, Existence and uniqueness of solutions of some abstract degenerate nonlinear equations, Differential Integral Equations 12 (1999), 343–394.
- 194. A. Favini, G. A. Sviridyuk, N. A. Manakova, *Linear Sobolev type equations with relatively p-sectorial operators in space of "noise"*, Abstract Appl. Anal. **2015** (2015), Article ID 697410, 8 pp.
- 195. A. Favini, H. Tanabe, Degenerate Volterra equations in Banach spaces, in: International Conference on Differential Equations and Related Topics (Pusan, 1999). J. Korean Math. Soc. 37 (2000), 915–927.
- 196. A. Favini, H. Tanabe, Degenerate Volterra equations in Banach spaces, Differential Integral Equations 14 (2001), 613–640.
- 197. A. Favini, H. Tanabe, Laplace transform method for a class of degenerate evolution equations, Rend. Univ. Padova 3-4 (1979), 511–536.
- A. Favini, L. Vlasenko, Degenerate non-stationary differential equations with delay in Banach spaces, J. Differential Equations 192 (2003), 93–110.
- 199. A. Favini, A. Yagi, *Degenerate Differential Equations in Banach Spaces*, Chapman and Hall/CRC Pure and Applied Mathematics, New York, 1998.
- 200. A. Favini, A. Yagi, Multivalued linear operators and degenerate evolution equations, Annali di Mat. Pura Appl. 163 (1993), 353–384.
- A. Favini, A. Yagi, Quasilinear degenerate evolution equations in Banach spaces, J. Evolution Equ. 4 (2004), 353–384.
- 202. V. E. Fedorov, A generalization of the Hille-Yosida theorem to the case of degenerate semigroups in locally convex spaces, Siberian Math. J. 46 (2005), 333–350 (Russian).

- V. E. Fedorov, A class of second-order equations of Sobolev type and degenerate groups of operators, Vestn. Chelyab. Gos. Univ. Mat. Mekh. Inform. 26(13) (2011), 59–75.
- V. E. Fedorov, Holomorphic solution semigroups for Sobolev-type equations in locally convex spaces, Mat. Sb. 195 (2004), 131–160.
- V. E. Fedorov, Strongly holomorphic groups of linear equations of Sobolev type in locally convex spaces, Differ. Uravn. 40 (2004), 753–765.
- V. E. Fedorov, A. S. Avilovič, The abstract Cauchy problem for fractional differential equations with Riemann-Liouville derivatives, Siberian Math. J. 60 (2019), 463–477 (Russian).
- 207. V. E. Fedorov, L. V. Borel, Solvability of loaded linear evolution equations with a degenerate operator at the derivative, St. Petersburg Math. J. 26 (2015), 487–497.
- V. E. Fedorov, P. N. Davydov, On nonlocal solutions of semilinear equations of the Sobolev type, Differ. Uravn. 49 (2013), 326–335.
- V. E. Fedorov, P. N. Davydov, Global solvability of some Sobolev type semilinear equations, Vestnik Chelyabinsk. Univ. Ser. 3 Mat. Mekh. Inform. 12 (2010), 80–87.
- V. E. Fedorov, A. Debbouche, A class of degenerate fractional evolution systems in Banach spaces, Differ. Uravn. 49 (2013), 1569–1576.
- V. E. Fedorov, D. M. Gordievskikh, Resolving operators of degenerate evolution equations with fractional derivative with respect to time, Russian Math. (Iz. VUZ) 1 (2015), 71–83.
- V. E. Fedorov, D. M. Gordievskikh, M. K. Plekhanova, Equations in Banach spaces with a degenerate operator under a fractional derivative, Differ. Uravn. 51 (2015), 1360–1368.
- V. E. Fedorov, M. Kostić, On a class of abstract degenerate multi-term fractional differential equations in locally convex spaces, Eurasian Math. J. 9 (2018), 33–57.
- V. E. Fedorov, M. Kostić, Degenerate fractional differential equations in locally convex spaces with a σ-regular pair of operators, UFA Math. J. 8 (2016), 100–113 (Russian).
- V. E. Fedorov, M. Kostić, Disjoint hypercyclic and disjoint topologically mixing properties of degenerate fractional differential equations, Russian Math. 7 (2018), 36–53.
- 216. V. E. Fedorov, A. V. Nagumanova, Inverse problem for evolutionary equation with the Gerasimov-Caputo fractional derivative in the sectorial case, Bulletin Irkutsk State University. Series Mathematics 28 (2019), 123–137.
- 217. V. E. Fedorov, E. A. Omel'chenko, Linear equations of the Sobolev type with integral delay operator, Siberian Math. J. 53 (2012), 335-344.
- V. E. Fedorov, E. A. Omel'chenko, Inhomogeneous degenerate Sobolev type equations with delay, Russian Math. (Iz. VUZ) 58 (2014), 60–69.
- V. E. Fedorov, O. A. Ruzakova, On solvability of perturbed Sobolev type equations, St. Petersburg Math. J. 4 (2009), 645–664.
- C. Foias, Remarques sur les semi-groupes distributions d'opérateurs normaux, Port. Math. 19 (1960), 227–242.
- M. Fuhrman, Sums of generators of analytic semigroups and multivalued linear operators, Annali di Mat. Pura Appl. 173 (1997), 63–105.
- 222. S. A. Gabov, A. G. Sveshnikov, Linear Problems in the Theory of Non-Steady-State Internal Waves, Nauka, Moskow, 1990.
- 223. R. Geel, Singular Perturbations of Hyperbolic Type, Mathematical Centre Tracts No. 98, Mathematische Centrum, Amsterdam, 1978.
- 224. R. M. Gethner, J. Shapiro, Universal vectors for operators on space of holomorphic functions, Proc. Amer. Math. Soc. 100 (1987), 281–288.
- M. I. Gil', Operator Functions and Localization of Spectra, Lecture Notes in Mathematics 1830, Springer-Verlag, Berlin, 2003.
- 226. S. G. Gindikin, L. R. Volevich, *Distributions and Convolution Equations*, Gordon and Breach Sci. Publ., 1992.
- 227. A.V. Glushak, Cauchy-type problem for an abstract differential equation with fractional derivatives, Math. Notes 77 (2005), 26–38.
- 228. A.V. Glushak, The problem of Cauchy-type for an abstract differential equation with fractional derivative, Vestn. Voronezh. Gos. Univ., Ser. Fiz. Mat. 2 (2001), 74–77 (Russian).

- 229. A. V. Glushak, The problem of Cauchy-type for an abstract inhomogeneous differential equation with fractional derivative, Vestn. Voronezh. Gos. Univ., Ser. Fiz. Mat. 1 (2002), 121–123 (Russian).
- A. V. Glushak, On the properties of a Cauchy-type problem for an abstract differential equation with fractional derivatives, Mat. Zametki 82 (2007), 665–677.
- A. V. Glushak, T. A. Manaenkova, Direct and inverse problems for an abstract differential equation containing Hadamard fractional derivatives, Differ. Uravn. 47 (2011), 1307–1317.
- A. Gomilko, Inverses of semigroup generators: a survey and remarks, Banach Center Publ. 112 (2017), 107–142.
- 233. A. Gomilko, H. Zwart, Y. Tomilov, On the inverse of the generator of C₀-semigroup, Mat. Sb. 198 (2007), 35–50 (in Russian); English transl.: Sb. Math. 198 (2007), 1095–1110.
- R. Gorenflo, Y. Luchko, F. Mainardi, Analytical properties and applications of the Wright function, Fract. Calc. Appl. Anal. 2 (1999), 383–414.
- R. Grimmer, J. Liu, Singular perturbations in linear viscoelasticity, Rocky Mountain J. Math. 24 (1994), 61–75.
- K.-G. Grosse-Erdmann, Universal families and hypercyclic operators, Bull. Amer. Math. Soc. 36 (1999), 345–381.
- 237. K.-G. Grosse-Erdmann, S.G. Kim, Bihypercyclic bilinear mappings, J. Math. Anal. Appl. 399 (2013), 701–708.
- 238. K.-G. Grosse-Erdmann, A. Peris, Linear Chaos, Springer-Verlag, London, 2011.
- 239. G. M. N'Guérékata, Topics in Almost Automorphy, Springer-Verlag, New York, 2005.
- G. M. N'Guérékata, Almost Automorphic and Almost Periodic Functions in Abstract Spaces, Kluwer Acad. Publ., Dordrecht, 2001.
- 241. G. M. N'Guérékata, M. Kostić, Generalized asymptotically almost periodic and generalized asymptotically almost automorphic solutions of abstract multiterm fractional differential inclusions, Abstract Appl. Anal., volume 2018, Article ID 5947393, 17 pages https://doi.org/10.1155/2018/5947393.
- G. M. N'Guérékata, A. Pankov, Stepanov-like almost automorphic functions and monotone evolution equations, Nonlinear Anal. 68 (2008), 2658–2667.
- 243. A. Guzman, Fractional-power semigroups of growth n, J. Funct. Anal. 30 (1978), 223–237.
- A. Guzman, Further study of growth of fractional-power semigroups, J. Funct. Anal. 29 (1978), 133–141.
- 245. M. Haase, The Functional Calculi for Sectorial Operators, Birkhauser-Verlag, Basel, 2006.
- 246. J. K. Hale, S. M. Verduyn Lunel, Introduction to Functional Differential Equations, Applied Mathematical Sciences 99. Springer-Verlag, New York, 1993.
- 247. S. Hariyanto, L. Aryati, Widodo, Generalized nonhomogeneous abstract degenerate Cauchy problem, Appl. Math. Sci. 7 (2013), 2441–2453.
- 248. M. Hasanov, Spectral problems for operator pencils in non-separated root zones, Turk. J. Math. 31 (2007), 43–52.
- H. R. Henríquez, On Stepanov-almost periodic semigroups and cosine functions of operators, J. Math. Anal. Appl. 146 (1990), 420–433.
- H. R. Henríquez, Asymptotically periodic solutions of abstract differential equations, Nonlinear Anal. 80 (2013), 135–149.
- 251. H. R. Henríquez, M. Pierri, P. Táboas, On S-asymptotically ω-periodic functions on Banach spaces and applications, J. Math. Appl. Anal. 343 (2008), 1119–1130.
- 252. D. Herceg, N. Krejić, Numerical Analysis, Stylos, Novi Sad, 1997 (Serbian).
- R. Hermann, Fractional Calculus: An Introduction for Physicists, World Scientific, 2nd edition, Singapore, 2014.
- M. Hieber, Integrated semigroups and differential operators on L^p, PhD. Thesis, Universität Tübingen, 1989.
- 255. R. Hilfer, Applications of Fractional Calculus in Physics, World Scientific Publ. Co., 2000.

- 256. Y. Hino, T. Naito, N. V. Minh, J. S. Shin, Almost Periodic Solutions of Differential Equations in Banach Spaces, Stability and Control: Theory, Methods and Applications, 15. Taylor and Francis Group, London, 2002.
- 257. L. Hörmander, Estimates for translation invariant operators in Lp spaces, Acta Math. 104 (1960), 93–140.
- 258. N.D. Ivanovna, Inverse problem for a linearized quasi-stationary phase field model with degeneracy, Vestnik YuUrGU. Ser. Mat. Model. Progr. 6 (2013), 128–132.
- L. Ji, A. Weber, Dynamics of the heat semigroup on symmetric spaces, Ergodic Theory Dynam. Systems 30 (2010), 457–468.
- 260. H. Jiang, F. Liu, I. Turner, K. Burrage, Analytical solutions for the multi-term time-space Caputo-Riesz fractional advection-diffusion equations on a finite domain, J. Math. Anal. Appl. 389 (2012), 1117–1127.
- E. Jordá, Vitali's and Harnack's type results for vector-valued functions, J. Math. Anal. Appl. 327 (2007), 739–743.
- M. Jung, Duality theory for solutions to Volterra integral equation, J. Math. Anal. Appl. 230 (1999), 112–134.
- T. Kalmes, Hypercyclic, mixing, and chaotic C₀-semigroups, PhD. Thesis, Universität Trier, 2006.
- T. Kalmes, Hypercyclic, mixing, and chaotic C₀-semigroups induced by semiflows, Ergodic Theory Dynam. Systems 27 (2007), 1599–1631.
- T. Kalmes, Hypercyclic C₀-semigroups and evolution families generated by first order differential operators, Proc. Amer. Math. Soc. 137 (2009), 3833–3848.
- 266. T. Kalmes, Hypercyclicity and mixing for cosine operator functions generated by second order partial diiferential operators, J. Math. Anal. Appl. 365 (2010), 363–375.
- 267. M. Kamenskii, V. Obukhovskii, On periodic solutions of differential inclusions with unbounded operators in Banach spaces, Novi Sad J. Math. 21 (1997), 173–191.
- M. Kamenskii, V. Obukhovskii, P. Zecca, Condensing Multivalued Maps and Semilinear Differential Inclusions in Banach Spaces, de Gruyter, Berlin-New York, 2001.
- 269. A. Karczewska, Stochastic Volterra equations of nonscalar type in Hilbert space, Transactions XXV International Seminar on Stability Problems for Stochastic Models (eds. C. D'Apice et al.). University of Salerno, 78-83, 2005.
- 270. M. D. A. Kassim, Well-posedness for a Cauchy fractional differential problem with Hilfer type fractional derivative, PhD. Thesis, King Fahd University of Petroleum and Minerals (Saudi Arabia), Dhahran, 2011.
- 271. T. D. Ke, C. T. Kinh, Generalized Cauchy problem involving a class of degenerate fractional differential equations, Dyn. Contin. Discrete Impuls. Syst. Ser. A: Math. Anal. 21 (2014), 449–472.
- 272. A.V. Keller, S.A. Zagrebina, Some generalizations of the Showalter-Sidorov problem for Sobolev-type models, Vestnik YuUrGU. Ser. Mat. Model. Progr. 8 (2015), 5–23.
- J. Kevorkian, J. Cole, Perturbation Methods in Applied Mathematics, Springer-Verlag, New York, 1981.
- 274. V. Keyantuo, C. Lizama, A characterization of periodic solutions for time-fractional differential equations in UMD spaces and applications, Math. Nachr. 284 (2011), 494–506.
- V. Keyantuo, C. Lizama, On a connection between powers of operators and fractional Cauchy problems, J. Evol. Equ. 12 (2012), 245–265.
- V. Keyantuo, C. Lizama, V. Poblete, Periodic solutions of integro-differential equations in vector-valued function spaces, J. Differential Equations 246 (2009), 1007–1037.
- 277. V. Keyantuo, C. Lizama, M. Warma, Spectral criteria for solvability of boundary value problems and positivity of solutions of time-fractional differential equations, Abstr. Appl. Anal. 2013 (2013), Article ID 614328, 11 pp.
- V. Keyantuo, P. J. Miana, L. Sánchez-Lajusticia, Sharp extensions for convoluted solutions of abstract Cauchy problems, Integral Equations Operator Theory 77 (2013), 211–241.

- A. A. Kilbas, H. M. Srivastava, J. J. Trujillo, Theory and Applications of Fractional Differential Equations, Elsevier Science B.V., Amsterdam, 2006.
- 280. M. Kim, Volterra inclusions in Banach spaces, Rocky Mountain J. Math. 32 (2002), 167-178.
- V. Kiryakova, Generalized Fractional Calculus and Applications, Longman Scientific & Technical, Harlow, 1994, copublished in the USA with John Wiley & Sons, Inc., New York.
- J. Kisyński, Distribution semigroups and one parameter semigroups, Bull. Polish Acad. Sci. 50 (2002), 189–216.
- 283. J. Kisyński, On second order Cauchy's problem in a Banach space, Bull. Acad. Polon. Sci. Ser. Sci. Math. Astronom. Phys. 18 (1970), 371–374.
- 284. C. Kitai, Invariant closed sets for linear operators, PhD. Thesis, University of Toronto, 1982.
- C. Knuckles, F. Neubrander, Remarks on the Cauchy problem for multi-valued linear operators, In: Partial Differential Equations (Hansur- Lesse, 1993), pages 174–187. Akademie-Verlag, Berlin, 1994.
- 286. H. Komatsu, Ultradistributions, I. Structure theorems and a characterization, J. Fac. Sci. Univ. Tokyo Sect. IA Math. 20 (1973), 25–105.
- 287. H. Komatsu, Ultradistributions, II. The kernel theorem and ultradistributions with support in a manifold, J. Fac. Sci. Univ. Tokyo Sect. IA Math. 24 (1977), 607–628.
- H. Komatsu, Ultradistributions, III. Vector valued ultradistributions. The theory of kernels, J. Fac. Sci. Univ. Tokyo Sect. IA Math. 29 (1982), 653–718.
- H. Komatsu, Operational calculus and semi-groups of operators, in: Functional Analysis and Related topics (Kyoto), Springer, Berlin, 213–234, 1991.
- 290. H. Komatsu, Fractional powers of operators, Pacific J. Math. 19 (1966), 285-346.
- M. Kostić, Generalized Semigroups and Cosine Functions, Mathematical Institute SANU, Belgrade, 2011.
- M. Kostić, Abstract Volterra Integro-Differential Equations, CRC Press, Boca Raton, FL, 2015.
- M. Kostić, Almost Periodic and Almost Automorphic Solutions to Integro-Differential Equations, W. de Gruyter, Berlin, 2019.
- 294. M. Kostić, Chaos for Linear Operators and Abstract Differential Equations, Nova Science Publishers, New York, 2020.
- 295. M. Kostić, (a, k)-regularized C-resolvent families: regularity and local properties, Abstr. Appl. Anal. 2009 (2009), Article ID 858242, 27 pp.
- 296. M. Kostić, Abstract Volterra equations in locally convex spaces, Sci. China Math. 55 (2012), 1797–1825.
- 297. M. Kostić, Some contributions to the theory of abstract Volterra equations, Int. J. Math. Anal. (Russe) 5 (2011), 1529–1551.
- 298. M. Kostić, Abstract time-fractional equations: existence and growth of solutions, Fract. Calculus Appl. Anal. 14 (2011), 301–316.
- M. Kostić, Generalized well-posedness of hyperbolic Volterra equations of non-scalar type, Ann. Acad. Rom. Sci. Ser. Math. Appl. 6 (2014), 19–45.
- M. Kostić, Hypercyclicity and mixing for abstract time-fractional equations, Dyn. Syst. 27 (2012), 213–221.
- M. Kostić, Hypercyclic and topologically mixing properties of abstract time-fractional equations with discrete shifts, Sarajevo J. Math. 9 (2013), 1–13.
- 302. M. Kostić, Hypercyclic and topologically mixing properties of certain classes of abstract time fractional equations, In: Difference Equations, Discrete Dynamical Systems and Applications ICDEA, Barcelona, Spain, July 2012; Lluís Alsedà i Soler et al Eds., Springer Proceedings in Mathematics & Statistics 180 (2016), pp. 150-170.
- M. Kostić, Regularization of some classes of ultradistribution semigroups and sines, Publ. Inst. Math., Nouv. Sér. 87 (2010), 9–37.
- M. Kostić, Abstract differential operators generating fractional resolvent families, Acta Math. Sin. (Engl. Ser.) 30 (2014), 1989–1998.

- 305. M. Kostić, On entire solutions of abstract degenerate differential equations of higher order, Funct. Anal. Appr. Comp. 8 (2016), 51–60.
- M. Kostić, Degenerate abstract Volterra equations in locally convex spaces, Filomat 31 (2017), 597–619.
- 307. M. Kostić, Degenerate multi-term fractional differential equations in locally convex spaces, Publ. Inst. Math., Nouv. Sér. 100 (2016), 49–75.
- M. Kostić, Hypercyclic and topologically mixing properties of degenerate multi-term fractional differential equations, Diff. Eqn. Dyn. Sys. 24 (2016), 475–498.
- M. Kostić, *D-Hypercyclic and D-topologically mixing properties of degenerate multi-term fractional differential equations*, Azerbaijan J. Math. 5 (2015), 78–95.
- M. Kostić, A note on semilinear fractional equations governed by abstract differential operators, An. Stiint, Univ. Al. I. Cuza Iaşi. Mat. (N.S.) 2 (2016), 757–762.
- M. Kostić, Entire and analytical solutions of certain classes of abstract degenerate fractional differential equations and their systems, Chelyabinsk Phy. Math. J. 4 (2019), 445–460.
- 312. M. Kostić, Abstract degenerate Volterra integro-differential inclusions in locally convex spaces, Electronic J. Diff. Equ., submitted.
- 313. M. Kostić, (C, B)-resolvents of closed linear operators, Novi Sad J. Math., submitted.
- M. Kostić, Degenerate k-regularized (C₁, C₂)-existence and uniqueness families, CUBO 17 (2015), 15–41.
- M. Kostić, Abstract degenerate multi-term fractional differential equations with Riemann-Liouville derivatives, Bull. Cl. Sci. Math. Nat. Sci. Math. 41 (2016), 1–20.
- M. Kostić, Approximation and convergence of degenerate (a, k)-regularized C-resolvent families, Bull. Cl. Sci. Math. Nat. Sci. Math. 42 (2017), 69–83.
- 317. M. Kostić, Abstract degenerate incomplete Cauchy problems, Tsukuba J. Math. 40 (2016), 29–53.
- M. Kostić, Abstract degenerate non-scalar Volterra equations, Chelyabinsk Phy. Math. J. 1 (2016), 104–112.
- M. Kostić, A note on semilinear degenerate relaxation equations associated with abstract differential operators, Chelyabinsk Phy. Math. 1 (2016), 85–93.
- M. Kostić, Distributionally chaotic properties of abstract fractional differential equations, Novi Sad J. Math. 45 (2015), 201–213.
- 321. M. Kostić, Hypercyclic and topologically mixing properties of abstract degenerate timefractional inclusions, Novi Sad J. Math. 47 (2017), 49–61.
- M. Kostić, The existence of distributional chaos in abstract degenerate fractional differential equations, J. Fract. Calc. Appl. 7 (2016), 53–74.
- M. Kostić, Perturbation results for abstract degenerate Volterra integro-differential equations, J. Fract. Calc. Appl. 9 (2018), 137–152.
- 324. M. Kostić, On a class of abstract degenerate fractional differential equations of parabolic type, Comment. Math. Univ. Carolin. 59 (2018), 81–101.
- M. Kostić, Abstract degenerate fractional differential inclusions, Appl. Anal. Discrete Math. 11 (2017), 39–61.
- 326. M. Kostić, Approximations and convergence of (a, k)-regularized C-resolvent families, Numerical Funct. Anal. Appl. 35 (2014), 1579–1606.
- 327. M. Kostić, Some contributions to the theory of abstract degenerate Volterra integrodifferential equations, J. Math. Stat. 12 (2016), 65–76.
- M. Kostić, Li-Yorke chaotic properties of abstract differential equations of first order, Appl. Math. Comput. Sci. 1 (2016), 15–26.
- 329. M. Kostić, Degenerate K-convoluted C-semigroups and degenerate K-convoluted C-cosine functions in locally convex spaces, Chelyabinsk Phy. Math. J. 3 (2018), 90–110.
- M. Kostić, Differential and analytical properties of semigroups of operators, Integral Equations Operator Theory 67 (2010), 499–557.
- 331. M. Kostić, Distribution cosine functions, Taiwanese J. Math. 10 (2006), 739–775.
- 332. M. Kostić, C-Distribution semigroups, Studia Math. 185 (2008), 201–217.

- 333. M. Kostić, Hypercyclic and chaotic integrated C-cosine functions, Filomat 26 (2012), 1-44.
- M. Kostić, Complex powers of non-densely defined operators, Publ. Inst. Math., Nouv. Sér 90 (2011), 47–64.
- 335. M. Kostić, Weyl-almost periodic and asymptotically Weyl-almost periodic properties of solutions to linear and semilinear abstract Volterra integro-differential equations, Math. Notes NEFU 25 (2018), 65–84.
- 336. M. Kostić, Erratum and addendum to the paper "Weyl-almost periodic and asymptotically Weyl-almost periodic properties of solutions to linear and semilinear abstract Volterra integro-differential equations", Mat. Zam. SVFU, 25(2) (2018), 65–84; Math. Notes NEFU 26 (2019), 60–64.
- 337. M. Kostić, Existence of generalized almost periodic and asymptotic almost periodic solutions to abstract Volterra integro-differential equations, Electron. J. Differential Equations, vol. 2017, no. 239 (2017), 1–30.
- M. Kostić, Besicovitch almost automorphic solutions of nonautonomous differential equations of first order, Adv. Oper. Theory 3 (2018), 491–506.
- 339. M. Kostić, Generalized almost periodic solutions and generalized asymptotically almost periodic solutions of inhomogenous evolution equations, Sarajevo J. Math. 14 (2018), 93–111.
- M. Kostić, Weyl-almost periodic solutions and asymptotically Weyl-almost periodic solutions of abstract Volterra integro-differential equations, Banach J. Math. Anal. 13 (2019), 64–90.
- M. Kostić, D. Velinov, A note on almost anti-periodic functions in Banach spaces, Kragujevac J. Math. 44 (2020), 287–297.
- 342. M. Kostić, Complex powers of multivalued linear operators with polynomially bounded Cresolvent, Chelyabinsk Phy. Math., in press.
- 343. M. Kostić, Abstract degenerate Volterra integro-differential equations: inverse generator problem, preprint.
- M. Kostić, Quasi-asymptotically almost periodic functions and applications, Bull. Braz. Math. Soc. (N.S.), accepted.
- 345. M. Kostić, C.-G. Li, M. Li, Abstract multi-term fractional differential equations with Riemann-Liouville derivatives, Acta Math. Sci. Ser. A Chin. Ed. **36A** (2016), 601-622.
- 346. M. Kostić, C.-G. Li, M. Li, On a class of abstract time-fractional equations on locally convex spaces, Abstr. Appl. Anal. 2012 (2013), Article ID 131652, 41 pp.
- 347. M. Kostić, C.-G. Li, M. Li, Abstract multi-term fractional differential equations, Krag. J. Math. 38 (2014), 51–71.
- M. Kostić, P.J. Miana, Relations between distribution cosine functions and almostdistribution cosine functions, Taiwanese J. Math. 11 (2007), 531–543.
- 349. M. Kostić, S. Pilipović, Global convoluted semigroups, Math. Nachr. 280 (2007), 1727-1743.
- M. Kostić, S. Pilipović, D. Velinov, On the exponential ultradistribution semigroups in Banach spaces, Funct. Anal. Appr. Comp. 7 (2015), 59–66.
- M. Kostić, S. Pilipović, D. Velinov, Structural theorems for ultradistribution semigroups, Siberian Math. J. 56 (2015), 83–91.
- 352. M. Kostić, S. Pilipović, D. Velinov, Degenerate C-distribution semigroups in locally convex spaces, Bull. Cl. Sci. Math. Nat. Sci. Math. 41 (2016), 101–123.
- 353. M. Kostić, S. Pilipović, D. Velinov, Degenerate C-ultradistribution semigroups in locally convex spaces, Bull. Cl. Sci. Math. Nat. Sci. Math. 42 (2017), 53–67.
- M. Kostić, S. Pilipović, D. Velinov, C-Distribution semigroups and C-ultradistribution semigroups in locally convex spaces, Siberian Math. J. 58 (2018), 476–492.
- 355. M. Kostić, S. Pilipović, D. Velinov, Quasi-equicontinous exponential families of generalized function C-semigroups in locally convex spaces, Rev. R. Acad. Cienc. Exactas Fís. Nat. Ser. A Math. RACSAM 113 (2019), 453–469.
- 356. A.S. Kovanko, Sur la compacié des sysémes de fonctions presque-périodiques généralisées de H. Weyl, C.R. (Doklady) Ac. Sc. URSS 43 (1944), 275–276.
- 357. A.S. Kovanko, Sur l'approximation des fonctions presque-périodiques généralisées, Mat. Sb. 36 (1929), 409–416.

- 358. A. S. Kovanko, On convergence of sequences of functions in the sense of Weyl's metric $D_{W_{\omega}}$, Ukrainian Math. J. **3** (1951), 465–476 (in Russian).
- 359. A.S. Kovanko, On compactness of systems of generalized almost-periodic functions of Weyl, Ukrainian Math. J. 5 (1953), 185–195 (in Russian).
- 360. A.I. Kozhanov, On boundary value problems for some classes of higher-order equations that are not solved with respect to the highest derivative, Siberian Math. J. 35 (1994), 324–340.
- S.G. Krein, Linear Differential Equations in Banach Space, Translations of Mathematical Monographs, AMS Providence, 1972.
- 362. Y. Kuang, Delay Differential Equations with Applications in Population Dynamics, Mathematics in Science and Engineering, 191. Academic Press, Inc., Boston, MA, 1993.
- P. C. Kunstmann, Stationary dense operators and generation of non-dense distribution semigroups, J. Operator Theory 37 (1997), 111–120.
- 364. P.C. Kunstmann, Distribution semigroups and abstract Cauchy problems, Trans. Amer. Math. Soc. 351 (1999), 837–856.
- 365. P.C. Kunstmann, Banach space valued ultradistributions and applications to abstract Cauchy problems, preprint.
- 366. C.-C. Kuo, Local K-convoluted C-semigroups and abstract Cauchy problems, Taiwanese J. Math. 9 (2015), 1227–1245.
- 367. C.-C. Kuo, Local K-convoluted C-cosine functions and abstract Cauchy problems, Filomat 30 (2016), 2583–2598.
- 368. A. Kuttler, A degenerate nonlinear Cauchy problem, Appl. Anal. 13 (1982), 307–322.
- 369. A. Kuttler, Implicit evolution equations, Appl. Anal. 16 (1983), 91-99.
- 370. R.K. Lamm, I.G. Rosen, An approximation theory for the estimation of parameters in degenerate Cauchy problems, J. Math. Anal. Appl. 162 (1991), 13–48.
- 371. N. T. Lan, Operator equation AX BXD = C and degenerate differential equations in Banach spaces, IJPAM 24 (2005), 383–404.
- 372. T. A. M. Langlands, B. I. Henry, S. L. Wearne, Fractional cable equation models for anomalous electrodiffusion in nerve cells: finite domain solutions, SIAM J. Appl. Math. 71 (2011), 1168–1203.
- 373. N.N. Lebedev, Special Functions and Their Applications, Selected Russian Publications in the Mathematical Sciences. Prentice-Hall, Inc., Englewood Cliffs, N. J., 1965.
- 374. W.-S. Lee, An algebraic-analytic framework for the study of intertwined families of evolution operators, PhD. Thesis, University of Pretoria, 2015.
- 375. Y. Lei, W. Yi, Q. Zheng, Semigroups of operators and polynomials of generators of bounded strongly continuous groups, Proc. London Math. Soc. 69 (1994), 144–170.
- B. Ya. Levin, *Lectures on Entire Functions*, Translations of Mathematical Monographs, AMS Providence, 1996.
- 377. M. Levitan, Almost Periodic Functions, G.I.T.T.L., Moscow, 1959 (in Russian).
- 378. F. Li, Mild solutions for abstract fractional differential equations with almost sectorial operators and infinite delay, Adv. Differ. Equ. 2013, 2013:327, 11 pp.
- 379. F. Li, J. Liang, H.-K. Xu, Existence of mild solutions for fractional integrodifferential equations of Sobolev type with nonlocal conditions, J. Math. Anal. Appl. 391 (2012), 510–525.
- 380. F.-B. Li, M. Li, Q. Zheng, Fractional evolution equations governed by coercive differential operators, Abstr. Appl. Anal. 2009 (2009), Article ID 438690, 14 pp.
- 381. F.-B. Li, M. Li, Q. Zheng, Fractional evolution equations governed by coercive differential operators, Abstr. Appl. Anal. 2009 (2009), Art. ID 438690, 14 pp.
- 382. K.-X. Li, J.-G. Peng, J.-X. Jia, Cauchy problems for fractional differential equations with Riemann-Liouville fractional derivatives, J. Funct. Anal. 263 (2012), 476–510.
- M. Li, J. Pastor, S. Piskarev, Inverses of generators of integrated fractional resolvent operator functions, Fract. Calc. Appl. Anal. 21 (2018), 1542–1564.
- 384. Y.-C. Li, S.-Y. Shaw, N-times integrated C-semigroups and the abstract Cauchy problem, Taiwanese J. Math. 1 (1997), 75–102.

- 385. Y.-C. Li, S.-Y. Shaw, On characterization and perturbation of local C-semigroups, Proc. Amer. Math. Soc.135 (2007), 1097–1106.
- 386. Y.-C. Li, S.-Y. Shaw, On local α-times integrated C-semigroups, Abstract Appl. Anal. 2007 (2007), Article ID 34890, 18 pp., DOI:10.1155/2007/34890.
- 387. Y.-N. Li, H.-J. Sun, Z. Feng, Integrated fractional resolvent operator function and fractional abstract Cauchy problem, Abstract Appl. Anal. 2014 (2014), Article ID 430418, 9 pp., http://dx.doi.org/10.1155/2014/430418.
- 388. Y.-N. Li, H.-J. Sun, Z. Feng, Fractional abstract Cauchy problem with order $\alpha \in (1, 2)$, Dynamics of PDEs 13 (2016), 155–177.
- J. Liang, J. Liu, T.-J. Xiao, Hyperbolic singular perturbations for integrodifferential equations, Appl. Math. Comput. 163 (2005), 609–620.
- 390. Y.-X. Liang, Z.-H. Zhou, Disjoint supercyclic powers of weighted shifts on weighted sequence spaces, Turkish J. Math. 38 (2014), 1007–1022.
- 391. J.L. Lions, Semi-groupes distributions, Port. Math. 19 (1960), 141-164.
- 392. J.H. Liu, Singular perturbations in a nonlinear viscoelasticity, J. Integral Equations Appl. 9 (1997), 99–112.
- 393. J. H. Liu, A singular perturbation problem in integrodifferential equations, Electronic J. Diff. Equ. 02 (1993), 1–10.
- 394. R. Liu, Is A⁻¹ an infinitesimal generator?, J. Appl. Math. Phy. 6 (2018), 1979–1987.
- 395. C. Lizama, Regularized solutions for abstract Volterra equations, J. Math. Anal. Appl. 243 (2000), 278–292.
- 396. C. Lizama, R. Ponce, Periodic solutions of degenerate differential equations in vector-valued function spaces, Studia Math. 202 (2011), 49–63.
- 397. C. Lizama, R. Ponce, Maximal regularity for degenerate differential equations with infinite delay in periodic vector-valued function spaces, Proc. Edinb. Math. Soc. 56 (2013), 853–871.
- C. Lizama, H. Prado, Singular perurbation for Volterra equations of convolution type, Appl. Math. Comp. 181 (2006), 1624–1634.
- 399. W. Long, S.-H. Ding, Composition theorems of Stepanov almost periodic functions and Stepanov-like pseudo-almost periodic functions, Adv. Difference Equ. 2011 (2011), Article ID 654695, 12 pp., doi:10.1155/2011/654695.
- 400. A. T. Lourêdo, G. Siracusa, C. A. Silva Filho, On a nonlinear degenerate evolution equation with nonlinear boundary damping, J. Appl. Math. 2015 (2015), Article ID 281032, 13 pp.
- 401. Y. Luchko, Asymptotics of zeros of the Wright function, J. Math. Anal. Appl. 19 (2000), 1–12.
- 402. C. R. MacCluer, Chaos in linear distributed spaces, J. Dynam. Systems Measurement Control 114 (1992), 322–324.
- 403. G. R. MacLane, Sequences of derivatives and normal families, J. Analyse Math. 2 (1952), 72–87.
- 404. R. Madbouly, A. G. Radwan, R. A. El Barkouky, On Some Fractional-Order Electromagnetic Problems, Lambert Academic Publishing, 2015.
- 405. F. Mainardi, Fractional Calculus and Waves in Linear Viscoelasticity. An Introduction to Mathematical Models, Imperial College Press, London, 2010.
- 406. I. Maizurna, Semigroup methods for degenerate Cauchy problems and stochastic evolution equations, PhD. Thesis, University of Adelaide, 1999.
- 407. L. Maniar, R. Schnaubelt, Almost periodicity of inhomogeneous parabolic evolution equations, Lecture Notes in Pure and Appl. Math. 234, Dekker, New York, 2003, 299–318.
- 408. A. S. Markus, Introduction to the Spectral Theory of Polynomial Operator Pencils, Translations of Mathematical Monographs, AMS Providence, 1998.
- Ö. Martin, Disjoint hypercyclic and supercyclic composition operators, PhD. thesis, Bowling Green State University, 2010.
- C. Martínez, M. Sanz, *The Theory of Fractional Powers of Operators*, North–Holland Math. Stud. 187, Elseiver, Amsterdam, 2001.

- C. Martínez, M. Sanz, J. Pastor, A functional calculus and fractional powers for multivalued linear operators, Osaka J. Math 37 (2000), 551–576.
- 412. C. Martínez, M. Sanz, F. Periago, Distributional fractional powers of the Laplacean. Riesz potentials, Studia Math. 135 (1999), 253–271.
- C. Martínez, M. Sanz, A. Redondo, Fractional powers of almost non-negative operators, Fract. Calc. Appl. Anal. 8 (2005), 201–230.
- 414. F. Martínez-Giménez, P. Oprocha, A. Peris, Distributional chaos for backward shifts, J. Math. Anal. Appl. 351 (2009), 607–615.
- 415. F. Martínez-Giménez, P. Oprocha, A. Peris, Distributional chaos for operators with full scrambled sets, Math. Z. 274 (2013), 603–612.
- 416. M. Matsui, M. Yamada, F. Takeo, Supercyclic and chaotic translation semigroups, Proc. Amer. Math. Soc. 131 (2003), 3535–3546.
- 417. M. Matsui, M. Yamada, F. Takeo, Erratum to "Supercyclic and chaotic translation semigroups", Proc. Amer. Math. Soc. 132 (2004), 3751–3752.
- M. M. Meerschaert, E. Nane, P. Vellaisamy, Fractional Cauchy problems on bounded domains, Ann. Probab. 37 (2009), 979–1007.
- 419. R. Meise, D. Vogt, Introduction to Functional Analysis, Translated from the German by M.S. Ramanujan and revised by the authors. Oxf. Grad. Texts Math., Clarendon Press, New York, 1997.
- I. V. Melnikova, The Cauchy problem for differential inclusion in Banach space and distribution spaces, Siberian Math. J. 42 (2001), 751–765.
- 421. I.V. Melnikova, The degenerate Cauchy problem in Banach spaces, Izv. Ural. Gos. Univ. Mat. Mekh. 10 (1998), 147–160.
- 422. I. V. Melnikova, M. A. Alshansky, Well-posedness of the Cauchy problem in a Banach space: regular and degenerate cases, J. Math. Sci. (New York) 87 (1997), 3732–3780.
- 423. I. V. Melnikova, U.A. Anufrievaa, V. YU. Ushkov, Degenerate distribution semigroups and well-posedness of the Cauchy problem, Integral Transforms Spec. Funct. 6 (1998), 247–256.
- 424. I. V. Melnikova, A. I. Filinkov, Abstract Cauchy Problems: Three Approaches, Chapman and Hall/CRC, Boca Raton, 2001.
- 425. Q. Menet, Linear chaos and frequent hypercyclicity, Trans. Amer. Math. Soc. 369 (2017), 4977–4994.
- 426. B. Messirdi, A. Gherbi, M. Amouch, A spectral analysis of linear operator pencils on Banach spaces with application to quotient of bounded operators, International J. Math. Anal. 7 (2015), 104–128.
- 427. G. Metafune, L^p-Spectrum of Ornstein-Uhlenbeck operators, Ann. Scuola Norm. Sup. Pisa Cl. Sci. 30 (2001), 97–124.
- 428. P.J. Miana, Almost-distribution cosine functions and integrated cosine functions, Studia Math. 166 (2005), 171–180.
- P. J. Miana, V. Poblete, Sharp extensions for convoluted solutions of wave equations, Differential Integral Equations 28 (2015), 309–332.
- V. Müller, J. Vršovský, Orbits of linear operators tending to infinity, Rocky Mountain J. Math. 39 (2009), 219–230.
- V.I. Nalimov, Degenerate second-order differential equations in Hilbert spaces, Siberian Math J. 35 (1994), 566–576.
- 432. E. Nane, Higher order PDE's and iterated processes, Trans. Amer. Math. Soc. 360 (2008), 2681–2692.
- F. Neubrander, Wellposedness of higher order abstract Cauchy problems, Trans. Amer. Math. Soc. 295 (1986), 257–290.
- 434. L. Nguyen, Periodicity of mild solutions to higher order differential equations in Banach spaces, Electronic J. Diff. Equ. 79 (2004), 1–12.
- 435. L. Nguyen, On the periodic mild solutions to complete higher order differential equations on Banach spaces, Surv. Math. Appl. 6 (2011), 23–41.

- 436. P. Q. H. Nguyen, On variable Lebesgue spaces, PhD. Thesis, Kansas State University, Pro-Quest LLC, Ann Arbor, MI, 2011. 63 pp.
- 437. V. Obukhovskii, P. Zecca, On boundary value problems for degenerate differential inclusions in Banach spaces, Abstr. Appl. Anal. 13 (2003), 769–784.
- 438. H. Oka, Linear Volterra equations and integrated solution families, Semigroup Forum 53 (1996), 278–297.
- K. B. Oldham, J. Spanier, *The Fractional Calculus*, Academic Press, New-York, London, 1974.
- 440. P. Oprocha, Distibutional chaos revisited, Trans. Amer. Math. Soc. 361 (2009), 4901-4925.
- 441. P. Oprocha, A quantum harmonic oscillator and strong chaos, J. Phys. A 39 (2006), 14559–14565.
- 442. A. P. Oskolkov, Initial-boundary value problems for equations of motion of Kelvin Voigt fluids and Oldroyd fluids, Trudy Mat. Inst. Steklov. 179 (1988), 126-164 (Russian).
- 443. A. P. Oskolkov, Nonlocal problems for one class of nonlinear operator equations, arising in the theory of Sobolev type equations, Zap. Nauchn. Sem. LOMI **198** (1991), 31–48 (Russian).
- 444. J. Pastor, On uniqueness of fractional powers of multi-valued linear operators and the incomplete Cauchy problem, Ann. Mat. Pura. Appl. 191 (2012), 167–180.
- 445. A. Pazy, Semigroups of Linear Operators and Applications to Partial Differential Equations, Springer-Verlag, New York, 1983.
- 446. F. Periago, Global existence, uniqueness, and continuous dependence for a semilinear initial value problem, J. Math. Anal. Appl. 280 (2003), 413–423.
- 447. F. Periago, B. Straub, A functional calculus for almost sectorial operators and applications to abstract evolution equations, J. Evol. Equ. 2 (2002), 41–68.
- 448. V. Petrović, Graph Theory, University of Novi Sad, Novi Sad, 1998 (Serbian).
- 449. S. Piskarev, H. Zwart, Crank-Nicolson scheme for abstract linear systems, Numer. Funct. Anal. Optim. 28 (2007), 717–736.
- 450. M. V. Plekhanova, Optimal control problems for degenerate semilinear equation of fractional order, preprint.
- 451. M. V. Plekhanova, V. E. Fedorov, Optimal Control of Degenerate Evolution Systems, Publication Center IIGU University, Chelyabinsk, 2013 (Russian).
- 452. I. Podlubny, Fractional Differential Equations, Academic Press, New York, 1999.
- 453. R. Ponce, Hölder continuous solutions for Sobolev type differential equations, Math. Nachr. 287 (2014), 70–78.
- 454. R. P. Cubillos, Bounded solutions to evolution equations in Banach spaces, PhD. Thesis, University of Talca, 2014.
- 455. R. Ponce, On the well-posedness of degenerate fractional differential equations in vector valued function spaces, Israel J. Math. 2 (2017), 727–755.
- 456. R. Ponce, M. Warma, Asymptotic behavior and representation of solutions to a Volterra kind of equation with a singular kernel, preprint. arXiv:1610.08750.
- 457. M. Pramanik, R. P. Sarkar, Chaotic dynamics of the heat semigroup on Riemannian symmetric spaces, J. Funct. Anal. 266 (2014), 2867–2909.
- 458. V. Protopopescu, Y. Azmy, Topological chaos for a class of linear models, Math. Models Methods Appl. Sci. 2 (1992), 79–90.
- J. Prüss, Evolutionary Integral Equations and Applications, Monogr. Math. 87, Birkhäuser, Basel, Boston, Berlin, 1993.
- 460. J. Prüss, Positivity and regularity of hyperbolic Volterra equations in Banach spaces, Math. Ann. 279 (1987), 317–344.
- 461. J. Prüss, On the spectrum of C₀-semigroups, Trans. Amer. Math. Soc. 284 (1984), 847–857.
- 462. Y. Puig, Disjoint Hypercyclicity along filters, preprint, arXiv:1411.7721.
- 463. L. Rodman, An Introduction to Operator Polynomials, Birkhäuser-Verlag, Basel, 1989.
- 464. S. Rolewicz, On orbits of elements, Studia Math. 32 (1969), 17–22.
- 465. W. M. Ruess, W. H. Summers, Asymptotic almost periodicity and motions of semigroups of operators, Linear Algebra Appl. 84 (1986), 335–351.

- 466. W. M. Ruess, W. H. Summers, Integration of asymptotically almost periodic functions and weak asymptotic almost periodicity, Dissertationes Math. (Rozprawy Mat.) 279 (1989), 35.
- 467. S. M. Rump, Polynomial minimum root separation, Math. Comp. 33 (1979), 327–336.
- 468. A.G. Rutkas, Solvability of semilinear differential equations with a singularity, Ukrainian Math. J. 60 (2008), 262–276.
- A.G. Rutkas, I.G. Khudoshin, Global solvability of one degenerate semilinear differential operator equation, Nonlinear Oscill. 7 (2004), 403–417.
- 470. H.R. Salas, Hypercyclic weighted shifts, Trans. Amer. Math. Soc. 347 (1995), 993-1004.
- 471. H. N. Salas, Dual disjoint hypercyclic operators, J. Math. Anal. Appl. 374 (2011), 106-117.
- 472. S.G. Samko, A.A. Kilbas, O.I. Marichev, Fractional Derivatives and Integrals: Theory and Applications, Gordon and Breach, New York, 1993.
- 473. A. M. Samoilenko, M. I. Shkil', V. P. Yakovets', Linear Degenerate Systems of Differential Equations, Vyshcha Shkola, Kyiv, 2000 (Ukrainian).
- 474. R. P. Sarkar, Chaotic dynamics of the heat semigroup on the Damek-Ricci spaces, Israel J. Math. 198 (2013), 487–508.
- 475. N. Sauer, Linear evolution equations in two Banach spaces, Proc. Royal Soc. Edinburgh 91A (1982), 287–303.
- 476. N. Sauer, Empathy theory and Laplace transform, Banach Center Publications 38 (1997), 325–338.
- 477. R. K. Saxena, A. M. Mathai, H. J. Haubold, *Reaction-diffusion systems and nonlinear waves*, Astrophys. Space Sci. **305** (2006), 297–303.
- 478. H. J. Schmeisser, H. Triebel, Topics on Fourier Analysis and Function Spaces, Wiley, 1987.
- 479. L. Schwartz, Théorie des distributions à valeurs vectorielles. I, Ann. Inst. Fourier (Grenoble) 7 (1957), 1–141.
- L. Schwartz, *Théorie des distributions à valeurs vectorielles. II*, Ann. Inst. Fourier (Grenoble)
 8 (1958), 1–209.
- 481. B. Schweizer, J. Smital, Measure of chaos and a spectral decomposition of dynamical systems on the interval, Trans. Amer. Math. Soc. 344 (1994), 737–754.
- 482. F. Schwenninger, Generalisations of semigroups of operators in the view of linear relations, Dipl. Thesis, Technischen Universiträt Wien, 2011.
- 483. S. Ya. Sekerzh-Zen'kovich, Construction of the fundamental solution for the operator of internal waves, J. Appl. Math. Mech. (01)45 (1981), 192–198.
- R. Servadeia, E. Valdinoci, On the spectrum of two different fractional operators, Proc. Royal Soc. Edinburgh 144 (2014), 831–855.
- 485. D.E. Shafranov, Cauchy problem for the equation of a free surface of filtered liquid, in space of k-forms defined on Riemannian manifolds without boundary, Proceedings of the Chelyabinsk Scientific Centre 3(37) (2007) (Russian).
- V. B. Shakhmurov, A. Sahmurova, Degenerate abstract parabolic equations and applications, Commun. Math. Anal. 18 (2015), 15–33.
- 487. S.-Y. Shaw, Cosine operator functions and Cauchy problems, Conf. Semin. Mat. Univ. Bari 287 (2002), 1–75.
- 488. R. Shiraishi, Y. Hirata, Convolution maps and semi-group distributions, J. Sci. Hiroshima Univ. Ser. A-I 28 (1964), 71–88.
- 489. S. Shkarin, A short proof of existence of disjoint hypercyclic operators, J. Math. Anal. Appl. 367 (2010), 713-715.
- R. E. Showalter, Hilbert Space Methods for Partial Differential Equations, Pitman, London, 1977.
- 491. R. E. Showalter, Monotone Operators in Banach Space and Nonlinear Partial Differential Equations, Math. Surveys and Monographs, AMS, Providence, 1997.
- 492. R. E. Showalter, Nonlinear degenerate evolution equations and partial differential equations of mixed type, SIAM J. Math. Anal. 6 (1975), 25–42.
- 493. R. E. Showalter, Nonlinear degenerate evolution equations in mixed formulation, SIAM J. Math. Anal. 42 (2010), 2114–2131.

- 494. R. E. Showalter, Degenerate parabolic initial-boundary value problems, J. Differential Equations 31 (1979), 296–312.
- 495. R. E. Showalter, Initial and final-value problems for degenerate parabolic evolution systems, Indiana Univ. Math. J. 28 (1979), 883–893.
- 496. D. Sidorov, Integral Dynamical Models: Singularities, Signals and Control, In: L.O. Chua (ed.), World Scientific Series on Nonlinear Science, Series A. 87 (2015), 258 pp., Singapore, London: World Scientific Publ.
- 497. N. A. Sidorov, General Regularization Questions in Problems of Bifurcation Theory (Obshchie Voprosy Regulyarizatsii v Zadachakh Teorii Vetvleniya) (Russian), Izdatel'stvo Irkutskogo Universiteta, Zbl 0703.58002, Irkutsk, 1982, 312 pp.
- 498. N. Sidorov, B. Loginov, B. Sinitsyn, M. Falaleev, Lyapunov-Schmidt Methods in Nonlinear Analysis and Applications, Springer Science+Business Media, Dordrecht, 2013.
- 499. A. Sklar, J. Smíntal, Distributional chaos on compact metric spaces via specification properties, J. Math. Anal. Appl. 241 (2000), 181–188.
- 500. D. Smith, Singular Perturbation Theory, Cambridge University Press, Cambridge, 1985.
- 501. O. J. Staffans, Periodic solutions of an integrodifferential equation in a Hilbert space, Proc. Amer. Math. Soc. 117 (1993), 745–751.
- 502. B. Straub, Fractional powers of operators with polynomially bounded resolvent and the semigroups generated by them, Hiroshima Math. J. 24 (1994), 529–548.
- 503. A.G. Sveshnikov, A.B. Al'shin, M.O. Korpusov, Yu. D. Pletner, *Linear and Nonlinear Equations of Sobolev Type*, Fizmatlit, Moscow, 2007 (Russian).
- 504. G. A. Sviridyuk, Morphology of the phase space of a class of semilinear equations of Sobolev type, Vestnik Chelyabinsk. Univ. Ser. 3 Mat. Mekh. **2(5)** (1999), 68–86 (Russian).
- 505. G. A. Sviridyuk, Phase spaces of Sobolev type semilinear equations with a relatively sectorial operator, St. Petersburg Math. J. 6 (1995), 1109–1126.
- 506. G. A. Sviridyuk, Semilinear equations of Sobolev type with a relatively sectorial operators, Russian Acad. Sci. Dokl. Math. 47 (1993), 612–617.
- 507. G. A. Sviridyuk, On the general theory of operator semigroups, Uspekhi Mat. Nauk 49 (1994), 47–74.
- 508. G. A. Sviridyuk, Semilinear equations of Sobolev type with a relatively sectorial operator, Dokl. Akad. Nauk **329** (1993), 274–277.
- 509. G. A. Sviridyuk, V. E. Fedorov, *Linear Sobolev Type Equations and Degenerate Semigroups* of Operators, Inverse and Ill-Posed Problems (Book **42**), VSP, Utrecht, Boston, 2003.
- G. A. Sviridyuk, N.A. Manakova, Regular perturbations of a class of linear equations of Sobolev type, Differ. Uravn. 38 (2002), 423–425.
- 511. G. A. Sviridyuk, N. A. Manakova, The dynamical models of Sobolev type with Showalter-Sidorov condition and additive 'noise', Bull. South Ural State University. Series: Mathematical Modelling, Programming and Computer Software 7 (2014), 70–75 (Russian).
- G. A. Sviridyuk, O. V. Vakarina, *Higher-order linear equations of Sobolev type*, Dokl. Akad. Nauk. **363** (1998), 308–310 (Russian).
- 513. G. A. Sviridyuk, O. V. Vakarina, The Cauchy problem for a class of higher-order linear equations of Sobolev type, Differ. Uravn. 33 (1998), 1415–1424.
- 514. G. A. Sviridyuk, S. A. Zagrebina, The Showalter-Sidorov problem as a phenomena of the Sobolev-type equations, IIGU Ser. Matematika 3 (2010), 104–125.
- 515. G. A. Sviridyuk, A. A. Zamyshlyaeva, The phase spaces of a class of higher-order linear equations of Sobolev type, Differ. Uravn. 42 (2006), 269–278.
- 516. V.E. Tarasov, Fractional Dynamics: Applications of Fractional Calculus to Dynamics of Particles, Fields and Media, Springer-Verlag, New York, 2011.
- 517. B. Thaller, S. Thaller, Factorization of degenerate Cauchy problems: the linear case, J. Operator Theory 36 (1996), 121–146.
- B. Thaller, S. Thaller, Semigroup theory of degenerate linear Cauchy problems, Semigroup Forum 62 (2001), 375–398.

- B. Thaller, S. Thaller, Approximation of degenerate Cauchy problems, SFB F0003, Optimierung und Kontrolle 76, University of Graz.
- 520. E. C. Titchmarsh, *The zeroes of certain integral functions*, Proc. London Math. Soc. **25** (1926), 283–302.
- 521. H. Triebel, Interpolation Theory. Function Spaces. Differential Operators, North-Holland Publ. Company, 1978.
- 522. T. Ushijima, On the abstract Cauchy problem and semi-groups of linear operators in locally convex spaces, Sci. Pap. Coll. Gen. Educ., Univ. Tokyo 21 (1971), 93–122.
- 523. D. Velinov, M. Kostić, S. Pilipović, Degenerate C-distribution cosine functions and degenerate C-ultradistribution cosine functions in locally convex spaces, Filomat **31** (2017), 3075–3089.
- 524. V. Vergara, R. Zacher, A priori bounds for degenerate and singular evolutionary partial integro-differential equations, Nonlinear Anal. 73 (2010), 3572–3585.
- 525. A.I. Volpert, S.I. Khudyaev, Cauchy's problem for degenerate second order quasilinear parabolic equations, Mat. Sb. (N.S.) 78 (1969), 374–396.
- 526. J. von Neumann, Functional Operators: The Geometry of Orthogonal Spaces, Vol. 2, Ann. of Math. Stud. 22, Princeton Univ. Press, Princeton NJ, 1950.
- 527. W. von Wahl, Gebrochene Potenzen eines elliptischen Operators und parabolische Differentialgleichungen in Räumen hölderstetiger Funktionen, Nachr. Akad. Wiss. Göttingen Math.-Phys. Kl. 11 (1972), 231–258.
- 528. M. Ju Vuvunikjan, The asymptotic resolvent, and theorems on generation of semigroups of operators in locally convex spaces, Mat. Zametki 22 (1977), 433–442.
- R.-N. Wang, D.-H. Chen, T.-J. Xiao, Abstract fractional Cauchy problems with almost sectorial operators, J. Differential Equations 252 (2012), 202–235.
- 530. S. Wang, Quasi-distribution semigroups and integrated semigroups, J. Funct. Anal. 146 (1997), 352–381.
- 531. S. Wang, Properties of subgenerators of C-regularized semigroups, Proc. Amer. Math. Soc. 126 (1998), 453–460.
- 532. S.W. Wang, M.C. Gao, Automatic extensions of local regularized semigroups and local regularized cosine functions, Proc. Amer. Math. Soc. 127 (1999), 1651–1663.
- 533. G.N. Watson, Theory of Bessel Functions, 2nd Ed., Cambridge Univ. Press, Cambridge, 1962.
- 534. H. Weyl, Integralgleichungen und fastperiodische Funktionen, Math. Ann. 97 (1926), 338–356.
- 535. G.B. Whitham, Linear and Nonlinear Waves, Wiley, Hoboken, 1974.
- 536. D.V. Widder, The Laplace Transform, Princeton University Press, 1946.
- 537. D. Wilcox, Multivalued semi-fredholm operators in normed linear spaces, PhD. Thesis, Universität of Cape Town, 2002.
- 538. D. Wilcox, Essential spectra of linear relations, Linaer Algebra Appl. 462 (2014), 110–125.
- R. Wong, Y.-Q. Zhao, Exponential asymptotics of the Mittag-Leffler function, Constr. Approx. 18 (2002), 355–385.
- J. Wu, Theory and Applications of Partial Functional Differential Equations, Applied Mathematical Sciences, 119. Springer-Verlag, New York, 1996.
- 541. T.-J. Xiao, J. Liang, The Cauchy Problem for Higher–Order Abstract Differential Equations, Springer–Verlag, Berlin, 1998.
- 542. T.-J. Xiao, J. Liang, Abstract degenerate Cauchy problems in locally convex spaces, J. Math. Anal. Appl. 259 (2001), 398–412.
- 543. T.-J. Xiao, J. Liang, Higher order degenerate Cauchy problems in locally convex spaces, Math. Comp. Modelling 41 (2005), 837–847.
- 544. T.-J. Xiao, J. Liang, Entire solutions of higher order abstract Cauchy problems, J. Math. Anal. Appl. 208 (1997), 298–310.
- 545. T.-J. Xiao, J. Liang, Higher order abstract Cauchy problems: their existence and uniqueness families, J. Lond. Math. Soc. 67 (2003), 149–164.

- 546. T.-J. Xiao, J. Liang, Laplace transforms and integrated, regularized semigroups in locally convex spaces, J. Funct. Anal. 148 (1997), 448–479.
- 547. T.-J. Xiao, J. Liang, J. V. Casteren, *Time dependent Desch-Schappacher type perturbations* of Volterra integral equations, Integral Equations Operator Theory **44** (2002), 494–506.
- 548. R. Xie and C. Zhang, Space of ω-periodic limit functions and its applications to an abstract Cauchy problem, J. Function Spaces, vol. 2015, Article ID 953540, 10 pages http://dx.doi.org/10.1155/2015/953540.
- 549. A. Yagi, Generation theorem of semigroup for multivalued linear operators, Osaka J. Math. **28** (1991), 385–410.
- 550. Q. Yang, Novel analytical and numerical methods for solving fractional dynamical systems, PhD. Thesis, Queensland University of Technology, Brisbane, 2010.
- 551. S.A. Zagrebina, On the Showalter-Sidorov problem, Russian Math. (Iz. VUZ) 51 (2007), 19–25.
- 552. S. A. Zagrebina, E. A. Soldatova, The linear Sobolev-type equations with relatively p-bounded operators and additive white noise, Bull. Irkutsk State University. Series Mathematics 6 (2013), 20–34 (Russian).
- 553. S. Zaidman, Well-posed Cauchy problem and related semigroups of operators for the equation $Bu'(t) = Au(t), t \ge 0$, in Banach spaces, Libertas Math. **12** (1992), 147–159.
- 554. A. A. Zamyshlyaeva, Stochastic incomplete linear sobolev type high-ordered equations with additive white noise, Bull. South Ural State University. Series: Mathematical Modelling, Programming and Computer Software 14 (2012), 73–82 (Russian).
- 555. A. A. Zamyshlyaeva, E. V. Bychkov, The phase space of the modified Boussinesq equation, Vestnik YuUrGU. Ser. Mat. Model. Progr. 12 (2012), 13–19.
- 556. A. A. Zamyshlyaeva, E. V. Bychkov, O. N. Tsyplenkova, Mathematical models based on Boussinesq-Love equation, Applied Math. Sci. 8 (2014), 5477–5483.
- 557. C. Zhang, Ergodicity and asymptotically almost periodic solutions of some differential equations, IJMMS 25 (2001), 787–800.
- 558. Z. Zhang, J.-B, Liu, J. Cao, W. Jiang, A. Alsaedi, F.E. Alsaedi, *Stability results for the linear degenerate fractional differential system*, Adv. Difference Equ. **216** (2016), DOI 10.1186/s13662-016-0941-0, 14), 17 pp.
- 559. Q. Zheng, Matrices of operators and regularized cosine functions, J. Math. Anal. Appl. 315 (2006), 68–75.
- Q. Zheng, M. Li, Regularized Semigroups and Non-Elliptic Differential Operators, Science Press, Beijing, 2014.
- 561. Q. Zheng, Y. Li, Abstract parabolic systems and regularized semigroups, Pacific J. Math. 182 (1998), 183–199.
- 562. M. Zubair, M. J. Mughal, Q. A. Naqvi, *Electromagnetic Fields and Waves in Fractional Dimensional Space*, Springer-Verlag, Berlin, 2012.
- 563. H. Zwart, Is A⁻¹ an infinitesimal generator?, In: Perspectives in Operator Theory, Banach Center Publ. 75, Inst. Math., Polish Acad. Sci. (2007), 303–313.
- 564. H. Zwart, Growth estimates for $\exp(A^{-1}t)$ on a Hilbert space, Semigroup Forum 74 (2007), 487–494.

The theory of linear degenerate Volterra integro-differential equations is still very undeveloped. The main purpose of this monograph is to provide an overview of the recent research on abstract degenerate Volterra integro-differential equations and abstract degenerate fractional differential equations in sequentially complete locally convex spaces. Some of our results seem to be new even for abstract non-degenerate differential equations in Banach spaces.

Fractional calculus and fractional differential equations are rapidly growing fields of research, having invaluable importance in modeling of various problems appearing in applied and theoretical sciences. The present monograph is probably the first research monograph that is specifically dedicated to the abstract degenerate (multi-term) fractional differential equations and the abstract degenerate Volterra integro-differential equations.

The book is not compulsively detailed and there are numerous important topics not analyzed here. Selection is based on the personal views and aspirations of the author, with the special attention paid to the applications of various types of convoluted or *C*-regularized families of solution operators in the analysis of abstract degenerate integro-differential equations.

